

CITY OF SEATTLE

City Council

Agenda

Tuesday, March 18, 2025 2:00 PM

Council Chamber, City Hall 600 4th Avenue Seattle, WA 98104

Sara Nelson, Council President Joy Hollingsworth, Member Robert Kettle, Member Cathy Moore, Member Alexis Mercedes Rinck, Member Maritza Rivera, Member Rob Saka, Member Mark Solomon, Member Dan Strauss, Member

Chair Info: 206-684-8809; Sara.Nelson@seattle.gov

Watch Council Meetings Live View Past Council Meetings

Council Chamber Listen Line: 206-684-8566

The City of Seattle encourages everyone to participate in its programs and activities. For disability accommodations, materials in alternate formats, accessibility information, or language interpretation or translation needs, please contact the Office of the City Clerk at 206-684-8888 (TTY Relay 7-1-1), <u>CityClerk@Seattle.gov</u>, or visit <u>https://seattle.gov/cityclerk/accommodations</u> at your earliest opportunity. Providing at least 72-hour notice will help ensure availability; sign language interpreting requests may take longer.



CITY OF SEATTLE

City Council Agenda

March 18, 2025 - 2:00 PM

Meeting Location:

Council Chamber, City Hall, 600 4th Avenue, Seattle, WA 98104

Committee Website:

http://www.seattle.gov/council

Members of the public may register for remote or in-person Public Comment to address the Council. Details on how to provide Public Comment are listed below:

Remote Public Comment - Register online to speak during the Public Comment period at <u>https://www.seattle.gov/council/committees/public-comment</u> Online registration to speak will begin one hour before the meeting start time, and registration will end at the conclusion of the Public Comment period during the meeting. Speakers must be registered in order to be recognized by the Chair.

In-Person Public Comment - Register to speak on the Public Comment sign-up sheet located inside Council Chambers at least 15 minutes prior to the meeting start time. Registration will end at the conclusion of the Public Comment period during the meeting. Speakers must be registered in order to be recognized by the Chair.

Submit written comments to all Councilmembers prior to 10 a.m. on the day of the meeting at <u>Council@seattle.gov</u> or at Seattle City Hall, Attn: Council Public Comment, 600 4th Ave., Floor 2, Seattle, WA 98104.

A. CALL TO ORDER

B. ROLL CALL

C. PRESENTATIONS

D. PUBLIC COMMENT

Members of the public may sign up to address the Council for up to 2 minutes on matters on this agenda; total time allotted to public comment at this meeting is 20 minutes.

E. ADOPTION OF INTRODUCTION AND REFERRAL CALENDAR:

Introduction and referral to Council committees of Council Bills (CB), Resolutions (Res), Appointments (Appt), and Clerk Files (CF) for committee recommendation.

IRC 471 March 18, 2025

Attachments: Introduction and Referral Calendar

F. APPROVAL OF THE AGENDA

G. APPROVAL OF CONSENT CALENDAR

The Consent Calendar consists of routine items. A Councilmember may request that an item be removed from the Consent Calendar and placed on the regular agenda.

Journal:

1. <u>Min 511</u> March 11, 2025

Attachments: Minutes

Bills:

2. <u>CB 120951</u> AN ORDINANCE appropriating money to pay certain claims for the week of March 3, 2025, through March 7, 2025, and ordering the payment thereof; and ratifying and confirming certain prior acts.

Supporting Documents: Summary and Fiscal Note

Appointments:

LIBRARIES, EDUCATION, AND NEIGHBORHOODS COMMITTEE:

City Council			Agenda	March 18, 2025
	3.	<u>Appt 03101</u>	Reappointment of Carmen A. Bendixen as r Seattle Public Library Board of Trustees, for April 1, 2030.	nember, [.] a term to
			The Committee recommends that Council the Appointment (Appt). In Favor: 4 - Rivera, Rinck, Moore, Solomo Opposed: None	confirm on
		<u>Attachments:</u>	Appointment Packet	
	4.	<u>Appt 03104</u>	Reappointment of Evan M. Smith as member Education, Preschool, and Promise Levy Ov Committee, for a term to December 31, 202	er, Families, /ersight 6.
			The Committee recommends that Council the Appointment (Appt). In Favor: 4 - Rivera, Rinck, Moore, Solomo Opposed: None	confirm on
		<u>Attachments:</u>	Appointment Packet	
	5.	<u>Appt 03105</u>	Reappointment of Linda Thompson-Black as Families, Education, Preschool, and Promis Oversight Committee, for a term to Decemb 2026.	s member, e Levy er 31,
			The Committee recommends that Council the Appointment (Appt). In Favor: 4 - Rivera, Rinck, Moore, Solomo Opposed: None	confirm on
		<u>Attachments:</u>	Appointment Packet	
н. сомміт	TEE R	EPORTS		
		Discussion and Appointments	d vote on Council Bills (CB), Resolutions (Res), (Appt), and Clerk Files (CF).	
<u>GOVERNAN</u>	<u>CE, A</u>	CCOUNTABILITY,	AND ECONOMIC DEVELOPMENT COMMIT	ITEE:

City Council	Agenda	March 18, 2025
1. <u>CB 120933</u>	AN ORDINANCE relating to land use and zoning; amendi 23.74.002 and 23.74.008 of the Seattle Municipal Code to residential uses in the Stadium Transition Area Overlay D	ng Sections allow istrict.
	The Committee recommends that City Council pass the Bill (CB). In Favor: 3 - Nelson, Rivera, Solomon Opposed: 2 - Kettle, Hollingsworth	e Council
<u>Supporting</u>		
<u>Documents:</u>	Summary and Fiscal Note	
	Summary Att 1 - Map of the Stadium Area Transition	
	Overlay District	
	Summary Att 2 - Seattle Industrial & Maritime Strategy	
	Final Environmental Impact Statement	
	Amendment A	
	Amendment B	
	Amendment C	
	Amendment D	
	Amendment E	
	Amendment F	
	Amendment G	
	Amendment H	
	Amendment I	
I. ITEMS REMOVED FI	ROM CONSENT CALENDAR	
J. ADOPTION OF OTH	IER RESOLUTIONS	

K. OTHER BUSINESS

L. ADJOURNMENT



Legislation Text

File #: IRC 471, Version: 1

March 18, 2025

6



Introduction and Referral Calendar

List of proposed Council Bills (CB), Resolutions (Res), Appointments (Appt) and Clerk Files (CF) to be introduced and referred to a City Council committee

Record No.		Title	Committee Referra		
	By: Strauss				
1.	<u>CB 120951</u>	AN ORDINANCE appropriating money to pay certain claims for the week of March 3, 2025, through March 7, 2025, and ordering the payment thereof; and ratifying and confirming certain prior acts.	City Council		
	By: Strauss				
2.	<u>CB 120955</u>	AN ORDINANCE relating to federal funding; authorizing department directors to accept certain grants and passthrough funding in 2025; and ratifying and confirming certain prior acts.	City Council		
	By: Nelson				
3.	<u>CB 120957</u>	AN ORDINANCE relating to publicly-financed election campaigns; providing for the submission to the qualified electors of the City at an election to be held on August 5, 2025, of a proposition authorizing the City to levy regular property taxes for up to ten years in excess of the limitation on levies in chapter 84.55 RCW for the purpose of funding the cost and administration of the City's Democracy Voucher program and other City purposes; outlining a process for contemplation of changes to the program; applying RCW 84.36.381's senior citizens and disabled persons exemption to such levy; and ratifying and confirming certain prior acts.	Governance, Accountability, and Economic Development Committee		
	By: Nelson				
4.	<u>Appt 03106</u>	Appointment of Vivian Vassall as member, Seattle Ethics and Elections Commission, for a term to December 31, 2027.	Governance, Accountability, and Economic Development Committee		
	By: Nelson				
5.	<u>Appt 03107</u>	Reappointment of Thomas J. Kelly as member, Joint Apprenticeship Training Committee, for a term to December 31, 2027.	Governance, Accountability, and Economic Development Committee		

By: Nelson

6. <u>Appt 03108</u>

Reappointment of Lisa Reager as member, Joint Apprenticeship Training Committee, for a term to December 31, 2027.

Governance, Accountability, and Economic Development Committee

Public Safety Committee

Public Safety

Committee

By: Kettle

7. <u>CB 120956</u> AN ORDINANCE relating to the regulation of after-hours nightlife lounges; defining after-hours nightlife lounges; establishing operational and safety requirements for such businesses; describing enforcement mechanisms; and adding a new Chapter 10.10 to the Seattle Municipal Code.

By: Saka,Kettle

8. <u>Res 32167</u> A RESOLUTION acknowledging that Seattle residents, workers, students, and visitors deserve to be safe and feel safe; recognizing and appreciating first responders from the Seattle Police Department, Seattle Fire Department, and the Community Assisted Response and Engagement Department; affirming the City's obligations to fully support, train, and equip first responders; committing to a diversified public safety response system; acknowledging the City's actions to reform the Police Department under the federal Consent Decree; committing to resolve the remaining issues of the Consent Decree; and affirming the essential services provided by the Police Department.

By: Rinck

9. CB 120952 AN ORDINANCE relating to the City Light Department; authorizing the Department to establish eligibility requirements for customers participating in the Renewable Plus Program and charging a rate for the delivery of dedicated renewable energy and its associated Renewable Energy Credits supplied by the Department under the Renewable Plus Program; and amending Section 21.49.089 of the Seattle Municipal Code. Sustainability, City Light, Arts and Culture Committee

Sustainability, City

Culture Committee

Light, Arts and

By: Rinck

10. <u>CB 120953</u> AN ORDINANCE relating to the City Light Department; accepting the following easements for electrical distribution rights in King County, Washington; placing said easements under the jurisdiction of the City Light Department; and ratifying and confirming certain prior acts.

By: Rinck

11. <u>CB 120954</u>
 AN ORDINANCE relating to the City Light Department; accepting the following easements for electrical distribution in King County, Washington; placing said easements under the jurisdiction of the City Light Department; and ratifying and confirming certain prior acts.
 Sustainability, City Light, Arts and Culture Committee



Legislation Text

File #: Min 511, Version: 1

March 11, 2025

9

SEATTLE CITY COUNCIL

600 Fourth Ave. 2nd Floor Seattle, WA 98104



Journal of the Proceedings of the Seattle City Council

Tuesday, March 11, 2025

2:00 PM

Council Chamber, City Hall 600 4th Avenue Seattle, WA 98104

City Council

Sara Nelson, Council President Joy Hollingsworth, Member Robert Kettle, Member Cathy Moore, Member Alexis Mercedes Rinck, Member Maritza Rivera, Member Rob Saka, Member Mark Solomon, Member Dan Strauss, Member

Chair Info: 206-684-8809; Sara.Nelson@seattle.gov

A. CALL TO ORDER

The City Council of The City of Seattle met in the Council Chamber in City Hall in Seattle, Washington, on March 11, 2025, pursuant to the provisions of the City Charter. The meeting was called to order at 2:02 p.m., with Council President Nelson presiding.

B. ROLL CALL

Present: 8 - Hollingsworth, Kettle, Moore, Nelson, Rinck, Saka, Solomon, Strauss

Late Arrival: 1 - Rivera

C. PRESENTATIONS

There were none.

D. PUBLIC COMMENT

The following individuals addressed the Council: Nancy Craig

Councilmember Rivera entered the Council Chamber at 2:03 p.m.

Lanning Mosher Bennett Haselton Alex Tsimerman Sam Montrose David Haines

E. ADOPTION OF INTRODUCTION AND REFERRAL CALENDAR:

IRC 470 March 11, 2025

By unanimous consent, the Introduction & Referral Calendar (IRC) was adopted.

In Favor: 9 - Hollingsworth, Kettle, Moore, Nelson, Rinck, Rivera, Saka, Solomon, Strauss

Opposed: None

F. APPROVAL OF THE AGENDA

By unanimous consent, the Agenda was adopted.

G. APPROVAL OF CONSENT CALENDAR

Motion was made by Council President Nelson, duly seconded and carried, to adopt the Consent Calendar.

Journal:

1. <u>Min 510</u> March 4, 2025

The Minutes (Min) were adopted on the Consent Calendar by the following vote, and the President signed the Minutes (Min):

In Favor: 9 - Hollingsworth, Kettle, Moore, Nelson, Rinck, Rivera, Saka, Solomon, Strauss

Opposed: None

Bills:

2. <u>CB 120947</u> AN ORDINANCE appropriating money to pay certain claims for the week of February 24, 2025, through February 28, 2025, and ordering the payment thereof; and ratifying and confirming certain prior acts.

The Council Bill (CB) was passed on the Consent Calendar by the following vote, and the President signed the Council Bill (CB):

In Favor: 9 - Hollingsworth, Kettle, Moore, Nelson, Rinck, Rivera, Saka, Solomon, Strauss

Opposed: None

H. COMMITTEE REPORTS

CITY COUNCIL:

1. <u>Res 32165</u> A RESOLUTION adopting the Statements of Legislative Intent for the 2025 Adopted Budget and 2025-2030 Adopted Capital Improvement Program.

Motion was made by Council President Nelson and duly seconded to adopt Resolution 32165.

The Motion carried, and the Resolution (Res) was adopted by the following vote, and the President signed the Resolution (Res): **In Favor:** 9 - Hollingsworth, Kettle, Moore, Nelson, Rinck, Rivera, Saka, Solomon, Strauss

Opposed: None

SELECT BUDGET COMMITTEE:

2. <u>CB 120946</u> AN ORDINANCE relating to the Seattle Department of Transportation; amending Ordinance 127156, which adopted the 2025 Budget, including the 2025-2030 Capital Improvement Program (CIP); and lifting a proviso.

> The Committee recommends that City Council pass the Council Bill (CB). In Favor: 8 - Strauss, Rivera, Hollingsworth, Kettle, Moore, Nelson, Rinck, Saka Opposed: None

The Council Bill (CB) was passed by the following vote, and the President signed the Council Bill (CB):

In Favor: 9 - Hollingsworth, Kettle, Moore, Nelson, Rinck, Rivera, Saka, Solomon, Strauss

Opposed: None

I. ITEMS REMOVED FROM CONSENT CALENDAR

There were none.

J. ADOPTION OF OTHER RESOLUTIONS

There were none.

K. OTHER BUSINESS

By unanimous consent Councilmember Saka was excused from the April 22, 2025, City Council meeting.

By unanimous consent, Councilmembers Moore and Strauss were excused from the April 1, 2025, City Council meeting.

L. ADJOURNMENT

There being no further business to come before the Council, the meeting was adjourned at 2:31 p.m.

Jodee Schwinn, Deputy City Clerk

Signed by me in Open Session, upon approval of the Council, on March 18, 2025.

Sara Nelson, Council President of the City Council



Legislation Text

File #: CB 120951, Version: 1

CITY OF SEATTLE

ORDINANCE

COUNCIL BILL

AN ORDINANCE appropriating money to pay certain claims for the week of March 3, 2025, through March 7, 2025, and ordering the payment thereof; and ratifying and confirming certain prior acts.
 BE IT ORDAINED BY THE CITY OF SEATTLE AS FOLLOWS:

Section 1. Payment of the sum of \$18,969,623.01 on PeopleSoft 9.2 mechanical warrants numbered 4100905120 - 4100906653 plus manual or cancellation issues for claims, e-payables of \$101,545.54 on PeopleSoft 9.2 9100015179 - 9100015189, and electronic financial transactions (EFT) in the amount of \$65,742,039.98 are presented to the City Council under RCW 42.24.180 and approved consistent with remaining appropriations in the current Budget as amended.

Section 2. RCW 35.32A.090(1) states, "There shall be no orders, authorizations, allowances, contracts or payments made or attempted to be made in excess of the expenditure allowances authorized in the final budget as adopted or modified as provided in this chapter, and any such attempted excess expenditure shall be void and shall never be the foundation of a claim against the city."

Section 3. Any act consistent with the authority of this ordinance taken prior to its effective date is ratified and confirmed.

Section 4. This ordinance shall take effect as provided by Seattle Municipal Code Sections 1.04.020 and 1.04.070.

Passed by the City Council the 18th of March, 2025, and signed by me in open session in authentication of its passage this 18th of March, 2025.

President	of the City Council
-----------	---------------------

Approved / returned unsigned / vetoed this _____ day of ______, 2025.

Bruce A. Harrell, Mayor

Filed by me this ______ day of ______, 2025.

Scheereen Dedman, City Clerk

(Seal)

SUMMARY and FISCAL NOTE

Department:	Dept. Contact:	CBO Contact:
Office of City Finance	Julie Johnson	Lorine Cheung

1. BILL SUMMARY

Legislation Title:

AN ORDINANCE appropriating money to pay certain claims for the week of March 3, 2025, through March 7, 2025, and ordering the payment thereof; and ratifying and confirming certain prior acts. Claims include all financial payment obligations for bills and payroll paid out of PeopleSoft for the covered.

Summary and Background of the Legislation:

RCW 42.24.180 requires that payment of certain claims be authorized by the City Council. This bill, prepared each week by the City Treasury, authorizes the payments of funds that were previously appropriated by the City Council, so the passage of this bill does not have a direct result on the City's budget.

2. CAPITAL IMPROVEMENT PROGRAM

Does this legislation create, fund, or amend a CIP Project?

3. SUMMARY OF FINANCIAL IMPLICATIONS

Does this legislation have financial impacts to the City?

This bill authorizes the payments of funds that were previously appropriated by the City Council, so the passage of this bill does not have a direct result on the City's budget.

If the legislation has costs, but they can be absorbed within existing operations, please describe how those costs can be absorbed. The description should clearly describe if the absorbed costs are achievable because the department had excess resources within their existing budget or if by absorbing these costs the department is deprioritizing other work that would have used these resources.

Please describe any financial costs or other impacts of *not* **implementing the legislation.** The legislation authorizes the payment of valid claims. If the City does not pay its legal obligations it could face greater legal and financial liability.

Yes No

4. OTHER IMPLICATIONS

a. Please describe how this legislation may affect any departments besides the originating department.

This type of legislation authorizes payment of bill and payroll expenses for all City departments.

- **b.** Does this legislation affect a piece of property? If yes, please attach a map and explain any impacts on the property. Please attach any Environmental Impact Statements, Determinations of Non-Significance, or other reports generated for this property. No.
- c. Please describe any perceived implication for the principles of the Race and Social Justice Initiative.
 - i. How does this legislation impact vulnerable or historically disadvantaged communities? How did you arrive at this conclusion? In your response please consider impacts within City government (employees, internal programs) as well as in the broader community. $N\!/\!A$
 - ii. Please attach any Racial Equity Toolkits or other racial equity analyses in the development and/or assessment of the legislation. N/A
 - iii. What is the Language Access Plan for any communications to the public? $N\!/\!A$
- d. Climate Change Implications
 - i. Emissions: How is this legislation likely to increase or decrease carbon emissions in a material way? Please attach any studies or other materials that were used to inform this response. N/A
 - Resiliency: Will the action(s) proposed by this legislation increase or decrease Seattle's resiliency (or ability to adapt) to climate change in a material way? If so, explain. If it is likely to decrease resiliency in a material way, describe what will or could be done to mitigate the effects. N/A
- e. If this legislation includes a new initiative or a major programmatic expansion: What are the specific long-term and measurable goal(s) of the program? How will this legislation help achieve the program's desired goal(s)? What mechanisms will be used to measure progress towards meeting those goals? N/A

5. CHECKLIST

	Is a public hearing required?
	Is publication of notice with <i>The Daily Journal of Commerce</i> and/or <i>The Seattle Times</i> required?
	If this legislation changes spending and/or revenues for a fund, have you reviewed the relevant fund policies and determined that this legislation complies?
	Does this legislation create a non-utility CIP project that involves a shared financial commitment with a non-City partner agency or organization?
6. A7	TACHMENTS

Summary Attachments: None.



Legislation Text

File #: Appt 03101, Version: 1

Reappointment of Carmen A. Bendixen as member, Seattle Public Library Board of Trustees, for a term to April 1, 2030.

The Appointment Packet is provided as an attachment.

City of Seattle Boards & Commissions Notice of Appointment

Appointee Name:			
Carmen A. Bendixen			
Board/Commission Name:		Position	Title:
Seattle Public Library Board of Trustees		Trustee	
	City Council Co	nfirmation re	quired?
Appointment <i>OR</i> 🛛 Reappointment	🛛 Yes		
	🗌 No		
Appointing Authority:	Term of Positio	n: *	
City Council	4/2/2025		
Mayor	to		
Other: Fill in appointing authority	4/1/2030		
	🗌 🗆 Serving remai	ning term of a	vacant position
Residential Neighborhood:	Zip Code:	Contact Phor	<u>n</u> e No. <i>:</i>
Green Lake	98103		

Background:

Bendixen was appointed to the Library Board by Mayor Jenny Durkan in April 2020 and has served as an active and engaged Trustee for five years, during which time she has held the positions of board president and vice president, as well as serving on the Executive Search Committee during the national search to hire a new Executive Director/Chief Librarian in 2022. She has played an active role in guiding policy and budget issues as well as development of the Library's 2024-2033 Strategic Plan.

Prior to serving on the Library Board of Trustees, Bendixen was a member of the Friends of The Seattle Public Library, where she served as board president, vice president, and secretary. On behalf of the Friends of the Library, Bendixen helped lead advocacy efforts during the 2012 and 2019 Library Levy campaigns. She has also been a Welcome Desk and events volunteer at the Central Library, and is a regular user of the Greenwood and Green Lake branches.

Bendixen has been employed at Washington State Ferries for over 8 years, where her responsibilities have included transportation and land use planning, community outreach, and public policy analysis. Bendixen has a bachelor's degree from Willamette University, and a master's degree in City Planning from the University of Pennsylvania.

Authorizing Signature (original signature):

Date Signed (appointed): February 13th., 2025 **Appointing Signatory:** *Bruce A. Harrell*

Mayor of Seattle

*Term begin and end date is fixed and tied to the position and not the appointment date.

Carmen A. Bendixen

Volunteer and Professional Activities

The Seattle Public Library, Board of Trustees, 2020-present

- President: 2022-2023
- Vice President: 2021

Friends of The Seattle Public Library, Board

- President: 2015-2017
- Vice President: 2014-2015
- Board Member: 2011-2020

Friends of The Seattle Public Library, FriendShop volunteer, 2007-2020 The Seattle Public Library, Central Library Events Volunteer, 2007-2020

Experience

Senior Transportation Planner,

Washington State Department of Transportation, Ferries Division (WSF), Seattle, WA

- Steward the development and implementation of WSF's plans, including 2040 Long Range Plan
- Supply data/analysis support for planning, service planning, and customer service areas
- Represent WSF on various internal and external workgroups and with regional planning organizations
- Analyze complex transportation planning data, perform analysis, and make recommendations to all levels of WSDOT and WSF groups and managers

Environmental Planner, CH2M Hill (now Jacobs), Bellevue, WA

- Assisted with preparation of reader-friendly environmental documents, including writing technical reports and sections for environmental impact statements (EISs) and other environmental compliance documents
- Served on project management team for high-profile EIS project
- Assisted with GIS and data coordination for EIS analyst team
- Assisted with quantitative and qualitative land use and transportation analysis for both public and private sector clients

Environmental Planner, Washington State Department of Natural Resources, Olympia, WA 2013-2015

- Served as a senior planner for the DNR's Long-Term Conservation Strategy (LTCS) for the marbled murrelet, a threatened species under the Endangered Species Act (ESA), and for DNR's other major policy planning project, the Sustainable Harvest Calculation
- Coordinated the development of an EIS to be utilized by DNR in its status as SEPA lead agency, including coordinating data needs for technical experts, and facilitating communications among team members
- Interpreted complicated SEPA and NEPA policy and procedural issues ranging from working on the LTCS EIS to providing interpretation of SEPA and NEPA to Executive Management

Transportation Planner, ICF International, Seattle, WA

- Served as project coordinator and supported the project team on major infrastructure projects, including but not limited to data management, meeting coordination, preparing presentation materials, and team communication
- Assisted with preparation of reader-friendly environmental documents, including writing transportation, land use, and parks sections for EISs, for projects in Washington and California
- Prepared GIS-based maps for various planning and environmental documents including transportation plans, EISs, and comprehensive plans
- Assisted with quantitative and qualitative air quality, noise, environmental protection, and transportation analysis for both public and private sector clients

Transportation Planner, Transpo Group, Kirkland, WA

- Prepared non-motorized transportation plans including ArcView spatial analyses of pedestrian and bicycle facilities
- Created GIS-based maps for various planning documents including transportation plans, concurrency policies,

2016-present

2015-2016

2009-2012

2006-2008

2

and transportation impact fee policies

- Collaborated on transportation plan updates and on SEPA documentation for public sector clients
- Prepared traffic impact studies, wrote memoranda and reports, assisted with Synchro analysis, performed field work for private and public development projects

Planning Intern, University City District, Philadelphia, PA

- Led weekly planning or urban history lesson and/or activity at local junior high school
- Collected and analyzed demographic and neighborhood information for annual report
- Carried out tasks and projects for Director of Research & Planning, such as analyzing new neighborhood development

Planning Intern, Division of Facilities & Real Estate Services, University of Pennsylvania, Philadelphia, PA 2004-2005

- Researched issues, including transportation and sustainable campus planning
- Prepared memoranda on topics such as long-term campus planning for the Office of the University Architect

Legislative Staff, Office of Senator Barbara Boxer, Washington, DC

Legislative Aide 2003-2004; Legislative Correspondent 2000-2003

- Drafted legislation and monitored legislative action related to transportation, energy, Native Americans, housing, and other issues as necessary
- · Wrote policy memoranda and summary documents for the Senator and senior legislative staff
- Prepared committee hearing and Senate floor materials, including witness questions and talking points
- Attended constituent and issue meetings with the Senator and senior staff
- Met with constituent and interest groups regarding a variety of issues
- Researched special topics for the legislative director
- Communicated information for the press office and state operations

Other Experience

- Congressional Hispanic Caucus Institute Legislative Fellow, Office of Rep. Jay Inslee, Washington, DC 2000
- Congressional Hispanic Caucus Institute Legislative Fellow, Office of Senator Patty Murray, Washington, DC 1999-2000
- AmeriCorps Volunteer & Assistant Program Coordinator, Hood Canal Watershed Project Center, Belfair, WA 1998-1999

Education

Master of City Planning, University of Pennsylvania, Philadelphia, Pennsylvania

Concentration: Land Use and Transportation Planning

Bachelor of Arts, Willamette University, Salem, Oregon

Major: Politics; Minor: African American Studies

2005-2006

2000-2004

Seattle Public Library Board of Trustees

5 Members: Pursuant to RCW 27.12, City Charter Article XII, all members subject to City Council confirmation, 5year terms:

- # City Council-appointed
- 5 Mayor-appointed
- # Other Appointing Authority-appointed (specify):

D		c	÷		r	•
	U	2	L	C		•

*D	**G	RD	Position No.	Position Title	Name	Term Begin Date	Term End Date	Term #	Appointed By
3	F	6	1.	Trustee	Carmen A. Bendixen	4/2/2025	4/1/2030	2	Mayor
1	F	7	2.	Trustee	Faith Pettis	4/2/2024	4/1/2029	1	Mayor
2	м	2	3.	Trustee	Dr. W. Tali Hairston PhD	4/2/2023	4/1/2028	2	Mayor
9	F	7	4.	Trustee	Yazmin Mehdi	4/2/2022	4/1/2027	1	Mayor
1	м	2	5.	Trustee	Ron Chew	4/2/2021	4/1/2026	2	Mayor

SELF-IDENTIFIED DIVERSITY CHART				(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	
	Male	Female	Transgender	NB/ O/ U	Asian	Black/ African American	Hispanic/ Latino	American Indian/ Alaska Native	Other	Caucasian/ Non- Hispanic	Pacific Islander	Middle Eastern	Multiracial
Mayor	2	3			2	1	1						1
Council													
Other													
Total													

Key:

*D List the corresponding Diversity Chart number (1 through 9)

**G List gender, M= Male, F= Female, T= Transgender, NB= Non-Binary O= Other U= Unknown

RD Residential Council District number 1 through 7 or N/A

Diversity information is self-identified and is voluntary.



Legislation Text

File #: Appt 03104, Version: 1

Reappointment of Evan M. Smith as member, Families, Education, Preschool, and Promise Levy Oversight Committee, for a term to December 31, 2026.

The Appointment Packet is provided as an attachment.

City of Seattle Boards & Commissions Notice of Appointment

Appointee Name:			
Evan M. Smith			
Board/Commission Name: Families, Educati	on, Preschool, and	Position Title:	
Promise Levy Oversight Committee		Member	
	City Council Conf	irmation required?	
Appointment OR 🛛 Reappointment	🖂 Yes		
	🗌 No		
Appointing Authority:	Term of Position:	*	
City Council	1/1/2025		
Mayor	to		
Other: Fill in appointing authority	12/31/26		
	Serving remainii	ng term of a vacant position	
Residential Neighborhood:	Zip Code: Co	ontact Phone No. <i>:</i>	
Leschi	98122		
Background:	·		

My interest in joining a Seattle Commission is twofold. Professionally, I have had the opportunity to work across the public, non-profit, private and political sectors. I have a wealth of experiences that inform a broader perspective of how each of these worlds overlap to form a healthy and dynamic ecosystem. Personally, as a partner in a multi-racial marriage, a father of a multi-racial daughter and a citizen of Seattle, I am driven to serve my community today to continue to shape an even better and more equitable future for my family and my fellow citizens.

Authorizing Signature (original signature):	Appointing Signatory:
	Bruce A. Harrell
$(\mathcal{A} \mathcal{A} \mathcal{M} \mathcal{O} \mathcal{V})$	Mayor of Seattle
V Innell. Howell	
Data Signad (appainted):	
Date Signed (appointed): November 26 th 2024	

EVAN M. SMITH

OVERVIEW

I am a leader of people and organizations with twenty years of diverse experience across functions, organizational sizes, sectors and geographies. The core of my work has been to lead people and organizations, in the public, nonprofit, and private sectors, through dramatic change in pursuit of radical results, inspiring teams to achieve more than they thought possible. My experience spans strategy leadership, operational accountability, organizational transformation, and delivering goals.

Vice President, Technology Strategy and Business Transformation

EXPERIENCE

2017 – Present Starbucks

Seattle, WA

- Direct report to CTO, serve as Chief of Staff, leading team of 30 (5 direct reports), responsible for developing and integrating technology strategy, driving business operations and internal and external communications.
- Drive annual strategic, operational, and financial planning to allocate nearly \$1B in technology spend against business and technology imperatives to direct the work of 1,700+ technologists.
- Built Inclusion and Diversity (I&D) strategy for the technology group, standing up I&D council and driving forward strategy to increase URM talent pipeline and enhance current employee experience.
- Achieved significant year-over-year growth in team satisfaction scores with my team's employee engagement scores significantly above the company average.

General Manager, Starbucks Delivery

- Operated Starbucks delivery program, owning day-to-day execution; directed partnership with Uber Eats, while coordinating 200+ Starbucks partners across marketing, retail operations, finance, product, packaging, R&D, technology, data, and analytics departments to drive responsible, high-impact program footprint and revenue growth; recognized for performance with leadership award from Starbucks CEO.
- Scaled delivery channel from one market to national scale, developing and driving the technological, operational, and marketing roadmaps to optimize program for enterprise execution and business impact, ultimately growing revenues from 0 to >5% of US revenues with line of sight to \$2B annual revenues.

Director, Global Corporate Strategy

- Designed and executed end-to-end Starbucks delivery pilot from conception to launch; built and secured approval of business case from CEO, COO, and CMO; drove from ideation to pilot launch in 75 days, managing and coordinating cross-functional team of 150+ Starbucks and Uber Eats partners; concept execution proved potential value of +\$2B for enterprise, leading CEO to fast-track national roll-out.
- Created annual corporate strategic plan, managing team and process to develop and distill C-suite vision into 5-year enterprise strategy for presentation to Board of Directors and broader enterprise alignment.
- Led strategic re-think of the Sourcing Department, designing, and directing cross-functional working teams of VPs to identify and ultimately achieve a 22% increase in annual sourced raw material savings.

2016 – 2017 Educents

Director, Strategy and Business Operations

- Member of a 7-person executive team with an early-stage education products, e-commerce start-up.
- Drove market and customer insights, honed and articulated focused company strategy, aligned company growth targets and strategies across finance, marketing, data and operations functions.
- Developed and directed quarterly key performance and objectives planning, and company target-setting.

2012 - 2015 McKinsey and Company

Engagement Manager 2014 - 2015

Associate 2012-2013

Led teams serving government and nonprofit clients in the education sector, focused on strategy and operations.

Operational improvement

Managed a team of senior City and School District leaders in a major urban school district to perform a
full operational and financial review, presenting opportunities to the mayor and a new superintendent.

Organizational design and transformation

• Led CEO and executive team of national education nonprofit to design dynamic organization and financial models, benchmarking best practices and developing financial model scenarios.

Strategy development

• Led a Dubai-based team to develop a national education system strategy, aligning public budgets with labor market needs and opportunities with senior government leaders in a major Middle Eastern country.

San Francisco, CA

Washington, DC

Redesigned and drove the annual budgeting process, building an automated online system to allocate balancing the needs of diverse stakeholders from various school communities. Managed a team of 50 supporting day-to-day school-level security, facilities, budget, and enrollment operations through a customer service delivery model, with school principals as primary customers. Director, School Opening Led planning and executing annual fall opening of all district schools; led cross-functional working teams to streamline critical operations, achieving fall opening described by the Washington Post as "uncommonly quiet" and a 30% year-over-year increase in measures of school opening readiness. Intergovernmental Liaison Mitch Landrieu for Mayor Campaign, Deputy Press Secretary New Orleans, LA Thibodaux, LA Federal Emergency Management Agency, Manager, Disaster Recovery Center New Orleans Public Schools, High School Social Studies Teacher New Orleans, LA **EDUCATION / DEVELOPMENT** Presidential Leadership Scholars Fellowship Various, USA 2023 Scholar Highly competitive national leadership development program sponsored by Presidents Clinton & Bush Darden Graduate School of Business Administration, University of Virginia Charlottesville, VA Master of Business Administration, 2012 Elected by peers as Student Body President Recipient of C. Stewart Sheppard Distinguished Service Award University of Cambridge Cambridge, England Master of Philosophy in Economic and Social History, High First Degree, August 2007 University of Pennsylvania Philadelphia, PA Bachelor of Arts in History and Urban Studies, Political Science minor, Cum Laude, May 2004 **VOLUNTEER / OTHER**

2007-2010

- Mayoral-Appointed Member, Seattle Families, Education, Preschool and Promise (FEPP) Levy Oversight Committee (2022 – present)
- Leadership Tomorrow Fellow, Class of 2022. Competitive Seattle-based leadership program.
- Board Officer, Schools First (2022 present). CBO focused on advocating for school levy passage.
- Board Director, College Success Foundation (2021 present)
 - Chairperson, Marketing and Development Committee
- Governor-Appointed Member, Washington State STEM Education Innovation Alliance (2020 present)

District of Columbia Public Schools Director, Office of School Operations

\$800M efficiently and accurately across all DC public schools aligned to district policy priorities, while

2006 2005 2004-2005

Families, Education, Preschool, and Promise Levy Oversight Committee

17 Members: Pursuant to Ordinance 125604, 12 members subject to City Council confirmation, staggered-year terms:

- 6 City Council-appointed 3-year terms, subject to City Council confirmation
- 6 Mayor-appointed 3-year terms, subject to City Council confirmation
 - 5 Other Appointing Authority-appointed (specify): Ordinance 125604

Roster:

*D	**G	RD	Position No.	Position Title	Name	Term Begin Date	Term End Date	Term #	Appointed By
1	F	2	1.	Member	Erin Okuno	1/1/22	12/31/25	3	Council
6	F		2.	Member	Natalie Beauregard	1/1/23	12/31/25	1	Council
3	F	2	3.	Member	Manuela Slye	1/1/21	12/31/24	1	Council
			4.	Member	Vacant				Council
2	М	2	5.	Member	Donald Felder	1/1/22	12/31/24	2	Council
			6.	Member	Vacant				Council
			7.	Member	Vacant				Mayor
2	F		8.	Member	Linda Thompson-Black	1/1/25	12/31/26	2	Mayor
1	F	N/A	9.	Member	Susan Yu Yi Lee	1/1/23	12/31/25	2	Mayor
2	М		10.	Member	Marques Gittens	1/1/23	12/31/25	2	Mayor
2	F	4	11.	Member	Stephanie R. Gardner	1/1/24	12/31/26	2	Mayor
6	м		12.	Member	Evan M. Smith	1/1/25	12/31/26	2	Mayor
9	м	N/A	13.	Mayor	Bruce Harrell	N/A	N/A	1	Ordinance 125604
3	F	N/A	14.	Library, Education and Neighborhoods Committee	Maritza Rivera	N/A	N/A	1	Ordinance 125604
2	м	N/A	15.	School District Superintendent	Brent Jones	N/A	N/A	1	Ordinance 125604
9	F	N/A	16.	School District Board Member	Sarah Clark	N/A	N/A	1	Ordinance 125604
1	F	N/A	17.	Chancellor of Seattle Colleges	Rosie Rimando-Chareunsap	N/A	N/A	1	Ordinance 125604

SELF	-IDEN	FIFIED I	DIVERSITY (CHART	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
	Male	Female	Transgender	NB/ O/ U	Asian	Black/ African American	Hispanic/ Latino	American Indian/ Alaska Native	Other	Caucasian/ Non- Hispanic	Pacific Islander	Middle Eastern	Multiracial
Mayor	2	3			1	3				1			
Council	1	4			1	1	1			1			
Other	2	3			1		1			1			1
Total	5	10			3	4	2			3			1

(-)

/ - >

....

()

()

Key:

*D List the corresponding *Diversity Chart* number (1 through 9)

**G List gender, M= Male, F= Female, T= Transgender, NB= Non-Binary O= Other U= Unknown

RD Residential Council District number 1 through 7 or N/A

Diversity information is self-identified and is voluntary.

SEATTLE CITY COUNCIL



Legislation Text

File #: Appt 03105, Version: 1

Reappointment of Linda Thompson-Black as member, Families, Education, Preschool, and Promise Levy Oversight Committee, for a term to December 31, 2026.

The Appointment Packet is provided as an attachment.

City of Seattle Boards & Commissions Notice of Appointment

Appointee Name:								
Linda Thompson-Black								
Board/Commission Name:	_	Position Title:						
Families, Education, Preschool, and Promise Levy Oversi	ght Committee	Member						
	City Council Co	nfirmation required?						
Appointment <i>OR</i> Reappointment	X Yes							
	No No							
Appointing Authority:	Term of Position: *							
City Council	1/1/2025							
Mayor	to							
Other: Fill in appointing authority	12/31/2026							
	🗆 Serving remai	ning term of a vacant position						
Residential Neighborhood:	Zip Code:	Contact Phone No.:						
Mount Baker	98118							
Background:	· ·							
I was on Mayor Norm Rice's team and led Education Initiatives. I staffed the original Education Summit								

after a contentious antibusing initiative passed and laid the groundwork for the first Families and Education Levy. It was my job to fulfill his commitment to bring the community together and develop an education agenda. I was recruited because of my previous work on the management teams of the zoo and aquarium when they were preparing for bond issues.

Authorizing Signature (original signature):	Appointing Signatory:
ρ ρ / ρ	Bruce A. Harrell
Burel. Hornell	Mayor of Seattle
Date Signed (appointed): February 12 th , 2025	

LINDA THOMPSON-BLACK



Strategic Operations Planner/ Collaborative Leader/ Financial and Program Administrator

Accomplished Education and Human Services professional with a strong reputation for: solving problems, successfully developing new organizations and restructuring existing organizations. Works well with stakeholders, visionary, and results oriented. Excellent communication skills committed to achieving organizational objectives. Areas of expertise:

- Resource Development
- Team Building
- Government Relations
- Project Management
- Program Development & Management
- Talent Development
- Data Driven Decision Making
- Network Investment Program
 Officer/Contract Management
- Strategic Planning
- Change Management
- Implementing a Total Quality Systems Program
- Non-profit Management
- Community Outreach & Development

PROFESSIONAL EXPERIENCE

UNCF Seattle Area Development Director 2016- to the present

UNCF is the nation's largest and most effective minority education organization started in 1944. In the role of Area Development Director I am the local community ambassador developing advocacy and resource development campaigns to support minority education. I develop partnerships with corporations, Foundations, Seattle Public Schools and Community Organizations across the City.

COMMUNITIES IN SCHOOLS NATIONAL, Arlington VA, 1995 - 2015 Director of Field Operations, (1999 - 2015) Associate Director, Senior Trainer, (1995 - 1999)

Started new affiliates and rebuilt struggling state and local affiliate organizations. Facilitated win/win solutions in politically charged situations. The network has annual expenditures and revenues of \$247 million with my portfolio ranging from \$15-25 million. Provided leadership and management of CIS network investments. CIS network investments total \$20 million. Established board development training, developed proposal criteria, led resource development campaigns, allocated investments, facilitated the contract process, and monitored compliance on budgets.

Led development activities as a local executive director. Implemented a diversified resource development plan. A major responsibility was cultivating individual donors. This included a variety of special events such as galas, gatherings at the homes of high net worth individuals, golf tournaments, walks and runs. As a regional director and I assisted state and local affiliates with establishing signature events such as the Education Ball with the Governor's Office, The Scare Crow Festival in Las Vegas, luncheon and breakfast events. I developed relationships with hundreds of potential donors and table captains. As the Seattle Executive Director I developed a relationship with COSTCO. They supported the CIS Seattle Homeless site. The relationship blossomed and they now provide a grant to the CIS network of approximately \$1million dollars annually. I wrote grants and taught resource development for the CIS network. I am also a member of the Greater Seattle Chapter of the Links Inc., and the Seattle Chapter of the Girlfriends. Seattle just hosted the Girlfriends Inc. in May 2016. There were 600 people in

attendance. This included amazing events such as the opening reception, The Emerald Martini Club, the All White Affair and the Emerald Ball. The Links have the annual Evening of Enchantment. The Girlfriends have the Paint the Town Red event. I have extensive experience with event planning.

- Led a network of 200 independent affiliates to implement a unified approach, moving to one consistent evidenced-based model and the use of best non-profit management practices.
- Effectively implemented a change management process, convincing affiliates to implement fifty standards that required proof of compliance. Developed a process to prepare them for accreditation and assembled accreditation teams to create a process for re-accreditation.
- Successfully implemented Total Quality Systems (TQS), accrediting 160 affiliates and provided strategic planning facilitation for over 30 affiliates resulting in more effective leadership, improved student outcomes, and increased funding.
- Led the replication and expansion process for CIS, creating a strategic plan, establishing a taskforce, board and fundraising plan and hiring a local liaison. As a result of these initiatives, the network grew by 260% (from serving 500,000 students to 1.3 million).
- Developed partnerships and partnership agreements to support students and families by working with pre-school providers, health and mental health organizations, universities, homeless agencies, mentor/tutor organizations, food banks, and gang prevention programs. As a result of this successful approach the CIS Network currently has over 20,000 partnerships across the country.
- Led a very challenging process to re-develop the CIS Nevada State office. Facilitated a leadership change of the state director and board, identified new leadership for the board

LINDA THOMPSON-BLACK

Page 2

and staff and worked with the new director to development a re-organization plan that lead to our national board chair, Elaine Wynn, providing a \$1 million investment.

- Facilitated data-driven/continuous improvement systems with internal and network-wide teams to elevate quality non-profit business practices.
- Worked with the National Center on Drop-out Prevention in Clemson to develop a wide array of summary briefs on the characteristics of best practice programs that address risk factors such as after school programs, and mentoring programs so that affiliates can assess the efficacy of potential partnering organizations. Information was developed to train affiliates on the use of these tools.
- Successfully implemented organizational capacity building plans to improve state and affiliate operations. Provided direct technical assistance and worked with consultants.
- Insured that there are appropriate fiscal controls for state offices and affiliates that conform to established fiscal management policies.
- Regularly monitored the performance and financial management of state directors and affiliates in the network. Reviewed budgets quarterly, establishing an early warning system to address financial challenges.
- Managed grants such as a federal OJJDP grant on gang prevention, and one on tutoring and mentoring in isolated communities, the Windsong Foundation and Costco grants.

ADDITIONAL PROFESSIONAL EXPERIENCE

SEATTLE COMMUNITIES IN SCHOOLS AFFILIATE, Seattle, WA Executive Director

CITY OF SEATTLE, Seattle, WA Special Assistant to the Mayor, Councilmember on Education and Human Services

WOODLAND PARK ZOO AND SEATTLE AQUARIUM, Seattle, WA Business Manager

CITY OF SEATTLE PARKS AND RECREATION, Seattle, WA Capital Projects Manager

EDUCATION

MPA, Organizational Development, Program Analysis and Budgeting, University of Washington, Seattle, WA

BA, Political Science and Communications, University of Washington, Seattle, WA

PROFESSIONAL DEVELOPMENT/AFFILIATIONS

Certificate, Board Development and Leadership Project Management Change Management Effectively Dealing With Difficult Clients and Customers Founding Member, Seattle Chapter of the Girlfriends Inc. Member, The Links Inc., Seattle Chapter

Families, Education, Preschool, and Promise Levy Oversight Committee

17 Members: Pursuant to Ordinance 125604, 12 members subject to City Council confirmation, staggered-year terms:

- 6 City Council-appointed 3-year terms, subject to City Council confirmation
- 6 Mayor-appointed 3-year terms, subject to City Council confirmation
 - 5 Other Appointing Authority-appointed (specify): Ordinance 125604

Roster:

*D	**G	RD	Position No.	Position Title	Name	Term Begin Date	Term End Date	Term #	Appointed By
									-
1	F	2	1.	Member	Erin Okuno	1/1/22	12/31/25	3	Council
6	F		2.	Member	Natalie Beauregard	1/1/23	12/31/25	1	Council
3	F	2	3.	Member	Manuela Slye	1/1/21	12/31/24	1	Council
			4.	Member	Vacant				Council
2	М	2	5.	Member	Donald Felder	1/1/22	12/31/24	2	Council
			6.	Member	Vacant				Council
			7.	Member	Vacant				Mayor
2	F		8.	Member	Linda Thompson-Black	1/1/25	12/31/26	2	Mayor
1	F	N/A	9.	Member	Susan Yu Yi Lee	1/1/23	12/31/25	2	Mayor
2	М		10.	Member	Marques Gittens	1/1/23	12/31/25	2	Mayor
2	F	4	11.	Member	Stephanie R. Gardner	1/1/24	12/31/26	2	Mayor
6	М		12.	Member	Evan M. Smith	1/1/25	12/31/26	2	Mayor
9	М	N/A	13.	Mayor	Bruce Harrell	N/A	N/A	1	Ordinance 125604
3	F	N/A	14.	Library, Education and Neighborhoods Committee	Maritza Rivera	N/A	N/A	1	Ordinance 125604
			4-	School District			N. / A		Ordinance
2	M	N/A	15.	Superintendent	Brent Jones	N/A	N/A	1	125604
9	F	N/A	16.	School District Board Member	Sarah Clark	N/A	N/A	1	125604
1	F	N/A	17.	Chancellor of Seattle Colleges	Rosie Rimando-Chareunsap	N/A	N/A	1	Ordinance 125604

SELF	-IDEN	FIFIED I	DIVERSITY (CHART	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
	Male	Female	Transgender	NB/ O/ U	Asian	Black/ African American	Hispanic/ Latino	American Indian/ Alaska Native	Other	Caucasian/ Non- Hispanic	Pacific Islander	Middle Eastern	Multiracial
Mayor	2	3			1	3				1			
Council	1	4			1	1	1			1			
Other	2	3			1		1			1			1
Total	5	10			3	4	2			3			1

(-)

/ - >

....

()

()

Key:

*D List the corresponding *Diversity Chart* number (1 through 9)

**G List gender, M= Male, F= Female, T= Transgender, NB= Non-Binary O= Other U= Unknown

RD Residential Council District number 1 through 7 or N/A

Diversity information is self-identified and is voluntary.



Legislation Text

File #: CB 120933, Version: 1

CITY OF SEATTLE

ORDINANCE

COUNCIL BILL _____

AN ORDINANCE relating to land use and zoning; amending Sections 23.74.002 and 23.74.008 of the Seattle Municipal Code to allow residential uses in the Stadium Transition Area Overlay District.
 BE IT ORDAINED BY THE CITY OF SEATTLE AS FOLLOWS:

Section 1. Section 23.74.002 of the Seattle Municipal Code, last amended by Ordinance 126862, is

amended as follows:

23.74.002 Purpose, intent, and description of the overlay district-Rezone requirement-Rezone criteria

A. Purpose and intent. The purpose of this Chapter 23.74 is to implement the City's Comprehensive Plan, including the neighborhood plan for the Greater Duwamish Manufacturing/Industrial Center, by establishing a Stadium Transition Area Overlay District for the area shown on Map A for 23.74.004. The Stadium Transition Area centers on large sports facilities and allows uses complementary to them. It is intended to contribute to a safer pedestrian environment for those attending events and permits a mix of uses, supporting the pedestrian-oriented character of the area as well as the surrounding industrial zone, while minimizing conflicts with industrial uses. Within the overlay district, use provisions and development standards are designed to: create a pedestrian connection with downtown; discourage encroachment on nearby industrial uses to the south; and create a pedestrian-friendly streetscape. Allowing a mix of uses, including office development, restaurants, lodging, <u>residential uses</u>, and maker uses and arts, is intended to encourage redevelopment and to maintain the health and vibrancy of the area during times when the sports facilities are not in operation.

* * *
File #: CB 120933, Version: 1

Section 2. Section 23.74.008 of the Seattle Municipal Code, last amended by Ordinance 126862, is amended as follows:

23.74.008 Uses.

Notwithstanding the use provisions of the underlying zone, the following use provisions apply:

- A. The following uses are permitted in buildings existing on June 1, 2023:
 - 1. Artist's studio/dwellings;
 - 2. Major institutions.
- B. The following uses are prohibited:
 - 1. Heavy manufacturing uses;
 - 2. High-impact uses;
 - 3. Solid waste management;
 - 4. Recycling uses;
 - 5. Animal shelters and kennels;
 - 6. Veterinary offices;
 - 7. Pet grooming;
 - 8. Airports, land and water based;
 - 9. Hospitals;
 - 10 Elementary and secondary schools;
 - 11. Drive-in businesses, including gas stations;
 - 12. Bus bases;
 - 13. Flexible-use parking¹; and
 - 14. Residential uses, except for those allowed under SMC 23.74.008.C. ((otherwise allowed as

an administrative conditional use in the Urban Industrial zone pursuant to subsection 23.50A.062.C)).

C. In areas zoned Urban Industrial, residential uses are permitted as a conditional use pursuant to the

File #: CB 120933, Version: 1

criteria contained in subsection 23.50A.062.C, except that criterion 23.50A.062.C.3 does not apply within the Stadium Transition Area Overlay District.

¹ Parking required for a spectator sports facility or exhibition hall is allowed and shall be permitted to be used for flexible-use parking or shared with another such facility to meet its required parking. A spectator sports facility or exhibition hall within the Stadium Transition Overlay Area District may reserve non-required parking only outside the overlay district and only if:

(a) The parking is owned and operated by the owner of the spectator sports facility or exhibition hall;

and

(b) The parking is reserved for events in the spectator sports facility or exhibition hall; and

(c) The reserved parking is south of South Royal Brougham Way, west of 6th Avenue South and north of

South Atlantic Street. Parking that is provided to meet required parking will not be considered reserved parking.

Section 3. This ordinance shall take effect as provided by Seattle Municipal Code Sections 1.04.020 and 1.04.070.

Passed by the City Council the	_day of		, 2025, and signed by
me in open session in authentication of its passa	ge this	day of	, 2025.

President _____ of the City Council

Approved / returned unsigned / vetoed this _____ day of ______, 2025.

Bruce A. Harrell, Mayor

Filed by me this ______ day of ______, 2025.

Scheereen Dedman, City Clerk

(Seal)

SUMMARY and FISCAL NOTE

Department:	Dept. Contact:	CBO Contact:
Legislative	Lish Whitson/206-615-1674	N/A

1. BILL SUMMARY

Legislation Title:

AN ORDINANCE relating to land use and zoning; amending Sections 23.74.002 and 23.74.008 of the Seattle Municipal Code to allow residential uses in the Stadium Transition Area Overlay District.

Summary and Background of the Legislation:

This bill would amend the Stadium Transition Area Overlay District (STAOD) to allow housing as a conditional use within the Urban Industrial (UI) zone. Most of the STAOD is zoned UI, with the exception of a block north of S Royal Brougham Way and west of 1st Avenue S, known as the WOSCA block, which is zoned MML.

According to Seattle Municipal Code Section 23.34.099, UI zoning is generally intended to provide "an integrated and healthy transition between core industrial areas and neighboring urban villages, residential, and mixed-use areas. These areas contain a mix of affordable, small-scale places for light industry, makers, brewing and distilling, creative arts, and industry supporting ancillary retail, office, or research activity. This area also provides limited opportunities for workforce housing that supports industrial uses. The area functions as a place for residents and workers from nearby urban villages or centers to patronize and experience unique local industrial businesses."

With the adoption of Industrial and Maritime zoning in 2023, housing is now allowed as a conditional use in most UI zones. However, the STAOD regulations, in SMC section 23.74.008 prohibit housing in the district. In addition, the conditional use criteria for housing in UI districts, SMC 23.50A.062.C, prohibit housing within 200 feet of a Major Truck Street. Within the STAOD all of the arterials are designated as Major Truck Streets including 1st Avenue S, S Royal Brougham Way, Edgar Martinez Drive S, and S Holgate Street, meaning that most of the STAOD is within 200 feet of major truck streets. During consideration of Ordinance 126862, which adopted these regulations, it was argued that housing within the STAOD would conflict with efficient movement of goods through the area and may have negative impacts on the Port of Seattle's operations.

This bill would remove the prohibition on housing in the STAOD and would allow housing within 200 feet of a major truck street within the STAOD. As a result, housing would be permitted in the STAOD if it complies with the other conditions of SMC 23.50A.060:

A. The use shall be determined not to be materially detrimental to the public welfare or injurious to property in the zone or vicinity in which the property is located.

B. The benefits to the public that would be provided by the use shall outweigh the negative impacts of the use.

C. Landscaping and screening, vehicular access controls, and other measures shall insure the compatibility of the use with the surrounding area and mitigate adverse impacts.

D. The conditional use shall be denied if it is determined that the negative impacts cannot be mitigated satisfactorily. However, adverse negative impacts may be mitigated by imposing requirements or conditions deemed necessary for the protection of other properties in the zone or vicinity and the public interest.

And the conditions of SMC 23.50A.062.C:

Residential use in UI zones. Residential uses are permitted as an administrative conditional use in UI zones if all of the following criteria are met. The residential use may be part of a Major Phased Development.

- 1. The residential use shall not exceed a density limit of 50 dwelling units per acre; and
- 2. The residential use shall not be located within 200 feet of a shoreline; and
- 3. [NOTE: Subsection 3, related to major truck streets, would not apply.]
- 4. All dwelling units shall have sound-insulating windows sufficient to maintain interior sound levels at 60 decibels or below in consideration of existing environmental noise levels at the site. The applicant shall submit an analysis of existing noise levels and documentation of the sound insulating capabilities of windows as part of the conditional use permit application; and
- 5. All dwelling units shall have a permanently installed air cooling system and a balanced ventilation system, which may be combined. The ventilation system shall filter any outdoor air supply through filters rated MERV 13 or higher as determined by the American Society of Heating, Refrigerating, and Air Conditioning Engineers (ASHRAE). The air cooling and ventilation systems shall be indicated on the plan; and
- 6. The residential use shall be located, designed, and configured in a manner to reduce potential conflict with adjacent existing industrial business operations; and
- 7. The owner(s) of a building seeking a conditional use for the residential use must sign and record a covenant and equitable servitude, on a form acceptable to the Director, that acknowledges that the owner(s) and occupants of the building accept the industrial character of the neighborhood and agree that existing or permitted industrial uses do not constitute a nuisance or other inappropriate or unlawful use of land. Such covenant and equitable servitude must state that it is binding on the owner(s)' successors, heirs, and assigns, including any lessees of the residential use; and
- 8. The residential use shall be a part of a mixed-use development that includes nonresidential uses permitted in UI zones, and the residential use component shall not exceed 50 percent of the total floor area of the mixed use development; and

- 9. Occupancies of dwelling units are voluntarily limited by the building owner to support the availability of housing that is affordable to area workers, such that the residential use consists of either:
 - a. All dwelling units are live-work units in which the commercial activity qualifies as industrial, or are caretakers' quarters associated with a business on the same site provided no single business shall have more than three associated caretakers' quarters; or
 - b. A minimum of 50 percent of the dwelling units are made available at affordable rent or affordable sale price for a period of 75 years beginning January 1 of the year following final certificate of occupancy to eligible households with annual incomes at or below 60 percent of median income for SEDUs, 80 percent of median income for studio and one bedroom units, and 90 percent of median income for two-bedroom and larger units. Standardized procedures and definitions established by the Office of Housing for administration of Chapter 5.73 shall apply. Dwelling units eligible for the multifamily housing tax exemption may be counted towards the minimum 50 percent.

The intent of the bill is to allow caretaker's quarters and residential uses serving a mix of incomes, including low-income households, as a conditional use and in turn such housing can help support additional activity within the STAOD, particularly on nights and weekends when there are not games or other events at Lumen Field or T-Mobile Park. More residences in the area, combined with non-residential uses can create a livelier Stadium District throughout the year.

2. CAPITAL IMPROVEMENT PROGRAM	
Does this legislation create, fund, or amend a CIP Project?	🗌 Yes 🖂 No
3. SUMMARY OF FINANCIAL IMPLICATIONS	
Does this legislation have financial impacts to the City?	🗌 Yes 🖂 No
3.d. Other Impacts	

Does the legislation have other financial impacts to The City of Seattle, including direct or indirect, one-time or ongoing costs, that are not included in Sections 3.a through 3.c? If so, please describe these financial impacts.

If the legislation has costs, but they can be absorbed within existing operations, please describe how those costs can be absorbed. The description should clearly describe if the absorbed costs are achievable because the department had excess resources within their existing budget or if by absorbing these costs the department is deprioritizing other work that would have used these resources.

This legislation affects the Seattle Department of Construction and Inspections (SDCI) as SDCI staff will need to be made aware of the code amendment for the purposes of permit review.

Lish Whitson LEG Stadium District Housing SUM D1a

However, this may not create a meaningful fiscal impact on SDCI. SDCI's fees are structured to cover the actual costs of reviewing permits. Additional review required for administrative conditional uses should be able to be covered by the standard fee schedule.

Please describe any financial costs or other impacts of *not* **implementing the legislation.** None identified

4. OTHER IMPLICATIONS

a. Please describe how this legislation may affect any departments besides the originating department.

The Seattle Department of Construction and Inspections (SDCI) regulates development and would need to update materials and systems to reflect the proposed zoning change. Central Staff is ready to consult with SDCI to determine whether this change would require any changes to software that could have fiscal impacts.

- b. Does this legislation affect a piece of property? If yes, please attach a map and explain any impacts on the property. Please attach any Environmental Impact Statements, Determinations of Non-Significance, or other reports generated for this property. This legislation applies to the Urban Industrial zone that is included within the Stadium Transition Overlay district. See attachment 1.
- c. Please describe any perceived implication for the principles of the Race and Social Justice Initiative.
 - i. How does this legislation impact vulnerable or historically disadvantaged communities? How did you arrive at this conclusion? In your response please consider impacts within City government (employees, internal programs) as well as in the broader community.

According to the 2022 Final Environmental Impact Statement for the Seattle Industrial and Maritime Strategy (FEIS), the STAOD is within one of the areas of the city with the highest disadvantages. FEIS attached hereto. The proposal would allow for additional housing, including potentially low-income housing, near both high-capacity transit and living wage jobs that are accessible to people without college degrees. Alternative 4 in the 2022 FEIS analyzed the likelihood of housing via caretaker's quarters and mixed-income housing in the SODO district and concluded that it may result in up to 990 units (see pages 2-43, 3-330, and 3-377-378).

However, living in the STAOD may present challenges to future residents. It is an area with few amenities for residents, including a lack of access to groceries, schools, or parks. The criteria for allowing housing include measures to mitigate the indoor air quality and noise impacts of living near major transportation corridors, but exterior air quality near the Port of Seattle, Interstate 90, State Route 99 and major truck routes can be poorer than in other areas. In addition to traffic noise, the stadiums, industrial activity, and railroads can create a noisier street environment than is found in other areas.

Lacking trees and open space, and predominantly developed with large open areas of concrete, the STAOD is one of the hottest areas in the city. The conditions for approval of housing in the UI zone include requirements for air conditioning. In addition, most development in the UI zone will be required to add street trees and comply with the green factor landscaping standards, adding some relief to the area as it redevelops.

There are also natural hazards in this area. The STAOD is in a liquefaction zone, which means that buildings constructed in the area will need to be carefully designed to avoid severe damage during an earthquake. Similarly, portions of the STAOD are susceptible to sea level rise, and building designs will need to consider those impacts.

The north Duwamish area, including the STAOD, has seen a high number of pedestrian and bicycle deaths in recent years. The current street environment in the area includes blocks without pedestrian or bicycle infrastructure. Encouraging redevelopment of the area would bring with it pedestrian facilities, including sidewalks, pedestrian lighting, and street trees in areas where they don't exist. It can also lead to a vulnerable population being added to a risky environment for pedestrians and bicyclists.

- Please attach any Racial Equity Toolkits or other racial equity analyses in the development and/or assessment of the legislation.
 See the analysis of Race and Social Equity included in the 2022 FEIS.
- What is the Language Access Plan for any communications to the public? SDCI regularly provides translated material describing changes to zoning on their website.

d. Climate Change Implications

i. Emissions: How is this legislation likely to increase or decrease carbon emissions in a material way? Please attach any studies or other materials that were used to inform this response.

According to the 2022 FEIS, allowing housing in the STAOD is likely to lead to slightly higher greenhouse gas emissions than maintaining the current zoning due to increased transportation to and from future homes and businesses that would be developed under the proposal. However, under either current zoning or the proposal, emissions are likely to decrease compared to current conditions.

ii. Resiliency: Will the action(s) proposed by this legislation increase or decrease Seattle's resiliency (or ability to adapt) to climate change in a material way? If so, explain. If it is likely to decrease resiliency in a material way, describe what will or could be done to mitigate the effects.

The STAOD was originally mud flats at the outlet of the Duwamish River into Elliott Bay. The area's current conditions are a result of significant channelization of the Duwamish River in the early 20th Century. As a result, the STAOD has unstable soils and is in a liquefaction zone that is prone to significant impacts during earthquakes.

Portions of the STAOD may be at increased risk of sea level rise. Seattle's codes and engineering requirements are well developed to address liquefaction risk, although the city has little residential development in its most liquefaction-prone areas. The area where residential development is most likely to occur – south of the Stadiums and east of 1st Avenue S is not the area identified on maps as particularly susceptible to sea level rise. However, the City could prohibit residential development at ground level, or otherwise require above-grade residential units to further mitigate possible sea level rise impacts on future residents of the area.

e. If this legislation includes a new initiative or a major programmatic expansion: What are the specific long-term and measurable goal(s) of the program? How will this legislation help achieve the program's desired goal(s)? What mechanisms will be used to measure progress towards meeting those goals? Not applicable.

5. CHECKLIST

Please click the appropriate box if any of these questions apply to this legislation.

☑ Is a public hearing required?

- **Is publication of notice with** *The Daily Journal of Commerce* and/or *The Seattle Times* required?
- If this legislation changes spending and/or revenues for a fund, have you reviewed the relevant fund policies and determined that this legislation complies?
- **Does this legislation create a non-utility CIP project that involves a shared financial commitment with a non-City partner agency or organization?**

6. ATTACHMENTS

List Summary Attachments (if any):

Summary Attachment 1 – Map of the Stadium Area Transition Overlay District Summary Attachment 2 – Seattle Industrial & Maritime Strategy Final Environmental Impact Statement



Stadium Transition Area Overlay District (STAOD)



Seattle Industrial & Maritime Strategy Final Environmental Impact Statement





September 29, 2022

Dear Community Members,

We are pleased to release the Final Environmental Impact Statement (Final EIS) for the Industrial and Maritime Strategy. This Final EIS illustrates how we can use our land use policies to help achieve our goals of securing a bright future for Seattle industrial and maritime sectors and supporting more equitable access to good paying jobs, while preparing to take advantage of emerging opportunities, and improving environmental health.

This EIS studies new industrial zones and land use policies that were recommended through a 2-year stakeholder process that concluded in May 2021. The stakeholder process engaged a broad cross section of interests including representatives from industrial and maritime sectors, labor, residents living in proximity to industrial areas, workforce development agencies, developers, and transportation agencies and advocates. A Draft EIS was released in December 2021 and was the subject of a public engagement process and extended comment period which resulted in the submission of 137 comments from community members.

The alternatives in this EIS review potential implementation of "future of industry" land use concepts intended to: strengthen and grow our city's core maritime, manufacturing, and logistics sectors, and assure long-term resilience for these important activities; support innovative employment-dense transit-oriented development (TOD) for industrial areas to maximize the benefits of new or expanded light rail stations in our industrial zones; and create healthier, more integrated transitions from industrial areas to nearby neighborhoods and urban villages. Throughout our analysis, we have an eye towards mitigating climate change and addressing existing environmental injustices.

Your comments helped refine the analysis of potential impacts of the alternatives studied and shaped a Preferred Alternative that is introduced in the Final EIS. The Preferred Alternative has several key differences from Draft EIS alternatives including: the location and amount of industry-supportive housing, stronger incentives for TOD, and neighborhood-specific nuances for zoning implementation in areas including Georgetown and South Park.

Moving forward, the City's Office of Planning & Community Development will develop a legislative proposal to amend Comprehensive Plan policies and amend the zoning code to implement these recommendations. The EIS process is an important tool for the public and decision-makers to understand the full effects of the proposal before any action is taken by the City. We believe that a combination of the changes studied in this EIS could lead to increased



economic opportunity for a broad range of households and ultimately increase the sustainability and resilience of our city. Thank you for your participation in this process and we look forward to future community engagement as these proposals move forward.

Sincerely,

8

Rico Quirindongo Acting Director

FACT SHEET

Project Title

Environmental Impact Statement (EIS) for the Seattle Industrial & Maritime Strategy

Proposed Action & Alternatives

Seattle's industrial and maritime policies are more than 35-years old and during that time, the trends and technologies impacting industrial and maritime users have experienced significant change. To reflect those changes as part of a comprehensive strategy to strengthen and grow Seattle's industrial and maritime sectors for the future, the City of Seattle is studying a proposal to update its industrial and maritime policies and industrial zoning. The proposal is informed by recommendations from community input, including an Industrial and Maritime Strategy Council, which resulted in an Industrial and Maritime Strategy Report that the City of Seattle released in June 2021.

This Environmental Impact Statement (EIS) studies five alternatives illustrating different potential futures for the city's industrially-zoned lands. The alternatives evaluate the effects of potential changes to Comprehensive Plan policies and changes to zoning over a 22-year time horizon (to 2044). The first alternative is a No Action alternative that is required by SEPA and is a basis for comparison. The Action Alternatives (alternatives 2, 3, and 4 and the Preferred Alternative) all apply proposed "future of industry" land use concepts that are based on community input and intended to respond to issues, challenges, and opportunities for the maritime and industrial sectors and adjacent communities.

Those future of industry land use concepts consist of three proposed new industrial zones:

- Maritime Manufacturing and Logistics (MML)—This zone would focus on strengthening land use protections for core and legacy industrial and maritime areas to better prevent the encroachment of development that is incompatible with industrial and maritime uses. This zone is particularly applicable within Seattle's Manufacturing/ Industrial Centers (MICs), near the shoreline or deep-water port, rail and freight infrastructure, and around existing clusters of industrial or maritime suppliers and services.
- Industry / Innovation (II)—This zone aims to encourage new development in multi-story buildings that accommodate industrial businesses mixed with other dense employment

51

iii

uses such as research, design, offices, and technology. By creating density bonuses for employment uses (i.e., office, R&D, etc.) if coupled with industrial uses in the same project, this type of modern industrial development would support high-density employment near transit stations and near existing industrial-commercial areas.

 Urban Industrial (UI)—This zone is designed to foster increased employment and entrepreneurship opportunities with a vibrant mix of affordable, small-scale places for light industry, makers, and creative arts, as well as industry supporting ancillary retail or housing spaces to create better, integrated, and healthier transitions at the edges between industrial areas and neighboring urban villages, residential, and mixed-use areas.

To implement the future of industry land use concepts in each of the Action Alternatives the City of Seattle would:

- Amend the comprehensive plan to add new text policies describing the intent and vision for how these concepts would be applied, including land use, environment, and transportation;
- Amend the industrial zoning section of the land use code to create new zone designations and corresponding development standards replacing the existing industrial zones;
- Apply new industrial zone classifications to industrial land; and
- Adopt new subarea plans for both the Ballard Interbay Northend and Greater Duwamish MICs.

However, each of the alternatives evaluated in this EIS pose different percentages of the future land use concepts in industrial and manufacturing lands for the purpose of strengthening and growing Seattle's industrial and maritime sectors in the future. The multi-faceted objectives of the proposal are listed in **Section 1.5.1** of this EIS.

The following is summary of the studied alternatives:

- Alternative 1—No Action: The SEPA-required alternative that would retain current Comprehensive Plan policies, development standards, or zoning maps.
- Alternative 2—Future of Industry Limited: Alternative 2 retains current MIC boundaries. Alternative 2 would implement future of industry land use concepts with a greater emphasis on strengthening protections for core and legacy industrial and maritime activities. The proposed MML zone would cover approximately 90% of industrial lands. Application of the proposed II and UI zones would be limited in scope, covering approximately 10% of current industrial areas. II zoning would be focused on existing Industrial Commercial (IC) zones and areas within approximately 1/4 mile of light rail stations. UI zoning would be focused on existing Industrial Buffer (IB) zones and the existing Stadium Transition Area Overlay. There are no changes to housing allowances in Alternative 2.
- Alternative 3—Future of Industry Targeted: Alternative 3 would strengthen protections for core industrial uses in the MML zone on approximately 86% of industrial lands. It applies a mix of the proposed II and UI zones in targeted geographies covering 14% of industrial lands. Compared to Alternative 2, II zoning is expanded to include areas an estimated 1/2 mile from light rail stations and UI zoning would be applied in additional areas in Ballard and the north shore of Lake Union. Alternative 3 creates limited flexibility for additional

iv

industry-supportive housing in UI zone that would result in an estimated 610 new homes in industrial zones. Alternative 3 removes focused land in Georgetown / South Park from the MIC and converts it to a non-industrial mixed-use zone.

- Alternative 4—Future of Industry Expanded: Alternative 4 would strengthen protections for core industrial uses in the MML zone on approximately 87% of industrial lands. Similar to Alternative 3, Alternative 4 would mainly apply II zoning in existing IC areas and within approximately a 1/2 mile from light rail stations, though with a greater expansion of the II zone in areas in Ballard and SODO. Compared to Alternative 3, the UI zone would be applied to a larger area in SODO, but to fewer areas in Ballard. This alternative includes additional flexibility for industry-supportive housing that could result in an estimated new 2,195 new homes in industrial zones. Just like Alternative 3, Alternative 4 removes focused land in Georgetown / South Park from MICs and convert it to a non-industrial mixed-use zone.
- Preferred Alternative—Future of Industry Balanced: The Preferred Alternative incorporates features of multiple Draft EIS alternatives. It includes modifications to address comments on the Draft EIS and reduce impacts identified for Draft EIS alternatives. The Preferred Alternative would implement the proposed land use concepts, and strengthen policy protections for industrial lands in MICs, while affording some greater flexibility for lands outside of MICs.
 - The MML zone would cover approximately 85% of industrial lands, while proposed II and UI zones would be targeted in scope and cover approximately 14% of current industrial areas. Unlike alternatives 2, 3, and 4, the Preferred Alternative would retain existing IC zoning only in areas outside of MICs.
 - The Preferred Alternative would allow limited industry-supportive housing in the UI zone as a conditional use subject to additional criteria to minimize potential conflicts. Overall, a lower amount of industry-supportive housing production would result compared to Draft EIS alternatives 3 and 4 within MICs in the UI zone (1,475 units). Concepts to remove focused land from the MIC in Georgetown and South Park are carried forward, and mixed use housing could occur as with alternatives 3 and 4. Housing in commercial zones could also occur outside the MIC in west Ballard, and Judkins Park. Though a greater amount of new unrestricted housing is projected outside of the MICs than any Draft EIS alternative (1,534 units), the combined growth of housing (3,009) would be less than Alternative 4.
 - The Preferred Alternative includes a more nuanced zoning approach for a proposed mixed use zone in central Georgetown, and greater application of UI zoning around Georgetown to create more neighborhood cohesion.
 - The Preferred Alternative features a reduced total amount of job growth, most similar to Draft EIS Alternative 2. Projections are adjusted downward to reflect conditions in commercial/office occupancy post-COVID and timelines for new light rail construction. The adjusted projections acknowledge that It will likely take longer to achieve levels of employment growth.
 - More information on mitigation measures is provided concurrent with the Preferred Alternative in response to suggestions and comments from community.

V

Proponent & Lead Agency

City of Seattle Office of Planning and Community Development

Location

The proposal addresses all lands zoned Industrial General (IG1 and IG2) zones, the Industrial Commercial (IC) zone, and the Industrial Buffer (IB) zone and land within two Manufacturing Industrial Centers (MIC): Seattle's Greater Duwamish Manufacturing and Industrial Center (Greater Duwamish MIC) and its Ballard Interbay North Manufacturing Industrial Center (BINMIC).

Tentative Date of Implementation

Summer/Fall 2022

Responsible SEPA Official

Rico Quirindongo Acting Director, Office of Planning & Community Development Mailing Address: P.O. Box 94788, Seattle, WA, 98124-7088 206-580-9509 <u>Rico.Quirindongo@seattle.gov</u>

Contact Person

Jim Holmes Planning and Community Development Mailing Address: P.O. Box 94788, Seattle, WA, 98124-7088 206-684-8372 jim.holmes@seattle.gov

Required Approvals

The proposal includes the development of legislative proposals for the Comprehensive Plan, municipal code, and subarea plans. The proposals will be reviewed by the Planning Commission and considered for approval by the City Council. The proposals will be reviewed by the Washington Department of Commerce for a 60-day period prior to City action.

54

vi

Principal EIS Authors & Contributors

Under the direction of the City of Seattle, the consultant team prepared the EIS as follows:

- <u>BERK Consulting</u> (prime consultant): SEPA documentation, Light and Glare, Housing, Open Space and Recreation, Public Services
- Fehr & Peers: Transportation
- <u>Herrera</u>: Soils/Geology, Air Quality and Greenhouse Gas, Water Resources, Plants and Animals, Contamination, Noise, Utilities
- Historical Research Associates: Historic, Archaeological, and Cultural Resources

Additional contributors included:

- City of Seattle. <u>Office of Planning and Community Development</u>: Alternatives and Land and Shoreline Use
- <u>Ramboll</u>: Air Quality and Noise level data collection

Draft EIS Date of Issuance

December 16, 2021

Draft EIS Comment Period

The City of Seattle requested comments from citizens, agencies, tribes, and all interested parties on the Draft EIS from December 16, 2021 to January 31, 2022. The comment period was extended to March 2, 2022. Additional engagement was conducted with the Georgetown and South Park communities through April 15, 2022.

All written comments were directed to:

Jim Holmes Office of Planning & Community Development Mailing Address: P.O. Box 94788, Seattle, WA, 98124-7088 206-684-8372 | PCD Industry and Maritime@seattle.gov

Please see the project website for information about other public comment opportunities: <u>https://www.seattle.gov/opcd/ongoing-initiatives/industrial-and-maritime-strategy</u>.

Comment opportunities were offered two virtual public hearings.

- Public Hearing January 11, 2021 at 10:00 a.m.
- Public Hearing January 12, 2021 at 6:00 p.m.

A link to these hearings was available at: Industrial and Maritime Strategy—OPCD | seattle.gov.

55

vii

Date of Final EIS Issuance

September 29, 2022

Date of Final Action

Anticipated Winter 2023

Prior Environmental Review

The study area was reviewed as part of the citywide Comprehensive Plan EIS completed in 2016:

Final Environmental Impact Statement for the Seattle Comprehensive Plan Update, May 5, 2016.

Location of Background Data

You may review the City of Seattle website for more information at <u>https://www.seattle.gov/opcd/ongoing-initiatives/industrial-and-maritime-strategy</u>. If you desire clarification or have questions, please see the contact person above.

Purchase/Availability of Final EIS

The Final EIS can be downloaded from the City of Seattle's website at

https://www.seattle.gov/opcd/ongoing-initiatives/industrial-and-maritime-

<u>strategy#projectdocuments</u>. Interested parties may arrange to obtain an electronic copy of the Final EIS on a USB flash drive free of charge, or purchase a hard copy at the cost of printing by contacting <u>Jim.Holmes@Seattle.gov</u>.

DISTRIBUTION LIST

The Final EIS has been issued with a notice of availability and methods of publication required in SMC 25.05.460 Issuance of FEIS.

Federal & Tribal Agencies

Muckleshoot Indian Tribe Suquamish Tribe Tulalip Tribes of Washington National Oceanic and Atmospheric Administration Fisheries, National Marine Fisheries Service U.S. Army Corps of Engineers U.S. Department of Commerce Economic Development Administration U.S. Department of Fish & Wildlife Services U.S. Department of Housing & Urban Development U.S. Environmental Protection Agency

USDA-Wildlife Services Division

State Agencies

Department of Archaeology & Historic Preservation Department of Commerce Department of Commerce, Growth Management Services Department of Ecology Department of Fish & Wildlife Department of Fisheries Habitat Department of Health Department of Natural Resources Department of Social & Health Services Department of Transportation

Regional and County Agencies

King County Community and Human Services

57

ix

King County Department of Natural Resources King County Department of Natural Resources, Parks Division King County Department of Permitting and Environmental Review King County Department of Transportation King County Executive's Office King County Metro Transit King County Regional Water Quality Committee King County Wastewater Treatment Division Port of Seattle Puget Sound Clean Air Agency Puget Sound Regional Council Seattle-King County Department of Public Health Sound Transit

Seattle, Adjacent Jurisdictions, Service Providers

See regional providers above and following.

City of Shoreline City of Tukwila Seattle City Light Seattle Housing Authority Seattle Public Library, Public Review Documents Seattle Public Utilities Seattle School District Southwest Suburban Sewer District

Seattle City Council Legislative Department Seattle Department of Education and Early Learning Seattle Department of Neighborhoods Seattle, Department of Neighborhoods, Historic Preservation Program Seattle Department of Transportation Seattle Fire Department Seattle Fleet Management Seattle Indian Services Commission Seattle Landmarks Preservation Board Seattle Law Department Seattle Office of Arts and Culture Seattle Office of Economic Development Seattle Office of Emergency Management Seattle Office of Housing Seattle Office of Planning & Community Development Seattle Office of the Mayor

Х

Seattle Parks and Recreation Seattle Police Department

Community Organizations & Individuals

Duwamish Tribe Industrial and Maritime Strategy Council Georgetown / South Park Council Ballard Council Interbay Council SODO Council Black Indigenous and Persons of Color (BIPOC) Youth Engagement Partners

Persons providing scoping comments and Draft EIS comments (see **Appendix A** and **Chapter 4** of this Final EIS).

59

xi

CONTENTS

1	Sum	mary	1-1
1.1	Purpo	se	1-2
1.2	Study	Area	1-5
1.3	Planni	ng Context & Outreach	1-7
	1.3.1	Emerging Factors Affecting Seattle's MICs	1-7
	1.3.2	Equity & Environmental Justice	1-7
	1.3.3	Mayor's Industrial & Maritime Strategy	1-8
1.4	SEPA F	Process	1-8
	1.4.1	Environmental Review	1-8
	1.4.2	Public Comment Opportunities	1-10
1.5	Object	tives, Proposal, & Alternatives	1-12
	1.5.1	Objectives	1-12
	1.5.2	Proposal	1-13
	1.5.3	Land Use Concepts	1-13
	1.5.4	Regulatory Concepts	1-19
	1.5.5	Alternative 1—No Action	1-20
	1.5.6	Alternative 2—Future of Industry Limited	1-23
	1.5.7	Alternative 3—Future of Industry Targeted	1-25
	1.5.8	Alternative 4—Future of Industry Expanded	1-27
	1.5.9	Preferred Alternative—Future of Industry Balanced	1-29
	1.5.10	Comparison of Alternatives	1-32
	1.5.11	Alternatives Considered & Not Carried Forward	1-37
1.6	Key Is	sues & Options	1-38
1.7	Summ	ary of Impacts & Mitigation Measures	1-39
	1.7.1	Soils/Geology	1-39
	1.7.2	Air Quality & GHG	1-41
	1.7.3	Water Resources	1-44
	1.7.4	Plants & Animals	1-46
	1.7.5	Contamination	1-48
	1.7.6	Noise	1-50
	1.7.7	Light & Glare	1-53
	1.7.8	Land & Shoreline Use	1-57

	1.7.9 1.7.10 1.7.11 1.7.12 1.7.13 1.7.14 1.7.15	Housing Transportation Historic, Archaeological, & Cultural Resources Open Space & Recreation Public Services Utilities Equity & Environmental Justice Considerations	1-61 1-64 1-69 1-73 1-75 1-78 1-79
2	Prop	osal & Alternatives	2-1
2.1	Introd	uction	2-2
	2.1.1	Overview of the Proposal	2-2
	2.1.2	Study Area	2-5
	2.1.3	Objectives of the Proposal	2-7
2.2	Planni	ng Context & Outreach	2-8
	2.2.1	Emerging Factors Affecting Seattle's MICs	2-8
	2.2.2	Equity & Environmental Justice	2-10
	2.2.3	Mayor's Industrial & Maritime Strategy	2-13
2.3	SEPA F	Process	2-15
	2.3.1	Environmental Review Process	2-15
~ (2.3.2	Public Comment Opportunities	2-15
2.4	Propo	sed Action & Alternatives	2-17
	2.4.1	Land Use Concepts	2-17
	2.4.2	Regulatory Concepts	2-22
	2.4.3	Alternative 1—No Action	2-28
	2.4.4	Alternative 2—Future of Industry Limited	2-32
	2.4.5	Alternative 3—Future of Industry Targeted	2-30
	2.4.0	Preferred Alternative—Future of Industry Expanded	2-40 2-44
	2.4.7	Summary of Alternatives	2-48
	2.4.9	Alternatives Considered & Not Carried Forward	2-54
2.5	Benefi	its & Disadvantages of Delaying the Proposed Action	2-54
3	Envi	ronment, Impacts, & Mitigation Measures	3-1
3.1	Soils/C	Geology	3-3
	3.1.1	Affected Environment	3-4
	3.1.2	Impacts	3-11
	3.1.3	Mitigation Measures	3-17
	3.1.4	Significant Unavoidable Adverse Impacts	3-18
3.2	Air Qu	ality & GHG	3-19
	3.2.1	Affected Environment	3-21
	3.2.2	Impacts	3-45

	3.2.3	Mitigation Measures	3-84
2 2	5.2.4 Water	Resources	3-00 3-90
5.5	221	Affected Environment	3-91
	337	Impacts	3-100
	3.3.3	Mitigation Measures	3-108
	3.3.4	Significant Unavoidable Adverse Impacts	3-111
3.4	Plants	& Animals	3-112
	3.4.1	Affected Environment	3-113
	3.4.2	Impacts	3-132
	3.4.3	Mitigation Measures	3-135
	3.4.4	Significant Unavoidable Adverse Impacts	3-137
3.5	Contai	mination	3-138
	3.5.1	Affected Environment	3-139
	3.5.2	Impacts	3-152
	3.5.3	Mitigation Measures	3-159
2.0	3.5.4	Significant Unavoidable Adverse Impacts	3-160
3.0	Noise		3-101
	3.6.1	Affected Environment	3-162
	3.6.2	Impacts Mitigation Moasuros	3-1//
	2.0.5 2.6.1	Significant Unavoidable Adverse Impacts	2-100
37	Light 8	Significant on avoidable Adverse impacts	3-189
5.7	271	Affected Environment	3-100
	3.7.1	Impacts	3-190
	3.7.3	Mitigation Measures	3-254
	3.7.4	Significant Unavoidable Adverse Impacts	3-255
3.8	Land 8	& Shoreline Use	3-256
	3.8.1	Affected Environment	3-258
	3.8.2	Impacts	3-295
	3.8.3	Mitigation Measures	3-344
	3.8.4	Significant Unavoidable Adverse Impacts	3-349
3.9	Housir	ng	3-351
	3.9.1	Affected Environment	3-352
	3.9.2	Impacts	3-370
	3.9.3	Mitigation Measures	3-382
2.4.0	3.9.4	Significant Unavoidable Adverse Impacts	3-384
3.10	Transp	portation	3-385
	3.10.1	Affected Environment	3-386
	3.10.2	Impacts	3-435
	3.10.3	Mitigation Measures	3-487

	3.10.4 Significant Unavoidable Adverse Impacts	3-497
3.11	Historic, Archaeological, & Cultural Resources	3-499
	3.11.1 Affected Environment	3-500
	3.11.2 Impacts	3-521
	3.11.3 Mitigation Measures	3-531
	3.11.4 Significant Unavoidable Adverse Impacts	3-535
3.12	Open Space & Recreation	3-536
	3.12.1 Affected Environment	3-537
	3.12.2 Impacts	3-548
	3.12.3 Mitigation Measures	3-559
	3.12.4 Significant Unavoidable Adverse Impacts	3-560
3.13	Public Services	3-561
	3.13.1 Affected Environment	3-562
	3.13.2 Impacts	3-585
	3.13.3 Mitigation Measures	3-597
	3.13.4 Significant Unavoidable Adverse Impacts	3-599
3.14	Utilities	3-600
	3.14.1 Affected Environment	3-601
	3.14.2 Impacts	3-609
	3.14.3 Mitigation Measures	3-615
	3.14.4 Significant Unavoidable Adverse Impacts	3-623
4	Comments & Responses	4-1
4.1	Introduction	4-2
4.2	Response to Common Comment Themes	4-7
	4.2.1 Economic & Market Analysis	4-7
	4.2.2 Non-Conforming Uses	4-7
	4.2.3 Industry Supportive Housing	4-8
	4.2.4 Light Rail Coordination	4-8
	4.2.5 Georgetown Arts & Culture	4-9
	4.2.6 Georgetown Buffer Areas	4-9
	4.2.7 Commitment to Mitigation Measures	4-10
	4.2.8 Community Engagement	4-10
	4.2.9 Consider Strategy & Comprehensive Plan	4-11
	4.2.10 Housing Instead of Industrial Uses	4-11
_	4.2.11 Modify MIC Boundaries or Uses	4-12
4.3	Individual Responses to Comments	4-13

4.3 Individual Responses to Comments4.3.1 Written Comments & Responses4.3.2 Public Hearing Verbal Comments & Responses

4.4 Marked Comment Letters & Public Hearing Transcripts 4-106

4-13

4-104

5	Acronyms & References	5-1
5.1	Acronyms	5-2
5.2	References	5-5
6	Appendices	6-1
А	Scoping Notice & Comment Summary	6-1
В	Industrial & Maritime Strategy Council Report	6-2
С	Alternative Future Land Use Zoning Maps	6-3
D	Draft Comprehensive Plan Goal & Policy Language	6-4
Е	Industrial Development Regulations	6-5
F	Shoreline Master Program Development Regulations	6-6
G	Preferred Alternative Preliminary Regulations	6-7
Н	Air Quality and Noise Technical Memo	6-8
	Transportation Screenline Information	6-9
J	EIS Mitigation Measures List	6-10
Κ	Presentation Slides from the Draft EIS Public Hearings	6-11

EXHIBITS

Exhibit 1.2-1	Study Area	1-6
Exhibit 1.4-1	EIS Process	1-9
Exhibit 1.4-2	Comparison of Project and Non-Project Environmental Review	1-9
Exhibit 1.5-1	Objectives of the Proposal	1-12
Exhibit 1.5-2	Maritime Manufacturing and Logistics Proposed Land Use Concept	1-14
Exhibit 1.5-3	Industry and Innovation Proposed Land Use Concept	1-15
Exhibit 1.5-4	Urban Industrial Proposed Land Use Concept	1-16
Exhibit 1.5-5	Development Standards by Land Use Concept	1-19
Exhibit 1.5-6	Alternative 1—No Action Zoning Districts (Acres)	1-20
Exhibit 1.5-7	Alternative 1—No Action Zoning Map	1-21
Exhibit 1.5-8	Alternative 2—Future of Industry Limited Zoning Districts (Acres)	1-23
Exhibit 1.5-9	Alternative 2—Future of Industry Limited	1-24
Exhibit 1.5-10	Alternative 3—Future of Industry Targeted Zoning Districts (Acres)	1-25
Exhibit 1.5-11	Alternative 3—Future of Industry Targeted	1-26
Exhibit 1.5-12	Alternative 4—Future of Industry Expanded Zoning Districts (Acres)	1-27
Exhibit 1.5-13	Alternative 4—Future of Industry Expanded	1-28
Exhibit 1.5-14	Preferred Alternative—Future of Industry Balanced Zoning Districts	
	(Acres)	1-29
Exhibit 1.5-15	Preferred Alternative—Future of Industry Balanced	1-30
Exhibit 1.5-16	Summary of Land Use Concepts by Alternatives	1-32
Exhibit 1.5-17	Comparison of Alternatives by Land Use/Zoning Acres	1-34
Exhibit 1.5-18	Industrial and Non-Industrial Job Share	1-35
Exhibit 1.5-19	Share of Industrial and Non-Industrial Jobs	1-35
Exhibit 1.5-20	Comparison of Combined Industrial and Population Growth by	
	Alternative	1-36
Exhibit 1.5-21	Percent Growth in Employment by Subarea	1-37
Exhibit 1.7-1	Increase in dBA Over Existing Conditions, All Alternatives	1-52
Exhibit 1.7-2	Summary of Significant Transportation Impacts	1-65

Exhibits

Exhibit 1.7-3	Impacted Study Corridors—Ballard Interbay Northend MIC, 2044	1-66
Exhibit 1.7-4	Impacted Study Corridors—Greater Duwamish MIC, 2044	1-67
Exhibit 1.7-5	Net Open Space and Recreation Acres to Meet LOS Standards, All	
	Alternatives	1-74
Exhibit 1.7-6	Student Generation by Subarea based on Net Change in Population	1-76
Exhibit 1.7-7	Seattle Racial and Social Equity Index	1-80
Exhibit 1.7-8	Equity and Environmental Justice Matrix of Topics	1-90
Exhibit 2.1-1	Study Areas	2-6
Exhibit 2.1-2	Objectives of the Proposal	2-7
Exhibit 2.2-1	Duwamish Valley Action Plan Racial Equity Outcomes	2-12
Exhibit 2.2-2	Industrial and Maritime Strategy Stakeholder Recommendations	2-14
Exhibit 2.3-1	EIS Process	2-15
Exhibit 2.4-1	Maritime Manufacturing and Logistics Proposed Land Use Concept	2-18
Exhibit 2.4-2	Industry and Innovation Proposed Land Use Concept	2-19
Exhibit 2.4-3	Urban Industrial Proposed Land Use Concept	2-20
Exhibit 2.4-4	Development Standards by Land Use Concept	2-22
Exhibit 2.4-5	Alternative 1—No Action Zoning Districts (Acres)	2-28
Exhibit 2.4-6	Alternative 1—No Action Zoning Map	2-29
Exhibit 2.4-7	Alternative 1—No Action Jobs and Housing Units, Existing and 2044	2-30
Exhibit 2.4-8	Current and Alternative 1—No Action Employment Mix by Subarea	2-30
Exhibit 2.4-9	Current and Alternative 1—No Action Housing Units in Industrial Zones	2-31
Exhibit 2.4-10	Alternative 1—No Action Jobs and Population Growth by Subarea	2-31
Exhibit 2.4-11	Alternative 2—Future of Industry Limited Zoning Districts (Acres)	2-32
Exhibit 2.4-12	Alternative 2—Future of Industry Limited	2-33
Exhibit 2.4-13	Alternative 2 Jobs and Housing Units, Existing and 2044	2-34
Exhibit 2.4-14	Current and Alternative 2 Employment Mix by Subarea	2-34
Exhibit 2.4-15	Current and Alternative 2 Housing Units in Industrial Zones	2-35
Exhibit 2.4-16	Alternative 2 Jobs and Population Growth by Subarea	2-35
Exhibit 2.4-17	Alternative 3—Future of Industry Targeted Zoning Districts (Acres)	2-36
Exhibit 2.4-18	Alternative 3—Future of Industry Targeted	2-37
Exhibit 2.4-19	Alternative 3 Jobs and Housing Units, Existing and 2044	2-38
Exhibit 2.4-20	Current and Alternative 3 Employment Mix by Subarea	2-38
Exhibit 2.4-21	Current and Alternative 3 Housing in Industrial Zones	2-39
Exhibit 2.4-22	Alternative 3 Jobs and Population Growth by Subarea	2-39
Exhibit 2.4-23	Alternative 4—Future of Industry Expanded Zoning Districts (Acres)	2-40
Exhibit 2.4-24	Alternative 4—Future of Industry Expanded	2-41

Exhibit 2.4-25	Alternative 4 Jobs and Housing Units, Existing and 2044	2-42
Exhibit 2.4-26	Current and Alternative 4 Employment Mix by Subarea	2-42
Exhibit 2.4-27	Current and Alternative 4 Housing Units in Industrial Zones	2-43
Exhibit 2.4-28	Alternative 4 Jobs and Population Growth by Subarea	2-43
Exhibit 2.4-29	Preferred Alternative—Future of Industry Balanced Zoning Districts	
	(Acres)	2-44
Exhibit 2.4-30	Preferred Alternative—Future of Industry Balanced	2-45
Exhibit 2.4-31	Preferred Alternative Jobs and Housing Units, Existing and 2044	2-46
Exhibit 2.4-32	Current and Preferred Alternative Employment Mix by Subarea	2-46
Exhibit 2.4-33	Current and Preferred Alternative Housing Units in Industrial Zones	2-47
Exhibit 2.4-34	Summary of Land Use Concepts by Alternatives	2-48
Exhibit 2.4-35	Comparison of Alternatives by Land Use/Zoning Acres	2-50
Exhibit 2.4-36	Industrial and Non-Industrial Job Share	2-51
Exhibit 2.4-37	Share of Industrial and Non-Industrial Jobs	2-51
Exhibit 2.4-38	Comparison of Combined Industrial and Population Growth by	
	Alternative	2-52
Exhibit 2.4-39	Percent Growth in Employment by Subarea	2-53
Exhibit 2.4-40	Employment Totals by Subarea and Alternative	2-53
Exhibit 3.1-1	Summary of Geologic Hazards Mapped in the BINMIC and Greater Duwamish MIC by Subarea	3-7
Exhibit 3.1-2	Geologic Hazards Mapped in the BINMIC	3-8
Exhibit 3.1-3	Geologic Hazards Mapped in the Greater Duwamish MIC	3-9
Exhibit 3.1-4	Existing and Net Employment Building Space by Alternative	3-12
Exhibit 3.1-5	Total Housing in Study Area by Alternative	3-13
Exhibit 3.2-1	National Ambient Air Quality Standards	3-23
Exhibit 3.2-2	Seattle Air Quality Monitoring Stations and Criteria Pollutants	3-28
Exhibit 3.2-3	Air Quality Monitoring Locations	3-29
Exhibit 3.2-4	Ambient Criteria Pollutant Concentration Levels Measured for the four	
	Seattle Sites (10 th & Weller, Beacon Hill, Duwamish, and South Park) from	0.04
		3-31
Exhibit 3.2-5	Ambient PM10 Concentration Levels Measured in 2021	3-32
Exhibit 3.2-6	Detected Pollutants and Measured Concentration Levels in 2021	3-32
Exhibit 3.2-7	Estimated Industrial and Non-Industrial Square Footage for All	
	square feet)	3-48
Exhibit 3.2-8	Estimated VMT During the PM Period for Action Alternatives (2044)	- 10
	Compared to Existing (2019) and Alternative 1 No Action (2042)	3-49

Exhibits

Exhibit 3.2-9	Estimated Tons of Criteria Pollutant Emissions from Road Transportation for Action Alternatives (2044) Compared to Existing and Alternative 1 No Action (2042)	3-49
Exhibit 3.2-10	Estimated Tons of Criteria Pollutant Emissions from Road Transportation in BINMIC, All Alternatives	3-50
Exhibit 3.2-11	Estimated Tons of Criteria Pollutant Emissions from Road Transportation in Greater Duwamish MIC, All Alternatives	3-51
Exhibit 3.2-12	Maritime Activities Air Emissions, Tons per Year, 2016	3-52
Exhibit 3.2-13	Estimated Road Transportation GHG Emissions for All Alternatives Compared to Existing Conditions (2019) and Alternative 1 No Action (2042) (MTCO2e)	3-54
Exhibit 3.2-14	Estimated GHG Emissions from Maritime Activities, 2016 (MTCO2e)	3-56
Exhibit 3.2-15	Estimated Building-Related GHG Emissions for Action Alternatives Compared to Existing Conditions (2017) and Alternative 1 No Action (2042) (MTCO2e)	3-57
Exhibit 3.2-16	Estimated Housing-Related GHG Emissions for All Alternatives Compared to Existing Conditions (2021) and Alternative 1 No Action (2042) (MTCO2e)	3-57
Exhibit 3.2-17	Estimated Waste-Related GHG Emissions for All Alternatives Compared to Existing Conditions and Alternative 1 No Action (MTCO2e)	3-58
Exhibit 3.2-18	Estimated VMT For the Baseline Year (2019) And Alternative 1 No Action (2042)	3-60
Exhibit 3.2-19	Estimated Tons of Criteria Pollutant Emissions from Road Transportation for Alternative 1 No Action (2042) Compared to Existing Conditions (2019)	3-61
Exhibit 3.2-20	Estimated Number of Housing Units for Industrial Subareas Under Alternative 1 No Action (2044) Compared to the Current Conditions (2021)	3-62
Exhibit 3.2-21	Total Estimated Annual MTCO2e Emissions Under Alternative 1 No Action Compared to Existing Conditions	3-64
Exhibit 3.2-22	Estimated VMT For Alternative 2 (2044) Compared to Alternative 1 No Action (2042)	3-65
Exhibit 3.2-23	Estimated Tons of Criteria Pollutant Emissions from Road Transportation for Alternative 2 (2044) Compared to Alternative 1 No Action (2042)	3-65
Exhibit 3.2-24	Estimated Number of Housing Units for Industrial Subareas Under Alternative 2 (2044) Compared to Alternative 1 No Action (2044)	3-67
Exhibit 3.2-25	Total Estimated Annual MTCO2e Emissions Under Alternative 2 Compared to Alternative 1 No Action	3-68

Exhibit 3.2-26	Estimated VMT for Alternative 3 (2044) Compared to Alternative 1 No Action (2042)	3-69
Exhibit 3.2-27	Estimated Tons of Criteria Pollutant Emissions from Road Transportation for Alternative 3 (2044) Compared to Alternative 1 No Action (2042)	3-70
Exhibit 3.2-28	Estimated Number of Housing Units for Industrial Subareas Under Alternative 3 (2044) Compared to Alternative 1 No Action (2044)	3-72
Exhibit 3.2-29	Total Estimated Annual MTCO2e Emissions Under Alternative 3 Compared to Alternative 1 No Action	3-73
Exhibit 3.2-30	Estimated VMT For Alternative 4 (2044) Compared to Alternative 1 No Action (2042)	3-74
Exhibit 3.2-31	Estimated Tons of Criteria Pollutant Emissions from Road Transportation for Alternative 4 (2044) Compared to Alternative 1 No Action (2042)	3-75
Exhibit 3.2-32	Estimated Number of Housing Units for Industrial Subareas Under Alternative 4 (2044) Compared to Alternative 1 No Action (2044)	3-77
Exhibit 3.2-33	Total Estimated Annual MTCO2e Emissions Under Alternative 4 Compared to Alternative 1 No Action	3-78
Exhibit 3.2-34	Estimated VMT For the Preferred Alternative (2044) Compared to Alternative 1 No Action (2042)	3-79
Exhibit 3.2-35	Estimated Tons of Criteria Pollutant Emissions from Road Transportation for the Preferred Alternative (2044) Compared to Alternative 1 No Action (2042)	3-80
Exhibit 3.2-36	Estimated Number of Housing Units for Industrial Subareas Under the Preferred Alternative (2044) Compared to Alternative 1 No Action (2044)	3-82
Exhibit 3.2-37	Total Estimated Annual MTCO2e Emissions Under the Preferred Alternative Compared to Alternative 1 No Action	3-83
Exhibit 3.3-1	Surface Water Bodies Located in each Subarea	3-92
Exhibit 3.3-2	Surface Water Bodies in the Primary Study Area	3-93
Exhibit 3.3-3	Location of Surface Water Bodies and Watersheds of Natural Streams	3-94
Exhibit 3.3-4	Comparison of Relative Water Resource Improvements Between	0.400
	Alternatives	3-103
EXNIDIT 3.3-5	Total Jobs by Subarea Current and Future	3-103
EXHIDIT 3.3-6	Share of Job Growth by Subarea Compared to Existing	3-104
EXHIDIL 3.4-1	Croater Duwamich MIC Study Area and Critical Areas 2021	5-114 2 115
Exhibit 2.4-2	Enderal State, and Local Pegulations and Permits Pelated to the	5-115
	Protection of Plants and Animals	3-116
Exhibit 3.4-4	Special Status Species and Habitats Occurring in the Study Areas	3-123
Exhibit 3.4-5	BINMIC Study Areas PHS Mapping, 2021	3-125

Exhibit 3.4-6	Greater Duwamish MIC Study Areas PHS Mapping, 2021	3-126
Exhibit 3.4-7	Critical Areas—Ballard Subarea, 2021	3-127
Exhibit 3.4-8	Critical Areas—Interbay Dravus Subarea, 2021	3-128
Exhibit 3.4-9	Critical Areas—Interbay Smith Cove Subarea, 2021	3-129
Exhibit 3.4-10	Critical Areas—SODO/Stadium Subarea, 2021	3-130
Exhibit 3.4-11	Critical Areas—Georgetown/South Park Subarea, 2021	3-131
Exhibit 3.5-1	Federal, State, and Local Arars Potentially Applicable for Cleanup Actions at Contaminated Sites Within the Full Study Area	3-142
Exhibit 3.5-2	Confirmed or Suspected Contaminated Sites Within 0.25-mile of the BINMIC	3-145
Exhibit 3.5-3	Hazardous Waste and Toxics Reduction Sites Within 0.25-mile of the BINMIC	3-146
Exhibit 3.5-4	Confirmed or Suspected Contaminated Sites Within 0.25-mile of the Greater Duwamish MIC	3-147
Exhibit 3.5-5	Hazardous Waste and Toxics Reduction Sites Located Within 0.25-mile of the Greater Duwamish MIC	3-148
Exhibit 3.5-6	Summary of Toxics Cleanup Sites and Hazardous Waste and Toxics Reduction Sites Within the BINMIC and Greater Duwamish MIC and Subareas	3-149
Exhibit 3.5-7	Existing and Net Employment Building Space by Alternative	3-154
Exhibit 3.5-8	Total Housing in Study Area by Alternative	3-154
Exhibit 3.6-1	WSDOT Noise Abatement Criteria by Land Use Category	3-166
Exhibit 3.6-2	Maximum Permissible Noise Levels: Seattle Noise Control Ordinance	3-167
Exhibit 3.6-3	Construction Noise Time Limits for Public Projects in Commercial Zones Under the City of Seattle Noise Ordinance	3-168
Exhibit 3.6-4	Boeing Field Noise Exposure	3-170
Exhibit 3.6-5	Typical Sound Levels	3-171
Exhibit 3.6-6	Ambient Noise Level Data at Ecology/PSCAA Seattle Monitoring Stations	3-173
Exhibit 3.6-7	Sound Level Measurements (dBA) in the Seattle MIC Areas (2021)	3-173
Exhibit 3.6-8	Noise Monitoring Locations	3-174
Exhibit 3.6-9	Location of Potential Noise Sensitive Areas	3-176
Exhibit 3.6-10	PM Peak Hour Traffic Volumes for Existing Conditions and All Alternatives	3-178
Exhibit 3.6-11	Increase in dBA Over Existing Conditions, All Alternatives	3-179
Exhibit 3.7-1	Seattle SEPA Scenic Routes Map—North	3-194
Exhibit 3.7-2	Seattle SEPA Scenic Routes Map—South	3-195
Exhibit 3.7-3	Nighttime Illumination, 2015	3-198

Exhibit 3.7-4	Industrial Subarea Viewshed, 2021	3-199
Exhibit 3.7-5	Nighttime Illumination—Ballard Interbay Northend MIC, 2015	3-201
Exhibit 3.7-6	Nighttime Illumination—Greater Duwamish MIC, 2015	3-205
Exhibit 3.7-7	Land Use Concepts Viewshed—Alternative 1	3-210
Exhibit 3.7-8	Ballard, Interbay Dravus, and Interbay Smith Cove Viewshed—Alternative	
	1	3-212
Exhibit 3.7-9	SODO/Stadium Viewshed—Alternative 1	3-215
Exhibit 3.7-10	Georgetown/South Park Viewshed—Alternative 1	3-217
Exhibit 3.7-11	Increase in Viewshed—Alternative 2	3-219
Exhibit 3.7-12	Increase in Viewshed (Ballard, Interbay Dravus, and Interbay Smith Cove)—Alternative 2	3-221
Exhibit 3.7-13	Increase in Viewshed (SODO/Stadium)—Alternative 2	3-223
Exhibit 3.7-14	Increase in Viewshed (Georgetown/South Park)—Alternative 2	3-225
Exhibit 3.7-15	Increase in Viewshed—Alternative 3	3-227
Exhibit 3.7-16	Increase in Viewshed (Ballard, Interbay Dravus, and Interbay Smith	
	Cove)—Alternative 3	3-229
Exhibit 3.7-17	Increase in Viewshed (SODO/Stadium)—Alternative 3	3-231
Exhibit 3.7-18	Increase in Viewshed (Georgetown/South Park)—Alternative 3	3-233
Exhibit 3.7-19	Increase in Viewshed—Alternative 4	3-235
Exhibit 3.7-20	Increase in Viewshed (Ballard, Interbay Dravus, and Interbay Smith Cove)—Alternative 4	3-237
Exhibit 3.7-21	Increase in Viewshed (SODO/Stadium)—Alternative 4	3-239
Exhibit 3.7-22	Increase in Viewshed (Georgetown/South Park)—Alternative 4	3-241
Exhibit 3.7-23	Increase in Viewshed—Preferred Alternative	3-243
Exhibit 3.7-24	Increase in Viewshed (Ballard, Interbay Dravus, and Interbay Smith Cove)—Preferred Alternative	3-245
Exhibit 3.7-25	Increase in Viewshed (SODO/Stadium)—Preferred Alternative	3-247
Exhibit 3.7-26	Increase in Viewshed (Georgetown/South Park)—Preferred Alternative	3-249
Exhibit 3.7-27	Summary of Light and Glare Impacts—Action Alternatives	3-250
Exhibit 3.8-1	Seattle's Shoreline Over Time	3-260
Exhibit 3.8-2	Commercial Map of Greater Seattle With "Grade Of Security"	
	Designations, 1936	3-263
Exhibit 3.8-3	Future Land Use Map for Industrial Areas Within and Outside MICs	3-278
Exhibit 3.8-4	Existing Zoning by MIC, Outside MICs, and Citywide	3-279
Exhibit 3.8-5	Industrial Land Uses by Area	3-280
Exhibit 3.8-6	Non-industrial Uses by Area	3-281
Exhibit 3.8-7	North Industrial Land Use	3-282

Exhibit 3.8-8	South Industrial Land Use	3-283
Exhibit 3.8-9	Industrial Subareas	3-285
Exhibit 3.8-10	The Duwamish Longhouse	3-291
Exhibit 3.8-11	Industrial Areas Employment by Economic Sector	3-294
Exhibit 3.8-12	Industrial and Non-Industrial Employment by Sub-Area, Current Conditions (2018)	3-295
Exhibit 3.8-13	Employment by Subarea, Current Conditions and Alternative 1	3-299
Exhibit 3.8-14	Employment by Subarea, Current Conditions and Alternative 2	3-308
Exhibit 3.8-15	Employment by Subarea, Current Conditions and Alternative 3	3-318
Exhibit 3.8-16	Employment by Subarea Current Conditions and Alternative 4	3-327
Exhibit 3.8-17	Employment by Subarea Current Conditions and the Preferred	
	Alternative	3-336
Exhibit 3.8-18	Summary of Land Use Impacts by Subarea and Alternative	3-344
Exhibit 3.9-1	Study Area Housing Units by Type by Subarea, 2021	3-353
Exhibit 3.9-2	Housing Type by Structure and Units, Ballard	3-354
Exhibit 3.9-3	Housing Type by Structure and Units, Interbay Dravus	3-354
Exhibit 3.9-4	Housing Type by Structure and Units, Interbay Smith Cove	3-354
Exhibit 3.9-5	Housing Type by Structure and Units, SODO/Stadium	3-355
Exhibit 3.9-6	Housing Type by Structure and Units, Georgetown	3-355
Exhibit 3.9-7	Housing Units by Year Built, Study Area	3-356
Exhibit 3.9-8	New Housing Added by Permit Class, 2000-2021	3-357
Exhibit 3.9-9	Displacement Risk Index	3-359
Exhibit 3.9-10	Access to Opportunity Index	3-361
Exhibit 3.9-11	Distance Traveled by Workers in Study Area, 2018	3-362
Exhibit 3.9-12	Top 25 Places of Worker Residence by Count/Percent	3-363
Exhibit 3.9-13	Home Location of Workers with Jobs in the Study Area, 2018	3-364
Exhibit 3.9-14	Washington Environmental Health Disparities Map	3-366
Exhibit 3.9-15	Air Quality: Environmental Exposure Map	3-367
Exhibit 3.9-16	Population Near Heavy Traffic Noise	3-368
Exhibit 3.9-17	Proximity to Hazardous Waste Sites	3-369
Exhibit 3.9-18	Alternative 1—No Action Jobs and Housing, Existing and 2044	3-371
Exhibit 3.9-19	Alternative 1—No Action Housing by Subarea	3-372
Exhibit 3.9-20	Alternative 2 Jobs and Housing, Existing and 2044	3-373
Exhibit 3.9-21	Alternative 2 Housing by Subarea	3-373
Exhibit 3.9-22	Alternative 3 Jobs and Housing, Existing and 2044	3-374
Exhibit 3.9-23	Alternative 3 Housing by Subarea	3-375
Exhibit 3.9-24	Alternative 4 Jobs and Housing Existing and 2044	3-376
-----------------	--	----------------
Exhibit 3.9-25	Alternative 4 Housing by Subarea	3-378
Exhibit 3.9-26	Preferred Alternative Jobs and Housing, Existing and 2044	3-379
Exhibit 3.9-27	Preferred Alternative Housing by Subarea	3-380
Exhibit 3.10-1	Study Area, 2021	3-387
Exhibit 3.10-2	LOS Thresholds for Travel Speeds and Travel Time	3-388
Exhibit 3.10-3	Study Corridors—Ballard Interbay Northend MIC, 2021	3-389
Exhibit 3.10-4	Study Corridors—Greater Duwamish MIC, 2021	3-390
Exhibit 3.10-5	Mode Share Sectors and Screenlines	3-392
Exhibit 3.10-6	LOS Thresholds for Screenlines	3-393
Exhibit 3.10-7	Existing Roadway Network, 2021	3-400
Exhibit 3.10-8	Existing PM Peak Hour LOS	3-402
Exhibit 3.10-9	Existing SOV Mode Share—PM Peak Period	3-405
Exhibit 3.10-10	Existing PM Peak Hour LOS	3-405
Exhibit 3.10-11	Existing Roadway Freight Network—Ballard Interbay Northend MIC, 2021	3-408
Exhibit 3.10-12	Existing Roadway Freight Network—Greater Duwamish MIC, 2021	3-409
Exhibit 3.10-13	Existing Rail Freight Network—Ballard Interbay Northend MIC, 2021	3-411
Exhibit 3.10-14	Existing Rail Freight Network—Greater Duwamish MIC, 2021	3-412
Exhibit 3.10-15	Existing Active Transportation Facilities—Ballard Interbay Northend MIC,	2 115
Eybibit 2 10 16	2021	3-415
EXHIDIL 3.10-10	Existing Active Transportation Facilities—Greater Duwarnish Mic, 2021	3-410 2 417
EXHIDIL 3.10-17	Existing Transit Network, 2021	3-417
EXHIDIL 3.10-18	Existing Transit Network, 2021	3-420
EXHIBIT 3.10-19	Passenger Load Factors on Bus Route across Transit Screenlines	3-421
EXNIDIT 3.10-20	Modal Conflicts in Industrial Areas	3-431
EXNIDIT 3.10-21	Collisions—Ballard Interbay Northend MIC, 2016-2020	3-432
EXNIDIT 3.10-22	Collisions—Greater Duwamish MiC, 2016-2020	3-433
EXNIDIT 3.10-23	Northend MIC, 2044	3-436
Exhibit 3.10-24	Planned Transportation Network Improvements—Greater Duwamish	
	MIC, 2044	3-437
Exhibit 3.10-25	2044 Alternative 1 No Action Person Trips in Study Area—PM Peak Hour	3-446
Exhibit 3.10-26	PM Peak Hour Travel Time LOS—Alternative 1 No Action	3-450
Exhibit 3.10-27	2044 Alternative 1 No Action SOV Mode Share—PM Peak Period	3-453
Exhibit 3.10-28	Screenline Volume-to-Capacity Ratio—Alternative 1 No Action	3-454
Exhibit 3.10-29	PM Peak Hour Passenger Load Factors—Alternative 1 No Action	3-455

Exhibit 3.10-30	2044 Alternative 2 Person Trips in Study Area—PM Peak Hour	3-457
Exhibit 3.10-31	PM Peak Hour Travel Time LOS—Alternative 2	3-459
Exhibit 3.10-32	2044 Alternative 2 SOV Mode Share—PM Peak Period	3-461
Exhibit 3.10-33	Screenline Volume-to-Capacity Ratio—Alternative 2	3-462
Exhibit 3.10-34	PM Peak Hour Average Passenger Load Factors—Alternative 2	3-463
Exhibit 3.10-35	2044 Alternative 3 Person Trips in Study Area—PM Peak Hour	3-463
Exhibit 3.10-36	PM Peak Hour Travel Time LOS—Alternative 3	3-466
Exhibit 3.10-37	2044 Alternative 3 SOV Mode Share—PM Peak Period	3-468
Exhibit 3.10-38	Screenline Volume-to-Capacity Ratio—Alternative 3	3-469
Exhibit 3.10-39	PM Peak Hour Average Passenger Load Factors—Alternative 3	3-470
Exhibit 3.10-40	2044 Alternative 4 Person Trips in Study Area—PM Peak Hour	3-470
Exhibit 3.10-41	PM Peak Hour Travel Time LOS—Alternative 4	3-473
Exhibit 3.10-42	2044 Alternative 4 SOV Mode Share—PM Peak Period	3-475
Exhibit 3.10-43	Screenline Volume-to-Capacity Ratio—Alternative 4	3-476
Exhibit 3.10-44	PM Peak Hour Average Passenger Load Factors—Alternative 4	3-477
Exhibit 3.10-45	2044 Preferred Alternative Person Trips in Study Area—PM Peak Hour	3-477
Exhibit 3.10-46	PM Peak Hour Travel Time LOS—Preferred Alternative	3-479
Exhibit 3.10-47	2044 Preferred Alternative SOV Mode Share—PM Peak Period	3-481
Exhibit 3.10-48	Screenline Volume-to-Capacity Ratio—Preferred Alternative	3-482
Exhibit 3.10-49	PM Peak Hour Average Passenger Load Factors—Preferred Alternative	3-483
Exhibit 3.10-50	Summary of Significant Transportation Impacts	3-484
Exhibit 3.10-51	Impacted Study Corridors—Ballard Interbay Northend MIC, 2044	3-485
Exhibit 3.10-52	Impacted Study Corridors—Greater Duwamish MIC, 2044	3-486
Exhibit 3.11-1	National Register of Historic Places Listed Architectural Properties and	
	Districts	3-509
Exhibit 3.11-2	Seattle Designated Landmarks	3-510
Exhibit 3.11-3	Recorded Archaeological Resources	3-513
Exhibit 3.11-4	Map Showing Archaeological Sensitivity from DAHP Model	3-514
Exhibit 3.11-5	Maritime Washington Heritage Area that Occurs Within the Study Area	3-516
Exhibit 3.11-6	Acres of Zoning or Land Use Concept and Qualitative Relationship to	2 5 2 2
	City of Contural Resources	3-523
EXHIDIL 3.12-1	City of Seattle Parks, Recreation, and Public Shoreline Access	3-539
EXHIBIL $3.12-2$	Traiks in Study Area	3-540
EXHIBIL $3.12-3$	Charalina Assass Daints	3-54U
EXNIDIT 3.12-4	Shoreline Access Points	3-541
EXNIBIT 3.12-5	Parks in and Around the Study Area	3-542

Exhibit 3.12-6	Trails in and Around the Study Area	3-543	
Exhibit 3.12-7	Exhibit 3.12-7 Shoreline Access Points in and Around the Study Area		
Exhibit 3.12-8	Potential Equity Access Goals	3-547	
Exhibit 3.12-9	Heat Watch and King County Results	3-550	
Exhibit 3.12-10	Access to Public Space in Georgetown and South Park	3-552	
Exhibit 3.12-11	Open Space and Recreation Acres Required for Alternative 1	3-553	
Exhibit 3.12-12	Open Space and Recreation Acres Required for Alternative 2	3-554	
Exhibit 3.12-13	Open Space and Recreation Acres Required for Alternative 3	3-555	
Exhibit 3.12-14	Open Space and Recreation Acres Required for Alternative 4	3-556	
Exhibit 3.12-15	Open Space and Recreation Acres Required for Preferred Alternative	3-558	
Exhibit 3.13-1	Fire Battalions and Stations	3-564	
Exhibit 3.13-2	SFD Facility Locations, Equipment, and Staffing for Stations Serving the	3-565	
Exhibit 3 13-3	Seattle Fire Department Emergency Response Incidents 2018-2020	3-567	
Exhibit 3.13-4	Calls for Fire and FMS Services by Subarea 2016-2020	3-567	
Exhibit 3 13-5	Total Calls for Fire and FMS Services in the Study Area and Surrounding	5 507	
EXHIBIT 5.15 5	Vicinity, 2016-2020	3-568	
Exhibit 3.13-6	Response Statistics, 2016-2020	3-569	
Exhibit 3.13-7	SPD Station Locations and Areas Served	3-572	
Exhibit 3.13-8	Police Precinct, Sector, and Beat Boundaries	3-573	
Exhibit 3.13-9	Port of Seattle Properties Near the Study Area, 2020	3-574	
Exhibit 3.13-10	Seattle Police Department Citywide Calls for Service, 2016–2020	3-576	
Exhibit 3.13-11	Seattle Police Department Calls for Service by Area, 2016 and 2019	3-576	
Exhibit 3.13-12	Median Response Times for Priority One Calls Citywide and in Sectors		
	Serving the Study Area, 2016–2020	3-577	
Exhibit 3.13-13	Top 5 Safety Concerns by MCPP in the Study Area in Ranked Order, 2020	3-579	
Exhibit 3.13-14	Port of Seattle Police Department Patrol Team Calls for Service, 2019– 2020	3-581	
Exhibit 3.13-15	Schools and Libraries in or Near the Study Area	3-582	
Exhibit 3.13-16	School Attendance Boundaries: Elementary and Middle Schools	3-583	
Exhibit 3.13-17	Total Housing in Study Area by Alternative	3-589	
Exhibit 3.13-18	Student Generation Rate	3-589	
Exhibit 3.13-19	Student Generation by Subarea based on Net Change in Population	3-590	
Exhibit 3.14-1	West Point Wastewater Treatment Plant Treatment Capacity	3-602	
Exhibit 3.14-2	Length of Wastewater Infrastructure	3-603	
Exhibit 3.14-3	Wastewater and Combined Sewer System	3-604	
_			

Exhibits

Exhibit 3.14-4	Length of Stormwater Infrastructure and Adjacent CSO Outfalls in the Study Area by Subarea	3-605
Exhibit 3.14-5	Stormwater System in the Study Area	3-606
Exhibit 3.14-6	Electrical Transmission Lines by Subarea	3-607
Exhibit 3.14-7	Power Infrastructure in Study Area	3-608
Exhibit 3.14-8	Current and Future Wastewater Service Population in the West Point Wastewater Treatment Facility Service Area Compared to Population in the Study Area	3-609
Exhibit 4.1-1	List of Written Commenters	4-2
Exhibit 4.1-2	List of Verbal Commenters from the Public Hearings	4-6
Exhibit 4.3-1	Written Comments and Responses	4-13
Exhibit 4.3-2	Public Hearing Verbal Comments and Responses	4-104



1.1 Purpose

Seattle's industrial and maritime policies are more than 35-years old, and during that time, the trends and technologies impacting industrial and maritime users have experienced significant change. To reflect those changes as part of a comprehensive strategy to strengthen and grow Seattle's industrial and maritime sectors for the future, the City of Seattle is studying a proposal to update its industrial and maritime policies and industrial zoning. The proposal is informed by recommendations from community input, including an Industrial and Maritime Strategy Council, which resulted in an Industrial and Maritime Strategy Report (Appendix B) that the City of Seattle released in June 2021.

This Environmental Impact Statement (EIS) studies four<u>five</u> alternatives illustrating different potential futures for the city's industrially-zoned lands. The four<u>five</u> alternatives evaluate the effects of potential changes to Comprehensive Plan policies and changes to zoning over a 22-year time horizon (to 2044).

The first alternative is a No Action Alternative that is required by the State Environmental Policy Act (SEPA) and is a basis for comparison. The threefour Action Alternatives (alternatives 2, 3, and 4 and the <u>Preferred Alternative</u>) all apply proposed "future of industry" land use concepts that are based on community input and intended to respond to issues, challenges, and opportunities for the maritime and industrial sectors and adjacent communities. Those future of industry land use concepts consist of three proposed new industrial zones:

 Maritime Manufacturing and Logistics (MML)—This zone would focus on strengthening land use protections for core and legacy industrial and maritime areas to

What is an Alternative?

Alternatives are different ways of achieving objectives that allow decisionmakers to compare the effects of different options. The No Action Alternative is based on current plans, policies, and regulations and is a benchmark against which other alternatives can be measured. Action Alternatives can test a range of ideas, implications, and benefits. The Alternatives in the EIS consider Comprehensive Plan policy amendments and different configurations for possible zoning changes and development standards to achieve the Maritime and Industrial Land Strategy objectives.

better prevent the encroachment of development that is incompatible with industrial and maritime uses. This zone is particularly applicable within Seattle's Manufacturing/Industrial Centers (MICs), near the shoreline or deep-water port, rail and freight infrastructure, and around existing clusters of industrial or maritime suppliers and services.

 Industry / Innovation (II)—This zone aims to encourage new development in multi-story buildings that accommodate industrial businesses mixed with other dense employment uses such as research, design, offices, and technology. By creating density bonuses for employment uses (i.e., office, R&D, etc.) if coupled with industrial uses in the same project, this type of modern industrial development would support high-density employment near transit stations and near existing industrial-commercial areas.

 Urban Industrial (UI)—This zone is designed to foster increased employment and entrepreneurship opportunities with a vibrant mix of affordable, small-scale places for light industry, makers, and creative arts, as well as industry supporting ancillary retail or housing spaces to create better, integrated, and healthier transitions at the edges between industrial areas and neighboring urban villages, residential, and mixed-use areas.

To implement the future of industry land use concepts in each of the Action Alternatives the City of Seattle would:

- Amend the comprehensive plan to add new text policies describing the intent and vision for how these concepts would be applied, including land use, environment, and transportation;
- Amend the industrial zoning section of the land use code to create a new zone designations and corresponding development standards replacing the existing industrial zones;
- Apply new industrial zone classifications to industrial land; and
- Adopt new subarea plans for both the Ballard Interbay Northend and Greater Duwamish MICs.

However, each of the alternatives evaluated in this EIS pose different percentages of the future land use concepts in industrial and manufacturing lands for the purpose of strengthening and growing Seattle's industrial and maritime sectors in the future. The multi-faceted objectives of the proposal are listed in **Section 1.5.1** below.

The following is a summary of the <u>fourfive</u> alternatives, which are described further in **Section 1.5** below.

- **Alternative 1—No Action:** The SEPA-required alternative that would retain current Comprehensive Plan policies, development standards, or zoning maps.
- Alternative 2—Future of Industry Limited: Alternative 2 retains current MIC boundaries. Alternative 2 would implement future of industry land use concepts with a greater emphasis on strengthening protections for core and legacy industrial and maritime activities. The proposed MML zone, would cover approximately 90% of industrial lands. Application of the proposed II and UI zones would be limited in scope, covering approximately 10% of current industrial areas. II zoning would be focused on existing Industrial Commercial (IC) zones and areas within approximately ¼ mile of light rail stations. UI zoning would be focused on existing Industrial Buffer (IB) zones and the existing Stadium Transition Area Overlay. There are no changes to housing allowances in Alternative 2.
- Alternative 3—Future of Industry Targeted: Alternative 3 would strengthen protections for core industrial uses in the MML zone on approximately 86% of industrial lands. It applies a mix of the proposed II and UI zones in targeted geographies covering 14% of industrial lands. Compared to Alternative 2, II zoning is expanded to include areas an estimated ½ mile from light rail stations and UI zoning would be applied in additional areas in Ballard and the north shore of Lake Union. Alternative 3 creates limited flexibility for additional industry-supportive housing in the UI zone that would result in an estimated 610 new homes in industrial zones. Alternative 3 removes focused land in Georgetown/South Park from the MIC and converts it to a non-industrial mixed-use zone.

- Alternative 4—Future of Industry Expanded: Alternative 4 would also strengthen
 protections for core industrial uses in the MML zone on approximately 87% of industrial
 lands. Similar to Alternative 3, Alternative 4 would mainly apply II zoning in existing IC zones
 and within a ½ mile from light rail stations, though with a greater expansion of the II zone
 in areas in Ballard and SODO. Compared to Alternative 3, the UI zone would be applied to
 a larger area in SODO, but to fewer areas in Ballard. This alternative includes additional
 flexibility for industry-supportive housing that could result in an estimated 2,195 new
 homes in industrial zones. Just like Alternative 3, Alternative 4 removes focused land in
 Georgetown/South Park from MICs and convert it to a non-industrial mixed-use zone.
- Preferred Alternative—Futuxe of Industry Balanced: The Preferred Alternative incorporates features of multiple Draft EIS Action Alternatives. It includes modifications to address comments on the Draft EIS and reduce impacts identified for Draft EIS alternatives. The Preferred Alternative would implement the proposed land use concepts, and strengthen policy protections for industrial lands in MICs, while affording some greater flexibility for lands outside of MICs.
 - <u>The MML zone would cover approximately 85% of industrial lands, while proposed II and UI zones would be targeted in scope and cover approximately 14% of current industrial areas. II zoning would be focused on existing Industrial Commercial (IC) zones inside of MICs and areas within approximately 1/2 mile of light rail stations. UI zoning would be focused on existing Industrial Buffer (IB) zones. Unlike alternatives 2, 3, and 4, the Preferred Alternative would retain existing IC zoning only in areas outside of MICs.
 </u>
 - The Preferred Alternative would allow limited industry-supportive housing in the UI zone as a conditional use subject to additional criteria to minimize potential conflicts. Concepts to remove focused land from the MIC in Georgetown and South Park are carried forward. Additionally, new areas for housing in mixed use zones are added in the Preferred Alternative outside MICs (west Ballard, and Judkins Park). Overall, a lower amount of industry-supportive housing production would result compared to Draft EIS alternatives 4 within MICs in the UI zone (1,475 units). Though a greater amount of new unrestricted housing is projected outside of the MICs than any Draft EIS alternative (1,534 units), the combined growth of housing would be less than Alternative 4.
 - <u>The Preferred Alternative includes a more nuanced zoning approach for a proposed</u> <u>mixed use zone in central Georgetown to preserve arts spaces and historic structures</u>, <u>and greater application of UI zoning around Georgetown to create more neighborhood</u> <u>cohesion</u>.
 - <u>The Preferred Alternative features a reduced total amount of job growth, most similar</u> to Draft EIS Alternative 2. Projections are adjusted downward to reflect conditions in commercial/office occupancy post-COVID and timelines for new light rail construction. The adjusted projections acknowledge that it will likely take longer to achieve levels of employment growth.

This chapter is the first of a series of chapters contained in the Draft EIS that provide a summary and more in-depth environmental review of the proposal and alternatives. The Draft <u>Final</u> EIS is organized as follows:

- Chapter 1 Summary
- Chapter 2 Proposal & Alternatives
- Chapter 3 Environment, Impacts, & Mitigation Measures
- Chapter <u>4</u> Comments & Responses
- <u>Chapter 5</u> Acronyms & References
- <u>Chapter 6</u> Appendices

1.2 Study Area

Most industrial land in Seattle is located within two Manufacturing Industrial Centers (MIC): Seattle's Greater Duwamish Manufacturing and Industrial Center (Greater Duwamish MIC) and Ballard Interbay North Manufacturing Industrial Center (BINMIC), important as a freshwater harbor. Within the MICs, subareas are defined—Ballard, Interbay Dravus, Interbay Smith Cove, SODO/Stadium, and Georgetown/South Park. The Greater Duwamish MIC and BINMIC contain 12% of Seattle's total land area. Other industrially zoned land that is outside a MIC is included in the study area, most of which is on shorelines of Lake Union and by Judkins Park. See **Exhibit 1.2-1**.



1.3 Planning Context & Outreach

1.3.1 Emerging Factors Affecting Seattle's MICs

MICs are regional designations and are defined in the City's Comprehensive Plan as home to the city's thriving industrial businesses. Like urban centers, they are important regional resources for retaining and attracting jobs and for maintaining a diversified economy. Seattle's manufacturing and maritime sectors generate middle-wage jobs that are cornerstones of a thriving and livable city. There are currently around 98,500 industrial jobs (2018) or about 15% of total jobs in the city—about two-thirds of these jobs are available with only a high school diploma, and over half of the jobs in the maritime sector are available to persons with no formal educational training. Average earnings per worker are over 70% of the Area Median Income (AMI) in the construction, aerospace/aviation, and logistics sectors, and a high number of jobs in logistics, maritime, and manufacturing sectors remain unionized and provide high quality benefits.

Since MICs were established in 1994 there have not been large_scale alterations to their geographic boundaries. Today, zoning within MICs must be one of four industrial zones in the Seattle Municipal Code (SMC). Those zones regulate the uses and activities that can take place in industrial areas, limiting them to prioritize manufacturing and industrial activities envisioned by the comprehensive plan. While manufacturing and maritime sectors today are strong, emerging factors affecting them include those listed below. See **Chapter 2** for a description of each of the emerging factors:

- Pressures to convert Industrial lands
- Emerging technologies and processes
- Unintended development
- Pending port, transportation, and new industrial building typology
- Environment and climate change
- Equity and accessibility

1.3.2 Equity & Environmental Justice

The study area includes territories of indigenous tribes <u>including the Muckleshoot, Snoqualmie,</u> <u>Suquamish, Duwamish, and Tulalip Tribes</u>; Euro-American <u>and other non-native</u> settlement and industrial development altered the natural character of this area and impacted tribal treaty rights. Since settlement the study area has had a growing industrial and maritime economy connected to the Puget Sound Region and West Coast.

Current conditions information indicates that the study area contains few housing units but is bordered by residential areas and nearby schools; the study area also contains parks that visitors use. These residents and users of the study area have a higher relative exposure to air

emissions, noise, and light and glare. Some lands in the study area contain hazardous waste or cleanup sites. These environmental conditions also affect the large numbers of workers that come every day to the study area and then commute to homes either elsewhere in Seattle or in King County and beyond.

Equity and environmental justice are considered throughout the EIS. **Chapter 2** describes existing environmental justice principles and actions that are under consideration as the alternatives are reviewed.

Section 1.7 addresses findings of the alternatives and relationship to environmental justice and equity. **Chapter 3, Section 3.8** addressing land use includes an overview of past land use policies and other actions that had inequitable outcomes.

1.3.3 Mayor's Industrial & Maritime Strategy

In 2019 Mayor Durkanthe City convened an Industrial and Maritime Strategy Advisory Council to chart a blueprint for the future of industrial land in Seattle with a focus on providing equitable access to high-quality, family-wage jobs and entrepreneurship opportunities. The Advisory Council included representation from citywide stakeholders and stakeholders from four neighborhood subareas. Stakeholders represented a diverse range of interests including maritime and industrial businesses, labor, residents of adjacent neighborhoods, developers, and industry groups.

In May 2021 the Advisory Council recommended 11 broad strategy statements to guide future actions to support the maritime and industrial sectors, and advance equitable access to family-wage employment, particularly for Black, Indigenous, and people of color (BIPOC) youth. **Chapter 2** describes the Advisory Council process and recommendations, and the Mayor's Industrial and Maritime Strategy Report is at **Appendix B**.

The key land use recommendations of the stakeholders informed the EIS alternatives.

1.4 SEPA Process

1.4.1 Environmental Review

Process

Under SEPA agencies conduct environmental review of actions that could affect the environment. For actions that have the potential for significant impacts, preparation of an EIS is required. An EIS is a useful tool that provides detailed information to the public, agencies, tribes, and City decision-makers about the environmental effects of a plan or project before a decision is made.

84

The EIS process involves the following steps: (1) scoping the contents of the EIS with agencies, tribes, and the public; (2) preparing a draft EIS with a comment period; (3) responding to comments and developing a preferred alternative; and (4) developing legislation. With the issuance of the Draft EIS, the EIS process is in phase 2. See **Exhibit 1.4-1**.





Note: This diagram was revised to change proposed legislation timing from Summer/Fall 2022 to Winter 2022/2023. Source: BERK, 2022+.

Non-Project EIS

This document is a non-project EIS that analyzes the proposals and alternatives broadly across the study area. See **Exhibit 1.4-2** below for features of a non-project EIS. SEPA identifies that a non-project EIS is more flexible and studies a range of alternatives comparatively to support the consideration of plans, policies, or programs (WAC 197-11-442). A non-project EIS does not provide site-specific detailed analysis.

Feature	Project Environmental Review	Non-Project Environmental Review (WAC 197-11-442, -774)
Location	Site-specific	Areawide
Analysis Level of Detail	Detailed	Broad / order-of-magnitude
Alternatives	Specific construction proposals	Conceptual based on vision
Mitigation	Specific, alters project, project proponent responsibility	Broader; changes policies, plans, or code. City or future developer responsibility.
Future Environmental Review	No additional SEPA review	Subject to additional SEPA Review

Exhibit 1.4-2 Comparison of Project and Non-Project Environmental Review

Sources: WAC 197-11-442, 2021; BERK, 2021.

1.4.2 Public Comment Opportunities

Scoping

The scoping process is intended to identify the range of potential significant impacts on the built and natural environment that should be considered and evaluated in the EIS. The City issued a Scoping Notice on July 8, 2021 with a 30-day public comment period that ran through August 9, 2021. Virtual scoping meetings were held during the comment period at 9:00 a.m. on July 21 and 6:00 p.m. on July 26, 2021. The City also published an <u>information website and</u> <u>online survey</u> as part of scoping.

The input received during the scoping period included:

- Written Comments: 105 letters and emails by 103 commenters
- Survey: 46 participants
- Public meeting participants: 7 participants

See **Appendix A** for the scoping report.

As part of scoping, the City identified a range of topics to explore in the EIS:

- Natural and Biological Resources and Resiliency: Soils/Geology, Air Quality/Greenhouse Gas, Water Resources, Plants and Animals
- Environmental Health and Compatibility: Contamination, Noise, Light and Glare
- Working, Living, and Mobility: Land and Shoreline Use, Housing, and Transportation
- Cultural and Recreational Resources: Historic, Archaeological & Cultural Resources, Open Space and Recreation
- Public Services and Utilities: Police, Fire, Schools, Libraries, Wastewater, Stormwater, and Power

Scoping comments indicated that air quality/greenhouse gas, contamination, transportation, and land and shoreline use were most important to address in the EIS. Commenters also gave input on alternatives to be studied, typically by indicating which of the scoping alternatives fit their views of the area or properties, or requesting adjustments. In response to the scoping comments one alternative was modified to include an evaluation of potentially increasing the size of use limit on indoor recreation facilities from 10,000 square feet to 50,000 square feet. A full response to scoping comments can be found in the Scoping Report.

This Final EIS includes comments gathered during the Draft EIS Comment period described further below. See **Chapter 4 Comments & Responses**.

Draft <u>& Final</u>EIS

This<u>The</u> Draft EIS <u>identifies</u><u>identified</u> environmental conditions, potential impacts, and measures to reduce or mitigate any unavoidable adverse impacts that could result from an update to policies and zoning for Seattle's maritime and industrial sectors. The Draft EIS

86

Port of Seattle; picture replaced per responses to comments in Chapter 4.



alternatives and topics were developed based on a review of scoping comments and prior Industrial and Maritime Strategy engagement results.

Public and agency comments arewere invited on theis Draft EIS. Written and verbal comments arewere invited during the 45-day public comment period (December 16, 2021 to January 31, 2022) following issuance of theis Draft EIS. The City extended the comment period to March 2, 2022 to allow more time for review.

The City will hold future held public engagement events during or following the 45-day comment period to help refine its preferred alternative. In addition, the City conducted a series of meetings with the South Park and Georgetown community members in neighborhood locations and included comments from these communities through April 15, 2022. Public comments will be are considered and addressed in the Final EIS in **Chapter 4**. Please see the Fact Sheet at the beginning of this Draft Einal EIS for the dates of the public comment period, as extended, and public meeting. Meetings and comment periods regarding the proposals are described on the City's project webpage: Industrial and Maritime Strategy—OPCD | seattle.gov.

Final EIS & Proposed Legislation

A Final EIS <u>will bewas</u> issued in 2022 and <u>will</u>include<u>s</u> responses to public comments received during the Draft EIS comment period. <u>Changes to the Draft EIS are shown in strikeout and</u> <u>underline. The Final EIS also studies a Preferred Alternative that responds to the comments.</u>

Following the EIS process, the City will develop specific policy and zoning proposals that will be the subject of public meetings and public hearings by the City Council.

87

1.5 Objectives, Proposal, & Alternatives

1.5.1 Objectives

The State Environmental Policy Act (SEPA) requires a statement of proposal objectives and the purpose and need to which the proposal is responding. Alternatives are different means of achieving the objectives.

The proposal would update Comprehensive Plan policies concerning industrial land and update the city's industrial zoning. The objectives behind this proposal are multi-faceted and seek to address the City's industrial and maritime sectors holistically. The objectives are informed by the recommendations of an <u>Industrial and Maritime Strategy</u> stakeholder process. Objectives are identified in four overlapping categories of people, place, and production and process. See **Exhibit 1.5-1**.

Exhibit 1.5-1 Objectives of the Proposal

People A. Increase the quantity of living wage jobs generated from activity on Seattle's currently designated industrial lands. B. Improve equitable access to the living wage jobs from these lands by increasing the proportion of the jobs held by: racial minorities, women, and persons without traditional 4-year college diplomas. C. Improve environmental health for people who live or work in or near industrial areas—especially at transitions to residential areas or urban villages. Place D. Provide long-term predictability to stakeholders that will support renewed investment in facilities, buildings, and infrastructure. E. Promote mutually reinforcing mixes of activities at the transitions between industrial areas and urban villages or residential neighborhoods. F. Support industrially compatible employment dense transit oriented development at existing and future high capacity transit stations. G. Increase access to workforce and affordable housing for employees in industrial maritime sectors, without creating land use conflicts that displace industrial uses. Production H. Position Seattle's industrial areas to respond competitively to new industrial and manufacturing processes and practices. I. Ensure available and adequate locations for components of regional and statewide supply chains and regional economic clusters. J. Increase the amount and accessibility of space for prototyping, entrepreneurship, and business incubation. K. Strengthen economic resiliency with the capacity to produce products locally and ensure stable distribution networks Process L. Develop Comprehensive Plan policies based on the Industrial and Maritime Strategy. M. Develop a subarea plan for the MICs that supports VISION 2050, accommodates growth targets, and the Puget Sound Regional Council Regional Centers Framework for MICs.

Source: City of Seattle, 2021.

1.5.2 Proposal

The proposal considers Comprehensive Plan policy amendments and changes to zoning and development standards that could help meet the objectives defined in **Section 1.5.1**. The EIS includes three<u>four</u> future of industry alternatives (alternatives 2, 3, and 4<u>and the Preferred</u><u>Alternative</u>) that would make different geographic combinations of zoning changes and degrees of change to development standards in industrial zones. A No Action Alternative with no changes to policies or zoning is also considered. The EIS addresses land use compatibility, and consistency with City and State plans and regulations.

1.5.3 Land Use Concepts

The future of industry alternatives (alternatives 2, 3, and 4<u>and the Preferred Alternative</u>) would apply proposed new land use concepts that are based on community input and intended to respond to issues, challenges, and opportunities for the maritime and industrial sectors and adjacent communities. The application of the concepts in the study area is provided in areawide maps in **Section 1.5.5** through **1.5.91.5.8**. Close ups of the land use/zoning maps are in **Appendix C Alternative Future Land Use Zoning Maps**.

Three proposed land use concepts are integrated to different degrees in the future of industry alternatives and include:



A description of concept is provided below and following that a full description of each alternative and how it assimilates the land use concepts.

Maritime, Manufacturing, and Logistics (MML)

The Maritime, Manufacturing, and Logistics (MML) land use concept would intend to strengthen established economic clusters and expand equitable access to jobs. <u>There would be a high likelihood that a substantial proportion of jobs in MML would be union represented.</u> Seattle's industrial areas host valuable economic clusters including fishing, logistics, maritime, aerospace, brewing and distilling, and others that depend on access to water or other irreplaceable supporting infrastructure. MML would be applied in locations near such infrastructure and would strengthen the policy and zoning protections for maritime and industrial uses. See **Exhibit 1.5-2**.





Challenges Addressed

- Market pressure for conversion away from industrial land.
- Vulnerabilities due to the interdependence of business within clusters.
- A pattern of "one off" zoning decisions that have removed industrial land.
- Encroachment of non-industrial uses in industrial zones.

Features/Development Standards

- Strictly limit allowable uses to industrial, manufacturing, maritime and similar uses.
- Do not allow new residential uses.
- Strict maximum size of use limits on non-industrial uses such as retail, office, and restaurants.

Source: City of Seattle, 2021.

Industry and Innovation (II)

The Industry and Innovation (II) land use concept would intend to support economic innovation and capitalize on emerging opportunities including expanded or new light rail stations in industrial areas. It would intend to support emerging formats for industrial activity that are more design and research oriented than traditional industrial uses. It would intend to introduce nodes of high-density employment and multi-modal access near transit. Industry and Innovation would also intend to encourage new investment in high quality industrial space. See **Exhibit 1.5-3**.



Exhibit 1.5-3 Industry and Innovation Proposed Land Use Concept

Source: City of Seattle, 2021.

Urban Industrial (UI)

The Urban Industrial (UI) land use concept would intend to foster vibrant districts that support a mix of local manufacturing, production, arts, and a sense of place. Urban Industrial would be located in areas adjacent to Seattle's designated urban villages. UI would intend to create thoughtful integration between the edges of Seattle's MICs and adjacent neighborhoods. It would seek to improve environmental health, walkability, and comfort in these areas. The UI concept would seek to leverage the industrial aesthetic, including adaptive reuse of buildings. In some alternatives, UI could allow a limited amount of new industry-supportive housing. See **Exhibit 1.5-4**.



Exhibit 1.5-4 Urban Industrial Proposed Land Use Concept

Challenges Addressed

Features/Development Standards

- Environmental health impacts that affect residents near industrial areas.
- Uncomfortable conditions for pedestrians, cyclists, and transit riders.
- Strong demand for worker housing near jobs.
- Lack of small or affordable space for makers, creatives, and artists.
- Strict maximum size of use limits for stand-alone non-industrial uses.
- Flexibility for larger size of use for retail or office space that is combined with a production or making use on-site.
- A moderate increase in allowed floor area compared to existing industrial zones.
- Development standards such as setbacks and landscaping that are more urban in nature, compared to the existing industrial buffer zones.
- Standards for pedestrian and cyclist-oriented frontage improvements.
- Expanded allowances for limited industry-supportive housing such as caretakers' quarters and maker studios (alternatives 3 and 4 only).

Source: City of Seattle, 2021.

Comprehensive Plan Policy Amendments

The Action Alternatives include new goals and policies relating to the industrial and maritime sectors that would be adopted into the City's Comprehensive Plan. The proposed amendments would establish a new land use framework to implement the concepts discussed above, and new policies concerning transition to clean fuels.

Below is a summary for how the new policies would be integrated into the existing Comprehensive Plan. Specific draft goal and policy language can be found in **Appendix D**.

- Add two new land use Goals in the industrial areas section, in addition to existing Land Use Goal 10:
 - Support employment-dense activities and emerging industries that require greater flexibility in the range of on-site uses and activities.
 - Develop transitions between industrial areas and adjacent neighborhoods that support healthy communities, reduce adverse environmental impacts, and minimize land use conflicts.
- Introduce new land use **Policies** that would support implementation of the new goals.
 Policy amendments would include a new land use framework for the MML, II, and UI zones, establishing their intent and purpose and locational guidance.
- Introduce a new policy to limit changes in MIC boundaries to major updates of the Comprehensive Plan or following a comprehensive city-led study.
- Establish the city's intent to work with owners or future owners of the Washington Oregon Shippers Cooperative Association (WOSCA) and Interbay Armory sites on a master planning process for future reuse according to the goals and policies for MICs.
- Introduce new or strengthened policies into chapters of the Comprehensive Plan that may include the Transportation, Environment, or Container Port elements encouraging transitions to clean fuels and decarbonization of industrial and maritime activities.

Port of Seattle Pier 86 Grain Terminal; picture replaced per responses to comments in Chapter 4.



Manufacturing Industrial Center Subarea Plan

The Puget Sound Regional Council's VISION 2050 and the Regional Centers Framework calls for jurisdictions to adopt subarea plans for regional centers. The City of Seattle anticipates updating existing subarea plans for the two MICs that were prepared in the late 1990s.

The subarea plans should provide or address:

- A Center Plan Concept/Vision and be the product of Regional Collaboration
- Demonstrate Environmental Protection, Climate Change Adaptation and Mitigation, and Vulnerable Community Protection
- Center Size and Boundaries and Land Use / Development Patterns
 - Industrial Employment Centers should have at least 10,000 existing jobs and plan for at least 20,000 jobs.
 - Regional manufacturing/industrial centers must retain a minimum 50% industrial employment.
 - The plan should include policies and identify programs that retain at least 75% of industrially zoned land for core industrial uses (e.g., manufacturing, transportation, warehousing, and freight).
- Economy and Market Potential
- Multimodal and Intermodal Transportation
- Public Services
- Innovation, Engagement, and Racial Equity

More information and evaluation are included in **Section 3.8 Land & Shoreline Use** addressing the relationship of the alternatives to plans and policies.

1.5.4 Regulatory Concepts

In the Action Alternatives, the proposal would implement the land use concepts by applying new Maritime Manufacturing and Logistics (MML), Industry and Innovation (II), and Urban Industrial (UI) zones. The new zones would replace existing industrial zones on the official land use map, and the new zones would be new development standards in the text of the Seattle Municipal Code. The new zones would have the intention of achieving the features of the proposed land use concepts.

Exhibit 1.5-5 is a brief overview of the proposed zones. A more complete description of the zones and how they would work to a level of detail sufficient for assessing environmental impacts is found in **Chapter 2**. See also development standards in **Appendix G**.

Development Standard	Maritime Manufacturing and Logistics (MML)	Industry and Innovation (II)	Urban Industrial (UI)
Locational Criteria	 Within a M/IC. Large parcel sizes. Proximate to water and port facilities. Proximate to rail or other freight infrastructure. Buffered from urban villages and residential zones. 	 Within ¼-½ mile walkshed of an existing or planned high capacity transit station. Within a MI/C or land previously in an industrial zone outside a MI/C. 	 Within a designated M/ IC, or an area with existing industrial/manufacturing/ maritime uses. Proximate to an urban village, or an existing agglomeration of residential uses.
Summary	 Wide range of light and heavy industrial uses permitted. Strict size of use and maximum FAR limits for non-industrial uses. Maximum FAR of 2.5, similar to existing industrial zones. 	 An incentive bonus system allowing dense non- industrial employment uses contingent on the construction of bona-fide new light industrial space. Substantially higher height limits and FAR limits than existing industrial zones. No expansion of housing allowances in any alternative. 	 Increased allowances for ancillary retail and restaurant spaces with on- site industrial uses. Higher FAR limits than existing industrial zones, and decreased setback requirements for more urban structures. Increased multi-modal frontage improvement requirements and urban landscaping requirements. Expansion of some limited industry-supportive housing allowances in alt<u>ernatives</u>: 3 and 4 <u>and the Preferred Alternative</u> only.

Exhibit 1.5-5 Development Standards by Land Use Concept

Source: City of Seattle, 2021.

95

1.5.5 Alternative 1—No Action

The No Action Alternative is required by SEPA. No change to current Comprehensive Plan policies, development standards, or zoning maps are included under this alternative. The existing zone classifications established in 1987—the Industrial General (IG1 and IG2) zones, the Industrial Commercial (IC) zone, and the Industrial Buffer (IB) zone—would remain. IG is the core industrial zone that prioritizes industrial and maritime uses and covers most of the MICs. IC allows for a mix of industrial and commercial uses. IB offers development standards intended to buffer industrial uses from adjacent neighborhoods and includes a focus on setbacks, limited heights, and landscaping. See **Exhibit 1.5-7**. The No Action Alternative retains the following:

- No change to IG zones that cover 90% of industrially zoned areas.
- No change to IC zone that cover 5% of industrially zoned areas.
- No change to IB zone that cover 5% of industrially zoned areas.
- Residential uses are prohibited with the exception of one caretaker quarters per industrial business, artist studio housing in existing structures, and housing that predates industrial zoning.

See Exhibit 1.5-6 with acres and percent of zones.

Exhibit 1.5-6 Alternative 1—No Action Zoning Districts (Acres)

Zoning Districts	Acres	Share
Industrial General (IG1/IG2)	6,273	90.4%
Industrial Buffer (IB)	316	4.6%
Industrial Commercial (IC)	347	5%
Total	6,936	100%

Source: City of Seattle, 2021.



Exhibit 1.5-7 Alternative 1—No Action Zoning Map

Sources: City of Seattle, 2021; BERK, 2021.

Port of Seattle; picture replaced per responses to comments in Chapter 4.



The City of Seattle will be planning for total citywide job growth of 169,500 jobs over the 20-year planning horizon of the One Seattle Comprehensive Plan major update. Employment growth of 23,500 projected under Alternative 1 in the study area would represent about 14% of total citywide job growth. The study area contains the MICs and additional industrial zoned areas outside of MICs. The 14% share of total citywide job growth under Alternative 1 is an increase to the share of job growth planned for industrial areas during the previous Seattle 2035 20-year planning horizon, which estimated 8% of the city's job growth in MICs (and not including industrial zoned lands outside of MICs).

Current jobs are majority industrial (55%). The total number of jobs is expected to increase by 23,500 with just over half of that industrial. When added to base jobs, the share of industrial jobs in 2044 would slightly decrease (54%). The current number of dwellings is small and is only projected to increase by 75 units, assumed to be caretakers' units and artist/studio quarters. Detailed summaries of projected employment mix and housing by sub-areas are included in **Chapter 2**.

Under Alternative 1—No Action, most industrial jobs as well as total jobs are located in the SODO/Stadium and Georgetown/South Park subareas, with less in the Ballard, Interbay Dravus, and Interbay Smith Cove subareas.

1.5.6 Alternative 2—Future of Industry Limited

Alternative 2—Future of Industry Limited applies the proposed land use concepts with relatively less Industry and Innovation and Urban Industrial than the other two Action Alternatives. See **Exhibit 1.5-9**.

Alternative 2 proposes the following:

- Updates industrial land use policies to anticipate future innovations and trends.
- Strengthens protections for industrial uses in MML zone covering 90% of industrial lands.
- Applies a mix of II and UI zone concepts in 10% of current MIC areas, including an estimated ¼ mile from light rail stations.
- No expansion of housing allowances.
- Does not remove any land from MICs.

See zoning district acres in **Exhibit 1.5-8**.

Exhibit 1.5-8 Alternative 2—Future of Industry Limited Zoning Districts (Acres)

Zoning Districts	Acres	Share
 Maritime, Manufacturing, and Logistics (MML) 	6,251	90.1%
Urban Industrial (UI)	222	3.2%
Industry and Innovation (II)	463	6.7%
Total	6,936	100%

Source: City of Seattle, 2021.

The total number of jobs is expected to increase by 34,400 with 72% of that industrial in nature; the total share of industrial jobs in 2044 would increase from 55% in 2018 to 60% in 2044. Employment growth of 34,400 projected under Alternative 2 in the study area would represent about 20% of total citywide job growth that the City would be planning for during the 20-year planning horizon of the One Seattle Comprehensive Plan major update. This would represent a shift of a moderately greater share of the city's expected employment growth into industrial areas compared to past trends and the previous 20-year Comprehensive Plan planning horizon.

The number of dwellings is projected to increase by 80 units and assumed to be caretakers' quarters and some artist/studios.





Sources: City of Seattle, 2021; BERK, 2021.

1.5.7 Alternative 3—Future of Industry Targeted

Alternative 3—Future of Industry Targeted applies the proposed land use concepts with a greater share of Industry and Innovation and Urban Industrial than Alternative 2. See **Exhibit 1.5-11**.

Alternative 3 proposes the following:

- Updates industrial land use policies to anticipate future innovations and trends.
- Strengthens protections for industrial uses in MML zones covering 86% of industrial lands.
- Applies a mix of II and UI zone concepts in 14% of current MIC areas, including an estimated ½ mile from light rail stations.
- Expansion of limited industry-supportive housing in UI zone concept.
- Removes focused land in Georgetown/South Park from MICs.

Acres by zoning are shown in **Exhibit 1.5-10**.

Exhibit 1.5-10 Alternative 3—Future of Industry Targeted Zoning Districts (Acres)

Zoning Districts	Acres	Share
Maritime, Manufacturing, and Logistics (MML)	5,968	86.0%
Urban Industrial (UI)	426	6.1%
Industry and Innovation (II)	516	7.4%
Mixed-Use Commercial	26	0.4%
Total	6,936	100%

Source: City of Seattle, 2021.

The total number of jobs would increase by 57,400 with 60% of those industrial jobs; the total share of industrial jobs in 2044 would slightly decrease from 55% in 2018 to 54% in 2044. This level of employment growth would shift a sizeable share of Seattle's total employment growth into MICs compared to historic growth rates in MICs. Employment growth of 57,400 projected under Alternative 3 in the study area would represent about 34% of total citywide job growth that the City is planning for during the 20-year planning horizon of the One Seattle Comprehensive Plan major update. This would represent a substantial shift of the total share of the city's expected employment growth into MICs and industrial areas compared to past trends and the previous 20-year Comprehensive Plan planning horizon.

The number of dwellings is projected to increase by 610 units <u>in industrial zones</u>, with a combination of caretakers' quarters and makers studios under modified allowances for industry-supportive housing in the UI zone.

In addition to the housing in industrial zones, some more new housing would result in focused areas in Georgetown and South Park that would be removed from the MIC and placed in a mixed-use zone. This would result in a total of 1,078 housing units over the study time horizon on land that is removed from industrial zoning under Alternative 3.





Sources: City of Seattle, 2021; BERK, 2021.

1.5.8 Alternative 4—Future of Industry Expanded

Alternative 4—Future of Industry Expanded applies the proposed land use concepts with a greater share of Industry and Innovation and Urban Industrial than Alternative 2. This alternative expands limited housing allowances compared to Alternative 3. See **Exhibit 1.5-13**.

Alternative 4 proposes the following:

- Updates industrial land use policies to anticipate future innovations and trends.
- Strengthens protections for industrial uses in MML zones covering 87% of industrial lands.
- Applies a mix of II and UI zone concepts in 13% of current MIC areas, including an estimated ½ mile from light rail stations.
- Expansion of limited industry-supportive housing in UI zone concept.
- Removes focused land in Georgetown/South Park from the MIC.
- Increases maximum size of use limit for indoor sports and recreation uses.

The zoning districts by acres is listed in **Exhibit 1.5-12**.

Exhibit 1.5-12 Alternative 4—Future of Industry Expanded Zoning Districts (Acres)

Zoning Districts	Acres	Share
Maritime, Manufacturing, and Logistics (MML)	6,035	87.0%
Urban Industrial (UI)	279	4.0%
Industry and Innovation (II)	600	8.7%
Mixed-Use Commercial	22	0.3%
Total	6,936	100%

Source: City of Seattle, 2021.

The total number of jobs would increase by 59,200 with 49% of those industrial jobs; the total share of industrial jobs in 2044 would slightly decrease from 55% in 2018 to 53% in 2044. Like Alternative 3, this level of employment growth would shift a sizeable share of Seattle's total employment growth into MICs compared to historic growth rates in MICs. Employment growth of 59,500 projected under Alternative 4 in the study area would represent about 35% of total citywide job growth that the City would be planning for during the 20-year planning horizon_ of the One Seattle Comprehensive Plan major update. Similar to Alternative 3, this would represent a substantial shift of the total share of the city's expected employment growth into MICs and industrial areas compared to past trends and the previous 20-year Comprehensive Plan planning horizon.

The number of dwellings is projected to increase by 2,195 units in industrial zones, with a combination of caretakers' quarters and makers studios under modified allowances for industry-supportive housing in the UI zone. An additional 1,078 dwellings would be located in new mixed use areas as in Alternative 4.





Sources: City of Seattle, 2021; BERK, 2021.

1.5.9 <u>Preferred Alternative—Future of Industry Balanced</u>

The Preferred Alternative—Future of Industry Balanced applies the proposed land use concepts with a combination of features of alternatives 2, 3, and 4, and new features and refinements to address comments and reduce impacts. The Preferred Alternative proposes the following:

- <u>Updates industrial land use policies to anticipate future innovations and trends like the Draft</u> <u>EIS action alternatives.</u>
- <u>Strengthens protections for industrial uses in maritime, manufacturing and logistics zones</u> <u>covering 85% of industrial lands like the Draft EIS action alternatives.</u>
- Applies a mix of II and UI zone concepts in 14% of current MIC areas, including an estimated ½ mile from light rail stations like Draft EIS Alternative 4.
- Expands limited industry-supportive housing in the UI zone, subject to a conditional use process and more location and performance criteria than Draft EIS alternatives 3 or 4, and maintaining a limit on density as in alternatives 3 or 4.
- The industry-supportive housing criteria could be met in one of two ways—either by limiting occupancy to caretakers or makers (as in alternatives 3 and 4), or by providing a minimum of 50% of any housing units that are created to households with incomes at 90% of AMI or below.
- Removes focused land in Georgetown/South Park from the MIC similar to alternatives 3 and 4.
- <u>Retains Industrial Commercial zoning for some existing industrially-zoned areas outside of</u> <u>MICs like Alternative 1.</u>
- Converts focused areas of industrial zoning outside of MICs to new mixed use zones that would allow housing. Together with the change around Georgetown/South Park the new mixed use zones would equal about 1% of the study area.
- Applies the MML zone to the WOSCA and Armory sites until site specific master planning can be completed.
- Increases the amount of proposed UI zoning around Georgetown to create more neighborhood cohesion.
- Applies a nuanced approach to a proposed mixed use zone in central Georgetown that reflects community priorities including preserving arts space and historic aged structures.
- Increases incentives for development feasibility in the II zone compared to Draft EIS alternatives.
- Increases maximum size of use limit for indoor sports and recreation uses like Alternative 4.

The zoning districts by acres is listed in **Exhibit 1.5-14**. See **Exhibit 1.5-15** for a map.

Exhibit 1.5-14 Preferred Alternative—Future of Industry Balanced Zoning Districts (Acres)

Zoning Districts	<u>Acres</u>	<u>Share</u>
Maritime, Manufacturing, and Logistics (MML)	5,895	<u>85.0%</u>
Urban Industrial (UI)	<u>376</u>	<u>5.4%</u>
Industry and Innovation (II)	<u>612</u>	<u>8.8%</u>
Mixed-Use Commercial	<u>53</u>	<u>0.8%</u>
Total	<u>6,936</u>	<u>100%</u>

Source: City of Seattle, 2022.





Sources: City of Seattle, 2022; BERK, 2022.

The total number of jobs would increase by 35,545 with 46% of those industrial jobs. The absolute number of industrial jobs would be greater than the No Action Alternative. The total share of jobs in 2044 that are industrial would slightly decrease from 55% in 2018 to 53% in 2044. The Preferred Alternative would make a moderate shift of Seattle's total employment growth into MICs compared to historic growth rates in MICs. Employment growth of 35,545 projected under the Preferred Alternative in the study area would represent about 18% of the net citywide job growth that the city would be planning for during the 20-year planning horizon of the One Seattle Comprehensive Plan major update. Projections are adjusted downward compared to the Draft EIS alternatives to reflect conditions more realistically in commercial/ office demand post-COVID and in consideration of Sound Transit's timeline for completion of light rail construction of the West Seattle to Ballard line. The adjusted projections acknowledge that it will likely take longer to achieve levels of employment growth.

The number of dwellings in industrial areas is projected to increase by 1,475 units in the UI zone, 33% less than the amount studied in Draft EIS Alternative 4. These would be industrysupportive housing units—either caretakers' quarters and makers studios, or having at least half of the homes restricted to be affordable to households with incomes common for jobs in industrial sectors. Criteria for the location and performance of any industry-supportive housing in the UI zone would be more limited than alternatives 3 and 4 in the Draft EIS and subject to a conditional use approval process. This housing would make up about 1.8% of new units planned for citywide over the time horizon of the One Seattle Comprehensive Plan major update.

Two new areas outside the MICs in west Ballard and Judkins Park would be converted to mixed use zoning allowing housing, in addition to the proposed mixed-use areas in Georgetown and South Park studied in Draft EIS alternatives. Overall, a higher total amount of housing production outside of MICs would result compared to Draft EIS alternatives—an additional 1,534 dwellings, 42% more than alternatives 3 and 4.

Most industrial jobs and total jobs (66%) are located in the SODO/Stadium Subarea and the Georgetown/South Park Subarea and these subareas would continue to have the greatest total growth. Relative to other alternatives, the Preferred Alternative places slightly more jobs in Ballard and Interbay subareas.

More nuanced specific development standards are proposed under the Preferred Alternative for the triangular area of Georgetown bounded by Airport Way, Corson Avenue S, and Carleton Avenue S. The standards integrate Georgetown priorities for historic preservation, antidisplacement, and arts spaces.

The collective change in industry supported housing including within industrial areas, areas removed from the MIC, and rezoned areas converted to mixed use zoning outside of the MIC would equal 3,009 households, about 8% less than Alternative 4. The combination of employment and population growth is lower than both alternatives 3 and 4.

1.5.10 Comparison of Alternatives

Exhibit 1.5-16 below summarizes the four<u>five</u> alternatives studied in this EIS. In summary, the alternatives are arranged with an increasing degree of land use change from 1 to 4, with Alternative 4 having the greatest degree of change <u>of the Draft EIS alternatives</u>. Higher number alternatives have larger geographic areas rezoned to the II or UI zone, and greater magnitudes of projected employment growth. The Preferred Alternative combines features of the Draft EIS alternatives and additional refinements to the proposed development standards to reduce potential impacts and respond to comments. The Preferred Alternative has an overall amount of employment growth similar to Alternative 2, and a pattern of zoning changes similar to Alternative 4. A legislative proposal will be developed once the EIS process is complete which will likely be similar to the Preferred Alternative but could also be a hybrid of the alternatives described below.

No Action Alternative	New Land Use Concepts	Alt 2—Future of Industry Limited	Alt 3—Future of Industry Targeted	Alt 4—Future of Industry Expanded	Preferred Alt— Future of Industry Balanced
Industrial General Zones: 90% of land area	 Maritime Manufacturing and Logistics (MML) Zone 	90% with stronger protections.	86% with stronger protections.	87% with stronger protections.	86% with stronger. protections.
Industrial Commercial Zones: 5% of land area	Industry and Innovation (II) Zone	7% of land area. Located up to approximately ¼ mile around transit stations and all land currently zoned industrial commercial.	7% of land area. Located approximately up to ½ mile around transit stations and all land currently zoned Industrial Commercial.	9% of land area. Located greater than ½ mile around transit stations and all land currently zoned Industrial Commercial. Includes land near potential Ballard ST3 station and the Stadium ST3 station.	8% of land area. Applies the II zone within approximately ½ mile around transit stations, and includes IC zoning that is retained outside of MICs.
Industrial Buffer Zone: 5% of land area	Urban Industrial (UI) Zone	3% of land area. Located generally in transition areas between MML or II zones and nonindustrial areas.	6% of all land area. Expanded transition area in Ballard.	4% of land area. Expanded transition area in Stadium district.	<u>5% of land area.</u>
Areas removed from MIC and placed in mixed-use zone		None.	Small nodes in Georgetown/ South Park to advance community goals_ (1,078 units).	Small nodes in Georgetown/South Park to advance community goals_ (1,078 units).	Small nodes in Georgetown/South Park to advance community goals. Similar area removed as in alternatives 3 and 4. Less dwellings due to nuanced development standards (686 units).

Exhibit 1.5-16 Summary of Land Use Concepts by Alternatives
No Action Alternative	New Land Use Concepts	Alt 2—Future of Industry Limited	Alt 3—Future of Industry Targeted	Alt 4—Future of Industry Expanded	<u>Preferred Alt—</u> Future of Industry <u>Balanced</u>
Areas outside of MIC in West Ballard and Judkins Park (currently zoned IG)		Applies MML in West Ballard and II in Judkins Park.	Applies MML in. West Ballard and II in Judkins Park.	Applies II in both West Ballard and Judkins Park.	Applies a mixed use (Neighborhood Commercial) zone resulting in 848 projected units.
Only new caretaker's quarters, artist housing, and existing non- conforming: approx. 413 units	Housing in Industrial Zones	No expanded allowances.	Expanded industry- supportive in UI zones: approx. 610 units.	Larger expansion of Industry-supportive in UI zones: approx. 2,195 units.	Expansion of Industry- supportive housing in. UI zones by conditional use only; approx. 1,475 units. 50% affordable at 90% AMI or below option.
Lodging Prohibited	Stadium Overlay	No change.	Allow lodging.	All lodging with larger size of use limits.	Same as Alternative 4.
Size of Use Limits	Non-Industrial uses.	Varies by zone. Expanded non-industrial ancillary uses_ (UI). Reduced stand-alone non- industrial size of use limits (MML). No limit in bonus space (II).	Varies by zone. Expanded non-industrial ancillary uses_ (UI). Reduced stand-alone non- industrial size of use limits (MML). No limit in bonus space (II).	Varies by zone. Expanded non- industrial ancillary uses <u>(UI)</u> . Reduced stand-alone non- industrial size of use limits <u>(MML)</u> . No limit in bonus <u>space (II)</u> . Expanded size of use limit for indoor recreational facilities.	Same as Alternative 4.
MIC Subarea Plans	Current Plans	Update MIC Subarea Plans per VISION 2050.	Update MIC Subarea Plans per VISION 2050.	Update MIC Subarea Plans per VISION 2050.	<u>Update MIC Subarea</u> Plans per VISION 2050.
Comprehensive Plan Policies	Current Policies	Amend Comprehensive Plan Policies to establish new land use framework, limit MIC boundary changes to Periodic Update, establish City's intent to work with State of Washington on a masterplan for the Armory and WOSCA Sites.	Amend Comprehensive Plan Policies to establish new land use framework, limit MIC boundary changes to Periodic Update, establish City's intent to work with State of Washington on a masterplan for the Armory and WOSCA Sites.	Amend Comprehensive Plan Policies to establish new land use framework, limit MIC boundary changes to Periodic Update, establish City's intent to work with State of Washington on a masterplan for the Armory and WOSCA Sites.	Amend Comprehensive Plan Policies to establish new land use framework, limit MIC boundary changes to Periodic Update, establish City's intent to work with State of Washington on a masterplan for the Armory and WOSCA Sites.

Sources: City of Seattle, 2022+; BERK, 2022+.

A comparison of zoned acres is listed below. In all alternatives, the majority of the study area would be dedicated for industrial and manufacturing uses (IG or MML). Some areas zoned for industrial and manufacturing uses today would be designated instead for transitional zoning (UI) or dense employment (II) under the Action Alternatives. See **Exhibit 1.5-17**.

Zoning Districts	Alt 1	Land Use Concept	Alt 2	Alt 3	Alt 4	<u>Pref.</u>
Industrial General (IG1/IG2)	6,035	Maritime, Manufacturing, and Logistics (MML)	6,251	5,968	6,035	<u>5,895</u>
Industrial Buffer (IB)	279	Urban Industrial (UI)	222	426	279	<u>376</u>
Industrial Commercial (IC)	600	Industry and Innovation (II)	463	516	600	<u>612*</u>
Mixed-Use Commercial	22			26	22	<u>53</u>
Total	6,936		6,936	6,936	6,936	6,936

Exhibit 1.5-17	Comparison of Alternatives by	y Land Use/Zoning Acres

*Includes some retention of IC outside the MIC. Sources: City of Seattle, 20224; BERK, 20224.

Exhibit 1.5-18 summarizes total projected employment growth in the study area for the base year and by alternative, with a breakout of industrial¹ and non-industrial employment. The No Action Alternative and all three of the Action Alternatives result in employment growth. Overall employment growth is strongest under alternatives 3 and 4, which would result in 58% and 60% employment growth from the base year of 2018 over the time horizon to 2044. This would be substantially more job growth in Seattle's MICs than has occurred in the last 20-year period due to the proposed changes. Total employment growth under the Preferred Alternative would be less than alternatives 3 and 4 and would be an increase of 36% from the base year. Employment projections are moderated under the Preferred Alternative to reflect more realistic conditions in demand for employment spaces post-COVID and timelines for completion of new Sound Transit light rail lines. The overall number of industrial jobs would grow in all of the alternatives—ranging from +11,900 under No Action to +28,800 under Alternative 4. The percentage of the jobs that are industrial however would decrease incrementally from 55% in the base year to 53% under Alternative 4<u>or the Preferred Alternative</u>. See **Exhibit 1.5-19**.

¹ Industrial employment estimated based on the 2019 share of industrial employment by sector based on the 2015 PSRC Industrial Lands Study NAICs-based definition of industrial activities. This uses classification of what counts as an industrial job consistent with Puget Sound Regional Council criteria, including jobs in Information Computer Technology (ICT). Projections show strong job growth in ICT under the Action Alternatives. Consistency with PSRC classifications is appropriate given the need to fit VISION 2050 and Regional Centers Framework. A more conservative classification of which jobs are industrial, especially in ICT would show a steeper decline in the % of industrial jobs under most studied alternatives.



Exhibit 1.5-18 Industrial and Non-Industrial Job Share

Note: This chart was updated to include the Preferred Alternative. Source<u>s</u>: City of Seattle, 202<u>2</u>1; BERK, 202<u>2</u>1.



Exhibit 1.5-19 Share of Industrial and Non-Industrial Jobs

Industrial Employment
Non-industrial Employment

Note: Regional manufacturing/ industrial centers must retain a minimum 50% industrial employment per PSRC.

Note: This chart was updated to include the Preferred Alternative. Sources: City of Seattle, 20221; BERK, 20221.

The total combined employment and population growth is illustrated on the graph below for each alternative. Considered in combination, the total jobs and population by alternative shows the highest total job and population growth under Alternative 4 and the lowest under Alternative 1. The Preferred Alternative has a total that is slightly more than Alternative 2 and less than Alternative 3. See **Exhibit 1.5-20**.





Sources: City of Seattle, 2022; BERK, 2022.

Exhibit 1.5-21 shows percentage of employment growth by subarea to display which subareas would have relatively greater employment growth over the base amount. The north subareas of Ballard, Interbay Dravus, and Interbay Smith Cove would have the highest employment growth on a percentage basis, most notably under alternatives 3 and 4 where employment growth is projected to increase by over 70% for each of these three northern areas. <u>Under the Preferred</u>. Alternative, employment growth in the Ballard and Interbay Dravus subareas would be between the amount for Alternative 2 and alternatives 3 and 4, and growth in the Interbay Smith Cove Subarea would be similar to Alternative 2. Employment growth in the SODO/Stadium and South Park/Georgetown subareas would be between the amounts of alternatives 1 and 2.

While the greatest percent change in jobs is in the northern BINMIC subareas, the number of new jobs is greater in the Greater Duwamish MIC southern subareas.



Exhibit 1.5-21 Percent Growth in Employment by Subarea

Note: This chart was updated to include the Preferred Alternative. Sources: CAI, 2021; City of Seattle, 20221; BERK, 20221.

1.5.11 Alternatives Considered & Not Carried Forward

Following scoping, the City made some adjustments to the <u>Draft EIS</u> alternatives (see **Appendix A** for the scoping report) such as considering the sizing of recreation uses in some zones. Other ideas were considered but not carried forward.

The City considered scoping comments requesting more extensive changes to MIC boundaries, or requests for zoning allowing residential or mixed-uses across the study area at particular sites, and considered an alternative that would have de-designated the BINMIC as a MIC. However, the city determined that these approaches would not be likely to advance towards the proposal's objectives and would not be in keeping with the intent of City decisionmakers and policymakers. Therefore, the City largely retained the focus of alternatives on industrial and maritime purposes.

- The EIS represents an implementation action of the recently completed Industrial and Maritime Strategy and the alternatives are heavily informed by the recommendations of that strategy, including adding no significant new housing in industrial areas, and rather focusing primarily on industrial uses consistent with regional and city plans.
- The proposal includes a policy change calling for collaborative master planning of the Armory site. The site is within the MIC, and the proposal is that updated MIC policies and industrial zone designations will apply to the site. Should the State and partners wish to

pursue non-industrial future uses, that would have to be determined in the master plan in partnership with the City and other entities.

The EIS does consider a policy to allow for individual MIC boundary adjustments during the periodic review or during the annual amendment process.

The City considered Draft EIS comments in developing the Preferred Alternative, as described in **Section 1.5.9**. Some features of the Preferred Alternative that were added directly in response to comments are noted in the description of the Preferred Alternative. Additional mitigation measures in response to comments are added in the Final EIS. Additional details about proposed development standards are added in the Final EIS in response to comments.

1.6 Key Issues & Options

The key issues facing decision makers are summarized below:

- Adjustments to land use regulation that will affect future industrial job growth, including the amount of growth and mix of job types.
- The extent of industry-supportive housing—such as caretakers' quarters and maker studios—and the best location for such housing.
- Revisions to the MIC boundary in focused areas of Georgetown and South Park.
- Revisions or retention of zoning on industrially zoned land outside the MICs.
- Level of investment in transitions between the MICs and adjacent residential neighborhoods or urban villages.
- Level of investment within the MICs to address equity and environmental justice.

1.7 Summary of Impacts & Mitigation Measures

1.7.1 Soils/Geology

How did we analyze Soils/Geology?

We conducted a desktop analysis of existing information sources on soils and geologic conditions and evaluated potential impacts of the various alternatives. Geologists used best professional judgement to determine the impacts on soils and geology that would occur from each alternative within the study area.

What impacts did we identify?

The study area is located within the Puget Sound Region, an area susceptible to moderately high

What are geologic hazards?

Geologically hazardous areas include areas susceptible to erosion, sliding, earthquake, or other geological events. (WAC 365-190-120(1)) In order to promote safe, stable, and compatible development, Seattle regulates liquefaction-prone areas, landslide-prone areas, peat settlement-prone areas, seismic hazards areas, and volcanic hazard areas. Landslide areas include steep slope erosion hazard areas. (SMC 25.09.012)

seismic activity. During a seismic event, the study area might be subjected to high-level ground motions and areas with steep slopes might experience seismic slope stability problems.

Portions of the Ballard and Interbay Dravus subareas, and all of the Interbay Smith Cove, SODO/Stadium, and Georgetown/South Park subareas are susceptible to liquefaction. During an earthquake, vertical and lateral displacements of structures, embankments, and paved areas might occur due to seismic liquefaction hazard.

A peat settlement-prone area in the southwest portion of the Georgetown/South Park Subarea could limit the possibility of development and maintenance of existing structures with any of the alternatives. In this area, compressible soils might need to be excavated and replaced, or planned structures, embankments, and pathways might need to be supported on deep foundations. All alternatives would allow development that could disturb soils.

Development on or adjacent to any of the five historical landfills located within the study areas would require special planning and design. This could include assuring the integrity of any existing landfill cap, installing methane barriers or appropriate ventilation and designing structures to account for poor or unpredictable soil characteristics that could cause settling, preventing water from entering the historic landfills (capping with an engineered or bentonite cap barrier), and/or managing any leachate as water percolates through the historical landfill areas.

What is different between the alternatives?

Under Alternative 1 No Action, humans and animals could potentially feel the greatest impacts from geologic hazards in all subareas because fewer aging buildings and infrastructure

would be upgraded to modern building codes to withstand geologic conditions including seismic events compared to Action Alternatives.

Alternative 2 would rezone about 10% of the MICs to an UI or II zone, increasing the likelihood that development there would upgrade structures to modern building codes, resulting in less potential damage from geologic conditions or seismic events.

Alternatives 3 and 4 would rezone greater portions of the MICs (14% and 13%, respectively) to the II or UI zones. <u>The Preferred Alternative would have</u> <u>about 14% in the new II or UI zones, or retained in</u> <u>the IC zone outside of MICs; it would rezone the</u> <u>most land to Seattle Mixed.</u> This <u>level of change</u> <u>under these Action Alternatives would result in the</u>



Duwamish River

most development and the most benefit from structures built to modern building codes and least potential damage from geologic conditions or seismic events. <u>In addition to structures, all</u> water, wastewater, transportation, and other infrastructure associated with new development and redevelopment would be carefully designed with input from site-specific geotechnical investigations to lessen and withstand the effects of earthquakes and liquefaction.

What are some solutions or mitigation for impacts?

Although the proposal would allow development at sites in areas prone to landslides, liquefaction, settlement, or similar geologic hazards, modern building codes <u>and conditions of</u> <u>approval from site-specific geotechnical investigations</u> mitigate the risk of injury or economic losses. <u>Tribal notification prior to commencing site-specific subsurface investigations of soils</u> <u>should be required</u>. Erosion control measures per suggested best management practices (BMPs) would be prescribed in Construction Stormwater Pollution Prevention Plans prepared for each development project. Development on or adjacent to landfills within the study areas would include special controls and design as needed to mitigate for methane gas or account for poor or unpredictable soil characteristics that could cause settling and manage any leachate.

With mitigation, what is the ultimate outcome?

The Action Alternatives would generally have positive long-term benefits. The greatest benefits would be associated with alternatives 3 and 4 <u>followed by the Preferred Alternative</u> because they would result in the most sites developed to modern building codes.

Development in the study area, as with most locations in Central Puget Sound, would expose population and structures to geologic hazards, and would disturb soils. These impacts can be mitigated to a less than significant level by designing development to the City's adopted construction codes and applying any site-specific conditions (e.g., methane mitigation systems for buildings built near historic landfills) required by the City during permit review.

1.7.2 Air Quality & GHG

How did we analyze Air Quality & GHG?

Eight sites within the BINMIC and Greater Duwamish MIC were monitored directly to provide site-specific baseline data on ambient air quality conditions for this EIS. Criteria pollutant emissions and greenhouse gas (GHG) emissions were estimated under the Alternatives for future industrial, non-industrial, and housing development, changes in vehicle miles traveled by residents and employees, natural gas usage in buildings, maritime activities, and solid waste generation. Estimated increases in vehicle miles travelled (VMT) were predicted in the transportation analysis (**Section 3.10 Transportation**) and based on emission factors reflecting future improvements to the vehicle fleet using the AFLEET tool (2020 version) and data from the EPA MOVES2014b model.

Evaluating Air Quality and Greenhouse Gas (GHG)

The air quality evaluation considers air quality standards and conditions, with a focus on carbon monoxide (CO), particulate matter (PM) emissions, ozone precursors, and Toxic Air Pollutants (TAPs). The evaluation considers potential sensitive populations in and near the industrial and maritime areas of Seattle.

At a planning level the analysis indicates increases in greenhouse gases (GHGs) in comparison to local or regional goals or targets for GHG reductions and identifies mitigation to reduce impacts.

The growth in square footage and number of households was used to forecast 2040 GHG emissions using the City of Seattle's Energy Benchmarking data, and CO2 emission coefficients from the U.S. Energy Information Administration (EIA) and the EPA. These emissions were then adjusted to account for use of natural gas only, as electricity supplied by Seattle City Light is carbon neutral. The increase in residents and employees under each alternative was used to estimate emissions from the increase in solid waste generation using emission factors from the EPA's WARM model and the most recent (2018) waste generation rates from Seattle Public Utilities.

What impacts did we identify?

The analysis found that ambient air concentrations of monitored pollutants in the study area met the national ambient air quality standards under existing conditions, when excluding wildfire smoke. Air pollutants related to land uses changes, transportation, building uses, and maritime activities would all likely decrease in the future compared to existing conditions. This is due to the combination of existing requirements for industrial operating permits from Puget Sound Clean Air Agency (PSCAA), and ongoing requirements for improvements in vehicle emissions control, fuel economy, and technology improvements, and overall changes in fleet and fuel mix toward electrification and cleaner fuels, respectively. The Action Alternatives would be slightly higher in criteria air emissions than No Action due to increases in jobs and residents anticipated under each.

GHG emissions would all likely decrease in the future compared to existing conditions; the Action Alternatives would be slightly higher in GHG emissions than No Action due to increases in jobs and residents anticipated under each. These emissions would combine with emissions across the city, state, country, and planet to cumulatively contribute to global climate change.

West Seattle Bridge



Transportation systems contribute to climate change primarily through the emissions of carbon dioxide (CO2) and nitrous oxide (N2O) from gasoline and diesel fuels used to operate passenger cars, trucks, buses, and construction equipment. Land use changes contribute to climate change through construction and operational use of natural gas and waste production.

The proposal and alternatives would support more efficient growth patterns, consistent with regional planning as well as the long-term planning goals of the City's Comprehensive Plan, 2013 CAP, and 2018 Climate Action Strategy which are expected to assist in controlling GHG emissions (and which would have a similar effect on criteria air pollutants). The alternatives would help Seattle achieve its goals for accommodating industrial and maritime growth in areas that are well served by transit and within walking distance to a broad range of services and employment opportunities. However, because the proposal and alternatives would result in a net increase in GHG emissions generated in MICs compared to No Action, mitigation measures are warranted to maintain consistency with the long-term planning goals.

What is different between the alternatives?

The Action Alternatives would reduce air pollutant and GHG emissions below current conditions, while each Action Alternative adds additional emissions compared to No Action though not significantly. The relative difference in the magnitude of these increases is directly attributable to the level of industrial and non-industrial growth, housing growth, and vehicle miles traveled. For example, alternatives 3 and 4 would provide more industrial and non-industrial space and housing units in the study area, and hence accommodate more employees and people. Therefore, the operational criteria pollutant and GHG emissions resulting from those alternatives would be incrementally greater than those of Alternative 1 No Action and Alternative 2. The Preferred Alternative has employment space similar to Alternative 2 and housing less than Alternative 4. It would have GHG similar to Alternative 2 altogether.

Not considered in the quantification of GHG emissions is the fact that if growth accommodated in the proposal and alternatives were to be developed in other peripheral areas of the city or region with fewer transit options, overall transportation related GHG emissions would likely be far greater.

What are some solutions or mitigation for impacts?

Future development under the alternatives would be implemented while benefitting from ongoing improvements in vehicle emissions control, fuel economy, and technology improvements, and likely, enhancements to the Seattle Energy Code and updated actions under the 2013 Seattle CAP and 2018 Strategy. These codes and policies regulate and guide the energy-use features of new and remodeled buildings, including requirements with respect to building envelopes for roofs, walls, and windows; heating, ventilation, and air conditioning efficiency mandates; and water heating equipment efficiency. Other mitigation measures related to waste diversion, green building standards, and building demolition waste reduction are recommended to ensure consistency with Clean Air Act standards, PSCAA requirements, Washington's GHG emissions reduction policies, and the City's Comprehensive Plan and 2013 CAP.

To further mitigate the impact of emissions from trucks, the City, Port of Seattle, and partners could adopt regulations for the study area that support the placement of infrastructure for charging of electric vehicles (including commercial and industrial vehicles) and explore the creation of a city-owned electrical vehicle charging facilities in intended for drayage trucks infrastructure. Independent of mitigation for this Plan, the Port of Seattle is participating in the Northwest Ports Clean Air Strategy with a goal to get to zero emissions by 2050. This effort will contribute significant reductions in existing air pollutant and GHG emissions from Port operations by focusing on implementing shore power at Port container and cruise terminals, transitioning nonroad cargo handling equipment fleets to electric and/or hydrogen, and working to help the drayage industry transition to zero emission technologies. To further mitigate the impact of emissions from marine vessels, the City, Port of Seattle, and private partners could accelerate the extension of shore power to terminals and docks throughout the Seattle waterfront where no current plans exist.

Potential for exposure of existing and new employees, residents, and visitors to potential air emissions in areas around arterials, along industrial buffers, and near port operations should be considered in future planning. Policy measures could include separating residences and other sensitive land uses (i.e., schools, day care) be separated from freeways, railways, and port facilities, and new MML, II, and UI zones by a buffer area (e.g., 500 feet+), include enhanced air filtering and circulation, add landscaping and tree canopy, etc.

The City and partner agencies could improve coordination and improve the user experience for community members registering complaints or requesting information about enforcement related to emissions, noise, or contamination from sites or businesses.

With mitigation, what is the ultimate outcome?

With identified mitigation, the proposal and alternatives would be consistent with air pollution and GHG reduction and climate change planning in the City of Seattle, reducing the severity of

the identified cumulative impact. While the residual impact of all alternatives would still be a net increase in GHG emissions generated from growth and development in the MICs, the regional benefit of capturing development that might otherwise occur in other areas of the city or region would serve to offset these impacts. No significant unavoidable adverse impacts related to air quality and greenhouse gas emissions are anticipated.

1.7.3 Water Resources

How did we analyze Water Resources?

We conducted a desktop analysis of existing information sources to characterize existing surface water, groundwater, and sea level rise conditions and analyzed impacts for all alternatives and impacts for each subarea. Mitigation measures were determined based upon city, state and federal regulations, codes, plans and policies. Water resources scientists used best professional judgement to determine how each alternative would affect water resources.

What impacts did we identify?



Duwamish River

Short-term impacts could result from redevelopment including discharge of sediment or spills during construction. These construction projects would need to comply with the Seattle Stormwater Code, which requires temporary erosion and sediment controls.

Longer-term impacts may result from increased stormwater contamination from metals, organics and other pollutants related to industrial activities and traffic. However, higher levels of redevelopment would result in more stormwater control, such as onsite stormwater management, flow control, and water quality treatment, relative to existing conditions. Therefore, all Alternatives are expected to improve water resources.

Low lying areas adjacent to tidally-influenced water bodies (Puget Sound, Elliott Bay, the Duwamish River, and the mouths of Longfellow Creek and Puget Creek) have the potential to be affected by sea level rise. All alternatives may increase vulnerability to sea level rise more than No Action by bringing more people into vulnerable areas. Redevelopment that complies with SMP and frequently flooded areas requirements, and where adaptation measures are implemented, may decrease vulnerability to sea level rise relative to existing conditions.

What is different between the alternatives?

The alternatives differ in the amount of area that would be subject to stormwater mitigation during redevelopment. Alternatives with greater redevelopment, whether it is expansion of an existing industrial site or additional dense employment, would result in greater improvements to water quality and/or increased flow control.

Alternatives 3 and 4 <u>and the Preferred Alternative</u> create more housing than alternatives 1 and 2. The housing is concentrated in the Ballard and SODO/Stadium subareas<u>and under the Preferred</u><u>Alternative the West Ballard and Judkins Park areas outside the MICs</u>. The increased housing will bring more permanent residents. Impacts include increased pets and pet waste with the potential to contribute fecal coliform bacteria to adjacent surface waters. New residents in caretakers' quarters and makers studios, as well as areas removed from the MIC in Georgetown/South Park for mixed-use residential would also be exposed to potential sea level rise.

What are some solutions or mitigation for impacts?

As redevelopment occurs, most projects would be required to implement onsite stormwater management, water quality treatment, and flow control, which would improve stormwater management relative to existing conditions. Compliance with these regulations is anticipated to result in a net benefit to water resources under all Alternatives, with the greatest benefits occurring for Alternatives with the most redevelopment.

During construction, stormwater control BMPs would prevent sediment and contaminants from coming in contact with drainage water or being discharged to the drainage system, public combined sewer, or directly into receiving waters.

Surface and groundwater quality at industrial and business sites are protected through ongoing inspection programs, which also applies to new development. Industrial permits issued and managed by the Washington State Department of Ecology and held by individual properties are inspected and required to implement source control BMPs.

An increased emphasis on pet waste management through education and outreach and increased pet waste disposal stations should be implemented in areas surrounding housing developments to prevent impacts on water quality.

Under all Alternatives, proposed development in areas that are near the shoreline or in known flooding areas would be required to comply with critical areas regulations for frequently flooded areas, which is regulated through the City's Environmentally Critical Areas (ECA) Code and the requirements of the Shoreline Master Program (SMP; Seattle Municipal Code 23.60A). Compliance with these codes would likely reduce vulnerability of those developments to sea level rise impacts relative to existing conditions. Additional reduction in vulnerability will be achieved upon implementation of planning and programmatic adaptation strategies specified in the City of Seattle 2017 Preparing for Climate Change including conducting a detailed coastal study of the Duwamish River to better assess the flood risk and identify mitigation strategies The City should also evaluate vulnerability of underground infrastructure to higher groundwater levels.

With mitigation, what is the ultimate outcome?

Redevelopment of previously developed areas would lead to improvement of stormwater management relative to existing conditions. If all minimization and mitigation measures are implemented, no significant unavoidable adverse impacts to water resources are anticipated.

Shorelines in the Study Area



1.7.4 Plants & Animals

How did we analyze Plants & Animals?

We conducted a desktop analysis of existing information sources to characterize plants and animals in the study area and analyze potential impacts of the various alternatives. We looked at city, state and federal GIS data, aerial photos, studies and reports on environmental conditions, and peer-reviewed literature. Biologists used best professional judgement to determine how each alternative would affect habitats and species within the study area.

What impacts did we identify?

Short-term impacts could occur during construction that stems from rezoning that encourages redevelopment. Noise and disturbance from construction activities could disturb wildlife nearby, causing minor disruptions of normal behaviors. Species in the study area are already adapted to high levels of human disturbance and are unlikely to be adversely affected by additional construction.

Stormwater runoff from active construction sites can mobilize sediments that have the potential to degrade water quality in receiving water bodies. Best management practices (BMPs) implemented during construction, such as erosion control, would minimize potential impacts.

Impacts to special status habitats, such as wetland and riparian areas, are expected to be minimal, as these habitats are protected, and mitigation measures would be implemented to offset those impacts. Conversion of undeveloped sites to residences or other buildings could reduce wildlife habitat. Because the study area is highly urbanized, impacts to unprotected habitat types (such as landscaped areas and undeveloped parcels) would be minor, and existing habitat is already degraded. Redevelopment of developed parcels could increase the creation of landscaped areas and other green spaces, resulting in a slight increase of habitat for urban-adapted species.

Stormwater runoff from developed land contains various pollutants that have the potential to degrade aquatic habitat and adversely affect aquatic species. Increasing residential or other development in the study area could increase those pollutants. Construction of green spaces, as well as redevelopment of developed parcels, provides opportunities to implement stormwater treatment where none currently exists, which would improve water quality in the study area.

What is different between the alternatives?

The alternatives differ in the amount of area that would be rezoned as well as the number of residential units that would be constructed. The No Action Alternative would not change existing zoning and would have fewer impacts to terrestrial habitat provided by existing landscaped and undeveloped parcels. However, there would also be less habitat created by increasing landscaping and green spaces. Less development would reduce the potential for increased pollutant loading to receiving water bodies, but also would not present new opportunities for providing increased stormwater treatment that would improve water quality.

Alternative 2 would rezone a portion (10%) of the MICs to allow denser development in the UI or II zones, increasing some development as well as landscaped and green areas. New development could result in minor increases to degraded wildlife habitat provided by undeveloped parcels, but this impact would likely be offset by new landscaping and green spaces. Stormwater infrastructure and treatment BMPs could also be implemented during redevelopment, potentially improving water quality in the study area. There would be less residential development than under alternatives 3 and 4, reducing pollution stemming from that type of development.

Alternatives 3 and 4 would also rezone a portion of the MICs to allow denser development in the UI or II zones (14% and 13%, respectively), which could increase the amount of landscaped and green spaces within the MICs, potentially increasing minor amounts of wildlife habitat and providing opportunities for reducing stormwater runoff and improving stormwater quality. The Preferred Alternative would rezone land to UI and II or retain IC in an amount totaling 14% of the study area. It would also provide for the most housing opportunities outside the MIC. It would have a potential for redevelopment in the range of the alternatives under study.

What are some solutions or mitigation for impacts?

Mitigation would be provided by incorporating green spaces into the II and UI zones for all of the Action Alternatives. Impacts would be avoided and minimized per existing city, state, and federal regulations, and compensatory mitigation would be provided for all protected areas. Water quality treatment would be provided for redeveloped areas.

With mitigation, what is the ultimate outcome?

If all minimization and mitigation measures are implemented, no significant unavoidable adverse impacts to plants and animals are anticipated. The study area is already highly

urbanized and existing habitat is degraded. Terrestrial species are tolerant of disturbance and are not likely to be adversely affected by additional development.

Redevelopment of previously developed areas provides opportunities to create additional landscaped and green spaces that provide wildlife habitat, as well as reduce urban runoff and pollutant loading to aquatic habitat, potentially contributing to improved water quality in the study area. Improved water quality would benefit aquatic species habitat.

1.7.5 Contamination

How did we analyze Contamination?

We conducted a desktop analysis of existing information sources to identify sites with confirmed or suspected contamination in soil, sediment, and groundwater, sites where hazardous materials are used or stored, and sites with historical landfills. Environmental scientists used best professional judgement to determine the impacts on human health and the environment that would occur from each alternative within the study area.

What impacts did we identify?

Development under any of the alternatives may encounter hazardous materials such as contaminated soil, groundwater, or surface water. The greatest potential for impacts associated with contamination would occur during construction when sites are disturbed. Construction activities could release hazardous materials due to ground disturbing, dewatering, and demolition activities. Development within the study area, especially where known hazardous material sites are located, would address the removal of hazardous materials, which could include contaminated soils, groundwater, surface water, and, in older structures, the potential for lead-based paints and asbestos-containing materials (ACMs).



A semi truck accident and fire resulted in a spill to the Duwamish Waterway. SPU deployed a water and land crew to mitigate the spill.

Contaminated soils excavated during construction activities would require special handling, transport, storage, and off-site disposal. Depending on groundwater depth and the type of hazardous materials, it is possible that contaminants from historic spills or releases may have infiltrated into groundwater becoming leachate and migrated, requiring additional cleanup. Short-term exposures to hazardous materials could occur during cleanup actions at contaminated sites. Because documented contamination requiring cleanup would be removed or contained prior to new development, it is assumed there would be no significant health and safety impacts on those living, working, or visiting the area, or impacts on the intended uses of properties within the study area.

As growth occurs in the study area, there is potential for hazardous material spills associated with petroleum products to increase as traffic and the potential for accidents increases. With growth there is also the potential for increased risk of spills from industrial activities, industrial processes, or use of industrial chemicals.

What is different between the alternatives?

The alternatives differ in the amount of area that would be rezoned as well as the number of residential units that would be constructed. The No Action Alternative would not change existing zoning and would have fewer impacts on contaminated sites that are redeveloped or cleaned up.

Alternative 2 would rezone a portion (10%) of the MICs to allow denser development in the UI or II zones. Increased development would increase the short-term risk of exposure to contaminants as sites are cleaned up but result in a long-term benefit of lower concentrations of chemicals after sites are cleaned up. With the increases in industrial jobs and industrial space added there would be an increased risk of chemical exposures and industrial spills related to industrial processes.

Alternatives 3 and 4 would also rezone a portion of the MICs to allow denser development in the UI or II zones (14% and 13%, respectively). <u>Similarly, the Preferred Alternative would rezone</u> <u>some areas as UI or II or retain IC-zoned land in an amount totaling 14% of the study area;</u> <u>it would allow more mixed uses outside the MIC.</u> This would result in the most development and short-term risk of exposure to contaminants as sites are cleaned. However, under these alternatives, there would be the most long-term benefits of lower concentrations of chemicals in soils, groundwater, and surface water after sites are cleaned up. With the most industrial jobs added and industrial space created there would be an increased risk of chemical exposures and industrial spills related to industrial processes.

What are some solutions or mitigation for impacts?

All site development projects would be required to comply with applicable federal, state, and local regulations. Existing regulations establish standards for site characterization, cleanup of hazardous materials, and disposal of hazardous waste, as well as mitigation measures for development on or adjacent to historic landfills. Development of known or suspected contaminated sites would require a Phase I Environmental Site Assessment and potentially a Phase II Environmental Site Assessment (with soil, sediment, and/or groundwater sampling) prior to construction-related activities, including demolition. Prior to renovation or demolition of structures, hazardous building material surveys (HBMS) would be conducted, and abatement of lead-based paints and asbestos, if present, would be required by the Puget Sound Clean Air Agency (PSCAA) and other agencies and laws. To the extent possible, the amount of contamination at a site with known contamination would be verified prior to construction, to minimize exposure to hazardous materials.

In Washington State, strict cleanup standards to ensure human health and the environment are not compromised, and stringent regulations ensure that non-hazardous and hazardous

solid wastes are properly managed from cradle to grave at industrial sites and other properties to prevent impacts to human health and the environment. Compliance with the regulations results in low levels of contamination after site cleanup and redevelopment.

Hazardous materials are regulated through the International Building Code and the International Fire Code and new development would need to meet requirements prior to permits being issued for construction. Development and implementation of Construction Stormwater Pollution Prevention Plans would be required by the City to minimize the potential for release of hazardous materials to soil, groundwater, or surface water during construction.

During construction, contingency plans would be required to help manage hazardous substances, protect worker health and safety, prevent spills, and prevent stormwater pollution.

The City and partner agencies could improve coordination and improve the user experience for community members registering complaints or requesting information about enforcement related to contamination from sites or businesses.

With mitigation, what is the ultimate outcome?

The risk of release of contaminants or of hazardous chemicals being used or causing conditions that result in health or safety impacts or impede future development is considered significant for all alternatives but avoidable with mitigation.

1.7.6 Noise

How did we analyze Noise?

A desktop survey using aerial photography, Google Earth, ArcGIS, and existing and proposed City of Seattle Comprehensive Plan land use designations and zoning was used to determine locations of noise sensitive land uses in the Study Area. Eight sites within the BINMIC and Greater Duwamish MIC were monitored directly to provide site-specific baseline data on existing noise levels for the analysis. Noise levels were modeled using the Federal Highway Administration (FHWA)/Federal Transit Administration (FTA) Noise Impact Assessment spreadsheet model under the alternatives for future increased traffic volumes at roadways adjacent to monitoring sites. After describing existing noise levels and the methods used for the impact analysis, each alternative was analyzed to determine how the resulting noise levels compared to the City's <u>Noise Control regulations (SMC 25.08) and the effects on noise sensitive land uses within the</u> Study Area. This includes primarily increased noise levels associated with increases in traffic, but also addresses potential noise associate<u>d</u> with construction, and stationary industrial activities.

What impacts did we identify?

Existing data show that ambient noise levels in maritime and industrial areas of the city can be higher than other developed areas of the city. Noise monitoring of existing conditions within



two of the subareas, Georgetown and SODO/Lander, was found to exceed a 24-hour day night average of 65dba—a Department of Housing and Urban Development standard for acceptable exterior noise levels for residential areas. Under all alternatives there would be temporary impacts in noise during construction. Construction activities would be temporary in nature, and it is anticipated the majority of the activities would occur during daytime working hours.

Future industrial and non-industrial developments could use stationary mechanical equipment that, unless properly designed or controlled, could exceed the allowable City noise ordinance limits intermittently. Depending on the location, this could impact new residential uses within and adjacent to some areas of the MICs.

Under all alternatives, traffic volumes on roads, including truck traffic, are expected to continue to be a primary source of noise in and near the MICs <u>and</u> are expected to increase due to expected development and associated population increase. These increased volumes would lead to very slight increases in roadway noise in some areas, but insufficient (less than 3 dBA) to generate noticeable increases in roadway noise compared to the existing condition or No Action.

What is different between the alternatives?

Traffic could increase roadway noise very slightly. The existing noise levels range from 51 to 69 dBA, and the increases over existing conditions in the alternatives range from 0 to 2 dBA, with most of the increasing bye 0 dBA. In specific areas, Alternative 2 and the Preferred Alternative haves greater impacts than Alternative 1 No Action, and alternatives 3 and 4 have greater impacts than alternatives 1 and 2. See **Exhibit 1.7-1**. However, an increase of 1-2 dBA is not perceptible to the average person and a 3 dBA increase is barely perceptible. Thus, impacts under any alternative would not be significant.

	Existing	Increase in dBA—PM Peak Hour Volume									
	24-Hour	2042 No	o Action	2044	Alt. 2	2044	Alt. 3	2044	Alt. 4	<u>2044 Pı</u>	r <mark>ef. Alt.</mark>
Geographic Area	Ldn	NB/EB	SB/WB	NB/EB	SB/WB	NB/EB	SB/WB	NB/EB	SB/WB	<u>NB/EB</u>	<u>SB/WB</u>
Ballard	62.5	0.0	0.0	0.0	0.0	0.0	0.0	1.0	1.0	<u>0.0</u>	<u>0.0</u>
Interbay/Dravus	59	0.0	0.0	1.0	0.0	2.0	0.0	2.0	0.0	<u>1.0</u>	<u>0.0</u>
Interbay/Armory	59	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	<u>0.0</u>	<u>0.0</u>
Stadium	69	-0.2	0.0	0.0	0.0	0.0	0.0	0.0	0.0	<u>0.0</u>	<u>0.0</u>
Georgetown	68.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	<u>0.0</u>	<u>0.0</u>
South Park 1	60.5	1.0	0.0	0.0	0.0	0.0	1.0	0.0	0.0	<u>0.0</u>	<u>0.0</u>
SODO/Lander	67.8	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	<u>0.0</u>	<u>0.0</u>
South Park 2	59.5	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	<u>0.0</u>	<u>0.0</u>

Exhibit 1.7-1 Increase in dBA Over Existing Conditions, All Alternatives

Sources: Fehr and Peers, 2022+; Herrera, 2022+.

What are some solutions or mitigation for impacts?

Current regulations and commitments include:

- SMC Chapter 25.08.410 provides specific noise controls and allowable community noise limits (expressed as dBA levels) for EDNA receivers.
- SMC Chapter 25.08.490 includes nuisance provisions.
- SMC Chapter 25.08.425 limits hours of construction to daytime periods.
- The SEPA review process allows the City to consider potential noise impacts. A noise impact study may be required to forecast future noise levels for some developments and identify mitigation measures.
- WSDOT Traffic Noise Abatement Protocol sets requirements to evaluate and abate traffic noise impacts, for roadway improvement projects that use state or federal funding. Construction noise measures include requiring a noise control plan where the contractor will be required to comply with all federal, state, and local regulations relating to construction noise.

The City could require each industrial facility proposed for construction within 500 feet of residentially zoned parcels to conduct a project-specific noise impact assessment to demonstrate compliance with the community noise limits set by the City's noise ordinance (SMC Chapter 25.08).

Zoning land use criteria or boundaries could be established, while meeting other planning goals, to limit the proximity of new residential development to known or anticipated sources of high noise levels.

Under alternatives 3 and 4<u>and the Preferred Alternative</u>, which would allow the development of new residential, the City could impose greater noise reduction standards in residential buildings (e.g., acoustically rated windows and doors, wall and roof insulation, dampers on vents, etc.) where exterior noise levels greater than 65 dBA are likely to occur or where other uses occupying the same structure would likely contribute to excessive noise levels (above 45 dBA) within residences.

The City and partner agencies could improve coordination and improve the user experience for community members registering complaints or requesting information about enforcement related to noise from sites or businesses.

Noise from tire-pavement interactions is the dominant contributor to roadway noise. A longterm mitigation program to reduce noise in noise-sensitive areas within the study area would be to install noise reducing pavement on major arterials and roadways that experience relatively high traffic volumes and speeds.

With mitigation, what is the ultimate outcome?

The potential increases in noise are not expected to increase 10 dBA over existing conditions nor would they be the cause of a failure to comply with SMC maximum allowable sound levels for receivers and based upon the modeling would increase by no more than 3 dBA. Considering the level of noise change as well as mitigation measures, no significant, unavoidable adverse impacts are anticipated.

1.7.7 Light & Glare

How did we analyze Light & Glare?

The EIS documents light and glare patterns in the study area, including a summary of existing development patterns and major sources of light emissions. The analysis uses digital topographic data maintained by the City of Seattle to calculate a potential viewshed area for the existing zoning pattern and each of the proposed alternatives to assess visibility of future development, based on allowed maximum building heights. The EIS also identifies sensitive locations and resources within these viewsheds that could potentially be impacted by additional light and glare emissions associated with future development, such as residential neighborhoods, parks and trails, or scenic views.

Nighttime Lights at the Port



What impacts did we identify?

Urban development, including development of a non-industrial nature, generates light and glare emissions associated with occupation and operation, and the precise nature of these emissions and impacts vary based on building design, location, and shielding/screening measures employed, but future growth under any of the alternatives will generate at least some increase in light and glare. These increased light emissions are most likely to affect residential areas north of the BINMIC, residential areas in Beacon Hill (east of the Greater Duwamish MIC), and the South Park neighborhood, which is adjacent to the southern end of the Greater Duwamish MIC. Lesser impacts may occur on the south slope of Queen Anne, southeast Magnolia, and eastern portions of West Seattle.

Additionally, some of these areas may experience increased visibility of development in industrial areas due to taller building heights under the Action Alternatives. However, the development typologies employed in these locations would typically employ less extensive outdoor lighting than existing industrial uses, which may result in reduced light and glare emissions at these locations.

What is different between the alternatives?

The No Action Alternative would preserve existing zoning and development regulations, resulting in future industrial development patterns similar to existing conditions. Future light

and glare impacts under the No Action Alternative would effectively be an intensification of existing conditions as additional development occurs in the study area.

The Action Alternatives create new land use concepts with new development standards:

- The MML land use concept is focused on traditional industrial and manufacturing uses, as well as shipping, logistics, and port facilities. Similar to the IG zone, major sources of light and glare would include outdoor illumination at storage yards and cargo staging areas. Manufacturing facilities that use exterior lights for operations and safety during nighttime hours would also be sources of light and glare. The MML land use concept would include zoning requirements for streetscape improvements, but on-site vegetation is anticipated to be sparse due to the intensive nature of development and the operational needs of shipping and logistics facilities, which are the primary anticipated uses. This lack of on-site vegetation would result in minimal screening of light sources.
- The II land use concept promotes higher-density industrial uses, including mixed-use development. The II land use concept is focused on a mix of uses that incorporates contemporary industrial methods and creates opportunities for combining light industrial and technology-oriented uses with associated office space. Compared to existing industrial areas, the II concept would exhibit taller building heights (up to 160 feet, including bonuses) and greater development density with fewer outdoor storage and/or staging areas. The integration of transit and bicycle/pedestrian connections would also result in fewer large parking areas. Without extensive outdoor areas requiring night-time lighting, exterior building illumination would be less intense, though taller allowable building heights could make buildings visible from farther away.
- The UI land use concept focuses on a mix of smaller-scale industrial uses (such as fabrication shops, artist and maker spaces, and light industry) and limited non-industrial uses, such as retail, offices, or industry-supportive housing. These areas would also include bicycle and pedestrian transportation facilities. Development in UI areas is anticipated to generate relatively lower light emissions compared to existing industrial typologies and the proposed MML and II land use concepts, due to the smaller scale of development and a greater emphasis on vegetation and green space. The UI land use concept would allow building heights up to 75 feet, which would represent a height increase in some industrial areas. Though less pronounced than potential height increases under the II land use concept, taller building heights may result in development being visible from farther away than current conditions.

By subarea, the anticipated light and glare impacts are noted for the Action Alternatives:

Ballard: Compared to other Action Alternatives, Alternative 2 locates greater MML along the waterfront and near Ballard Avenue Landmark District. MML zoning standards would allow larger buildings and less vegetation similar to the Alternative 1 IG zone. The increase in MML zoning along the waterfront could increase the potential for light emissions there. Alternatives 3 and 4 would have more UI zoned acres than Alternative 2. The UI zone standards would allow smaller footprints and greater screening through landscaping and design concepts and less impacts than MML type zoning. Compared to the IB zone under Alternative 1, the UI zone allows for improved transitions to residential areas such as in the northeast Ballard and Gas Works Park area. The II zone would have taller buildings more visible to surrounding areas.

Alternatives 2, 3, and 4 would result in more II zoned areas, particularly Alternative 4. <u>The</u> <u>Preferred Alternative would retain Industrial Commercial zoning in areas outside the MIC</u> <u>and implement mixed use zoning along the northwestern edge of the subarea, resulting in</u> <u>impacts similar to the No Action Alternative in these locations.</u>

- Interbay Dravus: Alternatives 2, 3, and 4 provide for MML along the Ballard Locks similar to the IG zone under Alternative 1. There could be light and glare impacts without mitigation. Under a<u>A</u>lternatives 2, 3, and 4<u>and the Preferred Alternative</u> have a small area of UI which could reduce light and glare emissions and better address transitions to residential areas on northwest Queen Anne. Compared to the other action alternatives, the Preferred Alternative would slightly increase building heights in the UI zone, increasing visibility from surrounding areas.
- Interbay Smith Cove: Alternatives 2, 3, and 4 and the Preferred Alternative replace IC with II and could reduce light emissions compared to Alternative 1, but taller building heights would increase visibility in Southeast Magnolia and South Queen Anne. Alternative 3 has a slightly smaller footprint of II than alternatives 2 and 4 and may affect a smaller viewshed. The Preferred Alternative would use the same land use pattern as Alternative 4, but heights in the II zone are anticipated to be lower, reducing the effective viewshed compared to the other Action Alternatives.
- SODO/Stadium: MML zone would replace IG zone in most areas, and development style and light emissions similar in nature, with Alternative 2 having a higher share of MML zone than alternatives 3 and 4 which have reduced footprints of MML. The II zone would bring taller building heights and visibility from Beacon Hill and surrounding areas with all Action Alternatives; alternatives 3 and 4 have greater II zoned areas and greater visibility than Alternative 2. For alternatives 2 and 3, the UI zone would reduce light emissions and create transition areas in targeted locations near the stadium district/downtown. Alternative 4 has a greater area of UI south and west of stadiums. South of stadiums, Alternative 4 applies UI which would slightly increase heights and visibility but would reduce light emissions. The Preferred Alternative applies MML west of the stadiums, which would have similar uses and light and glare effects to the No Action Alternative. The Preferred Alternative would also expand the II area south and east of the stadium UI corridor, and south of S Lander Street. The Preferred Alternative would also apply Industrial Commercial and Mixed Use zoning in the portion of the subarea outside the MIC in southern Judkins Park.
- Georgetown/South Park: Alternative 2 applies MML in place of IG with light emissions similar in nature and location as Alternative 1. Under alternatives 3 and 4 increased light emissions in the area between Corson Ave and Ellis Ave would increase due to conversion of current IB zoning to MML. Compared to Alternative 2 and No Action, alternatives 3 and 4 would have increased visibility of MML and UI areas removed from MIC due to taller building heights under SM zoning, though light emissions would likely be reduced. The Preferred Alternative would expand the SM zoning in this location, compared to alternatives 3 and 4. The Preferred Alternative would also expand the Georgetown UI node westward along S Orcas Street and S Homer Street, potentially reducing light and glare emissions compared to the other action alternatives.

What are some solutions or mitigation for impacts?

Existing City of Seattle development regulations include design standards that govern the placement of exterior illumination and requirements for shielding of light sources. The City also maintains SEPA policies that would require evaluation of light and glare impacts on sensitive resources for any site-specific development in the study area undergoing SEPA review. The land use concepts proposed under the Action Alternatives also include provisions for landscaping and greenspace that would help screen light sources from surrounding areas.

Additional mitigation to be considered could include additional design standards to regulate placement, light output, direction, and shielding of any exterior illumination above a given height to reduce light and glare emissions to adjacent non-industrial areas.

With mitigation, what is the ultimate outcome?

Any future growth in the study area, regardless of the specific uses or building design, will generate at least some increase in light and glare. Though unavoidable, these effects can be minimized and reduced to less than significant levels through application of design standards and the mitigation measures described in this EIS.

1.7.8 Land & Shoreline Use

How did we analyze Land & Shoreline Use?

The EIS uses an inventory of existing land uses based on parcel level GIS data that was updated with manual scans by City staff and consultants and input from stakeholders. We reviewed existing and projected employment information from a 2021 CAI Inc. study. We reviewed applicable state, regional and local land use policies. We anticipated the type and character of development that would be likely under existing and proposed zoning and analyzed potential impacts of the expected land use composition under each of the studied alternatives in four broad categories: inconsistencies with land use policies, conflicts resulting from incompatible land uses within industrial areas, employment mix impacts, and impacts resulting from inadequate



W Galer St Overpass and the Magnolia Brdige

transitions from industrial to nonindustrial areas. Impacts were analyzed for the study area as a whole and within the five subareas where appropriate—Ballard, Interbay Dravus, Interbay Smith Cove, SODO/Stadium, and Georgetown/South Park.

What impacts did we identify?

We identified some land use impacts under all alternatives and found that land use impact would vary by subarea, but none of the impacts would be significant adverse impacts. We characterized the severity of impact as minor or moderate in the categories described above.

What is different between the alternatives?

The alternatives differ in the geographic pattern of zoning changes and development standard allowances for industry supportive housing. The alternatives would result in differing amounts and patterns of future employment and housing growth, and the future type and character of expected development. The analysis showed no significant adverse impacts but did identify different levels of potential minor and moderate land use impacts resulting from the expected future land use pattern, including the potential locations of dense employment, and increased industry-supportive housing.

Inconsistency with Plans and Policies: Some degree of inconsistency between the expected land use pattern and plans and policies was found for all the alternatives. Since consistency of land use patterns with plans and policies requires interpretation and balancing with many policies, it is common for some inconsistency to exist, while maintaining an overall predominant level of consistency. Alternative 1—No Action would have moderate inconsistencies due to the likely continuing trend of stand-alone retail and office development and mini-storage locating in industrial zones and MICs under existing zoning. This is inconsistent with certain policies prioritizing industrial and maritime uses in these areas. Moderate inconsistencies would be present under alternatives 3 and 4 and the Preferred Alternative due to the introduction of increased amount of industry-supportive housing, which can be viewed as inconsistent with some regional and local policies limiting residential uses in MICs. Alternative 2 would have the fewest, and only minor, inconsistencies because Alternative 2 would reduce the prevalence of non-industrial uses in industrial areas through new standards in the proposed MML zone in larger areas than alternatives 3 and 4 and the Preferred Alternative, and Alternative 2 does not include expanded allowances for housing.

Incompatible Land Uses: Moderate incompatible use impacts are expected in all subareas under Alternative 1 due to the potential for stand-alone retail and office developments and mini-storage to locate in industrial areas causing potential incompatibility with industrial uses. Alternatives 3 and 4 <u>and the Preferred Alternative</u> would see moderate incompatible use impacts in some subareas—most notably Ballard, SODO/Stadium, and Georgetown/South Park—where introduction of new buildings with dense employment in the II zone and industry-supportive housing in the UI zone could create incompatibilities between new activity patterns and adjacent areas of continued industrial uses. Alternative 2 would have the fewest, and only minor, land use incompatibilities since the application of the II and UI zones would be more limited in scale.

Inadequate Transitions: Potential for inadequate transitions from industrial to nonindustrial areas is highest for the Ballard and Interbay Dravus subareas. Moderate impacts at transitions would be expected in <u>the Ballard and</u> Interbay Dravus <u>subareas</u> under all the alternatives,



and in Ballard under alternatives 1, 2, and 3. In general, portions of the study area that abut residential and urban village locations without strong physical edge features such as greenbelts, major roadways or topographical changes have greater potential for inadequate transition. Future land use under the UI zone is expected to assuage potentially inadequate transitions to residential and urban village areas, thus Alternative 4, which includes more UI zoning in the Ballard Subarea would have moderate transition impacts. Minor transition impacts are identified for the Georgetown/South Park <u>s</u>ubareas under all the alternatives, and for the <u>north portion of the SODO/Stadium Subarea</u> under <u>all</u> alternatives <u>1, 2, and</u> 3. NoMinor transition impacts are expected for Interbay Smith Cove under any alternative primarily because of the strong physical edges around the subarea.

Employment Mix Impacts: With <u>onetwo</u> exception<u>s</u>, no employment mix impacts are expected. In all subareas and under all alternatives, the projected employment mix would remain 50% or more industrial—one of the threshold criteria for regional designation as a MIC. A minor employment mix impact was identified in Alternative 4 <u>and the Preferred Alternative</u> for the Ballard subarea, where the percentage of industrial employment is projected to fall to a level approaching the 50% threshold.

What are some solutions or mitigation for impacts?

Numerous mitigation measures are incorporated plan features of the proposal including adoption of the proposed Comprehensive Plan policies and adoption of zoning regulations that

reduce the size of use limits for non-industrial uses and that prohibit new mini-storage facilities in industrial zones. For alternatives 3 and 4 and the Preferred Alternative, the proposed requirement for new housing occupants to have a connection to industrial activity in the area mitigates the potential impact. Proposed development regulations in the UI zone including application of frontage improvement standards, green factor landscaping requirements and setback standards to encourage urban character buildings would mitigate potential transition impacts where industrial areas abut residential areas or urban villages.

The Preferred Alternative includes additional detail or nuance to proposed development standards that are incorporated plan features intended to mitigate impact or respond to Draft EIS comments. This includes unique development standards for the proposed mixed use area in central Georgetown, a workforce development space incentive in the proposed II zone, and added flexibility for existing non-conforming uses in the proposed MML zone.

Existing regulatory commitments provide mitigation. Shoreline Master Program regulations would continue to apply to areas within 200' of shorelines providing additional guidance and regulation for appropriate shoreline uses. Future development under all alternatives would be subject to project level SEPA review.

Additional mitigation measures that could be considered to reduce the identified land use impacts include:

- Apply the maximum size of use limits and mini-storage prohibition of the proposed MML zone, to the existing Industrial General zones of Alternative 1, should a No Action Alternative be selected.
- Limit the geography of industry-supportive housing allowances to a pilot area of the proposed Urban Industrial zone locations, and closely monitor the production and impact of resultant housing.
- Update zoning at edge areas outside of the study area in the future, including the potential application of the Urban Industrial zone to locations outside of MICs and current industrially-zoned areas.
- Expand contributions by public agencies and private partners towards equitable development especially in locations historically impacted by industrial activities.
- Amend practices or procedures for determining when a substantial alteration threshold is exceeded for adaptive reuse of existing industrial structures.

With mitigation, what is the ultimate outcome?

Land use impacts are expected under all alternatives to varying degrees but none of the impacts are expected to be significant adverse impacts. Numerous mitigation measures are included as an integrated part of the proposed zoning, development standards, and comprehensive plan amendments under alternatives 2, 3, and 4 and the Preferred Alternative. Identified land use impacts could be further mitigated to an even lower level if a pilot approach to industry-supportive housing in the UI zone under alternatives 3 or 4 or the Preferred Alternative were adopted, and with future actions supporting equitable development and future adjustments to zoning at edge areas outside of the study area.

1.7.9 Housing

How did we analyze Housing?

This EIS considers housing inventory, production trends, and challenges and needs (including public health, access to opportunity and displacement risk) based on U.S. Census American Community Survey, City of Seattle, and King County Assessor data. Projected levels of residential and employment growth under each of the alternatives are compared to existing conditions. Impacts of redevelopment are considered significant if they would:

- Result in a loss of housing due to redevelopment and insufficient development capacity, tools, or programs to address displacement of dwellings and population,
- Increase households' exposure to air pollution, noise pollution, or environmental hazards in census tracts identified as having high environmental health disparities (e.g., exposure to diesel emissions and ozone or proximity to hazardous waste sites) and with sensitive populations (e.g., poverty, cardiovascular disease) based on the Washington Department of Health Environmental Health Disparities Index, or
- Create a demand for housing that cannot be accommodated within the city in adjacent districts or areas where housing is planned.

What impacts did we identify?

There is limited housing of 413 dwellings in the nearly 11 square mile study area. The City of Seattle Displacement Risk Index identifies areas of Seattle where displacement of marginalized populations may be more likely. It combines data about demographics, economic conditions, and the built environment into a composite index of displacement risk. Overall, parcels within the study area are at low or moderate risk for displacement. Under all alternatives additional growth and development will occur in the study area, with small changes to housing patterns. No significant loss of existing housing due to redevelopment is anticipated under any of the alternatives.

The Action Alternatives limit new housing in industrial zones to formats that are supportive of industrial uses (caretaker's quarters, live/work units, etc.). Alternatives 3 and 4<u>and the</u> <u>Preferred Alternative to a lesser dense level</u>, also add mixed-use housing opportunities near the Georgetown/South Park Subarea. As well, the Preferred Alternative rezones areas in West <u>Ballard and Judkins Park to mixed use allowing housing instead of industrial uses on these</u> <u>lands that lie outside the MIC.</u> Given the health impacts of housing proximity to industrial areas, especially the Duwamish area based on exposure of sensitive populations to air emissions and hazardous materials per the Washington Environmental Health Disparities Map, it is important to limit housing in these areas. Increases in housing under the alternatives, especially alternatives 3 and 4<u>and the Preferred Alternative</u>, will place residential uses in proximity to air quality and noise emissions. The Action Alternatives include new zoning standards that will provide amenities for residents of the study area. UI zoning is intended to create thoughtful integration between the edges of these industrial areas and adjacent neighborhoods. UI zoning would seek to improve environmental health, walkability, and comfort in these areas.

Increases in employment growth in the study area could shift some of the overall expected citywide employment growth into industrial areas. This could have an impact on housing, especially if additional new employment were added to industrial areas not subject to the Mandatory Housing Affordability (MHA) regulations. There may be shifts in housing demand in areas adjacent to or within easy access to the industrial employment centers. However, the increment of employment growth in all alternatives is within the citywide amount that the City will plan for in the 2024 Major Comprehensive Plan update.

What is different between the alternatives?

Each of the alternatives is consistent with City and regional policy that limits housing in industrial areas. None of the alternatives allow significant new housing growth on industrial lands. Alternative 2 would not change housing allowances and would only add 80 units to increase the total housing units to 493 units. Alternative 3 changes caretakers' and makers studio allowances and would add 610 units for a total of 1,023 dwelling units. Likewise, with greater zoning allowances, Alternative 4 adds 2,195 caretakers' and makers studio units for a total of 1,023 dwelling units for a total of 2,608 dwelling units. <u>The Preferred Alternative adds 1,475 caretakers' and makers studio units in the range of alternatives 3 and 4.</u> Both alternatives 3 and 4 also add mixed-use housing opportunities (an estimated 1,078 units) near the Georgetown/South Park Subarea in land to be removed from the MIC. <u>The Preferred Alternative allows about 686 units in mixed-use housing opportunities in the Georgetown/South Park Subarea in land Judkins Park areas outside the MIC.</u>

What are some solutions or mitigation for impacts?

- Increases in housing units under alternatives 2, 3, and 4<u>and the Preferred Alternative</u> will be subject to the development standards developed under the UI zone. These include pedestrian and cyclist-oriented frontage improvements, development of green public spaces, access to planned transit and non-motorized transportation connections that support new development. The integration of public green open spaces, pedestrian-oriented amenities, and the access to transit helps to soften potential impacts of locating housing in areas of intensive industrial activity and employment growth.
- Seattle's Plans and City Code help to address and avoid potential displacement. Examples include Seattle's Tenant Relocation Assistance Ordinance, Notice of Intent to Sell Ordinance, and Rental Registration and Inspection Ordinance.
- The City could consider applying MHA regulations to the to the proposed new II zone. Applying MHA to the proposed new II zone can mitigate shifts in demand related to employment growth in the industrial areas.
- See the Air Quality & GHG and Noise sections for mitigation meant to address housing compatibility and health.

The City will plan for the citywide amount of housing growth in the Comprehensive Plan EIS on a citywide scale. As part of this ongoing commitment, the City could consider:

• Adding additional capacity for housing in urban villages and residential areas in locations that will have fast access to the new II zones to help address the shifts in demand for housing in

response to employment growth in industrial areas. The II zones are in the closest locations to light rail ($\frac{1}{4}$ - $\frac{1}{2}$ mile), and light rail will provide good access to these areas.

 Adding additional capacity for housing in urban village and residential areas in locations adjacent to new UI zones to address the shifts in demand for housing in response to employment growth in the industrial areas.

With mitigation, what is the ultimate outcome?

Each of the alternatives allows for additional growth and development, including modest numbers of housing units. Under all alternatives additional growth and development will occur in the study area, with the potential for small changes in housing patterns. This change is unavoidable but is not considered significant or adverse within a changing urban area designated as an employment center in the Comprehensive Plan. However, with existing and new development regulations, and anti-displacement programs currently in place, no significant adverse impacts are anticipated.

Residential uses will be in proximity to air quality and noise emissions, particularly alternatives 3 and 4<u>and the Preferred Alternative</u>. With the application of air quality and noise mitigation measures, no significant unavoidable adverse housing impacts would occur under any of the alternatives.



The Bemis Building in SODO with Artist Studios

Increases in employment growth in the study area could shift some of the overall expected citywide employment growth into industrial areas. This could shift some demand for housing into areas adjacent to or within easy access of the industrial areas. With the application of mitigation measures, including the application of MHA regulations to the II zone, and citywide planning for housing capacity through the Comprehensive Plan, no significant unavoidable impacts would occur under any of the alternatives.

1.7.10 Transportation

How did we analyze Transportation?

Existing transportation conditions are documented throughout the study area and present findings related to current transportation and circulation. This includes travel time data along study corridors, passenger load data on existing buses and light rail trains, peak period volumes, and collision data. GIS files maintained by the City were used to map and describe existing pedestrian and bicycle facilities.

A version of the PSRC model developed for the West Seattle and Ballard Link Extension (WSBLE) project and the Ballard-Interbay Regional Transportation (BIRT) System project was used to estimate future year volumes. This version of the PSRC model is consistent with the growth and transportation network anticipated through 2042. While the No Action Alternative reflects land uses anticipated through 2042, the potential land use changes under the Action Alternatives extend slightly farther to a 2044 horizon year. This provides a conservative basis to evaluate potential impacts of the Action Alternatives compared to Alternative 1 No Action.

What impacts did we identify?

By 2044, traffic volumes and travel times would increase due to the land use growth within the Study Area and in other parts of the city as well as regional growth. There would be more people walking, biking, and riding transit, resulting in some impacts to those modes due to incomplete networks and potentially crowded buses. The Study Area is not expected to meet its SOV mode share target. Impacts to travel time, parking, and safety were also identified.

What is different between the alternatives?

Exhibit 1.7-2 summarizes the impacts among the alternatives. The impacts of the Action Alternatives are assessed against Alternative 1 No Action. Impacts identified under Alternative 1 No Action would remain throughout the Action Alternatives even if those alternatives would not result in additional impacts.

Type of Impact	Alternative 1 No Action	Alternative 2	Alternative 3	Alternative 4	<u>Preferred</u> Alternative
Active Transportation	Yes	Yes	Yes	Yes	
Auto & Freight					
Travel Time	10 LOS F corridors	1 impacted corridor	3<u>4</u> impacted corridors	3 <u>4</u> impacted corridors	<u>2 impacted</u> corridors
Mode Share	3 sectors	No	1 impacted sector	1 impacted sector	No
Screenline	No	No	No	No	No
Active Transportation	Yes	Yes	Yes	Yes	<u>Yes</u>
Transit	1 screenline	No	No	No	No
Parking	Yes	Yes	Yes	Yes	<u>Yes</u>
Safety	Yes	Yes	Yes	Yes	<u>Yes</u>

Exhibit 1.7-2 Summary of Significant Transportation Impacts

Source: Fehr & Peers, 202<u>2</u>1.

In summary, Alternative 1 No Action is expected to have significant impacts to active transportation, auto, and freight in terms of travel time, mode share, transit, parking, and safety. Alternative 2 is expected to result in additional significant impacts to autos and freight on one corridor as well as impacts to active transportation, parking, and safety. Alternatives 3 and 4 are expected to result in additional significant impacts to auto and freight on threefour corridors and one mode share sector as well as impacts to active transportation, parking, and safety. The Preferred Alternative is expected to result in additional significant impacts to auto and freight on two corridors as well as impacts to active transportation, parking, and safety. The locations of the corridors impacted by the Action Alternatives are mapped in **Exhibit 1.7-3** and **Exhibit 1.7-4**.



Exhibit 1.7-3 Impacted Study Corridors—Ballard Interbay Northend MIC, 2044

Note: This map was updated to include results of the Preferred Alternative analysis. Source: Fehr & Peers, 202<u>2</u>+.





Note: This map was updated to include results of the Preferred Alternative analysis and two additional study corridors (18 and 19). Source: Fehr & Peers, 20221.

What are some solutions or mitigation for impacts?

Under all alternatives, the City could implement solutions related to Transportation Systems Management and Operations (TSMO), travel demand management (TDM), pedestrian and bicycle system improvements, and parking management strategies. In combination, these measures could help reduce the SOV mode share for non-freight types of trips which is key to limiting the potential severity of transportation impacts. Lowering SOV mode share when possible would not only reduce travel time, mode share, and parking demand impacts, but is consistent with numerous other goals and policies in the Comprehensive Plan. <u>The City could</u> also pursue a variety of operational and capital projects aimed at addressing particular freight. bottlenecks to improve freight mobility and access. Representative projects that could improve freight mobility and access include: truck-only or joint-use freight and truck lanes, rail corridor grade separation, intersection geometry improvements to address turn radii challenges for trucks, channelization improvements, signal phasing or timing modifications, wayfinding and signage, intelligent transportation systems (ITS) strategies, and dedicated pedestrian and bicycle facilities to separate vulnerable users from freight.

Location-specific mitigation measures were identified for the travel time impacts along 15th Avenue W (between Magnolia Bridge and NW Leary Way) and W Dravus Street (between 15th Avenue W and 20th Avenue W). For 15th Avenue West, the measures include intersection operations refinements, adaptive signal system installation, transit and freight only lanes, and replacement of the Ballard Bridge. For W Dravus Street, the measures include signal operations improvements, roadway striping/channelization modifications, access management enhancements, and replacement and/or widening of the W Dravus Street bridges. No locationspecific mitigation measures addressing the travel time impact along I-5 between Madison Street and SR 599 have been identified.

The Preferred Alternative (developed based on feedback regarding potential impacts of the Draft EIS alternatives) would have less employment density than alternatives 3 and 4. The land uses proposed under the Preferred Alternative were analyzed using the regional travel demand model, which suggests there would be no significant travel time impacts to either I-5 or SR 509 under the Preferred Alternative.

Regarding land use mix and trips, under alternatives 3 and 4, the City could consider the balance of employment uses and plan for greater industrial jobs, and a smaller share of non-industrial jobs (e.g., retail, services, office) in the Greater Duwamish MIC to reduce trips. The City could consider a preferred alternative that has less of the employment dense Industry and Innovation zone than is found in alternatives 3 and 4 but more than Alternative 2 but that still avoids significant adverse impacts on I-5.

With mitigation, what is the ultimate outcome?

If mitigation measures are implemented, it is expected that the travel time impacts on 15th Avenue W and W Dravus Street could be brought to a less-than-significant level in relation to Alternative 1 No Action. Because no location-specific capital facility-based mitigation measures
along I-5 <u>or SR 509</u> are expected to fully mitigate the travel time impact to autos, freight, and buses, a significant travel time impact is expected under alternatives 3 and 4 on I-5. <u>MHowever,</u> <u>the land use m</u>odifications to alternatives 3 and 4 that<u>proposed for the Preferred Alternative</u> <u>are expected to reduce the total amount of future employment in the SODO/Stadium subareacould potentially</u> mitigate the impact to I-5 and SR 509 if the reduction in trips is below the threshold of significance.

Some combination of the TDM strategies could be implemented to reduce the magnitude of SOV travel. Given the small magnitude of difference projected between Alternative 1 No Action and alternatives 3 and 4, it is expected that the mode share impact could be reduced to a less-than-significant level. The land use modifications proposed for the Preferred Alternative are expected to mitigate the mode share impact below the threshold of significance.

Parking impacts are also anticipated to be brought to a less-than significant level by implementing a range of possible mitigation strategies. While there may be short-term impacts as individual developments are completed (causing on-street parking demand to exceed supply), it is expected that with mitigation, the on-street parking situation would reach a new equilibrium as residents, employees, and visitors adjust to the new context. Therefore, no significant unavoidable adverse impacts to parking are expected.

Significant impacts were identified to both active transportation and safety due to the projected increase in people walking, and biking, and rolling in areas with network gaps and the increased potential for vehicle conflicts (particularly trucks) and rail with vulnerable users. While the City can pursue a variety of mitigation measures to improve active transportation facilities for people walking and biking and pursue supplemental funding through federal or state programs, it is not expected that all network gaps can be addressed given the number of locations needing improvement and the limited funding available. Therefore, it is expected that the Action Alternatives could have significant unavoidable adverse impacts to active transportation and safety.

1.7.11 Historic, Archaeological, & Cultural Resources

How did we analyze Historic, Archaeological, & Cultural Resources?

To analyze historic, archaeological, and cultural resources in the study areas for the purposes of this report, we used a wide variety of sources to obtain information on the environmental, archaeological, and historical backgrounds of the project vicinity, and developed useful contexts for analysis. We gathered data from the King County Assessor's website, the Department of Archaeology and Historic Preservation's (DAHP's) online database, the Washington Information System for Architectural and Archaeological Records Data (WISAARD), and the City of Seattle's Landmarks List.

Using this data, our GIS Specialist created maps indicating parcels that contained historicperiod architectural resources (buildings, structures, objects, sites, and districts), had a historic property inventory form (HPI) in WISAARD, were eligible for or listed in the National Register of Historic Places (NRHP), or listed in the Washington Heritage Register (WHR), Washington Heritage Barn Register (WHBR), or was a designated Seattle Landmark (SL).

Additionally, the GIS Specialist gathered data on cultural resource survey reports, archaeological site records, and cemetery records in the MICs/ project subareas, and created maps that plotted recorded archaeological and cultural resources. These maps were analyzed by an archaeologist, who also reviewed the environmental characteristics,



Ballard Bridge

ethnographic data, and the distribution of known cultural resources within the MICs, reviewed DAHP's predictive model, and formulated expectations about the probability of impacts to known and as-yet unknown archaeological and cultural resources.

What impacts did we identify?

All the alternatives have the potential to affect districts, sites, buildings, structures, or objects (BSOs) that have been listed in the NRHP and other historic registers (Washington Heritage Register [WHR], and Seattle Landmarks [SL]), and those determined eligible for listing in the NRHP. Additionally, the alternatives could potentially affect the numerous BSOs and undiscovered archaeological sites that have yet to be surveyed and assessed for eligibility to the NRHP.

Impacts to historic, archaeological, and cultural resources in the study areas from the No Action and threefour Action Alternatives were identified by assessing potential for both aboveand below-ground changes. Such impacts generally include physical alteration, damage, or destruction of all or part of a resource that would affect its eligibility to qualify for inclusion in the NRHP.

What is different between the alternatives?

Alternative 1—No Action maintains the status quo within the existing industrial zones, with no changes to current Comprehensive Plan policies, development standards, or zoning. Impacts would be similar to those described above—physical alteration, damage, or destruction—due to no additional protections or improvements in planning for consideration of impacts to historic, archaeological, and cultural resources.

Under Action Alternatives, changes to zoning that allows a wider range of industrial or nonindustrial uses could spur redevelopment in those locations. Even where there are no formally designated historic landmarks, there are numerous properties with historic period buildings, or a very high or high risk of archaeological resources. Alternatives 2, 3, and 4<u>and the Preferred Alternative</u> would impact historic, archaeological, and cultural resources similar to Alternative 1, but would also increase the probability of inadvertent discovery of archaeological and cultural resources due to the need for substantial foundation work needed for multi-story buildings. Alternatives 2, 3, and 4<u>and the Preferred Alternative</u> feature different amounts of land rezoned to the proposed new UI and II zones that would allow denser development—with alternatives 3 and 4 <u>and the Preferred Alternative</u> having more land rezoned to II or UI. In general, areas would experience relatively greater pressure for redevelopment, which could impact historic resources. Additionally, without design guidelines or review, allowed adaptive reuse projects could impact historic-period architectural resources.

Under alternatives 3 and 4 and the Preferred Alternative, focused areas in Georgetown/South Park would be removed from the MIC to allow for mixed-use development including some areas where few surveys have been done. In addition, the Preferred Alternative proposes mixed use housing in industrial areas outside the MIC in West Ballard and Judkins Park. This may add to demolitions of historic-period architectural resources.

What are some solutions or mitigation for impacts?

Examples of solutions to avoid impacts to historic and cultural resources would be:

- Upon completion of the management plan (scheduled for 2022), Washington's designated National Maritime Heritage Area (NMHA) may help raise awareness of the importance of local history and historic resources, increase heritage tourism, strengthen relationships between heritage groups, and may allow for the receipt of grants and other federal funds, should funding be available.
- Implementation of a cultural resources survey and inventory (historic-period architectural, archaeological, and cultural resources) within the study area for the proposed project. Such surveys are recommended to take place during project development planning, so impacts to historic properties can be eliminated, minimized, or avoided, should historic properties be found within the survey area.
- An archaeological resources monitoring and inadvertent discovery plan (MIDP) should be developed for each project that includes ground disturbing activities, based on DAHP's archaeological predictive model. <u>All cultural resources survey and archaeological work</u> will follow best practices and standard archaeological techniques in the discovery and preservation of cultural and historical artifacts.
- When elimination, minimization, or avoidance of impacts to cultural resources is impossible, mitigation should be developed in accordance with DAHP Mitigation Options and Documentation Standards and in coordination with the area's Tribes, the lead agency, and all other consulting parties.

Some examples of mitigation for impacts are:

- Archaeological excavation and/or collection of artifacts for conservation.
- DAHP Level I (Historic American Building Survey/Historic American Engineering Record [HABS/HAER]) Documentation.
- DAHP Level II Documentation.

Other potential mitigation measures include:

- Funding City-initiated proactive landmark nominations for properties and potential historic districts identified in new neighborhood surveys.
- Prioritizing City funding for retrofitting Unreinforced Masonry (URM) buildings to those properties that meet eligibility requirements for designation as a landmark or for listing in the National Register of Historic Places. Development of cultural landscape contexts, including within historically marginalized communities.
- Developing histories of the study area including Indigenous perspectives. The City could work with tribes and others to develop context statements. A context statement focused on Historical Planning and Land Use Decisions is drafted in Section 3.8 Land & Shoreline Use.
- Funding City-led thematic historic context inventories that focus on marginalized or underrepresented immigrant communities and preparing thematic context statements relating to those resources.
- Supporting neighborhood survey and inventory projects within underrepresented or marginalized communities.
- Considering potential impacts to historic resources during development review specifically that are associated with marginalized or underrepresented immigrant communities as part of project level SEPA review, or during the design review process.
- Including a development incentive for preservation of architectural resources including adaptive reuse projects in the proposed Urban Industrial zone, such as an exemption from the floor area ration calculation, or flexibility for allowable uses within the structure. Such adaptive reuse projects could follow the Secretary of the Interior Standards for Rehabilitation or the City could develop new rehabilitation guidelines for adaptive reuse.
- For alternatives 3 and 4<u>and the Preferred Alternative</u>, exploring or studying the possible addition of a new Seattle Landmark District for the mixed-use area of Georgetown.
- Unique development standards for the Georgetown mixed-use area are included for the Preferred Alternative. The new mixed use zone could incentivize preservation and retention of historic structures by exempting floor area in historic-aged character structures from FAR maximums and providing other code flexibilities when a structure is preserved.
- Adding regulatory authority to identify resource-specific mitigation before demolition occurs.
- Requiring project proponents to nominate buildings for landmark review when demolition
 of properties that are over 50 years old is proposed, regardless of City permitting
 requirements, by modifying the SEPA exemptions thresholds in the Seattle Municipal Code
 at Table A for section 25.05.800, and Table B for section 25.05.800.

With mitigation, what is the ultimate outcome?

Advanced planning to eliminate, minimize, or avoid impacts to cultural resources is key. There is potential for significant adverse impact under all alternatives but with appropriate and meaningful mitigation significant impacts are avoidable. The ultimate outcome with mitigation is to moderate the adverse impacts of historic, archaeological, or cultural resources before they are lost or significantly altered. With mitigation, significant adverse impacts to historic, archaeological, and cultural resources can be avoided.

Smith Cove Park



1.7.12 Open Space & Recreation

How did we analyze Open Space & Recreation?

Impacts to open space and recreation were assessed based on the City of Seattle's adopted Level of Service (LOS) standard of 8 acres of open space for every 1,000 residents. Additional parkland required under each alternative to meet the LOS standards was then assessed in relation to the City's existing plans, policies, and regulations.

The thresholds of significance utilized in the impact analysis include:

- Insufficient parks, open space, and trail capacity to serve expected population or employment based on levels of service.
- Inconsistencies with shoreline public access policies.
- Have the potential to decrease public access to parks and open space or shoreline access in census tracts identified as high disadvantage in the Seattle Racial and Social Equity Composite Index. See Exhibit 1.7-7 later in this chapter.

What impacts did we identify?

Anticipated impacts on open space and recreation as a result of the alternatives include increased demand on existing parks, demand for new park land, and potential changes to the transportation network and/or transportation behavior.

What is different between the alternatives?

The difference between the alternatives for open space and recreation is the number of acres required to meet the LOS standard: 1.22 additional acres are required under Alternative 1, 1.30 additional acres under Alternative 2, 27.68 additional acres under Alternative 3, and 53.68 additional acres under Alternative 4, <u>and 49.36 additional acres under the Preferred Alternative</u> (see **Exhibit 1.7-5**). Alternative 1 No Action and Alternative 2 require the least amount of land to

meet the City's adopted LOS standard while Alternative 4 <u>and the Preferred Alternative</u> requires the most acres of land. The net park acres required under Alternative 4 would exceed the number of acres expected in the City's 2017 Parks, Recreation, and Open Space Plan.

Alternative	Net Population Growth	Net Open Space to Meet LOS Standard (Acres)
Alternative 1 No Action	153	1.22
Alternative 2	163	1.30
Alternative 3	3,460	27.68
Alternative 4	6,710	53.68
Preferred Alternative	<u>6,171</u>	<u>49.36</u>

Exhibit 1.7-5	Net Open Space and Recreation Acres to Meet LOS Standards, All Alternatives

Source: BERK, 202<u>2</u>+.

Alternatives 3 and 4 and the Preferred Alternative includes the removal of portions of two blocks of land adjacent to Duwamish Waterway Park and two blocks of land adjacent to Terminal 117/Duwamish River People's Park from the MIC designation and industrial zoning, and would apply a mixed-use zone. The higher housing and population growth anticipated under alternatives 3 and 4 and the Preferred Alternative would likely also require additional connectivity to/from open spaces for residents living in the area. Future development in the mixed-use zone has a higher potential for increasing integration with and access to the two open spaces from the South Park residential community. The change will increase the amount of required open space in new development near the parks and will increase the likelihood of future visual and/or physical access to river front land from privately owned parcels.

The Preferred Alternative would have demand for parks in the West Ballard and Judkins Park areas outside the MICs where NC zoning would be applied allowing mixed uses.

What are some solutions or mitigation for impacts?

The new land use concepts proposed under the Action Alternatives features design principles that would help mitigate impacts to open space and recreation, including standards for frontage improvements (sidewalks, pedestrian lighting, etc.), trees and landscaping, maximum limits on vehicle parking areas, and circulation routes that could be used as trails.

The City of Seattle regularly identifies and plans for open space and recreation needs. Relevant plans include Seattle Parks and Recreation's Recreation Demand Study, Community Center Strategic Plan, 2017 Parks, Recreation, and Open Space Plan, and 2020-2032 Strategic Plan. Additional open space and recreation needs and commitments are identified in annual reports from the Seattle Park District Annual Reports, the Seattle Comprehensive Plan, and the Duwamish Valley Action Plan. In addition to these plans, the Seattle Land Use Code (Seattle Municipal Code Title 23) contains development regulations related to open space and recreation, including standards governing the design and placement of exterior site and building

illumination. Future development in the study area will be required to comply with the standards established for industrial zones in SMC Chapter 23.50 and 23.49 as it pertains to open space.

While parks are a great source of open space, the combination of existing uses and new land use concepts within the alternatives may present challenges that may not be resolved with new parks. Other potential mitigation measures the City could explore outside of creating new parks include creating linear parks and trails, increasing frequency of maintenance to offset an increase in park usage, and building resilient parks. The City could also explore improving transportation to and from parks and potentially increase connectivity between existing and future parks. Finally, the City might explore the use of community gardens (permitted on some rooftops in individual zones) as a way to provide open space and an urban agricultural use.

With mitigation, what is the ultimate outcome?

While population and employment growth would occur under all studied alternatives, there are opportunities to meet the City's level of service for parkland through implementation of the Seattle plans and current and proposed development regulations. No significant unavoidable adverse impacts to open space and recreation are anticipated as a result of the alternatives.

1.7.13 Public Services

How did we analyze Public Services?

The public service evaluation considers the effect of the alternatives on fire/emergency medical services, police, school, and library services. Data from service providers is compiled for the study area. A focus is on the ability to meet levels of service or effects on capacity to provide services.

What impacts did we identify?

Growth in worker and residential populations in the study area is expected to lead to an increased number of calls for emergency services. Existing ladder trucks at stations in and near the study area are equipped to provide services to buildings of the heights proposed under all alternatives. Additional industrial development under all the alternatives could increase the amount or prevalence of hazardous materials in the study area. All new development would be required to meet the Seattle Fire Code which includes provisions for hazardous materials. Additional growth would increase traffic volumes which may in turn increase the response time of emergency vehicles.

Relative changes in population density by police beat and sector may generate more workload in some areas of the city but are not anticipated to impact police service or response times under any of the alternatives. Potential construction activities under all the alternatives could result in an increase in demand for police services. Existing Departmental resources are anticipated to be sufficient to handle such an increase. Future traffic volumes or changes to the transportation network in the study area could impact first responders' ability to respond rapidly to emergency

calls. SPD's staffing model factors in response time to determine appropriate staffing levels in each precinct. The Department would likely adjust staffing levels to improve response times if future increased traffic volumes or changes to the street network negatively impact police services.

Regular planning by SFD and SPD are anticipated to address incremental increased demand for fire, emergency medical, and police services. Any potential future facility, staffing, or equipment needs as a result of increased demand for services, traffic volumes, or changes to the transportation network could be included as part of the City's annual Budget and Capital Improvement Program process.

All alternatives to a lesser or greater degree may generate students that will attend schools, and residents of all ages that need library services.

What is different between the alternatives?

The demand for schools and libraries will be in proportion to the increase in housing under each alternative, which shows less growth in alternatives 1 and 2 and more under alternatives 3 and 4 and the Preferred Alternative. Based on the net change in dwellings and population, and a conservative assumption that 7.1% of the population are students, the number of potential students is shown in **Exhibit 1.7-6**. Since proposed housing in industrial zones would be limited to industry supportive types of live/work units and caretakers' units, the proportion of households with children could be lower. MThe most housing units and associated population are anticipated under Alternative 4 and the least under Alternative 1. TheNew students would have more effect on schools in Ballard, SODO/Stadium, and Georgetown/South Park.

Subarea	Alt. 1	Alt. 2	Alt. 3	Alt. 4	<u>Preferred</u>
Ballard	1	1	38	115	<u>75</u>
Interbay Dravus	1	1	11	25	<u>17</u>
Interbay Smith Cove	1	1	2	-	—
SODO/Stadium	4	5	29	144	<u>94</u>
Georgetown/South Park	3	3	9	35	<u>30</u>
Total: Ind Zone Housing (Caretaker/Artist)	11	12	89	319	<u>214</u>
With MIC Adjustments—Seattle Mixed-Use Zone Housing	_		157	157	<u>223</u>
Grand Total Students in Study Area	11	12	245	476	437

Exhibit 1.7-6 Student Generation by Subarea based on Net Change in Population

Source: BERK, 202<u>2</u>1.

Alternative 3 would affect demand at the South Park Library, and particularly schools like Wing Luke (capacity 351) and Concord (capacity 333) schools. This number of students would be about 45% of an elementary school capacity. However, the plan is a 20-year plan, and it is likely that not all housing would be developed at one time, and students would not start all at once and would be spread across grades.

Impacts under Alternative 4 are similar to Alternative 3 except that there would be substantially more caretakers' quarters/makers' studios, with most units and potential students in SODO/ Stadium and Ballard subareas. Like Alternative 3 there would be growth in the Georgetown/ South Park Subarea in mixed-use zones. All together there would be an increase in population of 6,710 including 476 students. Local libraries in Ballard and South Park would likely see an increase in demand for services. Schools serving Ballard, SODO/Stadium, and Georgetown/South Park could have increased demand at 33-45% of a typical elementary school capacity (~350).

The Preferred Alternative has similar impacts to Alternative 4 with more demand in Ballard and Judkins Park.

What are some solutions or mitigation for impacts?

- Compact growth in proximity to SFD and SPD services could result in more efficient service delivery and ability to meet LOS objectives.
- City fire codes govern inspection and operation of businesses and new construction (Title 22 Subtitle VI Fire Code of the Seattle Municipal Code, which has local amendments to the International Fire Code (IFC) with state adopted amendments).
- The Seattle Police Department enforces and is subject to various City of Seattle regulations such as Title 10 Healthy and Safety and Title 11 Vehicles and Traffic.
- Ongoing City of Seattle capital improvement planning and budgeting efforts are anticipated to address fire and police facility needs, including potential needs for future improvements.
- Ongoing Seattle School District capital facilities management planning is anticipated to be sufficient to address increases in student population. The Seattle School District prepares capital plans and projects are funded by levies.
- SDOT provides a Safe Routes to School program. In addition to education, there are walkway projects to make routes safer.
- The Seattle Public Library has a strategic plan and operations plan that guide the provisions of library services.
- The II and UI zones include potential changes to streetscape standards and could enhance walking routes to schools in areas with added housing.

With mitigation, what is the ultimate outcome?

All studied alternatives would increase the demand for public services in the study area with alternatives 2, 3, and 4<u>and the Preferred Alternative</u> increasing jobs above No Action. The increase in industrial jobs could result in a greater need for fire and emergency services in the study area. Increased non-industrial jobs would require apparatus for taller structures in the case of fire or rescue.

All alternatives, particularly alternatives 3 and 4<u>and the Preferred Alternative</u> would increase housing and increase demand for school and library services.

No significant unavoidable adverse impacts to fire and emergency medical services, police, or schools and libraries are anticipated with application of mitigation measures and regular capital planning.



1.7.14 Utilities

How did we analyze Utilities?

Utilities were analyzed by considering how the proposed alternatives, including changes in population, dwelling units, and jobs would affect wastewater generation (including CSOs), the quantity of stormwater runoff, and electrical demand. Stormwater quality is discussed in the <u>Section 3.3</u> Water Resources section.

What impacts did we identify?

The growth in population and employment may result in changes to the amount of wastewater flows and stormwater runoff generated as well as CSO frequency. Electrical demand could also increase due to an increase in population and employment.

What is different between the alternatives?

Generation of wastewater is scalable with population and employment. As a result, Alternative 4 would have the greatest increase in wastewater generation because it would cause the largest increase in employment and housing compared to the other alternatives. Conversely, with more development, stormwater management increases due to the implementation of stormwater management at development sites. For this reason, Alternative 4 would have the greatest reduction in the rate of stormwater runoff during the planning period and Alternative 1 No Action would have the least reduction in stormwater runoff amongst all alternatives.

Alternative 4 would also have the greatest reduction in CSO frequency during the planning period due to greater reduction in the rate of stormwater runoff to the combined system and the greatest increase in electrical demand due to increased population and employment. The Preferred Alternative is in the range of Action Alternatives considering combined employment and population.

What are some solutions or mitigation for impacts?

Seattle Public Utilities (SPU) manages the public wastewater and stormwater drainage in the City of Seattle. King County Wastewater Treatment Division (WTD) manages all the wastewater treatment plants and wet weather treatment facilities within the City of Seattle and surrounding King County. Together, SPU and WTD manage the combined sewer system. Seattle City Light (SCL) manages the electric power generation, transmission, and distribution services in the City of Seattle. Each utility plans, manages, and delivers capital projects that could mitigate the impact of all alternatives. The Seattle Stormwater Code also requires on-site management of stormwater, which could help mitigate the impact of stormwater runoff from all alternatives.

With mitigation, what is the ultimate outcome?

No significant unavoidable adverse impacts are anticipated for the wastewater, stormwater, CSOs, or electrical utilities under any of the alternatives. The levels of development proposed under all alternatives will be managed by existing, ongoing processes such as capital improvement planning and code requirements.

1.7.15 Equity & Environmental Justice Considerations

The City of Seattle has developed a Racial and Social Equity Index (posted January 2020) representing 5-year American Community Survey data, which provides information on race, ethnicity, and related demographics to consider areas where socioeconomic and health disadvantages. The index has three sub-indices: race/language/origins, socioeconomic, and health disadvantage. The Study Area boundaries and results of the index are shown in **Exhibit 1.7-7**.



Exhibit 1.7-7 Seattle Racial and Social Equity Index

Although the study area has a relatively low residential population density with only 413 existing residential homes, the results show where populations have higher or lower levels of disadvantages. Consideration is also given to where the study area abuts residential districts. More populations with higher disadvantages reside in the Greater Duwamish MIC than in the BINMIC. Within the Greater Duwamish MIC, the SODO/Stadium Subarea, and a portion of the Georgetown/South Park Subarea west of the Duwamish Waterway have the highest disadvantage. Other areas have middle or low disadvantage. Similar results are found within the Washington Environmental Health Disparities Map (see **Section 3.9 Housing** for selected maps). Both sources of socioeconomic and health data are considered in this EIS.

The EIS also considers how the alternatives advance the City's Equity and Environment Agenda and the City's Duwamish Valley Program and Action Plan described in **Section 1.3.2**. The alternatives are screened by whether they would increase, exacerbate, or impede mitigation of:

- Adverse impacts to air and water quality, soil contamination, noise pollution, and climate change, exacerbating **residents' and workers' exposures to environmental hazards**.
- Adverse impacts to achieve a safe, connected, and accessible neighborhood. Consider community conditions (transit, housing, food access/ insecurity, parks, sidewalks, cultural hubs, etc.).
- Adverse impacts regarding **displacement risk of EEI Populations**.
- Adverse impacts regarding access to education or pathways out of poverty through jobs and careers.

These screening criteria are addressed under EIS topics below. **Exhibit 1.7-8** at the end of this section summarizes the equity and environmental justice topics addressed in this EIS.

Natural & Biological Resources & Resiliency

Screening Criteria: Adverse impacts exacerbating residents' and workers' exposures to environmental hazards.

Summary of Impacts: The alternatives have the potential to allow for industrial and nonindustrial uses in areas of high disadvantage which may expose existing or new populations to air emissions. Current and new populations could be exposed to damage from sea level rise. Current and new populations would be exposed to risk of geologic hazards. Alternative 1 would have the lowest employment growth and least industry-related housing, and Alternative 4 would have the most with other alternatives in between <u>including the Preferred</u> <u>Alternative</u>. While greater development could result in more impacts, it can also result in more redevelopment meeting modern building and flood codes and improving conditions in the area (e.g., tree canopy, climate adaptation measures).

Mitigation and Investment: Mitigation measures include application of federal and state air emission standards (e.g., for vehicles), buffers between air emission sources and sensitive uses, interior air filtration, added tree canopy, and application of building and flood hazard

codes. Investment in climate adaptation measures could benefit current populations at risk of sea level rise as well as allow new development. Planting tree canopy in existing areas and redeveloping areas would benefit both existing and new populations and employees.

Topic-specific Summaries

Soils/Geology: Under any of the Action Alternatives, the primary equity and environmental justice concern for the proposal would be if development on lands subject to geologic hazards carries the risk of injury or damage to structures due to seismic activity. Although the proposal would allow development at sites in areas prone to landslides, liquefaction, or similar geologic hazards, modern building codes mitigate the risk of injury or economic losses for vulnerable communities.

Air Quality/Greenhouse Gas: While air quality impacts under all alternatives are expected to be less than significant, the primary equity and environmental justice concern for the proposal would be the emissions associated with industrial activities and road transportation emissions on vulnerable communities in the study area, on the periphery of industrial zones, and alongside higher-volume transportation routes. Populations with preexisting conditions that make them more sensitive to air pollution could be at greater risk from the activities associated with the alternatives. Potential mitigation measures consider buffers of sensitive land uses from emission sources, enhanced air filtration systems, and dense tree canopies.

The incremental traffic-related emissions of the proposed alternatives would represent a minor portion of all traffic emissions on any transportation route near vulnerable communities. In addition, due to EPA emission standards for motor vehicles and clean fuel standards, the total emissions from road transportation are expected to drop even as traffic levels increase in the study area. Thus, exposures to air pollution in the study area are expected to continue trending downward.

Water Resources—Water Quality: Increases in impervious surface can negatively affect surface water quality, which can disproportionately affect populations with a higher reliance on water resources for sustenance, such as subsistence fishers or Tribes. Poor water quality also poses health risks for populations that come in physical contact with surface water bodies. The Seattle Stormwater Code (SMC Title 22, Subtitle VIII) requires redevelopment projects in the Study Area to implement on-site stormwater management to infiltrate, disperse, and retain stormwater runoff to the maximum extent feasible. All Alternatives are expected to result in a net improvement in water quality and therefore reduce negative impacts on these populations as they relate to water resources.

Water Resources—Sea Level Rise: The Seattle Mapping Inventory of Changing Coastal Flood Risk provides a screening level picture of the impacts of sea level rise on Seattle. The analysis reveals that the communities most impacted by flooding are also disproportionately characterized by high levels of social vulnerability, most notably in the Georgetown/South Park Subarea. Under all Alternatives, proposed development in areas that are susceptible to impacts from extreme high tides would be required to comply with critical areas regulations



for frequently flooded areas. Compliance with these codes and implementation of adaptation measures may reduce vulnerability of those developments to sea level rise impacts relative to existing conditions.

Plants and Animals: New zones promote new streetscape and green space standards; the adaptation of impervious areas to increased tree canopy and green factor can increase shade and modestly improve habitat such as for birds and urban-adapted wildlife as well as for humans. Improvements to water quality and flow control would benefit fish and aquatic invertebrate species, many of which are harvested for human consumption.

Environmental Health & Compatibility

Screening Criteria: Adverse impacts exacerbating residents' and workers' exposures to environmental hazards.

Summary of Impacts: Cleanup of contaminated sites could cause temporary adverse effects from potential exposure of workers, nearby residents, and animals to contaminated soil, groundwater, surface water, fugitive dust, or spilled hazardous materials. Construction and increased activity under any of the alternatives has the potential to exacerbate residents' and workers' exposure to increased noise. Increased light and glare emissions would be particularly visible in South Park, an area of high disadvantage. There is more likelihood of construction activity in the Action Alternatives with high amounts of new jobs and with alternatives 3 and 4

<u>and the Preferred Alternative</u> that have the most residential uses. Opportunities include greater long-term health with more sites cleaned and with extended tree canopy.

Mitigation and Investment: Mitigation measures include detailed construction health and safety plans, noise reduction measures during construction, and construction standards to reduce noise. Additional landscaping, screening, setback, and lighting standards could reduce impacts both for existing residents and new workers.

Topic-specific Summaries

Contamination: Under any of the Action Alternatives, the primary equity and environmental justice concern for the proposal would be that cleanup of contaminated sites could cause temporary adverse effects from potential exposure of workers, nearby residents, and animals to contaminated soil, groundwater, surface water, fugitive dust, or spilled hazardous materials if mitigation measures are not fully implemented. Although all alternatives would likely result in short-term adverse effects on this determinant of equity and social justice, the Action Alternatives would generally have positive long-term benefits. In order to mitigate potential exposure to contaminants, all workers would be issued personal protective equipment and protected by measures implemented under the contractor's site-specific health and safety plan. Other mitigation measures include preparing a comprehensive contingency and hazardous substances management plan, a worker health and safety plan, a spill prevention control and countermeasures plan, and a Construction Stormwater Pollution Prevention Plan.

Noise: Construction and increased activity under any of the alternatives has the potential to exacerbate residents' and workers' exposure to increased noise. Limiting proximity of new residential and associated development to high noise sources would limit exposure to excessive noise. In addition, noise reduction measures can be mandated for construction activities and adequate noise reduction measures also mandated for new residential construction, in high noise environments within industrial areas. The City could impose greater noise reduction standards in residential buildings where exterior noise levels greater than US HUD standards.

Light and Glare: Exposure to light and glare emissions, are location-dependent and not equally distributed throughout the city. Due to market forces, historical practices regarding siting of industrial facilities, and historical restrictions on housing for people of color, residential areas near industrial centers are often home to communities of color and lower-income populations. Increased light and glare emissions would be particularly visible in South Park, an area of high disadvantage. Mitigation measures could include: additional landscaping, screening, and setback requirements in locations adjacent to residential zones, public lands, park and recreation facilities, and areas outside the BINMIC or Greater Duwamish MIC, and additional development standards to address maximum height of exterior illumination.

Working, Living, & Mobility

Screening Criteria: Adverse impacts to achieve a safe, connected, and accessible neighborhood (transit, housing, food access, parks, sidewalks, cultural hubs, etc.).

Displacement risk of EEI Populations.

Access to education or pathways out of poverty through jobs and careers.

Summary of Impacts: The risk of housing displacement is low due to the limited quantity of housing in the study area. The limited housing added under Action Alternatives could marginally assist with housing costs including rent due to expanded supply of housing and commuting costs if the additional homes are available to workers in the area. The Action Alternatives improve transitional standards for compatibility. Growth can bring impacts of traffic and delays in the study area including in areas with disadvantaged populations, but increased development can improve multimodal investments to create safe, connected, and accessible neighborhoods.

Each of the Action Alternatives includes an increase in projected employment in the study area, with substantially higher quantities of new employment under alternatives 3 and 4. An employment mix of greater than 50% industrial jobs is projected under all alternatives. A high proportion of industrial jobs are accessible without traditional four-year college degrees, and many industrial jobs remain unionized with high quality benefits.

Mitigation and Investment: Disadvantaged communities are disproportionately burdened by displacement. Given this, the City may be able to strengthen its anti-displacement efforts, and existing programs and enhancements are referenced.

The City's current plans and the Action Alternatives would provide improved transit, bicycle, pedestrian, and freight connections, as well as transportation demand management. However, city streets will remain congested during peak periods as growth continues to occur, and mitigation measures have been identified to improve particular corridors.

Topic-specific Summaries

Land and Shoreline Use: While shoreline and land use impacts are expected to be less than significant under all alternatives, some of the identified impacts could have equity and environmental justice considerations. Land use transition impacts would raise environmental justice concerns where residents of nonindustrial areas in or adjacent to the study area could be adversely affected by inadequate transitions at the edges of industrial areas. In areas of inadequate transitions, impacts from noise, odors and truck access and circulation associated with industrial land uses could affect communities of color and economically disadvantaged people. Impacts of increased building height, bulk and scale at transitions could also affect vulnerable populations. The neighborhoods of Georgetown, SODO, and South Park are vulnerable because there are land use transition impacts, and they have populations with higher levels of disadvantage. There is potential for new employees or residents in the rezoned

Fisherman's Terminal



areas to be vulnerable populations at a relatively higher rate. Adverse localized impacts on these community members could result from increased exposure to freight traffic and other challenges of working or living in the area. In general, it is expected that the proposal will have positive equity affects related to the employment mix, with greater levels of jobs having accessible education requirements and higher wages as noted above. While impacts on vulnerable communities are identified, a range of existing regulations and commitments and potential mitigation strategies will reduce the harmful impacts of the proposal related to land and shoreline use.

Housing: Key elements of housing displacement, supply, cost, health, and compatibility are addressed.

Displacement: There is limited existing housing in the study area, and therefore no potential for large amounts of displacement, although displacement risk is present for those living in existing homes. Displacement risk for smaller areas within these larger neighborhoods is hard to predict. Housing production trends show that, citywide, older single-family units are the most likely type of housing to be demolished to make way for new development. The industrial zoned areas in Ballard and South Park currently have very small proportions of the older single-family units most likely to be redeveloped. Some communities, and demographic groups, including low-income households, people of color, renters, seniors, and low and moderate-income families with children, are disproportionately burdened by displacement. Given this, the City may be able to strengthen its anti-displacement efforts.

Supply and Costs: Increases in supply can moderate home prices and rents so that housing is more affordable for households with lower incomes. However, the housing growth envisioned in the study area is not significant compared to city construction trends. The continued support for housing and the slight increases in housing envisioned in alternatives 3 and 4 <u>and the</u> <u>Preferred Alternative</u> will add to the housing supply and will allow some workers to live close to where they work. This can reduce the costs of commuting.

Housing and Health: The Action Alternatives limit new housing in industrial zones to caretakers' quarters and live/work studios and focus primarily on industrial uses. Alternatives 3 and 4 <u>and</u> <u>the Preferred Alternative</u> also add mixed-use housing opportunities near Georgetown/South Park. Given the health impacts of housing proximity to industrial areas, especially the Duwamish area, limiting the amount of housing in these areas has positive impacts on health equity.

Compatibility and Livability: Action Alternatives promote new zoning standards. UI zoning is intended to create thoughtful integration between the edges of these industrial areas and adjacent neighborhoods. UI zoning would seek to improve environmental health, walkability, and comfort in these areas. These changes tied to zoning are likely to ensure that the limited amount of housing allowed within the UI zone is accompanied by changes that add amenities to the area.

Transportation: The Action Alternatives—particularly alternatives 3 and 4<u>and the</u> Preferred Alternative—would result in more land use growth compared to Alternative 1 No Action particularly in the SODO/Stadium and South Park neighborhoods. With respect to transportation, this growth could provide both beneficial and adverse impacts to equity and environmental justice. Additional growth would bring increased traffic volumes, which in turn may bring impacts to the safety of people walking and biking, parking availability, and travel time delays to areas with high proportions of priority populations. At the same time, increased development could also bring improved infrastructure to neighborhoods with histories of longterm underinvestment. This is particularly the case for areas that would be rezoned as Industry & Innovation and Urban Industrial because those land use concepts would have development standards requiring frontage improvements such as sidewalks, pedestrian lighting, and street trees—all of which could be beneficial in progress toward more safe, connected, and accessible neighborhoods.



SODO Light Rail (Left) and Rail Tracks at Lander (Right)

Cultural & Recreational Resources

Screening Criteria: Adverse impacts to achieve a safe, connected, and accessible neighborhood (transit, housing, food access, parks, sidewalks, cultural hubs, etc.).

Summary of Impacts: Development has the potential to affect historic and cultural resources in historically marginalized neighborhoods. Added growth from the alternatives, particularly alternatives 3 and 4 <u>and the</u> <u>Preferred Alternative</u> could allow for more park demand and need in marginalized neighborhoods, which could prompt new park investments.

Mitigation and Investment: Applying state and federal standards, and engaging EEI populations in equitable development and redevelopment would limit impacts to historic and cultural resources. Regarding parks, the City could create linear parks and trails, increase frequency of maintenance to offset an increase in park usage, and build resilient parks. The City could explore improving transportation to and from parks to increase connectivity between parks. Community gardens (permitted on some rooftops in individual zones) could provide open space and urban agriculture.

Topic-specific Summaries

Historic, Archaeological & Cultural Resources: In the study areas, the alternatives have the potential to affect historic and cultural resources in historically marginalized neighborhoods. If impact minimization, or avoidance of impacts to historic, archaeological, and cultural resources is impossible, appropriate and meaningful mitigation should be developed in accordance with DAHP Mitigation Options and Documentation Standards and in coordination with the area's Tribes, the lead agency, and all other consulting parties. Equitable development and redevelopment should include the voices of the EEI populations to share in the decision-making process.

Open Space and Recreation: The Greater Duwamish MIC vicinity has higher levels of heat. Adding trees in streetscapes, private properties, and parklands can help



Duwamish Tribal Longhouse and Cultural Center



Kayaker on the North Shore of the Ship Canal

reduce the heat island effect Implementing a "pathway to equity" in the Seattle Parks and Recreation could address historical racial inequities in parks and open space. In Georgetown/ South Park, the neighborhoods have nearby parks, but the total acreage per capita is half the citywide average and there may be park congestion caused by added population. Meeting the City's level of service policy would mean adding parkland in appropriate areas. In the Study Area, most demand would be in Georgetown/South Park as well as the Ballard and SODO/ Stadium subareas.

Public Services & Utilities

Screening Criteria: Adverse impacts to achieve a safe, connected, and accessible neighborhood (transit, housing, food access, parks, sidewalks, cultural hubs, etc.).

Summary of Impacts: Additional growth could affect emergency vehicles response times including in underserved neighborhoods. Additional growth could add substantial new students at local schools including in Georgetown/South Park.

<u>Mitigation and Investment</u>: Compact growth, Water conservation, local power generation, and energy conservation measures are proposed.

Topic-specific Summaries

Public Services: Additional growth would increase traffic volumes which may in turn increase the response time of emergency vehicles in areas with high proportions of priority populations. However, increased development in areas with histories of long-term underinvestment could bring improved infrastructure to those neighborhoods. The increase in housing could generate students attending local schools in the Georgetown/South Park Subarea, particularly under alternatives 3 and 4 and the Preferred Alternative, which has a higher proportion of disadvantaged households. Ensuring access to schools with safe travel routes would help all local students in these areas.



Emergency Personnel at a Drill at Terminal 5

<u>Utilities:</u> Under all alternatives, minor impacts to utility services could occur during construction of individual development projects. All alternatives are likely to lead to utility improvements in the study area. There is no indication that the improvements are likely to cause adverse impacts to low income and other underserved populations in the study area as long as the utility improvements avoid displacement of these populations. Utility improvements could potentially benefit low income and other underserved populations in the study area, such as in portions of the SODO/Stadium and Georgetown/South Park subareas.

Equity and Environmental Justice Element	Natural and Biological Resources and Resiliency	Environmental Health and Compatibility	Working, Living, and Mobility	Cultural and Recreational Resources	Public Services and Utilities
Adverse impacts exacerbating residents' and workers' exposures to environmental hazards.	Potential exposure to environmental hazards (air quality, water quality) and risk of exposure to geologic hazards and sea level rise. Mitigation measures and investments can avoid impacts and improve conditions.	Temporary exposure to contamination and noise during construction. Longer- term exposure to light and glare from development, e.g., in South Park. Mitigation can address worker and resident safety and design standards can address light and glare.	New employees or residents could be exposed to environmental hazards. Mitigation measures address design and buffering of residential uses, addition of landscaping and tree canopy, implementation of sea level rise adaptation measures, and application of federal, state, and local laws regarding air quality, noise, hazardous materials handling, etc.	Residents, workers, and visitors may use parks and recreation facilities in the study area. Recreation areas are sensitive receptors for noise, and noise mitigation may be needed. Parks along shorelines may be affected by sea level rise and adaptation may be needed. Parks are potential locations for improvement of vegetation and canopy benefiting air quality.	Not applicable.
Adverse impacts to achieve a safe, connected, and accessible neighborhood (transit, housing, food access, parks, sidewalks, cultural hubs, etc.).	See above.	See above.	Growth can bring impacts of traffic and delays in the study area including in areas with disadvantaged populations, but increased development can improve multimodal investments to create safe, connected, and accessible neighborhoods.	Development may affect historic and cultural resources in historically marginalized neighborhoods. Applying state and federal standards, and engaging EEI populations in equitable development would limit impacts to resources. Added growth, particularly alternatives 3 and 4 <u>and</u> the Preferred Alternative, could increase park demand and need in marginalized neighborhoods, and could prompt new investments (parks, linear trails, community gardens, etc.).	Additional growth could affect emergency vehicles response times including in underserved neighborhoods. However, increased development in areas with histories of long- term underinvestment could bring improved infrastructure to those neighborhoods.

Exhibit 1.7-8 Equity and Environmental Justice Matrix of Topics

Equity and Environmental Justice Element	Natural and Biological Resources and Resiliency	Environmental Health and Compatibility	Working, Living, and Mobility	Cultural and Recreational Resources	Public Services and Utilities
Displacement risk of EEI Populations*	See above.	See above.	The risk of housing displacement is low due to the limited housing in the study area. The limited housing added under Action Alternatives could marginally assist with housing costs including rent and commuting. The Action Alternatives improve transitional standards for compatibility.	See above.	Not applicable.
Access to education or pathways out of poverty through jobs and careers	Not applicable.	Not applicable.	Increase in projected employment in the study area, with substantially higher quantities of new employment under alternatives 3 and 4. Proportion of industrial jobs are accessible without traditional four-year college degrees, and many industrial jobs remain unionized with high quality benefits. EEI populations could benefit from increased employment in industrial and nonindustrial sectors.	Not applicable.	Additional growth could add substantial new students at local schools including in Georgetown/South Park. Coordinated district capital and service planning should ensure capacity. Ensuring access to schools with safe travel routes would help all local students in these areas.

Note: Based on the Seattle Equity and Environment Agenda and Duwamish Valley Program & Action Plan.

* Equity & Environment Initiative (EEI) Populations: Communities of color, immigrants and refugees, people with low incomes and limited English-proficiency individuals. Youth from these communities are also a priority.

[This page is intentionally blank]



2.1 Introduction

2.1.1 Overview of the Proposal

Seattle's industrial and maritime policies are more than 35-years old and during that time, the trends and technologies impacting industrial and maritime users have experienced significant change. To reflect those changes as part of a comprehensive strategy to strengthen and grow Seattle's industrial and maritime sectors for the future, the City of Seattle is studying a proposal to update its industrial and maritime policies and industrial zoning. The proposal is informed by recommendations from community input, including an Industrial and Maritime Strategy Council, which resulted in an <u>Industrial and Maritime Strategy Report</u> that the City of Seattle released in June 2021 (**Appendix B**).

This Environmental Impact Statement (EIS) studies four five alternatives illustrating different potential futures for the city's industrially-zoned lands. The four five alternatives evaluate the effects of potential changes to Comprehensive Plan policies and changes to zoning over a 22-year time horizon (to 2044). The first alternative is a No Action alternative that is required by the State Environmental Policy Act (SEPA) and is a basis for comparison. The three four Action Alternatives (alternatives 2, 3, and 4 and the Preferred <u>Alternative</u>) all apply proposed "future of industry" land use concepts that are based on community input and intended to respond to issues, challenges, and opportunities for the maritime and industrial sectors and adjacent communities.

What is an Alternative?

Alternatives are different ways of achieving objectives that allow decisionmakers to compare the effects of different options. The No Action Alternative is based on current plans, policies, and regulations and is a benchmark against which other alternatives can be measured. Action alternatives can test a range of ideas, implications, and benefits. The Alternatives in the EIS will consider Comprehensive Plan policy amendments and different configurations for possible zoning changes and development standards to achieve the maritime and industrial land objectives.

The future of industry land use concepts consist of three proposed new industrial zones:

- Maritime Manufacturing and Logistics (MML)—This zone would focus on strengthening land use protections for core and legacy industrial and maritime areas to better prevent the encroachment of development that is incompatible with industrial and maritime uses. This zone is particularly applicable within Seattle's Manufacturing/Industrial Centers (MICs), near the shoreline or deep-water port, rail and freight infrastructure, and around existing clusters of industrial or maritime suppliers and services.
- Industry / Innovation (II)—This zone aims to encourage new development in multi-story buildings that accommodate industrial businesses mixed with other dense employment uses such as research, design, offices, and technology. By creating density bonuses for employment uses (i.e., office, R&D, etc.) if coupled with industrial uses in the same project,

this type of modern industrial development would support high-density employment near transit stations and near existing industrial-commercial areas.

 Urban Industrial (UI)—This zone is designed to foster increased employment and entrepreneurship opportunities with a vibrant mix of affordable, small-scale places for light industry, makers, and creative arts, as well as industry supporting ancillary retail or housing spaces to create better, integrated, and healthier transitions at the edges between industrial areas and neighboring urban villages, residential, and mixed-use areas.

To implement the future of industry land use concepts in each of the Action Alternatives the City of Seattle would:

- Amend the comprehensive plan to add new text policies describing the intent and vision for how these concepts would be applied, including land use, environment, and transportation;
- Amend the industrial zoning section of the land use code to create a new zone designations and corresponding development standards replacing the existing industrial zones;
- Apply new industrial zone classifications to industrial land; and
- Adopt new subarea plans for both the Ballard Interbay Northend and Greater Duwamish Manufacturing Industrial Centers (MICs).

However, each of the alternatives evaluated in this EIS pose different percentages of the future land use concepts in industrial and manufacturing lands for the purpose of strengthening and growing Seattle's industrial and maritime sectors in the future, as set out in **Section 2.4.1** of the EIS. The multi-faceted objectives of the proposal are listed in **Section 2.2** below.

The following is a brief summary of the <u>four five</u> alternatives, which are described further in **Section 2.4** below.

- Alternative 1—No Action: The SEPA-required alternative that would retain current Comprehensive Plan policies, development standards or zoning maps.
- Alternative 2—Future of Industry Limited: Alternative 2 retains current MIC boundaries. Alternative 2 would implement future of industry land use concepts with a greater emphasis on strengthening protections for core and legacy industrial and maritime activities. The proposed MML zone would cover approximately 90% of industrial lands. Application of the proposed II and UI zones would be limited in scope, covering approximately 10% of current industrial areas. II zoning would be focused on existing Industrial Commercial (IC) zones and areas within approximately 1/4 mile of light rail stations. UI zoning would be focused on existing Industrial Buffer (IB) zones and the existing Stadium Transition Area Overlay. There are no changes to housing allowances in Alternative 2.
- Alternative 3—Future of Industry Targeted: Alternative 3 would strengthen protections for core industrial uses in the MML zone on approximately 86% of industrial lands. It applies a mix of the proposed II and UI zones in targeted geographies covering 14% of industrial lands. Compared to Alternative 2, II zoning is expanded to include areas an estimated 1/2 mile from light rail stations, and UI zoning would be applied in additional areas in Ballard and the north shore of Lake Union. Alternative 3 creates limited flexibility for additional industry-supportive housing in UI zone that would result in an estimated 610 new homes in

industrial zones. Alternative 3 removes focused land in Georgetown / South Park from the MIC and converts it to a non-industrial mixed-use zone.

- Alternative 4—Future of Industry Expanded: Alternative 4 would also strengthen protections for core industrial uses in the MML zone on approximately 87% of industrial lands. Similar to Alternative 3, Alternative 4 would mainly apply II zoning in existing IC zones and within approximately a 1/2 mile from light rail stations, though with a greater expansion of the II zone in areas in Ballard and SODO. Compared to Alternative 3, the UI zone would be applied to a larger area in SODO, but to fewer areas in Ballard. This alternative includes additional flexibility for industry-supportive housing that could result in an estimated new 2,195 new homes in industrial zones. Just like Alternative 3, Alternative 4 removes focused land in Georgetown / South Park from MICs and convert it to a non-industrial mixed-use zone.
- Preferred Alternative—Future of Industry Balanced: The Preferred Alternative incorporates features of multiple Draft EIS Action Alternatives. It includes modifications to address comments on the Draft EIS and reduce impacts identified for Draft EIS alternatives. The Preferred Alternative would implement the proposed land use concepts, and strengthen policy protections for industrial lands in MICs, while affording some greater flexibility for lands outside of MICs.
 - <u>The MML zone would cover approximately 85% of industrial lands, while proposed II and UI zones would be targeted in scope and cover approximately 14% of current industrial areas. II zoning would be focused on existing Industrial Commercial (IC) zones inside of MICs and areas within approximately 1/2 mile of light rail stations. UI zoning would be focused on existing Industrial Buffer (IB) zones. Unlike alternatives 2, 3, and 4, the Preferred Alternative would retain existing IC zoning only in areas outside of MICs.
 </u>
 - <u>The Preferred Alternative would allow limited industry-supportive housing in the UI zone</u> as a conditional use subject to additional criteria to minimize potential conflicts. Additionally, new areas for housing in mixed use zones are added in the Preferred Alternative outside MICs (west Ballard, and Judkins Park). Overall, a lower amount of industry-supportive housing production would result compared to Draft EIS alternatives 4 within MICs in the UI zone (1,475 units). However, a greater amount of new unrestricted housing is projected outside of the MICs than any Draft EIS alternative (1,534 units). The combined growth of housing would be less than Alternative 4.
 - <u>Concepts to remove focused land from the MIC in Georgetown and South Park are</u> <u>carried forward. The Preferred Alternative includes a more nuanced zoning approach for</u> <u>a proposed mixed use zone in central Georgetown to preserve arts spaces and historic</u> <u>structures, and greater application of UI zoning around Georgetown to create more</u> <u>neighborhood cohesion.</u>
 - <u>The Preferred Alternative features a reduced total amount of job growth, most similar to</u> <u>Draft EIS Alternative 2. Projections are adjusted downward to reflect conditions in</u> <u>commercial/office occupancy post-COVID and timelines for new light rail construction.</u> <u>The adjusted projections acknowledge that it will likely take longer to achieve levels of</u> <u>employment growth.</u>

2.1.2 Study Area

Most industrial land in Seattle is located within two Manufacturing Industrial Centers (MIC): Seattle's Greater Duwamish Manufacturing and Industrial Center (Greater Duwamish MIC) and Ballard Interbay North Manufacturing Industrial Center (BINMIC), important as a freshwater harbor. Within the MICs, subareas are defined—Ballard, Interbay Dravus, Interbay Smith Cove, SODO/Stadium, and Georgetown/South Park. The Greater Duwamish MIC and BINMIC contain 12% of Seattle's total land area. Other industrially zoned land that is outside a MIC is included in the study area, most of which is on shorelines of Lake Union and by Judkins Park See **Exhibit 2.1-1**.



Exhibit 2.1-1 Study Areas

Source: BERK, 2021.

2.1.3 Objectives of the Proposal

The State Environmental Policy Act (SEPA) requires a statement of proposal objectives and the purpose and need to which the proposal is responding. Alternatives are different means of achieving the objectives.

The proposal would update Comprehensive Plan policies concerning industrial land and update the city's industrial zoning. The objectives behind this proposal are multi-faceted and seek to address the City's industrial and maritime sectors holistically. The objectives are informed by the recommendations of an Industrial and Maritime Strategy stakeholder process. Objectives are identified in four overlapping categories of people, place, and production and process. See **Exhibit 2.1-2**.

Exhibit 2.1-2 Objectives of the Proposal

Рео	ple
А.	Increase the quantity of living wage jobs generated from activity on Seattle's currently designated industrial lands.
В.	Improve equitable access to the living wage jobs from these lands by increasing the proportion of the jobs held by: racial minorities, women, and persons without traditional 4-year college diplomas.
C.	Improve environmental health for people who live or work in or near industrial areas—especially at transitions to residential areas or urban villages.
Pla	ce
D.	Provide long-term predictability to stakeholders that will support renewed investment in facilities, buildings, and infrastructure.
E.	Promote mutually reinforcing mixes of activities at the transitions between industrial areas and urban villages or residential neighborhoods.
F.	Support industrially compatible employment dense transit oriented development at existing and future high capacity transit stations.
G.	Increase access to workforce and affordable housing for employees in industrial maritime sectors, without creating land use conflicts that displace industrial uses.
Pro	duction
H.	Position Seattle's industrial areas to respond competitively to new industrial and manufacturing processes and practices.
I.	Ensure available and adequate locations for components of regional and statewide supply chains and regional economic clusters.
J.	Increase the amount and accessibility of space for prototyping, entrepreneurship, and business incubation.
К.	Strengthen economic resiliency with the capacity to produce products locally and ensure stable distribution networks.
Pro	cess
L.	Develop Comprehensive Plan policies based on the Industrial and Maritime Strategy.
М.	Develop a subarea plan for the MICs that supports VISION 2050, accommodates growth targets, and the Puget Sound Regional Council Regional Centers Framework for MICs.

Source: City of Seattle, 2021.

2.2 Planning Context & Outreach

2.2.1 Emerging Factors Affecting Seattle's MICs

MICs are regional designations and are defined in the City's Comprehensive Plan as home to the city's thriving industrial businesses. Like urban centers, they are important regional resources for retaining and attracting jobs and for maintaining a diversified economy. Seattle's manufacturing and maritime sectors generate middle-wage jobs that are cornerstones of a thriving and livable city. There are currently around 98,500 industrial jobs (2018) or about 15% of total jobs in the city—about two-thirds of these jobs are available with only a high school diploma, and over half of the jobs in the maritime sector are available to persons with no formal educational training. Average earnings per worker are over 70% of the Area Median Income (AMI) in the construction, aerospace/aviation, and logistics sectors, and a high number of jobs in logistics, maritime, and manufacturing sectors remain unionized and provide high quality benefits.

Since MICs were established in 1994 there have not been large<u>-</u>scale alterations to their geographic boundaries. Today, zoning within MICs must be one of four industrial zones in the Seattle Municipal Code (SMC). Those zones regulate the uses and activities that can take place in industrial areas, limiting them to prioritize manufacturing and industrial activities envisioned by the comprehensive plan. While manufacturing and maritime sectors today are strong, emerging factors affecting them include:

- Pressures to convert Industrial lands
- Emerging technologies and processes
- Unintended development
- Pending port, transportation, and new industrial building typology
- Environment and climate change
- Equity and accessibility

Pressures to Convert Industrial Lands

On a consistent basis, the City receives requests to remove parcels of land from a MIC designation from one of the industrial zones during the annual Comprehensive Plan amendment process. While growth rates in industrial rents were the highest in the world in 2017, average rental rates for commercial space are about three times higher than for industrial space. The requests amount to continual pressure to convert industrial land to other uses. This continues to create significant economic pressures to rezone industrial land for other uses.

Emerging Technologies & Processes

Definitions of Industrial uses in the Comprehensive Plan and in the Seattle Municipal Code may not reflect current standards of industrial activity. Some observers argue that new technologies and economic processes warrant reconsideration of definitions for what industrial and maritime use means, and/or reconsideration of the potential for compatible mixes of industrial uses with other activities.

Unintended Development

In recent years, some development in designated MICs was not intended to be allowed by zoning and is not compatible with the stated policy goals for industrial areas. For example, large retail stores do not complement the function of an industrial area, have no need to be in an industrial area, and often displace industrial uses.

Pending Port, Transportation, & New Industrial Building Typology

The City is experiencing several catalysts for further change in industrial areas, including:

- The Port of Seattle's plans to redevelop Terminal 46 to hold the world's largest cruise ships and the U.S. Coast Guard's proposed expansion of its Base Seattle onto portions of Terminal 46;
- Sound Transit's development of new light rail stations in Ballard, Dravus, Smith Cove (Interbay), and SODO. This EIS considers potential future light rail station locations being considered by Sound Transit in the West Seattle to Ballard Link Extension (WSBLE). Action alternatives are constructed to reflect what is known about WSBLE at this time. Since land use concepts address broad areas the action alternatives are anticipated to be responsive to multiple potential final stations locations or configurations.;
- The State's intentions for the sale and redevelopment of the armory site in Interbay and potentially the WOSCA site in the Duwamish MIC.; and
- New industrial development in non-traditional, vertical development.

Climate Change

Seattle's industrial areas that are undergoing economic change and infrastructure investment and its neighboring communities are also facing acute risk from rising sea levels, increased floods, and extreme heat.

Equity & Accessibility

Historically, unequal access to the career opportunities provided by maritime and other industrial sectors has been a barrier to people of color to share in the benefits of this activity. Providing entryways to these careers for Black, Indigenous, and people of color (BIPOC)

communities and women is one way that Seattle can advance its commitment to an equitable economy. Maintaining a strong industrial economy is a prerequisite to providing these opportunities, but other strategies including outreach to BIPOC youth and workforce training investments are key parts of the industry and maritime strategy.

2.2.2 Equity & Environmental Justice

The study area includes territories of indigenous tribes; Euro-American settlement and industrial development altered the natural character of this area and impacted tribal treaty rights. Since settlement the study area has had a growing industrial and maritime economy connected to the Puget Sound Region and West Coast.

Current conditions information indicates that the study area contains few housing units but is bordered by residential areas and nearby schools; the study area also contains parks that visitors use. These residents and users of the study area have a higher relative exposure to air emissions, noise, and light and glare. Some lands in the study area contain hazardous waste or cleanup sites. These environmental conditions also affect the large numbers of workers that come every day to the study area and then commute to homes either elsewhere in Seattle or in King County and beyond.

The economic, social and health conditions during COVID-19 illustrated the essential role of the maritime workforce and infrastructure to the city, regional and state economy.

This section describes some of the environmental justice principles and actions that are under consideration as the alternatives are reviewed.

Equity & Environment Agenda

The City of Seattle has committed to environmental justice for persons of color, low-income households, and others disparately affected by historic decisions on land uses and infrastructure that affect housing, health, and other aspects of quality of life. The City has created an Environmental Justice Committee. The body has developed an Equity and Environment Agenda with the following vision:

We are steadfast in our pursuit of Environmental Justice, redefining our environment as not just the natural environment, but also where we work, worship, play, learn and live. We believe in a world that respects communities' histories and cultures, and that uplifts self-determination and full participation. We know that communities of color are creative, resourceful and resilient, and deeply care about the environments in which they live. Given that, we believe in environmental solutions that connect to and create economic and educational opportunities so that all communities can thrive. To do this necessitates addressing past systemic injustice while creating proactive, transformational solutions for the future. The Equity and Environment Agenda is also based on the following principles:

- Community Driven Strategies: We believe in community self-determination, influence and leadership. We know that communities are resilient and resourceful, and that tapping into their own collective cultural cornerstones of environmental sustainability is key to ownership of initiatives and other efforts, as well as reducing invisibility.
- The Influence and Decision-Making of Those Most Affected: We believe that communities who are deeply affected by environmental issues should be highly involved throughout decision-making processes in meaningful and culturally appropriate ways.
- Strong Accountability: We believe that affected communities deserve strong, accountable, transparent, accessible, and culturally appropriate solutions that include ongoing oversight of government and other entities to address the negative impacts they have experiences.
- Solutions That Recognize Complexity and Interdependence: We believe in doing no harm, here or anywhere. We recognize that all places and people are interconnected, and commit to an approach of collective liberation, which recognizes that the liberation of each person is the liberation of all people.

Chapter 1, Section 1.7.15, of this Draft EIS addresses findings of the alternatives and relationship to environmental justice and equity. **Chapter 3, Section 3.8** addressing land use includes an overview of past land use policies and other actions that had inequitable outcomes.

Duwamish Valley Program & Action Plan

The City's <u>Duwamish Valley Program</u> worked with the Duwamish Valley Action Team (DAT) to develop the Duwamish Valley Action Plan to advance environmental justice and equitable development.

The Action Plan promotes racial equity outcomes through mid-term actions:

- Healthy Communities
- Thriving Neighborhoods
- Prosperity in Place
- Employment and Economic Opportunity
- Equitable Access to City Resources, Accountability, and Decision-making
- Community Leadership and Capacity Building

Each outcome is defined in **Exhibit 2.2-1**. Detailed actions for each outcome are in the Action Plan.

Exhibit 2.2-1 Duwamish Valley Action Plan Racial Equity Outcomes

RACIAL EQUITY OUTCOMES

The Duwamish Valley Action Plan *is advancing racial equity through mid-term actions. Find these icons throughout the* Action Plan.



1. Healthy Communities

Reduce health disparities and cumulative impacts present in the Duwamish Valley related to air and water quality, soil contamination, noise pollution, access to healthy food, and climate change adaptation that disproportionately affect Duwamish Valley residents and workers, including communities of color, immigrants, refugees, Native peoples, people with low incomes, youth, and limited English proficiency individuals.



2. Thriving Neighborhoods

A safe, connected, and accessible Duwamish Valley, with a focus on South Park and Georgetown, with amenities and physical improvements that benefit Duwamish Valley residents and workers, including communities of color, immigrants, refugees, Native peoples, people with low incomes, youth, limited English proficiency individuals, women- and minority-owned businesses, and people of color-led organizations.



3. Prosperity in Place

City policies and investments in the Duwamish Valley proactively prevent displacement risk so Duwamish Valley residents and workers, including communities of color, immigrants, refugees, Native peoples, people with low incomes, youth, and limited English proficiency individuals, women- and minority-owned businesses, and people of color-led organizations, enjoy a robust life and prosper in place.



4. Employment and Economic Opportunity

Duwamish Valley residents and workers, including communities of color, immigrants, refugees, Native peoples, people with low incomes, youth, limited English proficiency individuals, womenand minority-owned businesses, and people of color-led organizations, have economic mobility and opportunity through access to education, training, funding, and support programs, as well as pathways out of poverty through jobs and careers related to environmental policy and program and project development.



5. Equitable Access to City Resources, Accountability, and Decision-making

Duwamish Valley residents and workers, including communities of color, immigrants, refugees, Native peoples, people with low incomes, youth, limited English proficiency individuals, and women and people of color-owned businesses, meaningfully influence the design and participate in decision-making processes regarding City policies, programs, and services benefitting/affecting the Duwamish Valley. City decision-makers are responsive and accountable to Duwamish Valley residents and workers.



6. Community Leadership and Capacity Building

City programs, projects, investments, and engagement strategies are led, centered, and support the diverse cultures, stories, and experiences of Duwamish Valley residents and workers; a specific focus will be given to communities of color, immigrants, refugees, Native peoples, people with low incomes, youth, limited English proficiency individuals, and women and people of color-owned businesses. City policies and programs in the Duwamish Valley invest directly in leadership development, capacity building, and community-led solutions.

Source: Duwamish Valley Action Plan, 2018.
2.2.3 Mayor's Industrial & Maritime Strategy

In 2019 Mayor Durkan convened an Industrial and Maritime Strategy Advisory Council to chart a blueprint for the future of industrial land in Seattle with a focus on providing equitable access to high-quality, family-wage jobs and entrepreneurship opportunities. The Advisory Council included representation from citywide stakeholders and stakeholders from four neighborhood subareas. Stakeholders represented a diverse range of interests including maritime and industrial businesses, labor, residents of adjacent neighborhoods, developers, and industry groups.

- The Citywide Advisory Council consisted of representatives from industry, neighborhoods, and labor from across the city. It met more than ten times over a year and a half and included various phases and levels of dialogue. At each stage, council meetings were supplemented with individual outreach and dialogue between members of the strategy council, city staff, and the facilitator.
- The four neighborhood-based advisory councils consisted of representatives from subareas within and adjacent to Manufacturing Industrial Centers. Neighborhood Advisory Councils were convened for Ballard, Georgetown and South Park, Interbay, and SODO. Top issues and vision statements from each subarea were distilled to key themes. Neighborhood Advisory Council members were also attended and provided input at the full Citywide meetings.
- In parallel with the advisory councils, City staff worked with youth serving organizations to design and conduct engagement specifically targeted to **BIPOC youth**. This engagement resulted in direct dialogue, and a pre- and post-survey with over a hundred BIPOC youth to learn about their experiences accessing education, training, or employment opportunities in industrial maritime sectors.

In May 2021 the Advisory Council recommended **11 broad strategy statements to guide future actions** to support the maritime and industrial sectors, and advance equitable access to family-wage employment, particularly for BIPOC youth. See **Exhibit 2.2-2**.

The key land use recommendations of the stakeholders informed the EIS alternatives. Some of the strategies could be mitigation measures for impacts that are identified. Other strategies from the process that are not related to land use would be implemented through other City actions outside of the proposal studied in this EIS.

Exhibit 2.2-2 Industrial and Maritime Strategy Stakeholder Recommendations

Investment Strategies

1. Workforce Investments to Support Access to Opportunity for BIPOC, Youth, and Women: Create, expand, and support initiatives that increase access to opportunity and economic prosperity for Black, Indigenous, and People of Color, youth, and women through manufacturing, maritime, and logistics careers.

2. Public Safety Partnership to Support Maritime and Industrial Areas: Work closely with local business and community organizations to develop and implement a proactive public safety response to elevated levels of crime within maritime and industrial lands.

3. Transportation Priorities to Improve the Movement of People and Goods: Improve the movement of people and goods and make transit and freight networks work for industrial and maritime users with better service and facilities; improved last mile connections for active transportation, transit, and freight, including large truck access to shoreline and railroad uses; and advocating for a tunnel alignment for Ballard and Interbay future light rail.

4. Environmental Justice and Climate Action: Address environmental inequities and protect industrial adjacent communities from environmental harms, transition to a climate pollution free freight network, and prepare for a changing climate.

Land Use Strategies

5. Stronger Protections for Industrially Zoned Land: Strengthen protections for industrially zoned lands within Seattle by establishing higher thresholds to remove industrial land designations and closing loopholes that have allowed significant non-industrial development within industrially zoned lands.

6. High Density Industrial Development: Encourage modern industrial development that supports high density employment near transit stations and near existing industrial-commercial areas by creating density bonuses for employment uses (i.e., office, R&D, etc.) if coupled with industrial uses in the same project.

7. Healthy Transitional Areas near Urban Villages: Foster increased employment and entrepreneurship opportunities with a vibrant mix of affordable, small-scale places for light industry, makers, and creative arts, as well as industry supporting ancillary retail.

8. No New Residential Uses: No new residential uses on industrial and maritime lands. Limited adjustments to existing allowances in transitional zones to support industry and arts entrepreneurship opportunities. Any limited adjustments to existing allowances in transitional zones would be determined after additional study of potential impacts, including an Environmental Impact Statement (EIS).

9. Georgetown and South Park Neighborhood Goals: Remove a few small, focused locations from industrial zoning in Georgetown and South Park and convert them to mixed-use zoning to achieve neighborhood goals.

Action Strategies

10. Master Planning for WOSCA and Armory Sites: Recognizing the time limitations of this process and the specialized nature of these sites, partner with agencies of the State of Washington, Department of Transportation (WOSCA), and Department of Commerce (Armory), or future owners on a master planning process for industrial redevelopment specifically designed for each site based on the guiding principles of this workgroup.

11. Ongoing Stewardship Entities to Champion this Vision: Identify and grow ongoing stewardship entities with a complete range of stakeholders to champion the vision of the Industrial and Maritime Strategy, ensure its long-term implementation, and develop appropriate assessment metrics to help guide future policy decisions. In different neighborhoods, this could be an existing organization with a modified charter and/or a new organization.

Source: Industrial and Maritime Strategy Council Recommendations, May 2021.

2.3 SEPA Process

2.3.1 Environmental Review Process

Under SEPA agencies conduct environmental review of actions that could affect the environment. For actions that have the potential for significant impacts, preparation of an EIS is required. An EIS is a useful tool that provides detailed information to the public, agencies, tribes, and City decision-makers about the environmental effects of a plan or project before a decision is made. As described in Chapter 1, this document is a non-project EIS that analyzes the proposals and alternatives broadly across the study area. (WAC 197-11-442)

The EIS process involves the following steps: (1) scoping the contents of the EIS with agencies, tribes, and the public; (2) preparing a draft EIS with a comment period; (3) responding to comments and developing a preferred alternative; and (4) developing legislation. With the issuance of the Draft EIS, the EIS process is in phase 2. See **Exhibit 2.3-1**.



Source: BERK, 202<u>2</u>4

2.3.2 Public Comment Opportunities

Scoping

The scoping process is intended to identify the range of potential significant impacts on the built and natural environment that should be considered and evaluated in the EIS. The City issued a Scoping Notice on July 8, 2021 with a 30-day public comment period that ran through August 9, 2021 and was extended on request to August 23, 2021. Virtual scoping meetings were held during the comment period at 9:00 a.m. on July 21 and 6:00 p.m. on July 26, 2021. The City also published an <u>information website and online survey</u> as part of scoping.

The input received during the scoping period included:

- Written Comments: 105 letters and emails by 103 commenters
- Survey: 46 participants
- Public meeting participants: 7 participants

See **Appendix A** for the scoping report.

As part of scoping, the City identified a range of topics to explore in the EIS:

- Natural and Biological Resources and Resiliency: Soils/Geology, Air Quality/Greenhouse Gas, Water Resources, Plants and Animals
- Environmental Health and Compatibility: Contamination, Noise, Light and Glare
- Working, Living, and Mobility: Land and Shoreline Use, Housing, and Transportation
- Cultural and Recreational Resources: Historic, Archaeological & Cultural Resources, Open Space and Recreation
- Public Services and Utilities: Police, Fire, Schools, Libraries, Wastewater, Stormwater, and Power

Scoping comments indicated that air quality/greenhouse gas, contamination, transportation, and land and shoreline use were most important to address in the EIS. Commenters also gave input on alternatives to be studied, typically by indicating which of the scoping alternatives fit their views of the area or properties, or requesting adjustments. In response to the scoping comments one alternative was modified to include an evaluation of potentially increasing the size of use limit on indoor recreation facilities from 10,000 square feet to 50,000 square feet. A full response to scoping comments can be found in the Scoping Report.

This Final EIS includes comments gathered during the Draft EIS Comment period described further below. See **Chapter 4 Comments & Responses**.

Draft <u>& Final</u>EIS

This-The Draft EIS identified identifies environmental conditions, potential impacts, and measures to reduce or mitigate any unavoidable adverse impacts that could result from an update to policies and zoning for Seattle's maritime and industrial sectors. The Draft EIS alternatives and topics were developed based on a review of scoping comments and prior Industrial and Maritime Strategy engagement results.

Public and agency comments are were invited on th<u>e</u>is Draft EIS. Written and verbal comments are were invited during the 45-day public comment period (December 16, 2021 to January 31, 2022) following issuance of th<u>e</u>is Draft EIS. <u>The City extended the comment period to March 2, 2022 to allow more time for review.</u>

The City will hold futureheld public engagement events during or following the 45-day comment period to help refine its preferred alternative. In addition, the City conducted a series of meetings with the South Park and Georgetown community members in neighborhood

<u>locations and included comments from these communities through April 15, 2022.</u> Public comments will beare considered and addressed in the Final EIS. Please see the Fact Sheet at the beginning of this <u>Draft-Final</u> EIS for the dates of the public comment period, <u>as extended</u>, and public meeting. Meetings and comment periods regarding the proposals are described on the City's project webpage: <u>Industrial and Maritime Strategy—OPCD | seattle.gov</u>.

Final EIS & Proposed Legislation

A Final EIS will bewas issued in 2022 and will-includes responses to public comments received during the Draft EIS comment period. <u>Changes to the Draft EIS are shown in strikeout and underline. The Final EIS also studies a Preferred Alternative that responds to the comments.</u>

Following the EIS process, the City will develop specific policy and zoning proposals that will be the subject of public meetings and public hearings by the City Council.

2.4 **Proposed Action & Alternatives**

The proposal considers Comprehensive Plan policy amendments and changes to zoning and development standards that could help meet the objectives defined in **Section 2.1.3**. The EIS includes three four future of industry alternatives (alternatives 2, 3, and 4 and the Preferred Alternative) that would make different geographic combinations of zoning changes and degrees of change to development standards in industrial zones. A No Action Alternative with no changes to policies or zoning is also considered. The EIS addresses land use compatibility, and consistency with City and State plans and regulations.

2.4.1 Land Use Concepts

The future of industry alternatives (alternatives 2, 3, and 4<u>and the Preferred Alternative</u>) would apply proposed new land use concepts that are based on community input and intended to respond to issues, challenges, and opportunities for the maritime and industrial sectors and adjacent communities.

Three proposed land use concepts are integrated to different degrees in the future of industry alternatives and include:

- Maritime, Manufacturing, and Logistics (MML)
- Industry and Innovation (II)
- Urban Industrial (UI)

A description of concept is provided below and following that a full description of each alternative and how it assimilates the land use concepts.

Maritime, Manufacturing, and Logistics (MML)

The Maritime, Manufacturing, and Logistics (MML) land use concept would intend to strengthen established economic clusters and expand equitable access to jobs. <u>There would be a high</u> <u>likelihood that a substantial proportion of jobs in MML would be union represented.</u> Seattle's industrial areas host valuable economic clusters including fishing, logistics, maritime, aerospace, brewing and distilling, and others that depend on access to water or other irreplaceable supporting infrastructure. MML would be applied in locations near such infrastructure and would strengthen the policy and zoning protections for maritime and industrial uses. See **Exhibit 2.4-1**.

Exhibit 2.4-1 Maritime Manufacturing and Logistics Proposed Land Use Concept



Challenges Addressed

- Market pressure for conversion away from industrial land.
- Vulnerabilities due to the interdependence of business within clusters.
- A pattern of "one off" zoning decisions that have removed industrial land.
- Encroachment of non-industrial uses in industrial zones.

Features/Development Standards

- Strictly limit allowable uses to industrial, manufacturing, maritime and similar uses.
- Do not allow new residential uses.
- Strict maximum size of use limits on non-industrial uses such as retail, office, and restaurants.

Industry and Innovation (II)

The Industry and Innovation (II) land use concept would intend to support economic innovation and capitalize on emerging opportunities including expanded or new light rail stations in industrial areas. It would intend to support emerging formats for industrial activity that are more design and research oriented than traditional industrial uses. It would intend to introduce nodes of high-density employment and multi-modal access near transit. Industry and Innovation would also intend to encourage new investment in high quality industrial space. See Exhibit 2.4-2.



Challenges Addressed

Exhibit 2.4-2

- Industrial zoning hasn't been updated to reflect contemporary industrial methods.
- Lack of new investment (buildings & infrastructure) in industrial areas.
- Integration of high-capacity transit in industrial areas (ST3).
- High rent for office and tech uses make it difficult for industrial businesses to find space affordable to them.
- Lower density of jobs in distribution / warehouse uses.

Features/Development Standards

- An incentive structure allowing some non-industrial office or technology uses if a new bona-fide industrial space is included in the same development. Industrial uses would be likely to locate on the ground floor and/or second floor.
- A substantial increase in allowed floor area and height limits compared to existing industrial zones that would allow dense multi-story buildings.
- Minimum construction standards for bona-fide industrial space such as freight elevators, minimum clear ceiling heights, and load-bearing floors.
- Standards for pedestrian and cyclist-oriented frontage improvements.
- Vehicle parking maximums and strong commute trip reduction program requirements.

Urban Industrial (UI)

The Urban Industrial (UI) land use concept would intend to foster vibrant districts that support a mix of local manufacturing, production, arts, and a sense of place. Urban Industrial would be in areas adjacent to Seattle's designated urban villages. UI would intend to create thoughtful integration between the edges of Seattle's MICs and adjacent neighborhoods. It would seek to improve environmental health, walkability, and comfort in these areas. The UI concept would seek to leverage the industrial aesthetic, including adaptive reuse of buildings. In some alternatives, UI could allow a limited amount of new industry-supportive housing. See **Exhibit 2.4-3**.



Exhibit 2.4-3 Urban Industrial Proposed Land Use Concept

Challenges Addressed

Features/Development Standards

- Environmental health impacts that affect residents near industrial areas.
- Uncomfortable conditions for pedestrians, cyclists, and transit riders.
- Strong demand for worker housing near jobs.
- Lack of small or affordable space for makers, creatives, and artists.
- Strict maximum size of use limits for stand-alone non-industrial uses.
 - Flexibility for larger size of use for retail or office space that is combined with a production or making use on-site.
 - A moderate increase in allowed floor area compared to existing industrial zones.
 - Development standards such as setbacks and landscaping that are more urban in nature, compared to the existing industrial buffer zones.
- Standards for pedestrian and cyclist-oriented frontage improvements.
- Expanded allowances for limited industry-supportive housing such as caretakers' quarters and maker studios (alternatives 3 and 4 only).

Comprehensive Plan Policy Amendments

The Action Alternatives include new goals and policies relating to the industrial and maritime sectors that would be adopted into the City's Comprehensive Plan. The proposed amendments would establish a new land use framework to implement the concepts discussed above, and new policies concerning transition to clean fuels.

Below is a summary for how the new policies would be integrated into the existing Comprehensive Plan. Specific draft goal and policy language can be found in **Appendix D**.

- Add two new land use **Goals** in the industrial areas section, in addition to existing Land Use Goal 10:
 - Support employment-dense activities and emerging industries that require greater flexibility in the range of on-site uses and activities.
 - Develop transitions between industrial areas and adjacent neighborhoods that support healthy communities, reduce adverse environmental impacts, and minimize land use conflicts.
- Introduce new land use **Policies** that would support implementation of the new goals.
 Policy amendments would include a new land use framework for the MML, II, and UI zones, establishing their intent and purpose and locational guidance.
- Introduce a new policy to limit changes in MIC boundaries to major updates of the Comprehensive Plan or following a comprehensive city-led study.
- Establish the city's intent to work with owners or future owners of the WOSCA and Interbay Armory sites on a master planning process for future reuse according to the goals and policies for MICs.
- Introduce new or strengthened policies into chapters of the Comprehensive Plan that may include the Transportation, Environment, or Container Port elements encouraging transitions to clean fuels and decarbonization of industrial and maritime activities.

Manufacturing Industrial Center Subarea Plan

The Puget Sound Regional Council's VISION 2050 and the Regional Centers Framework calls for jurisdictions to adopt subarea plans for regional centers. The City of Seattle anticipates developing a subarea plan for the two MICs.

The subarea plans should provide or address:

- A Center Plan Concept/Vision and be the product of Regional Collaboration
- Demonstrate Environmental Protection, Climate Change Adaptation and Mitigation, and Vulnerable Community Protection
- Center Size and Boundaries and Land Use / Development Patterns
 - Industrial Employment Centers should have at least 10,000 existing jobs and plan for at least 20,000 jobs.

- Regional manufacturing/industrial centers must retain a minimum 50% industrial employment.
- The plan should include policies and identify programs that retain at least 75% of industrially zoned land for core industrial uses (e.g., manufacturing, transportation, warehousing, and freight).
- Economy and Market Potential
- Multimodal and Intermodal Transportation
- Public Services
- Innovation, Engagement, and Racial Equity

More information and evaluation is included in **Chapter 3**, **Section 3.8 Land & Shoreline Use** addressing the relationship of the alternatives to plans and policies.

2.4.2 Regulatory Concepts

Concurrent with implementation of the proposed zones, the City would clarify uses that are "industrial" or "non-industrial". Land uses would still be categorized in specific use categories similar to the existing SMC, but with some consolidation and simplification of terms. Additionally, the City would provide guidance or code language to identify specific use categories that would fall into a broader identification as industrial or non-industrial for the purposes of implementing intended zoning tools. In the Preferred Alternative, the City would create a new use definition for "Information Computer Technology (ICT)" and ICT would be given special consideration for occupying light industrial space in the Industry and Innovation zone.

The development standards in **Exhibit 2.4-4** are general, describe the overall intent of the zone and how it would work to a level of detail sufficient for assessing environmental impacts of the proposal. Specific code language would be drafted at the time of a future legislative proposal. Minor modifications or adjustments are expected and would be similar to the evaluation of alternatives in this EIS. Additional detail on proposed development standards including identification of industrial and non-industrial uses is provided in the development standards in **Appendix G**.

Development	Maritime Manufacturing	Industry and	Urban
Standard	and Logistics (MML)	Innovation (II)	Industrial (UI)
Locational Criteria	 Within a M/IC Large parcel sizes Proximate to water and port facilities Proximate to rail or other freight infrastructure Buffered from urban villages and residential zones 	 Within ¼–½ mile walkshed of an existing or planned high capacity transit station Within a MI/C or land previously in an industrial zone outside a MI/C. 	 Within a designated M/IC, or an area with existing industrial/manufacturing/ma ritime uses Proximate to an urban village, or an existing agglomeration of residential uses

Exhibit 2.4-4 Development Standards by Land Use Concept

Development Standard	Maritime Manufacturing and Logistics (MML)	Industry and Innovation (II)	Urban Industrial (UI)
Height Limit	None	<u>85125</u> –160 feet (with exemptions for industrial equipment, antennas etc.)	Variable with tiers at 45', 60', and 75' <u>, and 85' in the STAOD</u>
Floor Area Ratio (FAR) The FAR limit is for all uses in total in a development (whether office, manufacturing etc.)	2.5 Maximum FAR total. 0.4 Maximum for non- industrial uses.	 Base and Bonus Limits: Development of floor area up to the base amount must be built and dedicated for industrial uses. Development exceeding the base is allowed through a ratio whereby 3 square feet of non-industrial use space may be built for each additional square foot of dedicated industrial space that is built. There is a total Maximum FAR limit on all development. Mixed Development with Bonus: 3 (6 Preferred Alt.) sq. ft. of bonus floor area for non- industrial space for each 1 sq. ft. of industrial space above a base FAR 0.5 of industrial development. Total maximum FAR Limit: 4-6 (depends on location) Industrial-only development: Development that only include industrial uses with no bonus development have a max FAR of 2.5. Configuration: Industrial development must be in the same building (i.e., first two floors), or in a separate building on the same site as bonus development or another site in the same MIC in the Preferred Alt. A close-to- maximum development would be about 1/3 industrial, and 2/3 non-industrial (1/6, and 5/6 Preferred Alt.). 	3.0 for 45' heights; 4.0 for 60' heights, and 4.5 for 75' height, and 85' in the STAOD
Permitted Principal Uses The list is a general summary to describe	Industrial Uses Permitted outright with no maximum size of use limits or additional restrictions.	Industrial Uses—Base Same permitted as for the MML zone <u>, except ICT allowed</u> in the Preferred Alternative- Non-Industrial Uses—Bonus	Industrial Uses—Base Permitted outright with no maximum size of use limits or additional restrictions, but the

Development Standard	Maritime Manufacturing and Logistics (MML)	Industry and Innovation (II)	Urban Industrial (UI)
the overall intent and is not exhaustive.	A broad range of heavy and light Manufacturing uses. A broad range of warehousing / distribution, marine and logistics Transportation uses A broad range of Utility uses Outdoor Storage and Warehouse Uses (but mini- Storage Warehouses are prohibited) Laboratory, and research and development with physical processes Food processing and craft work Automotive uses Non-Industrial Uses Permitted as a principal use only when subject to strict maximum size of use limits and FAR sub-limit. Commercial sales and services Office Institutional Uses Lodging Entertainment Uses (#) Information Computer Technology (ICT)	Only allowed as bonus development. (2– <u>6</u> 3 sq. ft. allowed per each additional sq. ft. of industrial use space above the base FAR of 0.5 of industrial use space.)	heaviest / most impactful industrial uses are not allowed. Light Manufacturing uses. Warehousing / distribution, marine and logistics Transportation uses Some lower-impact utility uses Outdoor Storage and Warehouse Uses (but mini- Storage Warehouses are prohibited) Laboratory, and research and development with physical processes Food processing and craft work Automotive uses Non-Industrial Uses Permitted subject to strict maximum size of use limits. (Note—greater flexibility for ancillary uses below). Commercial sales and services Office Institutional Uses Entertainment Uses (1) Information Computer Technology (ICT)
Prohibited Uses <i>This is not a</i> <i>comprehensive list.</i>	Mini storage Principal use parking	Mini storage Principal use parking	Mini storage Principal use parking Heavy manufacturing Some intensive utility uses Some intensive transportation uses
Ancillary Uses Ancillary uses are functions associated with or related to the principal permitted use. Rules concerning ancillary uses would be clarified.	Non-Industrial activities that are ancillary to an Industrial Use are limited to 30% of the floor area or activity area of the use.	Non-Industrial activities that are ancillary to an Industrial Use are limited to 30% of the floor area or activity area of the use, or else the use would be classified as Non-Industrial / Bonus development.	Non-Industrial activities that are ancillary to an Industrial Use may occupy up to 80% of the floor area, with 20% of floor area in the industrial use. The intent is to allow large spaces for activities such as tasting rooms, retail and office when associated with a bona- fide on-site or nearby industrial use.

Development Standard	Maritime Manufacturing and Logistics (MML)	Industry and Innovation (II)	Urban Industrial (UI)
Maximum Size of Use Limits Limits pressure from non-industrial uses, and provides services intended to support workforce in the same building or general area as a principally allowed industrial uses.	 10,000 sq. ft. Major durables sales, service Office Lodging (#) Medical services Entertainment (#) 7.500 sq. ft. General retail sales and service 3.000 sq. ft. Restaurants / Bars 	None. Principal non-industrial uses are allowed without a size limit, subject to the incentive bonus system.	Maximum size of use limits are for stand-alone principal non- industrial uses. Note increased flexibility for ancillary uses, which could allow larger-sized spaces if combined with an industrial use. 25.000 sq. ft. 25.000 sq. ft. Lodging Medical services Entertainment 15.000 sq. ft. Major durables sales, service Office 7.500 sq. ft. General retail sales and service 3.000 sq. ft. Restaurants / Bars
Residential Uses	No change to existing, narrow allowances for caretakers' quarters (1 per business); and artist/studio housing (existing structures only, 800 sq. ft. max.)	No change to existing, narrow allowances for caretakers' quarters (1 per business); and artist/studio housing (existing structures only, 800 sq. ft. max.)	Alternatives 3: increased allowance for industry supportive housing: Up to 2 caretakers'/workers' quarters per on-site industrial business. Artist/studio/maker housing allowed in new buildings, no max. unit size. Maximum density of 25 dwelling units / acre. Residential may not exceed 40% total floor area. Alternatives 4: increased allowance for industry supportive housing: Up to 3 caretakers'/workers' quarters per on-site industrial business. Artist/studio/maker housing allowed in new buildings, no max. unit size. Maximum density of 50 dwelling units / acre. Residential may not exceed 60% total floor area.

Development Standard	Maritime Manufacturing and Logistics (MML)	Industry and Innovation (II)	Urban Industrial (UI)
			Additional conditions apply. (See Housing and Land & Shoreline Use sections). <u>Preferred:</u> By conditional use permit with criteria (See development standards appendix). Density limit same as Alt. 4.
Parking Requirements	No Minimum Parking	No minimum parking Maximum parking: 1 per 1,000 sq. ft. (Parking maximum is provided to minimize SOV trips. Other Transportation Demand Management requirements may be explored to minimize SOV trips.)	No minimum
Setbacks			If abutting a residential zone 10' ground level setback from abutting property line. If abutting a residential zone, an additional 5' upper-level setbacks at 30' of building height, and an additional 5' building setback for each additional 10' of building height above 30'.
Frontage and Landscaping and Design Requirements	Street improvements No design review required	Multi-modal frontage improvements (sidewalks, pedestrian lighting, street trees etc.) No design review required	Multi-modal frontage improvements (sidewalks, pedestrian lighting, street trees etc.) <u>See development</u> <u>standards Appendix G.</u> Green Factor of 0.2 required No design review required
Indoor Sports and Recreation (An entertainment use)	Alt. 4 only Increase max size of use for indoor sports and recreation uses to 50,000 sq. ft. subject to locational criteria near edges of MIC, and away from shorelines.	Alt. 4 only Increase max size of use for indoor sports and recreation uses to 50,000 sq. ft. subject to locational criteria near edges of MIC, and away from shorelines.	Alt. 4 and Preferred Alternative only Increase max size of use for indoor sports and recreation uses to 50,000 sq. ft. subject to locational criteria near edges of MIC, and away from shorelines.
Stadium Transition Area Overlay District STAOD would be retained, and unique allowances and requirements would	Not Applicable	Not Applicable	Amend STAOD so lodging is a permitted use, and no design review is required. Increased maximum size of use limits:

Development Standard	Maritime Manufacturing and Logistics (MML)	Industry and Innovation (II)	Urban Industrial (UI)
modify the underlying UI zone in that area in action alts. Including changes from existing STAOD standards.			Office: 75,000 Restaurants/bars: No Limit Lodging: 75,000 General retail sales: 20,000 Maximum size of use limits do not apply if 0.4 FAR or more industrial space is provided on site. For other standards including height, see development standards Appendix G.
Non-Conforming Uses and Structures		Existing single use non- industrial structures such as offices rezoned into the II zone shall be considered an allowed use and not classified as non- conformingAdditional flexibility for non-conforming uses added for all zones in the Industrial land use code chapter. See development standards Appendix G.	

Source: City of Seattle, 202<u>2</u>1.

2.4.3 Alternative 1—No Action

The No Action Alternative is required by SEPA. No change to current Comprehensive Plan policies, development standards, or zoning maps are included under this alternative. The existing zone classifications established in 1986—the Industrial General (IG1 and IG2) zones, the Industrial Commercial (IC) zone, and the Industrial Buffer (IB) zone—would remain. IG is the core industrial zone that prioritizes industrial and maritime uses and covers most of the MICs. IC allows for a mix of industrial and commercial activities, but in recent years has been developed primarily with office and commercial uses. IB offers development standards intended to buffer industrial uses from adjacent neighborhoods and includes a focus on setbacks, limited heights, and landscaping. See **Exhibit 2.4-6**.

The No Action Alternative retains the following:

- No change to IG zones that cover 90% of industrially zoned areas.
- No change to IC zone that cover 5% of industrially zoned areas.
- No change to IB zone that cover 5% of industrially zoned areas.
- Residential uses are prohibited with the exception of one caretaker quarters per industrial business, artist studio housing in existing structures, and housing that predates industrial zoning.

See **Exhibit 2.4-5** with acres and percent of zones.

Exhibit 2.4-5 Alternative 1—No Action Zoning Districts (Acres)

Zoning Districts	Acres	Share
Industrial General (IG1/IG2)	6,273	90.4%
Industrial Buffer (IB)	316	4.6%
Industrial Commercial (IC)	347	5%
Total	6,936	100%



Exhibit 2.4-6 Alternative 1—No Action Zoning Map

The City of Seattle will be planning for total citywide job growth of 169,500 jobs over the 20-year planning horizon <u>of the One Seattle Comprehensive Plan major update</u>. Employment growth of 23,500 projected under Alternative 1 in the study area would represent about 14% of total citywide job growth. The study area contains the MICs and additional industrial zoned areas outside of MICs. The 14% share of total citywide job growth under Alternative 1 is an increase to the share of job growth planned for industrial areas during the previous Seattle 2035 20-year planning horizon, which estimated 8% of the city's job growth in MICs (and not including industrial zoned lands outside of MICs).

Current jobs are majority industrial (55%). The total number of jobs is expected to increase by 23,500 with just over half of that industrial. When added to base jobs, the share of industrial jobs in 2044 would slightly decrease (54%). The current number of dwellings is small and is only projected to increase by 75 units, assumed to be caretakers' units and artist/studio quarters. See **Exhibit 2.4-7**.

	Existing	2044
Industrial Jobs	54,500*	66,400
Commercial Jobs	44,400*	55,600
Residential Dwellings	413**	488

Exhibit 2.4-7 Alternative 1—No Action Jobs and Housing Units, Existing and 2044

Notes: *2018, ** 2021 Sources: CAI, 2021; City of Seattle, 2021.

Under Alternative 1—No Action, most industrial jobs as well as total jobs are located in the SODO/Stadium and Georgetown/South Park subareas, with less in the Ballard, Interbay Dravus, and Interbay Smith-Cove subareas. See **Exhibit 2.4-8**.

	Current Conditions (2018)		Alternative 1 No Action— Existing Policies (2044)			
Subarea	Industrial Emp	Total Emp	Percent Industrial	Industrial Emp	Total Emp	Percent Industrial
Ballard	9,400	17,100	55.0%	11,600	22,300	52.0%
Interbay Dravus	3,400	5,600	60.7%	3,900	6,800	57.4%
Interbay Smith Cove	3,900	6,000	65.0%	4,700	7,400	63.5%
SODO/Stadium	23,000	43,900	52.4%	28,200	53,500	52.7%
Georgetown/South Park	14,900	25,900	57.5%	18,000	32,000	56.3%
Total	54,500	98,500	55.3%	66,400	122,000	54.4%

Exhibit 2.4-8 Current and Alternative 1—No Action Employment Mix by Subarea

Under Alternative 1 No Action, the expected increment in caretakers' quarters is proportional to the percent increase in employment growth, and there would be an estimated average annual growth rate of 3 artist/studio workspace conversions per year. The number of new units is expected to be 75. See **Exhibit 2.4-9**.

Subarea	Existing (2021)	No Action Total (2044)	No Action Growth (2018-2044)
Ballard	192	200	8
Interbay Dravus	3	11	8
Interbay Smith Cove	1	9	8
SODO/Stadium	21	51	30
Georgetown/South Park	196	219	23
Total	413	488	75

Exhibit 2.4-9 Current and Alternative 1—No Action Housing Units in Industrial Zones

Source: City of Seattle, 2021.

Over two thirds of the increase in jobs (67%) is anticipated to be in the Greater Duwamish MIC and one third in the BINMIC. The increase in population assumes the 2020 citywide household size of 2.05,² and is about 154 persons. See **Exhibit 2.4-10**.

Exhibit 2.4-10 Alternative 1—No Action Jobs and Population Growth by Subarea

Study Area	Job Increase 2018-2044	Population Increase 2021-2044
Ballard	5,200	15
Interbay Dravus	1,200	15
Interbay Smith Cove	1,400	15
SODO/Stadium	9,600	62
Georgetown/South Park	6,100	46
Total	23,500	154

² See 2020 US Census data: <u>https://www.seattle.gov/opcd/population-and-demographics/about-seattle#population</u>. The 2019 household size for the zip code including the Ballard Subarea is 1.96 and for the Georgetown/South Park Subarea is 2.76, or 2.35 average between them. Since it is expected that caretakers' quarters and live/work units may have smaller household sizes the citywide household size is used.

2.4.4 Alternative 2—Future of Industry Limited

Alternative 2—Future of Industry Limited applies the proposed land use concepts with relatively less Industry and Innovation and Urban Industrial than the other two Action Alternatives. See **Exhibit 2.4-12**.

Alternative 2 proposes the following:

- Updates industrial land use policies to anticipate future innovations and trends.
- Strengthens protections for industrial uses in MML zone covering 90% of industrial lands.
- Applies a mix of II and UI zone concepts in 10% of current MIC areas, including an estimated 1/4 mile from light rail stations.
- No expansion of housing allowances.
- Does not remove any land from MICs.

See zoning district acres in **Exhibit 2.4-11**.

Exhibit 2.4-11 Alternative 2—Future of Industry Limited Zoning Districts (Acres)

Zoning Districts	Acres	Share
Maritime, Manufacturing, and Logistics (MML)	6,251	90.1%
Urban Industrial (UI)	222	3.2%
Industry and Innovation (II)	463	6.7%
Total	6,936	100%



Exhibit 2.4-12 Alternative 2—Future of Industry Limited

The total number of jobs is expected to increase by 34,400 with 72% of that industrial in nature; the total share of industrial jobs in 2044 would increase from 55% in 2018 to 60% in 2044. Employment growth of 34,400 projected under Alternative 2 in the study area would represent about 20% of total citywide job growth that the City would be planning for during the 20-year planning horizon of the One Seattle Comprehensive Plan major update. This would represent a shift of a moderately greater share of the city's expected employment growth into industrial areas compared to past trends and the previous 20-year Comprehensive Plan planning horizon.

The number of dwellings is projected to increase by 80 units and assumed to be caretakers' quarters and some artist/studios. See **Exhibit 2.4-13**.

Exhibit 2.4-13 Alternative 2 Jobs and Housing Units, Existing and 2044

	Existing	2044
Industrial Jobs	54,500*	79,400
Commercial Jobs	44,000*	53,500
Residential Dwellings	413**	493

Notes: *2018, ** 2021 Sources: CAI, 2021; City of Seattle, 2021.

Most industrial jobs and total jobs are located in the SODO/Stadium and Georgetown/South Park subareas and would continue to have the greatest growth (67%). See **Exhibit 2.4-14**.

	Current Conditions (2018)		Alternative 2—Future of Indus Limited (2044)			
Subarea	Industrial Emp	Total Emp	Percent Industrial	Industrial Emp	Total Emp	Percent Industrial
Ballard	9,400	17,100	55.0%	13,600	23,600	57.6%
Interbay Dravus	3,400	5,600	60.7%	4,900	7,700	63.6%
Interbay Smith Cove	3,900	6,000	65.0%	5,800	8,600	67.4%
SODO/Stadium	23,000	43,900	52.4%	33,700	57,700	58.4%
Georgetown/South Park	14,900	25,900	57.5%	21,400	35,300	60.6%
Total	54,500	98,500	55.3%	79,400	132,900	59.7%

Exhibit 2.4-14 Current and Alternative 2 Employment Mix by Subarea

Under Alternative 2, the expected increment in caretakers' quarters is proportional to the percent increase in employment growth, and there would be an estimated average annual growth rate of 3 artist/studio workspace conversions per year. With these assumptions the number of units is expected to be 80. See **Exhibit 2.4-15**.

Subarea	Existing	Alternative 2 Total (2044)	Alternative 2 Growth
Ballard	192	200	8
Interbay Dravus	3	11	8
Interbay Smith Cove	1	9	8
SODO/Stadium	21	53	32
Georgetown/South Park	196	220	24
Total	413	493	80

Exhibit 2.4-15 Current and Alternative 2 Housing Units in Industrial Zones

Source: City of Seattle, 2021.

Two thirds of the increase in jobs is anticipated to be in the Greater Duwamish MIC and one third in the BINMIC. The increase in population assumes the 2020 citywide household size of 2.05,³ and is about 164 persons. See **Exhibit 2.4-16**.

Exhibit 2.4-16 Alternative 2 Jobs and Population Growth by Subarea

Subarea	Job Increase 2018-2044	Population Increase 2021-2044
Ballard	6,500	16
Interbay Dravus	2,100	16
Interbay Smith Cove	2,600	16
SODO/Stadium	13,800	66
Georgetown/South Park	9,400	49
Total	34,400	164

³ See 2020 US Census data: <u>https://www.seattle.gov/opcd/population-and-demographics/about-seattle#population</u>. The 2019 household size for the zip code including the Ballard Subarea is 1.96 and for the Georgetown/South Park Subarea is 2.76, or 2.35 average between them. Since it is expected that caretakers' quarters and live/work units may have smaller household sizes the citywide household size is used.

2.4.5 Alternative 3—Future of Industry Targeted

Alternative 3—Future of Industry Targeted applies the proposed land use concepts with a greater share of Industry and Innovation and Urban Industrial than Alternative 2. See **Exhibit 2.4-18**.

Alternative 3 proposes the following:

- Updates industrial land use policies to anticipate future innovations and trends.
- Strengthens protections for industrial uses in MML zones covering 86% of industrial lands.
- Applies a mix of II and UI zone concepts in 14% of current MIC areas, including an estimated 1/2 mile from light rail stations.
- Expansion of limited industry-supportive housing in UI zone concept.
- Removes focused land in Georgetown/South Park from MICs.

Acres by zoning are shown in **Exhibit 2.4-17**.

Exhibit 2.4-17 Alternative 3—Future of Industry Targeted Zoning Districts (Acres)

Zoning Districts	Acres	Share
Maritime, Manufacturing, and Logistics (MML)	5,968.00	86.0%
Urban Industrial (UI)	426	6.1%
Industry and Innovation (II)	516	7.4%
Mixed-Use Commercial	26	0.4%
Total	6,936	100%



Exhibit 2.4-18 Alternative 3—Future of Industry Targeted

The total number of jobs would increase by 57,400 with 60% of those industrial jobs; the total share of industrial jobs in 2044 would slightly decrease from 55% in 2018 to 54% in 2044. This level of employment growth would shift a sizeable share of Seattle's total employment growth into MICs compared to historic growth rates in MICs. Employment growth of 57,400 projected under Alternative 3 in the study area would represent about 34% of total citywide job growth that the city is planning for during the 20-year planning horizon of the One Seattle <u>Comprehensive Plan major update</u>. This would represent a substantial shift of the total share of the city's expected employment growth into MICs and industrial areas compared to past trends and the previous 20-year Comprehensive Plan planning horizon.

The number of dwellings is projected to increase by 610 units <u>in industrial zones to a total of 1,023 units</u>, with a combination of caretakers' quarters and makers studios under modified allowances for industry-supportive housing in the UI zone. <u>There would also be 1,078 units in mixed use zones removed from the MIC.</u> See **Exhibit 2.4-19**.

Exhibit 2.4-19 Alternative 3 Jobs and Housing Units, Existing and 2044

	Existing	2044
Industrial Jobs	54,400*	83,500
Commercial Jobs	44,000*	72,400
Residential Dwellings (Industrial zones)	112++	<u>1,023</u> 610
Residential Dwellings (new mixed-use commercial zones)	413**	1, <u>078</u> 491
Notoc: #2018 ##2021		

Notes: *2018, **2021 Sources: CAI, 2021; City of Seattle, 2021.

As with today's conditions, most industrial jobs and total jobs are located in the SODO/Stadium and Georgetown/South Park subareas and would continue to have the greatest growth. See **Exhibit 2.4-20**.

	Current Conditions (2018)		Alternative Ta	e 3—Future Irgeted (204	of Industry 4)	
Subarea	Industrial Emp	Total Emp	Percent Industrial	Industrial Emp	Total Emp	Percent Industrial
Ballard	9,400	17,100	55.0%	15,900	31,100	51.1%
Interbay Dravus	3,400	5,600	60.7%	5,500	9,900	55.6%
Interbay Smith Cove	3,900	6,000	65.0%	6,300	10,500	60.0%
SODO/Stadium	23,000	43,900	52.4%	34,700	66,000	52.6%
Georgetown/South Park	14,900	25,900	57.5%	21,100	38,400	54.9%
Total	54,500	98,500	55.3%	83,500	155,900	53.6%

Exhibit 2.4-20 Current and Alternative 3 Employment Mix by Subarea

Sources: CAI, 2021; City of Seattle, 2021.

Under Alternative 3, there would be greater allowance for housing in the UI zone. Focused land in Georgetown/South Park would be removed from MICs and could be developed for housing. With these collective changes, the number of dwellings in industrial zones would increase to about 610. Most of the housing in industrial zones would be in Ballard and the SODO/Stadium subareas. See **Exhibit 2.4-21**.

Subarea	Existing (2021)	Alternative 3 Total (2044)	Alternative 3 Growth
Ballard	192	452	260
Interbay Dravus	3	78	75
Interbay Smith Cove	1	16	15
SODO/Stadium	21	221	200
Georgetown/South Park (industrial zones)	196	256	60
Total	413	1,023	610

Source: City of Seattle, 2021.

In addition to the housing in industrial zones, some more new housing would result in focused areas in Georgetown and South Park that would be removed from the MIC and placed in a mixed-use zone. In Alternative 3, a total of 784 dwelling units in mixed-use developments are estimated for the triangular area of Georgetown bounded by Airport Way, Corson Avenue S, and Carleton Avenue S, and 294 dwelling units are estimated for the two small areas of South Park that would be removed from the MIC near the Duwamish River. This would result in a total of 1,078 housing units over the study time horizon on land that is removed from industrial zoning under Alternative 3.

About 60% of the increase in jobs is anticipated to be in the Greater Duwamish MIC and one 40% in the BINMIC. The increase in population assumes the 2020 citywide household size of 2.05,⁴ and is about 1,251 persons. See **Exhibit 2.4-22**. The areas removed from the MIC would be zoned for mixed-uses and have capacity for about 2,210 people beyond the population in the industrial zones addressed in **Exhibit 2.4-21**.

Within the study area the collective change in population including within industrial areas and the MIC reduction areas would equal 3,460 persons.

Subarea	Job Increase 2018-2044	Population Increase 2021-2044
Ballard	14,000	533
Interbay Dravus	4,300	154
Interbay Smith Cove	4,500	31
SODO/Stadium	22,100	410
Georgetown/South Park	12,500	123
Total	57,400	1,251

Exhibit 2.4-22 Alternative 3 Jobs and Population Growth by Subarea

⁴ See 2020 US Census data: <u>https://www.seattle.gov/opcd/population-and-demographics/about-seattle#population</u>. The 2019 household size for the zip code including the Ballard Subarea is 1.96 and for the Georgetown/South Park Subarea is 2.76, or 2.35 average between them. Since it is expected that caretakers' quarters and live/work units may have smaller household sizes the citywide household size is used.

2.4.6 Alternative 4—Future of Industry Expanded

Alternative 4—Future of Industry Expanded applies the proposed land use concepts with a greater share of Industry and Innovation and Urban Industrial than Alternative 2. This alternative expands limited housing allowances compared to Alternative 3. See **Exhibit 2.4-24**.

Alternative 4 proposes the following:

- Updates industrial land use policies to anticipate future innovations and trends.
- Strengthens protections for industrial uses in maritime, manufacturing and logistics zones covering 87% of industrial lands.
- Applies a mix of II and UI zone concepts in 13% of current MIC areas, including an estimated 1/2 mile from light rail stations.
- Expansion of limited industry-supportive housing in UI zone concept.
- Removes focused land in Georgetown/South Park from the MIC.
- Increases maximum size of use limit for indoor sports and recreation uses.

The zoning districts by acres is listed in **Exhibit 2.4-23**.

Exhibit 2.4-23 Alternative 4—Future of Industry Expanded Zoning Districts (Acres)

Zoning Districts	Acres	Share
Maritime, Manufacturing, and Logistics (MML)	6,035	87.0%
Urban Industrial (UI)	279	4.0%
Industry and Innovation (II)	600	8.7%
Mixed-Use Commercial	22	0.3%
Total	6,936	100%



Exhibit 2.4-24 Alternative 4—Future of Industry Expanded

The total number of jobs would increase by 59,200 with 49% of those industrial jobs; the total share of industrial jobs in 2044 would slightly decrease from 55% in 2018 to 53% in 2044. Like Alternative 3, this level of employment growth would shift a sizeable share of Seattle's total employment growth into MICs compared to historic growth rates in MICs. Employment growth of 59,500 projected under Alternative 4 in the study area would represent about 35% of total citywide job growth that the city would be planning for during the 20-year planning horizon of the One Seattle Comprehensive Plan major update. Similar to Alternative 3, this would represent a substantial shift of the total share of the city's expected employment growth into MICs and industrial areas compared to past trends and the previous 20-year Comprehensive Plan planning horizon.

The number of dwellings is projected to increase by 2,195 units <u>in industrial zones</u>, with a combination of caretakers' quarters and makers studios under modified allowances for industry-supportive housing in the UI zone. <u>An additional 1,078 dwellings would be located in new mixed use areas as in Alternative 4.</u> See **Exhibit 2.4-25**.

	Existing	2044
Industrial Jobs	54,400*	83,300
Commercial Jobs	44,000*	74,400
Residential Dwellings (industrial zones)	410++	2, <u>608</u> 195
Residential Dwellings (new commercial mixed-use zones)	413**	1, <u>078</u> 491

Exhibit 2.4-25 Alternative 4 Jobs and Housing Units, Existing and 2044

Notes: *2018, ** 2021

Sources: CAI, 2021; City of Seattle, 2021.

Most industrial jobs and total jobs (59%) are located in the SODO/Stadium and Georgetown/South Park subareas and would continue to have the greatest total growth. Relative to other alternatives, Alternative 4 places more jobs in Ballard and Interbay subareas. See **Exhibit 2.4-26**.

Exhibit 2.4-26 Current and Alternative 4 Employment Mix by Subarea

	Current Conditions (2018)			Alternative 4—Future of Industry Expanded (2044)		
Subarea	Industrial Emp	Total Emp	Percent Industrial	Industrial Emp	Total Emp	Percent Industrial
Ballard	9,400	17,100	55.0%	16,000	32,000	50.0%
Interbay Dravus	3,400	5,600	60.7%	5,600	10,200	54.9%
Interbay Smith Cove	3,900	6,000	65.0%	6,300	10,700	58.9%
SODO/Stadium	23,000	43,900	52.4%	34,400	66,300	51.9%
Georgetown/South Park	14,900	25,900	57.5%	21,000	38,500	54.5%
Total	54,500	98,500	55.3%	83,300	157,700	52.8%

Under Alternative 4, there would be greater allowance for housing in the UI zone. As a result, residential dwelling units in industrial zones would increase to about 2,195. Most would be in Ballard and the SODO/Stadium subareas. See **Exhibit 2.4-27**.

Subarea	Existing (2021)	Alternative 4 Total (2044)	Alternative 4 Growth
Ballard	192	982	790
Interbay Dravus	3	178	175
Interbay Smith Cove	1	1	0
SODO/Stadium	21	1011	990
Georgetown/South Park	196	436	240
Total	413	2,608	2,195

Exhibit 2.4-27	Current and Alternative 4 Housing Units in Industrial Zones
----------------	--

Source: City of Seattle, 2021.

In addition to the housing in industrial zones, some more new housing would result in focused areas in Georgetown and South Park that would be removed from the MIC and placed in a mixed-use zone. These quantities are the same in Alternative 4 as in Alternative 3: 784 dwelling units in the triangular area of Georgetown bounded by Airport Way, Corson Avenue S, and Carleton Avenue S, and 294 dwelling units in the two small areas of South Park near the Duwamish River, for a total of 1,078 housing units during the study time horizon. This would potentially add another 2,210 in population beyond the added population in the industrial zones addressed in **Exhibit 2.4-27**.

About 59% of the increase in jobs is anticipated to be in the Greater Duwamish MIC and 41% in the BINMIC; more jobs are in the Ballard Subarea than the Georgetown/South Park Subarea. The increase in population assumes the 2020 citywide household size of 2.05,⁵ and is about 4,500 persons. See **Exhibit 2.4-28**. Within the study area the collective change in population including within industrial areas and the MIC reduction areas would equal 6,710 persons.

Subarea	Job Increase 2018-2044	Population Increase 2021-2044
Ballard	14,900	1,620
Interbay Dravus	4,600	359
Interbay Smith Cove	4,700	
SODO/Stadium	22,400	2,030
Georgetown/South Park	12,600	492
Total	59,200	4,500

Exhibit 2.4-28 Alternative 4 Jobs and Population Growth by Subarea

⁵ See 2020 US Census data: <u>https://www.seattle.gov/opcd/population-and-demographics/about-seattle#population</u>. The 2019 household size for the zip code including the Ballard Subarea is 1.96 and for the Georgetown/South Park Subarea is 2.76, or 2.35 average between them. Since it is expected that caretakers' quarters and live/work units may have smaller household sizes the citywide household size is used.

2.4.7 Preferred Alternative—Future of Industry Balanced

<u>The Preferred Alternative—Future of Industry Balanced applies the proposed land use concepts</u> with a combination of features of alternatives 2, 3, and 4, and new features and refinements to address comments and reduce impacts. The Preferred Alternative proposes the following:

- Updates industrial land use policies to anticipate future innovations and trends like the Draft EIS action alternatives.
- Strengthens protections for industrial uses in maritime, manufacturing and logistics zones covering 85% of industrial lands like the Draft EIS action alternatives.
- Applies a mix of II and UI zone concepts in 14% of current MIC areas, including an estimated 1/2 mile from light rail stations like Draft EIS Alternative 4.
- Expands limited industry-supportive housing in the UI zone, subject to a conditional use process and more location and performance criteria than Draft EIS alternatives 3 or 4, and maintaining a limit on density as in alternatives 3 or 4.
- The industry-supportive housing criteria could be met in one of two ways—either by limiting occupancy to caretakers or makers (as in alternatives 3 and 4), or by providing a minimum of 50% of any housing units that are created to households with incomes at 90% of Area Median Income (AMI) or below.
- Removes focused land in Georgetown/South Park from the MIC similar to alternatives 3 and 4.
- Retains Industrial Commercial zoning for some existing industrially-zoned areas outside of MICs like Alternative 1.
- Converts focused areas of industrial zoning outside of MICs to new mixed use zones that would allow housing. Together with the change around Georgetown/South Park the new mixed use zones would equal about 1% of the study area.
- Applies the MML zone to the WOSCA and Armory sites until site specific master planning can be completed.
- Increases the amount of proposed UI zoning around Georgetown to create more neighborhood cohesion.
- Applies a nuanced approach to a proposed mixed use zone in central Georgetown that reflects community priorities including preserving arts space and historic aged structures.
- Increases incentives for development feasibility in the II zone compared to Draft EIS alternatives.
- Increases maximum size of use limit for indoor sports and recreation uses like Alternative 4.

The zoning districts by acres is listed in **Exhibit 2.4-29**. For a map please see **Exhibit 2.4-30**.

Exhibit 2.4-29 Preferred Alternative—Future of Industry Balanced Zoning Districts (Acres)

Zoning Districts	<u>Preferred</u>	<u>Share</u>
Maritime, Manufacturing, and Logistics (MML)	<u>5,895</u>	<u>85.0%</u>
<u>Urban Industrial (UI)</u>	<u>376</u>	<u>5.4%</u>
Industry and Innovation (II)	<u>612</u>	<u>8.8%</u>
Mixed-Use Commercial	<u>53</u>	<u>0.8%</u>
Total	<u>6,936</u>	<u>100%</u>



Exhibit 2.4-30 Preferred Alternative—Future of Industry Balanced

The total number of jobs would increase by 35,545 with 46% of those industrial jobs. The absolute number of industrial jobs would be greater than the No Action Alternative. The total share of jobs in 2044 that are industrial would slightly decrease from 55% in 2018 to 53% in 2044. The Preferred Alternative would make a moderate shift of Seattle's total employment growth into MICs compared to historic growth rates in MICs. Employment growth of 35,545 projected under the Preferred Alternative in the study area would represent about 18% of the net citywide job growth that the city would be planning for during the 20-year planning horizon of the One Seattle Comprehensive Plan major update. Projections are adjusted downward compared to the Draft EIS alternatives to reflect conditions more realistically in commercial/office demand post-COVID and in consideration of Sound Transit's timeline for completion of light rail construction of the West Seattle to Ballard line. The adjusted projections acknowledge that it will likely take longer to achieve levels of employment growth.

Exhibit 2.4-31 Preferred Alternative Jobs and Housing Units, Existing and 2044

	Existing	<u>2044</u>
Industrial Jobs	<u>54,400*</u>	<u>70,853</u>
Commercial Jobs	<u>44,000*</u>	<u>63,192</u>
Residential Dwellings (industrial zones)	410++	<u>1,888</u>
Residential Dwellings (new commercial mixed-use zones)	413***	<u>1,534</u>

Notes: *2018, **2021

Sources: CAI, 2021; City of Seattle, 2022.

Most industrial jobs and total jobs (66%) are located in the SODO/Stadium Subarea and the Georgetown/South Park Subarea and these subareas would continue to have the greatest total growth. Relative to other alternatives, the Preferred Alternative places slightly more jobs in Ballard and Interbay subareas. See **Exhibit 2.4-32**.

	Current Conditions (2018)			Preferred Alternative Future of Industry—Balanced (2044)		
<u>Subarea</u>	<u>Industrial</u> <u>Emp</u>	<u>Total</u> <u>Emp</u>	<u>Percent</u> Industrial	<u>Industrial</u> <u>Emp</u>	<u>Total</u> <u>Emp</u>	<u>Percent</u> Industrial
Ballard	<u>9,400</u>	<u>17,100</u>	<u>55.0%</u>	<u>13,685</u>	<u>27,479</u>	<u>49.8%</u>
Interbay Dravus	<u>3,400</u>	<u>5,600</u>	<u>60.7%</u>	<u>4,784</u>	<u>8,713</u>	<u>54.9%</u>
Interbay Smith Cove	<u>3,900</u>	<u>6,000</u>	<u>65.0%</u>	<u>5,130</u>	<u>8,713</u>	<u>58.9%</u>
SODO/Stadium	<u>23,000</u>	<u>43,900</u>	<u>52.4%</u>	<u>29,122</u>	<u>55,897</u>	<u>52.1%</u>
Georgetown/South Park	<u>14,900</u>	<u>25,900</u>	<u>57.5%</u>	<u>18,133</u>	<u>33,243</u>	<u>54.5%</u>
Total	<u>54,500</u>	<u>98,500</u>	<u>55.3%</u>	70,853	<u>134,045</u>	<u>52.9%</u>

Exhibit 2.4-32 Current and Preferred Alternative Employment Mix by Subarea

The number of dwellings in industrial areas is projected to increase by 1,475 units in the UI zone, 33% less than the amount studied in Draft EIS Alternative 4. These would be industry-supportive housing units—either caretakers' quarters and makers studios, or having at least half of the homes restricted to be affordable to households with incomes common for jobs in industrial sectors. Criteria for the location and performance of any industry-supportive housing in the UI zone would be more limited than alternatives 3 and 4 in the Draft EIS and subject to a conditional use approval process. This housing would make up about 1.8% of new units planned for citywide over the time horizon of the One Seattle Comprehensive Plan major update.

<u>Subarea</u>	<u>Existing (2021)</u>	<u>Preferred</u> <u>Alternative</u> <u>Total (2044)</u>	<u>Preferred</u> <u>Alternative</u> <u>Growth</u>
Ballard	<u>192</u>	706	<u>514</u>
Interbay Dravus	<u>3</u>	<u>117</u>	<u>114</u>
Interbay Smith Cove	1	<u>1</u>	<u>0</u>
<u>SODO/Stadium</u>	<u>21</u>	<u>665</u>	<u>644</u>
Georgetown/South Park	<u>196</u>	<u>400</u>	<u>204</u>
Total	<u>413</u>	<u>1,888</u>	<u>1,475</u>

Exhibit 2.4-33 Current and Preferred Alternative Housing Units in Industrial Zones

Source: City of Seattle, 2022.

Two new areas outside the MICs in west Ballard and Judkins Park would be converted to mixed use zoning allowing housing, in addition to the proposed mixed-use areas in Georgetown and South Park studied in Draft EIS alternatives. Overall, a higher total amount of housing production outside of MICs would result compared to Draft EIS alternatives—an additional 1,534 dwellings, 42% more than alternatives 3 and 4.

The collective change in population including within industrial areas, areas removed from the MIC, and rezoned areas converted to mixed use zoning outside of the MIC would equal 3,009 households, about 8% less than Alternative 4. The combination of employment and population growth is lower than both alternatives 3 and 4.

More nuanced specific development standards are proposed under the Preferred Alternative for the triangular area of Georgetown bounded by Airport Way, Corson Avenue S, and Carleton Avenue S. The standards integrate Georgetown priorities for historic preservation, antidisplacement, and arts spaces.

In the Preferred Alternative, the new mixed use zone in the triangle area of Georgetown would be the Neighborhood Commercial zone with a 55 foot height limit, (NC3-55). Additional standards would be added in the specific area development standards sections to provide the following features to incentivize the retention and restoration and reuse of a.) historic character structures, and b.) arts organization and/or arts studios. Floor area that is retained in a historic character structure, or floor area of an arts organization or studio would be exempt from FAR Exhibit 2 4-34-20

limits. The amount of the exempted floor area could be allocated to a same site or adjacent parcel. For any development that includes retained historic character structure or arts organization or arts studio, height limit can be increased to 65 feet. A Mandatory Housing Affordability (M1) suffix would be applied to the zone.

2.4.72.4.8 Summary of Alternatives

Exhibit 2.4-34 below summarizes the land use concepts under each of the four alternatives studied in this EIS. It is important to keep in mind that these are not zoning proposals when reviewing the alternatives. A legislative proposal will be developed once the EIS process is complete which will likely be a hybrid of the alternatives described below and may include refinements to detailed aspects of the development standards.

Summary of Land Use Concents by Alternatives

		,			
No Action Alternative	New Land Use Concepts	Alt 2—Future of Industry Limited	Alt 3—Future of Industry Targeted	Alt 4—Future of Industry Expanded	Preferred Alt— Future of Industry Balanced
Industrial General Zones: 90% of land area	Maritime Manufacturi ng and Logistics (MML) Zone	90% with stronger protections.	86% with stronger protections.	87% with stronger protections.	<u>86% with stronger</u> protections.
Industrial Commercial Zones: 5% of land area	Industry and Innovation (II) Zone	7% of land area. Located up to approximately ¼ mile around transit stations and all land currently zoned industrial commercial.	7% of land area. Located approximately up to ½ mile around transit stations and all land currently zoned Industrial Commercial.	9% of land area. Located greater than ½ mile around transit stations and all land currently zoned Industrial Commercial. Includes land near potential Ballard ST3 station and the Stadium ST3 station.	8% of land area. Applies the II zone within approximately ½ mile around transit stations, and includes IC zoning that is retained outside of MICs.
Industrial Buffer Zone: 5% of land area	Urban Industrial (UI) Zone	3% of land area. Located generally in transition areas between MML or II zones and nonindustrial areas.	6% of all land area. Expanded transition area in Ballard.	4% of land area. Expanded transition area in Stadium district.	<u>5% of land area.</u>
Areas removed from MIC and placed in mixed- use zone		None.	Small nodes in Georgetown/Sout h Park to advance community goals (1.078 units).	Small nodes in Georgetown/South Park to advance community goals (1.078 units).	Small nodes in Georgetown/South Park to advance community goals. Similar area removed as in alternatives 3 and
No Action Alternative	New Land Use Concepts	Alt 2—Future of Industry Limited	Alt 3—Future of Industry Targeted	Alt 4—Future of Industry Expanded	<u>Preferred Alt—</u> <u>Future of Industry</u> <u>Balanced</u>
--	-----------------------------------	--	--	--	--
					4. , L i ess dwellings due <u>to nuanced</u> <u>development</u> standards (686 units).
Areas outside of MIC in West Ballard and Judkins Park (currently zoned IG)		Applies MML in West Ballard and II in Judkins Park.	Applies MML in West Ballard and II in Judkins Park.	Applies II in both West Ballard and Judkins Park.	Applies a mixed use (Neighborhood Commercial) zone resulting in 848 projected units.
Only new caretaker's quarters, artist housing and existing non- conforming: approx. 413 units	Housing in Industrial Zones	No expanded allowances.	Expanded industry- supportive in UI zones: approx. 610 units.	Larger expansion of Industry-supportive in UI zones: approx. 2,195 units.	Expansion of Industry- supportive housing in UI zones by conditional use only; approx. 1,475 units. 50% affordable at 90% AMI or below option.
Lodging Prohibited	Stadium Overlay	No change.	Allow lodging.	All lodging with larger size of use limits.	Same as Alternative 4.
Size of Use Limits	Non- Industrial uses.	Varies by zone. Expanded non- industrial ancillary uses <u>(UI)</u> . Reduced stand-alone non- industrial size of use limits <u>(MML)</u> . No limit in bonus space (II).	Varies by zone. Expanded non- industrial ancillary uses (UI). Reduced stand-alone non- industrial size of use limits (MML). No limit in bonus space (II).	Varies by zone. Expanded non- industrial ancillary uses (UI). Reduced stand-alone non- industrial size of use limits (MML). No limit in bonus space (II). Expanded size of use limit for indoor recreational facilities.	Same as Alternative 4.
MIC Subarea Plans	Current Plans	Update MIC Subarea Plans per VISION 2050	Update MIC Subarea Plans per VISION 2050	Update MIC Subarea Plans per VISION 2050	<u>Update MIC Subarea</u> Plans per VISION 2050
Comprehensive Plan Policies	Current Policies	Amend Comprehensive Plan Policies to establish new land use framework, limit MIC boundary changes to Periodic Update, establish City's intent to work with State of Washington on a	Amend Comprehensive Plan Policies to establish new land use framework, limit MIC boundary changes to Periodic Update, establish City's intent to work with State of Washington on a	Amend Comprehensive Plan Policies to establish new land use framework, limit MIC boundary changes to Periodic Update, establish City's intent to work with State of Washington on a masterplan for the	Amend Comprehensive Plan Policies to establish new land use framework, limit MIC boundary changes to Periodic Update, establish City's intent to work with State of Washington on a masterplan for the

No Action Alternative	New Land Use Concepts	Alt 2—Future of Industry Limited	Alt 3—Future of Industry Targeted	Alt 4—Future of Industry Expanded	<u>Preferred Alt—</u> <u>Future of Industry</u> <u>Balanced</u>
		masterplan for the Armory and WOSCA Sites.	masterpian for the Armory and WOSCA Sites.	Armory and WUSCA Sites.	Armory and WUSCA Sites.

Sources: BERK, 20224; City of Seattle, 20224.

A comparison of zoned acres is listed below. In all alternatives, the majority of the study area would be dedicated for industrial and manufacturing uses (IG or MML). Some areas zoned for industrial and manufacturing uses today would be designated instead for transitional zoning (UI) or dense employment (II) under the Action Alternatives. See **Exhibit 2.4-35**.

Exhibit 2.4-35-30 Comparison of Alternatives by Land Use/Zoning Acres							
Zoning Districts	Alt 1	Land Use Concept	Alt 2	Alt 3	Alt 4	<u>Pref.</u>	
Industrial General (IG1/IG2)	6,273	Maritime, Manufacturing, and Logistics (MML)	6,251	5,968	6,035	<u>5,895</u>	
Industrial Buffer (IB)	316	Urban Industrial	222	426	279	<u>376</u>	
Industrial Commercial (IC)	347	Industry and Innovation	463	516	600	<u>612*</u>	
Mixed-Use Commercial				26	22	<u>53</u>	
Total	6,936		6,936	6,936	6,936	<u>6,936</u>	
the share a second part of LC successible the NAUC							

*Includes some retention of IC outside the MIC.

Sources: City of Seattle, 202<u>2</u>1; BERK, 202<u>2</u>1.

Exhibit 2.4-36 summarizes total projected employment growth in the study area for the base year and by alternative, with a breakout of industrial⁶ and non-industrial employment. The No Action Alternative and all three of the Action Alternatives result in employment growth. Overall employment growth is strongest under alternatives 3 and 4, which would result in 58% and 60% employment growth from the base year of 2018 over the time horizon to 2044. This would be substantially more job growth in Seattle's MICs than has occurred in the last 20-year period due to the proposed changes. Total employment growth under the Preferred Alternative would be less than alternatives 3 and 4 and would be an increase of 36% from the base year. Employment projections are moderated under the Preferred Alternative to reflect more realistic conditions in demand for employment spaces post-COVID and timelines for completion of new Sound Transit light rail lines. The overall number of industrial jobs would grow in all of the alternatives—ranging from +11,900 under No Action to +28,800 under

⁶ Industrial employment estimated based on the 2019 share of industrial employment by sector based on the 2015 PSRC Industrial Lands Study NAICs-based definition of industrial activities. This uses classification of what counts as an industrial job consistent with Puget Sound Regional Council criteria, including jobs in Information Computer Technology (ICT). Projections show strong job growth in ICT under the Action Alternatives. Consistency with PSRC classifications is appropriate given the need to fit VISION 2050 and Regional Centers Framework. A more conservative classification of which jobs are industrial, especially in ICT would show a steeper decline in the % of industrial jobs under most studied alternatives.

Alternative 4. The percentage of the jobs that are industrial however would decrease incrementally from 55% in the base year to 53% under Alternative 4<u>or the Preferred</u> <u>Alternative</u>. See **Exhibit 2.4-37**.



Exhibit 2.4-36-31 Industrial and Non-Industrial Job Share

Note: This chart was updated to include the Preferred Alternative. Source: City of Seattle, 202<u>2</u>4.





Industrial Employment

Non-industrial Employment

Note: Regional manufacturing/ industrial centers must retain a minimum 50% industrial employment per PSRC.

Note: This chart was updated to include the Preferred Alternative. Sources: City of Seattle, 20224; BERK, 20224.

The total combined employment and population growth is illustrated on the graph below for each alternative. Considered in combination, the total jobs and population by alternative shows

the highest total job and population growth under Alternative 4 and the lowest under Alternative 1. The Preferred Alternative has a total in that is slightly more than Alternative 2 and less than Alternative 3. See **Exhibit 2.4-38**.



Exhibit 2.4-39 shows percentage of employment growth by subarea to display which subareas would have relatively greater employment growth over the base amount. The north subareas of Ballard, Interbay Dravus, and Interbay Smith Cove would have the highest employment growth on a percentage basis, most notably under alternative<u>s</u> 3 and 4 where employment growth is projected to increase by over 70% for each of these three northern areas. <u>Under the Preferred Alternative</u>, employment growth in the Ballard and Interbay Dravus subareas would be between the amount for Alternative 2 and alternatives 3 and 4, and growth in the Interbay Smith Cove Subarea would be similar to Alternative 2. Employment growth in the <u>SODO/Stadium and South Park/Georgetown subareas would be between the amounts of alternatives 1 and 2.</u>

While the greatest percent change in jobs is in the northern BINMIC subareas, the number of new jobs is greater in the Greater Duwamish MIC southern subareas. See **Exhibit 2.4-40**.



Exhibit 2.4-39-33 Percent Growth in Employment by Subarea

Note: This chart was updated to include the Preferred Alternative. Sources: CAI, 202<u>2</u>4; City of Seattle, 202<u>2</u>4.





Sources: CAI, 202<u>2</u>1; City of Seattle, 202<u>2</u>1.

2.4.82.4.9 Alternatives Considered & Not Carried Forward

Following scoping, the City made some adjustments to the alternatives (see **Appendix A** for the scoping report) such as considering the sizing of recreation uses in some zones. Other ideas were considered but not carried forward.

The City considered scoping comments requesting more extensive changes to MIC boundaries, or requests for zoning allowing residential or mixed-uses across the study area at particular sites, and considered an alternative that would have de-designated the BINMIC as a MIC. However, the city determined that these approaches would not be likely to advance towards the proposal objectives and would not be in keeping with the intent of City decisionmakers and policymakers. Therefore, the City largely retained the focus of alternatives on industrial and maritime purposes.

- The EIS represents an implementation action of the recently completed Industry and Maritime Strategy and the alternatives are heavily informed by the recommendations of that strategy, including adding no significant new housing in industrial areas, and rather focusing primarily on industrial uses consistent with regional and city plans.
- The proposal includes a policy change calling for collaborative master planning of the Armory site. The site is within the MIC, and the proposal is that updated MIC policies and industrial zone designations will apply to the site. Should the State and partners wish to pursue non-industrial future uses, that would have to be determined in the master plan in partnership with the City and other entities.

The EIS does consider a policy to allow for individual MIC boundary adjustments during the periodic review or during the annual amendment process.

The City considered Draft EIS comments in developing the Preferred Alternative, as described in **Section 2.4.7**. Some features of the Preferred Alternative that were added directly in response to comments are noted in the description of the Preferred Alternative. Additional mitigation measures in response to comments are added in the Final EIS. Additional details about proposed development standards are added in the Final EIS in response to comments.

2.5 Benefits & Disadvantages of Delaying the Proposed Action

Benefits of the proposed action include strengthened protections for core industrial and maritime uses in established economic clusters, opportunities for emerging formats of industrial activity, higher levels of industrial and non-industrial job growth over time and expanded equitable access to living wage jobs (particularly for BIPOC youth), provisions for industry-supportive housing (such as caretakers' quarters and artist loft/maker studios) in targeted locations, and improved transportation conditions for multi-modal travel. In addition,

the proposed action would improve transitions between the MICs and adjacent residential neighborhoods or urban villages and introduce nodes of high-density employment and multimodal access near existing and planned high-capacity transit. Revisions to the MIC boundary in focused areas of Georgetown and South Park would also add additional mixed-use housing opportunities in these neighborhoods. <u>Under the Preferred Alternative, there would be</u> additional mixed use housing outside the MIC on land currently zoned industrial in West <u>Ballard and Judkins Park.</u>

The proposed action may increase vulnerability to sea level rise and exposure to other environmental health hazards—such as noise, air pollution, and GHG emissions—by bringing more people into vulnerable areas, particularly in the Greater Duwamish MIC which has large geographic areas vulnerable to sea level rise impacts. Redevelopment that complies with requirements of the Shoreline Master Program and frequently flooded areas, along with adaptation measures, may decrease vulnerability to sea level rise relative to existing conditions. The proposed action would reduce air emissions below current levels though not as much as delaying the proposed action. Increasing employment density in the MICs, could contribute to regional efforts to limit vehicular GHG emissions.

Delaying the proposed action would limit the addition of industry-supportive housing or mixeduse housing in the small areas removed from the MIC, and delaying the corresponding increase in demand for parks and schools. Disadvantages of delaying the proposed action may limit the pace of potential investments in parks and streetscapes that tend to be implemented with residential or mixed-use development.

Delaying the proposed action would continue the present built environment conditions and result in lower levels or job growth over time. This may result in continued loss of industrial land to non-industrial uses because of existing market pressures to convert industrial land, "one off" zoning decisions, and encroachment of non-industrial uses in industrial zones. There would also be slightly lower demand for public services and utilities.

Delaying the proposed action would not integrate recommendations from the Industrial and Maritime Strategy advisory council into the Comprehensive Plan or zoning and development standards. Updated Subarea Plans for the MICs per VISION 2050 may also be delayed. [This page is intentionally blank]



This chapter describes the affected environment, potential impacts, and mitigation measures for the following topics:

- Section 3.1 Soils/Geology
- Section 3.2 Air Quality & GHG
- Section 3.3 Water Resources
- Section 3.4 Plants & Animals
- Section 3.5 Contamination
- Section 3.6 Noise
- Section 3.7 Light & Glare
- Section 3.8 Land & Shoreline Use
- Section 3.9 Housing
- Section 3.10 Transportation
- Section 3.11 Historic, Archaeological, & Cultural Resources
- Section 3.12 Open Space & Recreation
- Section 3.13 Public Services
- Section 3.14 Utilities

Following a description of current conditions (affected environment), the analysis compares and contrasts the alternatives and provides mitigation measures for identified impacts. It also summarizes whether there are significant unavoidable adverse impacts.

The analysis is broad, areawide, and comparative, considering the non-project proposals. (WAC 197-11-442) Where there is a potential for more than a moderate adverse impact on environmental quality (WAC 197-11-794), existing or potential mitigation measures are posed. Consistent with the non-project analysis, mitigation measures are policy, plan, regulation, or program activities that the City could undertake to limit impacts.

Section 3.1 Soils/Geology



This chapter describes the affected environment for soils/geology conditions and presents the analysis completed to compare and contrast the alternatives. Mitigation measures for identified impacts and any significant unavoidable adverse impacts are also summarized. The study area for Soils/Geology is defined as the area that could be directly or indirectly affected by the construction activities or land uses that result from implementation of the industrial and maritime strategy.

Impacts of the alternatives on soils/geology conditions are considered significant if they result in:

- Erosion that could not be contained on future development sites.
- Exposure of people to risk of injury or substantial damage to structures and infrastructure due to the creation or acceleration of a geologic hazard, such as slope failure, liquefaction, settlement.

3.1.1 Affected Environment

Data & Methods

The project team collected data from the following sources to support analysis of existing soils and geologic conditions and potential effects of the project alternatives:

- Seattle Department of Construction & Inspections GIS (Seattle, City of 2021)
- Geology of Seattle, Washington (Galster and Laprade 1991)
- Quaternary geology of Seattle (Troost et al. 2003)
- Geologic Map of Seattle (Troost et al. 2005)
- Geology of Seattle and the Seattle area, Washington (Troost and Booth 2008)

Current Policy & Regulatory Frameworks

Geologic hazard areas and historical landfills that can impact site development are defined in the City's environmentally critical areas code (Seattle Municipal Code (SMC) 25.09.012 and include:

- Seismic hazard areas (include liquefaction-prone areas, areas subject to ground shaking from seismic hazards addressed by Building and Construction Codes under <u>Title 22</u>, the Seattle Fault Zone, shorelines that could be impacted by Tsunamis, and waterbodies that could be impacted by a seiche [a standing wave oscillating in a body of water])
- Sleep slopes (areas with an incline of 40% or more within a vertical elevation change of at least 10 feet).
- Landslide-prone areas (areas with indications of past landslide activity, and areas with signs
 of potential landsliding).

- Liquefaction-prone areas (areas typically underlain by cohesionless soils of low density, usually in association with a shallow groundwater table, which lose substantial strength during earthquakes)
- Peat-settlement-prone areas (sites containing peat and organic soils that may settle when the area is developed, or the water table is lowered)
- Historical landfills (includes areas with buried solid waste identified by the Seattle-King County Health Department, and areas within 1,000 feet of methane-producing landfills [Seattle-King County Department of Public Health. 1984])
- Seattle Municipal Code 25.09.220 (Environmentally Critical Areas Code) indicates that development on historical landfills is subject to Seattle-King County Health Department requirements. The code also specifies methane barriers or appropriate ventilation per Title 22, Subtitle I, Building Code, and the Seattle King County Health Department regulations.
- The Title 10 King County Board of Health Solid Waste Regulation governs construction standards and methane controls on historical landfills. Authority is established under RCW Chapter 70.05 and Washington State Administrative Code WAC 173-304, Minimal Functional Standards for Solid Waste Handling, and WAC 173-351, Criteria for Municipal Solid Waste Landfills.

Current Conditions

Geology

Seattle is located within the southwestern portion of the Puget Sound Lowland physiographic region, a basin located between the Olympic Mountains to the west and the Cascade Range to the east (Troost et al., 2003; Troost and Booth, 2008). Seattle's geology has been shaped by multiple processes with movement of materials caused by glaciers, rivers, volcanoes, earthquakes, landslides, coastal deposition and erosion, and human activities. A high degree of geological complexity and variation is frequently encountered on development projects within Seattle and subsurface conditions often change significantly and unpredictably over short distances. These conditions cause challenges for project planners who must consider multiple geological concerns for a single project.

At least seven glaciations have impacted the Seattle area within the last 2.4 million years (Troost and Booth, 2008). Near-surface geology in Seattle is dominated by sediments associated with the advance and retreat of Vashon Glaciation, the most recent icesheet that reshaped our region's topography around 15,000 to 13,500 years ago (Galster and Laprade 1991). As this icesheet advanced and retreated over the Puget Sound Lowland, it left behind a complex mix of geologic materials including advance outwash deposits (silt, sand, and gravel); dense glacial till (a random mixture of clay, silt, sand, and gravel); and recessional outwash (stratified deposits of sand and gravel).

The Ballard Subarea includes areas with Vashon till, recessional outwash, and artificial fill overlying the till, recessional outwash, and alluvium deposits. The Interbay Dravus Subarea

includes Pre-Fraser glacial deposits of firm interbedded sand, gravel, and silt on the north; alluvium deposits along the ship canal on the northeast, and large areas of artificial fill overlying tideflat deposits in the central part of the subarea. The Interbay Smith Cove Subarea is dominated almost entirely by artificial fill overlying tideflat deposits, with very small areas of Vashon till, recessional outwash, or other geologic units. The SODO/Stadium Subarea is similarly dominated by artificial fill overlying tideflat deposits, peat, and alluvium. The Georgetown/South Park Subarea is dominated by artificial fill overlying alluvium deposits, including younger alluvium containing peat lenses.

All of the subareas contain areas dominated by or with some history of artificial fill. These areas tend to contain alluvial or sandy soil conditions that could be subject to greater movement and/or liquefaction during major earthquake events.

Geologic Hazards or Limitations

Geologic hazards defined under **Current Policy & Regulatory Frameworks** above are found in each of the subareas as summarized in **Exhibit 3.1-1**. Maps of the BINMIC and Greater Duwamish MIC and geologic hazards are shown in **Exhibit 3.1-2** and **Exhibit 3.1-3**. Descriptions of the hazards follow the table and maps.

	by Subarca
Subarea	Geologic Hazards or Limitations
Ballard	 Short steep slope area along Shilshole Avenue NW Known areas of historical artificial fill A small liquefaction-prone area south of Leary Way NW One historical landfill located just south of Shilshole Avenue NW (no methane buffer)
Interbay Dravus	 Several steep slopes and landslide-prone-areas along the east and west edges of the study area Known areas of historical artificial fill Nearly all of the study area is prone to liquefaction The Interbay Landfill located adjacent to the MIC at Interbay Golf Course, with 1,000-foot methane buffer extending into the MIC
Interbay Smith Cove	 Several steep slopes and landslide-prone-areas along the east and west edges of the study area Known areas of historical artificial fill Nearly all of the study area is prone to liquefaction The Interbay Landfill located adjacent to the MIC at Interbay Golf Course, with 1,000-foot methane buffer extending into the MIC
SODO/Stadium	 A few steep slopes along the west side of Harbor Island Known areas of historical artificial fill Nearly all of the study area is prone to liquefaction Two historical landfills: the West Seattle Landfill along Harbor Avenue SW (with 1,000-foot methane buffer), and a second unnamed landfill that straddles 6th Avenue South.
Georgetown/ South Park	 Several steep slopes and landslide-prone areas along the east and west edges of the study area Known areas of historical artificial fill Nearly all of the study area is prone to liquefaction One peat-settlement-prone area near the far southeast corner, just west of State Route 99 The South Park Landfill located south of the South Transfer Station with 1,000-foot methane buffer

Exhibit 3.1-1 Summary of Geologic Hazards Mapped in the BINMIC and Greater Duwamish MIC by Subarea

Source: Herrera, 2021.



Exhibit 3.1-2 Geologic Hazards Mapped in the BINMIC

Source: Seattle, City of 2021.



Exhibit 3.1-3 Geologic Hazards Mapped in the Greater Duwamish MIC

Source: Seattle, City of 2021.

Seismic Hazards

Seismic hazards exist within the study area. Seattle and the surrounding region are located-in a seismically active region and Seattle sits atop the Seattle Fault Zone (SFZ), a major east-west trending fault zone (WDNR 2020a; USGS 2014). The SFZ consists of a series of closely spaced east-west faults with the exact locations unknown because few clear surface features are visible. The SFZ runs roughly parallel to Interstate 90 from southern Bainbridge Island, through south Seattle, across Lake Washington, and into the Bellevue area and beyond (Exhibit 3.1-2 and Exhibit 3.1-3).

Earthquake recurrence in the Puget Lowland is also influenced by the Cascadia Subduction Zone (CSZ), where the ocean crust off the Pacific Coast is sinking beneath the North American continental plate approximately 70–100 miles off the shoreline. The CSZ has four segments, with the Juan de Fuca plate off the coasts of Washington and Oregon being the segment located closest to CHRLF. The magnitude of an earthquake located along the CSZ varies depending on how many sections of the plate boundary fault are involved, the depth and location of the earthquake epicenter, and the amount of seismic displacement (Rogers 1988; WGCEP 2003).

Steep Slopes

Steep slopes are mapped in several places along the east and west edges of the Interbay Dravus and Interbay Smith Cove subareas (i.e., along the edges of Southeast Magnolia, North Queen Anne, and West Queen Anne). Steep slopes are mapped only in a few small areas in the Ballard Subarea along Shilshole Avenue NW. A few steep slopes are mapped along the west side of Harbor Island in the SODO/Stadium Subarea, and several steep slopes are mapped along the east and west edges of the Georgetown/South Park Subarea above Airport Way South and West Marginal Way, respectively.

Landslide-Prone-Areas

Landslide-prone-areas overlap closely with the steep slope areas described above except for Harbor Island, but they are more extensive in the north-south extents where they present hazards to development.

Liquefaction-Prone Areas

Mapped liquefaction-prone areas include a small portion of the Ballard Subarea south of Leary Way NW, and nearly all of the Interbay Dravus, Interbay Smith Cove, SODO/Stadium, and Georgetown/South Park subareas.

Peat-Settlement-Prone Areas

Only one peat-settlement-prone area is mapped near the far southeast corner of the Georgetown/South Park Subarea, just west of State Route 99.

Historical Landfills

Five historical landfills are mapped within or directly adjacent to the subareas. An unnamed landfill is located in the Ballard Subarea just south of Shilshole Avenue NW and does not include a 1,000-foot methane buffer. The Interbay Landfill is located beneath Interbay Golf Course and includes a 1,000-foot methane buffer that extends into the Interbay Dravus and Interbay Smith Cove subareas. The West Seattle Landfill and an unnamed landfill are located in the SODO/Stadium Subarea along harbor Avenue SW and straddling 6th Avenue South, respectively. The West Seattle Landfill has a 1,000-foot methane buffer, while the unnamed landfill beneath 6th Avenue South does not. And finally, the South Park Landfill is located along West Marginal Way and 5th Avenue South in the Georgetown/South Park Subarea.

The methane buffer is meant to allow for methane gas monitoring/mitigation. Landfills and other areas containing solid waste, refuse, or artificial fill soils, or lands substantially modified by humans can be challenging to develop due to poor or unpredictable soil characteristics. The construction potential of artificial fill areas depends on construction techniques and material type of the fill. Fill material unsuitable for construction may need to be removed or remediated to prevent problems such as settlement or expansion. Landfills may be unable to support the weight of buildings or structures and methane mitigation and monitoring may be required on and within 1,000 feet of landfills.

3.1.2 Impacts

Impacts Common to All Alternatives

None of the alternatives would accelerate or create geologic hazards; future development would need to be designed to respond to potential hazards consistent with adopted building codes to reduce risk of damage or injury. The study area is located within the Puget Sound Region, an area susceptible to moderately high seismic activity. During a seismic event, the study area might be subjected to high-level ground motions. Areas with steep slopes might experience seismic slope stability problems.

Portions of the Ballard and Interbay Dravus subareas, and all of the Interbay Smith Cove, SODO/Stadium, and Georgetown/South Park subareas are susceptible to liquefaction. During an earthquake, vertical and lateral displacements of structures, embankments, <u>water and</u> <u>sewer lines, roads,</u> and paved areas might occur due to seismic liquefaction hazard. The liquefaction potential of mapped liquefaction hazard areas would be confirmed during the design stage of proposed development, regardless of the alternative. <u>In addition to structures</u>, <u>all water</u>, <u>wastewater</u>, transportation, and other infrastructure associated with new <u>development and redevelopment would be carefully designed with input from site-specific</u> <u>geotechnical investigations to lessen and withstand the effects of earthquakes and liquefaction</u>. Development on or adjacent to any of the five historical landfills located within the study areas would require special planning and design. This could include installing methane barriers or appropriate ventilation per Title 22 of the Seattle Municipal Code 25.09.220 and the Seattle King County Health Department regulations. In addition, geotechnical studies would be completed to inform the design of structures and account for poor or unpredictable soil characteristics that could cause settling. These structural features can include the use of pile-supported or floating foundations, depending on the building type.

A peat settlement-prone area in the southwest portion of the Georgetown/South Park Subarea could limit the possibility of development and maintenance of existing structures with any of the alternatives. In this area, compressible soils might need to be excavated and replaced, or planned structures, embankments, and pathways might need to be supported on deep foundations.

All alternatives would allow development that could disturb soils, but erosion would be minimized using erosion control measures per suggested BMPs prescribed in Construction Stormwater Pollution Prevention Plans prepared for each development project.

Alternative 1 would allow the least new jobs and housing and Alternative 4 the most in each subarea and across the whole subarea. See **Exhibit 3.1-4** and **Exhibit 3.1-5**.



Exhibit 3.1-4 Existing and Net Employment Building Space by Alternative

Note: Existing based on Assessor Records. Alternatives assume 700 square feet per industrial employee and 250 square feet per nonindustrial employee similar to buildable lands assumptions. <u>This chart was updated to include the Preferred Alternative</u>. Source: City of Seattle, 202<u>24</u>; BERK, 202<u>24</u>.



Exhibit 3.1-5 Total Housing in Study Area by Alternative

Ballard

The Ballard Subarea would have the lowest growth under the Alternative 1 No Action and greatest under Alternative 4. This subarea has a small area prone to liquefaction and an historical landfill. The risk of erosion that could not be contained, or risk of damage to structures or injury from landslides, settlement, or seismic events is considered significant but avoidable with mitigation.

Interbay Dravus

The Interbay Dravus Subarea would have the lowest growth under Alternative 1 No Action and the most under Alternative 4. Approximately half of this subarea is prone to liquefaction and there are two areas with steep slopes and one area with potential landslide hazards. The southern portion of this subarea also lies within the 1,000-foot methane buffer of the Interbay Landfill. The risk of erosion that could not be contained, or risk of damage to structures or injury from landslides, settlement, or seismic events is considered significant but avoidable with mitigation.

Interbay Smith Cove

The Interbay Smith Cove Subarea would have the lowest growth under Alternative 1 No Action and the most under Alternative 4. All of this subarea is prone to liquefaction and potential landslide areas are located along the east and west edges. The Interbay Landfill and a large portion of the associated 1,000-foot methane buffer is located in the northern part of this subarea. The risk of erosion that could not be contained, or risk of damage to structures or injury from landslides, settlement, or seismic events is considered significant but avoidable with mitigation.

Source: City of Seattle, 202<u>2</u>4; BERK, 202<u>2</u>4.

SODO/Stadium

The SODO/Stadium Subarea would have the lowest growth under Alternative 1 No Action and the most under Alternative 4. All of this subarea is prone to liquefaction and both known and potential landslide areas are located along the east and west edges. Two landfills are located within this subarea; the West Seattle Landfill has a 1,000-foot methane buffer, while the unnamed landfill beneath 6th Avenue South does not. The risk of erosion that could not be contained, or risk of damage to structures or injury from landslides, settlement, or seismic events is considered significant but avoidable with mitigation.

Georgetown/South Park

The Georgetown/South Park Subarea would have the lowest growth under Alternative 1 and the greatest under Alternative 4. All of this subarea is prone to liquefaction. Known and potential landslide areas are located along the east and west edges, and steep slopes are located along the west edge. The South Park landfill with 1,000-foot methane buffer, and a peat settlement-prone area are both located within this subarea. The risk of erosion that could not be contained, or risk of damage to structures or injury from landslides, settlement, or seismic events is considered significant but avoidable with mitigation.

Equity & Environmental Justice Considerations

The population in the BINMIC portion of the study area are less disadvantaged than the population in the Greater Duwamish MIC which has the highest and middle disadvantage per the Seattle Racial and Social Equity Index. See **Exhibit 1.7-7**.

Under any of the Action Alternatives, the primary equity and environmental justice concern for the proposal would be if development on lands subject to geologic hazards carries the risk of injury or damage to structures due to seismic activity. Although the proposal would allow development at sites in areas prone to landslides, liquefaction, or similar geologic hazards, modern building codes mitigate the risk of injury or economic losses for vulnerable communities.

Under Alternative 1 No Action, humans and animals could potentially feel the greatest impacts from geologic hazards in all subareas due to potentially less redevelopment of aging buildings and infrastructure not built to modern building codes to withstand seismic events compared to Action Alternatives.

The Ballard Subarea is less susceptible to seismic impacts than other subareas given nature of the geology that includes deposits of Vashon till, recessional outwash, and artificial fill overlying the till, recessional outwash, and alluvium deposits. The other four subareas are more susceptible to seismic impacts such as liquefaction given the prevalence of large areas of artificial fill overlying tideflat deposits and alluvium deposits, including younger alluvium containing peat lenses. The Action Alternatives would generally have positive long-term benefits. The greatest benefits would be associated with Alternative 4 because it would result in the most sites developed to international building code standards.

Impacts of Alternative 1 No Action

Under Alternative 1 No Action, there would be similar building forms as found today with gradual densification in parts of all subareas. A total of 8,330,000 square feet (SF) of industrial space and 2,900,000 SF of non-industrial space would be developed. Existing dwellings would increase slightly from 413 to 488, or 75 net new units.

Due to the least amount of planned growth and development under the Alternative 1 No Action, there would be the least amount of soil disturbance but also the least number of structures built to modern building codes. The risk of damage or injury would be less in new buildings developed to international building code standards, but fewer buildings would be constructed to the latest standards compared to alternatives 2, 3, and 4.

Impacts of Alternative 2

The impacts of Alternative 2 are similar to those described under Impacts Common to All Alternatives. The total square feet of industrial space developed within the subareas would more than double, from 8,330,000 SF under the No Action Alternative to 17,430,000 SF under Alternative 2; there would be less non-industrial space of 2,375,000 SF under Alternative 2 compared to 2,900,000 SF with Alternative 1 No Action. In addition, the total housing units would increase from 488 under Alternative 1 No Action to 493 under Alternative 2 (80 above existing units, 5 more than Alternative 1 No Action).

This would mean more workers in industrial spaces and slightly more residents living in housing in the subareas. However, there should be less risk of injury or structural damage from geologic hazards than Alternative 1 No Action because structures would be designed to minimize risks consistent with <u>modern</u> building <u>codes</u> and construction standards.

Compared to Alternative 1 No Action, Alternative 2 could create more cut material to be hauled due to taller buildings that might require deeper foundations and potential increase in underground parking needs due to larger buildings. Cut materials in the area are potentially contaminated which would require special handling, storage, transportation, and off-site hauling. The cut materials in the region are known to be moisture sensitive (meaning difficult to compact if they are allowed by become wet) and therefore if not contaminated, cut material should be kept covered to facilitate reuse.

All these impacts together are considered significant but avoidable with mitigation.

Impacts of Alternative 3

The impacts of Alternative 3 are similar to those described under Impacts Common to All Alternatives and under Impacts of Alternative 2. Zoning would change to allow more development of residential properties and non-industrial mixed-use properties. Another 2,870,000 SF of industrial space, 4,725,000 SF of non-industrial space above Alternative 2 (total new 20,300,000 SF industrial and 7,100,000 SF non-industrial).

As well, 2,101 housing units would be developed within the subareas (610 new caretakers' quarters/makers' studios and 1,078 new units in mixed-use in areas removed from the MIC).

This would mean more workers in industrial spaces and more residents living in housing, and more structures that could be exposed to geologic hazards than Alternative 1 No Action, but structures would be designed to minimize risks consistent with <u>modern</u> building <u>codes</u> and construction standards.

All these impacts together are considered significant but avoidable with mitigation.

Impacts of Alternative 4

The impacts of Alternative 4 are similar <u>as-to</u> those described above under Impacts Common to All Alternatives and under Impacts of Alternative 3. The total square footage of industrial space would decrease slightly compared to Alternative 3, but an additional 500,000 SF of nonindustrial space is possible (total new 20,160,000 SF of industrial space and 7,600,000 SF of nonindustrial space). Additionally, 3,273 new housing units would be developed within the subareas (2,195 new caretakers' quarters/makers' studios and 1,078 new units in mixed-use in areas removed from the MIC).

Under Alternative 4, the greatest level of development could be subject to geologic hazards, compared to Alternative 1 No Action, but structures would be designed to minimize risks consistent with <u>modern</u> building <u>codes</u> and construction standards.

All these impacts together are considered significant but avoidable with mitigation.

Impacts of the Preferred Alternative

<u>Under the Preferred Alternative, increases in employment are expected to be similar to</u> <u>Alternative 2, while increases in housing are expected to be similar to alternatives 3 and 4. The</u> <u>share of job growth is increased in the Ballard and Interbay Dravus subareas and greater than</u> <u>alternatives 1 through 4. Job growth is reduced in the Georgetown/South Park Subarea.</u>

<u>Total residential growth under the Preferred Alternative-—including within industrial areas,</u> <u>areas removed from the MIC, and rezoned areas converted to mixed use zoning outside of the</u> <u>MIC—is similar to but lower than Alternative 4. The number of dwellings in industrial areas is</u> <u>projected to increase by 1,475 units in the UI zone, 33% less than the amount studied in Draft</u> EIS Alternative 4, and is concentrated in the Ballard and SODO/Stadium subareas. Outside of the MICs, two new areas in west Ballard and Judkins Park would be converted to mixed use zoning allowing housing, in addition to the proposed mixed-use areas in Georgetown and South Park studied in Draft EIS alternatives 3 and 4; overall, a higher total amount of housing production outside of MICs would result compared to Draft EIS alternatives—an additional 1,534 dwellings, 42% more than alternatives 3 and 4. Under the Preferred Alternative, more development could be subject to geologic hazards, than under alternatives 1, 2, and 3 but less than Alternative 4. However, structures would be designed to minimize risks consistent with modern building codes and construction standards.

With mitigation, all these impacts together would not be considered significant.

3.1.3 Mitigation Measures

Incorporated Plan Features

There are no incorporated plan features related to geology and soils.

Regulations and Commitments

Building and Construction Codes under Title 22 contains construction code standards, including the International Building Code, which ensure buildings are designed to meet seismic safety standards.

Seattle Municipal Code 25.09.220 (Environmentally Critical Areas Code) indicates that development on historical landfills is subject to Seattle-King County Health Department requirements. The code also specifies methane barriers or appropriate ventilation per Title 22, Subtitle I, Building Code, and the Seattle King County Health Department regulations.

The Title 10 King County Board of Health Solid Waste Regulation governs construction standards and methane controls on historical landfills. Authority is established under RCW Chapter 70.05 and Washington State Administrative Code WAC 173-304, Minimal Functional Standards for Solid Waste Handling, and WAC 173-351, Criteria for Municipal Solid Waste Landfills.

Other Potential Mitigation Measures

Geotechnical investigations are required as part of the design phase for new development, especially for those buildings with greater heights or in close proximity to artificially created slopes. Prior to commencing site-specific subsurface investigations of soils, the Duwamish tribe should be notified to ensure that an archaeologist can observe the work. Standard archaeological techniques should be used during excavation and drilling for the potential discovery and preservation of cultural and historical artifacts related to the indigenous tribes.

Any evidence gathered should be presented and turned over to the Duwamish Tribe at the Duwamish Longhouse & Cultural Center.

Specific recommendations for liquefaction mitigation, subgrade preparation, roadway embankment, cut and fill, slope stability, foundation design, retaining structures, and dewatering measures would be prepared prior to construction. Appropriate waste sites for unsuitable excavated soils would be identified prior to construction.

Potential impacts of soil liquefaction could be mitigated by removing and replacing the loose materials with compacted fill materials, by densifying or reinforcing the in-situ soils, or by supporting the proposed facilities on deep foundations or piles. The need for liquefaction mitigation would be evaluated on a case-by-case basis for the individual structural elements potentially impacted.

Potential impacts of vapor intrusion from historical landfills within the study area would be investigated by performing site-specific vapor intrusion assessments and/or by installing passive or active methane mitigation systems in structures developed on historical landfills, or within the 1,000-foot methane buffer.

3.1.4 Significant Unavoidable Adverse Impacts

Development in the study area, as with most locations in Central Puget Sound, would expose population and structures to geologic hazards, and would disturb soils. These impacts can be mitigated to a less than significant level by designing development to the City's adopted construction codes and applying any site-specific conditions (e.g., methane mitigation systems for buildings built near historical landfills) required by the City during permit review.

Section 3.2 Air Quality & GHG



This section assesses the potential air quality and greenhouse gas (GHG) emission impacts associated with implementing the alternatives under consideration.

The air quality section includes a description of regulatory standards for air quality, air emission sources and individual criteria pollutants of concern, with a focus on carbon monoxide (CO), particulate matter (PM) emissions, ozone precursors, and Toxic Air Pollutants (TAPs). The chapter also includes a discussion of potential sensitive populations in and near the industrial and maritime areas of Seattle, the methods used to assess air quality and impacts from those emissions, and an assessment of impacts associated with each alternative, as well as potentially feasible mitigation measures where appropriate. This analysis evaluates air quality conditions and potential impacts for each MIC on an area-wide cumulative basis and, and for PM2.5 and TAPs, a localized analysis is provided at specific areas to identify potential public health impacts from locating new sensitive receptors closer to or within MIC areas.

Under the SEPA Rules (see WAC 197-11-330, WAC 197-11-440 and WAC 197-11-794), the evaluation of the significance of potential impacts considers whether there is a reasonable likelihood of more than a moderate adverse impact on environmental quality (WAC 197-11-794). In making this assessment, the following are considered:

- The context of the proposal, including the physical setting
- The intensity of the impact, which depends on its magnitude and duration
- The likelihood of the impact's occurrence
- The duration of the impact.

In many cases, regulatory thresholds are used to judge significance, that is, if actions would meet regulatory thresholds (e.g., surface water quality standards, wetland/stream buffers, noise standards, endangered species) then the determination is typically that the level of impact is unlikely to be significant. For the purposes of this programmatic impact analysis, air quality is analyzed by examining whether:

• The alternative would prevent or deter achieving the National Ambient Air Quality Standards (NAAQS) for criteria pollutants.

The GHG section includes a description of community goals for GHG emissions and climate change, transportation, and land use emission sources in the industrial and maritime areas of Seattle, the methods used to measure GHG emissions, and how implementation of the alternatives considered may contribute to global climate change. This section also identifies potentially feasible emissions mitigation measures where appropriate. This analysis evaluates potential GHG emission impacts from each alternative on a cumulative basis.

There is no standard significance threshold for GHG emissions in the SEPA rules (WAC 197-11-330). However, Chapter 173-441 WAC requires mandatory GHG reporting for facilities that emit at least 10,000 metric tons of GHGs per year in Washington. For the purposes of this programmatic impact analysis, GHG emissions are analyzed by examining whether:

- The alternative would prevent or deter efforts to reduce emissions in comparison to local or regional goals or targets for GHG reductions.
- The alternative would cause the cumulative difference in GHG emissions between an alternative and Alternative 1 No Action to exceed Washington Department of Ecology's GHG reporting threshold of 10,000 metric tons per year.

The analysis confirms that changes to the MIC areas do not prevent or deter from meeting the National Ambient Air Quality Standards (NAAQS) for criteria pollutants. It illustrates increases in greenhouse gases (GHGs) in comparison to local or regional goals or targets for GHG reductions and identifies mitigation that, if implemented and tracked, could reduce impacts to a less than significant level.

This chapter relies on information that is contained in the Seattle Comprehensive Plan (Seattle 2035) EIS, which incorporated by reference herein. (City of Seattle 2016)

The study area for air quality is defined as the area that could be directly or indirectly affected by the construction activities or land uses that result from implementation of the industrial and maritime strategy. Given that air emissions cross county and state lines, the assessment here is considered to apply to air quality effects over the entire Seattle-King County area. With respect to GHG emissions and its effect on climate, the study area is the global environment. The study area for indirect impacts is the area affected by the transport of construction workers and materials to the project area.

3.2.1 Affected Environment

Data & Methods

The project team collected data from the following sources to support analysis of existing air quality conditions and potential effects of the project alternatives:

- U.S. Environmental Protection Agency (EPA) Greenbook (EPA 2021)
- Puget Sound Clean Air Agency (PSCAA) and Ecology Air Monitoring Network
- 2019 PSCAA Air Quality Data Summary (PSCAA 2019)
- 2016 Puget Sound Maritime Emissions Inventory (PSMEI 2018)
- Duwamish Valley Regional Modeling and Health Risk Assessment (WDOH 2008)
- Washington State Greenhouse Gas Emissions Inventory 1990–2018 (Ecology 2021)
- 2018 Community Greenhouse Gas Emissions Inventory (Seattle 2018)
- Direct monitoring of eight sites within the BINMIC and Greater Duwamish MIC during 2021

Air Quality

Current Policy & Regulatory Frameworks

Air quality in the Puget Sound region is regulated and enforced by federal, state, and local agencies—the Environmental Protection Agency (EPA), Department of Ecology (Ecology), and the Puget Sound Clean Air Agency (PSCAA); each have their own role in regulating air quality.

<u>U.S. EPA</u>

The 1970 Clean Air Act established National Ambient Air Quality Standards (NAAQS), with primary and secondary standards, to protect the public health and welfare from air pollution. As required by the Clean Air Act, EPA identified Ozone, CO, PM, nitrogen dioxide (NO2), sulfur dioxide (SO2) and lead as the six criteria air pollutants. Since then, subsets of PM have been identified for which permissible levels have been established. These include PM10 (particles that are less than or equal to 10 microns in diameter) and PM2.5 (particles that are less than or equal to 2.5 microns in diameter).

The NAAQS set limits on concentration levels of the criteria pollutants in the air. Concentration levels of the criteria pollutants must not exceed the NAAQS over specified time periods. These ambient air quality standards are designed to protect those segments of the public most susceptible to respiratory distress, including asthmatics, the very young, the elderly, people whose health is compromised from other illness or disease, or those engaged in strenuous work or exercise. Areas of the U.S. that do not meet the NAAQS for any pollutant are designated by the EPA as nonattainment areas. Areas that were once designated nonattainment but are now achieving the NAAQS are termed maintenance areas. Areas that have air pollution levels below the NAAQS are termed attainment areas. In nonattainment areas, states must develop plans to reduce emissions and bring the area back into attainment of the NAAQS.

The Clean Air Act also requires the EPA to regulate toxic air pollutants (or air toxics) from mobile sources and large industrial facilities. Air toxics are air pollutants known or suspected to cause health problems, including cancer. EPA's primary effort focuses on developing standards for controlling the emissions of air toxics from sources in industry groups (or source categories). These maximum achievable control technology (MACT) standards are based on emissions levels that are already being achieved by the controlled and low emitting sources in an industry.

Washington Department of Ecology

Ecology maintains an air quality program with a goal of safeguarding public health and the environment by overseeing the development and conformity of the State Implementation Plan (SIP), which is the state's plan for meeting and maintaining NAAQS. In addition to the NAAQS standards, Ecology has adopted state ambient air quality standards for 1-hour ozone

concentrations and its own more stringent air quality standards for annual NO2, SO2 and PM concentrations. Ecology also monitors air quality in the Puget Sound Region by measuring the levels of criteria pollutants found in the atmosphere and comparing them with the NAAQS. Ecology has also monitored 17 air toxics since 2000 in Seattle at a site on Beacon Hill.

Puget Sound Clean Air Agency

The PSCAA has local authority for setting regulations and permitting of stationary air pollutant sources and construction emissions. PSCAA also maintains and operates a network of ambient air quality monitoring stations measuring the levels of criteria pollutants found in the atmosphere throughout its jurisdiction. The NAAQS are summarized in **Exhibit 3.2-1**.

Pollutant	Averaging Times	Primary NAAQS	Secondary NAAQS	
Carbon monoxide	8-hour a	9 ppm (10 mg/m3)	None	
	1-hour a	35 ppm (40 mg/m3)	None	
Lead	Rolling 3-Month Average	0.15 µg/ m3	Same as Primary	
	Quarterly Average	1.5 µg/ m3	Same as Primary	
NO2	Annual (Arithmetic Mean)	0.053 ppm (100 µg/m3)	Same as Primary	
	1-hour b	0.100 ppm (188 ug/m3)	Same as Primary	
PM10	24-hour c	150 µg/m3	Same as Primary	
PM2.5	Annual d (Arithmetic Mean)	12.0 µg/m3	Same as Primary	
	24-hour e	35 μg/m3	Same as Primary	
Ozone	8-hour f	0.075 ppm (2008 std.)	Same as Primary	
	8-hour f	0.070 ppm (2015 std.)	Same as Primary	
SO2	3-hour a	none	0.5 ppm (1,300 μg/m3)	
	1-hour g	0.075 ppm (196 ug/m3)	Same as Primary	

Exhibit 3.2-1 National Ambient Air Quality Standards

Notes: $\mu g/m3 = micrograms$ per cubic meter; NO2 = nitrogen dioxide; PM2.5 = particulate matter less than or equal to 2.5 micrometers in diameter; PM10 = particulate matter less than or equal to 10 micrometers in diameter; ppm = parts per million; ppb = parts per billion; SO2 = sulfur dioxide.

A Not to be exceeded more than once per year.

B Standard is attained when the 3-year average of the eighth-highest daily maximum 1-hour average NO2 concentration does not exceed 0.100 ppm (100 ppb).

C Not to be exceeded more than once per year on average over 3 years.

D To attain this standard, the 3-year average at any monitor must not exceed 12.0 μ g/m3.

E To attain this standard, the 3-year average of the 98th percentile of 24-hour concentrations at each population-oriented monitor within an area must not exceed 35 µg/m3. PSCAA maintains a stricter standard for PM2.5 of 35 µg/m3

f To attain this standard, the 3-year average of the fourth-highest daily maximum 8-hour average ozone concentrations measured at each monitor within an area over each year must not exceed the standard. While both the 2008 and 2015 standards are still in place, the 2015 standard is the controlling one, given its greater stringency.

G Standard is attained when the 3-year average of the fourth-highest daily maximum 1-hour average NO2 concentration does not exceed 0.100 ppm (100 ppb).

Source: 40 Code of Federal Regulations part 50, EPA 2016.

Pollutants of Concern

This section discusses the main pollutants of concern and their impact on public health and the environment. Air quality is affected by pollutants that are generated by both natural and human sources. In general, the largest human sources of air emissions are transportation vehicles and power-generation, both of which typically burn fossil fuels. Criteria air pollutants are carbon monoxide (CO); particulate matter (PM); ozone, and the ozone precursors (volatile organic compounds [VOCs] and oxides of nitrogen [NOX]); sulfur dioxide (SO2); and lead. Both federal and state standards regulate these pollutants. Industrial sources such as metal processing are currently the primary source of lead emissions.

The largest contributors of pollution related to land development activity are construction equipment, motor vehicles and off-road construction equipment. The main pollutants emitted from these sources are CO, PM, ozone precursors (VOC and Nox), GHGs, and mobile source air toxics (MSATs). Motor vehicles and diesel-powered construction equipment also emit pollutants that contribute to the formation of ground-level ozone.

Carbon Monoxide

CO is an odorless, colorless gas usually formed as the result of the incomplete combustion of fuels. The largest sources of CO are motor vehicle engines and traffic, and industrial activity and woodstoves. Exposure to high concentrations of CO reduces the oxygen-carrying capacity of the blood and can cause headaches, nausea, dizziness, and fatigue; impair central nervous system function; and induce angina (chest pain) in persons with serious heart disease. Very high levels of CO can be fatal. The federal CO standards have not been exceeded in the Puget Sound area for the past 20 years (PSCAA 2019).

<u>Lead</u>

Lead is a highly toxic metal that was used for many years in household products such as paints, transportation fuel, and industrial chemicals. With lead now excluded from paint and most fuels, most lead emissions nationally are industrial processes and battery manufacturers though lead found in aviation fuel used by small aircraft remains a concern nationally. In October 2008, EPA strengthened the lead standard from 1.5 µg/m3 to 0.15 µg/m3 (rolling three-month average; PSCAA 2020).

<u>Ozone</u>

Ozone is a secondary air pollutant produced in the atmosphere through a complex series of photochemical reactions involving VOCs and Nox. The main sources of VOC and Nox—ozone precursors—are combustion processes (including motor vehicle engines) and the evaporation of solvents, paints, and fuels. Ozone levels are usually highest in the afternoon because of the intense sunlight and the time required for ozone to form in the atmosphere. Ecology currently monitors ozone from May through September because this is the period of concern for

elevated ozone levels in the Pacific Northwest. No violations of the NAAQS for ozone have occurred at the Seattle monitoring station since monitoring commenced there in 1999.

Elevated concentrations of ground-level ozone can cause reduced lung function and respiratory irritation and can aggravate asthma. Ozone has also been linked to immune system impairment. People with respiratory conditions should limit outdoor exertion if ozone levels are elevated. Even healthy individuals may experience respiratory symptoms on a high-ozone day. The Puget Sound region is designated as an attainment area for federal ozone standards.

Nitrogen Dioxide

Nitrogen dioxide (NO2) is a reddish brown, highly reactive gas that forms from the reaction of nitrogen oxide (NO) and free radicals in the atmosphere. NO2 can cause coughing, wheezing and shortness of breath in people with respiratory diseases such as asthma and long-term exposure can lead to respiratory infections.⁷

The term Nox is defined as NO + NO2. Nox participates in a complex chemical cycle with volatile organic compounds (VOCs) which can result in the production of ozone. Nox can also be oxidized to form nitrates, which are an important component of fine particulate matter. On-road vehicles such as trucks and automobiles and off-road vehicles such as construction equipment, marine vessels and port cargo-handling equipment are the major sources of Nox in Seattle's industrial areas. Industrial boilers and processes, home heaters, and gas stoves also produce Nox (PSCAA 2020).

Particulate Matter

PM is a class of air pollutant that is a mix of solid and liquid particles from human and natural sources. PM is measured in two size ranges: PM10 and PM2.5. Fine particles are emitted directly from a variety of sources, including wood burning (both outside and indoor wood-burning stoves and fireplaces; and wildfire), vehicles (both vehicle emissions and from generation of fugitive roadway dust) and industry. They also form when gases from some of these same sources react in the atmosphere.

Exposure to particle pollution is linked to a variety of significant health problems, such as increased hospital admissions and emergency department visits for cardiovascular and respiratory problems, including non-fatal heart attacks and premature death. People most at risk from fine and coarse particle pollution exposure include those with chronic heart and lung disease (like asthma, bronchitis, and emphysema), children, and the elderly. It worsens these diseases, which can lead to hospitalization or even early death. Pregnant women, newborns, and people with certain health conditions, such as obesity or diabetes, also may be more susceptible to PM-related effects.

⁷ EPA Airnow, NOX Chief Causes for Concern; <u>www.epa.gov/airquality/nitrogenoxides/.</u>

The federal annual PM2.5 standard has not been exceeded in the Puget Sound area since the U.S. EPA established its NAAQS in 2007. The daily federal PM2.5 standard has not been exceeded in the Puget Sound dating back to the initiation of monitoring for this pollutant in 2001 (PSCAA 2014). The U.S. EPA adopted a more stringent federal standard for PM2.5 in December 2012, and Seattle-King County is designated as an attainment area. Portions of the Puget Sound region, including an area encompassing the Greater Duwamish MIC, were designated as a maintenance area for PM10 through May 2021.

Sulfur Dioxide

Sulfur dioxide (SO2) is a colorless, reactive gas produced by burning fuels containing sulfur, such as oil, coal, and diesel, and by industrial processes. Historically, the greatest sources of SO2 were industrial facilities that derived their products from raw materials such as metallic ore, coal, and crude oil, or that burned coal or oil to produce process heat (petroleum refineries, cement manufacturing and metal processing facilities). Marine vessels, on-road vehicles, and diesel construction equipment are the main contributors to SO2 emissions today. Historically, Washington has measured very low levels of SO2. Because the levels were so low, most monitoring was stopped.

SO2 may cause people with asthma who are active outdoors to experience bronchial constriction, the symptoms of which include wheezing, shortness of breath and tightening of the chest. People should limit outdoor exertion if SO2 levels are high. SO2 can also form sulfates in the atmosphere, a component of fine particulate matter (PSCAA 2020).

<u>Toxic Air Pollutants</u>

Air toxics are defined by Washington State and PSCAA to include hundreds of chemicals and compounds that are associated with a broad range of adverse health effects, including cancer. Many air toxics are a component of either particulate matter or volatile organic compounds (a precursor to ozone).

There are no ambient air quality standards for toxic air pollutants. PSCAA is working with state, local, and tribal governments to reduce air toxics releases. While there are no ambient standards, there are several regulatory tools that are used to reduce air toxics emissions. These tools include: national regulations on industrial sources that require emission reducing technology, "new source review" for sources in Washington State, local regulations for specific industries that require specific technology, and national regulations to reduce emissions from mobile sources (including cars, trucks, and buses as well as marine vessels and locomotives; WDOH 2008)

Ecology began monitoring air toxics at the Seattle Beacon Hill site in 2000. The Clean Air Act identifies 188 air toxics; the U.S. EPA later identified 21 of these air toxics as mobile source air toxics (MSATs) and then a subset of seven priority MSATs: benzene, formaldehyde, diesel particulate matter/diesel exhaust organic gases, acrolein, naphthalene, polycyclic organic matter, and 1,3-butadiene. Exposure to these pollutants for long durations and sufficient

concentrations increases the chances of cancer, damage to the immune system, neurological problems, reproductive, developmental, respiratory, and other serious health problems.

Diesel particulate matter poses the greatest potential cancer risk (70% of the total risk from air toxics) in the Puget Sound area (PSCAA 2011). This pollution comes from diesel-fueled trucks, cars, buses, construction equipment, rail, marine and port activities. Particulate matter from wood smoke (a result of burning in woodstoves and fireplaces or outdoor fires) presents the second-highest potential cancer health risk. Wood smoke and auto exhaust also contain formaldehyde, chromium, benzene, 1,3-butadiene and acrolein. Chromium is also emitted in industrial plating processes.

Current Conditions

Puget Sound Climate & Air Quality

The City of Seattle is in the Puget Sound lowland and the region has a relatively mild, marine climate with cool summers and mild, wet, and cloudy winters. The prevailing wind direction in the summer is from the north or northwest. The average wind velocity is less than 10 miles per hour. Persistent high-pressure cells often dominate summer weather and create stagnant air conditions. This weather pattern sometimes contributes to the formation of photochemical smog. During the wet winter season, the prevailing wind direction is south or southwest.

Although the Puget Sound region contains some of the most densely populated and industrialized areas in Washington, there is sufficient wind most of the year to disperse air pollutants released into the atmosphere. Air pollution is usually most noticeable in the late fall and winter, under conditions of clear skies, light wind, and a sharp temperature inversion. Temperature inversions occur when cold air is trapped under warm air, thereby preventing vertical mixing in the atmosphere. These can last several days. If poor dispersion persists for more than 24 hours, the PSCAA can declare an "air pollution episode" or local "impaired air quality."

Regionally, weather conditions such as temperature, fog, rain, and snowfall can vary within short distances, influenced by such factors as the distance from Puget Sound, the rolling terrain, and air from the ocean moving inland. Wildfires typically occur during the warmer, drier summer months and recent years have seen increased incidence of more dense smoke episodes lasting days or weeks. Wildfire smoke carries the same health risks as wood smoke because of the presence of small particles, which can be especially dangerous for infants, children, and people over 65, or those that are pregnant, have heart or lung diseases (such as asthma or COPD), respiratory infections, diabetes, stroke survivors, and those suffering from COVID-19. (PSCAA 2021)

Full Study Area

Both Ecology and PSCAA operate ambient air quality monitoring stations to assess the levels of regulated pollutants and to verify continued compliance with the NAAQS. The monitoring

stations used for this analysis are the nearest to the BINMIC and Greater Duwamish MIC areas and shown in **Exhibit 3.2-2** along with the criteria pollutants monitored.

Site	Owner	PM2.5	Ozone	CO	SO2	Nox
10 th & Weller	Ecology	٠		•		٠
Beacon Hill Site, 4103 Beacon Ave S	Ecology	٠	•	٠	٠	•
Duwamish Site, 4700 East Marginal Way	PSCAA	٠				
South Park Site, 8201 10 th Ave S	PSCAA	•				
Tukwila Allentown Site, 11675 44 th Ave E	PSCAA	٠				

Source: PSCAA, 2021.

In addition, eight sites within the BINMIC and Greater Duwamish MIC were monitored directly to provide additional baseline data on ambient air quality conditions for this EIS. The sites are described below and **Exhibit 3.2-3** shows the site locations. They were selected due to the location of potential zoning changes in alternatives at the time of Scoping or due to their proximity to air quality emission sources.

- 1. Ballard: 5007 14th Avenue Northwest. This site is also close to the future Sound Transit light rail station.
- 2. Interbay/Dravus: 3425 16th Avenue West. This is also close to a future Sound Transit light rail station, a BNSF rail yard, and facilities.
- 3. Interbay/Armory site: 1561 W Armory Way. This is a site that is close to the BNSF rail yard.
- 4. Stadium area: 1730 1st Avenue South
- 5. Georgetown: 5707 Airport Way South.
- 6. South Park 1: 8620 16th Avenue South. An area close to the King County airport
- 7. SODO/Lander: 2437 6th Avenue South. An existing light rail station.
- 8. South Park 2: 8100 8th Avenue South. An area in proximity to SR 99 and SR 509.


Exhibit 3.2-3 Air Quality Monitoring Locations

Source: Herrera, 2021.

Air Quality Information and Trends

According to PSCAA, over the last two decades, many pollutant levels have declined, and air quality has improved overall. For Seattle area monitoring stations closest to the MICs, as it is within the Puget Sound area overall, the following trends exist:

- Carbon monoxide: CO has been declining, primarily due to improvements made to emission controls on motor vehicles and the retirement of older, higher-polluting vehicles. Reductions in motor vehicle emissions have occurred despite comparative increases in demographics (i.e., population, licensed drivers, registered vehicles) over the past 40+ years.
- Lead: Since the phase-out of lead in most fuels and the closure of the Harbor Island secondary lead smelter in Seattle in 1984, levels of lead in ambient air have decreased substantially.
- Ozone, and the ozone precursors (volatile organic compounds (VOCs) are at their highest concentrations in the communities downwind of large urban areas. In the Puget Sound region, the hot sunny days favorable for ozone formation also tend to have light north-to-northwest winds. Ozone levels remain a concern in the region, as measured ozone concentrations have remained fairly static since 2010 (PSCAA 2020)
- Nox: Motor vehicle and non-road engine manufacturers have been required by EPA to reduce Nox emissions from cars, trucks, and non-road equipment. As a result, emissions have declined dramatically since the 1970s. Nitrogen dioxide levels in the Puget Sound region, as currently monitored by Ecology, are typically below (cleaner than) EPA's 1-hour standard and are trending slightly downward in the last 10 years (PSCAA 2020). (PSCAA 2020)
- Particulate matter (PM): Elevated fine particle levels (PM2.5) pose the greatest air quality challenge in the region. Fine particle levels met the EPA's health-based standard of 35 micrograms per cubic meter in 2019 when days with wildfire smoke are excluded, though when wildfire is included some monitoring sites exceeded the federal standard in 2017 and 2018. There were no wildfire-impacted days in 2019.
- PSCAA's more stringent local PM2.5 health goal of 25 micrograms per cubic meter was exceeded on 22 days in winter months at Seattle monitoring sites (PSCAA 2020).
- Sulfur dioxide (SO2): The Puget Sound area has experienced a significant decrease in SO2 because control measures were added for some sources (e.g., cement plants), some larger SO2 sources shut down (e.g., pulp mills and smelters) and the sulfur content of gasoline and diesel fuel was cut by nearly 90% (Ecology 2011b) and continues to be far below the federal NAAQ Standard. (PSCAA 2020).
- Air toxics: Some air toxics continue to be measured at levels known to cause adverse health effects. These health effects include, but are not limited to, increased cancer risk, respiratory effects, and developmental effects. (PSCAA 2020)

Overall, the air quality in the Puget Sound has continued to improve to meet the standards, though the number of wildfire-impacted days has increased in the last five years.

Ambient air concentrations of the monitored pollutants for years 2018 through 2020 are summarized in **Exhibit 3.2-4** and shows that the air pollutant concentration trends for these pollutants remain below the NAAQS when wildfire is excluded. Ecology and PSCAA no longer collect particulate matter smaller than 10 microns in diameter (PM10) data in the Puget Sound Region.

Exhibit 3.2-4 Ambient Criteria Pollutant Concentration Levels Measured for the four Seattle Sites (10th & Weller, Beacon Hill, Duwamish, and South Park) from 2018-2020

	During a way (A				fire Incl	uded	Wildf	ire Excl	uded
Pollutant	Secondary	Time	NAAQS	Form	2018	2019	2020	2018	2019	2020
Carbon	primary	8 hours	9 ppm	Not to be exceeded more than once per year	1.4	1.7	1.8	nc	nc	nc
Monoxide (CO)		1 hour	35 ppm			2.3	2.1	nc	nc	nc
Lead (Pb)	primary and secondary	Rolling 3 month average	0.15 ug/m3	Not to be exceeded	nm	nm	nm	nm	nm	nm
Nitrogen Dioxide (NO2)	primary	1 hour	100 ppb	98 th percentile of 1-hour daily maximum concentrations, averaged over 3 years	62.6*	62.1*	59.2	nc	nc	nc
	primary and secondary	1 year	53 ppb	Annual mean		18.1	15.8	nc	nc	nc
Ozone (O3)	primary and secondary	8 hours	0.070 ppm	Annual fourth-highest daily maximum 8-hour concentration, averaged over 3 years	0.045	0.046	0.052	nc	nc	nc
PM2.5	primary	1 year	12.0 ug/m3	annual mean, averaged over 3 years	8.9**	9.3	9.1	7.9**	8.2	8.0
	secondary	1 year	15.0 ug/m3		8.9**	9.3	9.1	7.9**	8.2	8.0
	primary and secondary	25 hours	35 ug/m3	98 th percentile, averaged over 3 years	37.2**	36.7	37.5	20.7	21.5	19.3
PM10	primary and secondary	24 hours	150 ug/m3	Not to be exceeded more than once per year on average over 3 years	nm	nm	nm	nm	nm	nm
SO2	primary	1 hour	75 ppb	99 th percentile of 1-hour daily maximum concentrations, averaged over 3 years	6.0***	7.0***	6.0	nm	nm	nm
	secondary	3 hours	0.5 ppm	Not to be exceeded more than 1x per year	0.011	0.006	0.037	nm	nm	nm

nc



Not

measured

nm Meets standard Does not meet standard *<75% data completeness for one quarter in 2017 **<75% data completeness for one quarter in 2016 ***<75% data completeness for one quarter in 2016 and 2017

Source: Herrera, 2021.

Not

calculated

Ambient air concentrations of PM10 at target sites throughout the MICs for 2021 are summarized in **Exhibit 3.2-5** and show that the PM10 concentration for these pollutants remain below the NAAQS.

Pollutant	Station	Averaging Time	2021 Concentration	NAAQS
Particulate Matter (PM10)	Ballard	24-Hour (µg/m3)	17.25	150
	Interbay/Dravus	24-Hour (µg/m3)	16.46	150
	Interbay/Armory	24-Hour (µg/m3)	19.42	150
	Stadium	24-Hour (µg/m3)	20.17	150
	Georgetown	24-Hour (µg/m3)	14.96	150
	South Park 1	24-Hour (µg/m3)	8.92	150
	SODO/Lander	24-Hour (µg/m3)	8.33	150
	South Park 2	24-Hour (µg/m3)	7.08	150

Note: Results represent the singular 24-hour PM10 concentrations for the respective sample day and location. Source: Herrera and Ramboll, 2021.

Ambient air concentrations of detected metals and VOCs at target sites throughout the MICs for 2021 are summarized in **Exhibit 3.2-6** and show that the concentration for these pollutants remain below the NAAQS.

Exhibit 3.2-6	Detected Pollutants and Measured Concentration Levels in 20
Exhibit 3.2-6	Detected Pollutants and Measured Concentration Levels in 20

Pollutant	Station	Constituent	2021 Max Concentration	NAAQS/RSL
Metals	Ballard	Lead	ND	0.15 µg/m3
		Arsenic	ND	0.00<u>16</u>2 µg/m3
		Chromium	0.0021	n/a
		Nickel	ND	0.015*
	Interbay/Dravus	Lead	ND	0.15 µg/m3
		Arsenic	ND	0.00<u>16</u>2 µg/m3
		Chromium	ND	n/a
		Nickel	ND	0.015*
	Interbay/Armory	Lead	ND	0.15 µg/m3
		Arsenic	ND	0.00<u>16</u>2 µg/m3
		Chromium	0.0025	n/a
		Nickel	0.0018	0.015*
	Stadium	Lead	0.0033	0.15 µg/m3
		Arsenic	ND	0.00<u>16</u>2 µg/m3

Ch.3 Environment, Impacts, & Mitigation Measures • Air Quality & GHG

Pollutant	Station	Constituent	2021 Max Concentration	NAAQS/RSL
		Chromium	0.0032	n/a
		Nickel	0.001	0.015*
	Georgetown	Lead	0.0018	0.15 µg/m3
		Arsenic	ND	0.00<u>16</u>2 µg/m3
		Chromium	0.0026	n/a
		Nickel	ND	0.015*
	South Park 1	Lead	0.0014	0.15 µg/m3
		Arsenic	ND	0.002 µg/m3
		Chromium	ND	n/a
		Nickel	ND	0.015*
	SODO/Lander	Lead	0.0015	0.15 µg/m3
		Arsenic	ND	0.00<u>16</u>2 µg/m3
		Chromium	0.0022	n/a
		Nickel	ND	0.015*
	South Park 2	Lead	ND	0.15 µg/m3
		Arsenic	ND	0.00<u>16</u>2 µg/m3
		Chromium	0.0024	n/a
		Nickel	0.0009	0.015*
VOCs	Ballard	Ethanol	15	n/a
		2-Proponal	ND	211<u>20.1</u> μg/m3
		Toluene	ND	520 μg/m3
		Heptane	ND	42 µg/m3
	Interbay/Dravus	Ethanol	ND	n/a
		2-Proponal	ND	211<u>20.1</u> μg/m3
		Toluene	2.7	520 μg/m3
		Heptane	ND	42 µg/m3
	Interbay/Armory	Ethanol	16	n/a
		2-Proponal	24	211<u>20.1</u> μg/m3
		Toluene	ND	520 μg/m3
		Heptane	ND	42 µg/m3
	Stadium	Ethanol	ND	n/a
		2-Proponal	ND	211<u>20.1</u> μg/m3
		Toluene	ND	520 μg/m3
		Heptane	ND	42 µg/m3
	Georgetown	Ethanol	13	n/a
		2-Proponal	36	211<u>20.1</u> μg/m3
		Toluene	ND	520 μg/m3

Ch.3 Environment, Impacts, & Mitigation Measures • Air Quality & GHG

Pollutant	Station	Constituent	2021 Max Concentration	NAAQS/RSL
		Heptane	ND	42 µg/m3
	South Park 1	Ethanol	ND	n/a
		2-Proponal	ND	211<u>20.1</u> µg/m3
		Toluene	ND	520 μg/m3
		Heptane	ND	42 µg/m3
	SODO/Lander	Ethanol	38	n/a
		2-Proponal	8.5	211<u>20.1</u> µg/m3
		Toluene	3.7	520 µg/m3
		Heptane	3.5	42 µg/m3
	South Park 2	Ethanol	ND	n/a
		2-Proponal	10	211<u>20.1</u> µg/m3
		Toluene	ND	520 µg/m3
		Heptane	ND	42 µg/m3

NAAQS=National Ambient Air Quality Standard; RSL=EPA Region 9 Regional Screening Level; ND= Not Detected * Represents the RSL for Nickel Subsulfide

Note: RSLs are available at EPA's Regional Screening Levels website (https://www.epa.gov/risk/regional-screening-levels-rsls). The noncarcinogenic screening levels with a target hazard quotient of 0.1 are provided.

Source: Herrera and Ramboll, 202<u>2</u>1.

An area remains a nonattainment area for a particular pollutant until concentrations are in compliance with the NAAQS. Only after measured concentrations have fallen below the NAAQS can the state apply for redesignation to attainment, and it must then submit a 10-year plan for continuing to meet and maintain air quality standards that follow the Clean Air Act. During this 10-year period, the area is designated as a maintenance area.

The Puget Sound region, including the industrial and maritime areas of Seattle, is currently classified as an attainment area for ozone, Nox, lead, particulate matter and SO2. The region was designated as a maintenance area for CO until recently and is now considered in attainment. The U.S. EPA designated Seattle Duwamish area as a maintenance area for PM10 in 2000 and in 2002; the area is now in attainment.⁸ Tacoma is currently classified as attainment with maintenance provisions for PM2.5.

The Puget Sound Regional Council estimates that by 2050, the Puget Sound region population will grow by 1.6 million people, increasing 38%, to reach a population of 5.8 million people (PSRC 2021). The highest population increase is estimated to be in King County. Estimates such as this indicate that CO, PM2.5 and ozone emissions will increase, which could lead to future challenges meeting the NAAQS.

⁸ EPA 2021, <u>https://www3.epa.gov/airquality/greenbook/anayo_wa.html</u>

Air toxic pollutant emissions remain a concern because of the projected growth in vehicle miles traveled. The U.S. EPA has been able to reduce benzene, toluene, and other air toxics emissions from mobile sources by placing stringent standards on tailpipe emissions and requiring the use of reformulated gasoline.

Sources of Air Pollution in Seattle's MIC Areas

For this analysis, the existing air pollution sources in the BINMIC and Greater Duwamish MIC are divided into several categories: transportation sources such as surface vehicle traffic; rail operations including freight and commuter trains, shipping and marine terminal operations, and aircraft overflights; point sources such as commercial/industrial equipment and processes; and construction vehicles and equipment sources.

Transportation sources include vehicles on highways and major arterial roadways, particularly those supporting a high percentage of diesel truck traffic. These include routes such as Interstate 5 (I-5), State Route 99 (SR 99), State Route 599 (SR 599), and the major arterials that traverse the MICs such as E. Marginal Way S., W. Marginal Way SW, and Airport Way S. in the Greater Duwamish MIC, and 15th Avenue W in the BINMIC. Diesel-fueled trucks, particularly older trucks that emit more pollutants than newer trucks, are the focus of federal, state, and local effort to reduce pollutant emissions (see previous section). Drayage trucking (local trucking that moves shipping containers between Port of Seattle ship terminals and distribution centers in Seattle, Kent Valley, and elsewhere) represent a sizeable portioncomponent of local trucking in the MICs. These trucks, which are often older and independent operations, are often required to queue and idle near port facilities. Older truck fleets are undergoing turnover to newer truck fleets and cleaner burning fuels. Port of Seattle staff have invested more than \$15 Million in the last ten years, and significant staff time to transition the drayage fleet to year 2007 model or newer truck engines. As of 2019, more than 400 trucks had been scrapped and replaced with the help of federal, state, and local funding, and all trucks serving the Port's international container terminals had a 2007 or newer engine.

MIC areas in Seattle are also affected by air pollution from freight and passenger rail operations. Additional transportation sources include railway lines supporting diesel locomotive operations BNSF Railway Company (BNSF) owns and operates a mainline dual-track from Portland through the Greater Duwamish MIC to Seattle, and then extends north from downtown Seattle through the BINMIC to Snohomish County. A connecting spur, operated by the Ballard Terminal Rail Company, serves the Ballard and the western ship canal area. Union Pacific owns and operates a single mainline track with two-way train operations between Tacoma and Seattle that also passes through the Greater Duwamish MIC. While these operations generate air emissions in the immediate vicinity of the railways, train operations, including both freight and Commuter rail such as Sound Transit's Sounder system are intermittent. The contribution of air emissions from rail compared to the overall ambient air quality environment in the Seattle MIC areas is relatively minor compared to other sources such as traffic. However, areas near train yards may experience higher exposure to air emissions from assembling railcars into long trains and idling engines (WDOH 2008). Shipping and marine terminal operations include emissions from ocean-going vessels, harbor support vessels, ferries, and cargo-handling equipment at marine facilities near Interbay (Pier 90), along the Seattle waterfront, alongside Harbor Island, and in the Duwamish waterway. These marine sources also contribute to regional and localized pollutant concentrations. These vessels typically use a range of fuels, including marine diesel oil, Intermediate fuel oil, medium fuel oil, and heavy fuel oil (also known as bunker fuel). Implementation in 2015 of the North American Emissions Control Area (ECA) established by the International Maritime Organization (IMO) requiring that ocean going vessels use fuels with 0.1% sulfur within 200 miles from the U.S. coast rather than the typical higher sulfur content bunker fuel (2.7%), SO₂ and diesel particulate emissions have been significantly reduced (PSMEI 2018).

Aircraft using King County International Airport, also known as Boeing Field and Seattle-Tacoma International Airport (Sea-Tac) frequently fly over Seattle MICs, with some arriving and departing flight paths at lower altitudes, depending on atmospheric conditions. These operations contribute to the overall ambient air quality environment. Atmospheric conditions may contribute to the direction of aircraft operations (flow) and affect aircraft emissions distribution.

Point sources of air pollution in the Seattle MICs include a wide variety of industrial and other non-transportation air emissions sources and are almost always required to have a permit to operate from PSCAA. These include manufacturing plants, and other heavy and general industrial facilities, and others. Industrial turbines, paper and packaging manufacturing, building materials manufacturing, steel manufacturing and fabrication, airplane manufacturing and assembly, and cement manufacturing plants are examples of point sources of air pollution in the MICs. Wood smoke is also considered an important point source contributor, either from wood-burning fireplaces or wildfire.

Construction vehicles and equipment sources include diesel-powered construction equipment such as excavators, dump trucks, pile drivers, cranes, and small equipment such as conveyors, generators, and mixers. Industrial and equipment sources include industrial boilers, cleaning/solvent use, coating and printing, wastewater systems, VOC processes, cooling towers, leaking components, flares, storage tanks, and combustion.

Sensitive Populations in and Around Seattle's MIC Areas

A health risk assessment conducted by the Washington State Department of Health (DOH) found that point sources (e.g., manufacturing facilities, cement plants) make up only about 4% of the overall long-term health risk associated with air pollution in the region. Mobile sources (i.e., cars, trucks, buses, ships, planes, trains) and wood stove/fireplace emissions likely make up the bulk of air pollution health risk in the region. Diesel particulate, benzene and formaldehyde from car and truck emissions, and wood smoke were identified as being the toxic air pollutants that make up the bulk of risk (WDOH 2008). These on-road mobile sources contribute to the highest cancer and non-cancer risks near major roadways over a large area of

south Seattle and those risks and hazards are greatest near major highways and drop dramatically about 200 meters (656 feet) from the center of highways (WDOH 2008).

However, residential communities that border industrial areas like the BINMIC and Greater Duwamish MIC may be at risk of increased impact from pollutants due to their proximity to both transportation and point sources of pollution. The majority of land use in the BINMIC and Duwamish Valley are commercial or industrial, with most areas surrounding those industrial and maritime areas zoned as residential. The exception is the two residential communities of Georgetown and South Park, which are in the Duwamish Valley and surrounded by industrial uses.

Populations that are more sensitive to the health effects of air pollutants include the elderly and the young; groups with higher rates of respiratory disease, such as asthma; and those with other environmental or occupational health exposures (e.g., indoor air quality) that affect cardiovascular or respiratory diseases. Therefore, land uses and facilities such as schools, children's daycare centers, hospitals and nursing and convalescent homes are considered to be more sensitive than the general public to poor air quality because the population groups associated with these uses are more susceptible to respiratory distress.

Open spaces and playgrounds are considered moderately sensitive to poor air quality because those engaged in strenuous work or exercise have increased sensitivity to poor air quality; however, exposure times are generally shorter in parks and playgrounds than in residential locations and schools. Residential areas are considered more sensitive to air quality conditions compared to commercial and industrial areas because people generally spend longer periods of time at their residences, with proportionally greater exposure to ambient air quality conditions. Workers are not considered sensitive receptors because all employers must follow regulations set forth by the Occupational Safety and Health Administration to ensure the health and well-being of their employees with regard to their own operations.

Maps indicating disparities in the potential exposure of populations in census tracts in the subarea are addressed in **Section 3.9 Housing**.

Greenhouse Gases & Climate Change

Background

The accumulation of greenhouse gases (GHGs) has been identified as a driving force in global climate change. Definitions of climate change vary between and across regulatory authorities and the scientific community. In general, however, climate change can be described as the changing of earth's climate caused by natural fluctuations and anthropogenic activities (i.e., activities relating to, or resulting from the influence of human beings) that alter the composition of the global atmosphere.

The principal GHGs of concern are Carbon dioxide (CO2), methane (CH4), nitrous oxide (N2O), sulfur hexafluoride (SF6), perfluorocarbons (PFCs) and hydrofluorocarbons (HFCs). Electric

utilities use SF6 in electric distribution equipment. Each of the principal GHGs has a long atmospheric lifetime (one year to several thousand years). In addition, the potential heat-trapping ability of each of these gases varies significantly. CH4 is 25 times as potent as CO2 at trapping heat, while SF6 is 23,900 times more potent than CO2. Conventionally, GHGs have been reported as CO2 equivalents (CO2e). CO2e takes into account the relative potency of non-CO2 GHGs and converts their quantities to an equivalent amount of CO2 so that all emissions can be reported as a single quantity.

The primary human-made processes that release GHGs include combustion of fossil fuels for transportation, heating, and electricity generation; agricultural practices that release CH4, such as livestock production and crop residue decomposition; industrial processes that release smaller amounts of high global warming potential gases such as SF6, PFCs, and HFCs, and waste decomposition that releases CH4. Deforestation and land cover conversion have also been identified as contributing to global warming by reducing the earth's capacity to remove CO2 from the air and altering the earth's albedo (surface reflectance) thus allowing more solar radiation to be absorbed.

Global mean temperatures in the United States (U.S.) have warmed during the 20th century and continue to warm into the 21st century. According to data compiled by NOAA, the rate of warming for the entire period of record (1880–2020) is 0.13°F per decade across the contiguous 48 States. The 10 warmest years on record have all occurred since 2005, and 7 of the 10 have occurred just since 2014. (NOAA 2021)

Ecology estimated that in 2018, Washington produced about 99.6 million gross metric tons (MMTCO2e; about 109.7 million U.S. tons) of CO2e (Ecology 2021). Ecology found that transportation is the largest source, at 45% of the state's GHG emissions; followed by residential, commercial, and industrial (RCI) energy use at 23% and electricity generation (instate and purchased from out-of-state) at 16%. The sources of the remaining 16% of emissions are fossil fuel and industrial processes, agriculture, and waste management.

Current Policy & Regulatory Frameworks

<u>U.S. EPA</u>

The U.S. EPA regulates emission of GHGs through two approaches: the first establishes Corporate Average Fuel Economy (CAFÉ) and GHG emission standards for light-duty vehicles (passenger cars and trucks), for medium and heavy duty commercial trucks and buses, and for commercial marine diesel engines above 30 Liters per cylinder (Category 3 Engines), which include large marine engines; the second covers GHG emissions from the largest stationary sources (buildings, structures, facilities, or installations) by the Prevention of Significant Deterioration (PSD) and title V Operating Permit Programs under the Clean Air Act (40 CFR 52.21(b)(3)).

Because the Action Alternatives propose land use changes to the Seattle MICs and do not propose construction of specific facilities or use of specific types of vehicles, federal regulatory requirements are not applicable to this impact analysis, though these standards will help

reduce anticipated emissions in the future compared to existing conditions. Individual facilities and vehicle manufacturers will be responsible to ensure compliance in the MICs with EPA rules regarding GHG emissions.

Washington State

Washington has adopted a variety of regulations, programs, and initiatives designed to reduce GHG emissions.

Chapter 173-441 WAC—Reporting of Emissions of Greenhouse Gas, as adopted by Ecology, requires some facilities and transportation fuel suppliers to annually report their GHG emissions. The program uses the same emission calculation methods as EPA's GHG reporting program, and include:

- Facilities that emit at least 10,000 metric tons of carbon dioxide equivalent (CO2e) per year in Washington.
- Suppliers of liquid motor vehicle fuel, special fuel, or aircraft fuel that provide products equivalent to at least 10,000 metric tons of carbon dioxide per year in Washington.

In 2020, the Washington Legislature set new GHG emission limits (RCW 70A.45.020) in order to combat climate change. Under the law, the state is required to reduce emissions levels:

- 2020—reduce to 1990 levels.
- 2030—45% below 1990 levels.
- 2040—70% below 1990 levels.
- 2050—95% below 1990 levels and achieve net zero emissions.

The State Agency Climate Leadership Act (RCW 70.235.050 and 060) requires some state agencies to reduce their GHG emissions. The Act was updated in 2020 to require state agencies to reduce their carbon pollution to these targets:

- 2020—15% below 2005 levels
- 2030—45% below 2005 levels
- 2040—70% below 2005 levels
- 2050—95% below 2005 and achieve net-zero GHG emissions.

The 2019 Clean Energy Transformation Act (CETA) (SB 5116) requires all electric utilities in Washington to transition to carbon-neutral electricity by 2030 and to 100% carbon-free electricity by 2045. The Washington Department of Commerce and the Washington Utilities and Transportation Commission (UTC) are leading the implementation efforts.

The Motor Vehicle Emission Standards—Zero Emission Vehicles (ZEV) bill (RCW 70A.30.010) directs Ecology to adopt California vehicle emission standards, including zero emission vehicle standards that require a percentage of the vehicles sold in Washington to be zero emission. The 2021Clean Fuel Standard will require fuel suppliers to reduce the carbon intensity of their fuels 20% by 2038.

The 2021 Climate Commitment Act establishes a "cap and invest" program that sets a limit on the amount of GHGs that can be emitted in Washington (the cap) and then auctions off allowances for companies and facilities that emit GHGs until that cap is reached. Over time, the cap will be reduced, allowing total emissions to fall to match the GHG emission limits set in state law. Rulemaking will begin in 2021, and the program's first compliance period will begin in 2023.

Ecology adopted a rule in 2019 to transition away from using GHGs known as hydrofluorocarbons (HFCs) in products and equipment starting in 2020. A law passed in 2021 expands on that program, establishing a program to reduce leaks from large air conditioning and refrigeration equipment, limiting the impacts for refrigeration chemicals, and requiring Ecology to recommend options for capturing HFCs when equipment reaches the end of its useful life.

The Clean Buildings for Washington law (HB 1257), establishes energy use intensity (EUI) targets for large commercial buildings (over 50,000 square feet), which will be updated over time. Owners of these buildings must first meet these energy performance standards between 2026 and 2028, depending on square footage of the building.

There is no standard significance threshold for GHG emissions in the Washington SEPA rules (Washington Administrative Code [WAC] 197-11-330).

Seattle Climate Change Policies

Seattle is a member of the Carbon Neutral Cities Alliance, a collaboration of cities working to cut GHG emissions by 80-100% by 2050 or sooner—the most aggressive GHG reduction targets undertaken anywhere by any city. The City of Seattle is also a member of the King County-Cities Climate Collaboration (K4C). This Collaboration is working toward achieving shared countywide GHG reduction targets that reduce direct countywide sources of GHG emissions by at least 50% by 2030, and 80% by 2050, compared to a 2007 baseline. The City of Seattle is also a steering committee member of the Puget Sound Climate Preparedness Collaborative, a network of local and tribal governments, public agencies, and organizations working together towards regional climate resiliency.

Seattle Climate Action Plan

In 2011, the City Council adopted a long-term climate protection vision for Seattle (through Resolution 31312) which included achieving net zero GHG Emissions by 2050 and preparing for the likely impacts of climate change. To achieve these goals the City prepared a Climate Action Plan (2013 CAP) which detailed the strategy for realizing this vision. In 2017, the City Council adopted Resolution 31757, affirming Seattle's commitment to the goals established in the Paris Agreement, and resulting in the updated 2018 Climate Action Strategy, which identifies the actions necessary to limit atmospheric warming to 1.5 degrees Celsius.

City actions in the 2013 CAP and the updated 2018 Strategy focus on those sources of emissions where City action and local community action will have the greatest impact: road

transportation and building energy, which comprise the majority of local emissions. With 2008 as the baseline year, the 2013 CAP identifies the following as targets by 2030. These goals remained unchanged by the updated 2018 Strategy:

- 82% reduction in passenger vehicle emissions
- 20% reduction in vehicle miles traveled
- 75% reduction in GHG emissions per mile of Seattle vehicles
- 45% reduction in commercial building emissions
- 10% reduction in commercial building energy use
- 32% reduction in residential building emissions
- 20% reduction in residential building energy use
- 39% reduction in building energy emissions
- 25% reduction in combined commercial and residential building energy use

City of Seattle Comprehensive Plan 2015-2035

The current City of Seattle Comprehensive Plan, Seattle 2035 addresses climate change within its Environmental Element (City of Seattle, 2020). Climate change-related goals and policies contained within the environmental element of the current Comprehensive Plan are listed below.

Goal EN G3 Reduce Seattle's greenhouse gas emissions by 58 percent from 2008 levels by 2030, and become carbon neutral by 2050.

- Policy EN 3.1 Expand transit, walking, bicycling, and shared-transportation infrastructure and services to provide safe, affordable and effective options for getting around that produce low or zero emissions, particularly for lower-income households and communities of color.
- Policy EN 3.2 Implement the urban village strategy with the goal of meeting the growing demand for conveniently located homes and businesses in pedestrianfriendly neighborhoods where residents can walk to a variety of recreation and service offerings, in order to increase the number of trips that do not require automobile use and increase access to opportunity for lower-income households and communities of color.
- Policy EN 3.3 Implement innovative policies, such as road pricing and parking management, that better reflect the true cost of driving and therefore lead to less automobile use, while employing strategies that mitigate impacts on low-income residents.
- Policy EN 3.4 Encourage energy efficiency and the use of low-carbon energy sources, such as waste heat and renewables, in both existing and new buildings.
- Policy EN 3.5 Reduce the amount of waste generated while at the same time increasing the amount of waste that is recycled and composted.

- Policy EN 3.6 Reduce the emissions associated with the life cycle of goods and services by encouraging the use of durable, local products and recycled-content or reused materials, and recycling at the end of products ' lives.
- Policy EN 3.7 Support a food system that encourages consumption of local foods and healthy foods with a low carbon footprint, reduces food waste, and fosters composting.

Goal EN G4 Prepare for the likely impacts of climate change, including changing rain patterns, increased temperatures and heat events, shifting habitats, more intense storms, and rising sea level.

- Policy EN 4.1 Consider projected climate impacts when developing plans or designing and siting infrastructure, in order to maximize the function and longevity of infrastructure investments, while also limiting impacts on marginalized populations and fostering resilient social and natural systems.
- Policy EN 4.2 Prioritize actions that reduce risk and enhance resilience in populations nearest the likely impacts of climate change, including actions that are driven by the communities most impacted by climate change.
- Policy EN 4.3 Focus strategies to address the impacts of climate change, in particular, on the needs of marginalized populations and seniors, since these groups often have the fewest resources to respond to changing conditions and therefore may be more severely impacted.
- Policy EN 4.4 Partner with communities most impacted by climate change to identify local community assets, including infrastructure, cultural institutions, community centers, and social networks that can be supported and leveraged in adaption planning.

Building & Energy Policies

In 2021, the City of Seattle adopted new energy code updates for commercial and large multifamily buildings that:

- Eliminate all gas and most electric resistance space heating systems
- Eliminate gas water heating in large multifamily buildings and hotels
- Improves building exteriors to improve energy efficiency and comfort
- Requires electrical infrastructure necessary for future conversion of any gas appliances in multifamily buildings

Energy code updates do not apply to single family homes or low-rise multifamily homes, as the state prohibits city amendments to the residential energy code; nor does it apply to equipment used by a manufacturing, industrial or commercial process other than for conditioning spaces or maintaining comfort and amenities for the occupants (Seattle 2021c). Seattle also has a variety of other policies and programs specific to reductions in building energy use, including:

- Energy Benchmarking Program requires owners of non-residential and multifamily buildings (20,000 sf or larger) to track energy performance and annually report to the City of Seattle.
- Tune-ups aim to optimize energy and water performance by identifying low- or no-cost actions related to building operations and maintenance, that generate 10-15% in energy savings, on average.
- Passage of a new law to help phase out oil heat by 2028 in order to reduce climate pollution, prevent soil and groundwater contamination, and improve air quality.
- Adoption of policies addressing new construction and major renovations, as well as day-today operations of buildings owned and maintained by the City.

Maritime Policies

Seattle City Light and the Port of Seattle are committed to reducing the GHG emissions from marine activities. In 2020, the Northwest Ports, of which the Port of Seattle <u>and the Northwest</u> <u>Seaport Alliance areis a</u> member<u>s</u>, committed to phase out seaport related air and GHG emissions and transition to zero-emission operations by 2050 as part of the Northwest Ports Clean Air Strategy (NWP 2020). The commitment<u>, independent of the Industrial and Maritime Strategy</u>, covers all of the activities that are included in each participating port's emissions inventory, which includes direct emissions from port operations, as well as emissions from seaport-related activities.

Greenhouse Gas Emissions in Seattle

Seattle updated its GHG emissions inventory in 2018, documenting 5.7 million metric tons (MMTCO2e; about 6.3 million U.S. tons) of CO2e. Primary sources (core emissions) of GHG emissions include on-road buses, cars, light/medium/heavy duty trucks, residential and commercial building energy use, waste (residential, commercial, and self-haul) generation, and credits for offsets. Expanded sources of GHG emissions include core emissions plus marine, rail, and air transportation, waste (construction and demolition, wastewater) generation, industrial energy use and processes, and credits for offsets and sequestration of waste.

Overall, total emissions rose from 5.75 MMTCO2e in 2008 to 5.76 MMTCO2e in 2018, a 0.2% increase, despite an overall increase in population of over 25%. Per capita emissions dropped from 9.7 metric tons (MTCO2e) in 2008 to 7.7 MTCO2e per person in 2018, a decrease of over 20%. Core GHG emissions of GHGs declined from 3.2 MMTCO2e in 2008 to 3.1 MMTCO2e CO2e in 2018, a 4% decline (Seattle 2020).

Like Washington State, emissions in Seattle from transportation represent the largest percentage of overall emissions at 61%. The second largest emission source is building energy use at 24%, followed by emissions from industrial processes at 18%. City Light achieved GHG neutrality in 2005 through eliminating and reducing emissions, inventorying remaining emissions and purchasing offsets to offset the remaining emissions (SCL 2012) and has maintained GHG neutrality since that date.

Transportation Related GHG Emissions

Core transportation emissions decreased around 3% since 2008—from 2 MMTCO2e in 2008 to 1.94 million MMTCO2e in 2018. Road transportation has been the largest category of emissions since Seattle started tracking emissions in 1990. Total emissions in this sector increased through 2008; however, they have been decreasing since 2008 due to changes in the fuel economy of vehicles and changes in miles traveled. Most emissions from road transport, greater than 85%, are from gasoline fuel sources. Advances in vehicle technology have increased the average fuel economy for cars and light-duty trucks (including SUVs) in Seattle from about 20 miles per gallon of fuel in 2008 to about 23.6 miles per gallon in 2018 (Seattle 2020). Medium and heavy-duty truck diesel fuel sources contributed about 15% of the road transport emissions in 2008 and have increased about 2.5%—from 0.289 MMTCO2e in 2008 to 0.297 million MMTCO2e in 2018. This increase has occurred despite freight emissions per mile decreased 8% between 2008 and 2018, due largely to more vehicle miles traveled. Expanded GHG emissions increased almost 10% since 2008, with most of the increase attributed to greater air travel. Air transport emissions increased by 40% from 972,000 MTCO2e to 1.37 MMTCO2e in 2018 (Seattle 2020).

Shipping and marine terminal operations include GHG emissions from ocean-going vessels, harbor support vessels, ferries, and cargo-handling equipment at marine facilities near Interbay (Pier 90), along the Seattle waterfront, alongside Harbor Island, and in the Duwamish waterway.

Building Related GHG Emissions

Core building GHG emissions decreased 5.9% since 2008—from 1.27 MMTCO2e to 1.19 MMTCO2e in 2018. Expanded building emissions decreased 1.9% since 2008—from 1.43 MMTCO2e in 2008 to 1.40 MMTCO2e in 2018. However, both core and expanded building sector emissions increased by about 8% between 2016 and 2018, primarily as a result of an increase in fossil gas use.

About 90% of the electricity that Seattle City Light (SCL) provides to consumers in Seattle comes from low-carbon hydroelectric dams. SCL purchases local carbon offsets equal to the GHG emissions resulting from all other aspects of SCL's operations, including those created by fossil fuels included in the mix of power the utility buys, employees' travel, and the trucks and other equipment used in its operations. Because of variation in hydroelectricity production from year to year, SCL's external power purchases and the consequent amount of carbon offsets purchases varies annually. While electricity consumption is trending down, it is the largest source of energy for Seattle's buildings (54%) but is responsible for only 9% of emissions in the building sector before offsets. Fossil gas is currently responsible for 86% of building sector emissions, none of which are offset. (Seattle 2020)

Industrial Emissions

Industry emissions decreased 22.6% since 2008—from 1.36 MMTCO2e in 2008 to 1.05 MMTCO2e in 2018. This decrease in process emissions was largely due to reduction in cement

process emissions which was halved since 2008. Meanwhile fossil gas use has increased 24.9% since 2008 from .27 to .33 million MTCO2e (Seattle 2020).

Maritime Activities Related Emissions

Maritime activities taking place within and adjacent to the MICs emit GHG emissions, including from ocean-going vessel hoteling (operations while stationary at dock) and maneuvering, harbor vessel movements, ferry transits, recreational vessels, and shore-side cargo handling equipment. Additional context and information for maritime emissions in general, and in relation to the MIC areas affected by the proposal, can be found in the 2016 Puget Sound Maritime Emissions Inventory (PSMEI 2018), which is incorporated into this EIS by reference.

SCL is working with the Port of Seattle, Washington Department of Transportation (WSDOT), and the U.S. Coast Guard (USCG) to install electrical infrastructure along the Seattle waterfront (including in the MIC areas), at Fisherman's Terminal, and in the Port to provide shore power to cargo vessels, cruise ships, ferries, USCG vessels, and some recreation/commercial fishing vessels. This work will eliminate the necessity for those vessels to run their engines while dockside. The U.S. EPA indicates that under the right circumstances when a vessel is connected to shore power, overall pollutant emissions can be reduced by up to 98% when utilizing power from the regional electricity grid (EPA 2017). The Port of Seattle is also actively replacing diesel-powered cargo handling equipment with electric power equipment over time.

3.2.2 Impacts

Air quality impacts related to each alternative were evaluated by reviewing proposed land use changes and anticipated changes in employment, vehicle miles traveled (VMT), and commercial, industrial, and housing construction and post-construction activities. Because construction is considered a temporary activity, a qualitative analysis of construction impacts common to all alternatives is presented.

For impacts related to longer-term changes in land use, the proposed alternatives would increase housing, employment, and vehicle miles traveled (VMT) in the study area in increments through the horizon year (2044) compared to the baseline year (2021). The projected area-wide increases in vehicle miles traveled (VMT) for the p.m. peak periods were used as a basis for comparison of the alternatives to the base year.

This section also describes how implementation of any of the Action Alternatives could affect GHG emissions in the study area compared to Alternative 1 No Action, primarily through changes in transportation patterns and land uses. Transportation systems contribute to climate change primarily through the emissions of carbon dioxide (CO2), methane (CH4), and nitrous oxide (N2O) primarily from gasoline and diesel fuels used to operate cars, trucks, and rail vehicles. Land use changes contribute to climate change through construction and operational use of electricity and natural gas. GHG emission impacts related to each alternative were evaluated by reviewing proposed land use changes and anticipated changes in employment,

vehicle miles traveled (VMT), and commercial, industrial, and housing construction and postconstruction activities.

For impacts related to longer-term changes in land use, the proposed alternatives would increase housing, employment, industrial and non-industrial building space, and VMT in the study area in increments through the horizon year (2044) compared to the baseline year (2018-2021 depending on source). The projected area-wide increases in VMT for the p.m. peak periods were used as a basis for calculation of road transportation sources of GHG. The projected total and incremental increases in industrial and non-industrial building space and housing units were used as a basis for calculation of building related GHG emissions. The sum of these emissions were used as a basis for comparison of the alternatives to the No Action.

Impacts Common to All Alternatives

Air Quality

This discussion of impacts common to all alternatives covers all of the industrial lands subareas due to the regional nature of air quality, the mobility of transportation sources, and the dispersion of air pollutants. Air quality impacts specific to industrial lands subareas and for the locations targeted for air sampling, are discussed in the individual alternative discussions.

Construction Related Emissions

Future growth under any alternative would result in development of new maritime, industrial, design and research, and office uses, and some industry-supportive housing. Most development projects in the study area would entail a combination of demolition and removal of existing structures or parking lots, excavation and site preparation, construction of new buildings, and retrofit or adaptive reuse of existing buildings. Emissions generated during construction activities would include exhaust emissions <u>and associated odors</u> from construction equipment, commuting workers, trucks used to haul construction materials to and from sites, asphalt paving and painting, as well as fugitive dust emissions associated with soil-disturbing activities, demolition and construction work, and grading. Increased vehicle emissions associated with increased traffic congestion during construction could also occur.

The pollutants of concern from fugitive dust are PM2.5 and PM10. The PSCAA requires dust emission control measures on construction projects through Article 9, Section 9.15, including:

- 1. Using control equipment, enclosures or wet suppression techniques, and curtailment during high winds
- 2. Surfacing roadways and parking areas with asphalt, concrete, or gravel as soon as possible
- 3. Treating construction sites with water or chemical stabilizers, reducing vehicle speeds, installing pavement rip rap exit aprons, and cleaning vehicle undercarriages before entering public roadways
- 4. Covering or wetting truck loads or providing freeboard in truck loads.

With implementation of these requirements, impacts related to construction dust are expected to be less than significant.

Criteria air pollutants would be emitted during construction activities from construction equipment, much of it diesel fueled. Other emissions during construction would result from trucks used to haul construction materials to and from sites, and from vehicle emissions generated during worker travel to and from construction sites. Engine and motor vehicle exhaust produce emissions of VOCs, NOX, PM2.5, PM10, air toxics, and GHGs (assessed in Section 3.2.4). The primary emissions of concern with regard to construction equipment and trucking are Nox and PM2.5. Nox is primarily an air quality concern with respect to its role in (regional) ozone formation.

A number of federal regulations require emission and fuel standards that have or will lead to cleaner light- and heavy-duty truck and nonroad diesel equipment emissions. U.S. EPA Tier 3 Motor Vehicle Emission and Fuel Standards, established in 2014, set new vehicle emissions standards and a new gasoline sulfur standard beginning in 2017. The vehicle emissions standards reduce both tailpipe and evaporative emissions from passenger cars, light-duty trucks, medium-duty passenger vehicles, and some heavy-duty vehicles. Tier 4 emission standards, established in 2004 and fully phased in by 2014, targeted a reduction in Nox and PM emissions of more than 90% from nonroad diesel engines and sulfur reductions in nonroad diesel fuel (U.S. EPA 2004).

The Puget Sound air shed is currently designated as an attainment area with respect to ozone. Construction-related Nox emissions are not expected to generate significant adverse air quality impacts nor lead to violation of standards under any of the alternatives. The same conclusion is reached for diesel-related emissions of PM2.5, which could generate temporary localized adverse impacts within a few hundred feet of construction sites.

Consequently, given the intermittent and temporary nature of construction-related emissions and regulatory improvements that have been or are scheduled to be phased in, construction related emissions associated with all alternatives would be considered only a minor adverse air quality impact.

Land Use Change-Related Emissions

Under all alternatives, redesignation of some areas from strictly industrial land uses to those that support increased employment density, multi-story mixed-uses, and multi-modal access around future light rail stations would change growth and development patterns.

Anticipated total square footage of building space for industrial and non-industrial uses in each MIC under existing conditions and each of the four-alternatives are presented in **Exhibit 3.2-7**.

	Exis	ting	Alt. Act	1 No ion	Alt	t. 2	Alt	t. 3	Alt	t. 4	<u>Pref</u>	<u>. Alt.</u>
Geographic Area	1	NI	I	NI	I	NI	I	NI	I	NI	Ī	<u>NI</u>
BINMIC	6.8	5.4	9.2	6.5	12.1	6.3	14.5	8.3	14.6	8.6	<u>11.6</u>	<u>7.7</u>
Greater Duwamish MIC	34.6	13.9	40.4	15.7	46.7	15.4	47.2	18.1	46.9	18.3	<u>41.2</u>	<u>16.4</u>
Total	41.4	19.3	49.7	22.1	58.8	21.6	61.7	26.4	61.6	26.9	<u>52.8</u>	<u>24.1</u>

Exhibit 3.2-7 Estimated Industrial and Non-Industrial Square Footage for All Alternatives Compared to the Existing Conditions (2019), 2044 (million square feet)

Estimates for the MIC areas under all alternatives are approximate. Rounding error may cause total not to sum. Industrial employment estimated based on the 2019 share of industrial employment by sector based on the 2015 PSRC Industrial Lands Study NAICs-based definition of industrial activities.

I=Industrial; NI=Non-Industrial

Sources: CAI, Herrera, 202<u>2</u>4.

Anticipated development resulting from all alternatives would alter the proximity and number of future workers in the study area to mobile and stationary sources of air toxics and particulate matter PM2.5. The degree of potential for adverse impacts on sensitive receptors would depend on proximity to sources, the emissions from these sources and the density of future development. In addition, areas surrounding the MICs could be subject to any emissions from increased employment density, new industrial development, and any additional traffic arising from worker commute or commercial transportation activity. However, because all the alternatives include some focus on increased employment density and land uses changes around light rail stations, some emission increases associated with growth in background traffic, worker commuting, and commercial activities may be muted.

Vehicle emissions for all of the alternatives would be minor relative to the overall regional vehicle emissions in the Puget Sound air shed. Photochemical smog (the regional haze produced by ozone and fine particles) is caused by regional emissions throughout the Puget Sound region, rather than localized emissions from any individual neighborhood. As discussed previously, the Puget Sound region was designated a maintenance area for ozone, with the 20-year maintenance period ending in 2016. Since that time, the region has been a designated attainment area for ozone. In addition, the U.S. EPA Tier 3 Motor Vehicle Emission and Fuel Standards and Tier 4 Control of Emissions of Air Pollution From Nonroad Diesel Engines and Fuel (discussed in the previous section) have reduced vehicular emissions further. During the maintenance period, regional transportation emission budgets were set for three pollutants: CO, nitrogen oxides (Nox), and PM2.5. Based on the latest Puget Sound Regional Council (PSRC) air quality conformity analysis, forecasted regional emissions for its 2040 planning year are below the allowable budgets (PSRC 2018):

- CO: 38% of 2040 budget
- Nox: 62% of 2040 budget
- PM2.5: 83% of 2040 budget

Numerical forecasts of increased area wide vehicle miles traveled (VMT) during the PM Period are shown in **Exhibit 3.2-8**, below. Estimated road transportation emissions for each alternative are presented in the individual alternative's sections.

Geographic Area	2019 Existing	2042 No Action	2044 Alt. 2	2044 Alt. 3	2044 Alt. 4	<u>2044</u> <u>Pref. Alt.</u>
BINMIC	54,840	56,100	56,900	58,540	58,980	<u>57,600</u>
Greater Duwamish MIC	641,560	643,440	648,480	658,050	<u>659,520</u> 657,900	<u>649,950</u>
Seattle	2,964,540	<u>3,089,000</u> 3,083,140	<u>3,100,740</u> 3,094,870	<u>3,126,670</u> 3,121,270	<u>3,130,700</u> 3,121,420	<u>3,107,430</u>

Exhibit 3.2-8 Estimated VMT During the PM Period for Action Alternatives (2044) Compared to Existing (2019) and Alternative 1 No Action (2042)

Sources: Fehr & Peers, 20221; Herrera, 20221.

Population growth and VMT can be used as indicators of future transportation-related emissions. For each alternative, the forecasted VMT from the MIC area-wide modeling (see Transportation Chapter) is only a small fraction of the Puget Sound regional totals. Therefore, the forecasted similar VMT for all the Action Alternatives compared to Alternative 1 No Action would not alter PSRC's conclusion that future Puget Sound regional emissions will be less than the allowable emission budgets that were mandated by the air quality maintenance plans when they were in effect. It appears that neither of the alternatives would result in a significant impact on regional air quality.

Road transportation-related air pollutant emissions in each MIC under existing conditions and each of the four-alternatives are presented in **Exhibit 3.2-9**, **Exhibit 3.2-10**, and **Exhibit 3.2-11**. Anticipated for Seattle overall are shown for comparison. These emissions are based on existing and projected VMT.

Geographic Area	Pollutant	2019 Existing	2042 No Action	2044 Alt 2	2044 Alt 3	2044 Alt 4	<u>2044 Pref.</u> <u>Alt.</u>
BINMIC	СО	85.7	58.2	59.2	60.7	61.2	<u>59.8</u>
	Nox	19.8	15.9	16.5	16.6	16.7	<u>16.4</u>
	PM10	3.7	3.7	3.8	3.9	3.9	<u>3.8</u>
	PM2.5	0.7	0.7	0.7	0.7	0.7	<u>0.7</u>
	VOC	4.3	3.2	3.2	3.3	3.3	<u>3.2</u>
	Sox	0.2	0.2	0.2	0.2	0.2	<u>0.2</u>

Exhibit 3.2-9 Estimated Tons of Criteria Pollutant Emissions from Road Transportation for Action Alternatives (2044) Compared to Existing and Alternative 1 No Action (2042)

Ch.3 Environment, Impacts, & Mitigation Measures • Air Quality & GHG

Geographic Area	Pollutant	2019 Existing	2042 No Action	2044 Alt 2	2044 Alt 3	2044 Alt 4	<u>2044 Pref.</u> <u>Alt.</u>
Greater	СО	1,078.1	794.5	800.7	809.6	<u>811.0</u> 809.5	<u>800.4</u>
Duwamish MIC	Nox	641.2	552.8	557.1	557.2	<u>557.3</u> 557.2	<u>552.4</u>
	PM10	58.0	57.2	57.7	58.2	<u>58.3</u> 58.2	<u>57.6</u>
	PM2.5	15.0	12.5	12.5	12.6	<u>12.6</u> 12.6	<u>12.5</u>
	VOC	62.5	47.2	47.6	48.0	<u>48.1</u> 48.0	<u>47.5</u>
	Sox	3.8	3.4	3.4	3.4	<u>3.4</u> 3.4	<u>3.4</u>
Seattle	СО	4,783.0	<u>3,465.8</u> <u>3,459.5</u>	<u>3,480.5</u> <u>3,474.2</u>	<u>3,504.7</u> <u>3,498.9</u>	<u>3,508.5</u> 3,499.0	<u>3,484.8</u>
	Nox	1,900.8	<u>1,645.8</u> 1,643.6	<u>1,656.7</u> 1,654.4	<u>1,657.1</u> 1,654.8	<u>1,657.2</u> 1,654.8	<u>1,651.4</u>
	PM10	229.6	<u>234.9</u> 234.5	<u>236.0</u> 235.6	<u>237.4</u> 237.1	<u>237.7</u> 237.1	<u>236.2</u>
	PM2.5	52.9	<u>47.0</u> 46.9	<u>47.2</u> 47.1	<u>47.4</u> 47.4	<u>47.5</u> 47.4	<u>47.2</u>
	VOC	256.6	<u>196.7</u> 196.3	<u>197.6</u> 197.2	<u>198.8</u> 198.5	<u>199.0</u> 198.5	<u>197.7</u>
	Sox	14.7	<u>13.1</u> 13.1	<u>13.2</u> 13.2	<u>13.3</u> 13.2	<u>13.3</u> 13.2	<u>13.2</u>

All measurements in Tons.

Sources: Fehr & Peers, 20221; Herrera, 20221.



Exhibit 3.2-10 Estimated Tons of Criteria Pollutant Emissions from Road Transportation in BINMIC, All Alternatives

Note: This chart was updated to include the Preferred Alternative. Source; Herrera, 202<u>2</u>4.





<u>Note: This chart was updated to include the Preferred Alternative.</u> Source; Herrera, 202<u>2</u>4.

The Seattle Comprehensive Plan (Seattle 2035) EIS discusses the health risk associated with stationary emissions sources, including those near maritime uses where ship emissions and diesel locomotive emissions and diesel forklift emissions can all occur. Likewise, distribution centers that involve relatively high volume of diesel truck traffic can also represent a risk hazard to nearby sensitive land uses. That discussion is relevant to the proposal for the MICs and is incorporated here by reference. Land use changes that promote new or additional industrial and maritime uses of this type could add to the associated health risk of increased emissions associated with these uses, including the potential for criteria air pollutants and TAPs. Subarea plans developed for the MIC areas could consider setbacks for adjacent sensitive land uses from industrial sources and identify measures for receptors proposed in areas nearby such sources to reduce the potential risk.

The Washington State Department of Health (DOH) health disparities map (DOH 2021) indicates the Greater Duwamish MIC and the BINMIC census tracts rank among the highest for a comparison of pollution burden from Diesel Nox emissions and social factors that may contribute to disparities across the state. Where housing within the industrial zones is established under all alternatives, those residents would experience higher emissions resulting from industrial and other non-transportation air emissions. In addition, some of the housing units and anticipated growth could be placed near major highways, rail lines, or port facilities that produce greater vehicle emissions, particularly from diesel sources. Despite this potential, the combination of existing requirements for industrial operating permits from PSCAA, and ongoing requirements for improvements in vehicle emissions control, fuel economy, technology improvements, and overall fuel mix, local emissions <u>and associated odors</u> under the alternatives would result in a less than significant impact to air quality, and a moderate but less than significant impact on health related to air quality.

Local emissions of particulates could, however, impact residents of new residential development anticipated within the subareas, especially under alternatives 3 and 4 and the

<u>Preferred Alternative</u> if the new residential development occurs adjacent to major arterials. It would be prudent to consider risk-reducing mitigation strategies such as setbacks, improved building materials and structures, and improved air purification systems for residential and other sensitive land uses from major traffic corridors, rail lines, port terminals and similar point sources of particulates from diesel fuel.

Overall, given the regulatory improvements that have been or are schedule to be phased in, and the marginal increase in VMT associated with all of the alternatives, land use-related emissions would be considered only a less than significant impact adverse air quality impact.

Maritime Activities

Maritime activities taking place within and adjacent to the MICs, including ocean-going vessel (OGV) hoteling (operations while stationary at dock) and maneuvering, commercial harbor, and government vessel movements (including ferry transits), recreational vessels, and shore-side cargo handling equipment would continue to produce emissions under all alternatives.

Exhibit 3.2-12 shows 2016 air emissions in total annual tons associated with maritime sources in and adjacent to the study area.

Source	Nox	VOC	CO	SO2	PM10	PM2.5	Black Carbon*
OGV, hoteling**	450.2	15.2	40.8	22.9	10.5	9.9	0.6
OGV, maneuvering*	70.0	4.8	7.0	2.0	1.2	1.2	0.0
Recreational Vessels	138.2	13.6	87.7	0.1	4.6	4.2	3.2
Locomotives	167.0	10.7	29.1	0.1	5.1	4.7	3.6
Cargo-handling equipment	115.0	8.5	45.0	0.1	6.0	5.8	4.4
Heavy-duty vehicles	73.3	8.2	22.4	0.1	3.5	3.3	1.7
Fleet vehicles	1.9	0.4	6.6	0.0	0.0	0.0	0.0
Commercial Harbor/Govt. Vessel	2,105.0	92.0	599.0	1.0	77.0	71.0	54.0
Total	3,120.6	153.4	837.6	26.4	108.0	100.1	67.5

Exhibit 3.2-12 Maritime Activities Air Emissions, Tons per Year, 2016

Notes: *Black Carbon is soot, part of PM 2.5. **Ocean-going vessel (OGV) Source: 2016 Puget Sound Maritime Emissions Inventory.

The U.S. EPA has established Tier 4 emission standards for commercial marine diesel engines above 30 Liters per cylinder (Category 3 Engines), which align with International Maritime Organization (IMO) Annex VI marine engine Nox standards and low sulfur requirements. These standards require the use of exhaust aftertreatment technology, phased in between 2022 and 2024. In addition, SCL and the Port of Seattle are committed to reducing the air emissions from the marine activities they interact with and have embarked on a widespread effort to reduce or eliminate them by installing electrical infrastructure to provide shore power to cargo vessels, cruise ships, and ferries. The Port of Seattle is also actively replacing diesel-powered cargo handling equipment with electric power equipment over time to address existing emissions from Port operations. With these additional regulatory requirements and with local infrastructure improvements beyond what the Port is already planning, these maritime emissions are expected to drop significantly under all alternatives, even if cargo volumes and cruise ship visits increase.

Greenhouse Gases & Climate Change

Like the air quality section, this discussion of impacts common to all alternatives covers all of the industrial lands subareas due to the global nature of climate change, and the mobility and dispersion of GHG emissions. It is unlikely that a series of land use changes, even on the areawide scale of the alternatives under consideration, would have a perceptible impact on global climate change. It is more appropriate to conclude that GHG emissions from changes in future development in the Greater Duwamish MIC and the BINMIC would combine with emissions across the city, state, country, and planet to cumulatively contribute to global climate change.

Construction Related Emissions

Future growth under any alternative would result in development of new maritime, industrial, design and research, and office uses, and some industry-supportive housing. Most development projects in the study area would entail a combination of demolition and removal of existing structures or parking lots, excavation and site preparation, construction of new buildings, and retrofit or adaptive reuse of existing buildings. GHG emissions would occur as "embodied emissions" related to material extraction, manufacturing, transportation, building construction, maintenance, demolition or deconstruction, and disposal. Also included are emissions from demolition and construction sites. Increased vehicle emissions associated with increased traffic congestion during construction could also occur. Construction-related GHG emissions from any individual development project that may occur by 2044 would be temporary and would not represent an on-going source of emissions.

However, any accumulation of CO2 in the atmosphere, even if from a temporary source, can influence climate change when considered cumulatively with other global emissions. Over the course of the proposal's implementation, varying levels of construction activities within the MICs would be ongoing under any of the alternatives. Cumulatively, construction related emissions would be more than an insignificant contributor to GHG emissions within the study area between 2018 and 2044. An estimate of the GHG emissions resulting from 20 years of construction envisioned under the alternatives was calculated using research data from the Carbon Leadership Forum (CLF 2017) as a comparative tool. The total additional "embodied" emissions is estimated at between about 340,000 MTCO2e to 647,000 MTCO2e compared to Alternative 1 No Action, and includes emissions related to material extraction, manufacturing, transportation, building construction, maintenance, demolition or deconstruction, and disposal.

A number of federal regulations require emission and fuel standards that have or will lead to cleaner light- and heavy-duty truck and nonroad diesel equipment emissions (see Section 3.2.3.2.1). These standards also facilitate the adoption of new technologies necessary to meet GHG standards already promulgated by EPA (CRC 2014). The 2013 Seattle CAP and the updated 2018 Climate Action Strategy recognized the relevance of construction related GHG emissions and included several actions to be implemented by 2030 to address them, along with general actions to address transportation emissions.

Consequently, although construction related emissions would not be negligible, because of the combination of regulatory improvements and Climate Plan Actions under way, construction related GHG emissions associated with all alternatives would be considered a moderate adverse air quality impact.

Transportation Related GHG Emissions

Under all alternatives, redesignation of some areas from strictly industrial land uses to those that support increased employment density, multi-story mixed-uses, and some additional housing around future light rail stations would change growth and development patterns. These changes in development would result in changes in VMT, which were derived from the transportation analysis in **Section 3.10 Transportation**.

Existing and projected changes in VMT are estimated for cars, trucks, and buses and reflect all trips that start or end within the study area. GHG emissions from vehicle transportation were calculated based on estimated increases in VMT, emission factors reflecting future improvements to the vehicle fleet, and projected fuel economy for each vehicle class. Increased employment density and land uses changes around light rail stations may mute GHG emissions associated with worker commuting, and commercial activities, but these changes are reflected in VMT estimates.

Exhibit 3.2-13 shows GHG emissions in total annual metric tons of CO2e (MTCO2e) associated with road transportation sources in the study area under existing conditions and resulting from each of the four-alternatives.

0		,			, , , , , , , , , , , , , , , , , , , ,	
МІС	2019 Existing	2042 No Action	2044 Alt 2	2044 Alt 3	2044 Alt 4	<u>2044</u> <u>Pref. Alt.</u>
Ballard Interbay Northend	41,497	35,523	36,192	36,988	37,254	<u>36,470</u>
Greater Duwamish	662,025	577,635	582,056	586,450	586,381 <u>587,125</u>	<u>580,387</u>
Total	703,522	613,158	618,248	623,438	<u>624,379</u> 623,635	<u>616,857</u>
Seattle	2,582,481	<u>2,294,069</u> 2,290,282	<u>2,304,812</u> 2,300,999	<u>2,316,717</u> 2,313,120	<u>2,318,567</u> 2,313,189	<u>2,305,153</u>

Exhibit 3.2-13Estimated Road Transportation GHG Emissions for All Alternatives Compared to
Existing Conditions (2019) and Alternative 1 No Action (2042) (MTCO2e)

Sources: Fehr & Peers, 20221; Herrera, 20221.

The transportation analysis generally assumed continuation of current economic and demographic trends, with minor shifts toward shorter trips and more trips made by modes other than automobile travel. This reduces VMT per capita, but total VMT in the study area would continue to rise due to employment growth and some resident population growth.

A number of federal regulations require emission and fuel standards that have or will lead to cleaner light- and heavy-duty truck emissions (see **Section 3.2.1 Affected Environment**). These standards also facilitate the adoption of new technologies necessary to meet GHG standards already promulgated by EPA (CRC 2014). In addition, in August 2021, EPA proposed to revise existing national GHG emissions standards for passenger cars and light trucks for model years 2023–2026. The proposed standards would achieve significant GHG emissions reductions along with reductions in other criteria pollutants (U.S. EPA 2021). The proposed revised standards would result in substantial reductions in both GHG emissions and fuel consumption. According to the proposed standards, GHG emissions would decrease roughly 6% for new passenger cars and light trucks entering the vehicle fleet (U.S. EPA 2021).

Fuel economy for buses was also considered and fuel consumption were assumed to be reduced by 20% between 2018 and 2044. This is a conservative assumption given that King County Metro has targeted replacement of much of its fleet with battery-electric buses (Metro 2021).

All four-future year alternatives produce similar annual GHG emissions, as shown in **Exhibit 3.2-13**. Alternative 1 No Action is expected to have the lowest GHG emissions among the alternatives. Alternative 2, which includes limited land use changes, is expected to have the lowest GHG emissions among the proposed alternatives, with Alternative 4 having the highest. All of the 2044 alternatives are expected to generate lower road transportation GHG emissions than in 2019. This is because the projected improvements in fuel economy outweigh the projected increase in VMT.

When compared to the Alternative 1 No Action, road transportation emissions under <u>all the</u> <u>Action Aa</u>lternatives 2, 3 and 4 would be higher, but only <u>Aa</u>lternative<u>s 3 and 4</u> would exceed the 10,000 MTCO2e mandatory reporting threshold for the State of Washington for facilities <u>in</u> <u>the study area</u>.

Maritime Activities Related Emissions

Maritime activities taking place within and adjacent to the MICs, including ocean-going vessel hoteling (operations while stationary at dock) and maneuvering, commercial harbor, and government vessel movements (including ferry transits), recreational vessels, and shore-side cargo handling equipment would continue to produce GHG emissions under any of the alternatives. **Exhibit 3.2-14** shows current GHG emissions in total annual metric tons of CO2e (MTCO2e) associated with maritime sources in and adjacent to the study area.

Source	CO2e
OGV, hoteling	36,129
OGV, maneuvering	3,147
Recreational Vessels	8,616
Locomotives	10,894
Cargo-handling equipment	15,924
Heavy-duty vehicles	8,128
Fleet vehicles	463
Commercial Harbor / Government Vessel	138,019
Total	221,320

Exhibit 3.2-14 Estimated GHG Emissions from Maritime Activities, 2016 (MTCO2e)

Source: 2016 Puget Sound Maritime Emissions Inventory

Because changes to Comprehensive Plan policies, development standards and land use designations under all alternatives would protect and enhance industrial and maritime uses within the MICs, some of the increased employment and industrial and non-industrial space would likely include businesses that support maritime activities, which could indirectly increase GHG emissions from vessels, shore-side cargo handling equipment, and waterfront visitors. These potentially small and indirect increases are not quantified due to uncertainty.

With the existing and additional regulatory requirements and <u>with ongoing</u> local infrastructure improvements such as shore power, <u>existing and</u> future maritime GHG emissions are expected to decrease under all alternatives, even if cargo volumes and cruise ship visits increase.

Buildings & Energy Related Emissions

Under all alternatives, increased use of electricity could be generated in the MIC areas from any increases or changes in building space that result in heating and cooling, lighting, cooking and refrigeration, commercial and industrial equipment /machinery and processes, office equipment and computers, public transit operations (light rail), and streetlights and signal operations. In the MIC areas, all electricity is supplied by Seattle City Light. Seattle City Light is carbon neutral and, consistent with the 2013 CAP, no GHG emissions related to electricity would be generated from the alternatives and none are included in this analysis, as it is assumed that City Light would continue to produce carbon neutral electricity through 2044.

GHG emissions could be produced in the MIC areas from additional industrial and nonindustrial building space and housing that combusts natural gas for heating, cooking, or other industrial purposes. 2021 Seattle Energy Code changes that prohibit new natural gas connections would reduce GHG emissions from some of the anticipated development in the MIC where the code applies, such as commercial developments and some multi-family housing. GHG emissions from anticipated industrial and non-industrial building space, and housing units, for the alternatives was calculated using the City of Seattle's Energy Benchmarking data and CO2 emission coefficients from the U.S. Energy Information Administration (EIA). Because SCL is assumed to be carbon neutral through 2040, building emissions estimates include only those from combusted natural gas. The calculations use weather-normalized energy use intensity factors per square foot to estimate the GHG emissions from natural gas usage, adjusted to account for reductions due to planned and anticipated changes to Seattle's energy code.

Exhibit 3.2-15 shows existing and potential 2044 GHG Emissions from natural gas use in the study area under all alternatives.

Existing Conditions (2017) and Alternative 1 No Action (2042) (MTCO2e))	
Building Type	2017 Existing	2042 No Action	2044 Alt 2.	2044 Alt 3.	2044 Alt 4.	<u>2044</u> <u>Pref. Alt.</u>
Industrial	40,877	49,098	58,080	60,913	60,774	<u>52,175</u>
Non-Industrial	8,488	9,766	9,535	11,616	11,836	<u>10,602</u>
Total	49,365	58,864	67,615	72,528	72,610	<u>62,777</u>

Exhibit 3.2-15 Estimated Building-Related GHG Emissions for Action Alternatives Compared to Existing Conditions (2017) and Alternative 1 No Action (2042) (MTCO2e)

Source: Herrera, 202<u>2</u>1.

Exhibit 3.2-16 shows existing and potential 2044 GHG Emissions from housing units in the study area under all alternatives.

Exhibit 3.2-16	Estimated Housing-Related GHG Emissions for All Alternatives Compared to
	Existing Conditions (2021) and Alternative 1 No Action (2042) (MTCO2e)

Subarea	2021 Existing	2042 No Action	2044 Alt. 2	2044 Alt. 3	2044 Alt 4.	<u>2044</u> <u>Pref Alt</u>
Ballard	537	558	559	1,263	2,745	<u>1,973</u>
Interbay Dravus	8	29	31	218	498	<u>327</u>
Interbay Smith Cove	3	24	25	45	3	<u>3</u>
SODO/Stadium	59	143	148	618	2,826	<u>1,859</u>
Georgetown/South Park	548	611	615	716	1,219	<u>1,118</u>
Total	1,154	1,364	1,378	2,859	7,289	<u>5,280</u>
Added MU Housing						
With MIC Adjustments—Seattle Mixed-Use Zone Housing				3,013	3,013	<u>4,288</u>
Grand Total Source: Herrera, 202 2	1,154	1,364	1,378	5,872	10,302	<u>9,564</u>

Future building related GHG emissions from the use of natural gas are expected to increase under all alternatives, in line with increases in employment, building spaces, and housing. These results assume only the most recent changes to Seattle's energy code are in place in 2044, though it is reasonable to assume that future changes to the Code would further seek to reduce GHG emissions in line with updated climate action planning and that these future increases may be overestimated.

Other GHG Emissions

Because employment and some population would increase under all three Alternatives, waste generation and its associated GHG emissions would also increase. GHG emissions from solid waste generation were estimated using emission factors from the EPA's WARM model and the most recent (2018) waste generation rates from SPU. These emissions were then adjusted to account for waste diversion implemented through waste reduction, recycling, and composting fostered by the City's carbon-neutral goal target of 70% percent waste diversion by 2030.

Exhibit 3.2-17 shows existing and potential 2044 GHG Emissions from waste in the study area under all alternatives.

Subarea	Existing	2042 No Action	2044 Alt. 2	2044 Alt. 3	2044 Alt 4.	<u>2044</u> <u>Pref. Alt.</u>
C&D	—	<u>(332)(</u> 3)	<u>(586)(6)</u>	<u>(811)(8)</u>	<u>(821)(8)</u>	<u>(481)</u>
Industrial	<u>(526)(950)</u>	<u>(640)(1,176)</u>	<u>(766)(1,282)</u>	<u>(805)(1,503)</u>	<u>(803)(1,521)</u>	<u>(683)</u>
Non-Industrial	<u>(424)(526)</u>	<u>(536)(640)</u>	<u>(516)</u> (766)	<u>(698)(805)</u>	<u>(717)(803)</u>	<u>(609)</u>
Housing	<u>(2)</u> (424)	<u>(3)(536)</u>	<u>(3)(516)</u>	<u>(5)(698)</u>	<u>(14)(717)</u>	<u>(10)</u>
Total	<u>(952)(1,900)</u>	<u>(1,511)(2,356)</u>	<u>(1,870)(2,569)</u>	<u>(2,320)(3,015)</u>	<u>(2,356)(3,050)</u>	<u>(1,783)</u>

Exhibit 3.2-17 Estimated Waste-Related GHG Emissions for All Alternatives Compared to Existing Conditions and Alternative 1 No Action (MTCO2e)

Source: Herrera, 202<u>2</u>4.

Equity & Environmental Justice Considerations

Air Quality

While air quality impacts under all alternatives are expected to be less than significant, the primary equity and environmental justice concern for the proposal would be the emissions associated with industrial activities and road transportation emissions on vulnerable communities in the study area, on the periphery of industrial zones, and alongside higher-volume transportation routes. Depending on the transportation routes that are used, emissions of air pollutants from mobile sources could concentrate along routes that pass through vulnerable communities, leading to inequitable exposure to air pollution. Similar

effects could be experienced with activities related to employee and material transport during the construction phase of any of the alternatives.

At various thresholds of exposure, pollutants from mobile source operation can cause health effects such as cancer, asthma, and cardiovascular diseases, among others. Sensitivity to air pollution can depend on factors such as age, sex, and access to healthcare, the latter being correlated to income level. By race, asthma prevalence in the United States is greatest among American Indian/Alaska Natives and Black Americans (CDC 2019). Populations with preexisting conditions that make them more sensitive to air pollution could be at greater risk from the activities associated with the alternatives.

The incremental traffic-related emissions of the proposed alternatives would represent a minor portion of all traffic emissions on any transportation route near vulnerable communities. In addition, due to EPA emission standards for motor vehicles and clean fuel standards, the total emissions from road transportation are expected to drop even as traffic levels increase in the study area. Thus, exposures to air pollution in the study area are expected to continue trending downward.

Greenhouse Gases & Climate Change

GHG emissions under all alternatives are expected to have a potentially significant impact when combined with other global emissions, though mitigation opportunities, local and state climate actions, and expected continued regulatory changes would likely decrease the incremental contribution from the proposal to a moderate level of impact. The primary equity and environmental justice concern for the proposal would be the potential effect of emissions to accelerate climate change, which could disproportionately harm vulnerable communities in the study area. This could occur as the result of emissions from both the construction and operational phases of the proposal.

A new EPA analysis (EPA 2021) shows that the most severe harms from climate change fall disproportionately on vulnerable communities who are least able to prepare for, and recover from, exposure to extreme temperatures, poor air quality, flooding, sea level rise, and other impacts. EPA's analysis indicates that racial and ethnic minority communities are particularly vulnerable to the greatest impacts of climate change.

The incremental emissions of the proposed alternatives would represent a minor portion of all emissions that cumulatively contribute to climate change. However, planning for climate change should place emphasis on shoreline areas at risk from sea-level rise (see **Section 3.3 Water Resources**), among other risks, and prescribe adaptation measures that would help existing and new employees and residents, particularly vulnerable populations, in the MIC areas to reduce risks.

Impacts of Alternative 1 No Action

Air Quality

Under Alternative 1 future growth would continue based on current land use designations and comprehensive plan policies. No new land use concepts nor changes to MIC boundaries are proposed.

Transportation Related Emissions

Population and employment increases would continue, and area-wide VMT would increase in proportion. Projected changes in VMT were extracted from the projected travel demand model for cars, trucks, and buses. The travel demand model generally assumes existing economic and demographic trends continue with minor changes due primarily to mode share shifts and shortened trips due to increased traffic congestion. These changes cause projected VMT per capita to decline slightly by 2042. However, total VMT would continue to rise modestly due to population and employment growth.

The area wide estimated VMT for each of the MICs for the baseline year (2019) and the Alternative 1 No Action are presented in **Exhibit 3.2-18**.

		Р	PM Period VMT			PM Peak Hour VMT		
Geographic Area		2019 Existing	2042 No Action	Increase / Decrease	2019 Existing	2042 No Action	Increase / Decrease	
BINMIC	Cars	51,370	52,420	1,050	18,750	19,130	380	
	Trucks	2,550	2,760	210	930	1,010	80	
	Buses	920	920	0	340	340	0	
	Total	54,840	56,100	1,260	20,020	20,480	460	
Greater	Cars	531,320	516,020	-15,300	193,930	188,350	-5,580	
Duwamish MIC	Trucks	105,980	123,310	17,330	38,680	45,010	6,330	
	Buses	4,260	4,110	-150	1,550	1,500	-50	
	Total	641,560	643,440	1,880	234,160	234,860	700	

Exhibit 3.2-18 Estimated VMT For the Baseline Year (2019) And Alternative 1 No Action (2042)

PM Period = 3-6 PM

Net increase/decrease compares Alternative 1 with the Baseline year. Sources: Fehr & Peers, Herrera, 2021.

Under the Alternative 1 No Action, overall area-wide VMT could increase in the Greater Duwamish MIC by roughly 1,880 VMT during the PM period and 700 during the PM peak hour compared to the baseline year, and in the BINMIC by roughly 1,260 VMT during the PM period and 460 during the PM peak hour compared to the baseline year. In the Greater Duwamish MIC, the overall slight increase in total VMT includes an anticipated decrease in car VMT for the PM period and the PM peak hour, and a similar anticipated increase in truck VMT for the PM period and the PM peak hour. Overall slight increases in VMT for the BINMIC are also reflected across vehicle types.

Road transportation-related air pollutant emissions in each MIC for Alternative 1 No Action compared to existing conditions are presented in **Exhibit 3.2-19**. Anticipated for Seattle overall are shown for comparison.

Geographic Area	Pollutant	2019 Existing	2042 No Action	Increase / Decrease
BINMIC	СО	85.7	58.2	-27.5
	Nox	19.8	15.9	-3.9
	PM10	3.7	3.7	0.0
	PM2.5	0.7	0.7	-0.1
	VOC	4.3	3.2	-1.1
	Sox	0.2	0.2	0.0
Greater Duwamish MIC	СО	1,078.1	794.5	-283.6
	Nox	641.2	552.8	-88.5
	PM10	58.0	57.2	-0.8
	PM2.5	15.0	12.5	-2.6
	VOC	62.5	47.2	-15.3
	Sox	3.8	3.4	-0.5
Seattle	СО	4,783.0	<u>3,465.8</u> 3,459.5	<u>-1,317.2<mark>-1,323.4</mark></u>
	Nox	1,900.8	<u>1,645.8</u> 1,643.6	<u>-254.9</u> -257.2
	PM10	229.6	<u>234.9</u> 234.5	<u>5.3</u> 4.9
	PM2.5	52.9	<u>47.0</u> 4 6.9	<u>-6.0</u> -6.0
	VOC	256.6	<u>196.7</u> 196.3	<u>-59.9-60.3</u>
	Sox	14.7	<u>13.1</u> 13.1	<u>-1.6</u> -1.6

Exhibit 3.2-19 Estimated Tons of Criteria Pollutant Emissions from Road Transportation for Alternative 1 No Action (2042) Compared to Existing Conditions (2019)

Sources: Fehr & Peers, 2022; Herrera, 20224.

In addition to the road transportation emissions in **Exhibit 3.2-19**, vehicle travel would also generate PM2.5 through tire and brake wear and, more significantly, from entrained road dust. These non-vehicle emissions would not benefit from future improvements to the vehicle fleet as a whole or from improvements to fuel economy.

Regional emissions under Alternative 1 would be substantially lower than under existing background conditions. This is because the projected improvement in fuel economy, emission

reduction, and new technology implementation would offset the projected increase in VMT. Therefore, the No Action Alternative would result in a less than significant impact to air quality.

Land Use Change-Related Emissions

Under Alternative 1 No Action, existing Comprehensive Plan policies, development standards and zoning maps would dictate the patterns of development and the density of employment in the MIC areas. Alternative 1 No Action would result in continued growth in employment in the study area in 2044 compared to the baseline year of 2018 (see **Exhibit 3.8-13** in **Section 3.8 Land & Shoreline Use**). **Exhibit 3.2-7** on page 3-48 shows the square footage of industrial and non-industrial space in each MIC for existing conditions (2018) and anticipated under Alternative 1 No Action.

Where development occurs as current land use designations and Comprehensive Plan policies allow, and depending on the types of industry, those areas and employees would encounter the emissions resulting from existing and new industrial and other non-transportation air emissions. In addition, in areas with current industrial land use designations that maintain an industrial focus under new land use designations, residents or workers in adjacent areas with a residential or mixed-use focus could experience higher emissions resulting from industrial and other non-transportation air emissions. Areas particularly subject to these potential impacts include residential areas of Queen Anne and Magnolia adjacent to Interbay and commercial and mixed-use areas of Interbay itself, South Park, and Georgetown. However, with existing requirements for operating permits from PSCAA, these manufacturing plants, and other heavy and general industrial facilities are expected to remain compliant with air pollution control regulations that assure criteria air pollutant and TAP emissions meet standards, as they do currently.

Alternative 1 No Action would also result in some continued growth in housing in the study area in 2044 compared to the baseline year of 2018. **Exhibit 3.2-20** shows the number of housing units in each MIC for current conditions (2021) and anticipated under Alternative 1, No Action.

	Current Conditions (2021)		Alternative 1 No Action Existing Policies (2044)	
Subarea	Existing	Total	Growth	% Growth
Ballard	192	199	7	3.9%
Interbay Dravus	3	11	8	250.0%
Interbay Smith Cove	1	9	8	750.0%
SODO/Stadium	21	51	30	142.9%
Georgetown/South Park	196	218	22	11.5%
Total: Ind Zone Housing	413	488*	75*	18.2%

Exhibit 3.2-20 Estimated Number of Housing Units for Industrial Subareas Under Alternative 1 No Action (2044) Compared to the Current Conditions (2021)

Sources: Fehr & Peers, 2021; Herrera, 2021.

^{*}Rounded

Where housing within the industrial zones is <u>existing or would be</u> established, those residents would experience higher emissions resulting from industrial and other non-transportation air emissions. In addition, some of the housing units and anticipated growth, particularly in South Park <u>and Georgetown</u>, could be placed near major highways, rail lines, or port facilities that produce vehicle emissions in the highest concentrations. The DOH health disparities map (DOH 2021) indicates the South Park <u>and Georgetown</u> census tracts, including those surrounding SR 99 and SR509, as currently ranking <u>either a 9 or</u> a 10 out of 10 for a comparison of pollution burden from Diesel Nox emissions and social factors that may contribute to disparities across the state. Despite this potential, the combination of existing requirements for industrial operating permits from PSCAA, and ongoing requirements for improvements in vehicle emissions under Alternative 1 would be lower than under existing background conditions and Alternative 1 No Action would result in a less than significant impact to air quality, and a moderate but less than significant impact on health related to air quality.

Given this, it would be prudent to consider risk-reducing mitigation strategies such as setbacks for residential and other sensitive land uses from major traffic corridors, rail lines, port terminals and similar point sources of particulates from diesel fuel and/or to identify measures for sensitive populations proposed to be in areas near such sources.

Maritime Activities

Maritime activities that emit criteria pollutants within and adjacent to the MICs would be similar to those discussed and shown in **Exhibit 3.2-12**. With existing and planned regulatory requirements and local infrastructure improvements, these maritime emissions are expected to decrease under all alternatives, even if cargo volumes and cruise ship visits increase.

Greenhouse Gases & Climate Change

Changes in operational GHG emissions associated with development under Alternative 1 No Action would result from increases in VMT and improvements to the vehicle fleet, increased natural gas usage associated with new industrial and non-industrial development, and solid waste generation. These developments would be guided by existing Comprehensive Plan policies and existing land use designations. Potential operational GHG emissions from the Alternative 1 No Action are presented in **Exhibit 3.2-21**.

Source	Existing MTCO2e	2042 No Action MTCO2e
Transportation	703,522	613,158
Ind. And Non-Ind. Building—Gas	49,365	58,864
Housing	1,154	1,364
Waste	- <u>1,904</u> 3,799	- <u>2,690</u> 4,709
Total	<u>752,137</u> 750,242	6 <u>70,696</u> 68,677
Difference from Existing	0	-81, <u>441</u> 565
Difference from No Action	0	0
Source: Herrera, 202 <u>2</u> 1.		

Exhibit 3.2-21	Total Estimated Annual MTCO2e Emissions Under Alternative 1 No Action
	Compared to Existing Conditions

Total annual GHG emissions under Alternative 1 No Action could decrease by over 80,000 MTCO2e as compared to the baseline, which is the smallest increase in GHG emissions of all the alternatives when compared to existing conditions. However, this alternative contributes the least towards supporting growth and development for industrial and maritime uses, with less emphasis on development near existing and planned light rail transit. Growth that might otherwise be accommodated in the MIC buffer areas would occur in peripheral areas of the city or region where there are fewer jobs and services in close proximity, or fewer emission reduction policies driving change, resulting in greater net GHG emissions than are shown here.

Alternative 1 No Action would result in a less than significant impact for GHG emissions. None of the sources increases compared to the existing conditions by more than the 10,000 MTCO2e mandatory reporting threshold for the State of Washington for facilities. In fact, the increase in building natural gas emissions may be overestimated. Emissions associated with housing could also increase but by a small margin over existing conditions. In any case, taken as a whole, the individual source increases in GHG emissions are offset by decreases in all other source categories.

Impacts of Alternative 2

Air Quality

Alternative 2 could result in a very slight growth in overall VMT in the study area in 2044 compared to Alternative 1 No Action, and air quality impacts would be similar.

Transportation Related Emissions

Estimated VMT for the Greater Duwamish MIC and the BINMIC are presented in **Exhibit 3.2-22** comparing Alternative 1 No Action and Alternative 2.
		Ρ	PM Period VMT			PM Peak Hour VMT		
Geographic Area		2042 No Action	2044 Alt 2	Increase/ Decrease	2042 No Action	2044 Alt 2	Increase/ Decrease	
BINMIC	Cars	52,420	53,080	660	19,130	19,370	240	
	Trucks	2,760	2,900	140	1,010	1,060	50	
	Buses	920	920	0	340	340	0	
	Total	56,100	56,900	800	20,480	20,770	290	
Greater	Cars	516,020	520,080	4,060	188,350	189,830	1,480	
Duwamish MIC	Trucks	123,310	124,290	980	45,010	45,370	360	
	Buses	4,110	4,110	0	1,500	1,500	0	
	Total	643,440	648,480	5,040	234,860	236,700	1,840	

Exhibit 3.2-22 Estimated VMT For Alternative 2 (2044) Compared to Alternative 1 No Action (2042)

PM Period = 3-6 PM

Net increase/decrease compares Alternative 1 with the Baseline year.

Sources: Fehr & Peers, 2021; Herrera, 2021.

Under Alternative 2, VMT in the Greater Duwamish MIC could increase by roughly 5,040 in the PM period compared to Alternative 1 No Action and by 1,840 in the PM peak hour compared to Alternative 1. Most of those slight increases are from passenger cars. In the BINMIC, VMT could increase by roughly 800 in the PM period compared to Alternative 1 No Action and by 290 in the PM peak hour compared to Alternative 1.

Road transportation-related air pollutant emissions under Alternative 2 compared to Alternative 1 No Action are shown in **Exhibit 3.2-23** for both the Greater Duwamish MIC and the BINMIC. Anticipated for Seattle overall are shown for comparison.

Geographic Area	Pollutant	2042 No Action	2044 Alt 2	Increase/ Decrease
BINMIC	СО	58.2	59.2	1.0
	Nox	15.9	16.5	0.6
	PM10	3.7	3.8	0.1
	PM2.5	0.7	0.7	0.0
	VOC	3.2	3.2	0.1
	Sox	0.2	0.2	0.0
Greater Duwamish MIC	СО	794.5	800.7	6.2
	Nox	552.8	557.1	4.3
	PM10	57.2	57.7	0.4

Exhibit 3.2-23Estimated Tons of Criteria Pollutant Emissions from Road Transportation for
Alternative 2 (2044) Compared to Alternative 1 No Action (2042)

Ch.3 Environment, Impacts, & Mitigation Measures • Air Quality & GHG

Geographic Area	Pollutant	2042 No Action	2044 Alt 2	Increase/ Decrease
	PM2.5	12.5	12.5	0.1
	VOC	47.2	47.6	0.4
	Sox	3.4	3.4	0.0
Seattle	СО	<u>3,465.8</u> 3,459.5	<u>3,480.5</u> 3,474.2	<u>14.7</u> 14.7
	Nox	<u>1,645.8</u> 1,643.6	<u>1,656.7</u> 1,654.4	<u>10.9</u> 10.8
	PM10	<u>234.9</u> 234.5	<u>236.0</u> 235.6	<u>1.1</u> 1.1
	PM2.5	<u>47.0</u> 46.9	<u>47.2</u> 47.1	<u>0.2</u> 0.2
	VOC	<u>196.7</u> 196.3	<u>197.6</u> 197.2	<u>0.9</u> 0.9
	Sox	<u>13.1</u> 13.1	<u>13.2</u> 13.2	<u>0.1</u> 0.1

Sources: Fehr & Peers, 2022; Herrera, 20224.

Area wide road transportation pollutant emissions under Alternative 2 would also be substantially lower than under existing conditions, but slightly higher than Alternative 1. As with Alternative 1, this is because the projected improvement in fleet mix, emission reduction, and technology implementation due to fuel economy standards could offset this increase in VMT. <u>Generation of PM2.5 through tire and brake wear and from entrained road dust would also likely be greater with Alternative 2 than with Alternative 1 due to greater VMT. These nonvehicle emissions would not benefit from future improvements to the vehicle fleet as a whole or from improvements to fuel economy. Air emissions from the MIC areas under Alternative 2 as a percentage of overall City road transportation emissions would remain at or below that anticipated for Alternative 1 No Action. Therefore, Alternative 2 would likely result in a less than significant impact to air quality.</u>

Land Use Change-Related Emissions

Under Alternative 2, revised Comprehensive Plan policies, development standards and land use designations would result in generally more employment and additional development in the study area in 2044 compared to Alternative 1 No Action 2042. **Exhibit 3.2-7** on page 3-48 shows the square footage of industrial and non-industrial space in each MIC anticipated under Alternative 2 compared with Alternative 1 No Action, including the amount of anticipated growth.

As with Alternative 1 No Action, existing and new employees, depending on the types of businesses locating in the MICs, may encounter the emissions resulting from existing and new industrial and other non-transportation air emissions.

This alternative would place the emphasis for growth in industrial and maritime uses within appropriate land use zones, with a slight decrease in space devoted to non-industrial uses. Potentially a greater portion of projected growth in the MICs would be closer to and access major highway, rail line or port terminals, and contribute to the emissions from those sources. <u>As with Alternative 1 No Action, in areas with current industrial land use designations that maintain an industrial focus under new land use designations, residents or workers in adjacent areas with a</u>

residential or mixed-use focus could experience higher emissions resulting from industrial and other non-transportation air emissions in areas of Queen Anne and Magnolia, Interbay, South Park, and Georgetown. However, as shown in Exhibit 3.2-24, with existing requirements for operating permits from PSCAA, these manufacturing plants, and other heavy and general industrial facilities are expected to remain compliant with air pollution control regulations that assure criteria air pollutant and TAP emissions meet standards, as they do currently.

Alternative 2 would also result in some continued growth in housing in the study area in 2044 compared to Alternative 1 No Action 2044. Exhibit 3.2-24 shows the number of housing units in each MIC for Alternative 2 compared to those anticipated under Alternative 1, No Action.

	Alternative 1 No Action (2044)	Alternative 2 Future of Industry—Limited (2044)		
Subarea	Total Units	Total Units	Growth	% Growth
Ballard	199	200	1	0.3%
Interbay Dravus	11	11	0	4.8%
Interbay Smith Cove	9	9	0	5.9%
SODO/Stadium	51	53	2	3.9%
Georgetown/South Park	218	220	2	0.7%
Total: Ind Zone Housing	488*	493*	5	1.0%

Exhibit 3.2-24 Estimated Number of Housing Units for Industrial Subareas Under Alternative 2 (2044) Compared to Alternative 1 No Action (2044)

*Rounded

Source: City of Seattle, 2021.

Impacts to existing and new residents within and adjacent to the MICs under Alternative 2 would not be appreciably different from impacts under Alternative 1 No Action. Where housing within the industrial zones is existing or newly established, those residents would experience higher emissions resulting from industrial and other non-transportation air emissions. As with Alternative 1, the combination of existing requirements for industrial operating permits from PSCAA, and ongoing requirements for improvements in vehicle emissions control, fuel economy, technology improvements, and overall fuel mix, local emissions under Alternative 2 would be lower than under existing background conditions and Alternative 2 would result in a less than significant impact to air quality. Similar mitigation strategies should be considered.

Maritime Activities

Maritime activities and their impact on the Puget Sound air shed, including the MICs, would continue similarly as they would under Alternative 1 No Action. With existing and planned regulatory requirements and local infrastructure improvements, these maritime emissions are expected to decrease under all alternatives, even if cargo volumes and cruise ship visits increase.

Greenhouse Gases & Climate Change

GHG emissions under development of Alternative 2 were calculated using the same methodologies as those described previously but reflect the land use differences of increased industrial and non-industrial building space, added industry-supportive housing, and corresponding increased VMT in each of the MICs. These developments would be guided by changes to Comprehensive Plan policies and land use designations as outlined in the City's Industrial and Maritime Strategy and the resulting subarea plan policies to be developed. Operational GHG emissions from Alternative 2 are presented in **Exhibit 3.2-25**.

Source	No Action MTCO2e	Alt. 2 MTCO2e
Transportation	613,158	618,247
Ind. And Non-Ind. Building—Gas	58,864	67,615
Housing	1,364	1,378
Waste	- <u>2,690</u> 4, 709	- <u>3,154</u> 5 ,132
Total Difference from Existing Difference from No Action	6 <u>70,696</u> 68,677 -81, <u>441</u> 565 0	68 <u>4,0852,108</u> -68, <u>052134</u> 13, <u>389</u> 4 31

Exhibit 3.2-25 Total Estimated Annual MTCO2e Emissions Under Alternative 2 Compared to Alternative 1 No Action

Source: Herrera, 202<u>2</u>4.

Alternative 2 could decrease GHG emissions by approximately 68,000 MTCO2e per year compared to existing conditions but would represent an increase of over 13,000 MTCO2e compared to Alternative 1 No Action, which is above the 10,000 MTCO2e mandatory reporting threshold for the State of Washington. This is due largely to the GHG emissions associated with natural gas use with new industrial and non-industrial space increases compared to No Action conditions. As stated previously, these emissions may be overestimated.

Growth in the MICs that would otherwise be accommodated within other parts of the city would result in greater progress toward reducing overall transportation related emissions because the MICs have a high concentration of industrial and industry supporting jobs and services in close proximity with each other. This suggests that VMT per job could be lower in these areas than in most neighborhoods in the city. To the extent that Alternative 2 attracts growth that would otherwise occur outside of Seattle, it would result in an increase in total VMT within the city, making it more difficult to achieve City goals for a net reduction in citywide VMT over time.

It should be noted that despite the moderate increase in transportation-related emissions associated with VMT, Alternative 2 would support higher density growth patterns, particularly near planned light rail stations consistent with regional planning, as well as the long-term planning goals of the City's Comprehensive Plan and 2013 CAP, which are expected to assist in controlling GHG emissions. The Seattle Comprehensive Plan Final EIS (2016) presented analysis that showed that the VMT per job and resident in Seattle would be approximately 40% lower than VMT per job and resident outside of Seattle (City of Seattle, 2016b). Therefore, by increasing employment density in the MICs, Alternative 2 could contribute to regional efforts to limit vehicular GHG emissions.

Overall, Alternative 2 could result in an increase in GHG emissions compared to Alternative 1— No Action that could be considered potentially significant and additional mitigation measures would be warranted.

Impacts of Alternative 3

Air Quality

Alternative 3 could result in more robust growth in the study area in overall employment, industrial and non-industrial development, and in housing compared to Alternative 1 No Action and Alternative 2.

Transportation Related Emissions

Alternative 3 could result in a slight growth in overall VMT in the study area compared to Alternative 1 No Action and Alternative 2, but air quality impacts would be similar. Estimated VMT for the Greater Duwamish MIC and the BINMIC are presented in **Exhibit 3.2-26** comparing Alternative 1 No Action and Alternative 3.

		PM Period VMT		т	PM Peak Hour VMT		/MT
Geographic Area		2042 No Action	2044 Alt 3	Increase/ Decrease	2042 No Action	2044 Alt 3	Increase/ Decrease
BINMIC	Cars	52,420	54,700	2,280	19,130	19,970	840
	Trucks	2,760	2,920	160	1,010	1,070	60
	Buses	920	920	0	340	340	0
	Total	56,100	58,540	2,440	20,480	21,380	900
Greater	Cars	516,020	529,650	13,630	188,350	193,320	4,970
Duwamish MIC	Trucks	123,310	124,290	980	45,010	45,370	360
	Buses	4,110	4,110	0	1,500	1,500	0
	Total	643,440	658,050	14,610	234,860	240,190	5,330

Exhibit 3.2-26 Estimated VMT for Alternative 3 (2044) Compared to Alternative 1 No Action (2042)

PM Period = 3-6 PM

Net increase/decrease compares Alternative 1 with the Baseline year.

Sources: Fehr & Peers, Herrera, 2021.

Under Alternative 3, VMT in the Greater Duwamish MIC could increase by roughly 14,610 in the PM period compared to Alternative 1 No Action and by 2,440 in the PM peak hour compared to Alternative 1. Like Alternative 2, most of those increases are from passenger cars. In the BINMIC, VMT could increase by roughly 5,330 in the PM period compared to Alternative 1 No Action and by 900 in the PM peak hour compared to Alternative 1.

Road transportation-related air pollutant emissions under Alternative 3 compared to Alternative 1 No Action are shown in **Exhibit 3.2-27** for both the Greater Duwamish MIC and the BINMIC. Anticipated for Seattle overall are shown for comparison.

Geographic Area	Pollutant	2042 No Action	2044 Alt 3	Increase/ Decrease
BINMIC	CO	58.2	60.7	2.5
	Nox	15.9	16.6	0.7
	PM10	3.7	3.9	0.2
	PM2.5	0.7	0.7	0.0
	VOC	3.2	3.3	0.1
	Sox	0.2	0.2	0.0
Greater Duwamish MIC	СО	794.5	809.6	15.1
	Nox	552.8	557.2	4.4
	PM10	57.2	58.2	1.0
	PM2.5	12.5	12.6	0.2
	VOC	47.2	48.0	0.8
	Sox	3.4	3.4	0.0
Seattle	СО	<u>3,465.8</u> 3,459.5	<u>3,504.7</u> 3,498.9	<u>39.0</u> 39.4
	Nox	<u>1,645.8</u> 1,643.6	<u>1,657.1</u> 1,654.8	<u>11.3</u> 11.3
	PM10	<u>234.9</u> 234.5	<u>237.4</u> 237.1	<u>2.5</u> 2.5
	PM2.5	<u>47.0</u> 46.9	<u>47.4</u> 47.4	<u>0.5</u> 0.5
	VOC	<u>196.7</u> 196.3	<u>198.8</u> 198.5	<u>2.12.2</u>
	Sox	<u>13.1</u> 13.1	<u>13.3</u> 13.2	<u>0.10.1</u>

Exhibit 3.2-27 Estimated Tons of Criteria Pollutant Emissions from Road Transportation for Alternative 3 (2044) Compared to Alternative 1 No Action (2042)

Sources: Fehr & Peers, 20224; Herrera, 20224.

Area wide road transportation pollutant emissions under Alternative 3 would also be substantially lower than under existing conditions, but slightly higher than alternatives 1 and 2. As with the other alternatives, this is because the projected improvement in fleet mix, emission reduction, and technology implementation due to fuel economy standards could offset this increase in VMT. <u>Generation of PM2.5 through tire and brake wear and from entrained road</u> <u>dust would also likely be slightly higher than alternatives 1 and 2 due to greater VMT and</u> <u>because these emissions would not benefit from future improvements to the vehicle fleet as a</u> <u>whole or from improvements to fuel economy.</u> Air emissions from the MIC areas under Alternative 3 as a percentage of overall City road transportation emissions would remain at or belowsimilar to that anticipated for Alternative 1 No Action. Therefore, Alternative 3 would likely result in a less than significant impact to air quality.

Land Use Change-Related Emissions

Compared to Alternative 2, this alternative would increase the acreage within the MICs that would be redesignated for use in proposed Industry / Innovation and Urban Industrial zones in targeted geographies, including an estimated 1/2 mile from planned light rail stations. Some of the projected growth would likely be closer to existing and future sources of industrial, transportation, and non-transportation emissions and associated risks. Like the other alternatives, this growth includes new development for industrial and non-industrial employment. **Exhibit 3.2-7** on page 3-48 shows the square footage of industrial and non-industrial and non-industrial space in each MIC anticipated under Alternative 3 compared with Alternative 1 No Action, including the amount of anticipated growth.

As with the other alternatives, existing and new employees, depending on the types of businesses locating in the MICs, may encounter the emissions resulting from existing and new industrial and other non-transportation air emissions.

This alternative would also place the emphasis for growth in industrial and maritime uses within appropriate land use zones, as well as allowances for moderate growth in space devoted to non-industrial uses. Potentially a greater portion of projected growth in the MICs would be closer to and access major highway, rail line or port terminals, and contribute to the emissions from those sources. As with alternatives 1 and 2, in areas with current industrial land use designations that maintain an industrial focus under new land use designations, residents or workers in adjacent areas with a residential or mixed-use focus could experience higher emissions resulting from industrial and other non-transportation air emissions in areas of Queen Anne and Magnolia, Interbay, South Park, and Georgetown. However, as shown in **Exhibit 3.2-28**, with existing requirements for operating permits from PSCAA, these manufacturing plants, and other heavy and general industrial facilities are expected to remain compliant with air pollution control regulations that assure criteria air pollutant and TAP emissions meet standards, as they do currently.

Alternative 3 would result in a much greater growth in housing in the study area in 2044 compared to Alternative 2 and Alternative 1 No Action. **Exhibit 3.2-28** shows the number of housing units in each MIC for Alternative 3 compared to those anticipated under Alternative 1, No Action.

	Alternative 1 No Action (2044)	Alternative 3 Future of Industry—Targeted (2044)		
Subarea	Total Units	Total Units	Growth	% Growth
Ballard	199	452	253	126.6%
Interbay Dravus	11	78	67	642.9%
Interbay Smith Cove	9	16	7	88.2%
SODO/Stadium	51	221	170	333.3%
Georgetown/South Park	218	256	38	17.2%
Total: Ind Zone Housing	488*	1,023	535	109.6%

Exhibit 3.2-28 Estimated Number of Housing Units for Industrial Subareas Under Alternative 3 (2044) Compared to Alternative 1 No Action (2044)

*Rounded

Sources: City of Seattle, 2021.

In addition to increased industrial zone caretakers' quarters/makers' space of 535 units there would be an increase in residential development in land removed from the MIC that would be rezoned to Seattle Mixed. This would mean an increase in dwellings of 1,078 units in the Georgetown and South Park areas.

Impacts to existing and new residents within and adjacent to the MICs under Alternative 3 have the potential to be greater than the impacts under both Alternative 2 and Alternative 1 No Action. This is due mostly to the greater number of employees and residents within the MICs resulting from anticipated development. Where housing within the industrial zones is established, those residents would experience higher emissions resulting from industrial and other non-transportation air emissions. In SODO/Stadium, where over 30% of the housing growth is to occur is also adjacent to areas of high-capacity highways, major commute arterials, and a busy rail corridor. In Georgetown, where the triangular area bounded by Corson Avenue S, Carleton Avenue S and I-5 would be removed from the MIC and placed into a mixed-use zone and in the areas to be designated as Urban Industrial, existing or new residents would experience higher emissions resulting from nearby industrial, transportation, and other nontransportation air emissions, including the WSDOT Corson facility on Corson Avenue S.

However, as with alternatives 1 and 2, the combination of existing requirements for industrial operating permits from PSCAA, and ongoing requirements for improvements in vehicle emissions control, fuel economy, technology improvements, and overall fuel mix, local emissions under Alternative 3 would be lower than under existing background conditions. While rail emissions were not calculated for this assessment as they are not affected by the proposed action, they do contribute to the overall cumulative air emissions in the MICs. Nonetheless, Alternative 3 would likely result in a less than significant impact to air quality. Similar mitigation strategies as have been mentioned for the other alternatives should be considered.

Maritime Activities

Maritime activities and their impact on the Puget Sound air shed, including the MICs, would continue similarly as they would under Alternative 1 No Action. With existing and planned regulatory requirements and local infrastructure improvements, these maritime emissions are expected to decrease under all alternatives, even if cargo volumes and cruise ship visits increase.

Greenhouse Gases & Climate Change

GHG emissions under development of Alternative 3 reflect greater increases in industrial and non-industrial building space, added industry-supportive housing, added mixed-uses, and corresponding increased VMT in each of the MICs. These developments would also be guided by changes to Comprehensive Plan policies and land use designations as outlined in the City's Industrial and Maritime Strategy and the resulting subarea plan policies to be developed. Operational GHG emissions from Alternative <u>3</u>² are presented in **Exhibit 3.2-29**.

Source	No Action MTCO2e	Alt. 3 MTCO2e
Transportation	613,158	623,437
Ind. And Non-Ind. Building—Gas	58,864	72,528
Housing	1,364	5,872
Waste	- <u>2,690</u> 4 ,709	- <u>3,828</u> 6,022
Total Difference from Existing	6 <u>70,69668,677 -81,<u>441</u>565</u>	69 <u>8,010</u> 5,816 <i>-54,<u>127</u>425</i>
Difference from No Action	0	27 <u>,314</u> 139

Exhibit 3.2-29 Total Estimated Annual MTCO2e Emissions Under Alternative 3 Compared to Alternative 1 No Action

Source: Herrera, 202<u>2</u>1.

Alternative 3 could decrease GHG emissions by approximately 54,000 MTCO2e per year compared to existing conditions but would represent an increase of over 27,000 MTCO2e compared to Alternative 1 No Action, which is above the 10,000 MTCO2e mandatory reporting threshold for the State of Washington. As with Alternative 2, this is due largely to the GHG emissions associated with natural gas use with new industrial and non-industrial space but also includes increases from the addition of approximately 1,600 housing units compared to Alternative 1.

Like Alternative 2, reducing transportation related emissions through increasing density of employment growth in the MICs rather than in other Seattle neighborhoods or regionally would be consistent for Alternative 3. It should be noted for Alternative 3 also that despite the moderate increase in transportation-related emissions associated with VMT, Alternative 3 would support higher density growth patterns, particularly near planned light rail stations consistent with regional planning, as well as the long-term planning goals of the City's

Comprehensive Plan and 2013 CAP, resulting in contributions to regional efforts to limit vehicular GHG emissions.

Overall, Alternative 3 could result in an increase in GHG emissions compared to Alternative 1— No Action that could be considered potentially significant and additional mitigation measures would be warranted.

Impacts of Alternative 4

Air Quality

Alternative 4 could also result in more robust growth in the study area in 2044 in overall employment, industrial and non-industrial development, and the most growth in housing compared to Alternative 1—No Action and the other alternatives.

Transportation Related Emissions

Alternative 4 could result in a slight growth in overall VMT in the study area in 2044 compared to Alternative 1—No Action and Alternative 2 and similar to Alternative 3; air quality impacts would also be similar. Estimated VMT for the Greater Duwamish MIC and the BINMIC are presented in **Exhibit 3.2-30** comparing Alternative 1—No Action and Alternative 4.

			PM Period VMT			PM Peak Hour VMT		
Geographic Area		2042 No Action	2044 Alt 4	Increase/ Decrease	2042 No Action	2044 Alt 4	Increase/ Decrease	
BINMIC	Cars	52,420	55,110	2,690	19,130	20,120	990	
	Trucks	2,760	2,950	190	1,010	1,080	70	
	Buses	920	920	0	340	340	0	
	Total	56,100	58,980	2,880	20,480	21,540	1,060	
Greater Duwamish MIC	Cars	516,020	<u>531,120</u> 529,500	<u>15,100</u> 13,480	188,350	193,270	4,920	
	Trucks	123,310	124,290	980	45,010	45,370	360	
	Buses	4,110	4,110	0	1,500	1,500	0	
	Total	643,440	<u>659,520</u> 657,900	<u>16,080</u> 14,460	234,860	240,140	5,280	

Exhibit 3.2-30 Estimated VMT For Alternative 4 (2044) Compared to Alternative 1 No Action (2042)

PM Period = 3-6 PM

Net increase/decrease compares Alternative 1 with the Baseline year.

Sources: Fehr & Peers, 20221; Herrera, 20221.

Under Alternative 4, VMT in the Greater Duwamish MIC could increase by roughly <u>14,46016,080</u> in the PM period compared to Alternative 1 No Action and by <u>2,8805,280</u> in the PM peak hour compared to Alternative 1. Like the other alternatives, most of those increases are from passenger cars. In the BINMIC, VMT could increase by roughly <u>5,2802,880</u> in the PM period compared to Alternative 1 No Action and by 1,060 in the PM peak hour compared to Alternative 1.

Road transportation-related air pollutant emissions under Alternative 4 compared to Alternative 1 No Action are shown in **Exhibit 3.2-31** for both the Greater Duwamish MIC and the BINMIC. Anticipated for Seattle overall are shown for comparison.

Geographic Area	Pollutant	2042 No Action	2044 Alt 4	Increase/ Decrease
BINMIC	CO	58.2	61.2	3.0
	Nox	15.9	16.7	0.9
	PM10	3.7	3.9	0.2
	PM2.5	0.7	0.7	0.0
	VOC	3.2	3.3	0.2
	Sox	0.2	0.2	0.0
Greater Duwamish MIC	СО	794.5	809.5	15.0
	Nox	552.8	557.2	4.4
	PM10	57.2	58.2	1.0
	PM2.5	12.5	12.6	0.2
	VOC	47.2	48.0	0.8
	Sox	3.4	3.4	0.0
Seattle	СО	<u>3,465.8</u> 3,459.5	<u>3,508.5</u> 3,499.0	<u>42.7</u> 39.5
	Nox	<u>1,645.8</u> 1,643.6	<u>1,657.2</u> 1,654.8	<u>11.4</u> 11.3
	PM10	<u>234.9</u> 234.5	<u>237.7</u> 237.1	<u>2.7</u> 2.6
	PM2.5	<u>47.0</u> 46.9	<u>47.5</u> 47.4	<u>0.5</u> 0.5
	VOC	<u>196.7</u> 196.3	<u>199.0</u> 198.5	<u>2.32.2</u>
	Sox	<u>13.1</u> 13.1	<u>13.3</u> 13.2	<u>0.10.1</u>

Exhibit 3.2-31 Estimated Tons of Criteria Pollutant Emissions from Road Transportation for Alternative 4 (2044) Compared to Alternative 1 No Action (2042)

Sources: Fehr & Peers, 20224; Herrera, 20224.

Area wide road transportation pollutant emissions under Alternative 4 would also be substantially lower than under existing conditions, but slightly higher than the other alternatives. As with the other alternatives, this is because the projected improvement in fleet mix, emission reduction, and technology implementation due to fuel economy standards could offset this increase in VMT. <u>Generation of PM2.5 through tire and brake wear and from</u> <u>entrained road dust would also likely be slightly higher than the other alternatives due to</u> <u>greater VMT and because these emissions would not benefit from future improvements to the</u> <u>vehicle fleet as a whole or from improvements to fuel economy.</u> Air emissions from the MIC areas under Alternative 4 as a percentage of overall City road transportation emissions would remain at or below that anticipated for Alternative 1 No Action. Therefore, Alternative 4 would likely result in a less than significant impact to air quality.

Land Use Change-Related Emissions

Compared to Alternative 2, Alternative 4 would increase the acreage within the MICs that would be redesignated for use in proposed Industry / Innovation and Urban Industrial zones in targeted geographies, including an estimated 1/2 mile from planned light rail stations. Alternative 4 would designate slightly less than Alternative 3 in this regard. As with Alternative 3, some of the projected growth under Alternative 4 would likely be closer to existing and future sources of industrial, transportation, and non-transportation emissions and associated risks. Like the other alternatives, this growth under Alternative 4 includes new development for industrial and non-industrial employment. **Exhibit 3.2-7** on page 3-48 shows the square footage of industrial and non-industrial space in each MIC anticipated under Alternative 4 compared with Alternative 1 No Action, including the amount of anticipated growth.

As with the other alternatives, existing and new employees, depending on the types of businesses locating in the MICs, may encounter the emissions resulting from existing and new industrial and other non-transportation air emissions.

This alternative would also place the emphasis for growth in industrial and maritime uses within appropriate land use zones, as well as allowances for moderate growth in space devoted to non-industrial uses. Like Alternative 3, Alternative 4 projected growth in the MICs would be closer to and use access to major highway, rail line or port terminals, and contribute to the emissions from those sources. Like the other alternatives, in areas with current industrial land use designations that maintain an industrial focus under new land use designations, residents or workers in adjacent areas with a residential or mixed-use focus could experience higher emissions resulting from industrial and other non-transportation air emissions in areas of Queen Anne and Magnolia, Interbay, South Park, and Georgetown. However, as shown in **Exhibit 3.2-32**, with existing requirements for operating permits from PSCAA, these manufacturing plants, and other heavy and general industrial facilities are expected to remain compliant with air pollution control regulations that assure criteria air pollutant and TAP emissions meet standards, as they do currently.

Alternative 4 would result the greatest growth in housing in the study area in 2044 compared to the other alternatives and Alternative 1 No Action. **Exhibit 3.2-32** shows the number of housing units in each MIC for Alternative 4 compared to those anticipated under Alternative 1, No Action.

	Alternative 1 No Action (2044)	Alternative 4 Future of Industry—Expanded (2044)		
Subarea	Total Units	Total Units	Growth	% Growth
Ballard	199	982	783	392.2%
Interbay Dravus	11	178	167	1595.2%
Interbay Smith Cove	9	1	-8	-88.2%
SODO/Stadium	51	1011	960	1882.4%
Georgetown/South Park	218	436	218	99.5%
Total: Ind Zone Housing	488*	2,608	2,120	434.4%

Exhibit 3.2-32 Estimated Number of Housing Units for Industrial Subareas Under Alternative 4 (2044) Compared to Alternative 1 No Action (2044)

* Rounded

Sources: City of Seattle, 2021.

In addition to increased industrial zone caretakers' quarters/makers' studios of 2,120 units above Alternative 1 No Action there would be an increase in residential development in land removed from the MIC that would be rezoned to Seattle Mixed. This would mean an increase in dwellings of 1,078 units in the Georgetown and South Park areas.

Impacts to existing and new residents within and adjacent to the MICs under Alternative 4 have the potential towould likely be greater than the impacts under all-other alternatives and Alternative 1 No Action. This is due mostly to the highest number combination of employees and residents within the MICs resulting from anticipated development. Where housing within the industrial zones is established, those residents would experience higher emissions resulting from industrial and other non-transportation air emissions. In-The SODO/Stadium Subarea, where 45% of the housing growth is to occur, is also adjacent to areas of high-capacity highways, major commute arterials, and a busy rail corridor. As with Alternative 3, in Georgetown—where the triangular area bounded by Corson Avenue S, Carleton Avenue S, and I-5 would be removed from the MIC and placed into a mixed-use zone and in the areas to be designated as Urban Industrial—existing or new residents would experience higher emissions, including the WSDOT Corson facility on Corson Avenue S.

However, as with all other alternatives, the combination of existing requirements for industrial operating permits from PSCAA, and ongoing requirements for improvements in vehicle emissions control, fuel economy, technology improvements, and overall fuel mix, local emissions under Alternative 4 would be lower than under existing background conditions. Similar cumulative air emissions from rail would occur in the MICs under all alternatives. Nonetheless, Alternative 3 would likely result in a less than significant impact to air quality. Similar mitigation strategies as have been mentioned for the other alternatives should be considered.

Maritime Activities

Maritime activities and their impact on the Puget Sound air shed, including the MICs, would continue similarly as they would under Alternative 1 No Action. With existing and planned regulatory requirements and local infrastructure improvements, these maritime emissions are expected to decrease under all alternatives, even if cargo volumes and cruise ship visits increase.

Greenhouse Gases & Climate Change

GHG emissions under development of Alternative 4 reflect the greatest increases in industrysupportive housing, and office uses in places served by light rail within the MICs, and added mixed-uses slightly smaller than the Preferred Alternative, and, along with increases in industrial and non-industrial building space slightly smaller than Alternative 3. VMT increases for Alternative 4 are anticipated at about the same as Alternative 3 for the Greater Duwamish MIC and slightly greater than Alternative 3 for the BINMIC. Operational Total estimated GHG emissions from Alternative <u>42</u> are presented in **Exhibit 3.2-33**.

Source	No Action MTCO2e	Alt 4. MTCO2e
Transportation	613,158	<u>624,379</u> 623,635
Ind. And Non-Ind. Building—Gas	58,864	72,610
Housing	1,364	10,302
Waste	- <u>2,690</u> 4, 709	- <u>3,890</u> 6,091
Total	6 <u>70,696</u> 68,677	<u>703,401</u> 700,456
Difference from Existing	-81, <u>441</u> 565	- <u>48,736</u> 4 9,785
Difference from No Action	0	<u>32,705</u> 31,779
Courses Herrore 202 21		

Exhibit 3.2-33 Total Estimated Annual MTCO2e Emissions Under Alternative 43 Compared to **Alternative 1 No Action**

Source: Herrera, 202<u>2</u>1.

Alternative 4 could decrease GHG emissions by approximately 5049,000 MTCO2e per year compared to existing conditions but would represent an increase of almost more than 32,000 MTCO2e compared to Alternative 1 No Action, which is above the 10,000 MTCO2e mandatory reporting threshold for the State of Washington. Compared to the other alternatives, Alternative 4 results in increases in all source categories except waste, most notably different from the other alternatives those associated with increased housing.

Like alternatives 2 and 3, reducing transportation related emissions through increasing density of employment growth in the MICs rather than in other Seattle neighborhoods or regionally would be consistent for Alternative 4, despite the moderate increase in transportation-related emissions in the MIC areas.

Overall, Alternative 4 could result in an increase in GHG emissions compared to Alternative 1— No Action that could be considered potentially significant and additional mitigation measures would be warranted.

Impacts of the Preferred Alternative

Air Quality

The Preferred Alternative could also result in growth in the study area in 2044 in overall employment, industrial and non-industrial development, and growth in housing compared to Alternative 1—No Action and alternatives 2 and 3, but less than Alternative 4.

Transportation Related Emissions

The Preferred Alternative could result in a slight growth in overall VMT in the study area in 2044 compared to Alternative 1—No Action and similar to Alternative 2; air quality impacts would also be similar. Estimated VMT for the Greater Duwamish MIC and the BINMIC are presented in **Exhibit 3.2-34** comparing Alternative 1—No Action and the Preferred Alternative.

		PM Period VMT		<u>PM</u>	<u>MT</u>		
<u>Geographic</u> <u>Area</u>		<u>2042</u> <u>No Action</u>	<u>2044</u> <u>Pref. Alt.</u>	<u>Increase/</u> <u>Decrease</u>	<u>2042</u> <u>No Action</u>	<u>2044</u> <u>Pref. Alt.</u>	<u>Increase/</u> <u>Decrease</u>
<u>BINMIC</u>	<u>Cars</u>	<u>52,420</u>	<u>53,800</u>	<u>1,380</u>	<u>19,130</u>	<u>19,640</u>	<u>510</u>
	<u>Trucks</u>	<u>2,760</u>	<u>2,880</u>	<u>120</u>	<u>1,010</u>	<u>1,050</u>	<u>40</u>
	<u>Buses</u>	<u>920</u>	<u>920</u>	<u>0</u>	<u>340</u>	<u>340</u>	<u>0</u>
	<u>Total</u>	<u>56,100</u>	<u>57,600</u>	<u>1,500</u>	<u>20,480</u>	<u>21,030</u>	<u>550</u>
<u>Greater</u> Duwamish MIC	<u>Cars</u>	<u>516,020</u>	<u>522,640</u>	<u>6,620</u>	<u>188,350</u>	<u>190,760</u>	<u>2,410</u>
	<u>Trucks</u>	<u>123,310</u>	<u>123,200</u>	<u>-110</u>	<u>45,010</u>	<u>44,970</u>	<u>-40</u>
	<u>Buses</u>	<u>4,110</u>	<u>4,110</u>	<u>0</u>	<u>1,500</u>	<u>1,500</u>	<u>0</u>
	<u>Total</u>	<u>643,440</u>	<u>649,950</u>	<u>6,510</u>	<u>234,860</u>	237,230	<u>2,370</u>

Exhibit 3.2-34 Estimated VMT For the Preferred Alternative (2044) Compared to Alternative 1 No Action (2042)

PM Period = 3-6 PM

Net increase/decrease compares Alternative 1 with the Baseline year.

Sources: Fehr & Peers, 2022; Herrera, 2022.

<u>Under the Preferred Alternative, VMT in the Greater Duwamish MIC could increase by roughly</u> <u>6,510 in the PM period compared to Alternative 1 No Action and by 2,370 in the PM peak hour</u> <u>compared to Alternative 1. Like the other alternatives, most of those increases are from</u> passenger cars. In the BINMIC, VMT could increase by roughly 1,500 in the PM period compared to Alternative 1 No Action and by 550 in the PM peak hour compared to Alternative 1.

<u>Road transportation-related air pollutant emissions under the Preferred Alternative compared</u> to Alternative 1 No Action are shown in **Exhibit 3.2-35** for both the Greater Duwamish MIC and the BINMIC. Emissions anticipated for Seattle overall are shown for comparison.

<u>Geographic Area</u>	<u>Pollutant</u>	<u>2042 No Action</u>	2044 Pref Alt	<u>Increase/</u> Decrease
BINMIC	<u>CO</u>	58.2	<u>59.8</u>	<u>1.6</u>
	Nox	<u>15.9</u>	<u>16.4</u>	<u>0.5</u>
	<u>PM10</u>	<u>3.7</u>	<u>3.8</u>	<u>0.1</u>
	<u>PM2.5</u>	<u>0.7</u>	<u>0.7</u>	<u>0.0</u>
	VOC	<u>3.2</u>	<u>3.2</u>	<u>0.1</u>
	Sox	<u>0.2</u>	<u>0.2</u>	<u>0.0</u>
Greater Duwamish MIC	<u>CO</u>	<u>794.5</u>	<u>800.4</u>	<u>5.9</u>
	Nox	<u>552.8</u>	<u>552.4</u>	<u>-0.4</u>
	<u>PM10</u>	<u>57.2</u>	<u>57.6</u>	<u>0.3</u>
	<u>PM2.5</u>	<u>12.5</u>	<u>12.5</u>	<u>0.1</u>
	VOC	<u>47.2</u>	<u>47.5</u>	<u>0.3</u>
	<u>Sox</u>	<u>3.4</u>	<u>3.4</u>	<u>0.0</u>
Seattle	<u>CO</u>	<u>3,465.8</u>	<u>3,484.8</u>	<u>19.1</u>
	Nox	<u>1,645.8</u>	<u>1,651.4</u>	<u>5.5</u>
	<u>PM10</u>	<u>234.9</u>	<u>236.2</u>	<u>1.2</u>
	<u>PM2.5</u>	<u>47.0</u>	<u>47.2</u>	<u>0.2</u>
	VOC	<u>196.7</u>	<u>197.7</u>	<u>1.0</u>
	Sox	<u>13.1</u>	<u>13.2</u>	<u>0.1</u>

Exhibit 3.2-35Estimated Tons of Criteria Pollutant Emissions from Road Transportation for thePreferred Alternative (2044) Compared to Alternative 1 No Action (2042)

Sources: Fehr & Peers, 2022; Herrera, 2022.

Area wide road transportation pollutant emissions under the Preferred Alternative would also be substantially lower than under existing conditions, and nearly the same as with Alternative 2. Transportation related emission impacts would be similar to Alternative 2, and as a percentage of overall City road transportation emissions would remain at or below that anticipated for Alternative 1 No Action. Therefore, the Preferred Alternative would likely result in a less than significant impact to air quality.

Land Use Change-Related Emissions

Compared to Alternative 4, the Preferred Alternative would decrease the acreage within the MICs that would be redesignated for use in proposed Industry/Innovation and MML and increase the acreage for use in Urban Industrial zones in targeted geographies, including an estimated 1/2 mile from planned light rail stations. The Preferred Alternative would also designate additional acreage to mixed uses compared to all other alternatives, with the same increase as Alternative 4 in Georgetown and South Park, and new acreages in West Ballard and Judkins Park; overall, a higher total amount of housing production outside of MICs would result compared to Draft EIS alternatives—an additional 1,534 dwellings, 42% more than alternatives 3 and 4. Like the other alternatives, this growth under the Preferred Alternative includes new development for industrial and non-industrial employment. **Exhibit 3.2-7** on page 3-48 shows the square footage of industrial and non-industrial space in each MIC anticipated under the Preferred Alternative compared with Alternative 1 No Action.

As with the other alternatives, existing and new employees, depending on the types of businesses locating in the MICs, may encounter the emissions resulting from existing and new industrial and other non-transportation air emissions.

This alternative would also place the emphasis for growth in industrial and maritime uses within appropriate land use zones, as well as a more focused distribution of space devoted to non-industrial uses similar to Alternative 4 inside the MICs with elements of Alternative 1 outside the MICs. Like Alternative 4, projected growth in the MICs would be closer to and use access to major highway, rail line, or port terminals, and would contribute to the emissions from those sources. Like the other alternatives, in areas with current industrial land use designations that maintain an industrial focus under new land use designations, residents or workers in adjacent areas with a residential or mixed-use focus could experience higher emissions resulting from industrial and other non-transportation air emissions in areas of Queen Anne and Magnolia, Interbay, South Park, and Georgetown. However, as shown in **Exhibit 3.2-35**, with existing requirements for operating permits from PSCAA, these manufacturing plants and other heavy and general industrial facilities are expected to remain compliant with air pollution control regulations that assure criteria air pollutant and TAP emissions meet standards.

The Preferred Alternative would result in more housing growth in the study area in 2044 compared to Alternative 1 No Action and alternatives 2 and 3, but less than Alternative 4. **Exhibit 3.2-36** shows the number of housing units in each MIC for the Preferred Alternative compared to those anticipated under Alternative 1 No Action.

	<u>Alternative 1</u> <u>No Action (2044)</u>	<u>Preferred Alternative</u> <u>Balanced (2044)</u>		
<u>Subarea</u>	<u>Total Units</u>	<u>Total Units</u>	<u>Growth</u>	<u>% Growth</u>
Ballard	<u>199</u>	<u>706</u>	<u>507</u>	<u>254%</u>
Interbay Dravus	<u>11</u>	<u>117</u>	<u>107</u>	<u>1014%</u>
Interbay Smith Cove	<u>9</u>	<u>1</u>	<u>-8</u>	<u>-88%</u>
SODO/Stadium	<u>51</u>	<u>665</u>	<u>614</u>	<u>1204%</u>
Georgetown/South Park	<u>218</u>	<u>400</u>	<u>182</u>	<u>83%</u>
Total: Ind Zone Housing	<u>488*</u>	<u>1,888</u>	<u>1,400*</u>	<u>287%</u>

Exhibit 3.2-36 Estimated Number of Housing Units for Industrial Subareas Under the Preferred Alternative (2044) Compared to Alternative 1 No Action (2044)

*Rounded

Sources: City of Seattle, 2022.

In addition to increased industrial zone caretakers' quarters/makers' studios of 1,400 units above Alternative 1 No Action, there would be an increase in residential development in land removed from and outside the MIC that would be rezoned to Seattle Mixed. This would mean an increase in dwellings of 1,534 units in the Georgetown, South Park, west Ballard, and Judkins Park areas.

Impacts to existing and new residents within and adjacent to the MICs under the Preferred Alternative have the potential to be greater than the impacts under alternatives 2 and 3 and Alternative 1 No Action, but less than under Alternative 4. This is due mostly to the number of residents within and adjacent to the MICs resulting from anticipated development. Where housing within the industrial zones is established, those residents would experience higher emissions resulting from industrial and other non-transportation air emissions. The areas where the housing growth is to occur may also be adjacent to areas of high-capacity highways, major commute arterials, and a busy rail corridor.

In Georgetown, where the triangular area bounded by Corson Avenue S, Carleton Avenue S, and I-5 and a smaller node north of the triangle would be removed from the MIC and placed into a mixed-use zone and in the areas to be designated as Urban Industrial, existing or new residents would experience higher emissions resulting from nearby industrial, transportation, and other non-transportation air emissions, including the WSDOT Corson facility on Corson Avenue S.

However, as with all other alternatives, the combination of existing requirements for industrial operating permits from PSCAA, and ongoing requirements for improvements in vehicle emissions control, fuel economy, technology improvements, and overall fuel mix, local emissions under the Preferred Alternative would be lower than under existing background conditions. Similar cumulative air emissions from rail would occur in the MICs under all alternatives. Nonetheless, the Preferred Alternative would likely result in a less than significant impact to air quality. Similar mitigation strategies as have been mentioned for the other alternatives should be considered.

Maritime Activities

Maritime activities and their impact on the Puget Sound air shed, including the MICs, would continue similarly as they would under Alternative 1 No Action. With existing and planned regulatory requirements and local infrastructure improvements, these maritime emissions are expected to decrease under all alternatives, even if cargo volumes and cruise ship visits increase.

Greenhouse Gases & Climate Change

Compared to Alternative 1 No Action, GHG emissions under the Preferred Alternative reflect the increases in industry-supportive housing within the MICs between alternatives 3 and 4; increased office and non-industrial uses within the MICs between alternatives 2 and 3; added mixed-uses within and adjacent to the MICs larger than all alternatives; and increases in industrial building space between Alternative 1 and Alternative 2. VMT increases for the Preferred Alternative are anticipated slightly higher than Alternative 2 for both the Greater Duwamish MIC and the BINMIC. Total estimated GHG emissions from the Preferred Alternative are presented in **Exhibit 3.2-37**.

<u>Source</u>	No Action MTCO2e	Pref. Alt. MTCO2e
Transportation	<u>613,158</u>	<u>616,896</u>
Ind. And Non-Ind. Building—Gas	<u>58,864</u>	<u>62,777</u>
Housing	<u>1,364</u>	<u>9,564</u>
<u>Waste</u>	<u>-2,690</u>	<u>-3,086</u>
Total	<u>670,696</u>	<u>686,151</u>
Difference from Existing	<u>-81,441</u>	<u>-65,986</u>
Difference from No Action	<u>0</u>	<u>15,455</u>
Source: Herrera, 2022.		

Exhibit 3.2-37	Total Estimated Annual MTCO2e Emissions Under the Preferred Alternative
	Compared to Alternative 1 No Action

The Preferred Alternative could decrease GHG emissions by approximately 66,000 MTCO2e per year compared to existing conditions but would represent an increase of more than 15,000 MTCO2e compared to Alternative 1 No Action, which is above the 10,000 MTCO2e mandatory reporting threshold for the State of Washington. Compared to the other alternatives, the Preferred Alternative results in increases in all source categories except waste.

Like the other Action Alternatives, reducing transportation related emissions through increasing density of employment growth in the MICs and in mixed use zones within the MICs and new mixed use zones rather than in other Seattle neighborhoods or regionally would be consistent for the Preferred Alternative, despite the moderate increase in transportationrelated emissions in the MIC areas. Overall, the Preferred Alternative could result in an increase in GHG emissions compared to Alternative 1—No Action that could be considered potentially significant and additional mitigation measures would be warranted.

3.2.3 Mitigation Measures

It is notable that it is anticipated that the amount of development and activity projected under the alternatives, if confined within the <u>MICsstudy area</u>, would result in less GHG emissions than if that same development and activity were spread out to other parts of the city or region. While Alternative 1 No Action would result in lower GHG emissions within the MICs, it is likely that the population and employment growth anticipated to occur under the alternatives would occur elsewhere and those GHG emissions are not quantified but are expected to be greater than if focused in the MICs as proposed by the industrial and maritime strategy alternatives. The alternatives under the Industrial and Maritime Strategy serve to structure residences, employment, and activities in relatively efficient ways so that the GHG emission associated with their activities are less than what they would be if those people and jobs were more dispersed, and their activities conducted less efficiently.

Nonetheless, GHG emissions from future projects need to be mitigated so that future projects do not result in a significant environmental impact. A list of potential mitigation measures is given below; some measures would need to be integrated into Subarea Plan policies or codes as requirements and incentives to apply to future development.

Incorporated Plan Features

Air Quality

All Action Alternatives would change land use designations and development regulations applicable to the study area to target enhancement of industrial and maritime uses, and to allow a wider latitude of commercial/industrial development and industry supportive housing, while protecting adjacent residential areas. Increasing density in some areas of the MICs around light rail stations and with access to multiple mobility options could lead to more use of public transportation, biking, and walking, and less use of cars. These policies and actions recognize the value of planning for the type and density of future industries and employment as a way to reduce the need for future residents and workers to travel by automobile, thereby reducing transportation-related emissions in the region.

Greenhouse Gases & Climate Change

All alternatives—in particular alternatives 3 and 4—contribute to increased GHG emissions through future growth and development in the study area. All Action Alternatives result in GHG emissions above the 10,000 MTCO2e mandatory reporting threshold compared to Alternative 1 No Action.

All Action Alternatives would change land use designations and development regulations applicable to the study area to target enhancement of industrial and maritime uses, and to allow a wider latitude of commercial/industrial development and industry supportive housing, while protecting adjacent residential areas. These policies and actions recognize the value of planning for the type and density of future industries and employment as a way to optimize the coordination of complementary industries, and to reduce the transportation demand of businesses activities. The policies also allow increasing density in some areas of the MICs around light rail stations and with access to multiple mobility options, which could lead to more use of public transportation, biking, and walking, and less use of cars; and to reduce the need for future residents and workers to travel by automobile, thereby reducing transportationrelated emissions in the region.

The Industrial and Maritime Strategy includes policy concepts particularly relevant to Air Quality/GHG:

Introduce new or strengthened policies into chapters of the Comprehensive Plan that may include the Transportation, Environment, or Container Port elements encouraging transitions to clean fuels and decarbonization of industrial and maritime activities.

Regulations & Commitments

Air Quality

Several federal, state, and regional regulations <u>or efforts</u> apply to construction and allowed land uses (see also **Section 3.2.1 Affected Environment**):

- NAAQS: As described above, the EPA established NAAQS and specifies future dates for states to develop and implement plans to achieve these standards.
- Washington State: Ecology established state ambient air quality standards for the same six pollutants (CO, VOCs, NO2, PM, SO2, and ozone) that are at least as stringent as the national standards.
- PSCAA Regulations: All construction sites in the Puget Sound region are required to implement emission controls to minimize fugitive dust and odors during construction, as required by PSCAA Regulation 1, Section 9.15, Fugitive Dust Control Measures.

PSCAA manages permitting of stationary air pollutant sources and all industrial and commercial air pollutant sources in the Puget Sound region are required to register with the PSCAA.

Greenhouse Gases & Climate Change

- Washington State Energy Code: Development in the study area would be subject to the requirements of the Washington State Energy Code, which regulates the energy-use features of new and remodeled buildings.
- The City's 2013 CAP and the 2018 Climate Action Strategy includes strategies and actions to limit atmospheric warming to 1.5 degrees Celsius. The strategies and actions focus on road

transportation and building energy, which comprise the majority of local emissions, and which drive the GHG emissions in the study area.

- All buildings with 50,000 square feet or more of nonresidential space (excluding parking) must comply with the Building Tune-Ups requirement every five years (SMC 22.930). Building Tune-Ups involve assessment and implementation of operational and maintenance improvements to achieve energy (and water) efficiency, which helps to reduce GHG emissions.
- The Port of Seattle, independent of the Industrial and Maritime Strategy, is increasing shore power available at terminals to reduce maritime emissions (Starcrest 2018). Upcoming projects within the SODO/Stadium Subarea include planned shore power improvements in Terminal 155, Terminal 18, and possibly the electrification of Terminal 30 and the Coast Guard Station.

Other Potential Mitigation Measures

Air Quality

Mitigation strategies are not required due to a lack of significant adverse impacts, however potential for exposure of existing and new employees, residents, and visitors to potential air emissions in areas around arterials, along industrial buffers, and near port operations should be considered in future planning:

The Seattle Comprehensive Plan and MIC Subarea Plans could:

- Include policy guidance that recommends that residences and other sensitive land uses (i.e., schools, day care) be separated from freeways, railways, and port facilities, and new MML, II, and UI zones by a buffer area of no less than 500 feet, and possibly as much as 1,000 feet, depending on the height of the source, to reduce the potential exposure of sensitive populations to air toxics. (US Department of Transportation 2015)
- Include policy guidance that recommends and <u>funds</u> support<u>fors</u> the electrification of industrial and maritime activities that currently rely on fossil fuels, including the transportation related assets that are an integral part of those land uses.
- Incorporate new development standards that include requirements that recommend that residences and other sensitive land uses (i.e., schools, day care) include enhanced air filtering and circulation to address pollutant transportation generated particulates. Specifically, U.S. EPA identifies that mechanical ventilation/filtration systems with a Minimum Efficiency Reporting Value (MERV) of 9 through 12 are adequate for removing 25 to 80% of automobile emission particles (U.S. EPA 2009a).
- Consider locations for schools, daycares, and residential uses that increases buffers from high-volume roadways or other measures to reduce exposure to criteria pollutant emissions.
- Assure design standards for parks in proximity to high-volume roadways and industrial areas incorporate landscaping with full bottom to top of canopy coverage, higher canopy

heights, and multiple rows of vegetation types, including denser tree canopies, that help reduce exposure to criteria pollutant emissions.

- Add a denser tree canopy near high-volume roadways and industrial areas.
- Incorporate standards for more frequent street sweeping to reduce roadway dust and prevent emissions of PM2.5 in fugitive dust associated with increased vehicle miles traveled.
- Consider inclusion of a City-owned and operated air monitoring station in Ballard-Interbay and the Duwamish Valley to provide the public with access to daily air monitoring data.
- Where the City has authority to do so, consider designating truck routes serving industrial and manufacturing areas away from residential areas, particularly those residential areas with vulnerable populations.

Greenhouse Gases & Climate Change

- Subarea Plan Policies: As part of Subarea Plan development, the City could establish policies that:
 - <u>-il</u>ncentivize use of electrical infrastructure to serve industrial process needs, industrial, commercial, and residential space heating needs, rather than natural gas.
 - Strengthen climate resiliency requirements and City support for business engagement and continuity planning for developments throughout the MICs.
 - Expand City-sponsored development and training pathways for workers in resilient industries who locate in the MICs.
 - Incentivize industries focused on clean technologies or processes to locate within the MICs.
- Green Building Standards: To lower the GHG contribution from industrial and commercial uses, policies that encourage or mandate new construction projects in the study area to:
 - Achieve one of the following green building standards: LEED In Motion: Industrial Facilities, Built Green, the Living Building Challenge, or the Evergreen Sustainable Development Criteria.
 - Use low-embodied carbon construction material types, such as low-carbon concrete mixes.
 - Limit carbon-intensive materials or incentivize use of lower carbon alternatives such as a wood structure instead of steel and concrete, or agricultural products that sequester carbon.
 - Salvage materials like brick, metals, broken concrete, or wood.
 - Use high-recycled content materials.
 - Prioritize adaptive reuse for existing buildings to avoid additional embodied carbon emissions.
 - Include embodied carbon goals in building codes (AIA, 2021).
- Building Demolition Waste Reduction: The City could consider programs to require or encourage building deconstruction rather than bulk demolition for older industrial buildings demolished in the study area.

Ch.3 Environment, Impacts, & Mitigation Measures • Air Quality & GHG

- Puget Sound Energy (PSE): Seattle is served by PSE for natural gas service. PSE has established a target to reach net zero carbon emissions for natural gas used in customer homes and businesses by 2045, with an interim target of a 30% emissions reduction by 2030. The City could promote or incentivize PSE and/or study area employers to integrate greater volumes of renewable natural gas into their systems or processes. Coordination with King County Wastewater Treatment Division and with SPU' Solid Waste Division could enhance efforts.
- Electric Vehicles: The City could adopt regulations for the study area that support the placement of infrastructure for charging of electric vehicles (including commercial and industrial vehicles) in applicable new developments. Seattle Public Utilities is exploring the creation of a city-owned electrical vehicle charging facility in the Duwamish MIC intended for drayage trucks. The City and Port of Seattle could expand on the effort to establish multiple such facilities in strategic locations in proximity to Port terminals that require drayage.
- Trees: The City could adopt regulations/incentives for the study area that preserve and/or replace on-site trees and encourage planting of more trees. Trees and shrubs can provide shade and lower temperatures in urban areas and can assist with GHG reductions.
- Expand electrification of marine terminals: The City, Port of Seattle and private partners could accelerate the extension of shore power to terminals and docks throughout the Seattle waterfront, including at Coleman Dock and Terminals 5, 18, 30, 46, and 66, and where appropriate for US Coast Guard vessels, and other research vessel berths. Consider commitment of public funding for the infrastructure investment. Consider regulations requiring vessels to connect to shore power if it is present.
- Where the City of Seattle has authority, consider imposing restrictions on maritime air emissions for ocean-going vessels while underway.
- Consider commitment of public funding for the necessary infrastructure to expand availability of shore power, and electrify cargo and passenger handling equipment to include those areas and ships not covered by the Port of Seattle's existing plans.
- The City and partner agencies could improve coordination and improve the user experience for community members registering complaints or requesting information about enforcement related to emissions from sites or businesses.

3.2.4 Significant Unavoidable Adverse Impacts

Because of the combination of existing requirements for industrial operating permits from PSCAA, and ongoing requirements for improvements in vehicle emissions control, fuel economy, and technology improvements, and overall changes in fleet and fuel mix toward electrification and cleaner fuels, respectively, no significant unavoidable adverse impacts to air quality are anticipated.

Potentially significant impacts to GHG emissions could be expected for all alternatives as they could have the potential for increased GHG emissions above the 10,000 MTCO2e mandatory

reporting threshold. However, through mitigation implementation, local and state climate actions, and expected continued regulatory changes, the alternatives may result in a decrease of the growth in GHG emissions such that the impacts from future development allowed by the changes in plans and zoning could be considered less than significant for SEPA. As proposed, the alternatives would not prevent or deter efforts to reduce emissions in comparison to local or regional goals or targets for GHG reductions.

While each alternative would create a net increase in GHG emissions generated from growth and development in the study area, the region-wide benefit of capturing development that might otherwise occur in peripheral areas of the city or region could serve to offset these impacts.

Section 3.3 Water Resources



This section discusses water resources in the study area, including:

- Longfellow, Puget, and Wolfe Creeks
- Elliott Bay
- Duwamish River
- Ship Canal / Salmon Bay
- Groundwater

Impacts described in the following sections are broad evaluations based upon the details available at the time of analysis; each future planned action will be subject to City of Seattle code, regulations, and ordinances and will need to demonstrate consistency with applicable critical area requirements.

Thresholds of significance utilized in this impact analysis include:

- Development that results in discharges to surface waters that do not meet water quality or flow control standards.
- Development that eliminates groundwater recharge or results in groundwater that does not meet water quality standards.
- Development that increases vulnerability to sea level rise.

3.3.1 Affected Environment

Study Areas

The study area consists of the primary and secondary study areas. The primary study area encompasses all industrial land in the city and includes the Ballard Interbay Northend Manufacturing Industrial Center (BINMIC) and the Greater Duwamish Manufacturing and Industrial Center (Greater Duwamish MIC). The primary study area is divided into five subareas:

- Ballard
- Interbay Dravus
- Interbay Smith Cove
- SODO/Stadium
- Georgetown/South Park

The primary study area also includes other industrial zones lands within the city.

The secondary study area is defined as the area 500 feet from the primary study area because development of the Seattle Industrial and Maritime Lands could affect adjacent water resources.

Exhibit 3.3-1 lists surface water bodies located in each of the subareas, and **Exhibit 3.3-2**, **Exhibit 3.3-3** shows surface water bodies and watersheds of natural streams relative to the Secondary Study Area.

Exhibit 3.3-1	Surface Water	Bodies	Located i	in e	each 🗄	Subarea
	Sallace Hatel	Doalco	Located		CACILI	Sabarca

Subarea	Surface Water Bodies
Ballard	Ship Canal/Salmon Bay
Interbay Dravus	Ship Canal/Salmon Bay, Wolfe Creek
Interbay Smith Cove	North Elliott Bay, Puget Sound
SODO/Stadium	Duwamish River, Longfellow Creek, Puget Creek
Georgetown/South Park	Duwamish River



Exhibit 3.3-2 Surface Water Bodies in the Primary Study Area

Sources: Herrera, 2021.



Exhibit 3.3-3 Location of Surface Water Bodies and Watersheds of Natural Streams

Sources: Herrera, 2021.

Data & Methods

Current water quality was determined based upon the Ecology list of Category 5 impaired waters, and existing focus studies of surface, groundwater, and climate change performed in the study area and more broadly in the region.

The project team collected data from the following sources to support analysis of surface and groundwater conditions:

- Ship Canal Water Quality Project Final Facility Plan Prepared for Seattle Public Utilities (CH2M March 2017)
- Seattle Creeks State of the Waters Report (City of Seattle 2007)
- Ecology Water Quality Assessment Database (Ecology 2014)
- Duwamish Basin Groundwater Pathways Conceptual Model Report (Hart Crowser, Inc. 1998)
- Draft EIS Magnolia Bridge Replacement (KPFF Consulting Engineers Shannon and Wilson, Inc. 2005)
- Assessment of Existing Groundwater Quality Data in the Green-Duwamish Watershed, Washington Report 2019-1131 (USGS 2019)
- Wolfe Creek Drainage Feasibility Study Final Report (WR Consulting 2008)
- Projected Sea Level Rise for Washington State (Miller 2018)
- Preparing for Climate Change (City of Seattle 2017)

Current Policy & Regulatory Frameworks

Surface water quality is based upon the Washington State Department of Ecology beneficial uses for each water body in the plan area. Use designations differ for marine and fresh waters. Designated uses for marine waters in WAC 173-201A and for freshwaters in WAC 173201A-600. Aquatic life use is rated higher in Puget Sound (extraordinary) than Elliott Bay (excellent). These marine water bodies are both designated for shellfish harvesting and primary contact recreation (such as swimming), although shellfish harvesting is prohibited for all marine beaches in Seattle due to potential contamination by fecal bacteria and other pollutants.

Aquatic life and recreational uses for the freshwater bodies are highest (core summer habitat and extraordinary primary contact) for the Ship Canal/Lake Union, and the Duwamish River (rearing/migration and secondary contact). All freshwater bodies are designated for water supply uses with the exception that the Duwamish River is not designated for domestic water supply.

Water quality standards developed by Ecology under the Washington Administrative Code 173-201A set limits that are intended to protect aquatic life and recreational uses. The standards depend on the specific use designation for each water body, and they vary for fresh waters (streams, rivers, and lakes) and marine waters (Ecology, 2012a). Numeric standards are established for conventional parameters (common pollutants such as high temperature, low dissolved oxygen, pH, and turbidity), some toxic substances (mostly metals and some organic chemicals), and fecal bacteria. Under Section 303(d) of the Clean Water Act, Ecology is required to prepare a water quality assessment and develop a list of surface waters (marine and fresh water) that are impaired. This list is periodically prepared by Ecology and submitted to EPA for review and approval. The current active list was published in 2014 (Ecology, 2014).

The Section 303(d) list identifies five categories of water quality impairments:

- Category 1—meets tested standards for clean waters
- Category 2—waters of concern
- Category 3—insufficient data
- Category 4—polluted waters that do not require the establishment of a total maximum daily load (TMDL) for targeted pollutant(s) to allow the achievement of the surface water quality standards
- Category 5—polluted waters that require a TMDL program to establish maximum allowable pollutant discharges.

Groundwater quality is regulated by the Washington State Department of Ecology (Ecology) under the Water Quality Standards for Groundwaters of the State of Washington (Washington Administrative Code 173-200). These standards list the maximum concentrations of contaminants that are allowed in groundwater and prohibit further groundwater contamination.

Shoreline development is regulated at the local level by the Shoreline Master Program (SMP; Seattle Municipal Code 23.60A), which mandates that all shoreline modifications be constructed and managed to achieve no net loss of ecological functions. Shoreline setbacks in the SMP are based on the Ordinary High Water Mark (OHWM) as defined by WAC 173-22-030. The Washington Department of Ecology also provides regulatory oversight of shoreline development through the State Environmental Policy Act (WAC 197-11) and Habitat Project Approval process (WAC 220-660), both of which also use the OHWM as a jurisdictional boundary.

Section 10 of the Rivers and Harbors Act, administered by the US Army Corps of Engineers (USACE), provides for permitting of any work in, over, or under navigable waters of the United Sates, or which affects the course, location, conditions, or capacity of such waters. Regulated activities include docks and piers, marinas, intake and outfall pipes, transmission lines, and dredging. The USACE Seattle District recently redefined its jurisdictional boundary to be the High Tide Line, defined as the "maximum height reached by a rising tide," which encompasses spring high tides, but not storm surge.

The City of Seattle adopted the 2013 Climate Action Plan (CAP), 2018 Climate Action Strategy and 2017 Preparing for Climate Change includes City actions that will increase resilience to the likely impacts of climate change. Acknowledging that preparing for climate change impacts is a complex challenge, the CAP includes proactive planning for major infrastructure to include future projected conditions to prevent costly repairs or retrofits. The CAP also provides for community equity in planning for climate impacts, with priority given to actions that help vulnerable populations moderate potential impacts. Sea level rise projections that apply to the BINMIC and Greater Duwamish MIC are described below. **Section 3.14 <u>Utilities</u>** summarizes stormwater related policies and regulations that pertain to new development and redevelopment within the BINMIC and Greater Duwamish MIC, including City policies related to accounting for climate change in utility planning.

Current Conditions

Full Study Area

Surface Water

Water bodies located solely in the BINMIC study area include the Ship Canal (and Salmon Bay) and Wolfe Creek. Water bodies located solely in the Greater Duwamish MIC study area include the Duwamish River, Wolfe Creek, Longfellow Creek, and Puget Creek.

Puget Sound is a fjord-like estuary that stretches from Hood Canal to north of Admiralty Inlet. Industrial and maritime activities and their influences are not restricted to the nearshore environment adjacent to the study area but have regional impacts due to water circulation in <u>Puget Sound</u>. Freshwater flows influence water circulation in this portion of Puget Sound. Two main freshwater bodies flow into Puget Sound in the study area, the Green/Duwamish River, which enters Elliott Bay, and the Cedar River (Lake Washington drainage basin), which flows into the Sound through Lake Washington and the Ship Canal.

Elliott Bay is a partially enclosed embayment that is bordered on the north, east, and south sides by urbanized areas of Seattle and by Puget Sound on the west. The northern shoreline borders the Interbay Smith Cove Subarea, and the southern shoreline borders the SODO/Stadium Subarea. Both the southern and northern portions of Elliott Bay are heavily altered by industrial facilities.

The Lake Washington Ship Canal system is an 8.6-mile-long navigable waterway, completed in 1934, connecting Shilshole Bay in Puget Sound to Union Bay in Lake Washington. The system is bordered by the Ballard Subarea to the north and Interbay Dravus Subarea to the south. The Ship Canal includes several interconnected waterways—Hiram M. Chittenden Locks (Ballard Locks), Salmon Bay, Salmon Bay Waterway, Fremont Cut, Lake Union, Portage Bay, and Montlake Cut. The Ship Canal borders the Ship Canal Neighborhoods on the west end and the Lake Washington Neighborhoods on the east end. Lake Union is a freshwater lake that receives most of its inflow from Lake Washington via the Montlake Cut and Portage Bay.

Wolf Creek is a small stream located in the Dravus Bay subarea, with a watershed of approximately 90 acres, located in the Magnolia Neighborhood, which flows into Salmon Bay. It is highly modified with approximately 3,100 feet of open channel.

The Duwamish River originates at the confluence of the Green and Black Rivers near Tukwila and flows northwest for approximately 12 miles, splitting at the southern end of Harbor Island to form the East and West Waterways before discharging into Elliott Bay. The Duwamish River extends through both the SODO/Stadium and Georgetown/South Park subareas. The downstream portion of the Duwamish River serves as a major shipping route for bulk and containerized cargo. A portion of the lower Duwamish River is maintained as a federal navigation channel by the Corps of Engineers.

Longfellow Creek is approximately 3.5 miles in length and is a tributary of the Duwamish River discharging to the Duwamish River in the SODO/Stadium Subarea. It is one of the four largest streams in the City of Seattle with a watershed size of 2,685 acres. The creek originates in the Roxhill Park neighborhood, flows north along the valley of the Delridge Neighborhood of West Seattle, and then flows into the Duwamish Waterway.

Puget Creek is located in the SODO/Stadium Subarea on the eastern side of West Seattle and drains to the Duwamish River.

<u>Groundwater</u>

Because of the presence of a municipal water system in the Seattle area and the sources not located in the study area, groundwater use is generally limited to emergency and industrial supply wells for non-drinking use. No drinking water wells, wellhead protection areas, critical aquifer recharge areas, or sole source aquifers are identified in the study area. Numerous observation and test groundwater quality monitoring wells are present in the study areas due to historical industrial contamination and monitoring of clean-up projects.

<u>Sea Level Rise</u>

Seal levels in Elliot Bay have been monitored by the National Oceanic and Atmospheric Administration since 1899 (gauge #9447130). Observations are representative of the unrestricted tidal regions in the study area, but not the waterways within the Ship Canal system, which are controlled by the system of locks. Sea levels at the gauge have historically risen at a rate of 0.68 feet in 100 years. By comparison, recent work by the UW Climate Impacts Group (Miller et al. 2018) provide central to high-end estimates of future sea level rise of 2.3-5.1 feet by 2100. Sea level rise projections apply to all tidally influenced water bodies including Puget Sound, Elliott Bay, and the Duwamish River and may also affect water levels near the outlets of creeks in the primary study area. Ship Canal and Lake Union are above the Hiram M. Chittenden Locks so they are not affected by sea level rise. Sea level rise may also affect groundwater levels in the study area, which has the potential to cause flooding and affect underground infrastructure, including the wastewater, combined sewer, and stormwater infrastructure described in **Section 3.14 Utilities**.

Subareas

Ballard & Interbay Dravus

King County has characterized water in the Ship Canal and Lake Union as "fair" for most parameters important to fish and wildlife (temperature, dissolved oxygen, pH, and nutrients) and to humans (fecal coliform bacteria). The water in these areas is flushed rapidly with good quality outflow from Lake Washington. Salmon Bay is on the 303(d) list for total phosphorus, fecal coliform bacteria, lead, and aldrin.

There is no summary water quality data for Wolfe Creek and no 303(d) category 5 listings.

Groundwater elevation in the Ship Canal area is generally a shallow confined aquifer and ranges from 10 to 30 feet below the ground surface. Groundwater discharge from the shallow unconfined aquifer is primarily into the Ship Canal. Shallow groundwater wells have shown contamination for petroleum hydrocarbons (oil and gasoline), heavy metals (such as arsenic, chromium, lead, and mercury), dry cleaning and degreasing solvents (such as trichloroethylene and tetrachloroethylene) and asbestos.

Interbay Smith Cove

In general, the overall water quality in Puget Sound and Elliott Bay is good based on water quality parameters such as bacteria, nutrients, temperature, chlorophyll, dissolved oxygen, solids, and transparency. However, fecal coliform bacteria have exceeded allowable levels in some areas of these marine waters, most notably Elliott Bay, that are included on the 303(d) list of impaired waters.

The predominant groundwater flow system area consists of a shallow unconfined aquifer system. A lesser predominant system includes a deep artesian aquifer located approximately 300-400 feet below sea level (KPFF Consulting Engineers Shannon and Wilson, Inc. 2005). Groundwater elevations in the north Elliott Bay area is generally a shallow unconfined aquifer that ranges from 5 to 15 feet below the ground surface. Groundwater discharge from the shallow unconfined aquifer is primarily into Elliott Bay to the south. Contaminants detected in shallow groundwater include petroleum hydrocarbons, volatile organic compounds (VOC's), polycyclic aromatic hydrocarbons (PAH's), polychlorinated biphenyls (PCBs), and dissolved metals (KPFF Consulting Engineers Shannon and Wilson, Inc. 2005).

SODO/Stadium

The Duwamish River is included on Ecology's 303(d) category 5 list as impaired waters for fecal coliform bacteria, temperature, pH, and dissolved oxygen.

Longfellow Creek is included on Ecology's 303(d) category 5 list as impaired waters for fecal coliform bacteria, temperature, pH, and dissolved oxygen.

There is no summary surface water quality data for Puget Creek and no 303(d) category 5 listings.

The groundwater flow system is common between the Georgetown/South Park and SODO/Stadium subareas. Groundwater is generally a regional discharge due to its low elevation and surface water outlet at Elliott Bay. Groundwater is typically 5 to 15 feet below the ground surface. Tidal influence is present within 300 to 500 feet of the river where groundwater may fluctuate several feet and may rise in elevation as a result of sea level rise. Groundwater flow is generally to the Duwamish River (Hart Crowser, 1998). A recent summary of shallow

groundwater wells by USGS showed contamination for all classes of chemicals selected for research. Contaminants researched and confirmed were petroleum hydrocarbons (oil and gasoline), heavy metals (arsenic, zinc, and copper), polychlorinated biphenyls (PCBs), and phthalates (USGS 2019).

The shoreline of the SODO/Stadium Subarea surrounding the Lower Duwamish River and the mouths of Longfellow Creek and Puget Creek are vulnerable to sea level rise.

Georgetown/South Park

The Duwamish River and groundwater system extends through both the Georgetown/South Park and SODO/Stadium subareas so that the current conditions described above for that subarea apply in the Georgetown/South Park Subarea. Significant portions of both the Georgetown and South Park neighborhoods are susceptible to sea level rise. Areas in Georgetown are primarily vulnerable to rising groundwater levels, including areas northeast and southwest of Marginal Way, while South Park is primarily vulnerable to water overtopping the banks of the Duwamish River.

3.3.2 Impacts

Impacts Common to All Alternatives

Surface Water & Groundwater

Rainfall runoff from a portion of the Study Area discharges to natural streams including, Longfellow Creek, Puget Creek, and Wolf Creek, which are sensitive to increased flow rates or water quality impacts that could result from increases in impervious surfaces. Other water bodies including the Duwamish River, Puget Sound, and Ship Canal / Salmon Bay are only sensitive to changes in water quality that could be caused by increases in impervious surfaces or changes in land use. However, nearly all the Study Area that is feasible to develop has already been covered with a high percentage of impervious surface. Therefore, redevelopment expected under all Alternatives is not expected to significantly increase total impervious area or result in significant increases in flow rates or water quality impacts.

The Seattle Stormwater Code (SMC Title 22, Subtitle VIII) requires redevelopment projects in the Study Area to implement on-site stormwater management to infiltrate, disperse, and retain stormwater runoff to the maximum extent feasible. Where the developed site's stormwater flow rates or pollutant generation potential is expected to exceed the allowable thresholds, flow control and/or water quality treatment are required. As a result of these requirements and given that much of the existing development predates modern stormwater requirements, it is expected that there would be a reduction in uncontrolled flow rates and an increase in water quality in the Primary Study Area under all of the alternatives where new construction is anticipated.
The 2021 Stormwater Code also supports incentives for retrofitting existing development, such as opportunities for property owners to reduce their drainage rate if they install flow control and/or treatment facilities designed per the Code, which can include reducing impervious surfaces.

Under all alternatives, including Alternative 1 No Action, implementation of on-site stormwater management and continuation of retrofit incentives would continue to reduce adverse impacts to all surface water bodies in the Study Area, even if future rainfall patterns are more intense than historic rainfall patterns. Areas that do not redevelop, including areas that are currently industrial in nature, may not experience the same stormwater improvements as sites that redevelop; however, these sites would still be required to implement stormwater source control measures, even if no redevelopment occurs.

Under all alternatives, development and redevelopment projects have the potential to generate stormwater pollution during construction. The Seattle Stormwater Code requires all projects to implement Temporary Erosion and Sediment Control (TESC) stormwater management best management practices (BMPs) during construction that will minimize these impacts.

Under all alternatives, traffic would increase within the Study Area, which has the potential to introduce metals and other pollutants to ground surfaces, which could contribute to surface water or groundwater pollution. In portions of the Study Area where stormwater discharges to the combined sewer system, these pollutants would be treated by the West Point Wastewater Treatment Plant, so no water resource impacts are expected from these areas. In areas where stormwater discharges to surface water bodies, improvements in vehicle standards and the application of stormwater requirements described above as parcels and roadways are redeveloped and upgraded is expected to offset the increase in traffic and potentially lead to a net decrease in surface water pollution.

-Activities involving vessel traffic in the Puget Sound, including the cruise industry, are prohibited from discharge of sewage, treated or untreated, in compliance with the "No Discharge Zone" effective May 10, 2018, WAC 173-228. Similarly, the release of oils and gases, trash, and toxins associated with vessel maintenance are prohibited by The Marine Pollution Act (MARPOL), the Oil Pollution Control Act (OPA), and the Washington State Pollution Control Act. Therefore, increases in vessel traffic are not expected to result in significant impacts to water quality under any of the alternatives.

The Seattle Stormwater Code requires redevelopment projects in the Study Area to consider infiltration as a means of managing stormwater, which could improve groundwater recharge under all alternatives. The Code also requires review of the existing site conditions for potential soil or groundwater contamination, which would make infiltration infeasible in cases where the infiltration could mobilize existing pollutants in the soil (see **Section 3.5 Contamination**). In places where infiltration is feasible, the 2021 Stormwater Code requires infiltration facilities to protect groundwater quality.

With growth there is the potential for increased risk of spills from industrial activities, industrial processes, or use of industrial chemicals or other organics (see **Section 3.5 Contamination**).

The Seattle Stormwater Code and Washington State Industrial Stormwater General Permit require implementation of source control measures for developments that store liquids that could be spilled and impact groundwater. The use of source control BMPs would limit that risk, and any spills would be cleaned up quickly consistent with applicable state and local requirements and no significant impacts to surface or groundwater are anticipated. None of the alternatives are expected to reduce groundwater recharge, increase the potential for groundwater contamination, or increase mobilization of groundwater pollutants relative to existing conditions.

Sea Level Rise

Under all alternatives, low-lying areas adjacent to tidally influenced water bodies (Puget Sound, Elliott Bay, the Duwamish River, and the mouths of Longfellow Creek and Puget Creek) have the potential to be affected by sea level rise. These areas include portions of the Interbay Smith Cove, SODO/Stadium, and Georgetown/South Park subareas. Sea level rise vulnerability mapping is available from the <u>City of Seattle</u> and through the project <u>StoryMap</u>. Both maps portray results of the 2018 Washington Coastal Resilience Project report ("Projected Sea Level Rise for Washington State"). King County has infrastructure in these areas including wastewater pump stations, wastewater regulator stations, and wastewater odor control facilities that could be affected by sea level rise. The Interbay Dravus and Ballard subareas are above the Hiram M. Chittenden Locks so they are not affected by sea level rise.

Under all alternatives, proposed development in areas that are susceptible to impacts from extreme high tides would be required to comply with critical areas regulations for frequently flooded areas, which is regulated through the City's Environmentally Critical Areas (ECA) Code; the requirements of the Shoreline Master Program (SMP; Seattle Municipal Code 23.60A) also apply to development along the shoreline. Compliance with these codescurrent and future regulations of development within the study area may reduce vulnerability of those developments to sea level rise impacts relative to existing conditions, particularly in locations that are currently not compliant with current regulations.

Subarea Impacts

As described above, all alternatives are expected to result in a net improvement in water resources as newer development with modern stormwater management facilities replaces older development that lacks onsite stormwater management or flow control and water quality facilities. In general, alternatives that would result in more redevelopment would result in more improvements to water resources. Based on the square footage of new employment space and housing units added under each alternative, improvements to water resources are expected to be highest under Alternative 4 and lowest under the No Action Alternatives (see **Exhibit 3.3-4**).

	Alternative 1	Alternative 2	Alternative 3	Alternative 4	<u>Preferred</u> <u>Alternative</u>
Square Footage of New Employment Space	11,230,000	19,805,000	27,400,000	27,760,000	<u>16,245,000</u>
Housing Units Added	75	80	1,688	3,273	<u>3,009</u>
Relative Rank of Improvements to Water Resources	<u>5th</u> 4th	4 th 3rd	2 nd	1 st	<u>3rd</u>

Exhibit 3.3-4 Comparison of Relative Water Resource Improvements Between Alternatives

Sources: City of Seattle, 202<u>2</u>4; BERK, 202<u>2</u>4; Herrera, 202<u>2</u>4.

Total jobs in each subarea shows that the SODO/Stadium and Georgetown/South Park subareas have the most jobs currently and would still have the most jobs in the future. See **Exhibit 3.3-5**. The Ballard Subarea would increase its share of jobs particularly in alternatives 3 and 4 compared to other alternatives. See **Exhibit 3.3-6**. To a smaller degree the Interbay Dravus and Interbay Smith Cove subareas would also increase their share of jobs under the Action Alternatives compared to current or Alternative 1 No Action conditions.

Exhibit 3.3-5 Total Jobs by Subarea Current and Future



Note: This chart was updated to include the Preferred Alternative.

Sources: City of Seattle, 20221; BERK, 20221.



Exhibit 3.3-6 Share of Job Growth by Subarea Compared to Existing

<u>Ballard</u>

Salmon Bay is listed as an impaired water body for total phosphorus and fecal coliform bacteria. Redevelopment sites may be required to provide phosphorus treatment if discharging to Salmon Bay. Water quality treatment at redevelopment sites will reduce fecal bacteria impacts at sites that redevelop. Ballard is not expected to be vulnerable to sea level rise because it is above the Ballard Locks.

Interbay Dravus

Salmon Bay is listed as an impaired water body for total phosphorus and fecal coliform bacteria. Redevelopment sites may be required to provide phosphorus treatment if discharging to Salmon Bay. Water quality treatment at redevelopment sites will reduce fecal bacteria impacts at sites that redevelop. Interbay Dravus is not expected to be vulnerable to sea level rise because it is above the Ballard Locks.

Interbay Smith Cove

Elliott Bay is listed as an impaired water body for fecal coliform bacteria. Water quality treatment at redevelopment sites will improve fecal bacteria impacts at sites that redevelop. Minor portions of Interbay Smith Cove at Pier 90 and Elliott Avenue are vulnerable to sea level rise.

Note: This chart was updated to include the Preferred Alternative. Sources: City of Seattle, 20224; BERK, 20224.

SODO/Stadium

The Duwamish River and Longfellow Creek are each listed as an impaired water bodies for fecal coliform bacteria, temperature, pH, and dissolved oxygen. Water quality treatment at redevelopment sites will reduce fecal bacteria and other pollutant impacts. The shoreline of the SODO/Stadium Subarea surrounding the Lower Duwamish River and the mouths of Longfellow Creek and Puget Creek are vulnerable to sea level rise and all alternatives, including the No Action Alternative, would increase the concentration of people in these vulnerable areas. Compliance with requirements of the SMP and frequently flooded areas requirements at redevelopment sites, in addition to adaptation measures listed in the mitigation section, may help reduce vulnerability to sea level rise in some portions of the subarea.

Georgetown/South Park

The Duwamish River and Longfellow Creek are each listed as an impaired water body for fecal coliform bacteria, temperature, pH, and dissolved oxygen. Water quality treatment at redevelopment sites will reduce fecal bacteria and other pollutant impacts at sites that redevelop.

Significant portions of both Georgetown and South Park neighborhoods are susceptible to sea level rise and all alternatives, including the No Action Alternative, would increase the concentration of people in these vulnerable areas. Compliance with requirements of the SMP and frequently flooded areas requirements at redevelopment sites, in addition to adaptation measures listed in the mitigation section, may help reduce vulnerability to sea level rise in some portions of the subarea.

Other Industrial Zoned Lands

Growth would result in mitigation of stormwater at redevelopment sites. Lake Union is listed as an impaired water body for fecal coliform bacteria and temperature. Elliott Bay is listed as an impaired water body for fecal coliform bacteria. Water quality treatment at redevelopment sites will reduce fecal bacteria and other pollutant impacts at sites that redevelop.

Equity & Environmental Justice Considerations

Increases in impervious surface can negatively affect surface water quality, which can disproportionately affect populations with a higher reliance on water resources for sustenance, such as subsistence fishers or Tribes. Poor water quality also poses health risks for populations that come in physical contact with surface water bodies. As described above, all alternatives are expected to result in a net improvement in water quality and therefore reduce negative impacts on these populations as they relate to water resources.

The Seattle Mapping Inventory of Changing Coastal Flood Risk provides a screening level picture of the impacts of sea level rise on Seattle. The analysis reveals that the communities most impacted by flooding are also disproportionately characterized by high levels of social vulnerability, most notably in the Georgetown/South Park Subarea.

Impacts of Alternative 1 No Action

Surface Water & Groundwater

Impacts resulting from Alternative 1 No Action would be the same as described in the discussion of Impacts Common to All Alternatives. Compared to the Action Alternatives, there is likely to be less redevelopment in the Primary Study Area and the least improvements in surface water and groundwater that would result from installation of onsite stormwater management, flow control, and water quality treatment at redevelopment sites.

Sea Level Rise

Impacts resulting from Alternative 1 No Action would be the same as described in the discussion of Impacts Common to All Alternatives.

Impacts of Alternative 2

Surface Water & Groundwater

Alternative 2 includes greater change and densification of industrial zones than Alternative 1 which would result in increased implementation of on-site stormwater management and improvements to water resources on sites that redevelop. Alternative 2 would apply a mix of II and UI zone concepts in approximately 10% of current MIC areas. These concepts would increase the number of trees and landscaping, and green spaces, which would provide opportunities for stormwater treatment and water resource improvements. Water quality and flow control improvements would be less than alternatives 3 and 4.

Sea Level Rise

Alternative 2 includes more growth in the SODO/Stadium and Georgetown/South Park subareas than Alternative 1. These areas are substantially susceptible to sea level rise so Alternative 2 may increase vulnerability to sea level rise more than Alternative 1 by bringing more people into vulnerable areas. Through compliance with SMP and frequently flooded areas requirements, some of the development could reduce sea level rise vulnerability in areas near the shoreline more than Alternative 1, but less than alternatives 3 and 4.

Impacts of Alternative 3

Surface Water & Groundwater

Alternative 3 increases job growth and housing units in industrial and non-industrial areas more than alternatives 1 and 2 but less than Alternative 4. Implementation of on-site stormwater management at redevelopment sites would continue to reduce adverse impacts to

all surface water bodies in the Study Area. Alternative 3 would apply a mix of II and UI zone concepts in approximately 14% of current MIC areas, the most of any alternative, which would increase the number of trees and landscaping, and green spaces, which would provide opportunities for stormwater treatment and water resource improvements. Alternative 3 has greater residential growth than alternatives 1 or 2 but less than Alternative 4. With increased residential units, pet waste and fecal coliform pollution may be increased.

Relative water resource improvement under Alternative 3 would be greater than alternatives 1 and 2 but less than Alternative 4.

Sea Level Rise

Alternative 3 includes more growth in the SODO/Stadium and Georgetown/South Park subareas than alternatives 1 and 2. These areas are substantially susceptible to sea level rise so Alternative 3 may increase vulnerability to sea level rise more than alternatives 1 and 2 by bringing more people into vulnerable areas. Through compliance with SMP and frequently flooded areas requirements, and incorporation of adaptation measures, some of the development could reduce sea level rise vulnerability in areas near the shoreline more than alternative <u>s</u> 1 and 2, but less than Alternative <u>4</u>.

Impacts of Alternative 4

Surface Water & Groundwater

Alternative 4 has the greatest increase of job growth and housing units in industrial and nonindustrial areas. Because this alternative has the highest potential for redevelopment, it would also likely have the highest increase in on-site stormwater management flow control or water quality treatment, which could result in the greatest improvements in surface water and groundwater. Alternative 4 would apply a mix of II and UI zone concepts in approximately 13% of current MIC areas, only slightly less than Alternative 3, and would result in the creation of green spaces and landscaped areas that provide similar opportunities for stormwater retrofits and water resource improvements.

Alternative 4 has the greatest increase in residential units and therefore the highest potential for pet waste and fecal coliform pollution.

Sea Level Rise

Alternative 4 targets the highest growth in the SODO/Stadium and Georgetown/South Park subareas. These areas are substantially susceptible to sea level rise so Alternative 4 may increase vulnerability to sea level rise more than other alternatives bringing the most people into vulnerable areas. Through compliance with SMP and frequently flooded areas requirements, and incorporation of adaptation measures, some of the development could reduce sea level rise vulnerability in areas near the shoreline more than the other alternatives.

Impacts of the Preferred Alternative

Surface Water & Groundwater

The Preferred Alternative increases job growth similar to Alternative 2 for total number of jobs. The share of job growth is increased in the Ballard and Interbay Dravus subareas and greater than alternatives 1 through 4. Job growth is reduced in the Georgetown/South Park Subarea under the Preferred Alternative. See **Exhibit 3.3-5** and **Exhibit 3.3-6**. Job growth will increase traffic at a higher rate in the Ballard and Interbay Dravus subareas.

Total residential growth under the Preferred Alternative—including within industrial areas, areas removed from the MIC, and rezoned areas converted to mixed use zoning outside of the MIC—is similar to but lower than Alternative 4. See **Exhibit 3.3-4**. The number of dwellings in industrial areas is projected to increase by 1,475 units in the UI zone, 33% less than the amount studied in Draft EIS Alternative 4, and is concentrated in the Ballard and SODO/Stadium subareas. Outside of the MICs, two new areas in west Ballard and Judkins Park would be converted to mixed use zoning allowing housing, in addition to the proposed mixed-use areas in Georgetown and South Park studied in Draft EIS alternatives 3 and 4; overall, a higher total amount of housing production outside of MICs would result compared to Draft EIS alternatives—an additional 1,534 dwellings, 42% more than alternatives 3 and 4. The Preferred Alternative would result in increased implementation of on-site stormwater management and improvements to water resources on sites that are redeveloped. These concepts would increase the number of trees and landscaping, and green spaces, which would provide opportunities for stormwater treatment and water resource improvements. Water quality and flow control improvements would be more than alternatives 1 and 2 and less than alternatives 3 and 4.

Sea Level Rise

The Preferred Alternative includes more growth in the SODO/Stadium and Georgetown/South Park subareas than Alternative 1. These areas are substantially susceptible to sea level rise so the Preferred Alternative may increase vulnerability to sea level rise more than Alternative 1 by bringing more people into vulnerable areas. Through compliance with SMP and frequently flooded areas requirements, some of the development may reduce sea level rise vulnerability in areas near the shoreline more than Alternative 1.

3.3.3 Mitigation Measures

Incorporated Plan Features

There are no incorporated plan features.

Regulations & Commitments

Regulatory requirements for addressing water resource impacts would be met under each Alternative, as discussed above in **Section 3.3.1 Affected Environment**, below, and in the Utilities Section. If thresholds listed in the City's stormwater management standards are exceeded as redevelopment occurs, projects would be required to provide BMPs to the maximum extent feasible to infiltrate, disperse, or retain stormwater runoff. Projects would also be required to provide water quality treatment to reduce pollution levels in stormwater, and flow control to reduce flow rates as thresholds are exceeded. Compliance with these regulations is anticipated to result in a net benefit to water resources under all alternatives.

A majority of development and redevelopment projects would be parcel-based and require source control BMPs to the extent necessary to prevent prohibited discharges and to prevent contaminants from coming in contact with drainage water or being discharged to the drainage system, public combined sewer, or directly into receiving waters (City of Seattle Stormwater Manual, Volume 1, Chapter 2).

An individual project's plan for the type of surface that is new and replaced determines the requirement for water quality treatment. In general, pollution-generating hard surfaces (vehicular traffic, industrial activities, storage of wastes or chemicals) require a higher level of treatment over pollution-generating pervious surfaces (lawns, landscaping areas, parks).

Development and redevelopment projects would be required to conduct a downstream analysis of the runoff leaving the project site. This analysis is based upon the receiving water or point of discharge and is subject to review and approval or disapproval by the SPU Director. Due to the complexity of the City drainage system (creeks, ditches, combined sewer with capacity, combined sewer without capacity, small lakes, and designated receiving water) each project will be unique for the analysis and result.

Surface and groundwater quality at industrial and business sites are protected through ongoing inspection programs, which also applies to new development. Industrial permits issued and managed by the Washington State Department of Ecology and held by individual properties are inspected and held to source control BMPs. In some cases, depending on the industrial activity, properties are held to chemical discharge limits. Seattle Public Utilities conducts site inspections of all industrial and business properties with the potential to pollute surface and groundwaters through its NPDES Stormwater Phase 1 permit-requirements and local code (SMC 22.803.040).

Several regulations prohibit pollution-causing activities of marine vessels in Puget sound. Puget Sound is designated as a "No Discharge Zone" effective May 10, 2018, WAC 173-228. All vessels are prohibited from discharge of sewage, treated or untreated. The Marine Pollution Act (MARPOL), enforced by the U.S. Coast Guard, requires a written Waste Management Plan for vessels over 26 feet long and states that no trash may be thrown overboard within Puget Sound. The Oil Pollution Control Act (OPA) prohibit the discharge of oil. The Washington State Pollution Control Act prohibits underwater cleaning of boat hulls with anti-fouling paint and the discharge of soaps, paints, and dust from sanding.

Other Potential Mitigation Measures

Alternatives 3 and 4<u>and the Preferred Alternative</u>, result in the greatest increase in housing in portions of the Ballard and SODO/Stadium subareas, which could create a larger concentration of pets and associated animal waste and a potential to impact local surface water quality. An increased emphasis on pet waste management through education and outreach and increased pet waste disposal stations should be implemented in areas surrounding these housing developments to prevent negative impacts on water quality.

All alternatives, including the No Action Alternative, would increase the concentration of people in SODO/Stadium and Georgetown/South Park subareas, which have large geographic areas that are vulnerable to sea level rise impacts. The City of Seattle Office of Sustainability and Environment (2017) has identified the following adaptation strategies that should be prioritized by the City and partner agencies as a means of reducing vulnerability to sea level rise in the Study Area:

- Explore further opportunities to incentivize or require existing building upgrades to improve preparedness for future climate conditions, including consideration of regulations that require design of buildings, structures, and industrial and manufacturing sites to consider the sea level rise projected to occur during the life of the facility.
- Develop mechanisms to incorporate climate preparedness and passive survivability into the planning and development processes for new development.
- Consider the disproportionate impacts of climate change on communities of color and lower income communities in planning, policies, and programs, and prioritize programs and incentives that mitigate those impacts.
- To reduce flood risk and reduce flood insurance rates, evaluate the benefits and costs of participating in the National Flood Insurance Community Rating System program.
- Evaluate the requirements of the Floodplain Development Ordinance to identify additional opportunities to reduce food hazards, including the base flood elevation threshold, the definition of a substantial improvement, and the regulation of footbridges and other potential obstructions to stream flow.
- Regularly update flood prone area maps to incorporate the latest data near creeks, shorelines, and other emerging urban flooding areas.
- Conduct a detailed coastal study of the Duwamish River to better delineate the current and increasing risk of flooding and identify a range of mitigation strategies to pursue.
- Assess the benefits of incorporating rolling easements into the next update of the Shoreline Master Plan.
- Continue to incorporate Green Stormwater Infrastructure (GSI) into development regulations.
- The City should also evaluate vulnerability of underground infrastructure to higher groundwater levels.

3.3.4 Significant Unavoidable Adverse Impacts

Under all proposed alternatives, any redevelopment or new development will require compliance with all applicable regulations to avoid, minimize, or mitigate any impacts to water resources. Development will need to meet stormwater requirements to protect surface and groundwater from increased flow or water quality impacts. Therefore, no significant unavoidable adverse impacts are anticipated on water resources under any of the proposed alternatives.

Section 3.4 Plants & Animals



The study area is highly urbanized, but still provides habitat for numerous plant and animal species. Many of these are nonnative introduced species, and most of them are well-adapted to the urban environment and high levels of human disturbance.

Thresholds of significance used for this impact analysis include:

- The potential to reduce or damage rare, uncommon, unique, or exceptional benthic, marine, wetland, riparian, or fish and wildlife habitat.
- The potential to harass, harm, wound or kill any species listed as federally threatened or endangered.
- The potential to adversely affect critical habitat for any federally threatened or endangered species.
- The potential to block migration corridors for special status species.
- Terrestrial noise levels generated exceed any established injury thresholds for any specialstatus species.

3.4.1 Affected Environment

Study Area

The study area consists of primary and secondary study areas. The primary study area encompasses all industrial land in the City and includes the Ballard Interbay North Manufacturing Industrial Center (BINMIC; **Exhibit 3.4-1**) and the Greater Duwamish Manufacturing and Industrial Center (Greater Duwamish MIC; **Exhibit 3.4-2**). The primary study area is divided into five subareas as follows:

- Ballard
- Interbay Dravus
- Interbay Smith Cove
- SODO/Stadium
- Georgetown/South Park

The primary study area also includes other industrial zones lands within the city.

The secondary study area is defined as the area 500 feet from the primary study area, including any waterward areas because development of the Seattle Industrial and Maritime Lands could affect species in the nearshore (**Exhibit 3.4-1** and **Exhibit 3.4-2**). Water quality affecting plants and animals is discussed below as well as in **Section 3.3 Water Resources**.



Exhibit 3.4-1 BINMIC Study Area and Critical Areas, 2021



Exhibit 3.4-2 Greater Duwamish MIC Study Area and Critical Areas, 2021

Data & Methods

To characterize plants and animals for each alternative, the project team reviewed GIS data for the primary and secondary study areas identified for each alternative. Data sources included aerial imagery, national wetlands inventory, the City's GIS data for environmentally critical areas (wetlands, streams, wildlife habitats and riparian corridors) and the Washington Department of Fish and Wildlife's Priority Habitats and Species (PHS) information, as well as existing reports.

This review is a general summary for the purposes of identifying plants and animals that could be affected by implementation of the program. As with most construction projects conducted in the city, projects proposed under the program would require site-specific analysis to determine the presence of sensitive or protected plants, habitats, fish, or wildlife.

Current Policy & Regulatory Frameworks

Several federal, state, and local regulations and permits relate to the protection of plants and animals within the study areas (**Exhibit 3.4-3**). Projects that involve federal funding, land, or permits from a federal agency trigger the need to comply with federal regulations.

Statute	Lead Agency	Regulated Activity
Federal		
Endangered Species Act	National Marine Fisheries Service (NMFS) and U.S. Fish and Wildlife Service (USFWS)	Protects species identified as endangered or threatened along with critical habitat required for the conservation of those species. NMFS has authority over anadromous fishes, marine mammals, marine reptiles, and other fish species, while the USFWS has authority over terrestrial wildlife and resident fish species that inhabit inland waters. Requires that federal actions do not jeopardize the continued existence of any threatened, endangered, or proposed species or result in the destruction or adverse modification of critical habitat. To comply with the Act, project proponents are required to consult with the federal agencies regarding the effect of their projects on listed species.
Magnuson-Stevens Fishery Conservation Act	NMFS	Requires federal agencies to consult with NMFS on activities that may adversely affect Essential Fish Habitat for federally managed fish species within a 200-mile zone offshore of the United States.
Marine Mammal Protection Act	NMFS	Prohibits injury or harm to marine mammals in U.S. waters. NMFS has authority over whales, dolphins, porpoises, seals, and sea lions, while the USFWS has authority over otters. The USDA is responsible for managing marine mammals in captivity.
Migratory Bird Treaty Act	USFWS	Protects many of the most common birds in the study area as well as birds that are listed as threatened or endangered. USFWS has authority to regulate most aspects of the taking, possession, transportation, sale, purchase, barter, exportation, and importation of migratory birds. As of

Exhibit 3.4-3 Federal, State, and Local Regulations and Permits Related to the Protection of Plants and Animals

Statute	Lead Agency	Regulated Activity
		March 2010, there are 1,007 species protected under the Act (Federal Register Vol. 75, No. 39). Species whose occurrences in the United States are strictly the result of intentional human introduction are not protected under the Act. Of particular concern are activities that affect birds nesting on bridges, buildings, signs, illumination poles, and other structures in areas planned for construction.
Bald and Golden Eagle Protection Act	USFWS	Specifically protects bald and golden eagles and makes it unlawful to take, import, export, sell, purchase, or barter any bald or golden eagles, their parts, products, nests, or eggs. "Take" includes pursuing, shooting, poisoning, wounding, killing, capturing, trapping, collecting, molesting, or disturbing eagles. To avoid potential disturbance to bald eagles, the National Bald Eagle Management Guidelines (USFWS, 2007) provide recommendations that will likely avoid take for a list of activities.
Fish and Wildlife Conservation Act	USFWS	This Act authorizes financial and technical assistance for states to develop, revise, and implement conservation plans and programs for nongame fish and wildlife.
Section 404 of the Clean Water Act	U.S. Army Corps of Engineers	Regulates the placement of dredged or fill material into waters of the United States, including special aquatic sites such as wetlands.
State of Washington		
State Hydraulic Code (Chapter 220-110 WAC)	Washington Department of Fish and Wildlife (WDFW)	Protects fish and their habitat through regulation of activities in streams and lakes. WDFW administers state rules through its Hydraulic Project Approval (HPA) program. An HPA must be obtained from WDFW before work is conducted that uses, obstructs, diverts, or changes the natural flow or bed of state waters. The conditions of an HPA can be designed to protect fish, shellfish, and their habitat.
Priority Habitats and Species Program	WDFW	Provides information on documented locations of fish and aquatic resources, terrestrial plants and animals, and habitats that are listed or defined as priority. Priority species are those species that are: state endangered, threatened, sensitive, or candidate species; animal aggregations considered vulnerable; and species of recreational, commercial, or tribal importance that are vulnerable (WDFW, 2008). Priority habitats are habitat types or elements of habitat with unique or significant value to a diverse assemblage of species. A priority habitat may consist of a unique vegetation type (e.g., shrub-steppe) or dominant plant species, a described successional stage (e.g., old-growth forest), or a specific habitat feature (e.g., cliffs).
Natural Heritage Program	Washington Department of Natural Resources (WDNR)	Provides information for listed plant species or those that are defined as rare. Also maintains information on rare ecological communities and priority species.
Clean Water Act Section 401	Washington State Department of Ecology (Ecology)	Requires certification for any projects that may result in a discharge into waters of the United States to ensure that the discharge complies with applicable state water quality requirements.

Ch.3 Environment, Impacts, & Mitigation Measures • Plants & Animals

Statute	Lead Agency	Regulated Activity
Washington State Water Pollution Control Act (RCW 90.48)	Ecology	Regulates placement of dredge or fill material within non-federally regulated wetlands or waters of the State
Furbearer Regulations	WDFW	Furbearers may not be taken from the wild and held alive for sale or personal use without a permit pursuant to WAC 232 12 064.
Water Quality Standards for Surface Waters of the State of Washington	Ecology	Aquatic life uses are designated based on the presence or protection of species. Ecology provides general water quality standards based on aquatic life use categories.
Washington Regulations for Fish and Wildlife	WDFW	Washington State has its own criteria for listing species as endangered, threatened, sensitive, and candidate. Washington has developed rules to provide for additional protection of some species and their habitat. The state has defined suitable habitat, dispersal habitat, habitat buffers, critical habitat, and critical nesting season and nesting areas.
City of Seattle		
Environmentally Critical Areas Ordinance (Seattle Municipal Code [SMC] 25.09)	City of Seattle Department of Planning and Development (DPD)	 Protects and regulates activities on or adjacent to critical areas in the City. Critical areas include geologic hazard areas, flood-prone areas, wetlands, fish and wildlife habitat conservation areas (FWHCAs), and abandoned landfills. FWHCAs are wildlife habitats that are mapped or designated by WDFW, corridors connecting priority habitats, or areas that support species of local importance. FWHCAs and wetlands are typically protected by a buffer in which development, including clearing and other land disturbing activities, is prohibited or restricted. Riparian corridors, a type of FWHCA, include all areas within 100 feet of the ordinary high water mark of a watercourse. Parcels containing riparian corridors and shoreline habitat are also subject to the general development standards in SMC 25.09.060 and specific development regulations in SMC 25.09.200, as well as regulations regarding tree and vegetation alteration and pesticide use.
Shoreline Master Program (SMC 23.60)	DPD	Regulates water bodies above a threshold size as well as lands within 200 feet of the ordinary high water mark of those water bodies. Regulations include restrictions on development in the shoreline zone, requirements for maintaining native vegetation, and development standards.
Tree Protection Ordinance (SMC 25.11) and specific environmental policies related to trees (SMC 25.05.675)	DPD	Trees in Seattle are specifically valued and legally protected under various regulations in addition to the environmentally critical areas code. "Exceptional trees" are specifically protected and defined as a tree or group of trees that constitutes an important community resource because of its unique historical, ecological, or aesthetic value. Prior to construction at any site, a survey for exceptional trees would need to be conducted by a licensed arborist as required under SMC 25.11.
SEPA Plants and Animals Policy (SMC 25.05.675.N)	DPD	City policy to minimize or prevent loss of wildlife habitat. Allows <u>SDCI DPD</u> to grant, condition or deny construction and use permit applications for public or private proposal that are subject to environmental review.

Source: Herrera, 2021.

Full Study Area

Current conditions for plants and animals are defined as the conditions that exist within the study area in 2021 when the desktop analysis was conducted. Mapping for critical areas within the study areas are shown in **Exhibit 3.4-1** and **Exhibit 3.4-2**.

Plants

The heavily urbanized habitats in the study areas include streets, parking lots, commercial and industrial properties, high-density residential buildings, and railroad rights of way. Over the last 150 years, urban development has eliminated nearly all the native vegetation. Small pockets of native vegetation remain within protected park areas, protected shorelines, and undeveloped steep slopes. Additional vegetation exists as street trees and related streetscape vegetation in the right of way, and yards associated with private homes. Streetscape vegetation has been installed and is maintained by the City's Urban Forestry section or by private development projects under permit from SDOT.

Non-native invasive species, such as English ivy (*Hedera helix*) and Himalayan blackberry (*Rubus armeniacus*), are common in unmaintained portions of the study areas. These invasive species are well adapted to urban environments and out-compete native plant species. Non-vascular plants, such as mosses and lichens, grow on a variety of hard surfaces such as concrete, treated wood, and occasionally metal in the study areas.

The study areas for the BINMIC can be broken down into the following landscapes:

- approximately 20% vegetation
- approximately 62% hardscape
- approximately 18% water

The study areas for the Greater Duwamish MIC can be broken down into the following landscapes:

- approximately 22% vegetation
- approximately 68% hardscape
- approximately 10% water

Shorelines and nearshore areas within the study area include streams and riparian corridors, lakes, estuaries, and marine waters, as described below. Upland habitat consists of forests, natural areas, and landscaped areas.

Riparian Corridors

Riparian corridors are vegetated corridors present along streams. Within the study areas, riparian corridors are typically vegetated with deciduous trees and shrubs with a few conifer trees. Native plants common to riparian corridors in the study areas include red alder (*Alnus rubra*), big-leaf maple (*Acer macrophyllum*), Indian plum (*Oemleria cerasiformis*), vine maple (*Acer*

circinatum), willow (*Salix* spp.) and horsetail (*Equisetum* spp.). Common aquatic plants include rushes (family Juncaceae), sedges (family Cyperaceae), common cattail (*Typha latifolia*), duckweed (*Lemna* spp.), water lily, and pondweed. Nonnative invasive aquatic plants such as Eurasian watermilfoil (*Myriophyllum spicatum*) are present in some areas.

Some riparian corridors in the City are wide and densely vegetated, but most are narrow and constrained by urban development. Riparian areas provide important wildlife habitat including forage, cover, and complex habitat structure. This habitat supports a wide variety of terrestrial species such as songbirds, woodpeckers, and raptors. Riparian corridors also benefit aquatic habitats by providing shade, large wood, and organic material to streams. Streams in the study area are fed by surface runoff, groundwater, and drainage pipes that convey stormwater from impervious surfaces (Seattle 2010).

Riparian corridors are identified by the City in both the BINMIC and the Greater Duwamish MIC. Corridors within the BINMIC are connected streams that discharge to the Lake Washington Ship Canal and those in the Greater Duwamish MIC are connected to streams that discharge into the Duwamish Waterway.

Freshwater Wetlands

Freshwater wetlands in Seattle are associated with lake edges, streams and their riparian corridors, and scattered low-lying areas. Emergent, scrub-shrub, and forested wetlands are present. Plant species common to emergent wetlands include reed canarygrass (*Phalaris arundinacea*; nonnative), common cattail, and soft rush (*Juncus effusus*). Scrub-shrub and forested wetlands support many of the same plant species as riparian corridors, but also include red-osier dogwood (*Cornus sericea*), willow, and other water-tolerant species.

Freshwater wetlands are identified in both the BINMIC and the Greater Duwamish MIC study areas.

<u>Lakes</u>

The BINMIC study areas contain portions of Lake Union and the Ship Canal. These are open freshwater environments that have aquatic vegetation associated with them such as pondweeds (*Potamogeton* spp.) and hornwort (*Ceratophyllum demersum*). Eurasian watermilfoil and Brazilian elodea (*Egeria densa*) are invasive aquatic plants also well established in this area. The Ship Canal connects the Puget Sound to Lake Union and provides a corridor for aquatic species to travel between these two environments. Lake Union and the Ship Canal are on the Washington Department of Ecology (Ecology) 303(d) list for bacteria, temperature, and pesticides (Ecology 2021).

<u>Estuaries</u>

Estuaries are semi-enclosed bodies of water where freshwater and marine water mix (Hobbie 2000). These ecosystems are shaped by tidal fluctuations and freshwater flows and are among

the most highly productive and complex ecosystems in the state where quantities of sediments, nutrients and organic matter are exchanged among terrestrial, freshwater, and marine communities. In Puget Sound, salinity fluctuates with seasons and tides, making it difficult to differentiate between marine habitat and estuarine habitat. Marine nearshore areas within the study area can all generally be characterized as estuarine habitat and include Elliott bay and the Duwamish Waterway (Encyclopedia of Puget Sound 2020).

Shorelines in Elliott Bay and the Duwamish Waterway have been extensively modified by the placement of seawalls, bulkheads, and levees (Seattle 2015). Both the bay and the waterway are on the Washington Department of Ecology 303(d) list for water quality and sediment due to elevated contaminant concentrations (Ecology 2021). Estuarine wetlands in Seattle are associated with Puget Sound marine nearshore areas where enough light penetrates the water to support persistent aquatic vegetation. Estuarine wetlands are identified around Port of Seattle Terminal 91 and Smith Cove within the BINMIC study areas and in restored areas of the lower Duwamish Waterway within the Greater Duwamish MIC study areas. The Washington Department of Natural Resources identifies the presence of eelgrass (*Zostera marina*) in or around Smith Cover and the Duwamish Waterway (DNR 2021). Eelgrass provides important habitat for numerous Puget Sound species.

Forests

Forested communities are present in scattered patches throughout the city. Forests can be dominated by conifers (such as Douglas fir [*Pseudotsuga menziesii*]) or deciduous trees (such as big-leaf maple) or support a mixture of conifer and deciduous species. City of Seattle has mapped tree canopy coverage throughout the City. Forested areas are typically associated with steep slopes, top of bluffs, greenbelts, parks, and other pockets of undeveloped land. Tree canopy mapped by the City of Seattle also includes street trees. Plant species common to forested habitats in Seattle include Douglas fir, western red cedar (*Thuja plicata*), vine maple, and sword fern (*Polystichum munitum*). Forested habitats are important for woodpeckers, raptors, songbirds, crows, and jays. These forested areas are generally identified by City of Seattle critical area mapping as riparian corridors or wildlife habitat areas.

The patches of forest occur primarily within restored areas along the Duwamish Waterway, along the western edge of the Interbay neighborhood, and along W. Commodore Way leading to Commodore Park and Kiwanis Memorial Preserve Park.

Natural Areas

Natural areas support intact or natural vegetation (both native and nonnative) that is not formally landscaped. Parks and other public lands in the City support natural areas. Natural areas can contain mapped and unmapped riparian corridors and wetlands as well as forested habitats, but they can also contain grass or shrub areas that are not maintained or mowed.

Landscaped Areas

Landscaped areas provide some habitat for wildlife despite their level of development and human presence. Landscaped gardens, golf courses, and recreational parks provide food and water sources, shelter, and other habitat elements important for terrestrial wildlife. Species that use landscaped areas are usually those that can tolerate some level of ongoing human disturbance.

Animals

The study area contains a variety of fish and wildlife habitats and species. Terrestrial animals in the study areas are generally limited to those well adapted to living in a highly altered urban landscape. Examples include birds and mammals that tolerate or benefit from human disturbance, urban habitat features, and trash, such as various gulls (Family Laridae), crows (*Corvus brachyrhynchos*), coyotes (*Canis latrans*) racoons (*Procyon lotor*), opossums (*Didelphis virginiana*). Both marine and freshwater environments are present in the study areas, resulting in substantial diversity for aquatic species.

Special status species are identified in **Exhibit 3.4-4** with PHS mapping shown in **Exhibit 3.4-5** and **Exhibit 3.4-6**. Several of these species are listed as endangered or threatened under the Endangered Species Act. Lake Union, the Ship Canal, and nearshore areas of Elliott Bay are designated critical habitat for bull trout (*Salvelinus confluentus*) and Chinook salmon (*Oncorhynchus tshawytscha*), and the Duwamish Waterway provides critical habitat for bull trout, Chinook, and steelhead (*O. mykiss*). Elliott Bay is also designated critical habitat for yelloweye rockfish (*Sebastes paucispinis*) and bocaccio (*Sebastes paucispinis*). Deeper waters (great than 20 feet deep) of Elliott Bay are designated critical habitat for the Southern Resident killer whale (*Orcinus orca*) (NMFS 2021), but the species itself is extremely unlikely to occur in the study area.

The Ship Canal, Lake Union, and Elliott Bay are Essential Fish Habitat (EFH) for groundfish, Chinook, and coho salmon (*O. kisutch*). Elliott Bay and the Duwamish Waterway are EFH for Chinook, coho, pink salmon (*O. gorbuscha*), and coastal pelagic species.

Bald eagles (*Haliaeetus leucocephalus*), which are protected under the Bald and Golden Eagle Protection Act, forage in Lake Union, the Ship Canal, Elliott Bay, and the Duwamish River. Almost all other bird species are protected under the Migratory Bird Treaty Act. Although PHS data list historical occurrences of western pond turtle (*Actinemys marmorata*) in the study area, this species is extremely rare and highly unlikely to occur in the study area.

Exhibit 3.4-4 Special Status Species and Habitats Occurring in the Study Areas

Common Name	Scientific Name	Federal Status/Protection	State Status	Use of Study Area	Occurrence in Study Area
Dungeness crab	Cancer magister	N/A	N/A	Presence	BINMIC
Pacific Herring	Clupea pallasi	N/A	Candidate	Breeding Area	BINMIC
Dolly Varden/Bull Trout	Salvelinus malma/S. confluentus	Threatened	Candidate	Foraging/Migration	BINMIC Greater Duwamish MIC
Bull trout critical habitat	N/A	N/A	N/A	N/A	BINMIC Greater Duwamish MIC
Chinook	Oncorhynchus tshawytscha	Threatened	Candidate	Foraging/Migration	BINMIC Greater Duwamish MIC
Chinook critical habitat	N/A	Designated	N/A	N/A	BINMIC Greater Duwamish MIC
Chum	Oncorhynchus keta	Not Warranted	N/A	Foraging/Migration	Greater Duwamish MIC
Resident Coastal Cutthroat	Oncorhynchus clarki	N/A	N/A	Foraging/Migration	BINMIC Greater Duwamish MIC
Coho	Oncorhynchus kisutch	Candidate	N/A	Foraging/Migration	BINMIC Greater Duwamish MIC
Pink Salmon	Oncorhynchus gorbuscha	N/A	N/A	Foraging/Migration	Greater Duwamish MIC
Steelhead	Oncorhynchus mykiss	Threatened	Candidate	Foraging/Migration	BINMIC Greater Duwamish MIC
Steelhead critical habitat	N/A	Designated	N/A	N/A	BINMIC Greater Duwamish MIC
Sockeye	Oncorhynchus nerka	Not Warranted	Candidate	Foraging /Migration	BINMIC Greater Duwamish MIC
Pacific Sand Lance	Ammodytes hexapterus	N/A	N/A	Breeding Area	BINMIC
Yelloweye rockfish	Sebastes ruberrimus	Threatened	N/A	Presence	BINMIC Greater Duwamish MIC

Common Name	Scientific Name	Federal Status/Protection	State Status	Use of Study Area	Occurrence in Study Area
Yelloweye rockfish critical habitat	N/A	Designated	N/A	N/A	BINMIC Greater Duwamish MIC
Bocaccio	Sebastes paucispinis	Endangered	N/A	Presence	BINMIC Greater Duwamish MIC
Bocaccio critical habitat	N/A	Designated	N/A	N/A	BINMIC Greater Duwamish MIC
Purple martin	Progne subis	MBTA ¹	N/A	Foraging/Nesting	BINMIC Greater Duwamish MIC
Great blue heron	Ardea herodias	MBTA	N/A	Foraging/Nesting	BINMIC Greater Duwamish MIC
Bald eagle	Haliaeetus leucocephalus	BGEPA ²	N/A	Foraging	BINMIC Greater Duwamish MIC
Other bird species	N/A	MBTA	N/A	Foraging, nesting	BINMIC Greater Duwamish MIC
Southern resident killer whale critical habitat	N/A	Designated	N/A	N/A	BINMIC Greater Duwamish MIC

¹ MBTA = Migratory Bird Treaty Act ² BGEPA = Bald and Golden Eagle Protection Act Source: Herrera, 2021.





Source: Herrera, 2021.



Exhibit 3.4-6 Greater Duwamish MIC Study Areas PHS Mapping, 2021

Ballard

Critical areas identified within the Ballard Subarea are mapped in **Exhibit 3.4-7**. For further descriptions of plants in these areas please see the **Plants** section above. Areas that provide animal habitat are discussed in the **Animals** section above.



Exhibit 3.4-7 Critical Areas—Ballard Subarea, 2021

Source: Herrera, 2021.

Interbay Dravus

Critical areas identified within the Interbay Dravus Subarea are mapped in **Exhibit 3.4-8**. For further descriptions of plants in these areas please see the **Plants** section above. Areas that support animals are discussed in the **Animals** section above.





Interbay Smith Cove

Critical areas identified within the Interbay Smith Cove Subarea are mapped in **Exhibit 3.4-9**. For further descriptions of plants in these areas please see the **Plants** section above. The presence of animals and animal habitats in this subarea is discussed in the **Animals** section above.



Exhibit 3.4-9 Critical Areas—Interbay Smith Cove Subarea, 2021

SODO/Stadium

Critical areas identified within the SODO/Stadium Subarea are mapped in **Exhibit 3.4-10**. For further descriptions of plants in these areas please see the **Plants** section above. Areas that provide animal habitat are discussed in the **Animals** section above.





Source: Herrera, 2021.

Georgetown/South Park

Critical areas identified within the Georgetown/South Park Subarea are mapped in **Exhibit 3.4-11**. For further descriptions of plants in these areas please see the **Plants** section above. The presence of animals and animal habitats in this subarea are discussed in the **Animals** section above.



Exhibit 3.4-11 Critical Areas—Georgetown/South Park Subarea, 2021

3.4.2 Impacts

Impacts Common to All Alternatives

Noise & Disturbance

All alternatives involve construction activities that would generate noise and disturbance that could temporarily displace bird species listed in **Section 3.4.1 Affected Environment** from preferred nesting, foraging, and/or migration sites.

The amount and intensity of construction is expected to be greater under the Action Alternatives, particularly alternatives 3 and 4, which allow for the greatest industry-associated caretakers' quarters and makers' space, as well as remove focused land in the Georgetown subarea that could be developed for housing. In particular there would be an increase housing in the UI zone in the Ballard and SODO/Stadium subareas. The Georgetown/ South Park subarea would be a focus for 20+ acre rezone to Seattle Mixed where alternatives 3 and 4 would allow for greater attached housing.

All studied alternatives add employment space over current conditions with Alternative 1 No Action the least and alternatives 3 and 4 the most. Given the high levels of existing human activity and noise levels in these industrial zones, construction activities would not be likely to increase noise and disturbance to an extent that would adversely affect birds in the study area. These species are already adapted to high levels of human activity and any disturbance would be minor. These species would likely return to normal activity levels shortly following construction.

None of the alternatives affect shoreline land use regulations or propose changes to regulations governing in-water work; accordingly, the studied alternatives would not result in direct noise or disturbance impacts to aquatic habitats or species.

Construction Stormwater Runoff

Stormwater runoff from active construction sites has the potential to adversely affect water quality in receiving water bodies, primarily by increasing sediment and turbidity. Best management practices (BMPs) implemented during construction per City of Seattle regulations would be protective of water quality. Refer to **Section 3.3 Water Resources**, for a more detailed discussion of temporary impacts related to construction.

Lack of Redevelopment

Some existing properties cause detrimental environmental effects due to the presence of contamination, overwater cover, or lack of stormwater treatment, all of which can reduce water quality and negatively affect fish, benthic invertebrates, and other aquatic life. If these properties are not redeveloped, with associated mitigation of environmental impacts, these consequences would persist into the future and continue to degrade aquatic habitat.

Equity & Environmental Justice Considerations

The Action Alternatives would result in greater tree canopy cover in landscaped areas and green spaces that promote environmental health, provide safe, non-motorized transit options, encourage walkability and access to the outdoors, and improve comfort. This is through street frontage/street tree and green factor requirements in the II and UI zones. Alternatives 3 and 4 <u>and the Preferred Alternative</u> have the greatest share of land in II and UI zones (14% <u>and _</u>13%, <u>and 14%</u> respectively), where trees, landscaping, and green spaces would be concentrated. Under Alternative 2, about 10% of land within industrial areas would be zoned as II or UI. The No Action Alternative does not include II or UI zoning and does not have a plan for conversion of currently developed areas to landscaped areas or green spaces. The adaptation of impervious areas to increased tree canopy and green factor can increase shade and modestly improve habitat such as for birds and urban-adapted wildlife as well as for humans.

Focusing such street and landscaping improvements in SODO/Stadium and Georgetown/South Park areas would assist disadvantaged populations as identified in Seattle's Racial and Social Equity Index.

The Action Alternatives also have the potential to improve water quality in the study area. Older development that lacks modern stormwater infrastructure and treatment would be replaced with newer infrastructure that provides water quality treatment, thereby reducing pollutant loading to receiving water bodies. Similarly, flow control would be provided for discharges to flow-sensitive water bodies, reducing adverse effects of high flows. Improvements to water quality and flow control would benefit fish and aquatic invertebrate species, many of which are harvested for human consumption.

Impacts of Alternative 1 No Action

The No Action Alternative involves less redevelopment of previously developed parcels, and areas would not be rezoned for II and UI uses. Less redevelopment would result in fewer opportunities <u>for to</u> implementing stormwater treatment and creat<u>eing</u> landscaped areas and green spaces that improve water quality. Existing pollutant loading to receiving water bodies would continue at current levels and continue to degrade aquatic habitat. Pollutants in stormwater runoff can cause avoidance of preferred habitat by aquatic species, reduced foraging efficiency of fish, and direct toxicity to fish species and their prey (NMFS 2020).

Except where protected by critical area and shoreline regulations, some minor amounts of habitat (such as landscaped or unpaved areas) may be converted to developed areas, which would decrease habitat available to species found in the study area. Because this alternative maintains existing zoning, there would be less development and therefore less habitat loss compared to other alternatives. Impacts to protected habitats, such as riparian corridors and

wetlands, would be minimized to the extent possible per Seattle Municipal Code. Compensatory mitigation would be provided for permanent unavoidable impacts.

Impacts of Alternative 2

Alternative 2 would apply a mix of II and UI zone concepts in approximately 10% of current MIC areas. These concepts would increase the number of trees and landscaping, and green spaces, which would provide opportunities for stormwater treatment as well as terrestrial wildlife habitat. Stormwater treatment would reduce pollutant loading to receiving water bodies.

This alternative would result in a small increase of approximately 80 residential units, mostly in the SODO/Stadium and Georgetown/South Park subareas. Development on currently undeveloped parcels would increase impervious surfaces and resulting stormwater runoff, which could further degrade water quality. However, conversion of previously developed areas also provides opportunities for stormwater retrofits that would improve water quality.

Depending on where these units are located, and the degree of shoreline and critical area regulations protection, new construction has the potential to reduce wildlife habitat by converting minor amounts of landscaping or other unpaved areas to developed areas. Appropriate siting of new housing, as well as adherence to existing regulations regarding protected habitats, would minimize habitat impacts.

Impacts of Alternative 3

Alternative 3 would apply a mix of II and UI zone concepts in approximately 14% of current MIC areas, the most of any alternative. Residential dwelling would increase within the MIC and within focused areas removed from the MIC by approximately 1,688 net units, primarily within the Ballard, SODO/Stadium, and South Park/Georgetown subareas. As discussed under Alternative 2, II and UI zone concepts promote development of green spaces that provide opportunities for stormwater treatment and wildlife habitat.

Although residential development could degrade wildlife habitat by developing undeveloped properties, and creating new and additional sources of contamination (see **Section 3.3 Water Resources**), redevelopment of previously developed areas could provide opportunities for more advanced stormwater treatment, thereby improving water quality in the study area.

Impacts of Alternative 4

Alternative 4 would apply a mix of II and UI zone concepts in approximately 13% of current MIC areas, only slightly less than Alternative 3, and would result in the creation of green spaces and landscaped areas that provide similar opportunities for stormwater retrofits.

This alternative would increase residential units by approximately 3,273 net units, more than the other alternatives, primarily in the Ballard, SODO/Stadium, and Georgetown/South Park

subareas. Although this increase has the potential to result in more pollutant sources and greater pollutant loading to receiving water bodies, redevelopment of a larger area also provides greater opportunities for stormwater retrofits that could improve water quality within the study area.

Increasing residential units could result in greater conversion of minor amounts of wildlife habitat provided by landscaped and unpaved areas to developed areas. However, existing habitat within the study area is limited, and habitat impacts would be minimal. Mitigation measures proscribed by existing regulations would avoid, minimize, and compensate for impacts to special status habitats (refer to Mitigation Measures below).

Impacts of the Preferred Alternative

The Preferred Alternative applies a mix of II and UI zone concepts in 14% of current MIC areas. Similar to other alternatives, these concepts promote development of green spaces that provide opportunities for stormwater treatment and wildlife habitat. This alternative would result in a slight expansion of limited industry-supportive housing in the UI zone concept, with associated loss of minor amounts of degraded wildlife habitat, as described under other alternatives.

3.4.3 Mitigation Measures

Incorporated Plan Features

Development regulation proposals include some elements of streetscape and "green factor" in the II and UI zones for Action Alternatives.

Regulations & Commitments

The proposed alternatives would incorporate impact avoidance and minimization measures during construction and operation in accordance with the regulations described in this section. Construction impact avoidance and minimization measures would include the management of noise, dust, and runoff caused by construction activities. The proposed alternatives would include stormwater management measures during the operation of all constructed features to treat stormwater in compliance with all applicable regulations.

Existing environmental regulations including the City of Seattle Code, Washington State Law, and Federal Laws, aim to reduce the potential impacts of projects and would apply to all alternatives. These regulations ensure impacts to the environment are avoided, minimized, documented, and mitigated to the greatest extent possible. The procedures associated with these regulations create opportunities for public notice and comment on projects prior to implementation. Environmentally sensitive areas are designated as environmentally critical

areas and are protected from avoidable development impacts. The principal existing regulations that protect ecosystem resources include the following:

- Federal Clean Water Act. Federal review by the United States Army Corps of Engineers (USACE) is required for to any project affecting waters of the United States (WOTUS). The USACE requires avoidance, minimization, and mitigation for impacts to WOTUS, endangered species, and cultural resources.
- State of Washington Laws. State review by the Washington Department of Ecology and/or the Washington Department of Fish and Wildlife is required for any project which affects waters of the state. The state requires projects demonstrate avoidance, minimization, and mitigation measures for any impacts to waters of the state and/or fish and wildlife.
- City of Seattle Municipal Code (SMC) Chapter 25.09 Regulations for Environmentally Critical Areas. Environmentally critical areas are protected by the SMC because they provide unique environmental functions that are difficult to replace. SMC 25.09 designates geologic hazard areas, steep slope erosion hazard areas, flood-prone areas, wetlands, fish and wildlife habitat conservation areas, and abandoned landfills as environmentally critical areas. Buffers and structure setbacks are also designated by SMC and are required to protect the functions of these environmentally critical areas.
- **Stormwater Regulations.** The City of Seattle ensures development complies with stormwater standards during the construction and operation phases of projects.
- Environmental Health Regulations. The Model Toxics Control Act of the State of Washington defines limits of contamination. Any project activities and related disturbances will need to address these limits based on the type of activity and proposed use of the parcel. The standards for voluntary cleanup for lower levels of contaminants are incorporated into new development or redevelopment parcels that have been noted to have contamination potential.

<u>Changes to the shoreline environment would need to comply with these and other federal,</u> <u>state, and local environmental regulations.</u> These environmental regulations condition development proposals to avoid, minimize, and/or mitigate potential impacts. Residual impacts are possible even with these environmental regulations and should be evaluated and avoided during project development.

Other Potential Mitigation Measures

- Mitigation measures would be developed on a case-by-case basis related to specific projects to comply with applicable federal, state, and City permitting requirements.
- Additional stormwater treatment would be integrated into new development or redevelopment as feasible including but not limited to green roofs, enhanced BMPs, and pervious pavement alternatives.
- New development or redevelopment could plant vegetation adjacent to streams and lakes to provide shade and organic inputs.
3.4.4 Significant Unavoidable Adverse Impacts

If all minimization and mitigation measures are implemented, no significant unavoidable adverse impacts are anticipated to plants and animals. The study area is already highly urbanized. Most plant species are nonnative introduced species common in urban environments. Development on industrial lands would not significantly reduce available habitat, particularly rare or unique habitat.

Terrestrial animal <u>species in the study area species</u> are adapted to urban conditions and have a high tolerance for human disturbance. Additional noise and disturbance that would be generated under the different alternatives would not be likely to adversely affect <u>these</u> species in the study area. The project does involve changes to shoreline or critical area policies or regulations regarding in-water work and is not anticipated to result in direct noise and disturbance to aquatic species.

Redevelopment of previously developed areas provides opportunities to reduce urban runoff and pollutant loading to aquatic habitat, potentially contributing to improved water quality in the study area. Improved water quality would benefit special status aquatic species and critical habitat, as well as other animals that prey on aquatic species.

Section 3.5 Contamination



This chapter describes the affected environment for contamination and presents the analysis completed to compare and contrast impacts from the alternatives. Mitigation measures for identified impacts and any significant unavoidable adverse impacts are also summarized.

Thresholds of significance utilized in this impact analysis include:

- Release or contamination of soils, groundwater, or surface water that requires removal and disposal.
- Hazardous chemicals or conditions that might result in health or safety impacts or impede future development.

Many different terms may be used to describe contamination at a site. The term hazardous material (or hazardous substance) is typically used to describe chemical contaminants in soils, groundwater, surface water, or other media at a site that have the potential to harm humans, animals, or the environment. Once the hazardous material is excavated or removed from the ground, it is considered a hazardous waste that must then be tested to determine how it would be properly disposed offsite at a licensed landfill or treatment facility. These terms are discussed further in **Section 3.5.3**.

3.5.1 Affected Environment

Primary & Secondary Study Areas

The study area for Contamination is defined as areas within 0.25-mile of the boundaries of the BINMIC and Greater Duwamish MICs that could be directly or indirectly affected by the construction activities or land uses that result from implementation of the industrial and maritime strategy. The secondary study area extends 0.25 miles from the full study area. ⁹

Data & Methods

The project team collected data from the following sources to support analysis to identify sites with confirmed or suspected contamination in soil, sediment, and groundwater, and sites where hazardous materials are used or stored; locate historical landfills; and evaluate potential effects of the project alternatives:

- Washington State Department of Ecology (Ecology) Facilities/Sites of Environmental Interest Geodatabase (Ecology 2021).
- Abandoned Landfill Study in the City of Seattle (Seattle-King County Department of Public Health 1984).

The initial list of confirmed or suspected contaminated sites, and sites that use or store hazardous materials within the full study area was developed from the Ecology geodatabase

⁹ Maps show the 0.25-mile buffer, but tabular data and text refer to the hazardous sizes inside the primary study area.

that lists all known facilities and sites of environmental interest in Washington State. The geodatabase includes information on:

- State cleanup sites
- Federal Superfund cleanup sites
- Solid waste facilities
- Underground storage tanks and leaking underground storage tanks
- Dairies
- Enforcement actions
- Hazardous waste generators

To focus the analysis on contamination for the EIS, the geodatabase was pared down to include only those sites that fall within two program areas overseen by Ecology: 1) Toxics Cleanup, and 2) Hazardous Waste and Toxics Reduction. The Toxics Cleanup Program tracks sites with confirmed or suspected contamination of soil, sediment, groundwater, or other media, and the Hazardous Waste and Toxics Reduction Program tracks sites where hazardous chemicals are used or stored and where spills to the environment could potentially occur.

The geodatabase was downloaded and then sorted to include those sites located within 0.25mile of the BINMIC and Greater Duwamish MIC (see **Exhibit 3.5-1** and **Exhibit 3.5-2**). The 0.25mile distance was selected as the boundary of the secondary study area as an appropriate minimum search distance typically used for environmental site assessments to identify current or historical conditions that could cause soil, groundwater, or other contamination on or adjacent to a property per the American Society of Testing and Materials (ASTM) standard practice ASTM E 1527-13 (ASTM 2013). The 0.25-mile search radius also relates to the maximum distance that groundwater contamination is likely to travel for the majority of sites with groundwater contamination.

Available information regarding historical landfills located within the full study area was reviewed in the 1984 abandoned landfill study (Seattle-King County Department of Public Health 1984).

Current Policy & Regulatory Frameworks

Model Toxics Control Act

The Model Toxics Control Act (MTCA) Cleanup Regulation (Washington Administrative Code [WAC] 173-340-710) is one of several environmental laws in Washington. Known as the state's cleanup law, MTCA authorizes the Washington State Department of Ecology (Ecology) to adopt cleanup standards for soil, groundwater, surface water, and air at sites where hazardous substances are present, and establishes processes for identifying, investigating, and cleaning up these sites. The term "site" in this context generally refers to the property where the hazardous substances are present but can extend onto adjacent properties. MTCA's main purpose is to prevent the creation of future hazards due to improper disposal of toxic wastes into the state's lands and waters. MTCA Cleanup Regulations apply to all cleanups, whether they're upland cleanups on land or in groundwater, or sediment cleanups in freshwater or marine environments.

Comprehensive Environmental Response, Compensation, and Liability Act

The Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA), known also as Superfund, is a federal law (40 CFR Parts 300-311, 355, and 373) used to identify sites where hazardous materials threaten the environment and or public health because of leaks, spills, or general mismanagement, and identifies the responsible party. CERCLA authorizes Superfund cleanup responses in two ways: short-term removal and long-term environmental remediation. These actions are conducted only at sites listed on EPA's National Priorities List (NPL). CERCLA powers and responsibilities overlap with the Resource Conservation and Recovery Act (RCRA) (see below), the Clean Water Act, and the Safe Drinking Water Act. CERCLA and RCRA share jurisdiction with respect to hazardous materials, and underground storage tanks containing petroleum products. CERCLA was amended by the Superfund Amendments and Re-authorization Act (SARA) in 1986.

Resource Conservation & Recovery Act & Washington State Dangerous Waste Regulations

The Resource Conservation and Recovery Act (RCRA) is a federal law (40 CFR Parts 239 through 282) that creates the framework for proper management of non-hazardous and hazardous solid waste. Washington State's Dangerous Waste Regulations under WAC 173-303 are based on the federal RCRA law, but Washington's regulations are more protective and include more wastes. Per WAC 173-350-021, solid waste is defined as "all putrescible and non-putrescible solid and semisolid wastes including, but not limited to, garbage, rubbish, ashes, industrial wastes, swill, sewage sludge, demolition and construction wastes, abandoned vehicles or parts thereof, contaminated soils and contaminated dredged material, and recyclable materials."

During construction on a contaminated site, a cleanup contractor (also referred to as a remediation contractor) would typically screen and classify soils as they are excavated and select one of the following appropriate types of landfills for off-site disposal:

- Inert landfills accept clean soil with no detectable concentrations of contaminants, or clean waste with some organic debris/wood waste and trace amounts of detectable petroleum hydrocarbons, volatile organic compounds, metals, or other contaminants that are below MTCA cleanup levels.
- Subtitle D landfills accept solid waste, including contaminated soils with concentrations of contaminants detected above MTCA cleanup levels (includes hazardous waste but does not include contaminants at concentrations that trigger Washington's Dangerous Waste Regulations)

 Subtitle C landfills accept waste designated as dangerous waste and have special controls such as double liners, double leachate collection and removal systems, and leak detection systems to prevent release of contaminants to the environment.

Seattle Municipal Code 25.09.220 (Environmentally Critical Areas Code) indicates that development on historical landfills is subject to Public Health—Seattle & King County requirements. The code also specifies methane barriers or appropriate ventilation per Title 22, Subtitle I, Building Code, and Public Health—Seattle & King County regulations.

The Title 10 King County Board of Health Solid Waste Regulation governs construction standards and methane controls on historical landfills. Authority is established under RCW Chapter 70.05 and Washington State Administrative Code WAC 173-304, Minimal Functional Standards for Solid Waste Handling, and WAC 173-351, Criteria for Municipal Solid Waste Landfills.

General requirements for complying with federal, state, and local Applicable or Relevant and Appropriate Requirements (ARARs) for cleanup actions under MTCA are listed in WAC 173-340-710-745. A summary of potentially applicable federal, state, and local ARARs identified for cleanup actions and potential soil, groundwater, and surface water contamination at sites within the full study area is included in **Exhibit 3.5-1**.

Regulatory Program or Policies	Lead Agency	Description
The Federal Clean Water Act (33 USC Section 1251)	Ecology	The Federal Clean Water Act establishes the basic structure for regulating discharges of pollutants into the waters of the United States and regulating quality standards for surface waters.
The Washington Water Pollution Control Act (Chapter 90.48 RCW; Chapter 173 201A WAC; Chapter 173- 200 WAC)	Ecology	The Washington Water Pollution Control Act requires the use of all known available and reasonable methods by industries and others to prevent and control the pollution of the waters of the state of Washington.
Comprehensive Environmental Response, Compensation, and Liability Act and All Appropriate Inquiries (40 CFR Part 312)	Ecology	Commonly known as Superfund, this federal regulation governs cleaning up abandoned or uncontrolled hazardous waste sites.
Sediment Management Standards (Chapter 173-204 WAC)	Ecology	Standards developed for Washington state to reduce and ultimately eliminate adverse effects on biological resources and significant threats to human health from surface sediment contamination.
The Resource Conservation and Recovery Act (40 CFR Parts 239 through 282)	Ecology	RCRA is a federal law that creates the framework for the proper management of hazardous and non-hazardous solid waste.

Exhibit 3.5-1 Federal, State, and Local Arars Potentially Applicable for Cleanup Actions at Contaminated Sites Within the Full Study Area

Regulatory Program or Policies	Lead Agency	Description
Dangerous Waste Regulations (Chapter 173 303 WAC) and the Washington Hazardous Waste Management Act (Chapter 70.105 RCW)	Ecology	The Dangerous Waste Regulations implement the Washington Hazardous Waste Management Act and establish requirements for generators, transporters, and facilities that manage dangerous waste.
Federal and State Clean Air Acts (42 USC 7401 et seq.; 40 CFR 50; RCW 70.94; WAC 173-400, 403)	Puget Sound Clean Air Agency	These federal and state laws regulate air emissions from stationary and mobile sources, including construction sites.
The State Environmental Policy Act (RCW 43.21C; WAC 197-11	Ecology	SEPA ensures environmental values are considered during decision-making by state and local agencies when issuing permits for private projects; constructing public facilities; or adopting regulations, policies, or plans.
The Occupational Safety and Health Act (29 CFR 1910); Washington industrial Safety and Health Act (296- 800 WAC)	Washington Department of Labor and Industries	These federal and state rules regulate the safety and health of workers in the workplace, including construction sites.
General Occupational Health Standards (Chapter 296-62 WAC)	Washington Department of Labor and Industries	These rules are designed to protect the health of employees and help to create a healthy workplace by establishing requirements to control health hazards.
Safety Standards for Construction Work (Chapter 296-155 WAC)	Washington Department of Labor and Industries	These safety and health standards help protect workers at construction sites.
Minimum Standards for Construction and Maintenance of Wells (Chapter 173-160 WAC)	Ecology	These standards contain requirements for installation, maintenance, and decommissioning of groundwater monitoring wells.
Industrial Waste Discharge to Metropolitan King County Sewer System	King County Industrial Waste Program	This program regulates the discharge of industrial/commercial wastewater, including construction dewatering water, to the King County sanitary sewer system.

Source: Herrera, 2021.

Current Conditions

Full Study Area

A total of 710 Toxics Cleanup sites with confirmed and suspected contamination were identified within the full study area (Ecology 2021). Of these, 159 sites are located in the BINMIC and 551 are located in the Greater Duwamish MIC (see **Exhibit 3.5-2** and **Exhibit 3.5-4**, respectively). These sites have undergone various stages of investigation and cleanup. Some sites are still awaiting cleanup, others have been investigated to determine the nature and extent of contamination, and some sites have been satisfactorily cleaned up to the point where Ecology has issued a No Further Action letter.

In addition, a total of 1,537 Hazardous Waste and Toxics Reduction sites were identified within the full study area (Ecology 2021). Of these, 276 sites are located in the BINMIC and 1,261 are located in the Greater Duwamish MIC (see **Exhibit 3.5-3** and **Exhibit 3.5-5**, respectively). These sites typically range from well-managed, well-kept facilities with few if any historic spills or enforcement actions by Ecology, to facilities where violations and/or spills to the environment have occurred. Spills, whether documented or not, can cause soil, groundwater, or surface water to become contaminated if not cleaned up properly and promptly.

A total of five historical landfills were identified within the study area. All the landfills have documented soil and/or groundwater contamination as well as potential constraints for construction on or adjacent to the sites due to the poor structural support provided or settlement, and risk of methane intrusion into structures that may require mitigation. Three landfills have prescribed 1,000-foot methane buffers.

Four federal Superfund sites were identified within the study area, all within the Greater Duwamish MIC. These sites have undergone various stages of investigation and cleanup. Three sites have had cleanup mostly completed or completed and are undergoing long-term monitoring to ensure the cleanup activities are protective to human health and the environment. One site has been investigated to determine the nature and extent of contamination and has had five Early Action Area (EAA) cleanups. The remaining areas are the subject of phased design and cleanup actions.

Exhibit 3.5-6 provides a summary of the total number of Toxics Cleanup Sites, and Hazardous Waste and Toxics Reduction Sites within the BINMIC and Greater Duwamish MIC and subareas. Note that because some sites are tracked by Ecology's Toxics Cleanup Program in multiple categories, the total number of Toxics Cleanup Program sites listed is not equal to the sum of all sites shown in each program subcategory in **Exhibit 3.5-6**.



Exhibit 3.5-2 Confirmed or Suspected Contaminated Sites Within 0.25-mile of the BINMIC



Exhibit 3.5-3 Hazardous Waste and Toxics Reduction Sites Within 0.25-mile of the BINMIC

Source: Ecology, 2021.

Seattle Industrial & Maritime Strategy • September 2022 • Final Environmental Impact Statement



Exhibit 3.5-4 Confirmed or Suspected Contaminated Sites Within 0.25-mile of the Greater Duwamish MIC



Exhibit 3.5-5 Hazardous Waste and Toxics Reduction Sites Located Within 0.25-mile of the Greater Duwamish MIC

Source: Ecology, 2021.

MIC	Subarea	Enforcement Action	Federal Superfund	Independent Cleanup / Remedial Action	LUST	UST	Sediments	State Cleanup	VCP	Total Number of Toxics Cleanup Program Sitesª	Total Nu Hazardo and Toxics Progra	umber of us Waste Reduction m Sites
BINMIC	Ballard	3	0	9	19	44	2	22	29	73	143	276
	Interbay Dravus	1	0	5	11	21	1	16	13	38	79	
	Interbay Smith Cove	0	0	5	17	35	1	14	16	48	54	
Greater	SODO/Stadium	5	2	32	126	234	12	112	73	331	672	1,261
Duwamish MIC	Georgetown/South Park	20	0	26	76	141	4	81	51	220	589	
				Grand	Totals	Within	the Fu	III Study	y Area	710	1,!	537

Exhibit 3.5-6 Summary of Toxics Cleanup Sites and Hazardous Waste and Toxics Reduction Sites Within the BINMIC and Greater Duwamish MIC and Subareas

^a Because some sites are tracked by Ecology's Toxics Cleanup Program in multiple categories, the total number of Toxics Cleanup Program sites listed is not equal to the sum of all sites shown in each program subcategory.

LUST: leaking underground storage tank

UST: underground storage tank

VCP: voluntary cleanup program

Source: Ecology, 2021.

Ballard

A total of 73 Toxics Cleanup sites and 143 Hazardous Waste and Toxics Reduction were identified in the Ballard Subarea (see **Exhibit 3.5-2** and **Exhibit 3.5-3**).

A historical landfill is located within the Ballard Subarea, adjacent to the south of Shilshole Avenue NW, along Salmon Bay (see **Exhibit 3.5-2**; City of Seattle 2021). Limited information regarding the landfill is available and it was not identified in the 1984 Abandoned Landfill Study (Seattle-King County Department of Public Health 1984). The landfill likely began operating in the early 1900s, covers approximately 10.5 acres, and is now developed with industrial and office buildings. Development within the former landfill area is subject to special engineering and construction management requirements to prevent damage from methane gas buildup, subsidence, and earthquake-induced ground shaking. Development in this area must comply with critical areas regulations.

Interbay Dravus

A total of 38 Toxics Cleanup sites and 79 Hazardous Waste and Toxics Reduction were identified within the Interbay Dravus Subarea (see **Exhibit 3.5-2** and **Exhibit 3.5-3**).

The Interbay Landfill is located adjacent to the west of the Interbay Dravus Subarea ((see **Exhibit 3.5-2**). The landfill is situated along 15th Avenue West, south of West Dravus Street and north of West Wheeler Street. A 1,000-foot methane buffer for the landfill overlaps with the southern portion of the Interbay Dravus secondary study area. The landfill consists of approximately 55 acres of land presently occupied by the Interbay Golf Center. The landfill, also known as the Interbay Dump or Sanitary Landfill No. 2, was established by the City in 1911 and continued to be used off and on until 1968 (Seattle-King County Department of Public Health 1984). Municipal solid waste from local homes and businesses was dumped at the south end, the north end was operated as a fire dump and received combustible wastes including wastes from local industries and the military. The landfill contains a wide range of putrescible and non-putrescible solid waste. The landfill is prone to settlement and is still producing methane gas. High groundwater and leachate formation are also concerns at this site.

Interbay Landfill and areas within a 1,000-foot radius are regulated as an Abandoned Landfill environmentally critical area (Landfill ECA). Specifically, Seattle Building Code (SBC) 1811— *Methane Mitigation Measures* requires that all construction within a Landfill ECA be protected from accumulation of methane within or under the enclosed portion of a building. Methane mitigation systems typically consist of passive or active venting systems installed in subslab /crawlspace areas coupled with monitoring systems in enclosed interior spaces.

Interbay Smith Cove

A total of 48 Toxics Cleanup sites and 54 Hazardous Waste and Toxics Reduction were identified within the Interbay Smith Cove Subarea (see **Exhibit 3.5-2** and **Exhibit 3.5-3**).

The northern portion of the Interbay Smith Cove Subarea is also located within the 1,000-foot methane buffer for the Interbay Landfill (see **Exhibit 3.5-2**). As previously mentioned, areas within this buffer are subject to the methane mitigation measures outline under SBC 1811 to prevent accumulation of methane within or under the enclosed portion of a building.

SODO/Stadium

A total of 331 Toxics Cleanup sites and 672 Hazardous Waste and Toxics Reduction were identified within the SODO/Stadium Subarea (see **Exhibit 3.5-4** and **Exhibit 3.5-5**).

The West Seattle Landfill, previously known as the West Hanford Street Landfill, is located within the SODO-Stadium Subarea. The landfill is situated along Harbor Avenue SW, just south of SWA Florida Street (see **Exhibit 3.5-4**; City of Seattle 2021). The landfill is approximately 20 acres in size, built on former tidelands, and operated from 1939 until 1966. The landfill has a 1,000-foot methane buffer and areas within the buffer are subject to the methane mitigation measures outline under SBC 1811 (City of Seattle 2021).

The West Seattle Landfill accepted municipal solid waste as well as industrial wastes from local industries associated with lumber yards and mills, ship building, creosote pile treating, pesticide manufacturing, and a steel mill. The landfill historically had problems with fires and the Seattle Fire department also used a portion of the site for its oil fire control school (Seattle-King County Department of Public Health 1984).

A second landfill is located within the SODO/Stadium Subarea (see **Exhibit 3.5-4**; City of Seattle (2021). The landfill is approximately 51 acres in size, straddles 6th Avenue South, and extends from South Forest Street on the north end to South Charlestown Street on the south end. The landfill was not identified in the 1984 abandoned landfill study conducted by the Seattle-King County Department of Public Health. The former landfill area is densely developed with industrial/commercial buildings.

Three federal Superfund sites in the SODO-Stadium area have undergone cleanup. These include the Pacific Sound Resources and Lockheed West Seattle sites on what is now the Terminal 5 property on the west side of the west Duwamish waterway. The Harbor Island Superfund site is comprised of seven operable units—smaller areas to make cleanup easier and more manageable—with five having completed cleanup and two (the East Waterway and Todd Shipyards sediment areas awaiting cleanup estimated in 2023-2024).

Georgetown/South Park

A total of 220 Toxics Cleanup sites and 589 Hazardous Waste and Toxics Reduction were identified within the Georgetown/South Park Subarea (see **Exhibit 3.5-4** and **Exhibit 3.5-5**).

The South Park landfill located within the Georgetown/South Park Subarea covers approximately 96 acres and is bounded on the east by West Marginal Way and 5th Avenue South; on the north by Kenyon Street; on the west by 2nd Avenue South and Occidental Avenue; and on the south by Sullivan Street (see **Exhibit 3.5-4**; City of Seattle 2021). It began operating

after 1945 and closed in 1966 when the site was converted to a solid waste landfill (Seattle-King County Department of Public Health 1984). The landfill was used primarily for non-putrescible wastes and sawdust in the early years and operated as fire dump by the City where combustible refuse was burned until 1961. The landfill was also used to dispose putrescible waste as well as industrial wastes from nearby industries. An investigation in 1983 revealed fill soils with various debris, scattered organics, and an oily sheen and odors (Seattle-King County Department of Public Health 1984).

The Lower Duwamish Waterway (LDW) federal Superfund site extends 5 miles from the mouth of the Duwamish waterway in the SODO-Stadium area to the southern extent of the waterway where it becomes the Duwamish River in the Georgetown/South Park subarea. The LDW site encompasses upland sources of contamination as well as contamination within the waterway. The EPA is responsible for administering the cleanup of sediments in the Waterway, and Ecology is responsible for controlling sources of pollution to the Waterway. Most of the human health risk comes from polychlorinated biphenyls (PCBs), arsenic, carcinogenic polycyclic aromatic hydrocarbons (cPAHs), as well as dioxins and furans. As a result, consumption of resident fish and shellfish, as well as contact with contaminated sediments, pose a risk to human health (EPA 2021).

3.5.2 Impacts

Impacts Common to All Alternatives

Development under any of the alternatives may encounter hazardous materials such as contaminated soil, <u>soil vapor</u>, groundwater, surface water, or sediments. The greatest potential for impacts associated with contamination would occur during construction when sites are disturbed. Construction activities could release hazardous materials due to ground disturbing, dewatering, and demolition activities. Development within the study area, especially where known hazardous material sites are located, would address the removal of hazardous materials, which could include contaminated soils, groundwater, surface water, and, in older structures, the potential for lead-based paints and asbestos-containing materials (ACMs).

A soil and groundwater management plan could be necessary for construction activities in areas with known or suspected contamination. Contaminated soils excavated during construction activities would require special handling, transport, storage, and off-site disposal. If soils are not contaminated, excavations at many sites would still require off-site hauling if the soils cannot be relocated and placed onsite. If there is concurrent construction requiring earth fill in close proximity, excavated materials could be transported to the nearby site as long as the excavated material is protected from precipitation and surface water runoff.

Depending on groundwater depth and the type of hazardous materials, it is possible that contaminants from historic spills or releases may have infiltrated and migrated, requiring

additional cleanup. Cleanup efforts implemented before or during construction would reduce potential short-term and long-term impacts.

For contaminated soil, MTCA generally requires residential land uses to use the most protective cleanup levels established under MTCA Method A or B cleanup levels (WAC 173-340-740). These requirements apply to most land uses except those that meet the definition of "industrial property" as defined in WAC 173-340-200 and 173-340-745. For industrial properties, MTCA allows less restrictive soil cleanup levels established under MTCA Method A or C (WAC 173-340-745) based on adult worker exposure scenarios only and including the use of institutional controls.¹⁰ Access to industrial properties by the public, especially children, or even proximity to residential areas may limit use of the less restrictive standard. All sites being redeveloped and needing cleanup under MTCA would be assessed for the nature of the contamination, the complexity and location of the site, and the current and potential land use to determine appropriate cleanup standards. Because documented contamination requiring cleanup would be removed or contained prior to new development, it is assumed there would be no significant health and safety impacts on those living, working, or visiting the area, or impacts on the intended uses of properties within the study area.

As growth occurs in the study area, there is potential for hazardous material spills associated with petroleum products to increase as traffic and the potential for accidents increases. With growth there is also the potential for increased risk of spills from industrial activities, industrial processes, or use of industrial chemicals. Any spills would be cleaned up consistent with applicable state and local requirements and no significant impacts are anticipated.

Alternative 1 would allow the least new jobs and housing and Alternative 4 the most in each subarea and across the whole subarea. See **Exhibit 3.5-7** and **Exhibit 3.5-8**.

¹⁰ Measures undertaken to limit or prohibit activities that may interfere with the integrity of an interim action or cleanup action or that may result in exposure to hazardous substances at a site.





Note: This chart was updated to include the Preferred Alternative. Source: City of Seattle, 20224; BERK, 20224.



Exhibit 3.5-8 Total Housing in Study Area by Alternative

Note: This chart was updated to include the Preferred Alternative. Source: City of Seattle, 20224; BERK, 20224.

Ballard

Alternative 1 would allow the least new jobs and housing and Alternative 4 the most. The Ballard Subarea would see the third highest growth in jobs and the second highest in housing under the alternatives of all the subareas. This subarea also has 73 Toxics Cleanup Program sites and 143 Hazardous Waste and Toxics Reduction Program sites. The risks of release of contaminants or of hazardous chemicals being used or causing conditions that result in health or safety impacts or impede future development are potentially higher than with the two Interbay subareas, but less than the SODO/Stadium and Georgetown/South Park subareas. Although these risks are considered significant, they are avoidable with mitigation.

Interbay Dravus

Alternative 1 would allow the least new jobs and housing and Alternative 4 the most. The Interbay Dravus Subarea would see modest growth in jobs and housing under the alternatives compared to the other subareas. This subarea has 38 Toxics Cleanup Program sites and 79 Hazardous Waste and Toxics Reduction Program sites. The risks of release of contaminants or of hazardous chemicals being used or causing conditions that result in health or safety impacts or impede future development are less than the Ballard, SODO/Stadium and Georgetown/South Park subareas. These risks are considered significant but avoidable with mitigation.

Interbay Smith Cove

Alternative 1 would allow the least new jobs and housing and Alternative 4 the most. The Interbay Smith Cove Subarea would also see modest growth in jobs but minimal growth in housing under the alternatives compared to the other subareas. This subarea has 48 Toxics Cleanup Program sites and 54 Hazardous Waste and Toxics Reduction Program sites. The risks of release of contaminants or of hazardous chemicals being used or causing conditions that result in health or safety impacts or impede future development are also less than the Ballard, SODO/Stadium and Georgetown/South Park subareas. These risks are considered significant but avoidable with mitigation.

SODO/Stadium

Alternative 1 would allow the least new jobs and housing and Alternative 4 the most. The SODO/Stadium Subarea would see the most growth in jobs and housing under the alternatives compared to the other subareas. This subarea also has 331 Toxics Cleanup Program sites and 672 Hazardous Waste and Toxics Reduction Program sites. The risks of release of contaminants or of hazardous chemicals being used or causing conditions that result in health or safety impacts or impede future development are greater than the other subareas. These risks are considered significant but avoidable with mitigation.

Georgetown/South Park

Alternative 1 would allow the least new jobs and housing and Alternative 4 the most. The Georgetown/South Park Subarea would see the second highest growth in jobs and third highest growth in housing compared to the other subareas. This subarea also has 220 Toxics Cleanup Program sites and 589 Hazardous Waste and Toxics Reduction Program sites. The risks of release of contaminants or of hazardous chemicals being used or causing conditions that result in health or safety impacts or impede future development are greater than other subareas except the SODO/Stadium Subarea. These risks are considered significant but avoidable with mitigation.

Equity & Environmental Justice Considerations

Under any of the Action Alternatives, the primary equity and environmental justice concern for the proposal would be that cleanup of contaminated sites could cause temporary adverse effects from potential exposure of workers, nearby residents, and animals to contaminated soil, groundwater, surface water, fugitive dust, or spilled hazardous materials if mitigation measures are not fully implemented. This could lead to exposure of vulnerable communities, including lower-wage or under-represented workers, to inequitable exposure to contamination.

Under the Alternative 1 No Action, humans, plants, and animals could potentially be exposed to contaminants at existing contaminated sites in all subareas.

The greatest impacts would be associated with Alternative 4 because it would result in the most sites disturbed and cleaned up, housing units created, and workers living and working in the subareas. However, after completion of cleanup actions for projects under all the Action Alternatives, nearby residents would benefit from reduced risk of potential exposure to contaminants.

In order to mitigate potential exposure to contaminants, all workers would be issued personal protective equipment and protected by measures implemented under the contractor's site-specific health and safety plan.

Although all alternatives would likely result in short-term adverse effects on this determinant of equity and social justice, the Action Alternatives would generally have positive long-term benefits.

Impacts of Alternative 1 No Action

Under Alternative 1 No Action, contaminated sites and spills would still be investigated and cleaned up in accordance with MTCA and other applicable local, state, and federal laws. Industrial facilities and other sites would continue to manage hazardous and non-hazardous solid wastes as required under RCRA and Washington's Dangerous Waste Regulations to prevent human exposures and releases to the environment. A total of 8,330,000 square feet (SF) of industrial space and 2,900,000 SF of non-industrial space would be developed. This development would increase the short-term risk of exposure to contaminants as sites are cleaned up but result in a long-term benefit of lower concentrations of chemicals after sites are cleaned up. With the increases in industrial jobs (described below by subarea) and industrial space there would be an increased risk of chemical exposures and industrial spills related to industrial processes.

Impacts of Alternative 2

The impacts of Alternative 2 are similar to those described above under **Impacts Common to All Alternatives**, but the increased development under Alternative 2 increases the likelihood of encountering contaminated sites and for hazardous chemicals to cause impacts on health and safety or cause project delays. Under Alternative 2, the number of industrial jobs in the subareas would increase above Alternative 1 No Action by 2,000 in Ballard, 1,000 in Interbay Dravus, 1,100 in Interbay Smith Cove, 5,500 in SODO/Stadium, and 3,400 in Georgetown/South Park. In addition, the total square feet of industrial space developed within the subareas would more than double, from 8,330,000 SF under the No Action Alternative to 17,430,000 SF under Alternative 2.

With more industrial jobs and more than double the square footage of industrial space, there would be an increase in the number of Toxics Cleanup Program sites developed and cleaned up and an increase in the number of new Hazardous Waste and Toxics Reduction Program sites where chemicals are used. With the increase in industrial jobs and industrial space there would be an increased risk of chemical exposures and industrial spills related to industrial processes.

There would <u>be a slight decrease</u>also be an increase in non-industrial jobs of <u>2,100</u>9,500 in new building space of 2,375,000 square feet, slightly lower than Alternative 1 No Action; the development of non-industrial space has the potential to increase the risk of potential chemical exposures.

The increase in total housing units from 488 under the No Action Alternative <u>1 No Action</u> to 493 under Alternative 2 would also mean slightly more residents living in the subareas who could be exposed to contamination. The increased development would result in increases in traffic, which would increase the potential for hazardous material spills related to traffic accidents.

All these impacts together are considered significant but avoidable with mitigation.

Impacts of Alternative 3

The impacts of Alternative 3 are similar <u>to</u> as those described above under Impacts Common to All Alternatives and under Impacts of Alternative 2. The zoning and development of residential properties and non-industrial mixed-use properties would require more restrictive cleanup levels under MTCA. This would have the positive benefit of removing more contamination to achieve lower cleanup levels and further reduce potential exposures.

Under Alternative 3, there would be slight increases in the number of industrial employees added in each of the subareas and Ballard and SODO/Stadium subareas would see the largest increases in number of housing units created.

The number of industrial jobs would increase above Alternative 1 No Action by <u>24</u>,300 in Ballard, <u>1,</u>600 <u>each</u> in Interbay Dravus<u>and</u>, <u>500 in</u> Interbay Smith Cove, and <u>6,500</u>1,000 in SODO/Stadium, and would decrease by 300<u>3,100</u> in Georgetown.

Beyond Alternative 1 No Action, Alternative 3 would result in 2,87011,970,000 SF of industrial space, 4,725200,000 SF of non-industrial space. Additionally, 2,101-195 housing units would be developed above Alternative 1 within the subareas. As with Alternative 2, the increases

industrial jobs added, and industrial and non-industrial space added under Alternative 3 would increase the risk of potential chemical exposures.

All these impacts together are considered significant but avoidable with mitigation.

Impacts of Alternative 4

The impacts of Alternative 4 are similar <u>to</u> as those described above under Impacts Common to All Alternatives and under Impacts of Alternative 3. The zoning and development of residential properties and non-industrial mixed-use properties would require more restrictive cleanup levels under MTCA. As with Alternative 3 this would have the positive benefit of removing more contamination to achieve lower cleanup levels and further reduce potential exposures.

Under Alternative 4, the number of industrial jobs would increase by 100-above Alternative 1 No Action by a few thousand in each subarea in the Ballard and Interbay Dravus subareas, remain the same in the Interbay Smith Cove Subarea, and decrease by 300in the SODO/Stadium Subarea and 100 in the Georgetown/South Park Subarea. The total square footage of industrial space would decrease slightly, but an additional 500,000 SF of non-industrial space, and 3,686 housing units would be developed within the subareas. With the slight increases in the number of industrial employees working in the Ballard and Interbay Dravus subareas and increases in residents living in the developed housing units in the Ballard, Interbay Dravus, SODO/Stadium, and Georgetown/South Park subareas, potential exposures to contaminants or chemicals would increase due to the number of people working and living there.

All these impacts together are considered significant but avoidable with mitigation.

Impacts of the Preferred Alternative

Under the preferred alternative, increases in employment are expected to be similar to Alternative 2, while increases in housing are expected to be similar to alternatives 3 and 4. The impacts of the Preferred Alternative are similar to those described above under **Impacts Common to All Alternatives** and under Impacts of alternatives 3 and 4. More restrictive MTCA cleanup levels required for development of residential properties and non-industrial mixed-use properties would have the positive benefit of removing more contamination and further reducing potential exposures.

Under the Preferred Alternative, the number of industrial jobs would increase by 2,085 above Alternative 1 No Action in the Ballard Subarea, by 884 in Interbay Dravus, by 430 in Interbay Smith Cove, by 922 in SODO/Stadium, and 133 in the Georgetown/South Park Subarea. The total square footage of industrial space would increase by 3,117,176, and an additional 1,897,973 SF of non-industrial space, and 3,009 housing units would be developed within the subareas. With the slight increases in the number of industrial employees working in the subareas and increases in residents living in the developed housing units in the subareas, potential exposures to contaminants or chemicals would increase due to the number of people working and living there.

All these impacts together are considered significant but avoidable with mitigation.

3.5.3 Mitigation Measures

Incorporated Plan Features

There are no incorporated plan features related to contamination.

Regulations & Commitments

All site development projects would be required to comply with applicable federal, state, and local regulations. The existing regulations described under *Current Policy and Regulatory Frameworks* in **Section 3.5.1 Affected Environment** establish standards for site characterization, cleanup of hazardous materials, and disposal of hazardous waste, as well as mitigation measures for development on or adjacent to historical landfills.

Site contamination and remediation are addressed at the time of development or redevelopment through existing processes under MTCA and other regulations. SEPA documentation submitted with project applications requires disclosure of known or suspected contamination of soil, soil vapor, groundwater, and sediment.

Development of known or suspected contaminated sites would require a Phase I Environmental Site Assessment per ASTM 1527 and potentially a Phase II Environmental Site Assessment (soil, sediment, and/or groundwater sampling) prior to construction-related activities, including demolition. Prior to renovation or demolition of structures, hazardous building material surveys (HBMS) would be conducted, and abatement of lead-based paints and asbestos, if present, would be required by the Puget Sound Clean Air Agency (PSCAA) and other agencies and laws. To the extent possible, the amount of contamination at a site with known contamination would be verified prior to construction, to minimize exposure to hazardous materials.

For contaminated sites with current industrial land use designations that maintain an industrial focus under new land use designations, cleanup will not likely happen until redevelopment occurs, or there is a property sale that triggers site characterization and remediation activities in order to secure project financing.

MTCA sets strict cleanup standards to ensure human health and the environment are not compromised. Washington's Dangerous Waste Regulations ensure that non-hazardous and hazardous solid wastes are properly managed from cradle to grave at industrial sites and other properties to prevent impacts to human health and the environment. Compliance with the regulations results in low levels of contamination after site cleanup and redevelopment.

The Seattle Department of Construction and Inspections and Seattle Fire Department regulate hazardous materials through the International Building Code and the International Fire Code. New development would need to meet the requirements prior to permits being issued for construction. Development and implementation of Construction Stormwater Pollution Prevention Plans would be required by the City to minimize the potential for release of hazardous materials to soil, groundwater, or surface water during construction.

Other Potential Mitigation Measures

During construction, the following measures would minimize potential impacts of accidental releases of hazardous material:

- Preparing a comprehensive contingency and hazardous substances management plan, a worker health and safety plan, a spill prevention control and countermeasures plan, and a Construction Stormwater Pollution Prevention Plan.
- Managing and disposing of hazardous or contaminated materials in accordance with applicable laws and regulations.
- Prior to commencing site-specific subsurface investigations of soils, the Duwamish tribe should be notified to ensure that an archaeologist can observe the work. Standard archaeological techniques should be used during excavation and drilling for the potential discovery and preservation of cultural and historical artifacts related to the indigenous tribes. Any evidence gathered should be presented and turned over to the Duwamish Tribe at the Duwamish Longhouse & Cultural Center.
- The City and partner agencies could improve coordination and improve the user experience for community members registering complaints or requesting information about enforcement related to contamination from sites or businesses.

3.5.4 Significant Unavoidable Adverse Impacts

No significant unavoidable adverse impacts would occur with the implementation of mitigation measures. Hazardous materials sources would not impede redevelopment. Federal, state, and local regulations are in place to require cleanup of sites and to promote spill prevention.

Section 3.6 Noise



This chapter assesses the potential noise impacts associated with implementing the alternatives under consideration. The chapter includes a description of noise and noise levels in general, regulatory standards for noise, noise sources and potential sensitive noise receptors in the maritime and industrial areas of Seattle, the methods used to assess noise and impacts from noise, and an assessment of noise impacts associated with each alternative, as well as potentially feasible noise mitigation measures where appropriate. This analysis evaluates noise conditions and potential impacts for each MIC on an area-wide cumulative basis and, and in specific areas where the alternatives consider greater degrees of change.

Under the SEPA Rules (see WAC 197-11-330, WAC 197-11-440 and WAC 197-11-794), the evaluation of the significance of potential impacts considers whether there is a reasonable likelihood of more than a moderate adverse impact on environmental quality (WAC 197-11-794). In making this assessment, the following are considered:

- The context of the proposal, including the physical setting.
- The intensity of the impact, which depends on its magnitude and duration.
- The likelihood of the impact's occurrence.
- The duration of the impact.

In many cases, regulatory thresholds are used to judge significance. If actions would meet regulatory thresholds, then the determination is typically that the level of impact is unlikely to be significant. For the purposes of this programmatic impact analysis, noise is analyzed by examining whether:

- The alternative would cause future traffic noise levels of 10 dBA or more above existing noise levels.
- After application of mitigation, the alternative fails to comply with SMC Maximum Allowable Sound Level for receivers.

3.6.1 Affected Environment

Primary & Secondary Study Areas

The study area used in the noise assessment encompasses areas where construction noise or land uses that result from implementation of the industrial and maritime strategy would have the potential to affect nearby noise-sensitive land uses. The Primary Study Area includes the full study area and subareas referenced in **Chapter 2**, as well as a Secondary Study Area that includes areas extending 500 feet from the Primary Study Area boundary. Areas within the 500foot radius include portions of the City of Seattle, City of Tukwila, and unincorporated King County.

Data & Methods

The project team used the following data sources for this construction noise assessment:

- Highway Construction Noise Handbook (FHWA 2006)
- City of Seattle Municipal Code (SMC Chapter 25)
- State of Washington Administrative Code (Chapter 173-60 WAC)

Background

Noise is defined as excessive or undesired sound. Human sensitivity to sound depends on its intensity, frequency composition and duration. Sound waves are received by the human ear as variations in pressure through a medium such as air over time. The loudest sounds typically encountered by humans are a million times greater in pressure than faint sounds at the threshold of hearing. Because of this large scale, noise intensity is measured on a scale whose units are termed decibels (dB) which use a logarithmic scale to compress the range of pressure fluctuations to a more usable noise metric. A logarithmic loudness scale with 0 dB corresponds roughly to the threshold of human hearing and 120 dB to 140 dB corresponds to the threshold of pain.

The greater sensitivity of the human ear to certain frequencies is approximated by skewing (or weighing) the decibel scale towards those frequencies. The weighted decibel scale which best approximates the response of the human ear is known as the A- weighted scale (dBA). The A-scale deemphasizes low frequency noise, slightly emphasizes mid-high frequency noise, and slightly de-emphasizes high frequency noise.

Community noise levels are typically measured over an extended period of time to characterize a community noise environment and evaluate cumulative sound impacts. This time-varying characteristic of environmental noise is described using statistical noise descriptors. Commonly used technical noise terms used in this Chapter include:

Decibel (dB)—A logarithmic unit, which expresses the ratio of the measured sound pressure level to a standard reference level. Each increase in 10 dB equates to a tenfold increase in the magnitude of sound energy.

A-weighted Sound Level (dBA)—Sound level meters are usually equipped with weighting circuits, which filter out selected frequencies. The A-scale on a sound level meter best approximates the frequency response of the human ear.

Equivalent Noise Level (Leq)—The level of a constant sound having the same sound energy as the fluctuating levels measured over a period of time.

Ambient Noise Level—The composite of noise from all sources near and far. The normal or existing level of environmental noise at a given location (FWHA 2011).

Maximum Noise Level (Lmax)—The Lmax is the instantaneous maximum noise level measured during a measurement period of interest. This is the noise metric used when comparing a project's impacts to the City of Seattle Maximum Permissible Sound Levels.

Day-Night Average Sound Level (Ldn)—The day-night average sound level is the energy average of the A-weighted sound levels occurring during a 24-hour period, accounting for the greater sensitivity of most people to nighttime noise by weighting ("penalizing") nighttime noise levels by adding 10 dBA to noise between 10:00 p.m. and 7:00 a.m. (Seattle 2015).

Steady-state sound is typically described using the Leq descriptor. Impulse sound is sound generated over a relatively short duration period (e.g., a car horn or back-up alarm). Impulsive sound is typically characterized using the Lmax descriptor. Seattle's Noise code defines "Impulsive sound" where the peak of the sound level is less than one (1) second and short compared to the occurrence rate; the onset is abrupt; the decay rapid; and the peak value exceeds the ambient level by more than ten (10) dB(A) (Seattle 2015).

Effects of Noise on People

The effects of noise on people can be listed in three general categories:

- Subjective effects of annoyance, nuisance, and dissatisfaction
- Interference with activities such as speech, sleep, and learning
- Physiological effects such as startling and hearing loss

In most cases, environmental noise typically produces effects in the first two categories. Workers in industrial plants sometimes experience noise in the third category. There is no completely accurate way to measure the subjective effects of noise or the corresponding reactions of annoyance and dissatisfaction. This lack of a standard is primarily because of the wide variation in individual thresholds of annoyance and habituation to noise. Thus, an important way of determining a person's subjective reaction to a new noise is to compare it to the existing or "ambient" environment to which that person has adapted. In general, the more a new noise exceeds the previously ambient noise level, the less acceptable the new noise will be judged by listeners.

With regard to increases in A-weighted noise levels, the following relationships occur:

- People generally perceive a 10 dBA increase as a doubling of loudness and can cause an adverse response. For example, a 70 dBA sound will be perceived by an average person as twice as loud as a 60 dBA sound.
- People generally cannot detect differences of 1 dBA to 2 dBA between noise sources, but under typical listening conditions, differences of 3 dBA can be noticeable.
- A 5 dBA change would probably be perceived by most people under normal listening conditions.

Because of their logarithmic nature, decibels do not arithmetically add. For example, if two sound levels are added with one sound level being 10 dB louder than the other, the combined sound level is only 0.4 dB more than the louder sound level.

Exhibit 3.6-5 shows typical A-weighted noise levels and human response.

When distance is the only factor considered, sound levels from isolated point sources of noise typically decrease by about 6 dBA for every doubling of distance from the noise source. When the noise source is a continuous line (for example, vehicle traffic on a highway), noise levels decrease by about 3 dBA for every doubling of distance away from the source. Noise levels at receptor locations can also be affected by factors other than the distance from the noise source. For example, topographic features and physical barriers can increase or decrease noise levels by absorbing, reflecting, or scattering sound waves. Atmospheric conditions (wind speed and direction, humidity levels, and temperatures) can affect the degree to which sound is attenuated over distance. Temperature inversions and wind conditions can also diffract and focus a sound wave to a location at considerable distance from the noise source. The degree of impact also depends on the individual sensitivity of people listening and on ambient sound levels. For example, where background noise levels are high, introducing a new noise source tends to have less impact than in an environment where background noise levels are low.

Current Policy & Regulatory Frameworks

Federal Guidelines

The U.S. Department of Housing and Urban Development (HUD) promulgates noise standards (24 CFR Part 51, Subpart B) applicable to federally residential construction. These standards are widely used where federal involvement is not a factor to assess the significance of noise impacts in residential communities.

Under HUD standards, noise levels within residences should not exceed a day night average sound level (Ldn) of 45 dB (typically expressed as dBA). Because interior noise levels in typical residential construction are about 20 dBA below exterior levels, HUD standards classify sites where community exterior noise levels exceed 65 dB as noise-impacted areas and require additional sound attenuation to bring interior noise levels within the 45 dB standard.

A major source of noise in urban environments is from vehicles traveling on roads. Growth or changes in land use also can lead to additional traffic, and the potential for an increase in noise. Federal aid projects—transportation facilities receiving federal funding—are subject to federal noise guidelines. Washington State Department of Transportation (WSDOT) 2020 Traffic Noise Policy and Procedures (WSDOT 2020) are consistent with those of the Federal Highway Administration (FHWA) (23 Code of Federal Regulations 772) and have been approved by FHWA for use on federal-aid projects in Washington. FHWA guidelines state that noise abatement must be considered when a noise impact affects a particular land use or Activity Category. The FHWA Activity Categories B and C noise abatement criteria (NAC) of 67 dBA apply to residences

(single- and multi-family), places of worship, schools, recreation areas, and similar land use activities. **Exhibit 3.6-1** describes WSDOT's NAC by land use category. Other developed lands (e.g., hotels/motels, offices, restaurants/bars, or other developed lands) are included in Activity Category E, with a NAC of 72 dBA. FHWA determines a noise impact to occur when predicted future traffic noise levels "approach" or exceed the established FHWA NAC for a given Activity Category. WSDOT defines "approach" as within 1 dBA of the FHWA NAC (66 dBA for Activity Categories B and C or 71 dBA for Category E).

Activity Category	Leq(h)* (dBA) at Evaluation Location	Description of Activity Category
A	57 (exterior)	Lands on which serenity and quiet are of extraordinary significance and serve an important public need and where the preservation of those qualities is essential if the area is to continue to serve its intended purpose.
В	67 (exterior)	Residential (single and multi-family units)
С	67 (exterior)	Active sport areas, amphitheaters, auditoriums, campgrounds, cemeteries, day care centers, hospitals, libraries, medical facilities, parks, picnic areas, places of worship, playgrounds, public meeting rooms, public or nonprofit institutional structures, radio studios, recording studios, recreation areas, Section 4(f) sites, schools, television studios, trails, and trail crossings
D	52 (interior)	Auditoriums, day care centers, hospitals, libraries, medical facilities, places of worship, public meeting rooms, public or nonprofit institutional structures, radio studios, recording studios, schools, and television studios.
E	72 (exterior)	Hotels, motels, offices, restaurants/bars, and other developed lands, properties or activities not included in A-D or F. Includes undeveloped land permitted for these activities.
F		Agriculture, airports, bus yards, emergency services, industrial, logging, maintenance facilities, manufacturing, mining, rail yards, retail facilities, shipyards, utilities (water resources, water treatment, electrical), and warehousing
G		Undeveloped lands that are not permitted

Exhibit 3.6-1	WSDOT Noise Abatement Criteria by Land Use Category
---------------	---

Source: WSDOT, 2020.

Washington State

The State of Washington authorized the establishment of rules to abate and control noise pollution (Revised Code of Washington 70.107). The regulations (Washington Administrative Code (WAC) 173-60-040) establish maximum permissible noise levels (termed "Environmental Designation for Noise Abatement" or EDNA), which vary depending upon the land uses of the noise source and the receiving property. The maximum permissible noise level is the decibel

level of noise generated by the project as measured at the property line of adjacent land uses; it is not the combined noise of a project and background. Maximum Permissible Environmental Noise Levels apply to a variety of activities and facilities including residences, hospitals, commercial services, storage, warehouse and distribution facilities, and industrial property, and exempts electrical substations, certain industrial installations, mobile noise sources and vehicles traveling in public right of way, as well as safety warning devices (i.e., bells). The state provisions have been adopted by most cities around the state, including the City of Seattle (SMC 25.08).

City of Seattle

Operational Noise Standards

Chapter 25.08 of the Seattle Municipal Code (SMC) establishes exterior sound level limits for specified land use zones or "districts," which vary depending on the district of sound source and the district of the receiving property (see **Exhibit 3.6-2**).

	EDNA Receiver of Noise (Maximum Allowable Sound Level in dBA Leq				
EDNA Source of Noise	Residential	Commercial	Industrial		
Class A Residential	55	57	60		
Class B Commercial	57	60	65		
Class C Industrial	60	65	70		

Exhibit 3.6-2	Maximum Permissible Noise Levels: Seattle Noise Control Ordinance

Source: Seattle Municipal Code (SMC) 25.08.410.

Between the hours of 10pm and 7am on weekdays and 10pm and 9am during weekends, the maximum limits for receivers within residential zones are to be reduced by 10 dBA. For noises of short duration these limits can be exceeded by a maximum of 5 dBA for 15 minutes/hour, 10 dBA for 5 minutes/hour or 15 dBA for 1.5 minutes/hour.

Construction Noise Standards

The City's Noise Control code allows the exterior sound level limits to be exceeded by certain types of construction equipment operating in many commercial districts between 7 a.m. and 10 p.m. on weekdays and between 9 a.m. and 10 p.m. on weekends and legal holidays (SMC 25.08.425; see **Exhibit 3.6-3**). The types of equipment that would usually exceed the exterior sound level limit of 60 dBA are tractors, loaders, excavators, and cranes. This equipment may exceed the applicable standard by up to 25 dBA (an 85 dBA standard) when measured at a reference distance of 50 feet. Use of impact equipment, such as a pile driver, is restricted to 8 a.m. to 5 p.m. on weekdays and 9 a.m. to 5 p.m. on weekends and holidays and limited to a

continuous noise level of 90 dBA and a maximum noise level of 99 dBA Lmax when measured at a reference distance of 50 feet.





Source: City of Seattle, 2021.

Criteria for Increases in Noise Levels

As discussed in **Effects of Noise on People**, the following general relationships exist between noise levels and human perception:

- An increase in sound levels of 1- or 2-decibels is not perceptible to the average person.
- An increase in sounds levels of 3-decibels is just barely perceptible to the human ear.
- An increase in sounds levels of 5-decibels is readily perceptible to the human ear.
- An increase in sounds levels of 10-decibels is perceived as a doubling in loudness to the average person.

In addition, FHWA and WSDOT consider a traffic noise impact to occur if future predicted noise levels substantially exceed the existing noise levels. The WSDOT guidance indicates that a predicted future traffic noise level of 10 dBA or more above existing noise levels constitutes a substantial increase.

Current Conditions

Noise Sources in Seattle

For this analysis, the existing noise environments in the BINMIC and Greater Duwamish MIC are divided into several categories of noise sources: surface vehicle traffic; rail operations, including freight, light rail, and commuter trains; aircraft operations; commercial/industrial equipment; construction equipment and any other sources not associated with the transportation of people or goods.

<u>Traffic Noise</u>

Traffic noise exposure is primarily a function of the volume of vehicles per day, the speed of those vehicles, the number of those vehicles represented by medium and heavy trucks, the distribution of those vehicles during daytime and nighttime hours and the proximity of noise-sensitive receivers to the roadway. Existing traffic noise exposure adjacent to interstate highways is expected to be as high as 75 dB Ldn (FTA 2006). Bus transit can also make a meaningful contribution to roadway noise levels. Traffic noise assessment in this analysis is inclusive of bus transit, as buses are an assumed percentage of overall roadway volumes used in the calculation of roadside noise levels.

<u>Rail Noise</u>

MIC areas in Seattle are also affected by noise from freight and passenger rail operations. While these operations generate significant noise levels in the immediate vicinity of the railways, train operations are intermittent and area railways are widely dispersed. Commuter rail such as Sound Transit's light rail system operate with more frequency than standard gauge rail operations but electrification and lower speeds result in lower noise levels. The contribution of rail noise to the overall ambient noise environment in the Seattle MIC areas is relatively minor compared to other sources such as traffic. However, areas near train yards from assembling railcars into long trains and idling engines frequently experience high noise levels (FTA 2006).

<u>Aircraft Noise</u>

The King County International Airport, also known as Boeing Field, which generates approximately 500 aircraft operations a day. In addition to the numerous daily aircraft operations originating and terminating at Boeing Field, aircraft originating from other airports such as Seattle-Tacoma International Airport (Sea-Tac) frequently fly over Seattle. All of these operations contribute to the overall ambient noise environment. In general, like rail noise, the proximity of the receiver to the airport and aircraft flight path determines the noise exposure. Other contributing factors include the type of aircraft operated, altitude of the aircraft and atmospheric conditions. Atmospheric conditions may contribute to the direction of aircraft operations (flow) and affect aircraft noise propagation. **Exhibit 3.6-4** presents the noise contours for Boeing Field as of the 2010 (King County 2010).



Exhibit 3.6-4 Boeing Field Noise Exposure

Source: Herrera, 2021.

Construction Noise

Construction activities for new development and transportation improvements can create high noise levels of relatively short duration. Noise production from construction equipment varies depending on factors such as operation being performed and equipment type, model, age, and condition. Noise from heavy equipment diesel engine operations often dominates the noise environment in the vicinity of construction sites. Stationary sources such as generators, pumps and compressors may also produce a significant contribution. However, if present, operations from impact equipment (e.g., pile driving, pavement breaking) will generally produce the highest noise levels and may also produce significant vibration in the vicinity. Maximum noise exposure from typical construction equipment operations is approximately 75–100 dB (Lmax at 50 feet) with noise from heavy demolition and pile driving operations having the highest noise production.

Industry and Other Non-Transportation Noise

A wide variety of industrial and other non-transportation noise sources are located in Seattle MICs. These include manufacturing plants, marine shipping facilities, and other heavy and general industrial facilities, and others. Noise generated by these sources varies widely, but in many cases may be a significant contributor to a local noise environment.

For comparative purposes, a list of common A-weighted noise levels is shown in **Exhibit 3.6-5**. Decibel levels and common subjective responses to that sound level are also presented in the table. The table also depicts how persons commonly describe sound level differences of 10 dB as being twice as loud or half as loud.

Example	dBA	Qualitative Evaluations		
	140			
Threshold of Pain	135			
Jet Engine 200 feet	125	Destening		
	120	Dealerning	32	
Rock Band	115			3
Accelerating Motorcycle a few ft. away	110		16) dB/
Noisy Urban Street/Heavy City Traffic	100		8	re 7(
Jack Hammer at 50 feet	95	VorvLoud		pno
	90	very Loud	4	as L
Heavy Truck at 50 feet	85			mes
	80	Madarataly Laud		F
Vacuum Cleaner at 10 feet	75	would all y Loud		

Exhibit 3.6-5 Typical Sound Levels

Ch.3 Environment, Impacts, & Mitigation Measures - Noise

Example	dBA	Qualitative Evaluations	
Near freeway auto traffic	70		1
Business Office	60		1/2
	50	Quiet	1⁄4
	45		
Quiet urban nighttime	40		1⁄8
Soft whisper at 5 ft.	30	Faint	
	25		
Motion picture studio	20	Von (Epipt	
Human breathing	10	veryraint	
Threshold of human hearing	0		

Source: Herrera, 2021.

Existing Noise Levels in Seattle MIC Areas

This section presents current noise levels in the BINMIC and Greater Duwamish MIC study areas. Three methods are used to characterize existing noise levels in the Seattle MIC areas:

- A compilation of available noise data near the MICs was collected from publicly available documents to provide a sampling of noise environments near the areas of interest.
- Noise levels were measured at 8 locations in specific areas where the alternatives consider greater degrees of change and the potential for noise impacts is higher.
- A desktop survey using aerial photography, ArcGIS, and the City of Seattle Comprehensive Plan and zoning was used to determine locations of noise sensitive land uses in the MIC areas.

Existing noise levels are presented in **Exhibit 3.6-6** and **Exhibit 3.6-7** and the location of the measurements is presented in **Exhibit 3.6-8**. Measurements indicate that portions of the SODO/Stadium and Georgetown/South Park subareas exceed HUD's 65 dBA standard and would be classified as noise-impacted areas needing additional noise attenuation for residential structures.

These data show that ambient noise levels in maritime and industrial areas of the city (locations in **Exhibit 3.6-8**) are higher than other developed areas of the city. Larger traffic volumes on local roadways, rail and aircraft operations, and transit bus operations are largely responsible for this condition. Typical urban areas with low roadway volumes can regularly experience typical ambient noise levels below 50 dBA. Locations adjacent to freeways and highways can experience daytime ambient noise levels of 65–75 dBA, L50 (Caltrans 2009).
Exhibit 3.6-6	Ambient Noise Level	Data at Ecology/PSCAA	Seattle Monitoring Stations
---------------	---------------------	-----------------------	------------------------------------

	2018		20)19	20)20	2021	
Location	LEQ	LMax	LEQ	LMax	LEQ	LMax	LEQ	LMax
11-Beverly Park School	64.1	71.0	63.7	69.0	62.0	68.0	62.9	66.0
9-Beacon Hill	64.0	70.0	91.2	110.0	73.6	99.0	62.8	69.0
6-Hamilton Viewpoint Park	57.0	74.0	57.4	64.0	57.0	66.0	53.8	64.0
4b-Catherine Blaine School (Magnolia)	53.2	64.0	53.1	64.0	54.5	71.0	66.8	85.0

Sources: Port of Seattle, 2021 (https://www.portseattle.org/page/aircraft-noise-monitoring-system); Ramboll, 2021.

Exhibit 3.6-7 Sound Level Measurements (dBA) in the Seattle MIC Areas (2021)

	2021							
Location	24-Hour Day-Night Ldn	Hourly Leq Range— Daytime Hours	Hourly Leq Range— Nighttime Hours					
Ballard	62.5	55.6-66.7	47.4-60.2					
Interbay/Dravus	58.8	51.6-57.1	50.4–53.6					
Interbay/Armory	58.5	52.1-56.7	50.6-52.3					
Stadium	69.2	61.5-69.0	55.7-68.0					
Georgetown	68.1	62.8-67.6	55.2-66.0					
South Park 1	60.5	53.9–59.9	51.0-56.3					
SODO/Lander	67.8 ^a	57.4-72.2ª	53.1-61.2					
South Park 2	59.5	53.9-63.7	44.5-54.1					

^a At SODO/Lander location, sound levels during daytime hours between 7 a.m. and 2 p.m. were influenced by interference of a generator operating nearby. As estimate of the 24-hour Ldn sound level during this time period is approximately 65 dBA, approximately 3 dBA lower than presented in this table.

All measurements collected between August 23 and August 27, 2021. Source: Ramboll, 2021.



Exhibit 3.6-8 Noise Monitoring Locations

Noise-Sensitive Receptors

Noise-sensitive land uses are generally defined as locations where people reside or where the presence of unwanted sound could adversely affect the use of the land. Noise-sensitive land uses typically include residences, hospitals, schools, transient lodging, libraries, and certain types of recreational uses such as parks. Noise-sensitive residential receivers are found adjacent to many of the MIC areas. Residential areas of the Georgetown neighborhood occur within and surrounded by industrial land uses of the Greater Duwamish MIC.

Exhibit 3.6-9 shows the locations of the noise sensitive areas used in the analysis. The locations were selected because they are associated with areas of future land uses under one or more of the alternatives under consideration during Scoping, and would include a higher density of office workers, live/work uses, or in some cases, mixed-uses that may support residential and open space areas.

- 1. Ballard: 5007 14th Avenue Northwest. This site is also close to the future Sound Transit light rail station.
- 2. Interbay/Dravus: 3425 16th Avenue West. This is also close to a future Sound Transit light rail station, a BNSF rail yard, and facilities.
- 3. Interbay/Armory site: 1561 W Armory Way. This is a site that is close to the BNSF rail yard.
- 4. Stadium area: 1730 1st Avenue South
- 5. Georgetown: 5707 Airport Way South
- 6. South Park 1: 8620 16th Avenue South. An area close to the King County airport
- 7. SODO/Lander: 2437 6th Avenue South. An existing light rail station.
- 8. South Park 2: 8100 8th Avenue South. An area in proximity to SR 99 and SR 509.



Exhibit 3.6-9 Location of Potential Noise Sensitive Areas

3.6.2 Impacts

Impacts Common to All Alternatives

Ballard

Noise Sensitive Receivers

In the Ballard Subarea portion of the BINMIC, existing sensitive receivers potentially impacted by noise include residences and schools primarily adjacent to the north and east sides of the Ballard Subarea but also at scattered locations within the subarea.

Recreation sites and facilities in and adjacent to this subarea include: a boat ramp, wading pool, outdoor sports courts, play areas, sports fields, the Burke-Gilman multi-use trail, and Fremont Canal Park. As urban recreation facilities they are only moderately noise sensitive, and the likeliest adverse impacts would result from noise from nearby construction activities.

Construction

Construction activities associated with development or redevelopment of industrial, commercial, and residential land uses would result in temporary, localized increases in noise that could affect nearby sensitive receivers.

Construction activities most likely to lead to increased noise involve excavation and related site preparation, and construction of foundation and building structure and exterior. These activities typically involve the use of heavy on-site equipment. Construction is also typically associated with a temporary increase in truck traffic as material is brought to and from the construction site. As indicated earlier in this section construction activities typically generate noise levels of 75–100 dB (Lmax at 50 feet) with noise from heavy demolition and pile driving operations typically having the highest noise production.

As described earlier in this section, the City's Noise Control regulations (SMC 25.08) would serve to limit noise impacts from construction by restricting the times when construction activity can exceed standard noise limitations.

Vibration is a distinct noise-related effect resulting from some construction activities, such as pile-driving, that can adversely affect the integrity of nearby structures and cause annoyance to nearby residents and other sensitive receptors. The City has not adopted quantitative standards limiting vibration. Potential vibration impacts, where anticipated, would be assessed in project-specific environmental review documents.

Noise from Stationary Operations

Industrial activities include manufacturing plants, marine shipping facilities, and other heavy and general industrial facilities and other stationary activities and land uses that generate noise. These facilities could use outdoor loading docks and outdoor material storage areas that, unless properly designed and controlled, could also generate of noise in the surrounding community. Noise generated by these sources varies widely and are often periodic but can exceed 80 dBA close to the source for some activities (**Exhibit 3.6-5**) and can be a substantial contributor to localized levels of noise. Depending on location, new residential uses within the Ballard Subarea could experience noise impacts from stationary industrial operations. Such uses would be subject to the noise limits of SMC Chapter 25.08.

Traffic Noise

Under all alternatives, traffic volumes on roads in and near the Ballard Subarea are expected to increase due to expected development and associated population increase in the overall Seattle area. Roadways in the study area are expected to experience a relatively high volume of light and heavy trucks. **Exhibit 3.6-10** shows PM peak hour volumes for all alternatives at roadways adjacent to monitoring locations.

		PM Peak Hour Volume											
Geographic	Adjacent Model Roadway	2019 Existing		2042 No Action		2044 Alt. 2		2044 Alt. 3		2044 Alt. 4		2044 Pref. Alt.	
Area		NB/EB	SB/WB	NB/EB	SB/WB	NB/EB	SB/WB	NB/EB	SB/WB	NB/EB	SB/WB	<u>NB/EB</u>	<u>SB/WB</u>
Ballard	14th Ave NW	110	110	110	100	110	100	140	130	160	160	<u>130</u>	<u>120</u>
Interbay/Dravus	W Dravus St	1,180	1,220	1,150	1,210	1,240	1,230	1,390	1,260	1,410	1,260	<u>1,290</u>	<u>1,240</u>
Interbay/Armory	15th Ave NW	1,610	1,210	1,670	1,280	1,690	1,280	1,670	1,270	1,680	1,270	<u>1,680</u>	<u>1,280</u>
Stadium	1st Ave S	1,140	2,230	970	2,230	1,000	2,220	1,020	2,180	1,020	2,170	<u>1,000</u>	<u>2,190</u>
Georgetown	Airport Way S	510	1,580	760	1,590	780	1,600	820	1,620	830	1,650	<u>810</u>	<u>1,600</u>
South Park 1	14th Ave S	470	1,140	610	1,160	620	1,210	680	1,280	690	1,240	<u>620</u>	<u>1,190</u>
SODO/Lander	6th Ave S	250	320	230	720	230	720	230	700	250	720	<u>240</u>	<u>720</u>
South Park 2	8th Ave S	280	350	290	340	310	350	310	360	300	340	<u>310</u>	<u>350</u>

Exhibit 3.6-10 PM Peak Hour Traffic Volumes for Existing Conditions and All Alternatives

Source: Fehr and Peers, 20224.

These increased volumes would lead to very slight increases in roadway noise, if any. **Exhibit 3.6-11** shows estimated increases in modeled total noise exposure for all geographic areas for existing conditions, no action, and alternatives at the locations shown in **Exhibit 3.6-8**, above.

		Increase in dBA— PM Peak Hour Volume										
Geographic Area	Existing 24- Hour Day- Night Ldn	2042 No Action		2044	2044 Alt. 2		2044 Alt. 3		2044 Alt. 4		2044 Pref. Alt.	
		NB/EB	SB/WB	NB/EB	SB/WB	NB/EB	SB/WB	NB/EB	SB/WB	<u>NB/EB</u>	<u>SB/WB</u>	
Ballard	62.5	0.0	0.0	0.0	0.0	0.0	0.0	1.0	1.0	<u>0.0</u>	<u>0.0</u>	
Interbay/Dravus	59	0.0	0.0	1.0	0.0	2.0	0.0	2.0	0.0	<u>1.0</u>	<u>0.0</u>	
Interbay/Armory	59	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	<u>0.0</u>	<u>0.0</u>	
Stadium	69	-0.2	0.0	0.0	0.0	0.0	0.0	0.0	0.0	<u>0.0</u>	<u>0.0</u>	
Georgetown	68.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	<u>0.0</u>	<u>0.0</u>	
South Park 1	60.5	1.0	0.0	0.0	0.0	0.0	1.0	0.0	0.0	<u>0.0</u>	<u>0.0</u>	
SODO/Lander	67.8	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	<u>0.0</u>	<u>0.0</u>	
South Park 2	59.5	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	<u>0.0</u>	<u>0.0</u>	

Exhibit 3.6-11 Increase in dBA Over Existing Conditions, All Alternatives

Source: Fehr and Peers, 20221; Herrera, 20221.

Anticipated increases in traffic within the Ballard Subarea would be insufficient (less than 3 dBA) to generate noticeable increases in roadway noise compared to the existing condition. In addition, over the next several decades technology of vehicles, both car and truck, is likely to reduce average vehicle noise. The result of expected limited increases coupled with likely technology changes would be that existing noise-sensitive land uses adjacent to the Ballard Subarea are unlikely to be adversely affected under any of the alternatives. Roadway noise could, however, adversely impact residents of new residential development anticipated within the subarea, especially under alternatives 3 and 4 if the new residential development occurs adjacent to major arterials.

Land Use Compatibility

Land use compatibility issues can arise when noise-sensitive uses, especially residences, are located near to industrial activities or heavily travelled roadways that generate high levels of noise. A common exterior noise standard for residences is 65 Ldn noise level, because exterior noise at that level can be reduced to an interior level of 45 dBA Ldn (the accepted maximum interior noise level for residential uses) using standard construction techniques. In the Ballard Subarea, land use compatibility impacts contributed to by subarea noise could occur in residential areas adjacent to the periphery of the subarea or in new residential and associated uses anticipated to be developed in the northern portion of the subarea (**Exhibit 3.6-9**).

Interbay Dravus

Noise Sensitive Receivers

In the vicinity of the Interbay Dravus Subarea, existing sensitive receivers potentially impacted by noise include residences and schools located primarily on the flanking hillsides to the east and west of the subarea, but also including multi-family residences and outdoor recreation facilities on the valley floor adjacent to the subarea boundary.

Recreation sites and facilities in and adjacent to this subarea include: Interbay Athletic Fields and Interbay Golf Course and the Ship Canal multi-use trail. As urban recreation facilities they are only moderately noise sensitive, and the likeliest adverse impacts would result from noise from nearby construction activities.

Construction

Construction noise impacts would be similar to those described for the Ballard Subarea. However, much of this subarea is underlain by fill, and special foundation construction (for example, over-excavation, pile-driving) may be necessary for some development leading to comparatively greater levels and/or durations of some construction noise.

Noise from Stationary Operations

Noise impacts from stationary sources would be similar to those described for the Ballard Subarea. However, historically, rail operations at the Balmer Yard (previously the Interbay Yard) have generated noticeable periodic noise. Depending on location, new residential uses within the Interbay Dravus Subarea could experience noise impacts from stationary industrial operations.

Traffic Noise

In the Interbay Dravus Subarea, impacts from traffic noise would be similar to those described for the Ballard Subarea.

Land Use Compatibility

Land use compatibility issues in the Interbay Dravus Subarea would be similar to those described for the Ballard Subarea. In the Interbay Dravus Subarea, land use compatibility impacts could occur in residential areas adjacent to the periphery of the subarea or in new residential and associated uses anticipated to be developed within the subarea (**Exhibit 3.6-9**).

Interbay Smith Cove

Noise Sensitive Receivers

In the vicinity of the Interbay Smith Cove Subarea, existing sensitive receivers potentially impacted by noise include residences and schools located primarily on the flanking hillsides to the east and west of the subarea, but also including multi-family residences and outdoor recreation facilities on the valley floor adjacent to the subarea boundary.

Recreation sites and facilities in and adjacent to this subarea include: sports fields, the Elliott Bay multi-use trail, and Smith Cove and Elliott Bay parks. As urban recreation facilities they are only moderately noise sensitive, and the likeliest adverse impacts would result from noise from nearby construction activities.

Construction

Construction noise impacts would be similar to those described for the Ballard Subarea. However, much of this subarea is underlain by fill, and special foundation construction (for example, over-excavation, pile-driving) may be necessary for some development leading to comparatively greater levels and/or durations of some construction noise.

Noise from Stationary Operations

Noise impacts from stationary sources would be similar to those described for the Ballard Subarea. However, historically, rail operations at the Balmer Yard (previously the Interbay Yard) have generated noticeable periodic noise. Depending on location, new residential uses within the Interbay Smith Cove Subarea could experience noise impacts from stationary industrial operations.

Traffic Noise

In the Interbay Smith Cove Subarea, impacts from traffic noise would be similar to those described for the Ballard Subarea.

Land Use Compatibility

Land use compatibility issues in the Interbay Smith Cove Subarea would be similar to those described for the Ballard Subarea. In the Interbay Smith Cove Subarea, land use compatibility impacts could occur in residential areas adjacent to the periphery of the subarea or in new residential and associated uses anticipated to be developed within the subarea (**Exhibit 3.6-9**)

SODO/Stadium

Noise Sensitive Receivers

In the SODO/Stadium Subarea portion of the Greater Duwamish MIC, existing sensitive receivers potentially impacted by noise include residences just west of the subarea in West Seattle and residences adjacent to the north periphery of the subarea. The noise environment of the primarily residential Beacon Hill area east of the subarea is dominated by noise from Interstate-5, and this area is unlikely to be affected substantially by noise from subarea development.

Recreation sites and facilities in and adjacent to this subarea include: a community center, the West Seattle Bridge and Duwamish River multi-use trails, and the West Duwamish Greenbelt and Puget Park adjacent to the west side of the subarea. As urban recreation facilities they are only moderately noise sensitive, and the likeliest adverse impacts would result from noise from nearby construction activities.

Construction

Construction noise impacts would be similar to those described for the Ballard Subarea. As in the Interbay subarea, much of the SODO/Stadium Subarea is underlain by fill, and special foundation construction (for example, over-excavation, pile-driving) may be necessary for some development resulting in comparatively greater levels and/or durations of some construction noise.

Noise from Stationary Operations

In the SODO/Stadium Subarea, noise impacts from stationary sources would be similar to those described for the Ballard Subarea. Rail operations at the Argo Yard in the southeast portion of the subarea periodically generate high frequency and impulsive noise. Depending on location, new residential uses within SODO/Stadium Subarea could experience noise impacts from stationary industrial operations.

<u>Traffic Noise</u>

In the SODO/Stadium Subarea, impacts from traffic noise would be similar to those described for the Ballard Subarea.

Land Use Compatibility

Land use compatibility issues in the SODO/Stadium Subarea would be similar to those described for the Ballard Subarea. In the SODO/Stadium Subarea, land use compatibility impacts could occur in residential areas adjacent to the periphery of the subarea or in new residential and associated uses anticipated to be developed primarily in the vicinities of the stadiums and the SODO light rail station (**Exhibit 3.6-9**). Noise monitoring at locations within the subarea suggests that much of the subarea currently experiences noise levels above 65

dBA (Ldn) and new residential and associated noise sensitive development could be significantly adversely impacted by noise.

Georgetown/South Park

Noise Sensitive Receivers

In the Georgetown/South Park Subarea, existing sensitive receivers potentially impacted by noise include residences just west of the subarea in Delridge. The noise environment of the primarily residential Beacon Hill area east of the subarea is dominated by noise from Interstate-5 and Boeing Field (**Exhibit 3.6-4**), and this area is unlikely to be affected substantially by noise from subarea development.

Recreation sites and facilities in and adjacent to this subarea include: a hand launch site on the Duwamish River, a wading pool, a community center, outdoor sports courts, play areas, the Duwamish River and S Henderson Street multi-use trails, and the Duwamish Greenbelt and Puget Park adjacent to the west side of the subarea and Westcrest Park on the southeast. As urban recreation facilities they are only moderately noise sensitive, and the likeliest adverse impacts would result from noise from nearby construction activities.

Construction

Construction noise impacts would be similar to those described for the Ballard Subarea. Portions of the Georgetown/South Park Subarea are underlain by fill, and special foundation construction (for example, over-excavation, pile-driving) may be necessary for some development resulting in comparatively greater levels and/or durations of some construction noise.

Noise from Stationary Operations

In the Georgetown/South Park Subarea, noise impacts from stationary sources would be similar to those described for the Ballard Subarea. Depending on location, new residential uses within the Georgetown/South Park Subarea could experience noise impacts from stationary industrial operations.

Traffic and Aircraft Noise

In the Georgetown/South Park Subarea, impacts from traffic noise would be similar to those described for the Ballard Subarea. As shown in **Exhibit 3.6-4**, parts of the Georgetown portion of the subarea are within the Boeing Field Noise Exposure area. Areas of proposed Urban Industrial land within Georgetown lie within the exposure area, and new residential uses could experience relatively high levels of aircraft noise.

Land Use Compatibility

Land use compatibility issues in the Georgetown/South Park Subarea would be similar to those described for the Ballard Subarea. In the Georgetown/South Park Subarea, land use compatibility impacts could occur in residential areas adjacent to the periphery of the subarea or in new residential and associated uses anticipated to be developed primarily northwest of the north end of Boeing Field and in South Park (**Exhibit 3.6-9**). Noise monitoring at locations within the subarea suggests that the area northwest of Boeing Field may currently experience noise levels at or above 65 dBA (Ldn) and new residential and associated by noise.

Other Industrial Zoned Lands

Other industrial lands include areas within the Ballard Subarea, but outside of the BINMIC, along the north side of the Ship Canal from the east end of the Ballard Subarea to about the Interstate-5 bridge and a confined area along the southeast shore of Lake Union. Also, other industrial lands include two small areas within the SODO/Stadium neighborhoods, but outside of the Greater Duwamish MIC, located north and east of the north end of SODO/Stadium Subarea. The other industrial lands in the Ballard Subarea are adjacent to residential areas to the north and east and noise impacts would be similar to those described for Ballard Subarea are located adjacent to primarily commercial land uses which would not be noise sensitive and adverse noise impacts would be minimal.

Recreation sites and facilities in and adjacent to these other industrial lands include Gasworks Park at the north end of Lake Union. This is an urban recreation facility that is only moderately noise sensitive, and the likeliest adverse impact would result from noise from nearby construction activities.

Equity & Environmental Justice Considerations

Construction and increased activity under any of the alternatives has the potential to exacerbate residents' and workers' exposure to increased noise. Within the study area, the City has identified the SODO/Stadium and Georgetown/South Park subareas as having a Higher Disadvantage ranking in its Racial and Social Equity Index (City of Seattle 2017), and noise impacts to residents in those subareas may disproportionately affect low-income and minority communities.

Actual noise exposure is highly dependent on location, and in developing zoning boundaries to implement the selected industrial and maritime strategy, limiting proximity of new residential and associated development to high noise sources would limit exposure to excessive noise. In addition, noise reduction measures can be mandated for construction activities and adequate noise reduction measures also mandated for new residential construction, irrespective of market value, in high noise environments within industrial areas. These measures are addressed below under **Section 3.6.3**.

Impacts of Alternative 1 No Action

Based primarily on the expected increase in employment over the planning period, which is the lowest among the alternatives, the amount of construction and extent of new development would be less than other alternatives. Noise from stationary sources, which is typically dominated by heavy industrial exterior operations, would probably not increase substantially. As discussed above, expected traffic increases (see **Exhibit 3.6-10**) would be insufficient (less than 3dBA) to be noticeable (see **Exhibit 3.6-11**). Construction noise could adversely impact nearby locations, but impacts would be temporary and limited by the City's timing restrictions on construction activities. Overall, noise impacts should not be significant.

Impacts of Alternative 2

Based on the expected increase in employment and traffic over the planning period, noise increases would be similar to Alternative 1 in most locations, with minor increases in the Interbay Dravus Subarea; and similar to the other alternatives in most locations, but with less increase than Alternative 4 in the Interbay Dravus Subarea.

Impacts of Alternative 3

Based on the expected increase in employment and traffic over the planning period, noise increases would be similar to alternatives 1 and 2 in most locations, but slightly greater in the Interbay Dravus Subarea and portions of South Park; and similar to Alternative 4 but less than portions of South Park. Alternative 3 would include an expansion of housing allowances, which would expose new residents to potential noise impacts that could be significant without mitigation. Locations of new housing where residents would be particularly susceptible to adverse noise impacts include locations in proximity to one or more of the following: Interstate-5, active heavy or light rail lines, Boeing Field and its approach paths, and major activity centers. Locations having these characteristics occur in Interbay (heavy rail and future light rail); Stadium (Interstate-5, heavy and light rail, major activity center); SODO (Interstate-5, heavy and light rail); and Georgetown (Interstate-5, heavy rail, Boeing Field). In Georgetown, where the triangular area bounded by Corson Avenue S, Carleton Avenue S, and I-5 would be removed from the MIC and placed into a mixed-use zone and in the areas to be designated as Urban Industrial, existing or new residents may experience greater noise impacts resulting from nearby industrial and transportation activities, including the WSDOT Corson facility on Corson Avenue S. Some of these locations experience high existing ambient noise levels (Leq) up to about 69 dBA (Exhibit 3.6-7 Permissible EDNA levels are 60 dBA for residential receivers and up to 70 dBA for industrial receivers (SMC 25.08.410). However, some noise sources, for example rail and plane noise, are periodic and/or infrequent, and their contribution to hourly or daily noise metrics may not capture the extent to which their noise adversely affects noise-sensitive receptors.

Impacts of Alternative 4

Based on the expected increase in employment and traffic over the planning period, noise increases would be greater than alternatives 1 and 2 in the Ballard and Interbay Dravus subareas and less than some portions of South Park, but similar at all other locations. Alternative 4 would be similar to Alternative 3 in most locations, but greater in the Ballard Subarea <u>and in</u> <u>Georgetown where the triangular area bounded by Corson Avenue S, Carleton Avenue S, and I-5 would be removed from the MIC and placed into a mixed-use zone and in the areas to be designated as Urban Industrial. Alternative 4 impacts would likely be <u>and</u>-less <u>than Alternative 3</u> impacts in portions of South Park. Alternative 4 would include the largest expansion of housing allowances among the alternatives, which would expose the greatest number of new residents to potential noise impacts, the nature of which is discussed above under Alternative 3.</u>

Impacts of the Preferred Alternative

Based on the expected increase in employment and traffic over the planning period, noise increases overall would be similar or slightly greater than alternatives 1 and 2, and lower than alternatives 3 and 4. This pattern holds true for all locations, except SODO/Stadium and South Park and Georgetown where employment growth would be less than alternatives 2, 3, and 4 but still greater than Alternative 1 No Action. With the more concentrated amount of new housing in two new areas outside the MICs to mixed use zoning in west Ballard and Judkins Park, more housing would result with the Preferred Alternative compared to the Draft EIS alternatives in those locations, though total increases in industry supported housing and mixed use housing are below Alternative 4. However, because new housing would be targeted to reduce conflicts with existing and proposed industrial uses, noise impacts for these residents under the Preferred Alternative are expected to be less than alternatives 3 and 4. The Preferred Alternative would also remove focused land from the MIC in Georgetown and South Park as with the other alternatives, but also alter the zoning approach for the proposed mixed use zone in central Georgetown, and include more UI zoning around Georgetown. As a result, existing or new residents in these areas under the Preferred Alternative would likely experience less noise impacts resulting from nearby industrial and transportation activities, including the WSDOT Corson facility on Corson Avenue S, than under alternatives 3 and 4.

3.6.3 Mitigation Measures

Incorporated Plan Features

The recommendations of the Industrial and Maritime Strategy include the following features that relate to noise attenuation:

 Inclusion of circulation routes for non-motorized travel would reduce motorized traffic and associated noise.

- Incentivizing the use of transit and discouraging the use of single-occupancy vehicles would reduce overall traffic volumes and associated noise.
- Inclusion of green open spaces within Urban Industrial and Industry and Innovation districts would create greater separation between uses and decrease exterior noise levels.

Regulations & Commitments

City noise regulations establish exterior sound level limits for various land use zones with the limits varying depending on the source zone and the receiving zone (**Exhibit 3.6-2**). These limits are intended to result in acceptably low interior noise levels for residences and other sensitive noise receptors. City noise regulations also address construction noise, limiting the times during the day when construction noise, both impact and non-impact, can exceed exterior noise limits (**Exhibit 3.6-3**).

Other Potential Mitigation Measures

Zoning land use criteria or boundaries could be established, while meeting other planning goals, to limit the proximity of new residential development to known or anticipated sources of high noise levels.

To limit the impacts of temporary construction noise, in addition to restrictions on the hours of construction other mitigation that could be applied includes:

- installing barriers to shield noise sensitive receptors and enclosing stationary work
- selecting haul routes to avoid noise sensitive areas
- using alternative methods to pile-driving (e.g., hydraulic or vibration pile insertion or auguring/drilling holes for piles)
- using fully baffled compressors, or preferably electric compressors
- using fully mufflered construction equipment

Under alternatives 3 and 4<u>and the Preferred Alternative</u>, which would allow the development of new residential, the City could impose greater noise reduction standards in residential buildings where exterior noise levels greater than 65 dBA are likely to occur or where other uses occupying the same structure would likely contribute to excessive noise levels (above 45 dBA) within residences. These standards could include:

- installation of acoustically rated windows and doors that include high quality elastomeric caulking, multiple sashes, multiple panes, increased glass thickness, and increased airspace between glass panes
- installation of additional wall and attic/roof insulation
- installation of dampers and baffles on exterior vents, flues, and chimneys

Noise from tire-pavement interactions is the dominant contributor to roadway noise. A long-term mitigation program to reduce noise in noise-sensitive areas within the study area would

be to install noise reducing pavement on major arterials and roadways that experience relatively high traffic volumes and speeds.

The City and partner agencies could also improve coordination and improve the user experience for community members registering complaints or requesting information about enforcement related to noise from sites or businesses.

3.6.4 Significant Unavoidable Adverse Impacts

Under the studied alternatives, increased employment growth could result in increased traffic volumes, though the resulting noise increases are not anticipated to exceed 3dBA, the threshold of change that is perceptible. The location of noise sensitive receivers like residential uses near industrial or traffic noise sources could occur under all alternatives, particularly alternatives 3 and 4 and the Preferred Alternative. Implementation of residential noise mitigation described in the preceding subsection should adequately reduce noise experienced by noise sensitive receivers. With the application of mitigation measures described above, no significant unavoidable adverse noise impacts would occur under any of the alternatives.

Section 3.7 Light & Glare



This section discusses light and glare conditions in the study area and considers the impact of development under each of the alternatives on future conditions. The existing conditions and impacts analysis primarily use spatial data published by the City of Seattle, supplemented with King County and Federal sources.

Impacts of the alternatives on light and glare are considered significant if:

 Light and glare from new development has the potential to affect substantial numbers of residents, shoreline views, or protected scenic views (e.g., scenic routes, designated parks).

3.7.1 Affected Environment

This section discusses existing lighting and glare conditions in the study area, including major sources of exterior illumination and nearby high-sensitivity locations, such as residential areas, public open spaces, and scenic views.

Data & Methods

This section primarily uses spatial data published by the City of Seattle, supplemented with King County and Federal sources. Data sources include:

- City of Seattle Geographic Information Systems
 - City of Seattle 10-foot topographic contours (2016)
 - City of Seattle Parks and Trails inventory (2020)
 - City of Seattle Zoning (2021)
- King County Assessor
 - Existing land use property classifications (2020)
- National Aeronautics and Space Administration (NASA)
 - International Space Station nighttime light emission imagery of Seattle metropolitan area (2015)

Viewshed Calculation

To determine potential visibility areas, City-published elevation contours were processed using GIS software to create a digital elevation raster model of the city. The study area was then subdivided using a grid of 100-foot by 100-foot cells. The centroids of these equal-area cells were designated as "observer" points in the viewshed calculation. This created approximately 4,900 observer locations, equally distributed throughout the study area. To account for the visibility of buildings above ground level, each observer point was assigned an above-ground height offset based on the maximum structure height allowed in the applicable zoning district.

Lines of sight were calculated for each observer point and combined to generate a consolidated viewshed image that indicates relative visibility. Areas of the map highlighted as having high

visibility are visible from a greater number of observer points; lower visibility areas are visible from fewer observer points.

Nighttime Light Emission Mapping

Maps of nighttime lighting conditions used NASA orbital imagery captured by the International Space Station in 2015, the most recent year for which a nighttime image of Seattle was available. The image was reoriented and cropped using photo editing software and then georeferenced using GIS software. Due to image resolution limitations, the resulting maps are likely to contain a minor amount of spatial positioning error and are intended to illustrate relative brightness of nighttime light emissions across the city.

Current Policy & Regulatory Frameworks

Comprehensive Plan Goals & Policies

Seattle 2035, Seattle's comprehensive plan, establishes goals and policies related to urban design and aesthetics, including light and glare.

- Land Use Element Policy LU 5.14: Establish controls on the placement, direction, and maximum height of lighting and on the glare from reflective materials used on the exterior of structures in order to limit impacts on surrounding uses, enhance the character of the city, and encourage energy conservation.
- Eastlake Community Design Policy EL-P3: Anticipate and minimize, through zoning regulations and/or design review guidelines, to be prepared for the Eastlake area, the potential for impacts on residential uses from the close proximity, orientation, or incongruent scale of commercial development, including the loss of privacy, sunlight, or air, or increased noise, artificial light, or glare.

Seattle Municipal Code

SEPA Policies

The City of Seattle Municipal Code Chapter 25.05 codifies environmental policies and procedures. Section 25.05.675.K contains provisions related to light and glare.

- K. Light and glare
 - 1. Policy background
 - a. Development projects sometimes include lighting and/or reflective surface materials which can adversely affect motorists, pedestrians, and the surrounding area. Such adverse impacts may be mitigated by alternative lighting techniques and surface materials.
 - *b.* The City's Land Use Code specifically addresses the issue of light and glare control associated with commercial and industrial projects.

- 2. Policies.
 - *a.* It is the City's policy to minimize or prevent hazards and other adverse impacts created by light and glare.
 - b. If a proposed project may create adverse impacts due to light and glare, the decisionmaker shall assess the impacts and the need for mitigation.
 - c. Subject to the overview policy set forth in Section 25.05.665, the decisionmaker may condition or deny a proposed project to mitigate its adverse impacts due to light and glare.
 - d. Mitigating measures may include, but are not limited to:
 - 1) Limiting the reflective qualities of surface materials that can be used in the development;
 - 2) Limiting the area and intensity of illumination;
 - *3) Limiting the location or angle of illumination;*
 - 4) Limiting the hours of illumination; and
 - 5) Providing landscaping.

Seattle Municipal Code Section 25.05.675.P contains provisions related to public view protection.

- P. Public view protection
 - 1. Policy background
 - a. Seattle has a magnificent natural setting of greenery, mountains, and water; visual amenities and opportunities are an integral part of the City's environmental quality.
 - b. The City has developed particular sites for the public's enjoyment of views of mountains, water, and skyline and has many scenic routes and other public places where such views enhance one's experience.
 - c. Obstruction of public views may occur when a proposed structure is located in close proximity to the street property line, when development occurs on lots situated at the foot of a street that terminates or changes direction because of a shift in the street grid pattern, or when development along a street creates a continuous wall separating the street from the view.
 - d. Authority provided through Chapter 25.12 is intended to preserve sites and structures which reflect significant elements of the City's historic heritage and to designate and regulate such sites and structures as historic landmarks.
 - *e.* The Land Use Code provides for the preservation of specified view corridors through setback requirements.
 - *f.* The Land Use Code attempts to protect private views through height and bulk controls and other zoning regulations but it is impractical to protect private views through project-specific review.
- 2. Policies

a. 1) It is the City's policy to protect public views of significant natural and human-made features: Mount Rainer, the Olympic and Cascade Mountains, the downtown skyline, and major bodies of water including Puget Sound, Lake Washington, Lake Union and the Ship Canal, from public places consisting of the specified viewpoints, parks, scenic routes, and view corridors, identified in Attachment 1. (Attachment 1 is located at the end of this Section 25.05.675.) This subsection 25.05.675.P.2.a.i does not apply to the Space Needle, which is governed by subsection 25.05.675.P.2.c.

2) The decisionmaker may condition or deny a proposal to eliminate or reduce its adverse impacts on designated public views, whether or not the project meets the criteria of the overview policy set forth in Section 25.05.665; provided that downtown projects may be conditioned or denied only when public views from outside of downtown would be blocked as a result of a change in the street grid pattern.

b. 1) It is the City's policy to protect public views of historic landmarks designated by the Landmarks Preservation Board that, because of their prominence of location or contrasts of siting, age, or scale, are easily identifiable visual features of their neighborhood or the City and contribute to the distinctive quality or identity of their neighborhood or the City. This subsection does not apply to the Space Needle, which is governed by subsection 25.05.675.P.2.c.

2) A proposed project may be conditioned or denied to mitigate view impacts on historic landmarks, whether or not the project meets the criteria of the overview policy set forth in Section 25.05.665.

- c. It is the City's policy to protect public views of the Space Needle from the following public places. A proposed project may be conditioned or denied to protect such views, whether or not the project meets the criteria of the overview policy set forth in Section 25.05.665.
 - 1) Alki Beach Park (Duwamish Head)
 - 2) Bhy Kracke Park
 - 3) Gasworks Park
 - 4) Hamilton View Point
 - 5) Kerry Park
 - 6) Myrtle Edwards Park
 - 7) Olympic Sculpture Park
 - 8) Seacrest Park
 - 9) Seattle Center
 - 10) Volunteer Park

Designated scenic routes identified in SMC 25.05.675.P.2.a.1 are shown in **Exhibit 3.7-1** and **Exhibit 3.7-2**.



Exhibit 3.7-1 Seattle SEPA Scenic Routes Map—North

Source: Seattle Municipal Code Chapter 25.05 Subchapter VII—Attachment 1, 1987.





Source: Seattle Municipal Code Chapter 25.05 Subchapter VII—Attachment 1, 1987.

Development Standards

The Seattle Land Use Code (Seattle Municipal Code Title 23) contains development regulations for each of Seattle's zoning districts. These regulations establish light and glare standards for residential, commercial, and industrial zones that govern the design and placement of exterior site and building illumination, including effects on surrounding properties. As described in **Section 3.8 Land & Shoreline Use**, land in the study areas is primarily zoned Industrial; light and glare standards for Industrial Buffer (IB) or Industrial Commercial (IC) zones are established in SMC Chapter 23.50.046.

- A. Exterior lighting shall be shielded and directed away from lots in adjacent residential zones.
- *B.* Interior lighting in parking structures shall be shielded, to minimize nighttime glare affecting lots in adjacent residential zones.
- *C.* When nonconforming exterior lighting in an Industrial Buffer (IB) or Industrial Commercial (IC) zone is replaced, new lighting shall conform to the requirements of this section.
- *D. Glare diagrams which clearly identify potential adverse glare impacts on residential zones and on arterials shall be required when:*
 - 1. Any structure is proposed to have facades of reflective coated glass or other highly reflective material, and/or a new structure or expansion of an existing structure greater than sixty-five (65) feet in height is proposed to have more than thirty (30) percent of the facades comprised of clear or tinted glass; and
 - 2. The facade(s) surfaced or comprised of such materials either:
 - *a.* Are oriented towards and are less than two hundred (200) feet from any residential zone, and/or
 - b. Are oriented towards and are less than four hundred (400) feet from a major arterial with more than fifteen thousand (15,000) vehicle trips per day, according to Seattle Department of Transportation data.
- *E.* When glare diagrams are required, the Director may require modification of the plans to mitigate adverse impacts, using methods including but not limited to the following:
 - 1. Minimizing the percentage of exterior facade that is composed of glass;
 - 2. Using exterior glass of low reflectance;
 - *3. Tilting glass areas to prevent glare which could affect arterials, pedestrians or surrounding structures;*
 - 4. Alternating glass and nonglass materials on the exterior facade; and
 - 5. Changing the orientation of the structure.

Current Conditions

Full Study Area

As described in **Section 3.8 Land & Shoreline Use**, the study area consists primarily of industrially-zoned properties occupied by a variety of commercial and industrial uses. This style of development is often characterized by larger lot sizes and buildings than lower-intensity commercial or residential properties and a higher level of exterior building and site illumination.

Exhibit 3.7-3 shows nighttime illumination levels across Seattle, including the study area and adjacent neighborhoods. These visible light sources are a combination of streetlights, vehicles, and on-site exterior lighting. As shown on the map, nighttime illumination is brightest along major transportation corridors and in areas characterized by high-density commercial or industrial development, including Downtown, Uptown, the University District, Ballard, and the Greater Duwamish MIC. Adjacent residential neighborhoods appear darker by comparison, partially due to the lower level of lighting present and partially due to greater tree canopy presence, which can shield and screen light sources.

In general, the Greater Duwamish MIC (including the SODO/Stadium and Georgetown/South Park subareas) exhibits higher levels of light and glare than the Ballard and Interbay subareas. In particular, Harbor Island and the northwestern corner of the SODO/Stadium Subarea exhibit high levels of illumination comparable to the nearby Downtown core, with slightly lower levels of illumination present in the Georgetown/South Park Subarea to the south. The Ballard and Interbay subareas exhibit lower levels of light and glare, though still brighter than surrounding residential areas.

Exhibit 3.7-4 shows a topographic viewshed of the study area, based on City of Seattle 2016 elevation contours and maximum structure heights allowed by zoning. This viewshed provides an estimate of locations where portions of the study area are visible to observers and where light and glare generated by new and existing development could be perceived. The map also highlights locations that are likely to be highly sensitive to light and glare emissions; such locations include residential populations, scenic viewpoints, public parks and recreation areas, and open space and wildlife habitat areas. Major light sources and high-sensitivity locations in each subarea are described in more detail in the following sections.



Exhibit 3.7-3 Nighttime Illumination, 2015

Source: NASA, 2015; City of Seattle, 2021.



Exhibit 3.7-4 Industrial Subarea Viewshed, 2021

Source: City of Seattle, 2016. City of Seattle, 2021.

Ballard

Major Sources of Light & Glare

As shown in **Exhibit 3.7-5**, the Ballard Subarea occupies the northern shore of Salmon Bay and the Lake Washington Ship Canal. Much of the light and glare generated in the subarea comes from waterfront facilities, including docks and several small marinas, as well as the non-water oriented commercial/industrial area east of 15th Avenue NW. This area is characterized by small-scale commercial industrial properties, generally 1-2 stories in height. The area includes several breweries, multiple grocery stores and small-scale shopping centers, and limited large-format retail (Fred Meyer).

The Ballard Subarea also include several non-contiguous areas along the northern and eastern shores of Lake Union in Fremont and Eastlake, respectively. These areas consist primarily of docks and boat moorages; the Eastlake area includes a drydock, a seaplane dock, and several water-related industrial businesses.

High-Sensitivity Locations

Locations that are potentially sensitive to increases in light and glare associated with industrial development in the Ballard Subarea include the following:

- Burke-Gilman Trail: This major bicycle and pedestrian trail runs through the eastern portion of the main Ballard Subarea, as well as the non-contiguous portion of the subarea along the northern shore of Lake Union.
- **Gas Works Park:** One of Seattle's most popular parks, Gas Works Park provide approximately 19 acres of recreation opportunities and open space. The central hill offers views south to Downtown, as well as east and west along the ship canal.
- Ballard Locks: The Ballard Locks, one of Seattle's most popular tourist attractions, is located at the western end of the Ballard Subarea. The locks and their associated waterfront parks offer views eastward along the ship canal toward Lake Union, including the marine industry that lines the waterway.
- Ballard Avenue Landmark District: This historic district is home to a wide variety of hospitality, retail, office, and manufacturing uses and serves as an entertainment center for the Ballard neighborhood. The district is adjacent to the northern edge of the Ballard Subarea.



Exhibit 3.7-5 Nighttime Illumination—Ballard Interbay Northend MIC, 2015

Source: NASA, 2015; City of Seattle, 2021.

Interbay Dravus

Major Sources of Light & Glare

Industrial development in the Interbay Dravus Subarea consists of marine-related facilities along the south shore of Salmon Bay (Fisherman's Terminal and associated businesses) and railroad-related facilities generally located between W Emerson Place and W Dravus Street. The southern portion of the subarea, including the BNSF rail yard and the industrial development between the railroad and 15th Avenue W, is the primary source of light and glare; the area contains extensive on-site lighting and outdoor storage and parking areas, particularly along Thorndyke Avenue W. Several commercial businesses, including a grocery store and restaurant, as well as an apartment complex, also contribute to light generation in this portion of the subarea.

High-Sensitivity Locations

Locations that are potentially sensitive to increases in light and glare associated with industrial development in the Interbay Dravus Subarea include the following:

 Ballard Locks: The Ballard Locks, one of Seattle's most popular tourist attractions, is located at the western end of the Interbay Drave subarea. The locks and their associated waterfront parks offer views eastward along the ship canal toward Lake Union, including the marine industry that lines the waterway.

Interbay Smith Cove

Major Sources of Light & Glare

Primary light sources in the Interbay Smith Cove Subarea are concentrated in the western and southern portions of the study area. The Interbay rail yard forms the north-south spine of the subarea, with several large packing and shipping facilities located west of the rail yard. These facilities include large outdoor areas for loading, parking, and storage with extensive exterior lighting. The area east of the rail yard consists primarily of large-format commercial development, including a car wash, self-storage, a grocery store, a shopping center, and an Army National Guard facility with extensive outdoor storage. The Smith Cove Waterway, located south of the Magnolia Bridge, includes the Smith Cove Cruise Terminal (Pier 91) and several other port facilities. As shown in **Exhibit 3.7-5**, the cruise terminal and associated piers generate the highest levels of light and glare in the subarea.

High-Sensitivity Locations

Locations that are potentially sensitive to increases in light and glare associated with industrial development in the Interbay Smith Cove Subarea include the following:

• **Southeast Magnolia:** The southeast slope of Magnolia overlooks the Interbay rail yard and Smith Cove terminal. This area along Thorndyke Avenue W is characterized by a mix of

small-lot single family and moderate-density multifamily residential development, as well as the Magnolia Greenbelt, which occupies the steeply sloped hillside.

- Queen Anne South Slope: The south slope of Queen Anne Hill, above Uptown, is characterized by moderate to high-density urban housing and offers picturesque views of Downtown, Elliott Bay, Mount Rainier, and Harbor Island. In particular, two parks (Kinnear Park and Kerry Park) are popular with visitors and local photographers because of their exceptional views.
- Downtown Waterfront: Seattle's waterfront contains some of the city's most popular tourist attractions, such as the Seattle Aquarium, the Edgewater Hotel, Pier 66, and the Seattle Great Wheel, as well as lodging and restaurants. The waterfront provides visitors with panoramic views of southern Magnolia, Elliott Bay, Harbor Island, and West Seattle.

SODO/Stadium

Major Sources of Light & Glare

Due to the presence of extensive Port of Seattle facilities and associated private industrial development, the SODO/Stadium Subarea contains the most intense sources of light and glare in the study area, as shown in **Exhibit 3.7-6**. Harbor Island, located at the mouth of the Duwamish Waterway, and the surrounding facilities at Terminals 5, 25, 30, 37, 42, and 46, are characterized by large shipping facilities with extensive outdoor storage and staging areas. Compared with other portions of the study area, these locations include relatively few buildings; these facilities consist primarily of large open spaces where cargo can be staged and loaded, and the outdoor illumination necessary for operations generates large amounts of light and glare with few obstructions.

The portions of the SODO/Stadium Subarea east and south of the harbor also contribute to light and glare conditions, though to a lesser degree than the Harbor Island facilities. The industrial land use pattern in these areas consists of a mix of warehousing and manufacturing uses with large building footprints and limited outdoor storage or staging space.

High-Sensitivity Locations

Locations that are potentially sensitive to increases in light and glare associated with industrial development in the SODO/Stadium Subarea include the following:

 West Duwamish Greenbelt: Seattle's largest contiguous forest, the West Duwamish Greenbelt provides over 550 acres of recreation opportunities, open space, and wildlife habitat and runs roughly north-south along the western edge of both the SODO/Stadium and Georgetown/South Park subareas. The greenbelt provides a buffer between industrial development in the Greater Duwamish MIC and the residential neighborhoods of High Point and Delridge to the west.

- Magnolia Viewpoint: The Magnolia Viewpoint is a small park on the southwest side of Magnolia, along Magnolia Boulevard. This viewpoint offers unobstructed views of Downtown, Harbor Island, and West Seattle.
- Pigeon Point Neighborhood: This West Seattle residential neighborhood is located south of the West Seattle Bridge and west of West Marginal Way. The neighborhood occupies a hill overlooking Harbor Island and much of the northern Greater Duwamish MIC.
- West Seattle Viewpoints: Several parks and viewpoints in West Seattle offer scenic views looking eastward to Downtown, including Harbor Island and the Port of Seattle. Specific locations include:
 - Hamilton Viewpoint Park
 - Admiral Viewpoint
 - Northeast Alki Beach
- West Seattle—Harbor Avenue SW: Harbor Avenue SW runs along the northeastern edge of West Seattle between the West Seattle Bridge and Duwamish Head. The road runs along the shoreline, providing views of Elliott Bay and Downtown Seattle. The shoreline is the location of several public waterfront parks. As shown in Exhibit 3.7-2, this portion of Harbor Avenue SW is designated as a scenic route for purposes of SEPA under Seattle Municipal Code Chapter 25.05.675.P.2.a.1.
- Beacon Hill: This residential neighborhood occupies the eastern side of I-5 south of I-90. The north end of Beacon Hill overlooks both the Greater Duwamish MIC to the west and Downtown to the northwest. The neighborhood is separated from the Greater Duwamish MIC by the western slope of Beacon Hill and the I-5 corridor, but residences along the western edge of the hill have expansive views of the Duwamish Waterway, Elliott Bay, and West Seattle beyond. The central portion of Beacon Hill (south of Jefferson Park) has intermittent views of the Greater Duwamish MIC along the western edge of the neighborhood.
- Downtown Waterfront: Seattle's waterfront contains some of the city's most popular tourist attractions, such as the Seattle Aquarium, the Edgewater Hotel, Pier 66, and the Seattle Great Wheel, as well as lodging and restaurants. The waterfront provides visitors with panoramic views of southern Magnolia, Elliott Bay, Harbor Island, and West Seattle.



Exhibit 3.7-6 Nighttime Illumination—Greater Duwamish MIC, 2015

Source: NASA, 2015; City of Seattle, 2021.

Georgetown/South Park

Major Sources of Light & Glare

The Georgetown/South Park Subarea contains a mix of both large and small-scare industrial properties, as well as commercial and a small amount of residential development. The Duwamish Waterway divides the subarea, and the two sides differ in development pattern and intensity. The west side of the waterway features generally smaller lots with limited outdoor storage space or exterior illumination. The east side of the waterway features larger lots and buildings and more outdoor space for parking and storage. The eastern edge of the subarea is also the location of the King County International Airport (Boeing Field) and associated aviation-related industries. As shown on **Exhibit 3.7-6**, the airport runways themselves contribute very little illumination, but the adjacent terminals, hangars, and aircraft tie-down areas generate substantial light emissions.

High-Sensitivity Locations

Locations that are potentially sensitive to increases in light and glare associated with industrial development in the Georgetown/South Park Subarea include the following:

- West Duwamish Greenbelt: Seattle's largest contiguous forest, the West Duwamish Greenbelt provides over 550 acres of recreation opportunities, open space, and wildlife habitat and runs roughly north-south along the western edge of both the Georgetown/South Park and SODO/Stadium subareas. The greenbelt provides a buffer between industrial development in the Greater Duwamish MIC and the residential neighborhoods of High Point and Delridge to the west.
- South Park Neighborhood: The residential South Park neighborhood abuts the southern edge of the Georgetown/South Park Subarea on the west side of the Duwamish Waterway. The area features primarily moderate-density single-family and low-density attached housing, along with several parks and playgrounds, a school, and a branch of the Seattle Public Library. The neighborhood is bound on all sides by either a state highway, industrial development, or the Duwamish Waterway.
- South Beacon Hill: This residential neighborhood is located across I-5 from Boeing Field. The more southerly portions of the neighborhood are screened from views of the airport and MIC by vegetation, but the more northerly areas (north of S Kenyon Street) have little vegetation screening along the western periphery.

3.7.2 Impacts

The threshold of significance utilized in this impact analysis is as follows:

 Light and glare from new development that has the potential to affect substantial numbers of residents, shoreline views, or protected scenic views (e.g., scenic routes, designated parks).

Impacts Common to All Alternatives

Light and glare impacts associated with development depend on a variety of factors, including the type of development proposed, outdoor illumination needs of the specific uses proposed, elevation of the development site relative to surrounding areas, the density and size of on-site vegetation, and the architectural and site design characteristics of the structures and lighting elements specific to the development site. This combination of factors makes predicting potential impacts at an area-wide, programmatic scale challenging.

As described in **Chapter 2**, the proposed alternatives employ a combination of either existing land use designations (No Action Alternative) or new land use concepts (alternatives 2, 3, and 4 <u>and the Preferred Alternative</u>). Though development on individual sites may vary, these land use concepts define a baseline development typology for industrial development in the areas where they are applied, including factors such as allowed building size and height, allowed land use mix, and architectural and landscaping design requirements. The following impact analysis evaluates the potential light and glare impacts associated with each of the proposed land use concepts at the programmatic level, followed by analysis of the individual subareas under each of the alternatives.

Light & Glare Effects of Proposed Land Use Concepts

Maritime, Manufacturing, & Logistics (MML)

Overall, light and glare conditions on sites designated Maritime, Manufacturing, and Logistics (MML) would be similar in nature to existing industrial areas, though the intensity of light emissions would depend on specific site characteristics. Similar to existing General Industrial zones, the MML land use concept is focused on traditional industrial and manufacturing uses, as well as shipping, logistics, and port facilities. As illustrated in **Exhibit 2.4-1** and **Exhibit 2.4-4**, development patterns will be similar to existing industrial areas, characterized by large parcels, substantial outdoor storage and staging areas, and relatively low building heights.

Light and glare impacts associated with this land use concept are likely to be similar to existing heavy manufacturing and port-related industrial development typologies, extensive examples of which can be seen in the Greater Duwamish MIC. Major sources of light and glare associated with this land use concept would include outdoor illumination at storage yards and cargo staging areas. Manufacturing facilities that use exterior lights for operations and safety during nighttime hours would also be sources of light and glare. The MML land use concept would include zoning requirements for streetscape improvements, but on-site vegetation is anticipated to be sparse due to the intensive nature of development and the operational needs of shipping and logistics facilities, which are the primary anticipated uses. This lack of on-site vegetation would result in minimal screening of light sources. Similar to existing industrial development, the magnitude of light and glare impacts would depend on the specific design of on-site facilities and the proximity of high-sensitivity locations.

Industry & Innovation (II)

The Industry and Innovation (II) land use concept promotes higher-density industrial uses, including mixed-use development, as illustrated in **Exhibit 2.4-2** and **Exhibit 2.4-4**. Areas designated II are intended to be employment centers integrated with the high-capacity transit network. As such, the II land use concept is focused on a mix of uses that incorporates contemporary industrial methods and creates opportunities for combining light industrial and technology-oriented uses with associated office space. Compared to existing industrial areas, the II concept would exhibit taller building heights (up to 160 feet, including bonuses) and greater development density with fewer outdoor storage and/or staging areas. The integration of transit and bicycle/pedestrian connections would also result in fewer large parking areas.

Light and glare impacts associated with this land use concept are anticipated to be more similar to a commercial or mixed-use district than existing industrial areas. Without extensive outdoor areas requiring night-time lighting, exterior building illumination would be less intense, though taller allowable building heights could make buildings visible from farther away, depending on location and relative elevation.

Urban Industrial (UI)

The Urban Industrial (UI) land use concept focuses on a mix of smaller-scale industrial uses (such as fabrication shops, artist and maker spaces, and light industry) and limited nonindustrial uses, such as retail, offices, or industry-supportive housing. These areas would also include bicycle and pedestrian transportation facilities, and landscaped open spaces to promote environmental health. UI areas would be designed to include flexibility of uses and development standards that promote compatibility with nearby residential uses. See **Exhibit 2.4-3** and **Exhibit 2.4-4**.

Development in UI areas is anticipated to generate relatively lower light emissions compared to existing industrial typologies and the proposed MML and II land use concepts, due to the smaller scale of development and a greater emphasis on vegetation and green space, which can screen exterior illumination from surrounding areas. The UI land use concept would allow building heights up to 75 feet, which would represent a height increase in some industrial areas. Though less pronounced than potential height increases under the II land use concept, taller building heights may result in development being visible from farther away than current conditions, depending on location and relative elevation.
Equity & Environmental Justice Considerations

This EIS recognizes that impacts associated with industrial development, including exposure to light and glare emissions, are location-dependent and not equally distributed throughout the city. Due to market forces, historical practices regarding siting of industrial facilities, and historical restrictions on housing for people of color, residential areas near industrial centers are often home to communities of color and lower-income populations. The following impact analysis examines the potential for the alternatives to adversely affect residential populations, public spaces, and park and recreation facilities through exposure to increased light and glare emissions. The analysis also identifies instances where such impacts are likely to specifically affect vulnerable populations.

Impacts of Alternative 1 No Action

The No Action Alternative would preserve existing zoning and development regulations, resulting in future industrial development patterns similar to existing conditions. The No Action Alternative is anticipated to produce up to 11.23 million square feet of new employment-generating building space. Light and glare impacts associated with such development would be similar in nature to existing conditions, though the additional anticipated growth would increase overall light emissions as development occurs. **Exhibit 3.7-7** shows the viewshed and industrial zoning in the study area under the No Action Alternative.

Under the No Action Alternative, future industrial growth would generate additional light and glare emissions that could be perceived by non-industrial areas surrounding the study area, including high-sensitivity locations described in **Section 3.7.1 Affected Environment**. The following sections describe potential location-specific impacts.



Exhibit 3.7-7 Land Use Concepts Viewshed—Alternative 1

Ballard

Anticipated industrial development in the BINMIC would generate additional light and glare emissions that could be perceived by surrounding non-industrial areas (see **Exhibit 3.7-7** and **Exhibit 3.7-8**). The high-sensitivity areas primarily affected in the Ballard Subarea would include the western portions of the Burke-Gilman Trail and the Ballard Locks due to their close proximity to industrial development; the Ballard Locks would potentially be impacted by light and glare emissions from both the Ballard and Magnolia sides of the ship canal. However, use of these park and trail facilities is relatively low during nighttime hours, when light and glare emissions would be most evident.

Increased light and glare emissions from the BINMIC would potentially be visible to nonindustrial areas north of the Ballard Subarea, including the Ballard Avenue Landmark District. The landmark district itself is unlikely to experience significant impacts due to its location in the commercial center of Ballard, where nighttime illumination is already extensively used, though the portion of the district closest to industrial uses along Shilshole Avenue could experience impacts from the more intense lighting on industrial properties. Residential neighborhoods to the north at higher elevations could potentially observe the increased light and glare, though the effect would be attenuated with distance.

Industrial development at the eastern end of the Ballard Subarea could also potentially increase light and glare emissions observed at Gas Works Park, though potential increases in exposure at this location are likely to be reduced relative to other portions of the Ballard Subarea due to the smaller amount of adjacent industrial land. Likewise, the Eastlake portion of the Ballard Subarea is likely to experience minimal impacts; visibility of other industrial lands is relatively low, and the major concentrations of new industrial development in the BINMIC and Greater Duwamish MIC are screened by topography.



Exhibit 3.7-8 Ballard, Interbay Dravus, and Interbay Smith Cove Viewshed—Alternative 1

Interbay Dravus

Additional light and glare associated with new development in the Interbay Dravus Subarea would primarily be visible on immediately adjacent properties and along the Ballard waterfront, due to topography screening by nearby Magnolia and Queen Anne hillsides (see **Exhibit 3.7-7** and **Exhibit 3.7-8**). As described above, development in Interbay Dravus would contribute to light emissions observed at the Ballard Locks, which could potentially be impacted by light and glare emissions from both the Ballard and Magnolia sides of the ship canal. As described in **Chapter 2**, Interbay Dravus is anticipated to receive the smallest share of future employment growth under the No Action Alternative, so the increase in light and glare emissions is likely represent only an incremental increase compared to existing conditions.

Interbay Smith Cove

As shown in **Exhibit 3.7-7** and **Exhibit 3.7-8**, additional light and glare emissions in Interbay Smith Cove would primarily affect Southeast Magnolia and the South Slope of Queen Anne. As described in **Section 3.7.1 Affected Environment**, these areas contain a mix of single- and multi-family housing, exposing local residential populations to increased light and glare during nighttime hours. As described in **Chapter 2**, Interbay Smith Cove is anticipated to receive the second smallest share of future employment growth under the No Action Alternative (slightly more than Interbay Dravus), so the increase in light and glare emissions is likely represent a minor increase compared to existing conditions. Though minor, these increased light and glare emissions would be visible to a larger population than the northern portion of the Interbay corridor.

SODO/Stadium

As described in **Section 3.7.1 Affected Environment**, the SODO/Stadium Subarea is the largest and most intensely developed industrial area, and it produces the highest levels of light and glare emissions, due to the presence of the Port of Seattle and associated private industrial facilities. As shown in **Exhibit 3.7-7** and **Exhibit 3.7-9**, light and glare emissions from this study area have wide visibility, including residential areas in Beacon Hill and West Seattle (Pigeon Point, Alki) and public spaces in West Seattle (West Duwamish Greenbelt, Hamilton Viewpoint Park<u>Alki Beach, Harbor Avenue SW</u>), Downtown, and Magnolia. Under the No Action Alternative, the SODO/Stadium Subarea would absorb the greatest share of future employment growth, generating additional light and glare emissions as development occurs.

Increased light and glare under the No Action Alternative would be most perceptible to nearby residential areas in Pigeon Point and Beacon Hill due to their close proximity and higher elevation relative to the study area. Because future development would include a similar mix of industrial uses and facility types as existing conditions, the increase in light and glare emissions may not be perceptible at greater distances, such as Downtown or south Magnolia.

Light and glare emissions would also be visible from the West Duwamish Greenbelt, which runs along the western edge of the Greater Duwamish MIC. Those recreational use of the greenbelt occurs primarily during daylight hours when light and glare emissions are least perceptible, the greenbelt also include wildlife habitat areas that could be affected by nighttime light and glare. An analysis of potential impacts of the proposal on wildlife is contained in **Section 3.4 Plants & Animals**.



Exhibit 3.7-9 SODO/Stadium Viewshed—Alternative 1

Georgetown/South Park

As shown in **Exhibit 3.7-7** and **Exhibit 3.7-10**, light and glare emissions from the Georgetown/South Park Subarea would primarily affect South Beacon Hill, the South Park neighborhood, and the West Duwamish Greenbelt. South Park and South Beacon Hill are primarily residential areas and generally exhibit lower household incomes and higher populations of persons of color than other areas of Seattle. Increased light and glare emissions would be particularly visible in South Park, which is surrounded on three sides by portions of the Georgetown/South Park Subarea.



Exhibit 3.7-10 Georgetown/South Park Viewshed—Alternative 1

Impacts of Alternative 2

Alternative 2 applies the proposed land use concepts with relatively less Industry and Innovation and Urban Industrial land use than the other two Action Alternatives; the bulk of industrial land would be classified as MML, which would allow a mix of industrial uses and building typologies similar to the existing Industrial General zone. Sources of light and glare emissions would consist primarily of outdoor illumination for streets, storage and staging areas, as well as exterior operations and safety lighting for shipping and manufacturing facilities. Of the three Action Alternatives, Alternative 2 is the most similar to the No Action Alternative in terms of development type and distribution of light and glare sources and effects. **Exhibit 3.7-11** shows the land use concepts and potential viewshed for Alternative 2.

As described in **Chapter 2**, Alternative 2 is anticipated to produce up to 19.8 million square feet of new employment-generating building space. Overall light and glare emissions, though similar in nature and distribution to the No Action Alternative, are anticipated to be greater in intensity due to more extensive development of the study area. The following sections describe potential location-specific impacts and how the alternative differs from the No Action Alternative.



Exhibit 3.7-11 Increase in Viewshed—Alternative 2

Ballard

Light and glare emissions under Alternative 2 would be similar to the No Action Alternative; the majority of the study area would be classified MML, which would produce development types and lighting conditions similar to existing Industrial General zoning. Alternative 2 would incorporate the Urban Industrial and Industry & Innovation land use concept on the edges of the Ballard Subarea to serve as transition zones between MML areas and surrounding non-industrial development, as shown in **Exhibit 3.7-11** and **Exhibit 3.7-12**. These areas would generally allow greater building heights than current zoning, particularly in the II area on the northern edge of the subarea, where building heights could reach up to 160 feet. As described for the No Action Alternative, these increased heights would increase visibility of new buildings for residential areas to the north.

Though the II and UI areas would increase visibility of new buildings, development typologies in these areas would include fewer outdoor storage and staging areas, resulting in less use of intense exterior nighttime lighting, which would reduce light and glare emissions compared to the No Action Alternative. In particular, application of the UI land use concept to the area around the Ballard Avenue Landmark District would provide a buffer from more intense lighting conditions along the waterfront to the south.

Alternative 2 would implement the Industry & Innovation land use concept in the eastern portion of the Ballard Subarea, near Gas Works Park. Greater building heights would make this development more visible to the residential neighborhoods to the north, as well as from Lake Union itself. However, as described above, this land use concept places less emphasis on outdoor operations, reducing site lighting needs and resulting in reduced light and glare emissions compared to the No Action Alternative.

Exhibit 3.7-12 Increase in Viewshed (Ballard, Interbay Dravus, and Interbay Smith Cove)— Alternative 2



Interbay Dravus

Under Alternative 2, the Interbay Dravus Subarea would consist primarily of the MML land use concept, which would allow similar development types and intensities as the No Action Alternative, resulting in similar light and glare emissions and effects on high-sensitivity locations, such as the Ballard Locks. See **Exhibit 3.7-11** and **Exhibit 3.7-12**.

Interbay Smith Cove

Under Alternative 2, the southeastern portion of the Interbay Smith Cove Subarea, currently zoned Industrial Commercial, would be converted to Industry & Innovation (see **Exhibit 3.7-11** and **Exhibit 3.7-12**). The Industry & Innovation land use concept would promote greater development density and a wider mix of office and commercial uses than the current Industrial Commercial zone. With fewer outdoor storage and operations areas, light emissions would generally be reduced in this area compared to the No Action Alternative. However, the II land use concept would allow a substantial increase in building heights, resulting in greater visibility to surrounding areas, particularly Southeast Magnolia and the South Slope of Queen Anne.

SODO/Stadium

Similar to the No Action Alternative, the SODO/Stadium Subarea would absorb the greatest share of future employment growth under Alternative 2, generating additional light and glare emissions as development occurs. Most of the study area would be designated MML, resulting in similar building types and lighting features as under the No Action alternative. As shown in **Exhibit 3.7-11** and **Exhibit 3.7-13**, Alternative 2 would introduce the Urban Industrial land use concept in targeted locations on the edge of the Greater Duwamish MIC to create transition areas to surrounding neighborhoods (i.e., Pigeon Point and the Stadium District). In the area surrounding the stadiums, this would result in a slight increase in maximum building heights, increasing the visibility of development, but light emissions from the UI land use concepts are anticipated to be lower than MML or existing industrial zones. In the areas adjacent to Pigeon Point, application of the UI land use concept would implement lower building heights and reduce light and glare emissions on surrounding residential areas.

Alternative 2 would also implement the Industry & Innovation land use concept in the northern portion of subarea, near the stadiums and the I-5/I-90 interchange. As previously described, increased building heights would make development in these areas more visible, but light emissions are anticipated to be lower compared to the No Action Alternative.



Exhibit 3.7-13 Increase in Viewshed (SODO/Stadium)—Alternative 2

Georgetown/South Park

Alternative 2 would apply the Urban Industrial land use concept in most portions of the Georgetown/South Park Subarea currently zoned Industrial Buffer, providing a transition space to areas not within the Greater Duwamish MIC boundary. As described previously, this would slightly increase building heights and visibility of development in these locations, though the proposed land use mix of the UI designation would generate less intense light and glare emissions than the No Action Alternative. In particular, the South Park neighborhood is likely to experience reduced light and glare exposure compared to the No Action Alternative. See **Exhibit 3.7-11** and **Exhibit 3.7-14**.



Exhibit 3.7-14 Increase in Viewshed (Georgetown/South Park)—Alternative 2

Impacts of Alternative 3

As described in **Chapter 2**, Alternative 3 would apply the proposed land use concepts with a greater share of Industry & Innovation and Urban Industrial than Alternative 2. **Exhibit 3.7-15** shows the land use concepts and potential viewshed for Alternative 3.

As discussed previously, the building typologies and land use mix allowed under these land use concepts would generally reduce light and glare emissions from those areas due to a reduced focus on large-scale outdoor operations that require extensive lighting. However, Alternative 3 is anticipated to produce up to 27.4 million square feet of new employment-generating building space, and overall light and glare emissions from future development is likely to be greater than both the No Action Alternative and Alternative 2. Potential location-specific impacts are described in the following sections.





Ballard

As shown in **Exhibit 3.7-15** and **Exhibit 3.7-16**, Alternative 3 would implement the Urban Industrial land use concept more widely in the Ballard Subarea, specifically in the areas north of NW Leary Way and NW Market Street. Compared to Alternative 2, this change would increase building heights in this area (except for the small area designated Industry & Innovation under Alternative 2), though it would reduce light and glare emissions. This would create a transition zone between the MML area along the waterfront and reduce impacts on residential areas north of the subarea.

In the eastern portion of the subarea near Gas Works Park, the areas designated MML under Alternative 2 would be designated UI under Alternative 3. As described above, this would increase building heights and visibility of development, but it would result in lower light and glare emissions, reducing impacts on residential areas to the north, as well as the Burke-Gilman Trail, which travels through the area.

Exhibit 3.7-16 Increase in Viewshed (Ballard, Interbay Dravus, and Interbay Smith Cove)— Alternative 3



Interbay Dravus

As shown in **Exhibit 3.7-15** and **Exhibit 3.7-16**, Alternative 3 would implement the same land use concept pattern in the Interbay Dravus Subarea as Alternative 2, resulting in similar light and glare impacts.

Interbay Smith Cove

Alternative 3 would implement the same land use concept pattern in Interbay Smith Cove as Alternative 2, with the exception of the southwest slope of Queen Anne, where Alternative 3 would implement a greater amount of Urban Industrial instead of Industry & Innovation. Light and glare impacts in Interbay Smith Cover under Alternative 3 are therefore anticipated to be similar to, or less than, Alternative 2. See **Exhibit 3.7-15** and **Exhibit 3.7-16**.

SODO/Stadium

As shown in **Exhibit 3.7-15** and **Exhibit 3.7-17**, Alternative 3 would implement the same land use concept pattern in the SODO/Stadium Subarea as Alternative 2, with the exception of a larger node of Industry & Innovation south of S Holgate Street. Compared to Alternative 2, this change would result in a slight increase in visibility due to taller building heights in this location, though light and glare emissions would be less than the surrounding MML land use. As such, light and glare impacts in the SODO/Stadium Subarea under Alternative 3 are anticipated to be similar to, or less than, Alternative 2.



Exhibit 3.7-17 Increase in Viewshed (SODO/Stadium)—Alternative 3

Georgetown/South Park

Alternative 3 would apply a similar land use concept pattern in the Georgetown/South Park Subarea as Alternative 2 with the following changes:

- Removal of three targeted areas from the Greater Duwamish MIC, shown on Exhibit 3.7-15 and Exhibit 3.7-18 as Seattle Mixed:
 - One area approximately bounded by Corson Avenue S, <u>Carleton Avenue SS Michigan</u> <u>Street</u>, and Airport Way S. Removal of this area from the MIC would result in future development of this location for commercial and multifamily residential uses instead of industrial facilities. Light and glare emissions would be reduced compared to the MML land use proposed under Alternative 2, which would reduce potential impacts on the nearby Georgetown Playfield and Spraypark, located across Corson Avenue from the removal area.
 - Two areas adjacent to the South Park Neighborhood along the Duwamish Waterway. Removal of these areas and rezoning to Seattle Mixed would affect the uses allowed, but the building typologies and scale of development would be similar to the Urban Industrial land use concept proposed under Alternative 2, resulting in similar light and glare emissions.
- Designation of the eastern side of Ellis Avenue S north of S Myrtle Street as MML instead of UI. The use mix and building typologies allowed by the MML land use concept would potentially generate greater light and glare emissions than the UI land use concept proposed under Alternative 2. These impacts would primarily affect existing residential uses west of Ellis Avenue S, which are not included in the MIC.



Exhibit 3.7-18 Increase in Viewshed (Georgetown/South Park)—Alternative 3

Impacts of Alternative 4

As described in **Chapter 2**, Alternative 4 would implement a land use pattern similar to alternatives 2 and 3, but with a greater share of Industry & Innovation and Urban Industrial than Alternative 2. Compared to Alternative 3, Alternative 4 would have slightly higher shares of Maritime, Manufacturing, & Logistics and Industry & Innovation, and a lower share of Urban Industrial. **Exhibit 3.7-19** shows the land use concepts and potential viewshed for Alternative 4.

As described in **Chapter 2**, Alternative 4 would produce up to 27.8 million square feet of employment-generating building space, the highest of the four alternatives. Of the four alternatives, Alternative 4 also proposed the most extensive use of the Industry & Innovation land use concept, which would allow building heights up to 160 feet. As described in the introduction to this chapter, these increased heights would increase the visibility of industrial development to surrounding areas, though the building typologies allowed would likely generate less light and glare emissions due to less focus on outdoor operation and storage areas that require extensive outdoor lighting. Overall light and glare emissions are anticipated to be similar to or slightly higher than Alternative 3 due to the higher overall developed square footage and slightly greater share of land designated MML. Potential location-specific impacts are described in the following sections.



Exhibit 3.7-19 Increase in Viewshed—Alternative 4

Ballard

In the Ballard Subarea, Alternative 4 would implement two areas of Industry & Innovation on the north side of the subarea, as shown in **Exhibit 3.7-19** and **Exhibit 3.7-20**:

- At the northwest corner of the subarea, along NW Market Street; and
- North of NW Market Street on either side of 14th Avenue NW.

Implementation of the Industry & Innovation land use concept would allow increased building heights up to 160 feet, increasing the visibility of development to surrounding residential areas, particularly neighborhoods north of the subarea, which are located at higher elevations. While development under the II land use concept would generate lower light and glare emissions compared to the MML development proposed for these locations under Alternative 2, the increased height would expose a greater number of residents to light and glare effects than under alternatives 2 and 3.

Other portions of the subarea would implement the same land use concept pattern as Alternative 3 and would generate the same potential impacts.

Exhibit 3.7-20 Increase in Viewshed (Ballard, Interbay Dravus, and Interbay Smith Cove)— Alternative 4



Interbay Dravus

Alternative 4 would implement the same land use concept pattern in the Interbay Dravus Subarea as alternatives 2 and 3, resulting in similar light and glare impacts. See **Exhibit 3.7-19** and **Exhibit 3.7-20**.

Interbay Smith Cove

Alternative 4 would implement the same land use concept pattern in the Interbay Smith Cove Subarea as Alternative 2, <u>except for MML areas east of the railroad</u>, resulting in similar light and glare impacts <u>for most of the subarea</u>. Compared to <u>alternatives 2 and 3</u><u>Alternative 3</u>, Alternative 4 would implement more Industry & Innovation on the southwest slope of Queen Anne, resulting in taller building heights and increased visibility of development in western Queen Anne and Southeast Magnolia. See **Exhibit 3.7-19** and **Exhibit 3.7-20**.

SODO/Stadium

In the SODO/Stadium Subarea, Alternative 4 would apply a similar land use concept pattern as Alternative 3 with the following changes:

- Expand the Industry & Innovation node east of the stadiums northward to the I-5/I-90 interchange. This would increase building heights and visibility to residential populations in North Beacon Hill but reduce light and glare emissions compared to the MML land use concept.
- Expand the Urban Industrial node that encompasses the stadiums southward along 1st Avenue S Similar to above, this would increase allowed building heights but potentially reduce light and glare emissions.
- Convert the area west of Lumen Field bounded by Alaskan Way S, S Royal Brougham Way, and 1st Avenue S from Industry & Innovation to Urban Industrial. This would reduce building heights and visibility to the adjacent portions of Downtown.
- Incorporate additional Urban Industrial along Harbor Avenue SW in West Seattle and W Marginal Way in Pigeon Point. This change could increase building heights in this location, but effects on visibility to populations to the west in West Seattle and Pigeon point would be minimal due to steep terrain. Light and glare emissions would likely be reduced compared to the MML land use concept proposed under alternatives 2 and 3 in this area.

Overall, these changes would result in greater visibility of development to surrounding areas, particularly Downtown and North Beacon Hill, but reduced light and glare emissions, relative to alternatives 2 and 3. See **Exhibit 3.7-19** and **Exhibit 3.7-21**.



Exhibit 3.7-21 Increase in Viewshed (SODO/Stadium)—Alternative 4

Georgetown/South Park

In the Georgetown/South Park Subarea, Alternative 4 would apply a similar land use concept pattern as Alternative 3 with the following changes (see **Exhibit 3.7-19** and **Exhibit 3.7-22**):

- Designation of a small area near the intersection of Padilla Place S and S Orcas Street as MML instead of UI. This change could increase light and glare emissions and associated impacts on nearby residential properties not included in the MIC, as well as the nearby Georgetown Playfield and Spraypark to the northeast (which is within the MIC).
- Designation of the eastern side of Ellis Avenue S north of S Myrtle Street as UI instead of MML (similar to Alternative 2). Compared to Alternative 3, this would reduce light and glare emissions and effects on residential properties outside the MIC, west of Ellis Avenue S.
- Designation of the MIC area east of 14th Avenue S as MML instead of UI (as proposed for alternatives 2 and 3). This location is currently occupied by a Boeing facility and other manufacturing and warehouse uses consistent with the MML land use concept, so future light and glare emissions in this area would be similar to the No Action Alternative and greater than alternatives 2 and 3. Residential and commercial areas on the west side of 14th Avenue S in the South Park neighborhood would be most affected.
- Designation of a small area bounded by W Marginal Way, S Director Street, and 12th Avenue S as UI instead of MML. Due to the small size of this area, effects on overall light and glare emissions would be small, but it would create a transition area and reduce localized impacts on non-MIC residential properties in South Park north of S Director Street.



Exhibit 3.7-22 Increase in Viewshed (Georgetown/South Park)—Alternative 4

Impacts of the Preferred Alternative

As described in **Chapter 2**, the Preferred Alternative represents a blend of the other alternatives, resulting in a land use mix similar to alternatives 3 and 4, but with a greater share of Mixed Use Commercial and Industrial Commercial for areas outside the MICs. **Exhibit 3.7-23** shows the land use concepts and potential viewshed for the Preferred Alternative. The Preferred Alternative viewshed uses a combination of district-wide maximum building heights for each land use concept and more detailed height limits based on location-specific criteria. These represent a refinement of the height assumptions used for alternatives 2, 3, and 4. Where height assumptions for the same land use concept differ between alternatives, this is identified in the analysis.

Similar to Alternative 3, the Preferred Alternative would have reduced focus on large-scale outdoor operations, relative to alternatives 2 and 4, resulting in less usage of intense outdoor lighting. Similar to all the other Action Alternatives, the Preferred Alternative is anticipated to produce an increase in employment-generating building space above the No Action Alternative. Overall light and glare emissions from future development under the Preferred Alternative would be similar to those generated by alternatives 3 and 4, depending upon location. Potential location-specific impacts are described in the following sections.



Exhibit 3.7-23 Increase in Viewshed—Preferred Alternative

<u>Ballard</u>

In the Ballard Subarea, the Preferred Alternative would implement the same land use pattern as Alternative 4 for areas within the MIC, with the exception of one block east of 14th Avenue NW and north of NW Leary Way, where the MML land use concept would be applied instead of Industry & Innovation (see **Exhibit 3.7-24**). Given the small area of change, potential impacts from development within the MIC under the Preferred Alternative are anticipated to be the same as Alternative 4.

Outside the MIC, the Preferred Alternative would implement a combination of Urban Industrial, Mixed Use Commercial, and Industrial Commercial land use concepts.

- Along the northern edge of the MIC, west of 15th Avenue NW, existing Industrial Commercial zoning would be maintained where it currently applies, resulting in similar impacts as the No Action Alternative.
- The area northwest of the MIC along NW Market Street currently zoned Industrial Buffer would shift to Mixed Use Commercial. This would increase maximum allowed building heights to 75 feet, compared to 45 feet for alternatives 2, 3, and the No Action Alternative. The Preferred Alternative would reduce maximum building heights compared to Alternative 4 (160 feet).
- From Gasworks Park eastward, the Preferred Alternative would apply the Urban Industrial land use concept to the same locations as Alternative 4, resulting in similar building heights, typologies, and potential impacts.
- Portions of the subarea east of the MIC along the northern shore of the ship canal and Lake Union not designated as Urban Industrial would be zoned Industrial Commercial. As shown in Exhibit 2.4-6, the Industrial Commercial zone currently applies to much of this area. Potential impacts of the Preferred Alternative in these locations would therefore be similar to the No Action Alternative.
- The Preferred Alternative would rezone the portion of the subarea between the eastern MIC boundary and Evanston Avenue N from a mixture of Industrial General and Industrial Buffer to Industrial Commercial. However, allowed building heights would remain similar to existing zoning. Along the northern edge of this area, the Preferred Alternative would reduce building heights compared to alternatives 2, 3, and 4.




Source: City of Seattle, 2016; City of Seattle, 2021; BERK, 2022.

Interbay Dravus

The Preferred Alternative would implement the same land use concept pattern in the Interbay Dravus Subarea as alternatives 2, 3, and 4, resulting in similar light and glare impacts. See **Exhibit 3.7-23** and **Exhibit 3.7-24**. Compared to alternatives 2, 3, and 4, the Preferred Alternative would slightly increase maximum heights in the Urban Industrial area west of 15th Avenue NW (from 75 feet to 85 feet), resulting in increased visibility.

Interbay Smith Cove

The Preferred Alternative would implement the same land use concept pattern in the Interbay Smith Cove Subarea as Alternative 4, except for an area of II on the southwest slope of Queen Anne adjacent to the east side of the railroad, which would instead be MML. The Preferred Alternative also assumes lower maximum building heights in the portions of the subarea zoned for Industry & Innovation compared to Alternative 4 (85 feet versus 160 feet), resulting in reduced visibility and reduced light and glare impacts overall. See **Exhibit 3.7-23** and **Exhibit 3.7-24**.

SODO/Stadium

In the SODO/Stadium Subarea, the Preferred Alternative would apply a similar land use concept pattern as Alternative 4 with the following changes (see **Exhibit 3.7-23** and **Exhibit 3.7-25**):

- The area west of Lumen Field bounded by Alaskan Way S, S Royal Brougham Way, and 1st Avenue S would be designated Maritime, Manufacturing & Logistics instead of Urban Industrial. Maximum building heights in this area would be 85 feet, representing a small increase over Alternative 4 (75 feet). This would increase visibility to the adjacent portions of Downtown compared to Alternative 4 and the No Action Alternative, but reduce visibility compared to alternatives 2 and 3.
- The non-contiguous portion of the subarea east of I-5 and north of I-90 would be designated as a combination of Industrial Commercial and Mixed Use. This land use pattern (and associated building heights) would be similar to the No Action Alternative and would result in reduced visibility compared to alternatives 2, 3, and 4.
- The area east of 3rd Avenue S, north of S Holgate Street, south of S Royal Brougham Way, and west of I-5 would be designated as Maritime, Manufacturing & Logistics. Potential impacts would be consistent with alternatives 2 and 3 and would result in reduced building heights and visibility compared to Alternative 4.
- The Industry & Innovation node south of S Holgate Street is expanded southward along the east side of 4th Avenue S to S Horton Street and westward from Utah Avenue S to Colorado Avenue S (north of S Lander Street). The II zone is also expanded south of S Lander Street from Colorado Avenue S to Occidental Avenue S and north os S Forest Street. This would increase maximum building heights (from 85 to 160 feet) and visibility in these locations relative to all other alternatives.
- West of the Duwamish Waterway, the Preferred Alternative applies the same land use concepts as alternatives 2 and 3, except for the Urban Industrial area at the northwest corner of the subarea along Harbor Avenue SW (west of Port of Seattle Terminal 5). In all Urban Industrial areas west of the Duwamish Waterway, the Preferred Alternative would reduce maximum building heights to 45 feet, which would reduce building visibility relative to all other alternatives.



Exhibit 3.7-25 Increase in Viewshed (SODO/Stadium)—Preferred Alternative

Source: City of Seattle, 2016; City of Seattle, 2021; BERK, 2022.

Georgetown/South Park

In the Georgetown/South Park Subarea, the Preferred Alternative would apply a similar land use concept pattern as Alternative 4 with the following changes (see **Exhibit 3.7-23** and **Exhibit 3.7-26**):

- Designation of a small area near the intersection of Padilla Place S and S Orcas Street as UI instead of MML, consistent with alternatives 2 and 3. This change could reduce light and glare emissions and associated impacts on nearby residential properties and the nearby Georgetown Playfield and Spraypark compared to Alternative 4.
- Designation of a small area bounded by W Marginal Way, S Director Street, and 12th Avenue S as MML, consistent with alternatives 2 and 3. Due to the small size of this area, effects on overall light and glare emissions would be minimal.
- Expansion of the Seattle Mixed node between Corson Avenue S and Airport Way S:
 - Northwest across Corson Avenue S to include the Georgetown Playfield and Spraypark. This change would be unlikely to result in any change in the current use of the property as a park and playfield; light and glare conditions would be consistent with alternatives 2, 3, and 4.
 - Southward to include the areas between Corson Avenue S and Carleton Avenue S, north of S Bailey Street. Future development in this area would be mixed-use rather than urban industrial, but building heights would be similar to alternatives 2, 3, and 4, resulting in similar light and glare conditions.
- Expansion of the Urban Industrial corridor along S Orcas Street and S Homer Street westward to include areas west of 6th Avenue S. The use mix and building typologies allowed by the UI land use concept would potentially generate reduced light and glare emissions compared to the MML land use concept proposed under alternatives 2, 3, and 4. Additionally, the Preferred Alternative would reduce maximum building heights to 75 feet in this area, compared to 85 feet under the other alternatives, further reducing building visibility.





Source: City of Seattle, 2016; City of Seattle, 2021; BERK, 2022.

Summary of Impacts

Exhibit 3.7-27 provides a summary of impacts and comparison of the alternatives.

Subarea	Land Use Concept	Alternative 2	Alternative 3	Alternative 4	<u>Preferred</u> <u>Alternative</u>
Ballard	Maritime, Manufacturing, & Logistics	 Development style and light emissions similar in nature and location to existing Industrial General zones. Higher level of development would increase overall light emissions, especially along waterfront and near Ballard Avenue Landmark District. 	 Smaller MML footprint (compared to Alternative 2), resulting in reduced light emission exposure, particularly in areas northeast of the subarea. 	• Further reduced MML footprint, resulting in reduced light & glare emissions away from the waterfront.	 Within in the MIC, similar MML footprint as Alternative 4— except one block of II that would become MML. Outside the MIC, applies Industrial Commercial or Mixed Use to areas zoned MML under Alternatives 2, 3, and 4.
	Industry & Innovation	 Taller buildings would increase visibility in residential neighborhoods to the north. More office/commercial building typologies would reduce exterior light emissions. 	 See Alternative 2 	 Larger II footprint would increase visibility of buildings from surrounding neighborhoods. Largest potential viewshed of the alternatives. 	 Within the MIC, similar to Alternative 4 (see above). No II outside the MIC— applies IC and SM to areas zoned II under alternatives 2, 3, and 4.

Exhibit 3.7-27-23 Summary of Light and Glare Impacts—Action Alternatives

Subarea	Land Use Concept	Alternative 2	Alternative 3	Alternative 4	<u>Preferred</u> <u>Alternative</u>
	Urban Industrial	 Small increases in building heights would increase visibility in limited areas. Reduced light emissions and greater screening through landscaping and design concepts. 	 Increased UI footprint (compared to Alternative 2), providing more transitions to residential neighborhoods to the northeast and near Gas Works Park. Limited increases in height and visibility. 	• Smaller UI footprint than Alternative 3, but otherwise similar to Alternative 3.	 Inside MIC. same footprint as Alternative 4. Reduced footprint outside MIC, Industrial Commercial applied in downtown Ballard and between the MIC and Gas Works Park.
Interbay Dravus	Maritime, Manufacturing, & Logistics	 Development style and light emissions similar in nature and location to existing Industrial General zones. Light & glare emissions along the waterfront (including Ballard Locks) similar to No Action. 	 See Alternative 2 	 See alternatives 2 & 3 	• <u>See alternatives 2, 3, and 4.</u>
	Industry & Innovation	 N/A 	 N/A 	 N/A 	■ <u>N/A</u>
	Urban Industrial	 Small UI area would provide reduced emissions and transition to residential areas on northwest Queen Anne. 	 See Alternative 2 	 See alternatives 2 & 3 	 Same footprint as alternatives 2, 3, and 4. Slight height increase (10 feet): slight increase in visibility compared to alternatives 2, 3, and 4.
Interbay Smith Cove	Maritime, Manufacturing, & Logistics	 Development style and light emissions similar in nature and location to existing Industrial General zones. 	 <u>Reduced</u> <u>footprint of</u> <u>MML compared</u> to Alternative 2 <u>east of the</u> <u>railroad.See</u> <u>Alternative 2</u> 	 See Alternative <u>23</u> 	 <u>Greater MML footprint</u> <u>east of the railroad than</u> <u>alternatives 3 or 4, but</u> <u>less than Alternative 2.</u>

Subarea	Land Use Concept	Alternative 2	Alternative 3	Alternative 4	<u>Preferred</u> <u>Alternative</u>
	Industry & Innovation	 Would replace existing Industrial Commercial zoning in southeastern subarea. Reduced light emissions compared to No Action, but taller building heights would increase visibility in Southeast Magnolia and South Queen Anne. 	 Reduced II footprint compared to Alternative 2. Light emissions similar to Alternative 2, but smaller viewshed. 	 Greater II footprint than alternatives 2 or 3. Light emissions similar to alternatives 2 and 3, but greater viewshed.See Alternative 2 	 Reduced II footprint compared to alternatives 3 and 4, but greater than Alternative 2. Reduced height assumptions for II areas, reducing viewshed compared to other action alternatives.
	Urban Industrial	 <u>Convert IB areas</u> <u>east of railroad to</u> <u>UI. Reduced light</u> <u>emissions</u> <u>compared to</u> <u>IB.N/A</u> 	 Would create transition areas on southwest slope of Queen Anne. Light emissions would be similar to Alternative 2, but viewshed would be reduced. 	 <u>See Alternative</u> <u>2</u>N/A 	See Alternative 2
SODO/ Stadium	Maritime, Manufacturing, & Logistics	 Development style and light emissions similar in nature and location to existing Industrial General zones. Higher level of development would increase overall light emissions. 	 See Alternative 2. MML footprint reduced relative to Alternative 2 in area south of stadiums. 	 See Alternative 2. MML footprint further reduced relative to alternatives 2 and 3. 	 See Alternative 2. MML footprint increased west and southeast of stadiums. MML reduced in West Seattle.

Subarea	Land Use Concept	Alternative 2	Alternative 3	Alternative 4	<u>Preferred</u> <u>Alternative</u>
	Industry & Innovation	 Taller building heights in small area south of stadium district would increase visibility from surrounding areas, including Beacon Hill. Reduced light emissions in this location due to less intense exterior lighting. 	 See Alternative 2. Increased footprint compared to Alternative 2, further increasing visibility in surrounding areas. 	 II node east of stadiums expanded relative to alternatives 2 and 3, further increasing visibility in surrounding areas. II reduced west of the stadiums, reducing building heights and visibility relative to alternatives 2 and 3. 	 II removed southeast of stadiums (between I-90 and S Holgate Street), but expanded south along 4th Avenue S. II removed west of stadium. II in Judkins Park (alternatives 2, 3, and 4) converted to SM and IC. reducing heights.
	Urban Industrial	• Would reduce light emissions and create transition areas in targeted locations near the stadium district/downtown.	 See Alternative 2. 	 Increased UI footprint south and west of stadiums compared to alternatives 2 and 3. Conversion of MML to UI south of stadiums would slightly increase heights and visibility but would reduce light emissions. 	 <u>UI removed west of stadium, reducing visibility but increasing light/glare emissions.</u> <u>UI increased in West Seattle (Harbor Avenue SW).</u> <u>Reduced UI heights west of Duwamish, reducing visibility.</u>

Subarea	Land Use Concept	Alternative 2	Alternative 3	Alternative 4	<u>Preferred</u> <u>Alternative</u>
Georgetown/ South Park	Maritime, Manufacturing, & Logistics	 Development style and light emissions similar in nature and location to existing Industrial General zones. 	 Increased light emissions in the area between Corson Ave and Ellis Ave due to conversion of current Industrial Buffer zoning to MML. Compared to Alternative 2 and No Action, increased visibility of MML areas removed from MIC due to taller building heights under SM zoning. 	 Light emissions in the area between Corson Ave and Ellis Ave similar to Alternative 2 and No Action. Compared to Alternative 2 and No Action, increased visibility of MML areas removed from MIC due to taller building heights under SM zoning. 	 Reduced MML footprint in Georgetown compared to alternatives 2, 3, and 4, Similar footprint to Alternative 4 in South Park. Reduced overall light emissions compared to other alternatives.
	Industry & Innovation	■ N/A	 N/A 	 N/A 	• <u>N/A</u>
	Urban Industrial	 Implementation of UI along edges of the MIC would reduce light emission exposure 	 Compared to Alternative 2, increased visibility of UI areas removed from MIC due to taller building heights under SM zoning. 	• See Alternative 2.	 Expanded UI west along <u>S Orcas Street and S</u> <u>Homer Street, reducing</u> light emissions. <u>Convert UI at</u> <u>Georgetown Playfield to</u> <u>SM—similar heights as</u> <u>alternatives 2, 3, and 4.</u>

Source: City of Seattle, 2016. City of Seattle, 20224. BERK, 20224.

3.7.3 Mitigation Measures

Incorporated Plan Features

As described in **Chapter 2**, the Industry & Innovation and Urban Industrial land use concepts include several design principals that would limit light and glare impacts:

 The Industry & Innovation land use concept would include standards for frontage improvements, trees and landscaping, and maximum limits on vehicle parking areas.

- The Urban Industrial land use concept would incorporate open space and landscaping, which can reduce or screen light and glare emissions from surrounding areas.
- All proposed land use concepts would prohibit principal use parking areas, which often require extensive outdoor illumination. The Urban Industrial land use concept would also prohibit heavy manufacturing uses, which likewise may generate substantial light emissions due to operational and safety needs.
- The Urban Industrial land use concept includes standards for ground-level and upper-story setbacks from adjacent residential zones to create transition areas and reduce impacts.

Regulations & Commitments

- As described in Section 3.7.1 Affected Environment, Seattle Municipal Code Chapter 25.05.675 codifies environmental policies related to light and glare and public view protection. Future site-specific development projects requiring SEPA review will be evaluated for consistency with these policies.
- The Seattle Land Use Code (Seattle Municipal Code Title 23) contains development regulations, including standards governing the design and placement of exterior site and building illumination. Future development in the study area will be required to comply with the standards established for industrial zones in SMC Chapter 23.50, or their successor zones.

Other Potential Mitigation Measures

Consider implementation of additional development standards to address maximum height of exterior illumination. The Industry & InnovationII land use concept would allow buildings up to 160 feet in height, and the MML land use concept does not impose a maximum height, only a maximum Floor Area Ratio (FAR). These standards should address placement, light output, direction, and shielding of any exterior illumination above a given height to reduce light and glare emissions to adjacent non-industrial areas.

3.7.4 Significant Unavoidable Adverse Impacts

Urban development, including development of a non-industrial nature, generates light and glare emissions associated with occupation and operation. The precise nature of these emissions and impacts vary based on building design, location, and shielding/screening measures employed, but any future growth in the study area, regardless of the specific uses or building design, will generate at least some increase in light and glare. Though unavoidable, these effects can be minimized and reduced to less than significant levels through application of design standards and the mitigation measures described in this analysis.

Section 3.8 Land & Shoreline Use



This section summarizes the affected environment—including the historical context of planning and land use decisions, current land use plan and policy framework, and current land and shoreline uses in the study area—and compares impacts of the alternatives on land and shoreline use in the study area.

Four impact categories were used to identify potential adverse land use impacts in the study area and at a subarea level (where applicable): consistency with plans and policies, land use compatibility, employment mix, and land use transitions. The alternatives are expected to result in a land use impact if:

- Consistency with plans and policies. The action would result in an inconsistency between the predominant land use pattern and the stated land use goals and policies in the Comprehensive Plan and/or the VISION 2050 regional growth plan, Countywide Planning Policies, or Shoreline Master Program. The action would introduce a land use pattern that would foreclose future opportunities to reach goals and polices.
- Land use compatibility. The action would cause an increase in the prevalence of disparate activity levels and use patterns that would result in incompatibilities within industrial zones. Incompatibilities could undermine industrial and maritime operations, or the comfort and safety of employees or residents. Incompatibilities could be related to time of day/night activity, noise levels, odors, and conflicting movements by vehicles and other modes.
- **Employment mix.** The action would lead to changes to employment mix that would decrease the percentage and total quantity of jobs related to or supportive of industrial and maritime sectors, in Manufacturing Industrial Centers (MICs). The action would cause a high likelihood of voluntary or involuntary economic displacements of businesses in industrial maritime sectors widely throughout a subarea. It would preclude new opportunity for expansion of industrial and maritime employment through business formation and retention.
- Land use transitions. The action would create a land use pattern where high intensity / high impact uses would be likely to abut or encroach on adjacent non-industrial uses and concentrations of residential populations. These impacts can result from noise, light and glare, odor, or height, bulk, and scale of taller buildings adjacent to nonindustrial areas.

Land use impacts of the alternatives are considered significant if:

- There is an acute/severe adverse impact within one of the impact categories defined above.
- There are cumulative land use impacts in multiple categories within one of the defined subareas.

Within industrial areas that have limited residential populations and a utilitarian industrial context, impacts related to height, bulk, and scale, and aesthetics are not considered adverse impacts. Other areas of the city, outside of MICs or industrial zones are more sensitive to aesthetic and height/bulk/scale impacts. Therefore, within this EIS adverse impacts related to aesthetics and height/bulk/scale are focused on the transition areas and addressed as part of the land use transitions impacts analysis.

Mitigation measures and a summary of any significant unavoidable adverse impacts are included following the impacts analysis.

3.8.1 Affected Environment

Overview of Historical Planning & Land Use Decisions

Prior to the presence of White settlers in the region the study area was inhabited extensively by Coast Salish peoples for thousands of years. Before European contact, the region was one of the most populated centers in North America. The Indians of the Eastern Puget Sound lived in relatively small, autonomous villages and spoke variations of the Lushootseed (*txwəlšucid*, *dxwləšúcid*), one of the Coast Salish languages. Many tribes were affiliated through intermarriage, political agreement, trade, and material culture. Indigenous people lived in permanent villages of longhouses or winter houses, and traditionally left their winter residences in the spring, summer, and early fall in family canoes to travel to temporary camps at fishing, hunting, and gathering grounds. At the time of the first White settlements around 1850, natives were living in more than 90 longhouses, in at least 17 villages, in modern-day Seattle and environs including in the study area. See also **Section 3.11 Historic**, **Archaeological**, **& Cultural Resources**.

Waterways were central to the cultures and livelihoods of native people. Duwamish "Duwamish" is the Anglo-Europeanized word which meant "people of the inside", *dxwdəw?abš*, referencing the interior waters of the Duwamish, Black and Cedar rivers. The Suquamish take their name from the Lushootseed phrase for "people of the clear salt water", and the people living around Lake Washington were collectively known as *hah-choo-AHBSH* or *hah-chu-AHBSH* or *Xacuabš*, People of *HAH-choo* or *Xachu*, "People of a Large Lake" or "Lake People".

Physical alteration of the land and waterways by white settlers is important context for a discussion of land use today. Most present-day manufacturing and industrial centers are along the Duwamish River's historic meandering flood plain, Elliott Bay, Lake Union, and Salmon Bay. Prior to the Lake Washington ship canal and other alterations, the land and waterways looked much different. In the location of present-day Lake Union there were a series of separate lakes that natives transited with over-land portages. The Lushootseed name for present day Lake Union was *tenas Chuck* or *XáXu7cHoo* ("small great-amount-of-water"), present day Lake Washington was called *hyas Chuck* or *Xacuabš* ("great-amount-of-water"), and the present-day area of the Montlake Cut was called "Carry a Canoe".

Construction of a system of locks and cut waterways connecting east to west began in 1911 and culminated in 1916 (see **Exhibit 3.8-1**). Waters were connected from Lake Washington's Union Bay to Lake Union, to Salmon Bay though a series of locks to Shilshole Bay. As a result, the waters of Lake Washington were partially drained, lowering the level of that lake by 8.8 ft and drying up more than 1,000 acres of wetlands.

Changes to river flows at the south end of Lake Washington resulted from construction of the ship canal and locks. Prior to the alterations, Lake Washington emptied from its south end into the Black River (which no longer exists). The Black River connected to the Duwamish River, which outlets as it does today to Elliott Bay. The Cedar River, which had previously flowed into

the Black River in Renton, was diverted in 1912 directly into the south end of Lake Washington to reduce flooding in Renton. In 1916, when Lake Washington's level dropped, the remaining portion of the Black River dried up. Several indigenous villages were located near the confluence of the Black and Duwamish rivers and the area was long used as a place of refuge. When the Black River vanished, natives were displaced from the area.

During the first decades of the 20th century hundreds of acres of tide flats were also filled in to create dry land as depicted below in Exhibit 3.8-1. After the completion of the man-made Harbor Island in 1909, the mouth of the Duwamish River was divided into two channels. A series of major public works projects were undertaken to straighten and dredge the Duwamish riverbed, both to open the area to commercial use and to alleviate flooding. The City of Seattle formed the Duwamish Waterway Commission in order to oversee the re-channeling of the river and beginning in 1913 the river was altered to remove oxbows and meanders to maintain high water flows and turning ships. By 1920, 4¹/₂ miles of the Duwamish Waterway had been dredged to a depth of 50 feet, with 20 million cubic feet of mud and sand going into the expansion of Harbor Island. The shallow, meandering, nine-mile-long river became a five-mile engineered waterway capable of handling ocean-going vessels. The Duwamish basin became Seattle's industrial and commercial core area. Activities included cargo handling and storage, marine construction, ship and boat manufacturing, concrete manufacturing, paper and metals fabrication, food processing, and many other industrial operations. Boeing Plant 1 was established on the Lower Duwamish in 1916, and Boeing Plant 2 further upriver in 1936. Through the 1930's and 1940's Boeing's operations and footprint expanded greatly to support United States war efforts.

Native villages on the Duwamish were completely supplanted by white settlement and commercial use through the massive alterations of the land and waterways, the destruction of wildlife and fish habitats it caused, by the occupation of land. There was also deliberate removal of native settlements evidenced by burning of Indian longhouses in 1893. Duwamish people continued to work and fish in the area, using man-made "Ballast Island" on the Seattle waterfront as a canoe haul-out and informal market, but by the mid-1920's, most remnants of traditional life along the river had disappeared.



Exhibit 3.8-1 Seattle's Shoreline Over Time

Historical data from USCGS Topographic Sheet T-1406, 1875; USGS Land Classification Sheet, Seattle Quadrangle, 1897; USGS Topographic map, Seattle Quadrangle, 1909; USGS Topographic map, 1943; The Waterline's Project

Source: Burke Museum, The Waterlines Project.

With the spread of ecological concerns in the 1970s, various environmental, tribal, and community organizations became interested in the severely polluted Duwamish. Kellogg Island, the last remnant of the original river, was declared a wildlife preserve, and nearby terminal T-107 was converted into a park, creating a substantial natural area near the mouth of the river. T-107 is the site of the Duwamish village of xa?əpus. Intervention by Native people was instrumental in the reclamation of T-107 and Kellogg Island as natural areas that remain hotspots of biodiversity instead of additional industrial uses along the river.

In 2009, the Duwamish Longhouse and Cultural Center was opened on the west bank of the river. The Duwamish Longhouse, Herring's House Park, Kellogg Island have an important presence in the study area today as vestige of connection to the natural state of the river and of ownership and residence by the native Duwamish people.

Due to 20th century industrial contamination, the lower 5 miles of the Duwamish was declared a superfund site by the United State Environmental Protection Agency. Cleanup and restoration efforts are ongoing. The Duwamish River Community Coalition (DRCC) was established in 2001 to help monitor cleanup of the river. DRCC promotes place-keeping and prioritizes community capacity and resilience and is actively promoting improvements and investments in the greater Duwamish Valley that will benefit communities there.

The steady expansion of industrial and commercial enterprises on land in the study area led to some displacement of some non-native settlements. This history is evident in areas including Georgetown, South Park, and portions of northeast Ballard.

Due to dredging and rerouting of the Duwamish River, parts of the Georgetown and South Park neighborhoods once on riverbanks found themselves inland. Georgetown had early rail connections at the location of the present-day Union Pacific Argo Yard and operated as its own small city from 1904 to 1910 before being annexed by Seattle. Industrial and commercial activity expanded in the first decades of the 20th century with establishment of the public airport south of Georgetown (present day King County Airport), expansion of Boeing's aircraft assembly plants in the 1930's and 1940's and varied industrial and warehouse businesses on filled tidelands accessed by rail spurs. Residential elements declined in Georgetown by the 1950's and civic features such as a public library branch and movie theatre were shuttered. Evidence of isolation of former residential uses can be seen in the vicinity today where residential structures over 75 years old remain within a broader industrial context.

South Park, on the west bank of the Duwamish was similarly affected by the historical expansion of industry. In the late 1800s and early 1900s, South Park was largely a farming community. Italian and Japanese families farmed the alluvial plain of the Duwamish and brought goods for sale in Seattle at Pike Place Market. During the War era, South Park's residential population increased as a place for workers. However, in the late 1950s and 1960s Seattle sought to expand industrial zoning throughout South Park. Protests by residents resulted in most of present-day South Park retaining residential zoning and a residential presence.

Historical land use decisions also led to the location of multi-family housing in areas bordering industrial lands that caused environmental justice harms. Seattle's first zoning ordinance in 1923 and its major update in 1956 located multi-family residential districts at the edges of rail lines, industrial districts, and manufacturing districts. <u>These land use decisions were racially motivated and caused harm to non-White households.</u>

Racially restrictive covenants came into popular use in Seattle after 1920. Covenants were used by property owners, subdivision developers, or realtors to bar the sale or rental of property to specified racial or ethnic groups. Property deeds in predominantly White neighborhoods or desirable areas of new housing development often explicitly stated that no Asian, Black, and Indian people shall be permitted to occupy the property. Seattle residential areas with restrictive covenants included but are not limited to Victory Heights, Queen Anne, Capitol Hill, Blue Ridge, and Hawthorne Hills. Such neighborhoods are located away from the city's industrial areas. By excluding all but White households from covenant-restricted residential areas eligible locations for homes for Black, Asian, and Indigenous households were more likely to be in close proximity to industrial areas, such as Delridge, South Park, and South Beacon Hill (Honig 2021; University of Washington 2020).

In the late 1930s the practice of redlining was used to discriminate against racial minorities as the federal Home Owners' Loan Corporation (HOLC) evaluated mortgage risks in cities across the country. It rated neighborhoods as "best," "still desirable," "definitely declining," and "hazardous" (**Exhibit 3.8-2**). Neighborhoods with concentrations of Black, Asian, and Indian households were deemed financially risky and were marked in red so that mortgage lenders were discouraged from financing property there. The HOLC maps promoted racial inequality because it made mortgages difficult to obtain and expensive for minority households to buy homes where they lived, preventing them from accumulating wealth. Additionally, lenders refused to provide mortgages for Black, Asian, and Indian households in predominantly White neighborhoods rated "best" or "still desirable." On the 1936 HOLC map of Seattle, neighborhoods adjacent to the Duwamish industrial areas including Delridge, South Park, and South Beacon Hill were rated "hazardous", while neighborhoods closely adjacent to the Ballard and Interbay industrial areas including the lower slopes of Magnolia, Queen Anne, and portions of Ballard were rated "definitely declining."

<u>The effect of the racially discriminatory housing practices was that Black, Asian, Indian, and</u> <u>Rr</u>elatively less affluent renters were exposed to noise and air quality and other impacts, while single family districts removed from the edges of industrial areas were not. The continued pattern of multi-family housing and zoning districts bordering MICs continues to be evident today in areas including Interbay and the northeast edge of Ballard.



Exhibit 3.8-2 Commercial Map of Greater Seattle With "Grade Of Security" Designations, 1936

Source: Honig, 2021 (HistoryLink Essay No. 21296).

Data and Methods

The Land Use Section uses an inventory of existing land uses based on parcel level GIS data that was updated with manual scans by City staff and consultants and input from stakeholders. Existing and projected employment information relies on a 2021 CAI Inc. study. In addition to data, state, regional and local land use policies were reviewed and evaluated.

Current Policy & Regulatory Framework

Identification of land use impacts requires consideration of the policy framework regulating land use in Seattle's industrial areas. The policy framework flows from the State of Washington Growth Management Act, the Puget Sound Regional Council's (PSRC's) Multi-County Planning Policies (MPPs), King County's County-Wide Panning Policies (CPPs) the City Comprehensive Plan (Seattle 2035), and implementation actions including development standards in the Seattle Municipal Code (SMC) and the City's Shoreline Master Program. Several other regulatory measures affect industrial land use including localized overlay districts and community agreements.

State & Regional Policy Framework

Growth Management Act

The Washington State Growth Management Act (GMA), adopted in 1990, is a body of planning regulations that establishes requirements for Counties and localities to plan for future growth.

- GMA requires local governments to manage growth by (among other things) preparing comprehensive plans and implementing them through capital investments and development regulations (zoning).
- The Washington State Department of Commerce, the Puget Sound Regional Council, and a Governor-appointed Hearings Board oversees whether local governments are in compliance.
- Local comprehensive plans must provide land use capacity to accommodate growth that is projected for 20 years.
- Cities in King County must demonstrate sufficient zoned capacity for housing <u>and</u> employment growth.

Consistent with the GMA, the City of Seattle prepares updates to its Comprehensive Plan to accommodate new 20-year growth projections every eight years and has an annual process to amend the plan between major updates. Seattle most recently completed a major update, Seattle 2035, in 2015 and is preparing for a major update in 2024 that will extend the planning horizon to the year 2044.

The GMA establishes planning requirements and procedures including mandating elements of the Comprehensive Plan that the City must address (discussed below)

Puget Sound Regional Council VISION 2050

The Puget Sound Regional Council (PSRC) is composed of nearly 100 members, including the four counties, cities and towns, ports, state and local transportation agencies, and Tribal governments within the region. PSRC develops policies and coordinates decisions about regional growth, transportation and economic development planning within King, Pierce, Snohomish, and Kitsap counties.

The GMA requires multi-county planning policies (MPPs) and cities and counties planning under GMA <u>mustto</u> develop Comprehensive Plan policies consistent with the MPPs. MPPs for King, Pierce, Snohomish, and Kitsap are adopted by PSRC in a long-range plan called VISION 2050,

the region's plan for growth. By 2050, the region's population is expected to reach 5.8 million people.

PSRC designates MICs for the Puget Sound Region. VISION 2050 establishes criteria for designation of MICs. MICs are primarily locations of more intense industrial uses and employment and are not appropriate for housing. VISION 2050 calls for the recognition and preservation of existing centers of intensive manufacturing and industrial activity and the provision of infrastructure and services necessary to support these areas. VISION 2050 discourages non-supportive land uses in MICs, such as large retail stores or non-related offices.

The Regional Centers Framework adopted by PSRC in 2018 lays out criteria for designation of MICs that address size, current and future employment, and mix of uses, the majority of which are expected to represent core industrial activities. Cities are expected to plan for each MIC through a subarea planning process or the equivalent. There are 10 total designated MICs in the four-county region, two of which are in Seattle: the Greater Duwamish MIC and the Ballard Interbay Northend MIC (BINMIC).

The criteria established by PSRC for designation or redesignation as a MIC are the following:

- Planned jobs: 20,000 minimum.
- Minimum 50% industrial employment.
- If MIC is within a transit service district, availability of existing or planned frequent, local, express, or flexible transit service. If MIC is outside a transit service district, documented strategies to reduce commute impacts through transportation demand management strategies consistent with the Regional Transportation Plan Appendix F (Regional TDM Action Plan).
- Presence of irreplaceable industrial infrastructure.
- At least 75% of land area zoned for core industrial uses.
- Industrial retention strategies in place.
- Regional role.

MIC designation is important not only for the regional recognition of the value of the City's industrial areas to the State, but it also makes these areas eligible for federal transportation funding.

Local Policy Framework

King County

Within the GMA framework, each county collaborates with its cities to adopt Countywide Planning Policies (CPPs) and develop local growth targets that set expectations for local comprehensive plans. The MICs are also designated at this countywide level. In July of 2021 the GMPC approved new CPPs, and they are now being considered by the King County Council. The updated policies are consistent with PSRC's newly adopted VISION 2050. It is anticipated that these policies will be adopted prior to issuance of a Final Industrial and Maritime Strategy EIS. The CPPs include two policies for MICs. These policies are as follows:

DP-38 Designate and accommodate industrial employment growth in a network of regional and countywide industrial centers to support economic development and middle-wage jobs in King County. The Generalized Land Use Categories Map in Appendix 1 shows the locations of the designated Manufacturing/Industrial Centers. Designate these centers based on nominations from cities and after determining that:

a) the nominated locations meet the criteria set forth in the King County Centers Designation Framework and the criteria established by the Puget Sound Regional Council for regional manufacturing/ Industrial Centers;

b) the proposed center's location will promote a countywide system of manufacturing/industrial centers with the total number of centers representing a realistic growth strategy for the county; and

c) the city's commitments will help ensure the success of the center.

DP-38 Minimize or mitigate potential health impacts of the activities in manufacturing/industrial centers on residential communities, schools, open space, and other public facilities.

City of Seattle 2035 Comprehensive Plan

Seattle's Comprehensive Plan establishes land use policies for industrial areas in Seattle. The Plan, subject to approval by PSRC for consistency with VISION 2050 and the CPP's, above, sets out Seattle's growth management strategy. Seattle 2035 includes a land use element, container port element, and shoreline areas element that each establish land use goals and policies for Seattle's industrial areas. Other elements that guide the City's investments and activities in industrial lands include the transportation, economic development, and environment elements.

This proposal includes amendments to the existing goals and policies in the land use element that will include a framework for the new proposed industrial zones that are analyzed in the three Action Alternatives, an amendment to strengthen existing protections for industrial land by limiting changes to MIC boundaries to major updates of the Comprehensive Plan, and an amendment that states the intent of the City to work with the owners of the Oregon Washington Shippers Cooperative Association (WOSCA) site and the Interbay Armory site on future master planning for future industrial redevelopment of those sites.

The land use policies, below, include both the existing policy framework and the proposed amendments to the Comprehensive Plan that are a part of this proposal. The proposed amendments are indicated with <u>underlined</u>, and deletions are in strikethrough. <u>Changes to the</u> <u>Draft EIS amendments are also highlighted grey</u>.

Land Use Element

<u>Goals</u>

LU G10 Provide sufficient land with the necessary characteristics to allow industrial activity to thrive in Seattle and protect the preferred industrial function of these areas from activities that could disrupt or displace them.

LU G11 Support employment-dense emerging industries that require greater flexibility in the range of on-site uses and activities.

LU G12 Develop transitions between industrial areas and adjacent neighborhoods that support healthy communities, reduce adverse environmental impacts, and minimize land use conflicts.

<u>Policies</u>

LU 10.1 Designate industrial zones generally where

1. the primary functions are industrial activity and industrial-related commercial functions,

2. the basic infrastructure needed to support industrial uses already exists, areas are large enough to allow a full range of industrial activities to function successfully, and

3. sufficient separation or special conditions exist to reduce the possibility of conflicts with development in adjacent less intensive areas.

LU 10.2 Preserve industrial land for industrial uses, especially where industrial land is near rail- or water-transportation facilities, in order to allow marine- and rail-related industries that rely on that transportation infrastructure to continue to function in the city.

LU 10.3 Ensure predictability and permanence for industrial activities in industrial areas by limiting changes in industrial land use designation. There should be no reclassification of industrial land to a non-industrial land use category **or amendments to the**

boundaries of manufacturing industrial centers except as part of a City-initiated comprehensive study and review of industrial land use policies or as part of a major update to the Comprehensive Plan.

LU 10.34 Accommodate the expansion of current industrial businesses and promote opportunities for new industrial businesses <u>and emerging industries</u> within Seattle to strengthen the city's existing industrial economy.

LU 10.45 Restrict to appropriate locations within industrial areas those activities that by the nature of materials involved or processes employed—are potentially dangerous or very noxious.

LU 10.56 Provide a range of industrial zones that address varying conditions and priorities in different industrial areas. Those priorities include maintaining industrial areas that have critical supporting infrastructure, <u>leveraging investments in high-capacity</u> <u>transit service</u>, providing transitions between industrial areas and less intensive areas, and promoting high-quality environments attractive to business expansion or to new industrial activities.

LU 10.7 Use-Transition to the following zones for industrial lands in Seattle:

- Maritime, Manufacturing and Logistics: This designation would be intended to support the city's maritime, manufacturing, logistics and other industrial clusters. Areas that have significant industrial activity, accessibility to major industrial infrastructure investments, or locational needs (Port facilities, shipyards, freight rail, and shoreline access) may be considered for the maritime, manufacturing, and logistics designation.
- Industry and Innovation: This designation would be intended to promote emerging industries and leverage investments in high-capacity transit. These industrial transitoriented districts may be characterized by emerging industries and high-density industrial employment that combine a greater mix of production, research and design, and offices uses found in multi-story buildings. Areas in MICs and are generally within one quarter and one-half mile of high-capacity transit stations may be considered for the industry and innovation designation.
- Urban Industrial: This designation would be intended to encourage a vibrant mix of uses and relatively affordable, small-scale industrial, makers and arts spaces. Areas located at transitions from industrial to commercial and residential areas traditionally zoned for buffer purposes may be considered for the Urban industrial designation.
- Industrial Commercial: This designation is for industrial land located outside of Manufacturing Industrial Centers and is intended to permit a range of activities such as light industrial uses, research and development uses, and offices.

LU 10.68 Prohibit new residential development in industrial zones, except for certain types of dwellings, such as caretaker units <u>or, potentially in urban industrial zones</u>, <u>dwellings</u> <u>for-targeted to workers that</u> are related to the industrial area and that would not restrict or disrupt industrial activity.

LU 10.79 Use the general industrial or <u>maritime</u>, <u>manufacturing</u>, <u>and logistics zones</u> to promote a full range of industrial activities and related support uses.

LU 10.810 Apply the general industrial zones or the maritime, manufacturing, and logistics zone mostly within the designated manufacturing/industrial centers, where impacts from industrial activity are less likely to affect residential or commercial uses. Outside of manufacturing/industrial centers, general industrial <u>or the maritime,</u> manufacturing, and logistics zones may be appropriate along waterways used for maritime uses. <u>Consider applying the maritime, manufacturing, and logistics</u> designation zone mostly within the designated manufacturing/industrial centers. and it may also be appropriate outside of manufacturing/industrial centers along waterways used for maritime uses.

LU 10.9<u>11</u> Avoid placing industrial zones within urban centers or urban villages. However, in locations where a center or village borders a manufacturing/industrial center, use of the industrial commercial <u>zone</u> within the center or village where it abuts the manufacturing/industrial center may provide an appropriate transition to help separate residential uses from heavier industrial activities. Consider using the <u>urban</u> industrial zone in locations <u>within or outside where aurban</u> centers</u> or villages that borders a manufacturing/industrial center, where it abuts the manufacturing/industrial center may to help provide an appropriate transition <u>and promote complimentary land</u> <u>use patterns between to help separate residential uses from heavier</u> industrial <u>and nonindustrial</u> activities.

LU 10.4012 Limit the density of development for nonindustrial uses in the manufacturing/industrial centers to reduce competition from nonindustrial activities that are better suited to other locations in the city, particularly urban centers and urban villages, where this Plan encourages most new residential and commercial development. Permit <u>a limited amount of stand-alone</u> commercial uses in industrial areas as workforce amenities. or only if they reinforce the industrial character, and Sstrictly limit the size of office and retail uses not associated with industrial uses, in order to preserve these areas for industrial development, except for areas eligible for the Industry and Innovation zone.

LU 10.11<u>13</u> Recognize the unique working character of industrial areas by keeping landscaping and street standards to a minimum <u>in the maritime, manufacturing and</u> <u>logistics zone</u> to allow flexibility for industrial activities, except along selected arterials where installing street trees and providing screening and landscaping can offset impacts of new industrial development in highly visible locations.

LU 10.1214 Set parking and loading requirements in industrial zones to provide adequate parking and loading facilities to support business activity, promote air quality, encourage efficient use of the land in industrial areas, discourage underused parking facilities, and maintain adequate traffic safety and circulation. Allow some on-street loading and occasional spillover parking. <u>Consider limiting parking in the industry and</u> <u>innovation zone located in the vicinity of high-capacity transit stations.</u>

LU 10.1315 Maintain standards for the size and location of vehicle curb cuts and driveways in industrial zones in order to balance the need to provide adequate maneuvering and loading areas with availability of on-street parking and safe pedestrian, <u>bike, and transit</u> access.

LU 10.14<u>16</u> Permit noise levels in industrial areas, except buffer areas, that would not be allowed in other parts of the city, in recognition of the importance and special nature of industrial activities.

LU 10.15<u>17</u> Classify certain industrial activities as conditional uses in industrial zones in order to accommodate these uses while making sure they are compatible with the zone's primary industrial function and to protect public safety and welfare on nearby sites. Require mitigation of impacts on industrial activity and on the immediate surroundings, especially nearby less intensive zones.

LU 10.1618 Prohibit uses that attract large numbers of people to the industrial area for nonindustrial purposes, in order to keep the focus on industrial activity and to minimize

potential conflicts from the noise, nighttime activity, and truck movement that accompanies industrial activity. Consider allowing such uses in the urban industrial zone only.

LU 10.19 In the industry and innovation zone, consider development regulations that are compatible with employment-dense transit-oriented development. Seek to establish minimum density development standards thatto ensure employment density at a level necessary to leverage transit investments. Consider upper level density limits to discourage higher value ancillary uses that are more appropriate in non-industrial areas.

LU 10.20 In the Industry and Innovation zone, consider development standards that promotes development that meets the needs of industrial businesses including load-bearing floors, freight elevators, and adequate freight facilities.

LU 10.21 In the industry and innovation zone, consider an incentive system whereby nonindustrial floor area may be included in a development as a bonus if new bona-fide industrial space is included.

LU 10.1722 Establish the industrial buffer Consider using the urban industrial or <u>industrial buffer</u> zones to provide an appropriate transition between industrial areas and adjacent residential or pedestrian-oriented commercial zones.

LU 10.23 In the urban industrial zone, consider allowing a range of ancillary nonindustrial uses. Recognize that industrial businesses in this zone have a greater need for a limited amount of space for such uses as tasting rooms and retail facilities that directly support the industrial activity of the business.

LU 10.24 In the urban industrial zone, consider establishing buffer standards to ease the transition from industrial areas to urban villages and other non-industrial parts of <u>Seattle.</u>

LU 10.25 Recognize the unique development opportunit<mark>yies</mark> that the Washington National Guard Armory in the BINMIC and the WOSCA (Washington Oregon Shippers Coopertaive Association) represents. Work with the State of Washington or other future owners of this site to develop a comprehensive industrial development plan. This plan should include green infrastructure, consolidated waste management programs, and workforce equity commitments.

LU 10.1826 Allow the widest possible range of manufacturing uses and related industrial and commercial activities within the industrial buffer zone, while ensuring compatibility the activity and physical character of neighboring less intensive zones.

LU 10.1927 Include development standards or performance standards for the industrial buffer zone that protect the livability of neighboring areas, promote visual quality, and maintain a compatible scale of development along zone edges. Apply these standards only in places where existing conditions do not adequately separate industrial activity from less intensive zones.

LU 10.2028 Limit the height of structures on the borders of industrial buffer zones where streets along the zone edge do not provide sufficient separation for a reasonable transition in scale between industrial areas and less intensive neighboring zones, taking into consideration the permitted height in the abutting less intensive zone.

LU 10.2129 Allow a wide mix of employment activities in the industrial commercial zones, such as light manufacturing and research and development.

LU 10.2230 Limit development density in industrial commercial and <u>maritime</u>, <u>manufacturing</u>, <u>and logistics zones</u> in order to reflect transportation and other infrastructure constraints, while taking into account other features of an area.

LU 10.2331 Include development standards in the industrial commercial zone designed to create environments that are attractive to new technology businesses and that support a pedestrian-oriented environment, while controlling structure height and scale to limit impacts on nearby neighborhoods.

LU 10.24<u>32</u> Provide a range of maximum building height limits in the industrial commercial zones in order to protect the distinctive features that attract new technology businesses to the area—such as views of water, shoreline access, and the neighborhood scale and character—to make sure that these features will continue to be enjoyed, both within the zone and from the surrounding area.

LU 10.2633 Assign height limits independently of the industrial zoning designation to provide flexibility in zoning-specific areas and to allow different areas within a zone to be assigned different height limits according to the rezone criteria.

LU 10.2634 Restrict or prohibit uses that may negatively affect the availability of land for industrial activity, or that conflict with the character and function of industrial areas.

LU 10.2735 Consider high value-added, living wage industrial activities to be a high priority.

LU 10.2836 Permit commercial uses in industrial areas to the extent that they reinforce the industrial character, and limit specified non-industrial uses, including office and retail development, in order to preserve these areas for industrial development.

Container Port Element Land Use Policies (from Seattle 2035)

The container port element contains land use, transportation, economic development, and environmental policies to guide and support container port activities in Seattle. The land use policies emphasize ensuring adequate land area needs for port expansion, avoiding land use conflicts. These policies focus more specifically on the maritime industry than the land use policies, above. Container Port Element land use policies are below:

CP 1.1 Help preserve cargo container activities by retaining industrial designations on land that supports marine and rail- related industries including industrial land adjacent to rail or water-dependent transportation facilities.

CP 1.2 Continue to monitor the land area needs, including for expansion, of cargo container related activities and take action to prevent the loss of needed land that can serve these activities.

CP 1.3 Discourage non-industrial land uses, such as <u>stand-alone</u> retail and residential, in industrially zoned areas to minimize conflicts between uses and to prevent conversion of industrial land in the vicinity of cargo container terminals or their support facilities.

CP 1.4 Consider how zoning designations may affect the definition of highest and best use, with the goal of maintaining the jobs and revenue that cargo container activities generate and to protect scarce industrial land supply for cargo container industries, such as marine and rail-related industries.

CP 1.5 Consider the value of transition areas at the edges of general industrial and <u>maritime manufacturing and logistics</u> zones which allow a wider range of uses while not creating conflicts with preferred cargo container activities and uses. In this context, zoning provisions such as locational criteria and development standards are among the tools for defining such edge areas.

Shoreline Areas Element (from Seattle 2035)

As part of the Shoreline Master Program (discussed below), the shoreline areas element contains land use policies for industrial land adjacent to Seattle's shorelines. These policies are implemented through the Shoreline Master Program which designates which shorelines are industrial in use and establishes development regulations for those uses within 200-feet of the shoreline.

SA P37 Support the retention and expansion of existing conforming water-dependent and water-related businesses and anticipate the creation of new water-dependent and water-related development in areas now dedicated to such use.

SA P38 Identify and designate appropriate land adjacent to deep water for industrial and commercial uses that require such condition.

SA P39 Provide regulatory and nonregulatory incentives for property owners to include public amenities and ecological enhancements on private property.

SA P40 Identify and designate appropriate land for water-dependent business and industrial uses as follows:

- 1. Cargo-handling facilities
- 2. Tug and barge facilities
- 3. Shipbuilding, boatbuilding, and repairs
- 4. Moorage
- 5. Recreational boating
- 6. Passenger terminals
- 7. Fishing industry

(See Seattle 2035 for Detailed policy guidance provided for each)

SA P41 Allow multiuse developments including uses that are not water dependent or water related where the demand for water-dependent and water-related uses is less than the land available or if the use that is not water dependent is limited in size, provides a benefit to existing water-dependent and water-related uses in the area, or is necessary for the viability of the water-dependent uses. Such multiuse development shall provide shoreline ecological restoration, which is preferred, and/or additional public access to the shoreline to achieve other Shoreline Master Program goals.

Comprehensive Plan Growth Strategy

The Comprehensive Plan includes the city's overall plan for accommodating housing and job growth over a 20-year planning horizon. Under GMA the plan must demonstrate the City's ability to accommodate expected additional jobs and housing. The plan includes estimations for where jobs and housing will be located in the city and seeks to steer the allocation of new jobs and housing to those areas with land use regulations. During the previous 20-year planning horizon of the Seattle 2035 Comprehensive Plan, MICs were planned to accommodate 9,000 of the city's estimated total job growth of 115,000, or about 8%. The City is currently embarking on a major update to the Comprehensive Plan to the year 2044 <u>titled the One Seattle Plan</u>, and the total citywide estimat<u>eion</u> of job growth for the new 20-year planning horizon is 169,500 additional jobs. Growth studied in this <u>Industrial and Maritime Strategy EIS</u> is expected to be integrated into the 2044 Comprehensive Plan major update. Land use policy updates and zoning changes contemplated in this Industrial and Maritime Strategy EIS will be considered and integrated into the One Seattle Comprehensive plan major update EIS. The One Seattle Comprehensive Plan major update EIS held its scoping phase between June 23, 2022 and August 22, 2022.

MIC Subarea Plans

PSRC MIC designation also requires Centers Plans (this is a requirement for other designated Urban Centers as well). Both the Greater Duwamish MIC and the BINMIC have subarea plans that were adopted in 2000 and 1998, respectively. As part of VISION 2050, PSRC is requiring the City to prepare updated subarea plans for the two MICs. These updates will update goals and policies consistent with this proposal and address VISION 2050 goals for Centers Plans.

Ballard Interbay Northend Neighborhood Plan

Applicable goals and policies include:

BI-P2 Preserve land in the BINMIC for industrial activities such as manufacturing, warehousing, marine uses, transportation, utilities, construction, and services to businesses.

Ch.3 Environment, Impacts, & Mitigation Measures - Land & Shoreline Use

BI-P8 Maintain the BINMIC as an industrial area and work for ways that subareas within the BINMIC can be better utilized for marine/fishing, high tech, or small manufacturing industrial activities.

BI-P9 Support efforts to locate and attract appropriately skilled workers, particularly from adjacent neighborhoods, to fill family-wage jobs in the BINMIC.

BI-P10 Support efforts to provide an educated and skilled labor workforce for BINMIC businesses.

BI-P11 Within the BINMIC, water-dependent and industrial uses shall be the highest priority use.

Greater Duwamish MIC Neighborhood Plan

Applicable goals and policies include:

GD-G3 Land in the Duwamish Manufacturing/ Industrial Center is maintained for industrial uses including the manufacture, assembly, storage, repair, distribution, research about or development of tangible materials and advanced technologies; as well as transportation, utilities, and commercial fishing activities.

GD-P5 Limit the location or expansion of nonindustrial uses, including publicly sponsored nonindustrial uses, in the Duwamish Manufacturing/Industrial Center.

GD-G8 The Duwamish Manufacturing/Industrial Center remains a manufacturing/industrial center promoting the growth of industrial jobs and businesses and strictly limiting incompatible commercial and residential activities.

Seattle Municipal Code Industrial Zones (SMC 23.50)

Seattle's industrial zones were last updated in 1987 when the current Industrial 1 (IG1), Industrial General 2 (IG2), Industrial Commercial (IC) and Industrial Buffer (IB) zones were established and have only been altered slightly since then. The functional intent of the zones is as follows:

- IG1: An area that provides opportunities for manufacturing and industrial uses and related activity, where these activities are already established and viable, and their accessibility by rail and/or waterway make them a specialized and limited land resource.
- IG2: An area with existing industrial uses, that provides space for new industrial development and accommodates a broad mix of activity, including additional commercial development, when such activity improves employment opportunities and the physical conditions of the area without conflicting with industrial activity.
- IC: The Industrial Commercial zone is intended to promote development of businesses which incorporate a mix of industrial and commercial activities, including light

manufacturing and research and development, while accommodating a wide range of other employment activities.

 IB: An area that provides an appropriate transition between industrial areas and adjacent residential zones, or commercial zones having a residential orientation and/or pedestrian character.

For a summary of the locational criteria and development regulations in the IG1, IG2, IC, and IB zones see **Appendix E**. Development standards include allowable uses, height limits, floor area ratio limits, and maximum size of use limits.

In 2007, the City passed Ordinance 122601 that took steps to reduce maximum size of use limits for non-industrial uses in industrial zones. It was preceded by studies that found industrial occupancy rates of industrial land to be very high and that non-industrial uses, such as offices and retail stores, were displacing industrial uses.

For an overview of proposed development regulations in a new set of industrial zones that would update and replace the existing zones see the description of alternatives in **Chapter 2**.

Stadium Transition Area Overlay District (STAOD) (SMC 23.74)

In addition to zoning regulations in the Seattle Municipal Code for industrial zones, the areas around professional sports stadiums are subject to the Stadium Transition Area Overlay District. In 2000, the City established the STAOD, which is a 93-acre area comprised of Lumen Field, T-Mobile Field and surrounding areas to the east, west and south of those stadiums. The overlay district applied additional zoning standards beyond the base zoning to achieve certain goals for the district, including improving the pedestrian environment and connections to Downtown, discouraging encroachment into industrial areas, and permitting a mix of uses to support the pedestrian-oriented character of the area. For a summary of development regulations in the STAOD see **Appendix E**.

Shoreline Management Act & Shoreline Master Program

The State of Washington requires Cities and Counties to plan for how shorelines in their jurisdiction will develop through a Shoreline Master Program (SMP). The SMP must address a wide range of physical conditions and development settings along areas of the shoreline. The SMP prescribes different environmental protection measures, allowable use provisions and development standards for each of these areas of the shoreline. The method to account for different shoreline conditions is to assign an environment designation to each distinct shoreline section. The environment designation assignments provide the framework for implementing shoreline policies and regulatory measures specific to the environment designations. The shoreline environments within Seattle's Shoreline District are divided into two broad categories; Conservancy and Urban and then subdivided further within these two categories. Within the Urban category are the Urban Industrial (UI) and Urban Maritime (UM) designations. These shoreline designations are found on sections of Lake Union, Salmon Bay, Elliott Bay (Terminal 92) and the Duwamish where adjacent land is zone for industrial use. In cases where the

development regulations in the SMP are more restrictive than the zoning regulations, the SMP supersedes. Shoreline Master program regulations provide additional controls and supports for the intended character and uses of unique shoreline lands. No amendments to the SMP are a part of the proposal studied in this EIS. See **Appendix F**.

Community Agreements

In addition to the above policy framework, some parts of Seattle's industrial lands are subject to community agreements. The Port of Seattle and the Magnolia Community Club and the Queen Anne Community Council have entered into a Short-Fill Redevelopment Agreement that establishes a neighborhood advisory committee to work with the port on disputes occurring during redevelopment activities and operations of Terminal 91 regarding light and traffic (**Appendix E**). This agreement does not regulate land use but is more of a mitigation vehicle for impacts resulting from T91 activities.

Planned Future Land Use

Exhibit 3.8-3 shows planned future land use for the study area. The Future Land Use Map (FLUM) is a required feature of the Comprehensive Plan under GMA. It indicates the city's policies and intent for guiding use of land in geographic areas over time. Seattle's industrial land (the study area) comprises approximately 12% of land citywide.

Manufacturing Industrial Centers (MICs). Most industrial land is within the two regionally designated MICs mapped with the MIC designation on the FLUM. The Greater Duwamish MIC is approximately 5,330 acres and stretches from the south end of downtown Seattle to the city's southern city limit. It includes land along the Duwamish River. The Ballard Interbay Northend MIC (BINMIC) is approximately 1,458 acres. It includes the lowlands along 15th Avenue and the rail tracks area stretching from north side of Elliott Bay to Salmon Bay. The BINMIC also include shoreline lands along Lake Union and the ship canal, as well as uplands adjacent to the Ballard urban village.

Land with a MIC FLUM designation is the subject of extensive policy guidance in the City's Comprehensive Plan and via regional VISION 2050 policy and designation criteria, and county-wide planning policies. The policy documents give these areas the highest priority for continued and ongoing future use primarily with industrial and maritime land uses. The policies prioritize industrial and maritime uses over stand-alone commercial and retail uses, and generally do not support residential uses. The City's practice has been to apply only industrial zone classifications within the MICs.¹¹

Industrially Areas Designated Outside MICs. Not all the city's industrial land (study area) is within MICs. There are limited lands with a FLUM designation of "Industrial Areas" outside of

¹¹ The only exception is one parcel of land zoned Commercial in the BINMIC in Interbay on the site of the GM Nameplate facility that was the result of an industrial use expanding over time onto a commercially zoned parcel.

the MICs. Pockets of designated Industrial Area are found on the north shore of Lake Union between the Fremont Urban Village and the University District Urban Center, between the I-90 ramps and Dearborn Street, and small collections of parcels north of NW Leary Way, by Nickerson Street, and north of Smith Cove Park near the Magnolia Greenbelt. Land in these areas is subject to the City's comprehensive plan policy guidance for "industrial areas" (policies LU10.1–10.31), but not the regional or county-wide policy framework for MICs. Similar to MICs, the City's practice has been to apply only industrial zone classifications to these areas.

Other Industrial Zoned Land. There are several areas of industrial zoned land in the study area in other FLUM classifications. These are areas with a history of industrial use adjacent to MICs or by shorelines that are now included in urban village growth areas. They include land in the west portion of the Ballard Hub Urban Village along NW Market Street, and a pocket of land south of the Swedish Medical Center. In the Fremont Hub Urban Village, a swath of industrially zoned land extends from NW 36th Street to the ship canal and near the base of Stone Way Avenue N. A small collection of parcels at the northwest corner of the University District Urban Center is zoned industrial. A pocket of industrial shoreline land on the east shore of Lake Union is in the Eastlake Residential Urban Village. The policy framework for industrially zoned land inside of urban villages is complex because industrial areas policies apply, but so do policies for urban villages. Urban village growth strategy policies are found in the Growth Strategy element and call for a mix of commercial and residential areas that can accommodate growth and are connected by transit. A wide range of commercial and mixed-use zones may be applied in areas designated as urban villages on the FLUM.



Exhibit 3.8-3-2 Future Land Use Map for Industrial Areas Within and Outside MICs

Source: City of Seattle, 2021.

Existing Zoning

Exhibit 3.8-4 displays the amount of existing zoning in the study area by zone classification. These figures also represent zoning under Alternative 1 No Action. The intent and features of the existing zone classifications are summarized above and in **Appendix E**.

Most of the study area is zoned either IG1 (52%) or IG2 (38%) reflecting how the IG zones are the foundation of the land use regulatory framework for the city's industrial areas. Only 10% of study area lands are in the IB and IC zoning classifications combined. A large majority of industrially zoned areas that are outside of designated MICs are zoned IC (86%). The IB zones only cover 5% of the study area in total and are found inside of the designated MICs.

The BINMIC has a greater share of land area in IC and IB zones (10% and 9% of the BINMIC respectively) compared to the Greater Duwamish MIC which is almost entirely zoned IG (95% of the Greater Duwamish MIC). This difference between the two MICs reflects the fact that the Duwamish has a greater degree of separation and physical boundaries at the MIC edges, while the BINMIC has a somewhat greater degree of physical integration with surrounding neighborhoods—befitting placement of the IB and IC zones.

	BINMIC		Duwamish MIC		Outside MICs		Citywide Total	
Zone	Percent	Acres	Percent	Acres	Percent	Acres	Percent	Acres
Industrial General 1	56.52%	824	52.31%	2,787	5.41%	8	52 %	3,612
Industrial General 2	24.69%	360	43.80%	2,282	8.11%	12	38 %	2,661
Industrial Commercial	9.67%	141	1.46%	78	86.49%	129	5%	347
Industrial Buffer	9.12%	133	3.43%	183	0%	0	5%	316
Total	100%	1,458	100%	5,330	100%	148	100%	6,936

Exhibit 3.8-4-3 Existing Zoning by MIC, Outside MICs, and Citywide

Source: City of Seattle, 2021.

Existing Land Use

This section characterizes existing land use conditions in the study area and breaks out land use features for the Greater Duwamish MIC and the BINMIC individually, and for the five EIS subareas where information is available and useful.

Exhibit 3.8-5 and **Exhibit 3.8-6** summarize the amount of existing industrial and non-industrial land uses in the study area for the BINMIC and north industrial areas and the Greater Duwamish MIC and south industrial areas. Existing land uses are the observed current activities on non-right of way land parcels. The assessment methodology for existing land use started with data provided by the King County assessor's office at the parcel level. However, sometimes assessor data is out of date or does not accurately reflect all the uses present. To address these

issues, input from stakeholders and manual scans by City staff and consultants were used to update the inventory. Data relied on are from the 2017 CAI Study that was updated and modified in 2020.¹²

Not all land designated for planning purposes as industrial, or that is zoned industrial has industrial land uses. The analysis shows on an area basis, how much of the study area is currently in use for industrial (**Exhibit 3.8-5**) and non-industrial (**Exhibit 3.8-6**) activity at the time of analysis. The data also displays the size of lands devoted to specific land use categories.

In the BINMIC and north industrial areas 628 acres or 59.4% of land by area is in an industrial category. The largest industrial uses by area include marine terminals and industrial heavy marine, reflecting the large presence of Port of Seattle Terminal 91 and Fisherman's terminal, and the BNSF railyard. See also **Exhibit 3.8-7**.

In the Greater Duwamish and south industrial areas 3,249 acres or 80.4% of land by area is in an industrial use category. The largest industrial uses by area include transportation terminals, marine terminals, and warehouses reflecting the large presence of Port Terminals the SIG and Argo Rail Yards, and the network of other warehouse uses. See also **Exhibit 3.8-8**.

Office is the largest non-industrial land use in the BINMIC occupying over 9% of the land area. In the Greater Duwamish MIC, office is also the largest non-industrial use, but it occupies only about 3% of the land there. See **Exhibit 3.8-6**.

	BINMIC and North Industrial Areas		Greater Duwa South Indus	mish MIC and strial Areas
Detailed Land Use	Net Acres	Percent	Net Acres	Percent
Industrial Gen. Purpose	48	4.5	295	7
Industrial Flex	0	0	2	0.5
Industrial Heavy	4	0.4	334	7.9
Industrial Light	32	3	122	2.9
Industrial Park	0	0	54	1.3
Industrial Staging	7	0.7	52	1.2
Distribution	2	0.2	27	0.6
Warehouse	61	5.8	577	13.6
Marine Terminal	157	14.8	665	15.7
Shipyard	32	3	20	0.4

Exhibit 3.8-5-4 Industrial Land Uses by Area

¹² The methodology is documented on page 7 of the November 2017 CAI report, Industrial Lands Land Use and Employment Study: <u>https://www.seattle.gov/Documents/Departments/OPCD/OngoingInitiatives/DuwamishIndustrialLandsStudy/OPCDIndustrialLandU</u> <u>seEmploymentStudy1.pdf</u>.
Ch.3 Environment, Impacts, & Mitigation Measures • Land & Shoreline Use

	BINMIC and North Industrial Areas		Greater Duwamish MIC and South Industrial Areas		
Detailed Land Use	Net Acres	Percent	Net Acres	Percent	
Industrial Heavy Marine	112	10.6	97	2.3	
Transpo Terminal	39	3.7	881	20.8	
Railroad	30	2.8	145	3.4	
Fleet Support Services	40	3.8	57	1.3	
Utilities	62	5.9	50	1.2	
Vocational Training	2	.2	13	0.3	
Subtotal	628	59.4%	3,249	80.4	

Source: City of Seattle, 2021.

Exhibit 3.8-6-5 Non-industrial Uses by Area

	BINMIC and North Industrial Areas		Greater Duwa South Indus	mish MIC and strial Areas
Detailed Land Use	Net Acres	Percent	Net Acres	Percent
Accommodation	0	0	2	0.4
Artists' Lofts	0	0	1	.03
Marina	59	5.5	3	0.1
Office	101	9.4	139	3.3
Retail Trade	47	4.4	95	2.3
Auto Repair / Trade	2	0.2	9	0.2
Auto Dealerships	1	0,1	10	0.2
Warehouse (Comm)	8	0.7	19	0.5
Healthcare / Social Services	10	1	2	0.05
Animal Services	1	0.1	0.4	0.01
Public Service Facilities	23	2.1	3	0.1
Education	1	.1	0	0
Mail Processing	2	.2	5	0.11
Entertainment and Arts	2	0.2	49	1.1
Outdoor Vehicle Storage	1	.1	11	0.3
Religious Inst.	1	0.1	1	0.02
Single Family	4	0.4	11	0.3
Multi-Family	3	0.3	4	0.1
Parking	12	1.1	107	2.5

	BINMI North Indus	C and strial Areas	Greater Duwamish MIC and South Industrial Areas		
Detailed Land Use	Net Acres	Percent	Net Acres	Percent	
Open Space	41	3.9	113	2.7	
Miscellaneous (water, vacant land, unknown)	112	10.5	204	4.8	
Subtotal	431	40.1	776.4	18.75	

Source: City of Seattle, 2021.



Exhibit 3.8-7-6 North Industrial Land Use

Source: CAI, 2017, updated 2020.



Exhibit 3.8-8-7 South Industrial Land Use

Source: CAI, 2017, updated 2020.

In addition to aggregate quantification of land uses, the qualitative analysis below highlights major features, important sites and uses, concentrations of activity, and notable adjacencies in the five EIS subareas (**Exhibit 3.8-9**). Characterizations inform a basis for identification of impacts in the EIS impact categories.



Exhibit 3.8-9-8 Industrial Subareas

<u>Ballard</u>

The Ballard Subarea consists of the land between the Salmon Bay shoreline and the Ballard Urban Village. For the purposes of this analysis the subarea also includes portions of the study area in the Fremont Urban Village and along the north and east shores of Lake Union.

The study area includes an extensive stretch of shoreline along the north shore of Salmon Bay. Shoreline lands are in the designated MIC from the Hiram Chittenden locks at the west to 3rd Avenue NW at the east. This portion of the shoreline contains a variety of maritime uses and marine services on a series of docks and piers that extend into Salmon Bay. There are industrial marine services and businesses primarily in the west portion including Trident Seafoods, Stabbert Marine Industrial shipyard, Waypoint Marine, and others. Closer to the Ballard bridge is a higher concentration of recreational marina services, and Seattle Maritime Academy.

The Ballard uplands south of Leary Way include a series of large parcels or whole blocks that developed with large footprint non-industrial uses. Seven non-industrial use developments are located in close proximity to one another: Ballard Blocks 1, Ballard Blocks 2, former New Seasons, UW Medical, Big 5 Sports, Office Depot, Fred Meyer. Together these non-industrial uses occupy about 22 acres of land. They contain retail and office activities unrelated to industrial and maritime sectors and draw volumes of users into the area. A wide variety of industrial uses are co-mingled and adjacent to or across the street. The variety of industrial activities includes car repair services, building/trades supply, and other light manufacturing. Other large-footprint uses of note in this area include the Quest church at Leary/14th Avenue NW, and the whole-block USPS mail distribution facility at 11th Avenue NW/ NW 46th Street.

The Ballard uplands north of Leary Way include a diverse array of industrial, commercial/retail, office storage and even some residential uses. There is a high concentration of breweries and tap rooms. Reuben's Brews, Urban Family Brewing, the Fremont Brewing production facility, Stoup Brewing, Fair Aisle Brewing, Bale Breaker and Yonder Cider Tap Room, <u>the former</u> Peddler Brewing Company and others are located here. Several large-scale industrial operations that occupy whole blocks are present including Rudd Company paint manufacturer and Bardahl Manufacturing, a maker of petroleum oil additives, lubricants and gasoline additives that are sold worldwide, and has operated in Ballard since 1939. There is an eclectic mix of retailers, many related to hardware and automotive. The large, new West Woodland building is a multi-story light industrial structure. A few scattered non-conforming residential single family and multi-family homes are found in blocks flanking 14th Avenue NW towards the north end of the subarea.

A portion of the subarea is in the Fremont Urban Village. Parcels fronting N 36th Street are small and only about 115 feet deep. The parcels have a high concentration of non-industrial uses especially bars and restaurants, which are generally accessed by patrons on foot from the N 36th Street frontage. A topography drop is present at the alley to the rear of those parcels, and from this alley south to the ship canal parcels sizes are generally larger. Land uses in the area include several large-scale office, software and technology uses including the Google and Adobe campuses, some of the Tableau offices, a biotechnical laboratory company, and the Burke Building offices. This cluster of uses is sometimes referred to as the Silicon Canal. There are no marine uses fronting this section of the ship canal, and the water's edge is primarily a recreational and open space feature experienced by users from the Burke Gilman trail, which runs along it. Industrial uses are mixed in this geographic area including a large footprint film/sound studio company, a distillery, craft manufacturers, and the Theo Chocolate company which includes production, and retail activities.

The study area includes waterfront land and adjacent uplands from the east edge of the Fremont Urban village to the southwest corner of the University District Urban Center. The shoreline has a consistent string of marine uses on a series of docks and piers extending into Lake Union. Recreational marine activities are present including three marinas, as well as industrial maritime activity such as the North Lake shipyard, a divers training school, and the police department harbor patrol site. Recreational and open space uses are integrated into the area with the presence of the 20-acre Gas Works parks and the Burke Gilman Trail. About four blocks of upland are included in the study area near the corner of the Stone Way N / N 35th Street Intersection. The only significant industrial activity in this pocket is the Seattle Public Utilities transfer station. Non-industrial uses include recently constructed offices of Brooks headquarters, and Tableau software's new structure between Woodlawn Avenue N and Densmore Avenue N. Other uses include restaurants, bars breweries and retail uses.

The study area includes one shoreline area on the east bank of Lake Union between E Newton Street and E Nelson Place. Waterfront uses are all marine uses with substantial dock infrastructure, including US Seafoods, and the Lake Union Dry Dock, and Seattle Seaplanes. The limited upland uses in this area are dominated by biotechnical / laboratory uses.

Interbay Dravus and Interbay Smith Cove

The Interbay Dravus and Interbay Smith Cove subareas consists of three distinct nodes— Fisherman's Terminal and vicinity, Dravus, and Smith Cove. This subarea stretches from the southern shoreline of Salmon Bay between the locks and ship canal on the north and Elliott Bay to the South. It is bound by the Queen Anne and Uptown neighborhoods to the east and Magnolia to the west. This Subarea contains a significant number of Port of Seattle facilities (Terminal 91, the Terminal 91 Uplands, and Seattle Fisherman's Terminal), the Washington State National Guard Armory, the BNSF switching yard and maintenance facility, and a mix of industrial, retail, and office uses.

The southern shoreline of Salmon Bay between the Hiram Chittenden Locks and 3rd Avenue NW is developed with significant maritime industries, general industrial uses, and the Port of Seattle's Fisherman's Terminal. These maritime uses include shipyards, marine terminals, fishing, and warehousing. Immediately adjacent to the shoreline uses is the BNSF switching yard creating a southern edge to this subarea. This land is zoned IG1 and is within the BINMIC. The Port of Seattle has recently completed work on the Maritime Innovation Center to incubate the next generation of maritime companies and has future plans for additional development of facilities to support the maritime industry.

South of the Fisherman's Terminal area and separated by BNSF tracks is the Dravus area. This area is 21 acres in size and includes 7 acres of mixed-use zoning and 14 acres of land zoned IG1 located within the BINMIC. This area is bound by the BNSF rail corridor to the west and north, 15th Avenue W to the East and the Interbay Golf Center to the south. The industrially zoned property is developed with a mix of manufacturing, warehousing, and office uses. In the future this area may be the location of a Sound Transit light rail station and a Seattle Storm practice facility. The future light rail station has the potential to substantially reduce the industrial capacity of this area depending on future decisions regarding station location and whether the crossing at salmon bay will be above ground or by way of a tunnel. In 2006, the seven acres south of the IG1 zoned area was rezoned from Commercial 2 (C2) to Seattle Mixed Dravus (SM-D). The C2 zone designation prohibited residential development unless approved by a conditional use permit. One condition was that the area is not proximate to an industrially zoned area. The rezone from C2 to SM-D allowed recent mixed-use residential development in this area.

Smith Cove is the southern boundary of the BINMIC. This area includes major port facilities (Terminal 91 and the Terminal 91 Uplands), the Washington National Guard Armory, the corporate headquarters for Expedia, and a diverse mix of maritime, industrial, commercial, and retail uses. Zoning in this area is IG 1, Industrial Buffer (IB), Industrial Commercial (IC). Smith Cove is also the site of a proposed Sound Transit light rail station and line. Major property owners in this area include the Port of Seattle, the State of Washington, and development companies that own office and retail projects in this area.

Port facilities in the Smith Cove area play an important and expanding role in Seattle's maritime sector. Terminal 91 provides short-term and long-term moorage for fishing and commercial vessels, including factory trawlers, long liners, tugs, barges, ferries, research vessels, and ships of state, military, and commercial vessels for lay-up or idle. Terminal 91 includes fish processing and cold storage facilities, access to vessel repair and services, fueling by barge, and on-terminal rail access. Upland from Terminal 91 is the Port of Seattle's Terminal 91 Uplands development project. Over the next 10-15 years, this two-phase project will construct flexible, light industrial building space to support maritime manufacturers and fishing industry suppliers in the BINMIC. Phase I will develop 100,000 square feet of light industrial space with minimal site infrastructure improvements. Phase 2 will involve construction of approximately 300,000 square feet of additional industrial space along with extensive utility improvements.

To the east of Terminal 91 is the Expedia Corporate Campus. This project is part of Seattle's technology sector and consists of several large office buildings and a significant parking garage. This land is zoned IC. Seattle adopted the IC zone in 1988 with the intention that it allow for industrial uses and importantly research and development offices. This zone in other areas of Seattle is home to technology companies including Google and Adobe in Fremont. Stretching south from Expedia along Elliott Avenue W, land is zoned primarily IC and is developed with multiple office buildings, warehouses, retail, and limited industrial uses.

Directly east of the Port of Seattle's Terminal Uplands project, separated by the BNSF rail corridor, is the Washington State Armory. This site is approximately 26 acres in size and is the staging facility for the National Guard emergency response and other activities. This site is zoned IG2 and is located within the BINMIC. Currently, the State of Washington is exploring options to relocate this facility and redevelop this site. Adjacent to this site to the north and east is significant retail development. These retail developments are allowed by existing zoning but not the intended use for the IG2 zone.

SODO/Stadium

The SODO/Stadium Subarea includes the mouth of the Duwamish River where it outlets to Elliott Bay. There are a concentration of maritime installations and terminals at and around the Duwamish River shoreline and Elliott Bay. This includes Harbor Island, with major shipyard terminals of Vigor Shipyards and Crowley Marine as well as Port of Seattle Terminals 5 and 18 that handle container cargo, and pier 30. Terminal 5 is completing major investment and upgrade. These locations feature on-terminal rail. Other marine activity includes the Coast Guard base, which is a homeport of arctic icebreakers. Port of Seattle's Terminal 46 at the north end of the subarea is currently vacantin an interim use. Potential plans for conversion to a cruise ship terminal are on holdon Terminal 46 are abandoned. Plans are currently being developed to potentially expand the Coast Guard base to the southern portion of Terminal 46, and the Northwest Seaport Alliance has signaled the intent to reestablish a container shipping terminal at the north end.

SODO contains the BNSF Stacy railyard. The Stacy Yard hosts transloading—a practice whereby containers are transferred from ships via short-haul trucking and loaded onto trains. SODO also is home to the Union Pacific Argo Yard, south of Spokane Street near Georgetown. The heavy rail line bisects SODO with tracks that carry train traffic to destinations north and south. Other rail-related facilities include the Amtrack maintenance facility, and some direct rail connectivity to logistics businesses.

The presence of rail and marine infrastructure supports a cluster of logistics focused businesses in SODO and other businesses dealing in heavy materials that are dependent on rail and/or marine infrastructure. Examples are numerous and include Ash Grove Cement, Alaskan Copper, and Nucor Steel, Alaska Marine Lines, MacMillan Piper, and Republic Services (a refuse transfer station and recycling facility). Rail and marine terminals have been a fixture in the area for at least 100 years. <u>Food production and distribution facilities are also present and active in</u> <u>the area including the Franz Bakery facility and Charlie's Produce distribution warehouse.</u>

The Stadium area is home to Seattle's professional football/soccer and baseball stadiums as well as other event venues, the WAMU theater, Showbox SODO. These facilities draw large volumes of visitors to a range of events. The stadiums are integrated functionally with Pioneer Square, Downtown and C/ID to the north. In the stadium area there is a more consistent presence of open spaces, sidewalks than in other parts of the subarea.

The WOSCA site is a notable vacant piece of land. It is approximately 6 acres located between the stadiums and SR 99 infrastructure and Terminal 46. The site was used as construction staging by WSDOT and is potentially eligible for future reuse.

Major non-industrial employers are in the SODO/Stadium Subarea. Starbucks corporate headquarters and the Seattle School District's John Stanford Center are two large offices located in the Lander Street corridor, and the Army Corps of Engineers has offices near Diagonal Avenue S. Significant non-industrial retail is located throughout SODO including the Home Depot, and Costco Wholesale at 4th Avenue S, south of Spokane Street.

The district hosts large public utility operations that occupy expansive swaths of land. The King County Metro Central Base is west of the stadiums, the Sound Transit Operations and Maintenance Facility is south of S Forest Street, and the Seattle City Light South Service Center is to the south of Spokane Street.

Throughout the SODO/Stadium Subarea there are numerous craft business and activities. A concentration is evident along the 1st Avenue S corridor to the south of the stadiums. The stretch includes maker businesses that attract visitors and have a sense of design orientation to customers. Examples include Macrina Bakery, Westland Distillery, Filson, and others.

There is a significant cluster of auto-oriented sales and service business in the Airport Way corridor. In blocks to the south of S Holgate Street large-sized auto dealerships for Honda, Toyota, and Mercedes Benz area present. The general vicinity also includes multiple auto maintenance and repair shops.

Important adjacencies include interfaces with Pioneer Square and Chinatown / ID at the north end, as well as the edges of the West Seattle and Delridge residential neighborhoods. However, most other edges of SODO have strong physical buffers to non-industrial areas. This include I-5 at the east and the steeply sloped and heavily wooded greenbelt to the west, and waters of Elliott Bay to the north.

The Duwamish Longhouse is located on west bank of Duwamish River south of Harbor Island, overlooking the Duwamish River Valley, near the village called hah-AH-poos, a major archeological site known as Duwamish Site No. 1. The Longhouse is among a cluster of open spaces that are some of the only remaining vestiges of natural shoreline conditions along the lower Duwamish River. Kellogg Island and Terminal 107 Park is an approximately 60-acre natural area owned by the Port of Seattle. Adjacent to the north of this green space is Herring's House Park, a 6.5-acre open space owned by the City of Seattle Department of Parks and Recreation. The Duwamish Longhouse is directly across W Marginal Way from these open space resources. More greenbelt land owned by Seattle Parks is behind the longhouse in the wooded and sloping areas of Pigeon Point Park, the West Duwamish Greenbelt and Puget Park. Duwamish Tribal Services hosts community gatherings, meetings at the longhouse and is seeking to expand the facility to support the social, cultural, and economic survival of the Duwamish Tribe. The organization intends to display artifacts and to create interpretive exhibits and tours to maximize its cultural and recreational public use. See **Exhibit 3.8-10**.



Exhibit 3.8-10-9 The Duwamish Longhouse

The Duwamish Longhouse

The Duwamish Longhouse is located on the west bank of Duwamish River south of Harbor Island, overlooking the Duwamish River Valley, near the village called hah-AH-poos, a major archeological site known as Duwamish Site No. 1. Duwamish Tribal Services hosts community gatherings and meetings at the longhouse, and is seeking to expand the facility to support the social, cultural, and economic survival of the Duwamish Tribe. The organization intends to display artifacts and to create interpretive exhibits and tours to maximize its cultural and recreational public use.

Source: BERK, 2021.

Georgetown/South Park

Georgetown is situated on the east bank of the Duwamish River. The riverfront contains numerous heavy industrial operations including cement, materials, recycling/refuse handling and logistics companies that rely on barging and water access, and the Ardagh glass manufacturing facility. Other notable shoreline uses include the Army Corps of Engineers offices at Diagonal Avenue S.

Georgetown's industrial uplands between the Union Pacific Argo Rail Yard and E Marginal Way S contain a high concentration of logistics and warehousing activities. Many of these buildings are characterized by warehouse structures with loading docks and bays and large access areas for truck turning. The Prologis Georgetown Crossing facility is a notably new warehouse and distribution center constructed in 2017, because it includes multiple levels of stacked

warehouse space. Interspersed among logistics operations there is a wide variety of small and medium sized industrial supply businesses, small offices, manufacturers, and makers including the Equinox Studios campus, and South Seattle College. The area also contains multiple breweries and distilleries including Georgetown Brewing at the east end of Lucille Street.

Boeing's campus and the King County International Airport / Boeing Field are located at the south edge of industrial Georgetown and extend south outside of Seattle's city limit on both sides of E Marginal Way. The airport averages 180,000 takeoffs and landings each year. The airport serves small commercial passenger airlines, cargo carriers, private aircraft owners, helicopters, corporate jets, and military and other aircraft. It's also home to various Boeing Company operations.

The study area surrounds two residential neighborhoods areas in Georgetown—the Van Asselt district between Ellis Avenue S and Corson Avenue S and a roughly four-block residential district between S Homer Street and S Fidalgo Street Both include townhomes, single family and multifamily housing including some new construction. Residents of these areas are closely adjacent to the surrounding industrial activities. The study area also surrounds blocks of commercially zoned land along 4th Avenue S and Lucille Street, and the S Albro Place corridor that contain a variety of retail and service uses. At the time of this writing a mixed residential development was proposed for the commercial areas on 4th Avenue S.

The triangular area bounded by Corson Avenue S, Carleton Avenue S and I-5 contains a high concentration of retail and restaurant businesses fronting onto Airport Way S. This stretch contains a string of notable brick historic structures including the historic Georgetown Brewery complex that backs up to the rail line on the east side of Airport Way. These structures are now occupied by a variety of small business. Several historic storefronts on the west side of Airport Way contain restaurants and coffee shops and the Georgetown Ballroom. The area attracts visitors and events unrelated to industrial activities. The west portion of the blocks in this triangle (off of the Airport Way) include construction and building supply firms, warehouse structures, and other light industrial uses.

A little-used Union Pacific Rail spur track 101 bisects the triangular area described above. The track 101 spur plays a role in activities at the Argo Railyard in the building and assembly of trains. Switching activities using railyard ramps 6-10 are facilitated by the use of track 101 to extend trains to the south of the railyard up to about 3,000 feet. The spur has at-grade crossings of the active streets of S Lucille Street, Corson Avenue S, and S Carstens Place, as well as at grade crossings of quieter streets of S Homer Street, and S Nebraska Street. Trains are extended onto this track intermittently depending on needs of rail carriers, causing noise and vibrations to affect nearby businesses and homes.

There is a large presence by arts organizations and businesses in Georgetown. Equinox Studios at 5th Avenue S and S River Street contains over 100,000 sq. ft. of spaces for artists and makers in industrial and warehouse style structures, making space available to hundreds of artists. There are approximately a dozen private commercial and retail art spaces and galleries and the event space at venues such as the Georgetown Ballroom and MiniMart City Park. A concentration of private galleries is found in commercial and industrial spaces generally within the Airport Way corridor. The neighborhood has organizations dedicated to promoting arts including Equinox Studios and the Georgetown Arts and Cultural Center, and there is a regular GeorgetownArtAttack art walk event. The School of Acrobatics and Circus Acts is located at 7th Avenue S and S Homer Street. The Seattle Design Center, which is a showplace for furniture and building materials is at 4th Avenue S between S Orcas Street and S Mead Street and numerous building materials supply businesses are clustered in that general vicinity. Oxbow Park prominently features the Hat and Boots art sculptures within the public space. Arts have thrived in Georgetown in part due to the availability of industrial style spaces with high ceilings and clear spans, that may also be relatively affordable for rent compared to commercial space in some other Seattle neighborhoods.

South Park is situated on the west bank of the Duwamish River. The study area contains the industrial lands that surround the South Park neighborhood, which is a mixed-use neighborhood that is designated residential urban village in Seattle's Comprehensive Plan.

Like Georgetown's riverfront, the South Park riverfront in the study area contains numerous heavy industrial operations that rely on marine transport including the Duwamish Shipyard, materials handling and logistics companies, and marine services. Riverfront operations south of State Route 509 are on smaller shoreland parcels, while operations north of SR 509 are large on shoreland parcels 20 acres or larger.

Upland uses in the study area that are north of the South Park urban village include a variety of distribution and logistics activities, small manufacturing, construction related businesses, small offices, and marine and industrial supply companies. This mix of light and heavy industrial uses closely borders the north edge of the South Park urban village. Duwamish Waterway Park is a 1.26-acre open space bordering the river at the northwest corner of the South Park Urban Village affording some river access. Lands near the SR 509 ramps at S Cloverdale Street, and S Holden Street contain large footprint uses of the South Transfer Station and the First Student bus parking yard, and Waste Management services.

Portions of the study area to the south of the South Park Urban Village flank the offramps of SR99 at 14th Avenue S. A high concentration of land uses here appear to be transportation oriented including distribution and warehousing, materials supply, and building materials. Many structures have loading docks and truck access and circulation. Other land uses include union hall offices and the currently vacant Boeing Radiation Effects Lab and Boeing South Park facilities, which closely border the edge of the urban village. In addition to industrial marine activities on the riverfront, the Duwamish Yacht Club is located on the riverfront here.

Land at the base of the South Park bridge, bordering the river and the urban village is outside of Seattle city limits and outside the study area. For reference, that land contains a mix of neighborhood-residential uses, the South Park Marina, and Port of Seattle's Terminal 117, which is being converted into a 2+ acre river front park. Adjacent to the study area, residential uses inside the urban village are primarily single-family homes, with some multifamily housing near arterial roadways. In several locations such as the vicinity of S Southern Street, 8th Avenue S, 5th Avenue S and others, residential uses are closely adjacent to industrial activities such as a transit van company, a portable toilet company, and an equipment supply company as examples.

Existing Employment Mix

Employment mix is addressed in the land use section because City and regional land use policies encourage employment in industrial and maritime sectors. A chief intention of industrial lands policies is to foster living wage employment opportunities and economic development associated with industrial and maritime sectors, and for diversification of the economy. (See plans and policies description above.) Existing employment on study area land in thirteen industry sectors is characterized in **Exhibit 3.8-11** and **Exhibit 3.8-12**. There is a total of 98,500 jobs. The analysis characterizes employment in industry sectors of interest, where there are agglomerations of related supporting economic activity. Methodology is from the 2019 CAI study.¹³

Industry	2018 Industrial Areas Employment
Hospitality & Tourism	6,700
Construction & Utilities	13,700
ICT	8,200
Distribution & E-commerce	8,500
Food & Beverage Production	3,800
Aerospace	6,300
Transportation & Logistics	5,500
Maritime	8,600
Other Manufacturing	5,900
All Other Retail	3,400
All Other Services	21,400
Government	5,300
Education	1,200
Total	98.400

Exhibit 3.8-11-10 Industrial Areas Employment by Economic Sector

¹³ The methodology is documented on page 7 of the November 2017 CAI report, Industrial Lands Land Use and Employment Study: <u>https://www.seattle.gov/Documents/Departments/OPCD/OngoingInitiatives/DuwamishIndustrialLandsStudy/OPCDIndustrialLandU</u> <u>seEmploymentStudy1.pdf</u>.

Source: CAI, 2020.

Employment in the study area and subareas can also be analyzed according to the quantity of jobs in industrial vs. non-industrial classifications. It is not straightforward to classify jobs as industrial or non-industrial. Methods in this analysis are from the 2019 CAI study. As seen in **Exhibit 3.8-12**, 55.3% of all employment in the study area is industrial and the percentage of industrial employment in all subareas is above 50%.

Subarea	2018 Industrial Emp.	2018 Total Emp.	% Industrial
Ballard	9,400	17,100	55.0%
Interbay Dravus	3,400	5,600	60/7%
Interbay Smith Cove	3,900	6,000	65.0%
SODO/Stadium	23,000	43,900	52.4%
Georgetown/South Park	14,900	25,900	57.5%
Total	54,500	98,500	55.3%

Exhibit 3.8-12 -11	Industrial and Non-Industrial Employment by Sub-Area, Current Conditions
(2018)

Note: Methodology is documented on page 7 of the November 2017 CAI report, Industrial Lands Land Use and Employment Study: https://www.seattle.gov/Documents/Departments/OPCD/OngoingInitiatives/DuwamishIndustrialLandsStudy/OPCDIndustrialLandUseEm ploymentStudy1.pdf.

Source: CAI, 2020.

3.8.2 Impacts

As described in the introduction to this section, four impact categories were used to identify potential adverse land use impacts for the study area broadly and on a subarea level (where applicable): consistency with plans or policies, land use compatibility, employment mix, and land use transitions. The alternatives are expected to result in a land use impact if:

- Consistency with plans and policies. The action would result in an inconsistency between the predominant land use pattern and the stated land use goals and policies in the Comprehensive Plan and/or the VISION 2050 regional growth plan, Countywide Planning Policies, or Shoreline Master Program. The action would introduce a land use pattern that would foreclose future opportunities to reach goals and polices.¹⁴
- Land use compatibility. The action would cause an increase in the prevalence of disparate activity levels and use patterns that would result in incompatibilities within industrial zones. Incompatibilities could undermine industrial and maritime operations, or the comfort and

¹⁴ It is not practical to summarize consistency with every policy or goal. Select policies or goals with notable factors towards consistency or inconsistency are described, and a general summary of the level of consistency with the full range of policies is provided under each alternative.

safety of employees or residents. Incompatibilities could be related to time of day/night activity, noise levels, odors, and conflicting movements by vehicles and other modes.

- Employment mix. The action would lead to changes to employment mix that would decrease the percentage and total quantity of jobs related to or supportive of industrial and maritime sectors, in MICs. The action would cause a high likelihood of voluntary or involuntary economic displacements of businesses in industrial maritime sectors widely throughout a subarea. It would preclude new opportunity for expansion of industrial and maritime employment through business formation and retention.
- Land use transitions. The action would create a land use pattern where high intensity / high impact uses would be likely to abut or encroach or <u>create</u> impacts related to height, bulk, scale and aesthetics on adjacent non-industrial uses and concentrations of residential populations.

Not every adverse land use impact identified within the impact categories would result in a significant adverse impact as some impacts are an expected part of a changing urban environment. Land use impacts of the alternatives are considered significant if they would result in more than a moderate adverse impact regarding:

- An acute/severe adverse impact within one of the impact categories defined above.
- Cumulative land use impacts in multiple categories within one of the defined subareas.

The terms "**minor**" and "**moderate**" are also used in the assessment to describe relative levels of impact below the threshold of significance. Minor is used to describe a level of impact that is barely perceptible, de minimis or questionable as to whether it would materialize at all. Moderate is used to describe a level of impact that would clearly be perceptible, have a tangible influence, yet not exceed the threshold for significance.

Optional economic analysis. The City is not required to address purely economic impacts on individual businesses in environmental analysis (SMC 25.05.440 F.3). In scoping and Draft EIS comments, some commented that if certain land uses are not permitted under an alternative (i.e., unlimited housing) landowners would be less likely to invest in improvements and development, which would lead to economic blight. Comments are addressed in the scoping report (**Appendix A**), and responses to comments chapter. A wide variety of land uses would be allowed in the study area under all alternatives sufficient for robust economic use of property. However, purely economic factors for individual businesses are not an element of the environment to be analyzed and therefore are not considered a factor in determining significant impacts. The City includes at its option some non-environmental economic analysis of development feasibility. Development feasibility information was used by the City to refine and amend some of the development standards proposed in the Preferred Alternative, especially for the II zone to increase the likelihood of development feasibility and investment in the near term, while acknowledging that economic conditions are likely to change over time and feasibility of development on some sites will be further into the future.

Organization of the impacts analysis. The first portion of the impact analysis under each alternative describes the likely changes over the 20-year planning horizon under the alternative

in the topics areas of land use planning and policy context, future land use, zoning, land use, and employment mix. The changes are also described in the description of alternatives in **Chapter 2**, which should be read in conjunction with this Land Use Chapter. The assessment of impacts follow the descriptions.

Equity & Environmental Justice Considerations

While shoreline and land use impacts are expected to be less than significant under all alternatives, some of the identified impacts could have equity and environmental justice considerations.

Land use transition impacts would raise environmental justice concerns where residents of nonindustrial areas in or adjacent to the study area could be adversely affected by inadequate transitions at the edges of industrial areas. In areas of inadequate transitions, impacts from noise, odors, and truck access and circulation associated with industrial land uses could affect communities of color and economically disadvantaged people. Impacts of increased building height, bulk and scale at transitions could also affect vulnerable populations. The neighborhoods of Georgetown, SODO, and South Park are vulnerable because there are land use transition impacts and they have populations with higher levels of disadvantage as seen in **Exhibit 1.7-7**. However, the proposal includes features with potential to improve transitions as well, especially the expected development in the Urban Industrial zone. The Preferred Alternative includes new features to improve land use transition conditions based on extended public engagement during the Draft EIS comment period in the Georgetown and South Park neighborhoods.

Land use compatibility impacts could have equity and environmental justice considerations. Introduction of new buildings with dense employment in the II zone and industry-supportive housing in the UI zone could create incompatibilities between new activity patterns and adjacent areas of continued industrial uses. There is potential for new employees or residents in the rezoned areas to be vulnerable populations at a relatively higher rate. Adverse localized impacts on these community members could result from increased exposure to freight traffic and other challenges of working or living in the area. <u>The Final EIS includes additional</u> <u>mitigation measures and more information on potential mitigation to address potential</u> <u>compatibility impacts, especially in the Georgetown and South Park areas based on extended</u> <u>public engagement during the Draft EIS comment period.</u>

In general, it is expected that the proposal will have positive equity affects related to the employment mix. Under all alternatives, employment in the study area would increase including industrial employment. A high proportion of jobs in industrial and maritime sectors are accessible without a traditional four-year degree and many remain unionized with high quality benefits. With increased employment training opportunities focused on equitable access, vulnerable populations could benefit from increased employment in industrial and nonindustrial sectors.

While impacts on vulnerable communities are identified, a range of existing regulations and commitments and potential mitigation strategies will reduce the harmful impacts of the proposal related to land and shoreline use.

Impacts of Alternative 1 No Action

Likely Changes Over the 20-year Planning Horizon

Land Use Planning & Policy Context. Under Alternative 1 the planning and policy context would be unchanged from existing conditions. No changes to the Comprehensive Plan or policy framework would be enacted. Existing Comprehensive Plan land use policies (without amendment) summarized above in **Local Policy Framework** would continue. There would be no updates to the currently adopted Sub Area Plans for the Greater Duwamish MIC and BINMIC.

Future Land Use. Under Alternative 1 the future Land Use Map for the study areas would be unchanged. Future land use categories would be the same as shown in the existing conditions section in **Exhibit 3.8-3** above. No specific lands would be added to or removed from the MICs under Alternative 1. However, because regulations allow for annual amendment proposals to the Comprehensive Plan, some land could be removed from MICs over the 20-year planning horizon as a result of individually proposed annual amendments.

Zoning. Under Alternative 1 zoning would be unchanged. Development standards for the city's four existing industrial zones (IG1, IG2, IB, IC) would be unchanged from those summarized above in **Local Policy Framework**. No changes to the zoning maps would be proposed. However, because regulations allow for annual amendment proposals to the Comprehensive Plan and contract rezones, some land could be removed from MICs over the 20-year planning horizon as a result of individually proposed zoning changes.

Land Use. Under Alternative 1 land use would continue to evolve over the planning period according to current trends and the parameters of existing zoning. Some notable expected changes could include.

- Continued conversion to office and retail uses in IG zoned areas. Consistent with recent trends, more stand-alone retail and office structures similar to the Armory Way shopping center or Ballard Blocks would be anticipated. Developments would maximize current IG zone maximum size of use limits for offices and retail. Areas that could see increased concentrations of such development would be in Fremont, Ballard, Interbay Smith Cove, and Georgetown/South Park.
- Continued development of large offices in IC zoned areas. Consistent with recent trends, more large office development would occur in IC zoned areas with no inclusion of industrial uses. This would be expected in the stadium area and the Elliott Avenue corridor and areas of Ballard.
- Interim timeframe, some lack of investment. In the first half of the planning horizon some disinvestment could be expected for land parcels close to future ST station areas at

SODO/Lander, W Dravus Street, and Ballard as landholders would not be likely to invest in new development in areas of aging infrastructure on large parcels near stations, in anticipation of future rail.

- Continued addition of distribution and warehouse facilities. Continued addition of some new distribution and warehouse facilities would be expected in the study area.
- Little or no new housing. Only about 75 new homes would be added in caretakers' quarters and artist/studios.
- Maintenance of maritime and industrial base. Most long standing maritime and logistics uses would continue on waterfront lands and industrial lands near infrastructure, especially in the Duwamish.
- Armory Site Redevelopment. Under Alternative 1 the Armory site would be developed with light industrial and flex space of a relatively low-density nature after relocation of the Army National Guard to North Bend, WA.
- Piecemeal conversions of parcels from industrial to non-industrial. Annual comprehensive plan applications for amendment would allow for piecemeal removal of parcels of land from the MIC and conversions to non-industrial zoning. The location and amount are not known.

Employment Mix. Under Alternative 1 employment is projected to grow incrementally in proportions similar to trends from the last 10-20 years. A total of 23,500 additional jobs are projected for the study area, an increase of 24%. Job growth in the study area would be estimated to be about 14% of expected citywide job growth over the 20-year planning horizon. The percentage of industrial employment would decrease slightly by 0.9% points. Both MICs would continue to contain many more than the minimum number of industrial jobs required to meet PSRC's regional criteria for MIC designation (20,000). See **Exhibit 3.8-13**.

	Current Conditions (2018)			Alternative 1 No Action (2044)		
Subarea	Ind. Emp.	Total Emp.	% Ind.	Ind. Emp.	Total Emp.	% Ind.
Ballard	9,400	17,100	55.0%	11,600	22,300	52.0%
Interbay Dravus	3,400	5,600	60.7%	3,900	6,800	57.4%
Interbay Smith Cove	3,900	6,000	65.0%	4,700	7,400	63.5%
SODO/Stadium	23,000	43,900	52.4%	28,200	53,500	52.7%
Georgetown/South Park	14,900	25,900	57.5%	18,000	32,000	56.3%
Total	54,500	98,500	55.3%	66,400	122,000	54.4%

Exhibit 3.8-13-12 Employment by Subarea, Current Conditions and Alternative 1

Source: City of Seattle, 2021.

Consistency with Plans & Policies

Although there would be no changes to plans and policies under Alternative 1, some inconsistencies with plans and policies are expected to increase due to the evolution of land use during the study time horizon under Alternative 1.

Conditions in both the Greater Duwamish MIC and BINMIC would still meet PSRC's regional criteria for designation as an Employment Center MIC.

- 75% land zoned for core industrial uses. Under Alternative 1 the IC zone would not be considered a core industrial zone satisfying the PSRC criteria because zone development under existing standards would increasingly be dominated by office-only uses. Nonetheless, 90% of land in the BINMIC would be in core industrial zones, and 97% of land in the Greater Duwamish MIC would be in core industrial zones.
- Employment would remain over 50% industrial.
- Employment would remain far above 20,000 jobs.

Land use changes under Alternative 1 would continue to be consistent with most of the planning goals and policies of the City's Comprehensive Plan, SMP and regional plans. However, an incremental degree of inconsistency would arise with respect to select policies, because of development trends towards continued conversion to office and retail uses in IG zoned areas, and continued development of large offices in IC zoned areas. The resulting land use trend would be somewhat inconsistent with policy 10.2 (preserve land for industrial uses), and 10.17 (avoid attracting large numbers of visitors), and Container Port Element policy CP3 (discouraging retail and residential uses).

Localized areas where the inconsistency would increase would be upland areas in the Ballard, Interbay Dravus, Interbay Smith Cove, and SODO/Stadium subareas. In general inconsistency with policies would be largest in areas in proximity to the future light rail station and in proximity to areas that have strong demand for residential development. This would result in a **moderate** impact in this category due to inconsistency with plans for portions of the study area.

Land Use Compatibility

Under Alternative 1 No Action, land use incompatibilities would be similar to those observed today but would become more severe over time with continuing trends. Expected incompatibilities in localized areas are summarized below:

<u>Ballard</u>

Conflicts in the Ballard uplands in the 14th Avenue corridor north of NW Leary would increase and would manifest as increased difficulty for larger and long-standing industrial operations due to access and congestion constraints as a result of increasing non-industrial office and retail uses. Similar pressure would be exerted on remaining shoreline industrial/marine activities, as visitors of non-industrial activities congest roadways and access points to shoreline operations. Noises, visual impacts, and odors received by an increased number of nonindustrial visitors to the area would also result. However, with limited opportunities for housing and the sporadic nature of nonindustrial visits the impact would be **moderate**.

Interbay Dravus

Incompatible use conflicts would be about the same as today in this area. Maritime and shoreline areas such as Fisherman's Terminal and areas along W Commodore Way would continue to be well-buffered from encroaching uses, and rail yards and facilities would not change substantially. The biggest land use changes would occur in areas near the future rail station between BNSF rail tracks and 15th Avenue W north of W Dravus Street where construction of light rail infrastructure and infill development under IG zoning would be expected. But this triangular area already contains few extensive heavy industrial uses, and few new conflicts exceeding today's level are expected. If parcels adjacent to BNSF tracks develop with non-industrial uses some minor impact due to noise and vibration as experience by future tenants could result but because prohibitions on new residential development would limit nonindustrial activity to commercial uses which are less sensitive to noise and vibration than residential uses, the impact would not be more than **minor**.

Interbay Smith Cove

Incompatible use conflicts would be about the same as today in this area. Maritime and shoreline areas such as Terminal 91 and its upland would continue to be well-buffered from encroaching uses, and rail yards and facilities would not change substantially. Terminal 91 uplands and the Armory site would develop in part or in whole with industrial uses such as distribution space flex, or light industrial space. The effects of such development would not increase incompatibility with adjacent retail and office uses over existing conditions. The biggest land use changes would occur in areas near the future rail station in the Elliott Avenue corridor where construction of light rail infrastructure and infill development under IC and IG zoning would be expected and could include substantial offices. This area already contains few extensive heavy industrial uses, and few new conflicts exceeding today's level are expected. If parcels adjacent to BNSF tracks develop with non-industrial uses some minor impact due to noise and vibration as experienced by future tenants could result but because prohibitions on new residential development would limit nonindustrial activity to commercial uses which are less sensitive to noise and vibration than residential uses, the impact would not be more than **minor**.

SODO/Stadium

Incompatible use conflicts would increase incrementally throughout greater SODO as current trends towards non-industrial retail and office under existing IG zone regulations result in infill on more sites scattered across the area. Existing heavy industrial land uses described above that are sources of noise, odors and glare would be expected to continue in SODO. (i.e., Republic Services, Port Terminals, SIG, and Argo Rail yards etc.). Proximity of these sources to new non-industrial users would create incrementally greater incompatibility. Large industrial

users upland at the center of SODO would have increased difficulty continuing operation due to access constraints as a result of increasing non-industrial office and retail uses.

Maritime and waterfront areas would continue to be well-buffered from encroaching uses in SODO, including Harbor Island, Lower Duwamish Waterway installations and Terminal 5.

Incompatibilities in the stadium area would increase only slightly as more office development in the existing IC zones flanking 1st Avenue materializes. The main source of incompatibility would be exposure of new users/tenants/visitors to heavily trafficked roadways (i.e., SR99) and loud truck traffic that would continue to transit the area. The area contains few extensive heavy industrial uses, and few new conflicts exceeding today's level are expected. If parcels adjacent to the SIG railyard develop with non-industrial uses some minor impact due to noise and vibration as experience by future tenants could result but would not be more than minor.

Overall, in this subarea incompatible use impacts would increase over time but because prohibitions on new residential development would limit non-industrial activity to commercial uses which are less sensitive to noise and vibration than residential uses, the impact and would be **minor**.

Georgetown/South Park

Incompatible use conflicts would increase incrementally in portions of the Georgetown/South Park Subarea. This would be due to current trends towards non-industrial retail and office development under existing IG zone regulations that would result in infill on more sites scattered across the area. Existing heavy industrial land uses described above in existing conditions that are sources of noise, odors and glare would be expected to continue in SODO. (i.e., Argo Rail yards, manufacturers etc.). Proximity of these sources to new non-industrial users would create incrementally greater incompatibility. Large industrial users would have increased difficulty continuing operation due to access constraints as a result of increasing nonindustrial office and retail uses.

Maritime and waterfront areas would continue to be well-buffered from encroaching uses in Lower Duwamish Waterway Georgetown.

Incompatible use impacts would increase incrementally over time, particularly in areas proximate to residential uses, but due to the limited amount of housing the impacts and would be **minor**.

Employment Mix

As seen above in **Exhibit 3.8-13**, the overall employment mix would change only slightly. The employment mix would remain over 50% industrial in both MICs and the study area. Employment projections estimate an addition of 23,500 total jobs, of which about 11,900 would be industrial and 11,600 would be non-industrial. Alternative 1 would result in no more than a **minor** impact to employment mix.

Land Use Transitions

Under Alternative 1 land use transitions are expected to be similar to how they are today. Transition areas are industrial areas with uses that are less intense than core/heavy industrial areas and adjoin areas that are planned for non-industrial areas such as residential neighborhoods or mixed-use commercial areas. Abrupt transitions occur when non-industrial adjacencies are impacted by neighboring high intensity/high impact industrial activities that result in excessive noise, air pollution, noxious odor, or impacts resulting from height bulk and scale of taller buildings in the IC zone where it abuts nonindustrial areas.

Much of Seattle's industrial land has well defined edges (I-5, rail corridors, green belts, waterways) separating industrial areas from non-industrial uses eliminating the potential for encroachment of high intensity/high-impact uses adjacent on residential areas. However, some industrial areas directly abut residential and mixed-use commercial areas. In many cases, these areas are zoned IB which is intended to create a transition from industrial areas through a more limited set of permitted industrial uses and development standards such as setbacks, additional height limits, and landscaping requirements. In some places, the IB zone is applied with a shallow depth, limiting its effectiveness as a transition, and limiting development potential. Similarly, development in IC zones in some areas provides effective transition from core industrial areas to nonresidential areas. In some places where the IC zone abuts nonindustrial areas potential impacts related to height, bulk, scale, and aesthetics exist however, development standards intended to reduce these impacts on adjacent residential zones reduce these potential impacts to a level of insignificance.

<u>Ballard</u>

Industrial zones in the Ballard Subarea directly abut residential zones and mixed-use commercial areas resulting in long-term unavoidable impacts. North of Leary Way, the eastern edge of Ballard land zoned IB provides a transition from core industrial areas, developed with a mix of legally non-conforming residential, warehouse, industrial, storage, and retail uses. On the northern edge of the central portion of the BINMIC, east and west of 14th Avenue W, the transition is abrupt with significant industrial activity adjacent to mixed-use and residential commercial areas. This development is a mix of light industrial, warehouse, parking, and nonconforming residential uses. This land is currently zoned IG2 and while there are currently no high intensity/high impact uses, current zoning would allow such uses over the next 20 years. In northwest Ballard, an abrupt transition exists at the northwest corner of 24th Avenue NW and NW Market Street where maritime activity directly abuts mixed-use zoning as part of the Ballard Hub Urban Village. This area is an important location for maritime industries, including ship servicing, seafood processing, and other activity dependent on the critical maritime infrastructure that is Salmon Bay. An adequate transition exists existing moving west from the mixed-use zoning. The Nordic Museum and other property in the IC zoning on the south side of Market Street and IB zoning on the north side of Market Street provide a strong transition from

the core industrial area. Continued development in industrial areas is expected to be consistent with the existing development pattern and not result in height, bulk, scale, or aesthetic impacts.

The adjacency of IG zoned land and the range of existing and permitted uses in the Ballard Subarea results in a **moderate** impact.

Interbay Dravus and Interbay Smith Cove

The majority of Interbay is defined by hard edges, but long-term unavoidable impacts occur in the area west of the BNSF rail corridor (which in some places directly abuts residential areas) and in the Interbay Dravus Subarea (where industrial uses directly abut multifamily residential development). Operations of the BNSF rail corridor and switching yard results in noise and exhaust from train assembly, and idling locomotives are a high intensity/high impact use. The Interbay Dravus Subarea is a compact node and although the lack of transition will continue in the No Action Alternative, it is confined to a small area and somewhat mitigated by the presence of commercial uses fronting on both sides of Dravus Street. This subarea includes conditions where IG1 zoning directly abuts a mixed-use commercial area substantially developed with housing. The industrial part of this subarea is developed primarily with 1 and 2 story buildings with outdoor storage or parking which is typical of this zone classification. In the Interbay Smith Cove Subarea, This is different than the type of development expected in IC zoning present there will be with 3-4 story, bulkier office buildings although the scale of this development is consistent with other development in the that can result in height, bulk, scale, and aesthetic impacts to adjacent nonindustrial areas. In IC zoned areas, as well as areas at the west of the study area, the transition impacts would be lessened by the buffering features of topography, the green belt, and separation by major roadways. CTherefore, continued development in these industrial areas is expected to be consistent with the existing development pattern and not result in height, bulk, scale, or aesthetic impacts.

T<u>Overall, t</u>he adjacency of the BNSF rail corridor to residential areas in the Interbay Dravus Subarea combined with the lack of transition between industrial and non-industrial uses contributes to a transitions impact that is **a moderate** impact. The lack of transition in the Interbay Dravus Smith Cove Subarea impacts from adjacency to the BNSF rail corridor are present in some areas, but other transition impacts are lessened by mitigating features resultings in an overall minor impact.

SODO/Stadium

The majority of the SODO portion of the Subarea is defined by hard edges including I-5 and the parallel green belt to the east and steep topography and a green belt adjacent to W Marginal Way. However, potential impacts could occur over 20 years on land in Delridge adjacent to the Nucor Steel and on currently vacant land adjacent to Harbor Blvd SW. The Nucor Steel mill is a heavy manufacturing facility adjacent to residential development but transitions in the form of Longfellow Creek Green Space and IB zoning developed with a mix of office and mini-storage that reduces the scale of this ongoing moderate impact. North of Delridge, Harbor Avenue SW

separates the SODO portion of the MIC from mixed residential and commercial development to the west. Most of the industrial land adjacent to Harbor Avenue SW is vacant, used as outdoor storage, or developed as park land and currently provides a transition from adjacent industrial areas including Terminal 5. It is possible over the next 20 years that industrial development could occur on vacant land in the IG2 zone that would introduce high impact/high intensity uses thereby eroding the existing transition conditions and resulting in a **moderate** impact.

To the north, the Stadium district and its focus on spectator sports facilities provides a transition to the Pioneer Square Neighborhood. While the IC zoning to the northeast end of the Greater Duwamish MIC adjacent to the CID is currently developed with a mix of office, transportation, and industrial uses, it is likely in the next 20 years there will be continued office development in the IC zoned parcels in this area creating a stronger transition from core industrial areas to the CID. The IC zoned parcels in this area allow for substantially larger buildings than are found in existing industrial development (up to 175 feet), however, development standards for these IC zones intended to regulate bulk, scale, and aesthetic impacts mean future development on these sites will reduce impacts to insignificant levels.

Adjacent to Nucor Steel and Harbor Blvd the transition impact is **moderate**. Adjacent to Pioneer Square and CID the impact is **minor**.

Georgetown/South Park

Both the Georgetown and South Park neighborhoods abut industrial areas. A transitional strip of IB zoning separates the residential areas from core industrial areas resulting in a **moderate** land use impact. Land uses in the transition area include vehicle storage (WSDOT), a community college just east of Georgetown, and a mix of industrial uses north of South Park. The IB areas represent a shallow transition from core industrial areas and this pattern is expected to continue under the No Action Alternative. Continued development in industrial areas is expected to be consistent with the existing development pattern and not result in height, bulk, scale, or aesthetic impacts.

Impacts from a lack of transition in both areas are **minor**.

Other Industrial Zoned Lands

Industrial land outside the MICs include land in Fremont, the north shore of Lake union, the Southeast shore of Lake Union and the area bound by I-90, Rannier Avenue S, and S Dearborn.

Industrial land in Fremont is zoned with a mix of IB, IC, and IG2 zoning. On the south side of N 36th Street, land is zoned IB and is developed with a mix of commercial uses. An area south of the strip of IB zoned land and fronting N 36th Street is zoned IG2, with land further south adjacent to the ship canal zoned IC. The area currently zoned IG2 is developed with a mix of industrial, commercial, warehouse, and legally nonconforming residential uses. The relative size of the IG2 land compared to core industrial areas and the surrounding zones means it is unlikely to result in development of high impact/high intensity uses that will encroach on or abut non-industrial areas. Continued development in industrial areas is

expected to be consistent with the existing development pattern and not result in height, bulk, scale, or aesthetic impacts. The impact due to lack of transition in this area is low.

- Industrial land on the shoreline south of N Northlake Way is zoned IB. This land falls substantially within the shoreline area and is subject to provisions of the Shoreline Master Program. This area is developed with a mix of office, marina, marine terminal, warehouse, public safety, and park uses. There is little to no potential for high intensity/high impact uses to encroach on nonresidential areas. Development regulations including height limits, FAR limits, and view corridor requirements of the Shoreline Master Program means impacts resulting from height, bulk, scale, and aesthetics are not anticipated. North of N Northlake Way, land contiguous to the IB zoned land to the south at Stone Way N is zoned IC and is developed with a range of office and retail uses. The industrial area defined by N Northlake Way, I-90, and N Pacific Street is zoned IC. This area is currently developed with a mix of warehouse, office, light industrial, and parking uses. There is limited potential for high intensity/high impact uses to encroach on nonindustrial areas in all of these areas. Although IC zoning allows for development 3 to 4 story office buildings with greater bulk and scale than is typical of other industrial zones, development regulations for development projects in the IC zone that abut residential areas mean impacts related to height, bulk, scale, and aesthetics are not anticipated.
- The industrial area near I-90 is currently zoned IC and is developed with a mix of recreational, office, and warehouse uses. Because of the compact nature of this area, the hard edge of I-90 to the west and south, and significant arterials to the east and north that separate this area from nonindustrial areas and the IC zoning of this land, there is no potential for high intensity/high impact uses to encroach on nonresidential areas or for impacts resulting from incompatible height, bulk, scale, or aesthetics.
- The industrial area on the shoreline of southeast Lake Union is zoned with a mix of IG1 and IC. The IG1 portion of this area is currently developed with a seafood processing company and a drydock facility and falls substantially within the shoreline zone. Development regulations including height limits, FAR limits, and view corridor requirements of the Shoreline Master Program means impacts resulting from height, bulk, scale, and aesthetics are not anticipated. IC land up shore from the IG1 land is developed with R&D facilities. There is no potential for impacts resulting from encroachment of high intensity/high impact uses at this location because the IG1 land is shoreline and water and is subject to the provisions of the SMP which would preclude such impactful uses. The IC area is developed with R&D and office uses which do not encroach on non-industrial areas.

Impacts resulting from inadequate transition for industrial to nonindustrial areas outside of the MICs is **minor**.

Impacts of Alternative 2

Likely Changes Over the 20-year Planning Horizon

Land Use Planning & Policy. Under Alternative 2—Future of Industry Limited, the planning and policy context would be changed to enact the Comprehensive Plan policy amendments described above in **Local Policy Framework**. The City would also adopt updates to the currently adopted Sub Area Plans for the Greater Duwamish MIC and BINMIC which include the land use concepts identified in this proposal.

Future Land Use. Under Alternative 2 the future Land Use Map would not change. Boundaries of the BINMIC and Greater Duwamish MIC would not be altered, as no land is removed from MICs under Alternative 2. "Industrial Areas" designation on the FLUM outside of MICs would not be changed. Industrial zoned land within the FLUM designated urban villages would remain in that designation.

Zoning. Under Alternative 2 zoning would be changed to apply the proposed new Maritime, Manufacturing and Logistics (MML), Industry and Innovation (II), and Urban Industrial (UI) zones, instead of the existing zones. The Seattle Municipal Code would be amended to add the development standards in the MML, II and UI zones as described in **Chapter 2**, including retention of a Stadium Area Overlay District. The location of the zones in Alternative 2 is mapped as shown in **Chapter 2** and **Appendix C**.

Alternative 2 applies the proposed new industrial zones with relatively less Industry and Innovation and Urban Industrial than the other two Action Alternatives. Under Alternative 2:

- The maritime, manufacturing and logistics zones would cover 89% of industrial lands.
- A mix of Industry and Innovation and Urban Industrial Zones would cover 11% of the study area including an estimated ¼ mile from light rail stations.
- There would be no expansion of housing allowances in the UI zone

Land Use. Under Alternative 2, land use would change over the planning period according to current trends and as a result of the proposed zoning changes. Some notable expected changes include.

- Decreased rate of conversion to stand-alone office and retail uses in MML zoned areas. The new MML zone would have stricter size of use and FAR limits for stand-alone office and retail uses. As a result, there would be fewer conversions to stand-alone office and retail than past trends and under Alternative 1.
- Continued distribution and warehouse facilities. Strong demand for new warehouse and distribution space is expected to continue, resulting in the addition of new distribution and warehouse facilities in MML zoned areas.
- Maintenance of maritime and industrial base. Most long term maritime and logistics
 uses would continue on waterfront and industrial lands near infrastructure, especially in the
 Greater Duwamish MIC. New Comprehensive Plan policies limiting the removal of land from
 MICs will provide existing industrial land uses with the kind of long-term predictability that
 will increase onsite reinvestment for continued industrial use could be expected at a greater
 rate than under Alternative 1.

- Denser employment including new industrial space near future light rail station in the limited II zoned areas. The proposed II zone regulations combined with expected strong market interest due to increased access provided by light rail stations is likely to result in development with a high density of employment in ICT and Office development sectors in these areas. However, the mapped locations of the II zone are limited in the alternative. New development in the II zoned areas would include new light industrial space at ground level. Much higher levels of employment, and general activity by employees and visitors is expected. Associated frontage improvements and infrastructure upgrades would also be expected with the changed character and activity pattern in these nodes.
- Increased development of mixed-use, flex, and light industrial uses in UI zoned areas. The proposed UI zone regulations combined with expected strong market interest due to proximity to population centers will lead to incremental addition of new buildings with light industry, office, and flex space in areas at the edges of MICs near urban villages. Increased ancillary uses for breweries, retail showrooms and similar will incrementally increase use of the area by non-industrial populations. Frontage improvements, infrastructure upgrades, and increased landscape would be expected. The physical character in these edge areas would become more urban in nature with more buildings built to lot lines.
- Armory Site Redevelopment. Under Alternative 2 the Armory site would be developed with light industrial and flex space of a relatively low-density nature or remain vacant after relocation of the Army National Guard to North Bend, WA.
- Little or no new housing. Only an estimated 80 new homes would be added in caretakers' quarters and artist/studios.

Employment Mix. Under Alternative 2, employment is projected to grow substantially more than under Alternative 1 No Action. A total of 34,400 additional jobs are projected for the study area, an increase of 35%. This would represent about 20% of the projected citywide employment growth over the 20-year planning horizon. The mix of industrial employment would increase by 4.4% points compared to the No Action Alternative, up to 59.7%. Both MICs would continue to contain much more than the minimum number of industrial jobs required to meet PSRC's regional criteria for MIC designation (20,000). The percentage of industrial employment would remain at roughly 58% or greater in every subarea under the alternative. See **Exhibit 3.8-14**.

	Current Conditions (2018)			Alternative 2—Future of Industry Limited (2044)		
Subarea	Ind. Emp.	Total Emp.	% Ind.	Ind. Emp.	Total Emp.	% Ind.
Ballard	9,400	17,100	55.0%	13,600	23,600	57.6%
Interbay Dravus	3,400	5,600	60.7%	4,900	7,700	63.6%
Interbay Smith Cove	3,900	6,000	65.0%	5,800	8,600	67.4%
SODO/Stadium	23,000	43,900	52.4%	33,700	57,700	58.4%
Georgetown/South Park	14,900	25,900	57.5%	21,400	35,300	60.6%
Total	54,500	98,500	55.3%	79,400	132,900	59.7%

Exhibit 3.8-14-13 Employment by Subarea, Current Conditions and Alternative 2

Source: City of Seattle, 2021.

Consistency with Plans & Policies

Under Alternative 2, conditions in both the Greater Duwamish MIC and the BINMIC would still meet PSRC's regional criteria for designation as an Employment Center MIC.

- 75% land zoned for core industrial uses: Under Alternative 2, the new zones (MML, II, and UI) would be considered core industrial zones satisfying the PSRC criteria, because development under the standards in all three zones would include industrial development. Therefore, 100% of the land in the BINMIC and Greater Duwamish MIC would be zoned for core industrial purposes.
- Employment would remain over 50% industrial for the MICs as a whole and for all subareas.
- Employment would remain far above 20,000 jobs.

Land use changes over the time horizon would be consistent with Comprehensive Plan goals and policies. Updates to goals and policies are an integrated part of Alternative 2 and the new zones and development that would occur in them would be crafted to advance those policies.

Land use patterns would be consistent with the plan's goals and policies concerning protections for industrial and maritime uses in core areas, such as land use goal 10, and policies 10.2, 10.3 and 10.4. Future development in the MML zone would afford stronger protections (compared to existing IG zones) for industrial uses such as lower maximum size of use limits and FAR limits for non-industrial uses and prohibition of mini-storage uses. Limiting removal of land from MICs to major plan updates would also provide stronger protection in accord with these policies.

Land use changes expected over time under the new II and UI zones would be consistent with the plan's amended goals and policies including LUG11 and LUG12. New or amended policies including 10.6 address integration of land use with high-capacity transit. Development in the II zone would be consistent with new policies supporting dense employment and emerging industries near transit, including policies 10.7 and 10.19. New or amended plan policies would promote transitions at edges of MICs that integrate with nearby urban villages including 10.7, and 10.22-24. Development in the II zone would be built with reduced setback requirements, large ancillary size of use limits, and urban landscaping standards that would cause new buildings to augment transitions in line with the policy intent.

However, an incremental degree of inconsistency could arise with respect to select policies under Alternative 2, because there is some increased potential for denser development in the II and UI zones to adversely affect traditional heavy industrial uses. If robust development under the new II and UI zones occurs, there could be some incremental inconsistency with policies 10.12 (concerning limiting density in MICs), policy 10.13 (concerning limiting landscaping requirements in industrial areas), and 10.18 (concerning avoiding attracting large numbers of visitors), and Container Port Element CP3 (concerning discouraging retail and residential uses).

Alternative 2 would increase the share of projected employment growth in industrial areas to about 20% of total citywide job growth that the city would be planning for during the 20-year planning horizon. This would represent a shift of a moderately greater share of the city's

expected employment growth into industrial areas compared to past trends and the previous 20-year Comprehensive Plan planning horizon.

Overall consistency with regional plans would be maintained, and consistency with the City's Comprehensive Plan goals and policies would increase compared to Alternative 1. Although there is potential for slight inconsistency with a few policies, land uses under Alternative 2 would be strongly consistent with most policies and impacts related to consistency are **minor**.

Land Use Compatibility

<u>Ballard</u>

Most land in the Ballard uplands in the 14th Avenue corridor north of NW Leary would be placed in the MML zone. Over time some use conflicts would likely be reduced here because stricter maximum size of use limits for non-industrial would reduce pressure to convert land to standalone retail and offices, compared to Alterative 1. A stronger and more consistent industrial use pattern would evolve over time, and longstanding industrial operations would be afforded relative ease of operation concerning truck movements, and insulation from complaints about noises and odors. There is some potential for use conflict between an increasingly consistent industrial use pattern south of NW 53rd Street and increased volumes of passersby through the area to a transit station. See also **Section 3.10 Transportation**.

Dense employment in multi-story buildings would likely be added in the two blocks of the II zone between NW 53rd and NW 54th Street near a potential future light rail station, and II zoned areas in Fremont that already contain a high concentration of dense employment. No major use conflict would be expected in these locations.

Due to conversion to the MML zone for lands abutting the shoreline, incompatible use pressures in areas of Ballard south of Leary Way would be lessened over time, compared to Alternative 1. Existing use incompatibilities in the BINMIC due to proximities between retail and office land uses and industrial and maritime activity related to noise, congestion etc., would continue near existing levels, but are not expected to increase markedly.

Increased infill development with light industrial uses and brewers/makers with large ancillary spaces is expected in areas at the edges of Ballard, and along NW 36th Street in Fremont. New zone standards would allow smaller parcels to accommodate new structures. Uses that appeal to visitors from nearby urban villages such as showrooms and tasting rooms are expected to occupy new structures. Resulting land use patterns and expected times of day for activity would be consistent with the adjacent areas outside of the study area. There is some potential for increased volumes of visitors to create minor use conflicts with remaining heavy industrial uses in MML zones at the interior.

Overall impacts resulting from land use conflicts in Ballard would be **minor** under Alternative 2.

Interbay Dravus

Under Alternative 2, all the shoreline and adjacent lands including Fisherman's Terminal, W Commodore Way lands, and the BNSF railyard would be placed in the MML zone. Stricter maximum size of use limits would reduce pressures for conversion to non-industrial uses in these areas, and over time a stronger and more consistent industrial use pattern would emerge compared to Alternative 1. Use conflicts between operation of maritime and heavy industry related to noise, local truck access and similar would be reduced compared to Alternative 1.

Land north of Dravus Street along Thorndyke Avenue W that would be in the UI zone would likely receive incremental infill development with light industrial uses, brewers/makers with large ancillary spaces, including on some smaller parcels. The uses would appeal to visitors from nearby residential areas and by those using light rail transit. Some land would likely be used for light rail construction and operation. These changes would cause an overall transition of the 14-acre area to an urban mixed-use pattern (albeit without housing). Impact of this transition would be minor, since no very heavy or largescale industrial uses are located in the small area, and the area is contained by defined edges of the rail track.

Overall use impacts resulting compatible land uses in the Interbay Dravus Subarea would be **minor** under Alternative 2.

Interbay Smith Cove

Under Alternative 2, land use patterns in the Interbay Smith Cove Subarea would not change markedly from current conditions, and use incompatibilities are not expected to increase in severity compared to Alternative 1. The W Armory Way corridor has been developed with a mix of retail ministorage uses that are expected to remain in place, and since few adjacent heavy industrial activities remain, there is not a high degree of use conflict at present.

The Armory site would be in the MML zone and would likely be developed with light industrial and flex space of a relatively low-density nature including activities such as distribution and warehousing. Such uses are not expected to conflict with the surrounding context due in part to the large site that can contain activities and provide buffering at edges. Some minor use incompatibilities could arise due to increased volumes of truck entering and exiting the large site via routes including W Armory Way which also provides access to the non-industrial retail uses.

Marine Terminals and T91 uplands would be placed in the MML zone. Marine activities, and industrial use similar to existing conditions will continue on those lands and would not create additional land use conflicts.

Areas zoned Industry and Innovation not already developed with offices in the Elliott / 15th Avenue W corridor would be likely to receive some additional dense employment development in multi-story buildings. The use pattern by daytime employees would be similar to adjacent uses such as the Expedia campus.

Overall use compatible impacts in the Interbay Dravus-<u>Smith Cove</u> Subarea would be **minor** under Alternative 2.

SODO/Stadium

Under Alternative 2, all the shoreline and adjacent lands including Port Terminals, and expansive stretches of land currently zoned IG would be placed in the MML zone. Stricter maximum size of use limits would reduce pressures for conversion to non-industrial uses in these areas, and over time a stronger and more consistent industrial use pattern would emerge compared to Alternative 1. In MML zoned areas land use conflicts between operation of maritime and heavy industry related to noise, local truck access and similar would be reduced compared to Alternative 1.

Under Alternative 2 limited areas of the Industry and Innovation zone are added in a close ¼ walking areas to the SODO/Lander Street station, on the WOSCA site and on land north of I-90, all of which would be likely to receive some additional dense employment development in multi-story buildings, with light industrial uses integrated at ground level.

Minor new use conflicts would be likely in the area around the SODO/Lander Street light rail station, as a significantly higher volume of daytime workers, unrelated to industrial operations would be present. Minor conflicts could include new exposures of pedestrians and workers using outside spaces to loud truck traffic and industrial equipment and to odors from industrial operation such as the Republic Transfer station. Presence of workers could increase difficulty of fluid freight movement including difficulty with operation of loading docks and site access. However, since the geographic area zoned II is tightly limited to the area around the station, impacts would be minor.

The II zoned area between 4th Avenue S and I-90, the Rainier Avenue S corridor, and the WOSCA site would be likely to receive additional dense employment development in multi-story buildings and an increased volume of daytime workers. However, the use pattern by daytime employees would be like the adjacent existing uses such as Union Station, Home Plate Center, and other development in the permitting process. New development and uses are expected to be compatible with adjacent Chinatown/ID and Little Saigon neighborhoods. Ground floors in II developments would include new light industrial space, and there is a demand for such space in Chinatown/ID and Little Saigon by distributors of goods including produce and restaurant supply. Employees of office, R&D, and ICT uses would likely provide increased customer base for restaurant and service uses in Chinatown/ID.

Land in the stadium area in the UI zone would likely receive incremental infill development with light industrial uses, brewers/makers with large ancillary spaces, including on some smaller parcels. Some additional lodging uses would be expected due to the change to permit lodging in the Stadium Area Overlay District (STAOD). Continued addition of large-sized office and retail uses are expected in the STAOD. The uses would appeal to visitors from nearby residential areas and patrons of stadium events. These changes would cause an overall transition of the area fronting 1st Avenue to an urban mixed-use pattern (albeit without housing). Some minor impacts could result due to an incremental increase in exposure of pedestrian activity near trucks transiting on 1st Avenue and accessing I-90 and SR99 freeways.

Overall use compatibility impacts in the SODO/Stadium Subarea would be **minor** under Alternative 2.

Georgetown/South Park

Under Alternative 2, all riverfront lands including Port Terminals, marine operations, and expansive stretches of land currently zoned IG would be placed in the MML zone. Stricter maximum size of use limits would reduce pressures for conversion to non-industrial uses in these areas, and over time a stronger and more consistent industrial use pattern would emerge compared to Alternative 1. In MML zoned areas use conflicts within the MIC between operation of maritime and heavy industry related to noise, local truck access and similar would be reduced compared to Alternative 1.

Increased infill development with light industrial uses, brewers/makers, and small manufacturers with large ancillary spaces is expected in areas at the edges of South Park Urban Village and the Georgetown residential areas. New zone standards would allow smaller parcels to accommodate new structures. Uses that appeal to visitors from nearby urban villages such as showrooms, tasting rooms and similar are expected to occupy new structures. Resulting use patterns, and times of day for activity, would become more consistent with the adjacent areas outside of the study area in South Park. There is some potential for increased volumes of visitors to create minor use conflicts with heavy industrial uses in MML zones at the interior.

Existing use conflicts would persist in the triangular area of Georgetown bounded by Corson Avenue S, Carleton Avenue S, and I-5 where there are a high concentration of retail uses on Airport Way S. A primarily industrial character would remain and increase in the areas in the western portion of the triangle due to stricter limits on non-industrial uses in the MML zone. Land use incompatibilities with the existing track 101 spur would remain the same or potentially decrease over time under Alternative 2. This Alternative 2 would solidify a break in the continuity between the two residential portions of Georgetown neighborhood that exists today, which is a **minor** adverse land use impact.

Employment Mix

As seen above in **Exhibit 3.8-14** the overall employment mix would change incrementally. The mix of industrial employment would increase by 4.4% points compared to No Action, up to 59.7%. The percentage of industrial employment would remain at roughly 58% or greater in every subarea under the alternative. Although there would be an increase in non-industrial employment in office and ICT sectors, the increase in industrial employment due to stronger protections in the MML zones, inclusion of new light industrial space in II zone development, and industrial redevelopment of the Armory site would result in bigger increases in industrial employment than Alternative 1. No adverse impact is expected.

Land Use Transitions

Land use transitions under Alternative 2 remain much as they are under Alternative 1—No Action alternative. Most IB zoning is replaced with the new UI zone to create a scale of development and uses compatible with nearby non-residential areas and provide a transition from high intensity/high impact industrial uses in core industrial areas. Development in this

zone would be higher density than the IB zone with a finer grained development pattern consisting of makers spaces, light industrial uses. In addition to less intense industrial activity, these areas will generate pedestrian activity by including opportunities for more ancillary retail and showroom space. Because Alternative 2 applies the UI zone in the same pattern as the IB zone in the No Action Alternative, the narrow application of this zone in some areas limits the degree to which these areas developed as intended. However, it is expected that the UI zone will allow for an increased amount of infill development on small sites due to decreased setbacks compared to the IB zone.

Alternative 2 also replaces limited portions of land in the current IC zone classification with the core industrial zone, the MML zone, in locations that abut nonindustrial areas. This change could result in high impact/high intensity uses adjacent to nonindustrial areas resulting in an incremental impact due to lack of transition.

<u>Ballard</u>

In northeast Ballard the existing IB zoning is replaced with UI zoning and provides a narrow transition between the core industrial area to the west and non-industrial areas to the east and north. In the Central part of Ballard introduction of the II zone in the area of 14th Avenue NW and NW 54th Street could develop with a mix of industrial and commercial uses providing a narrow transition from the core industrial area. Because the UI zone will larger buildings than is typical of industrial areas, there is potential for impacts related to height, bulk, scale, and aesthetics where it abuts residential areas in northeast Ballard. In the western portion of the Ballard Subarea, expansion of the core industrial zone into areas currently zoned IC could result in introduction of high intensity/high impact uses adjacent to non-residential uses in the Ballard Hub Urban Village. Due to the limited size of this condition, this is a **moderate** impact.

<u>Interbay</u>

Like Alternative 1—No Action, most of this subarea is defined by hard edges except for the node anchored by W Dravus Street and the area adjacent to the BNSF rail corridor <u>in some places</u>. A stronger transition occurs in the Dravus area by applying the UI zone where land is currently in a core industrial zone. This will result in finer grained development of light industrial and makers spaces and anticipation of increased pedestrian activity that provides better compatibility with the adjacent residential development. The nature of the UI zone of encouraging pedestrian activity, and structures of a similar bulk and scale as the adjacent mixed-use zoning means there will be minimal impacts related to height bulk and scale. THowever, the presence of residential uses adjacent to the BNSF rail corridor will continue to result in long-term unavoidable impacts in the Interbay Dravus Subarea and in limited locations at the west edge of the Interbay Smith Cove Subarea. Similar to Alternative 1—No Action, in the scale of development in the II zone would be similar to that of surrounding areas in the Smith Cove Subarea, and features such as greenbelts, topography and separation by major roadways counteract the potential for transition impacts and no impacts.

-<u>For the Interbay Dravus Subarea due to the Aa</u>djacen<u>cy of residential uses </u>to the BNSF rail corridor<u>and some potential for transition impacts in the UI zoned area</u>, the <u>overall transition</u> impact is **moderate**, while transition impacts would be **minor** for the Interbay Smith Cove <u>Subarea</u>. In Dravus the impact is low.

SODO/Stadium

The impacts due to a lack of transition from core industrial areas to nonindustrial areas for Alternative 2 are similar to <u>but somewhat less than</u> Alternative 1—No Action. The IB zone in Alternative 1—No Action adjacent to the Nucor Steel plant will be rezoned to UI and the area along Harbor Boulevard would change from existing IG2 and limited IB zoning to the MML zone with a similar range of permitted uses and scale of development as existing zoning. Likewise, to the north the areas adjacent to Pioneer Square and the CID would see zoning changes from IC zoning to a mix of UI and II zoning with a similar reduction of impacts overtime as redevelopment occurs. The potential impact from encroaching or abutting high intensity/high impact uses <u>in the Delridge area would remain **moderate**, while in other areas included adjacencies to South Downtown or from-impacts related to heigh, bulk, scale, and aesthetics inn nonresidential areas is<u>result in a</u> **minor** overall transition impact levellow.</u>

Georgetown/South Park

The impacts in the subarea are expected to be similar to Alternative 1—No Action. Alternative 2 mirrors the existing zoning pattern by changing areas adjacent to Georgetown and South Park from IB to UI. Because the UI zone will allow for taller structures with ancillary housing than is allowed in the current IB zone there is potential for impacts related to height, bulk, scale, and aesthetics. However, due to the shallow depth of this zoning in areas where it abuts nonresidential areas the potential impact from encroaching or abutting high intensity/high impact uses on nonresidential areas or the potential for impacts associated with height, bulk, scale, and aesthetics is **minor**low.

Other Industrial Zoned Lands

In Fremont the impacts from lack of transition are the same as Alternative 1—No Action. Alternative 2 proposes to leave the land use pattern unchanged with UI replacing the IB zone, MML replacing the IG2 zone, and II replacing the IC zone. On the north shore of lake union, Alternative 2 proposes changing the IB zone to the more intensive MML zone but the narrowness of the strip and development regulations of the SMP preclude the potential for development of high intensity/high impact uses in this area. IC zoning is proposed to be changed to II and will result in the same level of impact as Alternative 1—No Action. The southeast Lake Union industrial area will continue to not have impacts resulting from inadequate transition from core uses. Alternative 2 proposes changing the IG1 zoning to MML and the IC zoning to II. The potential impact from encroaching or abutting high intensity/high impact uses or impacts related to height, bulk, scale, and aesthetics on nonresidential areas is **lowminor**.

Impacts of Alternative 3

Likely Changes Over the 20-year Planning Horizon

Land Use Planning & Policy. Under Alternative 3—Future of Industry Targeted, the planning and policy context would be changed to enact the Comprehensive Plan policy amendments described above in **Local Policy Framework**. The City would also adopt updates to the currently adopted Sub Area Plans for the Greater Duwamish MIC and BINMIC which include the land use concepts identified in this proposal.

Future Land Use. Under Alternative 3 the Future Land Use Map would be amended slightly. Boundaries of the Greater Duwamish MIC would be altered to remove focused land near Georgetown and South Park from the MIC designation, as indicated on the map in **Chapter 2** and **Appendix C**. In Georgetown, the triangular area bounded by Corson Avenue S, Carleton Avenue S, and I-5 would be placed into the Commercial/Mixed-Use Areas designation. In South Park select parcels at the northeast and southeast corners of the urban village adjacent to the Duwamish River would be removed from the MIC and placed in the South Park Urban Village. The total area of lands removed from the MIC would be 26 acres. No land would be removed from the BINMIC, and no other Comprehensive Plan FLUM designations would change.

Zoning. Under Alternative 3, zoning would be changed to apply the proposed new MML, II, and UI zones, instead of the existing zones. The Seattle Municipal Code would be amended to add the development standards in the MML, II and UI zones as described in **Chapter 2**, including the retention of a Stadium Transition Area Overlay District The location of the zones in Alternative 3 is mapped as shown in **Chapter 2** and **Appendix C**.

Alternative 3 applies the proposed new industrial zones with a greater share of II and UI zones than Alternative 2. Alternative 3:

- Applies the MML zones covering 86% of industrial lands.
- Applies a mix of II and UI zones in 14% of the study area including an estimated 1/2 mile from light rail stations.
- Expands allowances for limited industry-supportive housing in UI Zone concept with a maximum density of 25/dwelling units per acre.
- Applies mixed-use zoning to the areas of Georgetown and South Park that are removed from the MIC. Neighborhood Commercial with a 75' height limit or a 55' height limit could be applied. The higher scale 75' zone is analyzed for impact analysis purposes. An MHA (M1) suffix are assumed for analysis.

Land Use. Under Alternative 3, land use would change over the planning period according to current trends and as a result of the zoning changes of the alternative. Some notable expected changes include.

 Decreased rate of conversion to stand-alone office and retail uses in MML zoned areas. The new MML zone would have stricter size of use and FAR limits for stand-alone office and retail uses and a prohibition on mini-storage. As a result, there would be fewer
conversions to stand-alone office, retail, and mini-storage than past trends and under Alternative 1.

- Continued additions of distribution and warehouse facilities. Strong demand for new warehouse and distribution space is expected to continue, resulting in the addition of new distribution and warehouse facilities in MML zoned areas.
- Maintenance of maritime and industrial base. Most long standing maritime and logistics uses would continue on waterfront lands and industrial lands near key industrial infrastructure, especially in the Greater Duwamish MIC. Incremental renewal of facilities and buildings for industrial use could be expected at a greater rate than under Alternative 1.
- Denser employment including new industrial space, near future light rail station in II zoned areas. The proposed II zone standards combined with expected strong market interest due to increased access provided by light rail stations is likely to result in development with a high density of employment in new buildings for Information Computer Technology and offices in these areas. The development would also include new light industrial space at ground level. Much higher levels of employment, and activity pattern of employees and visitors than Alternative 1—No Action is expected. Associated frontage improvements and infrastructure upgrades would be expected.
- Increased development of mixed-use, flex, and light industrial uses in UI zoned areas. The proposed UI zone regulations combined with expected strong market interest due to proximity to population centers will lead to incremental addition of new buildings with light industry, office, and flex space in areas at the edges of MICs near urban villages. Incremental infill development will add density of activity and employment, serving nonindustrial populations. Frontage improvements and infrastructure upgrades and increased landscaped areas would be expected. The physical character in these edge areas would become more urban in nature with more buildings built to lot lines.
- Introduction of some new industry-supportive housing. Under Alternative 3 about 610 new homes would be added in UI zoned portions of industrial areas due to increased flexibility for caretakers' quarters and artist/studios. With Introduction of more housing changes use in activity patterns are expected, as more 24-hour presence of residents living in areas at the edges of MICs.
- Additional new housing in areas removed from the Greater Duwamish MIC. More housing would be added in mixed-use buildings in areas removed from MICs in Georgetown and South Park. This housing would contribute to the expansion of a mixed-use, urban neighborhood character in both locations. The added housing capacity is anticipated at 1,078 units.
- Armory Site Redevelopment. Under Alternative 3 the Armory site would be redeveloped with a mix of ICT/office and include new light industrial space at ground level after relocation of the Army National Guard to North Bend, WA. The site would contain a substantial amount of employment density in a new campus-like setting with integrated open space features and new roadway, utilities, and infrastructure, including integration of green infrastructure.

Employment Mix. Under Alternative 3, employment is projected to grow substantially more than under Alternative 1 No Action and more than Alternative 2. A total of 57,400 additional jobs are projected for the study area, an increase of 58%. This would represent 34% of the city's total expected job growth over the 20-year planning horizon. The mix of industrial employment would decrease by 1.7%% points compared to Alternative 1—No Action, down to 53.6%. Both MICs would continue to contain much more than the minimum number of industrial jobs required to meet PSRC's regional criteria for MIC designation (20,000). The percentage of industrial employment would decrease close to the 50% threshold in the Ballard (51.1%) and SODO/Stadium (52.6%) subareas. See **Exhibit 3.8-15**.

	Currei	Current Conditions (2018)			Alternative 3—Future of Industry Limited (2044)		
Subarea	Ind. Emp.	Total Emp.	% Ind.	Ind. Emp.	Total Emp.	% Ind.	
Ballard	9,400	17,100	55.0%	15,900	31,100	51.1%	
Interbay Dravus	3,400	5,600	60.7%	5,500	9,900	55.6%	
Interbay Smith Cove	3,900	6,000	65.0%	6,300	10,500	60.0%	
SODO/Stadium	23,000	43,900	52.4%	34,700	66,000	52.6%	
Georgetown/South Park	14,900	25,900	57.5%	21,100	38,400	54.9%	
Total	54,500	98,500	55.3%	83,500	155,900	53.6%	

Exhibit 3.8-15-14 Employment by Subarea, Current Conditions and Alternative 3

Source: City of Seattle, 2021.

Consistency With Plans & Policies

Impacts regarding consistency with plans and policies under Alternative 3 are the same as described for Alternative 2 with additional impacts related to housing. Alternative 3 includes an estimated additional 610 limited industry supportive housing units in industrial zones. The housing would be available to business owners or employees of an on-site business that is an industrial use, or available to artists/makers with a business license in live-work spaces. Live/workspaces contain area for production/art/making activities that are physically connected to residential space. Limitations on who may occupy the housing is expected to mitigate the impact of the introduction of residential use (see discussion in **Mitigation Measures**). the homes are considered residential uses for the purposes of environmental analysis in this section.

Alternative 3 would increase the share of projected employment growth in industrial areas to about 34% of total citywide job growth that the city is planning for during the 20-year planning horizon. This would represent a substantial shift of the city's expected employment growth into industrial areas compared to past trends and the previous 20-year Comprehensive Plan planning horizon. This could have the effect of curtailing recently high rates of job growth in other areas of the City such as Downtown and South Lake Union compared to past

comprehensive planning period. Or, if the city receives greater job growth than the 20-year citywide estimate, robust employment growth in the study area would provide the benefit of absorbing some of the city's stronger than expected growth. Since the Comprehensive Plan major update is expected to integrate and plan for the changes contemplated in this EIS the share of employment growth in the study area is not considered an adverse impact.

An incremental increase in conflicts would arise with regional multi-county and PSRC policies that discourage location of new housing in MICs, including (e.g., MPP-Ec-22 and MPP-DP-). Similarly, a degree of inconsistency would arise with the City's Comprehensive Plan policies including LU Policy 10.12 (competition for industrial land by non-industrial uses) and Container Port CP.3 (discouraging retail and residential uses). Overall impacts to consistency with plans and policies due to introduction of housing would result in **moderate** impacts.

Land Use Compatibility

Ballard

Under Alternative 3 land in the Ballard uplands in the 14th Avenue NW corridor north of NW Leary would be placed in the UI zone, and the zone would allow industry supportive housing at a maximum density of 25 dwelling units / acre. A substantial amount of increased infill development with light industrial uses, brewers/makers with large ancillary spaces is expected throughout this area and along NW 36th Street in Fremont. Proximity to light rail would fuel demand. New zone standards would allow small parcels to accommodate new structures. An additional 260 housing units are estimated, and they would typically be located on an upper floor of a 3-4 story mixed-use development. Ground floor uses would appeal to visitors from nearby urban villages such as showrooms, tasting rooms and similar.

These changes would cause an overall and thorough transition of the area to an urban mixeduse pattern with some 24-hour residences interspersed sporadically throughout. Compatibility impacts would increase between remaining longstanding industrial operations and the evolving context due to factors such as impediments to local truck access, and increased exposure of new residents and patrons/visitors to industrial noises and other effects. These compatibility impacts would likely increase pressure on intensive or heavy industrial activities to relocate form the area over time and would rise to the level of moderate. However, use compatibility could also improve to some degree as new investment brings associated streetscape improvements and landscaping that would reduce conflicts between pedestrians and freight movement by increasing amenity features and vegetation that would buffer non-industrial visitors or residents from effects of heavier industrial uses. The resulting use patterns, and times of day for activity, would become more consistent with the adjacent areas outside of the study area.

Due to conversion to the MML zone for lands abutting the shoreline, incompatible use pressures in areas of Ballard south of Leary Way would be lessened over time, compared to

Alternative 1. Existing land use incompatibilities here would continue near existing levels and would be the same as described above for Alternative 2.

Dense employment in multi-story buildings would likely be added in the two blocks of the II in Fremont that already contain a high concentration of dense employment. Similar to Alternative 2, No major land use conflict would be expected in these locations.

Overall use compatibility impacts in Ballard would be **moderate** under Alternative 3.

Interbay Dravus

Under Alternative 3, all of the shoreline and adjacent lands including Fisherman's Terminal, W Commodore Way lands, and the BNSF railyard would be placed in the MML zone. Use land use conflicts in industrial areas would be the same as described above under Alternative 2 and reduced compared to Alternative 1—No Action.

Land north of Dravus Street along Thorndyke Avenue W that would be in the UI zone as in Alternative 2, however in Alternative 3 the zone would allow for supportive housing at a maximum density of 25 dwelling units / acre. An additional 75 housing units are estimated, and they would typically be located on an upper floor of a 3-4 story mixed-use development. Similar to Alternative 2, the areas would likely receive incremental infill development with light industrial uses, brewers/makers with large ancillary spaces, including on some smaller parcels. Some land would likely be used for light rail construction. These changes would cause an overall transition of the area to an urban mixed-use pattern with housing units sporadically introduced throughout. However, no very heavy or largescale industrial uses are in the small area, and the area is contained by defined edges of the rail track and 15th Avenue W.

The limited size of land in this node and the relatively small number of projected homes are factors that keep overall use compatible impacts in the Interbay Dravus Subarea to a degree of **minor** under Alternative 3.

Interbay Smith Cove

Under Alternative 3 the Armory site would be in the II zone and would likely be developed with a significant amount of dense employment in multistory structures, including some towers, with dedicated space for ground floor light industry. Development would be coordinated through master planning to create a campus like setting with interconnected circulation, open spaces, and infrastructure. Since development would be coordinated, light industry would be integrated such that potential use conflict are minimized with respect to factors such as noise, access, glare, and odors. Such redevelopment would contribute to a strong agglomeration of daytime employment uses in conjunction with the existing Expedia campus and offices in the Elliott Way corridor. The resulting use pattern would be largely compatible with surrounding context towards the Queen Anne, Uptown urban villages, and the W Armory Way corridor that has already been converted to include a prevalence of retail uses. There is some potential for land use incompatibility at the west edge of the Armory site abutting BNSF rail tracks where

vibrations, noise from trains could impact new office uses. Due to integrated design of the Armory site, and findings in other sections of this EIS, such impacts would not be more than minor.

As with Alternative 2, the T91 Marine Terminals and T91 uplands would be placed in the MML zone. Marine activities, and industrial uses similar to existing today would continue on these lands and would not create additional land use conflict.

For other parts of Interbay /Smith Cove, use compatibility aspects of Alternative 3 would be similar to Alternative 2. However, there is increased potential for incompatibility in UI zoned areas in the four blocks along 15th Avenue NW where an estimated 15 housing units would be located. Some land use conflicts resulting from a high volume of truck traffic and presence of 24-hour residences could occur, but the small overall quantity of residences would keep impact to a minor level.

Overall use compatible impacts in the Interbay Smith Cove Subarea would be **minor** under Alternative 3.

SODO/Stadium

Under Alternative 3, of the shoreline areas and adjacent lands including Port Terminals, and land currently zoned IG would be placed in the MML zone. Compatibility impacts there would be similar to Alternative 2 and reduced compared to Alternative 1—No Action.

Under Alternative 3 a larger area of the II zone is added in areas approximately ½ mile walking distance to the SODO/Lander station. A higher amount of new land use conflicts would be likely in the area around the SODO/Lander Street light rail station compared to Alternative 2 as more land would potentially generate higher volumes of daytime workers unrelated to industrial operations. Conflicts could include new exposures of pedestrians and workers using outside spaces to loud truck traffic and industrial equipment and to odors from industrial operation such as the Republic Transfer station. Presence of workers could increase difficulty of fluid movement of freight and other industrial vehicles, including difficulty operating loading docks and site access. Under Alternative 3 new dense employment would abut heavy rail tracks on the east and west and would be closer to rail yards. There is a higher potential for new employees or tenants in the area to levy complaints against longstanding heavy industrial activities in the vicinity. Impacts rise to the level of moderate.

Use compatibility impacts for The II zoned area between 4th Avenue S and I-90, and in the Rainier Avenue S corridor, and the WOSCA site would be the same as under Alternative 2, including the described relationships to adjacent Chinatown/ID and Little Saigon neighborhoods.

Under Alternative 3 land in the stadium area in the UI zone could receive an estimated 200 industry-supportive housing units. The area would also receive incremental infill development with light industrial uses, brewers/makers with large ancillary spaces, including on some smaller parcels—similar to Alternative 2. Some additional lodging uses would be expected due

to the change to permit lodging in the Stadium Area Overlay District (STAOD). Continued addition of large-sized office and retail uses are expected in the STAOD. The uses would appeal to visitors from nearby residential areas and patrons of stadium events. These changes would cause an overall transition of the area fronting 1st Avenue to an urban mixed-use land use pattern, with some homes interspersed on upper stories of new buildings. The duration of hours and times when visitors unrelated to industry are present would increase in the stadium area outside of event times. This could result in use compatibility impacts due to an incremental increase in exposure of pedestrian activity near trucks transiting 1st Avenue increasing the potential for complaints levied against long standing industrial users. Such impacts would rise to the level of moderate. However, there is also potential for decreases in use conflict as the stadium area transitions to an internally cohesive mixed-use environment with more regular patterns of patronage outside of event times.

Overall use compatibility impacts in the Stadium area would be **moderate** under Alternative 3.

Georgetown/South Park

Under Alternative 3, all of the riverfront lands including Port Terminals and marine operations, and expansive stretches of land currently zoned IG would be placed in the MML zone. Use compatibility impacts there would be similar to Alternative 2 and reduced compared to Alternative 1.

Under Alternative 3 (as in Alternative 2) edges of South Park and Georgetown residential areas would be zone UI, and increased infill development with light industrial uses, brewers/makers, and small manufacturers with large ancillary spaces is expected. However, the zone would enable an estimated 60 industry supportive residential units interspersed in these areas. Resulting use patterns, and times of day for activity that would become more consistent with the adjacent areas outside of the study area in South Park. There is some potential for increased volumes of visitors to create minor use conflicts with heavy industrial uses in MML zones, including the potential for increased complaints levied against industrial users. In some locations, residences could directly view industrial layout spaces and storage yards, truck loading docks, and other industrial development.

Under Alternative 3, the triangular area of Georgetown bounded by Corson Avenue S, Carleton Avenue S and I-5 would be removed from the MIC and placed into a mixed-use zone. The area would likely develop with a high concentration of urban mixed-use structures with ground level retail and residential above. An estimated 1,078 housing units could be added. Land use incompatibility would contribute to pressure for existing industrial businesses to relocate, and by the end of the study time horizon the area would likely transition to <u>a</u> mixed-use area similar to an urban village. The new activity pattern would complement the existing use pattern of restaurants, bars, and retail that fronts Airport Way S and could create a cohesive district. The break in the continuity between the two residential portions of the Georgetown neighborhood that exists today would be removed, which could increase land use compatibility. <u>Use</u> compatibility impact between new development and the existing track 101 spur would increase

under the Alternative 3. There would be increased incompatibility with respect to noise and vibration affecting non-industrial uses and potentially increase safety risks. The new land uses could make it less practical for rail operators to use the spur, causing incremental pressure on the railyard to consider abandoning the spur.

Land removed from the MIC at the edges of South Park would be placed in a mixed-use zone. Some of it would likely redevelop with mixed-use structures including housing on upper floors. The new activity patterns could complement existing use pattern of community uses, local businesses and housing that is inside the South Park urban village. Land added to the South Park urban village adjacent to Duwamish Waterway Park would support community goals to add community uses and residents near open space, provide better physical connection of community members to the Duwamish River. Similarly, conversion to mixed residential and commercial use in the area adjacent to Terminal 117 could alleviate the perception of disconnectedness of South Park community to the Duwamish River. Addition of residences and mixed-use structures would, however, create a period of moderate land use conflict between operation of light industrial businesses and new users.

Overall, while there is potential to reduce land use conflicts by creation of more cohesive mixed-use districts in Georgetown and South Park, the process would result in interim increases in **moderate** land use incompatibilities.

Employment Mix

As seen above in **Exhibit 3.8-15**, the overall employment under Alternative 3 would increase by 57,000 jobs. The mix of industrial employment would decrease by 1.7% points compared to 53.6% in the No Action Alternative, but total industrial employment would increase by 29,000 jobs. The percentage of industrial employment would remain at roughly 51% or greater in every subarea under the alternative. Although there would be an increase in non-industrial employment in office and ICT sectors, the increase in industrial employment due to stronger protections in the MML zones, inclusion of new light industrial space in II zone development, and industrial redevelopment of the Armory site would result in bigger increases in industrial employment. No adverse impact is expected.

Land Use Transitions

Alternative 3 results in impacts due to a lack of transition similar to Alternative 2 except for Ballard and Georgetown where changes to land currently in the IG1 zone are rezoned as UI or in the case of Georgetown and South Park where small areas are removed from the MIC and placed in a mixed-use commercial zone.

<u>Ballard</u>

The area north of Leary Way and east of 15th Avenue W are removed from a core industrial zone and rezoned UI. This change further reduces the potential for high intensity/high impact uses occurring in proximity to nonindustrial areas. The IC zoned area in northwest Ballard is

removed from the IC zone and the core industrial zone is extended to the north Side of Market Street. Potential impacts related to height, bulk, and scale in Alternative 3 are similar to those in Alternative 2. The IC zoned area in northwest Ballard is removed from the IC zone and the core industrial zone is extended to the north Side of Market Street. This has the potential to introduce high intensity/high impact uses. The relative size of this change means that the impact due to a lack of transition is **moderate**.

Interbay Dravus and Interbay Smith Cove

Alternative 3 proposes the same land use pattern as Alternative 2. Existing unavoidable impacts occur parallel to the BNSF rail corridor where it is close to residential areas, and some transition impacts due to the close placement of industrial and non-industrial uses in the Dravus area would occur, though they would be reduced potential impacts occur in the Dravus areacompared to the No Action Alternative. Impacts adjacent to the BNSF rail corridorDue to this combination, overall transition impacts in the Interbay Dravus Subarea are **moderate**, and in the Interbay Smith Cove Subarea transition impacts would be **minor** due to the same factors cited under alternatives 1 and 2-and in Dravus are low.

SODO/Stadium

The land use pattern in SODO/Stadium is the same as Alternative 2 and will result in the same impacts as Alternative 2.

Georgetown/South Park

Alternative 3 proposes the removal of some land from both the Georgetown and South Park neighborhoods from the MIC and rezoned with a mixed-use commercial zone. In Georgetown land zoned IG2 bound by Corson Avenue S to the north, Airport Way S to the east and the commercial and mixed-use commercial zoning to the south is removed from the MIC. This change will not erode the existing transition from the core industrial areas and removes the potential for high impact/high intensity to encroach or abut nonindustrial areas. Potential impact from this proposal is **minor**low.

Other Industrial Zoned Lands

The only difference between the proposed changes for land outside the MICs between Alternative 2 and 3 occurs on the northern shoreline of Lake Union. Alternative 3 proposes changing the IB zoning in this area to UI with no potential for high intensity/high impact uses. The potential for impacts with this proposal is low.

Impacts of Alternative 4

Likely Changes Over the 20-year Planning Horizon

Land Use Planning & Policy. Under Alternative 4—Future of Industry Expanded, the planning and policy context would be changed to enact the Comprehensive Plan policy amendments described above in **Local Policy Framework**. The City would also adopt updates to the currently adopted Sub Area Plans for the Greater Duwamish MIC and BINMIC which include the land use concepts identified in this proposal.

Future Land Use. Under Alternative 4, the Future Land Use Map would be amended slightly. Boundaries of the Greater Duwamish MIC would be altered to remove 19 acres near Georgetown and 7 acres adjacent to South Park from the MIC designation, as indicated on the map in **Chapter 2** and **Appendix C**. In Georgetown, the triangular area bounded by Corson Avenue S, Carleton Avenue S, and I-5 would be placed into the Commercial/Mixed-Use Areas designation. In South Park select parcels at the northeast and southeast corners of the urban village adjacent to the Duwamish River would be removed from the MIC and placed in the South Park Urban Village. The total area of lands removed from the MIC would be 26 acres. No land would be removed from the BINMIC, and no other Comprehensive Plan FLUM designations would change.

Zoning. Under Alternative 4, zoning would be changed to apply the proposed new MML, II, and UI zones, instead of the existing zones. The Seattle Municipal Code would be amended to add the development standards in the MML, II and UI zones as described in **Chapter 2**, including retention of a Stadium Transition Area Overlay District. The location of the zones in Alternative 4 is mapped as shown in **Chapter 2**, and **Appendix C**.

Alternative 4 applies the proposed land use concepts with a greater share of Industry and Innovation and Urban Industrial than Alternative 2. This alternative expands limited housing allowances to the greatest degree of any of the alternatives. Alternative 4:

- Applies the MML zone covering 87% of industrial lands.
- Applies a mix of the II and UI zones to 13% of the study area includes an estimated 1/2 mile from light rail stations.
- Expands limited industry-supportive housing in UI zone with a maximum density of 50 Dwelling Units / Acre
- Applies mixed-use zoning to the areas of Georgetown and South Park that are removed from the MIC. Neighborhood Commercial with a 75' height limit or a 55' height limit could be applied. The higher scale 75' zone is analyzed for impact analysis purposes. An MHA (M1) suffix are assumed for analysis.

Alternative 4 includes a maximum size of use limit for indoor sports and recreation uses up to 50,000 sq. ft. in all proposed industrial zones. This would be an increase over the 10,000 sq. ft. size limit of the existing IG zones, but a decrease from the existing limit of 75,000 in IB and IC zones. The SMC also already includes a special allowance for indoor sports and recreation

facilities up to 50,000 sq. ft. in the BINMIC subject to limiting locational criteria of SMC 23.50.027.H.

Land Use. Under Alternative 4, land use would change over the planning period according to current trends and as a result of the zoning changes of the alternative. Some notable expected changes include.

- Decreased rate of conversion to stand-alone office and retail uses in MML zoned areas. The new MML zone would have stricter size of use and FAR limits for stand-alone office and retail uses and a prohibition on new mini-storage facilities. As a result, there would be fewer conversions to stand-alone office and retail than past trends and under Alternative 1.
- Continued additions of distribution and warehouse facilities. Strong demand for new warehouse and distribution space is expected to continue, resulting in the addition of new distribution and warehouse facilities in MML zoned areas.
- Maintenance of the maritime and industrial base. Most long standing maritime and logistics uses would continue on shorelines and industrial lands near industrial infrastructure, especially in the Greater Duwamish MIC. Incremental renewal of facilities and buildings for industrial use could be expected at a greater rate than under Alternative 1.
- Denser employment including new industrial space, near future light rail station in II zoned areas. The proposed II zone standards combined with expected strong market interest due to increased access provided by light rail stations is likely to result in development with a high density of employment in new buildings for Information Computer Technology and offices in these areas. The development would also include new light industrial space at ground level. Much higher levels of employment, and activity pattern of by employees and visitors. Associated frontage improvements and infrastructure upgrades would be expected.
- Increased development of mixed-use, flex and light industrial uses in UI zoned areas. The proposed UI zone regulations combined with expected strong market interest due to proximity to population centers will lead to the incremental addition of new buildings with light industry, office, and flex space in areas at the edges of MICs near urban villages. Incremental infill development will add density of activity, employment, serving nonindustrial populations. Frontage improvements and infrastructure upgrades and increased landscaped areas would be expected. The physical character in these edge areas would become more urban in nature with more buildings built to lot lines.
- Introduction of some new industry-supportive housing. Under Alternative 4 about 2,195 new homes would be added in UI zoned portions of industrial areas due to increased flexibility for caretakers' quarters and artist/studios. Introduction of housing changes use patterns, as more 24-hour presence of residents living in areas at the edges of MICs.
- Additional new housing in areas removed from the Greater Duwamish MIC. More housing would be added in mixed-use buildings in areas removed from MICs in Georgetown and South Park. This housing would contribute to the expansion of a mixed-use, urban

neighborhood character in both locations. The number of new units in the mixed-use areas removed from the MIC equal about 1,078.

Armory Site Redevelopment. Under Alternative <u>3-4</u> the Armory site would be redeveloped with a mix of ICT/office and include new light industrial space at ground level after relocation of the Army National Guard to North Bend, Washington. The site would contain a substantial amount of employment density in a new campus-like setting with integrated open space features and new roadway and utilities infrastructure, including integration of green infrastructure.

Employment Mix. Under Alternative 4, employment is projected to grow substantially more than under Alternative 1 No Action and Alternative 2, and by a similar amount to Alternative 3. A total of 59,200 additional jobs are projected for the study area, an increase of 59%. This would represent 35% of the total projected citywide employment growth over the 20-year planning horizon. The mix of industrial employment would decrease by 2.5% points compared to No Action, down to 52.8%. Both MICs would continue to contain much more than the minimum number of industrial jobs required to meet PSRC's regional criteria for MIC designation (20,000). The percentage of industrial employment would decrease close to the 50% threshold in the Ballard (50.0%) and SODO/Stadium (51.9%) subareas. See **Exhibit 3.8-16**.

	Current Conditions (2018)			Alternative 4—Future of Industry Limited (2044)		
Subarea	Ind. Emp.	Total Emp.	% Ind.	Ind. Emp.	Total Emp.	% Ind.
Ballard	9,400	17,100	55.0%	16,000	32,000	50.0%
Interbay Dravus	3,400	5,600	60.7%	5,600	10,200	54.9%
Interbay Smith Cove	3,900	6,000	65.0%	6,300	10,700	58.9%
SODO/Stadium	23,000	43,900	52.4%	34,400	66,300	51.9%
South Park/Georgetown	14,900	25,900	57.5%	21,000	38,500	54.5%
Total	54.500	98,500	55.3%	83.300	157.700	52.8%

Exhibit 3.8-16-15 Employment by Subarea Current Conditions and Alternative 4

Source: City of Seattle, 2021.

Consistency With Plans & Policies

Impacts regarding consistency with plans and policies under Alternative 4 are similar in nature to those described under Alternative 3 above. However, the anticipated impact is greater under Alternative 4 because Alternative 4 introduces a greater quantity of industry-supportive housing (an estimated 2,195 limited industry supportive housing units in industrial zones). The housing would be available in caretakers' quarters or artist/maker live/workspaces as described for Alternative 3.

Similar to Alternative 3, Alternative 4 would represent a substantial shift of the city's expected employment growth into industrial areas compared to past trends and the previous 20-year Comprehensive Plan planning horizon, with 35% of expected job growth in the study area. For reasons described above for Alternative 3 however, this would not result in an adverse impact.

An incremental increase in conflicts would arise with regional multi-county and PSRC policies that discourage location of new housing in MICs. Similarly, a degree of inconsistency would arise with the City's Comprehensive Plan policies including LU Policy 10.12 (competition for industrial land by non-industrial uses) and Container Port CP.3 (discouraging retail and residential uses). Overall impacts to consistency with plans and policies due to introduction of housing would be greater than under Alternative 3 but would result in **moderate** impacts.

Land Use Compatibility

<u>Ballard</u>

Under Alternative 4 land in the Ballard uplands in the 14th Avenue corridor north of NW Leary would be placed in a combination of the II zone and the UI zone. The UI zone would allow a greater density of industry supportive housing at a maximum density of 50 dwelling units / acre.

The blocks zoned II would likely be developed with a significant amount of dense employment in multistory structures, including some towers, with dedicated space for ground floor light industry. Development pressure would be fueled by proximity to light rail. The redevelopment would contribute to an agglomeration of daytime employment uses in conjunction with nearby activity in the Ballard Urban Village. New uses would generate higher volumes of daytime workers unrelated to industrial operations. Conflicts could include new exposures of pedestrians and workers using outdoor spaces to loud truck traffic and industrial equipment and to odors from long-standing industrial operations in the area. Presence of workers could increase difficulty of fluid movement of freight and other industrial vehicles, including difficulty with loading and site access.

Other areas that are north of NW Leary Way NW and in Fremont north of 36th Street would be placed in the UI zone and would likely receive a substantial amount of increased infill development with light industrial uses, brewers/makers with large ancillary spaces. New zone standards would allow small parcels to accommodate new structures. An additional 790 housing units are estimated and would typically be located on several upper floors of a 4-6 story mixed-use development. Ground floor uses would appeal to visitors from nearby urban villages such as showrooms, tasting rooms and similar uses. Residents would be in view of storge and loading areas of industrial business. The likelihood of complaints levied against industrial businesses would increase.

These changes would cause an overall and thorough transition of the area to an urban mixeduse pattern with some 24-hour residences interspersed throughout. Compatibility impacts would likely increase pressure on intensive or heavy industrial activities to relocate form the area over time and would rise to the level of moderate impact. However, light industrial spaces would be integrated at ground level and some area businesses could have access to new light industrial space.

Due to conversion to the MML zone for lands abutting the shoreline, incompatible use pressures in areas of Ballard south of Leary Way could be lessened over time. However, the magnitude of new residences and employees in areas north of Leary Way could exhibit spillover use compatibility pressure on some land south of Leary Way. Therefore, use incompatibilities here would be greater than the other alternatives, and rise to the level of **moderate**.

Dense employment in multi-story buildings would likely be added in the two blocks of the II in Fremont that already contain a high concentration of dense employment. Similar to Alternative 2, no major use conflict would be expected in these locations.

Overall use compatible impacts in Ballard would be the greatest of any of the alternatives and would be **moderate** under Alternative 4.

Interbay Dravus

Under Alternative 4, all shoreline and adjacent lands including Fisherman's Terminal, W Commodore Way lands, and the BNSF railyard would be placed in the MML zone. Use conflicts in these areas would be the same as described in alternatives 2 and 3 and reduced compared to Alternative 1—No Action.

Land north of W. Dravus Street along Thorndyke Avenue W would be zoned UI as in alternatives 2 and 3, but in Alternative 4 the zone would allow for industry supportive housing at a maximum density of 50 dwelling units per acre. An additional 175 housing units are estimated, and they would typically be located on an upper floor of a 4-6 story mixed-use development. These changes would cause an overall transition of the area to an urban mixed-use pattern with housing units interspersed, which could lead to the type of land use conflicts described in Alternative 3 where housing is introduced. However, no very heavy or largescale industrial uses are in the small area, and it is contained by defined edges of the rail track and 15th Avenue W.

As with Alternative 3, the limited size of land in this node limits the degree of potential impact. However, the greater density of homes increases likelihood of land use conflict compared to Alternative 3. Therefore, use compatibility impacts in the Interbay Dravus Subarea for Alternative 4 would be **moderate**.

Interbay Smith Cove

Under Alternative 4 the Armory site would be in the II zone, as it is in Alternative 3. The land use compatibility impacts would be the same as described for Alternative 3.

As with alternatives 2 and 3, T91 Marine Terminals and T91 uplands would be placed in the MML zone. Marine activities, and industrial use similar today would continue on those lands and would not create additional use conflict.

No additional housing is expected in the Interbay Smith Cove Subarea under Alternative 4 because of the small application of the UI zone on parcels unlikely to redevelop.

Overall use compatible impacts in Interbay Smith Cove would be **minor** under Alternative 4.

SODO/Stadium

Under Alternative 4, all shoreline areas and adjacent lands including Port Terminals, and expansive stretches of land currently zoned IG would be placed in the MML zone. Compatibility impacts there would be similar to Alternative 2, and 3, and reduced compared to Alternative 1—No Action.

Under Alternative 4 a larger area of the Industry and InnovationII zone (more than alternatives 2 or 3) is added in an expanded ½ mile walking areas to the SODO/Lander station, including blocks along 6th Avenue S north of S Holgate Street. A higher amount of new land use conflict than alternatives 2 or 3 would be likely in the area as more land would potentially generate higher volumes of daytime workers unrelated to industrial operations. Land use compatibility conflicts would be similar to those described for Alternative 3 but greater in scale. Land added to the II zone in Alternative 4 would abut heavy rail tracks, freeway ramp infrastructure onramps to I-90, and the King County Metro Central Base exposing future occupants to close contact with regular effects of bus noise and emissions.

Use compatibility impacts for the II zoned area between 4th Avenue S and I-90, in the Rainier Avenue S corridor, and the WOSCA site would be the same as under alternatives 2 and 3 including the described relationships to the adjacent Chinatown/ID and Little Saigon neighborhoods.

Under Alternative 4, land in the stadium area would be zoned UI, and the UI zone would be extended further south along 1st Avenue to Starbucks Center. This would allow the area to receive an estimated 990 industry-supportive housing units. The area would also receive incremental infill development with light industrial uses, brewers/makers with large ancillary spaces, including on some smaller parcels—similar to alternatives 2 and 3. Some additional lodging uses would be expected due to the change to permit lodging in the Stadium Transition Area Overlay District (STAOD). Continued addition of large-sized office and retail uses are expected in the STAOD. These changes, including the higher proportion of housing would cause an overall transition of the area fronting 1st Avenue to an urban mixed-use pattern, with homes interspersed on upper stories of new buildings. Use compatibility conflicts would be similar to those described for Alternative 3, but greater in scale. Such impacts would rise to the level of moderate. However, there is also potential for decreases in land use conflict as the stadium area transitions to a more internally cohesive mixed-use environment with more regular patterns of patronage outside of event times.

Overall use compatibility impacts in the SODO/Stadium area in Alternative 4 would be greater than Alternative 3 but would be **moderate.**

Georgetown/South Park

Under Alternative 4, all of the riverfront lands including Port Terminals and marine operations, and expansive stretches of land currently zoned IG would be placed in the MML zone. Land use compatibility impacts there would be similar to alternatives 2 and 3 and reduced compared to Alternative 1.

Under Alternative 4 (as in Alternative 2) edges of South Park and Georgetown residential areas would be zoned UI, and increased infill development with light industrial uses, brewers/makers, and small manufacturers with large ancillary spaces is expected. However, the zone would enable an estimated 240 industry supportive residential units interspersed in these areas. Use compatibility conflicts would be similar to those described for Alternative 3 but greater in scale.

Land use changes and resulting compatibility impacts in the triangular area that would be removed from the MIC in Georgetown and the land removed from the MIC at the edges of South Park and placed into a mixed-use zone would be <u>similar to the same as under</u> Alternative 3.

Overall, while there is potential to reduce land use conflicts by creating cohesive mixed-use districts in Georgetown and South Park over time, the process would result in interim land use compatibility impacts that rise to the level of **moderate**.

Employment Mix

As seen above in **Exhibit 3.8-16**, the overall employment under Alternative 4 would increase by 59,200 jobs. The mix of industrial employment would decrease by 2.5% points compared to 53.6% in the Alternative 1—No Action, but total industrial employment would increase by 28,800 jobs. The percentage of industrial employment would remain at roughly 50% or greater in every subarea under the alternative. Although there would be an increase in non-industrial employment in office and ICT sectors, the increase in industrial employment due to stronger protections in the MML zones, inclusion of new light industrial space in II zone development, and industrial redevelopment of the Armory site would result in bigger increases in industrial employment. Since the employment mix of industrial would drop to 50.0% in Ballard—at the threshold for percentage of industrial employment in MICs per regional criteria—a minor adverse impact in employment mix is present for the Ballard Subarea.

Land Use Transitions

Alternative 4 has the greatest amount of proposed change but at the transitions from core industrial areas to nonindustrial areas result in the same or fewer impacts than Alternative 3. In this alternative some areas that are proposed to be zoned UI in Alternative 3 are proposed to be II which, like UI, precludes the potential for high impact/high intensity uses to abut or encroach on nonresidential areas but could result in some impacts related to height, bulk, scale, and aesthetics.

<u>Ballard</u>

The central part of the Ballard Subarea bisected by 14th Avenue W is proposed to be rezoned from IG1 to II. This change is expected to result in a mix of light industrial and commercial development. The change from IG to II will provide a better transition to nonindustrial areas to the north by reducing the likelihood of high impact/high intensity uses encroaching or abutting nonresidential areas. In northwest Ballard where industrial land abuts nonindustrial land, the proposal is to rezone existing IC zoned land to II providing a similar transition as Alternative 1— No Action and maintaining the unlikely potential for high intensity/high impact uses to abut or encroach on nonindustrial areas. However, larger and taller buildings anticipated by the II, particularly near transit stations, have the potential for impacts related to height, bulk, scale, and aesthetics adjacent to nonindustrial areas. Impacts of the proposal in Ballard are **moderate**.

Interbay Dravus and Interbay Smith Cove

The proposed land use changes in the Interbay Dravus and Interbay Smith Cove subareas are identical to those proposed in Alternative 3 and the resulting impacts are the same.

SODO/Stadium

The proposed land use changes in the SODO/Stadium Subarea are the very similar to same as Alternative 3. except for <u>A</u>a small node of land on the west side of Harbor Avenue SW which is proposed to be changed from I<u>G</u>B to UI instead of the MML zone. This adjustment could slightly lessen the potential for a transition impact at the location in the Delridge area, however the overall potential for a moderate transition impact remains the same as the other alternatives. <u>Minor</u> The impacts are similar to impacts identified in the other alternatives would remain for other parts of the SODO Subarea.

Georgetown/South Park

The proposed land use changes in Interbay Georgetown/South Park are identical to those proposed in Alternative 3 and the resulting impacts are the same.

Other Industrial Zoned Lands

Alternative 4 proposes the same changes for land outside the MICs that Alternative 3 proposes and there are no impacts due to lack of transition between core industrial areas and nonresidential areas.

Impacts of The Preferred Alternative

Likely Changes Over the 20-year Planning Horizon

Land Use Planning & Policy. Under the Preferred Alternative, the planning and policy context would be changed to enact the Comprehensive Plan policy amendments described above in

Local Policy Framework. The City would also adopt updates to the currently adopted Centers Plans for the Greater Duwamish MIC and BINMIC which include the land use concepts identified in this proposal.

Future Land Use. Under the Preferred Alternative, the Future Land Use Map would be amended slightly. Boundaries of the Greater Duwamish MIC would be altered to remove 29 acres near Georgetown and 7 acres adjacent to South Park from the MIC designation, as indicated on the map in **Chapter 2** and **Appendix C**.

In Georgetown, the triangular area bounded by Corson Avenue S, Carleton Avenue S, and I-5 and the Georgetown Playfield and Park north of Corson Street would be placed into the Commercial/Mixed-Use Areas designation. This change would comprise a larger removal of land from the MIC than in any other alternative, and it would create a new contiguous area connecting to both of the areas designated Single Family or Multi-Family Residential on the Future Land Use Map. In South Park select parcels at the northeast and southeast corners of the urban village adjacent to the Duwamish River would be removed from the MIC and placed in the South Park Urban Village. The total area of lands removed from the MIC would be 36 acres. No land would be removed from the BINMIC, and no other Comprehensive Plan FLUM designations would change.

Outside the MIC and BINMIC boundaries the Industrial Future Land Use Map designation would remain for most areas where the Industrial Commercial zone would be retained including in Fremont, North Lake Union and for some of Judkins Park. However, a portion of the land in Judkins Park (11.5 acres) that is currently in the Industrial designation would be changed to the Mixed Use Commercial designation.

Zoning. Under the Preferred Alternative, zoning would be changed to apply the proposed new MML, II, and UI zones, instead of the existing zones within the MICs. Unlike Alternatives 2,3 and 4, under the Preferred Alternative the existing IC zone would be retained for several areas outside the MICs, and the IC zone would be retained in the land use code. The Seattle Municipal Code would be amended to add the development standards in the MML, II and UI zones as described in **Chapter 2**, including retention of a Stadium Transition Area Overlay District. The location of the zones in the Preferred Alternative is mapped as shown in **Chapter 2**, and **Appendix C**. The Preferred Alternative applies a similar amount of the II (and IC outside of MICs) as Alternative 4 and slightly less Urban Industrial than Alternative 3 but more than Alternative 4.

<u>Under the Preferred Alternative, there are several modifications and refinements to the</u> <u>development standards for the Industry and Innovation zone and the Urban Industrial zone</u> <u>compared to the Draft EIS alternatives. In the II zone modifications are intended to enhance</u> <u>propensity for development including an increase of the bonus ratio of non-industrial space to</u> <u>light industrial space, a new use definition for Computer Information and Technology (ICT), and</u> <u>allowing offsite performance for industrial space within the same MIC. In the Urban Industrial</u> <u>zone modifications would make housing allowances a conditional use and would provide the</u> <u>option for 50% of any housing to be built at a workforce affordable housing level (up to 90%</u> <u>AMI), in addition to the option of limiting tenancies to caretakers' quarters or maker studios.</u> <u>These amendments would result in housing within MICs at levels greater than Alternative 3 but</u> <u>less than Alternative 4.</u>

Housing opportunities would be expanded for the portions of the study area outside the MICs in Judkins Park and Ballard at levels greater than Alternative 4 where zoning would be changed to Neighborhood Commercial. Housing allowances in new mixed use areas of South Park would be similar to alternatives 3 and 4. Within Georgetown, the new mixed use area would have unique development standards to encourage preservation of historic structures and arts spaces and minimize potential displacement, which lower the projected amount of future housing development compared to Alternative 4 (see Development Standards Appendix # for detail on Preferred Alternative development standards).

The Preferred Alternative:

- Applies the MML zone covering 85% of industrial lands.
- Applies a mix of the II and UI zones to 14% of the study area within an estimated 1/2 mile from light rail stations.
- Provides additional incentives for development in the II zone.
- Allows limited industry-supportive housing in UI zone as a conditional use with a maximum density of 50 Dwelling Units / Acre subject to tenancy limits or a 50% workforce housing affordability limit.
- Applies mixed-use zoning to the area of Georgetown that is removed from the MIC.
 Neighborhood Commercial 3 with a 55' height limit could be applied and the zone would have unique features to incentivize preservation of historic structures and arts spaces. An MHA (M1) suffix is assumed for analysis.
- Applies mixed-use zoning to the areas of South Park that are removed from the MIC.
 Neighborhood Commercial 3 with a 75' height limit could be applied to accommodate a wide variety of community-supportive uses. An MHA (M1) suffix is assumed for analysis.
- Applies mixed-use zoning to existing Industrial Buffer in northwest Ballard and existing Industrial Commercial near Judkins Park. The Neighborhood Commercial 3 zone with a 75' height limit could be applied with an MHA (M1) suffix assumed for analysis.
- The Preferred Alternative includes a maximum size of use limit for indoor sports and recreation uses up to 50,000 sq. ft. in all proposed industrial zones. This would be an increase over the 10,000 sq. ft. size limit of the existing IG zones, but a decrease from the existing limit of 75,000 in IB and IC zones. The SMC also already includes a special allowance for indoor sports and recreation facilities up to 50,000 sq. ft. in the BINMIC subject to limiting locational criteria of SMC 23.50.027.H.4.
- <u>Retains the Stadium Transition Overlay District and adds flexibility for larger size of use</u> <u>limits for stand-alone retail, commercial and restaurant/bar activities than would be allowed</u> <u>in the UI zone otherwise, and allows lodging uses.</u>

Land Use. Under the Preferred Alternative, land use would change over the planning period according to current trends and as a result of the zoning changes of the alternative. Overall

projected employment growth under the Preferred Alternative of 35,545 is similar to Alternative 2 and substantially less than alternatives 3 and 4, due to updates of expectations about recovery in demand for office and ICT uses. The amount of housing in MICs under the Preferred Alternative (1,475 projected housing units) would be appreciably less than Alternative 4, while there simultaneously be an increase in housing production outside of MICs (to 1,534 projected housing units). Housing in the Stadium/SODO area (644 projected units) would be a slight decrease from Alternative 4. The result of these adjustments is that many of the same types of land use changes would be expected to occur under the Preferred Alternative as draft alternatives 3 and 4, but to a slightly lesser degree or intensity.

Some notable expected changes include:

- Decreased rate of conversion to stand-alone office and retail uses in MML zoned areas. The new MML zone would have stricter size of use and FAR limits for stand-alone office and retail uses and a prohibition on new mini-storage facilities. As a result, there would be fewer conversions to stand-alone office and retail than past trends and under Alternative 1.
- Continued additions of distribution and warehouse facilities. Strong demand for new warehouse and distribution space is expected to continue, resulting in the addition of new distribution and warehouse facilities in MML zoned areas.
- Maintenance of the maritime and industrial base. Most long standing maritime and logistics uses would continue on shorelines and industrial lands near industrial infrastructure, especially in the Greater Duwamish MIC. Incremental renewal of facilities and buildings for industrial use could be expected at a greater rate than under Alternative 1.
- Denser employment including new industrial space, near future light rail station in II zoned areas. The proposed II zone standards combined with access provided by light rail stations is likely to result in development with a high density of employment in new buildings for Information Computer Technology and offices in these areas. The development would also include new light industrial space and ICT uses at ground level. Much higher levels of employment, and activity pattern of by employees and visitors would result. Associated frontage improvements and infrastructure upgrades would be expected.
- Increased development of mixed-use, flex, and light industrial uses in UI zoned areas. The proposed UI zone regulations combined with expected strong market interest due to proximity to population centers will lead to the incremental addition of new buildings with light industry, office, and flex space in areas at the edges of MICs near urban villages. Incremental infill development will add density of activity, employment, serving nonindustrial populations. Frontage improvements and infrastructure upgrades and increased landscaped areas would be expected. The physical character in these edge areas would become more urban in nature with more buildings built to lot lines.
- Introduction of some new industry-supportive housing. Under the Preferred Alternative about 644 new homes would be added to the SODO/Stadium District and in UI zoned portions of industrial areas due to increased flexibility for caretakers' quarters and

artist/studios. Introduction of housing changes use patterns, as more 24-hour presence of residents living in areas at the edges of MICs.

- Additional new housing in areas removed from the Greater Duwamish MIC. More
 housing would be added in mixed-use buildings in areas removed from MICs in Georgetown
 and South Park. This housing would contribute to the expansion of a mixed-use, urban
 neighborhood character in both locations. The projected number of new units in the mixed use areas outside of the MIC equals about 1,534 dwelling units.
- Armory Site Redevelopment. Under the Preferred Alternative the Armory site would be redeveloped with a mix of ICT/office and include new light industrial space at ground level after relocation of the Army National Guard to North Bend, Washington. The site would contain a substantial amount of employment density in a new campus-like setting with integrated open space features and new roadway and utilities infrastructure, including integration of green infrastructure.

Employment Mix. The total number of jobs would increase by 35,545. 52.8% of all jobs would be industrial jobs, a lesser proportion than Draft EIS Alternatives 1,2 and 3, and about the same as Alternative 4. The absolute number of industrial jobs (70,853) would be greater than the No Action Alternative 1, so even as the share of industrial employment would decrease, the City would gain industrial jobs. The Preferred Alternative would make a moderate shift of Seattle's total employment growth into MICs compared to historic growth rates in MICs. Employment growth of 35,545 projected under the Preferred Alternative in the study area would represent about 18% of the net citywide job growth that the city would be planning for during the 20-year planning horizon (2019-2044). Projections are adjusted downward compared to the Draft EIS alternatives to reflect conditions more realistically in commercial/office demand post-COVID. The adjusted projections acknowledge that it will likely take longer to achieve levels of employment growth. See **Exhibit 3.8-17**.

	<u>Current Conditions (2018)</u>			<u>Preferred Alternative—Future of</u> <u>Industry Balanced (2044)</u>			
<u>Subarea</u>	<u>Ind. Emp.</u>	<u>Total Emp.</u>	<u>% Ind.</u>	<u>Ind. Emp.</u>	<u>Total Emp.</u>	<u>% Ind.</u>	
<u>Ballard</u>	<u>9,400</u>	<u>17,100</u>	<u>55.0%</u>	<u>13,685</u>	<u>27,479</u>	<u>49.8%</u>	
Interbay Dravus	<u>3,400</u>	<u>5,600</u>	<u>60.7%</u>	<u>4,784</u>	<u>8,713</u>	<u>54.9%</u>	
Interbay Smith Cove	<u>3,900</u>	<u>6,000</u>	<u>65.0%</u>	<u>5,130</u>	<u>8,714</u>	<u>58.9%</u>	
SODO/Stadium	<u>23,000</u>	<u>43,900</u>	<u>52.4%</u>	<u>31,922</u>	<u>55,897</u>	<u>52.1%</u>	
South Park/Georgetown	<u>14,900</u>	<u>25,900</u>	<u>57.5%</u>	<u>18,133</u>	<u>33,243</u>	<u>54.5%</u>	
<u>Total</u>	<u>54,500</u>	<u>98,500</u>	<u>55.3%</u>	<u>70,853</u>	<u>134,045</u>	<u>52.8%</u>	

Exhibit 3.8-17 Employment by Subarea Current Conditions and the Preferred Alternative

Source: City of Seattle, 2022.

Consistency With Plans & Policies

Impacts regarding consistency with plans and policies under the Preferred Alternative are similar in nature to those described under Alternative 3 above. The Preferred Alternative represents a moderate increase in the share of citywide growth occurring in MIC's over the 20year Comprehensive Plan time horizon with 18% of expected job growth in the study area. This level of employment growth would not result in an adverse impact.

An incremental increase in conflicts would arise with regional multi-county and PSRC policies that discourage location of new housing in MICs. However, the Preferred Alternative would only allow housing in the UI zone as a conditional use subject to additional criteria compared to Alternatives 3, and 4. This affords the City more control during the permitting process to disallow new housing in inappropriate circumstances.

As with alternatives 3 and 4, a degree of inconsistency would arise with the City's Comprehensive Plan policies including LU Policy 10.12 (competition for industrial land by nonindustrial uses) and Container Port CP.3 (discouraging retail and residential uses). Overall impacts to consistency with plans and policies due to introduction of housing would be greater than under Alternative 3 and less than Alterative 4 resulting in **moderate** impacts.

Land Use Compatibility

Ballard

<u>Under the Preferred Alternative land in the Ballard uplands in the 14th Avenue corridor north of</u> <u>NW Leary would be placed primarily in the II zone. Adjacent land to the east currently in the IB</u> <u>zone would be placed in the UI zone. The UI zone would allow industry supportive housing at a</u> <u>maximum density of 50 dwelling units / acre with locational location and performance criteria.</u>

The blocks zoned II would likely be developed with a significant amount of dense employment in multistory structures, including some towers, with dedicated space for ground floor light industry. Development pressure would be fueled by proximity to light rail. The redevelopment would contribute to an agglomeration of daytime employment uses in conjunction with nearby activity in the Ballard Urban Village. New uses would generate higher volumes of daytime workers unrelated to industrial operations. Conflicts could include new exposures of pedestrians and workers using outdoor spaces to loud truck traffic and industrial equipment and to odors from long-standing industrial operations in the area. Presence of workers could increase difficulty of fluid movement of freight and other industrial vehicles, including difficulty with loading and site access. Since the overall amount of employment growth and overall pace of development in the II zone would be less than alternatives 3 and 4, these effects are expected to be somewhat less than those alternatives.

<u>Areas that are outside of the MIC, north of NW Leary Way NW near Swedish Hospital in Ballard</u> and areas in Fremont north and south of 36th Street would be placed in an IC zone, or retain the existing IC zone. This is a change from alternatives 3 and 4 that apply the II zone in some of these locations. Compared to alternatives 3 and 4 these locations will likely experience infill development with a greater proportion of offices and ICT uses without inclusion of new light industrial space. The locations could be expected to transform more completely to urban village mixed use environments lessening use conflicts compared to alternatives 3 and 4 over time. Like Alternative 2, no major use conflict would be expected in these locations.

Due to conversion to the MML zone for lands abutting the shoreline, incompatible use pressures in areas of Ballard south of Leary Way could be lessened over time. However, the magnitude of new residences and employees in areas north of Leary Way could exhibit spillover use compatibility pressure on some land south of Leary Way. Therefore, use incompatibilities here would be greater than Alternative 2, and rise to the level of **moderate**.

Under the Preferred Alternative the II zone would allow for off-site performance within an MML zone in the same MIC, by a developer when building industrial space required to access bonus development. A potential effect of this provision is an increased amount of new stand-alone bona-fide industrial space in more locations within core industrial areas. This could strengthen the industrial protection aspects of the MML zone and further reduce incompatible use conflicts there over time in BINMIC MML-zoned areas (as well as other MML zoned areas throughout the study area).

Overall use compatible impacts in Ballard would be similar to, but somewhat less than Alternative 4 and would be **moderate** under the Preferred Alternative.

Interbay Dravus

Impacts under the Preferred Alternative will be similar to those under Alternative 4. All shoreline and adjacent lands including Fisherman's Terminal, W Commodore Way lands, and the BNSF railyard would be placed in the MML zone. Use conflicts in these areas would be the same as described in alternatives 2, 3, and 4 but reduced compared to Alternative 1—No Action.

Land north of W Dravus Street along Thorndyke Avenue W would be zoned UI as in alternatives 2 and 3 but in the Preferred Alternative would allow for industry supportive housing at a maximum density of 50 dwelling units per acre subject to locational criteria. This is like Alternative 4. An additional 114 housing units are estimated, and they would typically be located on an upper floor of a 4-6 story mixed-use development. These changes would cause an overall transition of the area to an urban mixed-use pattern with housing units interspersed, which could lead to the type of land use conflicts described in alternatives 3 and 4 where housing is introduced. However, no very heavy or largescale industrial uses are in the small area, and it is contained by defined edges of the rail track and 15th Avenue W.

<u>As with alternatives 3 and 4, the limited size of land in this node limits the degree of potential</u> <u>compatibility impact. However, the greater density of homes increases likelihood of land use</u> <u>conflicts compared to Alternative 3. Therefore, use compatibility impacts in the Interbay Dravus</u> <u>Subarea for the Preferred Alternative would be **moderate**.</u>

Interbay Smith Cove

Under the Preferred Alternative the Armory site would be in the MML zone, and adjacent land to the north and east fronting on 15th Avenue W will be in the II zone. The land use compatibility impacts would be similar to alternatives 3 and 4. As with alternatives 2, 3, and 4, the Armory Site would be expected to undergo coordinated master planning at some future point that would mitigate the potential for land use incompatibilities. Some potential remains for impacts related to a daytime office population in this vicinity interacting with traditional industrial activity at the rail yard, or any remaining traditional industrial users.

As with alternatives 2, 3, and 4, T91 Marine Terminals and T91 uplands would be placed in the MML zone. Marine activities, and industrial use similar today would continue on those lands and would not create additional use conflict.

No additional housing is expected in the Interbay Smith Cove Subarea under the Preferred Alternative because of the small application of the UI zone on parcels unlikely to redevelop.

Overall use compatible impacts in Interbay Smith Cove would be **minor** under the Preferred Alternative.

SODO/Stadium

<u>Under the Preferred Alternative, all shoreline areas and adjacent lands including Port</u> <u>Terminals, and expansive stretches of land currently zoned IG would be placed in the MML</u> <u>zone. Compatibility impacts there would be like alternatives 2, 3, and 4 and reduced compared</u> <u>to Alternative 1—No Action.</u>

Under the Preferred Alternative, a larger area of the II zone (more than alternatives 2 or 3) is added in an expanded ½ mile walking areas to the SODO/Lander station. Under the Preferred Alternative the II zoned area would include blocks along 4th Avenue S and 1st Avenue South, south of S Lander Street. These new extensions of the II zone could create different areas of potential land use conflict than alternatives 2, 3, or 4 as new development in those blocks could generate higher volumes of daytime workers unrelated to industrial operations. Land use compatibility conflicts would be like those described for Alternative 4 but in some additional places south of S Lander Street, where they could approach existing distributing, outdoor storage, and truck-heavy uses there. Proximities of office and ICT employment activity to rail tracks, the Republic transfer station, and outdoor industrial yard activities could be in more areas south of S Lander Street, exposing future occupants to close contact with regular effects of noise and emissions.

Some factors would mitigate the compatibility impacts. Because the geography of the II zones would be in contiguous and linear extensions from the Lander Street Light rail station, there would be a high potential for development of improved pedestrian and bicycle pathways on 1st and 4th avenues or interior to the blocks or along the SODO busway. There is strong potential for coordinated development integrated with the rail station. Additionally, the pace and total amount of employment growth in the II zone under the Preferred Alternative is less than

alternatives 3 and 4 so the intensity or prevalence of localized compatibility conflicts could be relatively less than alternatives 3 or 4.

<u>Use compatibility impacts for the II zoned area between 4th Avenue S and I-90, in the Royal</u> <u>Brougham Way corridor, are the same as under alternatives 2, 3, and 4 including the described</u> <u>relationships to the adjacent Chinatown/ID and Little Saigon neighborhoods. Under the</u> <u>Preferred Alternative the WOSCA site is placed in the MML zoning on T46 immediately to the</u> <u>west.</u>

Under the Preferred Alternative, land in the stadium area would be zoned UI, and the UI zone would be extended south to S Stacy Street. This would allow the area to receive an estimated 644 industry-supportive housing units. The area would also receive incremental infill development with light industrial uses, brewers/makers with large ancillary spaces, including on some smaller parcels—like alternatives 2, 3, and 4. Some additional lodging uses would be expected due to the change to permit lodging in the Stadium Transition Area Overlay District (STAOD). Continued addition of large-sized office and retail uses are expected in the STAOD. Use compatibility conflicts would be like those described for Alternative 3, but greater in scale. Such impacts would rise to the level of moderate.

Under the Preferred Alternative the II zone would allow for off-site performance within an MML zone in the same MIC, by a developer when building industrial space required to access bonus development. A potential effect of this provision is an increased amount of new stand-alone bona-fide industrial space in more locations within core industrial areas. This could strengthen the industrial protection aspects of the MML zone and further reduce compatible use conflicts there over time in SODO MML-zoned areas (as well as other MML zoned areas throughout the study area).

Overall use compatibility impacts in the SODO/Stadium area in the Preferred Alternative would be greater than Alternative 2, and similar to alternatives 3 and 4. Compatibility impacts would be incrementally less than Alternative 4 but would be **moderate.**

Georgetown/South Park

Under the Preferred Alternative, all the riverfront lands including Port Terminals and marine operations, and expansive stretches of land currently zoned IG would be placed in the MML zone. Land use compatibility impacts there would be like alternatives 2 and 3 and reduced compared to Alternative 1.

Under the Preferred Alternative (as in alternatives 2, 3, and 4) edges of South Park and Georgetown residential areas would be zoned UI. Under the Preferred Alternative new areas near Georgetown would be brought into the UI zone in the vicinity of S Orcas Street connecting to areas of existing Commercial zoning on 4th Avenue S. These areas would be expected to see increased infill development with light industrial uses, brewers/makers, and small manufacturers with large ancillary spaces, and industry supportive housing is expected. Use compatibility conflicts would be like those described for Alternative 3, but in some new areas. Due to the increased connectivity of lands zoned UI with existing commercial or residentially zoned areas in Georgetown, over the long term the degree of land use incompatibility could decrease if the area transitions to a more cohesive urban village-like environment.

Land use changes and resulting compatibility impacts in the triangular area that would be removed from the MIC in Georgetown and the land removed from the MIC at the edges of South Park and placed into a mixed-use zone would be similar to Alternative 3. However, compared to Alternative 3, the degree of incompatibility would be reduced because the Preferred Alternative includes unique zoning provisions for the Georgetown area. Standards would incentivize the preservation of historic character structures and arts spaces. The scale of development would also be limited to 55'. These features would reduce the estimated housing production in Georgetown to 392 housing units, compared to over 700 in alternatives 3 and 4. Use compatibility impact between new development and the existing track 101 spur would be reduced in the Preferred Alternative compared to alternatives 3 and 4. There would still be incompatibility impacts with respect to noise and vibration affecting non-industrial uses and potentially increased safety risks. New land uses could make it less practical for rail operators to use the spur, causing incremental pressure on the railyard to consider abandoning the spur. However, conditional use criteria on new housing would mitigate potential incompatibilities by ensuring improved configurations for developments with housing and sound insulation.

Overall, while there is potential to reduce land use conflicts by creating cohesive mixed-use districts in Georgetown and South Park over time, the process would result in interim land use compatibility impacts that rise to the level of **moderate**.

Employment Mix

As seen above in **Exhibit 3.8-17**, the overall employment under the Preferred Alternative would increase by 35,545 jobs. The mix of industrial employment would decrease by 1.5% points compared to 54.4% in the Alternative 1—No Action, but total industrial employment would increase by 4,453 jobs. The percentage of industrial employment would remain at 50% or greater in every subarea under the alternative except for the Ballard Subarea, which dip just slightly below to 49.8%. Although there would be an increase in non-industrial employment in office and ICT sectors, the increase in industrial employment due to stronger protections in the MML zones, inclusion of new light industrial space in II zone development, and mixed industrial redevelopment of the Armory site would result in bigger increases in industrial employment. Since the employment mix of industrial would drop to 49.8% in Ballard—at the threshold for percentage of industrial employment in MICs per regional criteria—a minor adverse impact in employment mix is present for the Ballard Subarea.

Land Use Transitions

<u>The Preferred Alternative creates slightly less pronounced changes with respect to transitions</u> <u>than Alternative 4 and would result in similar levels of land use transition impact to Alternative</u> <u>4. Like Alternative 4, zoning transitions generally preclude the potential for high impact/high</u> intensity uses to abut or encroach on residential areas but could result in some impacts related to height, bulk, scale, and aesthetics.

<u>Ballard</u>

The central part of the Ballard Subarea bisected by 14th Avenue W is proposed to be rezoned from IG1 to II. This change is expected to result in a mix of light industrial and commercial development and ICT uses. The change from IG to II will provide a better transition to nonindustrial areas to the north by reducing the likelihood of high impact/high intensity uses encroaching or abutting residential areas. In northwest Ballard where industrial land abuts nonindustrial land, the proposal is to rezone existing IC zoned land to II providing a similar transition as Alternative 1—No Action and maintaining the unlikely potential for high intensity/high impact uses to abut or encroach on nonindustrial areas. North of NW Market Street and west of 24th Ave NW, land currently zone Industrial Buffer would be rezoned to a mixed-use commercial zone. Larger and taller buildings anticipated under the proposed II zone, particularly near transit stations, have the potential for impacts related to height, bulk, scale, and aesthetics adjacent to nonindustrial areas. Impacts of the proposal in Ballard are **moderate**.

Interbay Dravus and Interbay Smith Cove

With respect to land use transitions, the proposed land use changes in the Interbay Dravus and Interbay Smith Cove subareas are identical to those proposed in Alternative 3 and the resulting impacts are the same.

SODO/Stadium

With respect to land use transitions, the proposed land use changes in the SODO/Stadium Subarea are the same as Alternative 3 except for a small node of land on the west side of Harbor Avenue SW which is proposed to be changed from IB to UI. The impacts are as described for alternatives 3 and 4 above. Potential impacts stemming from the addition of II zoned land south of S Lander Street. In the Preferred Alternative are addressed in the compatibility discussion above.

Georgetown/South Park

Similar to alternatives 3 and 4 the Preferred Alternative proposes the removal of some land from both the Georgetown and South Park neighborhoods from the MIC and rezones it with a mixed-use commercial zone. In the Preferred Alternative the triangular area in the center of Georgetown that is removed from the MIC and placed in a mixed-use zone is slightly larger than in alternatives 3 and 4 because it includes the Georgetown Playfield and Park. This change will not erode the existing transition from the core industrial areas and removes the potential for high impact/high intensity to encroach or abut nonindustrial areas. Additionally, more area in Georgetown is changed to the Urban Industrial zone under the Preferred Alternative and would connect between the commercially zoned stretch on 4th Avenue S to the residential part of the Georgetown neighborhood along S Orcas Street. Over time, land uses in these areas of change in Georgetown would become more commercial and residential in character. Creation of a contiguous geography for the mixed use neighborhood could incrementally improve land use transitions compared to all of the other Alternatives including No Action. Although the potential for land use transition impacts is reduced in the Preferred Alternative there is still potential under the proposal for **minor** impact as discussed for alternatives 3 and 4.

Other Industrial Zoned Lands

For industrially zoned areas outside of the MICs, the Preferred Alternative differs from alternatives 2, 3, and 4 where it would rezone two areas from an existing Industrial Commercial zone and apply a Neighborhood Commercial mixed-use zone (west edge of Ballard along S Market Street, and Judkins Park between S Poplar Street and Rainier Avenue S). There is strong potential for additional housing development in these localized areas in mixed use buildings with ground level retail and multiple stories of apartments above. In west Ballard this change could cause a minor transition impact with respect to height, bulk, and scale, if multistory new development occurs adjacent to other lowrise multifamily development at the townhouse scale directly to the north. However, this pattern would be consistent with other areas in Ballard and is not expected to be more than minor. The area in Judkins Park that would allow for new mixed use development would not cause a transition impact because it is bordered by the wide roadway of Rainier Avenue S, and mixed light industrial uses to its west would not be adversely affected by the increased scale of infill development. Otherwise, the Preferred Alternative treats industrial lands outside of MICs similarly to Alternative 4, and no more transition impact would be realized.

Summary of Impacts

Exhibit 3.8-18 summarizes adverse impacts under each alternative by subarea. The degree of impact varies within subareas and may only manifest in a subset of locations. The greatest adverse impact identified within each subarea is listed in **Exhibit 3.8-18**, below.

Category of Land Use Impact		Alt. 1	Alt. 2	Alt. 3	Alt. 4	Pref. Alt.
Consistency with Plans & Policies	Ballard		Minor	Moderate	Moderate	<u>Moderate</u>
	Interbay Dravus	Moderate				
	Interbay Smith Cove					
	SODO/Stadium					
	Georgetown/South Park					
Compatible	Ballard	Moderate	Minor	Moderate	Moderate	<u>Moderate</u>
Uses	Interbay Dravus	Moderate	Minor	Minor	Moderate	<u>Moderate</u>
	Interbay Smith Cove	Moderate	Minor	Minor	Minor	<u>Minor</u>
	SODO/Stadium	Moderate	Minor	Moderate	Moderate	<u>Moderate</u>
	Georgetown/South Park	Moderate	Minor	Moderate	Moderate	<u>Moderate</u>
Transitions	Ballard	Moderate	Moderate	Moderate	Moderate	<u>Moderate</u>
	Interbay Dravus	Moderate	Moderate	Moderate	Moderate	<u>Moderate</u>
	Interbay Smith Cove	None <u>Minor</u>	None <u>Minor</u>	None <u>Minor</u>	None <u>Minor</u>	<u>Minor</u>
	SODO/Stadium	<u>Moderate/</u> Minor	<u>Moderate/</u> Minor	<u>Moderate/</u> Minor	<u>Moderate/</u> <u>Minor</u> None	<u>Moderate/</u> <u>Minor</u>
	Georgetown/South Park	Minor	Minor	Minor	Minor	Minor
Employment Mix	Ballard	None	None	None	Minor	<u>Minor</u>
	Interbay Dravus	None	None	None	None	<u>None</u>
	Interbay Smith Cove	None	None	None	None	<u>None</u>
	SODO/Stadium	None	None	None	None	<u>None</u>
	Georgetown/South Park	None	None	None	None	None

Exhibit 3.8-18 -16	Summary of Land L	Jse Impacts b	y Subarea and	Alternative
-------------------------------	-------------------	---------------	---------------	-------------

Source: City of Seattle, 20221.

3.8.3 Mitigation Measures

Incorporated Plan Features

Many of the potential land use impacts are mitigated down to non-significant level by incorporated plan features that are a part of the proposal. These aspects are described elsewhere in the Chapter; especially important mitigating features are highlighted below.

 Reduced maximum size of use limits. Proposed MML zone standards include maximum size of use limits of 10,000 sq. ft for offices, medical services (and others), a 7,500 limit for general retail sales, and 3,000 sq. ft. for bars and restaurants. These are significant reductions compared to current IG zones. The proposed UI zone also includes reduced maximum size of use limits for stand-alone non-industrial uses. These reductions reduce the potential for incompatible use and employment mix impacts.

- Incentive structure in the II zone. The incentive bonus system would ensure that any new non-industrial development includes bona-fide, newly constructed industrial space. New development of high value uses supports the construction of new space for industrial uses. This contrasts with the existing IC zone, in which new development frequently includes no industrial space. This feature mitigates potential plan consistency, incompatible use, and employment mix impacts. Under the Preferred Alternative the industrial space required to achieve the bonus could be built off-site within an MML zone in the same MIC. This could strengthen the preservation of core industrial areas for industrial uses because it would incentivize new investment in solely industrial uses in MML-zoned areas.
- Limits on changes to MIC boundaries. The proposed Comprehensive Plan policy to limit changes to MIC boundaries will mitigate potential future impacts related to incompatible land uses in all alternatives.
- Limitations on Occupancy of Industry-Supportive Housing. The limitation on occupancy mitigates potential incompatible use and policy inconsistency impacts. With a limitation on residents to persons engaged in and familiar with industrial operations and/or making/arts, new residents introduced into industrial areas would have greater understanding of the impacts (noise, odors etc.) compared to the general population. Residents would have better understanding of safety protocols and potential hazards of an industrial area. They would be less likely to levy complaints against industrial businesses, and more likely to use protections in appropriate situation such as safety glasses and hearing protection. Potential residents would be more likely to have full awareness of any potential hazards when choosing whether to live there compared to the general population. These factors significantly reduce adverse effects typically associated with introduction of residences into an industrial area. In the Preferred Alternative, the housing allowance in the UI zone would be by conditional use permit only, affording the City more opportunity to prevent incompatible locations or configurations for new homes. The Preferred Alternative also provides the option to provide the housing as 50% affordable to households with income at or below 90% AMI. This workforce affordable housing option would still provide some of the mitigating benefits of occupancy limitations because there is a likelihood that area workers in the target income bands would be relatively more likely to choose the housing that is made available.
- Comprehensive Plan Policy Amendments. Comprehensive Plan goal and policy amendments set a new vision and guidance for the city's industrial areas and address new aspects such as high-capacity transit. Without the foundational policy amendments all the Action alternatives would likely have significant adverse impacts on consistency with the current Comprehensive Plan policy framework. Since the plan amendments are an integrated part of the proposal, policy inconsistency is mitigated down to a non-significant level for all Action Alternatives.

- Development standards in the UI zone. Reduced setbacks would allow for construction on more small sites, which provides buffering affects. The standards also include increased requirements (compared to the IB zone) for urban style streetscape improvements and would introduce the green factor landscaping requirement. Development standards for development in the UI zone These features reduce the potential for transition impacts.
- Completion of MIC subarea plans. The proposal includes completion of subarea plan updates for the Greater Duwamish MIC and BINMIC. The plans would be updated to reflect the umbrella policy updates in the Comprehensive Plan, and the land use and zoning changes described in this EIS. Data and information included in Chapters of this EIS and related studies would be integrated into the subarea plans. Additionally, the plans would address more location specific strategies for integration of amenity features, open spaces, configuration of circulation improvements and other non-land use features. Completion of the plans will mitigate potential use compatibility, transition impacts.
- Georgetown Unique Development Standards. The Preferred Alternative includes
 provisions to incentivize the preservation of historic structures and arts spaces in new
 mixed-use zoning areas in Georgetown (see Development Standards appendix). This feature
 would incrementally reduce the potential for incompatible use impacts because it would
 dampen the potential for dense new market rate housing, and it would encourage retention
 and growth of arts-oriented uses of which there is an existing concentration in Georgetown.
 The Preferred Alternative would apply the moderate height limit of 55' to the triangular area
 in Georgetown which would be less than some existing structures in the vicinity, mitigating
 the potential for any transition impacts due to building scale.
- Workforce Development Space Incentive. Under the Preferred Alternative the II zone would include a Floor Area Ratio exemption for space that is dedicated for use as workforce development or vocational training. The same space would qualify as light industrial space to achieve bonus development capacity. The feature could help encourage employment in industrially-related fields in MICs mitigating potential employment mix impacts.
- Improve At-Grade Rail Crossing Safety Features. Incompatible use impacts could be mitigated if the City and partners installed improved at-grade crossing safety features near the track 101 rail spur in Georgetown, such as electric crossing warning signals and gates. These features could mitigate safety risks for community members in the vicinity and lessen pressures on the rail operator to abandon the spur.

Regulations & Commitments

Many of the potential land use impacts are mitigated down to non-significant level by the presence of existing regulatory commitments that would apply with or without the proposal.

Shoreline Master Program (SMP). The existing SMP regulations are unchanged and will continue to apply to all new development. SMP regulations supersede underlying zoning. Many of the SMP regulations supporting protections for industrial maritime activities at the shorelines in industrial areas under all alternatives. These designations require water-

dependent and water-related uses at the shoreline and will provide protection from incompatible land uses for all alternatives for land that is within 200' of the shoreline.

- SEPA Project Level Review. The existing State Environmental Policy Act (SEPA) regulations are unchanged and will continue to apply to all new development at the time of project level review. SEPA project level review would apply to any development proposal that includes 4 or more residential units, or 12,000 or more sq. ft. of non-residential development. Site specific factors would be considered at the time of project level SEPA review, and development projects could be conditioned to address any localized impacts pursuant to Chapter 25 of the SMC and other State RCW 43.21C.
- Noise Ordinance. Application of the City's Noise Ordinance (SMC 25.08) can mitigate impacts from poor transitions from industrial areas to nonindustrial areas by limiting noise impacts to adjacent areas.
 - Nonconforming Use Provisions. Nonconforming uses are permitted to continue subject to provisions of the Seattle Land Use Regulations (SMC Subtitle III). Under existing regulations, a nonconforming use that has been discontinued for more than 12 consecutive months shall not be reestablished or recommenced (SMC 23.42.104(B)) and would need to adhere to the underlying zoning regulations if redeveloped. As a part of the proposal the City would add flexibility for nonconforming uses in the MML zone. In the MML zone special accommodation will be given to allow nonindustrial uses that exceed maximum size of use limits prior to the adoption of legislation establishing the MML zone to reestablish or recommence without a time limit. Additional flexibility would also be provided to allow for existing commercial office uses with an operational connection to an industrial use or an existing Information Computer Technology (ICT) use to expand beyond maximum size of use limits. The added flexibility mitigates the potential for unintended land use impacts of industrial or maritime businesses displacement because of difficulty expanding in the MML zone.

Other Potential Mitigation Measures

Though no significant adverse land use impacts are identified, it would be possible to further mitigate the identified moderate and minor land use impacts with the following actions. Incorporation of these actions would reduce the likelihood that any of the impacts could potentially become significant.

- Apply maximum size of use limits to industrial zones in Alternative 1. If Alternative 1— No Action is selected, expected use incompatibility impacts and policy conflict impacts could be reduced by incorporating maximum size of use reductions for office and retail uses (similar to the MML zone) into the existing Industrial General zones. This could be standalone legislation. The maximum size of use limits could be applied to areas only within designated MICs in order to provide continued flexibility for IG zoned areas outside of MICs.
- Limit the geography of industry-supportive housing and monitor. Incompatibility, transition, and policy inconsistency impacts could be mitigated to a lower level if the proposed industry supportive housing allowances are initially limited to a smaller

geography. Limits could test the concept in a pilot area, or the proposed UI zone could include versions with and without the expanded housing allowances. The City and partners could monitor the initial effects of the expanded housing allowances for an initial test period of 3–5 years, then consider applying to more areas. Stakeholders in industrial areas such as community organizations, Business Improvement Areas (BIAs) and trade groups could be involved in the monitoring process through formation of a stewardship group.

- Update zoning at edge areas outside of the study area in the future. Changes include limiting significant housing development in adjacent mixed-use zones to reduce potential impacts related to inadequate transitions from industrial to nonindustrial areas, particularly where core industrial zones are located close to these transitions. Changes could include application of the proposed Urban Industrial zone to more areas outside of industrial areas, including in some urban villages.
- Contributions towards equitable development. There have been historic impacts from industrial activities on populations including indigenous communities that preceded this proposal. It is plausible that continuation of land uses according to City and regional policies could perpetuate past harms according to some populations including indigenous peoples. As a voluntary measure unrelated to impacts of any of the proposal's alternatives, current owners of land could support equitable development for indigenous groups by developing a broad-based system of contributions to community building and resilience. The contributions could take forms such as donations to ongoing community development initiatives identified in the Duwamish Valley Action Plan, or participation in the Duwamish's Real Rent program. As infrastructure investments are made in the study area, promote equitable phasing and locations to reduce historic impacts with input from affected community members. Examples include improving parks and streetscapes to reduce heat island effects, improving existing transitions to residential areas, improving noise attenuation to residential areas, and reducing existing risks of sea level rise.
- Design Guidance for development in the UI and II zones when abutting nonindustrial areas. Non-codified design guidance to address impacts associated with height, bulk, scale, and aesthetics, and design treatments appropriate for the edges of industrial areas could be a resource for developers and community members alike in developing projects that abut nonindustrial areas.
- Amend Substantial Alteration Thresholds. The City could review and amend its practice of determining when the threshold for a building substantial alteration is exceeded in industrial zones, especially the UI zone. When a substantial alteration threshold is exceeded, construction must upgrade to current energy and seismic code standards. This can potentially disincentivize the adaptive reuse of older warehouse style structures that were common in industrial areas. To allow for adaptive reuse more often to achieve the intent of the UI zone, the City could consider more forgiving determinations of substantial alteration.

3.8.4 Significant Unavoidable Adverse Impacts

Across all alternatives the City found minor and moderate impacts related to land and shoreline use:

- Inconsistency with Plans and Policies: Some degree of inconsistency between the expected land use pattern and plans and policies was found for all the alternatives. Since consistency of land use patterns with plans and policies requires interpretation and balancing with many policies, it is common for some inconsistency to exist, while maintaining an overall predominant level of consistency. Alternative 1-No Action would have moderate inconsistencies due to the likely continuing trend of stand-alone retail and office development and mini-storage locating in industrial zones and MICs under existing zoning. This is inconsistent with certain policies prioritizing industrial and maritime uses in these areas. Moderate inconsistencies would be present under alternatives 3 and 4 and the Preferred Alternative due to the introduction of increased amount of industry-supportive housing, which can be viewed as inconsistent with some regional and local policies limiting residential uses in MICs. Alternative 2 would have the fewest, and only minor, inconsistencies because Alternative 2 would reduce the prevalence of non-industrial uses in industrial areas through new standards in the proposed MML zone in larger areas than alternatives 3 and 4 and the Preferred Alternative, and Alternative 2 does not include expanded allowances for housing.
- Incompatible Land Uses: Moderate incompatible use impacts are expected in all subareas under Alternative 1 due to the potential for stand-alone retail and office developments and mini-storage to locate in industrial areas causing potential incompatibility with industrial uses. Alternatives 3 and 4 and the Preferred Alternative would see moderate incompatible use impacts in some subareas—most notably Ballard, Stadium/SODO, and Georgetown/South Park—where introduction of new buildings with dense employment in the II zone and industry-supportive housing in the UI zone could create incompatibilities between new activity patterns and adjacent areas of continued industrial uses. Alternative 2 would have the fewest, and only minor, land use incompatibilities since the application of the II and UI zones would be more limited in scale.
- Inadequate Transitions: Potential for inadequate transitions from industrial to nonindustrial areas is highest for the Ballard and Interbay Dravus subareas. Moderate impacts at transitions would be expected in the Ballard and Interbay Dravus subareas under all the alternatives, and in Ballard under alternatives 1, 2, and 3. In general, portions of the study area that abut residential and urban village locations without strong physical edge features such as greenbelts, major roadways or topographical changes have greater potential for inadequate transition. Future land use under the UI zone is expected to assuage potentially inadequate transitions to residential and urban village areas, thus Alternative 4, which includes more UI zoning in the Ballard Subarea would have moderate transition impacts. Minor transition impacts are identified for the Georgetown/South Park Subarea under all the alternatives, and for the <u>SODO/</u>Stadium/SODO <u>Sub</u>area under alternatives 1, 2, and 3 and the Preferred Alternative. No transition impacts are expected for

Interbay Smith Cove under any alternative primarily because of the strong physical edges around the subarea.

Employment Mix Impacts: With one-two exceptions, no employment mix impacts are expected. In all subareas and under all alternatives, the projected employment mix would remain 50% or more industrial—one of the threshold criteria for regional designation as a MIC. A minor employment mix impact was identified in Alternative 4 and the Preferred Alternative for the Ballard subarea, where the percentage of industrial employment is projected to fall to a level approaching the 50% threshold.

Under all of the alternatives, any inconsistencies with plans and policies, incompatible land uses, undesired employment mixes, or inadequate land use transitions described above would be minimized and reduced to less than significant levels via incorporated plan features and existing regulations and commitments. No significant unavoidable adverse impacts to land or shoreline use are anticipated under any of the alternatives.

Section 3.9 Housing



This section summarizes the affected environment—including the current housing policy framework, and current housing in the study area—and compares impacts of the alternatives on housing in the study area.

Three impact thresholds were used to identify potential adverse housing impacts in the study area. Impacts of the alternatives on housing are considered significant if they:

- Result in **loss of housing due to redevelopment** and insufficient development capacity, tools, or programs to address displacement of dwellings and population.
- Potential to increase households' exposure to air pollution, noise pollution, or environmental hazards in census tracts identified as having high environmental health disparities and with sensitive populations.
- Creation of demand for housing that cannot be accommodated within the city in adjacent districts or areas where housing is planned.

Mitigation measures and a summary of any significant unavoidable adverse impacts are included following the impacts analysis.

3.9.1 Affected Environment

The study area consists of lands used and zoned for industrial purposes, primarily in the BINMIC and Greater Duwamish MIC. Though these areas are predominantly used for employment there remain scatted residential dwellings. Some are caretakers' quarters.

The data and methods considered in this section include: housing inventory, production trends, and challenges and needs (including public health, access to opportunity and displacement risk) based on U.S. Census American Community Survey, City of Seattle, and King County Assessor data.

Current Policy & Regulatory Framework

Existing housing patterns in the study area are influenced by the current land use policy and regulatory framework. This framework flows from the State of Washington Growth Management Act (GMA), the Puget Sound Regional Council's (PSRC's) Multi-County Planning Policies (MPPs), King County's County-Wide Panning Policies (CPPs) the City Comprehensive Plan (Seattle 2035), and implementation actions including development standards in the Seattle Municipal Code (SMC) and the City's Shoreline Master Program. Several other regulatory measures affect industrial land use including localized overlay districts and community agreements.

Detailed descriptions of the framework are included in **Section 3.8 Land & Shoreline Use**.

Housing Inventory & Production

This section characterizes existing housing patterns in the study area and breaks out housing patterns for the EIS subareas where information is available and useful.
Existing Housing Inventory

As of 2020, the study area included an estimated 413 housing units. More than half (54%) of housing units in the study area are in multi-unit apartment buildings while 32% of the area's housing units are in single-family buildings (as defined by the King County Assessor). Relatively smaller numbers of housing units are duplexes and 4-plexes. **Exhibit 3.9-1** below presents the units by housing type within the study area.

Housing Type	Ballard	lnterbay Dravus	Interbay Smith Cove	SODO/ Stadium	Georgetown	Total
Single-family*	49			9	78	136
Duplex	9				15	24
4-plex	20				12	32
Apartments	111	3	1	12	91	218
**Other	3					3
Total	192	3	1	21	196	413

Exhibit 3.9-1	Study Area Housing Units by Type by Suba	area, 2021
---------------	--	------------

*Detached single family may include some accessory dwelling units. King County Assessor does not track ADUs or DADUs separately so we cannot reliably summarize the number of ADUs in this inventory. It is also possible there are many additional units in ADUs that are not included in the totals. Between 1994 and 2020, Seattle permitted 862 DADUs and about 1,900 ADUs.

**Housing units classified as "Other" include unique residence types such as houseboats, caretaker quarters, housing attached to private schools and churches, and housing units in certain historic properties.

Source: King County Assessor, 2020; BERK, 2021.

Most of the housing in the study area is in the Ballard (46%) and Georgetown/South Park (47%) subareas.

<u>Ballard</u>

The Ballard Subarea consists of the land between the Salmon Bay shoreline and the Ballard Urban Village. For the purposes of this analysis the subarea also includes portions of the study area in the Fremont Urban Village and along the north and east shores of Lake Union.

Housing in this subarea is located along the northern edge where the industrial areas are adjacent to more residential and commercial areas in Ballard, primarily the scattered single family and multi-family homes in blocks flanking 14th Avenue NW.

There are roughly 192 housing units in the Ballard Subarea. More than half these units are apartments. Single-family homes constitute a little more than 20% of housing units in the subarea. There are a small number of duplexes and 4-plexes. See **Exhibit 3.9-2**.

Housing Type	Percentage of Residential Structures	Percentage of Units
Single-family	59.7%	22.4%
Duplex	5.6%	4.7%
4-plex	5.6%	10.4%
Apartments	26.4%	57.8%
Other	2.8%	1.6%

Exhibit 3.9-2 Housing Type by Structure and Units, Ballard

Source: King County Assessor, 2020; BERK, 2021.

Interbay Dravus and Interbay Smith Cove

The Interbay Dravus and Interbay Smith Cove subareas consists of three distinct nodes— Fisherman's Terminal and vicinity, Dravus, and Smith Cove. These subareas stretch from the southern shoreline of Salmon Bay between the locks and ship canal on the north and Elliott Bay to the South, and are bound by the Queen Anne and Uptown neighborhoods to the east and Magnolia to the west. Both subareas contain very little housing. The Interbay Dravus Subarea includes only three units characterized as apartments in the assessor data (**Exhibit 3.9-3**) and the Interbay Smith Cover Subarea includes one apartment building (**Exhibit 3.9-4**).

Exhibit 3.9-3 Housing Type by Structure and Units, Interbay Dravus

Housing Type	Percentage of Residential Structures	Percentage of Units
Apartments	100%	100%

Source: King County Assessor, 2020; BERK, 2021.

Exhibit 3.9-4 Housing Type by Structure and Units, Interbay Smith Cove

Housing Type	Percentage of Residential Structures	Percentage of Units
Apartments	100%	100%

Source: King County Assessor, 2020; BERK, 2021.

SODO/Stadium

The SODO/Stadium Subarea includes the mouth of the Duwamish River where it outlets to Elliott Bay. The SODO/Stadium Subarea includes 21 housing units. About one-half of the units are in apartments and the other half are single-family homes. The Subarea has no duplexes or 4-plexes. See **Exhibit 3.9-5**.

Housing Type	Percentage of Residential Structures	Percentage of Units
Single-family	90%	48%
Apartments	10%	52%

Exhibit 3.9-5 Housing Type by Structure and Units, SODO/Stadium

Source: King County Assessor, 2020; BERK, 2021.

Georgetown/South Park

The Georgetown portion of the subarea is situated on the east bank of the Duwamish River. The study area surrounds two residential areas in the Georgetown neighborhood—the Van Asselt district between Ellis Avenue S and Corson Avenue S and a roughly four-block residential district between S Homer Street and S Fidalgo Street. Both areas include townhomes, single family, and multifamily housing including some new construction. Residents of these areas are closely adjacent to the surrounding industrial activities.

The South Park portion of the study area is situated on the west bank of the Duwamish River. The study area contains only the industrial lands that surround the South Park neighborhood, which is a mixed-use neighborhood that is designated as a residential urban village in Seattle's Comprehensive Plan.

Approximately 196 housing units are scattered throughout the subarea, especially along the edges. Single-family homes constitute roughly 40% of the housing units in the subarea. There are a small number of duplexes and 4-plexes. See **Exhibit 3.9-6**.

Housing Type	Percentage of Residential Structures	Percentage of Units
Single-family	84%	40%
Duplex	7%	8%
4-plex	3%	6%
Apartments	6%	46%
Other	0%	0%

Exhibit 3.9-6 Housing Type by Structure and Units, Georgetown

This subarea includes three hotels/motels that are not included in the unit count. Source: King County Assessor, 2020; BERK, 2021.

Age of Existing Housing

The Study Area has seen little housing development in the past twenty years. Roughly 32% of the housing in the Study Area was built prior to 1950, 62% were built between 1950 and 2000, and 17% were built in and after 2000. See **Exhibit 3.9-7**.



Exhibit 3.9-7 Housing Units by Year Built, Study Area

Source: King County Assessor, 2020; BERK, 2021.

Housing Production Trends

Citywide Trends

Between 2010 and 2019, Seattle added over 69,000 new housing units and demolished nearly 6,000 older housing units, for a net gain of over 63,000 units in total. On average, the city gained 6,300 new units per year, with annual production increasing most years from a low of 2,340 in 2011 following the last economic recession to a high of 10,651 in 2019. Citywide, however, housing production has not kept pace with employment growth, leading to an increasing supply shortage (City of Seattle 2021).

Nearly all of Seattle's capacity for residential growth is in villages/centers and corridors with mixed-use and multifamily zoning. According to analysis of development (2010-2019) by year built in King County Assessor data by far, the largest share of new development is in the Greater Downtown market area, followed by the North Central area which stretches from Ballard in the west to northeast Seattle in the east (City of Seattle 2021).

Subarea Trends

City permit data shows that the industrial areas are not locations for significant housing development. A total of 62 housing units were added to the subareas between 2000 and 2021. Housing ancillary to units attached to commercial development accounted for the bulk of these units. See **Exhibit 3.9-8**.

	Ballard	Interbay Dravus	Interbay Smith Cove	SODO/ Stadium	Georgetown/ South Park	Total
Single Family/Duplex	1	0	0	0	0	1
Multifamily	0	0	0	1	0	1
Commercial	11	16	4	8	1	40
Industrial	3	0	2	1	3	9
Institutional	0	0	0	1	0	1
Vacant Land	0	0	2	0	8	10
Total	15	16	8	11	12	62

Exhibit 3.9-8 New Housing Added by Permit Class, 2000-2021

Source: City of Seattle permit data, 2021.

Housing Challenges, Needs, & Considerations

Displacement Risk

As a companion document to the Seattle 2035 Comprehensive Plan EIS, Seattle's Growth and Equity Analysis examined demographic, economic, and physical factors to evaluate the risk of displacement and access to opportunity for marginalized populations across Seattle neighborhoods. The findings are expressed as the Displacement Risk Index in this section and the Access to Opportunity Index in the following section.

The Displacement Risk Index identifies areas of Seattle where displacement of marginalized populations may be more likely. It combines data about demographics, economic conditions, and the built environment into a composite index of displacement risk. It focuses on displacement that affects marginalized populations, defined in the Seattle 2035 Comprehensive Plan as people of color, people with low incomes, English-language learners, and people with disabilities. It reflects data on vulnerability, amenities, development capacity, and rent to identify where displacement of those populations is more likely to occur. The map below shows areas of the city according to their level of displacement risk.

Exhibit 3.9-9 illustrates this index for Seattle and the study area. Overall, parcels within the study area are at low or moderate risk for displacement.



Exhibit 3.9-9 Displacement Risk Index

Access to Opportunity

Historic practices such as redlining, and more modern policies have shaped access to opportunity across the city. As a result, access to neighborhoods with large parks, more trees, and walkable streets varies significantly by race. Marginalized populations tend to live in areas (in Seattle or elsewhere) with fewer opportunities.

Seattle's Growth and Equity Analysis (2016) examined demographic, economic, and physical factors to evaluate the risk of displacement and access to opportunity for marginalized populations across Seattle neighborhoods. The findings are expressed as the Access to Opportunity Index in this section and the Displacement Risk Index in the previous section.

The analysis considers marginalized populations' access to some key determinants of social, economic, and physical well-being. This includes data in the following categories: education, economic opportunity, transit, civic infrastructure, and health. The index captures a broad range of indicators that measure access to some of the resources that residents need to succeed and thrive.

Exhibit 3.9-10 illustrates this index for Seattle and the Study Area. Overall, parcels within the study area have low or moderate access to opportunity. Some limited areas in the Ballard subarea are seen to have relatively higher access to opportunity.



Exhibit 3.9-10 Access to Opportunity Index

Source: City of Seattle, 2016; BERK, 2021.

Jobs/Housing Balance & Worker Demographics

Another indicator of housing challenges is the jobs/housing ratio. Data show that housing production has not kept pace with employment growth in Seattle. In 2005 there were 1.8 jobs for every one housing unit in Seattle. Between 2005 and 2019, the city gained about 169,000 net new jobs. Over the same time, Seattle would have needed to increase its housing production by an additional 9,000 units just to maintain its 2005 jobs to housing ratio of 1.8.

Balancing jobs and housing within a city can reduce commuting and improve traffic congestion and air quality. A jobs/housing imbalance can cause upward pressure on housing costs. In employment centers, local workers may have no choice but to pay higher prices to avoid longer commutes.

Based on 2019 Census On the Map information, study area workers are primarily aged 30-54 (56.2%), earn more than \$3,333 (65%), two thirds white and one third persons of color (34.7%), and two thirds male and one third female (34.3%). About 35% are earning less than \$3,333 per month, which at about \$40,000 would be less than 80% of the area median income. Lower wage workers are especially vulnerable to displacement risks. Those who move to more affordable communities further from employment centers face longer commutes. While not all Seattle workers may wish to live in the city, workers in low-wage jobs who are commuting very long distances are a good indicator of a lack of an adequate supply of affordable housing in the city.

Exhibit 3.9-11 shows the distance traveled by workers in industrial subareas. Roughly 37% of workers (29,543) travel 10-24 miles one-way to get to their jobs. The remainder travel more than 25 miles each way between home and work.

Distance	Count	Share
Less than 10 miles	31,471	39.7%
10 to 24 miles	29,543	37.3%
25 to 50 miles	10,592	13.4%
Greater than 50 miles	7,604	9.6%
Total All lobs	79,210	100.0%

Exhibit 3.9-11 Distance Traveled by Workers in Study Area, 2018

Source: Census LEHD Origin-Destination Employment Data, 2018; BERK, 2021.

Workers in industrial areas commute from homes in Seattle, other parts of King County, Snohomish County, and Pierce County. See **Exhibit 3.9-12** and **Exhibit 3.9-13**.

City	Count	Share
Seattle city, WA	22,769	28.7%
Kent city, WA	2,853	3.6%
Renton city, WA	2,452	3.1%
Burien city, WA	2,108	2.7%
Tacoma city, WA	1,937	2.4%
Federal Way city, WA	1,902	2.4%
Bellevue city, WA	1,841	2.3%
Shoreline city, WA	1,419	1.8%
Auburn city, WA	1,296	1.6%
Kirkland city, WA	1,154	1.5%
Everett city, WA	1,118	1.4%
Des Moines city, WA	924	1.2%
SeaTac city, WA	921	1.2%
Edmonds city, WA	905	1.1%
Tukwila city, WA	823	1.0%
Sammamish city, WA	741	0.9%
White Center CDP, WA	738	0.9%
Lynnwood city, WA	691	0.9%
Marysville city, WA	660	0.8%
Redmond city, WA	646	0.8%
Bothell city, WA	624	0.8%
Bryn Mawr-Skyway CDP, WA	554	0.7%
Mountlake Terrace city, WA	525	0.7%
South Hill CDP, WA	521	0.7%
Issaquah city, WA	501	0.6%
All Other Locations	28,587	36.1%

Exhibit 3.9-12 Top 25 Places of Worker Residence by Count/Percent

Source: Census LEHD Origin-Destination Employment Data, 2018; BERK, 2021.



Exhibit 3.9-13 Home Location of Workers with Jobs in the Study Area, 2018

Source: Census LEHD Origin-Destination Employment Data, 2018; BERK, 2021.

Public Health

The Washington Environmental Health Disparities Map (EHD Map) is an existing tool created by DOH and others that ranks environmental health disparities by census tract. It is an interactive tool that combines the most comprehensive data available to rank Washington communities according to the risk each faces from environmental factors that influence health outcomes. The EHD includes fossil fuel exposure as well as social and health vulnerability measures. The map shows pollution measures such as diesel emissions and ozone, as well as proximity to hazardous waste sites. In addition, it displays measures like poverty and cardiovascular disease.

The data on the map include 19 indicators and are divided into four themes:

- Environmental Exposures (NOx-diesel emissions; ozone concentration; PM2.5 Concentration; populations near heavy traffic roadways; toxic release from facilities (RSEI model))
- Environmental Effects (lead risk from housing; proximity to hazardous waste treatment, storage, and disposal facilities (TSDFs); proximity to National Priorities List sites (Superfund Sites); proximity to Risk Management Plan (RMP) facilities; wastewater discharge)
- Sensitive Populations (death from cardiovascular disease; low birth weight)
- Socioeconomic Factors (limited English; no high school diploma; poverty; race—people of color; transportation expense; housing cost burden; unemployment)

The EHD map ranks the risks communities face from environmental burdens including fossil fuel pollution and vulnerability to climate change impacts that contribute to health inequities. The EHD map is based on a conceptual formula of Risk = Threat x Vulnerability. Threat is comprised of both environmental effects and exposures, and vulnerability is comprised of socioeconomic factors and sensitive populations. It is a well-known vulnerability index for environmental health disparities and is being used by state processes to guide funding to reduce environmental health disparities.

Industrial areas in the Greater Duwamish MIC are ranked at high risk based on environmental factors that influence health. See **Exhibit 3.9-14**. This map is aligned with several studies that have documented the disproportionately high environmental health burdens and risks relative to the rest of Seattle that communities in the Duwamish Valley experience. Exposure to air pollution, noise pollution, and highways is higher in the Duwamish Valley than the city average and access to open space is lower. See **Exhibit 3.9-15** breaking down potential exposure to environmental exposures to NOx-Diesel emissions, Ozone, PM 2.5, and potential toxic releases from facilities. **Exhibit 3.9-16** illustrates census tract populations near heavy traffic roadways. **Exhibit 3.9-17** shows a moderate proximity to hazardous waste sites compared to other census tracts in Washington State.

The Duwamish River is a 5.5-mile Superfund site, and the City is working closely with the U.S. Environmental Protection Agency (EPA) on cleanup and source control efforts. While cleanup is ongoing, health advisories are still in place. The Duwamish Valley is also an area subject to flooding, which is anticipated to increase due to climate change.

The health impacts on residents of housing in or adjacent to industrial areas must be considered carefully to ensure equitable outcomes.



Exhibit 3.9-14 Washington Environmental Health Disparities Map







Exhibit 3.9-16 Population Near Heavy Traffic Noise



Exhibit 3.9-17 Proximity to Hazardous Waste Sites

3.9.2 Impacts

As described in the introduction to this section, three impact thresholds were used to identify potential adverse housing impacts in the study area and at a subarea level (where applicable). Impacts of the alternatives on housing are considered significant if they:

- Result in loss of housing due to redevelopment and insufficient development capacity, tools, or programs to address displacement of dwellings and population.
- Potential to increase households' exposure to air pollution, noise pollution, or environmental hazards in census tracts identified as having high environmental health disparities (e.g., exposure to diesel emissions and ozone or proximity to hazardous waste sites) and with sensitive populations (e.g., poverty, cardiovascular disease) based on the Washington Department of Health Environmental Health Disparities Index.
- Creation of demand for housing that cannot be accommodated within the city in adjacent districts or areas where housing is planned.

Equity & Environmental Justice Considerations

Jobs/Housing Balance

Housing production has not kept pace with employment growth in Seattle putting pressure on prices. While roughly 29% of workers in the study area live in Seattle, the majority of workers live in places across the region and travel long distances to get to their jobs. **Exhibit 3.9-11** shows the distance traveled by workers in industrial subareas. Roughly 37% of workers (29,543) travel 10-24 miles one-way to get to their jobs. More than 10,000 workers travel 25-50 miles one-way to get to their jobs. Some of these workers may prefer to live closer to their jobs if adequate and affordable housing were available.

The continued regulatory support for industry-related housing (caretakers' residences and artist lofts) and the slight increases in housing envisioned in alternatives 3 and 4 can add to the housing supply and allow some workers to live close to where they work. Applying the Mandatory Housing Affordability (MHA) regulations to the proposed new Industry & Innovation (II) zone can also mitigate some of the housing impacts on the study area. Additional housing supply near jobs can reduce the costs of commuting. In addition, adding capacity for additional housing in areas adjacent to or connected by transit to these employment centers can also mitigate the impacts of increased employment growth on housing.

Access to Opportunity

A key concern around adding housing to industrial areas is whether this would perpetuate historic patterns of increasing housing capacity in areas with low opportunities. The City's Access to Opportunity Index shows that parcels within the study area have low or moderate access to opportunity. No significant new housing in these areas of low or moderate opportunity is anticipated under any of the Alternatives. While there are slight increases in housing envisioned in alternatives 3 and 4, in the Ballard and SODO/Stadium subareas, these increases are tied to a change to zoning from the existing zones to Urban Industrial (UI) zoning. UI zoning is intended to create thoughtful integration between the edges of these industrial areas and adjacent neighborhoods. UI zoning would seek to improve environmental health, walkability, and comfort in these areas. These changes tied to zoning are likely to ensure that the limited amount of housing allowed within the UI zone is accompanied by changes that add amenities to the area.

Public Health

Residents of industrial areas in the Greater Duwamish MIC are at high risk of environmentrelated health problems. Exposure to air pollution, noise pollution, and highways is higher in the Duwamish Valley than the city average and access to open space is lower. In addition, health advisories are in place for the Duwamish River as the City works with the U.S. Environmental Protection Agency (EPA) on cleanup and source control efforts. The Duwamish Valley is also an area subject to flooding, which is anticipated to increase due to climate change.

The Action Alternatives limit new housing in industrial zones and focus primarily on industrial uses. Alternatives 3 and 4 add mixed-use housing opportunities near Georgetown/South Park, addressed by alternative below. Given the health impacts of housing proximity to industrial areas, especially the Duwamish area, limiting the amount of housing in these areas avoids impacts on health equity.

Impacts of Alternative 1 No Action

Loss of housing due to redevelopment and insufficient development capacity, tools, or programs to address displacement of dwellings and population. Under Alternative 1 No Action, the full study area would support 488 total housing units or an addition of 75 housing units from the existing 413 units. As the area grows, the mix of land uses under Alternative 1 will remain similar to the existing condition. There is likely to be some redevelopment in areas adjacent to Seattle's designated urban villages, in areas where the Industrial Commercial (IC) zone applies, but concentrated development of housing is not anticipated. See Exhibit 3.9-18.

	0 , 0,	<u> </u>
	Existing	2044
Industrial Jobs	54,500 (2018)	66,400
Total Jobs	98,500 (2018)	122,000
Residential Dwellings	413 (2021)	488

Exhibit 3.9-18	Alternative 1—N	lo Action Jobs and	Housing,	Existing and 2044
----------------	-----------------	--------------------	----------	-------------------

Sources: CAI, 2021; City of Seattle, 2021.

As noted earlier most of the modest increase in housing is anticipated to be in typologies that remain similar to the forms that exist today.

Under Alternative 1 No Action, most industrial jobs as well as total jobs are located in the SODO/Stadium and Georgetown/South Park subareas, with relatively less in the Ballard, Interbay Dravus, and Interbay Smith-Cove subareas. Since housing is limited to those connected to industrial activities, increases in housing are also anticipated to be concentrated in the SODO/Stadium and Georgetown/South Park subareas. See **Exhibit 3.9-19**.

Subarea		Existing (2021)	Total	Growth
Ballard	10%	192	199	7
Interbay Dravus	10%	3	11	8
Interbay Smith Cove	10%	1	9	8
SODO/Stadium	40%	21	51	30
Georgetown/South Park	30%	196	218	22
Grand Total Housing in Study Area		413	488	75

Exhibit 3.9-19 Alternative 1—No Action Housing by Subarea

Sources: CAI, 2021; City of Seattle, 2021.

The City's Displacement Risk Index identifies areas of Seattle where displacement of marginalized populations may be more likely. It reflects data on vulnerability, amenities, development capacity, and rent to identify where displacement of those populations is more likely to occur. Overall, parcels within the study area are at low or moderate risk for displacement.

Very little housing growth and related redevelopment is anticipated under Alternative 1. With a mix of land uses and housing typologies similar to existing conditions, there is unlikely to be any significant loss of housing due to redevelopment within the study area under Alternative 1.

Potential to increase households' exposure to air pollution, noise pollution, or environmental hazards in census tracts identified as having high environmental health disparities and with sensitive populations. Under Alternative 1, the number of dwellings is only projected to increase by 75 units, with most of this increase assumed to be in the form of caretakers' units and artist/studio quarters. Under this Alternative, housing is limited to those connected with industrial activities, and modest increases are anticipated in the SODO/Stadium and Georgetown/South Park subareas. While these are areas with high disparities, the increase in housing of 75 units is not considered significant.

Creation of demand for housing that cannot be accommodated within the city in adjacent districts or areas where housing is planned. Alternative 1 anticipates an increase in total jobs in the study area. Increases in employment growth envisioned under this Alternative could shift some of the overall expected citywide employment growth into industrial areas. This could have an impact on housing, especially if additional new employment were added to industrial areas not subject to the MHA regulations. Overall, the increased employment growth envisioned in Alternative 1 is addressed within the City's 2035 Comprehensive Plan and will be within the amount that the City will plan for in the 2024 major Comprehensive Plan update for 2044. Similarly, the City will evaluate the overall citywide demand for housing consistent with its growth targets.

Impacts of Alternative 2

Loss of housing due to redevelopment and insufficient development capacity, tools, or programs to address displacement of dwellings and population. Little new housing is envisioned in this Alternative. Under Alternative 2, housing units are expected to increase slightly by only 80 units to 493 from the existing 413 units. Similar to existing conditions, and Alternative 1 No Action, the housing types that are added are likely to be caretakers' quarters and some artist/studios. See Exhibit 3.9-20.

Exhibit 3.9-20 Alternative 2 Jobs and Housing, Existing and 2044

	Existing	2044
Industrial Jobs	54,500 (2018)	66,400
Total Jobs	79,400 (2018)	132,900
Residential Dwellings	413 (2021)	493

Sources: CAI, 2021; City of Seattle, 2021.

Modest increases in housing under Alternative 2 are anticipated to be concentrated in the SODO/Stadium and Georgetown/South Park subareas. See **Exhibit 3.9-21**.

Exhibit 3.9-21 Alternative 2 Housing by Subarea

Subarea	Total	Growth
Ballard	200	8
Interbay Dravus	11	8
Interbay Smith Cove	9	8
SODO/Stadium	53	32
Georgetown/South Park	220	24
Grand Total Housing in Study Area	493	80

Sources: CAI, 2021; City of Seattle, 2021.

As noted earlier the City's Displacement Risk Index shows the study area with low or moderate risk of displacement. While some changes to housing patterns may be possible under this Alternative, this is an expected part of a changing urban environment. There is unlikely to be any significant loss of housing due to redevelopment within the study area under Alternative 2.

Potential to increase households' exposure to air pollution, noise pollution, or

environmental hazards in census tracts identified as having high environmental health disparities and with sensitive populations. Housing growth is relatively higher in SODO/Stadium and Georgetown/South Park subareas under this Alternative. These are areas with high disparities. However, only an estimated 80 new homes would be added in caretakers' quarters and artist/studios under this Alternative. This modest addition is not considered significant.

Creation of demand for housing that cannot be accommodated within the city in adjacent districts or areas where housing is planned. Under Alternative 2, employment is projected to grow substantially more than under Alternative 1 No Action. A total of 34,400 additional jobs are projected for the study area, an increase of 35%.

Increases in employment growth envisioned under this Alternative could shift some of the overall expected citywide employment growth into industrial areas. This could have an impact on housing, especially if additional new employment were added to industrial areas not subject to the MHA regulations. Demand for new housing could be shifted to areas of the city closer to locations of dense employment growth (II zones), but outside of the study area. The II zones are in the closest locations to light rail (1/4–1/2 mile) and locations with fast access by light rail to these areas may see some shifts in demand.

Overall, the increased employment growth envisioned in Alternative 2 is within the citywide amount that the City will plan for in the 2024 major Comprehensive Plan update; similarly, the City will plan for its housing growth target and address the citywide demand for housing.

Impacts of Alternative 3

Loss of housing due to redevelopment and insufficient development capacity, tools, or programs to address displacement of dwellings and population. Under Alternative 3, housing units are projected to increase by 610 units in addition to 413 existing units. Housing types are expected to include caretakers' quarters and makers' studios as well as newer industry-supportive formats allowed under the UI zone such as live/work units, and housing connected to makers' studios. See **Exhibit 3.9-22**.

	Existing	2044
Industrial Jobs	54,500 (2018)	83,500
Total Jobs	98,500 (2018)	155,900
Residential Dwellings	413 (2021)	1,023 <u>*</u>

* Without MIC adjustments—Seattle Mixed-Use Zone Housing.

Sources: CAI, 2021; City of Seattle, 2021.

The following section describes the anticipated changes to housing by subarea under this Alternative. See **Exhibit 3.9-23**.

- Ballard. While Alternative 3 adds housing in the Ballard Subarea, it does so in limited locations along the edge or transition areas between industrial areas and the neighborhood. Land in the Ballard uplands in the 14th Avenue NW corridor north of NW Leary would be placed in the UI zone, and the zone would allow industry supportive housing at a maximum density of 25 dwelling units / acre. Housing allowed under the new UI zone would include development standards that limit the types of housing to those that are industry-supportive. An additional 260 units are anticipated.
- Interbay Dravus. Land north of Dravus Street along Thorndyke Avenue W would be in the UI zone as in Alternative 2. However, in Alternative 3 the zone would allow for supportive housing at a maximum density of 25 dwelling units / acre. An additional 75 housing units are estimated, and they would typically be located on an upper floor of a 3-4 story mixeduse development.
- Interbay Smith Cove. UI zoned areas in the four blocks along 15th Avenue NW would be the location for an estimated 15 housing units.
- **SODO/Stadium.** Under Alternative 3 land in the stadium area in the UI zone could receive an estimated 200 industry-supportive housing units.
- Georgetown/South Park. Under Alternative 3 edges of South Park and Georgetown residential areas would be zoned UI, which is anticipated to enable an estimated 60 industry supportive residential units interspersed in these areas. Under Alternative 3, the triangular area of Georgetown bounded by Corson Avenue S, Carleton Avenue S and I-5 would be removed from the MIC and placed into a mixed-use zone. The area would likely develop with a high concentration of urban mixed-use structures with ground level retail and residential above. An estimated 1,078 housing units could be added. Land removed from the MIC at the edges of South Park would be placed in a mixed-use zone. Some of it would likely redevelop with mixed-use structures including housing on upper floors. This would add capacity for a range of housing in these areas. These areas currently include a mix of industrial service and repair businesses, and small-scale commercial uses.

Subarea	Total	Growth
Ballard	452	260
Interbay Dravus	78	75
Interbay Smith Cove	16	15
SODO/Stadium	221	200
Georgetown/South Park	256	60
Total: Ind Zone Housing (Caretaker/Artist)	1,023	610
		Added MU Housing
With MIC Adjustments—Seattle Mixed-Use Zone Housing		1,078
Grand Total Housing in Study Area	2,101	1,688
Courses CAL 2021, City of Coottle, 2021		

Exhibit 3.9-23 Alternative 3 Housing by Subarea

Sources: CAI, 2021; City of Seattle, 2021.

Potential to increase households' exposure to air pollution, noise pollution, or

environmental hazards in census tracts identified as having high environmental health disparities and with sensitive populations. Alternative 3 adds housing in the SODO/Stadium and Georgetown/ South Park area and has the potential to add more residents in a census tract shown to have greater exposure to air pollution, noise sources and health disparities. Application of mitigation measures under air quality and noise (**Sections 3.2 and 3.6**) could help reduce potential impacts, e.g., building design, distance, landscaping, and others.

Creation of demand for housing that cannot be accommodated within the city in adjacent districts or areas where housing is planned. Overall employment under Alternative 3 would increase by 57,000 jobs.

Increases in employment growth envisioned under this Alternative could shift some of the overall expected citywide employment growth into industrial areas. This could have an impact on housing, especially if additional new employment were added to industrial areas not subject to the MHA regulations. Demand for new housing could be shifted to areas of the city closer to locations of dense employment growth (II zones), but outside of the study area. The II zones are in the closest locations to light rail (1/4–1/2 mile) and locations with fast access by light rail to these areas may see some shifts in demand.

Overall, the increased employment growth envisioned in Alternative 3 is within the citywide amount that the City will plan for in the 2024 major Comprehensive Plan update; likewise, the City will plan for its housing growth target in 2024 and address the citywide demand for housing.

Impacts of Alternative 4

Loss of housing due to redevelopment and insufficient development capacity, tools, or programs to address displacement of dwellings and population. Alternative 4 expands limited housing allowances to the greatest degree of any of the <u>Draft EIS</u> alternatives. Under Alternative 4 about 2,195 new homes would be added in UI zoned portions of industrial areas due to increased flexibility for caretakers' quarters and makers' studios. Housing types in this Alternative are likely to be a combination of existing and newly allowed formats such as caretakers' quarters, makers' studios, live/work units, and housing in conjunction with small production spaces. See **Exhibit 3.9-24**.

Exhibit 3.9-24 Alternative 4 Jobs and Housing Existing and 2044

	Existing	2044
Industrial Jobs	54,500 (2018)	66,400
Total Jobs	98,500 (2018)	157,700
Residential Dwellings	413 (2021)	2,608*

* With<u>out</u> MIC adjustments—Seattle Mixed-Use Zone Housing<u>.</u> Sources: CAI, 2021; City of Seattle, 2021. The following section describes the anticipated changes to housing by subarea under this Alternative. See **Exhibit 3.9-25**.

- Ballard. Under Alternative 4 land in the Ballard uplands in the 14th Avenue corridor north of NW Leary would be placed in a combination of the II zone and the UI zone. The UI zone would allow a greater density of industry supportive housing at a maximum density of 50 dwelling units / acre. Other areas that are north of NW Leary and in Fremont north of 36th Street would be placed in the UI zone and would likely receive a substantial amount of increased infill development. An additional 790 housing units are estimated and would typically be located on several upper floors of a 4-6 story mixed-use development.
- Interbay Dravus. Within the Interbay Dravus subarea, land north of Dravus Street along Thorndyke Avenue W would be zoned UI as in alternatives 2 and 3, but in Alternative 4 the zone would allow for industry supportive housing at a maximum density of 50 dwelling units per acre. An additional 175 housing units are estimated, and they would typically be located on an upper floor of a 4-6 story mixed-use development.
- Interbay Smith Cove. No additional housing is expected in the Interbay Smith Cove Subarea under Alternative 4 because of the small application of the UI zone on parcels unlikely to redevelop.
- **SODO/Stadium.** Under Alternative 4, land in the stadium area would be zoned UI, and the UI zone would be extended further south along 1st Avenue to Starbucks Center. This would allow the area to receive an estimated 990 industry-supportive housing units.
- Georgetown/ South Park. Under Alternative 4 (as in Alternative 2) edges the residential areas would be zoned UI, and increased infill development with light industrial uses, brewers/makers, and small manufacturers with large ancillary spaces is expected. However, the zone would enable an estimated 240 industry supportive residential units interspersed in these areas.

Similar to Alternative 2, under Alternative 4, the triangular area of Georgetown bounded by Corson Avenue S, Carleton Avenue S and I-5 would be removed from the MIC and placed into a mixed-use zone. An estimated 1,078 housing units could be added.

Land removed from the MIC at the edges of South Park would be placed in a mixed-use zone. Some of it would likely redevelop with mixed-use structures including housing on upper floors. This would add capacity for a range of housing in these areas. These areas currently include a mix of industrial service and repair businesses, and small-scale commercial uses.

Alternative 4 adds more housing than alternative<u>s</u> 1, 2, or 3. Housing added to the Ballard subarea would be part of mixed-use infill development. New zone standards would allow small parcels to accommodate new structures as well. Areas that are changing to the Urban Industrial Zone in SODO under Alternative 3 currently has no significant amounts of housing.

Redevelopment in the areas zoned for UI may be more likely to add housing under the industry-supportive housing formats allowed under UI zone rather than displace existing housing. As noted earlier the City's Displacement Risk Index shows the study area overall with low or moderate risk of displacement. While some loss of existing housing may be

possible under this Alternative this is an expected part of a changing urban environment. There is unlikely to be any significant loss of housing due to redevelopment within the study area under Alternative 4.

Subarea	Total	Growth
Ballard	982	790
Interbay Dravus	178	175
Interbay Smith Cove	1	0
SODO/Stadium	1,011	990
Georgetown/South Park	436	240
Total: Ind Zone Housing (Caretaker/Artist)	2,608	2,195
		Added MU Housing
With MIC Adjustments—Seattle Mixed-Use Zone Housing		1,078
Grand Total Housing in Study Area	3,686	3,273

Exhibit 3.9-25 Alternative 4 Housing by Subarea

Sources: CAI, 2021; City of Seattle, 2021.

Potential to increase households' exposure to air pollution, noise pollution, or

environmental hazards in census tracts identified as having high environmental health disparities and with sensitive populations. Similar to Alternative 3, adding housing in the Seattle Mixed zone under Alternative 4, particularly in the South Park area, and housing growth in the SODO/Stadium and Georgetown areas, could add more residents in a census tract shown to have greater exposure to air pollution, noise sources and health disparities. Similar to Alternative 3, the air quality and noise mitigation measures (**Sections 3.2** and **3.6**) could help reduce potential impacts of housing located in or near the study area, e.g., building design, distance, landscaping, and others.

Creation of demand for housing that cannot be accommodated within the city in adjacent districts or areas where housing is planned. Under Alternative 4, employment is projected to grow substantially more than under Alternative 1 No Action and Alternative 2, and by a similar amount to Alternative 3. A total of 59,2000 additional jobs are projected for the study area, an increase of 59%.

Increases in employment growth envisioned under this Alternative could shift some of the overall expected citywide employment growth into industrial areas. This could have an impact on housing, especially if additional new employment were added to industrial areas not subject to the MHA regulations. Demand for new housing could be shifted to areas of the city closer to locations of dense employment growth (II zones), but outside of the study area. The II zones are in the closest locations to light rail (1/4–1/2 mile) and locations with fast access by light rail to these areas may see some shifts in demand.

Overall, the increased employment growth envisioned in Alternative 4 is within the citywide amount that the City will plan for in the 2024 major Comprehensive Plan update; similarly, the City will plan for its housing growth target and address the citywide demand for housing.

Impacts of the Preferred Alternative

Loss of housing due to redevelopment and insufficient development capacity, tools, or programs to address displacement of dwellings and population. Under the Preferred Alternative, the number of housing units in industrial areas is projected to increase by 1,475 units—more than alternatives 1, 2, and 3 but less than the amount studied in Alternative 4 (720 less). The Preferred Alternative expands limited industry-supportive housing in the UI zone, subject to a conditional use process and more location and performance criteria than Draft EIS alternatives 3 or 4, and maintains a limit on density as in alternatives 3 or 4. The industrysupportive housing criteria could be met in one of two ways—either by limiting occupancy to caretakers or makers (as in alternatives 3 and 4), or by providing a minimum of 50% of any housing units that are created to households with incomes at 90% of AMI or below. See **Exhibit 3.9-26.**

	Existing	<u>2044</u>
Industrial Jobs	<u>54,500 (2018)</u>	70,853
Total Jobs	<u>98,500 (2018)</u>	<u>134,045</u>
Residential Dwellings	<u>413 (2021)</u>	<u>1,888*</u>

Exhibit 3.9-26 Preferred Alternative Jobs and Housing, Existing and 2044

* Without MIC adjustments—Seattle Mixed-Use Zone Housing. Sources: CAI, 2021; City of Seattle, 2022.

Sources: CAI, 2021; City of Seattle, 2022.

The following section describes the anticipated changes to housing by subarea under this Alternative. See **Exhibit 3.9-27.**

- Ballard. An additional 514 housing units are added to the subarea under the Preferred Alternative. This is within the range of the alternatives in the Draft EIS which looked at a growth of 8 units under Alternative 2 and 790 units in Alternative 4. Under the Preferred Alternative, land in the Ballard uplands in the 14th Avenue NW corridor north of NW Leary would be placed in the UI zone, and the zone would allow housing at a maximum density of 50 dwelling units / acre subject to tenancy limits or a 50% workforce housing affordability limit.
- Interbay Dravus. An additional 114 housing units are added to the subarea under the Preferred Alternative. This is within the range of the alternatives in the Draft EIS which looked at a growth of 8 units under Alternative 2 and 175 units in Alternative 4. Land north of Dravus Street along Thorndyke Avenue W would be in the UI zone. The Preferred Alternative would allow housing at a maximum density of 50 dwelling units / acre subject to tenancy limits or a 50% workforce housing affordability limit.

- Interbay Smith Cove. No additional housing is expected in the Interbay Smith Cove
 Subarea under the Preferred Alternative because of the small application of the UI zone on parcels unlikely to redevelop (the same as Alternative 4).
- SODO/Stadium. An additional 644 housing units are added to the subarea under the Preferred Alternative. This is within the range of the alternatives in the Draft EIS which looked at a growth of 32 units under Alternative 2 and 990 units in Alternative 4.
- Georgetown/South Park (industrial zones). An additional 204 housing units are added to the subarea under the Preferred Alternative. This is within the range of the alternatives in the Draft EIS which looked at a growth of 24 units under Alternative 2 and 240 units in Alternative 4.
- New Mixed Use Areas. Under the Preferred Alternative, areas of mixed use in South Park and Georgetown are like alternatives 3 and 4. An estimated 392 housing units could be added in Georgetown and 294 housing units in South Park under this Alternative. More nuanced specific development standards are proposed under the Preferred Alternative for the triangular area of Georgetown bounded by Airport Way, Corson Avenue S, and Carleton Avenue S. The standards integrate Georgetown priorities for historic preservation, antidisplacement, and arts spaces. Two new areas outside the MICs in west Ballard and Judkins Park would be converted to mixed use zoning allowing housing, in addition to the proposed mixed-use areas in Georgetown and South Park. Overall, a higher total amount of housing production outside of MICs would result compared to Draft EIS alternatives—an additional, 1,534 dwellings, 42% more than alternatives 3 and 4.

<u>Subarea</u>	<u>Total</u>	<u>Growth</u>
Ballard	706	<u>514</u>
Interbay Dravus	<u>117</u>	<u>114</u>
Interbay Smith Cove	<u>1</u>	<u>0</u>
SODO/Stadium	<u>665</u>	<u>644</u>
Georgetown/South Park (industrial zones)	<u>400</u>	<u>204</u>
Total: Ind Zone Housing	<u>1,888</u>	<u>1,475</u>
		Added MU Housing
With MIC Adjustments—Seattle Mixed-Use Zone Housing		<u>1,534</u>
Grand Total Housing in Study Area	<u>3,422</u>	<u>3,009</u>

Exhibit 3.9-27 Preferred Alternative Housing by Subarea

Sources: CAI, 2021; City of Seattle, 2022.

In the Preferred Alternative, the number of dwellings in industrial areas is projected to increase by 1,475 units—more than alternatives 1, 2, and 3 but less than the amount studied in Alternative 4 (720 less). Areas of mixed use in South Park and Georgetown are like alternatives

<u>3 and 4. Some land outside of the MICs in Georgetown, South Park, west Ballard, and Judkins</u> <u>Park would be zoned mixed use residential instead of mixed use industrial (II). Overall, a higher</u> <u>total amount of housing production outside of MICs would result compared to Draft EIS</u> <u>alternatives—an additional 1,534 dwellings, 42% more than alternatives 3 and 4.</u>

Redevelopment in the areas zoned for UI may be more likely to add housing under the housing formats allowed under UI zone rather than displace existing housing. The City's Displacement Risk Index shows the study area overall with low or moderate risk of displacement. While some loss of existing housing may be possible under this Alternative, this is an expected part of a changing urban environment. There is unlikely to be any significant loss of housing due to redevelopment within the study area under the Preferred Alternative.

Potential to increase households' exposure to air pollution, noise pollution, or

environmental hazards in census tracts identified as having high environmental health disparities and with sensitive populations. The Preferred Alternative adds housing in the SODO/Stadium and Georgetown/ South Park area, though less than Alternative 4, and has the potential to add more residents in a census tract shown to have greater exposure to air pollution, noise sources and health disparities. Application of mitigation measures under air quality and noise (**Sections 3.2 and 3.6**) could help reduce potential impacts (e.g., building design, distance, landscaping, and others).

<u>Creation of demand for housing that cannot be accommodated within the city in</u> <u>adjacent districts or areas where housing is planned</u>. Overall employment under the <u>Preferred Alternative would increase</u>. Considered in combination, the total jobs and population by alternative shows the highest total job and population growth under Alternative 4 and the <u>lowest under Alternative 1</u>. The Preferred Alternative has a total that is slightly more than <u>Alternative 2 and less than Alternative 3</u>.

Increases in employment growth envisioned under this Alternative could shift some of the overall expected citywide employment growth into industrial areas. This could have an impact on housing, especially if additional new employment were added to industrial areas not subject to the MHA regulations. Application of mitigation measures could help reduce the impact of employment shift in the industrial areas on housing.

Overall, the increased employment growth envisioned under the Preferred Alternative is within the citywide amount that the City will plan for in the 2024 major Comprehensive Plan update; likewise, the City will plan for its housing growth target in 2024 and address the citywide demand for housing.

3.9.3 Mitigation Measures

Incorporated Plan Features

The Seattle Comprehensive Plan designates the MICs as major industrial employment centers. While alternatives 2, 3, and 4 include some expansions in allowed housing, the scale of housing growth is significantly smaller than employment growth. The addition of small amounts of housing in limited locations is intended to foster vibrant industrial districts that support a mix of uses that include local manufacturing, production, arts. This mix has the potential to address the shortage of small or affordable space for makers and creatives.

Increases in housing units under alternatives 2, 3, and 4 will be subject to the development standards developed under the UI zone. These include pedestrian and cyclist-oriented frontage improvements, development of green public spaces, access to planned transit and non-motorized transportation connections that support new development. The integration of public green open spaces, pedestrian-oriented amenities, and the access to transit, helps to soften potential impacts of locating housing in areas of intensive industrial activity and employment growth. Access to open space is an amenity that can be used for recreation, community gathering, access to nature, and a variety of environmental benefits. Housing in proximity to transit can help potential employees in the industrial centers live closer to their jobs. See Other Potential Mitigation Measures regarding reducing health disparities.

Regulations & Commitments

Seattle's City Code contains regulations that help to address potential displacement. A summary of these regulations, which would mitigate impacts associated with the alternatives, is presented below.

SEPA Review

Section 25.05 of Seattle Municipal Code contains environmental procedures that govern the issues to be addressed during development review under the State Environmental Policy Act (SEPA). SEPA addresses issues related to height, bulk, scale, and land use compatibility. Future site-specific development would be subject to additional SEPA review.

Development Regulations

Title 23 contains Seattle's Land Use Code, which establishes zoning and development regulations. These development regulations contain provisions governing the design of buildings, site planning, and provisions for adaptive reuse of existing buildings. Industrial zones generally contain provisions relating to limits of housing designed in industry supportive formats. Regulations are in place to address housing development related to the implementation of Alternative 1.

Existing Programs to Address Potential Displacement

- Seattle's Tenant Relocation Assistance Ordinance. This provides relocation assistance to very low-income households and provide notice to all households prior to relocation. Renters are considered displaced when their housing is scheduled to be torn down or undergo substantial renovation, have its use changed (for example, from an apartment building to a hotel), or have certain rent or income restrictions removed (for example a property is no longer required to rent only to low-income renters under a federal program).
- Notice of Intent to Sell Ordinance. The Notice of Intent to Sell ordinance reauthorized by Council in 2019, provides the City with information about the intention to sell residential rental property with at least one unit rented at 80% of Area Median Income (AMI) or below. The City, in partnership with the Seattle Housing Authority and community partners, can use the notification information to evaluate properties and deploy a range of property preservation tools, including incentives and acquisition. The notice can also help residents seek tenant protections and relocation resources if necessary.
- Rental Registration and Inspection Ordinance. The Rental Registration and Inspection Ordinance (RRIO) helps ensure that all rental housing in Seattle is safe and meets basic housing maintenance requirements. All rental property owners in Seattle must register their properties with the City. Inspectors will make sure all registered properties comply with minimum housing and safety standards at least once every 5–10 years. RRIO helps improve and maintain the quality of Seattle's rental housing over time.

This patchwork of programs and regulations works to address displacement in the areas in which they are applied. These rules would be in place under all alternatives.

Other Potential Mitigation Measures

Impacts of anticipated residential growth under the alternatives are not significant based on the thresholds identified in the EIS.

Comprehensive Plan Update

The City will plan for the citywide amount of housing growth in the Comprehensive Plan EIS on a citywide scale. As part of this ongoing commitment, the City could consider

- Adding additional capacity for housing in urban villages and residential areas in locations that will have fast access to the new II zones to help address the shifts in demand for housing in response to employment growth in industrial areas. The II zones are in the closest locations to light rail (1/4–1/2 mile), and light rail will provide good access to these areas.
- Adding additional capacity for housing in urban village and residential areas in locations adjacent to new UI zones to address the shifts in demand for housing in response to employment growth in the industrial areas.

Mandatory Housing Affordability

Given the potential for employment growth to shift demand for housing, the City could consider the following mitigation measures:

- Apply MHA regulations to the to the proposed new Industry and Innovation zone. Increases in employment growth envisioned under the alternatives could shift some of the overall expected citywide employment growth into industrial areas. This could have an impact on housing, especially if additional new employment were added to industrial areas not subject to the MHA regulations. Applying MHA to the proposed new Industry and Innovation zone can mitigate this shift in demand.
- The City can also mitigate negative impacts of industrial development on nearby residents as follows (see Section 3.2 Air Quality & GHG and Section 3.6 Noise for details):
 - Include policy guidance that recommends that residences and other sensitive land uses be separated 500 feet or appropriate distance from freeways, railways, and port facilities.
 - Add a denser tree canopy near high-volume roadways and industrial areas.
 - Impose greater noise reduction standards in residential buildings where exterior noise levels greater than 65 dBA are likely to occur.
 - Install noise reducing pavement on major arterials and roadways that experience relatively high traffic volumes and speeds.

3.9.4 Significant Unavoidable Adverse Impacts

Under all alternatives additional growth and development will occur in the study area, with small changes in the mix of housing. This change is unavoidable but is not considered significant or adverse within an urban area designated as an employment center in the Comprehensive Plan. No significant loss of existing housing due to redevelopment is anticipated under any of the alternatives. The potential impacts related to these changes may differ in intensity and location in each of the alternatives. However, with existing and new development regulations, antidisplacement programs currently in place, no significant adverse impacts are anticipated.

Increases in housing, particularly under alternatives 3 and 4 <u>and the Preferred Alternative</u>, could increase households' exposure to air pollution, noise pollution, or environmental hazards in census tracts identified as having high environmental health disparities and with sensitive populations. With the application of air quality and noise mitigation measures, no significant unavoidable adverse noise impacts would occur under any of the alternatives.

Increases in employment growth in the study area may shift some demand for housing. The increment of employment growth in all alternatives is within the citywide amount that the City will plan for in the 2024 Major Comprehensive Plan update. With the application of mitigation measures, including the application of MHA regulations to the II zone, and citywide planning for housing capacity through the Comprehensive Plan, no significant unavoidable impacts would occur under any of the alternatives.

Section 3.10 Transportation



Sections of text related to freight and transit were reordered from the Draft EIS in response to comments. This chapter presents a multimodal transportation evaluation of the potential impacts of implementing the range of land use alternatives under consideration. The chapter presents existing transportation conditions within the study area and future transportation conditions under <u>four five</u> alternatives: Alternative 1 No Action representing a continuation of the City's adopted land use plan in the study area<u>and</u> three <u>Draft EIS</u> Action Alternatives reflecting varying increases in the amount of growth accommodated by 2044 as a result of the proposal<u>and the Preferred Alternative as part of this Final EIS</u>. Significant transportation impacts and potential mitigation strategies are identified for the Action Alternatives based on the policies and recommendations established in local plans.

Thresholds of significance utilized in this impact analysis include:

- Lengthy travel times on key corridors <u>designated as major truck streets</u>.
- Peak hour volumes on key corridors that cannot be accommodated by roadway capacity.
- Mode shares in conflict with City goals.
- Transit demand on key corridors that cannot be accommodated by planned service.
- Increases in pedestrian and bicycle demand in locations with network gaps or preclusion of planned pedestrian and bicycle improvements.
- Substantive increases in parking demand in excess of parking supply.
- Increases in serious and fatal crash rates in the study area.

More specific thresholds are described in **Section 3.10.2**.

3.10.1 Affected Environment

This section presents existing transportation conditions within the study area for all modes as well as the methodologies used to quantitatively evaluate the current performance of the transportation network. This includes evaluations of autos, freight, transit, people walking and biking, parking, and safety.

Primary & Secondary Study Areas

The study area includes the areas designated as Manufacturing/Industrial Centers (MICs) by the Puget Sound Regional Council (PSRC) as well as some nearby areas with similar uses. The study area is mapped in **Exhibit 3.10-1**. The Ballard Interbay Northend MIC (BINMIC) includes the secondary subareas of Ballard, Interbay Dravus, and Interbay Smith Cove. The Greater Duwamish MIC includes the secondary subareas of SODO/Stadium and Georgetown/South Park.

The locations of the MICs are unique in their transportation context and serve uses that could not be replaced elsewhere. In particular, the Duwamish MIC, adjacent to Elliott Bay's naturally deep harbor, is anchored by a marine container and breakbulk facilities and three major rail yards for the transfer of freight between rail, truck, and ship. These are essential facilities of the MIC transportation system and are critical to maintaining industrial activity and the local and regional economy and supply chain.





Data & Methods

A variety of data were collected and compiled to assess transportation conditions in the study area. This section describes the data and methods used to evaluate key transportation metrics.

Travel Time

Travel time along major arterials <u>(that</u> <u>are also part of the City's designated</u> <u>major truck street network)</u> was selected as a performance measure because it is easily relatable and addresses the fundamental concern of most travelers—the time it takes to move within and through the study area. This metric is relevant for autos, freight, and transit that travel along these corridors. To assess existing conditions, PM peak

EIS Analysis Years

This EIS considers two distinct time periods for analysis: 2019 as the baseline of existing conditions and 2044 as a horizon year at which the outcomes of the alternatives are compared. A variety of events have occurred over the past two years that have disrupted transportation patterns in the study area. These include global events like the COVID-19 pandemic which has changed longstanding commute patterns and created supply chain bottlenecks at West Coast ports including the Port of Seattle. Locally, the closure of the West Seattle Bridge has fundamentally changed travel patterns through the study area. For this reason, 2019 was selected as a more representative year for baseline travel conditions. While these factors are profoundly affecting the transportation system as of the publication of this EIS, it is assumed that they will be resolved in the next several years and therefore not meaningfully affect operations by the horizon year of 2044.

hour travel times were analyzed using October 2019 data; this time period represents conditions before the COVID-19 pandemic as well as before the West Seattle Bridge was closed for emergency repairs. Based on the data collected, 4:45-5:45 PM was found to be the peak hour of the PM period. Data for the month of October 2019 was obtained from Wejo, which supplies raw data collected from connected vehicle data. For all observed trips during the PM peak hour, the total travel time and distance traveled along each study corridor was summed, and then a 25th percentile speed was calculated for the entire corridor.

To provide context for the results, the concept of level of service (LOS) is used to describe traffic operations by assigning a letter grade of A through F, where A represents free-flow conditions. <u>B</u> represents free-flow conditions with some restrictions in lane changes, C is near free-flow conditions with a heavier flow, D is an unstable flow with minor queuing, E represents unstable flow with potentially extended queuing, and F represents highly congested conditions. This study uses concepts from the 6th Edition of the Highway Capacity Manual (HCM) to define thresholds for each LOS grade, as shown in **Exhibit 3.10-2**. The ranges shown in the table below represent the ratio between observed travel time and free-flow travel time (i.e., at the speed limit). For example, if you are traveling at half the free-flow speed, your travel time will be twice that of the free-flow travel time, which equates to the breakpoint between LOS C and LOS D. The travel time study corridors are shown in **Exhibit 3.10-3** and **Exhibit 3.10-4**.

Exhibit 3.10-2 LOS Thresholds for Travel Speeds and Travel Time

	LOS A	LOS B	LOS C	LOS D	LOS E	LOS F
Threshold for Ratio of PM Peak Hour Travel Time to Travel Time at Free-Flow Speed	<1.25	<1.5	<2.0	<2.5	<3.0	≥3.0
Source: Highway Capacity Manual, 6th Edition, 2016.						


Exhibit 3.10-3 Study Corridors—Ballard Interbay Northend MIC, 2021



Exhibit 3.10-4 Study Corridors—Greater Duwamish MIC, 2021

<u>Note: Map was updated to add two additional study corridors</u> Sources: Fehr & Peers, 202<u>2</u>4.

Mode Share

The *Seattle 2035* Comprehensive Plan uses the concept of mode share to evaluate Seattle's transportation network. Mode share is analyzed at a sector level rather than citywide; the analysis geographies are shown in **Exhibit 3.10-5**. For this EIS, mode share and single occupant vehicle (SOV) trips are evaluated for trips originating from or destined to the Northwest Seattle, Magnolia/Queen Anne, and Duwamish sectors during the PM peak period. All trip types are included in the analysis, and the existing mode share estimates are from the PSRC's most recently available Soundcast activity-based model which has a base year of 2014. Data from the PSRC 2017-2019 Household Survey sample was also reviewed but were found to have too small of a sample size at the sector level to estimate mode share. Mode share is used as one of the impact identification criteria as described in **Section 3.10.2**.

Screenlines

Prior to shifting to the mode share method, the City used a "screenline" methodology to evaluate transportation LOS for locally-owned arterials. Screenlines were used to evaluate autos, freight, and transit since buses usually travel in the same traffic stream as autos. A screenline is an imaginary line across which the number of passing vehicles <u>(including passenger cars, trucks, or buses)</u> is counted, often including multiple corridors. As stated in *Seattle 2035*, this methodology recognizes that no single intersection or arterial operates in isolation and motorists choose among multiple routes to minimize travel times, among other factors. This analytic methodology focuses on a "traffic-shed" where the screenlines measure groups of arterials among which drivers logically can choose to travel.

The City set an LOS threshold in the form of a volume-to-capacity (v/c) ratio: the number of vehicles crossing the screenline compared to the designated capacity of the roadways crossing the screenline. This method is also used to evaluate the magnitude of vehicles using the City's roadway network; this EIS focuses on the 11 screenlines most relevant to the study area. **Exhibit 3.10-5** and **Exhibit 3.10-6** summarize the location of each screenline, as well as its LOS threshold. Screenlines are used as one of the impact identification criteria as described in **Section 3.10.2**.



Exhibit 3.10-5 Mode Share Sectors and Screenlines

Sources: City of Seattle, 2020; Fehr & Peers, 2021.

Screenline	Location	Volume-to-Capacity Threshold
2	Magnolia	1.0
3.11	Duwamish River—West Seattle Bridge and Spokane Street	1.2
3.12	Duwamish River—1st Avenue S and 16th Avenue S	1.2
4.13	South City Limit—SR 99 to Airport Way S	1.0
5.11	Ship Canal—Ballard Bridge	1.2
5.12	Ship Canal—Fremont Bridge	1.2
5.13	Ship Canal—Aurora Bridge	1.2
7.11	West of Aurora Avenue—Fremont Place N to N 65th Street	1.0
8	South of Lake Union	1.2
9.12	South of Spokane Street—E Marginal Way to Airport Way S	1.0
10.11	South of S Jackson Street—Alaskan Way S to 4th Avenue S	1.0

Exhibit 3.10-6 LOS Thresholds for Screenlines

Source: Seattle 2035 Comprehensive Plan Transportation Appendix, 2020.

Transit Load Factor

In addition to considering the roadway conditions on which buses operates, this EIS also includes a metric to evaluate whether there is sufficient transit capacity to accommodate demand. Specifically, King County Metro guidelines are used to measure bus passenger loads on transit routes through the study areas. The King County Metro Strategic Plan Service Guidelines define overcrowded routes as trips with average maximum loads greater than the thresholds for the entire service change period, and routes with standing loads (the amount of time passengers on the bus exceed the number of seats) greater than 20 minutes.

For this EIS, overcrowding is identified when the average maximum load of a bus trip exceeds the passenger load threshold. It is calculated by dividing the average maximum number of passengers on a particular route by the number of seats on the bus plus the number of standing people that can fit on the bus, assuming a standing person uses 4 square feet of floor space. In other words, the calculation represents the average maximum load factor over the PM peak period at the highest ridership location along the route. For this study, transit load factor is calculated for all transit routes that cross five screenlines:

- A: East of 8th Avenue NW (NW Market Street to Leary Way NW)
- B: Ballard Bridge
- C: Elliott Avenue W north of W Mercer Place
- D: North of S Lander Street (SR 99 to Airport Way S)
- E: West Seattle Bridge

This report also summarizes light rail passenger load information from the Sound Transit 2020 Service Implementation Plan (reflecting ridership from the 2018-2019 pre-pandemic time period).

Current Policy & Regulatory Frameworks

Relevant policies related to transportation in Seattle are summarized below. The City of Seattle has a 10year strategic plan outlined in Move Seattle (2015). Seattle also has master plans for transit, freight, pedestrians, and bicycles. More detailed information is available in the specified documents.

Seattle Transportation Plan

The City has adopted citywide modal plans for pedestrian, bicycle, transit, and freight travel. SDOT will soon be embarking on a process to create a unified, multimodal **Seattle Transportation Plan** that will integrate the City's modal network visions into a single, holistic transportation plan.

Move Seattle

Move Seattle is a strategic document published in 2015 that guides SDOT's work over the following ten years with an updated workplan published in 2018. The plan identifies the following three key elements:

- Organizing daily work around core values: a safe, interconnected, vibrant, affordable, and innovative city.
- Integrating modal plans to deliver transformational projects: this includes creating a nearterm strategy to integrate recommendations from the freight, transit, walking, and bicycling 20-year modal plans.
- Prioritizing projects and work to identify funding: in 2015, voters approved a nine-year \$930 million Levy to Move Seattle. This funding source replaces the prior Bridging the Gap levy which expired in 2015. SDOT is using the levy funds to implement projects including safety improvements, new facilities, as well as maintenance of existing infrastructure.

SDOT provides annual reports summarizing accomplishments and delivery plans for the coming year.

Freight Master Plan

The Freight Master Plan (FMP) was adopted by the City in 2016. Its purpose is to ensure efficient and predictable goods movement in the region to promote economic activity and international trade. It analyzes the current freight facilities and their ability to accommodate future freight growth and overlays the truck street system with other modal systems with the goal of facilitating better understanding of the potential for modal conflicts. The plan identifies six main goals with a total of 92 actions that address economy, safety, mobility, state of good repair, equity, and the environment in order to create a comprehensive freight network. These include a list of freight supportive projects within the two MICs on corridors connecting the MICs to the freeway system and corridors connecting the MICs. This document is especially important for the two designated manufacturing and industrial centers, the BINMIC and Greater Duwamish MIC, and the Port of Seattle.

Transit Master Plan

The *Transit Master Plan (TMP)* is a 20-year plan that outlines the needs to meet Seattle's transit demand through 2030. It prioritizes capital investment to create frequent transit services that meet the needs of residents and workers. It outlines the high priority transit corridors and the preferred modes along each corridor. This document specifies capital projects to improve speed and reliability. Goals include:

- Meet sustainability, growth management and economic development goals.
- Make it easier and more desirable to take transit.
- Respond to needs of transit-reliant populations.
- Create great places where modes connect.
- Advance implementation within constraints.

The elements of the document include policies and programs, transit corridors and service, access and connections to transit, and funding and performance monitoring.

Pedestrian Master Plan

The *Pedestrian Master Plan* (PMP) envisions Seattle as the most walkable and accessible city in the nation. To achieve that vision, the following goals are identified:

- Reduce the number and severity of crashes involving pedestrians;
- Develop a connected pedestrian environment that sustains healthy communities and supports a vibrant economy;
- Make Seattle a more walkable city for all through public engagement, service delivery, accessibility, and capital investments that promote equity; and
- Get more people moving to improve health and increase mobility.

The plan documents existing pedestrian facilities and creates a Priority Investment Network to guide future improvements.

Bicycle Master Plan

The Seattle *Bicycle Master Plan* (BMP) provides guidance on future investments in bicycle facilities in Seattle, with a vision for bicycling as a safe and convenient mode for people of all ages and abilities on a daily basis. Goals include increasing bicycle ridership, safety, connectivity, equity, and livability. The document outlines the existing network and over 400 miles of planned future network for the city. Strategies for end-of-trip facilities, programs, maintenance, project prioritization and funding are included. SDOT publishes reports every two years to update the public on its progress toward implementing BMP projects and meeting the identified performance measures.

Freight Master Plan

The *Freight Master Plan* (FMP) was adopted by the City in 2016. Its purpose is to ensure efficient and predictable goods movement in the region to promote economic activity and international trade. It analyzes the current freight facilities and their ability to accommodate future freight growth. The plan identifies six main goals with a total of 92 actions that address economy, safety, mobility, state of good repair, equity, and the environment in order to create a comprehensive freight network. This document is especially important for the two designated manufacturing and industrial centers, the BINMIC and Greater Duwamish MIC, and the Port of Seattle.

Transportation Capital Improvement Program

For the 2021 to 2026 period, the Capital Improvement Program (CIP) plans to invest more than \$1.5 billion on developing, maintaining, and operating Seattle's transportation system. The CIP aims to promote safe and efficient movement of people and goods and to enhance the quality of life, environments and economy within the city and surrounding areas. Funding has been designated for <u>a subset of projects</u> in all four of the adopted modal plans. Highlighted improvement projects include:

- Heavy Haul Network Program
- East Marginal Way, Phase I (separated bicycle/pedestrian facility between S Atlantic St and Spokane St) and Phase II (roadway reconstruction, signal, and ITS enhancements and safety measures to reduce conflicts between freight and non-motorized users)
- West Marginal Way Safe Street and Accessibility Improvements—rail crossing improvements, street crossing improvements, and sidewalk connections in the vicinity of West Marginal Way SW and SW Alaska St.
- Freight Spot Improvement Program—small scale mobility improvements to connections between port facilities, railroad intermodal yards, industrial businesses, the regional highway system, and supply chain first and last miles such as pavement repairs in industrial areas, turning radius adjustments, channelization changes, left-turn improvements, and signage.
- West Seattle Bridge Program—early work on the Reconnect West Seattle multimodal strategy, emergency repairs and bridge stabilization, bridge replacement options analysis and design, and Spokane Swing Bridge repairs and enhancements.
- Georgetown to South Park Trail—shared use path between Georgetown and South Park neighborhoods.
- New sidewalks, particularly near schools;
- School safety improvements;
- Pedestrian crossing improvements and stairway rehabilitation;
- Neighborhood greenways, bicycle lanes, and bicycle parking;
- Madison Street Bus Rapid Transit;
- RapidRide Roosevelt and Multimodal Corridor;
- South Lander Street Grade Separation Project (completed in 2020)—grade separation of S

Lander Street roadway and the BNSF railroad tracks between First Avenue S and Fourth Avenue S;

- Bridge rehabilitation and replacement; and
- Alaskan Way Main Corridor and Overlook Walk and East-West Connections Project.

Complete Streets

This 2006 policy directs SDOT to consider roadway designs that balance the needs of all roadway users, including pedestrians, bicyclists, transit riders and people of all abilities, as well as automobiles and freight. Design decisions are based on data, such as the adjacent land uses and anticipated future transportation needs. There is no set design template for complete streets as every situation requires a unique balance of design features within the available right-of-way. With respect to Major Truck Streets, the Complete Streets Ordinance (Section 3 of Ordinance No. 122386) states, "Because freight is important to the basic economy of the City and has unique right-of-way needs to support that role, freight will be the major priority on streets classified as Major Truck Streets. Complete Street improvements that are consistent with freight mobility but also support other modes may be considered on these streets." However, the

SDOT has developed a Right-of-Way Improvements Manual, called Seattle Streets Illustrated, which helps property owners, developers, engineers, and architects who are involved in the design, permitting, and construction of local streets. Streets Illustrated sets standards for a variety of elements of the public right-of-way including sidewalks, landscaping, bicycle lanes, transit stop amenities, and vehicle lane widths.

Intelligent Transportation Systems (ITS) Strategic Plan

For the 2010-2020 period, the Intelligent Transportation Systems (ITS) Strategic Plan provides a 10-year approach for implementing ITS across Seattle. ITS employs electronic and communication technologies on the streets, as well as automated traffic systems, to enhance mobility for all modes by increasing the efficiency and safety of the transportation infrastructure. The goal of the strategic plan is to ensure the existing ITS infrastructure is maintained and preserved, maximize the value of the existing infrastructure, and expand ITS to provide additional geographic coverage and services to travelers.

PSRC Regional Centers Framework Update

<u>PSRC adopted a Regional Centers Framework Update in 2018 to support the regional centers</u> <u>concept defined in VISION 2040 and the Regional Growth Strategy. The Regional Centers</u> <u>Framework Update includes a revised structure and criteria for defining regional and</u> <u>countywide centers and direction on policy and procedure updates. Of particular relevance to</u> <u>the study area, the Regional Centers Framework Update recognizes that MICs preserve lands</u> <u>for living-wage jobs which may offer higher than average wages, provide employment growth</u> <u>opportunities, act as a critical regional resource, support national and international trade, and</u> generate revenue for local governments. With respect to transportation, PSRC notes the unique characteristics of MICs and the critical nature of infrastructure to move freight and goods via trucks, rail, marine, and air modes to support manufacturing and industrial activity and regional economic objectives.

PSRC Plan Review Manual

The Plan Review Manual provides guidance on PSRC's plan review and plan certification program, with the most recent update completed in 2021 to align with VISION 2050. The manual assists jurisdictions in developing, updating, or amending center plans and identifies planning expectations for Regional Manufacturing/Industrial Center plans. The Plan Review Manual includes the following relevant expectations for policies and programs in MICs:

- Identify strategies to address deficiencies in the center's transportation network.
- Prioritize transportation projects that provide access to freight intermodal facilities to optimize freight movement for local, regional, and national distribution.
- Reduce commute impacts through Transportation Demand Management (TDM) strategies.
- Support an integrated multimodal transportation network, including freight, transit, pedestrian, and bicycle facilities and linkages to adjacent neighborhoods and districts.
- Identify strategies to achieve a mode split goal that advances a more sustainable mix of auto, transit, and non-motorized trips.

Current Conditions

This section describes current transportation conditions for all modes in the study area. Where applicable, more detail is provided at the subarea level.

Roadway UsersAuto & Freight

The City of Seattle is served by a dense roadway system of principal, minor, and collector arterials, as shown in **Exhibit 3.10-7**. Auto and freight travel also access several state highways—I-5, SR 99, and SR 509—which run north-south through the city. Bridges in the study area play a central role in facilitating travel across waterways and steep topography; these include the Ballard Bridge, Magnolia Bridge, West Seattle Bridge, 1st Avenue S Bridge, and South Park Bridge. In addition, rail grade separations act as both both structural constraints and connections across railways; these include locations in the Duwamish MIC including 1st Avenue South, 4th Avenue South, Airport Way, South Lander Street, and the Spokane Street Viaduct. The study area includes some of the most constrained areas of the city given the nature and location of water crossings and maritime and industrial land uses.

The City has designated a major truck street network throughout the city as shown in **Exhibit 3.10-13**. In the study area, the major truck street network includes most major arterials, including SR 99, SR 509, W Marginal Way SW, E Marginal Way S, 1st Avenue S, 4th Avenue S, Elliott Way, 15th Avenue W, and Leary Way. The Seattle Zero Emissions Freight Study included an evaluation of multiple data sources to understand freight activity throughout the city. The study found that roughly 2% of all vehicles in the Interbay area are freight vehicles while roughly 5% of vehicles in SODO are freight vehicles. In both areas, approximately one-quarter of freight vehicles are light-duty commercial vehicles and over one-half are medium-duty trucks. Most delivery VMT within the city is generated by medium-duty trucks. An analysis of freight activity within the Greater Duwamish MIC found that 50-70% of all medium- and heavy-duty truck trips in the Duwamish Valley are pass-through trips while 75-80% of medium- and heavy-duty truck trips in South Park—where SR 99, SR 509, and the South Park Bridge are located—are pass-through trips.



Exhibit 3.10-712 Existing Roadway Network, 2021



Exhibit 3.10-13 Existing Freight Network, 2021

Travel Time

Using the HCM guidelines for defining LOS thresholds as described in the Data & Methods section, **Exhibit 3.10-8** summarizes the travel time conditions along each of the study corridors. The existing travel time was calculated using the 25th percentile speeds for PM peak hour (4:45-5:45pm) for each direction of the study corridors. In other words, the travel time estimates reflect a somewhat more congested condition than the average day. This analysis speaks to the relative congestion experienced by autos as well as freight, particularly as the study corridors were selected from the major street truck network. Traffic congestion is more difficult for freight to navigate and trucks typically travel at slower speeds than general auto traffic. However, much of the daily freight movement activity occurs in the midday when traffic congestion is less pronounced. Additional discussion related to freight mobility is provided in the subsequent section.

For facilities that have peak directional patterns, the AM peak hour is typically expected to have similar characteristics in the opposite direction than those shown for the PM peak hour. For example, 15th Avenue W shows longer travel times northbound in the PM peak hour so similar conditions are expected southbound during the AM peak hour. The travel times shown below are rounded to the nearest half minute.

<u>Exhibit 3.10-8-14</u>	Existing PM Peak Hour LOS

		<u>PM Peak Hour LOS</u>		<u>Observed Travel</u> <u>Time (Minutes)</u>	
<u>ID</u>	<u>Corridor</u>	<u>N/E</u>	<u>S/W</u>	<u>N/E</u>	<u>S/W</u>
Balla	ard Interbay Northend MIC				
<u>1</u>	15th Ave W from Magnolia Bridge to NW Leary Way	<u>E</u>	<u>A</u>	<u>11.5</u>	<u>4.5</u>
<u>2</u>	15th Ave NW from NW Leary Way to N 85th St	<u>E</u>	<u>C</u>	<u>9.5</u>	<u>6.5</u>
<u>3</u>	Leary Ave NW/ Leary Way NW/ N 36th S/ Fremont Bridge between NW Market St and W Nickerson St/Westlake Ave	<u>C</u>	<u>C</u>	<u>11.0</u>	<u>12.0</u>
<u>4</u>	Shilshole Ave NW between NW Market and 15th Ave NW	<u>B</u>	<u>D</u>	<u>2.5</u>	<u>3.5</u>
<u>5</u>	NW Market St/N 50th/ 46th St between 24th Ave NW and I-5	<u>C</u>	<u>D</u>	<u>14.0</u>	<u>16.5</u>
<u>6</u>	<u>W Nickerson St/Westlake Ave N between 15th Ave W and</u> <u>Mercer St</u>	<u>C</u>	<u>C</u>	<u>13.0</u>	<u>11.5</u>
<u>7</u>	W Dravus St between 15th Ave W and 20th Ave W	<u>E</u>	<u>D</u>	<u>2.0</u>	<u>1.5</u>
<u>8</u>	Elliott Ave W between Magnolia Bridge and Wall St	<u>B</u>	<u>B</u>	<u>5.5</u>	<u>6.0</u>
<u>9</u>	W Mercer St from Elliott Ave W to I-5	<u>E</u>	<u>E</u>	<u>32.0</u>	<u>22.0</u>
<u>10</u>	Denny Way from Elliott Ave W to I-5	<u>E</u>	<u>E</u>	<u>14.5</u>	<u>11.0</u>
<u>11</u>	Magnolia Bridge between 15th Ave W and Thorndyke Ave W	<u>B</u>	A	<u>2.5</u>	<u>1.5</u>
<u>12</u>	SR 99 between N 46th St and Denny Way ¹	<u>C</u>	<u>E</u>	<u>8.5</u>	<u>14.0</u>
<u>13</u>	W Emerson St between 15th Ave W and Gilman Ave W	E	E	<u>6.0</u>	<u>4.0</u>
<u>14</u>	N 85th St between 15th Ave NW and I-5	<u>E</u>	E	<u>13.0</u>	<u>14.5</u>

		PM Peak Hour LOS <u>Tin</u>		<u>Observe</u> <u>Time (N</u>	<u>served Travel</u> <u>me (Minutes)</u>	
<u>ID</u>	<u>Corridor</u>	<u>N/E</u>	<u>S/W</u>	<u>N/E</u>	<u>S/W</u>	
<u>15</u>	I-5 between N 85th Street and Madison Street ¹	<u>E</u>	<u>E</u>	<u>19.0</u>	<u>24.0</u>	
<u>Grea</u>	ater Duwamish MIC					
<u>1</u>	1st Ave S between S Royal Brougham Way and SR 99	<u>C</u>	<u>C</u>	<u>11.0</u>	<u>11.0</u>	
<u>2</u>	4th Ave S between Seattle Blvd S to E Marginal Way S	<u>C</u>	<u>C</u>	<u>12.0</u>	<u>12.5</u>	
<u>3</u>	6th Ave S between Seattle Blvd S to Spokane St Viaduct	<u>C</u>	<u>C</u>	<u>6.5</u>	<u>6.0</u>	
<u>4</u>	<u>Airport Way S/ Seattle Blvd S between S Royal Brougham Way</u> <u>to S Boeing Access Rd</u>	A	A	<u>16.5</u>	<u>15.5</u>	
<u>5</u>	West Seattle Bridge/Spokane St Viaduct between 35th Ave SW and I-5	<u>C</u>	<u>E</u>	<u>6.5</u>	<u>10.0</u>	
<u>6</u>	Spokane St Bridge between Harbor Ave SW and SR 99	<u>B</u>	<u>B</u>	<u>4.5</u>	<u>4.5</u>	
<u>7</u>	E Marginal Way S between SR 99 and S Boeing Access Rd	<u>C</u>	<u>D</u>	<u>8.5</u>	<u>10.5</u>	
<u>8</u>	Alaskan Way S from Broad St to SR 99	<u>D</u>	E	<u>9.0</u>	<u>13.0</u>	
<u>9</u>	S Royal Brougham Way between SR 99 and Airport Way S	<u>E</u>	<u>D</u>	<u>4.5</u>	<u>3.0</u>	
<u>10</u>	Edgar Martinez Dr S between SR 99 and 4th Ave	<u>E</u>	<u>E</u>	<u>2.5</u>	<u>2.5</u>	
<u>11</u>	<u>S Holgate St between 1st Ave and Airport Way S</u>	<u>D</u>	E	<u>3.0</u>	<u>4.5</u>	
<u>12</u>	<u>S Lander St between 1st Ave and Airport Way S</u>	<u>E</u>	<u>E</u>	<u>4.0</u>	<u>4.0</u>	
<u>13</u>	S Lucile St between SR 99 and Airport Way S	<u>D</u>	<u>E</u>	<u>4.0</u>	<u>5.0</u>	
<u>14</u>	W Marginal Way SW between West Seattle Bridge and 2nd Ave	Δ	Δ	<u>5.0</u>	<u>4.5</u>	
<u>15</u>	S Michigan St/ Corson Ave S between E Marginal Way S and I-5	<u>C</u>	<u>E</u>	<u>3.5</u>	<u>5.5</u>	
<u>16</u>	E Marginal Way S/SR 99 between S Atlantic Street and 1st Ave S Bridge	Δ	A	<u>9.0</u>	<u>9.0</u>	
<u>17</u>	I-5 between Madison Street and SR 5991	E	E	<u>25.5</u>	<u>30.0</u>	
<u>18</u>	SR 509 between SR 99 and SR 518 ¹	<u>A</u>	<u>C</u>	<u>6.0</u>	<u>9.0</u>	
<u>19</u>	SR 99/599 between SR 509 and I-5 ²	A	<u>B</u>	<u>6.0</u>	<u>8.0</u>	

 Note:
 1. WSDOT sets a LOS D standard on I-5, SR 509, and SR 99 north of SR 509.

 2. WSDOT sets a LOS E standard on SR 99/599 between SR 509 and I-5.

Source: Wejo, 2019; Fehr & Peers, 2021.

During the PM peak hour, most corridors operate at LOS E or better in both directions. Corridors operating at LOS F include:

- Both directions of W Mercer St from Elliott Avenue W to I-5
- Eastbound Denny Way from Elliott Avenue W to I-5
- Both directions of W Emerson St from Gilman Avenue W to 15th Avenue W
- Both directions of I-5 between N 85th Street and Madison Street
- Southbound Alaskan Way S from Broad St to SR 99
- Eastbound S Royal Brougham Way between SR 99 and Airport Way S

- Both directions of Edgar Martinez Dr S between SR 99 and 4th Avenue
- Westbound S Holgate St from Airport Way S to 1st Avenue
- Southbound I-5 from Madison Street to SR 599

<u>Ballard</u>

In the Ballard Subarea, principal arterials include 15th Avenue NW and Leary Way NW. These roadways, as well as Shilshole Avenue NW, carry high volumes of freight traffic in the area. Along 15th Avenue NW, the peak direction of travel during the PM peak hour is northbound with more balanced volumes on Leary Avenue NW and Shilshole Avenue NW. All study corridors in the Ballard Subarea operate at LOS E or better during typical conditions.

<u>Interbay Dravus</u>

The principal arterials and freight corridors in the Interbay Dravus Subarea include 15th Avenue W, W Dravus Street, W Emerson Street, and W Nickerson Street. All study corridors except W Emerson Street operate at LOS E or better in the Interbay Dravus study area during typical conditions.

Interbay Smith Cove

In the Interbay Smith Cove Subarea, the principal arterials and freight routes include 15th Avenue W, W Mercer Street, Denny Way, and Elliott Avenue W. The Magnolia Bridge is classified as a minor arterial as well as a freight route. Congestion stemming from the I-5 on-ramps affects travel times in the eastbound direction of both Denny Way and W Mercer St which operate at LOS F. Both routes typically have less congestion on the western ends closer to the study area, but congestion increases along the corridors as they near center city and I-5.

<u>SODO/Stadium & Georgetown</u>

In the SODO/Stadium Subarea, 1st Avenue S, 4th Avenue S, and E Marginal Way are primary arterials, and most other roadways are minor arterials. The West Seattle Bridge and the Spokane Street Bridge both span the Duwamish Waterway. The West Seattle Bridge has been closed since March 2020, resulting in major travel pattern changes and increased demand on alternate routes. However, the existing conditions discussed in this report focuses on the 2019 period, when operations were more "typical," both in terms of the available network and prepandemic travel demand.

Because of the predominantly industrial land uses, all arterials in the subarea are designated as freight routes. In particular, East Marginal Way S carries a high percentage of cargo trucks and provides access to multiple terminal entrances. Most corridors operate at LOS E or better during the PM peak hour, with the exception of the east/west corridors of S Holgate Street, S Royal Brougham Way, and Edgar Martinez Drive S.

Georgetown/South Park

In the Georgetown/South Park Subarea, all minor and principal arterials are designated freight corridors, including E Marginal Way S, 1st Avenue S, and S Michigan Street. Airport Way S is often used as a bypass of I-5 when the interstate is highly congested due to collisions or construction. As noted above, this area has been experiencing an increase in traffic volumes since March 2020 when the closure of the West Seattle Bridge caused motorists to seek alternate routes. Under typical 2019 conditions, almost all corridors operate at LOS E or better.

Mode Share

The existing SOV mode share in the City of Seattle is summarized by sector using the PSRC Soundcast model and is shown in **Exhibit 3.10-9**. Within the study area, the Duwamish sector has the highest share of PM peak period SOV trips at 53.5%. Magnolia/Queen Anne and Northwest Seattle have lower SOV percentages, as these sectors contain a larger mix of residential and commercial uses.

Sector	Existing SOV Share
Duwamish	<u>53.5%</u>
Magnolia/Queen Anne	<u>43.1%</u>
Northwest	<u>41.6%</u>

Exhibit 3.10-9-15 Existing SOV Mode Share—PM Peak Period

Source: PSRC, 2021; Fehr & Peers, 2021.

Screenlines

The City's screenline thresholds are in the form of a volume-to-capacity (v/c) ratio: the number of vehicles crossing the screenline compared to the designated capacity of the roadways crossing the screenline. **Exhibit 3.10-10** summarizes the location of the study area screenlines, as well as their LOS threshold and current v/c ratio. All screenline locations are currently under the LOS threshold defined by the City of Seattle. A table showing the number of vehicles expected to cross each studied screenline during the PM peak hour in shown in **Appendix I**.

Exhibit 3.10-10-16 Existing PM Peak Hour LOS

<u>enline</u>		Volume-to-	<u>2019 Pl</u> Period v	<u>M Peak</u> v/c Ratio
<u>Scre</u>	<u>Location</u>	<u>Capacity</u> <u>Threshold</u>	<u>N/E</u>	<u>s/W</u>
<u>2</u>	Magnolia	<u>1.0</u>	<u>0.51</u>	<u>0.54</u>
<u>3.11</u>	Duwamish River—West Seattle Bridge and Spokane Street	<u>1.2</u>	<u>0.57</u>	<u>0.53</u>
<u>3.12</u>	Duwamish River—1st Avenue S and 16th Avenue S	<u>1.2</u>	<u>0.54</u>	<u>0.51</u>

reenline		<u>Volume-to-</u> <u>Capacity</u>	<u>2019 PM</u> Period v	<u>M Peak</u> /c Ratio
Sci	Location	<u>Threshold</u>	<u>N/E</u>	<u>S/W</u>
<u>4.13</u>	South City Limit—SR 99 to Airport Way S	<u>1.0</u>	<u>0.40</u>	<u>0.45</u>
<u>5.11</u>	Ship Canal—Ballard Bridge	<u>1.2</u>	<u>1.01</u>	<u>0.75</u>
<u>5.12</u>	Ship Canal—Fremont Bridge	<u>1.2</u>	<u>0.59</u>	<u>0.66</u>
<u>5.13</u>	Ship Canal—Aurora Bridge	<u>1.2</u>	<u>0.30</u>	0.34
<u>7.11</u>	West of Aurora Avenue—Fremont Place N to N 65th Street	<u>1.0</u>	<u>0.54</u>	<u>0.62</u>
<u>8</u>	South of Lake Union	<u>1.2</u>	<u>0.62</u>	<u>0.69</u>
<u>9.12</u>	South of Spokane Street—E Marginal Way to Airport Way S	<u>1.0</u>	<u>0.47</u>	<u>0.48</u>
<u>10.11</u>	South of S Jackson Street—Alaskan Way S to 4th Avenue S	<u>1.0</u>	<u>0.58</u>	<u>0.66</u>

Source: City of Seattle count data, 2019; Fehr & Peers, 2021.

<u>Freight</u>

This section addresses the freight network and its users. This includes descriptions of the roadway network, rail network, and intermodal yards that support the industrial uses throughout the MICs. Related issues such as truck parking and travel time reliability are also discussed. Note that safety is discussed in a separate section as it inherently relates to multiple modes and their potential conflicts.

Roadway Network

The City has designated a truck street network throughout the city as shown in **Exhibit 3.10-11** and **Exhibit 3.10-12**. In the study area, the major truck street network includes most major arterials, including SR 99, SR 509, W Marginal Way SW, E Marginal Way S, 1st Avenue S, 4th Avenue S, Elliott Way, 15th Avenue W, and Leary Way. Many of these streets are also designated as Over-legal Routes which are designated routes that may accommodate oversized and overweight trucks. The City requires permits to operate such vehicles on the designated over-legal routes.

Travel Patterns of Industrial Workers

While the most congested transit conditions occur during conventional AM and PM peak periods, some industrial land uses generate different temporal patterns. For example, some workers need to commute during offpeak periods for their shifts when transit options are more limited. Moreover, workers within the study area commute from a wide geographic area. As summarized in Exhibit 3.9-11 and mapped in Exhibit 3.9-13, roughly 40% of study area workers commute less than 10 miles; 37% commute 10-24 miles; 13% commute 25-50 miles; and 10% commute more than 50 miles. Therefore, the challenge in accessing transit service for some industrial workers may be the availability or convenience of the service.

In October 2015, the City of Seattle approved legislation that established a Heavy Haul network of city streets to allow heavier cargo containers to be transported between the Port of Seattle, industrial businesses, and rail yards with appropriate permits. The measure also provides a framework and funding to repair and build roadways within the network, calls for semi-annual safety inspections of heavy haul trucks, and aligns city weight regulations with those of the state and other municipalities across the country.

Exhibit 3.10-11 and **Exhibit 3.10-12** also show the locations of the three intermodal facilities serving the MICs: Balmer Yard in the BINMIC (operated by BNSF), Seattle International Gateway (SIG, also operated by BNSF) in the Duwamish MIC, and the Seattle ARGO Terminal (operated by Union Pacific) in the Duwamish MIC. At these facilities, freight is transferred between truck and railcar. As this area serves a seaport, freight is also transferred between ships and trucks and/or rail. Streets frequently used as seaport intermodal connectors in the Duwamish MIC are mapped in **Exhibit 3.10-12**. These routes provide access between Port terminals and railroad intermodal facilities.

<u>Truck activity tends to be highest during the morning and midday and avoid the afternoon peak</u> which is the most congested period of the day, as described in the Data and Methods section. <u>To provide a conservative impact analysis, the most congested hour (4:45-5:45pm) is used for</u> analysis of total vehicle volumes (passenger vehicles, trucks, and buses) and travel times.



Exhibit 3.10-11 Existing Roadway Freight Network—Ballard Interbay Northend MIC, 2021



Exhibit 3.10-12 Existing Roadway Freight Network—Greater Duwamish MIC, 2021

Source: City of Seattle, 2022; Fehr & Peers, 2022.

Traffic Congestion and Reliability

The travel time analysis presented in the previous section speaks to the relative congestion experienced by freight during the PM peak hour which is the most congested time period of the day. However, much of the daily freight movement activity occurs in the midday when overall traffic congestion is less pronounced. While the travel time analysis indicates the general congestion levels of the study corridors, it should be noted that traffic congestion is more difficult for freight to navigate and trucks typically travel at slower speeds than general auto traffic.

<u>Freight corridors that experience substantial congestion are also more likely to have poor travel</u> <u>time reliability, a key concern for freight operators. Traffic congestion leads to worse reliability</u> <u>on the roadway network as the system is less resilient to recover from disruptions such as</u> <u>blocking incidents; moreover, freight operators have a more difficult time maneuvering in</u> <u>traffic congestion.</u>

Rail Network

Rail is also a critical mode for freight movement within the MICs, as shown in **Exhibit 3.10-13** and **Exhibit 3.10-14**. There are two Class 1 railroads in Seattle: BNSF and the Union Pacific Railroad (UP). The BNSF mainline extends north-south through Seattle and operates in a doubled-tracked tunnel through downtown, serving Balmer Yard in the BINMIC and SIG in the Duwamish MIC. The UP mainline only operates south of downtown Seattle and parallels the BNSF network, serving the Seattle ARGO Terminal. The MICs also include a variety of local rail spurs that provide direct rail service to businesses as well as on-dock rail at Port of Seattle terminals. Lastly, Sound Transit's light rail system has several at-grade crossings in the study area.

As shown in **Exhibit 3.10-13** and **Exhibit 3.10-14**, the BNSF and UP railroads cross roadways in many locations throughout the MICs. While at-grade crossings are more limited in the BINMIC, they are prevalent throughout the Duwamish MIC. When a train is passing through these locations, the crossing is closed to vehicle traffic resulting in delays to those on the roadway network, particularly truck freight in heavily industrial areas. Delays depend on the frequency and duration of the at-grade crossing closure and have been identified by the freight community as a key challenge for truck freight mobility.



Exhibit 3.10-13 Existing Rail Freight Network—Ballard Interbay Northend MIC, 2021



Exhibit 3.10-14 Existing Rail Freight Network—Greater Duwamish MIC, 2021

Truck Parking

<u>Truck parking is of particular importance in the MICs as drivers need places to stage for loading,</u> <u>store their truck, and take required rest periods which are critical to safety. Curb space for</u> <u>trucks is also needed to conduct deliveries to businesses throughout the study areas. The</u> <u>Freight Master Plan identified adequate truck parking as an issue and included two related</u> <u>actions which speak to both the availability of truck parking as well as community impact if</u> <u>truck parking is not provided in appropriate locations:</u>

- Action 3.5.1: Work with the Port of Seattle and other partners to determine suitable locations and technology to provide and manage additional truck parking.
- Action 5.2.2: Reduce long-term truck parking on residential streets through education and enforcement activities, and identify alternative truck parking locations.

Past reviews of truck parking supply and demand have estimated that there is demand for an additional 500 to 900 truck parking spaces in the city. Due to the shortage in parking areas suitable for trucks (and with supportive facilities nearby to serve driver needs), there is truck parking overflow from industrial areas resulting in some trucks parking illegally in residential neighborhoods. Truck parking is a regional, and even statewide, issue. WSDOT has a Joint Transportation Committee (JTC) Truck Parking Action Plan that is focused on understanding truck parking challenges and identifying solutions. City staff are actively engaged in these efforts. Locally, the City Council passed Ordinance 126647 in August 2022 which establishes the authority for SDOT to designate parking for truck tractors in public right-of-way and enforce such signage.

Active Transportation

The pedestrian network is composed of sidewalks, walkways, crosswalks, staircases, curb ramps, and multi-use trails. The presence, connectivity, and quality of the pedestrian network varies throughout the area often correlating with the prevailing land use. Industrial areas tend to have fewer pedestrian facilities and limited connectivity while adjacent commercial and residential areas usually have moderately dense pedestrian networks with sidewalks on at least one side of nearly all streets, and most intersections have marked crosswalks and curb ramps. Some pedestrian crossing locations have been enhanced with signage and/or curb extensions which shorten crossing distances. SDOT maintains an inventory of pavement condition which indicates that conditions tend to be poorer in more industrial areas such as SODO, South Park, and waterfront areas within the BINMIC.

The existing bicycle network is made up of bicycle lanes, cycle tracks (protected bike lanes), multi-use trails, signed routes, and shared streets known as Neighborhood Greenways designated with "sharrow" markings. Bicycle facilities are distributed throughout the city but are most prevalent in the Center City area situated between the MICs. The study area includes a variety of multi-use trails along waterways adjacent to industrial areas. This includes the Burke-Gilman Trail and Ship Canal Trail in the Ballard and Interbay Dravus areas; the Elliott Bay Trail connecting the Interbay Smith Cove Subarea to SODO; and the SODO Trail, West Seattle Bridge Trail, and Duwamish River Trail in the SODO/Stadium and Georgetown/South Park subareas. Pedestrian and bicycle facilities are mapped in **Exhibit 3.10-15** and **Exhibit 3.10-16**. The City of Seattle maintains data layers showing existing sidewalk and curb ramps; findings and trends from this data are described in the following sections. However, these data are not shown in the following exhibits due to legibility of the maps at the study area level. <u>To explore the detailed data, the City's interactive GIS database can be accessed here.</u>









Source: City of Seattle, 2020; Fehr & Peers, 2021.

<u>Ballard</u>

Within the Ballard Subarea, there are sidewalks on both sides of nearly all streets within the study area. However, sidewalks and pedestrian connectivity are more limited closest to the waterfront where there are large parcels of industrial uses. There are limited marked crosswalks in the study area, most of which are located on Leary Way NW at major intersections. Curb ramps are generally present within the street grid, but there are some missing stretches, particularly along 14th Avenue NW and NW 50th Street, as well as within the industrial areas along the waterfront.

The Ballard study area includes a portion of the Burke-Gilman Trail, which includes frequent marked crossings west of Leary Way NW. There are also separated bike lanes on NW 45th Street that connect to the Burke-Gilman Trail at 11th Avenue NW. The Ballard Subarea is home to the "missing link" of the Burke-Gilman Trail which stretches from 11th Avenue NW to the Ballard Locks. Construction is underway along Market Street to include a shared use trail and sidewalk with plans to complete the trail along Shilshole Avenue NW and NW 45th Street. To cross the Ship Canal, people walking and biking share narrow pathways on either side of the Ballard Bridge. Due to the limited width of the facilities, it is difficult for people to pass one another comfortably, as shown in **Exhibit 3.10-17**.

Exhibit 3.10-17-9 Pedestrian and Bicycle Facility Constraints



Note: Photo at left shows the Ballard Bridge and photo at right shows the Elliott Bay Trail between Terminal 91 and the BNSF Railyard. Source: Seattle Department of Transportation, 2020.

Interbay Dravus

Interbay Dravus has a relatively complete pedestrian network along the main arterials; however, most other roadways in this industrial area have no sidewalks. There are limited marked pedestrian crossings, with marked crosswalks and curb ramps only at the major intersections along W Dravus Street and W Emerson Street. W Dravus Street and W Emerson Place/Street serve as the only connections across the railway between the North Queen Anne and Southeast Magnolia neighborhoods. W Dravus Street provides sharrows from 15th Avenue W to 20th Avenue W as well as sidewalks along the bridge. The Ship Canal Trail transitions to a cycle track along W Emerson Place. Both facilities connect with separated bike lanes on Gilman Avenue W and 20th Avenue W that provide bicycle connections to the Elliott Bay Trail to the south.

Interbay Smith Cove

The Interbay Smith Cove Subarea has minimal public pedestrian facilities, as the Seattle Armory and Port of Seattle properties comprise most of the subarea. Outside of this industrial area, there are sidewalks on both sides of nearly all streets, including the major thoroughfare of 15th Avenue W/Elliott Avenue W. Marked crosswalks and curb ramps exist about every fifth of a mile along this corridor. However, pedestrian and bicycle comfort along the corridor is affected by the width, traffic volumes, and speeds along the roadway.

East-west connectivity across the subarea is very limited. Travelers can use the Elliott Bay Trail around the perimeter of Terminal 91 or the Magnolia Bridge, which is the only roadway that provides public access east/west in Interbay Smith Cove. The Elliott Bay Trail has a constrained section, shown in **Exhibit 3.10-17**, where the trail passes through the Terminal 91 area. The Magnolia Bridge has a narrow sidewalk on one side; the bridge can be used to connect to 16^r Avenue W beneath the Magnolia Bridge or to the Magnolia neighborhood to the west though the grade is steep.

SODO/Stadium

In the SODO/Stadium Subarea, the pedestrian network is generally complete north of the West Seattle Bridge, with sidewalks on both sides of nearly all streets. Marked crosswalks and curb ramps exist at the major intersections within the area, along the north/south corridors of 1st Avenue, 4th Avenue, 6th Avenue, and Airport Way. However, the major east/west corridors in the subarea are spaced about a half-mile apart, which limits crossing options and increases travel distances for people looking to cross the street between these intersections. South of the bridge, sidewalks only exist along E Marginal Way, 1st Avenue S, and 4th Avenue S, with very limited marked crossings. West of the waterway on Harbor Island and Terminal 5, sidewalks exist on portions of 16th Avenue SW and along the lower Spokane Street Bridge, but the only marked crosswalks are at the port access intersections along the Spokane Street Bridge.

In the subarea, there are minimal bicycle facilities, with sharrows along 1st Avenue S and S Lander Street. The multi-use SODO Trail provides a bicycle connection between the SODO and

Stadium Link Light Rail stations and there are bike lanes along E Marginal Way S connecting the Waterfront Trail and the West Seattle Bridge Trail.

Georgetown/South Park

The Georgetown/South Park Subarea has a less dense pedestrian network, with sidewalks on both sides of the streets along the arterials such as Airport Way S, Ellis Avenue S, and S Michigan Street. Sidewalks also exist in the residential neighborhood located between Corson Avenue S and Ellis Avenue S. However, there are limited pedestrian crossings in the area, as marked crosswalks and curb ramps exist at only a few major intersections.

The Duwamish River Trail runs along the west side of the subarea providing a north-south route along the west side of the waterway. Bicycle sharrows exist on some local streets within the subarea, and separated bike lanes are present on Ellis Avenue S and E Marginal Way S. Connections across the Duwamish Waterway are limited: there is a shared use facility running alongside the 1st Avenue Bridge and sidewalks are provided on both sides of the 16th Avenue S Bridge. The Georgetown to South Park connection, to be constructed in 2022, will link the two neighborhoods via E Marginal Way and 16th Avenue S.

Transit

The study area and surrounding neighborhoods are served by King County Metro and Sound Transit public transit including local, rapid, and express fixed route bus services as well as light rail.

- King County Metro operates a fixed route bus system that includes RapidRide, a separatelybranded set of frequent transit routes in West Seattle, Ballard, and Downtown.
- Sound Transit Express and Community Transit operate buses that provide service from outside the City of Seattle.
- Rail transit services include Sound Transit Link Light Rail, City-operated streetcars in South Lake Union and First Hill, and the Sounder commuter train that provides service from King Street Station north to Everett and south to Tacoma.

Sound Transit's expansion of Link Light Rail will provide expanded rail service to the SODO, Interbay Smith Cove, Interbay Dravus, and Ballard study areas. **Exhibit 3.10-18** displays the existing transit services as well as the five screenlines used to summarize demand along key transit corridors in the study area.

Sound Transit reports its ridership and passenger load trends in its annual Service Implementation Plan. According to the 2020 Service Implementation Plan, which reflects conditions in 2019 before the COVID-19 pandemic disrupted typical travel patterns, Link light rail between Angle Lake and the University of Washington had average weekday boardings of over 80,000. Peak loads typically occur between the CID and Pioneer Square stations just north of the SODO/Stadium Subarea. During the PM peak period, peak flows are in the southbound direction through the subarea as people travel outbound from center city. Sound Transit monitors the passenger loads on each trip and found only one trip consistently exceeding the loading standard.



Exhibit 3.10-18-10 Existing Transit Network, 2021

Source: King County Metro, 2021; Sound Transit, 2021; Fehr & Peers, 2021.

King County Metro ridership data for the PM peak period was summarized for each route that crosses a study area screenline. The average maximum load for each trip was extracted and compared to the capacity of the trip (i.e., the number of seats on the bus plus standing room) to determine if the trip exceeded King County Metro's crowding threshold. Note that the maximum load does not necessarily occur at the screenline. For instance, routes leaving downtown for outlying areas tend to have maximum loads occurring closer to the center city. The average maximum loads for the study area routes were aggregated at the screenline level and results are reported in **Exhibit 3.10-19**.

	Average Maximum Load Factor on Routes Crossing Screenline			
Screenline	Inbound	Outbound		
A: East of 8th Avenue NW	0.63	1.21		
B: Ballard Bridge	0.98	1.13		
C: North of W Mercer Place	0.86	1.08		
D: North of Lander St	0.51	0.93		
E: West Seattle Bridge	0.49	0.95		

Exhibit 3.10-19 -11	Passenger Load Factors on Bus Route across Transit Screenlines
--------------------------------	--

Note: Inbound refers to travel into the downtown area and outbound travel out of the downtown area. Source: King County Metro, Fall 2018; Fehr & Peers, 2021.

Because the analysis period is the PM peak period, the outbound load factors are higher than the inbound load factors; the inverse pattern would be present during the AM peak period. The data show that many of the routes traveling across the study area screenlines operate over their crowding threshold at some point along their trip. Specific routes are discussed below.

Ballard, Interbay Dravus, & Interbay Smith Cove

The Ballard Bridge screenline includes routes traveling north-south through the Interbay area and into areas of center city. Nearly all of the routes traveling across the bridge exceed their crowding threshold at some point for more than half of their PM peak period trips. This includes the D Line (both inbound and outbound), Route 15, and Route 18. The screenline east of 8th Avenue NW captures routes 28, 40, and 44. All three of those routes exceed their crowding threshold on most of their PM peak period trips; however, the highest loads tend to occur closer to downtown or the U District rather than in the study area.

The screenline north of Mercer Place captures routes traveling along the Elliott Way/15th Avenue NW corridor. Several of these routes also cross the Ballard Bridge as described above. This screenline also includes routes serving Magnolia, Uptown, Fremont, Wallingford, and the U District. In addition to the D Line, Route 15, and Route 18 as mentioned above, Route 32 exceeds its crowding threshold on the majority of its PM peak period trips, with the maximum load usually occurring nearer to the U District.

SODO/Stadium & Georgetown

The north of Lander Street screenline captures a large number of routes that travel through the SODO/Stadium area as they approach center city. Most routes generally operate below their crowding thresholds. The exceptions are the outbound C Line, Route 118, and Route 101. The West Seattle Bridge screenline captures a variety of routes; however, only the C Line and Route 118 exceed their crowding thresholds on a majority of PM peak period trips.

Auto & Freight

The City of Seattle is served by a dense roadway system of principal, minor, and collector arterials, as shown in **Exhibit 3.10-12**. Auto and freight travel also access several state highways—I-5, SR 99, and SR 509—which run north-south through the city. Bridges in the study area play a central role in facilitating travel across waterways and steep topography; these include the Ballard Bridge, Magnolia Bridge, West Seattle Bridge, 1st Avenue S Bridge, and South Park Bridge. The study area includes some of the most constrained areas of the city given the nature and location of water crossings and maritime and industrial land uses.

The City has designated a major truck street network throughout the city as shown in **Exhibit 3.10-13**. In the study area, the major truck street network includes most major arterials, including SR 99, SR 509, W Marginal Way SW, E Marginal Way S, 1st Avenue S, 4th Avenue S, Elliott Way, 15th Avenue W, and Leary Way.

The Seattle Zero Emissions Freight Study included an evaluation of multiple data sources to understand freight activity throughout the city. The study found that roughly 2% of all vehicles in the Interbay area are freight vehicles while roughly 5% of vehicles in SODO are freight vehicles. In both areas, approximately one-quarter of freight vehicles are light-duty commercial vehicles and over one-half are medium-duty trucks. Most delivery VMT within the city is generated by medium-duty trucks. An analysis of freight activity within the Greater Duwamish MIC found that 50-70% of all medium- and heavy-duty truck trips in the Duwamish Valley are pass-through trips while 75-80% of medium- and heavy-duty truck trips in South Park—where SR 99, SR 509, and the South Park Bridge are located—are pass-through trips.







Exhibit 3.10-13 Existing Freight Network, 2021
Travel Time

Using the HCM guidelines for defining LOS thresholds as described in the Data & Methods section, **Exhibit 3.10-14** summarizes the travel time conditions along each of the study corridors. The existing travel time was calculated using the 25th percentile speeds for PM peak hour (4:45-5:45pm) for each direction of the study corridors. In other words, the travel time estimates reflect a somewhat more congested condition than the average day. Traffic congestion is more difficult for freight to navigate and trucks typically travel at slower speeds than general auto traffic. However, much of the daily freight movement activity occurs in the midday when traffic congestion is less pronounced.

For facilities that have peak directional patterns, the AM peak hour is typically expected to have similar characteristics in the opposite direction than those shown for the PM peak hour. For example, 15th Avenue W shows longer travel times northbound in the PM peak hour so similar conditions are expected southbound during the AM peak hour. The travel times shown below are rounded to the nearest half minute.

		PM Peak	Hour LOS	Observe Time (N	d Travel linutes)
₽₽	Corridor	N/E	S/W	N/E	s/W
Balla	ard Interbay Northend MIC				
1	15th Ave W from Magnolia Bridge to NW Leary Way	Ę	A	11.5	4.5
2	15th Ave NW from NW Leary Way to N 85th St	Æ	e	<u>9.5</u>	6.5
3	Leary Ave NW/ Leary Way NW/ N 36th S/ Fremont Bridge between NW Market St and W Nickerson St/Westlake Ave	÷	e	11.0	12.0
4	Shilshole Ave NW between NW Market and 15th Ave NW	₽	Ð	2.5	3.5
5	NW Market St/N 50th/ 46th St between 24th Ave NW and I-5	Ę	Ð	14.0	16.5
6	W Nickerson St/Westlake Ave N between 15th Ave W and Mercer St	÷	Ę	13.0	11.5
7	W Dravus St between 15th Ave W and 20th Ave W	E	Ð	2.0	1.5
8	Elliott Ave W between Magnolia Bridge and Wall St	₽	B	5.5	6.0
9	W Mercer St from Elliott Ave W to I-5	F	F	32.0	22.0
10	Denny Way from Elliott Ave W to I-5	F	E	14.5	11.0
11	Magnolia Bridge between 15th Ave W and Thorndyke Ave W	₽	A	2.5	1.5
12	SR 99 between N 46th St and Denny Way	÷	E	8.5	14.0
13	W Emerson St between 15th Ave W and Gilman Ave W	F	F	6.0	4.0
14	N 85th St between 15th Ave NW and I-5	E	E	13.0	14.5
15	I-5 between N 85th Street and Madison Street	F	F	19.0	24.0

Exhibit 3.10-14 Existing PM Peak Hour LOS

		PM Peak	Hour LOS	Observe Time (M	e <mark>d Travel</mark> linutes)
₽Đ	Corridor	N/E	S/W	N/E	s/W
Grea	ater Duwamish MIC				
1	1st Ave S between S Royal Brougham Way and SR 99	e	e	11.0	11.0
2	4th Ave S between Seattle Blvd S to E Marginal Way S	÷	e	12.0	12.5
3	6th Ave S between Seattle Blvd S to Spokane St Viaduct	÷	e	6.5	6.0
4	Airport Way S/ Seattle Blvd S between S Royal Brougham Way to S Boeing Access Rd	A	A	16.5	15.5
5	West Seattle Bridge/Spokane St Viaduct between 35th Ave SW and I-5	Ę	E	6.5	10.0
6	Spokane St Bridge between Harbor Ave SW and SR 99	₿	B	4.5	4.5
7	E Marginal Way S between SR 99 and S Boeing Access Rd	Ę	Ð	8.5	10.5
8	Alaskan Way S from Broad St to SR 99	Ð	F	9.0	13.0
9	S Royal Brougham Way between SR 99 and Airport Way S	F	Ð	4 .5	3.0
10	Edgar Martinez Dr S between SR 99 and 4th Ave	F	F	2.5	2.5
11	S Holgate St between 1st Ave and Airport Way S	Ð	F	3.0	4.5
12	S Lander St between 1st Ave and Airport Way S	E	E	4.0	4.0
13	S Lucile St between SR 99 and Airport Way S	Ð	E	4.0	5.0
14	W Marginal Way SW between West Seattle Bridge and 2nd Ave SW	A	A	5.0	4.5
15	S Michigan St/ Corson Ave S between E Marginal Way S and I-5	Ę	Ē	3.5	<u>5.5</u>
16	E Marginal Way S/SR 99 between S Atlantic Street and 1st Ave S Bridge	A	A	9.0	9.0
17	I-5 between Madison Street and SR 599	E	F	25.5	30.0

Source: Wejo, 2019; Fehr & Peers, 2021.

During the PM peak hour, most corridors operate at LOS E or better in both directions. Corridors operating at LOS F include:

- Both directions of W Mercer St from Elliott Avenue W to I-5
- Eastbound Denny Way from Elliott Avenue W to I-5
- Both directions of W Emerson St from Gilman Avenue W to 15th Avenue W
- Both directions of I-5 between N 85th Street and Madison Street
- Southbound Alaskan Way S from Broad St to SR 99
- Eastbound S Royal Brougham Way between SR 99 and Airport Way S
- Both directions of Edgar Martinez Dr S between SR 99 and 4th Avenue
- Westbound S Holgate St from Airport Way S to 1st Avenue
- Southbound I-5 from Madison Street to SR 599

Ballard

In the Ballard Subarea, principal arterials include 15th Avenue NW and Leary Way NW. These roadways, as well as Shilshole Avenue NW, carry high volumes of freight traffic in the area. Along 15th Avenue NW, the peak direction of travel during the PM peak hour is northbound with more balanced volumes on Leary Avenue NW and Shilshole Avenue NW. All study corridors in the Ballard Subarea operate at LOS E or better during typical conditions.

Interbay Dravus

The principal arterials and freight corridors in the Interbay Dravus Subarea include 15th Avenue W, W Dravus Street, W Emerson Street, and W Nickerson Street. All study corridors except W Emerson Street operate at LOS E or better in the Interbay Dravus study area during typical conditions.

Interbay Smith Cove

In the Interbay Smith Cove Subarea, the principal arterials and freight routes include 15th Avenue W, W Mercer Street, Denny Way, and Elliott Avenue W. The Magnolia Bridge is classified as a minor arterial as well as a freight route. Congestion stemming from the I-5 on-ramps affects travel times in the eastbound direction of both Denny Way and W Mercer St which operate at LOS F. Both routes typically have less congestion on the western ends closer to the study area, but congestion increases along the corridors as they near center city and I-5.

SODO/Stadium & Georgetown

In the SODO/Stadium Subarea, 1st Avenue S, 4th Avenue S, and E Marginal Way are primary arterials, and most other roadways are minor arterials. The West Seattle Bridge and the Spokane Street Bridge both span the Duwamish Waterway. The West Seattle Bridge has been closed since March 2020, resulting in major travel pattern changes and increased demand on alternate routes. However, the existing conditions discussed in this report focuses on the 2019 period, when operations were more "typical," both in terms of the available network and prepandemic travel demand.

Because of the predominantly industrial land uses, all arterials in the subarea are designated as freight routes. In particular, East Marginal Way S carries a high percentage of cargo trucks and provides access to multiple terminal entrances. Most corridors operate at LOS E or better during the PM peak hour, with the exception of the east/west corridors of S Holgate Street, S Royal Brougham Way, and Edgar Martinez Drive S.

Georgetown/South Park

In the Georgetown/South Park Subarea, all minor and principal arterials are designated freight corridors, including E Marginal Way S, 1st Avenue S, and S Michigan Street. Airport Way S is often used as a bypass of I-5 when the interstate is highly congested due to collisions or construction. As noted above, this area has been experiencing an increase in traffic volumes

since March 2020 when the closure of the West Seattle Bridge caused motorists to seek alternate routes. Under typical 2019 conditions, almost all corridors operate at LOS E or better.

Mode Share

The existing SOV mode share in the City of Seattle is summarized by sector using the PSRC Soundcast model and is shown in **Exhibit 3.10-15**. Within the study area, the Duwamish sector has the highest share of PM peak period SOV trips at 53.5%. Magnolia/Queen Anne and Northwest Seattle have lower SOV percentages, as these sectors contain a larger mix of residential and commercial uses.

Sector	Existing SOV Share
Duwamish	53.5%
Magnolia/Queen Anne	43.1%
Magnona Queen Anne	13.170
Northwest	41.6%

Exhibit 3.10-15 Existing SOV Mode Share—PM Peak Period

Source: PSRC, 2021; Fehr & Peers, 2021.

Screenlines

The City's screenline thresholds are in the form of a volume-to-capacity (v/c) ratio: the number of vehicles crossing the screenline compared to the designated capacity of the roadways crossing the screenline. **Exhibit 3.10-16** summarizes the location of the study area screenlines, as well as their LOS threshold and current v/c ratio. All screenline locations are currently under the LOS threshold defined by the City of Seattle.

Exhibit 3.10-16 Existing PM Peak Hour LOS

eenline		Volume-to- Capacity	2019 PM Peak Period v/c Ratio	
Scr	Location	Threshold	N/E	s/W
2	Magnolia	1.0	0.51	0.54
3.11	Duwamish River—West Seattle Bridge and Spokane Street	1.2	0.57	0.53
3.12	Duwamish River—1st Avenue S and 16th Avenue S	1.2	0.54	0.51
4 .13	South City Limit SR 99 to Airport Way S	1.0	0.40	0.45
5.11	Ship Canal Ballard Bridge	1.2	1.01	0.75
5.12	Ship Canal—Fremont Bridge	1.2	0.59	0.66
5.13	Ship Canal—Aurora Bridge	1.2	0.30	0.34
7.11	West of Aurora Avenue—Fremont Place N to N 65th Street	1.0	0.54	0.62

reenline		Volume-to- Capacity	2019 PN Period v	/ Peak / c Ratio
Sci	Location	Threshold	N/E	<mark>S/₩</mark>
8	South of Lake Union	1.2	0.62	0.69
9.12	South of Spokane Street – E Marginal Way to Airport Way S	1.0	0.47	0.48
10.11	South of S Jackson Street Alaskan Way S to 4th Avenue S	1.0	0.58	0.66

Source: City of Seattle count data, 2019; Fehr & Peers, 2021.

Parking

The City of Seattle sets goals and policies related to parking in its Comprehensive Plan. Goals include managing the on-street parking supply to achieve auto trip reduction and improved air quality. In addition, the City recognizes that the primary transportation purpose of the arterial street system is to move people and goods. <u>See the **Freight** section for additional context</u> regarding the importance of truck parking to freight mobility.

The City regulates on-street parking by issuing on-street permits, charging by the hour, setting time limits, and defining loading zones. Some areas of the study area have time-limited paid parking, in effect between 8 AM and 6 or 8 PM, with rates between \$0.50 and \$5 per hour depending on location. Some blocks have free time-limited parking, unrestricted parking, carpool only parking, or freight loading only zones. In some locations in both MICs, parking supply is currently being limited by business operator placement of "ecology blocks" that limit access of City on-street parking to the public.

Conditions in specific subareas are described below. One common trend is that on-street parking tends to be more informal in industrial areas, with the frontage of many parcels lacking curbs or delineated spaces. This type of parking can create obstacles for pedestrians and bicycles. More formal parking configurations are typically implemented as frontage improvements occur.

<u>Ballard</u>

In the Ballard Subarea, most roadways have unrestricted parking. Portions of NW Leary Way and Shilshole Avenue NW have free, time-limited parking. The only paid parking is along streets within a few blocks north and south of NW Market St between 26th Avenue NW and 15th Avenue NW. Parking in the industrial areas tends to be informal, with no curbs or delineated spaces. The Freight Master Plan identifies Ballard as needing additional truck-sized loading zones to support goods delivery.

Interbay Dravus

The Interbay Dravus Subarea has unrestricted parking on most streets within the subarea except for W Commodore Way, Thorndyke Avenue W, and several blocks west of the W Dravus Street Bridge which have time-limited parking. There is no on-street parking permitted on 15th Avenue W north of W Dravus Street. There is no paid parking within the Interbay Dravus Subarea.

Interbay Smith Cove

The Interbay Smith Cove Subarea has unrestricted parking on the residential streets east of 15th Avenue W, and west of the Magnolia Bridge. There are stretches of time-limited parking along portions of 15th Avenue W/Elliott Way. The southbound curb lane is bus only during the AM commute period and the northbound curb lane is bus only during the PM commute period. Outside of those hours, on-street parking is permitted. There is no paid parking within the Interbay Smith Cove Subarea.

SODO/Stadium

Near the stadiums and within the SODO/Stadium Subarea, most streets have time-limited parking. Multiple blocks surrounding the stadiums, and along 1st Avenue S, 4th Avenue S, 6th Avenue S, and Airport Way S do not allow parking. The only on-street paid parking within the subarea is along 1st Avenue S and Occidental Avenue S just west of Lumen Field. The north/south arterials tend to have more formal parking in front of businesses, with curbs and delineated spaces. Along many industrial parcels, parking is more informal as those areas often lack curbs and delineated spaces. <u>Adequate parking for large trucks is a concern in this area as it surrounds the Duwamish MIC.</u>

Georgetown/South Park

In the Georgetown/South Park Subarea, a variety of streets offer time-limited parking; there is no paid parking. Many of the local roadways have no restrictions on parking. On the west side of the Duwamish waterway, West Marginal Way SW does not include on-street parking. Adjacent land uses include off-street parking throughout the corridor. <u>Adequate parking for</u> <u>large trucks is also a concern in this area as well as its potential effects on residents.</u>

Safety

The City has a Vision Zero policy that aims to reduce the number of fatalities and serious injuries to zero by 2030. The Vision Zero program includes a variety of strategies, including reduced speed limits, Safe Routes to Schools investments, safety improvements at high-risk locations, enforcement, and education. In 2019, there were 26 fatalities and 194 serious injuries in the city. Although fatalities on city streets had been on a downward trend, there has been a recent increase, a trend similar to what has been observed nationwide. Of the 26 fatalities resulting from collisions within the city in 2019, three occurred within the study area. These included a pedestrian/bus collision on SR 509, a bicyclist/vehicle collision at Alaskan Way & S

Spokane St, and a vehicle/vehicle collision at Airport Way/Hinds Street. Of the 194 serious injuries in the city, 20 occurred within the study area, with the SODO subarea accounting for just over half.

Modal conflicts between trucks, pedestrians, and bicyclists <u>(or micromobility users such as</u> <u>people riding scooters</u>) are of particular concern given the size and visibility of trucks and the vulnerability of people walking and biking. As documented in the Freight Master Plan, trucks typically represent a higher proportion of fatal collisions than any other type of collision.



Exhibit 3.10-20-17 Modal Conflicts in Industrial Areas

Sources: Seattle Department of Transportation, 2020.

Exhibit 3.10-21 and **Exhibit 3.10-22** are heat maps created using five years (2016-2020) of recent collision data. Within the study area, most fatal and serious injury collisions occur on the major arterials, including Leary Way, 15th Avenue W, 4th Avenue S, E Marginal Way S, and 1st Avenue S. Other hot spots for collisions of all severities include Spokane Street, Edgar Martinez Drive, and Emerson Place.

SDOT also completed a Bicycle and Pedestrian Safety Analysis (BPSA) in 2020 which identified locations that should be prioritized for improvements based on pedestrian and bicycle crash data. Findings related to each subarea are included below.



Exhibit 3.10-21-18 Collisions—Ballard Interbay Northend MIC, 2016-2020

Source: WSDOT, 2016-2020; Fehr & Peers, 2021.



Exhibit 3.10-22-19 Collisions—Greater Duwamish MIC, 2016-2020

Source: WSDOT, 2016-2020; Fehr & Peers, 2021.

Ballard, Interbay Dravus, & Interbay Smith Cove

Within the Ballard study area, collisions are most concentrated along Leary Way, in particular at the intersection of Leary Way NW and 15th Avenue NW. The majority of collisions throughout the Interbay subareas occur along the 15th Avenue NW corridor with multiple collisions near the intersections of the Galer Way Flyover and Magnolia Bridge.

The BPSA identified several locations in the study area as priority areas for improvements: the intersection of W Emerson Place and Gilman Avenue W, several locations along Leary Way, and a large cluster of locations in south Fremont near the waterfront (i.e., the vicinity of the Fremont Bridge and Burke-Gilman Trail).

SODO/Stadium & Georgetown/South Park

In the SODO/Stadium Subarea, collisions are most concentrated along the north/south arterials, including E Marginal Way S, 1st Avenue S, 4th Avenue S, and 6th Avenue S, with the greatest number of fatal and serious injury collisions on 4th Avenue S. The most pronounced "hotspots" are surrounding the intersection of Edgar Martinez Drive and 4th Avenue S, along the I-90/I-5 on-ramps, and along Spokane Street. The SODO area accounted for more than half of the serious injuries and fatalities that occurred within the study area in 2019.

In the Georgetown/South Park Subarea, collisions were most common along the major arterials, including E Marginal Way S and 1st Avenue S. The largest hotspot in the subarea is the intersection of the 1st Avenue S Bridge and E Marginal Way S.

The BPSA identified a substantial number of priority areas for improvements in the Greater Duwamish MIC. Locations including a large cluster in the Chinatown-International District; along 1st Avenue S between Downtown and the West Seattle Bridge; the convergence of Delridge Way, West Marginal Way SW, and the West Seattle Bridge; the SR 509/SR99 interchange area; and the southern end of the South Park Bridge.

3.10.2 Impacts

This section describes the potential impacts of each future year alternative. The impacts of the Action Alternatives are measured against conditions expected under Alternative 1 No Action.

Analysis Methodology & Planning Scenarios Evaluated

Four alternatives are evaluated under future year 2044 conditions: Alternative 1 No Action and three Action Alternatives. Alternative 1 No Action is consistent with the City's current zoning and adopted plans. The Action Alternatives would increase the amount of growth within the study area. A full description of the land use assumptions may be found in **Chapter 2**. All alternatives assume improvements included in current City and regional plans, as shown in **Exhibit 3.10-23** and **Exhibit 3.10-24**. Key projects include the West Seattle and Ballard Link light rail extensions, Waterfront Seattle improvements along Alaskan Way, and an expanded network of bicycle infrastructure.

To develop the future forecasts for this project, Fehr & Peers applied a version of the PSRC <u>regional trip-based travel demand</u> model developed for the West Seattle and Ballard Link Extension (WSBLE) project and the Ballard-Interbay Regional Transportation (BIRT) System project. <u>The model estimates the demand for person and freight travel across a range of travel modes</u>: private automobiles, trucks, transit vehicles, walking, and biking. The truck model defines a truck based on relative weight classes and separates medium and heavy trucks based on the definitions used by WSDOT for collecting truck counts.

This version of the PSRC model is an appropriate tool for this project given its level of detail in the study area (in terms of both land uses and transportation network), assumptions for transit investments, and future land use assumptions that are consistent with growth anticipated through 2042. While the No Action Alternative reflects land uses anticipated through 2042, the potential land use changes under the Action Alternatives extend slightly farther to a 2044 horizon year. This provides a conservative basis to evaluate potential impacts of the Action Alternatives 1 No Action.

The model contains household and employment forecasts consistent with regional assumptions from PSRC and the City's MHA growth distributions. The model also incorporates planned transportation facilities into the model network, such as the Link light rail extensions to Ballard and West Seattle. Note that the Alternative 1 No Action model reflects the current capacities and configurations for the Magnolia Bridge and Ballard Bridge.



Exhibit 3.10-23-20 Planned Transportation Network Improvements—Ballard Interbay Northend MIC, 2044

Source: Fehr & Peers, 2021.



Exhibit 3.10-24-21 Planned Transportation Network Improvements—Greater Duwamish MIC, 2044

Note: Map was updated to include minor revisions to planned improvements.

Source: Fehr & Peers, 202<u>2</u>4.

Thresholds of Significance

This section outlines the thresholds used to determine impacts of Alternative 1 No Action and the Action Alternatives; see **Data & Methods** section for background on these criteria. A transportation impact under *Alternative 1 No Action* is identified if:

- A corridor would have a travel time LOS grade of F.
- A screenline would exceed the threshold stated in the Seattle Comprehensive Plan by at least 0.01.
- A sector would have a percentage of SOV travel exceeding the target stated in the Seattle Comprehensive Plan.
- A transit screenline would have passenger load factor exceeding 1.0.

Potential impacts of Alternative 1 No Action related to active transportation, parking, and safety are discussed qualitatively.

<u>Thresholds of significance were developed based on typical City of Seattle SEPA practices and</u> with the intent of recognizing impacts of a magnitude beyond typical model variation and/or <u>daily observed variation</u>. A transportation impact is identified under an *action alternative* if:

- A study corridor that would operate at an acceptable travel time LOS under Alternative 1 No Action would operate at LOS F or the travel time along a study corridor identified as an impact under Alternative 1 No Action would increase by at least 5%.
- A screenline that would operate acceptably under Alternative 1 No Action would exceed the threshold **or** a screenline that is identified as an impact under Alternative 1 No Action would increase by at least 0.01.
- A sector that would operate acceptably under Alternative 1 No Action would exceed its mode share target or the mode share in a sector that is identified as an impact under Alternative 1 No Action would increase by at least 0.5%.
- A transit screenline that would operate acceptably under Alternative 1 No Action would exceed 1.0 or a transit screenline that is identified as an impact under Alternative 1 No Action would increase by at least 0.05.

Potential impacts of the Action Alternatives related to active transportation, parking, and safety are discussed qualitatively based on the following considerations:

- Active Transportation: A significant impact is identified if an action alternative would preclude planned pedestrian and bicycle investments or increase the number of people walking or biking compared to Alternative 1 No Action in locations with network gaps.
- Parking: A significant impact is identified if an action alternative is expected to result in parking demand exceeding supply for a sustained period and by a substantive amount compared to Alternative 1 No Action.
- Safety: A significant impact is identified if an action alternative is expected to increase the rate of serious and fatal collisions in the study area compared to Alternative 1 No Action.

Impacts Common to All Alternatives

Freight Mobility & Access

Corridor-specific travel time findings for roadway users—including freight—are presented by alternative subsequently in this chapter. Although freight uses the same facilities, traffic congestion is more difficult for large trucks to navigate, and trucks typically travel at slower speeds than general auto traffic. For those freight corridors that are projected to experience increased congestion compared to Alternative 1 No Action, it is also expected that travel time reliability may be affected. Traffic congestion leads to worse reliability on the roadway network as the system is less resilient to recover from disruptions such as blocking incidents; moreover, freight operators have a more difficult time maneuvering in traffic congestion. Therefore, lower LOS on study area roadways would also likely lead to lower reliability in travel times, a particular concern for freight operators.

This programmatic EIS addresses area-wide land use zoning changes, rather than a projectspecific proposal. The proposal may result in a wide range of individual projects implemented over a long timeframe and across a large geographic area. Because the specific locations and sizes of development are unknown at this time, it is not possible to know how freight may be impacted by changes to loading zones or access needs at particular locations. These are potentially significant impacts that would need to be analyzed and mitigated at the project level. See the Parking section for a discussion of potential impacts to overall parking supply and demand, which would affect the availability of truck parking in the study area.

The alternatives under consideration are not expected to materially affect rail operations. The railroads in the study area are privately operated and regularly adjust their operations to respond to changing needs. At-grade rail crossing safety is discussed in the Safety section.

Active Transportation

The City is continually planning and implementing improvements to active transportation facilities through the *Pedestrian Master Plan* (PMP), *Bicycle Master Plan* (BMP), and various subarea planning efforts. The City will soon be developing a citywide transportation plan that will bring together its individual modal plans into a single integrated document.

The PMP includes the identification of a Priority Investment Network (PIN) which designates street segments that should be prioritized for investment. However, the PIN identifies many more locations than can be improved in the near term given the high cost of infrastructure. Given the City's emphasis on prioritizing neighborhoods with historical underinvestment, areas within the Greater Duwamish MIC would likely be prioritized higher than areas in the BINMIC. Among many other factors, the planning process will consider development trends and changes in land use patterns for continued prioritization and phasing of infrastructure improvement projects.

SDOT publishes a BMP Implementation Plan every two years detailing the infrastructure projects that will be constructed over the following four years. Between 2016 and 2020, SDOT completed more than 45 miles of bicycle facility improvements including bike lanes, protected bike lanes, trails, and neighborhood greenways. It is assumed that the City will continue to implement its BMP network under whichever alternative is pursued, though the pace of improvements will vary over time depending on funding availability.

Sound Transit's light rail extensions to Ballard and West Seattle is planned to be complete by 2044, providing frequent, high-capacity service along the Elliott Avenue W/15th Avenue NW corridor in the BINMIC and across the Duwamish Waterway in the Greater Duwamish MIC. These Link extensions would construct three stations within the BINMIC—Ballard, Interbay (in the vicinity of Dravus Street), and Smith Cove—and one new station in Delridge on the western edge of the Greater Duwamish MIC. The Greater Duwamish MIC also includes the existing Stadium and SODO stations. The light rail expansion would include a new station at SODO for the West Seattle Link Extension and potentially relocating the Stadium Station for the Ballard Link Extension (depending on the alternative selected).

It is expected that pedestrian and bicycle activity will continue to increase compared to existing conditions, both due to overall growth in the study area as well as an increasing share of people walking and biking. Therefore, under Alternative 1 No Action, there would be more demand in areas that lack sidewalks, curb ramps, pedestrian crossings, and dedicated bicycle facilities, particularly in industrial areas (as detailed in the Affected Environment section). While many locations in the study area would benefit from improvements to make walking and biking more comfortable, capacity constraints on active transportation facilities are rare throughout the study area and are typically only a concern at network bottlenecks (for example the walkway along the Ballard Bridge) or areas of extremely high pedestrian activity. Specific areas that may experience substantial increases in the number of people walking and biking are discussed in the following sections.

The Action Alternatives are not expected to preclude any planned pedestrian and bicycle improvements and would likely result in improved infrastructure in the areas zoned as Industry & Innovation and Urban Industrial because they would be subject to development standards for pedestrian and cyclist-oriented frontage improvements. However, because the Action Alternatives would result in higher levels of growth than Alternative 1 No Action, there would likely be more people walking and biking in areas with existing network gaps. Therefore, a significant impact to pedestrian and bicycle travel is identified under alternatives 2, 3, and 4<u>and the Preferred Alternative</u>.

Ballard, Interbay Dravus, & Interbay Smith Cove

The BMP includes a variety of planned projects in the vicinity of the BINMIC including completion of the Burke-Gilman Trail "missing link" and multi-use trails on the Ballard Bridge and West Galer Street Flyover. Bicycle network connections are also recommended between

the Ballard Locks and the Ship Canal Trail, 20th Avenue W between Thorndyke Avenue W and the Elliott Bay Trail, and across the W Dravus Street bridge, among other areas.

Pedestrian and bicycle activity would increase substantially in the vicinity of the planned light rail stations as all riders would access the stations by walking, biking, transit, or pickup/dropoff (no on-site parking is being provided). Among the new stations, the highest numbers of people walking and biking would occur near the Ballard station as walking and biking access are more limited near the Interbay and Smith Cove stations due to topography, connectivity, and surrounding land uses. Key connections to the stations would include the Elliott Bay Trail, Ship Canal Trail, Magnolia Connector Trail, West Galer Street Flyover, and Helix pedestrian bridge depending on which station options are selected.

Based on the proposed development standards, the areas that would be zoned as Industry & Innovation and Urban Industrial would be the most likely to see substantial increases in people walking and biking. Within the Ballard, Interbay Dravus, and Interbay Smith Cove areas, the Action Alternatives would implement those development standards along the north side of Lake Union, slightly inland areas of Fremont and Ballard, and along the 15th Avenue W/Elliott Avenue W corridor.

SODO/Stadium & Georgetown

Among other projects, the BMP recommends improvements such as an extension of the SODO Trail south to Georgetown, a combination of protected bike lanes and a multi-use trail along E Marginal Way S between S Spokane Street and Ellis Avenue S, and extending the Duwamish River Trail from its current northern terminus to the West Seattle Bridge.

While the SODO and Stadium stations are already hubs of pedestrian and bicycle activity, the growing ridership with Sound Transit's system expansion will also increase the number of people walking and biking in the immediate vicinity of the stations. In contrast, the Delridge station will be a new hub of activity as the neighborhood is currently primarily residential. No on-site parking is being provided so all riders will access the stations by walking, biking, transit, or pickup/dropoff.

Based on the proposed development standards, the areas that would be zoned as Industry & Innovation, Urban Industrial, and Seattle Mixed would be the most likely to see substantial increases in people walking and biking. Within the SODO/Stadium and Georgetown/South Park subareas, the Action Alternatives would implement those development standards in areas of SODO north of S Lander Street, Georgetown, South Park, and north Delridge.

Parking

The City prioritizes the use of its public right-of-way to balance competing needs, including people walking, biking, taking transit, and driving whether for personal travel or for goods movement. The "flex zone" along the curb may be used for parking, bus stops, passenger loading, freight loading, travel lanes during peak times or other activating uses such as parklets

or play streets. Decisions on the use of the flex zone will continue to be made by the City as the context evolves throughout the study area. While the use of the flex zone will vary by location, it is unlikely that the overall supply of on-street parking in any subarea would increase under any of the alternatives. Industrial areas may be more likely to see changes in parking supply as redevelopment triggers frontage improvements such as adding curbs and delineating parking spaces in rights-of-way that were previously used for informal parking.

While parking demand varies throughout the study area, there are some localized areas where on-street parking demand exceeds parking supply<u>, particularly demand for truck parking given</u> the industrial nature of the MICs. Given projected growth throughout the city and that on-street parking is unlikely to increase in the future, a parking impact is expected under Alternative 1 No Action. With the increase in development expected under the Action Alternatives, parking demand would be higher than Alternative 1 No Action. Because the Action Alternatives are expected to increase demand in localized areas that already exceed supply, potentially for a sustained period and by a substantive amount compared to Alternative 1 No Action, significant adverse parking impacts are expected under all of the Action Alternatives. Impacts are expected to be greater under alternatives 3 and 4, which have higher levels of development planned than Alternative 2 or the Preferred Alternative.

The location and severity of impacts would vary by alternative depending on the concentrations of land use. The degree of parking supply impacts experienced in any given neighborhood would depend on many factors, including how much off-street parking is provided by future development projects, as well as varying conditions related to on-street parking patterns <u>for</u> <u>both freight trucks and passenger vehicles</u> and City regulations (e.g., pricing, enforcement, RPZ permits, etc.) within each neighborhood.

Ballard, Interbay Dravus, & Interbay Smith Cove

The Ballard, Interbay Dravus, and Interbay Smith Cove subareas will experience a substantial transportation change with the construction of new light rail stations for the Ballard Link light rail extension. The flex zones in the immediate vicinities of the stations (specific locations are still to be determined) are likely to experience changes as they will accommodate bus stops, pickup/dropoff areas, and potentially bus layover areas. None of the stations are planned to include parking facilities. While Sound Transit is planning for most access to occur via transit, walking, biking, and pickup/dropoff, some riders may drive to the station and seek parking in nearby areas, increasing baseline parking demand within walking distance of each station.

The City has already developed approaches to manage this type of "hide and ride" parking demand at new light rail stations, such as the U District and Roosevelt. It is assumed similar efforts will be made for the new light rail stations in the study area. The City's management strategies include on-street parking surveys before station openings to identify and implement appropriate mitigation elements prior to station opening; mitigation measures such as paid parking meters, time-limit signs, passenger drop-off/pick-up zones, truck and load/unload zones, and residential parking zones (RPZs) within a 1/4-mile radius of each station; ongoing

monitoring of parking controls after the system opens to determine if RPZ boundaries or other on-street controls are insufficient; and parking enforcement.

Based on the proposed development standards and locations, the areas that would be zoned as Industry & Innovation and Urban Industrial would be the most likely to see increases in parking demand that exceed parking supply. Within the Ballard, Interbay Dravus, and Interbay Smith Cove areas, the Action Alternatives would implement those development standards along the north side of Lake Union, areas of Fremont and Ballard, and along the 15th Avenue W/Elliott Avenue W corridor.

SODO/Stadium & Georgetown

The SODO/Stadium area includes two existing light rail stations which will have ridership increases with the expansion of the light rail system. Improvements at these stations may result in changes to the flex zones in the immediate vicinity but they are likely to be minor. The new Delridge station would be located near the edge of the study area and would result in changes to the flex zones in the immediate vicinity to accommodate uses such as bus stops and pickup/dropoff areas. No parking facility is planned for the station area. While Sound Transit is planning for most access to occur via transit, walking, biking, and pickup/dropoff, it is possible some riders may drive to the station and seek parking in nearby areas, increasing baseline parking demand within walking distance of each station. The management approaches described above for the Ballard Link light rail extension would also likely be in place for the station areas within the Greater Duwamish MIC.

Based on the proposed development standards and locations, the areas that would be zoned as Industry & Innovation, Urban Industrial, and Seattle Mixed would be the most likely to see increases in parking demand that exceed available supply. Within the SODO/Stadium and Georgetown/South Park subareas, the Action Alternatives would implement those development standards in areas of SODO north of S Lander Street, Georgetown, South Park, and north Delridge.

Safety

The City has a Vision Zero policy that aims to reduce the number of fatalities and serious injuries to zero by 2030. This goal, and the policies and strategies supporting it, will continue to be pursued under whichever land use alternative is selected. Recent examples of policy changes include widespread reductions in speed limits along city streets and the introduction of leading pedestrian intervals to make people walking more visible to vehicles (timing signals to give people walking a head start before the vehicles receive a green light). SDOT also regularly studies intersections and corridors that have been identified as needing safety improvements by the community or through collision data review. The types of location-specific measures that can be implemented depending on the context include traffic calming treatments, new traffic signals, separation of facilities for vulnerable users, and hardened centerlines (small rubber barrier that require drivers making left turns to slow down and make

squarer left turns). The City will continue to monitor traffic safety and take any necessary steps to address areas of high need particularly for the most vulnerable users. Over time, it is expected that the safety program will result in decreases to the number of traffic fatalities and serious injuries.

The Action Alternatives are expected to result in between 0.8 to 2.5% more vehicle miles traveled than Alternative 1 No Action in the Greater Duwamish MIC area and roughly 1.4 to 5.1% more vehicle miles travelled than Alternative 1 No Action in the BINMIC area. In terms of relative exposure among the alternatives, alternatives 3 and 4 are expected to have more substantive VMT increases in the Duwamish MIC area: 2.3% and 2.5% increases for Alternative 3 and 4, respectively compared to 0.8% and 1.0% for Alternative 2 and the Preferred Alternative, respectively. Likewise, VMT increases in the BINMIC area would be 4.3% and 5.1% under alternatives 3 and 4, respectively, compared to more limited increases of 1.4% and 2.5% under Alternative 2 and the Preferred Alternative.

This could potentially lead to an increase in the number of collisions. In addition, the Action Alternatives may result in an increased number of truck and vehicle conflicts with vulnerable users such as people walking<u>, and</u> biking<u>, and rolling</u> in industrial areas, as outlined in the 2020 Bicycle and Pedestrian Safety Analysis. <u>Risks to vulnerable users are heightened in areas with large truck activity which is inherent to industrial operations. Truck drivers have a limited range of sight distance and often encounter turning radii conflicts that are not expected from smaller <u>vehicles</u>. Because trucks represent a higher proportion of fatal collisions than any other type of collision (as documented in the *Freight Master Plan*), it is reasonably likely that the Action Alternatives could result in an increased rate of serious and/or fatal collisions in the study area. Likewise, the Action Alternatives would result in more vulnerable users in areas with routine atgrade train crossings, which has the potential to lead to an increase in the number of rail collisions.</u>

Due to the potential increase of collisions between trucks/vehicles/<u>rail</u> and vulnerable users, a significant impact is expected under the Action Alternatives. Site-specific issues cannot be addressed at this level of analysis. However, individual development applications would be reviewed through the City's permitting process, at which time the City may identify required safety features for the specific site.

Pavement Condition

As noted above, the Action Alternatives are expected to result in an increased number of vehicle miles traveled in the Greater Duwamish MIC and BINMIC, areas of which already have worn pavement condition due to regular heavy vehicle use. While increased use of these roadways may incrementally degrade pavement condition further, vehicles are subject to gas taxes and weight-based license fees that can be directed toward more frequent maintenance of facilities. Therefore, while the Action Alternatives may cause some impact to roadway pavement condition, it is not expected to rise to a level of significance.

State Highway System

<u>Per WSDOT Design Manual Chapter 1130.09(2)(a), published in September 2021, WSDOT</u> <u>considers any proposal that meets or exceeds either or both of the following thresholds to</u> <u>have a probable significant adverse impact to the state highway system:</u>

- Addition of 10 or more AM or PM peak hour vehicle trips assigned to an individual approach leg to a state highway intersection.
- Addition of 25 or more AM or PM peak hour vehicle trips assigned to a state highway segment (two-way travel) or intersection (total 25 trips all legs).

<u>Changes in traffic volumes in the regional travel demand model were reviewed across the state</u> <u>highway system in the study area vicinity. It was determined it was likely that the cumulative</u> <u>development proposed by each Action Alternative would exceed one or both of the above</u> <u>thresholds on I-5, SR 99, SR 509, and SR 599 throughout the city with alternatives 3 and 4</u> <u>having higher increases than Alternative 2 and the Preferred Alternative. This is not an impact</u> <u>threshold currently adopted by the City of Seattle. It is a potential impact with respect to</u> <u>consistency with the WSDOT Design Manual. The City of Seattle will coordinate further with</u> <u>WSDOT regarding the individual project review process to determine impacts to the state</u> <u>highway system of any specific development proposal.</u>

Equity & Environmental Justice Considerations

The City of Seattle developed a Racial and Social Equity Index that combines data on race, ethnicity, and socioeconomic and health disadvantages to identify neighborhoods with large proportions of priority populations as residents. Much of the SODO/Stadium Subarea as well as the South Park neighborhood were found to have among the highest disadvantages in the city.

The Action Alternatives—particularly alternatives 3 and 4—would result in more land use growth compared to Alternative 1 No Action particularly in the SODO/Stadium and South Park neighborhoods. With respect to transportation, this growth could provide both beneficial and adverse impacts to equity and environmental justice. Additional growth would bring increased traffic volumes, which in turn may bring impacts to the safety of people walking and biking, parking availability, and travel time delays to areas with high proportions of priority populations. At the same time, increased development could also bring improved infrastructure to neighborhoods with histories of long-term underinvestment. This is particularly the case for areas that would be rezoned as Industry & Innovation and Urban Industrial because those land use concepts would have development standards requiring frontage improvements such as sidewalks, pedestrian lighting, and street trees—all of which could be beneficial in progress toward more safe, connected, and accessible neighborhoods.

Impacts of Alternative 1 No Action

This section summarizes analysis results and environmental impacts of Alternative 1 No Action. Alternative 1 No Action serves as the baseline for the impact analysis of the Action Alternatives. It represents the operation of the transportation system if no zoning or network changes were made in the study area. However, growth would continue to occur under Alternative 1 No Action consistent with current adopted zoning. Alternative 1 No Action is expected to result in roughly 23,500 additional jobs in the study area compared to existing conditions. Residential development would be very minor—approximately 75 new dwellings over the study area. For both employment and residential uses, the growth is expected to be highest in the SODO/Stadium and Georgetown/South Park subareas.

Exhibit 3.10-25 summarizes the number of person trips expected to be generated during the PM peak hour by the land uses in the study area by mode of travel.

Mode	Alternative 1 No Action		
SOV (in passenger vehicles)	35,400	<u>40.5%</u>	
HOV (in passenger vehicles)	32,800	<u>37.5%</u>	
Transit	<u>4,800</u>	<u>5.5%</u>	
Walk	12,400	<u>14.2%</u>	
Bike	<u>2,100</u>	<u>2.4%</u>	
Total	<u>87,500</u>	<u>100%</u>	

Exhibit 3.10-25	2044 Alternative 1 No Action Person Tri	ps in Stud ^y	y Area—PM Peak Hour

Source: Fehr & Peers, 2022.

Transit

As noted in the Active Transportation section, the study area would experience a fundamental change in transit service by 2044. Sound Transit's light rail extensions to Ballard and West Seattle would be complete, providing frequent, high-capacity service along the Elliott Avenue W/15th Avenue NW corridor in the BINMIC and across the Duwamish Waterway in the Greater Duwamish MIC. In addition to these routes directly affecting the study area, Sound Transit's light rail system would also include extensions north to Everett, east to Redmond and Issaquah, and south to the Tacoma Dome. Fixed route bus service would be restructured, where appropriate, to better connect surrounding neighborhoods to light rail stations and have fewer routes running into the downtown core.

For the existing conditions evaluation, there is extremely granular data available identifying the maximum load that occurs along an entire route for every trip in the peak period. For the future conditions evaluation, transit load factors are estimated using average passenger loads, as that is the metric available from traditional travel demand modeling tools. On average

across the studied routes, the maximum passenger load is approximately 78% higher than the average passenger load, though typically only for a short segment of the transit route. To reflect an appropriately conservative capacity against which *average* passenger loads should be measured, Sound Transit's planning load of 148 passengers per car is used as the light rail capacity and the number of seats on each bus is used as the bus capacity. In other words, both types of transit vehicles are able to accommodate higher capacities than are used for this analysis. The forecasted passenger loads for Alternative 1 No Action are consistent with Sound Transit's ongoing planning for the West Seattle and Ballard Link Extensions.

Exhibit 3.10-22 summarizes the average PM peak hour passenger load factor for the transit routes that would operate along key corridors in the study area. The passenger load factors include both light rail and bus services.

Exhibit 3.10-22 PM Peak Hour Passenger Load Factors—Alternative 1 No Action

Note: Inbound refers to travel into the downtown area and outbound travel out of the downtown area. Source: King County Metro, Fall 2018; Fehr & Peers, 2021.

As is the case under current conditions, outbound transit (i.e., routes leaving Downtown or the U District) are expected to have higher passenger loads than inbound routes during the PM peak hour with the inverse true during the AM peak hour. While some of the routes traveling across the study area screenlines may operate over their crowding threshold for some individual trips, the load factors indicate that there would be adequate transit capacity across most of the transit screenlines. The exception is the outbound direction across 8th Avenue NW (from the U District to Ballard) which is expected to have a passenger load impact under Alternative 1 No Action. Specific routes are discussed below.

Ballard, Interbay Dravus, & Interbay Smith Cove

The Ballard Link Extension would construct three stations within the BINMIC: Ballard, Interbay (in the vicinity of Dravus Street), and Smith Cove. With trips running approximately every five minutes and each trip able to comfortably carry nearly 600 riders,¹⁵ transit capacity along the corridor would dramatically increase compared to existing conditions. This is reflected in the average outbound passenger load factor of 0.39 across the Ballard Bridge and 0.59 north of W Mercer Place which indicate that transit demand would be accommodated by the planned capacity.

The screenline east of 8th Avenue NW shows a different trend as it includes east-west bus service between Ballard and the U District which would not be replaced by high-capacity transit. Demand across that screenline is expected to grow in the future with average passenger loads exceeding seated capacity indicating some passengers would need to stand. In practice, King

⁴⁵ This assumes four-car trains at Sound Transit's planning load of 148 passengers per car.

County Metro continually reallocates resources based on demand and it is likely that frequencies would be increased if necessary for those crosstown routes to alleviate crowding.

SODO/Stadium & Georgetown/South Park

Both the Ballard-Tacoma and West Seattle-Everett Link lines, along with multiple bus routes, would cross the screenline north of Lander Street. With the enhanced capacity, average passenger loads are expected to be roughly 0.75 indicating most trips would operate within the planned capacity.

The majority of transit riders crossing the West Seattle Bridge would use the new Link light rail extension which is expected to run roughly every six minutes. With the large increase in capacity, passenger loads are expected to be well within planned capacity, at 0.35 for an average passenger load during the PM peak hour.

Roadway UsersAuto & Freight

Under Alternative 1 No Action, growth would continue throughout the city and region, resulting in increases in traffic volumes, including passenger vehicles and trucks. However, traffic volume growth rates within the study area are expected to be relatively low given that many facilities already operate with congestion during peak periods and new high-capacity transit options would be available, making non-auto modes increasingly competitive. This is consistent with traffic growth patterns over the past decade. According to SDOT's 2020 Traffic Report, average daily traffic volumes remained essentially flat over the 2009-

Magnolia Bridge Replacement

All alternatives assume that the Magnolia Bridge retains the same fundamental configuration and connections as exist today. However, the City is also considering an option that would instead replace the Magnolia Bridge with a new bridge along Armory Way connecting to Thorndyke Avenue W at W Halladay Street. Refer to the BIRT Report for more information.

2019 period despite a 24% increase in the City's population and a 23% increase in regional employment.¹⁶

<u>Travel Time</u>

Using the HCM guidelines for defining LOS thresholds as described in the Data & Methods section, **Exhibit 3.10-26** summarizes travel time conditions along each of the study corridors under the No Action Alternative. Travel times for 2019 are also shown to illustrate change over time under Alternative 1 No Action. Note that these results also represent indicate relative effects on both auto traffic and freight operations which travel in the same lanes as auto traffic.

¹⁶ Seattle Department of Transportation, 2020 Traffic Report. Available at:

https://www.seattle.gov/Documents/Departments/SDOT/About/DocumentLibrary/Reports/2020_Traffic_Report.pdf. Accessed September 7, 2021.

Although freight uses the same facilities, traffic congestion is more difficult for large trucks to navigate, and trucks typically travel at slower speeds than general auto traffic. While the actual travel times for large trucks may be higher, the magnitude of change is still reflective of how conditions will vary across alternatives. The travel times below are rounded to the nearest half minute.

<u>Travel time reliability is also a key concern for travelers, particularly for freight operators. Traffic congestion leads to worse reliability on the roadway network as the system is less resilient to recover from disruptions such as blocking incidents. Therefore, lower LOS on study area roadways would also likely lead to lower reliability in travel times</u>

		PM Peak Hour LOS / Travel Time (minutes)			inutes)
		Existing (Conditions	Alternative	1 No Action
ID	 Corridor	N/E	S/W	N/E	S/W
Ballar	d Interbay Northend MIC				
1	15th Ave W from Magnolia Bridge to NW Leary Way	E / 11.5	A / 4.5	E / 12.5	A / 5
2	15th Ave NW from NW Leary Way to N 85th St	E/9.5	C / 6.5	E/9.5	C / 6.5
3	Leary Ave NW/ Leary Way NW/ N 36th S/ Fremont Bridge between NW Market St and W Nickerson St/Westlake Ave	C / 11	C / 12	C / 11	D/13
4	Shilshole Ave NW between NW Market and 15th Ave NW	B / 2.5	D / 3.5	B / 2.5	D / 4
5	NW Market St/N 50th/ 46th St between 24th Ave NW and I-5	C / 14	D / 16.5	C / 14	D / 16.5
6	W Nickerson St/Westlake Ave N between 15th Ave W and Mercer St	C / 13	C / 11.5	C / 13.5	C / 12.5
7	W Dravus St between 15th Ave W and 20th Ave W	E / 2	D / 1.5	E / 2	D/1.5
8	Elliott Ave W between Magnolia Bridge and Wall St	B / 5.5	B / 6	B / 6	B / 6.5
9	W Mercer St from Elliott Ave W to I-5	F / 32	F / 22	F/32	F / 22.5
10	Denny Way from Elliott Ave W to I-5	F/14.5	E / 11	F/15	F/11.5
11	Magnolia Bridge between 15th Ave W and Thorndyke Ave W	B / 2.5	A / 1.5	B / 2.5	A / 1.5
12	SR 99 between N 46th St and Denny Way ¹	C / 8.5	E/14	D/10.5	F/14.5
13	W Emerson St between 15th Ave W and Gilman Ave W	F / 6	F / 4	F/6	F/4
14	N 85th St between 15th Ave NW and I-5	E / 13	E/14.5	E / 13.5	E/14.5
15	I-5 between N 85th Street and Madison Street ¹	F/19	F/24	F/22.5	F/26
Great	er Duwamish MIC				
1	1st Ave S between S Royal Brougham Way and SR 99	C / 11	C / 11	C / 11	C / 12
2	4th Ave S between Seattle Blvd S to E Marginal Way S	C / 12	C / 12.5	C / 12.5	C / 13.5

Exhibit 3.10-26-23 PM Peak Hour Travel Time LOS—Alternative 1 No Action

		PM Peak Hour LOS / Travel Time (minutes)			inutes)
		Existing C	Conditions	Alternative	1 No Action
ID	– Corridor	N/E	S/W	N/E	S/W
3	6th Ave S between Seattle Blvd S to Spokane St Viaduct	C / 6.5	C / 6	C / 6.5	C / 6.5
4	Airport Way S/ Seattle Blvd S between S Royal Brougham Way to S Boeing Access Rd	A / 16.5	A / 15.5	A / 16.5	A / 16
5	West Seattle Bridge/Spokane St Viaduct between 35th Ave SW and I-5	C / 6.5	E/10	C / 6.5	E/10
6	Spokane St Bridge between Harbor Ave SW and SR 99	B / 4.5	B / 4.5	B / 4.5	B / 4.5
7	E Marginal Way S between SR 99 and S Boeing Access Rd	C / 8.5	D / 10.5	C / 8.5	D / 10.5
8	Alaskan Way S from Broad St to SR 99	D/9	F/13	E/10.5	F/14.5
9	S Royal Brougham Way between SR 99 and Airport Way S	F/4.5	D/3	F / 4.5	D/3
10	Edgar Martinez Dr S between SR 99 and 4th Ave	F/2.5	F / 2.5	F/3	F / 2.5
11	S Holgate St between 1st Ave and Airport Way S	D/3	F/4.5	D/3	F / 4.5
12	S Lander St between 1st Ave and Airport Way S	E / 4	E / 4	E / 4	E / 4
13	S Lucile St between SR 99 and Airport Way S	D / 4	E / 5	D / 4	E / 5
14	W Marginal Way SW between West Seattle Bridge and 2nd Ave SW	A / 5	A / 4.5	A / 5	A / 5
15	S Michigan St/ Corson Ave S between E Marginal Way S and I-5	C / 3.5	E / 5.5	C / 3.5	E / 5.5
16	E Marginal Way S/SR 99 between S Atlantic Street and 1st Ave S Bridge	A / 9	A/9	A / 9	A / 9
17	I-5 between Madison Street and SR 599 ¹	E / 25.5	F/30	F / 27.5	F/31
<u>18</u>	SR 509 between SR 99 and SR 518 ¹	<u>A / 6.0</u>	<u>C /9.0</u>	<u>A / 6.0</u>	<u>D / 11.5</u>
<u>19</u>	SR 99/599 between SR 509 and I-5 ²	<u>A / 6.0</u>	<u>B / 8.0</u>	<u>A / 6.5</u>	<u>B / 8.0</u>

Note: Cells shown in bold indicate an impact.

1. WSDOT sets a LOS D standard on I-5, SR 509, and SR 99 north of SR 509.

2. WSDOT sets a LOS E standard on SR 99/599 between SR 509 and I-5.

Sources: Wejo, 2019; Fehr & Peers, 2021.

Overall, travel times along the study corridors are expected to remain relatively consistent between 2019 and 2044. During the PM peak hour under the 2044 Alternative 1 No Action, most corridors would continue to operate at LOS E or better in both directions with travel time increases of up to two minutes (the exception is northbound I-5 from Madison Street to N 85th Street with an increase of 3.5 minutes). Corridors operating at LOS F in both existing and 2044 Alternative 1 No Action, constituting an impact to auto and freight, include:

- Both directions of W Mercer Street from Elliott Avenue W to I-5
- Eastbound Denny Way from Elliott Avenue W to I-5
- Both directions of W Emerson Street from Gilman Avenue W to 15th Avenue W
- Both directions of I-5 between N 85th Street and Madison Street
- Southbound Alaskan Way S from Broad Street to SR 99
- Southbound Alaskan Way S from Broad Street to SR 99
- Eastbound S Royal Brougham Way between SR 99 and Airport Way S
- Both directions of Edgar Martinez Drive S between SR 99 and 4th Avenue
- Westbound S Holgate Street from Airport Way S to 1st Avenue
- Southbound I-5 from Madison Street to SR 599

In addition to these corridors, the following corridors that operated at LOS E or better under existing conditions would operate at LOS F under 2044 Alternative 1 No Action, constituting an impact to auto and freight:

- Westbound Denny Way from Elliott Avenue W to I-5
- Southbound SR 99 between N 46th Street and Denny Way
- Northbound I-5 from SR 599 to Madison Street

The following corridors are expected to have the largest increases in travel times, but would still operate at LOS E or better:

- Northbound SR 99 between N 46th Street and Denny Way
- Northbound Alaskan Way between SR 99 and Broad Street

Buses that operate on the impacted corridors would also experience the same travel time conditions.

Mode Share

The Alternative 1 No Action SOV mode share in the City of Seattle is summarized by sector using the PSRC model and is shown in **Exhibit 3.10-27**. The model predicts that SOV mode shares would decrease by 2044, with changes ranging from one to three percent depending on the sector. The smallest decrease is expected in the Duwamish sector while the Magnolia/Queen Anne sector would experience the largest decrease of drive-alone trips. Although all three sectors are expected to have lower SOV shares under the 2044 Alternative 1 No Action scenario than existing conditions, they are still expected to be two to three percentage points above the City's 2035 SOV targets. Therefore, there are expected to be mode share impacts for all three study area sectors under Alternative 1 No Action.

Sector	2035 SOV Target	Existing SOV Share	Alternative 1 No Action SOV Share
Duwamish	51%	53.5%	52.6%
Magnolia/Queen Anne	38%	43.1%	40.1%
Northwest	37%	41.6%	39.7%

Exhibit 3.10-27-24 2044 Alternative 1 No Action SOV Mode Share—PM Peak Period

Note: Cells shown in bold indicate an impact. Source: PSRC, 2021; Fehr & Peers, 2021.

Screenlines

The City's screenline thresholds are in the form of a volume-to-capacity (v/c) ratio: the number of vehicles crossing the screenline compared to the designated capacity of the roadways crossing the screenline. **Exhibit 3.10-28** summarizes the projected PM peak hour v/c ratios across each screenline in 2044. All screenline locations are forecasted to be under the LOS threshold defined by the City of Seattle, therefore no screenline impacts are expected under Alternative 1 No Action. There are no substantial capacity projects planned for construction within the study area between the existing and 2044 forecast year, so all changes in v/c ratios are due to traffic volume increases.

Within the study area, the greatest v/c ratio increases are seen at the South City Limit screenline, the Ballard Bridge, the Fremont Bridge, and south of Jackson Street. The Ballard Bridge screenline is the closest to reaching the City's screenline threshold.

ne			Р	M Peak Peri	od v/c Rat	io
eenli	via		Existing C	onditions	Alt. 1 N	o Action
Scre	Location	Threshold	N/E	S/W	N/E	S/W
2	Magnolia	1.0	0.51	0.54	0.51	0.54
3.11	Duwamish River—West Seattle Bridge and Spokane Street	1.2	0.57	0.53	0.57	0.53
3.12	Duwamish River—1st Avenue S and 16th Avenue S	1.2	0.54	0.51	0.53	0.52
4.13	South City Limit—SR 99 to Airport Way S	1.0	0.40	0.45	0.47	0.50
5.11	Ship Canal—Ballard Bridge	1.2	1.01	0.75	1.11	0.78
5.12	Ship Canal—Fremont Bridge	1.2	0.59	0.66	0.68	0.68
5.13	Ship Canal—Aurora Bridge	1.2	0.30	0.34	0.35	0.35
7.11	West of Aurora Avenue—Fremont Place N to N 65th Street	1.0	0.54	0.62	0.55	0.64
8	South of Lake Union	1.2	0.62	0.69	0.43	0.51
9.12	South of Spokane Street—E Marginal Way to Airport Way S	1.0	0.47	0.48	0.51	0.49
10.11	South of S Jackson Street—Alaskan Way S to 4th Avenue S	1.0	0.58	0.66	0.65	0.68

Exhibit 3.10-28-25 Screenline Volume-to-Capacity Ratio—Alternative 1 No Action

Source: City of Seattle count data, 2019; Fehr & Peers, 2021.

<u>Transit</u>

As noted in the Active Transportation section, the study area would experience a fundamental change in transit service by 2044. Sound Transit's light rail extensions to Ballard and West Seattle would be complete, providing frequent, high-capacity service along the Elliott Avenue W/15th Avenue NW corridor in the BINMIC and across the Duwamish Waterway in the Greater Duwamish MIC. In addition to these routes directly affecting the study area, Sound Transit's light rail system would also include extensions north to Everett, east to Redmond and Issaquah, and south to the Tacoma Dome. Fixed route bus service would be restructured, where appropriate, to better connect surrounding neighborhoods to light rail stations and have fewer routes running into the downtown core.

For the existing conditions evaluation, there is extremely granular data available identifying the maximum load that occurs along an entire route for every trip in the peak period. For the future conditions evaluation, transit load factors are estimated using average passenger loads, as that is the metric available from traditional travel demand modeling tools. On average across the studied routes, the maximum passenger load is approximately 78% higher than the average passenger load, though typically only for a short segment of the transit route. To

reflect an appropriately conservative capacity against which *average* passenger loads should be measured, Sound Transit's planning load of 148 passengers per car is used as the light rail capacity and the number of seats on each bus is used as the bus capacity. In other words, both types of transit vehicles are able to accommodate higher capacities than are used for this analysis. The forecasted passenger loads for Alternative 1 No Action are consistent with Sound Transit's ongoing planning for the West Seattle and Ballard Link Extensions.

Exhibit 3.10-29 summarizes the average PM peak hour passenger load factor for the transit routes that would operate along key corridors in the study area. The passenger load factors include both light rail and bus services.

	Alternative 1 No Action—Average Passenger Load Factor				
<u>Screenline</u>	Inbound	<u>Outbound</u>			
A: East of 8th Avenue NW	0.57	<u>1.28</u>			
<u>B: Ballard Bridge</u>	<u>0.09</u>	<u>0.39</u>			
C: North of W Mercer Place	<u>0.29</u>	<u>0.59</u>			
<u>D: North of Lander St</u>	<u>0.21</u>	<u>0.75</u>			
<u>E: West Seattle Bridge</u>	<u>0.12</u>	<u>0.35</u>			

Exhibit 3.10-29 PM Peak Hour Passenger Load Factors—Alternative 1 No Action

Note: Inbound refers to travel into the downtown area and outbound travel out of the downtown area. Source: King County Metro, Fall 2018; Fehr & Peers, 2021.

As is the case under current conditions, outbound transit (i.e., routes leaving Downtown or the U District) are expected to have higher passenger loads than inbound routes during the PM peak hour with the inverse true during the AM peak hour. While some of the routes traveling across the study area screenlines may operate over their crowding threshold for some individual trips, the load factors indicate that there would be adequate transit capacity across most of the transit screenlines. The exception is the outbound direction across 8th Avenue NW (from the U District to Ballard) which is expected to have a passenger load impact under Alternative 1 No Action. Specific routes are discussed below.

Ballard, Interbay Dravus, & Interbay Smith Cove

The Ballard Link Extension would construct three stations within the BINMIC: Ballard, Interbay (in the vicinity of Dravus Street), and Smith Cove. According to the WSBLE Draft EIS, the following levels of activity are expected at the light rail stations in the area during the PM peak hour: approximately 3,400 people boarding or alighting at the Ballard Station, 1,100 people boarding or alighting at the Interbay Station, and 700 people boarding or alighting at the Smith Cove Station. With trips running approximately every five minutes and each trip able to

comfortably carry nearly 600 riders,¹⁷ transit capacity along the corridor would dramatically increase compared to existing conditions. This is reflected in the average outbound passenger load factor of 0.39 across the Ballard Bridge and 0.59 north of W Mercer Place which indicate that transit demand would be accommodated by the planned capacity.

The screenline east of 8th Avenue NW shows a different trend as it includes east-west bus service between Ballard and the U District which would not be replaced by high-capacity transit. Demand across that screenline is expected to grow in the future with average passenger loads exceeding seated capacity indicating some passengers would need to stand. In practice, King County Metro continually reallocates resources based on demand and it is likely that frequencies would be increased if necessary for those crosstown routes to alleviate crowding.

SODO/Stadium & Georgetown/South Park

Both the Ballard-Tacoma and West Seattle-Everett Link lines, along with multiple bus routes, would cross the screenline north of Lander Street. With the enhanced capacity, average passenger loads are expected to be roughly 0.75 indicating most trips would operate within the planned capacity.

The majority of transit riders crossing the West Seattle Bridge would use the new Link light rail extension which is expected to run roughly every six minutes. According to the WSBLE Draft EIS, the following levels of activity are expected at the light rail stations in the area during the PM peak hour: approximately 3,700 people boarding or alighting at the SODO Station, 700 people boarding or alighting at the Stadium Station, and 8,100-9,300 people boarding or alighting at the CID Station. With the large increase in capacity, passenger loads are expected to be well within planned capacity, at 0.35 for an average passenger load during the PM peak hour.

Impacts of Alternative 2

This section summarizes analysis results and environmental impacts for Alternative 2 (Future of Industry—Limited) in 2044. Compared to Alternative 1 No Action, Alternative 2 would result in 10,900 additional jobs and residential growth would remain essentially flat. As with Alternative 1 No Action, most of the new growth would be concentrated in the Greater Duwamish MIC.

Exhibit 3.10-30 summarizes the number of person trips expected to be generated during the PM peak hour by the land uses in the study area by mode of travel.

¹⁷ This assumes four-car trains at Sound Transit's planning load of 148 passengers per car.

Mode	<u>Alternative</u>	ternative 1 No Action Alternative 2		ative 2
SOV (in passenger vehicles)	<u>35,400</u>	<u>40.5%</u>	<u>36,500</u>	<u>40.7%</u>
HOV (in passenger vehicles)	<u>32,800</u>	<u>37.5%</u>	<u>33,600</u>	<u>37.5%</u>
Transit	<u>4,800</u>	<u>5.5%</u>	<u>4,900</u>	<u>5.5%</u>
<u>Walk</u>	<u>12,400</u>	<u>14.2%</u>	<u>12,500</u>	<u>14.0%</u>
Bike	<u>2,100</u>	<u>2.4%</u>	<u>2,100</u>	<u>2.3%</u>
Total	87,500	<u>100%</u>	<u>89,600</u>	<u>100%</u>

Exhibit 3.10-30 2044 Alternative 2 Person Trips in Study Area—PM Peak Hour

Source: Fehr & Peers, 2022.

Transit

Exhibit 3.10-26 summarizes the average PM peak hour passenger load factor for a transit trip along key corridors under Alternative 2. The average passenger load factors include both light rail and bus services. Passenger load factors under Alternative 2 would be similar to those under Alternative 1 No Action. This reflects both the modest magnitude of growth between the two alternatives and also the type of growth as industrial employees are often less likely to commute by transit than those of other employment sectors. While some of the routes traveling across the study area screenlines may operate over their crowding threshold for some individual trips, load factors indicate there would generally be adequate transit capacity across most of the transit screenlines (with the exception of the 8th Avenue NW screenline). No significant impacts to transit load are expected under Alternative 2.

Exhibit 3.10-26 PM Peak Hour Average Passenger Load Factors—Alternative 2

Note: Inbound refers to travel into the downtown area and outbound travel out of the downtown area. Source: Fehr & Peers, 2021.

Roadway UsersAuto & Freight

Under Alternative 2, traffic volumes would be higher than Alternative 1 No Action though the magnitude of change would be relatively small in relation to the amount of background traffic in the city. The PM peak vehicle miles traveled (VMT) within the Greater Duwamish MIC would increase by roughly 0.8% and the PM peak VMT within the BINMIC would increase by roughly 1.4%. The effects of this additional traffic in terms of travel time, mode share, and screenline volumes, are detailed below.

Travel Time

Exhibit 3.10-31 summarizes travel time conditions along each of the study corridors under Alternative 2. Travel times for Alternative 1 No Action are also shown to illustrate how travel times would change compared to development expected to occur by 2044 under current zoning. The travel times below are rounded to the nearest half minute.

During the PM peak hour under the 2044 Alternative 2, most corridors would continue to operate at similar levels of congestion as under Alternative 1 No Action with travel times increasing by no more than 4% on any study segment. Based on the criteria for travel time impacts, one significant travel time impact is expected under Alternative 2:

- Eastbound W Dravus Street between 15th Avenue W and 20th Avenue W
- W Dravus Street would be impacted because the increase in travel time would cause the segment to fall from LOS E under Alternative 1 No Action to LOS F under Alternative 2 though the magnitude of change is expected to be less than 3%. Because freight operates on the same corridors as autos, a freight impact is also identified along eastbound W Dravus Street. Any buses operating on that corridor in the future would also be impacted.

Gameday Conditions

The study area includes large event venues including Lumen Field and T-Mobile Park. During event ingress and egress, large numbers of event attendees access these venues, bringing congestion to area roadways. This is an existing condition of the Affected Environment and there is an established Transportation Management Plan process through which event traffic is managed. The typical PM peak conditions studied in this EIS identify the relative congestion expected to be generated by each alternative and therefore provide an indication of how alternatives may compare to one another during event ingress or egress.

Peak Spreading

As growth throughout the city continues, the city will likely experience "peak spreading." Peak spreading refers to travelers shifting their departure times to avoid the heaviest traffic congestion. The result is that while the peak hour may retain similar characteristics, the length of the congested period may grow.

 All of the study segments expected to operate at LOS F under Alternative 1 No Action would continue to operate at LOS F and with slightly higher travel times under Alternative 2. However, these are not considered impacts caused by Alternative 2 because none of the travel time increases would reach the 5% impact threshold.

At this programmatic level of analysis, it is not possible to know how freight may be impacted by changes to loading zones or access needs. These are potentially significant impacts that would need to be analyzed and mitigated at the project level.

As noted in the Alternative 1 No Action section, increased traffic congestion leads to worse reliability on the roadway network, which is of particular importance for freight operators. Therefore, Alternative 2 may result in worse travel time reliability due to increased volumes and travel times.

		PM Peak Hour LOS / Travel Time (minutes)					
		Alternative 1 No Action		Alternative 2			
ID	- Corridor	N/E	S/W	N/E	S/W		
Ballard Interbay Northend MIC							
1	15th Ave W from Magnolia Bridge to NW Leary Way	E / 12.5	A / 5	E / 12.5	A / 5		
2	15th Ave NW from NW Leary Way to N 85th St	E / 9.5	C / 6.5	E/10	C / 6.5		
3	Leary Ave NW/ Leary Way NW/ N 36th S/ Fremont Bridge between NW Market St and W Nickerson St/Westlake Ave	C / 11	D/13	C / 11	D/13.5		
4	Shilshole Ave NW between NW Market and 15th Ave NW	B / 2.5	D / 4	B / 2.5	D / 4		
5	NW Market St/N 50th/ 46th St between 24th Ave NW and I-5	C / 14	D / 16.5	C / 14	D/16.5		
6	W Nickerson St/Westlake Ave N between 15th Ave W and Mercer St	C / 13.5	C / 12.5	C / 13.5	C / 12.5		
7	W Dravus St between 15th Ave W and 20th Ave W	E / 2	D / 1.5	F/2	D / 1.5		
8	Elliott Ave W between Magnolia Bridge and Wall St	B / 6	B / 6.5	B / 6	B / 6.5		
9	W Mercer St from Elliott Ave W to I-5	F/32	F / 22.5	F/32.5	F / 22.5		
10	Denny Way from Elliott Ave W to I-5	F/15	F/11.5	F/15	F/11.5		
11	Magnolia Bridge between 15th Ave W and Thorndyke Ave W	B / 2.5	A / 1.5	B / 2.5	A / 1.5		
12	SR 99 between N 46th St and Denny Way ¹	D / 10.5	F/14.5	D / 10.5	F/14.5		
13	W Emerson St between 15th Ave W and Gilman Ave W	F/6	F / 4	F/6	F / 4		
14	N 85th St between 15th Ave NW and I-5	E / 13.5	E/14.5	E / 13.5	E/14.5		
15	I-5 between N 85th Street and Madison Street ¹	F / 22.5	F / 26	F / 22.5	F/26		
Greater Duwamish MIC							
1	1st Ave S between S Royal Brougham Way and SR 99	C / 11	C / 12	C / 11	C / 12		
2	4th Ave S between Seattle Blvd S to E Marginal Way S	C / 12.5	C / 13.5	C / 12.5	C / 13.5		

Exhibit 3.10-31-27 PM Peak Hour Travel Time LOS—Alternative 2

		PM Peak Hour LOS / Travel Time (minutes)			
		Alternative 1 No Action		Altern	ative 2
ID	Corridor	N/E	S/W	N/E	S/W
3	6th Ave S between Seattle Blvd S to Spokane St Viaduct	C / 6.5	C / 6.5	C / 6.5	C / 6.5
4	Airport Way S/ Seattle Blvd S between S Royal Brougham Way to S Boeing Access Rd	A / 16.5	A / 16	A / 16.5	A / 16.5
5	West Seattle Bridge/Spokane St Viaduct between 35th Ave SW and I-5	C / 6.5	E/10	C / 6.5	E/10
6	Spokane St Bridge between Harbor Ave SW and SR 99	B / 4.5	B / 4.5	B / 4.5	B / 4.5
7	E Marginal Way S between SR 99 and S Boeing Access Rd	C / 8.5	D / 10.5	C / 8.5	D / 10.5
8	Alaskan Way S from Broad St to SR 99	E/10.5	F/14.5	E / 10.5	F/14.5
9	S Royal Brougham Way between SR 99 and Airport Way S	F/4.5	D/3	F/4.5	D/3
10	Edgar Martinez Dr S between SR 99 and 4th Ave	F/3	F / 2.5	F/3	F / 2.5
11	S Holgate St between 1st Ave and Airport Way S	D/3	F / 4.5	D/3	F / 4.5
12	S Lander St between 1st Ave and Airport Way S	E / 4	E / 4	E / 4	E / 4
13	S Lucile St between SR 99 and Airport Way S	D / 4	E / 5	D / 4	E / 5
14	W Marginal Way SW between West Seattle Bridge and 2nd Ave SW	A / 5	A/5	A/5	A / 5
15	S Michigan St/ Corson Ave S between E Marginal Way S and I-5	C / 3.5	E / 5.5	C / 3.5	E / 5.5
16	E Marginal Way S/SR 99 between S Atlantic Street and 1st Ave S Bridge	A / 9	A/9	A/9	A/9
17	I-5 between Madison Street and SR 599 ¹	F / 27.5	F/31	F/27.5	F/32
<u>18</u>	SR 509 between SR 99 and SR 518 ¹	<u>A/6</u>	<u>D / 11.5</u>	<u>A/6</u>	<u>D/12</u>
<u>19</u>	SR 99/599 between SR 509 and I-5 ²	<u>A / 6.5</u>	<u>B/8</u>	<u>A / 6.5</u>	<u>B/8</u>

Note: Cells shown in bold indicate an impact.

1. WSDOT sets a LOS D standard on I-5, SR 509, and SR 99 north of SR 509.

2. WSDOT sets a LOS E standard on SR 99/599 between SR 509 and I-5.

Source: Fehr & Peers, 2021.
Mode Share

The Alternative 2 mode share is summarized by sector using the PSRC model and is shown in **Exhibit 3.10-32Exhibit 3.10-32**. The model predicts that SOV mode shares would remain very similar between Alternative 1 No Action and Alternative 2. Therefore, as is the case under Alternative 1 No Action, all three sectors are expected to have higher SOV shares than the City's 2035 SOV targets.

The Duwamish and Northwest sectors, where the largest changes in industrial employment would be concentrated, are expected to have a slightly higher SOV share though the magnitude of change is less than the 0.5% threshold for a significant impact. Therefore, no significant mode share impacts are expected under Alternative 2.

Sector	2035 SOV Target	Alternative 1 No Action SOV Share	Alternative 2 SOV Share
Duwamish	51%	52.6%	52.8%
Magnolia/Queen Anne	38%	40.1%	40.1%
Northwest	37%	39.7%	39.8%

Exhibit 3.10-32-28 2044 Alternative 2 SOV Mode Share—PM Peak Period

Source: Fehr & Peers, 2021.

Screenlines

Exhibit 3.10-33Exhibit 3.10-33 summarizes the projected PM peak hour v/c ratios across each screenline under Alternative 2. Although traffic volumes would increase under Alternative 2, the PM peak hour v/c ratios are expected to remain very similar to those under Alternative 1 No Action. All screenline locations are forecasted to be under the LOS threshold defined by the City of Seattle; therefore, no significant screenline impacts are expected under Alternative 2.

Within the study area, the largest v/c ratio increases between Alternative 1 No Action and Alternative 2 are expected at the South City Limit screenline and the Ballard Bridge.

ne			PM Peak Period v/c Ratio			
enli		v/c Ratio	Alt. 1 N	o Action	Alt	2
Scre	Location	Threshold	N/E	S/W	N/E	S/W
2	Magnolia	1.0	0.51	0.54	0.52	0.54
3.11	Duwamish River—West Seattle Bridge and Spokane Street	1.2	0.57	0.53	0.57	0.53
3.12	Duwamish River—1st Avenue S and 16th Avenue S	1.2	0.53	0.52	0.53	0.52
4.13	South City Limit—SR 99 to Airport Way S	1.0	0.47	0.50	0.47	0.52
5.11	Ship Canal—Ballard Bridge	1.2	1.11	0.78	1.13	0.78
5.12	Ship Canal—Fremont Bridge	1.2	0.68	0.68	0.69	0.68
5.13	Ship Canal—Aurora Bridge	1.2	0.35	0.35	0.35	0.35
7.11	West of Aurora Avenue—Fremont Place N to N 65th Street	1.0	0.55	0.64	0.55	0.64
8	South of Lake Union	1.2	0.43	0.51	0.43	0.51
9.12	South of Spokane Street—E Marginal Way to Airport Way S	1.0	0.51	0.49	0.51	0.49
10.11	South of S Jackson Street—Alaskan Way S to 4th Avenue S	1.0	0.65	0.68	0.65	0.68

Exhibit 3.10-33-29 Screenline Volume-to-Capacity Ratio—Alternative 2

Source: Fehr & Peers, 2021.

<u>Transit</u>

Exhibit 3.10-34 summarizes the average PM peak hour passenger load factor for a transit trip along key corridors under Alternative 2. The average passenger load factors include both light rail and bus services. Passenger load factors under Alternative 2 would be similar to those under Alternative 1 No Action. This reflects both the modest magnitude of growth between the two alternatives and also the type of growth as industrial employees are often less likely to commute by transit than those of other employment sectors. While some of the routes traveling across the study area screenlines may operate over their crowding threshold for some individual trips, load factors indicate there would generally be adequate transit capacity across most of the transit screenlines (with the exception of the 8th Avenue NW screenline). No significant impacts to transit load are expected under Alternative 2.

	Alternative 1 No Action		<u>Alternative 2</u>	
<u>Screenline</u>	<u>Inbound</u>	<u>Outbound</u>	<u>Inbound</u>	<u>Outbound</u>
<u>A: East of 8th Avenue NW</u>	0.57	<u>1.28</u>	<u>0.58</u>	<u>1.28</u>
<u>B: Ballard Bridge</u>	<u>0.09</u>	<u>0.39</u>	<u>0.09</u>	<u>0.39</u>
<u>C: North of W Mercer Place</u>	0.29	0.59	0.30	0.59
<u>D: North of Lander St</u>	<u>0.21</u>	<u>0.75</u>	<u>0.21</u>	<u>0.76</u>
E: West Seattle Bridge	<u>0.12</u>	<u>0.35</u>	<u>0.12</u>	<u>0.35</u>

Exhibit 3.10-34 PM Peak Hour Average Passenger Load Factors—Alternative 2

Note: Inbound refers to travel into the downtown area and outbound travel out of the downtown area.

Source: Fehr & Peers, 2021.

Impacts of Alternative 3

This section summarizes analysis results and environmental impacts for Alternative 3 (Future of Industry—Targeted) in 2044. Compared to Alternative 1 No Action, Alternative 3 would result in 33,900 additional jobs and 535 additional dwelling units. As with Alternative 1 No Action, most of the new employment growth would be concentrated in the Greater Duwamish MIC; the Ballard Subarea would have the highest increase in residential growth.

Exhibit 3.10-35 summarizes the number of person trips expected to be generated during the PM peak hour by the land uses in the study area by mode of travel.

Exhibit 3.10-35	2044 Alternative 3 Person	Trips in Study	/ Area—PM Peak Hour

Mode	<u>Alternative</u>	1 No Action	<u>Alternative 3</u>	
SOV (in passenger vehicles)	<u>35,400</u>	<u>40.5%</u>	<u>39,600</u>	<u>41.3%</u>
HOV (in passenger vehicles)	<u>32,800</u>	<u>37.5%</u>	<u>35,700</u>	<u>37.2%</u>
<u>Transit</u>	<u>4,800</u>	<u>5.5%</u>	<u>5,200</u>	<u>5.4%</u>
Walk	<u>12,400</u>	<u>14.2%</u>	<u>13,200</u>	<u>13.8%</u>
Bike	<u>2,100</u>	<u>2.4%</u>	<u>2,200</u>	<u>2.3%</u>
Total	<u>87,500</u>	<u>100%</u>	<u>95,900</u>	<u>100%</u>

Source: Fehr & Peers, 2022.

Transit

Exhibit 3.10-30 summarizes PM peak hour average passenger load factors (including both light rail and bus) under Alternative 3. The largest increases in passenger load would occur eastbound across the 8th Avenue NW screenline toward the University District, and southbound on 15th Avenue NW toward Downtown. These increases reflect the expected travel

patterns of additional employees leaving the BINMIC area to travel home during the PM peak hour. Southbound travel demand across Lander Street would also increase slightly. Overall capacity across these screenlines is expected to be adequate for the demand—some routes traveling across the study area screenlines, however, may operate over their crowding threshold for some individual trips. Although a minor increase is expected westbound across 8th Avenue NW (which is already expected to have crowded transit routes under Alternative 1 No Action), the magnitude of change is less than the threshold for a significant impact. Therefore, no significant transit passenger load impacts are expected under Alternative 3.

Exhibit 3.10-30 PM Peak Hour Average Passenger Load Factors—Alternative 3

Note: Inbound refers to travel into the downtown area and outbound travel out of the downtown area. Source: Fehr & Peers, 2021.

Roadway UsersAuto & Freight

Under Alternative 3, traffic volumes would be slightly higher than Alternative 1 No Action. The PM peak vehicle miles traveled (VMT) within the Greater Duwamish MIC would increase by roughly 2.3% and the PM peak VMT within the BINMIC would increase by roughly 4.3%. The effects of this additional traffic in terms of travel time, mode share, and screenline volumes, are detailed below.

Travel Time

Exhibit 3.10-36 summarizes travel time conditions along each of the study corridors under Alternative 3 and compares them to travel times under Alternative 1 No Action. The travel times below are rounded to the nearest half minute.

During the PM peak hour under the 2044 Alternative 3, most corridors would continue to operate at similar levels of congestion as under Alternative 1 No Action with travel time increases of up to 2 minutes. Based on the criteria for travel time impacts, three four significant travel time impacts are expected under Alternative 3:

- Northbound 15th Avenue W from Magnolia Bridge to NW Leary Way
- Eastbound W Dravus Street between 15th Avenue W and 20th Avenue W
- Southbound I-5 from Madison Street to SR 599
- Southbound SR 509 between SR 99 and SR 518

The first two segments would be impacted because the increase in travel time would cause the segment to fall from LOS E under Alternative 1 No Action to LOS F under Alternative 3 and the fourth segment is because the increase in travel time would cause the segment to fall from LOS D to E, below WSDOT's LOS standard for SR 509. The I-5 segment is already expected to operate at LOS F under Alternative 1 No Action and under Alternative 3 is expected to experience a 6% increase in travel time compared to Alternative 1, exceeding the criteria for a significant impact. Because freight operates on the same corridors as autos, freight impacts are also identified along northbound 15th Avenue W, eastbound W Dravus Street, and southbound I-5. Any buses operating on those corridors in the future would also be impacted.

At this programmatic level of analysis, it is not possible to know how freight may be impacted by changes to loading zones or access needs. These are potentially significant impacts that would need to be analyzed and mitigated at the project level. As noted in the Alternative 1 No Action section, increased traffic congestion leads to worse reliability on the roadway network, which is of particular importance for freight operators. Therefore, Alternative 3 may result in worse travel time reliability due to increased volumes and travel times.

		PM Peak Hour LOS / Travel Time (minutes)			inutes)
		Alternative	1 No Action	Altern	ative 3
ID	Corridor	N/E	S/W	N/E	S/W
Ballar	d Interbay Northend MIC				
1	15th Ave W from Magnolia Bridge to NW Leary Way	E / 12.5	A / 5	F/13	A / 5
2	15th Ave NW from NW Leary Way to N 85th St	E / 9.5	C / 6.5	E/10	C / 6.5
3	Leary Ave NW/ Leary Way NW/ N 36th S/ Fremont Bridge between NW Market St and W Nickerson St/Westlake Ave	C / 11	D / 13	C / 11.5	D / 13.5
4	Shilshole Ave NW between NW Market and 15th Ave NW	B / 2.5	D / 4	B / 2.5	D / 4
5	NW Market St/N 50th/ 46th St between 24th Ave NW and I-5	C / 14	D / 16.5	C / 14	D / 16.5
6	W Nickerson St/Westlake Ave N between 15th Ave W and Mercer St	C / 13.5	C / 12.5	C / 13.5	C / 12.5
7	W Dravus St between 15th Ave W and 20th Ave W	E / 2	D / 1.5	F/2	D / 1.5
8	Elliott Ave W between Magnolia Bridge and Wall St	B / 6	B / 6.5	B / 6	B / 6.5
9	W Mercer St from Elliott Ave W to I-5	F/32	F/22.5	F/32.5	F/22.5
10	Denny Way from Elliott Ave W to I-5	F/15	F/11.5	F/15	E/11.5
11	Magnolia Bridge between 15th Ave W and Thorndyke Ave W	B/2.5	A / 1.5	B / 2.5	A / 1.5
12	SR 99 between N 46th St and Denny Way ¹	D / 10.5	F/14.5	D / 10.5	F/14.5
13	W Emerson St between 15th Ave W and Gilman Ave W	F/6	F / 4	F/6	F / 4
14	N 85th St between 15th Ave NW and I-5	E / 13.5	E/14.5	E / 13.5	E / 14.5
15	I-5 between N 85th Street and Madison Street ¹	F / 22.5	F/26	F/22.5	F / 26
Great	er Duwamish MIC				
1	1st Ave S between S Royal Brougham Way and SR 99	C / 11	C / 12	C /11	C / 12
2	4th Ave S between Seattle Blvd S to E Marginal Way S	C / 12.5	C / 13.5	C / 12.5	C / 14

Exhibit 3.10-36-31 PM Peak Hour Travel Time LOS—Alternative 3

		PM Peak Hour LOS / Travel Time (minutes)			inutes)
		Alternative	1 No Action	Altern	ative 3
ID	Corridor	N/E	S/W	N/E	S/W
3	6th Ave S between Seattle Blvd S to Spokane St Viaduct	C / 6.5	C / 6.5	C / 6.5	C / 7
4	Airport Way S/ Seattle Blvd S between S Royal Brougham Way to S Boeing Access Rd	A / 16.5	A / 16	A / 16.5	B / 17.5
5	West Seattle Bridge/Spokane St Viaduct between 35th Ave SW and I-5	C / 6.5	E/10	C / 6.5	E/10
6	Spokane St Bridge between Harbor Ave SW and SR 99	B / 4.5	B / 4.5	B / 4.5	B / 4.5
7	E Marginal Way S between SR 99 and S Boeing Access Rd	C / 8.5	D / 10.5	C / 8.5	D / 11
8	Alaskan Way S from Broad St to SR 99	E / 10.5	F/14.5	E / 10.5	F / 14.5
9	S Royal Brougham Way between SR 99 and Airport Way S	F/4.5	D/3	F/4.5	D/3
10	Edgar Martinez Dr S between SR 99 and 4th Ave	F/3	F / 2.5	F/3	F / 2.5
11	S Holgate St between 1st Ave and Airport Way S	D/3	F / 4.5	D/3	F / 4.5
12	S Lander St between 1st Ave and Airport Way S	E / 4	E / 4	E / 4	E / 4
13	S Lucile St between SR 99 and Airport Way S	D / 4	E / 5	D / 4	E / 5
14	W Marginal Way SW between West Seattle Bridge and 2nd Ave SW	A / 5	A / 5	A / 5	A / 5
15	S Michigan St/ Corson Ave S between E Marginal Way S and I-5	C / 3.5	E / 5.5	C / 3.5	E / 5.5
16	E Marginal Way S/SR 99 between S Atlantic Street and 1st Ave S Bridge	A / 9	A/9	A / 9	A / 9.5
17	I-5 between Madison Street and SR 599 ¹	F/27.5	F/31	F / 27.5	F/33
<u>18</u>	SR 509 between SR 99 and SR 518 ¹	<u>A/6</u>	<u>D / 11.5</u>	<u>A/6</u>	<u>E / 12.5</u>
<u>19</u>	<u>SR 99/599 between SR 509 and I-52</u>	<u>A / 6.5</u>	<u>B/8</u>	<u>A / 6.5</u>	<u>C / 8.5</u>

Note: Cells shown in bold indicate an impact.

1. WSDOT sets a LOS D standard on I-5, SR 509, and SR 99 north of SR 509.

2. WSDOT sets a LOS E standard on SR 99/509 between SR 509 and I-5.

Source: Fehr & Peers, 2021.

Mode Share

The Alternative 3 mode share is summarized by sector using the PSRC model and is shown in **Exhibit 3.10-37**. The model predicts that SOV mode shares under Alternative 3 would remain similar or slightly higher than Alternative 1 No Action. Therefore, as is the case under Alternative 1 No Action, all three sectors are expected to have higher SOV shares than the City's 2035 SOV targets.

The Duwamish sector is expected to have the same SOV share as Alternative 1 No Action and the Magnolia/Queen Anne and Northwest sectors are expected to have slightly higher SOV shares. Because the SOV mode share in the Magnolia/Queen Anne sector is expected to increase by 0.5% compared to Alternative 1 No Action, a significant mode share impact is expected in that sector.

Sector	2035 SOV Target	Alternative 1 No Action SOV Share	Alternative 3 SOV Share
Duwamish	51%	52.6%	52.6%
Magnolia/Queen Anne	38%	40.1%	40.6%
Northwest	37%	39.7%	39.9%

Exhibit 3.10-37-32 2044 Alternative 3 SOV Mode Share—PM Peak Period

Note: Cells shown in bold indicate an impact. Source: Fehr & Peers, 2021.

Screenlines

Exhibit 3.10-38 summarizes the projected PM peak hour v/c ratios across each screenline under Alternative 3. Although traffic volumes would increase under Alternative 3, all screenline locations are forecasted to be under the LOS threshold defined by the City of Seattle. Therefore, no significant screenline impacts are expected under Alternative 3.

Within the study area, the largest v/c ratio increases between Alternative 1 No Action and Alternative 3 are expected at the Magnolia screenline, South City Limit screenline, and the Ballard Bridge.

ne			Р	M Peak Peri	iod v/c Rat	io
enli		v/c Ratio	Alt. 1 No	o Action	Alt	t. 3
Scre	Location	Threshold	N/E	S/W	N/E	S/W
2	Magnolia	1.0	0.51	0.54	0.55	0.55
3.11	Duwamish River—West Seattle Bridge and Spokane Street	1.2	0.57	0.53	0.57	0.53
3.12	Duwamish River—1st Avenue S and 16th Avenue S	1.2	0.53	0.52	0.54	0.52
4.13	South City Limit—SR 99 to Airport Way S	1.0	0.47	0.50	0.48	0.56
5.11	Ship Canal—Ballard Bridge	1.2	1.11	0.78	1.15	0.77
5.12	Ship Canal—Fremont Bridge	1.2	0.68	0.68	0.69	0.69
5.13	Ship Canal—Aurora Bridge	1.2	0.35	0.35	0.35	0.35
7.11	West of Aurora Avenue—Fremont Place N to N 65th Street	1.0	0.55	0.64	0.55	0.64
8	South of Lake Union	1.2	0.43	0.51	0.43	0.52
9.12	South of Spokane Street—E Marginal Way to Airport Way S	1.0	0.51	0.49	0.52	0.50
10.11	South of S Jackson Street—Alaskan Way S to 4th Avenue S	1.0	0.65	0.68	0.65	0.68

Exhibit 3.10-38-33 Screenline Volume-to-Capacity Ratio—Alternative 3

Source: Fehr & Peers, 2021.

<u>Transit</u>

Exhibit 3.10-39 summarizes PM peak hour average passenger load factors (including both light rail and bus) under Alternative 3. The largest increases in passenger load would occur eastbound across the 8th Avenue NW screenline toward the University District, and southbound on 15th Avenue NW toward Downtown. These increases reflect the expected travel patterns of additional employees leaving the BINMIC area to travel home during the PM peak hour. Southbound travel demand across Lander Street would also increase slightly. Overall capacity across these screenlines is expected to be adequate for the demand—some routes traveling across the study area screenlines, however, may operate over their crowding threshold for some individual trips. Although a minor increase is expected westbound across 8th Avenue NW (which is already expected to have crowded transit routes under Alternative 1 No Action), the magnitude of change is less than the threshold for a significant impact. Therefore, no significant transit passenger load impacts are expected under Alternative 3.

	Alternative 1 No Action		Alternative 3	
<u>Screenline</u>	<u>Inbound</u>	<u>Outbound</u>	<u>Inbound</u>	<u>Outbound</u>
<u>A: East of 8th Avenue NW</u>	0.57	<u>1.28</u>	<u>0.64</u>	<u>1.29</u>
<u>B: Ballard Bridge</u>	<u>0.09</u>	<u>0.39</u>	<u>0.09</u>	<u>0.39</u>
C: North of W Mercer Place	0.29	0.59	0.34	0.58
<u>D: North of Lander St</u>	<u>0.21</u>	0.75	<u>0.21</u>	0.77
E: West Seattle Bridge	<u>0.12</u>	<u>0.35</u>	<u>0.12</u>	<u>0.35</u>

Exhibit 3.10-39	PM Peak Hour	Average Pass	senger Load I	Factors—Alternative 3

Note: Inbound refers to travel into the downtown area and outbound travel out of the downtown area.

Source: Fehr & Peers, 2021.

Impacts of Alternative 4

This section summarizes analysis results and environmental impacts for Alternative 4 (Future of Industry—Expanded) in 2044. Compared to Alternative 1 No Action, Alternative 4 would result in 35,700 additional jobs and 2,120 additional dwelling units. Most of the new employment growth would be concentrated in the Greater Duwamish MIC. The Ballard and SODO/Stadium subareas would have the highest increases in residential growth.

Exhibit 3.10-40 summarizes the number of person trips expected to be generated during the PM peak hour by the land uses in the study area by mode of travel.

Exhibit 3.10-40 2044 Alternative 4 Person Trips in Study Area—PM Peak Hour

Mode	<u>Alternative</u>	<u>1 No Action</u>	Alterna	ative 4
SOV (in passenger vehicles)	<u>35,400</u>	40.5%	<u>40,300</u>	<u>41.2%</u>
HOV (in passenger vehicles)	<u>32,800</u>	<u>37.5%</u>	<u>36,400</u>	<u>37.3%</u>
<u>Transit</u>	<u>4,800</u>	<u>5.5%</u>	<u>5,300</u>	<u>5.4%</u>
<u>Walk</u>	<u>12,400</u>	<u>14.2%</u>	<u>13,400</u>	<u>13.7%</u>
Bike	<u>2,100</u>	<u>2.4%</u>	<u>2,300</u>	<u>2.4%</u>
Total	87,500	<u>100%</u>	<u>97,700</u>	<u>100%</u>

Source: Fehr & Peers, 2022.

Transit

Exhibit 3.10-34 summarizes PM peak hour average passenger load factors under Alternative 4. The passenger load factors include both light rail and bus services. The largest increases in passenger load would occur eastbound across the 8th Avenue NW screenline toward the University District, and southbound on 15th Avenue NW toward Downtown. These increases reflect the expected travel patterns of additional employees leaving the BINMIC area to travel home during the PM peak hour. Southbound travel demand across Lander Street would also increase slightly. Overall capacity across these screenlines is expected to be adequate for the demand—some routes traveling across the study area screenlines, however, may operate over their crowding threshold for some individual trips. Although a minor increase is expected westbound across 8th Avenue NW (which is already expected to have crowded transit routes under Alternative 1 No Action), the magnitude of change is less than the threshold for a significant impact. Therefore, no transit passenger load impacts are expected under Alternative 4.

Exhibit 3.10-34 PM Peak Hour Average Passenger Load Factors—Alternative 4

Note: Inbound refers to travel into the downtown area and outbound travel out of the downtown area. Source: Fehr & Peers, 2021.

Roadway UsersAuto & Freight

Among the alternatives, traffic volumes would be highest under Alternative 4. The PM peak vehicle miles traveled (VMT) within the Greater Duwamish MIC would increase by roughly 2.5% and the PM peak VMT within the BINMIC would increase by roughly 5.1%. The effects of this additional traffic in terms of travel time, mode share, and screenline volumes, are detailed below.

Travel Time

Exhibit 3.10-41 summarizes travel time conditions along each of the study corridors under Alternative 4 and compares them to travel times under Alternative 1 No Action. The travel times below are rounded to the nearest half minute.

During the PM peak hour under the 2044 Alternative 4, most corridors would continue to operate at similar levels of congestion as under Alternative 1 No Action with travel times increases of up to 2 minutes. Based on the criteria for travel time impacts, three four significant travel time impacts are expected under Alternative 4:

- Northbound 15th Avenue W from Magnolia Bridge to NW Leary Way
- Eastbound W Dravus Street between 15th Avenue W and 20th Avenue W
- Southbound I-5 from Madison Street to SR 599
- Southbound SR 509 between SR 99 and SR 518

The first two segments would be impacted because the increase in travel time would cause the segment to fall from LOS E under Alternative 1 No Action to LOS F under Alternative 4 <u>and the fourth segment is because the increase in travel time would cause the segment to fall from LOS D to E, below WSDOT's LOS standard for SR 509</u>. The I-5 segment is already expected to operate at LOS F under Alternative 1 No Action and under Alternative 4 is expected to experience a 7% increase in travel time compared to Alternative 1, exceeding the criteria for a significant impact. Because freight operates on the same corridors as autos, freight impacts are also identified

along northbound 15th Avenue W, eastbound W Dravus Street, and southbound I-5. Any buses operating on those corridors in the future would also be impacted.

At this programmatic level of analysis, it is not possible to know how freight may be impacted by changes to loading zones or access needs. These are potentially significant impacts that would need to be analyzed and mitigated at the project level.

As noted in the Alternative 1 No Action section, increased traffic congestion leads to worse reliability on the roadway network, which is of particular importance for freight operators. Therefore, Alternative 4 may result in worse travel time reliability due to increased volumes and travel times.

		PM Peak Hour LOS / Travel Time (minutes)			inutes)
		Alternative	1 No Action	Alterr	native 4
ID	Corridor	N/E	S/W	N/E	S/W
Ballaro	d Interbay Northend MIC				
1	15th Ave W from Magnolia Bridge to NW Leary Way	E/12.5	A / 5	F/13	A / 5
2	15th Ave NW from NW Leary Way to N 85th St	E / 9.5	C / 6.5	E/10	C / 6.5
3	Leary Ave NW/ Leary Way NW/ N 36th S/ Fremont Bridge between NW Market St and W Nickerson St/Westlake Ave	C / 11	D/13	C / 11.5	D / 13.5
4	Shilshole Ave NW between NW Market and 15th Ave NW	B / 2.5	D / 4	B / 2.5	D / 4
5	NW Market St/N 50th/ 46th St between 24th Ave NW and I-5	C / 14	D / 16.5	C / 14	D / 16.5
6	W Nickerson St/Westlake Ave N between 15th Ave W and Mercer St	C / 13.5	C / 12.5	C / 13.5	C / 12.5
7	W Dravus St between 15th Ave W and 20th Ave W	E / 2	D / 1.5	F/2	D / 1.5
8	Elliott Ave W between Magnolia Bridge and Wall St	B / 6	B / 6.5	B / 6	B / 6.5
9	W Mercer St from Elliott Ave W to I-5	F / 32	F / 22.5	F / 32.5	F/22.5
10	Denny Way from Elliott Ave W to I-5	F/15	F/11.5	F/15	E/11.5
11	Magnolia Bridge between 15th Ave W and Thorndyke Ave W	B / 2.5	A / 1.5	B / 2.5	A / 1.5
12	SR 99 between N 46th St and Denny Way ¹	D / 10.5	F/14.5	D / 10.5	F/14.5
13	W Emerson St between 15th Ave W and Gilman Ave W	F/6	F / 4	F/6	F / 4
14	N 85th St between 15th Ave NW and I-5	E/13.5	E/14.5	E / 13.5	E/14.5
15	I-5 between N 85th Street and Madison Street ¹	F/22.5	F / 26	F/22.5	F/26.5
Greate	er Duwamish MIC				
1	1st Ave S between S Royal Brougham Way and SR 99	C / 11	C / 12	C / 11	C / 12
2	4th Ave S between Seattle Blvd S to E Marginal Way S	C / 12.5	C / 13.5	C / 12.5	C / 14

Exhibit 3.10-41-35 PM Peak Hour Travel Time LOS—Alternative 4

		PM Peak Hour LOS / Travel Time (minutes)			inutes)
		Alternative	1 No Action	Alterr	native 4
ID	Corridor	N/E	S/W	N/E	S/W
3	6th Ave S between Seattle Blvd S to Spokane St Viaduct	C / 6.5	C / 6.5	C / 6.5	C / 7
4	Airport Way S/ Seattle Blvd S between S Royal Brougham Way to S Boeing Access Rd	A / 16.5	A / 16	A / 16.5	B / 17.5
5	West Seattle Bridge/Spokane St Viaduct between 35th Ave SW and I-5	C / 6.5	E/10	C / 6.5	E/10
6	Spokane St Bridge between Harbor Ave SW and SR 99	B / 4.5	B / 4.5	B / 4.5	B / 4.5
7	E Marginal Way S between SR 99 and S Boeing Access Rd	C / 8.5	D/10.5	C / 8.5	D / 11
8	Alaskan Way S from Broad St to SR 99	E / 10.5	F/14.5	E / 10.5	F / 14.5
9	S Royal Brougham Way between SR 99 and Airport Way S	F/4.5	D/3	F / 4.5	D/3
10	Edgar Martinez Dr S between SR 99 and 4th Ave	F/3	F/2.5	F/3	F/2.5
11	S Holgate St between 1st Ave and Airport Way S	D/3	F/4.5	D/3	F / 4.5
12	S Lander St between 1st Ave and Airport Way S	E / 4	E / 4	E / 4	E / 4
13	S Lucile St between SR 99 and Airport Way S	D / 4	E / 5	D / 4	E / 5
14	W Marginal Way SW between West Seattle Bridge and 2nd Ave SW	A / 5	A / 5	A / 5	A / 5
15	S Michigan St/ Corson Ave S between E Marginal Way S and I-5	C / 3.5	E/5.5	C / 3.5	E / 5.5
16	E Marginal Way S/SR 99 between S Atlantic Street and 1st Ave S Bridge	A / 9	A/9	A / 9	A / 10
17	I-5 between Madison Street and SR 599 ¹	F/27.5	F/31	F/27.5	F/33
<u>18</u>	SR 509 between SR 99 and SR 518 ¹	<u>A/6</u>	<u>D / 11.5</u>	<u>A/6</u>	<u>E/13</u>
<u>19</u>	SR 99/599 between SR 509 and I-5 ²	<u>A / 6.5</u>	<u>B/8</u>	<u>A / 6.5</u>	<u>C / 8.5</u>

Note: Cells shown in bold indicate an impact.

1. WSDOT sets a LOS D standard on I-5, SR 509, and SR 99 north of SR 509.

2. WSDOT sets a LOS E standard on SR 99/599 between SR 509 and I-5.

Source: Fehr & Peers, 2021.

Mode Share

The Alternative 4 mode share is summarized by sector using the PSRC model and is shown in **Exhibit 3.10-42**. The model predicts that SOV mode shares under Alternative 4 would remain similar or slightly higher than Alternative 1 No Action. Therefore, as is the case under Alternative 1 No Action, all three sectors are expected to have higher SOV shares than the City's 2035 SOV targets.

The Duwamish sector is expected to have the same SOV share than <u>as</u> Alternative 1 No Action and the Magnolia/Queen Anne and Northwest sectors are expected to have slightly higher SOV shares. Because the SOV mode share in the Magnolia/Queen Anne sector is expected to increase by 0.5% compared to Alternative 1 No Action, a significant mode share impact is expected in that sector.

Sector	2035 SOV Target	Alternative 1 No Action SOV Share	Alternative 4 SOV Share
Duwamish	51%	52.6%	52.6%
Magnolia/Queen Anne	38%	40.1%	40.6%
Northwest	37%	39.7%	39.9%

Exhibit 3.10-42-36 2044 Alternative 4 SOV Mode Share—PM Peak Period

Note: Cells shown in bold indicate an impact. Source: Fehr & Peers, 2021.

Screenlines

Exhibit 3.10-43 summarizes the projected PM peak hour v/c ratios across each screenline under Alternative 4. Although traffic volumes would increase under Alternative 4, all screenline locations are forecasted to be under the LOS threshold defined by the City of Seattle. Therefore, no significant screenline impacts are expected under Alternative 4.

Within the study area, the largest v/c ratio increases between Alternative 1 No Action and Alternative 4 are expected at the Magnolia screenline, South City Limit screenline, and the Ballard Bridge.

ne		PM Peak Period v/c Ratio				
enli		v/c Ratio	Alt. 1 N	o Action	Alt	t. 4
Scre	Location	Threshold	N/E	S/W	N/E	S/W
2	Magnolia	1.0	0.51	0.54	0.55	0.55
3.11	Duwamish River—West Seattle Bridge and Spokane Street	1.2	0.57	0.53	0.57	0.53
3.12	Duwamish River—1st Avenue S and 16th Avenue S	1.2	0.53	0.52	0.54	0.52
4.13	South City Limit—SR 99 to Airport Way S	1.0	0.47	0.50	0.48	0.56
5.11	Ship Canal—Ballard Bridge	1.2	1.11	0.78	1.15	0.77
5.12	Ship Canal—Fremont Bridge	1.2	0.68	0.68	0.69	0.69
5.13	Ship Canal—Aurora Bridge	1.2	0.35	0.35	0.35	0.35
7.11	West of Aurora Avenue—Fremont Place N to N 65th Street	1.0	0.55	0.64	0.55	0.64
8	South of Lake Union	1.2	0.43	0.51	0.43	0.51
9.12	South of Spokane Street—E Marginal Way to Airport Way S	1.0	0.51	0.49	0.52	0.50
10.11	South of S Jackson Street—Alaskan Way S to 4th Avenue S	1.0	0.65	0.68	0.66	0.68

Exhibit 3.10-43-37 Screenline Volume-to-Capacity Ratio—Alternative 4

Source: Fehr & Peers, 2021.

<u>Transit</u>

Exhibit 3.10-44 summarizes PM peak hour average passenger load factors under Alternative 4. The passenger load factors include both light rail and bus services. The largest increases in passenger load would occur eastbound across the 8th Avenue NW screenline toward the University District, and southbound on 15th Avenue NW toward Downtown. These increases reflect the expected travel patterns of additional employees leaving the BINMIC area to travel home during the PM peak hour. Southbound travel demand across Lander Street would also increase slightly. Overall capacity across these screenlines is expected to be adequate for the demand—some routes traveling across the study area screenlines, however, may operate over their crowding threshold for some individual trips. Although a minor increase is expected westbound across 8th Avenue NW (which is already expected to have crowded transit routes under Alternative 1 No Action), the magnitude of change is less than the threshold for a significant impact. Therefore, no transit passenger load impacts are expected under Alternative 4.

	Alternative 1 No Action		Altern	ative 4
<u>Screenline</u>	<u>Inbound</u>	<u>Outbound</u>	<u>Inbound</u>	<u>Outbound</u>
<u>A: East of 8th Avenue NW</u>	<u>0.57</u>	<u>1.28</u>	<u>0.67</u>	<u>1.30</u>
<u>B: Ballard Bridge</u>	<u>0.09</u>	<u>0.39</u>	<u>0.09</u>	<u>0.39</u>
<u>C: North of W Mercer Place</u>	0.29	0.59	0.35	0.58
<u>D: North of Lander St</u>	<u>0.21</u>	<u>0.75</u>	<u>0.21</u>	<u>0.77</u>
E: West Seattle Bridge	<u>0.12</u>	<u>0.35</u>	<u>0.12</u>	<u>0.35</u>

Exhibit 3.10-44 PM Peak Hour Average Passenger Load Factors—Alternative 4

Note: Inbound refers to travel into the downtown area and outbound travel out of the downtown area.

Source: Fehr & Peers, 2021.

Impacts of the Preferred Alternative

This section summarizes analysis results and environmental impacts for the Preferred Alternative (Future of Industry—Balanced) in 2044, which is described in **Chapter 2**. The regional travel demand model was updated to include the Preferred Alternative's proposed land use to estimate how that growth would affect the transportation system throughout the study area. **Exhibit 3.10-45** summarizes the number of person trips expected to be generated by the land uses in the study area by mode of travel. PM peak hour person trips generated by the land uses in the study area are expected to increase from 87,500 to 92,600, a 6% increase. On a percentage basis, mode shares would remain essentially the same between the No Action Alternative and the Preferred Alternative.

Mode	<u>Alternative</u>	<u>1 No Action</u>	Preferred Alternative	
SOV (in passenger vehicles)	<u>35,400</u>	<u>40.5%</u>	<u>37,600</u>	<u>40.6%</u>
HOV (in passenger vehicles)	<u>32,800</u>	<u>37.5%</u>	<u>34,600</u>	<u>37.4%</u>
<u>Transit</u>	<u>4,800</u>	<u>5.5%</u>	<u>5,000</u>	<u>5.4%</u>
<u>Walk</u>	<u>12,400</u>	<u>14.2%</u>	<u>13,200</u>	<u>14.3%</u>
Bike	<u>2,100</u>	<u>2.4%</u>	<u>2,200</u>	<u>2.4%</u>
Total	<u>87,500</u>	<u>100%</u>	<u>92,600</u>	<u>100%</u>

Exhibit 3.10-45 2044 Preferred Alternative Person Trips in Study Area—PM Peak Hour

Source: Fehr & Peers, 2022.

Roadway Users

Among the alternatives, traffic volumes generated by the Preferred Alternative would be between those generated by alternatives 2 and 3. Relative to Alternative 1 No Action, the PM peak vehicle miles traveled (VMT) within the Greater Duwamish MIC would increase by roughly 1% and the PM peak VMT within the BINMIC would increase by roughly 2.7%. The effects of this additional traffic in terms of travel time, mode share, and screenline volumes, are detailed below.

Travel Time

Exhibit 3.10-46 summarizes travel time conditions along each of the study corridors under the Preferred Alternative and compares them to travel times under Alternative 1 No Action. The travel times below are rounded to the nearest half minute.

During the PM peak hour under the 2044 Preferred Alternative, most corridors would continue to operate at similar levels of congestion as under Alternative 1 No Action with travel times increases of up to one minute. Based on the criteria for travel time impacts, two significant travel time impacts are expected under the Preferred Alternative:

- Northbound 15th Avenue W from Magnolia Bridge to NW Leary Way
- Eastbound W Dravus Street between 15th Avenue W and 20th Avenue W

Both segments would be impacted because the increase in travel time would cause the segment to fall from LOS E under Alternative 1 No Action to LOS F under the Preferred Alternative. Because freight operates on the same corridors as autos, freight impacts are also identified along northbound 15th Avenue W and eastbound W Dravus Street. Any buses operating on those corridors in the future would also be impacted.

As noted in the Alternative 1 No Action section, increased traffic congestion leads to worse reliability on the roadway network, which is of particular importance for freight operators. Therefore, the Preferred Alternative may result in worse travel time reliability due to increased volumes and travel times.

Exhibit 3.10-46 PM Peak Hour Travel Time LOS—Preferred Alternative

		<u>PM Peak Hour LOS / Travel Time (minutes)</u>			<u>ninutes)</u>
		<u>Alternative</u>	<u>1 No Action</u>	Preferred	<u>Alternative</u>
<u>ID</u>	Corridor	<u>N/E</u>	<u>S/W</u>	<u>N/E</u>	<u>S/W</u>
Ballar	d Interbay Northend MIC				
<u>1</u>	15th Ave W from Magnolia Bridge to NW Leary Way	<u>E / 12.5</u>	<u>A/5</u>	<u>F / 12.5</u>	<u>A/5</u>
<u>2</u>	15th Ave NW from NW Leary Way to N 85th St	<u>E / 9.5</u>	<u>C / 6.5</u>	<u>E / 10</u>	<u>C / 6.5</u>
<u>3</u>	Leary Ave NW/ Leary Way NW/ N 36th S/ Fremont Bridge between NW Market St and W Nickerson St/Westlake Ave	<u>C / 11</u>	<u>D/13</u>	<u>C / 11.5</u>	<u>D / 13.5</u>
<u>4</u>	Shilshole Ave NW between NW Market and 15th Ave NW	<u>B/2.5</u>	<u>D/4</u>	<u>B/2.5</u>	<u>D/4</u>
<u>5</u>	NW Market St/N 50th/ 46th St between 24th Ave NW and I-5	<u>C / 14</u>	<u>D / 16.5</u>	<u>C / 14</u>	<u>D/16.5</u>
<u>6</u>	W Nickerson St/Westlake Ave N between 15th Ave W and Mercer St	<u>C / 13.5</u>	<u>C / 12.5</u>	<u>C / 13.5</u>	<u>C / 12.5</u>
<u>7</u>	W Dravus St between 15th Ave W and 20th Ave W	<u>E/2</u>	<u>D / 1.5</u>	<u>F/2</u>	<u>D / 1.5</u>
<u>8</u>	Elliott Ave W between Magnolia Bridge and Wall St	<u>B/6</u>	<u>B/6.5</u>	<u>B/6</u>	<u>B/6.5</u>
<u>9</u>	W Mercer St from Elliott Ave W to I-5	<u>F/32</u>	<u>F/22.5</u>	<u>F/32.5</u>	<u>F/22.5</u>
<u>10</u>	Denny Way from Elliott Ave W to I-5	<u>F/15</u>	<u>F / 11.5</u>	<u>F/15</u>	<u>F/11.5</u>
<u>11</u>	Magnolia Bridge between 15th Ave W and Thorndyke Ave W	<u>B/2.5</u>	<u>A / 1.5</u>	<u>B/2.5</u>	<u>A / 1.5</u>
<u>12</u>	SR 99 between N 46th St and Denny Way ¹	<u>D / 10.5</u>	<u>F / 14.5</u>	<u>D / 10.5</u>	<u>F/14.5</u>
<u>13</u>	W Emerson St between 15th Ave W and Gilman Ave W	<u>F/6</u>	<u>F/4</u>	<u>F/6</u>	<u>F/4</u>
<u>14</u>	N 85th St between 15th Ave NW and I-5	<u>E / 13.5</u>	<u>E / 14.5</u>	<u>E / 13.5</u>	<u>E / 14.5</u>
<u>15</u>	I-5 between N 85th Street and Madison Street ¹	<u>F / 22.5</u>	<u>F/26</u>	<u>F/22.5</u>	<u>F/26.5</u>
Great	er Duwamish MIC				
<u>1</u>	1st Ave S between S Royal Brougham Way and SR 99	<u>C / 11</u>	<u>C / 12</u>	<u>C / 11</u>	<u>C / 11.5</u>
<u>2</u>	4th Ave S between Seattle Blvd S to E Marginal Way S	<u>C / 12.5</u>	<u>C / 13.5</u>	<u>C / 12.5</u>	<u>C / 13.5</u>

		<u>PM Peak Hour LOS / Travel Time (minutes)</u>			<u>ninutes)</u>
		<u>Alternative</u>	<u>1 No Action</u>	<u>Preferred</u>	<u>Alternative</u>
<u>ID</u>	<u>Corridor</u>	<u>N/E</u>	<u>S/W</u>	<u>N/E</u>	<u>S/W</u>
<u>3</u>	6th Ave S between Seattle Blvd S to Spokane St Viaduct	<u>C / 6.5</u>	<u>C / 6.5</u>	<u>C / 6.5</u>	<u>C / 6.5</u>
<u>4</u>	Airport Way S/ Seattle Blvd S between S Royal Brougham Way to S Boeing Access Rd	<u>A / 16.5</u>	<u>A / 16</u>	<u>A / 16.5</u>	<u>A / 16.5</u>
<u>5</u>	West Seattle Bridge/Spokane St Viaduct between 35th Ave SW and I-5	<u>C / 6.5</u>	<u>E / 10</u>	<u>C / 6.5</u>	<u>E / 10</u>
<u>6</u>	Spokane St Bridge between Harbor Ave SW and SR 99	<u>B / 4.5</u>	<u>B / 4.5</u>	<u>B / 4.5</u>	<u>B / 4.5</u>
<u>7</u>	E Marginal Way S between SR 99 and S Boeing Access Rd	<u>C / 8.5</u>	<u>D / 10.5</u>	<u>C / 8.5</u>	<u>D / 10.5</u>
<u>8</u>	Alaskan Way S from Broad St to SR 99	<u>E / 10.5</u>	<u>F/14.5</u>	<u>E / 10.5</u>	<u>F / 14.5</u>
<u>9</u>	S Royal Brougham Way between SR 99 and Airport Way S	<u>F / 4.5</u>	<u>D/3</u>	<u>F / 4.5</u>	<u>D/3</u>
<u>10</u>	Edgar Martinez Dr S between SR 99 and 4th Ave	<u>F/3</u>	<u>F/2.5</u>	<u>F/3</u>	<u>F/2.5</u>
<u>11</u>	S Holgate St between 1st Ave and Airport Way S	<u>D/3</u>	<u>F/4.5</u>	<u>D/3</u>	<u>F/4.5</u>
<u>12</u>	S Lander St between 1st Ave and Airport Way S	<u>E/4</u>	<u>E/4</u>	<u>E/4</u>	<u>E/4</u>
<u>13</u>	S Lucile St between SR 99 and Airport Way S	<u>D/4</u>	<u>E/5</u>	<u>D/4</u>	<u>E/5</u>
<u>14</u>	W Marginal Way SW between West Seattle Bridge and 2nd Ave SW	<u>A/5</u>	<u>A/5</u>	<u>A/5</u>	<u>A/5</u>
<u>15</u>	S Michigan St/ Corson Ave S between E Marginal Way S and I-5	<u>C / 3.5</u>	<u>E/5.5</u>	<u>C/3.5</u>	<u>E / 5.5</u>
<u>16</u>	E Marginal Way S/SR 99 between S Atlantic Street and 1st Ave S Bridge	<u>A/9</u>	<u>A/9</u>	<u>A/9</u>	<u>A/9</u>
<u>17</u>	I-5 between Madison Street and SR 599 ¹	<u>F/27.5</u>	<u>F/31</u>	<u>F/27.5</u>	<u>F/32</u>
<u>18</u>	SR 509 between SR 99 and SR 518 ¹	<u>A/6</u>	<u>D / 11.5</u>	<u>A/6</u>	<u>D /12</u>
<u>19</u>	SR 99/599 between SR 509 and I-5 ²	<u>A / 6.5</u>	<u>B/8</u>	<u>A/6.5</u>	<u>B/8</u>

Note: Cells shown in bold indicate an impact.

1. WSDOT sets a LOS D standard on I-5, SR 509, and SR 99 north of SR 509. 2. WSDOT sets a LOS E standard on SR 99/599 between SR 509 and I-5. Source: Fehr & Peers, 2022.

Mode Share

The Preferred Alternative mode share is summarized by sector using the PSRC model and is shown in **Exhibit 3.10-47**. The model predicts that SOV mode shares under the Preferred Alternative would remain similar or slightly higher than Alternative 1 No Action. Therefore, as is the case under Alternative 1 No Action, all three sectors are expected to have higher SOV shares than the City's 2035 SOV targets.

The Duwamish sector is expected to have the same SOV share as Alternative 1 No Action and the Magnolia/Queen Anne and Northwest sectors are expected to have slightly higher SOV shares. However, the magnitude of change is less than the 0.5% threshold for a significant impact and therefore, no significant mode share impacts are expected under the Preferred Alternative.

Exhibit 3.10-47 2044 Preferred Alternative SOV Mode Share—PM Peak Period

<u>Sector</u>	2035 SOV Target	<u>Alternative 1 No</u> <u>Action SOV Share</u>	<u>Preferred Alternative</u> <u>SOV Share</u>
Duwamish	<u>51%</u>	<u>52.6%</u>	<u>52.6%</u>
Magnolia/Queen Anne	<u>38%</u>	<u>40.1%</u>	<u>40.4%</u>
<u>Northwest</u>	<u>37%</u>	<u>39.7%</u>	<u>39.8%</u>

<u>Note: Cells shown in bold indicate an impact.</u> Source: Fehr & Peers, 2022.

Screenlines

Exhibit 3.10-48 summarizes the projected PM peak hour v/c ratios across each screenline under the Preferred Alternative. Although traffic volumes would increase under the Preferred Alternative, all screenline locations are forecasted to be under the LOS threshold defined by the City of Seattle. Therefore, no significant screenline impacts are expected under the Preferred Alternative.

Within the study area, the largest v/c ratio increases between Alternative 1 No Action and the Preferred Alternative are expected at the Magnolia screenline, and the Ballard Bridge.

ne				<u>PM Peak Period v/c Ratio</u>			
enli		v/c Ratio	<u>Alt. 1 N</u>	o Action	Prefer	red Alt.	
Scre	<u>Location</u>	Threshold	<u>N/E</u>	<u>S/W</u>	<u>N/E</u>	<u>S/W</u>	
<u>2</u>	Magnolia	<u>1.0</u>	<u>0.51</u>	<u>0.54</u>	<u>0.53</u>	<u>0.54</u>	
<u>3.11</u>	Duwamish River—West Seattle Bridge and Spokane Street	<u>1.2</u>	<u>0.57</u>	<u>0.53</u>	<u>0.57</u>	<u>0.53</u>	
<u>3.12</u>	Duwamish River—1st Avenue S and 16th Avenue S	<u>1.2</u>	<u>0.53</u>	<u>0.52</u>	<u>0.54</u>	<u>0.52</u>	
<u>4.13</u>	South City Limit—SR 99 to Airport Way S	<u>1.0</u>	<u>0.47</u>	<u>0.50</u>	<u>0.47</u>	<u>0.51</u>	
<u>5.11</u>	Ship Canal—Ballard Bridge	<u>1.2</u>	<u>1.11</u>	<u>0.78</u>	<u>1.13</u>	<u>0.78</u>	
<u>5.12</u>	Ship Canal—Fremont Bridge	<u>1.2</u>	<u>0.68</u>	<u>0.68</u>	<u>0.69</u>	<u>0.69</u>	
<u>5.13</u>	Ship Canal—Aurora Bridge	<u>1.2</u>	<u>0.35</u>	<u>0.35</u>	<u>0.35</u>	<u>0.35</u>	
<u>7.11</u>	<u>West of Aurora Avenue—Fremont Place</u> <u>N to N 65th Street</u>	<u>1.0</u>	<u>0.55</u>	<u>0.64</u>	<u>0.55</u>	<u>0.65</u>	
<u>8</u>	South of Lake Union	<u>1.2</u>	<u>0.43</u>	<u>0.51</u>	<u>0.43</u>	<u>0.51</u>	
<u>9.12</u>	South of Spokane Street—E Marginal Way to Airport Way S	<u>1.0</u>	<u>0.51</u>	<u>0.49</u>	<u>0.51</u>	<u>0.49</u>	
<u>10.11</u>	South of S Jackson Street—Alaskan Way S to 4th Avenue S	<u>1.0</u>	<u>0.65</u>	<u>0.68</u>	<u>0.65</u>	<u>0.68</u>	

Exhibit 3.10-48 Screenline Volume-to-Capacity Ratio—Preferred Alternative

Source: Fehr & Peers, 2022.

<u>Transit</u>

Exhibit 3.10-49 summarizes PM peak hour average passenger load factors under the Preferred Alternative. The passenger load factors include both light rail and bus services. Similar to the other alternatives, the largest increases in passenger load would occur eastbound across the 8th Avenue NW screenline toward the University District, and southbound on 15th Avenue NW toward Downtown. These increases reflect the expected travel patterns of additional employees leaving the BINMIC area to travel home during the PM peak hour. Southbound travel demand across Lander Street and westbound demand across 8th Avenue NW would also increase slightly. Overall capacity across screenlines is expected to be adequate for the demand—some routes traveling across the study area screenlines, however, may operate over their crowding threshold for some individual trips. Although a minor increase is expected westbound across 8th Avenue NW (which is already expected to have crowded transit routes under Alternative 1 No Action), the magnitude of change is less than the threshold for a significant impact. Therefore, no transit passenger load impacts are expected under the Preferred Alternative.

	Alternative 1 No Action		<u>Preferred</u>	<u>Alternative</u>
<u>Screenline</u>	<u>Inbound</u>	<u>Outbound</u>	<u>Inbound</u>	<u>Outbound</u>
<u>A: East of 8th Avenue NW</u>	0.57	<u>1.28</u>	<u>0.64</u>	<u>1.31</u>
<u>B: Ballard Bridge</u>	<u>0.09</u>	<u>0.39</u>	<u>0.09</u>	<u>0.39</u>
<u>C: North of W Mercer Place</u>	<u>0.29</u>	<u>0.59</u>	<u>0.33</u>	<u>0.58</u>
<u>D: North of Lander St</u>	<u>0.21</u>	<u>0.75</u>	<u>0.21</u>	<u>0.76</u>
<u>E: West Seattle Bridge</u>	<u>0.12</u>	<u>0.35</u>	<u>0.12</u>	<u>0.35</u>

Exhibit 3.10-49 PM Peak Hour Average Passenger Load Factors—Preferred Alternative

Note: Inbound refers to travel into the downtown area and outbound travel out of the downtown area.

Source: Fehr & Peers, 2022.

Summary of Impacts

Exhibit 3.10-50 summarizes significant transportation impacts anticipated under each alternative. The purpose of this EIS is to disclose how potential actions by the City may impact the transportation system in comparison to what is expected to occur with currently adopted zoning codes and development standards. Therefore, the impacts of the Action Alternatives are assessed against Alternative 1 No Action. Impacts identified under Alternative 1 No Action would remain throughout the Action Alternatives even if those alternatives would not result in additional impacts. While the focus of the EIS is not to mitigate conditions under the currently adopted zoning code and development standards (i.e., Alternative 1 No Action), many of the mitigation measures identified for the Action Alternatives would also benefit conditions under Alternative 1 No Action.

In summary, Alternative 1 No Action is expected to have significant impacts to active transportation, auto, and freight in terms of travel time, mode share, transit, parking, and safety. Alternative 2 is expected to result in additional significant impacts to autos and freight on one corridor as well as impacts to active transportation, parking, and safety. Alternatives 3 and 4 are expected to result in additional significant impacts to auto and freight on <u>fourtwo</u> corridors and one mode share sector as well as impacts to active transportation, parking, and safety. <u>The Preferred Alternative is expected to result in additional significant impacts to auto and freight on two corridors as well as impacts to active transportation, parking, and safety. The locations of the corridors impacted by the Action Alternatives are mapped in **Exhibit 3.10-51** and **Exhibit 3.10-52**.</u>

Type of Impact	Alternative 1 No Action	Alternative 2	Alternative 3	Alternative 4	<u>Preferred</u> <u>Alternative</u>
Active Transportation	Yes	Yes	Yes	Yes	
Auto & Freight					
Travel Time	10 LOS F corridors	1 impacted corridor	3-<u>4</u> impacted corridors	3-<u>4</u> impacted corridors	<u>2 impacted</u> <u>corridors</u>
Mode Share	3 sectors	No	1 impacted sector	1 impacted sector	No
Screenline	No	No	No	No	<u>No</u>
Active Transportation	<u>Yes</u>	<u>Yes</u>	<u>Yes</u>	<u>Yes</u>	<u>Yes</u>
Transit	1 screenline	No	No	No	<u>No</u>
Parking	Yes	Yes	Yes	Yes	<u>Yes</u>
Safety	Yes	Yes	Yes	Yes	<u>Yes</u>

Exhibit 3.10-50-38 Summary of Significant Transportation Impacts

Source: Fehr & Peers, 20224.



Exhibit 3.10-51-39 Impacted Study Corridors—Ballard Interbay Northend MIC, 2044

Source: Fehr & Peers, 202<u>2</u>4.



Exhibit 3.10-52-40 Impacted Study Corridors—Greater Duwamish MIC, 2044

Source: Fehr & Peers, 202<u>2</u>4.

3.10.3 Mitigation Measures

The City of Seattle is committed to investing in supportive transportation investments to improve access, mobility, and safety to allow the industrial and maritime sector to strengthen and grow. Maintaining freight mobility is critical and requires both transportation infrastructure and transportation systems management in the MICs. Because many industrial and maritime-related trips will need more convenient transit services and alternative travel options in order to convert from remain as SOV due to the nature of the industry, reducing the SOV mode share for other types of trips is key to limiting the potential severity of transportation impacts. Lowering SOV mode share when possible would not only reduce travel time, mode share, and parking demand impacts, but is consistent with numerous other goals and policies in the Comprehensive Plan.

This section identifies a range of potential mitigation strategies that could be implemented to help reduce severity of the adverse impacts of the Action Alternatives identified in the previous section. These include impacts

Secondary Impacts

It should be noted that some transportation mitigation projects could have secondary impacts. For example, converting a general-purpose travel lane or a parking lane to a transit lane, truckonly lane, or cycle track would reduce capacity for autos to travel or park. As required, the City would prepare additional analysis and take public and stakeholder input into consideration before implementing specific transportation improvement projects. Given the programmatic nature of this study, this EIS simply lists the types of projects that could be considered to mitigate potential impacts of the proposed alternatives.

to active transportation, travel time along key arterial corridors, mode share, parking, and safety.

Incorporated Plan Features

The Action Alternatives propose three new land use concepts: Maritime, Manufacturing, and Logistics (MML), Industry and Innovation (II), and Urban Industrial (UI). Each concept includes <u>characteristics and/or</u> development standards, some of which would influence the transportation network and/or transportation behavior <u>in those areas</u>. These include:

- Standards for pedestrian and cyclist-oriented frontage improvements (sidewalks, pedestrian lighting, street trees, etc.)—Industry & Innovation and Urban Industrial
- Vehicle parking maximums and strong commute trip reduction program requirements— Industry & Innovation
- Proximity to a light rail station—Industry & Innovation

Regulations & Commitments

In addition to the development standards incorporated into the proposed land use concepts, the City of Seattle has numerous ongoing <u>plans and</u> strategies to support non-SOV travel modes and

increase the overall efficiency of the transportation system for all Seattle residents and employees. These strategies would be pursued as part of any of the future year alternatives. Strategies are discussed beginning with those expected to be most effective in mitigating impacts.

Transportation Systems Management and Operations (TSMO)

Transportation systems management and operations (TSMO) is a philosophy that encompasses strategies to optimize the existing transportation system by understanding the root causes of poor performance, improving collaboration, encouraging behavior changes through travel demand management, and using technology to manage how the system operates. TSMO strategies focus on cost-effective, near-term,

Project Highlight: East Marginal Way Corridor Improvement Project

The recently announced \$20 million federal grant for the East Marginal Way Corridor Improvement Project is an example of how TSMO strategies can be integrated with enhanced maintenance and safety projects. The grant will fund improvements including widening and strengthening the corridor to accommodate larger and heavier truck traffic; construction of dedicated space for people walking and biking along the corridor; and installation of more advanced traffic signals to reduce traffic congestion, particularly for freight accessing the Port.

multimodal improvements to better operate the City's infrastructure and systems.

Seattle has an ongoing program to improve the operations of traffic signals and provide drivers with more information about congestion and travel times in an effort to make more efficient use of the City's streets. The City will continue to implement new traffic signal systems, such as adaptive signal control which is already in place along the Mercer Street corridor and will soon be implemented along Denny Way. Adaptive signal control is a coordinated traffic signal system that gathers real-time vehicle demand data and dynamically adjusts signal timing to optimize traffic flow. These programs are designed to specifically reduce traffic congestion and improve freight and vehicle flow.

TSMO strategies can be targeted to high priority roadway users, including freight and transit. The *Transit Master Plan_*-and *Freight Master Plan_ and Seattle Industrial Areas Freight Access Project* identify speed and reliability improvements throughout the city that could benefit those particular modes. In particular, the Freight Master Plan identifies truck-only lanes on highly used truck routes as one potential strategy to improve freight mobility while the BIRT Study proposes joint-use Freight and Transit lanes along 15th Avenue W. SDOT is currently considering policy guidance on Freight-Transit Only Lanes and Truck-Only Lanes. Other potential strategies include:

- intelligent transportation systems (ITS) applications such as dynamic message signs to alert travelers to blocking incidents or give travel time information about route choices;
- truck-specific ITS notifications to inform truck drivers of incidents and major points of congestion;

- truck detection and signal priority to allow traffic signals to recognize an approaching truck so the green light may be extended to let the truck travel through the intersection (providing both freight mobility and safety benefits);
- wayfinding for trucks to improve route decisions and reduce illegal movements;
- geometric improvements at intersections to better design for key truck turning movements; and
- freight operations management to prioritize freight movements during certain times in certain locations.

Many of these<u>Some</u> types of improvements could be funded through the Move Seattle Levy which commits \$14 million over the nine-year life of the levy for the Freight Spot Improvements Project<u>while others would require partnering with regional and state agencies for comprehensive implementation</u>.

Freight Mobility and Access Strategies

Potentially significant impacts to freight mobility and access have been identified under all future year alternatives; all alternatives are expected to result in increased congestion affecting the roadway network as well as increased travel of other modes which may conflict with freight operator needs. To mitigate this impact, the City could pursue a variety of operational and capital projects aimed at addressing particular freight

Specific projects and high priority areas for improvement may be found in:

- Freight Master Plan
- Seattle Industrial Areas Freight Access Project
- Ballard-Interbay Regional Transportation (BIRT) System Report
- Georgetown Mobility Study

bottlenecks. The City can consider changing needs as new land uses develop and areas of need are identified.

The City has developed a citywide Freight Master Plan along with other studies addressing the MICs such as the Seattle Industrial Areas Freight Access Project, Ballard-Interbay Regional Transportation (BIRT) System Report, and Georgetown Mobility Study that propose a variety of projects that, if implemented, would improve freight mobility and access. Representative projects that could improve freight mobility and access include: truck-only or joint-use freight and truck lanes, rail corridor grade separation, intersection geometry improvements to address turn radii challenges for trucks, channelization improvements, signal phasing or timing modifications, wayfinding and signage, ITS strategies as described in the TSMO section, and dedicated pedestrian and bicycle facilities to separate vulnerable users from freight.

Travel Demand Management (TDM)

Managing demand for auto travel is an important element of reducing overall congestion impacts that affect auto, freight, transit, and parking demand. There are well-established travel demand management programs in place, including Transportation Management Programs (TMPs), the Commuter Benefit Ordinance, and the State's Commute Trip Reduction (CTR) program. Because CTR and TMPs typically focuses on large employers, the City could pursue expansions of those programs tailored to smaller employers and residential buildings or support the creation of Transportation Management Associations (TMAs).

A TMA is an organization that provides transportation services and information in a defined area (for example, an office or industrial park or a commercial district). TMAs are typically oriented around TDM programs and focused on commuters but can also serve shoppers, hospital visitors, or residents depending on the characteristics of area they serve and the needs of their members. In some cases, TMAs are developed to advance shared goals among members around sustainability, employee retention, and congestion management. Seattle currently has a TMA in the Downtown area (Commute Seattle) and previously had an active TMA in the Duwamish area (currently TDM programs and services are supported by the SODO Business Improvement Area). There is local precedent for compelling participation in a TMA through code requirements; however, to fully implement a robust TMA, this would also need to be paired with a budget action to establish a funding and governance structure.

Industrial areas can be challenging for TDM due to the characteristics of workers' schedules. For example, many workers need to commute during off-peak periods for their shifts when transit options are more limited and workers often live relatively far from worksites (see **Exhibit 3.9-11** and **Exhibit 3.9-13** for commute length data). Potential TDM measures suited to the study area could include last-mile shuttle systems between key transit nodes and the MICs; coordination with King County Metro and/or Sound Transit to provide off-peak transit service tailored to shift workers with irregular hours; subsidized vanpools; rideshare matching to limit the number of drive-alone commute trips; and micromobility options such as scooters or bicycles to make last-mile connections. In addition to addressing the unique needs of MICs in terms of commute timing, the City could also coordinate with King County Metro through their routine service planning process to explore adding transit service on corridors that serve many industrial and maritime workers.

The City could consider updating municipal code and/or Director's Rules related to Transportation Management Plans to tailor requirements for transportation demand management measures that are most effective in industrial settings. This may include membership in a TMA and discounted or free transit passes and/or car share and bike share memberships. For residential buildings, the City could also consider extending Transportation Management Plans or requiring travel options programs (such as GreenTRIP in California).

Research by the California Air Pollution Control Officers Association (CAPCOA), which is composed of air quality management districts in that state, has shown that implementation of travel demand management programs can substantially reduce vehicle trip generation, which in turn reduces congestion for transit, freight, and autos. Reduced auto travel can indirectly mitigate on-street parking impacts. The City could consider modifying specific measures described below or expanding current strategies. It should be noted that any changes to offstreet parking policies would be considered in consultation with stakeholders and in conjunction with improvements to make transit a more competitive option for workers.

- Parking maximums that would limit the number of parking spaces which can be built with new development.
- Review the parking minimums currently in place for possible revisions.
- Review on-street parking management strategies in concert with any adjustment to offstreet parking standards to reduce the impact of spillover parking.
- Unbundling of parking to separate parking costs from total property cost, allowing buyers or tenants to forgo buying or leasing parking spaces.
- Increased parking taxes/fees.
- Review and revise transit pass provision programs for employees.
- Encourage or require transit pass provision programs for residents—King County Metro has a Passport program for multifamily housing that is similar to its employer-based Passport program. The program discounts transit passes purchased in bulk for residences of multifamily properties.

Pedestrian & Bicycle System Improvements

Potentially significant impacts to active transportation have been identified under all future year alternatives because all are expected to result in more people walking and biking in areas with network gaps. To mitigate this impact, the City would need to improve the facilities provided for people walking<u>and</u> biking<u>and</u> rolling, with particular attention to areas that have safety concerns and areas of historic underinvestment. The City continually reevaluates

Specific projects and high priority areas for improvement may be found in:

- Pedestrian Master Plan
- Bicycle Master Plan
- Bicycle and Pedestrian Safety Analysis
- Ballard-Interbay Regional Transportation (BIRT) System Report
- Georgetown Mobility Study

its implementation prioritization and can consider changing needs as new land uses develop, for example to prioritize connections between bus stops/light rail stations and places of employment.

The City has developed a citywide *Pedestrian Master Plan* and citywide *Bicycle Master Plan* along with other subarea plans focused on particular neighborhoods. These plans and documents include myriad projects that, if implemented, would improve the environment for people walking and biking. Representative projects that could improve conditions for people walking and biking in the study areas include: facilities such as sidewalks, asphalt walkways, or painted walkways; signals to make crossing roadways easier; treatments such as rectangular rapid flashing beacons to alert drivers to people crossing the street; marked crosswalks; curb bulbs or extensions to shorten crossing distances and make people walking more visible to drivers; bicycle lanes (including protected and buffered bicycle lanes); and multi-use trails. This work will be refined and integrated into a single multimodal plan in the upcoming Seattle Transportation Plan which will include a holistic framework for system improvements. In addition, the City and Sound Transit are currently coordinating on transportation mitigation

around expanded and new light rail stations (coinciding with II zoning). While specific projects have not yet been identified, it is assumed that Sound Transit will be constructing improvements in the immediate vicinity of each station as part of their mitigation package. Additional improvements could also be implemented through Sound Transit's System Access Fund which awards funds to jurisdictions to design and construct improvements that make it easier and more convenient for people to reach transit. This could include capital projects such as sidewalks, bike lanes, shared use paths, transit integration, and pick-up/drop-off facilities,

SDOT also has ongoing safety programs that are aimed at reducing the number of collisions, benefiting both safety and reliability of the transportation system. Projects could be implemented through City-led efforts or in partnership with new development through the development review and permitting process.

In addition to creating a better connected and safer walking and riding environment <u>for those</u> <u>already using active transportation modes</u>, pedestrian and bicycle infrastructure investments would encourage <u>additional</u> travelers to choose walking<u>, or micromobility options</u> <u>such as scooters</u> rather than driving<u>, This creates the secondary benefit of</u> contributing toward mitigation of the mode share, travel time, and parking impacts.

Safety Strategies

Potentially significant impacts to safety have been identified under all future year alternatives due to the potential increase of collisions between trucks/autos/rail and vulnerable users, such as people walking, biking, or riding scooters. The pedestrian and bicycle system improvements described in the previous section would help to mitigate safety issues by providing dedicated facilities to separate vulnerable users from motorized traffic (particularly large trucks which inherently operate with higher-risk collisions) and/or adding design elements designed to make vulnerable users more visible to truck and auto drivers. These include: facilities such as sidewalks, asphalt walkways, or painted walkways; signals to make crossing roadways easier; treatments such as rectangular rapid flashing beacons to alert drivers to people crossing the street; marked crosswalks; curb bulbs or extensions to shorten crossing distances and make people walking more visible to drivers; bicycle lanes (including protected and buffered bicycle lanes); and multi-use trails. Projects pertaining to increasing safety at at-grade rail crossings could include: grade separation to avoid the modal conflict entirely and improvements to active warning devices such as bells, flashing lights or gates.

SDOT also has ongoing safety programs that are aimed at reducing the number of collisions, benefiting both safety and reliability of the transportation system. Projects could be implemented through City-led efforts or in partnership with new development through the development review and permitting process.

Parking Strategies

While parking demand varies throughout the study area, there are some localized areas where on-street parking demand exceeds parking supply, particularly demand for truck parking and

near commercial nodes and activity centers such as the Ballard brewery area and businesses along 1st Avenue South. Because the Action Alternatives are expected to increase demand in localized areas that already exceed supply, potentially for a sustained period and by a substantive amount compared to Alternative 1 No Action, significant adverse parking impacts are expected under all of the Action Alternatives. Impacts are expected to be greater under alternatives 3 and 4, which have higher levels of development planned than Alternative 2 or the Preferred Alternative.

The City has multiple ongoing programs to manage on-street parking including the Community Access and Parking Program, Performance-Based Parking Pricing Program, and Restricted Parking Zone (RPZ) Program. These approaches could be modified and/or applied at the neighborhood level to manage the increased demand for the city's limited parking supply. <u>The SDOT Curbside Management Team actively identifies and installs commercial vehicle and</u> general load unload zones in business districts throughout Seattle and would identify load zone needs with new development as needed or requested by development projects. SDOT is also working on potential policy changes to more actively install load zones and other curb access needs at new development during the City development review process.

SDOT's Community Access and Parking Program works with community members to identify parking challenges and opportunities within a neighborhood and implement changes. Parking recommendations could include new time-limit signs, load zones, paid parking, restricted parking zones, bicycle parking, or other changes.

The City is expected to continue managing on-street paid parking through SDOT's Performance-Based Parking Program which evaluates data to determine if parking rates, hours of operation and/or time limits could be adjusted to achieve the City's goal of one to two available spaces per block face throughout the day. The City could continue to manage on-street paid parking through existing programs, redefine subareas and manage them with time-of-day pricing, and/or institute paid parking in new areas.

The study area does not have any current RPZs defined. However, if SDOT determines a RPZ would be a beneficial tool to manage parking demand as growth continues, one or more RPZs could be created. RPZs have typically been implemented in residential neighborhoods where there is high parking demand generated by a use such as a business district, hospital, or school; RPZs allow short-term parking for customers or visitors but limit long-term use by employees or commuters. Within the context of the alternatives considered in this EIS, this situation would be most likely to arise in the denser, mixed-use Industry & Innovation and Urban Industrial zones that are located near light rail stations and/or urban villages. Subsequent management changes could include splitting existing RPZs into multiple zones, adding new RPZs, or adjusting RPZ boundaries. The City could also review the RPZ program and its policies in areas that are oversubscribed (where there are more permits issued than parking spaces) to limit the number of permits issued.

Truck parking management could require a complementary set of strategies. The City is also actively engaged in addressing truck parking needs in partnership with the Port of Seattle; the

City and Port of Seattle entered into an agreement in November 2021 to identify 200 on-street and/or off-street truck parking spaces for drayage drivers by the end of 2023. While this effort is focused on accommodating existing needs, continuing this partnership is a strategy to continue to address truck parking needs as they evolve.

WSDOT held a Washington State Truck Parking Workshop in June 2021 which resulted in development of potential solutions for future consideration.¹⁸ Among the strategies that could be applicable to the study area are:

- Encourage private businesses to develop truck parking in key areas.
- Investigate capacity access at unique facilities and partner with different stakeholders to foster this (examples: use of sports stadium, music venue, fairground, or boat launch parking lots when they are not in use).
- Determine future needs and opportunities for electric truck charging facilities. Looks for opportunities to increase parking availability when building out charging infrastructure.
- Incorporate truck parking into zoning codes and growth management policies. Develop minimum standards for truck and delivery vehicle staging for new developments.
- Assess targeted exceptions to land use provisions where suitable urban sites could be purchased by the state or a private company and used for truck parking.
- Partner with WSDOT and University of Washington on their Truck Parking Information and Management System (TPIMS) pilot project which allows information to be shared with truck drivers about where parking is available.

State Highway System

As described earlier, WSDOT considers any proposal that meets or exceeds a certain threshold of vehicle trips on state highway intersections and/or segments to have a probable significant adverse impact to the state highway system. The City of Seattle will coordinate further with WSDOT to determine how this threshold may be considered in the individual project review process to determine impacts to the state highway system of any specific development proposal.

¹⁸ WSDOT, 2021. 2021 Washington State Truck Parking Workshop, Overview and potential solutions for consideration. Available at: https://wsdot.wa.gov/sites/default/files/2021-12/Synopsis-2021-WA-Truck-Parking-Workshop.pdf

Potential Mitigation Measure Funding Options

Programs like the City's Business Improvement Area (BIA) are possible models for future funding sources. A BIA is an organization funded by property owners and businesses within a local district to collectively fund the maintenance and improvement of their area. There are currently ten BIAs established in the city, including the SODO and Ballard neighborhoods. BIAs can help to fund and promote TMAs that focus on tailored TDM strategies for the local context.

Through the Department of Construction and Inspection's permitting processes, the City can negotiate a proportional share developer contribution toward multimodal transportation improvements needed to mitigate impacts of the project. Given the temporal travel characteristics of industrial land uses (not necessarily following a conventional peak period travel pattern), a proportional share could be estimated based on the expected daily trips of the project.

To support delivery of multimodal projects, the City of Seattle could also implement a Growth Management Act (GMA) compliant multimodal Transportation Impact Fee (TIF) program. The City has already done some initial research into what a program could look like, including consideration of the projects it could fund, how to consider growth, and how development projects' impacts could be measured. Some of the initial findings include that a multimodal TIF program in Seattle could help fund a project list that includes complete streets, transit supportive infrastructure, freight network improvements, and investments to create a more complete network for walking and biking. To align with City's mode-share level of service policy, the TIF program would likely be based on person trips rather than vehicle trips given the strong nexus between new development and the need to expand the City's multimodal transportation network. To implement the program, the City would need to complete a rate study establishing a nexus between the impact fee project list and rates charged and the City Council would need to adopt an impact fee ordinance and associated code language that directs how impact fees would be assessed and spent. RCW 82.02.050–.110 and WAC 365-196-850 provide direction for how counties, cities, and towns planning under the Growth Management Act (GMA) can impose impact fees.

Other Potential Mitigation Measures

Location-specific mitigation measures are discussed for the following two travel time corridor and transit screenline impacts:

- 15th Avenue W between Magnolia Bridge and NW Leary Way
- W Dravus Street between 15th Avenue W and 20th Avenue W

Travel Time Impact: 15th Avenue W between Magnolia Bridge and NW Leary Way

A travel time impact is expected along 15th Avenue W between Magnolia Bridge and NW Leary Way under both alternatives 3 and 4. The BIRT Study analyzed the 15th Avenue NW corridor in detail and outlines potential investments, some of which would mitigate the travel time impacts. The scale of each project's potential efficacy in improving the transportation system is evaluated as either transformative or small. These include:

- Intersection operations refinements along 15th Avenue W at W Armory Way, Gilman Drive W and W Howe Street (transformative). This would include improvements such as turning radii adjustments to better accommodate frequent freight turning movements and signal phasing adjustments to shorten the amount of time needed for traffic flow crossing the 15th Avenue W corridor.
- Installation of an adaptive signal system along the corridor (transformative). Adaptive signal control is a coordinated traffic signal system that gathers real-time vehicle demand data and dynamically adjusts signal timing to optimize traffic flow.
- Joint-use of the existing bus-only lanes by both transit and freight on 15th Avenue W between Denny Way and Market Street during off-peak times (small). The City is currently planning a pilot project for Freight and Bus Lanes on Westlake Avenue which will provide information about benefits and implementation elsewhere, such as the 15th Avenue NW corridor.
- Replacement of the Ballard Bridge to improve northbound traffic flow (transformative). There are currently two options under consideration: a mid-level and a low-level replacement. The mid-level bridge would reduce the frequency of bridge span openings making travel times across the bridge more reliable and shorter on average while the low-level option would provide an easier grade for people walking and biking. Both options would include a Single Point Urban Interchange (SPUI) at W Nickerson Street/W Emerson Street which would improve travel time reliability for trucks entering and exiting the BINMIC.

Travel Time Impact: W Dravus Street between 15th Avenue W and 20th Avenue W

A travel time impact is expected along W Dravus Street between 15th Avenue W and 20th Avenue W under alternatives 2, 3, and 4. The BIRT Study outlines potential investments along the W Dravus Street corridor, some of which would mitigate the travel time impacts. These include:

- Signal operations improvements and ITS strategies (small). This could include optimizing traffic signal timing along W Dravus Street to support both general purpose traffic and freight reliability to and from the Terminal 91 North Gate if it reopens. Signal timing and hardware improvements at the 15th Avenue W and W Dravus Street ramps could also ensure vehicle queues on the bridge have cleared to give trucks adequate space to turn, minimizing the delays currently experienced at this location.
- Roadway striping/channelization modifications to remove geometric constraints for large trucks (small). This would include improving the turn radii at 15th Avenue W and W Dravus Street so trucks could more easily make the turn to and from the ramps, minimizing the delays currently experienced at this location.
- Access management enhancements at frequent and busy driveway access points (small).
- Replacement and/or widening of the W Dravus Street bridges (transformative). Options could include roadway rechannelization, conversion to a roundabout at 17th Avenue W, and/or widening the Dravus Street bridge west of 17th Avenue W.
Travel Time Impact: I-5 Between Madison Street & SR 599 <u>& SR 509 Between SR 99</u> <u>& SR 518</u>

A travel time impact is expected along I-5 between Madison Street and SR 599 (stretching along the east side of the Greater Duwamish MIC) <u>and SR 509 between SR 99 and SR 518</u> under alternatives 3 and 4. While the City of Seattle works closely with WSDOT regarding facilities running through the city limits, I-5 <u>and SR 509 are-is</u> owned and operated by the State. In 2019, WSDOT and the City of Seattle jointly applied for a federal grant to move planning efforts for the I-5 system forward; however, the project was not awarded any funding at that time. Both agencies continue to work toward securing funding for I-5 improvements, as well as coordinate with the PSRC on potential approaches to address congestion on regional highways. However, for the purposes of this EIS, no location-specific capital improvement-based mitigation measures are assumed that would address travel time impacts along I-5 <u>or SR 509</u>.

Regarding land use mix and trips, under alternatives 3 and 4, the City could consider the balance of employment uses and plan for greater industrial jobs, and a smaller share of non-industrial jobs (e.g., retail, services, office) in the Greater Duwamish MIC to reduce trips. The City could consider a preferred alternative that has less of the employment-dense Industry and Innovation zone than is found in alternatives 3 and 4 but more than Alternative 2 and still avoid significant adverse impacts on I-5.

The Preferred Alternative (developed based on feedback regarding potential impacts of the Draft EIS alternatives) would have less employment density than alternatives 3 and 4. The land uses proposed under the Preferred Alternative were analyzed using the regional travel demand model, which suggests there would be no significant travel time impacts to either I-5 or SR 509 under the Preferred Alternative.

3.10.4 Significant Unavoidable Adverse Impacts

This section describes the significant and unavoidable adverse impacts to transportation that would occur as a result of implementation of the Action Alternatives. Travel demand and associated congestion is expected to increase over time regardless of the alternative pursued. In addition to citywide transportation capacity improvements largely focused on improved transit, bicycle, pedestrian, and freight connections, the City will manage demand using policies, programs, and investments aimed at shifting travel to non-SOV modes. However, city streets will remain congested during peak periods as growth continues to occur. With respect to the three Action Alternatives studied in the<u>is</u> Draft EIS and the Preferred Alternative studied in the <u>Final EIS</u>, potentially significant adverse impacts are identified for active transportation, corridor travel times (affecting autos, freight, and buses), mode share, on-street parking, and safety.

Potential mitigation measures for the 15th Avenue W and W Dravus Street corridors impacted by the Action Alternatives are proposed above. If these measures are implemented, it is expected that the travel time impact could be brought to a less-than-significant level in relation to Alternative 1 No Action. At this time, no location-specific mitigation measures along I-5 <u>or SR</u> <u>509</u> are expected to fully mitigate the travel time impact to autos, freight, and buses under alternatives 3 and 4. <u>However, the land use Mm</u>odifications to alternatives 3 and 4. <u>However, the land use Mm</u>odifications to alternatives 3 and 4. <u>However, the land use Mm</u>odifications to alternatives 3 and 4. <u>However, the land use Mm</u>odifications to alternatives 3 and 4. <u>However, the land use Mm</u>odifications to alternatives 3 and 4. <u>However, the land use Mm</u>odifications to alternatives 3 and 4. <u>However, the land use Mm</u>odifications to alternatives 3 and 4. <u>However, the land use Mm</u>odifications to alternative 3. <u>Alternatives 3 and 4</u>. <u>However, the land use Mm</u>odifications to alternatives 3. <u>Alternatives 3 and 4</u>. <u>Therefore the Preferred Alternative are expected to reduce the total amount of future employment in the SODO subarea could potentially mitigate the impact to I-5 and SR 509 <u>-</u>. <u>Therefore, a significant travel time impact may be avoided on I-5 if the reduction in trips brings travel time increases</u> below the threshold of significance.</u>

Some combination of the travel demand management strategies discussed in **3.10.3 Mitigation Measures** could be implemented to reduce the magnitude of SOV travel. Given the small magnitude of difference projected between Alternative 1 No Action and alternatives 3 and 4, it is expected that the mode share impact could be reduced to a less-than-significant level. The land use modifications proposed for the Preferred Alternative are expected to mitigate the mode share impact below the threshold of significance.

Parking impacts are anticipated to be brought to a less-than significant level by implementing a range of possible mitigation strategies such as those discussed in **3.10.3 Mitigation Measures**. While there may be short-term impacts as individual developments are completed (causing onstreet parking demand to exceed supply), it is expected that with <u>new strategies for truck</u> <u>parking accommodation</u>, expanded paid parking zones, revised RPZ permitting, more sophisticated parking availability metrics, and continued expansion of non-auto travel options, the on-street parking situation <u>will-would</u> reach a new equilibrium as residents, employees, and visitors adjust to the new context. Therefore, no significant unavoidable adverse impacts to parking are expected.

Significant impacts were identified to both active transportation and safety due to the projected increase in people walking<u>, and</u> biking<u>, and rolling</u> in areas with network gaps and the increased potential for vehicle conflicts (particularly trucks) <u>and rail</u> with vulnerable users. While the City can pursue a variety of mitigation measures to improve <u>active transportation</u> facilities for people walking and biking and pursue supplemental funding through federal or state programs, it is not expected that all network gaps can be addressed given the number of locations needing improvement and the limited funding available. Therefore, it is expected that the Action Alternatives could have significant unavoidable adverse impacts to active transportation and safety.

Section 3.11

Historic, Archaeological, & Cultural Resources



This section details the current historic, archaeological, and cultural resources policy and regulatory frameworks, describes the current conditions (affected environment), analyzes the alternatives' potential impacts on historic, archaeological, and cultural resources, and suggests possible mitigation measures. Finally, it summarizes any significant unavoidable adverse impacts.

Impacts of the alternatives on historic, archaeological, and cultural resources are considered significant if they result in:

 Substantial adverse changes to, alteration, or loss of a resource that impacts its eligibility for inclusion in the National Register of Historic Places (NRHP) or the Washington Heritage Register (WHR), or as a City of Seattle Landmark (SL). Resources that are not eligible for these registers will not be adversely impacted by the proposed alternatives.

3.11.1 Affected Environment

This section describes the prehistoric, ethnographic, and historic contexts of the areas within the MICs as background by which to address the potential for impacts to historic, archaeological, and cultural resources.

Precontact Period Context

Based upon current scientific understandings of the archaeological record, the earliest human occupations in the Pacific Northwest were characterized by highly mobile bands of broad-spectrum foragers. The widespread Clovis culture, the first well-defined cultural complex in North America, has been dated to between 12,800 and 13,200 calibrated years before present (cal. B.P.) (Ames and Maschner 1999:65–66; Kirk and Daugherty 2007:13). Recent research suggests that large stemmed projectile points (i.e., Western Stemmed complex) may have been produced by populations pre-dating Clovis (e.g., Jenkins et al. 2012). These early Paleoindian cultures consisted of small, nomadic bands that specialized in hunting a variety of small- to large-sized game animals, including megafauna that went extinct across North America at the end of the Pleistocene (e.g., wooly mammoth [*Mammuthus primigenius*], mastodon [*Mammut americanum*], ancient bison [*Bison antiquus*]) (Kirk and Daugherty 2007:13).

Following the Clovis period, early and middle Archaic populations across western Washington produced large, willow leaf-shaped ("Olcott" phase) projectile points, in addition to lanceolate points and scrapers (Ames and Maschner 1999; Kopperl et al. 2016; Nelson 1990:483). Similar projectile points have been found in sites from the Fraser River Valley in British Columbia down to the margins of the Columbia River, indicating the wide dispersal of related groups across the broader Northwest Coast during this period. Sites containing Olcott material are most commonly documented well inland from the coast along rivers, suggesting that these populations were likely still subsisting largely upon terrestrial plant and animal resources and had not yet developed the extensive reliance upon riverine and coastal food resources observed among later Coast Salish peoples (Kopperl et al. 2016; Nelson 1990:483).

Between approximately 6400 and 2500 cal. B.P., there was a gradual shift across the Northwest Coast to an increasingly heavy reliance on marine and riverine resources for subsistence. This shift was coincident with a general trend toward increasing sedentism as more sites were settled along river courses, estuaries, and productive marine environments (Ames and Maschner 1999:93–94; Nelson 1990:483). During this period, settlements began to be occupied on a seasonal basis. Larger, denser artifact concentrations have been identified within sites dating from 6400 to 2400 cal. B.P., and deep shell middens have been dated to as early as 5,200 years ago (Larson and Lewarch 1995; Mierendorf 1986:57; Wessen 1988). It was during this time that coastal and neighboring inland communities developed their complex suites of lithic, bone, and antler tool technologies suited for marine mammal hunting, riverine fishing, and the further exploitation of terrestrial plant and animal resources (Ames and Maschner 1993:93–95; Blukis Onat et al. 1980:29–30; Kopperl et al. 2016:117–118).

Along with steady population growth and increasingly intensive resource utilization across the broader Northwest Coast, Late Pacific (2400–200 cal. B.P.) precontact archaeological sites in the region demonstrate the emergence of status differentiation and complex social hierarchies (Ames and Maschner 1999:95–96). Increased reliance on stored foods and controlled access to resources, including salmon and shellfish, also developed during this period. By this time, the general ethnographic (prior to Euroamerican influence) pattern observed along the Northwest Coast had become well-developed, although these societies saw swift and dramatic changes with the arrival of Euroamerican explorers, traders, and settlers beginning in the late 1700s (Ames and Maschner 1999:95–96, 112).

Ethnographic Background

This section presents an Ethnographic Background prepared by Historic Resources Associates to provide context for Historic, Archaeological, & Cultural Resources evaluated in this EIS. See **Section 3.8 Land & Shoreline Use** for an overview of historical planning and land use decisions developed by the City of Seattle in support of this EIS.

The EIS study area is within the traditional territory of the Lushootseed-speaking Duwamish people. The settlements of this ethnographically documented Coast Salish group were principally located along the Duwamish, Black, and Cedar Rivers, as well as along the coasts of Puget Sound and Lake Washington in the vicinity of present-day Seattle (Duwamish Tribal Services 2018; Ruby and Brown 1992:72). The Duwamish were part of the broader Southern Coast Salish culture, which was generally adapted toward the intensive utilization of marine and riverine resources (Suttles and Lane 1990). A principal division among the Duwamish existed between the SxwaldjaÉbc ("saltwater dwellers") who lived in settlements on Puget Sound and the XatcuaÉbc ("lake dwellers") who lived along the shores of Lake Washington. The latter, as well as Duwamish groups living along the interior rivers of the region, were considered to be poorer and lower-status than the coastal communities (Hilbert et al. 2001:45; Ruby and Brown 1992:72–73; Suttles and Lane 1990:485–486; Swanton 1952:26).

Like other Southern Coast Salish peoples, the Duwamish relied heavily upon salmon and other fish for subsistence and utilized a diverse suite of technologies to harvest them in different settings. They made use of trolling, seine, and gill net technologies to harvest fish in Puget Sound, while weirs, nets, gaff hooks, harpoons, and spears were all employed in rivers (Suttles and Lane 1990:488–489). Terrestrial mammals, especially black-tailed deer and elk were also hunted by the Duwamish and neighboring Tribes using the bow and arrow, and they gathered a great variety of plant foods, including edible roots, bulbs, and berries (Duwamish Tribal Services 2018; Gunther 1945; Suttles and Lane 1990:489).

The Duwamish lived a semi-sedentary lifestyle, spending part of the year in permanent winter villages and the warmer months in temporary encampments from which they fished, hunted, and gathered plant resources. Smaller bands would travel across their territory to hunt and forage for plant resources during the summer months, returning to their permanent settlements for the ceremonially rich winter season and to intensively fish in the spring and autumn (Duwamish Tribal Services 2018; Suttles and Lane 1990).

In 1855, members of the Duwamish and neighboring Puget Sound tribes signed the Treaty of Point Elliott, which directed the removal of Tribal members to reservations. The Duwamish were ordered to relocate to the Port Madison Reservation, along with the Suquamish (Lane 1975:3–4). Many Duwamish remained along the Black River in defiance of government orders but were removed by the early 1900s (Lewarch et al. 1996:3–13). The Duwamish Indian Tribe petitioned for federal recognition in 1979. In 2001, the federal government rejected the petition, reversing the decision of the previous administration to recognize its Tribal status. The Duwamish Indian community continues to pursue recognition, build their community, and maintain their cultural traditions (Duwamish Tribal Services 2018; Thrush 2007:196–197).

An important Duwamish village, šalšúl ("Tucked Away Inside"), was located at the west end of the Ballard portion of the Ballard Interbay Northend Manufacturing Industrial Center (MIC). The village site was situated along the northwestern shore of Salmon Bay and was destroyed during the construction of the Hiram M. Chittenden Locks in the 1910s (Hilbert et al. 2001:54-55; Thrush 2007:221–223). Several Duwamish villages were recorded within the Greater Duwamish MIC around the former mouth and lower reaches of the Duwamish River. The village of yalíq^wad ("basketry cap") was named for the distinctive woven hats worn by peoples such as the Yakama, perhaps because its residents participated in trade networks that spanned the Cascades (Dailey 2020; Hilbert et al. 2001:119; Thrush 2007:236-237). This village was located along the west bank of the Duwamish River west of Kellogg Island. Site 45KI23 (the Duwamish No. 1 Site) has been identified at this location, and likely represents the archaeological remains of the village. The village of tú?ul?altx^w ("where herring live" or "herring house"), was situated to the west of the mouth of the Duwamish River under the West Seattle bluff. An unknown Euroamerican settler burned the town down in 1893, and its name was eventually given to the Terminal 107 Park (Hilbert et al. 2001:46; Thrush 2007:234). A third village, dx^wq^wítad ("Place of the Fish Spear"), was located atop a large flat next to the Duwamish River at what is presently the north end of Boeing Field (Hilbert et al. 2001:47; Thrush 2007:240).

Historic Period Context

Early Settlement

European visitation to the Puget Sound Region began in 1792 when George Vancouver and his crew explored the region. Within the next 100 years, Native populations plummeted due to repeated outbreaks of introduced diseases such as smallpox, influenza, and typhoid fever (Boyd 1990; Suttles and Lane 1990). The Treaty of Washington in 1852 conveyed the territory to the United States, and the Donation Land Claim Act drew settlers into land occupied by the Duwamish and their neighbors. In 1855, members of the Duwamish and neighboring Puget Sound tribes signed the Treaty of Point Elliott, which provided for the removal of Tribal members to reservations, including the Port Madison Reservation (Suquamish/Fort Kitsap). Some Duwamish people continued to live in and around Seattle, maintaining friendly relations, working for, and trading with incoming settlers. Many others, meanwhile, relocated to the Port Madison Reservation, but due to undesirable conditions were compelled to leave. Many then attempted to return to their ancestral lands, and a few were able to claim or purchase land (Ruby and Brown 1992; Thrush 2007).

Tribal lands and fishing rights continued to be eroded through the late 1800s and 1900s, culminating, in the late 1900s, in a series of lawsuits and court cases that upheld certain treaty rights (Marino 1990; Ruby and Brown 1992). The federally-recognized Muckleshoot, Snoqualmie, Suquamish, and Tulalip Tribes are the descendant Tribes that represent the various tribes and bands with territorial interests in the portion of Seattle addressed by this EIS, that were signers of the Point Elliott Treaty. The Duwamish Tribe is not currently federally recognized but continues to fight for this distinction. See **Section 3.8 Land & Shoreline Use** for related information on historical planning and land use decisions developed by the City of Seattle in support of this EIS.

It was in 1851 that the first Euroamerican settlers arrived in what is now the Seattle area. They were the Denny Party, which included Arthur A. Denny and his brother David T., John N. Low, Carson D. Boren, William N. Bell, Charles C. Terry and his brother Lee, and their families (Denny 1888:7–13, 16–17; Fiset 2001; USSG 1856, 1863). These early settlers encouraged additional settlement by adjusting their claims to accommodate new arrivals, such as sawmill owner, Henry L. Yesler in 1852, and filed the first plat for the town of Seattle. Logging, which began with local men working with oxen and small timber mills, became the primary industry of this period (Caldbick 2014; Denny 1888:16–22; Fiset 2001). Over time, larger mills were constructed in the area and the industry offered steady employment for incoming settlers (Sanborn Map Co. 1884, 1888, 1893).

To the north, Dr. Henry Smith with his wife, mother, and sister settled in the Interbay area in 1853 and filed for land claims. More settlers followed, made claims, and supported themselves by farming and logging (Wilma 2001a). To the south, Luther Collins, Jacob Maple, Samuel Maple, and Henry Van Asselt settled along the Duwamish River on lands that now make up Georgetown, with farming the main industry in this area (Wilma 2001b).

By 1860, the population of settlers in Seattle had risen to 302, and many of them were working to grow the town into something more substantial. While most of the industry and commercial activity had grown along the eastern shore of Elliott Bay, sparse residential and family farms were beginning to pop up in the areas surrounding Seattle's central core (Bagley 1929).

Maritime Commerce & Industrial Development

In the 1870s, the discovery of large deposits of coal near present-day Newcastle and Renton, created a need for transportation to Seattle docks on Elliott Bay. Initially, the coal was transported on barges across Lake Washington, then unloaded to wagons and transported overland to Lake Union, where it would be loaded back onto barges and shipped southwest across the lake. Then the coal was once again unloaded onto wagons for the final leg of the route to Elliott Bay. In an attempt to simplify this onerous shipping system, a narrow-gauge rail line was constructed in 1872 between Lake Union's south shore to the coal dock on Elliott Bay. Five short years later, the line was abandoned as the Seattle and Walla Walla Railroad (S&WW) was constructed by the enterprising locals in Seattle from Elliott Bay south to the coal fields near Renton and then north to those near present-day Newcastle (Link 2004:3; MacIntosh and Crowley 1999). In 1884, the Northern Pacific Railroad built its line to Seattle, spurring additional growth (Chesley 2009).

Seattle's economy boomed with shipping, railroads, timber extraction and milling, coal mining and shipping, commercial and industrial manufacturing such as iron works, and service industry support. At this time, Seattle's economy was closely tied to other Pacific ports, especially those in California. At various times, a substantial percentage of lumber shipped from Seattle went to San Francisco to aid in its reconstruction from catastrophic fires and, later, the 1906 earthquake that was accompanied by a fire that destroyed some 25,000 buildings. The close connection between these ports can be seen in the creation of Ballast Island, an artificial landform on the Seattle waterfront, that is largely made up of rock mined from outcrops in San Francisco and dumped in Elliott Harbor to make space for the Seattle products shipped in return sailings. This rise in production created jobs and encouraged population growth.

In response to Seattle's growth, the pace of construction in the surrounding neighborhoods began accelerating in the late 1880s and early 1890s. Mills and other commercial ventures were built on the available lands, existing lumber mills and manufacturing companies expanded, and support services such as restaurants, hotels, breweries, laundries, creameries, soap works, and other similar enterprises were established throughout the neighborhoods. As well, houses were constructed to accommodate increasing numbers of employees, both management and labor, and business owners (Fiset 2001; Sanborn Map Co. 1884, 1888, 1893). Cable cars and electric streetcars crisscrossed Seattle's neighborhoods, ferries transported passengers across Lake Union, and systems of staircases, first constructed of wood and later of concrete, were built for ease of travel over the area's hilly topography (Fiset 2001; Thompson and Marr 2013). According to Sanborn maps, in 1884 the population of Seattle was 7,000 persons; this number more than doubled by 1888 to 16,000 (Sanborn Map Co. 1884, 1888).

Like many cities in the United States, Seattle was devastated by fire. The Great Seattle Fire occurred in 1889 and leveled the city's 18-block waterfront and 40 blocks of the city center. Destroyed were not only wood-frame buildings and structures, but those constructed of brick and stone, including wharves, piers, depots, mills, warehouses, businesses, offices, banks, stores, hotels, apartment buildings, and some residences. Rebuilding began almost immediately. The City widened some streets and raised others, implemented a new building code, banned wood buildings in the fire zone, and established a city water works (Caldbick 2020a, 2020b). Many of Seattle's sawmills that had been destroyed in the fire moved north to the north side of Salmon Bay, to what is now Ballard (Wilma 2001a).

After the fire, in the 1890s, the Great Northern Railway Company's president, James J. Hill, constructed docks, a grain terminal, grain elevator and warehouse at Smith Cove to facilitate maritime commerce with the Far East. Other private docks and warehouses were also built in the area (McClary 2013).

Around the turn of the twentieth century, construction in Seattle's neighborhoods included educational buildings, religious facilities, and multi-unit apartment buildings in support of the rapidly expanding population (Baist 1905; Fiset 2001). Additionally, religious organizations, commercial enterprises, and industrial operations were upgrading their wood-frame buildings with more substantial masonry versions in the wake of the fire (Link 2004:6). Industry boomed as well, spreading north and south of Seattle to more accommodating topography and expansive rail and waterway transportation systems (Langloe 1946). Private wharves, piers, warehouses, and mills were built south of the city, many were linked to the Northern Pacific lines to handle freight shipped into and out of Seattle. During this time, Georgetown's identity as Seattle's party area began to shift towards industry, especially after annexation by Seattle. By 1904, the population of Seattle had swelled to over 150,000. This number tripled to 456,000 by 1928 (Sanborn Map Co. 1905, 1928; Wilma 2001b).

The onset of the 1910s saw big changes for Seattle's maritime and industrial services. Between 1912 and 1917, the U.S. Army Corps of Engineers (USACE) constructed a canal between Puget Sound and Lake Washington following Ross Creek, which had been widened ca. 1885 for use as a log canal (Chrzastowski 1983:6). The Chittenden/Ballard Locks was completed in 1917, opening a major shipping route that connected Lake Washington, Lake Union, and Salmon Bay Waterway to Puget Sound. The project was funded by King County and the federal government. Simultaneous to the construction of the Canal, the City of Seattle completed bridge construction, street grading, and built the Third Avenue West Tunnel to provide a route for utilities to pass under the new Canal (Fiset 2001; Walton Potter 1977:12).

Other large projects during that time included the flattening of Denny Hill and streets north of downtown Seattle, known as regrades, which allowed for easier transportation routes in and out of the city (Link 2004:8). Much of the earth removed in the regrades was used to fill in wetlands and tidal flats. In 1912, the Great Northern docks at Smith Cove were sold to the newly created Port of Seattle for construction of a deep-sea terminal. The Port's comprehensive plan also included the construction of Fisherman's Terminal on Salmon Bay, the Bell Street Pier,

wharves and warehouses on the East Waterway pier and a second pier on the East Waterway, a public wharf and warehouse at the end of Bell Street, a grain elevator at Hanford Street, and a new ferry service on Lake Washington (Oldham 2020).

Additionally, man-made alterations along the Duwamish River—rerouting, straightening, and channelizing the river, and draining, dredging, and filling tidelands—and extensive logging, created land for agriculture and industry. The dredged material was used to construct Harbor Island, that split the mouth of the river into two channels. The Port of Seattle would later plan extensive terminals on Harbor Island. The renamed Duwamish Waterway supported shipping and large industrial complexes, such as shipbuilders, foundries, clay and coal plant, terracotta factory, antimony smelting and refining plant, iron works, flour mill, meat packer and slaughterhouse, creosoting works, lumber mills, warehouses, and Boeing Company's Plant 1 (Oldham 2020; Sanborn Map Co. 1905, 1928, 1950; Updegrave 2016). This industrial growth created additional employment opportunities and additional residences and apartment buildings were constructed to house the influx of workers (Sanborn Map Co. 1905, 1928).

Like most of the United States, the Great Depression hit Seattle hard, as the area's industries faltered, jobs were lost, and subsequently, the population fell (Fiset 2001; Link 2004:13). The arrival of World War II and the corresponding growth in war supporting industries slowed the decline. During this time, the city's earliest residential neighborhoods were in flux due to pressure of commercial and industrial interests. Additionally, the 1949 earthquake, which damaged numerous buildings, hastened the shift away from mixed residential and commercial neighborhoods towards those with a mix of commercial and industrial (Thompson and Marr 2013).

The gradual rebuilding began in the late 1950s, in part stimulated by the rezoning of the some of Seattle's neighborhoods to general manufacturing (Link 2004:14). Years in the planning, in 1959 work began on U.S. Interstate 5 (I-5) through Washington. The freeway aligned north–south along the east side of Eastlake Avenue E, cutting many neighborhoods in half, disrupting traffic patterns and routes, and introducing visual and auditory impacts. Much of I-5 through Seattle was completed in 1967, but the entire I-5 project was completed in 1969 (Dougherty 2010).

Although Seattle began as a sparsely populated region whose settlers supported nearby lumber mills, by the turn of the twentieth century, it had become the Pacific Northwest's powerhouse city with considerable commercial, transportation, industrial, and maritime industries. Today the city is home to modern hi-tech, retail, commercial, and multi-family infill construction in villages. While some single-family homes and small commercial ventures make way for denser urban infill most of the city's acres are still in low density residential use.

Current Conditions

Data & Methods

To analyze historic and cultural resources in the study areas for the purposes of this report, HRA's GIS Specialist gathered building data from the King County Assessor's website and the Department of Archaeology and Historic Preservation's (DAHP's) online database, the Washington Information System for Architectural and Archaeological Records Data (WISAARD), for cultural resource survey reports, archaeological site records, historic property inventory forms (HPIs), cemetery records, and National Register of Historic Places (NRHP)- and Washington Heritage Register (WHR)-listed and eligible resources in the MICs/project subareas. Additionally, HRA's architectural historian reviewed the Seattle Landmarks (SL) designated Landmarks List and Landmarks Districts map on the City's website.

For the architectural resources analysis, the GIS Specialist created maps showing the locations of the parcels that meet the 50-years or older threshold, properties that have been recorded on an HPI form, and NRHP-listed properties and districts.

HRA's in-house library was used to obtain information on the environmental, archaeological, and historical context of the project vicinity. HRA research staff also examined General Land Office (GLO) plats, available online through the Bureau of Land Management (BLM) website, to locate potential historical features. These nineteenth-century maps, arranged by township and range, indicate locations of then-extant historical structures, trails, and features. Although most of these structures are no longer extant, the maps indicate where historic period cultural resources could be encountered. Researchers reviewed additional historic maps (e.g., U.S. Geological Survey [USGS] maps, Sanborn Fire Insurance maps, County atlases) available through online resources.

Based on environmental characteristics, ethnographic data, and the distribution of previously recorded cultural resources, HRA formulated initial expectations about the sensitivity of the MICs for containing historic-period architectural and archaeological resources. DAHP's statewide predictive model layer was also reviewed for probability estimates of the presence of precontact cultural resources.

Full Study Area

Cultural resources identified in or adjacent to the Full Study Area include districts, sites, buildings, structures, or objects (BSOs) that are 45 years old or older, and listed or eligible for listing in the NHL Program, NRHP, WHR, WHBR, or the SL program, whose age threshold for inclusion is 25 years old or older.

Architectural Resources

Within the full study area, there is 1 NHL property and a number of properties that are listed in the NRHP, WHR, and SL. There are 3 NRHP-listed historic districts in the study area, 12 NRHP-

and WHR-listed historic properties, 5 properties that are listed in the WHR, and 15 historic properties designated Seattle Landmarks (**Exhibit 3.11-1**). There are no historic barns listed in the WHBR within the study area. There are several Seattle Landmarks in the Study Area, some of which are listed by the NRHP. See **Exhibit 3.11-2**.

According to the King County Tax Assessor, there are 865 historic-period buildings within the full study area, of which 774 are commercial/industrial buildings and the remaining 91 are residential buildings.

In contrast, DAHP online WISAARD records show 1,566 individual historic-period architectural resources within the full study area that have been previously recorded on HPI forms. Of these, 73 were determined eligible for listing in the NRHP and 154 were determined not eligible. The remaining 1,339 resources have no formal determinations of eligibility, and many were created by data transfer for an Assessors Data Project for King County (**Exhibit 3.11-2**). These resources were not formally surveyed and recorded and have neither eligibility recommendations nor determinations of eligibility.

The discrepancy between the Assessor's and DAHP's records are likely due to demolitions that alter County Tax Assessor's records but do not change the records in DAHP's WISAARD database.



Exhibit 3.11-1 National Register of Historic Places Listed Architectural Properties and Districts





Source: HRA, 2021.



Exhibit - Recorded Historic Period Buildings, Structures, and Objects in the Study Area

Source: HRA, 2021.

Archaeological Resources

Within the full study area, there are 31 archaeological sites recorded by 83 previous studies that included archaeological investigations (**Exhibit 3.11-3**). One precontact site is listed in the NRHP and WHR, one historic period site has been determined eligible for inclusion in the NRHP, eleven historic period sites have been determined not eligible for inclusion in the NRHP, and the remaining sites, all of which date to the historic period, have not been formally evaluated.

All of the project subareas are considered of High or Very High Risk to contain precontact archaeological resources by DAHP's precontact archaeological site probability model (**Exhibit 3.11-4**).



Exhibit 3.11-3 Recorded Archaeological Resources



Exhibit 3.11-4 Map Showing Archaeological Sensitivity from DAHP Model

Source: HRA, 2021.

4 - Survey Highly Advised: High Risk (Color: Pale Yellow)

5 - Survey Highly Advised: Very High Risk (Color: Brightest Yellow/Canary Yellow)

Maritime Washington National Heritage Area

The Maritime Washington National Heritage Area (MW NHA) was designated by Congress in 2019 as a place recognized for its nationally important natural, cultural, historic, and recreational resources, which combine to form a nationally important landscape. The MW NHA stretches along 3,000 miles of coastline from Grays Harbor County to the Canadian border. The MW NHA encompasses 18 federally recognized Tribes, 13 counties, 32 incorporated cities, and 30 port districts in Washington state. The MW NHA is non-regulatory but is controlled by grassroots organizations and is facilitated by the Washington Trust for Historic Preservation (WTHP), Washington's statewide nonprofit historic preservation organization, with technical assistance and funding from the National Park Service (NPS). The MW NHA is a cooperative organization with regional representation that is supportive of tourism and economic development, and functions to build partnerships to support communities in maintaining and sharing their unique resources and telling the stories of those places.

After receiving designation, the WTHP with partners and community stakeholders were tasked with developing a management plan that typically includes an education plan, rehabilitation strategy for historic sites or vessels, a tourism enhancement strategy, a strategy for improvement of local museums, and other related activities. After completion of the management plan, the MW NHA will be able to receive grants and other federal funds, should funding be available.

Exhibit 3.11-5 shows the portion of the MW NHA that occurs within the study area of this EIS. For more information, go to the WTHP website, <u>http://www.preservewa.org/wp-</u> <u>content/uploads/2019/04/NationalMaritimeHeritageAreaStudy.pdf</u>.



Exhibit 3.11-5 Maritime Washington Heritage Area that Occurs Within the Study Area

Ballard

There are three NRHP-listed historic districts and six individually listed resources within or adjacent to the BINMIC, all of which are found in the Ballard Subarea. Also, there is one WHR-listed resource within the area.

The first district is the Ballard Avenue Historic District, which is immediately adjacent to the MIC boundary. The District was designated a Seattle Landmark and listed in the NRHP in 1976. The District is associated with a pattern of events that contributed to the development of Ballard (Criterion A), under the themes of commerce, industry, politics/government, and transportation, and the District embodies the distinctive characteristics of modest commercial architecture (Criterion C), with a period of significance 1890–1930, and, when nominated, contained around 50 contributing resources.

The second is the Chittenden Locks and Lake Washington Ship Canal Historic District, which was listed in the NRHP in 1978. With a period of significance of 1906–1917, the District is significant under Criterion A for its contributions to commerce and politics/government, and under Criterion C under the themes landscape engineering, engineering, and architecture. The District encompasses about 16 contributing resources, including the dam, double locks, channels, and various associated accessory buildings/structures.

The third NRHP-listed historic district is the Gas Works Park Historic Landscape, which was listed was listed in 2013 with a period of significance 1950–1974, and 1975–2000. The landscape is significant under Criterion A, for the theme of industry, and under Criterion C under the themes of landscape architecture/engineering. The District contains 20 contributing resources, including sites, structures, objects, and buildings such as the north lawn, concrete railroad trestle, tanks, generator towers, the Foamite house, and others.

Also found within the Ballard Subarea is the NRHP-listed Ballard Bridge. Listed in 1982 under the Historic Bridges and Tunnels in Washington State Multiple Property Documentation form (MPD), the Ballard Bridge is significant under Criterion A for its contributions to transportation and under Criterion C under the theme of engineering as a double-leaf bascule bridge. The bridge has a period of significance of 1900–1924.

Three additional bridges adjacent to the Ballard Subarea were listed in the NRHP under the Historic Bridges and Tunnels in Washington State MPD in 1982. They are the University Bridge (1919), under Criterion C under the theme of engineering as a double-leaf trunnion bascule bridge; the Fremont Bridge (1919), under Criterion C under the theme of engineering as a double-leaf trunnion bascule bridge; and Aurora Avenue Bridge (1931), under Criterion C under the theme of engineering as a double-leaf trunnion bascule bridge; and Aurora Avenue Bridge (1931), under Criterion C under the theme of engineering as a cantilever truss bridge.

Two ships in the Ballard Subarea were listed in the NRHP. One is the *Wawona* schooner (1897), which was listed in 1977 under Criterion A for the themes of commerce, industry, and maritime transportation. The second is the *Zodiac* schooner (1924), which was listed in 1982, under Criterion C, for its architectural significance.

Listed in the WHR in 2001, was the tugboat, *Chickamauga*, with its period of significance 1915, the year it was built. The ship was listed for its significance related to events as first diesel powered tugboat in the U.S. (Criterion 3), for its association with Arthur McNealy (Criterion 6), for its engineering as a representative example of the transition from steam to diesel power (Criterion 7), and for its design by Leslie Edward "Ted" Geary (Criterion 8).

King County Tax Assessor records show that within the Ballard Subarea, there are 156 historicperiod buildings. Of these, 141 are commercial/industrial buildings, while the remaining 15 are residential buildings.

DAHP records show 274 individual historic-period architectural resources have been documented on HPI forms within the Ballard Subarea. Of these, only 9 were determined eligible for listing in the NRHP.

DAHP records show seven cultural resources studies that included archaeological resources investigations have been conducted within the Ballard Subarea. Two sites were recorded by these studies.

Interbay Dravus

There are two NRHP-listed resources found partially within the boundaries of the Interbay Dravus Subarea of the BINMIC. They are the aforementioned Chittenden Locks and Lake Washington Ship Canal Historic District, and the southern end of the NRHP-listed Ballard Bridge. Also found in the Interbay Dravus Subarea is one SL designated building, Alexander Hall.

According to the King County Tax Assessor, within the Interbay Dravus Subarea, there are 56 historic-period buildings, all of which are commercial or industrial buildings. DAHP records show 141 individual historic-period architectural resources have been recorded on HPI forms within the Interbay Dravus Subarea. Of these, 2 were determined eligible for listing in the NRHP.

DAHP records show three cultural resources studies that included archaeological resources investigations have been conducted within the Interbay Dravus Subarea. No sites were recorded by these studies.

Interbay Smith Cove

There are five SL-designated historic-period architectural resources within the Interbay Smith Cove Subarea of the BINMIC. These are the 14th Avenue W Group and include 2000, 2006, 2010, 2014, and 2016 14th Avenue W. There are no NRHP-, NHL-, or WHR/WHBR-listed architectural resources in this Subarea. Adjacent to the Interbay Smith Cove Subarea is one NRHP-listed architectural resource, the Admiral's House, 13th Naval District (Quarters A). Listed in 2013, the Admiral's House is significant under Criterion A, for its association with the U.S. Navy and its role in Seattle, and under Criterion C, as a representative example of the Colonial Revival style. The property has a period of significance of 1944–1960.

According to the King County Tax Assessor, within the Interbay Smith Cove Subarea, there are 35 historic-period buildings, of which, 23 are commercial or industrial buildings, and the remaining 12 are residential buildings. DAHP records show 96 individual historic-period architectural resources have been documented on HPI forms within the Interbay Smith Cove Subarea. Of these, 8 were determined eligible for listing in the NRHP.

DAHP records show seven cultural resources studies that included archaeological resources investigations have been conducted within the Smith Cove Subarea. Two historic period sites were recorded, but the sites have not been formally evaluated.

SODO/Stadium

Within the boundaries of the SODO/Stadium Subarea are nine historic properties listed in the various registers. Listed in 1976 in both the NRHP and SL is the Triangle Hotel and Bar, also known as the Flatiron Building. The building is significant under Criterion A for commerce and Criterion C for architecture, with a period of significance 1909–1910. The A. L. Palmer Building was listed in the NRHP in 2008 for its contributions to commerce and industry (Criterion A) and under the theme of architecture (Criterion C), with a period of significance of 1910. The Bay View Brewery was listed in the NRHP in 2013, under Criterion A for commerce and industry, Criterion B for its association with brewery owners and operators, Andrew Hemrich and Emil Sick, and Criterion C for architecture. The building's period of significance is 1886–1962. The Ford Motor Company Assembly Plant is also found in this subarea. Listed in 2013, this resource is significant for its contributions to industry and commerce (Criterion A), and for its architecture (Criterion C). The building has a period of significance of 1932, the date of its initial construction.

There are two WHR-listed architectural resources in the SODO/Stadium Subarea. One is the First Service Station Site, which was listed in 1970, as the World's First Service Station. It was listed under Criterion A, for commerce, industry, and transportation, with a period of significance of 1907, the date of its initial construction. The second is the *USS Nebraska* Launching (1904) and Skinner and Eddy Shipyard (1916–1920), which was listed in the WHR in 1970 for its significant contributions to Maritime and Naval history, industry, and transportation (Criterion A), and for engineering (Criterion C).

The SODO/Stadium Subarea also contains three SL-designated resources, including Fire Station #14, the Duwamish Railroad Bridge, and the Flatiron Building. Additionally, located immediately adjacent to the northern boundary of the SODO/Stadium Subarea of the Greater Duwamish MIC is the Pioneer Square Preservation District, an SL-designated district.

According to the King County Tax Assessor, within this subarea, there are 331 historic-period buildings, 310 of which are commercial or industrial buildings, and the remaining 21 are residential buildings. DAHP records show 620 individual historic-period architectural resources

have been documented on DAHP HPI forms within the SODO/Stadium Subarea. Of these, 38 were determined eligible for listing in the NRHP.

DAHP records show 40 cultural resources studies that included archaeological resources investigations within the SODO/Stadium Subarea. One precontact site, a 2.25-acre shell midden site, was discovered in 1975 when the landowner demolished houses on a portion of the site. Subsequent archaeological investigations led to the site being listed in the NRHP and WHR. The current Duwamish longhouse is located in the vicinity of this site (see **Exhibit 3.8-10**). Of the 15 historic period sites recorded, one has been determined eligible for inclusion in the NRHP and the WHR, nine have been determined not eligible for inclusion in the NRHP or the WHR, and five sites have not been formally evaluated.

Georgetown/South Park

Within the boundaries of the Georgetown/South Park Subarea are three historic properties listed in the NRHP and WHR. The Seattle Electric Company Georgetown Steam Plant was listed in the NRHP in 1978, for its significant contributions to the theme of engineering under Criterion C. Built in 1906, the property has a period of significance of 1900–1924. The property achieved NHL status in 1984. Listed in the WHR, are the Maple Donation Claim and Gorst Field. The Maple Donation Claim was listed in the WHR in 1970 for its significant contributions to local history (Criterion A), with a period of significance of 1851, the date the Donation Land Claim was staked. The final historic property in this subarea is Gorst Field. Listed in 1970 in the WHR, Gorst Field is significant for its contributions to commerce, industry, and transportation under Criterion A, and engineering under Criterion C. The field had a period of significance of 1920– 1928.

According to the King County Tax Assessor, within this subarea, there are 286 historic-period buildings—219 of these are commercial or industrial buildings, and the remaining 67 are residential buildings. DAHP records show 434 individual historic-period architectural resources have been documented on HPI forms within or immediately adjacent to the Georgetown/South Park Subarea. Of these, 15 were determined eligible for listing in the NRHP.

DAHP records show 26 cultural resources studies that included archaeological resources investigations have been conducted within the Georgetown/South Park Subarea. Eleven historic period sites have been recorded, two of which have been determined not eligible for inclusion in the NRHP or the WHR, and the remaining have not been formally evaluated.

3.11.2 Impacts

This section considers the impacts of the alternatives on historic, archaeological, and cultural resources within the study area.

Impacts Common to All Alternatives

Full Study Area

All the alternatives have the potential to affect districts, sites, buildings, structures, or objects (BSOs) that have been listed in the NRHP and other historic registers, including the WHR, WHBR, and SL, and those resources that have been determined eligible for listing in the NRHP. Additionally, the alternatives could potentially affect the numerous BSOs and undiscovered archaeological sites that have yet to be surveyed and assessed for potential eligibility to the NRHP.

Impacts to historic, archaeological, and cultural resources in the study areas from the No Action Alternative and three Action Alternatives were identified by assessing potential for both aboveand below-ground changes. Such impacts generally include physical alteration, damage, or destruction of all or part of a resource; alteration of the characteristics of the surrounding environment that contribute to the property's significance; and the introduction of visual or audible elements that are out of character with the property. In other words, actions that would alter, directly or indirectly, any of the characteristics of a historic property in such a way that would diminish its integrity of location, design, setting, materials, workmanship, feeing, and association, and would affect its eligibility to qualify for inclusion in the NRHP or other historic registers.

All Action Alternatives would result in the implementation of the Industrial and Maritime Strategy meant to support and retain maritime businesses that contribute to the maritime history of the study area. The strategy supports continued implementation of the Seattle Shoreline Master Program jointly adopted by the City and the Washington Department of Ecology according to the State Shoreline Management Act which promotes ports and shoreline industry, while protecting environmental and cultural resources. See a summary of the strategy in **Exhibit 2.2-2**.

The Action Alternatives also include proposed land use concepts such as incentivizing investments by industrial businesses to expand industrial sites; changes to development and landscaping standards addressing street frontages and parcels; incentivizing development and densification of multi-story buildings; limited caretakers' quarters and makers studios in industrial areas and some areas of mixed-use residential construction in selected locations (see **Exhibit 2.4-4**). Historic-period BSOs located in the study area could be subject to demolition for new construction, incompatible alterations/additions, and inappropriate renovation of existing buildings for reuse under all alternatives. Such demolition and construction projects could require substantial below-ground work, thus negatively and irreversibly impacting below-

ground archaeological and cultural resources. DAHP's archaeological predictive model used to establish probabilities for precontact cultural resources, depicts almost all the land within the MICs as within a Very High Risk area, primarily because of proximity of Puget Sound, Salmon Bay, Lake Union, Elliott Bay, and the Duwamish River, and the use history throughout the precontact and historic periods.

Since development may occur in any location in the study area under any alternative, it is possible that cultural resources could be impacted under each alternative. Changes to zoning that allows a wider range of industrial or non-industrial uses could spur redevelopment in those locations. This could occur, for example, where the Industry and Innovation or Urban Industrial Districts allow for more mixed industrial/office near station areas, or caretakers' quarters and makers studios for live/work options throughout the study area. This could also occur where areas are removed from the MIC and allowed for mixed-use residential near Georgetown and South Park. Even where there are no formally designated historic landmarks, there are numerous properties with historic period buildings, or a very high or high risk of archaeological resources. A qualitative summary of areas of zoning change are listed in **Exhibit 3.11-6** below.

Zoning Districts	Alt. 1 Acres	Land Use Concept	Alt. 2 Acres	Alt 2 Zone Acres Change Description	Alt. 3 Acres	Alt 3 Zone Acres Change Description	Alt. 4 Acres	Alt 4 Zone Acres Change Description	<u>Pref.</u> <u>Alt.</u> <u>Acres</u>	<u>Preferred</u> <u>Alternative</u>	Relationship to Mapped Resources
Industrial General (IG1/IG2)	6,273	Maritime, Manufacturi ng, and Logistics (MML)	6,251	Increase in Ballard near Lock. Small increase near West Marginal Way. Otherwise, similar to IG Zone.	5,968	Increase in Ballard near Lock. Small increase near West Marginal Way. Otherwise, similar to IG Zone. Reduced where UI or II is applied.	6,035	Increase in Ballard near Lock. Small increase near West Marginal Way. Otherwise, similar to IG Zone. Reduced where UI or II is applied.	<u>5.895</u>	Similar to Alternative 4. <u>Reduced</u> where UI or II is applied.	Some acres of zoning change near listed and mapped resources (e.g., National Register Resources, Historic Period Buildings, and Very/High Risk of Archaeological Sensitivity).
Industrial Buffer (IB)	316	Urban Industrial (UI)	222	Increase/relocation in Interbay Dravus south of Ballard Bridge and near Duwamish River at city limits in South Park. Other UI similar to IB zone location.	426	Increase in Ballard north of Leary and along Lake Washington (e.g., near Gas Works Park). Similar to Alternative 2 UI extent south of Ballard Bridge. Small increase in Interbay Smith Cove. Similar to IB zone extent elsewhere.	279	Greater area of UI than Alternative 2 but less than Alternative 3 in Ballard and Interbay. Similar to Alternative 2 in SODO.	<u>376</u>	Similar to alternatives 3 and 4. Some area of IB outside the MIC shown as Mixed Use in West Ballard.	Some acres of zoning change are near or encompass listed and mapped resources (e.g., National Register Resources, Historic Period Buildings, and Very/High Risk of Archaeological Sensitivity).
Industrial Commercial (IC)	347	Industry and Innovation (II)	463	Small area added in Ballard south of NW Market. Area added in SODO area near 4th Avenue. Mostly applied in similar locations as IC zone or in place of IB zone.	516	In Ballard and Interbay, mostly applied in similar locations as IC zone, except where UI is expanded. Expanded in SODO along 1st and 4th Avenues.	600	Increase in Ballard north of Leary Way. Mostly applied in similar locations as IC zone. Greatest expansion in SODO along 1st and 4th Avenues.	<u>612</u>	Similar to Alternative 4 except outside MIC some areas retained as IC. Judkins Park outside the MIC changed to Mixed Use.	Some acres of zoning change are near or encompass listed and mapped resources (e.g., National Register Resources, Historic Period Buildings, and Very/High Risk of Archaeological Sensitivity).

Exhibit 3.11-6 A	cres of Zoning or Land Us	e Concept and Qualitativ	e Relationship to Mapped Cu	ultural Resources
------------------	---------------------------	--------------------------	-----------------------------	-------------------

Ch.3 Environment, Impacts, & Mitigation Measures • Historic, Archaeological, & Cultural Resources

Zoning Districts	Alt. 1 Acres	Land Use Concept	Alt. 2 Acres	Alt 2 Zone Acres Change Description	Alt. 3 Acres	Alt 3 Zone Acres Change Description	Alt. 4 Acres	Alt 4 Zone Acres Change Description	<u>Pref.</u> <u>Alt.</u> <u>Acres</u>	<u>Preferred</u> <u>Alternative</u>	Relationship to Mapped Resources
Mixed-Use Commercial				Not applicable.	26	Increased in Georgetown and South Park.	22	Increased in Georgetown and South Park.	<u>53</u>	Increased in Georgetown and South Park. Added in West Ballard and Judkins Park outside the MIC.	Some acres of zoning change are near or encompass listed and mapped resources (e.g., Historic Period Buildings, Recorded Archeological Resources, and Very/High Risk of Archaeological Sensitivity).
Total	6,936		6,936		6,936		6,936			<u>6,936</u>	
Source: BERK, 202 <mark>24</mark> .											

Ballard

All alternatives have the potential to affect the known and unknown historic, archaeological, and cultural resources in the Ballard Subarea. The Ballard Subarea contains three NRHP-listed historic districts and six individually listed resources, one WHR-listed resource, and numerous historic-period buildings, some of which have been documented on HPI forms, and nine of those determined eligible for listing in the NRHP. Due to the area's concentration of historicperiod buildings, structures, and objects—many of which have yet to be surveyed—it is plausible that many could be determined eligible for listing in the NRHP and local registers. Two known archaeological sites have been previously recorded in the Ballard Subarea; however, due to the area's very high probability for archaeological and cultural resources, many more unknown sites could be present.

Interbay Dravus

All alternatives have the potential to affect the known and unknown historic, archaeological, and cultural resources in the Interbay Dravus Subarea. The Interbay Dravus Subarea contains a NRHP-listed historic district, an individually listed resource, one SL-designated resource, and numerous historic-period buildings and structures, many of which have been documented on HPI forms, with two of those determined eligible. Due to the area's concentration of historicperiod buildings, structures, and objects—many of which have yet to be surveyed—it is plausible that many could be determined eligible for listing in the NRHP and local registers. No archaeological sites have been previously recorded in the Interbay Dravus Subarea; however, due to the area's very high probability for archaeological and cultural resources, many more as yet unknown sites could be present.

Interbay Smith Cove

All alternatives have the potential to affect the known and unknown historic, archaeological, and cultural resources in the Interbay Smith Cove Subarea. While the Interbay Smith Cove Subarea contains no NRHP-, WHR-, WHBR-listed resources, there are five SL-designated historic-period architectural resources within this subarea, and numerous historic-period buildings and structures, many of which have been documented on HPI forms, with eight of those determined eligible. Also, immediately adjacent to the subarea's western boundary is a NRHP-listed resource. Due to the subarea's concentration of historic-period buildings, structures, and objects—many of which have yet to be surveyed—it is plausible that many could be determined eligible for listing in the NRHP and local registers. Two archaeological sites have been previously recorded in the Interbay Smith Cove Subarea but not formally evaluated; however, due to the area's very high probability for archaeological and cultural resources, many more as yet unknown sites could be present.

SODO/Stadium

All alternatives have the potential to affect the known and unknown historic, archaeological, and cultural resources in the SODO/Stadium Subarea. Four NRHP-listed, two WHR-listed, and three SL-designated historic-period architectural resources within the SODO/Stadium Subarea, and numerous historic-period buildings and structures, many of which have been documented on HPI forms, with 38 of those determined eligible. Also, immediately adjacent to the subarea's northern boundary is a SL-designated historic district. Due to the area's concentration of historic-period buildings, structures, and objects—many of which have yet to be surveyed—it is plausible that many could be determined eligible for listing in the NRHP and local registers. One precontact archaeological site was listed in the NRHP and WHR and 16 historic-period sites have been previously recorded in the SODO/Stadium Subarea. One has been determined eligible for the NRHP, nine have been determined not eligible, and five sites have not been formally evaluated. Due to the area's very high probability for archaeological and cultural resources, many more as yet unknown sites could be present.

Georgetown/South Park

All alternatives have the potential to affect the known and unknown historic, archaeological, and cultural resources in the Georgetown/South Park Subarea. The Georgetown/South Park Subarea contains one NRHP-listed resource that has achieved National Historic Landmark (NHL) status, two WHR-listed resources, and numerous historic-period buildings and structures, many of which have been documented on HPI forms, with 15 of those determined eligible. Due to the area's concentration of historic-period buildings, structures, and objects—many of which have yet to be surveyed—it is plausible that many could be determined eligible for listing in the NRHP and local registers. In the Georgetown/South Park Subarea, eleven archaeological sites have been previously recorded, with two determined not eligible and nine not formally evaluated. However, due to the area's very high probability for archaeological and cultural resources, many more as yet unknown sites could be present.

Equity & Environmental Justice Considerations

In 2015, Seattle established the City of Seattle Equity and Environment Initiative (EEI) to address the connection between race and social justice and the environment. The Community Partners Steering Committee (CPSC), working with City staff, defined EEI populations as people of color, immigrants, refugees, people with low incomes, and people with limited-English proficiency (CPSC 2016:1–8). Studies by the National Trust for Historic Preservation (NTHP) have noted that while rezoning and redevelopment can address some of the particular issues in neighborhoods with high EEI populations of historically marginalized communities, such as poor air and water quality, soil contamination, noise pollution, climate change, and unsafe, disconnected, and inaccessible neighborhoods, some of the land use concepts and strategies could lead to adverse impacts of economic displacement, and loss of locally owned small businesses, and potentially loss of fair and affordable housing. Equitable development and redevelopment should include the voices of the EEI populations to share in the decision-making process (Canaan, et al 2021:54–55; NTHP 2021:10; Rypkema 2004).

See **Chapter 2** for a description of the City's process to develop the Industrial and Maritime Strategy and to engage EEI populations. The scoping and Draft EIS comment periods are an opportunity to gain input from EEI populations as well.

Under all alternatives, should redevelopment occur within high EEI population neighborhoods in the study areas, benefits could be realized such as reinvestment in aging buildings, increased levels of homeownership/business ownership in newly rehabilitated buildings, and renovation/adaptive re-use of vacant and abandoned properties. However, there could also be adverse impacts from these benefits such as rising rents and property taxes, loss of "power" and "ownership" by long-term residents, and rising potential for conflicting priorities between new and long-term residents (Ryberg 2010:265–266; Rypkema 2004). These adverse impacts disproportionately affect EEI populations.

All alternatives have the potential to affect historic and cultural resources in historically marginalized neighborhoods in the study areas, such as the southern end of the Seattle-Chinatown International District, SODO/Industrial District, Highland Park, South Park, Greater Duwamish, and Georgetown (OPCD 2020:2). Specifically, impacts to historic-period architectural resources could occur under all alternatives as a result of alteration, demolition, damage, or destruction. In addition, development under all alternatives could increase the probability of inadvertent discovery of archaeological and cultural resources because of foundation, circulation, and landscaping work.

Additionally, Indigenous populations may lose access to both known and potentially unrecorded cultural or spiritual sites, due to redevelopment on their traditional lands in the study areas. As the locations of such resources are considered restricted information, specifics will not be discussed here without permission from the appropriate Tribes.

Impacts of Alternative 1 No Action

Alternative 1 No Action maintains the status quo within the existing industrial zones, with no changes to current Comprehensive Plan policies, development standards, or zoning. Redevelopment and development projects due to market pressures under Alternative 1 No Action would continue to affect historic, archaeological, and cultural resources, with such impacts as alteration, demolition, damage, or destruction. Alternative 1 No Action includes no additional protections or improvements in planning for consideration of impacts to historic, archaeological, and cultural resources.

Impacts of Alternative 2

Alternative 2 (Future of Industry—Limited) applies the proposed land use concepts of Maritime Manufacturing and Logistics (MML), Industry and Innovation (II) and Urban Industrial (UI).

Typically, the II and UI are located in places where similar IC and IB zoning is applied today but with expanded use allowances and development standards. These new II and UI zones could incentivize development to increase floor area and height limits that would allow construction of dense multi-story buildings. The UI zone would allow adaptive reuse of buildings and adds flexibility for larger size of use for combination industry-retail or industry-office space. Additionally, Alternative 2 expands non-industrial ancillary uses and reduces stand-alone non-industrial size of use limits. Some areas of zoning change include increased or altered boundaries of the UI zone in the Interbay Dravus area south of the Ballard Bridge, and near the Duwamish River near South Park. The II zone is added in Ballard south of NW Market Street. An area of II is added in SODO area near 4th Avenue S. As mapped in the Affected Environment and described in **Exhibit 3.11-6** some acres of zoning change abut listed historic or recorded archaeological resources or contain mapped resources sensitivity areas (e.g., Historic Period Buildings, and Very/High Risk of Archaeological Sensitivity).

Impacts to historic, archaeological, and cultural resources could occur under Alternative 2 as a result of alteration, demolition, damage, or destruction. In addition, development under Alternative 2 could increase the probability of inadvertent discovery of archaeological and cultural resources as compared to Alternative 1 No Action because of substantial foundation work needed for multi-story buildings. Additionally, without design guidelines, preservation incentives, or review, allowed adaptive reuse projects could impact historic-period architectural resources by allowing for inappropriate alterations, changes, additions, and loss of character-defining features and historic building materials. However, appropriate adaptive reuse projects guided by the *Secretary of the Interior Standards for Rehabilitation* or new city-level rehabilitation guidelines and incentives, could save some historic-period architectural resources from demolition.

Impacts of Alternative 3

Alternative 3 (Future of Industry—Targeted) also applies the MML, II, and UI land use concepts, but with a greater share than Alternative 2. This includes 7% of the land area and up to 0.50 mi around transit stations, expanding the transition area in Ballard, removing small nodes of land in Georgetown/South Park from the MIC for rezoning to mixed-use to advance community goals, allows lodging, and expands limited industry-supporting housing (610 units), such as new caretaker's quarters, makers studios, and existing non-conforming housing. Additionally, Alternative 3 expands non-industrial ancillary uses and reduces stand-alone non-industrial size of use limits.

Impacts to historic, archaeological, and cultural resources from alteration, demolition, damage, or destruction under Alternative 3 are similar to Alternative 2. Like Alternative 2, development under Alternative 3 could increase the probability of inadvertent discovery of archaeological and cultural resources as compared to Alternative 1 No Action because of substantial foundation work needed for new development and multi-story buildings.

Some areas of UI would increase in Ballard north of Leary Way NW and along Lake Washington (e.g., near Gas Works Park). Similar to Alternative 2 the UI would extend south of Ballard Bridge.

There is a small increase in Interbay Smith Cove. The II is expanded in SODO along 1st and 4th Avenues. These areas of change are near or encompass listed and mapped resources sensitivity areas (e.g., National Register Resources, Historic Period Buildings, and Very/High Risk of Archaeological Sensitivity).

Industry-supporting housing and those areas in the Georgetown/South Park Subarea removed from the MIC to allow for mixed-use—especially in those historic commercial areas rezoned to Seattle Mixed where few surveys have been done—could also add to demolitions of historic-period architectural resources. The areas of zoning change to Seattle Mixed are in areas mapped with Historic Period Buildings, Recorded Archeological Resources, and Very/High Risk of Archaeological Sensitivity. A 2014 Georgetown survey noted that the great majority of the historic residential and commercial properties exhibit some degree of alteration; however, they remain generally intact and continue to convey historic character (Krafft 2015).

Additionally, without design guidelines, incentives, and project review, allowed adaptive reuse projects could impact historic-period architectural resources by allowing for inappropriate alterations, changes, additions, and loss of character-defining features and historic building materials.

Impacts of Alternative 4

Alternative 4 (Future of Industry—Expanded) also applies the MML, II, and UI land use concepts of, but with a greater share than Alternative 3, and includes 8% of the land area and wider than 0.50 mi around transit stations including land near potential stations Ballard ST3 and Stadium ST3, expanding the transition area in the Stadium district, removing small nodes of land in Georgetown/South Park from the MIC and rezoned to mixed-use to advance community goals, and allows all lodging with larger size of use limits. This alternative also allows unlimited market housing in the areas removed from the MIC and industry-supporting housing (2,195 units).

Under Alternative 4, there would be a greater area of UI zoning than Alternative 2 but less than Alternative 3 in Ballard and Interbay. The extent of UI zoning would be similar to Alternative 2 in SODO. There would be an increase in II in Ballard north of Leary Way. II is mostly applied in similar locations as IC zone. The greatest extent of II is in SODO along 1st and 4th Avenues. Some acres of zoning change are near or encompass listed and mapped resources sensitivity areas (e.g., National Register Resources, Historic Period Buildings, and Very/High Risk of Archaeological Sensitivity).

Impacts to historic, archaeological, and cultural resources from alteration, demolition, damage, or destruction under Alternative 4 are similar to alternatives 2 and 3. Like alternatives 2 and 3, development under Alternative 4 could increase the probability of inadvertent discovery of archaeological and cultural resources as compared to Alternative 1 No Action because of substantial foundation work needed for new development, multi-story buildings, and new housing.

Due to market pressures, unlimited market housing in areas removed from MICs, especially in the Georgetown/South Park Subarea, and industry-supporting housing would invariably add to demolitions of historic-period architectural resources and impacts to archaeological and cultural resources. The greater allowances for caretakers/artist residences under Alternative 4 compared to all other alternatives may result in greater pressure for conversion of properties that may contain historic period structures, or that are mapped as having a high or very high risk of archaeological resources. The MIC reduction areas that would be rezoned to Seattle Mixed are near or encompass listed and mapped resources sensitivity areas (e.g., Historic Period Buildings, Recorded Archeological Resources, and Very/High Risk of Archaeological Sensitivity).

Additionally, without the implementation of design guidelines. Incentives, or project review, allowed adaptive reuse projects could impact historic-period architectural resources by allowing for inappropriate alterations, changes, additions, and loss of character-defining features and historic building materials.

Impacts of the Preferred Alternative

<u>The Preferred Alternative (Future of Industry—Balanced) also applies the MML, II, and UI land</u> <u>use concepts like the Draft EIS alternatives, incorporating many of their features and new</u> <u>features/refinements driven by community comments and to reduce impacts.</u>

Under the Preferred Alternative, only areas outside the MICs (such as in south Ballard, south Fremont, along the north and east shores of Lake Union, and in Judkins Park) would retain existing Industrial Commercial (IC) zoning, to promote development of mixed industrial and commercial activities. Additionally, this alternative would allow housing in commercial zones outside the MICs.

The Preferred Alternative adds additional areas of mixed use that would allow housing, but with a greater share than alternatives 3 and 4, in the Georgetown/South Park area, and in small areas outside the MICs (Judkins Park), This alternative applies the II and UI land use concepts in 14% of the land area in the MICs, and approximate 0.50 mi around transit stations and expanding the UI and II zoning in SODO in the Stadium Transition Area Overlay District (STAOD), and removing small nodes of land in Georgetown/South Park and along Harbor Boulevard SW from the MIC to advance community goals.

Under the Preferred Alternative, there would be a greater area of II and UI zoning than Alternative 2 but less than alternatives 3 and 4 in Ballard and Interbay. The extent of UI zoning would be similar to Alternative 4 in SODO. Some acres of zoning change are near or encompass listed and mapped resources sensitivity areas (e.g., National Register Resources, Historic Period Buildings, and Very/High Risk of Archaeological Sensitivity).

Impacts to historic, archaeological, and cultural resources from alteration, demolition, damage, or destruction under the Preferred Alternative are similar to alternatives 2 and 3. Like alternatives 2 and 3, development under the Preferred Alternative could increase the probability of inadvertent discovery of archaeological and cultural resources as compared to <u>Alternative 1 No Action because of substantial foundation work needed for new development,</u> <u>multi-story buildings, and new housing.</u>

Due to market pressures and the higher total of housing, the Preferred Alternative would invariably add to demolitions of historic-period architectural resources and impacts to archaeological and cultural resources. The greater allowances for caretakers/artist residences under the Preferred Alternative compared to all other alternatives may result in greater pressure for conversion of properties that may contain historic-period structures, or that are mapped as having a high or very high risk of archaeological resources.

However, in the Georgetown Triangle, the new mixed use zone standards would incentivize projects that adaptively reuse historic-period buildings by allowing for an exemption from FAR limits and increased height limits. These incentives could reduce the loss of historic-period buildings. As well, the application of a Mandatory Housing Affordability (M1) suffix to this zone could offset economic displacement and affordable housing through redevelopment of existing historic-period buildings.

The MIC reduction areas that would be rezoned to Seattle Mixed or UI, are near or encompass listed and mapped resources sensitivity areas (e.g., Historic Period Buildings, Recorded Archeological Resources, and Very/High Risk of Archaeological Sensitivity).

Additionally, without the implementation of design guidelines or project review, for example in the Neighborhood Commercial zone, the new mixed use zoning could allow adaptive reuse projects that could impact historic-period architectural resources by allowing for inappropriate alterations, changes, additions, and loss of character-defining features and historic building materials.

3.11.3 Mitigation Measures

Incorporated Plan Features

The Action Alternatives include some land use concepts that may mitigate adverse impacts to historic, archaeological, and cultural resources, such as expansion of new land use concepts and updates to industrial land use policies to anticipate future innovations and trends that may incentivize adaptive re-use of historic-period architectural resources.

Regulations & Commitments

Federal

Projects implemented under this EIS may require compliance with a number of federal, state, and local regulations, including the National Historic Preservation Act of 1966, as amended, Archaeological Resources Protection Act of 1979, National American Graves Protection and

Repatriation Act, National Environmental Protection Act of 1969, as amended, Washington Executive Order 21-02 (formerly 05-05), and the Washington State Environmental Protection Act.

- National Historic Preservation Act (NHPA) of 1966, as amended, commonly referred to as Section 106, has implementing regulations (36 CFR Part 800), that require Federal agencies (or others who have received Federal grants or funds, or a Federal permit or license) to take into account the effects of their undertakings on historic properties, by identifying historic properties, assessing adverse effects, and resolving those adverse effects.
 - The NHPA authorized the NRHP as the program to coordinate and support the Act. To be considered a historic property, resources must be determined eligible for listing in the NRHP by meeting at least one of the four established Criteria of Evaluation and retaining sufficient integrity to express its significance.
 - The National Historic Landmarks (NHL) Program functions to honor historic properties that are nationally and exceptionally significant in American history and culture.
 Properties must meet one of six NHL Criteria and possess a high degree of integrity.
- Archaeological Resources Protection Act (ARPA) of 1979, protects archaeological resources.
- National American Graves Protection and Repatriation Act (NAGPRA) creates protections for Native American burial sites, remains, and cultural objects.
- National Environmental Protection Act (NEPA) of 1969, as amended, requires federal agencies to assess whether a major federal action has the potential to significantly affect the human environment prior to making decisions. This is done through the preparation of an Environmental Assessment (EA) or an EIS.

State

- Washington Executive 21-02 (formerly 05-05) requires that impacts to cultural resources must be considered as part of any state-funded project or investment and must include consultation with DAHP and with Tribal governments.
- Washington State Environmental Protection Act (SEPA) has a process to identify and analyze environmental impacts to cultural resources associated with governmental decisions such as issuing permits, constructing public facilities, or adopting regulations, policies, and plans. This is accomplished through the SEPA Checklist.
- Washington State Archaeological Sites and Resources Protection Act (RCW 27.53) requires a permit to excavate or remove any archaeological resource located on public or Tribal lands.
- Registration of Historic Archaeological Resources on State-Owned Aquatic Lands (25-46 WAC) establishes to establish registration procedures for previously unreported historic archaeological resources discovered on, in, or under state-owned aquatic lands as provided for in chapter 27.53 RCW.
- The Washington Heritage Register (WHR) is an official state listing of significant sites and properties and is administered by DAHP. The list is honorary and the effects of listing in the WHR are parallel to the NRHP. Properties listed in the NRHP are automatically listed in the WHR.
The Washington Heritage Barn Register (WHBR) honors the barns of the State that are historically significant. Administered by DAHP, the heritage barn designation allows the property owners access to matching grant funds.

Local

King County's Historic Preservation Program (HPP) provides a number of preservation-related services including the Regional Preservation Program, Historic Resource Inventory, and the Landmarks Ordinance that is implemented through the county Landmarks Commission to ensure that the historic places, material culture, and traditions that reflect the region's history are preserved. County landmark designation and regulation is limited by law to the unincorporated area. The City of Seattle contracts with the county for archaeological review services (King County 2018). City of Seattle's Historic Preservation Program, through the Seattle Landmarks (SL) program, protects designated landmark sites, buildings, structures, objects, and districts city wide. Protections of designated landmarks is provided by design review of proposed alterations and the issuance of a Certificate of Approval.

Other Potential Mitigation Measures

When elimination, minimization, or avoidance of impacts to historic, archaeological, and cultural resources is impossible, appropriate and meaningful mitigation should be developed in accordance with DAHP Mitigation Options and Documentation Standards and in coordination with the area's Tribes, the lead agency, and all other consulting parties. Developing a mitigation plan should be an iterative and collaborative process using a diversity of lenses, which results in mitigation that improves the public's understanding and enriches technical knowledge of the impacted resource(s) (Douglass and Manney 2020).

Some examples of mitigation for impacts for architectural resources, might include:

- Preparing DAHP Level I (Historic American Building Survey/Historic American Engineering Record [HABS/HAER]) Documentation.
- Preparing DAHP Level II Documentation.
- Funding to DAHP for improvements to WISAARD to improve mapping of resources.
- Funding City-initiated proactive landmark nominations for properties and potential historic districts identified in new neighborhood surveys.
- Prioritizing City funding for retrofitting Unreinforced Masonry (URM) buildings to those properties that meet eligibility requirements for designation as a landmark or for listing in the National Register of Historic Places.
- Developing of cultural landscape contexts, including within historically marginalized communities.
- Preparing histories of the area including Indigenous perspectives. The City could work with tribes and others to develop context statements. A context statement focused on Historical Planning and Land Use Decisions is drafted in Section 3.8 Land & Shoreline Use.

- Funding City-led thematic historic context inventories that focus on marginalized or underrepresented immigrant communities and preparing thematic context statements relating to those resources.
- Conducting neighborhood survey and inventory projects within underrepresented or marginalized communities
- Considering potential impacts to historic resources during development review specifically that are associated with marginalized or underrepresented immigrant communities as part of project level SEPA review, or during the design review process.
- Including development incentives for preservation of architectural resources including adaptive reuse projects in the proposed Urban Industrial zone, such as an exemption from the floor area ration calculation, or flexibility for allowable uses within the structure. Such adaptive reuse projects could follow the *Secretary of the Interior Standards for Rehabilitation* or the City could develop new rehabilitation guidelines for adaptive reuse.
- For alternatives 3 and 4, exploring or studying the possible addition of a new Seattle Landmark District for the mixed-use area of Georgetown.
- Establishing new conservation districts in order to encourage preservation of older structures (referred to in SMC as "character structures"). Establishing Transfer of Development Rights (TDR) programs within new conservation districts to provide incentives for property owners to keep existing character structures.
- Adding regulatory authority to identify resource-specific mitigation before demolition occurs.
- Requiring project proponents to nominate buildings for landmark review when demolition
 of properties that are over 50 years old is proposed, regardless of City permitting
 requirements, by modifying the SEPA exemptions thresholds in the Seattle Municipal Code
 at Table A for section 25.05.800, and Table B for section 25.05.800.

Mitigation for adverse impacts to archaeological or cultural resources, could include:

- Prior to commencing site-specific subsurface investigations of soils, notifying the Duwamish tribe so an archaeologist can observe the work.
- Employ standard archaeological techniques such as <u>Aarchaeological testing</u>, excavation and <u>data recovery/collection of artifacts</u>, documentation, analysis, sharing evidence with the <u>Duwamish Tribe</u>, and archiving, possibly in a repository for future research.
- Archaeological testing, excavation and data recovery/collection of artifacts, documentation, analysis, and archiving, possibly in a repository for future research.
- Public education and outreach, including interpretive signage and/or a museum exhibit.
- Interpretive signage and educational programs for the National Maritime Heritage Area.
- Development of digital and other media content, including film, to share holistic stories of the impacted resource(s).

3.11.4 Significant Unavoidable Adverse Impacts

All the alternatives have the potential for significant adverse impacts to historic, archaeological, and cultural resources in the MICs. Such impacts can include physical alteration, damage, or destruction of all or part of a resource; alteration of the characteristics of the surrounding environment that contribute to the property's significance; and the introduction of visual or audible elements that are out of character with the property. Such impacts could alter the characteristics of a historic property in such a way as to diminish its integrity thus affecting its eligibility to qualify for inclusion in the NRHP. No additional significant adverse impacts are anticipated under the Action Alternatives as compared to Alternative 1 No Action.

Advanced planning to eliminate, minimize, or avoid impacts to cultural resources is crucial under all of the alternatives. Appropriate mitigation should be established and implemented by coordinating with the area's Tribes, the lead agency, and all other stakeholders and consulting parties in accordance with DAHP Mitigation Options and Documentation Standards. The ultimate outcome of such mitigation is to moderate the adverse impacts to historic, archaeological, or cultural resources before they are lost or significantly altered. With mitigation, significant impacts to historic, archaeological, and cultural resources can be avoided.

Section 3.12

Open Space & Recreation



Open space and recreation discussed in this section includes parks, trails, public shoreline access, and water access. The primary government agency offering these facilities is Seattle Parks and Recreation (SPR). The Port of Seattle also provides shoreline access and recreational opportunities in the study area(s). The Seattle Office of Sustainability and Environment (OSE) also provides partnership and coordination to advance equity and environmental justice goals. Open space and recreation facilities exist both within the Primary Study Area and in industrially zoned land in close proximity to the study area.

Impacts of the alternatives on open space and recreation are considered significant if they:

- Result in insufficient parks, open space, and trail capacity to serve expected population or employment based on levels of service.
- Feature inconsistencies with shoreline public access policies.
- Have the potential to decrease public access to parks and open space or shoreline access in census tracts identified as high disadvantage in the Seattle Racial and Social Equity Composite Index.¹⁹

3.12.1 Affected Environment

Data & Methods

Information about open space and recreation was collected from Seattle Parks and Recreation (SPR) and the Seattle Parks District. The plans and studies include the SPR Recreation Demand Study, Community Center Strategic Plan (2016), Parks, Recreation and Open Space (PROS) Plan (2017), Parks and Recreation Strategic Plan (2020). The annual reports from the Seattle Park District Annual Reports (2016-2019), Seattle Comprehensive Plan (2015), and Duwamish Valley Action Plan (2018) are also referenced.

Level of Service (LOS)

The City of Seattle sets level of service (LOS) standards for open space and recreation across the City. These standards are intended to help the City meet its "Citywide Open Space goal" or "Acceptable Open Space Guideline" to provide guidance and measure if park acres and facilities are meeting population growth and density, With the passage of several parks levies containing robust acquisition priorities, Seattle Parks and Recreation (SPR) has maintained and exceeded the Acceptable Population-based Open Space Goal of 1/3 acre per 100 residents since 2001 to 2016 (Seattle Parks and Recreation 2017).

With growth projections anticipating 120,000 new residents in the next 17 years, the 2017 Parks and Open Space Plan changed the Citywide acceptable guideline of 3.33 acres per 1,000

¹⁹ See the <u>Racial and Social Equity Index Interactive Map</u>, 2017.

residents to a new 8 acres per 1,000 residents LOS that is needed to help provide recreational opportunities (Seattle Parks and Recreation 2017). The assumption of 8 acres of park and recreation facilities per 1,000 residents is used throughout this impacts analysis to open space and recreation.

In addition to this measure, the PROS Plan (2017) included comments from the Seattle Planning Commission about additional measures. The Seattle Planning Commission listed the measures below as metrics that could be used to assess Seattle's open space and recreation needs.

- Size and percentage of City Land: The median size of parks and park acreage as a percentage of a city's land area are two additional metrics that are related to the amount of parkland.
- Park Pressure: Park pressure is a lesser known, but helpful metric that refers to the potential demand on a park, assuming that the residents in a "parkshed" use the park closest to them.
- Quantity and Variety of Park Amenities: Communities should regularly assess their amenities, including playgrounds, swimming pools, sport courts and playfields, skate parks, picnic shelters, splash pads, gymnasiums, recreation centers, senior centers, restrooms, etc.
- **Condition of Park Amenities:** The condition or quality of park amenities is a key measure of park adequacy.

Resources

Seattle Parks and Recreation (SPR) manages a 6,414-acre park system with over 485 parks and natural areas. This system includes athletic fields, play areas, gardens, trails, facilities and community centers, swimming pools, education centers, golf course, and skateparks. The SPR system comprises about 12% of Seattle's land area.

The study area, the subareas, and the parks and recreation facilities available are identified in the map below (see **Exhibit 3.12-1**).



Exhibit 3.12-1 City of Seattle Parks, Recreation, and Public Shoreline Access

Source: BERK, 2021.

Within the study area there are 22 parks, 19 trails, and 34 Shoreline access points. These facilities combine to provide 93.23 acres of parkland and 15.9 miles of trails. Together these facilities include greenbelts, multi-use trails, bike trails, rentable picnic shelters, picnic benches, green spaces, and playfields. There are also several parks that continue outside the study area totaling about 442 acres (see **Exhibit 3.12-2**).

Subarea	Total Acres	Subarea Acres
Ballard	27.81	25.22
Interbay Dravus	7.39	0.00
Interbay Smith Cove	42.31	29.59
SODO/Stadium	59.13	24.61
Georgetown/South Park	305.04	13.81
Total	441.68	92.01

Exhibit 3.12-2 Parks in Study Area

Source: Seattle GIS, 2021; BERK, 2021.

Multiuse trail miles are also found in each subarea (Exhibit 3.12-3).

Exhibit 3.12-3 Trails in Study Area

Subarea	Trail Length (Miles)
Ballard	2.2
Interbay Dravus	1.3
Interbay Smith Cove	3.5
SODO/Stadium	6.3
Georgetown/South Park	2.7
Total	15.9

Source: Seattle GIS, 2021; BERK, 2021.

Shoreline access is available in each subarea, with more access points in Ballard than in other subareas (**Exhibit 3.12-4**).

Exhibit 3.12-4 Shoreline Access Points

Subarea and Shoreline Access Type	Count
Ballard	11
Boat/Kayak Access	3
View Only	6
Water Access	2
Interbay Dravus	2
Boat/Kayak Access	1
View Only	1
Interbay Smith Cove	1
Water Access	1
SODO/Stadium	12
View Only	12
Georgetown/South Park	8
Boat/Kayak Access	1
View Only	5
Water Access	2
Total	34

Source: Seattle GIS, 2021; BERK, 2021.

Boat ramps include three in Ballard/Lake Union (two motorized and one hand carry), and one in Georgetown (hand carry)

- 14th Avenue NW Boat Ramp (Ballard)
- Sunnyside Avenue N Boat Ramp (Ballard)
- Fairview Walkway Boat Launch (Lake Union)
- Duwamish Waterway Park Boat Launch (Georgetown)

The exhibits below highlight the parks, trails, and shoreline access points within and around the study area (see **Exhibit 3.12-5**, **Exhibit 3.12-6**, and **Exhibit 3.12-7**, respectively). Instances where parks, trails, and shoreline access points are owned or managed by another agency are noted.

Seattle Parks	Size (Acres)	In Study Area (Y/N)
Ballard		
14th Ave NW Boat Ramp	0.018	Yes
Fremont Canal Park	0.095	Yes
Fairview Walkway	0.0017	Yes
Gas Works Park	21.35	Yes
Waterway 20 (managed by the Department of Natural Resources)	1.77	Yes
Northlake Park	0.016	Yes
Terry Pettus Park	0.097	Yes
Waterway 19 (managed by the Department of Natural Resources)	1.86	Yes
Interbay Dravus		
Interbay Athletic Field	2.05	No. Abutting.
Interbay Golf Center	45.00	No. Abutting.
Interbay Smith Cove*		
Myrtle Edwards Park/Centennial Park (managed by the Port of Seattle)	4.80	Yes
Open Water Park	14.03	Yes
Smith Cove Park	9.51	Yes
SW Queen Anne Greenbelt	0.06	Yes
SODO/Stadium		
Herrings House Park (Tulaltx)	15.24	Yes
Longfellow Creek Greenspace	0.48	Yes
Westbridge Shops	3.41	Yes
West Duwamish Greenbelt	5.47	Yes
Georgetown/South Park		
Duwamish Waterway Park	1.38	Yes
Georgetown Playfield	5.28	Yes
Georgetown Pump Station	0.20	Yes
West Duwamish Greenbelt	6.82	Yes
Westcrest Park	0.12	Yes
Total in Study Area:	92.01 acres	21 parks

	Exhibit 3.12-5	Parks in a	and Around	the Study	Area
--	----------------	------------	------------	-----------	------

Note: Park acres only includes parks that fall within the subarea boundary.

*West Central Grounds Maintenance is within the Interbay Smith Cove Subarea but is not a public park and is inaccessible to the public. Sources: Seattle GIS, 2021; BERK, 2021.

Seattle Trails	Trail Length (Miles)	In Study Area (Y/N)
Ballard		
Burke Gilman Trail (owned and maintained by SDOT)	2.1	Yes
Fremont Ave N	0.11	Yes
Interbay Dravus		
3rd Ave W	0.01	Yes
Ship Canal Trail	1.27	Yes
Interbay Smith Cove		
23rd Ave W	0.14	Yes
Elliott Bay Trail (owned and maintained by SDOT)	3.12	Yes
W Thomas St Overpass	0.21	Yes
SODO/Stadium		
Alki Trail	2.4	Yes
Duwamish River Trail	0.7	Yes
Portside Trail	0.6	Yes
SoDo Trail (owned and maintained by SDOT)	1.0	Yes
SW Alaska St	0.0	Yes
SW Spokane Br	0.4	Yes
SW Spokane St	0.2	Yes
W Sea Bridge Bike Trail	0.2	Yes
West Seattle Bridge Trail	0.6	Yes
Georgetown/South Park		
Duwamish River Trail (owned and maintained by SDOT)	1.8	Yes
S Portland St	0.4	Yes
West Marginal NB Way S	0.5	Yes
Total Tra	ails: 15.8 miles	19 trails

Exhibit 3.12-6 Trails in and Around the Study Area

Notes: Park acres only includes parks that fall within the subarea boundary. Sources: Seattle GIS, 2021; BERK, 2021.

Shoreline Access Points	Access Type	In Study Area (Y/N)
Ballard		
11th Ave NW and NW 45th St	View Only	Yes
20th Ave NW and Shilshole Ave NW	View Only	Yes
24th Ave NW and NW 54th St	Water Access	Yes
28th Ave NW and NW Market St	Boat/Kayak Access	Yes
Fremont Ave N and N 34th St	View Only	Yes
15th Ave NW and Ballard Br	Water Access	Yes
Eastlake Ave NE and University Br	View Only	Yes
14th Ave NW and Shilshole Ave NW	Boat/Kayak Access	Yes
Latona Ave NE and NE Northlake Way	View Only	Yes
Sunnyside and N and N Northlake Way	Boat/Kayak Access	Yes
3rd Ave NW and NW 39th St	View Only	Yes
Interbay Dravus		
6th Ave W and W Ewing St	Boat/Kayak Access	Yes
3rd Ave W and W Ewing N St	View Only	Yes
Interbay Smith Cove		
W Thomas St and Dead End	View Access	Yes
SODO/Stadium		
East Marginal Way S and S Spokane Sr St	View Only	Yes
Spokane St—W Sea B Rp and West Seattle Br Eb	View Only	Yes
SW Edmunds St and West Marginal Way SW	View Only	Yes
Diagonal Ave S and East Marginal Way S (Port of Seattle)	View Only	Yes
SW Alaska St and West Marginal Way SW	View Only	Yes
East Marginal Way S and S Idaho St	View Only	Yes
16th Ave SW and SW Lander St	View Only	Yes
Harbor Ave SW and SW Bronson Way	View Only	Yes
West Marginal Turn Rd and SW Spokane St	View Only	Yes
Klickitat Ave SW and Dead End (Port of Seattle)	View Only	Yes
26th Ave SW and Dead End 1	View Only	Yes
Chelan Ave SW and West Marginal Way SW	View Only	Yes
Georgetown/South Park	5	
5th Ave S and S Fontanelle St	View Only	Yes
7th Ave S and S Holden St	View Only	Yes
1st Ave S and SW Michigan St	Water Access	Yes

Exhibit 3.12-7 Shoreline Access Points in and Around the Study Area

Ch.3 Environment, Impacts, & Mitigation Measures • Open Space & Recreation

Shoreline Access Points	Access Type	In Study Area (Y/N)
1st Ave S and S Michigan S St	Boat/Kayak Access	Yes
East Marginal Way S and S Fidalgo St	View Only	Yes
10th Ave S and S Kenyon St	View Only	Yes
S Riverside Dr and Dead End 1	View Only	Yes
8th Ave S and S Portland St	Water Access	Yes
Total Shoreline Access Points:	34 access points	

Notes: Park acres only includes parks that fall within the subarea boundary. Shoreline Access points are owned by Seattle Parks and Recreation (SPR), the Seattle Department of Transportation (SDOT), the Port of Seattle, and King County. Sources: Seattle GIS, 2021; BERK, 2021.

The combination of parks, trails, and shoreline access points provide open space and recreation facilities within the industrially zoned areas of the study area. However, there are still gaps within the study area which have been identified.²⁰ As part of the 2017 SPR Gap Analysis Update, Seattle identified gaps in parks inside and outside of Urban Villages based on distance greater than 10-minutes to a park. As well, areas of greater population density were also considered.

Some of the gaps within the study area include:

- Ballard: There are limited gaps in walkability to parks in the subarea per the 2017 SPR Gap Analysis.²¹ There are portions of the subarea at the southern edge that are considered "gaps within of urban villages". Overall, the subarea is within the second lowest disadvantage per the 2017 SPR Gap Analysis.
- Interbay Dravus: There are no walkability gaps to parks per the 2017 SPR Gap Analysis.
 Overall, the subarea is within the second lowest disadvantage per the 2017 SPR Gap Analysis.
- Interbay Smith Cove: There are some gaps at the southern / southeastern edge of the subarea per the 2017 SPR Gap Analysis. Overall, the subarea is within the lowest and the second lowest disadvantage per the 2017 SPR Gap Analysis.
- **SODO/Stadium**: Nearly the full subarea is considered a "gap outside of urban villages" in the 2017 SPR Gap Analysis Update. In addition, parts of the subarea are considered "high disadvantage" within Seattle's Racial and Social Equity Composite Index.
- Georgetown/South Park: Nearly the full subarea is considered a "gap outside of urban villages" in the 2017 SPR Gap Analysis Update. In addition, per Seattle's Racial and Social Equity Composite Index parts of the subarea are considered "middle disadvantage," in Georgetown, and "second highest disadvantage" and "highest disadvantage" in South Park and areas along the west side of the Duwamish waterway.

²⁰ 2017 Gap Analysis Update Vol 1 (seattle.gov)

²¹ See 2017 Gap Analysis, available: <u>http://www.seattle.gov/ArcGIS/SMSeries_GapAnalysisUpdate2017/index.html</u>.

Demand

SPR anticipates parks demand in order to meet use of facilities for natural areas, trails and beaches, picnic shelters and community centers. SPR also considers the demand for sports fields which may be needed across seasons. These considerations are filtered through two methodologies based on the Recreation Conservation Office (RCO's) Planning Policies and Guidelines: Recreation Participation, and Community Satisfaction.

Measures of demand related to recreation participation include how many people use specific park facilities and the frequency of use within a year. From this information, SPR determines for each type of recreation/sports facility long-term need based on how people currently use facilities and any projected population changes.

Measures of demand related to community satisfaction include community rankings of different recreation services and facilities, feedback on resource allocation to different park types and facilities, facility use requests, and community priorities.

Goals

The City of Seattle PROS Plan (2017) provides open space and recreation goals citywide. The goals from this plan are outlined below.

PROS Plan Goals

- **Goal 1:** Provide a variety of outdoor and indoor spaces throughout the city for all people to play, learn, contemplate, and build community.
- **Goal 2:** Continue to provide opportunities for all people across Seattle to participate in a variety of recreational activities.
- **Goal 3:** Manage the city's park and recreation facilities to provide safe and welcoming places.
- **Goal 4:** Plan and maintain Seattle's parks and facilities to accommodate park users and visitors.
- **Goal 5:** Engage with community members on parks and recreation plans, and design and develop parks and facilities, based on the specific needs and cultures of the communities that the park is intended to serve.

These goals are not specific to the study area or subareas within. These goals also do not focus on open space and recreation in industrially zoned areas. The PROS plan does identify industrial lands as an opportunity for increasing the total available parkland in the City. There are, however, goals for the Georgetown/South Park Subarea which are outlined in the Duwamish Valley Action Plan (City of Seattle 2018).

The Action Plan is organized into seven priority areas: Healthy Environment, Parks & Open Spaces, Community Capacity, Mobility & Transportation, Economic Opportunity & Jobs,

Affordable Housing, and Public Safety. The Plan identifies the following goals for parks and open spaces:

- Increased area of parks and open space per capita in the Duwamish Valley.
- Culturally appropriate programming that meets the needs of the community members in the Duwamish Valley.
- Increased public access to the Duwamish River.

In the new SPR 2020-2032 Strategic Plan a "pathway to equity" is outlined as a commitment to creating an equitable parks and creation system. This commitment is supported by potential equity access goals (see **Exhibit 3.12-8**).

Facility Type	Target Goals
Community Centers	Every household in Seattle should be within 1-2 miles of a Community Center.
Aquatic Facilities	Every household in Seattle should have access to a swimming pool or swimming beach within 4 miles.
Outdoor Sports Courts and Facilities	80% of all residents will rate their access to desired outdoor facilities, such as tennis and basketball courts, as Good or Excellent.
Sports/Athletic Fields	Every household in Seattle should have access to sports fields within 2 miles.
Greenways	Continue to coordinate with SDOT on preferred routes and connections to enhance access to parks and open space.
Picnic Shelters	All picnic shelters should be ADA accessible.
Play Areas	All play areas should include facilities for a range of age groups.

Exhibit 3.12-8 Potential Equity Access Goals

Source: Seattle Parks and Recreation Open Space Plan, 2017.

These potential goals are coupled with SPR's equity commitments. These commitments include focusing work in a way that seeks to eliminate racial health disparities, seeks to minimize the impacts of climate change on those most vulnerable, strengthen outreach and engagement opportunities, and allocate resources strategically though a racial equity framework.

3.12.2 Impacts

This section considers the potential impacts to open space and recreation that may occur as a result of implementation of the alternatives. Impacts and resulting mitigation measures to open space and recreation have been assessed based on thresholds of significance.

The thresholds of significance utilized in this impact analysis include:

- Insufficient parks, open space, and trail capacity to serve expected population or employment based on levels of service.
- Inconsistencies with shoreline public access policies.
- Have the potential to decrease public access to parks and open space or shoreline access in census tracts identified as high disadvantage in the Seattle Racial and Social Equity Composite Index.²²

Impacts Common to All Alternatives

Changes driven by housing and employment are anticipated to increase population growth within the study area. As discussed in the affected environment above, the City of Seattle maintains a goal of 8 acres of parkland per 1,000 residents. This means that across all alternatives population growth will have an impact on the acres of parkland required within the study area and the subareas.

The primary possible impacts across alternatives would be demand on existing parks and demand for future parkland. Additional impacts specific to the subareas could be connectivity. These impacts are discussed below.

Population Growth

Anticipated population growth may add pressure on existing parks within the study area. Park pressure is a metric that refers to the potential demand on a park, assuming that the residents in a "parkshed" use the park closest to them (Seattle Parks and Recreation 2017). The number of park acres available per 1,000 people within the parkshed can be used to determine demand on existing parks. It is possible that population growth will decrease the number of park acres available per 1,000 people. This impact coupled with frequency of use and availability of park amenities contribute to an impact for all alternatives.

Based on the existing conditions and the City's current LOS standard for open space and recreation the City anticipates needing additional parkland. To meet the baseline of 8 acres per 1,000 residents the City is currently considering acquiring parkland through greenbelts, natural areas, and non-SPR owned open space such as plazas downtown, college and university campus land, and industrial lands (Seattle Parks and Recreation 2017).

²² See the <u>Racial and Social Equity Index Interactive Map</u>, 2017.

Connectivity/Transportation

The Action Alternatives propose three new land use concepts: Maritime, Manufacturing, and Logistics (MML), Industry and Innovation (II), and Urban Industrial (UI). Each concept includes development standards, some of which would influence the transportation network and/or transportation behavior. The proposals include standards for pedestrian and cyclist-oriented frontage improvements (sidewalks, pedestrian lighting, street trees, etc.) in the Industry & Innovation and Urban Industrial zones.

Open Space & Recreation Effects of Proposed Land Use Concepts

Maritime, Manufacturing, and Logistics (MML)

The Maritime, Manufacturing, and Logistics (MML) land use concept is designed to be applied in locations near infrastructure that supports fishing, logistics, maritime, aerospace, brewing and distilling activities. If adopted, this land use concept would amend the land use code to increase policy and zoning protections for maritime and industrial uses. This land use concept may have a minimal impact on open space and recreation in the form of parkland because it applies to existing industrial areas and uses. There are opportunities for shoreline public access where there is not a conflict with public safety.

Industry and Innovation (II)

The Industry and Innovation (II) land use concept is designed to support a combination of design and research industrial uses along with high-density employment and transit access. If adopted this land use concept would amend the land use code to support non-industrial office or technology uses and integration of high-capacity transit. Within this concept, open space and recreation could feature small greenspace increases through trees and landscaping. The location of a light rail station would increase foot traffic in and around the area and could lead to cyclist-oriented trails and plazas associated with employment buildings.

<u> Urban Industrial (UI)</u>

The Urban Industrial (UI) land use concept is designed to create industrial districts that can serve a mix of uses including manufacturing, production, and arts. This land use concept is also an opportunity to support place making and would be located in areas adjacent to Seattle's designated urban villages.

Within this concept, open space and recreation would be impacted in several different ways. This concept allows industrial uses to be integrated near urban villages which leads to the need for green open spaces, safe trails and routes that can be used for travel and as an industrial buffer, and park space to support any housing in new mixed-use buildings. If adopted, this land use concept would increase the opportunity for mixed-use housing leading to a more stable population in the area. This population would need access to open space and recreation.

Equity & Environmental Justice Considerations

Heat Islands

Based on a King County and City of Seattle Study of heat mapping, "surface-level temperatures in areas with paved landscapes, less tree canopy, and industrial activity are substantially higher during summer heat events compared to less urbanized areas." The study published in June 2021 shows that by evening, the Greater Duwamish MIC vicinity has higher levels of heat (see **Exhibit 3.12-9**). Adding trees in streetscapes, private properties, and parklands can help reduce the heat island effect.



Exhibit 3.12-9 Heat Watch and King County Results

Source: King County and City of Seattle, 2021.

Pathway to Equity

In the SPR 2020-2032 Strategic Plan, the City outlined a commitment to addressing historical racial inequities in parks and open space. In the plan a "pathway to equity" is used to describe this commitment to creating an equitable parks and recreation system. The pathway includes the following steps (Seattle Parks and Recreation 2020):

- Developing an SPR Equity and Engagement Plan to implement the City's equity goals.
- Developing an equity scorecard and map for resource allocation and planning that leverages data to identify and address disparities in underserved areas and for underserved groups.
- Revamping SPR's Race and Social Justice Initiative Outcomes, Strategies, and Actions (ROSA)10 to more intentionally ensure an equity lens is woven throughout SPR work.
- Training all SPR staff about the Pathway to Equity.
- Conducting robust and culturally responsive community outreach and engagement.
- Developing an equity dashboard and performance indicators as part of departmentwide performance management efforts.

A combination of these actions could improve equitable outcomes within the study area. A map for resource allocation, an equity dashboard, and community outreach and engagement would each provide opportunities for the City to assess current disparities and create solutions with the community.

Park Pressure & Park Access

The demand on existing parks was discussed above under Impacts Common to all Alternatives. In addition to park demand being an impact for the study area there are also equity implications of park pressure. Research has demonstrated that park pressure can be used to highlight racial inequities in park access, showing that people of color and low-income groups are more likely to live close to parks with higher potential park congestion (Seattle Parks and Recreation 2017). This is most notable for park access in the Georgetown/South Park Subarea.

In Georgetown and South Park neighborhoods (within and outside of the Georgetown portion of the Greater Duwamish MIC) access to public space is comparable and, in some cases, better than the City as a whole. Georgetown and South Park scored 77 and 80 (Public Space Access Score out of 100) respectively in comparison to Seattle which scored 73 (see **Exhibit 3.12-10**).

Exhibit 3.12-10 Access to Public Space in Georgetown and South Park

ACCESS TO	PUBLIC SPACE [®]
Public Space Acc	ess Score (out of 100)
	South Park 77 Georgetown 80 Seattle 73
Households with	in 5 Minute Walk of a Public Space
3 /	South Park 56% Georgetown 99% Seattle 53%
Households with	in 10 Minute Walk of a Public Space
	South Park 94% Georgetown 100% Seattle 90%
⁸ City of Seattle Data	

Source: Seattle Duwamish Valley Action Plan (Action Plan), 2018.

While the neighborhoods have nearby parks, the total acreage per capita is half the citywide average and there may be park congestion caused by added population. Another factor related to park pressure and park access is being able to travel to and from the parks.

Impacts of Alternative 1 No Action

Alternative 1 prohibits residential uses within industrial zones except for caretaker quarters per industrial business, artist studio housing, and housing the existed before industrial zoning. Only about 75 of these industrial zone related dwellings are projected.

Growth is still expected under Alternative 1 No Action from naturally occurring population growth in the city (under current zoning) with small amounts of housing in the study area. The 2017 PROS Plan includes an aspirational LOS standard needed to accommodate the projected 120,000 additional residents citywide by 2035 (Seattle Parks and Recreation 2017). The number

of acres of parkland needed to address population growth under Alternative 1 is presented in **Exhibit 3.12-11**.

	Current Conditions (2018)		Alternative 1 No Action— Existing Policies (2044)	
Subarea	Existing Pop	Existing Open Space (Acres)	Expected Pop Growth	Open Space for Net Growth (Acres)
Ballard	394	25.21	15	0.12
Interbay Dravus	6	0.00	15	0.12
Interbay Smith Cove	2	28.40	15	0.12
SODO/Stadium	43	24.60	62	0.50
Georgetown/South Park	402	13.80	46	0.37
Total	847	92.01	153	1.22

Source: BERK, 2021.

The summary presented in **Exhibit 3.12-11** assumes the City maintains its desired Level of Service (LOS) standard of 8 acres of parkland per 1,000 people. Under Alternative 1, the City would need to add an additional 1.22 acres of parkland to accommodate 153 additional residents within the study area.

Growth and associated acres of needed parkland are expected to be highest in the SODO/Stadium Subarea (0.50 acres) followed by the Georgetown/South Park Subarea (0.37) under Alternative 1. The remaining subareas—Ballard (0.12 acres), Interbay Dravus, (0.12) and Interbay Smith Cove (0.12)—would have the same need for additional acres. No impacts other than those described under Impacts Common to All Alternatives are anticipated under Alternative 1.

Impacts of Alternative 2

Alternative 2 includes less land zoned UI and II than the other two Action Alternatives. This alternative would result in more job creation and minimal residential growth.

Growth under Alternative 2 is anticipated to have a minimal increase on the population (163 people). The number of acres of parkland needed to address population growth under Alternative 2 is presented in **Exhibit 3.12-12**.

	Current Con	ditions (2018)	Alternative 2—Future of Industry Limited (2044)		
Subarea	Existing Pop	Existing Open Space (Acres)	Expected Pop Growth	Open Space for Net Growth (Acres)	
Ballard	394	25.21	16	0.13	
Interbay Dravus	6	0.00	16	0.13	
Interbay Smith Cove	2	28.40	16	0.13	
SODO/Stadium	43	24.60	66	0.53	
Georgetown/South Park	402	13.80	49	0.39	
Total	847	92.01	163	1.30	

Source: BERK, 2021.

The summary presented in **Exhibit 3.12-12** indicates a similar degree of change as seen in Alternative 1. Under Alternative 2, the City would need to add an additional 1.3 acres of parkland to accommodate 163 additional residents within the study area.

Similar to Alternative 1 No Action, growth and associated acres of needed parkland under Alternative 2 is expected to be highest in the SODO/Stadium Subarea (0.53 acres) followed by the Georgetown/South Park Subarea (0.39 acres). The remaining subareas—Ballard (0.13 acres), Interbay Dravus (0.13), and Interbay Smith Cove (0.13)—would each have the same need for additional acres of open space and recreation. No impacts other than those described under Impacts Common to All Alternatives are anticipated under Alternative 2. There will be impacts to existing open space and recreation facilities and a need for new facilities to meet anticipated demand.

Impacts of Alternative 3

Alternative 3 includes a higher degree of UI and II zoned land than Alternative 1 No Action and Alternative 2. Alternative 3 would result in a combination of industry/innovation and urban industrial zone concepts in existing areas industrially zoned and would expand limited industrysupportive housing in areas where the UI zone concept is featured. This UI zone concept is most featured in the Ballard, the SODO/Stadium, and pockets of the Georgetown/South Park subareas.

Growth under Alternative 3 is anticipated to have a larger increase in the population living in or near industrially zoned areas than alternatives 1 or 2. The number of acres of parkland needed to address population growth under Alternative 3 is presented in **Exhibit 3.12-13**.

	Current Conditions (2018)		Alternative 3—Future of Industry Targeted (2044)		
Subarea	Existing Pop	Existing Open Space (Acres)	Expected Pop Growth	Open Space for Net Growth (Acres)	
Ballard	394	25.21	533	4.26	
Interbay Dravus	6	0.00	154	1.23	
Interbay Smith Cove	2	28.40	31	0.25	
SODO/Stadium	43	24.60	410	3.28	
Georgetown/South Park	402	13.80	123	0.98	
With MIC Adjustments	0.00	0.00	2,210	17.68	
Total	847	92.01	3,461	27.68	

Exhibit 3.12-13	Open Space and	Recreation Acres	Required for	Alternative 3
-----------------	-----------------------	-------------------------	---------------------	---------------

Source: BERK, 2021.

The summary presented in **Exhibit 3.12-13** indicates a much greater degree of change compared to Alternative 1 and Alternative 2. Under Alternative 3, the City would need to add an additional 27.68 acres of parkland to accommodate 3,461 additional residents within the study area. This increase in acres of open space and recreation is slightly below the PROS Plan estimates—in the plan, SPR anticipated it would need to acquire at least 40 acres of parkland to meet the adopted LOS by 2035. However, that estimate was for the entire city and not the study area alone.

Alternative 3 also includes MIC adjustments that would result in population growth. The population growth anticipated from these MIC adjustments accounts for 17.68 acres of the total 27.68 acres of parkland needed under Alternative 3. The need for more open space and recreation is highest in the Ballard (4.26 acres) and SODO/Stadium (3.28 acres) subareas, followed by the Interbay Dravus (1.23 acres), Georgetown/South Park (0.98 acres), and Interbay/Smith Cove (0.25 acres) subareas.

In addition to the impacts described under Impacts Common to All Alternatives above there could be impacts under Alternative 3 caused by balancing industrial uses with housing and transportation. The impact of Alternative 3 may also limit the types of open space and recreation to facilities other than parks. Considering the Seattle Racial and Social Equity Composite Index, the SODO/Stadium Subarea is within the highest 20% disadvantage of census tracts and the Georgetown/South Park Subarea falls within the middle 40-60% of disadvantaged tracts.²³ An increase in population in the Georgetown/South Park Subarea may place additional pressure on existing parks and more parkland needs to be acquired and

²³ See the <u>Racial and Social Equity Index Interactive Map</u>, 2017.

developed to meet demand in the SODO/Stadium Subarea. For both of these subareas, there will not be a limit on park access if more park acres are acquired.

Alternative 3 includes the removal of portions of two blocks of land adjacent to Duwamish Waterway Park and two blocks of land adjacent to Terminal 117/Duwamish River People's Park from the MIC designation and industrial zoning and would apply a mixed-use zone. Future development in the mixed-use zone has a higher potential for increasing integration with and access to the two open spaces from the South Park residential community. The change will increase the amount of required open space in new development near the parks and will increase the likelihood of future visual and/or physical access to river front land from privately owned parcels.

Impacts of Alternative 4

Alternative 4 features a higher degree of UI and II land use concepts than the Alternative 1 and Alternative 2. This alternative would result in a combination of industry/innovation and urban industrial zone concepts in existing areas industrially zoned. Regarding residential development, Alternative 4 would expand limited industry-supportive housing in areas where the UI zone concept is featured. This UI zone concept is most featured in Ballard, the SODO/Stadium, and pockets of the Georgetown/South Park subareas.

Growth under Alternative 4 is anticipated to have a large increase in the population living in or near industrially zoned areas that is greater than alternatives 1 or 2, and similar to the amount in Alternative 3. The number of acres of parkland needed to address population growth under Alternative 4 is presented in **Exhibit 3.12-14**.

	Current Con	ditions (2018)	Alternative 4—Future of Industry Expanded (2044)		
Subarea	Existing Pop	Existing Open Space (Acres)	Expected Pop Growth	Open Space for Net Growth (Acres)	
Ballard	394	25.21	1,620	12.96	
Interbay Dravus	6	0.00	359	2.87	
Interbay Smith Cove	2	28.40	0	0.00	
SODO/Stadium	43	24.60	2,030	16.24	
Georgetown/South Park	402	13.80	492	3.94	
With MIC Adjustments	0.00	0.00	2,210	17.68	
Total	847	92.01	6,710	53.68	

Source: BERK, 2021.

The summary presented in **Exhibit 3.12-14** indicates a much larger degree of change compared to Alternative 1 and Alternative 2. Under Alternative 4, the City would need to add an additional 53.68 acres of parkland to accommodate 6,710 additional residents within the study area. This increase in acres of open space and recreation would exceed the PROS Plan's estimated 40 additional acres needed citywide.

Like Alternative 3, Alternative 4 includes MIC adjustments that would result in population growth. The population growth anticipated from these MIC adjustments accounts for 17.68 acres of the total 53.68 acres of parkland needed under Alternative 4. The need for more open space and recreation is highest in the SODO/Stadium (16.24) and Ballard (12.96 acres) subareas. The smallest increases in Alternative 4 would occur in the Georgetown/South Park (3.94 acres) and Interbay Dravus (2.87 acres) subareas. However, with the SM zoned areas, there would also need to be 17.68 acres in the Georgetown/South Park Subarea. Each of the alternatives feature a concentration of growth and subsequent demand for open space in SODO/Stadium Subarea.

In addition to the impacts described under Impacts Common to All Alternatives above there could be impacts under Alternative 4 that were discussed in Alternative 3. There is a need to balance industrial uses with housing and transportation. Most notably, in Alternative 4 the UI land use concept is featured throughout the study area which will result in new mixes of uses that may have been industrially zoned previously.

Considering the Seattle Racial and Social Equity Composite Index, the Ballard Subarea is within the lowest 40% of disadvantaged tracts across the city, while the SODO/Stadium Subarea is within the highest 20% disadvantage of census tracts.²⁴ For the SODO/Stadium Subarea in particular, there is an existing limit of available parkland; an increase in population would lead to the need for more park land in the subarea.

Alternative 4 includes the removal of portions of two blocks of land adjacent to Duwamish Waterway Park and two blocks of land adjacent to Terminal 117/Duwamish River People's Park from the MIC designation and industrial zoning and would apply a mixed-use zone. Future development in the mixed-use zone has a higher potential for increasing integration with and access to the two open spaces from the South Park residential community. The change will increase the amount of required open space in new development near the parks and will increase the likelihood of future visual and/or physical access to river front land from privately owned parcels.

Impacts of the Preferred Alternative

<u>The Preferred Alternative provides a combination of MML, II and UI zones with some</u> <u>allowances for industry-supportive housing in areas where the UI zone concept is featured.</u>

²⁴ See the <u>Racial and Social Equity Index Interactive Map</u>, 2017.

Population growth under the Preferred Alternative is anticipated to be greater than alternatives 1, 2, and 3 but less than Alternative 4. The number of acres of parkland needed to address population growth under the Preferred Alternative is presented in **Exhibit 3.12-15**.

	Current Conditions (2018)		Preferred Alternative (2044)		
<u>Subarea</u>	Existing Pop	<u>Existing Open</u> Space (Acres)	<u>Expected Pop</u> <u>Growth</u>	<u>Open Space for Net</u> <u>Growth (Acres)</u>	
Ballard	<u>394</u>	<u>25.21</u>	<u>1,054</u>	<u>8.43</u>	
Interbay Dravus	<u>6</u>	<u>0.00</u>	<u>234</u>	<u>1.87</u>	
Interbay Smith Cove	<u>2</u>	<u>28.40</u>	<u>0</u>	<u>0.00</u>	
SODO/Stadium	<u>43</u>	<u>24.60</u>	<u>1,320</u>	<u>10.56</u>	
Georgetown/South Park	<u>402</u>	<u>13.80</u>	<u>418</u>	<u>3.35</u>	
With MIC Adjustments	<u>0.00</u>	<u>0.00</u>	<u>3,145</u>	<u>25.16</u>	
Total	847	<u>92.01</u>	<u>6,168</u>	<u>49.36</u>	

Exhibit 3.12-15 Open Space and Recreation Acres Required for Preferred Alternative

Source: BERK, 2021.

The summary presented in **Exhibit 3.12-15** indicates the City would need to add an additional 49.36 acres of parkland to accommodate 6,168 additional residents within the study area. This increase in acres of open space and recreation would exceed the PROS Plan's estimated 40 additional acres needed citywide. This would be more than the demand under Alternative 3 but less than Alternative 4.

Where industry supportive housing is provided, there could be an increase in park demand especially in the Ballard and SODO/Stadium subareas. In addition to MIC adjustments to Seattle Mixed zoning in South Park and Georgetown, there would be mixed use Neighborhood Commercial zoning in West Ballard and Judkins Park creating more demand for parkland.

In addition to the impacts described under **Impacts Common to All Alternatives** above there could be impacts under the Preferred Alternative similar to alternatives 3 and 4. There is a need to balance industrial uses with housing and transportation.

<u>Considering the Seattle Racial and Social Equity Composite Index, the Ballard Subarea is within</u> <u>the lowest 40% of disadvantaged tracts across the city, while the SODO/Stadium Subarea is</u> <u>within the highest 20% disadvantage of census tracts.²⁵ For the SODO/Stadium Subarea in</u> <u>particular, there is an existing limit of available parkland; an increase in population would lead</u> <u>to the need for more park land in the subarea.</u>

²⁵ See the Racial and Social Equity Index Interactive Map, 2017.

<u>Like alternatives 3 and 4, the Preferred Alternative includes the removal of portions of two</u> <u>blocks of land adjacent to Duwamish Waterway Park and two blocks of land adjacent to</u> <u>Terminal 117/Duwamish River People's Park from the MIC designation and industrial zoning</u> <u>and would apply a mixed-use zone. Future development in the mixed-use zone has a higher</u> <u>potential for increasing integration with and access to the two open spaces from the South Park</u> <u>residential community. The change will increase the amount of required open space in new</u> <u>development near the parks and will increase the likelihood of future visual and/or physical</u> <u>access to river front land from privately owned parcels.</u>

3.12.3 Mitigation Measures

Incorporated Plan Features

The Action Alternatives propose three new land use concepts: Maritime, Manufacturing, and Logistics (MML), Industry and Innovation (II), and Urban Industrial (UI). Each concept features design principles that would help mitigate impacts to open space and recreation:

- The Industry & Innovation land use concept includes standards for frontage improvements (sidewalks, pedestrian lighting, etc.), trees and landscaping, and maximum limits on vehicle parking areas. This concept would also include a need for circulation routes which could be used as trails.
- The Urban Industrial land use concept incorporates open space and landscaping, which support open space and recreation demand and help meet LOS standards. This concept also includes standards for frontage improvements (sidewalks, pedestrian lighting, etc.) and could make use of landscaping on or around buildings.
- The Maritime, Manufacturing, and Logistics land use concept could result in the location of new boat ramps and shoreline access areas within the study area.
- Alternatives 3 and 4 remove land adjacent to parks in the South Park neighborhood from a MIC designation, increasing the likelihood for increasing integration with and access to river front open spaces from the South Park residential community.

Regulations & Commitments

The study area is located within King County in the City of Seattle. Open space and recreation in Seattle is managed by separate local governments with overlapping boundaries. Relevant plans include SPR's Recreation Demand Study, Community Center Strategic Plan, PROS Plan, and Parks and Recreation Strategic Plan. Additional open space and recreation needs and commitments are identified in annual reports from the Seattle Park District Annual Reports, the Seattle Comprehensive Plan, and the Duwamish Valley Action Plan. While not located in the study area, north of the Greater Duwamish MIC are plans for the Seattle Waterfront including a park promenade and bike path. Another potential concept includes the potential for Pier 48 as a park.

These various plans provide a framework for the City when assessing and planning for open space and recreation needs. The SPR Strategic Plan provides strategies arranged by healthy people, healthy environment, strong communities, and organizational excellence. The Duwamish Valley Action Plan builds upon the Equity & Environment Agenda and the Duwamish Valley Program, two commitments from the City to genuinely collaborate with communities to further social justice goals in policy and development. The PROS Plan outlines the City's existing open space and recreational facilities, capital funding, and projects being funded and a 6-year vision for the future.

In addition to these plans, the Seattle Land Use Code (Seattle Municipal Code Title 23) contains development regulations, including standards governing the design and placement of exterior site and building illumination. Future development in the study area will be required to comply with the standards established for industrial zones in SMC Chapter 23.50 and 23.49 as it pertains to open space.

Other Potential Mitigation Measures

While parks are a great source of open space, the combination of existing uses and new land use concepts within the alternatives may present challenges that may not be resolved with new parks. Other potential mitigation measures the City could explore outside of creating new parks include creating linear parks and trails, increasing frequency of maintenance to offset an increase in park usage, and building resilient parks. The City could also explore transportation to and from parks and potentially increase connectivity between parks. Finally, the City might explore the use of community gardens (permitted on some rooftops in individual zones) as a way to provide open space and an urban agricultural use.

3.12.4 Significant Unavoidable Adverse Impacts

No significant unavoidable adverse impacts to open space and recreation are anticipated. While population and employment growth would occur under all studied alternatives, there are opportunities to meet the City's level of service for parkland through implementation of the Seattle plans and current and proposed development regulations.

Section 3.13 Public Services



Public services discussed in this section include fire, police, school, and library services. The primary providers of these services for the study area are the Seattle Fire Department (SFD), the Seattle Police Department (SPD), the Port of Seattle Police Department (POSPD), Seattle Public Schools (SPS), and the Seattle Library System (SLS). The Primary Study Area includes industrially zoned lands both inside and outside of the manufacturing industrial centers. Secondary Study Areas include fire stations, police stations, schools, and libraries in proximity to the Primary Study Area.

Impacts of the alternatives on public services are considered significant if they:

- Negatively affect the response times for police and/or fire and emergency medical services.
- Increase demand for special emergency services beyond current operational capabilities of service providers.
- Result in increases in students and lack of facilities unanticipated in district plans or that would reduce adopted levels of service.

3.13.1 Affected Environment

Fire & Emergency Medical Services

Data & Methods

Information about fire and emergency medical services was collected from the Seattle Fire Department. SFD publishes an annual report each year which includes information about the department, incident response trends and response standards, preventative measures taken (e.g., fire code implementation), public events/education, and other notable highlights. The City of Seattle also publishes geolocated call data on its Open Data Portal. SFD's 2012-2017 Strategic Plan and the City's proposed 2022 Budget and 2022-2027 CIP were also referenced.

Services & Resources

The Seattle Fire Department provides fire and rescue response, fire prevention and public education, fire investigation, and emergency medical services (EMS) throughout the city, including the study area. Emergency medical services include basic life support (BLS) and advanced life support (ALS). SFD also has specially trained technical teams that provide technical and heavy rescue, dive rescue, tunnel rescue, marine fire/EMS response, and hazardous materials response. In addition, SFD provides mutual aid response to neighboring jurisdictions.

SFD provides emergency response services through five battalions consisting of 33 fire stations (plus Battalion 3/Medic One at Harborview Medical Center) strategically placed around the city to maximize coverage and minimize response time (see **Exhibit 3.13-1**). The study area is mostly within Battalions 4, 5, and 7 and is primarily served by the following stations:

- BINMIC: Stations 3, 5, 8, 9, 18, 20, and 41
- Greater Duwamish MIC: Stations 5, 10, 11, 13, 14, 26, 27, 29, 33, and 36

Marine fire response is provided by Station 3 at Fisherman's Terminal and Station 5 on Seattle's Waterfront (Station 5 is currently under construction concurrent with portions of the Seattle Waterfront project)—fire boats at these stations are prepared to respond to ship fires, marina fires, water rescues, and other water related emergencies. Other industrial lands along the north side of Salmon Bay are served by stations 9 and 17, and industrial lands in Eastlake are served by Station 22. Emergency support may come from other stations depending on resource needs and availability.

All SFD stations are staffed 24 hours a day, seven days a week, by four separate shifts of firefighters. There are 216 members responding to emergencies every day across the city (220 with upstaffing for 2 daytime aid cars). In total, SFD currently has 1,008 uniformed personnel and 77 civilian personnel—uniform personnel include 940 firefighter/EMTs (including 36 chiefs) and 68 firefighter/paramedics (Seattle Fire Department 2020).



Exhibit 3.13-1 Fire Battalions and Stations

Source: City of Seattle, 2021; BERK, 2021.

A variety of ladder trucks, fire engines, fireboats, aid cars (BLS), medic units (ALS), and other specialty teams are housed at stations serving the study area (see **Exhibit 3.13-2**). Ladder trucks and fire engines are staffed by teams of four personnel while aid cars are staffed by teams of two personnel. Medic One at Harborview Medical Center also provides the city with ALS activities that, in the past, could only be performed by licensed physicians. In addition to responding to medical emergencies, medic units respond to all working fires, hazardous materials, and rescue responses citywide.

Facility	Location	Equipment & Staffing
Headquarters*	301 2nd Avenue S	Does not serve as a working fire station, but houses the Executive Team, Deputy 1, Safety Chief, Fire Investigation Unit, and other administrative functions.
Medic One / Harborview Medical Center	325 9th Ave	Battalion 3, Medic 1, Medic 10, Medic 44, and Medic 55
3—Fisherman's Terminal	1735 W Thurman	Fireboat Chief Seattle, Fireboat 1, FB1, FB3 and FB4
5—Seattle Waterfront	925 Alaskan Way	Fireboat Leschi, Fireboat 2, Rescue Boat 5, Engine 5, and PT520. <i>Note: Station 5 is currently under construction.</i>
8—Queen Anne	110 Lee St	Fire Engine 8 and Ladder Truck 6
9—Fremont	3829 Linden Ave N	Fine Engine 9
10—International District	400 S Washington St	Fire Engine 10, Ladder Truck 1, Aid Car 5, Aid Car 10, Staff 10, and the Hazardous Materials Team—includes the city's Fire Alarm Center and Emergency Operation Center
11—Highland Park	1514 SW Holden St	Fire Engine 11
13—Beacon Hill*	3601 Beacon Ave S	Fire Engine 13 and Battalion 5
14—SODO District*	3224 4th Ave S	Ladder Truck 7, Aid Car 14, and Rescue One (Technical Rescue Team)
17—University District	1050 NE 50th St	Fire Engine 17, Ladder Truck 9, Medic 17, and Battalion 6
18—Ballard	1521 NW Market St	Fire Engine 18, Ladder Truck 8, Medic Unit (ALS) 18, Hose 18, and Battalion 4
20—West Queen Anne	2800 15th Ave W	Fire Engine 20
22—Roanoke	901 E Roanoke	Fire Engine 22, Command and Communications Van
26—South Park	800 S Cloverdale St	Fire Engine 26 and Medic Unit (ALS) 26**
27—Georgetown	1000 S Myrtle St	Fire Engine 27, REHAB1, and DECON1
29—Admiral District	2139 Ferry Ave SW	Fire Engine 29
33—Rainier Beach	9645 Renton Ave S	Fire Engine 33
36—Delridge & Harbor Island	3600 23rd Ave SW	Fire Engine 36 and Marine 1
41—Magnolia	3216 34th Ave W	Fire Engine 41

Exhibit 3.13-2	SFD Facility Locations,	Equipment, and	Staffing for Stations	Serving the Study Area
----------------	-------------------------	----------------	------------------------------	------------------------

Note: Ladder trucks and fire engines are staffed by teams of four personnel. Aid cars are staffed by teams of two personnel. *Indicates a historic building.

**SFD staffed an additional ladder truck (Ladder 13) and medic unit (Medic 26) to serve the residents of West Seattle in response to the closure of the West Seattle Bridge. Ladder Truck 13 is housed at Station 37 and Medic Unit 26 at Station 26. Source: Seattle Fire Department Annual Report, 2020; Seattle 2035 Capital Facilities Appendix, 2020.

Stations and associated equipment and staffing serving each of the subareas are summarized below:

- Ballard: The Ballard Subarea is in the service area of stations 9 and 18. Station 18—the primary station serving the Ballard portion of the study area—houses Fire Engine 18, Ladder Truck 8, Medic Unit (ALS) 18, Hose 18, and Battalion 4 while Station 9 in Fremont houses one fire engine.
- Interbay Dravus: The Interbay Dravus Subarea is in the service area of stations 3, 20, and 41. Station 3 at Fisherman's Terminal houses Fireboat Chief Seattle, Fireboat 1, FB1, FB3, and FB4. Station 20 in West Queen Anne and Station 41 in Magnolia each house a fire engine.
- Interbay Smith Cove: The Interbay Smith Cove Subarea is in the service area of stations 5, 8, and 20. Station 8 in Queen Anne houses a fire engine and ladder truck while Station 20 in West Queen Anne house a single fire engine. Station 5 houses two fire boats, one rescue boat, one fire engine, and the specialty unit PT520.
- SODO/Stadium: The SODO/Stadium Subarea is in the service area of stations 5, 10, 13, 14, 27, 29, and 36. Together these stations house two fire boats, six fire engines, two ladder trucks, three aid cars, and several specialty units, including SFD's Hazardous Materials Team, Rescue One (Technical Rescue Team), REHAB1, DECON1, Marine 1, and PT520.
- Georgetown/South Park: The Georgetown/South Park Subarea is in the service area of stations 11, 26, 27, and 33. Together these stations house four fire engines, one medic unit (ALS), REHBA1, and DECON1.
- Other Industrial Zoned Lands: Other industrial lands along the north side of Salmon Bay are served by stations 9 and 17, and industrial lands in Eastlake are served by Station 22. Stations 9 and 17 north of Salmon Bay house two fire engines, one ladder truck, one medic unit (ALS), and Battalion 6, while Station 22 in Eastlake houses one fire engine and SFD's Command and Communications Van.

Performance

Incident Response Trends

Between 2017 and 2020, total Seattle Fire Department incident responses ranged from 80,316 to 96,822. As shown in **Exhibit 3.13-3**, the number of total responses remained relatively constant in 2017 and 2018, then decreased in 2019 and 2020. The COVID-19 pandemic drove a decrease in EMS calls in 2020—a trend SFD believes resulted from fewer people being outside their homes coupled with a fear of being exposed to the virus—and a rise in fire responses. Total incident responses decreased from 2017-2019 by 5% and by 17% from 2017-2020.

Fire incident response increased 9% from 2017-2019. However, Seattle has fewer fires than the national average and of other cities with similar population size—Seattle averages 0.9 fires annually per 1,000 residents compared to the national average of 3.9 (Seattle City Budget Office 2021, 325). EMS incident responses decreased 7% from 2017-2019. The proportion of fire

incident responses compared to EMS incident responses has correspondingly increased. EMS calls still make up over three-quarters of total responses though, accounting for 81% of total responses in 2017 and 80% in 2019.

Year	EMS Incidents: BLS & ALS	Fire & Specialty Incidents	Other & Mutual Aid	Total
2017	78,758 (81%)	16,548 (17%)	1,111 (1%)	96,822
2018	76,484 (81%)	17,080 (18%)	1,128 (1%)	94,780
2019	72,980 (80%)	18,088 (20%)	648 (1%)	91,716
2020	61,717 (77%)	18,094 (23%)	505 (1%)	80,316

Exhibit 3.13-3 Seattle Fire Department Emergency Response Incidents, 2018-2020

Note: EMS incidents include BLS and ALS incidents. Other incidents include transfers to other agencies where a fire unit was also dispatched and, for 2020, includes responses where a single battalion chief was dispatched. Source: Seattle Fire Department Annuals Reports, 2019 and 2020.

SFD received fewer calls for service citywide and within the study area in 2020 than in the four years prior (see **Exhibit 3.13-4**), likely because of the overall decrease in EMS related calls as a result of the pandemic (EMS incidents make up about 80% of incidents overall). Within the study area, the fewest calls were received in the Ballard Subarea and the most were received in the SODO/Stadium Subarea. Calls for service in the study area decreased by 19% from 2019 to 2020 but stayed relatively constant citywide (increased by 0.4%). Less than 0.2% of calls for service citywide were located in the study area each year from 2016 to 2020. As shown in **Exhibit 3.13-5**, SFD calls for service from 2016-2020 were more heavily concentrated in non-industrial areas of the city, including Downtown, east of Downtown near the hospitals, and in areas with large institutions such as the University of Washington. The Georgetown/South Park Subarea received more calls from 2016-2020 than other parts of the study area.

Subarea	2016	2017	2018	2019	2020	Total
Ballard	17	12	13	8	12	62
Interbay Dravus	39	31	35	31	29	165
Interbay Smith Cove	27	19	13	35	12	106
SODO/Stadium	47	56	51	46	32	232
Georgetown/South Park	45	43	31	21	20	160
Study Area Total	175	161	143	141	105	725
Citywide Total	101,974	102,947	101,485	102,368	93,495	502,269

Exhibit 3.13-4 Calls for Fire and EMS Services by Subarea, 2016-2020

Note: Citywide calls for service are higher than the number of response incidents in **Exhibit 3.13-3** as not all calls for service result in in an emergency incident response.

Sources: Real Time Fire 911 Calls, 2021 (https://data.seattle.gov/Public-Safety/Seattle-Real-Time-Fire-911-Calls/kzim-xkqj); BERK, 2021.



Exhibit 3.13-5 Total Calls for Fire and EMS Services in the Study Area and Surrounding Vicinity, 2016-2020

Sources: Real Time Fire 911 Calls, 2021 (https://data.seattle.gov/Public-Safety/Seattle-Real-Time-Fire-911-Calls/kzjm-xkqj); BERK, 2021.
Response Time

Maintaining or improving emergency response times is the core of Seattle Fire Department operations (Seattle Fire Department 2012). SFD's response standards specify the minimum criteria needed to effectively and efficiently deliver fire suppression, special operations response, and emergency medical services (Seattle Fire Department 2020). The Capital Facilities Appendix of *Seattle 2035* establishes the following response time standards for the Department (City of Seattle 2020, 529-530):

- Call Processing Time: 60 seconds for phone answered to first unit assigned for 90% of calls.
- Fire Response Time: Arrival within 4 minutes for first-arriving engine at a fire for 90% of calls, and arrival within 8 minutes of the full first alarm assignment of 15 firefighters, for 90% of calls.
- Basic Life Support: Arrival within 4 minutes of the first medical unit with two EMTs, for 90% of calls.
- Advanced Life Support: Arrival within 8 minutes for 90% of calls.

Exhibit 3.13-6 shows the statistics the Department uses to measure response time performance. These statistics generally correspond with the Department's response time standards. Between 2016 and 2020 the Department fell short of meeting its response time standards, with the exception of meeting its call processing time standard in 2018 and its full first alarm assignment standard in 2018, 2019, and 2020.

Year	Call Processing Time within 60 Seconds	First Arriving Engine at Fire within 4 Minutes	Full First Alarm Assignment at Fire within 8 Minutes	First Arriving Unit for a BLS Incident within 4 Minutes	First Arriving Unit for an ALS Incident within 8 Minutes
Adopted Standard	90%	90%	90%	90%	90%
2016	85%	84%	86%	84%	89%
2017	84%	77%	71%	79%	86%
2018	92%	76%	93%	79%	86%
2019	64%	75%	94%	76%	86%
2020	66%	78%	92%	73%	81%

Exhibit 3.13-6 Response Statistics, 2016-2020

Note: SFD updated data for 2018 and 2019 in the 2020 Annual Report to reflect more accurate information from their system. 2016 and 2017 information are from the 2018 Annual Report.

Source: Seattle Fire Department Annual Report, 2018 and 2020.

Planning

Facilities

The Seattle Department of Finance & Administrative Services designs, builds, and maintains City-owned buildings, including fire facilities. They coordinate with SFD to ensure facility plans are consistent with strategic planning for fire services.

In 2003, a Fire Facilities and Emergency Response Levy was approved by Seattle voters. The levy provided funding for major facility improvements across the Seattle Fire Department including upgrades, renovations, or replacements of 32 neighborhood fire stations (including all stations serving the study area), construction of a new training facility, establishment of emergency preparedness facilities, renovation of the Chief Seattle Fireboat (located at Station 3 Fisherman's Terminal), and construction of 2 new fireboats (Department of Finance and Administrative Services 2021). New facilities were built with excess physical capacity (City of Seattle 2020).

Seattle's 2022-2027 proposed CIP includes funding for a new Station 31 in North Seattle,²⁶ replacement of the existing dock at Station 5 on the downtown waterfront, seismic assessments at five public safety facilities, and general maintenance to facilities system-wide (Seattle City Budget Office 2021). The City also anticipates it will need to replace Station 3 and the Fire Marshal office, acquire, or develop a new facility for SFD Headquarters, replace or expand the commissary and fire garage, develop a fire station in South Lake Union, and develop a freshwater marine fire suppression facility (City of Seattle 2020).

The 2022 Proposed Budget adds funding to enhance SFD operations in several areas including emergency responses, diversity recruitment, dispatch training, and IT system upgrades. In response to extensive research into community response models and on best practices gleaned from around the country, SFD will add a new specialized triage response program (Seattle City Budget Office 2021, 326).

Strategic Planning

The Department's 2012-2017 Strategic Plan is a road map for SFD and a guide for identifying priorities for emergency response services into the future. The plan identifies internal and external challenges facing the Department. Internal challenges include providing adequate leadership development and operations training and maintaining employee involvement and engagement. External challenges include financial constraints, growth of non-emergency calls, and changing demographics. The plan sets forth six goals and related strategies and action

²⁶ Station 31 in Northgate closed in June 2019 in response to air quality concerns. Units were temporarily reassigned to neighboring stations until an interim facility is established. The interim Fire Station 31 at 10503 Interlake Avenue North is planned to open in fall 2021 and will house Engine 31, Ladder 5, Aid 31, and Medic 31 (Seattle Fire Department 2021). Construction on a new permanent fire station is tentative but could be ready for general contractor bid in 2023 (Seattle Fire Department 2020). Station 31 is located north of the study area but could be called upon to provide emergency support if assistance is requested.

steps to address these challenges and to support the Department's mission. One of the goals is to maintain quality equipment, apparatus, facilities, and technology. The strategies and action steps under this step support facilities planning and coordination with the Department of Finance & Administrative Services.

Police

Data & Methods

Information about police services was collected from the Seattle Police Department, Port of Seattle Police Department, and Burlington Northern Santa Fe Railroad Police. SPD publishes calls for service, response times, and crime reports annually. Independent researches at Seattle University also collect data at the micro-community level through the annual Seattle Public Safety Survey (available via SPD's Survey Results Dashboard). SPD's 2019 Strategic Plan and the City's adopted 2021 Budget and 2021-2026 CIP were also referenced. Median response times by precinct were calculated from call data published on the City of Seattle's Open Data Portal.

The Port of Seattle Police Department publishes an annual report. BNSF Railway does not publish statistics about its police unit.

Services & Resources

Seattle Police Department

The Seattle Police Department (SPD) provides police protection services to the City of Seattle, including the study area. Its primary duties include foot, car, and bike patrols, harbor patrols, 911 calls, investigations, traffic enforcement, parking enforcement, homeland security, and specialty units such as Special Weapons and Tactics (SWAT), gang, bomb/arson, and canine units. SPD currently has 1,325 deployable sworn officers (1,433 total sworn officers) and 631 civilian employees (Seattle Police Department 2021).

The Department is divided into five precincts, each with a police station that serves as the base of operations for that precinct (see **Exhibit 3.13-7**). The BINMIC portion of the study area is in the North and West precincts while the Greater Duwamish MIC portion is primarily in the South and Southwest precincts (the area near the stadiums is in the West Precinct and a small portion near I-90 is in the East Precinct). Other industrial lands along the north side of Salmon Bay are served by the North Precinct, and industrial lands in Eastlake are served by the West Precinct. Precincts are further divided into smaller geographic areas called sectors and beats (there are three beats per sector; e.g., Ocean Sector is divided into beats O1, O2, and O3). Individual patrol officers are assigned responsibility based on beats (Seattle Police Department 2021). The location of the study area relative to police service areas is shown in **Exhibit 3.13-8**.

Precinct	Location	Primary Area Served	Sq. Ft.	Year Built
North	10049 College Way N (NE of the study area)	North of the Ship Canal to city limits	16,434	1984
West	810 Virginia St (E of the study area)	Queen Anne, Magnolia, the Downtown core, and the area west of I-5	46,231	1999
East	1519 12th Ave (E of the study area)	Eastlake and the area north of I-90 to the Ship Canal and east of I-5	61,580	1926 (updated 1985)
South	3001 S Myrtle St East (E of the study area)	South of I-90 to city limits and west of the Duwamish	13,688	1983
Southwest	2300 SW Webster St (W of the study area)	West Seattle and the Duwamish Industrial Area	28,531	2002

Exhibit 3.13-7 S	SPD Station	Locations and	Areas	Served
------------------	--------------------	---------------	-------	--------

Source: City of Seattle, 2020.



Exhibit 3.13-8 Police Precinct, Sector, and Beat Boundaries

Source: City of Seattle, 2021; BERK, 2021.

Port of Seattle Police Department

The Port of Seattle Police Department was created in 1972 and provides the primary law enforcement service to Seattle-Tacoma International Airport and the Port's seaport properties (see **Exhibit 3.13-9**). Port of Seattle Police patrol more than 30 miles of waterfront property, piers, marinas, and cargo and cruise ship terminals and are the primary first responders for all reported crimes and incidents within its jurisdiction. The Department's Waterfront Office is located in the study area at Terminal 30 (2715 East Marginal Way South, Building A-5). The POSPD has been internationally accredited by the Commission on Accreditation for Law Enforcement Agencies since 2011 (Port of Seattle Police 2020).

Exhibit 3.13-9 Port of Seattle Properties Near the Study Area, 2020



Source: (Port of Seattle Police 2020).

The POSPD currently consists of 115 commissioned police officers—including 1 chief, 2 deputy chiefs, 6 commanders, 18 sergeants, and 88 police officers—and 38 non-commissioned personnel—including 911 communications specialists who receive and coordinate all calls for service for both the Port of Seattle Fire and Police Departments and the Burlington Northern Santa Fe Railroad Police. The Department also has several specialized units, including a Marine Patrol Unit, a Dive Team, and a Commercial Vehicle Enforcement Unit to support seaport activities (Port of Seattle Police 2020).

The Transportation Security Administration, as an agency of the federal Department of Homeland Security, oversees the security efforts for all Port properties. Currently, the U.S. Coast Guard maintains responsibility for shoreline security for the Port. TSA provides support to the Coast Guard in its maritime security efforts and focuses primarily on passenger security and intermodal connectivity to ports. In partnership with the Coast Guard, TSA administers the Transportation Worker Identification Credential program, which is required for workers who need access to secure areas of the nation's maritime facilities and vessels (TSA 2016).

Burlington Northern Santa Fe Railroad Police

The Burlington Northern Santa Fe (BNSF) Railroad Police provide police services along the BNSF railway within the study area. Systemwide, BNSF Police's jurisdiction is 32,500 miles long and 100 feet wide, crisscrossing hundreds of local and state jurisdictions along the way. BNSF Police analyze statistical data to discover crime trends, use K-9 units and proactive uniformed patrol to combat trespassing and cargo thefts, and actively participate with SPD and the Port of Seattle Police Department to investigate crimes committed on railroad property (BNSF Railway 2021).

Performance

Seattle Police Department

Trends in Calls for Service and Response Times

In 2020, SPD received approximately 343,100 calls for service citywide, 100,000-130,000 calls lower than each of the previous 4 years. Total calls were likely lower in 2020 due to the COVID-19 pandemic. In 2019, SPD received 461,328 calls for service—approximately 66% of these were dispatched calls and 34% were on-view incidents (events that officers logged during routine patrols). Total calls for service increased by 5% from 2016 through 2019. **Exhibit 3.13-10** shows the total number of dispatched calls and on-views in the city during this time period. In comparison, the total calls for service in beats serving the study area increased by 17% (see **Exhibit 3.13-11**).

Year	Community Generated	Officer Generated	Total
2016	311,380 (71%)	129,496 (29%)	440,877
2017	307,904 (68%)	144,471 (32%)	452,321
2018	317,380 (69%)	142,072 (31%)	459,462
2019	306,586 (66%)	154,551 (34%)	461,328
2020	245,580 (72%)	91,364 (27%)	343,100

Exhibit 3.13-10 Seattle Police Department Citywide Calls for Service, 2016–2020

Note: Total calls is slightly higher than the sum of community generated (dispatched) and officer generated (on-view) calls as some calls are logged as "Unknown" for how they were received.

Source: Seattle Police Department Calls for Service Dashboard (<u>http://www.seattle.gov/police/information-and-data/calls-for-service-dashboard</u>), 2021.

Exhibit 3.13-11 Seattle Police Department Calls for Service by Area, 2016 and 2019

Area	Total Calls 2016	Total Calls 2019	Percent Change
Citywide	440,877	461,328	5%
Study Area Includes beats B1, B2, B3, Q1, Q2, Q3, W1, O1, O2, O3, F1, F3, and D3.	106,343	124,494	17%
Ballard In beats B1, B2, & B3	27,874	30,060	8%
Interbay Dravus Primarily in beats Q1 & Q2	14,488	15,580	8%
Interbay Smith Cove Primarily in beats Q1 & Q3	16,154	15,695	-3%
SODO/Stadium Primarily in beats W1, O1, & O2	26,726	35,283	32%
Georgetown Primarily in beats F1, F3, O2, & O3	24,685	35,349	43%
Other Industrial Lands North of Salmon Bay In beats B2 & B3	17,442	19,288	11%
Other Industrial Lands in Eastlake In beat D3	8,460	8,469	0%

Note: Study area total includes beats B1, B2, B3, Q1, W1, O1, O2, O3, F1, and F3.

Source: Seattle Police Department Calls for Service Dashboard (<u>http://www.seattle.gov/police/information-and-data/calls-for-service-dashboard</u>), 2021.

SPD tracks average response time for priority one calls by precinct and sector. **Exhibit 3.13-12** shows statistics from 2016 through 2020 for sectors serving the study area. Citywide, SPD met its seven-minute response time target all five years. The median response time citywide stayed

relatively constant from 2016-2019 (decreased by 1%) but increased by 11% from 2019 to 2020 (increased 36 seconds from 5 minutes 42 seconds to 6 minutes 18 seconds). Median response times within the six sectors serving the study area varied from year to year and from sector to sector, but all sectors saw an increase in median response time from 2019 to 2020. Sectors W and F saw the greatest increase in response time (nearly a minute for both) from 2019 to 2020. This was likely in part because of the closure of the West Seattle High-Rise Bridge (the high bridge) to all vehicle traffic on March 23, 2020. The Spokane St Swing Bridge (the low bridge) remained open to emergency vehicles, transit, and heavy freight at all times of the day but was not built to handle the same volumes of traffic as the high bridge (Seattle Department of Transportation 2021). Sector B in Ballard also saw a 51 second increase in median response time from 2019 to 2020.

Year	Citywide	Sector B North Precinct	Sector Q West Precinct	Sector K West Precinct	Sector D West Precinct	Sector W South Precinct	Sector F South Precinct	Sector O Southwest Precinct
2016	5:44	7:49	6:35	4:05	5:12	8:02	6:27	5:28
2017	5:40	7:34	6:27	4:13	5:10	8:00	6:28	5:16
2018	5:45	8:24	6:40	4:06	5:09	7:06	6:20	5:01
2019	5:42	8:45	6:30	4:09	4:59	6:59	5:38	4:44
2020	6:18	9:36	6:37	4:16	5:06	7:58	6:37	5:08

Exhibit 3.13-12 Median Response Times for Priority One Calls Citywide and in Sectors Serving the Study Area, 2016–2020

Note: The Seattle Police Department utilizes the median value of this dataset because it is less impacted by extreme values. Source: City of Seattle Open Data Portal, Call Data (<u>https://data.seattle.gov/Public-Safety/Call-Data/33kz-ixgy/data</u>), 2021; Seattle Police Department Calls for Service Dashboard (<u>http://www.seattle.gov/police/information-and-data/calls-for-service-dashboard</u>), 2021.

From 2016 to 2020, the Department has fallen short of meeting its seven-minute response time target for priority one calls in Sector B for all five years and for all but 2019 in Sector W.

Trends in calls for service and response time for sectors and beats serving each of the subareas are summarized below (see **Exhibit 3.13-8** above for the location of each subarea relative to police sectors and beats):

- Ballard: The Ballard Subarea is in Sector B in the North Precinct, and is primarily within the boundaries of beats B1 and B2. Calls for service increased by 8% in Sector B from 2016 to 2019 and the median response time increased by 23% from 2016 to 2020, with a 51 second increase in median response time from 2019 to 2020. The Department fell short of meeting its seven-minute response time target in Sector B from 2016 to 2020.
- Interbay Dravus: The Interbay Dravus Subarea is in Sector Q in the West Precinct, and is primarily within the boundaries of beats Q1 and Q2. Calls for service increased by 8% in these two beats from 2016 to 2019 with the greatest increases in Beat Q2 (15%). The median response time in Sector Q stayed nearly constant from 2016 to 2020 (increased by

1%). The Department met its seven-minute response time target in Sector Q from 2016 to 2020.

- Interbay Smith Cove: The Interbay Smith Cove Subarea is in Sector Q in the West Precinct and is primarily within the boundaries of beats Q1 and Q3. Calls for service decreased by 3% in these two beats from 2016 to 2019. The median response time in Sector Q stayed nearly constant from 2016 to 2020 (increased by 1%). The Department met its seven-minute response time target in Sector Q from 2016 to 2020.
- SODO/Stadium: The SODO/Stadium Subarea is in Sector O of the South Precinct and Sector W of the Southwest Precinct, and is primarily within the boundaries of beats W1, O1, and O2 (a small portion is also in sectors K and F). Calls for service increased by 32% in these three beats from 2016 to 2019 with the greatest increases in Beat O2 (66%). The average response time decreased by 6% in Sector O and by 1% in Sector W from 2016 to 2020. However, as noted above, median response time in Sector W increased by 59 seconds from 2019 to 2020, likely in part because of the closure of the West Seattle High-Rise Bridge to all vehicle traffic on March 23, 2020. The Department met its seven-minute response time target in Sector O and fell short of meeting its target in Sector W from 2016 to 2018 and in 2020 (the Department met its target in Sector W by 1 second).
- Georgetown/South Park: The Georgetown/South Park Subarea is in Sector O of the South Precinct and Sector F of the Southwest Precinct, and is primarily within the boundaries of beats F1, F3, O2, and O3 (a small portion is also in Sector S). Calls for service increased by 43% in these four beats from 2016 to 2019 with the greatest increases in beats O2 (66%) and O3 (62%). The average response time decreased by 6% in Sector O and increased by 3% in Sector F from 2016 to 2020. Median response time in Sector F increased by 59 seconds from 2019 to 2020, likely in part because of the closure of the West Seattle High-Rise Bridge to all vehicle traffic on March 23, 2020. The Department met its seven-minute response time target in sectors O and F from 2016 to 2020.
- Other Industrial Zoned Lands: Other industrial lands along the north side of Salmon Bay are within the boundaries of beats B2 and B3. Calls for service increased by 11% in these beats from 2016 to 2019 and the average response time increased in Sector B by 23% from 2016 to 2020, with a 51 second increase in median response time from 2019 to 2020. The Department fell short of meeting its seven-minute response time target in Sector B from 2016 to 2020.

Other industrial lands in Eastlake are within the boundaries of Beat D3. Calls for service in this beat did not change from 2016 to 2019 but the average response time decreased in Sector D by 2% from 2016 to 2020. The Department met its seven-minute response time target in Sector D for from 2016 to 2020.

MCPP Priorities

The Seattle Public Safety Survey collects data at the micro-community level about perceptions of crime and public safety, police-community interactions, and knowledge and understanding of the MCPPs. The top five citywide public safety concerns identified in the 2020 survey (in order) were police capacity, property crime, homelessness, drugs and alcohol, and community and public safety capacity. The top five public safety concerns in each micro-community serving the study area are listed in **Exhibit 3.13-13**—police capacity, property crime, and homelessness were among the top three for all but the South Beacon Hill MCPP.

МСРР	1st	2 nd	3 rd	4 th	5th
Ballard South	Homelessness	Property Crime	Police Capacity	Drugs & Alcohol	Community & Public Safety Capacity
Chinatown/ Int'l District	Homelessness	Police Capacity	Property Crime	Drugs & Alcohol	Violent Crime
Commercial Duwamish	Police Capacity	Homelessness	Property Crime	Drugs & Alcohol	Traffic Safety
Commercial Harbor Island	Property Crime	Homelessness	Police Capacity	Traffic Safety	Drugs & Alcohol
Eastlake—West	Property Crime	Police Capacity	Homelessness	Community & Public Safety Capacity	Public Order Crime
Fremont	Police Capacity	Homelessness	Property Crime	Traffic Safety	Community & Public Safety Capacity
Georgetown	Homelessness	Property Crime	Police Capacity	Drugs & Alcohol	Community & Public Safety Capacity
Magnolia	Police Capacity	Property Crime	Homelessness	Drugs & Alcohol	Community & Public Safety Capacity
Pioneer Square	Homelessness	Police Capacity	Property Crime	Drugs & Alcohol	Violent Crime
Queen Anne	Property Crime	Police Capacity	Homelessness	Traffic Safety	Community & Public Safety Capacity
SLU/Cascade	Homelessness	Police Capacity	Property Crime	Drugs & Alcohol	Community & Public Safety Capacity
SODO	Homelessness	Property Crime	Police Capacity	Drugs & Alcohol	Public Order Crime
South Beacon Hill	Police Capacity	Property Crime	Community & Public Safety Capacity	Traffic Safety	Violent Crime
South Park	Property Crime	Police Capacity	Homelessness	Traffic Safety	Drugs & Alcohol
Wallingford	Homelessness	Property Crime	Police Capacity	Traffic Safety	Community & Public Safety Capacity

Exhibit 3.13-13 Top 5 Safety Concerns by MCPP in the Study Area in Ranked Order, 2020

Source: Seattle Police Department Service Results Dashboard (<u>https://www.seattle.gov/police/information-and-data/mcpp-about/survey-results-dashboard</u>), 2021.

MCPP priorities for each subarea are summarized below:

- Ballard: The Ballard Subarea includes the Ballard South and Fremont MCPPs. The top five public safety concerns in these MCPPs as identified in the 2020 Seattle Public Safety Survey included homelessness, property crime, police capacity, and community and public safety capacity. Respondents in Ballard South also included drugs and alcohol among their top five concerns while those in Fremont included traffic safety.
- Interbay Dravus: The Interbay Dravus Subarea includes the Magnolia and Queen Anne MCPPs. The top five public safety concerns in these MCPPs as identified in the 2020 Seattle Public Safety Survey included police capacity, property crime, homelessness, and community and public safety capacity. Respondents in Magnolia also included drugs and alcohol among their top five concerns while those in Queen Anne included traffic safety.
- Interbay Smith Cove: The Interbay Smith Cove Subarea includes the Magnolia and Queen Anne MCPPs. The top five public safety concerns in these MCPPs as identified in the 2020 Seattle Public Safety Survey included police capacity, property crime, homelessness, and community and public safety capacity. Respondents in Magnolia also included drugs and alcohol among their top five concerns while those in Queen Anne included traffic safety.
- **SODO/Stadium:** The SODO/Stadium Subarea includes the following MCPPs by precinct:
 - West Precinct: Pioneer Square and Chinatown/International District
 - South Precinct: SODO and Georgetown
 - Southwest Precinct: Commercial Duwamish and Commercial Harbor Island.

The top five public safety concerns in these MCPPs as identified in the 2020 Seattle Public Safety Survey included homelessness, police capacity, property crime, and drugs and alcohol. Other top five concerns varied by MCPP: respondents in the West Precinct included violent crime, respondents in the Southwest Precinct included traffic safety, respondents in Georgetown included community and public safety capacity, and respondents in SODO included public order crime among their top five concerns.

- Georgetown/South Park: The Georgetown/South Park Subarea includes the Georgetown and South Beacon Hill MCPPs in the South Precinct and the Commercial Duwamish and South Park MCPPs in the Southwest Precinct. The top five public safety concerns in these MCPPs as identified in the 2020 Seattle Public Safety Survey included homelessness, property crime, police capacity, and community and public safety capacity. Drugs and alcohol were among the top five concerns in Ballard South while traffic safety was among the top five in Fremont.
- Other Industrial Zoned Lands: Other industrial lands along the north side of Salmon Bay are within the Fremont and Wallingford MCPPs, and other industrial lands in Eastlake are within the Eastlake—West and SLU/Cascade MCPPs. The top five public safety concerns in these MCPPs as identified in the 2020 Seattle Public Safety Survey included homelessness, property crime, police capacity, and community and public safety capacity. Respondents in the Fremont and Wallingford MCPPs also included traffic safety among their top five concerns while those in the Eastlake—West MCPP include public order crime and those in the SLU/Cascade MCPP included drugs and alcohol.

Port of Seattle Police Department

In 2020, the Port of Seattle Police Department's patrol team responded to 106,463 calls for service jurisdiction wide (airport and seaport properties), including 55,000 self-initiated contacts (area checks, subject contacts, traffic stops, and checkpoint alarm checks). This was about 15% more calls than in 2019 (106,463 vs. 92,186; see **Exhibit 3.13-14**).

Exhibit 3.13-14 Port of Seattle Police Department Patrol Team Calls for Service, 2019–2020

Year	Calls for Service	Self-initiated
2019	92,186	61,168
2020	106,463	55,000

Source: Port of Seattle Police Department Annual Report 2020.

Schools & Libraries

Data & Methods

The information about schools and libraries was collected from:

- Seattle Public Schools
- Seattle Public Libraries
- King County Assessor Parcel Records
- Seattle Comprehensive Plan
- Seattle Land Use Code

Services & Resources

The Seattle School District serves the city as a whole. It operates 106 schools and employs about 7,574 staff including about 6,173 educators that are school-based. There are about 25,528 Elementary, 12,025 Middle, and 14,828 high school students. The students are 46% white and 54% persons of color.²⁷

The Seattle School District Administrative offices are in the SODO/Stadium Subarea. See **Exhibit 3.13-15**. There are no public schools in the study area. There is one private school in Ballard. In the Secondary Study Area there are schools in proximity to industrial zones identified in relation to the nearest subareas. For the few residences in the study area, they would attend a variety of schools based on the service areas in **Exhibit 3.13-16**. Schools are allowed in existing buildings in industrial zones except in the Greater Duwamish MIC.

²⁷ Seattle Public Schools. 2020-21 Fast Facts & Figures Seattle Public Schools.

https://www.seattleschools.org/UserFiles/Servers/Server_543/File/District/Departments/Communications/seattle-public-schoolsquick_facts.pdf.



Exhibit 3.13-15 Schools and Libraries in or Near the Study Area

Source: King County GIS, 2021; CAI, 2021; BERK, 2021.



Exhibit 3.13-16 School Attendance Boundaries: Elementary and Middle Schools

Source: Seattle School District, 2020.

The Seattle Public Library system offers 27 locations. As of 2020, they provide access to 1.7 million print materials, 677,000 pieces of media, as well as 4.3 million e-books and 2.1 million streaming and downloadable medial. The system also offers 1,100 virtual classes, events, and activities including classes and to learn skills, find job resources and make social connections.²⁸ There are no libraries in the Primary Study Area, and several nearby in the Secondary Study area, described with the nearest subarea below. See **Exhibit 3.13-15**. The Seattle Industrial zones prohibit libraries.

Schools and libraries serving each subarea are summarized below:

- Ballard: The Ballard Subarea is served by BF Day Elementary and Adams Elementary Schools and Hamilton and Whitman Middle Schools based on service areas. There is one private school known as Modern Pilot, offering simulation-based flight training and curriculum, and located on Russel Avenue NW on property zoned IC-65 (M). It is operated in an industrial building on a 5,000 square foot property. The Assessor considers the property to be in an industrial use.
- Interbay Dravus: There are no mapped public or private schools in the Interbay Dravus Subarea. To the west is a public school, Lawton Elementary School, which is separated from the study area by topography and a strip of commercial and residential zones. The subarea is served by Lawton, Code, and Magnolia Elementary Schools and McClure Middle School. There are no libraries in the subarea.
- Interbay Smith Cove: There are no public or private schools or libraries in the subarea. The subarea is served by Magnolia, Code, and Hay Elementary Schools and McClure Middle School.
- SODO/Stadium: The John Stanford Center for Education Excellence and Seattle School District Administrative offices are in the SODO Stadium district on Lander Street on land zoned IG1 U/85. The building lies on about 6.9 acres and contains a 325,000 gross square foot building with two-thirds in office space and one third in storage/warehouse space. The district also owns a 4.3-acre parking lot to the north of the offices. There are no public or private schools or libraries in the subarea. The study area is served by Wing Luke Elementary and Mercer Middle School. The Puget Sound Community School, a private institution, lies on Dearborn Street in the International district and serves students between 11 and 18 years old (6-12 grades). North of the subarea lies the International District / Chinatown Library on Eighth Avenue S.
- Georgetown/South Park: There are no schools or libraries in the Georgetown/South Park Subarea. The MIC surrounds the Georgetown Urban Center/Village which contains the historic Concord International school and the South Park Library. The Georgetown/South Park Subarea is served by Concord International, Sanislo, and Wing Luke Elementary Schools and Mercer and Denny Middle Schools.
- **Other Industrial Zoned Lands:** In the Eastlake area abutting the IG1 U/45 zone on E Galer Street is a private school called the Fusion Academy offering one on one teacher/student

²⁸ The Seattle Public Library. 2021. 2020 Statistical and Financial Summaries. <u>https://www.spl.org/about-us/library-impact/2020-impact-report/2020-statistics</u>.

ratios for middle and high school students. The school is in an office building on a property zoned C1-75. In Eastlake, the industrial area is served by Montlake and Lowell Elementary Schools, and Meany Middle School. The scattered industrial areas along Salmon Bay and north Lake Union are served by BF Day, John Stanford International, and Laurelhurst Elementary Schools and Hamilton and Eckstein Middle Schools.

3.13.2 Impacts

Thresholds of significance utilized in this impact analysis include:

- Negatively affect the response times for police and/or fire and emergency medical services.
- Increase demand for special emergency services beyond current operational capabilities of service providers.
- Result in increases in students and lack of facilities unanticipated in district plans or that would reduce adopted levels of service.

Impacts Common to All Alternatives

Fire & Emergency Medical Services

Population Growth

Growth in worker and residential populations in the study area is expected to lead to an increased number of calls for emergency services. Growth is expected to occur incrementally under all alternatives, as individual development projects are constructed. The Seattle Fire Department would attempt to maintain response times consistent with or better than current performance levels as the population grows. Over time, additional staffing and equipment may be required in order to maintain performance levels.

As described under the Affected Environment, fire stations serving the study area were recently upgraded or replaced as part of the Fire Facilities and Emergency Response Levy and are not anticipated to need renovations in the near future. In addition, the Chief Seattle Fireboat at Station 3 Fisherman's Terminal was renovated as part of the levy and Station 5 (serving the downtown waterfront) is currently under construction concurrent with portions of the Seattle Waterfront project.

Any potential future fire facility, staffing, or equipment needs could be included as part of the City's annual Budget and Capital Improvement Program process.

Building Heights & Density

Existing ladder trucks at Stations 8, 10, 14, 17, and 18 and at other stations near the study area are equipped to provide services to buildings of the heights proposed under all alternatives.

Additionally, new buildings would be required to meet the Seattle Fire Code which requires sprinklers throughout. The City also applies standards for live/work units (like artists' lofts and caretakers' units) to ensure there are exits from sleeping rooms and fire-rated walls and doors between different uses. No impacts to fire services are anticipated due to increases in building height or density.

Hazardous Materials

Industrial uses often include hazardous materials or have the potential to produce hazardous waste. Hazardous materials are defined by the City of Seattle as "those that pose an unreasonable risk to the health and safety of operating or emergency personnel, the public, and the environment if not properly controlled during handling, storage, manufacture, processing, packaging, use, disposal, or transportation" (City of Seattle 2018).

Additional industrial development under all of the alternatives could increase the amount or prevalence of hazardous materials in the study area. All new development would be required to meet the Seattle Fire Code which includes provisions for hazardous materials (<u>Part V, Chapter 50-67</u>). Development proposals would be reviewed by the Seattle Department of Construction & Inspections as well as the SFD. Additional federal and state regulations also apply to development that includes hazardous materials or wastes—for example, the U.S. Environmental Protection Agency regulates hazardous waste in part 262 of title 40 of the Code of Federal Regulations, WSDOT regulates off-site transportation of hazardous materials, and the Washington State Department of Ecology requires additional permits and inspections for such facilities as underground storage tanks. No impacts to fire or EMS services are anticipated due to an increased amount of hazardous materials.

Construction

The Seattle Fire Department makes service calls related to inspection of construction projects and calls to respond to construction-related accidents. As such, increased construction activities associated with potential development under all alternatives could result in an increase in demand for fire services. Existing Fire Department staffing and equipment are anticipated to be sufficient to handle increased services needed for construction activities.

Transportation Network & Traffic Volumes

Use of the public right of ways is critical to SFD meeting their response goals as the Department is dependent upon the capability of the city's street network to handle traffic flows. No specific transportation projects or changes to emergency access routes are proposed under any of the alternatives, but changes to the street network over time has the potential to impact the mobility of fire response vehicles. Any street improvements must be consistent with the Seattle Fire Code Section 503 and Appendix D, which address fire apparatus access roads. Additionally, SFD reviews proposed street improvements on a project-by-project basis to identify potential negative impacts on response times. It is anticipated that these mitigation measures would

adequately address the potential impacts of future changes to the transportation network under any of the alternatives.

Traffic volumes are anticipated to increase under all of the alternatives. Travel times in the study area are expected to remain relatively consistent between 2019 and 2044 (see **Section 3.10 Transportation** and the impacts discussion under each alternative below). Regular planning by SFD is anticipated to address any needed changes to emergency access routes or any future facility, staffing, or equipment needs as a result of increased traffic volumes.

Ballard, Interbay Dravus, & Interbay Smith Cove

The Ballard Link Extension would construct three stations within the BINMIC: Ballard, Interbay (in the vicinity of Dravus Street), and Smith Cove. Transit capacity along the north-south corridor will dramatically increase compared to existing conditions making non-auto modes increasingly competitive.

SODO/Stadium & Georgetown/South Park

Terminal 5, the international marine cargo terminal operated by the Northwest Seaport Alliance (a partnership of the Ports of Seattle and Tacoma), is scheduled to open in early 2022. This opening will significantly increase the number of trucks that must use the West Seattle low bridge to reach the terminal. The opening of Terminal 5 and associated increase in truck traffic could negatively impact response times for emergency vehicles trying to access West Seattle. The City of Seattle is working closely with the Port of Seattle and Northwest Seaport Alliance to plan for more trucks on the low bridge and monitor the increase in workers traveling to the terminals for their shifts (Seattle Department of Transportation 2021, The Northwest Seaport Alliance 2021).

Police

Population Growth

Population growth in the study area may not necessarily result in increased crime and demand for police services. For example, total calls for service decreased by 3% in Beat Q3 from 2016 through 2019, while the population in the study area increased (PSRC 2020). While population growth and increases in urbanization can impact crime, many other factors are part of the equation including population characteristics, economic conditions, transportation conditions, climate, prevalent attitudes towards crime and crime reporting practices in the local population, and police department characteristics (Federal Bureau of Investigation 2013).

Since population and employment growth do not directly correlate to an increased demand for police services, none of the four growth alternatives would necessarily result in proportional increases in call volumes or incidence of major crimes. Therefore, no specific findings of adverse effects on response times or criminal investigations volumes are made. SPD will continue to analyze where best to focus its resources to respond to changes in demand for

police services regardless of which alternative is selected. Better site and building design such as with building placement, lighting, and visibility can reduce the potential for crime.

Building Heights & Density

No impacts to police services are anticipated due to increased building heights. Of the seven sectors serving the study area, Sector K consistently reported the fastest median response time for priority one calls from 2016-2020, ranging from 4 minutes 5 seconds to 4 minutes 16 seconds (see **Exhibit 3.13-11**). Sector K serves a portion of Downtown where there are many tall buildings. Conversely, other sectors serving the study area (such as Sectors B serving Ballard and Sector W) consistently reported the slowest median response time for priority one calls over the same time period.

Relative changes in population density by beat and sector may generate more workload in some areas of the city but are not anticipated to impact police service or response times under any of the alternatives. The Department's deployment model is adjusted for changes in workload. Increased city tax revenue generated by new businesses or households could help defray costs of increased police workload.

Construction

The Seattle Police Department responds to construction-related service calls such as construction site theft and vandalism. Potential construction activities under all the alternatives could result in an increase in demand for police services. Existing Departmental resources are anticipated to be sufficient to handle such an increase.

Transportation Network & Traffic Volumes

Future traffic volumes or changes to the transportation network in the study area could impact first responders' ability to respond rapidly to emergency calls. SPD's staffing model factors in response time to determine appropriate staffing levels in each precinct. The Department would likely adjust staffing levels to improve response times if future increased traffic volumes or changes to the street network negatively impact police services.

SODO/Stadium & Georgetown/South Park

As discussed under Fire & Emergency Medical Services, the opening of Terminal 5 in early 2022 and associated increase in truck traffic could negatively impact response times for emergency vehicles trying to access West Seattle. The City of Seattle is working closely with the Port of Seattle and Northwest Seaport Alliance to plan for more trucks on the low bridge and monitor the increase in workers traveling to the terminals for their shifts (Seattle Department of Transportation 2021).

Schools & Libraries

The demand for schools and libraries will be in proportion to the increase in housing under each alternative, which shows less growth in alternatives 1 and 2 and more under alternatives 3 and 4 and the Preferred Alternative. See **Exhibit 3.13-17**.



Exhibit 3.13-17 Total Housing in Study Area by Alternative

Source: City of Seattle, 202<u>2</u>4; BERK, 202<u>2</u>4.

Students are anticipated to be a similar share of the future population as today. Based on the State Office of Financial Management (OFM) population, and the Office of the Superintendent of Public Instruction (OSPI), the student enrollment for fall 2020 is about 7.1% of the total population. See **Exhibit 3.13-18**. In fall 2021, the School District saw a dip in enrollment; thus the 2020 rate is considered conservative and is retained for purposes of the EIS.

Exhibit 3.13-18 Student Generation Rate

	Number
Seattle School District Population (OFM 2020)	761,932
Enrollment OSPI 2020-2021	53,997
% of Pop	7.1%

Source: OFM, 2021; OSPI, 2021; BERK, 2021.

Based on the net change in dwellings and population, and assuming 7.1% of the population are students, the number of potential students is shown in **Exhibit 3.13-19**. <u>The m</u>Aost housing units and associated population are anticipated under Alternative 4 and the least under Alternative 1. <u>The New</u> students would have more effect on schools in Ballard, SODO/Stadium, and Georgetown/South Park.

Subarea	Alt. 1	Alt. 2	Alt. 3	Alt. 4	<u>Pref. Alt.</u>
Ballard	1	1	38	115	<u>75</u>
Interbay Dravus	1	1	11	25	<u>17</u>
Interbay Smith Cove	1	1	2	—	=
SODO/Stadium	4	5	29	144	<u>94</u>
Georgetown/South Park	3	3	9	35	<u>30</u>
Total: Ind Zone Housing (Caretaker/Artist)	11	12	89	319	<u>214</u>
With MIC Adjustments—Seattle Mixed-Use Zone Housing	_	_	157	157	<u>223</u>
Grand Total Students in Study Area	11	12	245	476	<u>437</u>

Exhibit 3.13-19 Student Generation by Subarea based on Net Change in Population

Source: BERK, 202<u>2</u>4.

Equity & Environmental Justice Considerations

The City of Seattle developed a Racial and Social Equity Index that combines data on race, ethnicity, and socioeconomic and health disadvantages to identify neighborhoods with large proportions of priority populations as residents. Much of the SODO/Stadium Subarea as well as the South Park neighborhood were found to have among the highest disadvantages in the city.

The Action Alternatives—especially alternatives 3 and 4—would result in more land use growth compared to Alternative 1 No Action particularly in the SODO/Stadium and South Park neighborhoods. Additional growth would increase traffic volumes which may in turn increase the response time of emergency vehicles in areas with high proportions of priority populations. However, increased development in areas with histories of long-term underinvestment could bring improved infrastructure to those neighborhoods. Development standards in areas rezoned as Industry & Innovation and Urban Industrial would require frontage improvements such as sidewalks, pedestrian lighting, and street trees that would likely result in safer, more connected, and more accessible neighborhoods.

The increase in housing in areas rezoned Seattle Mixed under alternatives 3 and 4<u>and under</u> <u>the Preferred Alternative</u> is anticipated to generate students attending local schools in the Georgetown/South Park Subarea which has a higher proportion of disadvantaged households. The caretakers' quarters and makers' studios may also house families with students though less likely. Ensuring access to schools with safe travel routes would help all local students in these areas.

Impacts of Alternative 1 No Action

Alternative 1 No Action is expected to result in roughly 23,500 additional jobs in the study area compared to existing conditions. Residential development would be very minor—approximately 75 new dwellings over the study area. For both employment and residential uses, growth is expected to be highest in the SODO/Stadium and Georgetown/South Park subareas.

Fire & Emergency Medical Services & Police Services

No impacts other than those described under Impacts Common to All Alternatives are anticipated under Alternative 1 No Action. Regular planning by SFD and SPD are anticipated to address incremental increased demand for fire, emergency medical, and police services.

Traffic volume growth rates within the study area are expected to be relatively low under Alternative 1 No Action given that many facilities already operate with congestion during peak periods and new high-capacity transit options would be available, making non-auto modes increasingly competitive. Travel times in the study area are expected to remain relatively consistent between 2019 and 2044 (see **Section 3.10 Transportation**)

Any potential future facility, staffing, or equipment needs as a result of increased demand for services, traffic volumes, or changes to the transportation network could be included as part of the City's annual Budget and Capital Improvement Program process.

Schools & Libraries

Population growth is anticipated to be the lowest under Alternative 1 at 154, and would have low demand for school and library services.

Two thirds of the small population growth would be in the SODO/Stadium and Georgetown/South Park subareas. The population would generate about 11 students. See **Exhibit 3.13-19**.

There could be a small increase in demand at the Concord International school and the South Park Library. Other schools with minimal changes in students could be Sanislo and Wing Luke Elementary Schools and Mercer and Denny Middle Schools.

Impacts of Alternative 2

Alternative 2 would result in 10,900 <u>more</u> jobs more than Alternative 1 No Action and residential growth would remain essentially flat (80 new housing units versus 75 under Alternative 1). As with Alternative 1 No Action, most of the new growth would be concentrated in the Greater Duwamish MIC.

Fire & Emergency Medical Services & Police Services

Alternative 2 applies a mix of Industry & Innovation and Urban Industrial Zone concepts in 10% of the current MIC areas, including an estimated ¼ mile from future light rail stations. These zones introduce nodes of high-density employment and multi-modal access near transit and create thoughtful integration between the edges of Seattle's MICs and adjacent neighborhoods. Compact growth in these areas in proximity to SFD and SPD services could result in more efficient service delivery and greater ability to meet LOS objectives than under Alternative 1 No Action.

Traffic volumes under Alternative 2 would be slightly higher than Alternative 1 No Action but the magnitude of change would be relatively small in relation to the amount of background traffic in the city. Travel times in the study area are expected to remain relatively consistent on most corridors between 2019 and 2044, with travel time increases of up to 4% over Alternative 1 No Action. One corridor—eastbound W Dravus Street between 15th Avenue W and 20th Avenue W—would also fall from LOS E under Alternative 1 No Action to LOS F under Alternative 2 (see **Section 3.10 Transportation**).

No other impacts aside from those described under Impacts Common to All Alternatives are anticipated under Alternative 2. Regular planning by SFD and SPD are anticipated to address incremental increased demand for fire, emergency medical, and police services. Any potential future facility, staffing, or equipment needs as a result of increased demand for services, traffic volumes, or changes to the transportation network could be included as part of the City's annual Budget and Capital Improvement Program process.

Schools & Libraries

Impacts are very similar to Alternative 1 No Action. There are only 5 more dwellings than Alternative 1 (about 80 total new) and 10 more people (about 164 total new population). Student generation is about 12 instead of 11. See **Exhibit 3.13-19**. Similar small demand could occur with schools and the library serving the Georgetown/South Park Subarea.

Impacts of Alternative 3

Alternative 3 would result in 33,900 <u>more jobs</u> more than Alternative 1 No Action. As with Alternative 1 No Action, most of the new employment growth would be concentrated in the Greater Duwamish MIC.

Alternative 3 also includes additional allowance for housing in the Urban Industrial Zone and new housing in focused areas removed from the MIC and placed in a mixed-use zone in Georgetown and South Park. Most of the additional 610 industry-supportive housing in industrial zones (535 more than Alternative 1 No Action) would be in the Ballard and SODO/Stadium subareas. An additional 784 dwelling units in mixed-use developments are estimated for the triangular area of Georgetown bounded by Airport Way, Corson Avenue S, and Carleton Avenue S, and 294 dwelling units are estimated for the two small areas of South Park that would be removed from the MIC near the Duwamish River. This would result in a total of 1,048 housing units over the study time horizon on land that is removed from industrial zoning under Alternative 3.

Fire & Emergency Medical Services

Alternative 3 applies a mix of Industry & Innovation, Urban Industrial, and Mixed-Use Commercial Zone concepts in 14% of the current MIC areas, covering more land area than under Alternative 2 and including an estimated ½ mile from future light rail stations. Similar to Alternative 2, these zones introduce nodes of high-density employment and multi-modal access near transit and create thoughtful integration between the edges of Seattle's MICs and adjacent neighborhoods. However, more industry-supportive housing would be allowed in the Urban Industrial Zone under Alternative 3 than Alternative 2; most of this housing would be in the Ballard and SODO/Stadium subareas. In addition, areas of land would be removed from the MICs in the Georgetown and South Park neighborhoods and placed in a mixed-use zone under Alternative 3. Compact growth in these areas—both inside and outside the MICs—in proximity to SFD and SPD services could result in more efficient service delivery and greater ability to meet LOS objectives under Alternative 3 than under Alternative 1 No Action or Alternative 2. New buildings would be required to meet the Seattle Fire Code, including standards for live/work units (like makers' studios and caretakers' units) to ensure there are exits from sleeping rooms and fire-rated walls and doors between different uses.

Traffic volumes under Alternative 3 would be higher than Alternative 1 No Action and Alternative 2—the PM peak vehicle miles traveled within the Greater Duwamish MIC would increase over Alternative 1 by roughly 2.3% and the PM peak VMT within the BINMIC would increase by roughly 4.3%. Travel times in the study area are expected to remain relatively consistent on most corridors between 2019 and 2044, with travel time increases of up to 1.5 minutes over Alternative 1. Two corridors—northbound 15th Avenue W from Magnolia Bridge to NW Leary Way and eastbound W Dravus Street between 15th Avenue W and 20th Avenue W—would also fall from LOS E under Alternative 1 No Action to LOS F under Alternative 3, southbound SR 509 between SR 99 and SR 518 would fall below WSDOT's LOS standard for SR 509 fall from LOS D to E, and southbound I-5 from Madison Street to SR 599 is expected to experience a 6% increase in travel time compared to Alternative 1 (see Section 3.10 Transportation).

No other impacts aside from those described under Impacts Common to All Alternatives are anticipated under Alternative 3. Regular planning by SFD and SPD are anticipated to address incremental increased demand for fire, emergency medical, and police services. Any potential future facility, staffing, or equipment needs as a result of increased demand for services, traffic volumes, or changes to the transportation network could be included as part of the City's annual Budget and Capital Improvement Program process.

Schools & Libraries

The increase in caretakers' quarters/makers' studios of 610 dwellings would primarily be in the Ballard and SODO/Stadium subareas, generating most of the potential 89 students. This could increase demand for schools, particularly BF Day, Adams, Beacon Hill, and Wing Luke.

In addition, about 1,078 dwellings are planned in the Georgetown/South Park Subarea generating about 2,210 people and 157 students. This could affect demand at the South Park Library, and particularly schools like Wing Luke (capacity 351) and Concord (capacity 333) schools. This number of students would be about 45% of an elementary school capacity. However, the plan is a 20-year plan and it is likely that not all housing would be developed at one time, and students would not start all at once and would be spread across grades.

Impacts of Alternative 4

Alternative 4 would result in 35,700 <u>more j</u>obs more than Alternative 1 No Action. As with Alternative 1 No Action, most of the new employment growth would be concentrated in the Greater Duwamish MIC.

Alternative 4 also includes the greatest allowance for housing in the Urban Industrial Zone and new housing in focused areas removed from the MIC and placed in a mixed-use zone in Georgetown and South Park. Most of the additional 2,195 industry-supportive housing in industrial zones (2,120 more than Alternative 1 No Action) would be in the Ballard and SODO/Stadium subareas. New housing in the focused areas in Georgetown and South Park that are removed from industrial zoning is the same as under Alternative 3 (1,048 housing units over the study time horizon).

Fire & Emergency Medical Services

Under Alternative 4, the potential for more efficient service delivery and greater ability of SFD and SPD to meet LOS objectives is similar to that described under Alternative 3. Alternative 4 applies a mix of Industry & Innovation, Urban Industrial, and Mixed-Use Commercial Zone concepts in 13% of the current MIC areas, including an estimated ½ mile from future light rail stations. The same areas of land would be removed from the MICs in the Georgetown and South Park neighborhoods and placed in a mixed-use zone under Alternative 4 as under Alternative 3. However, Alternative 4 includes the most industry-supportive housing in the Urban Industrial Zone of the Action Alternatives; most of this housing would be in the Ballard and SODO/Stadium subareas. New buildings would be required to meet the Seattle Fire Code, including standards for live/work units (like makers' studios and caretakers' units) to ensure there are exits from sleeping rooms and fire-rated walls and doors between different uses.

Traffic volumes under Alternative 4 would be slightly higher than Alternative 3. Associated impacts on travel times and corridor LOS are similar to those described above for Alternative 3 (see **Section 3.10 Transportation**).

No other impacts aside from those described under Impacts Common to All Alternatives are anticipated under Alternative 4. Regular planning by SFD and SPD are anticipated to address incremental increased demand for fire, emergency medical, and police services. Any potential future facility, staffing, or equipment needs as a result of increased demand for services, traffic volumes, or changes to the transportation network could be included as part of the City's annual Budget and Capital Improvement Program process.

Schools & Libraries

Impacts under Alternative 4 are similar to Alternative 3 except that there would be more caretakers' quarters/makers' studios at up to 2,195, with most in the SODO/Stadium and Ballard subareas. Like Alternative 3, there would be 1,078 dwellings in the Georgetown/South Park Subarea.

All together there would be an increase in population of 6,710 including 476 students. Local libraries in Ballard and South Park would likely see an increase in demand for services. Schools serving Ballard, SODO/Stadium, and Georgetown/South Park could have increased demand at 33-45% of a typical elementary school capacity (~350).

Impacts of the Preferred Alternative

The Preferred Alternative would result in 12,045 more jobs than Alternative 1 No Action, slightly more than under Alternative 2 but less than alternatives 3 and 4. Most of the new employment growth would be concentrated in the Greater Duwamish MIC within the SODO/Stadium and Georgetown/South Park subareas. Slightly more jobs would be located in the Ballard, Interbay Dravus, and Interbay Smith Cove subareas relative to other alternatives.

Total residential growth under the Preferred Alternatives is similar to but lower than Alternative 4. Most of the additional 1,475 industry-supportive housing in industrial zones (1,400 more than Alternative 1 No Action) would be in the Ballard and SODO/Stadium subareas. New housing in the focused areas in Georgetown and South Park that are removed from industrial zoning is the same as under alternatives 3 and 4. Two additional areas outside the MICs in west Ballard and Judkins Park would also be converted to mixed use zoning allowing housing under the Preferred Alternative. This would result in a total of 1,534 housing units over the study time horizon on land that is removed from industrial zoning under the Preferred Alternative (versus 1,078 under alternatives 3 and 4).

The collective change in population—including within industrial areas, areas removed from the MIC, and rezoned areas converted to mixed use zoning outside of the MIC—would equal 3,009 households, about 8% less than Alternative 4. Combined employment and residential growth under the Preferred Alternative is lower than both alternatives 3 and 4.

Fire & Emergency Medical Services & Police Services

<u>Under the Preferred Alternative, the potential for more efficient service delivery and greater ability</u> of SFD and SPD to meet LOS objectives is similar to that described under alternatives 3 and 4.

The Preferred Alternative applies a mix of Industry & Innovation, Urban Industrial, and Mixed-Use Commercial Zone concepts in 14% of the current MIC areas, covering more land area than any of the other Action Alternatives and including an estimated ½ mile from future light rail stations. Similar to the other Action Alternatives, these zones introduce nodes of high-density employment and multi-modal access near transit and create thoughtful integration between the edges of Seattle's MICs and adjacent neighborhoods.

The same areas of land would be removed from the MICs in the Georgetown and South Park neighborhoods and placed in a mixed-use zone under the Preferred Alternative as under alternatives 3 and 4. Two new areas outside the MICs in west Ballard and Judkins Park would also be converted to mixed use zoning allowing housing under the Preferred Alternative (in addition to the proposed mixed use areas in the Georgetown and South Park neighborhoods under alternatives 3 and 4). Compact growth in these areas—both inside and outside the MICs—in proximity to SFD and SPD services could result in more efficient service delivery and greater ability to meet LOS objectives under the Preferred Alternative than under the other alternatives.

Traffic volumes under the Preferred Alternative would be slightly higher than Alternative 2 but lower than alternatives 3 and 4—the PM peak vehicle miles traveled within the Greater Duwamish MIC would increase over Alternative 1 by roughly 1.0% and the PM peak VMT within the BINMIC would increase by roughly 2.7%. Travel times in the study area are expected to remain relatively consistent on most corridors between 2019 and 2044, with travel time increases of up to 1 minute over Alternative 1. Two corridors—northbound 15th Avenue W from Magnolia Bridge to NW Leary Way and eastbound W Dravus Street between 15th Avenue W and 20th Avenue W—would fall from LOS E under Alternative 1 No Action to LOS F under the Preferred Alternative (see **Section 3.10 Transportation**).

No other impacts aside from those described under Impacts Common to All Alternatives are anticipated under the Preferred Alternative. Regular planning by SFD and SPD are anticipated to address incremental increased demand for fire, emergency medical, and police services. Any potential future facility, staffing, or equipment needs as a result of increased demand for services, traffic volumes, or changes to the transportation network could be included as part of the City's annual Budget and Capital Improvement Program process.

Schools & Libraries

The Preferred Alternative has slightly fewer caretakers' quarters/makers' studios than Alternative 4 at up to 1,475, with most in the Ballard and SODO/Stadium subareas. Less than alternatives 3 and 4, there would be 686 dwellings in the Georgetown/South Park Subarea, but the Preferred Alternative adds mixed use dwellings in Judkins Park (284) and West Ballard (564). With the industry supportive housing and mixed use housing, there would be an increase in population of 6,168 including up to 437 students, more than alternatives 1, 2, or 3 but less than Alternative 4. Schools serving Ballard, SODO/Stadium, and Georgetown/South Park could have increased demand at 20-65% of a typical elementary school capacity (~350). Local libraries in Ballard and South Park would likely see an increase in demand for services.

3.13.3 Mitigation Measures

Incorporated Plan Features

Fire, Emergency Medical, & Police Services

 Compact growth in proximity to SFD and SPD services could result in more efficient service delivery and ability to meet LOS objectives.

Schools & Libraries

None.

Regulations & Commitments

Fire & Emergency Medical Services

- Rules governing fire prevention in the State of Washington and the City of Seattle are addressed in the International Fire Code (IFC) with state adopted amendments in <u>WAC</u> <u>Chapter 51-54A</u>. In addition to the requirements detailed in the 2018 IFC, the City of Seattle has also adopted its own local amendments that can be found in Title 22 Subtitle VI Fire Code of the Seattle Municipal Code. All new development in the primary and secondary study areas is required to meet City of Seattle development regulations as well as the International Building Code and IFC. The Fire Code provides minimum fire and life safety standards for buildings, access roads processes, and fire protection equipment installations. Adequate fire flow to serve potential development is required under the Fire Code. Potential development would also be required to comply with code requirements for emergency access to structures.
- The Seattle Fire Department enforces and is subject to various City of Seattle regulations such as Title 22 Subtitle VI Fire Code, Title 10 Healthy and Safety, Title 11 Vehicles and Traffic, and Title 23 Land Use Code.
- The City sends plans for building construction from the Seattle Department of Construction & Inspections to the Fire Department for review of fire apparatus access and other fire code related issues.

- The City applies standards for live/work units like artists' lofts and caretakers' units to ensure there are exits from sleeping rooms and fire-rated walls and doors between different uses.
- The City of Seattle maintains a Comprehensive Emergency Management Plan (CEMP) which unifies a series of all-hazards documentation to holistically describe the doctrines, strategies, and responsibilities through which the City of Seattle's emergency management system is organized and managed. The City's Disaster Recovery Framework further addresses how the City would partner with the community and coordinate with County, State, and Federal agencies in recovering from the effects of disaster (using a massive earthquake as the premise).

Police

- The Seattle Police Department enforces and is subject to various City of Seattle regulations such as Title 10 Healthy and Safety and Title 11 Vehicles and Traffic.
- Ongoing Seattle Police Department processes to evaluate where to best focus its resources are anticipated to help address future changes in demand for police services in the study area.
- Ongoing City of Seattle capital improvement planning and budgeting efforts are anticipated to address police facility needs, including potential needs for future improvements.

Schools & Libraries

- Ongoing Seattle School District capital facilities management planning is anticipated to be sufficient to address increases in student population. The Seattle School District prepares capital plans and projects are funded by levies.
- SDOT provides a Safe Routes to School program. In addition to education, there are walkway projects to make routes safer.

Other Potential Mitigation Measures

Fire & Emergency Medical Services

- Ongoing City operational and capital facilities planning efforts are anticipated to address incremental increases and other changes in demand for fire services.
- A portion of the tax revenue generated from potential redevelopment in the study area would accrue to the City of Seattle and could be used to help fund fire services.
- The City is considering an option to replace the Magnolia Bridge with a new bridge along Armory Way connecting to Thorndyke Avenue W at W Halladay Street. Replacing the bridge could improve emergency vehicle access to the study are and potentially lower response times.

Police

- A portion of the tax revenue generated from potential redevelopment in the study area would accrue to the City of Seattle and could be used to help fund police services.
- To reduce criminal activity and calls for service, site design principles can be employed such as orienting buildings towards the street, providing public connections between buildings, and providing adequate lighting and visibility.

Schools & Libraries

- The Seattle Public Library has a strategic plan and operations plan that guide the provisions of library services.
- The II and UI zones include potential changes to streetscape standards and could enhance walking routes to schools in areas with added housing.

3.13.4 Significant Unavoidable Adverse Impacts

All studied alternatives would increase the demand for public services with alternatives 2, 3, and 4 <u>and the Preferred Alternative</u> increasing jobs above No Action. The increase in industrial jobs could result in a greater need for fire and emergency services. Increased non-industrial jobs would require apparatus for taller structures in the case of fire or rescue.

All alternatives, particularly alternatives 3 and 4 <u>and the Preferred Alternative</u> would increase housing and increase demand for school and library services.

No significant unavoidable adverse impacts to fire and emergency medical services, police, or schools and libraries are anticipated with application of mitigation measures and regular capital planning.

Section 3.14 **Utilities**



This section documents the effected environment, impacts, mitigation measures, and significant unavoidable impacts of the public utilities that provide services to the study area. Utilities discussed in this section include the public wastewater system (including combined sewer), the stormwater drainage system, and the electrical system.

Impacts of the alternatives on utilities are considered significant if they:

- Are inconsistent with utility system planned growth and capital plans.
- Have the potential to require major new projects or initiatives for energy system upgrades to accommodate redevelopment.

Potable water is provided to the study area by Seattle Public Utilities (SPU). Seattle anticipated water service needs in its Final EIS for the Seattle Comprehensive Plan Update, May 5, 2016, hereby incorporated by reference. To plan for long-term needs and meet regulatory requirements, Seattle Public Utilities regularly updates its Water System Plan. The 2019 Water System Plan is the latest update. It describes near- and long-term plans for the regional water system. Through their water forecasting, asset management framework, and CIP, SPU employs a variety of strategies that allow them to anticipate and adjust to changing demands. Future developments would seek a water availability certificate (WAC) from SPU that confirms SPU water infrastructure exists to supply the parcel(s) (City of Seattle n.d.). The document identifies requirements, system improvements, and conditions necessary to provide water service to the parcel. With the Comprehensive Plan Final EIS, the current Water System Plan, and the WAC process, water services are addressed and not further considered in this EIS.

3.14.1 Affected Environment

Data & Methods

This section considers wastewater, stormwater, and power provider plans and studies. The section evaluates changes in population, dwelling units, and jobs and their effect on wastewater generation, the quantity of stormwater runoff, and electrical demand.

Service Providers

Seattle Public Utilities (SPU) manages the public wastewater and stormwater drainage in the City of Seattle. King County Wastewater Treatment Division (WTD) manages all the wastewater treatment plants and wet weather treatment facilities within the City of Seattle and surrounding King County. Together, SPU and WTD manage the combined sewer system. Seattle City Light (SCL) manages the electric power generation, transmission, and distribution services in the City of Seattle.

Wastewater & Combined Sewer

SPU Drainage and Wastewater Utility collects and conveys wastewater through a system of pipes, detention facilities, pump stations, outfalls, and treatment facilities. Most of the wastewater flows collected in the study area wastewater collection system are conveyed to King County for regional conveyance and treatment. The King County WTD operates the West Point Wastewater Treatment Plant (West Point) and Elliott West Wet Weather Treatment Facility (Elliott West), which serve the BINMIC and Greater Duwamish MICs and the subareas within. A small area in the southwest corner of the study area discharges to the Southwest Suburban Sewer District.

Exhibit 3.14-1 West Point Wastewater Treatment Plant Treatment Capacity

	Flow (mgd)
Dry Weather	90
Wet Weather	300 ¹

¹ primary treatment and disinfection for flows between 300 to 440 mgd. Source: Herrera, 2021.

As shown in **Exhibit 3.14-3**, the BINMIC has a combination of a partially separated and combined sewer system and the Greater Duwamish MIC has a combination of partially separated, combined sewer, and separated sewer systems. Both SPU and King County WTD operate combined sewer systems in the city. Combined sewer systems collect stormwater runoff and domestic wastewater in the same pipe and transport it to a wastewater treatment facility for treatment prior to discharge. In partially separated areas a portion of the runoff has been diverted in pipes to the separate drainage system. The primary objective of these separation projects was to reduce emergency overflows of untreated sewage into nearby waterbodies. **Exhibit 3.14-3** shows the partially separated areas in the study area. Areas of the system that were constructed as combined sewer but now function solely for wastewater conveyance have excess capacity because they were sized to convey stormwater, which no longer flows the system in these areas.

The installation of the combined sewer system is older; most pipes date back to the late 1800s and early 1900s. The partially separated system is more recent, with most pipes installed in the 1960s. The local collector pipes range from 8 to 12 inches in diameter and are primarily constructed of vitrified clay and concrete. As shown in **Exhibit 3.14-3**, wastewater lines primarily run north-south through the study area. During dry weather, the northern portion of the Elliott Bay Interceptor conveys wastewater from BINMIC to West Point via the Interbay Pump Station. Flow from the Greater Duwamish MIC is conveyed from either the West Duwamish Interceptor or the southern portion of Elliott Bay Interceptor via the Duwamish and Interbay Pump Stations to West Point.

During wet weather, combined wastewater and stormwater flows in combined sewer systems can exceed the system's capacity (**Exhibit 3.14-1Exhibit 3-53-1**). In the neighborhoods adjacent

to the BINMIC, these wet weather flows from the combined sewer systems are diverted to a 14foot diameter storage tunnel under Mercer Street. The Mercer Street Tunnel can store up to 7.2 million gallons until the Elliott Bay Interceptor has the capacity to transport the wastewater to West Point. Depending on the severity of the storm, stored flow in the tunnel is conveyed to West Point or the Elliott West Wet Weather Treatment Facility (Elliott West) for treatment prior to discharge. During the largest storms—on average, once a year—flows may exceed pumping capacity of Elliott West and are discharged untreated. This untreated flow is known as a "combined sewer overflow" (CSO). CSOs from regulated outfalls are allowed at times, when the system reaches capacity, and as permitted by agreements with the Washington Department of Ecology (Ecology) and the U.S. Environmental Protection Agency (EPA). SPU and King County WTD have made significant upgrades to the conveyance and detention capacity of the combined sewer system to limit these overflows. As the combined sewer system was designed to convey both wastewater and stormwater, during dry weather there is not a capacity issue for wastewater flow alone. More information about CSOs can be found in Section 3.14.3, **Regulations & Commitments** (see King County & City of Seattle Guidelines, Regulations for Wastewater & Combined Sewer).

Exhibit 3.14-2 summarizes the length of the combined, sanitary, and total systems in each subarea.

Subarea	Infrastructure Type	Total Pipe Length (ft) ¹
Ballard	Combined System	419
	Sanitary System	5,184
	Total System	5,604
Interbay Dravus	Combined System	4,492
	Sanitary System	310
	Total System	4,802
Interbay Smith Cove	Combined System	22,773
	Sanitary System	19,931
	Total System	42,705
SODO/Stadium	Combined System	21,719
	Sanitary System	46,897
	Total System	639,789
Georgetown/South Park	Combined System	15,291
	Sanitary System	18,733
	Total System	34,024

Exhibit 3.14-2 Length of Wastewater Infrastructure

¹ Infrastructure within the City of Seattle Right of Way (ROW) were not included in the calculations. Source: Herrera, 2021.



Exhibit 3.14-3 Wastewater and Combined Sewer System

Source: Herrera, 2021.
Stormwater

Stormwater runoff from impervious surfaces in the BINMIC and Greater Duwamish MICs is collected and conveyed from streets and properties, through the stormwater collection system. A portion of the system is managed by the Port of Seattle's Marine Stormwater Utility and much of the water is conveyed to receiving water bodies by the SPU storm drain system. This collection system includes the piping network, catch basins, and manholes that convey stormwater from the BINMIC and Greater Duwamish MICs to Elliott Bay (see **Exhibit 3.14-5**). Stormwater surrounding the MICs is collected and conveyed through SPU's combined and separated sewer systems. A small percentage of stormwater runoff from public rights-of-way is collected and conveyed in separate pipe networks within the partially separated portion of the surrounding neighborhoods (see **Exhibit 3.14-5**). The combined and partially separated systems are described in the wastewater discussion, above.

The stormwater drainage system within the partially separated areas includes a series of catch basins running along main drainage lines to take surface water runoff from roadways. In some areas, stormwater flows from these lines are conveyed back into the combined sewer system. In other areas, stormwater flows continue within the drainage system and discharge at outfalls to Elliott Bay. As with the wastewater system, SPU manages the storm drain system through asset-based management and operational standards.

Exhibit 3.14-4 summarizes the length of stormwater infrastructure<u>, including stormwater</u> system mainlines managed by SPU and private stormwater mainlines managed by the Port of <u>Seattle's Marine Stormwater Utility</u>, and number of adjacent CSO outfalls in each subarea.

	Total Pipe		
Subarea	<u>Stormwater System</u> <u>Mainlines</u>	<u>Private Stormwater</u> <u>Mainlines</u>	Adjacent CSO Outfalls ²
Ballard	3,993	4,438	10
Interbay Dravus	183	<u>2,864</u>	0
Interbay Smith Cove	28,101	<u>9,848</u>	2
SODO/Stadium	90,661	<u>16,062</u>	11
Georgetown/South Park	22,371	<u>51,283</u>	6

Exhibit 3.14-4 Length of Stormwater Infrastructure and Adjacent CSO Outfalls in the Study Area by Subarea

¹ Infrastructure within the City of Seattle Right of Way (ROW) were not included in the calculations.

Source: Herrera, 2021.

² King County and Seattle Public Utilities CSO outfalls within a 150-ft buffer of each subarea.



Exhibit 3.14-5 Stormwater System in the Study Area

Electrical Power

Seattle City Light (SCL), a municipal utility, supplies electrical power to customers in Seattle, including the BINMIC and Greater Duwamish MIC, and some portions of King County north and south of the city limits. Electric power infrastructure is shown in **Exhibit 3.14-7**. SCL's transmission system includes several high-voltage, 115.1-kilovolt (kV) and 230-kV transmission lines. These transmission lines run between electrical substations, which lower the voltage of the electricity before transferring it to the distribution lines. In the study area, the SCL system uses a combination of overhead and underground electrical transmission and distribution lines. The Broad Street Substation, located on 6th Avenue North between Broad Street and Thomas Street, is the electrical substation serving the BINMIC. The Massachusetts Substation, located on Utah Avenue S between Colorado Avenue S and S Massachusetts Street, is the electrical substation serving the Duwamish BIC.

SCL also has an ongoing program since 2007 to provide electrical service connections and related improvements within the Broad Street network areas. This program includes capacity additions work associated with service connections to customers. The program also replaces or installs network transformers, network protectors and specialty transformers, and performs other improvements. This program fluctuates with land use development (City of Seattle 2015b).

Exhibit 3.14-6 summarizes the approximate lengths of electrical lines in the subareas.

Exhibit 3.14-6	Electrical	Transmissio	on Lines	by Subarea

Subarea	Total Line Length (ft) ¹
Ballard	52,298
Interbay Dravus	18,787
Interbay Smith Cove	7,677
SODO/Stadium	118,042
Georgetown/South Park	85,752

¹ Infrastructure within the City of Seattle Right of Way (ROW) were not included in the calculations Source: Herrera, 2021.



Exhibit 3.14-7 Power Infrastructure in Study Area

Source: Herrera, 2021.

3.14.2 Impacts

Impacts Common to All Alternatives

While demand for utilities is expected to be similar for all alternatives, future development could result in adverse impacts to localized portions of the utility system. Seattle Public Utilities (SPU), King County WTD, and Seattle City Light (SCL) currently employ a variety of strategies to anticipate and adjust to changing demands. Both potential impacts and strategies employed by the utilities to respond to changing demand are discussed below.

Wastewater & Combined Sewer

Development under any of the alternatives could result in greater demands on the local wastewater collection system and on the downstream conveyance and treatment facilities. Increased wastewater flow is related to increased water consumption. Flow from the Primary Study Area to West Point (operated by King County WTD) represents only a small portion of the total West Point service area population (**Exhibit 3.14-8**), so increases in wastewater generation within the Primary Study Area under any of the alternatives are small compared to projected increases in flow already accounted for by King County WTD planning documents (King County 2014a). However, as some redevelopment of industrial areas is expected under all alternatives, impacts to the wastewater system should be evaluated for specific industries during future system planning efforts to assess whether historical loading rates and assumptions apply. Individual industries are required to get authorization from King County before discharging wastewater to the sewer system, which may involve on-site pretreatment. As noted in the Mitigation Measures section, development under the proposed alternatives is not expected to alter permitted use of King County facilities.

		Residential ¹		Commercial	Industrial	1
Population Category		Households	Population	Employment Population	Employment Population	Total Population
2018 Populatic	on Served by West Point ²	343,902	705,000	580,000	37,000	1,322,000
2044 Population Served by West Point ²		404,878	830,000	815,000	40,200	1,685,200
Existing	2018 Population ³	413	847	44,000	54,500	99,347
Conditions	Percent ⁴	0.1%	0.1%	7.6%	147.3%	7.5%
Alternative 1 No Action	2044 Population ³	488	1,000	55,600	66,400	123,000
	Percent ⁵	0.1%	0.1%	6.8%	165.2%	7.3%
Alternative 2	2044 Population ³	493	1,011	53,500	79,400	133,911
	Percent ⁵	0.1%	0.1%	6.6%	197.5%	7.9%

Exhibit 3.14-8 Current and Future Wastewater Service Population in the West Point Wastewater Treatment Facility Service Area Compared to Population in the Study Area

Ch.3 Environment, Impacts, & Mitigation Measures • Utilities

			Residential ¹		Industrial		
Population Category		Households	Population	Employment Population	Employment Population	Total Population	
Alternative 3	2044 Population ³	2,101	4,307	72,400	83,500	160,207	
	Percent ⁵	0.5%	0.5%	8.9%	207.7%	9.5%	
Alternative 4	2044 Population ³	3,686	7,556	74,400	83,300	165,256	
	Percent ⁵	0.9%	0.9%	9.1%	207.2%	9.8%	
Preferred Alternative	2044 Population ³	<u>3,422</u>	<u>7,015</u>	<u>63,192</u>	<u>70,853</u>	<u>141,060</u>	
	Percent ⁵	<u>0.9%</u>	<u>0.9%</u>	<u>7.8%</u>	<u>176.3%</u>	<u>8.4%</u>	

¹ Conversion between number of residential households and residential population assumes the 2020 citywide household size of 2.05 (CAI 2021; City of Seattle, 2021)

² Estimate of the total population served by the West Point Wastewater Treatment Plant in 2018 (Current Conditions) and 2044 (Future Conditions) (King County 2014a). These population assumptions represent the most recent publicly-available data. It is likely that King County is in the process of updating these projections to account for growth expected within the service area, including growth expected within the Primary Study Area as part of Alternative 1 No Action.

³ Population served with the Primary Study Area

⁴ Percent of the 2018 population served within the Primary Study Area when compared to the estimate of the total population served by the West Point Wastewater Treatment Plant in 2018 (King County 2014a).

⁵ Percent of the 2044 population served within the Primary Study Area when compared to the estimate of the total population served by the West Point Wastewater Treatment Plant in 2044 (King County 2014a).

Source: Herrera, 2021.

Under all alternatives, increases in employment and/or residential populations in portions of the Primary Study Area are expected to result in greater wastewater generation, which could locally impact the wastewater collection system operated by SPU. Although there may be a greater overall need for wastewater system capacity with increased density, new development can reduce per-capita demand, as newer, low- or no-flow plumbing fixtures and equipment replaces older, less efficient, installations. This could help reduce overall impact. Consistent with SPU's guiding plans and asset management framework, SPU employs a variety of strategies to anticipate and adjust to changing demands.

While there would be increased demand on the wastewater system under any of the alternatives, existing programs, such as SPU's asset management framework and the capital improvement program (CIP), are in place to identify and implement projects to address system capacity issues and to incorporate improvements and repairs in association with major redevelopment and projects. As a result of these ongoing programs and current planning, increased demand for wastewater service under any of the alternatives is not considered a significant impact.

Because combined sewers receive both wastewater and stormwater runoff during wet weather, impacts to the combined system result from changes to both wastewater generation and stormwater runoff. Redevelopment governed by current Stormwater Code standards would help control peak rates of stormwater through the local combined sewer systems and reduce the risk of combined sewer overflows. This could potentially result in less usage of King County's CSO treatment facilities, such as West Point and Elliott West for the Ballard and Interbay subareas and the future Georgetown Wet Weather Station in the Georgetown/South Park and SODO/Stadium subareas. More information about the impact of the current Stormwater Code is discussed in greater detail in the Stormwater section below.

Stormwater

In general, increases in impervious area result in higher peak flows and total runoff, but because the majority of the Primary Study Area is impervious, redevelopment expected under all alternatives is not expected to significantly increase total impervious area. As described in **Section 3.14.3 Mitigation Measures**, the 2021 Stormwater Code requires on-site stormwater management to infiltrate, disperse, and retain stormwater runoff to the maximum extent feasible. Where the developed site's stormwater flow is expected to exceed the allowable flow levels, stormwater flow control is required. As a result of these requirements, given that some of the existing development predates modern stormwater requirements, it is expected that there would be a reduction in uncontrolled runoff in the Primary Study Area under all of the alternatives where new construction is anticipated.

The 2021 Stormwater Code also supports incentives for retrofitting existing development, such as opportunities for property owners to reduce their drainage rate if they install flow control and/or treatment facilities designed per the Code, which can include reducing impervious surfaces. Redevelopment that replaces existing impervious surface and provides flow control can reduce runoff rates even below current levels.

Under all scenarios, including Alternative 1 No Action, implementation of on-site stormwater management and continuation of retrofit incentives would continue to reduce adverse impacts on both the combined sewer system and the drainage system. <u>This would be true even if future rainfall patterns are more intense than historic rainfall patterns.</u> No significant adverse location-specific impacts are identified in this review.

Electrical Power

Under all alternatives, including the No Action Alternative, future growth and development would increase demand for electrical energy. With the completion of the Denny Substation project in 2018 described in **Section 3.14.3 Mitigation Measures**, the existing Broad Street Substation and transmission infrastructure is expected to meet future needs through at least 2035.

Under any alternative, the local distribution system may need improvements or reconfiguration to meet future growth needs. Seattle City Light is actively planning to increase infrastructure along the central waterfront and in portions of both MIC areas to support conversion of cargo and cruise vessels to the use of shore power. Specific improvements would be addressed on a project-by-project basis. Currently, Seattle City Light is installing public electric vehicle charging stations in the Ballard and Georgetown/South Park subareas. No significant adverse impacts have been identified for any of the alternatives.

Equity & Environmental Justice Considerations

Under all alternatives, minor impacts to utility services could occur during construction of individual development projects. Construction could disturb existing utility lines; however, any disruptions would be temporary because the construction contractor would be required to establish connections to prevent any disruptions prior to construction and be required to communicate the disruptions to the public in advance. These temporary disruptions could be disproportionately felt by low income and other underserved populations in the study area.

All alternatives are likely to lead to utility improvements in the study area. There is no indication that the improvements are likely to cause adverse impacts to low income and other underserved populations in the study area as long as the utility improvements avoid displacement of these populations. Utility improvements could potentially benefit low income and other underserved populations in the study area, such as in portions of the SODO/Stadium and Georgetown/South Park subareas.

Impacts of Alternative 1 No Action

Wastewater & Combined Sewer

Impacts resulting from Alternative 1 No Action would be the same as described in the discussion of **Impacts Common to All Alternatives**. Compared to the Action Alternatives, there is likely to be less redevelopment in the Primary Study Area and the least amount of increased wastewater service demand and the least reduction in the rate of stormwater runoff to the combined sewer system during wet weather.

Stormwater

Impacts resulting from Alternative 1 No Action would be the same as described in the discussion of **Impacts Common to All Alternatives**. Stormwater runoff in the Primary Study Area would continue to be collected and directed through the stormwater drainage system for discharge to existing outfalls. Potential impacts of future, specific development proposals would be addressed through implementation of the regulations and project-specific environmental review as appropriate. As sites redevelop, implementation of on-site stormwater management required under the 2021 Stormwater Code would continue to reduce adverse impacts that would otherwise occur under existing conditions. However, there would potentially be less redevelopment and less implementation of on-site stormwater management under Alternative 1 No Action, resulting in less reduction of peak flows and total runoff compared to other alternatives.

Electrical Power

Impacts resulting from Alternative 1 No Action would be the same as described in the discussion of **Impacts Common to All Alternatives**. Even without changes to current

Comprehensive Plan policies, development standards, or zoning maps, the demand on the electrical system is likely to increase over time. However, compared to the Action Alternatives, there is likely to be less redevelopment pressure in the Primary Study Area resulting in the least change to electricity demand compared to the other alternatives.

Impacts of Alternative 2

Wastewater & Combined Sewer

There is likely to be a greater increase in wastewater service demand for this Alternative compared to Alternative 1 No Action due to the greater increase in industrial employment. Compared to alternatives 3 and 4, there is likely to be less redevelopment, resulting in less increases in wastewater generation and less reductions of the rate of stormwater runoff to the combined sewer system in the Primary Study Area.

Stormwater

Alternative 2 includes greater change and densification of industrial zones than Alternative 1 which could result in increased implementation of on-site stormwater management. Source control practices will need to be reevaluated by developers and City reviewers as land uses change to ensure that adequate treatment is occurring. Compared to alternatives 3 and 4, there is likely to be less redevelopment resulting in less reduction of the rate of stormwater runoff to the separated stormwater system.

Electrical Power

Assuming greater change and densification of industrial zones than Alternative 1, the demand on the electrical system is likely to be greater under Alternative 2 than Alternative 1, but less than alternatives 3 and 4.

Impacts of Alternative 3

Wastewater & Combined Sewer

There is likely to be a greater increase in wastewater service demand for Alternative 3 compared to alternatives 1 and 2 due to the greater increase in employment and housing, but due to greater redevelopment expected, the rate of stormwater runoff to the combined sewer system is likely to decrease due to the implementation of improved stormwater controls, and less wet weather flow in the combined system. Compared to Action Alternative 4, there is likely to be less increase in wastewater generation and less reduction of stormwater runoff in the Primary Study Area, which could reduce the frequency of CSO events. While increases in residential population are greater for this Alternative than for alternatives 1 and 2, particularly in the Ballard and SODO/Stadium subareas, the total residential population accounts for less

than 1% of the expected residential population served by West Point in 2044 (**Exhibit 3.14-8**Exhibit 3-58) and small when compared to the projected job increases in any given Subarea or the Study Area as a whole. Compared to Action Alternative 4, there is likely to be less increase in wastewater generation and less reduction of the rate of stormwater runoff to the combined sewer system.

Stormwater

Alternative 3 includes increased industrial and non-industrial redevelopment, which could result in increased implementation of on-site stormwater management compared to alternatives 1 and 2. This is likely to decrease the rate of discharge to the stormwater system relative to alternatives 1 and 2, but not as much as Alternative 4.

Electrical Power

Assuming greater change and densification of industrial zones than Alternative 1 and increased non-industrial land used compared to Alternative 2, the demand on the electrical system is likely to be greater for Alternative 3 than alternatives 1 and 2, but less than Alternative 4.

Impacts of Alternative 4

Wastewater & Combined Sewer

The greatest increase in wastewater service demand is expected for Alternative 4 due to the greater increase in employment and housing. Additionally, because the greatest redevelopment is expected under this alternative, the greatest improvements to stormwater flow rates to the combined sewer system are expected, resulting in the greatest reductions to wet weather flow in the combined system when compared to other alternatives. As with Alternative 3, though increases to the residential population are expected, particularly in the Ballard and SODO/Stadium subareas, the total residential employment population accounts for less than 1% of the expected residential population served by West Point in 2044 (**Exhibit 3.14-8Exhibit 3-58**) and small when compared to the projected job increases in any given Subarea or the Study Area and a whole.

Stormwater

Alternative 4 includes the greatest expected redevelopment, which could result in the most implementation of on-site stormwater management compared to the other alternatives. As discussed above, this is likely to decrease the rate of discharge to the stormwater system.

Electrical Power

The demand on the electrical system is likely to be the greatest for Alternative 4 compared to other studied alternatives.

Impacts of the Preferred Alternative

Wastewater & Combined Sewer

Under the Preferred Alternative, increases in employment are expected to be similar to Alternative 2, while increases in housing are expected to be similar to alternatives 3 and 4. Therefore, the increase in wastewater service demand expected for this Alternative is expected to be less than alternatives 3 and 4 and greater than Alternative 2. Redevelopment under this alternative, which is expected to reduce stormwater flow rates to the combined sewer system, is expected to reduce wet weather flow in the combined system more than Alternative 2 and less than alternatives 3 and 4. Though increases to the residential population are expected, the total residential population accounts for just under 1% of the expected residential population served by West Point in 2044 (**Exhibit 3.14-8**) and, as with alternatives 3 and 4, is small when compared to the projected job increases in any given Subarea or the Study Area and a whole.

Stormwater

<u>The Preferred Alternative includes more redevelopment than Alternative 2 and less</u> <u>redevelopment than alternatives 3 and 4, and is expected to result in implementation of on-site</u> <u>stormwater management. As discussed above, this is likely to decrease the rate of discharge to</u> <u>the stormwater system.</u>

Electrical Power

The demand on the electrical system is likely to be greater than Alternative 2 and less than alternatives 3 and 4 compared to other studied alternatives.

3.14.3 Mitigation Measures

Incorporated Plan Features

The Industrial and Maritime Strategy includes policy concepts relevant to Power and Air Quality/GHG:

- Introduce new or strengthened policies into chapters of the Comprehensive Plan that may include the Transportation, Environment, or Container Port elements encouraging transitions to clean fuels and decarbonization of industrial and maritime activities.
- Seattle Municipal Code (SMC 23.50.012) currently permits the use of currently zoned industrial areas for utility services by the King County Department of Natural Resources and Parks (DNRP). The proposed changes would not alter or prohibit currently permitted uses for these DNRP utility services.

Regulations & Commitments

Wastewater & Combined Sewer

SPU Drainage and Wastewater Utility and King County WTD are guided by several federal and state regulations as well as City of Seattle policies, programs, and plans. Regulations and guidance specific to wastewater are described below.

Federal Guidelines & Regulations

Federal guidelines for wastewater include the Clean Water Act (CWA). The 1977 CWA gave the EPA the authority to implement pollution control programs such as setting wastewater standards and regulating point discharges of pollutants. The EPA has the authority to delegate enforcement to the states, where state regulations are required to be at least as strict as federal regulations. The EPA has established minimum requirements for states to use in enacting regulations for wastewater reuse and reclamation. In the State of Washington, Ecology administers and enforces the CWA.

State of Washington Guidelines & Regulations

All wastewater treatment plants (WWTPs) in the State of Washington are regulated by Ecology. Ecology issues wastewater discharge permits, which regulate how WWTPs treat, control, and operate their facilities. WWTPs are required to control the quantity and quality of their discharges into surface or groundwater. These waters of the state include rivers, streams, bays, lakes, and aquifers. Chapter 173-221 of the Washington Administrative Code (WAC) defines WWTP discharge standards in further detail.

As discussed in previous sections, the BINMIC and Greater Duwamish MICs are served by the West Point WWTP. This facility is regulated under the National Pollutant Discharge Elimination System (NPDES) permit No. WA0029181. The permit requires that the West Point facility must not exceed the following design criteria:

- Maximum Month Design Flow (MMDF): 215 mgd
- BOD₅ Influent Loading for Maximum Month: 201,000 lbs/day
- Total Suspended Solids Influent Loading for Maximum Month: 218,000 lbs/day

As part of the renewal process, King County submits a CSO Control Plan approximately every 5 years. Under WAC 173-245, the plan must update Ecology on program achievements, CSO control projects for the next NPDES permit phase, and plan amendments.

King County & City of Seattle Guidelines, Regulations, & Commitments

Regulations on the local level consist of King County Code, King County Public Rules, and SPU's Side Sewer Code. Title 28 of King County Code regulates the disposal of industrial waste into

the sewer system. King County Public Rules PUT 8-13 – 8-16, 8-22, and 8-24 cover the following subjects:

- Local discharge limits
- Construction dewatering
- Discharge of contaminated groundwater to the sewer
- Discharge of cooling water to the sewer

SPU's Side Sewer Code regulates the design, construction, and permitting of privately-owned sewer pipe systems within private property and/or the right-of-way. To work on a side sewer project, SPU requires a Side Sewer Permit. This permit has fees dependent on the scope of work being performed.

Capital Improvement Programs

King County

Implementing capacity expansion projects at each of the County's regional treatment facilities would be initiated as required to meet population growth. Projects at West Point will have the greatest impact on the BINMIC and Greater Duwamish MIC, including near-term (by 2030) improvements to solids digestion.

City of Seattle

Guidance from SPU Drainage and Wastewater Utility includes SPU's 2015 Plan to Protect Seattle's Waterways and the utilities' 2015–2010 Strategic Business Plan (Seattle Public Utilities, 2015a) (Seattle Public Utilities, 2015b). The overriding goals of these plans is to construct and maintain facilities that:

- Reduce the frequency of flooding and sewer backups for customers
- Improve water quality and habitat in the environment
- Reduce sewage overflows and the impacts of stormwater pollution

Within SPU's asset management framework, SPU regularly inspects, repairs, and replaces pipe. As needed, new development may be required to make system improvements (Kelleher, 2016). SPU's Drainage and Wastewater CIP is the vehicle for identifying major projects and programs to rehabilitate, replace, improve, and expand system infrastructure (City of Seattle, 2015b). Projects are ranked based on a set of criteria to establish priority. This includes "level of service" criteria that address the provision of services to customers, including projects that address system capacity needs. Current Drainage and Wastewater CIP projects within the BINMIC include the Ballard Locks Improvements and the Ship Canal Water Quality Project (SCWQP). Flow from the Greater Duwamish MIC also impacts the SCWQP.

Within the CIP, SPU has an ongoing program, the Wastewater Capacity Improvement Program, to enhance sanitary sewer service to Seattle customers by addressing current and projected capacity limitations of the wastewater system through structural improvements. Such

improvements may include infiltration and inflow (I/I) reduction, increased conveyance capacity, and individual customer measures to reduce the risk that customers would experience backups of sewage into their homes and businesses during storm events.

As part of another ongoing program in the CIP, the Shared Cost Project Program, SPU works take better advantage of opportunities to incorporate improvements and repairs to the drainage and wastewater systems with major redevelopment and projects undertaken by others (e.g., private developers, other city departments, regional and state agencies). Due to increased project costs (\$5.4 million) in Waterfront CSO projects, the Shared Cost Projects budget was reduced by an overall \$9.2 million in 2021.

Stormwater

SPU Drainage and Wastewater Utility and the Port of Seattle's Marine Stormwater Utility are guided by several federal and state regulations as well as City of Seattle policies, programs, and plans. Regulations and guidance specific to stormwater are described below.

Federal Guidelines & Regulations

Federal guidelines for stormwater include the Federal Endangered Species Act (ESA). The ESA is intended to protect threatened or endangered species from extinction. The ESA prohibits the "take" of all listed species, including a take that could result from the Port's stormwater facility operations or private development stormwater management activities that are permitted by the Port.

State of Washington Guidelines & Regulations

The State of Washington requirements for stormwater management for the City of Seattle are described in the Western Washington NPDES Phase I Municipal Stormwater Permit (Phase I Permit) (Ecology 2019). The 2019-2024 Phase I Permit, issued by Ecology on July 1, 2019, and effective on August 1, 2019, addresses a variety of issues associated with stormwater runoff and requires the City to develop several distinct stormwater management program (SWMP) components:

- Municipal separate storm sewer system (MS4) permit mapping and documentation
- Public involvement and participation
- Controlling runoff from new development, redevelopment, and construction sites
- Stormwater planning
- Structural Stormwater Controls Program
- Source Control Program for Existing Development
- Illicit discharge detection and elimination (IDDE)
- Operations and Maintenance Program
- Education and Outreach Program
- Compliance with Total Maximum Daily Load (TMDL) requirements

- Monitoring and assessment
- Reporting requirements

The Port of Seattle is a secondary permittee under the Phase I Permit due to its ownership and operation of its stormwater system within the City of Seattle that drains to the Ship Canal, Shilshole Bay, Duwamish River, and Elliot Bay. The following requirements apply to the Port of Seattle:

- Education Program
- Public Involvement and Participation
- Illicit Discharge Detection and Elimination
- Construction Site Stormwater Runoff Control
- Post-Construction Stormwater Management for New Development and Redevelopment
- Operation and Maintenance Program
- Source Control in Existing Developed Areas
- Monitoring Program
- Compliance with TMDL requirements
- Monitoring and assessment
- Reporting requirements

Most of the Port's property is leased to commercial and industrial tenants. Approximately 70% of these properties are covered by an NPDES Industrial Stormwater General Permit, which includes additional requirements beyond those in the Phase I Permit. Maritime tenants play a crucial role in protecting water quality in Puget Sound. Any polluting activity has direct effects on the nearshore waters and Puget Sound. The Port is actively working with tenants to improve operations and manage stormwater runoff to protect the natural environment.

City of Seattle Guidelines & Regulations

As described in the Wastewater & Combined Sewer section above, SPU is guided by several federal regulations, City policies, and plans that address wastewater and stormwater drainage. SPU manages stormwater programs in the combined sewer area to improve water quality and habitat in the environment by reducing sewage overflows and the impacts of stormwater pollution. SPU also implements rules governing management of stormwater on private and public property through its current stormwater code (2021 Stormwater Code). The City's NPDES permit, issued in December 2005, requires implementation of stormwater pollution prevention programs in the combined sewer areas and is described in the section above (the permit was last modified issued on August 1, 2019).

Starting in 2009 and continuing with the 2021 Stormwater Code, Seattle has required on-site stormwater management (formerly green stormwater infrastructure) when feasible, as part of stormwater mitigation for all development and redevelopment projects. Examples of on-site stormwater management include permeable pavement, rainwater harvesting, rain gardens, infiltration facilities, bioretention facilities, and vegetated roofs. Individual projects are required

to manage on-site stormwater runoff in accordance with City requirements to ensure that a development properly regulates its stormwater runoff.

It also should be noted that as described above, both SPU and King County WTD are required by agreements with Ecology and the EPA to reduce combined sewer overflows, of which stormwater is a component.

Capital Improvement Programs

King County

King County's 2018 CSO Control Program Update (King County 2018) presents a series of projects to control King County's remaining uncontrolled CSO locations in collaboration with SPU. The plan includes projects that would be built in the BINMIC and others that would be built in the Greater Duwamish MIC.

King County entered a consent decree with the U.S. Department of Justice and EPA (filed July 3, 2013) that ensures its CSO Control Plan (King County 2012a) is completed by 2030. King County had already committed to limiting CSOs to one per year at each outfall by 2030 through its adopted policies and a 2011 Agreement with Ecology.

City of Seattle

SPU is preparing a comprehensive strategy, The Plan to Protect Seattle's Waterways (Plan), to reduce CSOs and stormwater pollutants. The goals of the Plan are to protect public health and the environment while complying with federal and state regulations. The Plan is being developed under a Consent Decree agreement with EPA, Ecology, and the U.S. Department of Justice. The Consent Decree was entered in United States District Court for Western District of Washington on July 3, 2013. The Plan will define projects to control a significant source of contamination and when implemented, the Plan will bring the City into compliance with the State and Federal requirements for CSO discharges. Specifically, the Plan will:

- Identify areas of Seattle where projects are needed to reduce combined sewer overflows.
- Evaluate alternatives for reducing combined sewer overflows in these areas.
- Identify additional areas where projects to control and treat polluted stormwater runoff will improve water quality.
- Recommend a schedule for designing and constructing projects.
- Estimate program costs and associated impacts on Seattle Public Utilities customer bills.
- Consider public and stakeholder input.

The Plan includes an Executive Summary (Volume 1), the Long-term Control Plan (Volume 2), the Integrated Plan (Volume 3), and the Environmental Impact Statement (Volume 4).

The Long-term Control Plan (LTCP) includes a ranking of the uncontrolled CSO basins with the largest negative impact on receiving water bodies and human health. The following basins are included within the BINMIC and Greater Duwamish MIC:

- Basins 174 and 147. Fremont/Wallingford
- Basins 107 and 111. East Waterway and Duwamish

SPU selected the Shared West Ship Canal Tunnel Option as the recommended LTCP option to provide the greatest benefit to receiving waterbodies and human health. The City would be the lead agency for construction and operation of the facility under the terms of a joint project agreement to be executed with King County. This project would impact the Freemont/Wallingford basins within the study area, which include portions of the Ballard Subarea.

The Integrated Plan identifies LTCP projects to be deferred until after 2025 so that the City can focus available resources on implementing the proposed stormwater projects. The Integrated Plan consists of implementing three stormwater projects by 2025 and deferring construction completion of six candidate LTCP projects until 2030. The three stormwater projects are as follows:

- Natural Drainage Systems (NDS) Partnering
- South Park Water Quality (WQ) Facility
- Street Sweeping Expansion Arterials

NDS Partnering would entail reconstructing City rights-of-way to manage flow and provide water quality treatment for urban runoff using primarily the green infrastructure practice of bioretention (i.e., engineered rain gardens). The South Park WQ Facility would provide active basic treatment for roughly 74 million gallons per year of stormwater runoff from a largely industrial area that discharges to the Lower Duwamish Waterway, thereby reducing the potential for recontamination of sediment remediation areas. This affects the SODO/Stadium and Georgetown/South Park subareas. The Street Sweeping Expansion Arterials would expand the area, frequency, and duration of the City's current arterial street sweeping efforts within the Primary Study Area.

Electrical Power

At the federal level, all electrical utilities are regulated by the 2020 National Electric Code (NEC). The State of Washington has adopted the 2020 NEC as of November 1, 2020 and can be found in WAC 296-46B. In addition to the NEC, the WAC also includes the International Energy Conservation Code, as provided in RCW 19.27A,020. This code has been adopted by the State Building Code Council in Chapter 51-11C and 51-11R WAC.

The City of Seattle adopts the 2020 NEC as part of their 2020 Seattle Electrical Code and the International Energy Conservation Code as part of their Seattle Energy Code. This code generally states that the State of Washington energy code shall be designed to construct increasingly energy efficient homes and buildings that help achieve the broader goal of building zero fossil-fuel greenhouse gas emission homes and buildings by the year 2031, and to require new buildings to meet a certain level of energy efficiency.

Capital Improvement Programs

SCL has recently completed two projects which affect the Primary Study Area: the Denny Substation and the Broad Street improvements. The Denny Substation project was completed in 2018 in response to the high electrical load density caused by rapid redevelopment in the South Lake Union area over the past 15 years. In addition to serving the current and future needs of the South Lake Union area, the project frees up capacity at the Broad Street Substation, providing more system flexibility to accommodate current and future growth in the BINMIC.

SCL has an ongoing program since 2007 to provide electrical service connections and related improvements within the Broad Street network areas. This program includes capacity additions work associated with service connections to customers. The program also replaces or installs network transformers, network protectors and specialty transformers, and performs other improvements. This program fluctuates with land use development (City of Seattle, 2015b).

The Port of Seattle is increasing shore power available at terminals to reduce maritime emissions (Starcrest, 2018). Upcoming projects within the SODO/Stadium Subarea include planned shore power improvements in Terminal 15, Terminal 18, and possibly the electrification of Terminal 30 and the Coast Guard Station.

Other Potential Mitigation Measures

Wastewater & Combined Sewer

 Water Conservation Measures: Redevelopments may reduce per-capita water demand (and therefore, wastewater service demand) by using newer, low- or no-flow plumbing fixtures and equipment.

Stormwater

• No additional mitigation is proposed.

Electrical Power

- Future service system needs could be identified and evaluated through collaborative planning between Seattle's Office of Planning & Community Development and Seattle City Light.
- Installation of photovoltaic and other local generating technologies would reduce the demand on the public generating and distribution facilities.
- Construction and operation of LEED compliant (or similar ranking system) buildings would reduce the level of increase required in power systems.
- The use of passive systems, such as building design which utilizes layout and materials for transfer of heat rather than electrical systems, and modern power saving units would reduce the use of power in building heating and cooling. This could include, but is not limited to upgraded levels of insulation, reduced air infiltration, and selection of energyefficient appliances.

3.14.4 Significant Unavoidable Adverse Impacts

Wastewater & Combined Sewer

No significant unavoidable adverse impacts on wastewater and combined sewer systems are anticipated. The levels of development proposed under all alternatives are expected to be managed through King County WTD and SPU's existing, ongoing processes for identifying CIP projects to address system capacity issues and reduce CSO frequency.

Stormwater

No significant unavoidable adverse impacts on the stormwater system are anticipated. New development allowed under any alternative would be required to meet City stormwater codes that would likely improve stormwater management (i.e., reduced flow rates and improved water quality) relative to existing conditions, and CIP projects identified in the Primary Study Area as part of SPU's asset management program would improve system capacity and performance.

Electrical Power

No significant unavoidable adverse impacts on the electrical system are anticipated. Recent SCL investments in the power system are anticipated to meet growth needs under all studied alternatives and development proposals the require specific improvements to the system would be addressed at a planning level through regular capital planning cycles as well as on a project-by-project basis.

[This page is intentionally blank]



4.1 Introduction

The City received 137 comments from the tribes, agencies, organizations, and individuals listed in **Exhibit 4.1-1** and **Exhibit 4.1-2** during the Draft EIS comment period. The Draft EIS was published on December 16, 2021 with a 45-day comment period from December 16, 2021 to January 31, 2022. The comment period was extended to March 2, 2022. Additional engagement was conducted with the Georgetown and South Park communities through April 15, 2022. Comments were received via e-mail, the online commenting survey, and at the January 11, 2022 and January 12, 2022 virtual public hearings.

The issues raised in each comment letter and verbal statement are numbered and provided correspondingly numbered responses. Comments that state preferences on alternatives or other matters are acknowledged with a response that the comment is noted and forwarded to City decision makers. Comments that address methods, analysis results, mitigation, or other matters are provided a response. See **Section 4.2** for the response to common comment themes, **Section 4.3** for individual responses to comments, and **Section 4.4** for the marked comment letters and public hearing transcripts.

Number	Last Name	First Name	Date	Agency/Organization
Tribes/Ind	digenous			
1	Hansen	Cecile	3/2/2022	The Duwamish Tribe
State/Reg	ional/Local Ager	ncies		
2	Cotten	Mike	2/28/2022	Washington State Department of Transportation
3	Curtis	Joshua et al.	3/2/2022	Washington State Major League Baseball Stadium Public Facilities District (PFD), Washington State Public Stadium Authority (PSA)
4	Felleman	Fred	3/2/2022	Port of Seattle The Northwest Seaport Alliance
5	Saganic	Erik	2/18/2022	Puget Sound Clean Air Agency
6	Inghram	Paul	3/2/2022	Puget Sound Regional Council
7	Panganiban	Justin	1/25/2022	Seattle Department of Transportation
8	Acutanza	Jeanne	3/2/2022	Seattle Freight Advisory Board
9	Mohler	Rick et al.	2/18/2022	Seattle Planning Commission
10	Gannon	Rob	2/28/2022	Seattle Public Schools
11	Persak	John	2/28/2022	The Office of Economic Development
Special In	terest Organizat	ions/Corporat	ions	
12	Brower	Joshua	3/2/2022	Brower Law, Salmon Bay Sand and Gravel Company
13	Burke	Suzanne	3/2/2022	Freemont Dock Company via Houlihan Law
14	Ciserella	Mike	3/3/2022	Cantera Development Group
15	Clark	Mel	3/9/2022	CleanTech Alliance

Exhibit 4.1-1 List of Written Commenters

Number	Last Name	First Name	Date	Agency/Organization
16	Clawson	Jessica	3/2/2022	Interbay Urban Investors
17	Clawson	Jessica	3/2/2022	Madisonian Manager, LLC
18	Clawson	Jessica	3/2/2022	AnMarCo
19	Daniels	Kevin	3/1/2022	First and Utah Street Associates, LLC
20	Ffitch	Eric	3/2/2022	BNSF Railway Company, Freezer Longline Coalition, ILWU Local 19, Inlandboatmen's Union of the Pacific, Manufacturing Industrial Council, North Seattle Industrial Association, Pacific Merchant Shipping Association, Port of Seattle, Puget Sound Pilots, Seattle Marine Business Coalition, The Northwest Seaport Alliance, Transportation Institute, Vigor, Washington Maritime Federation, Port of Seattle, Puget Sound Pilots, Seattle Marine Business Coalition, The Northwest Seaport Alliance, Transportation Institute, Vigor, Washington Maritime Federation
21	Fu	Peggi	3/2/2022	NAIOP Washington State
22	Gering	Dave	3/2/2022	Manufacturing Industrial Council
23	Gilder	Ginny	3/9/2022	Gilder Office for Growth, LLC
24	Gillespie	Robert	2/28/2022	Lander Street Partners, LLC
25	Gillespie	Steve	3/2/2022	Hess Callahan Grey
26	Goodman	Erin	2/15/2022	SODO BIA
	Goodman	Erin	1/24/2022	SODO BIA
27	Horn	Colleen	3/2/2022	MAK Management, LLC
28	Howard	Lisa	2/11/2022	Alliance for Pioneer Square
29	Johnson	Kathleen	2/27/2022	Historic South Downtown
30	Krohn	Herb	3/3/2022	SMART Transportation Division, United Transportation Union
31	Lehmann	Ted	3/2/2022	Industrial and Maritime Strategy Council
	Lehmann	Ted	2/23/2022	Industrial and Maritime Strategy Council
32	Loe	Laura	3/2/2022	Share The Cities Action Fund
33	Malshuk	Nicholas	3/1/2022	First South Properties, LLC
34	McCullough	John	3/2/2022	Seattle Industrial Lands Coalition
35	Nelson	Patty	3/3/2022	Elliott Way Partners, LLC
36	Nitze	Peter	2/23/2022	Nitze-Stagen
37	Printz	Peggy	3/2/2022	Seattle Cruise Control
38	Rivera	Fred	3/1/2022	Seattle Mariners
39	Ugles	Herald et al.	3/1/2022	International Longshore and Warehouse Union Locals 19, 52, and 98, Inland Boatmens Union
40	Selig	Jordan	3/2/2022	J Selig Real Estate, LLC
41	Trohimovich	Tim	1/31/2022	Futurewise
42	Tucker	Tarrance	2/28/2022	Pacific Christian Academy
43	Vaughn	Greg	2/22/2022	GPG&C Investment Group LLC
44	Weed	Mark	3/1/2022	SoDo Industrial Coalition

Number	Last Name	First Name	Date	Agency/Organization
Individua	ls			
45	Aggen	Angie	2/24/2022	Individual
46	Anane	Layla	2/24/2022	Individual
47	Anawalt	Bradley	3/4/2022	Individual
48	Baker	Dan	2/17/2022	Individual
49	Brubeck	Donald	3/1/2022	Individual
50	Burg	Jack	2/28/2022	Individual
51	Bush	Erica	1/26/2022	Individual
52	Clark	Justin	2/21/2022	Individual
53	Corbin	Lisa	3/2/2022	Individual
54	Dee	Katherine	2/28/2022	Individual
	Dee	Kate	2/19/2022	Individual
55	Devine	Paul	3/2/2022	Individual
56	Dickinson	Anne	2/28/2022	Individual
57	Dillon	Ann	3/1/2022	Individual
58	DiMartino	Janie	2/24/2022	Individual
59	Dubicki	Raymond	3/1/2022	Individual
60	Dunn	Kathleen	3/2/2022	Individual
61	Eldridge	Xen	2/19/2022	Individual
62	Fragada	Tony	3/2/2022	Individual
63	Frishholz	Christine	2/24/2022	Individual
64	Fiorito	Dan	1/31/2022	Individual
65	Graham	Kirsten	2/24/2022	Individual
66	Greene	Angela	2/28/2022	Individual
67	Hammerberg	Rita	3/1/2022	Individual
68	Hanlon	Robert	3/29/222	Individual
69	Huling	Sharon	2/24/2022	Individual
70	Kartchner	Dylan	2/25/2022	Individual
71	Katz	Andrew	3/2/2022	Individual
72	Kromm	Richard	2/28/2022	Individual
73	Lau	Wayne	2/28/2022	Individual
74	Lewis	Maggie	2/28/2022	Individual
75	Livingston	Robert	3/1/2022	Individual
76	Main	Bonnie	2/24/2022	Individual
77	Mathison	Jon	2/2/2022	Individual
78	Menin	Andrea	3/2/2022	Individual
79	Olofson	Bree	2/28/2022	Individual

Number	Last Name	First Name	Date	Agency/Organization
80	Perry	Chuck	3/2/2022	Individual
81	Personett	Wendy	2/26/2022	Individual
82	Phillips	Maria	2/28/2022	Individual
83	Robinson	Kathryn	2/24/2022	Individual
84	Shaffer	Brett	3/1/2022	Individual
85	Shaw	Aaron	2/24/2022	Individual
86	Standifer	Nancy	2/28/2022	Individual
87	Strohmeier	Jill	2/28/2022	Individual
88	Sundquist	Stephen	2/25/2022	Individual
89	Wood	Shawn	2/24/2022	Individual
90	Anonymous	Aiden	2/18/2022	Individual
Comment	ts via Georgetow	/n / South Park	Engagement	:
91	Schaefer	Rachel	4/15/2022	Cascade Bicycle Club
92	Fong	Alice	4/15/2022	Center for Ethical Leadership
93	_	_	4/15/2022	Duwamish River Accountability Group
94	Bush	Erica	4/15/2022	Duwamish Valley Safe Streets
95	Farrazaino	Samuel	4/15/2022	Equinox Development Unlimited LLC
96	Ramirez	George	4/14/2022	Georgetown Community Council, King County International Airport Community Coalition
97	Hampton- Clarridge	Adrienne	4/15/2022	Georgetown Community Council, King County International Airport Community Coalition, Duwamish River Community Coalition, Duwamish Valley Affordable Housing Coalition, Duwamish Valley Safe Streets
98	Davidson	Sara Ann	4/15/2022	Georgetown Merchants Association
99	Bookwalter	Jake	4/15/2022	Georgetown Youth Council
100	_	_	4/10/2022	Seattle Bicycle Advisory Board
101	Schwartz	Robin	4/14/2022	South Park Neighborhood Association (SPNA)
102	Simson	Cari	4/15/2022	Urban Systems Design
103	Benetua	Michelle	4/15/2022	Individual
104	Bookwalter	Emmett	4/14/2022	Individual
105	Bookwalter	Melissa	4/14/2022	Individual
106	Bushue	Cedar	4/11/2022	Individual
107	Carpenter	Karen Paola	4/15/2022	Individual
108	Claxton	Jo	4/15/2022	Individual
109	Cocking	Penny	4/15/2022	Individual
110	Dae	Tiffany	4/15/2022	Individual
111	Del Rio	Eleana	4/15/2022	Individual
112	Facundo	Victor	4/14/2022	Individual

Number	Last Name	First Name	Date	Agency/Organization
113	Gallagher	Erin	4/15/2022	Individual
114	Kirschenbaum	John	4/15/2022	Individual
115	Knowles	Melissa	4/14/2022	Individual
116	Krejci	Holly	4/15/2022	Individual
117	Lanen	Steve	4/15/2022	Individual
118	Madison	Tracy	4/15/2022	Individual
119	Medina	Rosario	4/15/2022	Individual
120	Miller	Kate	4/15/2022	Individual
121	Morrison	Кау	4/15/2022	Individual
122	Neil	Tim	4/15/2022	Individual
123	Nyland	Kathy	4/15/2022	Individual
124	Nyland	Kelsey	4/15/2022	Individual
125	Rajcich	Brooke	4/14/2022	Individual
126	Rivera	Melina	4/15/2022	Individual
127	Ryan	Maureen	4/15/2022	Individual
128	Schiffer	Andrew	4/15/2022	Individual
129	Smith	Ethan	4/15/2022	Individual
130	St John	Peter	4/12/2022	Individual
131	Sweet	M. Anne	4/15/2022	Individual
132	Terrenzio	Andrea	4/15/2022	Individual
133	Tilley	Joanne	4/14/2022	Individual
134	Veloria	Velma	4/15/2022	Individual
135	White	Мауа	4/15/2022	Individual
136	Woo	Anita	4/14/2022	Individual
137	Wright	Laura	4/15/2022	Individual

Exhibit 4.1-2 List of Verbal Commenters from the Public Hearings

Number	Last Name	First Name	Hearing Date	Agency/Organization
H1	Curtis	Josh	1/11/2022	Washington State Ballpark Public Facilities District
H2	Marchione	John	1/11/2022	Washington State Public Stadium Authority
H3	_	Scott	1/12/2022	Individual
H4	Williams Jr.	Dennis	1/12/2022	Individual
H5	Loe	Laura	1/12/2022	Share The Cities Action Fund
H6	_	Scott	1/12/2022	Individual

4.2 Response to Common Comment Themes

Below is a list of common comment themes and comprehensive responses. Individual responses to comments refer to relevant discussions.

4.2.1 Economic & Market Analysis

Comment Theme: Request for economic feasibility, cost-estimates, or market analysis.

Response: SEPA does not require cost-benefit or economic analysis (WAC 197-11-448 and 450). Separate from the EIS, the City considered economic feasibility information in preparation of draft zoning changes and/or Comprehensive Plan change proposal. The City also solicited input from property owner and development stakeholders when drafting development standards. Economic feasibility of development is affected by many factors, including unpredictable and frequently changing market conditions. The time horizon of the EIS is over 20 years, and factors that affect the short-term feasibility of development are likely to change over the study period. The City has considered the best available information on economic feasibility and will continue to conduct additional economic feasibility testing when preparing zoning change legislation. The Preferred Alternative includes refinements to development standards especially for the Industry and Innovation (II) zone in consideration of economic feasibility of development.

4.2.2 Non-Conforming Uses

Comment Theme: Concern about nonconforming uses under the proposed land use concepts.

Response: Nonconforming uses are permitted to continue subject to provisions of the Seattle Land Use Regulations (SMC Subtitle III). Under existing regulations, a nonconforming use that has been discontinued for more than 12 consecutive months shall not be reestablished or recommenced (SMC 23.42.104(B)) and would need to adhere to the underlying zoning regulations if redeveloped. As a part of the proposal the City would add flexibility for nonconforming uses in the Maritime Manufacturing and Logistics (MML) zone. In the MML zone special accommodation would be given to allow nonindustrial uses that exceed maximum size of use limits prior to the adoption of legislation establishing the MML zone to reestablish or recommence without a time limit. Additional flexibility would also be provided to allow for existing commercial office uses with an operational connection to an industrial use or an existing Information Computer Technology (ICT) uses to expand beyond maximum size of use limits.

4-7

4.2.3 Industry Supportive Housing

Comment Theme: Clarify the definition of industry supportive housing.

Response: In Draft EIS alternatives the concept of industry supportive housing is included in alternatives 3 and 4 in the Urban Industrial (UI) zone. Under Alternative 3 it would mean allowances for a.) up to two caretakers' quarters in which an owner or employee of an on-site business could reside, and b.) workspace studios in which a person who operates a making-use or arts business could live in a combined guarters with their workspace. Under Alternative 3 the maximum density of the total number of caretakers guarters and workspace studios is 25 per acre. Under Alternative 4 the same concept would apply with slightly more liberal allowances of up to three caretakers' guarters per business and a maximum density of 50 per acre. Note that in the Preferred Alternative housing would in the UI zone as a conditional use in criteria-limited locations would not be occupancy limited to the industry supportive housing concept. Under the Preferred Alternative the limited industry supportive housing standard could be met when a developer either a.) conforms to the same occupancy limitations as in Alternative 4, or b.) provides a minimum of 50% of the housing units at a level that is affordable to households with incomes at 90% of the Area Median Income (AMI) or below. The intent to make housing available to workers close to jobs is carried through all the alternatives, but the alternatives evaluate different variations of the development standards, which would result in slightly different quantities or types of homes.

4.2.4 Light Rail Coordination

Comment Theme: EIS doesn't adequately incorporate Link light rail analysis or results of the Sound Transit West Seattle and Ballard Link Extension (WSBLE) Draft EIS.

Response: The EIS does consider planning for the WSBLE light rail. The addition of new light rail service in WSBLE is one of the drivers of the proposed action as indicated in Objective F, and further described in **Section 2.2.1 Emerging Factors Affecting Seattle's MICs**. The locations and patterns of proposed zoning changes under the alternatives directly respond to the potential WSBLE station locations. The transportation analysis includes the effects of the future light rail on the transportation system. The WSBLE Draft EIS was released on January 28, 2022, and its analysis and information is considered in preparation of this Final EIS.

Comments were received from community members in and around the Georgetown neighborhood during an extended comment period provided to those community members. Several comment themes from those letters are addressed here and cross-referenced below.

4.2.5 Georgetown Arts & Culture

Comment Theme: The EIS should include more analysis of impacts on arts and cultural spaces, facilities, and communities in the Georgetown area. Concern that the alternatives studied threaten affordable arts and performance spaces.

Response: The retention and preservation of arts communities and resources in Georgetown is important to the City and has been elaborated further in the Preferred Alternative and this Final EIS. Additional description of these valuable communities is added in the existing land uses description. Potential impacts on arts communities from the Alternatives is included in discussion of potential displacement impacts in the Section 3.9 Housing and Section 3.8 Land & Shoreline Use. Additional detail on potential development standards to encourage retention and preservation of arts spaces is included in the Final EIS, as a component of the Mixed Use zone in the Preferred Alternative. The Mixed Use zone would include specific measures to incentivize the creation and/or preservation of arts space. Details of such development standards could continue to be refined with participation by community before any zoning changes are made. Non-land use actions to support arts and culture in Georgetown are ongoing outside of the proposals studied in this EIS. These efforts include City funding support for the Mini Mart City Park arts-centered community center, and City and State support for the authorization of tax-exempt revenue bonds for Equinox Studios / the Georgetown Community Development Authority to finance nonprofit facilities for the benefit of local artists and artisans, which may include housing, and funding to the Georgetown Merchants Association through the Economic Recovery Fund for activities including promotion and marketing.

4.2.6 Georgetown Buffer Areas

Comment Theme: Alternatives should include proposals to provide larger buffer areas between residential and mixed use areas of Georgetown, and heavier industrial areas. Accomplish this by studying conversion of more and larger geographic areas from MML zoning to the UI zone, or a Commercial 2 zone or mixed use zone.

Response: EIS alternatives study a range of potential geographic patterns of proposed zoning changes. In response to this comment, the Preferred Alternative includes a larger area of UI zoning in Georgetown compared to the Draft EIS alternatives. A new stretch of UI zoning is included for land in the vicinity of S Orcas Street to the northeast of 7th Avenue S and the existing Commercially zoned area on 4th Avenue S. The effect of this addition is to create a larger contiguous area of UI, Mixed Use, and residential zoning comprising a cohesive Georgetown neighborhood area. The Preferred Alternative also includes conversion of land from industrial zoning to a Mixed Use zone in a larger geography than Draft EIS alternatives. The City acknowledges that some Georgetown residential community members comments request even larger areas to be taken out of the MML zone. Alternatives are crafted in consideration of the proposal's objectives.

4.2.7 Commitment to Mitigation Measures

Comment Theme: Mitigation measures are only suggested. The City should provide firm commitments to enacting mitigation measures potentially including legislation to memorialize commitments to mitigation measures.

Response: The City and its consultant team prepared the non-project EIS in accordance with SEPA laws. The EIS identifies possible mitigation measures that will reduce or eliminate adverse environmental impacts of a proposal. These are detailed by topic in Chapter 3, and the other potential mitigation measures in addition to plans and codes are listed in Appendix J EIS Mitigation Measures List. Many mitigation measures respond to the evaluation of the proposal and are presented to decisionmakers (Mayor and Council) as optional actions they could choose to take in conjunction with the proposed action, in light of the impacts that are disclosed in the EIS. Decisionmakers determine what combination of the action (if any) and mitigation measures they wish to take. An action may still be taken even if there are impacts if decisionmakers believe it would be in the public interest after considering all the information. For some mitigation measures it is uncertain whether the measures are technically feasible or economically practical. The EIS only looks at mitigation measures that address impacts attributable to the adverse impacts of this proposal. It is an option for the Mayor and/or City Council to make a statement of commitment to certain mitigation measures in conjunction with their decision to implement land use policy or code changes. This could come in forms including but not limited to a Resolution or City budget line item. OPCD intends to continue working with community and decisionmakers after the Final EIS to explore potential commitments to mitigation measures.

4.2.8 Community Engagement

Comment Theme: Concern that the EIS process did not include enough community engagement, only engaged certain stakeholders, or did not adequately engage historically disproportionately impacted communities.

Response: The City extended the Draft EIS comment period specifically for Georgetown and South park residents an additional 45 days and held numerous in-person and online meetings to speak with those community members. The Draft EIS comment period had already been extended to 75 days prior to the special extension for Georgetown and South Park residents. The EIS process adhered to all of the SEPA required notice and comment period requirements, including required public hearings. The City translated Draft EIS executive summary material into Spanish, Vietnamese, Somali and Chinese. Interpretation was provided at community meetings in Georgetown and South Park. The EIS process is not the only opportunity for community engagement. The City will continue to work with communities after the Final EIS to engage them during the legislative phase to provide further input on possible policy or zoning changes before they are made. Community engagement separate from this land use proposal is also ongoing as part of the Duwamish Valley Action Plan and other project-specific efforts.

4.2.9 Consider Strategy & Comprehensive Plan

Comment Theme: Suggestion that the Industrial and Maritime Strategy should be addressed as a part of the Comprehensive Plan 10-year major update, and as an independent action or study.

Response: The contents of this proposal are being closely coordinated with the Comprehensive Plan major update. Land use concepts and zones proposed and studied in this EIS, and growth estimates in this EIS will be incorporated into the public engagement and proposed alternatives of the One Seattle Comprehensive Plan major update. The community will have additional opportunities to provide input on the City's overall growth strategy as part of the Comprehensive Plan major update. The City considers the Industrial and Maritime Strategy to be a distinct subject area worthy of a topic-specific study and land use policy proposals because there are unique attributes and issues related to industrial lands and designated Manufacturing and Industrial Centers.

4.2.10 Housing Instead of Industrial Uses

Comment Theme: There is a need for more housing in the City, and the City should allow more housing in the MIC study areas.

Response: The purpose of Manufacturing Industrial Centers (MICs) is to protect industrial uses that offer jobs important to the local and regional economy and that offer wages that can support households and are accessible to those without higher education. The Puget Sound Regional Council (PSRC) has designated the MICs and has adopted policies and requirements the City needs to honor in subarea planning. See **Section 3.8** of the EIS. These policies include limitations on residential uses:

- Establish strategies to avoid land uses that are incompatible with manufacturing/industrial uses, such as large retail uses, residential uses, or non-related office uses (other than as an accessory use).
- The MIC study areas contain little housing. The Draft EIS Action Alternatives consider industry-supported housing in the form of caretakers and artists/live-work housing, as well as small, targeted adjustments to the Greater Duwamish MIC in the Georgetown and South Park areas to address community needs. For the Preferred Alternative, two new areas outside the MICs in west Ballard and Judkins Park would be converted to mixed use zoning allowing housing, in addition to the proposed mixed-use areas in Georgetown and South Park studied in Draft EIS alternatives. Overall, a higher total amount of housing production outside of MICs would result compared to Draft EIS alternatives—an additional 1,534 dwellings, 42% more than alternatives 3 and 4. In total, nearly 3,000 new homes are projected to be generated under the Preferred Alternative compared to the No Action Alternative.

The EIS also offers mitigation measures to apply the MHA program to the new II zone to help address demand for housing and this action would generate funding from new development of commercial space in industrial zones for affordable housing.

4.2.11 Modify MIC Boundaries or Uses

Comment Theme: Allow for changes to the MIC Boundaries.

Response: MICs are drawn to identify areas where manufacturing and industrial uses predominate and to protect such uses for the long-term as unique job centers, consistent with VISION 2050 and the regional growth strategy. Recognizing the importance of the industrial and maritime jobs in the MICs, the EIS action alternatives propose a policy to consider MIC boundary adjustments comprehensively. The proposed policy included in EIS **Appendix C** is:

LU 10.3 Ensure predictability and permanence for industrial activities in industrial areas by limiting changes in industrial land use designation. There should be no reclassification of industrial land to a non-industrial land use category or amendments to the boundaries of manufacturing industrial centers except as part of a City-initiated comprehensive study and review of industrial land use policies or as part of a major update to the Comprehensive Plan.

The Action Alternatives consider altering about 22 acres and the Preferred Alternative about 53 acres to non-MIC status in the Georgetown and South Park areas to address compatibility and local community needs. Out of 6,936 acres in the study area this is less than 1%. The small amount of MIC boundary adjustments and the proposed policy to limit removal of land from MICs respond to the objectives of the proposal (**Exhibit 2.1-2**).

4.3 Individual Responses to Comments

The issues raised in each comment letter and at the public hearings are numbered on the letters and hearing transcripts in **Section 4.4**. Responses to individual comments are detailed below in **Exhibit 4.3-1** and **Exhibit 4.3-2**.

4.3.1 Written Comments & Responses

Exhibit 4.3-1	Written	Comments	and	Responses
---------------	---------	----------	-----	-----------

Number	Comment Summary	Response
1	Hansen	The Duwamish Tribe
1-1	Appreciate inclusion and would like to explore a community benefits agreement.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
1-2	Support the UI strategy to ensure stronger land use protections for core industrial and maritime areas, high density industrial near public transit, and affordable small-scale light industrial businesses in southwest Seattle.	The comment is noted and forwarded to City decision makers.
1-3	Recommend expanding the current footprint of the designated Urban Villages beyond the 6% allotment to include Duwamish Greenbelt properties adjacent or contiguous to the Duwamish Longhouse and Cultural Center, Ha Ah Poos Park, and Herring House. Allow for natural buffer	The comment is noted and forwarded to City decision makers. This proposal could be considered in the Comprehensive Plan Update as it addresses land use designations largely outside the MIC.
1-4	Request access and notification of any earthwork or groundwork performed in the SODO Stadium and Georgetown subareas. Request any evidence or artifacts gathered to be presented and turned over to the Duwamish Tribe.	All cultural resources survey and archaeological work will follow best practices and standard archaeological techniques in the discovery and preservation of cultural and historical artifacts. See revisions made to Section 3.11 Historic, Archaeological, & Cultural Resources that clarify this approach.
1-5	Request that any historic, archaeological, or historic resource uncovered during groundwork be preserved and presented and turned over to the Duwamish Tribe.	The City of Seattle Shoreline Master Program Regulations are referenced in Seattle Municipal Code section 23.60A. 23.60A.154 outlines the standards for archaeological and historic resources which are in line with the request by the Duwamish Tribe. Specifically, 23.60A.154.C requires:
		If any archaeological resources are uncovered during the proposed work, work shall be stopped immediately, and the applicant shall notify the City, affected tribes, and the Washington State Department of Archeology and Historic Preservation. The applicant shall submit a site inspection and evaluation report by a qualified professional archaeologist, approved by the City, that identifies all possible valuable archaeological data and makes recommendations on how to handle the data properly. When the report is prepared, the applicant shall notify affected tribes and the Washington State Department of Archaeology and Historic Preservation and provide them with copies of the report.

Number	Comment Summary	Response
		Also see language added to Sections 3.1.3 Solls/Geology, 3.5.3 Contamination, and 3.11.3 Historic, Archaeological, & Cultural Resources under Other Potential Mitigation Measures to address providing access and notification of earthwork to appropriate stakeholders.
1-6	EIS should address remediation of existing contaminated properties, especially in the context of equity and environmental justice.	Clean-up at existing contaminated properties is ongoing such as the Lower Duwamish Waterway Superfund site by the Lower Duwamish Waterway Group. The Duwamish River Community Coalition (DRCC) was established in 2001 to help monitor cleanup of the river as described in Section 3.9.1 .
		For contaminated sites with current industrial land use designations that maintain an industrial focus under new land use designations, cleanup will not likely happen until redevelopment occurs, or there is a property sale that triggers site characterization and remediation activities in order to secure project financing. This is clarified in Section 3.5 of the EIS.
		Site contamination and remediation are addressed at the time of development or redevelopment through existing processes under MTCA. SEPA documentation submitted with project applications require disclosure of known or suspected contamination of soil, soil vapor, groundwater, or other media, and lenders require Phase I and/or Phase II Environmental Site Assessments be completed before they will provide project funding.
1-7	Sound Transit's light rail proposals in West Seattle would result in permanent loss of Duwamish Greenbelt, an environmental asset and equity issue for the Tribe and Pigeon Point community. Request for Sound Transit to examine other light rail routes to avoid loss of greenspace and to consider the equity impacts.	The comment is noted and forwarded to City decision makers. The Sound Transit EIS is a different proposal from the Industrial Maritime Strategy. City staff are coordinating information and data from Sound Transit to the greatest extent possible. Sound Transit's West Seattle and Ballard Link Extension Draft EIS proposes three Duwamish Segment Alternatives—the North Crossing Alternative (DUW-2) would avoid any construction impacts to the West Duwamish Greenway. Sound Transit will mitigate impacts to greenspaces according to applicable regulations and permit conditions. The alternatives in the Industrial and Maritime EIS do not directly affect the greenspaces.
2	Cotten	Washington State Department of Transportation
2-1	Appreciate the opportunity to comment on the EIS and the positive and collaborative relationship between WSDOT and the City.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
2-2	Draft EIS uses old traffic data. WSDOT requests that impact of SR 99 tolling and Alaska Way Tunnel be reflected in trip pattern analysis.	The commenter correctly notes that the citywide travel time data available for this project includes a period before tolling began in the SR 99 tunnel. While the existing conditions patterns may differ to a degree, all future year analyses assume SR 99 tolling is in place. Because the traffic patterns used to evaluate impacts reflect post- toll patterns, the comparisons among the future year alternatives adequately represent the relative differences among the alternatives.
2-3	WSDOT requests additional specific measures to manage travel demand and supplement transit under alternatives 3 and 4.	The EIS includes a Mitigation Measures section including potential travel demand management (TDM) strategies for the study area. Among those strategies are last-mile shuttle systems between key transit nodes and the MICs; coordination with King County Metro and/or Sound Transit to provide off-peak transit service tailored to shift workers with irregular hours; subsidized vanpools; rideshare matching to limit the number of drive-alone commute trips; and

Number	Comment Summary	Response
		micromobility options such as scooters or dicycles to make last- mile connections.
2-4	 (1) WSDOT should be engaged for mitigation strategies for impacts to I-5 under alternatives 3 and 4. (2) SR 99, 509, and 599 should be investigated for transportation impacts. For vehicular trip thresholds, please see WSDOT Design Manual Chapter 1130.09(2)(a). 	(1) The City is committed to working with WSDOT through a variety of means, including the I-5 System Partnership, to consider the future needs for this critical regional corridor. The Preferred Alternative proposes growth more similar to Alternative 2 and would not exceed the City's non-project threshold of impacts like alternatives 3 and 4.
		(2) In addition to the projected conditions on SR 99 through the study areas, information related to projected conditions on SR 509 and SR 599 has been included in the Final EIS. The Preferred Alternative proposes growth more similar to Alternative 2 and would not exceed the City's non-project threshold of impacts like alternatives 3 and 4. With respect to the vehicular trip thresholds cited, the City is committed to working with WSDOT to determine how that impact threshold may be incorporated into the individual project review process to determine impacts to the state highway system of any specific development proposal.
2-5	All alternatives would rezone Corson Facility (6413 Corson Ave S) to UI. Concerned about facility becoming nonconforming, and potential for residential development to encroach into industrial lands and inhibit its functionality. Request for an evaluation of micro-mobility and freight conflicts.	Nonconforming uses are permitted to continue subject to provisions of the Seattle Land Use Regulations (SMC Subtitle III). See Response to Common Comment Theme at Section 4.2.2 . Additional clarification were added to Sections 3.2.2 and 3.6.2 to indicate that areas of Georgetown, around or near the Corson facility, slated for additional growth in housing of all types may be subject to increased air and noise impacts from vehicle traffic and industrial sources
		The EIS includes discussion of potential modal conflicts between freight trucks and vulnerable users including people walking, biking, and using micromobility modes such as scooters.
2-6	Appreciate the opportunity to comment on the EIS and the positive and collaborative relationship between WSDOT and the City.	The comment is noted and forwarded to City decisionmakers.
3	Curtis	Washington State Major League Baseball Stadium Public Facilities District (PFD), Washington State Public Stadium Authority (PSA)
3-1	Appreciate the opportunity to comment on the EIS. Summary of comments below.	Thank you for your letter. Comment is noted. See the response to comments 3-2 through 3-23 below.
3-2	Request that the Stadium Transition Area Overlay District (STAOD) be removed from the Duwamish MIC, and for the Final EIS to study the impacts of removing the STAOD from the Duwamish MIC as a separate alternative.	The Preferred Alternative includes expanded flexibilities to address unique conditions of the stadium area through the Stadium Transition Area Overlay District. More information on these flexibilities is provided in the development standards appendix. The City's proposed action intentionally limits removal of land from MICs to focused locations in the Georgetown and South Park neighborhoods.
3-3	Final EIS should analyze impacts in the STAOD (distinct from the balance of SODO) related to transportation, housing, and land use.	The STAOD is part of the evaluation of the MIC in transportation, housing, and land use. This EIS provides a non-project level of detail that is areawide, consistent with WAC 197-11-442.
		Alternatives' effects on transportation corridors in and near the STAOD are included; and the area is referenced in the land use evaluation and included on maps. The STAOD boundaries are added to the Preferred Alternative map to assist in viewing that portion of the study area. The industry-supportive housing that

Number	Comment Summary	Response
		would be located in the STAOD is evaluated for each alternative including effects on transportation, housing, and land use.
3-4	Final EIS should analyze the traffic impacts from the office and commercial uses under the No Action Alternative in comparison to other alternatives.	The land use of each alternative is compared in the EIS and included in transportation modeling and results on transportation networks inside and outside the STAOD are addressed at a non- project level of detail.
3-5	Final EIS should address transportation impacts from residential uses being allowed in the new UI Zone.	The land use of each alternative is compared in the EIS including effects on transportation networks inside and outside the STAOD at an areawide level of detail consistent with a non-project EIS. This includes considering the effect of housing. In general, where housing is part of mixed uses in proximity to jobs and other destinations there are fewer trips.
3-6	Final EIS should evaluate the marginal impact of adding residential uses to the STAOD, compared to office/commercial under the No Action Alternative.	See alternatives 3 and 4 and the Preferred Alternative that add more residential into the Greater Duwamish MIC with including in the STAOD.
3-7	Final EIS should evaluate an alternative that includes more housing near the stadiums.	Consistent with the PSRC criteria for designating MICs to focus industrial uses in the MIC, the EIS does not study allowing residential uses in the majority of the study area. Alternatives 3 and 4 and the Preferred Alternative consider limited additional flexibility of existing allowances for caretakers' units and artist/studio quarters in the proposed UI zone, and the Preferred Alternative allows some housing in the UI zone as a conditional use.
3-8	Final EIS should include an alternative that eliminates the concept of limiting occupancy of housing and instead concentrates housing in the STAOD. Consider Heartland's analysis of the economic feasibility of the City's proposed development prototypes.	The comment is noted and forwarded to City decision makers. Consistent with the PSRC criteria for designating MICs to focus industrial uses in the MIC, the EIS does not study allowing residential uses in the majority of the study area. Alternatives 3 and 4 consider limited additional flexibility of existing allowances for caretakers' units and artist/studio quarters in the proposed UI zone only. Within the UI zone, the Preferred Alternative would allow some expanded housing as a conditional use, and would present the option of providing housing at an affordability level instead of limiting the occupancy.
		SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450). Separate from the EIS, the City will consider economic feasibility; see Section 4.2.1 .
3-9	The City should eliminate the concept of tenant restrictions for housing and should analyze the impacts of a reasonable alternative in the Final EIS.	See response to comment 3-8.
3-10	City should increase the housing density limits in the STAOD to 200-220 DU/Acre and or a FAR of 4.25-4.75, and the results of that should be evaluated in the Final EIS.	Comment is noted and forwarded to City decision makers. See response to comment 3-8.
3-11	Final EIS should evaluate an alternative that applies the MFTE program to the STAOD.	Comment is noted and forwarded to City decision makers. Application of the MFTE program is noted in the development standards Appendix G regarding the Preferred Alternative.
3-12	Final EIS should evaluate an alternative that allows for 5 floors of residential in a mid-rise development, with the first 25' reserved for	See response to comment 3-8.
Number	Comment Summary	Response
--------	--	--
	industriai/maker spaces (as studied by Heartland).	
3-13	Land cost, commercial and residential rental assumptions, capitalization rates, financing considerations, parking, construction costs, and timing considerations should be corrected in the Final EIS.	SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450). The data points suggested in the comment are not a part of the EIS. Separate from the EIS, the City will consider economic feasibility; see Section 4.2.1 .
3-14	Request for the City to study an alternative that eliminates the citywide limit on residential units in industrial lands.	The comment is noted and forwarded to City decision makers. Consistent with the PSRC criteria for designating MICs to focus industrial uses in the MIC, the EIS does not study allowing significantly expanded residential uses in the majority of the study area. See response to comment 3-8. For the Preferred Alternative, two new areas outside the MICs in west Ballard and Judkins Park would be converted to mixed use zoning allowing housing, in addition to the proposed mixed-use areas in Georgetown and South Park. Overall, a total amount of housing production under alternatives 3, 4 and the Preferred Alternative would exceed housing production under No Action, but much of it would be outside of MICs, or in targeted locations selected to reduce conflicts.
3-15	Final EIS should evaluate an alternative where the height limit is capped at 85' (instead of 75') in the proposed UI Zone.	The proposed UI zone has height limit tiers of 45', 60' and 75'; however, under the Preferred Alternative the UI zone in the STAOD would be increased to 85'. See development standards Appendix G .
3-16	The 500' residential restriction from railroads should be lifted in the STAOD.	A 500' buffer requirement to housing from railroads would not apply in the STAOD under the Preferred Alternative. See development standards Appendix G .
3-17	Specific Comprehensive Plan goals should be developed for the STAOD if the STAOD is not removed from the Duwamish MIC.	The comment is noted. The studied alternatives retain the STAOD in the MIC. Policy intent can be clarified in the municipal code regarding the STAOD.
3-18	Final EIS should delineate what industrial/manufacturing/maritime uses are being referred to in the STAOD.	The expanded development standards Appendix G includes a table clarifying which specific land uses would qualify as industrial in the proposed zones.
3-19	Final EIS should explain its rationale for applying the II zone to WOSCA.	The alternatives in the EIS consider a range of different future zoning designations for the WOSCA site. A proposed comprehensive plan policy calls for site specific master planning of the site before its reuse.
3-20	Final EIS should demonstrate how its proposals and alternatives will impact the development around stadium station.	The EIS considers the programmatic level environmental impacts in all portions of the study area. Sub area analysis and localized potential impacts are discussed to practical and feasible extents; non-project EISs are areawide and not site specific in nature per WAC 197-11-442.
3-21	City should clarify how the requirement for new housing's connection to industrial activity is a mitigation measure for noise and other impacts in alternatives 3 and 4.	The occupancy limitation of housing to industry-supportive housing under alternatives 3 and 4 would mitigate impacts because occupants of the housing would be more accustomed to an industrial context than the general residential population. This mitigates the potential for noise complaints and other compatibility impacts because the residents would be less likely to levy complaints against nearby industrial operations, lessening the potential for disruption to industrial activities. With new housing occupants having a connection to industrial activity in the area, reduced traffic and VMT would also reduce noise associated with

Number	Comment Summary	Response
		venicies and associated impacts. Similar mitigation would result for housing provided under the affordable workforce option (Preferred Alternative) because occupants of the affordable housing would be more likely to be people holding nearby jobs in MICs.
3-22	Final EIS should explain how new open and green space would be funded/delivered in alternatives 3 and 4.	SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450); see Section 4.2.1 . The City's Parks, Recreation and Open Space Plan outlines the City's existing open space and recreational facilities, capital funding, and projects being funded and a vision for the future.
3-23	Final EIS should clarify how the new subarea plans would be conducted and what that process would entail.	The City will be working with stakeholders in the Duwamish MIC and the BINMIC to update their respective Centers Plans in 2023 and 2024.
4	Felleman	Port of Seattle The Northwest Seaport Alliance
4-1	Support Alternative 2. Reviewed the Draft EIS under the goal to support the next generation of industrial and maritime jobs and centered feedback on that principle (as opposed to trying to address the regional housing affordability crisis). Comments will be augmented by a technical appendix with comments from subject matter experts across the Port and NWSA.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
4-2	The equity of economic opportunity afforded by the industrial and maritime employment mix is the central reason to preserve industrial lands.	The comment is noted and is acknowledged in Section 1.3.1 and Section 2.2.1 . The EIS recognizes equity and accessibility as one of the six key emerging factors affecting Seattle's MICs, specifically access to maritime and other industrial career opportunities for BIPOC and women. Maintaining a strong industrial economy is a prerequisite to providing these opportunities, but other strategies including outreach to BIPOC youth and workforce training investments are key parts of the Industry and Maritime Strategy.
4-3	Port of Seattle has devoted substantial time and investment to addressing environmental justice issues in near-port communities. Concerned that alternatives 3 and 4 do not adequately mitigate the impacts of locating new residential communities near the industrially-zoned MICs.	Consistent with the PSRC criteria for designating MICs to focus industrial uses in the MIC, the EIS Alternatives do not study allowing significantly expanded residential uses in the majority of the study area. For alternatives, 3, 4 and the Preferred Alternative, overall, a higher total amount of housing production would result compared to No Action, and much of it would be outside of MICs, or in targeted locations selected to reduce conflicts. In addition to limiting the amount and location of housing and updating zoning standards per Land Use Mitigation Measures in Section 3.8.3 , other measures addressing compatibility include those designed to address noise and air pollution. See Sections 3.2.3 and 3.6.3 .
4-4	Request additional detail on the cost of mitigation requirements in alternatives 3 and 4.	SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450). See response to comment themes at Section 4.2.1 .
4-5	Freight routes and regional transportation planning must be a central focus (regional importance of freight mobility).	The Final EIS has been reorganized to include freight as a separate section together with additional information. It should also be noted that study corridors were selected based on the City's Major Truck Streets designation and include most Major Truck Streets within each study area. While freight mobility is critical in these areas, the EIS is intended to address all travel modes.

Number	Comment Summary	Response
		A section on the PSRC Regional Centers Framework and Plan Review Manual has been added to the Current Policy & Regulatory Frameworks portion of Section 3.10 Transportation .
4-6	Would like to see the Final EIS acknowledge that industrial workers' needs for affordable housing does not mean that housing should be within or adjacent to the MICs.	Comment is noted. Section 3.9 Housing discusses the balance of housing and jobs. The EIS discusses the impacts and effects of including housing in MICs in differing quantities and patterns, including no expansion of housing allowances, under the different alternatives. The EIS Alternatives focus on industrial uses and would not significantly expand residential uses in the majority of the study area. Caretakers' quarters and artist/live-work units are meant to provide industry supportive housing. Some alternatives would adjust some locations for mixed use housing that would either be outside of MICs, or in targeted locations selected to reduce conflicts.
4-7	Concerned the EIS is not easily accessible to the public and relies on technical jargon.	Comment is noted. The City prepared executive summaries and a story map in addition to EIS documents and provided presentations and meetings to interested groups. Translations of summaries were provided.
4-8	Please replace the photo on page 1-10 with an image of Seattle container operations (photo is of the East Blair Roll on Roll Off terminal at the Port of Tacoma).	The photo was replaced in this Final EIS with an image of Seattle container operations.
4-9	Recommend the EIS consider alignment with Port/NWSA development plans.	The EIS includes alignment with Port/NWSA development plans to the extent practical and feasible.
4-10	Recommend including data that confirms industrial workforce demand for housing in the MICs. Please replace the photo on page 1-17 with an image of Seattle container operations (photo is of the Husky Container Terminal at the Port of Tacoma).	Comment is noted. Currently housing is not a permitted use in the MICs and the City does not have data demonstrating the level of demand for workforce housing in the MICs. See response to comment 4-6 regarding the limited housing in the EIS alternatives. The photo was replaced in this Final EIS with an image of Seattle
		container operations.
4-11	Recommend including description of allowable development types under current zoning in Exhibit 1.5-5.	See adjustments in Section 1.5.5 to add allowable development types. See also Industrial Development Regulations appendix that summarizes development allowed under existing zoning designations.
4-12	Recommend including description of allowable development types under current zoning in Section 1.5.5.	See response to comment 4-11. The photo was replaced in this Final EIS with an image of workers at Seattle facilities.
	Please replace the photo on page 1-21 with an image of workers at Seattle facilities (photo is of the Pierce County Terminal at the Port of Tacoma).	
4-13	Inclusion of Northwest Ports Clean Air Strategy and distinction between those investments and mitigation for the revised zoning as presented in the Plan	Additional language has been added to Section 1.7.2 to add detail on the Northwest Ports Clean Air Strategy, and to distinguish between mitigation currently planned by the Port of Seattle for Port operations from mitigation proposed for implementation of the Industrial and Maritime Strategy.
4-14	Support Alternative 2.	Comment is noted and forwarded to City decision makers.

Number	Comment Summary	Response
4-15	Amend Draft EIS to reflect that Terminal 46 is a facility licensed to the NWSA for container cargo use and will not become a cruise terminal. Specific text edits recommended.	See revisions made to text in Section 2.2.1 (page 2-8) to reflect that a prior effort to convert the terminal to a cruise terminal has been abandoned.
4-16	Light rail/HCT: Ensure the Final EIS provides contingency concepts in the case of delayed development past 2037-39 of the Interbay and Ballard stations, and potential for alternative station sites, and that the Final EIS considers a similar analysis in the SODO area in case an interim terminus occurs at SODO/Lander.	The City is considering the most updated and recent information provided regarding light rail timing and planning as provided by Sound Transit. The City periodically reviews its growth and policies and adjustments are possible when appropriate.
4-17	Support the MML and II land use concepts. Recommend acknowledgment of incompatible uses in the UI zone.	Comment is noted and forwarded to City decision makers. The land use chapter evaluates potential land use compatibility impacts under all alternatives.
4-18	Request the inclusion of cost estimates for each mitigation measure identified in Chapter 3 and dispute the inclusion of mitigation measures that are already underway. Request expanded truck charging facilities (in Section 3.2 Air Quality & GHG) not be considered a mitigation measure under alternatives 3 and 4.	SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450). See response to comment themes Section 4.2.1 . Per SEPA, mitigation measures may include measures that could be implemented or might be required as well as those the City is committed to implement. If the technical feasibility or economic practicality of mitigation measures is uncertain, the mitigation measure may still be discussed but discussion of the uncertainties should be included. See WAC 197-11-440 and WAC 197-11-660. Additional language has been added to Section 3.2.1 to identify mitigation currently planned by the Port of Seattle for Port operations as not mitigation proposed for implementation of the Industrial and Maritime Strategy. Specific language was also deleted in Section 3.2.3 to add similar clarity.
4-19	Summary of air quality monitoring; description of current drayage trucking and conversion to newer truck models; and Northwest Seaport Alliance (NWSA) participation in the Northwest Ports Clean Air Strategy.	Exhibit 3.2-6 provides a summary of the detected pollutants and measured concentration levels for the eight sites monitored in 2021 for this EIS. Appendix H presents the Technical Memo, Summary of Air Quality and Noise Monitoring Results at 8 Locations Within The City of Seattle. The document presents a summary of air constituents monitored at those sites and what methodology was used in that monitoring. Text has been changed in Air Quality & GHG Section 3.2.1 to reflect drayage trucking as an unknown portion of all trucking in the MICs; added data and reporting from the NWSA's Clean Truck Program; and added clarifying language to indicate that the NWSA is a member of the Northwest Ports Clean Air Strategy.
4-20	Incorrect reference to Terminal 15; inclusion of funding along with policy guidance for industrial and maritime electrification	Draft EIS Section 3.2.3 reference to Terminal 15 was corrected to Terminal 5. Additional text has been added to Section 3.2.3 to include funding as a possible addition to the Seattle Comprehensive Plan and MIC Subarea Plans to support electrification of industrial and maritime activities. See Section 3.2 Air Quality & GHG .
4-21	Water Resources Impacts: Port/NWSA staff agree with the analysis; because Seattle's updated stormwater code will drive improvements to stormwater quality under all alternatives that involve re-development.	Comment is noted.

Number	Comment Summary	Response
4-22	Discussion should state that the City will adhere to PSRC's methodology for distinguishing industrial from non-industrial jobs.	Industrial employment estimates are based on the 2019 share of industrial employment by sector based on the 2015 PSRC Industrial Lands Study NAICs-based definition of industrial activities. This uses classification of what counts as an industrial job are consistent with PSRC criteria, including jobs in Information Computer Technology (ICT). Projections show strong job growth in ICT under the Action Alternatives. Consistency with PSRC classifications is appropriate given the need to fit VISION 2050 and Regional Centers Framework. A more conservative classification of which jobs are industrial, especially in ICT would show a steeper decline in the percent of industrial jobs under most studied alternatives. See footnote in Section 2.4.8 of the Final EIS.
4-23	Request that the EIS include a report on how much more housing can be put into the residential zones before it contemplates adding housing to industrial zones.	The comments are noted and forwarded to City decision makers. The 2021 King County Urban Growth Capacity Report determines the amount of land suitable for urban development, and evaluates the capacity for growth, based upon measurement of recent actual development activity. The City of Seattle's residential land supply and capacity is quantified on page 104 of that report. ²⁹
4-24	 (1) Final EIS transportation section should be substantially reworked to prioritize freight mobility consistent with the goals of the EIS. (2) Draft EIS should acknowledge role of essential facilities in the transportation system of the Duwamish MIC (marine containers, breakbulk facilities, and major rail yards for freight transfer). (3) Draft EIS should provide information on the impacts of alternatives on the freight system. The analysis should include Major Truck Streets, NHS Freight Intermodal Connectors, Critical Urban Freight Network, and Heavy Haul Network. Draft EIS should include map of City's truck network from City's Freight Master Plan. (4) Draft EIS should reflect City's Complete Streets Ordinance that prioritizes freight on the Major Truck Streets. (5) Final EIS should align with requirements for the MICs Subarea Plans to "Prioritize transportation projects that provide access to freight intermodal facilities to optimize freight movement for local, regional, and national distribution (including rail, trucking facilities, or waterways, as appropriate)." 	 (1) The Final EIS has been reorganized to include freight as a separate section together with additional information. It should also be noted that study corridors were selected based on the City's Major Truck Streets designation and include most Major Truck Streets within each study area. While freight mobility is critical in these areas, the EIS is intended to address all travel modes. (2) The recommended text has been added to the Primary and Secondary Study Areas introduction. (3) Draft EIS Exhibit 3.10-13 Existing Freight Network provides additional data displaying more elements of the freight system and the Final EIS has been reorganized to include freight as a separate section. (4) The recommended text has been added to the Complete Streets Ordinance summary in the Current Policy & Regulatory Frameworks portion of Section 3.10 Transportation. (5) The EIS evaluates the proposals and alternatives regarding regional plans and policies in Section 3.8 Land & Shoreline Use. The City will prepare a subarea plan for each Manufacturing Industrial Center and will address Centers requirements and the PSRC regional growth strategy.
4-25	 (1) Final EIS transportation section should reference the Seattle Industrial Areas Freight Access Project, including the provided data and strategies. (2) Unclear whether maps (which are missing the legend for the corridors under analysis) cover all major freight corridors. WSDOT's FGTS 	 Information from the Seattle Industrial Areas Freight Access Project has been added to the Freight section of the Final EIS. The commenter states that maps are missing legends, but no exhibit number is cited. Study corridor Exhibit 3.10-3 and Exhibit 3.10-4 include legends in the Draft and Final EIS; text descriptions of the study corridors are included in Draft EIS Exhibit 3.10-14 (Final EIS Exhibit 3.10-8), as well as the travel time results table for each

²⁹ See King County Urban Growth Capacity Report, 2021, available: <u>https://kingcounty.gov/~/media/depts/executive/performance-strategy-budget/regional-planning/UGC/KC-UGC-Final-Report-2021-Ratified.ashx?la=en</u>.

Number	Comment Summary	Response
	 data should be reviewed to identify these corridors. (3) Final EIS should include analysis of travel time reliability for freight. (4) Final EIS should include east-west screenlines in Duwamish MIC. 	alternative in the impacts section. Draft EIS EXHIDIT 3. 10-13 EXISTING Freight Network has been provided with additional data so the reader can compare with the study corridors (see Final EIS Exhibit 3.10-11 through Exhibit 3.10-14). Study corridors were selected based on the City's Major Truck Streets designation and include most Major Truck Streets within each study area. (3) Language has been added to Section 3.10.2 Impacts to note
		the relationship between traffic congestion and reliability, i.e., that increasing traffic congestion results in deteriorating reliability. The importance of this issues for freight operators is also reflected in the Final EIS.
		(4) The studied screenlines are consistent with those designated in the Seattle comprehensive plan and include two screenlines capturing east-west traffic across the Duwamish River: 3.11 Duwamish River—West Seattle Bridge and Spokane Street and 3.12 Duwamish River—1st Avenue S and 16th Avenue S. East of the Duwamish River, the study corridors used for analysis include seven east-west corridors in the Duwamish MIC.
4-26	(1) Add provided paragraph to section on the Freight Master Plan.	(1) The description of the Freight Master Plan has been provided with additional information from the commenter.
	(2) The CIP section should focus on the types of projects that are relevant for access to and	(2) Additional projects from the CIP have been added to the summary section per the commenter's suggestion.
	 movement within the MICs, including all phases of the East Marginal Way Rehabilitation project and West Seattle bridge projects. (3) Final EIS include information from PSRC's Regional Centers Framework and Plan Review Manual that provide detailed guidance on the importance for MIC plan development including prioritizing freight projects. 	 (3) A section on the PSRC Regional Centers Framework and Plan Review Manual has been added to the Current Policy & Regulatory Frameworks portion of Section 3.10 Transportation.
4-27	(1) Existing Conditions section should have a separate freight section presented first.	(1) The Final EIS has been reorganized to include freight as a separate section with additional information.
	(2) Freight Network map should add: presence of major marine and rail intermodal cargo facilities in the Duwamish; hierarchy of truck streets; and information on NHS Freight	(2) Draft EIS Exhibit 3.10-13 Existing Freight Network has been provided with additional data displaying more elements of the freight system. See Final EIS Exhibit 3.10-11 through Exhibit 3.10-14 .
	Intermodal Connectors and Critical Urban Freight Corridors.	(3) While peak hours vary to some degree by individual roadway, the PM peak hour is typically the most congested period. To
	(3) Final EIS should analyze the impacts of different alternatives in the MICS using AM conditions.(4) Expand the identification of truck streets with heavy truck volumes to other corridors.	confirm that approach, an observed travel time dataset for Seattle was reviewed and it was determined that 4:45-5:45pm represented the peak hour of travel in Seattle during the day. Therefore, the PM peak hour as selected for this EIS analysis to be conservative in identifying potential impacts.
	especially Atlantic and SW Spokane St. Maybe add map of the FGTS system. (5) Final EIS should add section on truck parking, both its availability and community impact	As stated in the Draft EIS on page 3-370, the AM peak hour is typically expected to have similar characteristics in the opposite
		direction than those shown for the PM peak hour. Therefore, the same locations that are identified as having traffic congestion during the PM peak hour would likely have traffic congestion during the AM peak hour.
		(4) Exhibit 3.10-13 Existing Freight Network has been provided with additional data so the reader can compare with the study corridors (see Final EIS Exhibit 3.10-11 through Exhibit 3.10-14). Study corridors were selected based on the City's Major Truck Streets designation with the goal of analyzing those facilities most likely to

Number	Comment Summary	Response
		be impacted by the alternatives. The study corridors include most Major Truck Streets within each study area.
		 (5) Language has been added to the Final EIS to more fully address truck parking needs in Sections 3.10.1 Affected Environment, 3.10.2 Impacts, and 3.10.3 Mitigation Measures.
4-28	Final EIS should describe how adding additional residential uses, potentially with commuter trip to the industrial areas, can be accommodated, given the scant resources to increase transit capacity.	The EIS includes an analysis of transit demand relative to capacity for each future year alternative. Based on that analysis, the EIS concludes that one screenline (across 8th Avenue NW east of the BINMIC) would be impacted by Alternative 1 No Action. For the action alternatives, while some routes traveling across the study area screenlines may operate over their crowding threshold for some individual trips, overall planned capacity is expected to adequately accommodate increasing demand relative to Alternative 1 No Action.
4-29	Concerned that the increased safety conflict between trucks and bike/peds will lead away from the City's Vision Zero goal to eliminate traffic fatalities.	The City is committed to ending deaths and serious injuries caused by traffic collisions. This commitment is reflected in the Vision Zero policy which is supported by a variety of strategies as described in the EIS. The EIS includes a Mitigation Measures section dedicated to Pedestrian & Bicycle System Improvements including the City's safety programs. However, the City also acknowledges that significant impacts to active transportation and safety may remain due to the projected increase in people walking and biking in areas with network gaps and the increased potential for vehicle conflicts (particularly trucks) with vulnerable users. While the City can pursue a variety of mitigation measures to improve facilities for people walking and biking and pursue supplemental funding through federal or state programs, it is not expected that all network gaps can be addressed given the number of locations needing improvement and the limited funding available.
4-30	 (1) Concerned about the use of "PSRC's Transit model" for this MIC-focused analysis. How does the PSRC Transit model account for truck trips on the system, and how are they classified? (2) Draft EIS should have a description of the rationale for choosing the criteria and thresholds of significance. (3) Critical freight corridors must be included for the LOS and travel time analysis, and should take into account the volume of freight moving along the corridor. (4) Travel time reliability should be analyzed for freight. (5) The Final EIS should take a similar approach to SDOT's Complete Corridors approach for transit to prioritize major truck streets and adjust the active transportation metric accordingly. 	 (1) The project team used a version of the PSRC regional trip-based travel demand model that was customized for Sound Transit's West Seattle to Ballard Link Extensions (WSBLE) environmental review and documentation. The model estimates the demand for person and freight travel across a range of travel modes: private automobiles, trucks, transit vehicles, walking, and biking. The truck model defines a truck based on relative weight classes and separates medium and heavy trucks based on the definitions used by WSDOT for collecting truck counts: Medium trucks are defined as single unit, six or more tires, two to four axles and 16,000 to 52,000 lbs. gross vehicle weight Heavy trucks are defined as double or triple unit, combinations, five or more axles, and greater than 52,000 lbs. gross vehicle weight (2) The criteria used to evaluate impacts is described in the Data & Methods section including explanations regarding sources such as the Highway Capacity Manual, <i>Seattle 2035</i> Comprehensive Plan, and King County Metro Strategic Plan Service Guidelines. Additional explanation regarding the thresholds of significance used to compare the No Action Alternative and action alternatives has been added to the Thresholds of Significance section of the Final EIS. (3) Exhibit 3.10-13 Existing Freight Network has been provided with additional data so the reader can compare with the study corridors (see Final EIS Exhibit 3.10-11 through Exhibit 3.10-14). Study corridors

Number	Comment Summary	Response
		designation and include most wajor Truck Streets within each study area.
		(4) See response to comment 4-25, part 3.
		(5) The City will apply its Streets Illustrated design manual and Complete Streets approach to corridors in the study area. If new zoning designations are adopted, SDCI will work with SDOT to develop updates to the Streets Illustrated manual reflecting street design standards tailored to the industrial context and level of expected pedestrian and bicycle activity. Updates will consider street typologies and design standards that can accommodate both freight activity and non-motorized uses with a focus on reducing potential conflicts.
4-31	(1) Add separate section on freight impacts to the "impacts common to all alternatives" section. In same section, potentially add opportunities for providing on-street truck	(1) The Final EIS has been reorganized to include freight as a separate section, including in the Impacts Common to All Alternatives section. This section will address on-street truck parking.
	(2) Einal EIS should evaluate degree to which	(2) See response to comment 11-17.
	different alternatives increase the potential for conflict for trucks and non-motorized users, and whether they can be mitigated without negative impacts to freight mobility.	(3) The Draft EIS addresses potential effects on pavement condition on page 3-388 and concludes that the action alternatives may cause some impact though it is not expected to rise to a level of significance in comparison to Alternative 1 No Action.
	(3) Final EIS should describe how the existing poor Pavement Condition Index ratings stemming from a lack of maintenance in light of existing gas taxes and license fees, would impact future development alternatives or be mitigated.	
4-32	For all alternatives:	(1) The Final EIS has been reorganized to include freight as a
	(1) Add a freight impact section to the top of the analysis	separate section, including in the Impacts sections. (2) See response to comment 4-27, part 3.
	(2) Carry out the analysis for the AM peak	(3) See response to comment 4-27, part 4.
	(3) Include all critical truck corridors in the	(4) See response to comment 4-30, part 1.
	analysis	(5) See response to comment 4-25, part 4.
	(4) Incorporate the increase in truck traffic into the analysis	(6) A discussion of potential effects on safety related to at-grade rail crossings has been added to the Final EIS.
	(5) Add east-west screenline in Duwamish	
	(6) Add at-grade rail crossing safety to safety criterion	
4-33	(1) Very concerned about mitigation for I-5 travel time impact and the suggestion to reduce jobs in SODO.	(1) All alternatives increase jobs in SODO including Alternatives 2, 3, and 4 and the Preferred Alternative. However, recognizing impacts of the highest increased job levels under alternatives 3 and 4, the
	 (2) Draft EIS does not account for traffic diversion that occurs on many corridors at LOS F; adding residential traffic to major truck streets does not support a growing industrial area. 	Draft EIS included a mitigation measure to address job levels closer to Alternative 2. Due to factors described in Chapter 2 , the Preferred Alternative features a lower amount of job growth than alternatives 3 and 4. Job growth under the Preferred Alternative is similar to alternatives 2.
	(3) Greater growth in alternatives 3 and 4 causes significant impacts to vehicle movement and travel time. Vehicles, buses, and trucks will get stuck in this congestion.	(2) The PSRC regional travel demand model that was used for this project covers the four-county region (King, Snohomish, Pierce, and Kitsap) and forecasts travel demand throughout the day. Therefore, the model reflects diversion to other facilities or time periods when capacity is reached.

Number	Comment Summary	Response
		(3) The commenter's concerns about the vehicle movement and travel time impact findings of the Draft EIS are noted.
4-34	 Impacts of No Action alternative indicate the need for additional capacity to support the MICs; the City should pay greater attention to the MICs to ensure continued support and economic/job growth goals. Alternative analysis sections should provide more information on active transportation, freight, or safety. 	(1) The commenter's concern about infrastructure needs under the No Action Alternative (i.e., current policies) is noted. SDOT is currently in the process of developing the Seattle Transportation Plan which will integrate the City's modal plans into a comprehensive vision for the citywide transportation network centered around the following values and goals: equity, safety, mobility, sustainability, livability, and excellence.
		(2) Dedicated freight sections have been added to the Final EIS. With respect to active transportation and safety, see response to comment 11-17.
4-35	(1) Add provided introduction to transportation mitigation measures section.	(1) Language has been added to the Mitigation Measures introduction to explicitly acknowledge freight mobility needs in the MICs.
	to collaborate to ensure that public funds are available to mitigate any negative freight	(2) The City is committed to continuing its partnership with the Port of Seattle and NWSA to implement freight mobility improvements.
	impacts, since the development standards in this section provide no direct benefits to freight mobility, parking, or delivery.	(3) The commenter's perspective about the need for an efficient and reliable freight system to support industrial jobs is noted and forwarded to City decision makers.
	(3) Jobs in MIC would not exist without an efficient and reliable freight system.(4) Add a section for freight mitigation in this section.	(4) The Final EIS has been reorganized to include freight as a separate section, including in Section 3.10.3 Mitigation Measures .
4-36	 (1) TSMO section should also include truck-specific notifications for incidents and major points of congestion. (2) Support for rules that tailor TDM requirements to those most effective in industrial settings. (3) Parking policies in the MIC must take the needs of workers, trucks, delivery and service vehicles, and business customers into account. (4) Draft EIS should list potentially significant gaps in ped/bike systems within and providing access to MICs. (5) Large truck parking and curb-side management is needed in parking strategies. (6) Final EIS should add safety subsection to mitigation section. (7) BIA, developer contributions, and TIFs are unlikely to address major transportation system improvement needs, let alone help reduce existing system gaps or maintenance and rehabilitation needs. (8) Proposed widening on Dravus Bridge has not been proposed in any funding planning, and the Ballard and Magnolia Bridges have been studied for years and are still not funded. Concerned about lack of bridge funding. 	 (1) Language has been added to Section 3.10.3 Mitigation Measures to reflect the commenter's suggestion. (2) The commenter's support for TDM requirements tailored to industrial settings is noted. (3) The commenter's perspective on parking policies supportive of membrane truels delivers and having and
		(4) A link has been added to the Final EIS so that readers can explore detailed data in the City's interactive GIS database within
		their areas of interest. (5) The Final EIS provides more language regarding truck parking and curb space management needs
		(6) The Final EIS mitigation measure text is rearranged so that safety is discussed its own section rather than being nested within
		 (7) The funding sources suggested in the Potential Mitigation Measure Funding section are some of the tools the City could pursue. In addition, the City has a biennial budget process through which transportation system improvements, maintenance, and
		rehabilitation needs are considered and funded as feasible.
		(8) The City has a biennial budget process through which transportation system improvements, maintenance, and rehabilitation needs are considered and funded as feasible. In addition to pursuing grant funding sources, the biennial budget is the process through which funding for bridge retrofit and replacement would be identified.
		(9) The commenter's perspective on transportation mitigation fees is noted and forwarded to City decision makers.

Number	Comment Summary	Response
	(9) Transportation mitigation tees will impact the development financial pro formas and risk the ability to fund such development.	
4-37	 (1) ITS and TSMO improvements will be needed on other corridors as well as W Dravus St and I- 5. (2) Draft EIS proposed TSMO, TDM, and ped/bike improvements to offset travel time impact and congestion; concerned that there is no effort to demonstrate how much the impacts can be mitigated, or the cost/funding to complete them. (3) Final EIS should address mitigation for travel time increase on I-5. (4) Value of freight/transit lanes should be analyzed on a case-by-case basis. (5) Gas tax and vehicle license fees have not been effective to date in resolving pavement issues in the Duwamish. Six years ago, the Port and City developed an intergovernmental agreement to fund the Heavy Haul Network for container drayage activity, but the funding identified has yet to be invested in new pavement. 	 (1) Comment noted. The Transportation Systems Management and Operations (TSMO) section within Section 3.10.3 Mitigation Measures lists the types of TSMO measures that could be implemented throughout the study areas. (2) SEPA does not require quantification of the magnitude to which each measure would mitigate impacts. This programmatic EIS addresses area-wide land use zoning changes, rather than a project-specific proposal. The proposal may result in a wide range of individual projects implemented over a long timeframe and across a large geographic area. Because the specific locations and sizes of development are unknown at this time, it would be speculative to identify specific mitigation measures. Individual development projects will undergo separate and more detailed SEPA review during which specific impacts and mitigation will be determined. The City is committed to seeking funding to implement these strategies as needed, but it would be speculative to quantify potential costs at this stage. (3) The City is committed to working with WSDOT through a variety of means, including the I-5 System Partnership, to consider the future needs for this critical regional corridor. Any mitigation measures within the City's sole control. See also response to comment 4-33(1). (4) Section 3.10.3 Mitigation Measures notes the possibility of freight/transit lanes as one potential strategy. SDOT would study any specific projects on a case by case basis to determine their benefit on a particular corridor. (5) The commenter's concerns about funding sources are noted. The City will continue to pursue partnerships to make improvements to the Heavy Haul Network, such as the recent Memorandum of Understanding to contribute funding to the East
		Marginal Way Corridor Improvement Project – North Segment. In addition, the City has a biennial budget process through which transportation system improvements, maintenance, and rehabilitation needs are considered and funded as feasible.
4-38	Final EIS should provide more detail on unavoidable adverse impacts, in particular those that affect freight mobility under alternatives 3 and 4. Scenarios detrimental to supporting maritime and industrial businesses in the MICs should not be considered. This is a major factor for preferring Alternative 2.	The commenter's preference for Alternative 2 due to the increased impacts to transportation/freight mobility under alternatives 3 and 4 is noted. This programmatic EIS evaluates proposed actions that are area- wide and programmatic in nature, rather than location-specific. Therefore, the methodologies used to evaluate potential changes and impacts to the transportation network are broad-based as is typical for the analysis of large-scale plan updates. Because the specific locations and sizes of development are unknown at this time, the location-specific impacts and mitigation projects that will be required are also unknown. Individual development projects will undergo separate and more detailed SEPA review during which

Number	Comment Summary	Response
4-39	Exhibits 3.14-4 and 3.15-5 appear to be missing stormwater infrastructure when compared with the Port's mapping records.	Exhibit 3.14-4 and Exhibit 3.14-5 have been updated to include private stormwater mains available in the City of Seattle mapping. See Section 3.14 Utilities .
5	Saganic	Puget Sound Clean Air Agency
5-1	Final EIS should address dust impacts from increased VMT in the study area.	Thank you for your letter. The potential health impacts of particulate matter is discussed in Pollutants of Concern in Section 3.2.1 . Additional text has been added to include fugitive roadway dust as a source of particulate matter. The potential for fugitive dust emissions associated with soil-disturbing activities, demolition and construction work, and grading are discussed in general in Section 3.2.2 , Construction Related Emissions. The potential for vehicle travel to generate PM2.5 from road dust is discussed under Transportation Related Emissions in Section 3.2.2 Impacts of Alternative 1 No Action . Discussion under Transportation Related Emissions for alternatives 2, 3, and 4 compare emissions to Alternative 1. Additional text is added in each of these sections to include the potential generation of dust associated with increased vehicle miles traveled. Additional text is added to Section 3.2.3 regarding increased street sweeping to prevent impacts from fugitive dust.
5-2	Exhibit 3.2-3.5 is unclear from what the text and figure descriptions provide. Clarification needed.	The results shown in Draft EIS Exhibit 3.2-5 and in Appendix H represent the singular 24-hour PM10 concentrations for the respective sample day and location. Each location had only one 24-hour sample collected. A note has been added to Exhibit 3.2-5 (see Section 3.2 Air Quality & GHG).
5-3	Exhibits 3.2-3.6 are unclear as to the source of the RSL. Source for each RSL should be included	RSLs provided in Draft EIS Exhibit 3.2-6 are available at EPA's Regional Screening Levels website (<u>https://www.epa.gov/risk/regional-screening-levels-rsls</u>). The noncarcinogenic screening levels with a target hazard quotient of 0.1 are used. A note has been added to Exhibit 3.2-6 (see Section 3.2 Air Quality & GHG).
5-4	Details and raw data from air sampling, including detection limits, should be shared publicly.	Appendix H , Technical Memo, "Summary of Air Quality and Noise Monitoring Results at 8 Locations Within the City of Seattle" has been added to the Final EIS and presents the raw data and detection limits used in that monitoring. None of the parameters had laboratory detection limits or reportable limits above the PSI s. There were two locations (SEA3 &
		SEA5) that had measurable concentrations above the RSL for 2- Propanol.
5-5	Incorrect reference to Tacoma attainment status for $PM_{2.5}$.	Additional text has been added in Section 3.2.1 to correct the reference.
5-6	Clarification of the location of denser housing in the Duwamish Valley and potential impacts associated with exposure to changes in air quality.	See Exhibit 2.4-6, Exhibit 2.4-12, Exhibit 2.4-18, and Exhibit 2.4-24 for maps of the MICs and designations for proposed land use changes under each of the alternative studied in the Draft EIS (see also Exhibit 2.4-30 for a map of proposed land use changes under the Preferred Alternative). Section 3.8 Land & Shoreline Use provides descriptions of uses within proposed land use designations, including those that will accept additional and denser housing. Given the non-project nature of this EIS, Section 3.2.1 provides an appropriate level of detail on anticipated sources of pollution that existing and new residents in the study area may be exposed to. Section 3.2.2 provides an appropriate level of detail on the potential air quality impacts to those residents. Section 3.2.3

Number	Comment Summary	Response
		provides an appropriate level of detail for available air quality impact mitigation options (see Section 3.2 Air Quality & GHG). Subsequent developments that may arise from the proposed land use changes in the Industrial and Maritime Strategy will be required to meet all applicable codes and regulations, and to conduct project-level SEPA review at that time, in which analysis will be conducted to assess site specific impacts and necessary mitigation measures.
6	Inghram	Puget Sound Regional Council
6-1	Appreciate the opportunity to comment and the City's work to develop a guiding strategy for its industrial areas. Encourage the City to be thoughtful in meeting regionally-adopted criteria so as to maintain regional designation while balancing a variety of other interests.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
6-2	Cities with centers are required to adopt or update subarea plans for their MICs prior to 2025 to demonstrate consistency with the Regional Centers Framework. Encourage the City to limit housing in MICs. Suggest reviewing PSRC's Industrial Lands Analysis for consistent classification of industrial vs. non-industrial jobs.	As part of VISION 2050, PSRC is requiring the City to prepare updated subarea plans for the two MICs. These updates will update goals and policies consistent with this proposal and address VISION 2050 goals for Centers Plans (see also Objective M of the proposal). Consistent with the PSRC criteria for designating MICs to focus industrial uses in the MIC, the EIS does not study allowing residential uses in the majority of the study area. Alternatives 3 and 4 and the Preferred Alternative consider limited additional flexibility of existing allowances for caretakers' units and artist/studio quarters, or other criteria-limited affordable housing, in the proposed UI zone only. Industrial employment estimates are based on the 2019 share of industrial employment by sector based on the 2015 PSRC Industrial Lands Study NAICs-based definition of industrial activities. This uses classification of what counts as an industrial job are consistent with PSRC criteria, including jobs in Information Computer Technology (ICT). Projections show strong job growth in ICT under the Action Alternatives. Consistency with PSRC classifications is appropriate given the need to fit VISION 2050 and Regional Centers Framework. A more conservative classification of which jobs are industrial, especially in ICT would show a steeper decline in the percent of industrial jobs under most studied alternatives. See footnote in Section 2.4.8 of the Final EIS.
6-3	Encourage a Comprehensive Plan policy to maintain consistency with adopted regional and county criteria for manufacturing/industrial centers.	Comment is noted. Section 3.8 Land & Shoreline Use discusses consistency with regional and county criteria for MICs.
6-4	Support the addition of LU 10.3. Policy could be further improved by referencing potential updates to city-adopted subarea plans for the MICs. Once the City has adopted subarea plans for the MICs, it is reasonable to contemplate land use changes in conjunction with those subarea plan updates.	The City will partner with communities to update subarea plans for the two MICs by the 2025 timeline provided by PSRC. Zoning changes studied in this EIS could be implemented in stages. It is possible that some or all of the zoning changes could occur after subarea planning processes.
7	Panganiban	Seattle Department of Transportation
7-1	Appreciate the opportunity to comment. Interested in proposed development standards	Thank you for your comments. Comment is noted.

Number	Comment Summary	Response
	and code language affecting KUW in the three land use concepts.	
7-2	More detailed exhibits should be shown in the Final EIS for curb ramps and sidewalk conditions in the UI and II zones where multi-modal development standards are proposed.	A link has been added to the Final EIS so that readers can explore detailed data in the City's interactive GIS database within their areas of interest.
7-3	Final EIS should consider and discuss code updates that can expand curb ramp requirements to improve access in the study area, as curb ramps are not required outside of specific development conditions currently.	Through the SDOT Americans with Disabilities Act Transition Plan for the Seattle Public Right-of-Way, the City is committed to install or remediate at least 1,250 curb ramp replacements each year. Locations within the study area will be considered through that prioritization process.
7-4	 (1) Final EIS should outline how the land use code requires new development to construct pedestrian improvements. (2) New zoning designations provide an opportunity for code updates on pedestrian access and circulation requirements. 	If new zoning designations are adopted, SDCI will work with SDOT to develop updates to the Streets Illustrated manual reflecting street design standards tailored to the industrial context and level of expected pedestrian and bicycle activity. Updates will consider street typologies and design standards that can accommodate both freight activity and non-motorized uses with a focus on reducing potential conflicts.
7-5	Will new zoning designations expand street tree requirements to the entire IC/II and IB/UI zone?	Per SMC 23.53.020.B.3 (Improvement requirements for existing streets in industrial zones), if a lot abuts a street designated on Map A for 23.50.016, street trees shall be provided along all designated frontages. These street tree requirements are limited to select streets in the Ballard-Interbay and Duwamish Industrial areas. Proposed development standards for the UI and II zones include street tree requirements on all streets in new development in those areas.
7-6	Will the list of industrial landscape streets and associated landscape standards be revised to align with future land use and transportation patterns in future MML zoning?	The industrial landscaped streets and standards will be revised to align. See also discussion in the development standards Appendix G .
7-7	In the MML zone, please clarify if streets improvements are intended to be consistent with what is currently required under IG zoning, or if more extensive development, standards will be developed to improve pedestrian access, circulation, and safety.	Landscape and street improvement standards will be modified more for the II and UI zones, than for the MML zone. However, some updates and modifications to the street improvement standards will occur for the MML zone. See also the development standards Appendix G .
7-8	The Draft EIS identifies modal conflicts and collisions near intersections. Does the analysis include documentation and analysis of curb cuts and vehicular access onto private property, and collision data related to turn movements onto private property?	The analysis includes all reported collisions within the study areas including crashes that related to turning movements to and from private property along the roadway. Characteristics of individual collisions were not analyzed for this programmatic evaluation.
7-9	Consider how standards developed within this body of work are coordinated with ST3 development standards and potential street design concepts for station frontages.	Comment is noted and forwarded to City decision makers. The Sound Transit EIS is a different proposal from the Industrial Maritime Strategy. City staff are coordinating information and data from Sound Transit to the greatest extent possible. Text has been added to the mitigation section of this EIS to note that the City and Sound Transit are coordinating on transportation mitigation around expanded and new light rail stations and notes the System Access Fund as a funding mechanism for station area improvements. See also Section 4.2.4 . Updated street design standards will also be developed for the upcoming update of the Streets Illustrated manual related to any new adopted zoning

Number	Comment Summary	Response
		designations as studied in this EIS. City code updates may also be implemented for station frontages. Sound Transit's West Seattle and Ballard Link Extensions Draft EIS also covers non-motorized mitigation measures. Section ES.4 Avoidance, Minimization, and Mitigation Measures reads as follows:
		"When maintaining a facility would not be feasible, Sound Transit would work with the City of Seattle to develop and implement a construction management plan to provide alternate facilities for non- motorized travel."
8	Acutanza	Seattle Freight Advisory Board
8-1	Summary of purpose and mission of the SFAB.	Thank you for your comments. The comment is noted and forwarded to City decision makers.
8-2	Additional analysis requested for the impacts of new land uses by all modal networks (heavy haul networks, rail systems, and intermodal yards supporting manufacturing uses). Additional questions about at-grade rail crossing impacts on the alternatives.	The Final EIS has been reorganized to include freight as a separate section throughout each element of the transportation section (Affected Environment, Impacts, Mitigation Measures, and Significant Unavoidable Adverse Impacts). This will include addressing intermodal yards, rail, and truck parking.
8-3	Additional analysis requested for issues related to demand for overnight truck parking and the impact on land uses. How do alternatives accommodate long-haul parking needs?	See response to comment 8-2.
8-4	Final EIS should acknowledge the heightened risk of impacts to pedestrians and cyclists from heavy and/or large vehicles. If alternatives increase conflicts with vulnerable users, impacts and mitigation measures should be identified.	The commenter requests that "the Final EIS in this industrial area should acknowledge the heightened risk of impacts to pedestrians, cyclists and scooter riders from heavy and/or large vehicles (like trucks, which are inherent to industrial operations)." Language to this effect was included in the Draft EIS (p. 3-388) and will be retained in the Final EIS. Supplemental language has also been added per the commenter's suggestion regarding truck drivers' limited range of sight distance and turning radii conflicts that aren't expected with smaller vehicles.
		The Mitigation Measures , Pedestrian & Bicycle System Improvements section of the EIS identifies the types of mitigation measures that would complete network gaps for vulnerable users and separate them from motorized traffic. These include "facilities such as sidewalks, asphalt walkways, or painted walkways; signals to make crossing roadways easier; treatments such as rectangular rapid flashing beacons to alert drivers to people crossing the street; marked crosswalks; curb bulbs or extensions to shorten crossing distances and make people walking more visible to drivers; bicycle lanes (including protected and buffered bicycle lanes); and multi- use trails." Language has been added to the mitigation section to reiterate that those measures would have safety benefits as they would separate vulnerable users from motorized traffic, particularly large trucks which inherently operate with higher-risk conflicts.
8-5	Mitigation in the Draft EIS is not applied or described in enough detail to know whether it will resolve the impacts mentioned—request the Final EIS to address the likelihood that mitigation would resolve or successfully lessen the negative impacts identified.	SEPA does not require quantification of the magnitude to which each measure would mitigate impacts and the non-project EIS addresses the qualitative effectiveness of the potential mitigation measures. This programmatic EIS addresses area-wide land use zoning changes, rather than a project-specific proposal. The proposal may result in a wide range of individual projects implemented over a long timeframe and across a large geographic

Number	Comment Summary	Response
		area. Because the specific locations and sizes of development are unknown at this time, it would be speculative to identify specific mitigation measures. Individual development projects will undergo separate and more detailed SEPA review during which specific impacts and mitigation will be determined. The City is committed to seeking funding to implement these strategies as needed, but it would be speculative to quantify potential costs at this stage. Further, SEPA does not require cost information for mitigation. The City develops cost and funding options for its capital improvement programs. Secondary impacts are discussed on page 3-419 of the Draft EIS: <i>"It</i> <i>should be noted that some transportation mitigation projects could</i> <i>have secondary impacts. For example, converting a general-purpose</i> <i>travel lane or a parking lane to a transit lane, truck-only lane, or cycle</i> <i>track would reduce capacity for autos to travel or park. As required, the</i> <i>City would prepare additional analysis and take public and stakeholder</i> <i>input into consideration before implementing specific transportation</i> <i>improvement projects. Given the programmatic nature of this study,</i> <i>this EIS simply lists the types of projects that could be considered to</i> <i>mitigate potential impacts of the programatic natures,</i> "
9	Mohler	Seattle Planning Commission
9-1	EIS must ensure any zoning proposals move to repair harms of the past and benefit affected communities through both public and private investment. Summary of key questions and concerns addressed in the letter.	Thank you for your comments. The EIS includes a section on historical land use and planning decisions that has an emphasis on past harms (see Section 3.8.1). A new subsection is added related to exclusionary zoning in the Final EIS. The EIS also includes a review of equity and environmental justice considerations in Chapter 1 and throughout the other chapters of the EIS. Where appropriate expanded discussion of mitigation measures is included with a focus on historically disproportionately impacted communities including Georgetown and South Park. See also response to comments 9-2 through 9-24 below.
9-2	Tribes should be consulted. Recommend explicit recognition of impacts to the cultural and historic importance of indigenous land, including the ancestral lands of the Duwamish, Suquamish, Stillaguamish, and Muckleshoot Tribes.	See response to comments 9-1 and 9-22. The overview of past planning and land use decisions section of Section 3.8 and Section 3.11 Historic, Archaeological, & Cultural Resources recognizes the historical and present importance of Tribal lands. Input from Tribes was solicited during the scoping and Draft EIS comment periods.
9-3	 (1) Additional analysis and requested mitigation related to equity and environmental justice. Specifically identify the key differences between the two MICs when documenting impacts and proposing mitigation measures for each. (2) Analyze environmental health impacts to both residents and workers in the Duwamish Valley from exposures to environmental hazards such as air pollution, contamination, and noise. (3) Recognize that more new jobs will be created in the BINMIC than in the Duwamish Valley under the proposed alternatives. Evaluate mitigation strategies that will enable BIPOC and gender-inclusive access to job opportunities in Ballard and Interbay and increase opportunities in the Duwamish Valley. 	 (1) The impacts analysis under each environmental topic considers impacts common to all industrial areas as well as those specific to each of the five subareas defined within the MICs (Ballard, Interbay Dravus, Interbay Smith Cove, SODO/Stadium, and Georgetown/South Park). (2) Please see Section 3.2.1, Pollutants of Concern for a discussion of health impacts associated with exposure to criteria air pollutants [carbon monoxide (CO); particulate matter (PM); ozone, and the ozone precursors (volatile organic compounds [VOCs] and oxides of nitrogen [NOX]); sulfur dioxide (SO2); and lead], or toxic air pollutants. Section 3.2.2 discusses potential impacts associated with each alternative, including potential increased exposure to these air pollutants. At this non-project level of analysis, more specific analysis of potential health impacts is not possible, as specific developments, development locations, site-specific conditions, exposure pathways and receptors are unknown.

displacement pressures, and identify the

benefits of anti-displacement measures and incentives. Analyze the regional economic impact of combining land usable for

manufacturing jobs with other uses as a result of the II and UI land use concepts. Analyze the economic impacts of the land use alternatives in light rail station areas, including an economic development feasibility analysis of the Industry and Innovation land use concept. Analyze impacts of locating makerspaces and other creative uses within non-industrial

neighborhoods, urban villages, and mixed-use zones. Analyze the economic feasibility of

Number	Comment Summary	Response
		use changes in the industrial and ivaritime strategy will be required to meet all applicable codes and regulations, and to conduct project-level SEPA review at that time, in which analysis will be conducted to assess site specific impacts and necessary mitigation measures. See Section 3.2 Air Quality & GHG .
		Please see Effects of Noise on People in Section 3.6.1 for a discussion of health impacts associated with exposure to environmental noise. At this non-project level of analysis, more specific analysis of potential health impacts is not possible, as specific developments, development locations, site-specific conditions, noise sources and receptors are unknown. Subsequent developments that may arise from the proposed land use changes in the Industrial and Maritime Strategy will be required to meet all applicable codes and regulations, and to conduct project-level SEPA review at that time, in which analysis will be conducted to assess site specific impacts and necessary mitigation measures.
		Site redevelopment activities in general have a positive effect on legacy contamination caused during previous decades of less stringent regulations, because sites must be characterized and remediated in order to receive financing, and/or to satisfy conditions of Consent Decrees or Administrative Orders. Please see Contamination Section 3.5.3 Mitigation Measures for a description of how environmental health impacts to both residents and workers from exposures to environmental hazards such as contamination would be mitigated under all alternatives.
		(3) The Draft EIS recognizes equity and accessibility as one of the six key emerging factors affecting Seattle's MICs, specifically access to maritime and other industrial career opportunities for BIPOC and women. Maintaining a strong industrial economy is a prerequisite to providing these opportunities, but other strategies including outreach to BIPOC youth and workforce training investments are key parts of the Industry and Maritime Strategy.
9-4	Additional analysis and requested mitigation related to Land & Shoreline Use. Identify how much total industrial space is needed for the City to reach its growth projections. Specifically identify which of the sub-areas studied will likely receive job growth and require additional investment and how this may create or exacerbate economic segregation impacts. Identify the impacts of protecting industrial and maritime lands, reference potential	SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450). See Section 4.2.1 . The EIS includes employment projections associated with each alternative including proportion of industrial and non-industrial jobs. Amounts of employment are estimated for subareas. The distribution of jobs by subarea is shown in Exhibit 2.4-40 . Alternatives assume 700 square feet per industrial employee and 250 square feet per non-industrial employee similar to buildable lands assumptions. Building space associated with each alternative is illustrated in Exhibit 3.1.4 and a

similar graph in **Exhibit 3.5-7**.

Number	Comment Summary	Response
	establishing higher standards for landscaping and multi-modal transportation to create healthier transitions within single-use industrial zones.	
9-5	Evaluate the City's Shoreline Master Program's effectiveness in maritime and industrial areas to strengthen protection of currently undeveloped shorelines and to promote strategies to improve water quality treatment and flood resiliency.	Shoreline Master Program regulations are summarized in Section 3.8 Land & Shoreline Use and Appendix F . No changes to Shoreline Master Program regulations are proposed as a part of this action.
9-6	In policies SA P37 and SA P39, consider building in a requirement for climate resiliency and consider removing the allowance of expansion of existing water -dependent facilities unless such expansion will provide ecological benefits. Recommend goals and policies codify language around BIPOC and gender-inclusive job training programs and access to opportunity for both the BINMIC and Greater Duwamish MIC subarea plans.	The Draft EIS recognizes equity and accessibility as one of the six key emerging factors affecting Seattle's MICs, specifically access to maritime and other industrial career opportunities for BIPOC and women. Maintaining a strong industrial economy is a prerequisite to providing these opportunities, but other strategies including outreach to BIPOC youth and workforce training investments are key parts of the broader Industry and Maritime Strategy's non land- use components that are not required to be analyzed in an EIS under SEPA, but are components the City can address in its economic development strategy implementation.
9-7	Additional analysis and requested mitigation related to housing. Concerned with the broad impacts on housing citywide and throughout the region resulting from increased employment growth under the Action Alternatives. Concerned proposed mitigation measures may not be sufficient to address the housing needs associated with the significant job growth. Suggest including a jobs/housing analysis, current and future housing capacity outside Seattle that will be accessible via light rail, impacts of residential uses in industrial areas through an environmental justice and public health lens, and trade-offs associated with allowing industry-supportive residential uses. Request appropriate mitigation measures for the many skilled workers that may need to commute long distances to new jobs and evaluation of tools such as impact fees to generate additional affordable housing options within Seattle.	Comment is noted. In the Preferred Alternative, the number of dwellings in industrial areas is projected to increase by 1,475 units—less than the amount studied in Draft EIS Alternative 4 (720 less). Allowances for caretakers' quarters and makers studios in the UI zone are more limited than Alternative 4 in the Draft EIS—only 2% of such new units are projected citywide (2019-2044). Two new areas outside the MICs in west Ballard and Judkins Park would be converted to mixed use zoning allowing housing, in addition to the proposed mixed-use areas in Georgetown and South Park. Overall, a slightly lower total amount of housing production would result compared to Draft EIS Alternative 4 (8%), but it would be outside of MICs, or subject to standards to reduce conflicts. Affordability requirements proposed with the Preferred Alternative are described in Appendix G . The City will plan for the citywide amount of housing growth in the Comprehensive Plan EIS on a citywide scale. Applying MHA to the proposed new II zone can also be a mitigation strategy.
9-8	Additional analysis and requested mitigation related to transportation.	See response to comments 9-9 through 9-12.
9-9	Clearly identify how future light rail stations will interact with the surrounding and/or adjacent industrial and maritime lands. Analyze the potentially competing demands of protecting industrial lands and robust ridership at all station locations. Reference estimates of job growth resulting from the zoning changes around each of the stations in industrial areas as well as ridership projections in Sound Transit's West Seattle and Ballard Link Extensions Draft EIS.	As described in Chapter 1 , the EIS analyzes alternatives representing different potential futures for the city's industrial lands with the aim of both strengthening land use projections for core and legacy industrial and maritime areas and encouraging denser development coupled with industrial businesses near transit stations. The ridership projections published in the WSBLE Draft EIS have been referenced in the Final EIS. The percent of job growth is higher in Ballard and Interbay where stations are planned compared with other areas. See Exhibit 1.5-21 of this Final EIS.

Number	Comment Summary	Response
9-10	Identify specific mitigation measures for impacts to freight mobility and logistics.	The EIS includes a Mitigation Measures section which describes the various plans that include specific projects and high priority areas for improvement. Those documents include: the Freight Master Plan, Transit Master Plan, Pedestrian Master Plan, the Bicycle Master Plan, the Bicycle and Pedestrian Safety Analysis, the Ballard-Interbay Regional Transportation (BIRT) System Report, and the Georgetown Mobility Study.
9-11	Conduct an equity analysis to identify impacts resulting from conflicts between freight traffic and other modes in communities without sufficient non-motorized infrastructure and identify appropriate additional mitigation measures. Consider mobility hierarchy through an equity lens when assessing mitigation measures.	The EIS includes an Equity & Environmental Justice Considerations section (see Section 3.10.2) describing which portions of the study area have large proportions of priority populations and how they could be affected by the alternatives. In particular, it references potential impacts to the safety of people walking and biking in neighborhoods with histories of long-term underinvestment.
		With respect to considering a mobility hierarchy through an equity lens, SDOT is currently in the process of developing the Seattle Transportation Plan which will integrate the City's modal plans into a comprehensive vision for the citywide transportation network centered around the following values and goals: equity, safety, mobility, sustainability, livability, and excellence.
9-12	Conduct an inventory and gap analysis of walking and biking facilities in industrial areas, especially around future light rail stations. Identify what types of transportation capital projects are required to keep pace with the change in jobs resulting from the Action Alternatives.	A link has been added to the Final EIS so that readers can explore detailed data in the City's interactive GIS database within their areas of interest.
		The EIS includes a Mitigation Measures section which describes the various plans that include specific projects and high priority areas for improvement. Those documents include: the Freight Master Plan, Transit Master Plan, Pedestrian Master Plan, the Bicycle Master Plan, the Bicycle and Pedestrian Safety Analysis, the Ballard-Interbay Regional Transportation (BIRT) System Report, and the Georgetown Mobility Study.
		The Mitigation Measures , Pedestrian & Bicycle System Improvements section of the EIS identifies the types of mitigation measures that would complete network gaps for vulnerable users and separate them from motorized traffic. These include "facilities such as sidewalks, asphalt walkways, or painted walkways; signals to make crossing roadways easier; treatments such as rectangular rapid flashing beacons to alert drivers to people crossing the street; marked crosswalks; curb bulbs or extensions to shorten crossing distances and make people walking more visible to drivers; bicycle lanes (including protected and buffered bicycle lanes); and multi- use trails."
9-13	Additional analysis and requested mitigation related to biological resources and resiliency.	See response to comments 9-14 through 9-18.
9-14	Clearly identify risks of all construction in liquefaction zones.	Text has been added to Section 3.1.2 Impacts to address the risks associated with construction of water, wastewater, and transportation infrastructure. See Section 3.1 Soils/Geology .
9-15	Additional analysis of air quality impacts on residential areas near industrial zones such as South Park and Georgetown; and of co-locating offices and other non-industrial uses above industrial spaces in the Industry and Innovation land use concept.	This non-project EIS provides an assessment of the existing levels of regulated pollutants and compliance with the NAAQS, and anticipated air emissions associated with potential land use changes based on two sources of baseline ambient air quality conditions data: 1) from Ecology- and PSCAA-operated ambient air quality monitoring stations; and 2) from air quality data collected directly by The City of Seattle at eight sites within the BINMIC and

Number	Comment Summary	Response
		Greater Duwamish MIC—selected due to the location of potential zoning changes in alternatives or due to their proximity to air quality emission sources. All data indicate that air pollutant concentration trends, and individual measurements, for these pollutants remain below the NAAQS when wildfire is excluded.
		SEPA's procedural provisions require the consideration of "environmental" impacts (see definition of "environment" in WAC 197-11-740 and of "impacts" in WAC 197-11-752), with attention to impacts that are likely, not merely speculative. (See definition of "probable" in WAC 197-11-782 and 197-11-080 on incomplete or unavailable information.
		The current level of analysis provides an appropriate level of detail for a non-project EIS (see WAC 197-11-442 for a description of the contents of an EIS on non-project proposals), and without more specific knowledge of development locations, site-specific conditions, exposure pathways and receptors at proposed developments, additional analysis would be overly speculative. Subsequent developments that may arise from the proposed land use changes in the Industrial and Maritime Strategy will be required to meet all applicable codes and regulations, and to conduct project-level SEPA review at that time, in which analysis will be conducted to assess site specific impacts and necessary mitigation measures.
9-16	 (1) Analyze and document future projections of rainfall and stormwater flows. Evaluate the extent of existing stormwater and water quality impacts to determine whether the Action Alternatives can provide significant beneficial impacts. (2) Identify opportunities for increasing innovative green infrastructure in industrial zones to protect water quality and mitigate 	(1) As stated in Sections 3.3 and 3.14 , development under any alternative will be required to meet current stormwater regulations which is expected to improve stormwater management relative to existing conditions. This conclusion applies to any land use type in the Study Area, including industrial and maritime areas. Additional text has been added to clarify that this is true even if rainfall patterns increase in intensity. Sections 3.3 and 3.14 provide an assessment of future impacts to water resources relative to existing conditions, which is appropriate for this EIS.
	climate change.	(2) Green infrastructure methods are standard for meeting on-site stormwater management as stated in Section 3.14 . Site specific analysis would be performed at the lot level during redevelopment projects. Redevelopment projects will result in improved water quality and flow control (if applicable and feasible).
9-17	Identify specific areas of SODO, South Park, Ballard, and Interbay at risk for sea level rise and evaluate the impacts of adding density to these areas.	Sea level rise is addressed through existing regulations as discussed in Section 3.3.2 . Subareas sensitive to sea level rise are discussed in this section, along with mitigation measures in Section 3.3.3 . Given the non-project nature of this EIS, Section 3.3 provides an appropriate level of detail on the risk and impact of development related to sea level rise. Subsequent developments that may arise from the proposed land use changes in the Industrial and Maritime Strategy will be required to meet all applicable codes and regulations, and to conduct project-level SEPA review at that time, in which analysis will be conducted to assess site specific impacts and necessary mitigation measures.
9-18	Identify the ecosystem benefits of adding green infrastructure and increasing trees and green landscaping in and near the MICs.	Analysis of impacts for each alternative in Section 3.4 Plants & Animals includes a discussion of how green infrastructure and increasing trees/landscaping provides opportunities for stormwater treatment and additional wildlife habitat.

Number	Comment Summary	Response
9-19	Additional analysis and requested mitigation related to environmental health and compatibility.	See response to comments 9-21 through 9-24.
9-20	 (1) Identify and analyze any potential contamination impacts on future residential uses in or near industrial areas. (2) Restore lands and shorelines with industrial contamination, including contaminants in fish from waterways adjacent to industrial areas. 	 (1) This non-project EIS is limited to a general discussion of potential contamination impacts of alternatives on future residential land uses near industrial areas. The current level of analysis provides an appropriate level of detail for a non-project EIS. Please refer to Section 3.5.3 that describes how redevelopment at individual parcels will have to comply with all regulatory requirements at that time such as SEPA, and MTCA which sets stricter cleanup levels for residential land uses. See Section 3.5 Contamination.
		(2) Comment acknowledged. Restoration of shorelines and remediation of contaminated sites is accomplished on a site- specific basis at the time of redevelopment and through the project permitting process. The process of site characterization, remediation, and preventing recontamination of the Lower Duwamish Waterway during site construction activities for example, is closely scrutinized by Ecology, EPA, and others.
9-21	Support the proposed mitigation measure to limit proximity of new residential development to known or anticipated sources of high noise levels.	Comment is noted.
9-22	Concerned that the list of data sources in Section 1.7.11 (page 1-62) does not include tribal consultation. Suggest codifying consultation with the Duwamish Tribe to redress historic exclusion, despite the tribe not yet being federally recognized. Request listing specific indigenous tribes as well as acknowledging other settlement in addition to Euro-American settlement.	The cultural resources consultant accessed WISAARD's archaeological records that contain known Tribal cultural sites. These records are considered restricted and confidential. Cultural resources review is a process that is done prior to the start of many projects and includes consultation with Tribes. Many federal, state, and local statutes and ordinances require notice and consultation with affected Tribes before, during, and after project review. The National Historic Preservation Act (NHPA) of 1966, was amended in 1986 with provisions for consultation with affected Tribes and 1992 to include and clarify the roles and responsibilities of Indian Tribes in Section 106 reviews.
		The Advisory Council on Historic Preservation (ACHP) adopted a Policy Statement Regarding the ACHP's Relationships with Indian Tribes in 2000. The policy was developed in consultation with some Tribes and inter-Tribal organizations, and addresses tribal sovereignty, government-to-government consultation, trust responsibilities, tribal participation in historic preservation, sympathetic construction, and respect for tribal religious and cultural values.
		The state of Washington has a government-to-government relationship with the 29 federally recognized Tribes in the state (RCW 43.376). Each Tribe is a sovereign nation and has its own definition of appropriate consultation.
		Input from Tribes was solicited during the scoping and Draft EIS comment periods.
		The statutes and ordinances specify consultation with federally recognized Tribes only. In addition, the City solicited input directly from the Duwamish.

Number	Comment Summary	Response
		In the Ethnographic Background section (Section 3.11.1), the Duwamish Tribe and significant cultural locations to the Tribe are specifically discussed. The Duwamish Tribe as well as the federally recognized Muckleshoot, Snoqualmie, Suquamish, and Tulalip Tribes are also addressed under Historic Period Context .
9-23	Increase and/or improve parks and open space in and near the MICs, especially in the Duwamish Valley, where appropriate in an industrial context. Request analysis and documentation of impacts related to the need for parks and open space resulting from future residential uses within industrial areas.	See Section 3.12 Open Space & Recreation for an analysis of additional need for parks and open space under each of the alternatives. It addresses the demand for parks by subarea with all housing types under each alternative.
9-24	Analyze the impacts and need for public services specifically related to future residential uses within industrial areas. We also request an assessment of the impacts and mitigation measures for organizations other than emergency services.	See Section 3.13 Public Services for an analysis of additional need for fire and emergency medical services, police, and schools and libraries under each of the alternatives related to both increases in residential and worker populations. The City identified specific public services to be studied in the EIS during scoping.
10	Gannon	Seattle Public Schools
10-1	Appreciates the opportunity to comment. SPS owns and operates the John Stanford Center for Educational Excellence in the SODO neighborhood. MML designation would render SPS's use of the Stanford Center as legally nonconforming limiting development flexibility in the future.	Thank you for your comments. See Section 4.2.2 concerning non- conforming uses. The different alternatives in the EIS consider different zoning designations on the referenced site. The Preferred Alternative includes the site in the II zone.
10-2	Alternatives considered could better address existing conditions and encourage both industrial and office development in a more flexible manner.	The studied alternatives are intended to promote industrial uses consistent with VISION 2050 MIC requirements and recognize evolving employment formats and supportive uses. See Appendix G of the Final EIS for the Preferred Alternative conceptual code.
10-3	Draft EIS greatly understates the environmental and community impacts of the Action Alternative by precluding development that exceeds its strict limitations on storage, offices, sales and services, restaurants.	The EIS studies the No Action Alternative as well as action alternatives. The MIC policies have for some time intended to maintain industrial uses as primary in zones. See also response to comment 10-2.
10-4	Alternative 2 could preclude the development of properties leaving them vacant and maintaining status quo, particularly in SODO.	Comment noted. See Section 4.2.2 concerning non-conforming uses.
10-5	Environmental cleanup will not occur if redevelopment is rendered infeasible, causing adverse impacts.	See Section 3.4.2 addressing impacts of a lack of redevelopment.
10-6	 (1) Assumption that there would be improved infrastructure in areas zoned as II needs further exploration; uses prohibited or made difficult by the Alternatives will be sited further from the people that use them, thus increasing the volume and length of vehicle trips and causing significant adverse transportation impacts. (2) Alternatives do not take advantage of light rail proximity and instead encourage heavy car usage. 	 (1) The EIS evaluates the uses allowed in the MIC, including industrial and non-industrial employment and limited housing. The results of the evaluation on all modes and needed mitigation measures at a planning level are provided in the EIS. (2) The II zone is considered for the site in alternatives 3 and 4 and the Preferred Alternative. It would allow mixed use with industrial, technology, and office in proximity to light rail. See response to comment 10-7 below for additional information. (3) The City is working closely with Sound Transit as the ST3 project moves forward.

Number	Comment Summary	Response
	(3) City should work with Sound Transit to ensure that Final EIS is aligned with ST3.	
10-7	Final EIS should provide alternatives that allow for non-industrial uses in SODO.	Comment is noted and forwarded to City decision makers. Alternatives studied in the EIS are consistent with the PSRC criteria for designating MICs to focus industrial uses in the MIC. Non- industrial uses in the proposed MML zone are permitted as a principal use only when subject to strict maximum size of use limits and FAR sub-limit. Non-industrial uses are permitted subject to strict maximum size of use limits only, and are only allowed as bonus development in the II zone. Non-industrial uses ancillary to an industrial use would be allowed in the proposed MML and UI zones under varying requirements. Alternatives 3 and 4 and the Preferred Alternative consider limited additional flexibility of existing allowances for caretakers' units and artist/studio quarters, or other criteria-limited affordable housing, in the proposed UI zone only. The II zone, applied in alternatives 3 and 4 and the Preferred Alternative, would allow for a significant amount of non- industrial uses through a development bonus system. The II zone would be applied under multiple alternatives to the area around the SODO/Lander St. station in the SODO area.
11	Persak	The Office of Economic Development
11-1	Want to advocate for actions which cumulatively will have the least Significant Impacts, and the lowest possible risk for Significant Unavoidable Adverse Impacts to the maritime, manufacturing, and logistics industry's supporting land use activities and transportation safety, so that these jobs remain for future generations.	Thank you for your comments. The comment is noted and forwarded to City decision makers.
11-2	Top priority is centering workforce development for BIPOC communities and women to benefit from more direct pathways into maritime, manufacturing, and logistics. Retention and expansion of "missing middle" livable wage jobs in Seattle can be achieved in maritime, manufacturing, and logistics within the footprint of Seattle's MICs if we can prioritize the functionality of these spaces.	The comment is noted and forwarded to City decision makers. The EIS recognizes equity and accessibility as one of the six key emerging factors affecting Seattle's MICs, specifically access to maritime and other industrial career opportunities for BIPOC and women. Maintaining a strong industrial economy is a prerequisite to providing these opportunities, but other strategies including outreach to BIPOC youth and workforce training investments are key parts of the Industry and Maritime Strategy.
11-3	Publicizing training, retention strategies, and partnering with employers and CBOs who are committed to equity in maritime, manufacturing, and logistics is more effective when there is a built environment that supports business longevity.	The comment is noted and forwarded to City decision makers.
11-4	Final EIS and adoption of Comprehensive Plan amendments should provide more policy stability for future job growth in maritime, manufacturing, and logistics. Appreciate the EIS public comment opportunities thus far.	The comment is noted and forwarded to City decision makers.
11-5	Future multiple opportunities for engagement will build on the results of this effort. Final EIS will be the necessary cornerstone to make progress on future planning.	The comment is noted and forwarded to City decision makers.

Number	Comment Summary	Response
11-6	Offers qualified support for Alternative 2.	The comment is noted and forwarded to City decision makers.
11-7	Emphasis on maritime, manufacturing, and logistics job growth is highest under Alternative 2.	The comment is noted and forwarded to City decision makers.
11-8	PSRC requires a 50% rate of "industrial" employment in the MIC. Alternatives 3 and 4 risk falling below that threshold.	The comment is noted and forwarded to City decision makers. Conditions in both the Greater Duwamish MIC and the BINMIC would still meet PSRC's regional criteria under all of the alternatives studied.
11-9	Study SR 509 through South Park, SR 599 feeding into SR 99, and 1st Avenue Bridge— extent of impacts is unknowable without additional study.	See response to comment 2-4.
11-10	Alternative 2 represents the highest land use capacity for maritime, manufacturing, and logistics while addressing some past limitations, and supports future TOD along light rail extensions.	The comment is noted and forwarded to City decision makers.
11-11	Alternatives 3 and 4 would incur substantial more costs to mitigate traffic congestion and safety, presenting a higher risk that mitigation does not actually occur in the long run due to financial constraints.	SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450). See Section 4.2.1 . Please note that the Preferred Alternative growth is more similar to Alternative 2 and would have lesser traffic impacts than alternatives 3 and 4.
11-12	Alternative 2 represents less future risk of protracted community conflicts over land use and supportive appropriate transportation modes for the MIC.	The comment is noted and forwarded to City decision makers.
11-13	Open space concepts in the Georgetown neighborhood should be studied under new UI zoning.	Section 3.12 discusses the effects on demand for and need for new open space resources under different land use and growth scenarios. Mitigation measures in the open space section consider approaches to providing open space.
11-14	In Alternative 2, there is no significant residential housing expansions in the UI zone under Alternative 2, whereas there is expansion in alternatives 3 and 4. The UI zone in each Action Alternative represent a one size fits all approach, despite the substantial differences in the needs and challenges of these areas.	The comment is noted and forwarded to City decision makers. Alternatives consider different patterns and location of zone changes in response to local conditions and needs. The Preferred Alternative includes a new pattern of zoning changes, including the extent of the UI zone, in response to comments on the Draft EIS.
11-15	Flexibility in the current STOAD provides ample opportunity for further in-fill development but the "buffering" potential has not been fully utilized. Moving ahead in the STOAD on UI as presented may induce demand for additional mixed-use south of the Overlay beyond what is already allowed. No data presented in Draft EIS to suggest community preference for new housing near freeways, major truck streets, and other heavy uses in SODO.	The EIS alternatives include analysis of potential impacts of varied amounts and concentrations of housing under different alternatives in Section 3.9 Housing. Section 3.8 Land & Shoreline Use evaluates potential land use impacts.
		Consistent with the PSRC criteria for designating MICs to focus industrial uses in the MIC, the EIS does not study allowing significantly expanded residential uses in the majority of the study area. The City has no data on the additional demand for mixed-use that would be induced because of potential zoning changes in the
11-16	Impacts of changing IG2 zoning in Georgetown to mixed use zoning demands a separate analysis. Alternative 2 should be modified to	STAOD. Additional detail regarding development standards to address the unique conditions in the proposed mixed use zoning in Georgetown are included under the Preferred Alternative, in the

Number	Comment Summary	Response
	create an overlay for Georgetown to recognize and preserve its distinct character.	development standards Appendix 6 . This includes incentive features to protect distinct character.
11-17	Document should disaggregate data for collisions between vehicles and bikes vs. trucks and bikes, and include discussion on risk factors of truck and bicycle/pedestrian collisions for each alternative.	Language distinguishing between the safety risks of cars and trucks was included in the Draft EIS (p. 3-388) and is retained in the Final EIS. This programmatic EIS addresses area-wide land use zoning changes, rather than a project-specific proposal. The proposal may result in a wide range of individual projects implemented over a long timeframe and across a large geographic area. Because the specific locations and sizes of development are unknown at this time, it would be speculative to identify how modal conflict risk factors may compare in particular locations. However, the VMT increase range discussed on page 3-388 of the Safety impacts section of the Draft EIS has been broken out by alternative to compare the relative exposure of vulnerable users. Individual development projects will undergo separate and more detailed SEPA review during which specific impacts and mitigation (including potential conflicts between trucks and people walking and biking) will be determined.
11-18	 (1) Additional data on truck parking, especially where capacity is at an equilibrium and at capacity for other vehicle needs to be considered, and realistic and achievable solutions identified. (2) Draft EIS should discuss peak game day traffic patterns in their impact on freight. (3) Draft EIS should analyze impact of the inventory and functionality of truck loading zones and other freight access points for all alternatives. 	 (1) [City input needed—do you have data regarding truck parking that could be referenced? We are adding general text about truck parking needs, but have not seen any quantitative demand data] Additional information about truck parking has been added throughout the transportation chapter of the Final EIS. (2) A text box has been added to the Final EIS referencing the large event venues in the study area and gameday conditions. (3) The EIS includes a Parking impacts section describing the competing needs for public curb space and acknowledges that the action alternatives are expected to result in significant adverse impacts to on-street parking absent mitigation measures. This programmatic EIS addresses area-wide land use zoning changes, rather than a project-specific proposal. The proposal may result in a wide range of individual projects implemented over a long timeframe and across a large geographic area. Because the specific locations and sizes of development are unknown at this time, it would be speculative to quantify truck loading demand in a particular location. Individual development projects will undergo separate and more detailed SEPA review during which specific impacts and mitigation (including on-street parking) will be determined. The SDOT Curbside Management Team actively identifies and installs commercial vehicle and general load unload zones in business districts throughout Seattle and would identify load zone needs with new development as needed or requested by development projects. SDOT is also working on potential policy changes to more actively install load zones and other curb access needs at new development during the City development review process.
11-19	Qualified support for Alternative 2.	The comment is noted and forwarded to City decision makers.
12	Brower	Brower Law, Salmon Bay Sand and Gravel Company
12-1	Generally support Alternative 3 but want to ensure the City doesn't continue trying to located incompatible uses in industrial areas.	Thank you for your comments. The comment is noted and forwarded to City decision makers.

Number	Comment Summary	Response
12-2	Seattle must stop trying to locate incompatible uses in maritime and industrial zones because doing so actively undermines existing maritime and industrial businesses.	Comment is noted. Section 3.8 Land & Shoreline Use discusses land use compatibility impacts under all alternatives including potential impacts of non-industrial uses on the ability of industrial uses to operate effectively.
12-3	OPCD must recognize how incompatible uses will undermine its industrial areas. OPCD should revise goal B1-G11 and B1-P15 in the 2020 comp plan to prohibit location and construction of recreational uses in the BINMIC.	Comment is noted. Section 3.8 Land & Shoreline Use discusses compatibility impacts.
12-4	Revise transportation figures in Seattle 2035 to eliminate incompatible uses/co-locations. Unsupportive of the missing link strategy through the BINMIC.	Comment is noted. The City will address the MIC Plan to address regional requirements and can consider consistency with other City policies as appropriate.
12-5	Supportive of alternatives 3 and 4 approach to make it harder if not impossible to rezone industrial lands to non-industrial uses.	Comment is noted and forwarded to City decision makers.
13	Burke	Freemont Dock Company via Houlihan Law
13-1	Appreciates the opportunity to comment. Fremont Dock Company owns properties in Ballard and Fremont within the study area, and is a member of the Ballard Council and North Seattle Industrial Association.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
13-2	Draft EIS is inadequate because the zoning changes and implementing development regulations should be considered together.	This is a programmatic level EIS. Sufficient detail about the proposed development standards to fully consider the potential for environmental impacts is included, such as preliminary zoning maps, tables of potential standards, etc. See Chapter 2 . The Draft EIS includes sufficient detail about proposed development standards and potential zoning changes. Based on the Draft EIS evaluation and mitigation measures, a Preferred Alternative has been developed, and finer grained preliminary development standards are included in this Final EIS. See Final EIS Appendix G .
13-3	Proposal is not sufficiently defined to allow meaningful environmental review because "industrial" is not defined.	See response to comment 13-2. The Final EIS includes a new table specifying which specific land uses would be qualifying as industrial under the proposed zones. See Appendix G .
13-4	Alternatives are inconsistent with the locational criteria and proposed polices.	Section 3.8 Land & Shoreline Use discusses the degree of consistency of the alternatives with policies.
13-5	Alternatives should be proposed, reviewed, and selected on a subarea basis.	All alternatives include detailed proposals with map information to specific boundaries for all subareas. Where feasible and practical, impacts are summarized on the basis of five subareas indicated on Exhibit 2.1-1 .
13-6	Draft EIS does not adequately consider Sound Transit's planned Ballard light rail extension.	Comment is noted and forwarded to City decision makers. The Sound Transit EIS is a different proposal from the Industrial Maritime Strategy. City staff are coordinating information and data from Sound Transit to the greatest extent possible. See responses to comment themes regarding light rail in Section 4.2.4 .
13-7	Draft EIS does not adequately assess impacts on Land & Shoreline Use.	Impacts are assessed in Section 3.8 Land & Shoreline Use.
13-8	City should assess the purely economic impacts on individual businesses and land owners.	SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450). See Section 4.2.1 .

Comment Summary	Response
All IC and IB zoned land east of 3 rd Ave NW to the Aurora bridge should remain or be changed to IC.	Land in the noted geography is retained in the IC zone under the Preferred Alternative. See Chapter 2 of the Final EIS and Appendix C .
All IB zoned land in the study area east of the Aurora bridge to I-5 should be zoned IC.	Comment is noted and forwarded to City decision makers.
Development regulations should allow bulk and dimension limitations to be met on a project- wide basis and not a parcel-by-parcel basis.	During development review development standards would be applied on the basis of the particular site consistent with current practices by SDCI.
Ciserella	Cantera Development Group
Asks OPCD to provide a true analysis of the existing conditions and consider alternatives that allow for more flexible development in light industrial zones. City should withdraw the Draft EIS and reissue a new Draft EIS.	Thank you for your comments. The comment is noted and forwarded to City decision makers. Action Alternatives include more flexible development regulations compared to existing regulations, especially in the proposed UI and II zones.
Clark	CleanTech Alliance
Appreciate the opportunity to comment. In favor of a comprehensive strategy to strengthen and grow Seattle's industrial and maritime sectors. Not offering specific comments on the various Action Alternatives—comments provide additional information that may assist in the selection of the best alternative.	Thank you for your comments. The comment is noted and forwarded to City decision makers.
Description of the CleanTech Alliance, general support for the Seattle Industrial & Maritime Strategy, and specific examples of events and programs aimed at accelerating cleantech innovations and related business development aligned with the strategy.	The comment is noted and forwarded to City decision makers.
Clawson	Interbay Urban Investors
Appreciate the opportunity to comment. Submitting on behalf of Interbay Urban Investors who own property at 2210 W Armory Way (zoned IG2 in the BINMIC). Draft EIS does not account for the existing realities of the south Interbay corridor (specifically portions that are primarily office/retail and no longer in industrial use).	Thank you for your comments. The comment is noted and forwarded to City decision makers and forwarded to City decision makers. EIS alternatives, including the Preferred Alternative would apply the proposed II zone to the noted geography, which would allow for significantly expanded development capacity and allowable uses compared to the existing IG2 zone.
Draft EIS does not address what will happen to properties in the south Interbay corridor that would become severely nonconforming.	Nonconforming uses are permitted to continue subject to provisions of the Seattle Land Use Regulations (SMC Subtitle III). See Section 4.2.2 for a comprehensive response.
City should complete an economic and affordability study that considers the impacts on housing supply and affordability of keeping land like this zoned industrial. Draft EIS must acknowledge the impact on housing displacement.	SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450). See Section 4.2.1 . The MIC requirements under VISION 2050 limit non-industrial uses including housing. The action alternatives evaluate industry supportive housing inside the MIC, and targeted areas of mixed uses outside of the MIC. One of the impact thresholds used to identify potential adverse housing impacts in the study area (see Section 3.9) and at a subarea level (where applicable) addresses displacement. Impacts
	Comment Summary All IC and IB zoned land east of 3rd Ave NW to the Aurora bridge should remain or be changed to IC. All IB zoned land in the study area east of the Aurora bridge to I-5 should be zoned IC. Development regulations should allow bulk and dimension limitations to be met on a project- wide basis and not a parcel-by-parcel basis. Ciserella Asks OPCD to provide a true analysis of the existing conditions and consider alternatives that allow for more flexible development in light industrial zones. City should withdraw the Draft EIS and reissue a new Draft EIS. Clark Appreciate the opportunity to comment. In favor of a comprehensive strategy to strengthen and grow Seattle's industrial and maritime sectors. Not offering specific comments on the various Action Alternatives—comments provide additional information that may assist in the selection of the CleanTech Alliance, general support for the Seattle Industrial & Maritime Strategy, and specific examples of events and programs aimed at accelerating cleantech innovations and related business development aligned with the strategy. Clawson Appreciate the opportunity to comment. Submitting on behalf of Interbay Urban Investors who own property at 2210 W Armory Way (zoned IG2 in the BINMIC). Draft EIS does not account for the existing realities of the south Interbay corridor (specifically portions that are primarily office/retail and no longer in industrial use). Draft EIS does not address what will happen to properties in the south Interbay corridor that would become severely nonconforming. City should complete an economic and affordability study that considers the impacts on housing supply and affordability of keeping land like this zoned industrial. Draft EIS must acknowledge the impact on housing displacement.

Number	Comment Summary	Response
		 Result in loss of nousing due to redevelopment and insufficient development capacity, tools, or programs to address displacement of dwellings and population.
		lower risk of displacement. With the II zone there is an opportunity to apply MHA regulations to address demand for and funding of affordable housing.
		See also Section 4.2.10.
16-4	Draft EIS does not analyze if the south Interbay corridor is well suited for industrial use under City and VISION 2050 criteria.	Section 3.8 Land & Shoreline Use discusses consistency of alternatives with City and regional policies, as well as land use compatibility impacts.
16-5	Draft EIS fails to analyze the impact on loss of multimodal transit, and the climate implications.	The City of Seattle agrees that a main contributor of climate gases in the Pacific Northwest is from transportation/cars, and that combining housing and transportation together is one of the main strategies to reduce climate emissions. Section 3.2 Air Quality & GHG evaluates the potential air quality and greenhouse gas impacts associated with the action alternatives compared with the No Action Alternative.
		As referenced in WAC 197-11-442(4), "The EIS's discussion of alternatives for a comprehensive plan, community plan, or other areawide zoning or for shoreline or land use plans shall be limited to a general discussion of the impacts of alternate proposals for policies contained in such plans, for land use or shoreline designations, and for implementation measures. The lead agency is not required under SEPA to examine all conceivable policies, designations, or implementation measures but should cover a range of such topics. The EIS content may be limited to a discussion of alternatives which have been formally proposed or which are, while not formally proposed, reasonably related to the proposed action."
		The City believes that the analysis of impacts and mitigation measures conforms to the requirements cited above, and that the analysis covers a reasonable range of actions that may result from implementation of the Industrial and Maritime Strategy, including the potential for a different land use designation of the property referenced in the comment.
16-6	Air quality and noise impacts on Interbay in general and for the property at 2210 W Armory Way, specifically.	Additional text has been added to Section 3.2.2 to reflect potential air quality impacts to adjacent residential and mixed-use land uses from areas that continue to maintain an industrial focus under the proposed alternatives (Section 3.2 Air Quality & GHG).
		See Section 3.6.2 for a discussion of potential noise impacts on residential or mixed use land uses adjacent to existing industrial areas or areas that will remain industrially focused in all MIC subareas under the proposal (Section 3.6 Noise).
16-7	Consider environmental and stormwater impacts if redevelopment does not occur in the south Interbay corridor and specifically on the property at 2210 W Armory Way.	Additional text has been added to Section 3.3.2 to reflect stormwater requirements, in general, for industrial parcels that do not redevelop (Section 3.3 Water Resources).
16-8	Draft EIS should consider the Sound Transit Draft EIS and light rail alignment option.	The comment is noted and forwarded to City decision makers. The Sound Transit EIS is a different proposal from the Industrial Maritime Strategy. City staff are coordinating information and data from Sound Transit to the greatest extent possible. See Section 4.2.4 .

Number	Comment Summary	Response
16-9	Draft EIS is not clear about what will happen to the armory property.	The proposal includes a policy change calling for collaborative master planning of the Armory site. The site is within the MIC, and the proposal is that updated MIC policies and industrial zone designations will apply to the site. Should the State and partners wish to pursue non-industrial future uses, that would be determined in the master plan in partnership with the City and other entities.
16-10	Alternatives should consider the relative impacts of removing the south Interbay corridor and Armory property from industrial designation as almost the entirety of the corridor is no longer in industrial use.	The comment is noted and forwarded to City decision makers. The proposal includes a policy change calling for collaborative master planning of the Armory site. The site is within the MIC, and the proposal is that updated MIC policies and industrial zone designations will apply to the site. Should the State and partners wish to pursue non-industrial future uses, that would be determined in the master plan in partnership with the City and other entities. An existing land use analysis is included in Section 3.8 Land & Shoreline Use . The City's proposed action intentionally limits removal of land from a MIC to focused locations in the South Park and Georgetown neighborhoods.
17	Clawson	Madisonian Manager, LLC
17-1	Appreciate the opportunity to comment. Submitting on behalf of Madisonian Manager who own property at 900 Poplar Place S (zoned IC-65(M) outside of an MIC). Draft EIS should study taking this property out of industrial zoning and allow housing (similar to adjacent properties).	Thank you for your letter. The comment is noted and forwarded to City decision makers. The Preferred Alternative applies a mixed use zone that would allow housing to a portion of the noted geography.
17-2	City should complete an economic and affordability study that considers the impacts on housing supply and affordability of keeping land like this zoned industrial. Draft EIS must acknowledge the impact on housing displacement and land use conflicts as IC zoning does not currently allow for residential uses.	SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450). See Section 4.2.1 , Section 4.2.10 , and Section 4.2.11 .
17-3	Draft EIS does not address land use conflicts if the property is kept industrial.	Section 3.8 Land & Shoreline Use discusses land use compatibility impacts under all alternatives including No Action.
17-4	Draft EIS does not analyze if this area is well suited for industrial use under City and VISION 2050 criteria.	Section 3.8 Land & Shoreline Use discusses consistency of alternatives with City and regional policies, as well as land use compatibility impacts
17-5	Draft EIS fails to analyze the impact on loss of multimodal transit, and the climate implications.	See response to comment 16-5.
17-6	Air quality, noise pollution, and environmental justice issues are not addressed in the context of this property.	See response to comment 16-6.
17-7	Consider environmental and stormwater impacts if redevelopment does not occur at 900 Poplar Place S.	Additional text has been added to Section 3.3.2 to reflect stormwater requirements, in general, for industrial parcels that do not redevelop (Section 3.3 Water Resources).
17-8	City must take the climate and housing crisis seriously when drafting these policies. The Judkins Park area is not suited for industrial uses.	Comment is noted and forwarded to City decision makers. The EIS recognizes climate change as one of the six key emerging factors affecting Seattle's MICs and addresses various climate change related impacts (sea level rise, increased floods, extreme heat) in

Number	Comment Summary	Response
		the analysis. Section 3.9 Housing address nousing impacts and proposed mitigation. The Preferred Alternative would allow for mixed use housing in Judkins Park.
18	Clawson	AnMarCo
18-1	Appreciate the opportunity to comment. Submitting on behalf of AnMarCo who own property at 2130 Harbor Ave SW ("Pier One" property zoned IG2 in the Duwamish MIC). Draft EIS should study taking this property out of industrial zoning or rezoned IC because of specific conditions.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See Section [].
18-2	Pier One property does not meet the criteria of "industrial land" defined in VISION 2050, the King County CPPs, and the City's own criteria.	Consistency of alternatives with city and regional policies is discussed in Section 3.8 Land & Shoreline Use .
18-3	Any jobs analysis that includes contribution from the Pier One property is faulty. The property has not created any jobs in 30 years.	Employment growth projections are for aggregated areas and specific quantities are not attributed to individual parcels. The overall quantity of redevelopable parcels in a subareas is one factor in the employment growth projections model.
18-4	Draft EIS does not consider changes that would need to be made to the shoreline environments to achieve any of the proposed alternatives for properties in the shoreline.	See minor revision to Section 3.4.3 acknowledging that development within the shoreline would need to comply with existing federal, state, and local regulations. The EIS lists relevant statutes and agencies (Exhibit 3.4-3). The degree of difficulty relating to industrial development depends greatly on the individual project and would be addressed during environmental review and permitting at the project level.
18-5	Consider the economic impacts of leaving Pier One property in the MIC and zoned industrial, including blight. Consider the environmental impacts associated with properties staying in the MIC and remaining undeveloped.	SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450). See Section 4.2.1 . Additional text has been added to Section 3.4.2 addressing impacts of failure to redevelop properties that have existing environmental impacts. See Section 3.4 Plants & Animals .
18-6	Consider the visual and aesthetic/view impacts of leaving Pier One property in the MIC and zoned industrial. Harbor Avenue SW is a designated SEPA view corridor.	Please see the discussion of scenic routes and the alternatives in Section 3.7 Light & Glare. The view from parks and view corridors in the West Seattle Area is addressed in the discussion of the SODO/Stadium Subarea under each alternative, including the Preferred Alternative.
18-7	Pier One property contains environmental contamination. Property will not be cleaned up if remains in the MIC and zoned industrial.	Comment is acknowledged. See response to comment 16-5 regarding the appropriate level of analysis completed under this EIS. For contaminated sites with current industrial land use designations that maintain an industrial focus under new land use designations, cleanup will not likely happen until redevelopment occurs, or there is a property sale that triggers site characterization and remediation activities to secure financing. Added text to this effect to Section 3.5 Contamination .
18-8	Consider the land use conflicts of leaving Pier One property in the MIC and zoned industrial.	Section 3.8 Land & Shoreline Use discusses land use compatibility.
18-9	Existing over-water structure at Pier One has negatively impacted the shoreline environment since 1905.	The non-project EIS considers future development allowed under the No Action Alternative as well as action alternatives and associated policies and regulations. With development or redevelopment, modern regulations addressing shorelines, stormwater, etc. could apply under any alternative. Section 3.4.2

Number	Comment Summary	Response
		nas been clarified regarding detrimental impacts of existing properties prior to redevelopment across the study area.
18-10	Pier One property does not have stormwater infrastructure on-site. Impact will remain without redevelopment.	See Section 3.3.2 which is clarified in the Final EIS to reflect stormwater requirements, in general, for industrial parcels that do not redevelop (Section 3.3 Water Resources). These sites would still be required to implement stormwater source control measures, even if no redevelopment occurs.
19	Daniels	First and Utah Street Associates, LLC
19-1	Appreciate the opportunity to comment. First & Utah has deep roots in SODO and owns several properties in the area.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
19-2	Support the II zone. Preferred Alternative should support legacy businesses near light rail investments and allow for modern industrial uses with an expanded and modified II zone.	The comment is noted and forwarded to City decision makers. See response to comments 19-3 through 19-9 below.
19-3	Maximum height limit in the II zone should be increased to a minimum of 180' to allow for innovative and sustainable mass timber construction types.	The comment is noted. A maximum height of 160' is studied for the II zone and could accommodate mass timber construction. See Section 2.4.2 and Appendix G .
19-4	Maximum FAR in the II zone achievable via the mixed development bonus program should be increased to at least 6-7 FAR to allow buildings to achieve the increased maximum height limit by stacking density to provide needed accompanying amenities.	The comment is noted. A maximum FAR of 6.0 is studied for the II zone. See Section 2.4.2 and Appendix G .
19-5	List of industrial uses in the current code should be used as the basis for uses qualifying for the mixed development bonus program in the II zone.	The comment is noted. Additional information regarding qualifying and bonus allowable uses in the II zone is provided in the Final EIS in the development standards Appendix G .
19-6	City should set rules around ancillary uses in the II zone that look at several factors like the actual function of spaces, use of technology, and the overall purpose of the business in a space (rather than just size of uses).	The comment is noted and forwarded to City decision makers.
19-7	City should publish the Joint Director's Rule contemplated by SMC 23.52.004.B. so that property owners can properly evaluate the available mitigation measures to help achieve the 51% SOV goal in the Duwamish MIC and similar areas.	The comment is noted and forwarded to City decision makers.
19-8	Retain the general exemption from design review in most industrial zones and extend this exemption to the II zone.	The comment is noted and forwarded to City decision makers. No expansion of design review to industrial zones is proposed.
19-9	Preferred Alternative zoning map should be amended to apply the II zone to all of First & Utah's property within a half mile of light rail.	The comment is noted and forwarded to City decision makers. Different alternatives include varying geographies for the II zone including coverage of noted properties.
20	Ffitch	BNSF Railway Company, Freezer Longline Coalition, ILWU Local 19, Inlandboatmen's Union of the Pacific, Manufacturing Industrial Council, North Seattle Industrial Association, Pacific Merchant Shipping Association, Port of Seattle, Puget Sound

Number	Comment Summary	Response
		Pliots, Seattle Marine Business Coalition, The Northwest Seaport Alliance, Transportation Institute, Vigor, Washington Maritime Federation, Port of Seattle, Puget Sound Pilots, Seattle Marine Business Coalition, The Northwest Seaport Alliance, Transportation Institute, Vigor, Washington Maritime Federation
20-1	Support Alternative 2.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
20-2	 (1) Document needs to address freight. Can the City engage the Freight Board to ensure that freight concerns are reflected in the final document? (2) Address mobility concerns between major truck streets and the connections to business 	(1) The Final EIS has been reorganized to include freight as a separate section with additional information.
		(2) The Final EIS has been reorganized to include freight as a separate section including the commenter's concerns about existing mobility challenges between major truck streets and the connections to business driveways.
	driveways. (3) Final EIS must differentiate between car vs truck safety and discuss safety issues posed by sight distance and turning radius conflicts between heavy trucks and bicycles and pedestrians.	(3) Language distinguishing between the safety risks of cars and trucks was included in the Draft EIS (p. 3-388) and is retained in the Final EIS. Supplemental language has also been added in the Final EIS per the commenter's suggestion regarding truck drivers' limited range of sight distance and turning radii conflicts.
20-3	City must adopt policies and regulations that implement elements of Alternative 2 to promote diversity of economic opportunity, as is represented by industrial jobs.	The comment is noted and forwarded to City decision makers. The EIS recognizes equity and accessibility as one of the six key emerging factors affecting Seattle's MICs, specifically access to maritime and other industrial career opportunities for BIPOC and women. Maintaining a strong industrial economy is a prerequisite to providing these opportunities, but other non-land use strategies including outreach to BIPOC youth and workforce training investments are key parts of the Industry and Maritime Strategy outside of topics required to be analyzed in this EIS under SEPA. The EIS estimates employment growth including estimation of the proportion of employment industrial and non-industrial categories.
20-4	Final EIS should affirm that increased density in current residential areas is preferable to bringing new residents into and alongside the MICs.	Comment is noted and forwarded to City decision makers. Consistent with the PSRC criteria for designating MICs to focus industrial uses in the MIC, the EIS does not study allowing residential uses in the majority of the study area. Alternatives 3 and 4 consider limited additional flexibility of existing allowances for caretakers' units and artist/studio quarters in the proposed UI zone only. The Preferred Alternative limits housing growth to less than Alternative 4.
21	Fu	NAIOP Washington State
21-1	Final EIS should consider the Port of Seattle and Boeing Field as separate industrial uses, delineate between industrial and uses that are heavy commercial or commercial, and acknowledge vacant or interim-use industrial buildings. Should also more robustly study the No Action Alternative.	Thank you for your letter. Section 3.8 Land & Shoreline Use includes an analysis of existing land use, including narrative descriptions of subarea land use patterns under existing conditions.
21-2	Existing code-based definition of "industrial use" is out of date.	The EIS considers three proposed new industrial zones based on community input that are intended to respond to issues, challenges, and opportunities for the maritime and industrial sectors and adjacent communities (MML, II, and UI zones). The action alternatives apply these proposed "future of industry" land use concepts to the city's industrial areas. The EIS will eventually

Number	Comment Summary	Response
		neip the City develop a proposal that will identity specific zone standards including uses.
21-3	Draft EIS fails to address that many industrially zoned areas in Seattle have few industrial uses.	Section 3.8 Land & Shoreline Use includes an analysis of existing land uses.
21-4	Draft EIS fails to address that many industrially zoned areas in Seattle have few industrial uses.	Section 3.8 Land & Shoreline Use includes an analysis of existing land use.
21-5	Draft EIS does not analyze the relationship between future light rail service and industrial zoned land with non-industrial uses impact on TOD.	The EIS alternatives include a range of additional employment densities at existing and future light rail stations with a focus on a land use concept of transit-oriented employment or industrial TOD (see also Objective F of the proposal). The II land use concept is intended to support economic innovation and capitalize on emerging opportunities including expanded or new light rail stations in industrial areas. Section 3.8 Land & Shoreline Use discusses the relationship of likely future land use with future light rail stations under each alternative.
21-6	Support continuing to not require design review in industrial areas.	The comment is noted and forwarded to City decision makers. No expansion of design review to industrial zones is proposed.
21-7	Final EIS should study increased density for all II zoned property near future and current light rail stations with height limits increased to 180'.	Comment is noted. Il zones are studied in alternatives in varied geographies near future light rail. Height limits up to 160' are studied. See Section 2.4.2 and Appendix G .
21-8	Limiting future removal of land in the MIC and BINMIC to every 8 years is onerous.	The comment is noted and forwarded to City decision makers.
21-9	Permitted light industrial uses need to be broader and more flexible.	Additional information about specific uses qualifying as industrial is included in the development standards Appendix G in the Final EIS. A new definition for Information Computer Technology (ICT) is proposed and would be eligible as an industrial use in the II zone.
21-10	Final EIS should explore alternatives that study: all urban industrial lands with residential allowances of Seattle Mixed Use zoning; Interbay and non-water dependent Ballard land within BINMIC as II; Interbay and non-water dependent Ballard land within BINMIC as UI; adding all non BINMIC Ballard lands as Seattle Mixed Use zoning; adding all non BINMIC Ballard lands as UI with housing option; adding land around Lake Union, outside of the BINMIC as II, UI, and Seattle Mixed; and the impact of removing non-industrial limitation caps in UI zones.	The comment is noted and forwarded to City decision makers. The EIS studies a range of varied patterns of the proposed UI and II zones. Different alternatives feature varied allowances for housing within the UI zone. See Section 4.2.10 .
21-11	Believe the current EIS falls short of analyzing several key components necessary for a comprehensive study of Seattle's industrial lands to be accurate and inform new zoning and land use codes.	The comment is noted and forwarded to City decision makers. See response to comments 21-1 through 21-10 above.
22	Gering	Manufacturing Industrial Council
22-1	The Draft EIS presents an opportunity for the City to build on success of an industrial career learning initiative already in place and ready to grow in the Seattle Public Schools.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See response to comments 22-2 through 22-9 below.

Number	Comment Summary	Response
22-2	Requests a meeting with the Mayor and his staff team regarding the opportunity to incorporate the Seattle Public Schools into the Seattle Industrial Maritime Strategy. Agree with letters submitted by the Seattle Freight Advisory Board and the Port of Seattle. Specific concerns with the significant increases in residential and worker populations under alternatives 3 and 4.	The comments are noted and forwarded to City decision makers.
22-3	Draft EIS fails to account for aging infrastructure, including updates needed to accommodate increased truck and rail traffic and potential impacts of a major earthquake.	Section 3.10.1 Transportation identifies the City's Transportation Capital Improvement Program which include developing, maintaining, and operating Seattle's transportation system including truck and freight as well as roads and bridges. During an earthquake, vertical and lateral displacements of structures, embankments, and paved areas might occur due to seismic liquefaction hazard. The liquefaction potential of mapped liquefaction hazard areas would be confirmed during the design stage of proposed development, regardless of the alternative (see Section 3.1 Soils/Geology). Text was added to Section 3.1.2 Impacts describing how structures, all water, wastewater, transportation, and other infrastructure associated with new development and redevelopment would be carefully designed with input from site-specific geotechnical investigations to lessen and withstand the effects of earthquakes and liquefaction. The City of Seattle maintains a Comprehensive Emergency Management Plan (CEMP) which unifies a series of all-hazards documentation to holistically describe the doctrines, strategies, and responsibilities through which the City of Seattle's emergency management system is organized and managed. In addition, the City's Disaster Recovery Framework addresses how the City would partner with the community and coordinate with County, State, and Federal agencies in recovering from the effects of disaster using a massive earthquake as the premise.
22-4	Draft EIS should more fully address climate concerns, including conflicts with residential uses from noise and light impacts.	As discussed in WAC 197-11-440, this non-project EIS is limited to a general discussion of the impacts of alternate proposals for policies contained in the proposed Industrial and Maritime Strategy. The City of Seattle concluded that as proposed, the alternatives would not prevent or deter efforts to reduce emissions in comparison to local or regional goals or targets for GHG reductions. See Section 3.2 Air Quality & GHG . The current level of analysis provides an appropriate level of detail for a non-project EIS. Subsequent developments that may arise from the proposed land use changes in the Industrial and Maritime Strategy will be required to meet all applicable codes and regulations, and to conduct project-level SEPA review at that time, in which analysis will be conducted to assess site specific impacts and necessary mitigation measures, including for climate change related issues. See Section 3.6.2 for a discussion of potential impacts associated with the location of noise sensitive receivers like residential uses near industrial or traffic noise sources under all alternatives, particularly alternatives 3 and 4 and the Preferred Alternative. The City of Seattle concluded that as proposed, implementation of the prescribed residential noise mitigation in general should

Number	Comment Summary	Response
		adequately reduce noise experienced by noise sensitive receivers. It is beyond the scope of this analysis to speculate about specific potential complaints or remedies.
		The current level of analysis provides an appropriate level of detail for a non-project EIS. Existing operations and subsequent developments that may arise from the proposed land use changes in the Industrial and Maritime Strategy are or will be required to meet all applicable codes and regulations, and to conduct project- level SEPA review at that time, in which analysis will be conducted to assess site specific impacts and necessary mitigation measures, including for noise.
		See Section 3.7.2 for a discussion of potential impacts and mitigation measures associated with light and glare. Future development could generate at least some increase in light and glare, but these effects can be minimized and reduced through application of design standards and the mitigation measures addressing placement, light output, direction, and shielding of any exterior illumination above a given height to reduce light and glare emissions to adjacent non-industrial areas.
22-5	Increased traffic will result in increased non- point source stormwater pollution from roadways with no mitigation offered in the Draft EIS.	Section 3.3.2 discusses the expected increase in traffic for all alternatives and states that improvements in vehicle standards and the application of stormwater requirements during redevelopment described in this and other sections of the EIS are expected to offset the increase in traffic and potentially lead to a net decrease in surface water pollution.
22-6	Industrial soil cleanup levels cannot be applied in areas near residential and other vulnerable populations. Parcels cleaned up to industrial standards must have a wide buffer zone and be protected from upzoning in the future.	As described in Section 3.5 Contamination , site characterization and remediation occur on a site-specific basis and the cleanup standards applied under MTCA are tied to the current land use. However, as described in Section 3.8 Land & Shoreline Use , one of the goals of the City of Seattle 2035 Comprehensive Plan is to develop better transitions between industrial areas and adjacent neighborhoods that support healthy communities, reduce adverse environmental impacts, and minimize land use conflicts.
22-7	Request the Mayor engage in Core Plus career learning opportunities at Seattle schools. Believe a leadership intervention is necessary to achieve stakeholder goals for more equitable access to high-wage industrial careers.	Comment is noted and request is forwarded to the mayor's office. Non-land use actions outside the scope of what is required to be analyzed under SEPA are being pursued in parallel with the proposed action. This includes workforce development and career pathway efforts largely led by Seattle's Office of Economic Development (OED).
23	Gilder	Gilder Office for Growth, LLC
23-1	Endorse comments in letter 34.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See responses to letter 34.
24	Gillespie	Lander Street Partners, LLC
24-1	Alternatives do not provide incentive for industrial development, TOD, or large scale redevelopment of existing structures.	Thank you for your letter. The comment noted and forwarded to City decision makers. The II zone includes significantly increased development capacity and flexibility compared to existing IG zoning standards. In response to comments II standards under the Preferred Alternative are modified to provide additional incentive. See development standards Appendix G .
25	Gillespie	Hess Callahan Grey

Number	Comment Summary	Response
25-1	Appreciate the opportunity to comment. Submitting on behalf of Hess Callahan Grey who develop and manage properties in Fremont's industrial areas.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
25-2	Wallingford, UW MIO, and Silicon Canal area in Fremont should be considered uniquely from the Ballard Subarea.	Unique land use conditions in Fremont and other noted areas are described in Section 3.8 Land & Shoreline Use . The Preferred Alternative applies an approach that is distinct from the Ballard areas, since the alternative proposes to retain Industrial Commercial zones for the geography noted in the comment. See Chapter 2 and Appendix C maps.
25-3	Silicon Canal area in Fremont should be considered separate from the Ballard Subarea.	See response to comment 25-2 above.
25-4	Proposal is not adequately defined. No SEPA review draft ordinance published by OPCD.	This is a programmatic level EIS. Sufficient detail about the proposed development standards to fully consider the potential for environmental impacts is included (see Section 2.4.2). The Draft EIS includes sufficient detail about proposed development standards and potential zoning changes to understand the scale and physical characteristics and likely use patterns from the development that would occur. It is not possible to predict the exact features of new development over a future 20-year time horizon on a wide range of sites and geographic areas. Additional detail beyond the level that would be required for a programmatic EIS is included about fine-grained development standards in this Final EIS. In association with the Preferred Alternative, detail about development standards is included in Appendix G , which are similar to the Draft EIS action alternative concepts.
25-5	Selection of alternatives does not highlight the environmental impacts of any proposed action and limits the choice of reasonable alternatives Council can consider.	Per WAC 197-11-442, a non-project EIS is "not required under SEPA to examine all conceivable policies, designations, or implementation measures but should cover a range of such topics." The alternatives include a range of different geographic patterns of proposed zoning designations. Development standards are also varied between alternatives. Action alternatives are compared to a No Action Alternative.
25-6	Draft EIS does not analyze if the market will support any development under UI size-of-use limits.	The comment is noted is forwarded to City decision makers. See Section 4.2.1 .
25-7	Draft EIS does not meaningfully analyze the interplay between the Action Alternatives and the Shoreline Master Program.	Section 3.8 Land & Shoreline Use discusses the interplay between proposed development standards and Shoreline Master Program regulations.
25-8	Draft EIS is inadequate because it needs a clear proposal and unique consideration of the Silicon Canal.	See response to comments 25-2 through 25-7 above.
26	Goodman	SODO BIA
26-1	Transportation section is missing the subject of freight including trucks and rail. Include truck and rail existing conditions, future no action, and future action conditions. Include relevant basis for analysis from the City of Seattle Freight Master Plan. Include potential future operating policies on rail lines	Thank you for your letter. The Final EIS has been reorganized to include freight as a separate section with additional information. Future operating policies on privately operated rail lines (for example, train speed, train horn noise, blocked/occupied at-grade rail crossings) is not within the purview of the City and this EIS.

Number	Comment Summary	Response
 (1) Final EIS should present daily trips general by the alternatives and the subsequent mode split throughout the day. (2) Final EIS should present the changes in data and PM peak hour traffic on study area street (3) Final EIS should present extent of peak hour. 	(1) Final EIS should present daily trips generated by the alternatives and the subsequent mode split throughout the day	(1) A trip summary table including daily trips and a more detailed breakdown of mode split has been added to the Final EIS.
	(2) Final EIS should present the changes in daily and PM peak hour traffic on study area streets.(3) Final EIS should present extent of peak hour spreading and show the daytime peak hour.	 studied screenline during the PM peak hour has been added to the Appendix I of the Final EIS, consistent with methodology from a prior Comprehensive Plan. However, it should be noted the travel time on individual roadways was used as the main indicator of congestion. That analysis indicated conditions would generally be LOS E and F in the SODO area with slightly higher travel times (i.e., more congestion) under the action alternatives because of higher vehicle trip generation. (3) As disclosed in the EIS, peak spreading is expected to occur in locations that are already at capacity. There is an inherent
		and the precise duration, while informative, would not change the identification of an impact.
26-3	Document should prepare text describing the operating conditions for each level of service.	Text qualitatively describing the operating conditions for each level of service has been added to the Final EIS.
26-4	Document should analyze impact of daily traffic generated by alternative conditions, midday conditions, and peak hour spreading.	See response to comment 26-2 regarding daily traffic and peak hour spreading. See response to comment 4-27, part 3 regarding analysis period.
26-5	 (1) Document should present rail operating conditions, operating policies, frequency, and length of time streets are blocked during daytime and PM peak hour conditions. Present existing conditions data for queuing and delays when streets are blocked, and future conditions that could occur through railroad action. (2) Future No Action should disclose the status of Holgate being removed for general-purpose traffic by the railroad. 	 (1) The Final EIS provides additional information about rail conditions in the study area. (2) The City is in communication with railroad operators regarding their future operational plans and how they could affect City roadway operations. This includes discussion of potential changes at the Holgate Street crossing. Should changes at Holgate Street move forward, SDOT will conduct a study of potential impacts to the area.
26-6	(1) Final EIS does not state embedded assumption that for alternatives 3 and 4, employees are traveling to work by transit, walking, or biking.	(1) The model does not assume that employees are only arriving by transit, walking, or biking. A trip summary table with a more detailed breakdown of vehicle trip growth and mode split has been added to the Final EIS.
	(2) Final EIS should discuss relevant conditions and traffic impacts in MICs before full buildout of Sound Transit Phase 3	(2) Text has been added to the Final EIS qualitatively addressing how interim conditions may compare to the EIS 2044 horizon year.
 (3) Draft EIS particular transit mode of the transit set of the transi	(3) Draft EIS page 3-366 includes a text box titled Travel Patterns of Industrial Workers which includes statistics about the geographic distribution of study area workers and acknowledges that accessing transit may be a challenge due to the availability and convenience of the transit service.	
26-7	Document should add personal safety for transit riders in the safety section.	Personal safety at transit stops is not expected to be adversely impacted by the action alternatives and could potentially result in a safety benefit by concentrating more land uses and activity near transit stops, i.e., more "eyes on the street" as the comment states. This concern among MIC employees has been added to the Travel Patterns of Industrial Workers text box as an additional existing challenge to transit use.
Number	Comment Summary	Response
--------	---	--
26-8	Final EIS should acknowledge the need for parking along street frontages and the limitations of a qualitative parking analysis.	The EIS includes a Parking impacts section describing the competing needs for public curb space and acknowledges that the action alternatives are expected to result in significant adverse impacts to on-street parking absent mitigation measures. This programmatic EIS addresses area-wide land use zoning changes, rather than a project-specific proposal. The proposal may result in a wide range of individual projects implemented over a long timeframe and across a large geographic area. Because the specific locations and sizes of development are unknown at this time, it would be speculative to quantify parking demand in a particular location. Individual development projects will undergo separate and more detailed SEPA review during which specific impacts and mitigation (including on-street parking) will be determined.
26-9	Final EIS should include an equity analysis focused on the quality of employment and access to that employment by alternative.	The EIS recognizes equity and accessibility as one of the six key emerging factors affecting Seattle's MICs, specifically access to maritime and other industrial career opportunities for BIPOC and women. Maintaining a strong industrial economy is a prerequisite to providing these opportunities, but other non-land use strategies including outreach to BIPOC youth and workforce training investments are key parts of the Industry and Maritime Strategy outside of topics required to be analyzed in this EIS under SEPA. The EIS estimates employment growth including estimation of the proportion of employment in industrial and non-industrial categories. Section 1.7.15 of the EIS is an equity and environmental justice review.
26-10	The transportation mitigation section is very general and not tangible to the average person.	This programmatic EIS addresses area-wide land use zoning changes, rather than a project-specific proposal. The proposal may result in a wide range of individual projects implemented over a long timeframe and across a large geographic area. Because the specific locations and sizes of development are unknown at this time, the specific mitigation projects that will be required are also unknown. Individual development projects will undergo separate and more detailed SEPA review during which specific impacts and mitigation will be determined.
26-11	Document should acknowledge that standards should be developed for industrial and maritime uses; there is risk in approaching the standard primarily for pedestrians and cyclists.	If new zoning designations are adopted, SDCI will work with SDOT to develop updates to the Streets Illustrated manual reflecting street design standards tailored to the industrial context and level of expected pedestrian and bicycle activity. Updates will consider street typologies and design standards that can accommodate both freight activity and non-motorized uses with a focus on reducing potential conflicts.
26-12	Document should provide text that acknowledges the parking and vehicular curbside access needs for commercial and industrial uses in mitigation section.	The parking and curbside access needs findings from the Impacts section has been summarized at the beginning of Parking Strategies in Section 3.10.3 Mitigation Measures in the Final EIS. The SDOT Curbside Management Team actively identifies and installs commercial vehicle and general load unload zones in business districts throughout Seattle and would identify load zone needs with new development as needed or requested by development projects. SDOT is also working on potential policy changes to more actively install load zones and other curb access needs at new development during the City development review process.

Number	Comment Summary	Response
26-13	 (1) Clarify intent of "proximity to a light rail station—Industry & Innovation" (2) In "Regulations and Commitments" section, TSMO, TDM, and Parking Strategies are system management, not mitigation. 	(1) The bulleted list on page 3-419 of the Draft EIS summarizes the transportation-related aspects of the proposals, i.e., that the II land use concept would be located within close proximity to light rail stations, making travel by transit more convenient. The sentence that precedes that list has been clarified in the Final EIS.
	(3) Prepare text that acknowledges the deteriorating conditions in the No Action alternative. Describe programmed projects that would mitigate future No Action conditions.	(2) Section 3.10.3 Mitigation Measures proposes a variety of strategies, not solely limited to street infrastructure. TSMO measures would mitigate traffic congestion impacts identified in the EIS by better operating the City's existing infrastructure and systems. TDM measures would mitigate traffic congestion impacts identified in the EIS by lowering the vehicle demand on the network. Parking Strategies in Section 3.10.3 Mitigation Measures describe the way the City can manage the public curbspace to meet competing demands for its use.
		(3) See page 4-416 of the Draft EIS for a discussion of conditions under the No Action Alternative and the purpose of this programmatic EIS: <i>"The purpose of this EIS is to disclose how potential</i> <i>actions by the City may impact the transportation system in</i> <i>comparison to what is expected to occur with currently adopted zoning</i> <i>codes and development standards. Therefore, the impacts of the Action</i> <i>Alternatives are assessed against Alternative 1 No Action. Impacts</i> <i>identified under Alternative 1 No Action would remain throughout the</i> <i>Action Alternatives even if those alternatives would not result in</i> <i>additional impacts. While the focus of the EIS is not to mitigate</i> <i>conditions under the currently adopted zoning code and development</i> <i>standards (i.e., Alternative 1 No Action), many of the mitigation</i> <i>measures identified for the Action Alternatives would also benefit</i> <i>conditions under Alternative 1 No Action.</i>
		In summary, Alternative 1 No Action is expected to have significant impacts to active transportation, auto, and freight in terms of travel time, mode share, transit, parking, and safety."
26-14	 Include temporary traffic signal at Forrest/4th Ave S as potential mitigation. Add mitigation measure to improve personal safety of transit riders. 	 (1) The commenter's request for a signal at the Forrest/4th Ave S intersection is noted. The City does not anticipate installing a signal in the near term, but will continue to monitor the location to determine if it meets a signal warrant in the future. (2) See response to comment 26-7.
26-15	Comments and requests for additional methodology, data, analysis of impacts, and mitigation are based on the missing information relative to the unique needs of commercial and industrial land uses in the Greater Duwamish MIC.	The comment is noted. See response to comments 26-1 through 26-14 above.
26-16	Include daily trips generated by the alternatives and mode split in absolute numbers, changes in daily and PM peak hour traffic on streets in the study area, and the extent of peak hour spreading.	See response to comment 26-2.
27	Horn	MAK Management, LLC
27-1	General background on MAK Management, LLC and the properties they represent.	Thank you for your letter. The comment noted and forwarded to City decision makers.
27-2	Development standards aren't fully disclosed in the Draft EIS.	This is a programmatic level EIS. Sufficient detail about the proposed development standards to fully consider the potential for

Number	Comment Summary	Response
		environmental impacts is included. The Draft EIS includes sufficient detail about proposed development standards and potential zoning changes to understand the scale and physical characteristics and likely use patterns from the development that would occur. It is not possible to predict the exact features of new development over a future 20-year time horizon on a wide range of sites and geographic areas. Additional detail beyond the level that would be required for a programmatic EIS is included about fine-grained development standards in this Final EIS. Detail about development standards is included in Appendix G .
27-3	No development feasibility analysis is included in the EIS.	The comment noted and forwarded to City decision makers. See Section 4.2.1 .
27-4	Supports zoning that would allow stacked mixed uses.	The comment is noted. The proposed UI zone would allow mixing of uses, and would allow large allowances for ancillary office and other non-industrial uses if affiliated with an industrial operation. The II zone would encourage investment in non-industrial uses if mixed in a development with light industrial uses. Overall development capacity in both zones would be increased compared to existing regulations in the Industrial General and Industrial Buffer zones.
27-5	Ancillary brewing/tasting rooms should be allowed on adjacent or other sites.	The comment is noted and forwarded to City decision makers. The Final EIS includes additional detail about development standards in Appendix G . In response to this comment Appendix G describes an allowance for ancillary brewing/tasting rooms to be located off- site within the same MIC.
27-6	Proposed size of use limits are too small.	See response to 27-5. See Section 4.2.2 concerning non- conforming uses. In the UI zone standards allow large ancillary spaces. In the II zone bonus non-industrial spaces would not be subject to a maximum size of use limit.
27-7	The suggested 1/1000 maximum parking limit for the II zone will create significant impacts for non-industrial uses away from transit.	The Draft EIS identified potential significant adverse impacts to on- street parking under all alternatives (p. 3-386). Decisions on the use of any particular flex zone (i.e., whether it's used for freight loading, passenger loading, bus stops, parking, etc.) will be made by SDOT depending on the specific context of the block face, including needs of adjacent land uses and the transportation activity/network in that location.
		However, the City also has a variety of strategies available to mitigate these potential impacts—see Parking Strategies in Section 3.10.3 Mitigation Measures . Other strategies such as travel demand management, continued expansion of transit service, and improvements to active transportation modes will also provide more travel options for people traveling to and from the study area. With a combination of those approaches tailored to each specific location's needs, it is expected that parking impacts could be brought to a less-than-significant level.
27-8	The EIS does not assess how proposed maximum size of use limits in the MML zone would affect surrounding areas with respect to creating more demand for office and other uses.	The comment noted and is forwarded to City decision makers. Section 3.8 Land & Shoreline Use includes analysis of transitions impacts, which addresses potential for impacts on adjacent areas. If size of use limits caused increased demand for non-industrial uses such as offices in other areas that are zoned for offices and non-industrial uses, this would not be considered an adverse impact. Additionally, the proposal creates new development

Number	Comment Summary	Response
		capacity under the action alternatives for offices, especially in the II zone, that could receive demand for offices.
27-9	Final EIS should assess how frontage and landscaping improvements might impact freight mobility.	As the commenter notes, frontage and landscaping requirements may result in a change of use in public rights-of-way that were previously used for informal parking and/or loading. This is disclosed in the Parking impacts section on p. 3-386 of the Draft EIS. The commenter's suggestions regarding modifications to those requirements to maintain freight mobility are noted. Additional detail on proposed frontage and landscaping requirements is included in the Final EIS in the development standards Appendix G . Standards vary between the proposed zones with higher requirements in the II and UI zones, and lesser requirements in the MML zone. If new zoning designations are adopted, SDCI will work with SDOT to develop updates to the Streets Illustrated manual reflecting street design standards tailored to the industrial context.
27-10	Do not add design review. Consider a TDR program within the BINMIC for industrial uses.	No expansion of design review into industrial areas is proposed. Comment noted.
27-11	Concern that non-conformities will be caused in the MML zone. Consider amending the substantial alteration thresholds.	See response to frequent comment themes concerning non- conforming uses in Section 4.2.2 . The development standards appendix includes additional detail, including a paragraph addressing potential amendments to the substantial alteration threshold (Appendix G).
27-12	Concern about creation of non-conforming uses and structures.	Comment noted. See response to frequent comment themes concerning non-conforming uses.
27-13	City should delay implementation of the proposal and Final EIS until Sound Transit selects the route for the planned light rail extension into Ballard.	Comment noted. See Section 4.2.4 concerning coordination with Sound Transit.
27-14	Information about sub-area planning was not included.	See responses to comments 6-2 and 6-4.
27-15	Study removing more land from MICs.	See Section I regarding MIC boundaries.
27-16	Study different zoning options for the areas zoned IB and IC in west Ballard along Market Street.	Comment is noted. The Preferred Alternative includes a different zoning designation for these areas compared to Draft EIS alternatives, converting a portion of it to a mixed-use (NC-75) zone in that alternative.
27-17	Property specific comment for 21 st Ave W, North of W Emerson Place and South of Commodore Way	Comment noted. See Section 4.2.2 concerning non-conforming uses and other responses to this letter.
27-18	Property specific request for 2715 W Fort St,	Comment noted. See Section 4.2.2 concerning non-conforming uses and other responses to this letter.
27-19	Property specific request for North side of NW 53rd St, Between 15th Ave NW and 14th Ave NW, 98107	Comment noted. The location is zoned II under multiple alternatives including the preferred alternative. Information on proposed development standard is included in the appendix.
27-20	Property specific request for 5010-5014 14th Ave NW, 98107	Comment noted. The location is zoned II or UI under multiple alternatives. Information on proposed development standards is included in Appendix G .
27-21	Property specific request for NW 50th and NW 52st between 14th Ave NW and 11th Ave NW	Comment noted. The location is zoned II or UI under multiple alternatives. Information on proposed development standards is included in Appendix G .

Number	Comment Summary	Response
27-22	Property specific request for 800 NW 46th St, 98107	Comment noted. The area is zoned UI in the Preferred Alternative. Information on proposed development standards is included in Appendix G .
27-23	Property specific request for NW Market St, 98107 west of 28th Ave NW	Comment noted. The Preferred Alternative includes this area in a mixed use (Neighborhood Commercial) zone.
27-24	The Proposal is a De Facto Zone Change and Must be Considered Together with the Specific Implementing Development Regulations	Detailed information about development standards is included in Appendix G . Sufficient information is present to fully understand the allowed scale, nature, and allowable uses in new development under the proposed zones. The Final EIS included added detail in the appendix in response to comment.
27-25	The "Action" or "Proposal" is not Sufficiently Defined to Allow Meaningful Environmental Review Because "Industrial" is not Defined	Comment noted. The Final EIS includes additional information and detail on proposed development standards, including a new table of uses with an indication of qualification as an industrial use (Appendix G). Sufficient information is provided to understand the potential for impacts under SEPA.
27-26	The Alternatives are not Reasonable because they are Inconsistent with the Locational Criteria and Proposed Policies (e.g. all alternatives designate land outside of MICs as	The proposal is legislative and the City has flexibility in defining and evaluating non-project proposals (WAC 197-11-442). The City will consider public comments to shape the preferred alternative and final legislative proposals.
	MML, small parcels are MML, maps are not clear).	The MML zone is conceptually identified in Section 2.4.1 and would apply to areas with established economic clusters and infrastructure or water. The MML zone would be commonly applied in areas currently zoned IG1/IG2 inside or outside of the MIC. The Preferred Alternative retains some existing zoning outside the MICs. A detailed zoning map proposal down to the parcel level is included for each alternative (Appendix C)
27-27	The Draft EIS is Inadequate Because the Alternatives are not Adequately Segregated or Assessed for Each Sub-Area in the Study Area	A detailed zoning map proposal down to the parcel level is included for each alternative (Appendix C). Where feasible and practical impacts are summarized on a subarea level.
27-28	The Draft EIS Does not Adequately Consider Sound Transit's Planned Ballard Light Rail Extension Project.	Comment noted see response to frequent comments concerning coordination with Sound Transit (Section 4.2.4).
27-29	The Draft EIS Does Not Adequately Assess Impacts on Land and Shoreline Use	Impacts in several impact categories are assessed in Section 3.8 Land & Shoreline Use.
27-30	The City Should Assess Purely Economic Impacts	Economic analysis is not required under SEPA. The City has considered economic information separately. See Section 4.2.1 .
28	Howard	Alliance for Pioneer Square
28-1	Submitting on behalf of Alliance for Pioneer Square. Appreciate the opportunity to comment and the objective to "promote mutually reinforcing mixes of activities at the transitions between industrial areas and urban villages or residential neighborhoods." Encouraged by the City's stated intent to work with owners or future owners of the WOSCA and Interbay Armory sites.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
28-2	Tailor the UI zone to allow opportunity to use upper floors of the WOSCA site for industry supportive or work force housing while	The comment is noted and forwarded to City decision makers. The proposal includes a policy for site-specific master planning of the

Number	Comment Summary	Response
	encouraging new spaces for makers, artists, and other uses appropriate for transitional industrial sites.	wOSCA site. Unique development standards and approaches could be arrived at through that future process.
29	Johnson	Historic South Downtown
29-1	Appreciate the opportunity to comment. Changes in the Stadium/SODO area of the Duwamish MIC border and intersect Historic South Downtown's areas of concern, and specific definition of the different functions that industrial lands serve in Seattle could benefit these areas. The UI and II zones that would define a transition area along the west side of the stadium area and the south side of the CID have the potential to benefit the edges of both historic neighborhoods.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
29-2	Requests that the city provide additional resources to the Pioneer Square Preservation Board to review changes to historic buildings for remaining industrial properties within the CID boundaries. City should prioritize retrofitting landmarked unreinforced masonry structures within SODO.	The comment is noted and forwarded to City decision makers. The study area does not include any land that is within the Pioneer Square or C/ID historic landmark districts.
29-3	For areas with increased residential units, the zoning should allow for provision of all services necessary for an increased residential population, particularly grocery stores and pharmacies located in reasonable walking or transit distances.	The comment is noted and forwarded to City decision makers. Commercial services would be allowed under proposed development standards in the UI zone up to maximum size of use limits, and in the II zone according to the incentive bonus development structure.
29-4	Requests additional information on the new II zoning area adjacent to C-ID, which should include an analysis of how increased need for housing, services and other zoning changes may affect the historic neighborhood.	Comment is noted. The II zone does not allow new housing development. Potential impacts on historic districts are discussed in Section 3.11 Historic, Archaeological, & Cultural Resources .
29-5	Pier 48 is currently omitted from the Draft EIS and should be addressed.	Pier 48 is not within the EIS study area. Potential for open space impacts to be addressed by future use of Pier 48 is included in Section 3.12 Open Space & Recreation, Mitigation Measures .
29-6	Would like to see mitigation recommendations for proactive survey on publicly-owned parcels of land, as well as on vacant lands, in the Duwamish MIC given the area's high potential for archaeological discovery.	Cultural resources review, including archaeological survey, is a process that is done prior to the start of many projects, and includes consultation with potentially affected Tribes. Many federal, state, and local statutes and ordinances require notice and consultation with affected Tribes before, during, and after project review. The National Historic Preservation Act (NHPA) of 1966, was amended in 1986 with provisions for consultation with affected Tribes and 1992 to include and clarify the roles and responsibilities of Indian Tribes in Section 106 reviews. All cultural resources survey and archaeological work will follow best practices and standard archaeological techniques in the discovery and preservation of cultural and historical artifacts.
		any project with rederal running, permits, or on rederal or state lands, or that use State capital funds have some cultural resources survey and inventory requirements that must be satisfied before construction activities can begin. An Inadvertent Discovery Plan (IDP) typically accompanies a cultural resources survey and inventory

Number	Comment Summary	Response
		report, which spells out the appropriate procedures to follow should an inadvertent discovery of cultural or archaeological resources occur. See also response to comment 1-4.
29-7	Final EIS would better serve its purpose with enhanced attention to specific equity and culture issues for areas adjacent to Pioneer Square and the CID.	Comment is noted. Section 1.7.15 includes a summary of race and social justice considerations. Other EIS sections including Section 3.8 Land & Shoreline Use integrate race and social justice analysis.
30	Krohn	SMART Transportation Division, United Transportation Union
30-1	Organization represents railroad workers with a substantially large workforce within the industrial areas of Seattle.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
30-2	Do not agree with the proposed upzone of MIC lands currently zoned IG-2 in Georgetown to non-industrial mixed-use zones under alternatives 3 and 4.	The comment is noted and forwarded to City decision makers. Mixed use is considered in the Preferred Alternative too.
30-3	Rezoning to increased residential and mixed- use development near UP Track 101 lead spur would increase safety risks to the public and railroad employees.	Comment is noted. Additional discussion of potential impacts is added in the Final EIS for the relevant alternatives.
30-4	Upzoning area adjacent to the Track 101 rail spur would result in additional pressure on the carrier to consider the possibility of abandonment.	The comment is noted and forwarded to City decision makers. Additional discussion of potential impacts and mitigation measures are added in the Final EIS for the relevant alternatives. Note that the Preferred Alternative includes conditional use criteria for the location of housing, which could improve designs and configurations to minimize potential conflict between the track spur and new uses.
30-5	Greatest concerns center on any zoning changes near, adjacent to, or affecting Union Pacific's track 101 lead spur.	The comment is noted and forwarded to City decision makers. Additional discussion of potential impacts and mitigation measures is added in the Final EIS for the relevant alternatives.
30-6	Eliminating the track 101 spur would cut off south-end yard access to intermodal loading ramps 6-10.	See response to comments 30-2 through 30-5 above. The proposal does not include an action to eliminate the track 101 spur.
30-7	Eliminating the track 101 spur would cut off south-end yard access to intermodal loading ramps 6-10.	See response to comments 30-2 through 30-5 above. The proposal does not include an action to eliminate the track 101 spur.
30-8	Abandonment of the track 101 spur would increase and transfer risk to other public crossings and onto railroad operating crew employees.	See response to comments 30-2 through 30-5 above. The proposal does not include an action to eliminate the track 101 spur.
30-9	Proposals to add residential in the area should be reconsidered.	The EIS alternatives differ as to whether residential uses would be allowed near the location. Impacts and mitigation measures are discussed for the relevant alternatives.
31	Lehmann	Industrial and Maritime Strategy Council
31-1	Support zoning changes concentrated along major commercial thoroughfares and around existing and planned light rail hubs to permit a broader range of commercial activities and the development of limited workforce housing.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Note the new zones, particularly II, is meant to provide additional mixed industrial/technology uses and employment density near light rail investments.

Number	Comment Summary	Response
31-2	Proposal as-is will result in no meaningful change to the status quo in SODO and a wasted opportunity to leverage light rail investments.	The comment is noted and forwarded to City decision makers. See response to comment 31-1.
31-3	Most current zoning in SODO dates back to the 2000 Greater Duwamish MIC Plan.	The comment is noted and forwarded to City decision makers.
31-4	Challenges in SODO include escalating land value, sites that are not conducive to large-scale industrial uses, and existing land uses that are predominantly non-industrial.	The comment is noted and forwarded to City decision makers. See response to comment 31-1.
31-5	East/west congestion is a challenge in SODO. Lack of adequate street infrastructure results in increasing conflicts between bikes and freight/auto.	Section 3.10 Transportation includes analysis of transportation impacts including safety impacts. The roadway network is considered holistically including east–west connections.
31-6	Contamination in SODO can impose extraordinary costs on new development.	Contamination is analyzed in Section 3.5 Contamination . The effects of contamination on development potential are noted in Section 3.8 Land & Shoreline Use .
31-7	Industrial development cannot underwrite the significant cost of ground improvement and foundation systems in the liquefiable soils of SODO.	The effects of contamination on development potential are noted in Section 3.8 Land & Shoreline Use . See also Section 4.2.1 concerning development feasibility.
31-8	SODO is home to a significant number of unreinforced masonry buildings.	The comment is noted. Historic aged masonry structures are discussed in Section 3.11 Historic, Archaeological, & Cultural Resources .
31-9	Current zoning restrictions in SODO do not capitalize on light rail.	The comment is noted. The proposal includes varied potential zoning changes in action alternatives intended in part to improve land use integration with transit.
31-10	Protective zoning in SODO precludes uses and development that can support new capital investment.	The comment is noted. See also response to Section 4.2.10 and D.
31-11	Lack of new office sites in Center City Seattle and the upcoming light rail expansion present an opportunity for SODO to help alleviate regional challenges.	The comment is noted. The proposal includes varied combinations of potential zoning changes in action alternatives that would allow for expanded capacity for office development in the II zone.
31-12	Little reinvestment expected in SODO with commercial FARs remaining so low.	The comment is noted and forwarded to City decision makers.
31-13	Expand EIS study to include greater commercial density and workforce housing and prove that concepts like II zones exist elsewhere.	The comment is noted and forwarded to City decision makers. See also Section 4.2.10 and []
31-14	Key to SODO's future is to attract capital investment that will support long-term industrial uses and address challenges of the area.	The comment is noted and forwarded to City decision makers.
31-15	Request for economic analysis, including demand for industrial property, square footage rents, and projected vacancy rates.	SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450). See also Section 4.2.1 .
32	Loe	Share The Cities Action Fund

Number	Comment Summary	Response
	Note: Comments overlap with comment letters 59 and 71. Responses here are primarily cross- referenced to letter 59.	
32-1	Ask for additional outreach and community engagement, specifically for non-English speaking residents.	Thank you for your letter. The comment is noted. The City translated Draft EIS executive summary material and held numerous workshops and community engagement events with interpretation into Spanish, Vietnamese, and Somali. Efforts were targeted to the South Park and Georgetown neighborhood areas.
32-2	ElS should address small business displacement, greater partnership with Indigenous communities, present a clear air quality monitoring strategy, highlight the unique importance of Ballard-Interbay as a freshwater harbor, consider BNSF's historical and continuing lack of transparency and accountability, clarify which existing and proposed uses in the industrial areas would be considered nonconforming, clarify the definition of industry supportive housing, include a complete list of the neighborhood-level comprehensive plan recommendations impacted by these zoning changes, connect Seattle's historic segregation, redlining, and exclusion to present-day location of industrial uses, complete a citywide zoning analysis looking at commercial and multi-family exclusion, and examine which recommendations and boundaries are carried over from older plans that have never been vetted for equity or impact.	See response to comments 59-2 through 59-7 and 71-1 through 71-7.
32-3	Examine comments submitted by the Duwamish River Community Coalition, Seattle Cruise Control, and the Georgetown/South Park Advisory Group. Requests additional scrutiny regarding the impacts of the systemic racist policies that created Seattle's industrial land and exacerbated the disparate impacts of pollution and disinvestment on nearby underserved neighborhoods of color.	See response to comments 59-1 and 59-7. Comments from the Duwamish River Community Coalition, Seattle Cruise Control, and Georgetown/South Park Advisory Group are addressed in letters 93, 37, and 96, respectively.
32-4	In the MML zone, code should clarify which existing and proposed uses will become nonconforming and should accommodate uses such as the WNBA Storm practice facility. In the UI zone, clarify the definition of industry supportive housing, provide examples from other locations of housing on top of industry, and propose thresholds for mixed use buildings.	See response to comments 59-2 and 59-3.
32-5	EIS does not examine where the II zone expressly contradicts existing neighborhood plans. EIS should include a complete list of the neighborhood-level comprehensive plan recommendations impacted by these zoning changes and analyze whether they conform or	See response to comment 59-4.

Number	Comment Summary	Response
	contradict the Dratt Comprenensive Plan Goal and Policy Language in Appendix D.	
32-6	Agree with how the EIS alternatives are organized, but the document can be clearer about the distinction. Support Alternative 4 only because there are no alternatives that more liberally use the UI and II zones across larger portions of the city. EIS must do a better job establishing why areas change under each of the alternatives, and which areas should be treated as a cohesive cluster. At the neighborhood level, the proposed maps do not offer a picture of cohesiveness—what does it mean if blocks are divided? Alternative 1 should be considered a non-starter.	See response to comments 59-5 and 59-6.
32-7	City's industrial boundaries carry the history of segregation that cannot be washed away with a cursory equity analysis.	See response to comment 59-7.
32-8	EIS doesn't consider how boundaries of the current industrial zones came to exist. Impossible to develop policies that address land use and zoning issues without considering large areas of the city devoted exclusively to single- family housing.	See response to comment 59-7.
32-9	More thoroughly consider equity impacts. Connect Seattle's historic segregation, redlining, and exclusion to present-day location of industrial uses. Complete a citywide zoning analysis looking at commercial and multi-family exclusion in other areas. Examine which recommendations and boundaries are carried over from older plans that have never been vetted for equity or impact.	See response to comment 59-7.
32-10	EIS must make robust efforts to understand history and the sources of inequity in shaping land use decisions.	See response to comment 59-8.
33	Malshuk	First South Properties, LLC
33-1	Own property at 7343 E Marginal Way S. Zoned IG-1 and would be rezoned as MML under the Action Alternatives.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
33-2	Request the Final EIS continue to recognize existing uses, increase flexibility for ancillary uses (from 30% to 49% limit), and broadly define industrial uses.	The comment is noted. Elements of the proposal increase flexibility for ancillary uses, especially in the UI zone. Details concerning qualification as industrial use under action alternatives in included in Appendix G of the Final EIS.
33-3	Urge the City to study and adopt maximum flexibility in the regulatory framework.	The comment is noted. Elements of the proposal would increase flexibility under actin alternatives especially in the proposed II and UI zones.
34	McCullough	Seattle Industrial Lands Coalition
34-1	Writing on behalf of Seattle Industrial Coalition.	Thank you for your letter. The comment is noted and forwarded to City decision makers.

Number	Comment Summary	Response
34-2	Coalition members own, manage, and develop industrially-zoned property in Seattle. Members are adversely affected because the current and future use of their property will be unreasonably restricted by the proposal.	The comment is noted and forwarded to City decision makers. All proposed zones allow a broad variety of uses and proposed development standards allow a variety of potential development.
34-3	Proposal is not described in terms of its objectives per WAC 197-11-060 but rather as specific zoning text amendments.	Objectives of the proposal are defined in EIS Section 1.5.1 . The objectives are informed by the recommendations of an <u>Industrial</u> and <u>Maritime Strategy</u> stakeholder process. Objectives are identified in four overlapping categories of people, place, and production and process.
34-4	Draft EIS is based on inadequate information and fails to disclose or evaluate the entire proposal. Draft EIS alternatives fail to meet the requirements of SEPA because they are not reasonable alternatives.	See response to comment 25-5.
34-5	Draft EIS manipulates the description of the existing condition to mask existing non- industrial uses.	Section 3.8 Land & Shoreline Use includes existing land use analysis in map format that is based on empirical study and available data, and the section also includes narrative summary of existing land use for all sub areas.
34-6	Draft EIS fails to address many industrially- zoned areas in Seattle that include few industrial uses or where industrial uses are likely to be replaced in the next decade.	Existing land use is analyzed in Section 3.8 Land & Shoreline Use , and future land use impacts are analyzed under each alternative.
34-7	Draft EIS ignores impact of light rail station area walksheds.	The EIS includes information about existing and future light rail station areas to the extent it is known. Section 3.10 Transportation includes future light rail expansion plans. Geographic configurations of potential zone changes under EIS action alternatives is informed by the locations of existing and future rail station areas.
34-8	Draft EIS alternatives have not been tested for financial feasibility, including cap rates, vacancy rates, development hard costs, environmental costs, land value, and infrastructure.	SEPA does not require cost-benefit or economic analysis (WAC 197-11-448 and 450). See Section 4.2.1 .
34-9	Draft EIS ignores the impacts of alternatives on blight.	The EIS analyzes numerous elements of the environment as required by the SEPA rules and a scoping process (i.e., Air Quality, Noise, Contamination, Land Use, Transportation safety etc.), and analysis of these environmental topics amounts to analysis of environmental health and livability impacts under different alternatives. No blight analysis is required in the SEPA rules under elements of the environment (197-11-444). Purely economic analysis is not required in an EIS. See Section 4.2.1 .
34-10	Proposal will result in significant adverse impacts to the built environment, including aesthetics and blight, environmental health, transportation, and land use.	The EIS analyses potential impacts on the built environment in sections including Sections 3.8 Land & Shoreline Use and 3.11 Historic, Archaeological, & Cultural Resources . Environmental health is addressed in multiple sections of the EIS in topical areas including Air Quality, Noise, Contamination, and Transportation, and in the Environmental Health and Compatibility subsection of Section 1.7.15 Equity & Environmental Justice Considerations .
34-11	Adoption of any Draft EIS alternatives will compound and exacerbate existing environmental problems.	Site contamination and remediation are addressed at the time of development or redevelopment through existing processes under MTCA. SEPA documentation submitted with project applications require disclosure of known or suspected contamination of soil, soil

Number	Comment Summary	Response
		vapor, groundwater, or other media, and lenders require Phase I and/or Phase II Environmental Site Assessments be completed before they will provide project funding. See Section 3.5 Contamination .
34-12	Draft EIS ignores regional impacts	The EIS discusses consistency with regional plans and policies in Section 3.8 Land & Shoreline Use . Where appropriate and feasible to analyze, potential impacts beyond City of Seattle borders in other parts of the region are studied or analyzed (including related to air quality/GHG and transportation).
34-13	Draft EIS fails to disclose prior planning efforts.	The EIS discusses historical planning and land use decisions (Section 3.8.1) along with the current policy and regulatory framework that features a summary of past planning efforts. The EIS also incorporates and references many other City plans that establish impact thresholds or levels of service such as parks plans, transportation plans and others.
34-14	Draft EIS must be withdrawn and reissued.	The non-project EIS was developed consistent with SEPA rules including WAC 197-11-442 and based on a scoping process consistent with WAC 197-11-360. See response to comment 25-5.
35	Nelson	Elliott Way Partners, LLC
35-1	Incorporate comments issued by Seattle Industrial Coalition and NAIOP Recommendations on land use alternatives to be studied and support for no design review requirements.	Thank you for your letter. The comment is noted. See also response to comments in letter 21.
36	Nitze	Nitze-Stagen
36-1	Entirely comfortable with the methodology applied by CAI once the revised, market-based assumptions are incorporated.	Economic development feasibility is not a part of the EIS. See Section 4.2.1 concerning development feasibility analysis.
36-2	The EIS should clearly document economic impacts such as demand for industrial property, square footage rents, and projected vacancy rates.	SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450). See Section 4.2.1 .
37	Printz	Seattle Cruise Control
37-1	Examine comments submitted by the Duwamish River Community Coalition, Seattle Cruise Control, and the Georgetown/South Park	Comments from the Duwamish River Community Coalition, Seattle Cruise Control, and Georgetown/South Park Advisory Group are addressed in letters 93, 37, and 96, respectively.
	Advisory Group. Ask the City to establish goals of near full employment and affordable housing to improve quality of life, protect the climate, and reduce traffic congestion. Prioritize climate protection and resiliency. Most support Alternative 4 of the alternatives proposed.	The Industrial and Maritime seeks to simultaneously advance environmental protection, addressing climate change risks, and strengthening and supporting Seattle's maritime and industrial sectors and ensuring the benefits of economic diversity and opportunity. In order to address all of these goals, Section 3.2.3 Air Quality & GHG provides mitigation measures that address the root causes of greenhouse gas emissions; fossil fuel combustion for both industrial and heating processes, and vehicle use, while not restricting industrial users who may have a history of fossil fuel use. Green infrastructure methods are standard for meeting on- site stormwater management as stated in Section 3.14 Utilities . Text has been added to Section 3.2.3 to strengthen potential mitigation measures aimed at climate resiliency and transformation of fossil fuel dependent industries

Number	Comment Summary	Response
37-2	Water Quality section must address impacts of cruise ships. Most air/water impacts are narrowly constrained to the study areas. Document mentions there are significant impacts to Puget Sound, but only refers to the Sound in two places with no listing of mitigation measures for that body of water.	Text has been added to Sections 3.3.2 and 3.3.3 which discusses the classification of the Puget Sound as a No Discharge Zone which prohibits the discharge of sewage, as well as other regulations which prohibit the discharge of oil, trash, and other pollutants. Text was also added to Sections 3.3.1 and 3.3.2 to reflect the wider regional impact of maritime activities including cruise ships. See Section 3.3 Water Resources .
37-3	The vague mention of "planned regulatory requirements" to achieve emission reduction outcomes comes across as misplaced faith that undermines our ability to plan realistically for the future. Statements about maritime emissions lack context.	The overall context for current maritime emissions for criteria air pollutants and GHG emissions can be found in Section 3.2.2, Impacts Common to All Alternatives, Exhibit 3.2-12 and Exhibit 3.2-14. In addition, text has been added to Section 3.2.1 to indicate that additional context and information for maritime emissions in general, and in relation to the MIC areas affected by the proposal, can be found in the 2016 Puget Sound Maritime Emissions Inventory (PSMEI 2018), which is now incorporated by reference. See Section 3.2 Air Quality & GHG.
		Section 3.2.2 , Maritime Emissions, includes a discussion of several regulatory changes that will decrease maritime emissions in alignment with IMO Annex VI. These regulatory changes, combined with anticipated though uncertain future improvements in both engine technology and emission requirements set by federal, state, and international regulatory entities, are expected to decrease future air emissions, particularly from diesel engines.
		Additional text has been added to Section 3.2.3 Air Quality & GHG to address the potential for state and local government to impose restrictions on maritime air emissions for ocean-going vessels while underway in US waters. Additional text has also been added to address the potential to expand availability of shore power to include those areas and ships not covered by the Port of Seattle's existing plans.
		As discussed in WAC 197-11-440, this non-project EIS is limited to a general discussion of the impacts of alternate proposals for policies contained in the proposed Industrial and Maritime Strategy. The City of Seattle concluded that as proposed, the alternatives would not prevent or deter efforts to reduce emissions in comparison to local or regional goals or targets for GHG reductions.
		The current level of analysis provides an appropriate level of detail for a non-project EIS. Subsequent developments that may arise from the proposed land use changes in the Industrial and Maritime Strategy will be required to meet all applicable codes and regulations, and to conduct project-level SEPA review at that time, in which analysis will be conducted to assess site specific impacts and necessary mitigation measures, including for maritime emissions related issues.
38	Rivera	Seattle Mariners
38-1	Encourage the City to recognize the unique character of the Stadium Transition Area Overlay District. Final EIS must recognize the stadiums and event center that draw more than six million visitors each year and make the Stadium District different than other industrial transitional areas. Most support Alternative 4 of the alternatives proposed.	The comment is noted and forwarded to City decision makers. The Preferred Alternative includes more distinct measures in the proposed development standards for the STAOD compared to the Draft EIS Alternatives.

Number	Comment Summary	Response
38-2	Support the proposed lodging use allowance in the Stadium District and within the UI zone. Request the Preferred Alternative allow lodging without a size limit in the Stadium District. Encourage the Final EIS to acknowledge positive impact lodging in the Stadium District will have on transportation patterns in the district by keeping event attendees in the neighborhood and off the roads before and after events.	The comment is noted. If a small portion (0.4 FAR) of a development includes light industrial uses, the maximum size of use limit would not apply.
38-3	Do not place size limits on activating uses within the Stadium District. City should use incentives to encourage smaller-scale spaces. Support the proposed size of use limit for office uses proposed in the Draft EIS.	The comment is noted. Please note that if a small portion of the development (0.4 FAR) includes light industrial uses, the maximum size of use limits would not apply.
38-4	Support additional density for compatible uses in the Stadium District.	The comment is noted. Note that the Preferred Alternative applies special allowances in the STAOD (density and other standards). Please see Appendix G .
38-5	The Final EIS should include a transportation study that examines the potential impacts should the existing IC-zoned and IG-zoned parcels in the Stadium District be developed to their maximum available density as office buildings under the proposed framework.	The EIS analyzes a 22-year future scenario under different land use alternatives. Growth and development patterns are projected in the aggregate and are not broken down to a parcel specific level. The action alternatives do evaluate for different concentrations of office and residential future land uses in the vicinity of the STAOD because the alternatives apply different land use regulatory schemes. Different transportation impacts associated with the different growth projections under the alternatives are a feature of the transportation analysis in Section 3.10 Transportation .
38-6	The Final EIS should analyze allowing workforce housing within the Stadium District.	The comment is noted. Some industry-supportive housing would be allowed in action alternatives. Please see Section 4.2.3 .
39	Ugles	International Longshore and Warehouse Union Locals 19, 52, and 98, Inland Boatmens Union
39-1	As union workers, and those most directly impacted by zoning changes within the City's MIC's, we hope you will give considerable weight to our support for Alternative 2,	Thank you for your letter. The comment is noted and forwarded to City decision makers.
39-2	Experiences during the pandemic made apparent how essential our maritime workforce and infrastructure are to the residents and industries of our state. The Final EIS should recognize and be informed by these experiences.	Chapter 2 in the Final EIS includes new text that describes the effects of the pandemic related to this proposal.
39-3	(1) Document should recognize the critical public infrastructure to the state's economy in the MICs.	(1) Language has been added to the Primary & Secondary Study Areas in Section 3.10.1 Affected Environment reflecting the commenter's suggestion.
	(2) Document should mention the quality of the jobs, particularly union jobs, created within the Alternatives.(3) Document should include separate section on freight movement in the MIC's, especially rail and truck.	 (2) Objectives for the action alternatives include increasing living wage jobs; see Section 1.5.1. The types of industrial uses promoted in each zone and the number of jobs expected for each alternative are included in Sections 1.5 and 2.4 of the EIS. Details of job types are not projected beyond industrial and non-industrial jobs consistent with the areawide programmatic analysis. (3) The Final EIS has been reorganized to include freight as a
		separate section with additional information.

Number	Comment Summary	Response
39-4	Draft EIS should describe marine terminals and Elliott Bay's naturally deep harbor as essential and irreplaceable to industrial activity, the economy, and maritime jobs/livelihoods.	Language has been added to Primary & Secondary Study Areas in Section 3.10.1 Affected Environment reflecting the commenter's suggestion.
39-5	We request that the Final EIS delineate the projected number of unionized jobs created in each Alternative; provide a definition of a quality job (versus simply a "living wage"), and that the objective of the EIS be restated to increase the quantity of quality jobs.	Comments is noted. Chapter 2 of the Final EIS includes edits to the text to describe the benefits of union jobs, with expanded discussion of the likelihood of projected employment to be unionized.
39-6	Draft EIS lacks the subject of freight, including trucks and rail. Freight should be a standalone subject in the Final EIS with analysis of freight movement, rail operations, and freight and passenger rail impacts. Auto & Freight sections only address vehicular traffic volumes and not conditions for freight movement or facilities.	The Final EIS has been reorganized to include freight as a separate section with additional information.
40	Selig	J Selig Real Estate, LLC
40-1	Appreciate the opportunity to comment. Currently proposing a rezone of a split-zoned parcel at 2501 NW Market St (currently IC and NC-3, requesting NC)) within the Ballard Urban Village and outside the BINMIC. Request the Final EIS consider this rezone.	Thank you for your letter. The comment is noted and forwarded to City decision makers. The EIS includes different zoning options for the site in the alternatives. In addition to the factors noted in the comment letter, the site is adjacent to shoreline lands with working maritime uses. See maps of the Preferred Alternative in Appendix C .
40-2	Requested rezone is consistent with draft LU Goal 12.	The comment is noted and forwarded to City decision makers.
40-3	Requested rezone is consistent with Comp Plan policy LU 10.9.	The comment is noted and forwarded to City decision makers.
40-4	Comp Plan expressly states the City should avoid placing industrial zones within urban villages.	The comment is noted and forwarded to City decision makers.
41	Trohimovich	Futurewise
41-1	Appreciate the opportunity to comment. Overall concludes the Draft EIS adequately explains the proposal, analyzes the alternatives, identifies and discloses environmental impacts, and identifies required and potential mitigation measures.	Thank you for your letter. Comment is noted.
41-2	Final EIS should consider designating truck routes serving industrial and manufacturing areas away from residential areas especially residential areas with vulnerable populations as an additional air quality and GHG mitigation measure.	An additional mitigation measure has been added to Section 3.2.3 .
41-3	Clarify sentence on page 3-94 regarding impacts common to alternatives under sea level rise (Water Resources) considering that Seattle's flood plain regulations and master program regulations will not protect against sea level rise overall and for the subareas.	As stated in Section 3.3.2 Water Resources , development in the study area will be required to comply with regulations which may reduce the vulnerability of those developments to sea level rise impacts relative to existing conditions, particularly in locations that are currently not compliant with current regulations. As flood regulations evolve based on the best available science,

Number	Comment Summary	Response
		requirements for development will be modified. Lext has been added to clarify that regulations for development in the study area may change. This impact is expected to apply to all alternatives proposed, including the No Action Alternative.
41-4	EIS should propose as a mitigating measure development regulation that require buildings, structures, and industrial and manufacturing sites to be elevated above the sea level rise projected to occur during the life of the facility.	See response to comment 41-3. Additional text has been added to Section 3.3.3 Water Resources to add consideration of sea-level rise in design of buildings, structures, and industrial and manufacturing sites.
41-5	EIS does not analyze the impacts of allowing more housing in the proposed Urban Industrial (UI) zone on nearby industrial and manufacturing uses.	The EIS includes a discussion of compatibility as one of the impact categories in Section 3.8 Land & Shoreline Use . See also Section 3.9 Housing that describes impacts of allowed industry supportive housing and other housing under each alternative including exposure to air pollution, noise pollution, or environmental hazards.
41-6	One of the most effective mitigating measures for cultural and archaeological resources is to require investigation by cultural and archaeological professionals working cooperatively with local Tribes and Native American groups to determine if a site contains cultural or archaeological resources before ground disturbing activities are allowed. EIS should add this as one of the required mitigation measures.	Cultural resources review, including archaeological survey, is a process that is done prior to the start of many projects, and includes consultation with Tribes. Many federal, state, and local statutes and ordinance require notice and consultation with affected Tribes before, during, and after project review. The National Historic Preservation Act (NHPA) of 1966, was amended in 1986 with provisions for consultation with affected Tribes and 1992 to include and clarify the roles and responsibilities of Indian Tribes in Section 106 reviews. All cultural resources survey and archaeological work will follow best practices and standard archaeological techniques in the discovery and preservation of cultural and historical artifacts. Any project with Federal funding, permits, or on federal or state lands, or that use State capital funds have some cultural resources survey and inventory requirements that must be satisfied before construction activities can begin. An Inadvertent Discovery Plan (IDP) typically accompanies a cultural resources survey and inventory report, which spells out the appropriate procedures to follow should an inadvertent discovery of cultural or archaeological resources occur. See also response to comment 1-4.
41-7	Example Becket Point project in Jefferson County regarding upfront archaeological investigations.	The comment is noted. See also response to comment 41-6.
42	Tucker	Pacific Christian Academy
42-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
43	Vaughn	GPG&C Investment Group LLC
43-1	Follow up to confirm receipt of letter 44.	Thank you for your letter. Comment is noted. See response to comments in letter 44.
44	Weed	SoDo Industrial Coalition
44-1	Further analysis requested on industrial land quantification.	Thank you for your letter. See Section 3.8.1 Land & Shoreline Use for an analysis of existing land use.

Number	Comment Summary	Response
44-2	Further detail and analysis requested regarding work force housing to support the vision/study.	See Section 4.2.3 . Under the Preferred Alternative more of the potential housing would be in the SODO/Stadium Subarea; there would be some limited opportunity elsewhere too.
44-3	Request quantification of the level of infrastructure investment, capital projects, and circulation improvements required. Suggest leveraging light rail commitments.	SEPA does not require cost-benefit or economic analysis (WAC 197- 11-448 and 450). See Section 4.2.1 .
		The EIS includes a Mitigation Measures section which describes the various plans that include specific projects and high priority areas for improvement. Those documents include: the Freight Master Plan, Transit Master Plan, Pedestrian Master Plan, the Bicycle Master Plan, the Bicycle and Pedestrian Safety Analysis, the Ballard-Interbay Regional Transportation (BIRT) System Report, and the Georgetown Mobility Study. SDOT is currently in the process of developing the Seattle Transportation Plan which will integrate the City's modal plans into a comprehensive vision for the citywide transportation network centered around the following values and goals: equity, safety, mobility, sustainability, livability, and excellence. The STP is considering station planning needs for Sound Transit's planned light rail extension.
		Text has been added to the mitigation section of this EIS to note that the City and Sound Transit are coordinating on transportation mitigation around expanded and new light rail stations and notes the System Access Fund as a funding mechanism for station area improvements.
44-4	Document should emphasize the importance of transit investments in the MICs and encourage TOD density.	The commenter's support for TOD is noted. The EIS addresses transit both from a capacity perspective as well as its benefits to mitigate traffic congestion as described in the Travel Demand Management (TDM) section of the Mitigation Measures section. No changes are requested with respect to the EIS transportation analysis.
44-5	Request to include a stated strategy and commitment for direct solicitation of input from potentially affected parties throughout the policy making process.	Following the EIS process, the City will develop specific policy and zoning proposals that will be the subject of public meetings and public hearings by the City Council.
45	Aggen	Individual
45-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. Comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
46	Anane	Individual
46-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. Comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
47	Anawalt	Individual
47-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. Comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
48	Baker	Individual

Number	Comment Summary	Response
48-1	Supports Alternative 4.	Thank you for your letter. Comment is noted and forwarded to City decision makers.
49	Brubeck	Individual
49-1	Strategy was not developed with Duwamish Tribe or other tribes.	Tribes were contacted through the scoping and Draft EIS comment period opportunities. See also letter 1 with Duwamish Tribal comments on proposals.
49-2	The Duwamish MIC map labeling and region naming should be revised to distinguish areas west of the Duwamish River by their established place names	The comment is noted and forwarded to City decision makers. The subareas are broadly defined for analysis purposes in the EIS.
49-3	(1) Document should include planned active transportation networks including 2014 BMP and BMIPs.	(1) Planned active transportation projects are shown in Draft ElS Exhibit 3.10-20 and Exhibit 3.10-21 (Final ElS Exhibit 3.10-23 and Exhibit 3.10-24).
	(2) Transit, biking, and walking routes are necessary through the Duwamish MIC.	(2) The commenter's support for improved transit, biking, and walking facilities in the Duwamish MIC is noted.
	(3) Mitigation measures should be included to close gaps in pedestrian and bike routes and avoid significant unavoidable adverse impact to active transportation and safety.	(3) The EIS includes a Mitigation Measures section dedicated to Pedestrian & Bicycle System Improvements which describes the various plans that include specific projects and high priority areas for improvement. Those documents include: the Pedestrian Master Plan, the Bicycle Master Plan, the Bicycle and Pedestrian Safety Analysis, the Ballard-Interbay Regional Transportation (BIRT) System Report, and the Georgetown Mobility Study. SDOT is currently in the process of developing the Seattle Transportation Plan which will integrate the City's modal plans into a comprehensive vision for the citywide transportation network centered around the following values and goals: equity, safety, mobility, sustainability, livability, and excellence.
49-4	List of proposed mitigation measures (1) Full implementation of BMP, PMP, TMP & FMP with priority to improvements at hazardous areas identified in the BPSA. Increase current funding to accomplish expedited implementation. (2) Implementation of Design Guidelines in Appendix C of FMP and add to Streets	(1) SDOT is currently in the process of developing the Seattle Transportation Plan which will integrate the City's modal plans into a comprehensive vision for the citywide transportation network. The City has a biennial budget process through which transportation system improvements, maintenance, and rehabilitation needs are considered and funded as feasible. In addition to pursuing grant funding sources, the biennial budget is the process through which funding for transportation improvements would be identified.
	 (3) Prioritize construction of sidewalks/paths between places of employment and bus stops/light rail stations. (4) Replacement or implementation of phase 2 	(2) If new zoning designations are adopted, SDCI will work with SDOT to develop updates to the Streets Illustrated manual reflecting street design standards tailored to the industrial context. Updates will consider designs that can accommodate both freight
	retrofit of Ballard Bridge to include shared use path meeting current design standards.	conflicts.
	(5) Initiation of transit service along streets such as West Marginal Way SE to serve employees and customers of industries and maritime businesses.	Bicycle System Improvements section to note how the City may prioritize new active transportation connections.
		(4) The EIS includes replacement of the Ballard Bridge as a potential mitigation measure (page 3-425 of the Draft EIS). The City recently
	(6) Implementation of safe bike routes from the First Ave S Bridge through Georgetown to Downtown.	completed the Ballard-Interbay Regional Transportation System project which studied two replacement options. The report has been submitted to the Washington State Legislature for consideration of planning/funding for design and engineering.

Number	Comment Summary	Response
	 (/) Improving pavement conditions/drainage should be strategy for safety/ease of active transportation and vehicles. (8) Active transportation should be included under TSMO as strategy to reduce SOV use and free up capacity for freight/transit. Seattle should implement entire BMP and strive for high bike mode share. 	(5) The commenter's support for transit service tailored to employees/customers of industries and maritime business is noted. The Mitigation Measures section includes language to this effect under the Travel Demand Management (TDM) section: "Potential TDM measures suited to the study area could include last-mile shuttle systems between key transit nodes and the MICs; coordination with King County Metro and/or Sound Transit to provide off-peak transit service tailored to shift workers with irregular hours;" The language has been clarified to note that service could be tailored not just in terms of timing, but also key corridors serving many industrial and maritime workers.
		(6) The EIS includes a Mitigation Measures section dedicated to Pedestrian & Bicycle System Improvements including the City's safety programs and Draft Exhibit 3.10-21 (Final EIS Exhibit 3.10-24) shows the currently planned network improvements, including bike lanes, multi-use trails, and neighborhood greenways in the area mentioned by the commenter. SDOT is currently in the process of developing the Seattle Transportation Plan which will integrate the City's modal plans into a comprehensive vision for the citywide transportation network.
		(7) Language regarding pavement conditions/safety has been added to the Mitigation Measures section of the Final EIS.
		 (8) The commenter's support for a more robust bike network to support increased travel by bike is noted. The Mitigation Measures section includes language to this effect under the Travel Demand Management (TDM) section which is focused on reducing demand for auto travel (resulting in a shift to other modes including bike travel) and the Pedestrian & Bicycle System Improvements section which discusses how the City could improve the network to attract more people to travel by bike.
49-5	Seattle should not accept death and serious injuries to people walking and biking. Revise to propose measures that eliminate adverse impacts to people using active transportation.	The City is committed to ending deaths and serious injuries caused by traffic collisions. This commitment is reflected in the Vision Zero policy which is supported by a variety of strategies as described in the EIS. The EIS includes a Mitigation Measures section dedicated to Pedestrian & Bicycle System Improvements including the City's safety programs. However, the City also acknowledges that significant impacts to active transportation and safety may remain due to the projected increase in people walking and biking in areas with network gaps and the increased potential for vehicle conflicts (particularly trucks) with vulnerable users. While the City can pursue a variety of mitigation measures to improve facilities for people walking and biking and pursue supplemental funding through federal or state programs, it is not expected that all network gaps can be addressed given the number of locations needing improvement and the limited funding available.
49-6	Land designated for industrial and maritime use is Duwamish Tribe land. Other tribes have rights for fishing in the area.	The City appreciates Mr. Brubeck's comments. The City agrees that developing histories centering on the Tribes' perspectives should include the active involvement of the Duwamish and other affected Tribes, and assumes that they have "no present or future." The strategy of context development from the Tribes' perspectives is one of using their input, their stories, and their voices to create narratives to inform others of not only the history of the region's Tribes but of their continued cultural ties to the areas in the MIC. Cultural resources review is a process that is done prior to the start of many projects, and includes consultation with Tribes. Many

Number	Comment Summary	Response
		rederal, state, and local statutes and ordinances require notice and consultation with affected Tribes before, during, and after project review. The National Historic Preservation Act (NHPA) of 1966, was amended in 1986 with provisions for consultation with affected Tribes and 1992 to include and clarify the roles and responsibilities of Indian Tribes in Section 106 reviews.
		The Advisory Council on Historic Preservation (ACHP) adopted a Policy Statement Regarding the ACHP's Relationships with Indian Tribes in 2000. The policy was developed in consultation with some Tribes and inter-Tribal organizations, and addresses tribal sovereignty, government-to-government consultation, trust responsibilities, tribal participation in historic preservation, sympathetic construction, and respect for tribal religious and cultural values. The state of Washington has a government-to-government relationship with the 29 federally recognized Tribes in the state (RCW 43.376). Each Tribe is a sovereign nation and has its own definition of appropriate consultation.
49-7	City should actively involve the Duwamish and other affected tribes in future planning for the area. Mitigation should include an emphasis on archeological investigations in consultation with the tribes.	See response to comment 49-6.
50	Burg	Individual
50-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
51	Bush	Individual
51-1	Live in SODO. Supports Action Alternatives. Desire for more mixed use, affordable housing, and safe walking and biking conditions.	Thank you for your letter. The commenter's support for the Action Alternatives and vision for SODO to be a comfortable walking and biking environment are noted. SDOT is currently in the process of developing the Seattle Transportation Plan which will integrate the City's modal plans into a comprehensive vision for the citywide transportation network including industrial areas like SODO.
52	Clark	Individual
52-1	Supports Alternative 4 and requests the alternative be taken farther and concentrate more housing around Link light rail stations.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See the definition of the Preferred Alternative in Chapter 2 which includes the II zone around station areas, and a focus of supportive housing in the Stadium District.
52-2	City needs a vision for what "future industrial" looks like, and implementation and follow through to match the vision. Need to leverage this huge transit investment in the City by creating station-adjacent uses that will attract riders day and night.	The comment is noted. The proposed new zoning designations are intended to support station adjacent land uses in an industrial context.
53	Corbin	Individual
53-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.

Number	Comment Summary	Response
54	Dee	Individual
54-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
54-2	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	See response to comment 54-1.
55	Devine	Individual
55-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
56	Dickinson	Individual
56-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
57	Dillon	Individual
57-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
58	DiMartino	Individual
58-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
59	Dubicki	Individual
	Note: Comments overlap with letters 32 and 71. Responses provided here are cross-referenced in responses to letters 32 and 71.	
59-1	Examine comments submitted by the Duwamish River Community Coalition, Seattle Cruise Control, and the Georgetown/South Park Advisory Group. Requests additional scrutiny regarding the impacts of the systemic racist policies that created Seattle's industrial land, underlying industrial zone boundaries, and exacerbated the disparate impacts of pollution and disinvestment on nearby underserved neighborhoods of color.	Thank you for your letter. Comments from the Duwamish River Community Coalition, Seattle Cruise Control, and Georgetown/South Park Advisory Group are addressed in letters 93, 37, and 96, respectively.
59-2	In the MML zone, code should clarify which existing and proposed uses will become nonconforming and should accommodate uses such as the WNBA Storm practice facility.	Comment is noted. Additional information is added in the Final EIS concerning non-conforming uses in the MML zone. See also Section 4.2.2 .
59-3	In the UI zone, clarify the definition of industry supportive housing, provide examples from	See Section 4.2.3 concerning the definition of industry-supportive housing.

Number	Comment Summary	Response
	otner locations of nousing on top of industry, and propose thresholds for mixed use buildings.	
59-4	EIS does not examine where the II zone expressly contradicts existing neighborhood plans. EIS should include a complete list of the neighborhood-level comprehensive plan recommendations impacted by these zoning changes and analyze whether they conform or contradict the Draft Comprehensive Plan Goal and Policy Language in Appendix D.	Comment is noted. Section 3.8 Land & Shoreline Use includes discussion of consistency with existing plans and policies at the City and regional level including MIC subarea plans. Neighborhood plans were developed generally in the 1990s and anticipated similar adjacent industrial uses as those in the II zone. The City will re-review neighborhood policies with the development of MIC plan updates consistent with regional requirements.
59-5	Agree with how the EIS alternatives are organized, but the document can be clearer about the distinction. Support Alternative 4 only because there are no alternatives that more liberally use the UI and II zones across larger portions of the city. EIS must do a better job establishing why areas change under each of the alternatives, and which areas should be treated as a cohesive cluster.	Comment is noted. See response to comment 71-11. Chapter 2 of the EIS describes the alternatives, including the overall intent and themes for each. A Preferred Alternative is added in the Final EIS. All Action Alternatives are different variations of application of the UI, II, and Maritime, Manufacturing and Logistics (MML) zones. General locational criteria and intent is described for each of the three proposed new zones in Chapter 2 .
59-6	At the neighborhood level, the proposed maps do not offer a picture of cohesiveness. What does it mean if blocks are divided? Alternative 1 should be considered a non-starter.	Comment is noted. The EIS Appendix C includes detailed maps depicting alternate zone changes with specific boundaries. A story map is also provided by the City which allows detailed review to a parcel-specific level. See the <u>storymap link here</u> .
59-7	City's industrial boundaries carry the history of segregation that cannot be washed away with a cursory equity analysis. EIS doesn't consider how boundaries of the current industrial zones came to exist. Impossible to develop policies that address land use and zoning issues without considering large areas of the city devoted exclusively to single-family housing. EIS must more thoroughly consider equity impacts, including connecting Seattle's historic segregation, redlining, and exclusion to the present-day location of industrial uses, completing a citywide zoning analysis looking at commercial and multi-family exclusion in other areas, and examining which recommendations and boundaries are carried over from older plans that have never been vetted for equity or impact.	Comment is noted. In the Final EIS a new subsection is added to the review of historical planning and land use decisions (see Section 3.8.1). The subsection includes the historic red lining map and a discussion of the map's implications related to this proposed action. The EIS also includes an Equity & Environmental Justice review in Section 1.7.15 .
59-8	EIS must make robust efforts to understand history and the sources of inequity in shaping land use decisions.	Comment is noted. See response to comment 59-5 and 59-7
59-9	Add documentation, analysis, and maps that connect Seattle's historic segregation, redlining, and exclusion to the present-day location of industrial uses. Complete a citywide analysis of zoning that looks specifically at the ways commercial and multi-family exclusions in other parts of the city lead to the competition for industrial land. Examine which recommendations and boundaries are carried	See response to comments 59-5 and 59-7.

Number	Comment Summary	Response
	over from older plans that have never been vetted for equity or impact, including transportation and public facilities	
59-10	Specify which groups of zoning changes within each alternative should be treated as divisible or as a cluster/group and describe why.	See response to comment 59-6.
59-11	Engage communities to explain the purpose of this EIS more clearly, the difference between the proposed zones and the Alternatives, and the legislative steps yet to come.	The comments are noted. Section 1.4.2 describes public comment opportunities to develop the proposals. The Draft EIS comment period of 45 days was extended several weeks, and more engagement was conducted in Georgetown and South Park. The City will continue to engage with communities after publication of the Final EIS related to potential legislation to make comprehensive plan policy amendments and/or zoning changes. The City will also engage with communities during updates to subarea plans.
59-12	Clarify which existing and proposed uses in the industrial areas will be considered nonconforming under the MML, II, and UI zones.	See response to comment 59-2.
59-13	Clarify the definition of "industry supportive housing," provide examples from other locations of mixed-use housing/industrial, and propose thresholds for mixed-use buildings.	See response to comment 59-3. See Section 4.2.3 concerning industry supportive housing.
59-14	Develop a complete list of the neighborhood- level comprehensive plan recommendations in areas that will be impacted by these zoning changes, and analyze whether they conform or contradict the Draft Comprehensive Plan Goal and Policy Language found in Appendix D.	See response to comment 59-4.
60	Dunn	Individual
60-1	Limited services in West Seattle and traffic on/off the peninsula is a major contributor to air and water pollution, unhealthy noise levels, and climate warming that will eventually exacerbate our growing climate crises.	Thank you for your letter. The commenter's perspective on existing traffic congestion and other environmental conditions is noted.
60-2	Riding bicycles should be made safer by slowing down freight and vehicle traffic on W Marginal Way, Spokane St, and E Marginal Way. Seattle should reduce southbound vehicle traffic to one lane on W Marginal Way between the West Seattle Bridge and Duwamish Longhouse to mitigate environmental impacts so that bicycle riders have a safe connection instead of riding on a sidewalk and develop safe routes throughout industrial and maritime areas.	The commenter's suggestion to implement traffic calming measures and improve bike facility connectivity along W Marginal Way is noted. That location is identified in Draft Exhibit 3.10-21 (Final EIS Exhibit 3.10-24) as having a planned multi-use trail. The City is currently considering options to fill the identified trail gap.
61	Eldridge	Individual
61-1	EIS should consider how future zoning counteracts the existing racialized exclusionary zoning history.	Comments is noted. The Final EIS includes additional information on this topic in Section 3.8 Land & Shoreline Use under the overview of historical planning and land use decisions subsection.
62	Fragada	Individual

Number	Comment Summary	Response
62-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
63	Frishholz	Individual
63-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
64	Fiorito	Individual
64-1	Support for UI zoning and opportunity for makers space.	Thank you for your letter. The comment is noted and forwarded to City decision makers. The Preferred Alternative identifies UI for the site similar to alternatives 3 and 4.
65	Graham	Individual
65-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
66	Greene	Individual
66-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
67	Hammerberg	Individual
67-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
68	Hanlon	Individual
68-1	In UI or II zones, how will adverse impacts (noise, traffic) be enforced after business hours? How does the SIMS address community concerns over code enforcement?	Thank you for your letter. City noise regulations (SMC 25.08) establish exterior sound level limits for various land use zones with the limits varying depending on the source zone and the receiving zone (see Exhibit 3.6-2). These limits are intended to result in acceptably low interior noise levels for residences and other sensitive noise receptors. City noise regulations also address construction noise, limiting the times during the day when construction noise, both impact and non-impact, can exceed exterior noise limits (see Exhibit 3.6-3). Noise limits are enforced by the City's noise abatement coordinators. The Seattle Police Department handles response to public nuisance noise—such as horns or sirens, music, amplified sound, motor vehicles, or watercraft—via the non-emergency line. A mitigation measure has been added to improve coordination and improve the user experience for community members registering complaints or requesting information about enforcement under air quality/ghg, noise, and contamination topics.
69	Huling	Individual

Number	Comment Summary	Response
69-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
70	Kartchner	Individual
70-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
71	Katz	Individual
	Note: Comments overlap with comment letters 32 and 59. Responses here are primarily cross- referenced to letter 59.	
71-1	Engage communities to explain the purpose of this EIS more clearly, the difference between the proposed zones and the Alternatives, and the legislative steps yet to come. Address small business displacement.	Thank you for your letter. Comment is noted. See response to comment 59-11.
71-2	Emphasize a greater partnership with Indigenous communities and Indigenous sovereignty.	The comment is noted. Cultural resources review is a process that is done prior to the start of many projects, and includes consultation with potentially affected Tribes. Many federal, state, and local statutes and ordinances require notice and consultation with affected Tribes before, during, and after project review. The National Historic Preservation Act (NHPA) of 1966, was amended in 1986 with provisions for consultation with affected Tribes and 1992 to include and clarify the roles and responsibilities of Indian Tribes in Section 106 reviews. The Advisory Council on Historic Preservation (ACHP) adopted a Policy Statement Regarding the ACHP's Relationships with Indian
		Tribes in 2000. The policy was developed in consultation with some Tribes and inter-Tribal organizations, and addresses tribal sovereignty, government-to-government consultation, trust responsibilities, tribal participation in historic preservation, sympathetic construction, and respect for tribal religious and cultural values.
		The state of Washington has a government-to-government relationship with the 29 federally recognized Tribes in the state (RCW 43.376). Each Tribe is a sovereign nation and has its own definition of appropriate consultation.
71-3	Present a clear path to support daily air monitoring in Ballard-Interbay.	As described in Section 3.2.1 Air Quality & GHG , the Puget Sound Clean Air Agency (PSCAA) has local authority for setting regulations and permitting of stationary air pollutant sources and construction emissions. PSCAA and Ecology maintain and operate a network of ambient air quality monitoring stations measuring the levels of criteria pollutants found in the atmosphere throughout the region, with the Ecology-operated site at 10th and Weller the closest network station to the Interbay-Ballard subarea (<u>https://secure.pscleanair.org/AirQuality/NetworkMap</u>). In addition, PSCAA maintains an air quality senor map that displays calibrated data for a variety of pollutants, measured by lower-cost portable air quality devices, including dust, fine particulate matter, carbon dioxide, carbon monoxide, ozone, nitrogen oxide, and others (<u>http://map.pscleanair.org/?lat=47.6768311&lon=-</u>

Number	Comment Summary	Response
		<u>122.4/56425&Z=9</u>). These air sensors are intended to be educational and are non-regulatory, meaning that they cannot be used for permitting, compliance, policy, or interpretation of health effects. The data from these sensors are not owned by PSCAA.
		Text has been added to Section 3.2.3 Air Quality & GHG to suggest consideration of a City-owned and operated air monitoring station in Ballard-Interbay to provide the public with access to daily air monitoring data.
71-4	Prioritize dramatic visual cues in built environment to get people who are driving vehicles to slow down on major arterials and urban freeways.	The comment is noted and forwarded to City decision makers.
71-5	Address the power and values imbalance caused by freight lobby's political pressure.	The comment is noted and forwarded to City decision makers.
71-6	Highlight the unique importance of Ballard- Interbay as a freshwater harbor.	The comment is noted. A reference noting the freshwater nature of the harbor is added in the description of the study area in Chapter 2 .
71-7	Highlight BNSF's historic and continuing lack of transparency and accountability.	The comment is noted and forwarded to City decision makers.
71-8	Clarify which existing and proposed uses in the industrial areas will be considered nonconforming under the MML, II, and UI zones.	See response to comment 59-2.
71-9	Clarify the definition of "industry supportive housing," provide examples from other locations of mixed-use housing/industrial, and propose thresholds for mixed-use buildings.	See response to comment 59-3.
71-10	Develop a complete list of the neighborhood- level comprehensive plan recommendations in areas that will be impacted by these zoning changes, and analyze whether they conform or contradict the Draft Comprehensive Plan Goal and Policy Language found in Appendix D.	See response to comment 59-4.
71-11	Specify which groups of zoning changes within each alternative should be treated as divisible or as a cluster/group and describe why.	See response to comment 59-6.
71-12	Add documentation, analysis, and maps that connect Seattle's historic segregation, redlining, and exclusion to the present-day location of industrial uses.	See response to comment 59-7.
71-13	Complete a citywide analysis of zoning that looks specifically at the ways commercial and multi-family exclusions in other parts of the city lead to the competition for industrial land	See response to comment 59-7.
71-14	Examine which recommendations and boundaries are carried over from older plans that have never been vetted for equity or impact, including transportation and public facilities.	See response to comment 59-7.

Number	Comment Summary	Response
71-15	Examine comments submitted by the	See response to comment 59-1.
	Duwamish River Community Coalition, Seattle Cruise Control, and the Georgetown/South Park Advisory Group.	Comments from the Duwamish River Community Coalition, Seattle Cruise Control, and Georgetown/South Park Advisory Group are addressed in letters 93, 37, and 96, respectively.
71-16	Requests additional scrutiny regarding the impacts of the systemic racist policies that created Seattle's industrial land and exacerbated the disparate impacts of pollution and disinvestment on nearby underserved neighborhoods of color.	See response to comments 59-1 and 59-7.
71-17	In the MML zone, code should clarify which existing and proposed uses will become nonconforming and should accommodate uses such as the WNBA Storm practice facility.	See response to comment 59-2.
71-18	In the UI zone, clarify the definition of industry supportive housing, provide examples from other locations of housing on top of industry, and propose thresholds for mixed use buildings.	See response to comment 59-3.
71-19	EIS does not examine where the II zone expressly contradicts existing neighborhood plans. EIS should include a complete list of the neighborhood-level comprehensive plan recommendations impacted by these zoning changes and analyze whether they conform or contradict the Draft Comprehensive Plan Goal and Policy Language in Appendix D.	See response to comment 59-4.
71-20	Agree with how the EIS alternatives are organized, but the document can be clearer about the distinction.	See response to comment 59-5.
71-21	Support Alternative 4 only because there are no alternatives that more liberally use the UI and II zones across larger portions of the city.	See response to comment 59-5.
71-22	EIS must do a better job establishing why areas change under each of the alternatives, and which areas should be treated as a cohesive cluster.	See response to comment 59-5.
71-23	At the neighborhood level, the proposed maps do not offer a picture of cohesiveness. What does it mean if blocks are divided?	See response to comment 59-6.
71-24	Alternative 1 should be considered a non-starter.	See response to comment 59-6.
71-25	City's industrial boundaries carry the history of segregation that cannot be washed away with a cursory equity analysis.	See response to comment 59-7.
71-26	EIS doesn't consider how boundaries of the current industrial zones came to exist.	See response to comment 59-7.
71-27	Impossible to develop policies that address land use and zoning issues without considering large areas of the city devoted exclusively to single-	See response to comment 59-7.

Number	Comment Summary	Response
	Tamily nousing. EIS must more thoroughly consider equity impacts, including connecting Seattle's historic segregation, redlining, and exclusion to the present-day location of industrial uses, completing a citywide zoning analysis looking at commercial and multi-family exclusion in other areas, and examining which recommendations and boundaries are carried over from older plans that have never been vetted for equity or impact.	
71-28	EIS must make robust efforts to understand history and the sources of inequity in shaping land use decisions.	Comment is noted. See response to comment 59-5.
72	Kromm	Individual
72-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
73	Lau	Individual
70-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
74	Lewis	Individual
74-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
75	Livingston	Individual
75-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
76	Main	Individual
76-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
77	Mathison	Individual
77-1	Does not support IB zoning designation in Ballard Subarea and desires more housing alternatives.	The comment is noted and forwarded to City decision makers. Under the action alternatives IB zoning in Ballard would be replaced by a combination of UI or II zones. The UI zone would allow some expansion of allowances for industry-supportive housing under some of the alternatives.
78	Menin	Individual
78-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative

Number	Comment Summary	Response
		incorporate an increase in the maximum size of use for indoor sports and recreation uses.
79	Olofson	Individual
79-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
80	Perry	Individual
80-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
81	Personett	Individual
81-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
82	Phillips	Individual
82-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
83	Robinson	Individual
83-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
84	Shaffer	Individual
84-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
85	Shaw	Individual
85-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
86	Standifer	Individual
86-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
87	Strohmeier	Individual
87-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative

Number	Comment Summary	Response
		incorporate an increase in the maximum size of use for indoor sports and recreation uses.
88	Sundquist	Individual
88-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
89	Wood	Individual
89-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
90	Anonymous	Individual
90-1	Compare historic segregation, redlining, and exclusion to present day location of industrial uses. Consider how constraints in non-industrial zones citywide lead to competition for industrial land.	The comment is noted. A new subsection in the Final EIS is added to the Overview of Historical Planning and Land Use Decisions in Section 3.8.1 .
90-2	Specify which groups of zoning changes within each alternative should be treated as divisible or as a cluster/group and why.	The EIS Action Alternatives include a range of different geographic patterns of zoning changes that take into account numerous context specific factors.
90-3	Ensure zoning around high capacity transit nodes extends out the full ½-mile in each direction.	The action alternatives apply II zoning in various extents from future transit stations to 1/2 mile and more in certain instances.
91	Schaefer	Cascade Bicycle Club
91-1	Process has not included citywide outreach and is happening independently from Comprehensive Plan update and STP processes. Request that industrial zoning changes should be wrapped into Comprehensive Plan process and that safety is paramount.	Thank you for your letter. See response to comment 4-29 regarding the ElS's approach and findings regarding safety. See Section 4.2.9 concerning coordination with the Comprehensive Plan major update.
91-2	(1) Critical to have feedback from people who	(1) See responses to comments 91-3 and 91-4.
	walk, roll, and bike through industrial areas.	(2) See response to comment 91-5, 91-6, and 91-7.
	modes opens opportunities to jobs and supports City mode shift goals.	(3) See response to comment 91-8.
	(3) Changes to industrial zoning addresses pollution and climate change issues.	
91-3	Concern that changes to land use will be made before wider outreach around Comprehensive Plan and STP. Strategy assumes that the preservation of industrial land uses is the best and only outcome.	See Section 4.2.9 concerning coordination with the Comprehensive Plan major update process. The proposed alternatives include different combinations of potential zoning changes, some of which reduce the amount of industrially zoned lands and/or increase flexibilities for uses other than traditional industrial activities in the study areas.
91-4	Industrial areas are of particular concern because of key cycling routes, lack of street improvements, conflicts with large trucks etc.	The commenter's concerns about the challenges in the study area are noted. The EIS acknowledges the biking and walking conditions in the study area, and concludes the network gaps and conflicts between cars/trucks and vulnerable users would be a significant impact.

Number	Comment Summary	Response
91-5	Safe bike routes are attractive to potential employees; all industrial jobs should be accessible by walking, biking, and transit.	Comment noted. The City shares the goal of allowing for improved travel by non-auto modes. All modes are addressed in the EIS including mitigation measures to encourage travel by transit, walking, and biking.
91-6	Development standards should be updated to require frontage improvements that increase safety for walking and biking and planting of trees to reduce heat island effects.	Development standards including street improvement requirements would be updated for the proposed new zones under the action alternatives. The UI and II zones would have higher standards for frontage improvements compared to the zones they would replace. See also Appendix G for a more detailed discussion of development standards provided in the Final EIS.
91-7	Conduct more detailed existing land use analysis and consider corridors that could have more UI zoning or other non-industrial uses which could support a safer biking corridor from Georgetown to downtown.	Section 3.8 Land & Shoreline Use includes analysis of existing land uses. See response to 91-5.
91-8	Pollution and climate change are poorly addressed by all options.	Section 3.3.2 discusses the expected increase in traffic for all alternatives and states that improvements in vehicle standards and the application of stormwater requirements during redevelopment described in this and other sections of the EIS are expected to offset the increase in traffic and potentially lead to a net decrease in surface water pollution. See Section 3.3 Water Resources .
		Section 3.2.1 acknowledges that industrial uses contribute to air quality emissions that can affect human health. That section also discusses the regulatory framework for limiting air emissions. Section 3.2.3 cites possible mitigation measures for air emissions that include changes to the Seattle Comprehensive Plan and future MIC Subarea Plans recommending residences and other sensitive land uses (i.e., schools, day care) be separated from freeways, railways, and port facilities, and new MML, II, and UI zones by a buffer area of no less than 500 feet, and possibly as much as 1,000 feet, depending on the height of the source, to reduce the potential exposure of sensitive populations to air toxics. See Section 3.2 Air Quality & GHG.
		Appendix G also shows potential conceptual development regulations associated with the Preferred Alternative. The Preferred Alternative includes a basic 200 feet between truck routes to housing. Through the permit review process or SEPA review of site- specific proposals, the City can consider building and site design, topography, traffic volumes, and level of air emissions or noise and require a greater distance at a project level.
91-9	Concern that there are not commitments to mitigation measures.	Comment noted. See Section 4.2.7 concerning mitigation measures.
92	Fong	Center for Ethical Leadership
92-1	Consider working more closely with community leaders living in the impacted neighborhoods such as, Georgetown Community Council, King County International Community Coalition, and many others. Create a holistic, sustainable, and community-driven industrial lands strategy that addresses affordability, environmental impacts, and equity across Seattle.	Thank you for your letter. The comment is noted. OPCD will continue to pursue close community engagement with community members in Georgetown and South Park and other areas. This will include ongoing engagement after the Final EIS is issued and before any changes to land use policies or zoning are made. See also Section 4.2.8 concerning community engagement.
02		Development of the second shifts of the second

Number	Comment Summary	Response
93-1	Concern that the Draft EIS did not consider impacts of industrial uses on residential community members in Georgetown and South Park. There are cultural artifacts in the Duwamish River flood plain. There need to be more green spaces and native trees in the area.	Thank you for your letter. The EIS contains an analysis of existing conditions and measurement of impacts under the alternatives for each element of the environment, such as Air Quality & GHG , Noise , etc. Section 3.11 Historic, Archaeological, & Cultural Resources includes an analysis of archaeological resources. The UI zone would have higher standards for landscaping and tree planting with new development than the zone it would replace under the alternatives.
94	Bush	Duwamish Valley Safe Streets
94-1	Isolated improvements fall quite short in providing the transformative vision for this area that is long overdue. The members of Duwamish Valley Safe Streets stand with our fellow community members in great concern that the process for this planning effort and strategy has not had a citywide outreach process and is happening independently from both the updates to the Comprehensive Plan and the new Seattle Transportation Plan.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See Section 4.2.9 .
94-2	Community members face conflicts with large vehicles, poorly defined and unimproved roadways, lack of sidewalk, rough railroad tracks, and poor air quality. Commenter requests: (1) Feedback from community members who walk, bike, and use other non-motorized modes through industrial areas (2) Land use decisions led by environmental historical inequities (3) Integrate better planning for pedestrian and bicycle routes and public transportation investments	The commenter's concerns about the challenges in the study area are noted. The EIS acknowledges the biking and walking conditions in the study area and concludes the network gaps and conflicts between cars/trucks and vulnerable users would be a significant impact. (1) See Section 4.2.8 . The Industrial and Maritime Strategy Council included persons with advocacy and expertise in transportation, including the Director of the Transportation Choices Coalition. (2) The EIS considers historic planning and inequities. See Sections 3.8.1 and 3.9.1 . (3) The EIS includes mitigation measures related to pedestrian and bicycle improvements as well as TDM measures that could include public transit programs gapted toward the unique people of the
	Investments	study area. Moreover, SDOT is currently in the process of the developing the Seattle Transportation Plan which will integrate the City's modal plans into a comprehensive vision for the citywide transportation network centered around the following values and goals: equity, safety, mobility, sustainability, livability, and excellence.
94-3	Pollution and climate change are poorly addressed by all options.	See responses to comments 97-4, 97-19, and 91-8.
94-4	While strategy provides for some adjustment in land uses in the industrial areas, the approach taken within this document falls short. We ask that any changes to industrial land uses should be wrapped into the process for the Comprehensive Plan and involve a more robust and equitable outreach effort.	The comment is noted and forwarded to City decision makers. See Section 4.2.9 .
95	Farrazaino	Equinox Development Unlimited LLC
95-1	Summary of comments. Appreciate opportunity to comment and extension of comment period. Support the Duwamish Tribe, Georgetown Community Council, etc,	Thank you for your letter. The comments are noted and forwarded to City decision makers.

Number	Comment Summary	Response
95-2	Pause this process and take the time and actions needed to authentically engage all of the stakeholders to either validate the premise and details of this Draft EIS or create a new one.	See Section 4.2.8.
95-3	Integrate the Industrial and Maritime Strategy and any potential Alternatives, including the No Action Alternative, into the Comprehensive Plan process.	See Section 4.2.9.
95-4	Study the intersectional and cumulative impacts of the 14 affected environments in the Draft EIS and plan for and enact mitigation measures to address these exponentially more intense impacts.	See Section 4.2.7.
95-5	Institute mechanisms to protect current community before, or in conjunction with, making zoning changes.	See Section 4.2.7.
95-6	Inventory actual use of all properties to determine efficacy of potential changes or effects of no action.	See Section 3.8 Land & Shoreline Use , which includes parcel- specific land use maps.
95-7	Study the actual financial implications and market conditions to validate the efficacy of your assumptions and adjust the alternatives, either in location or development capacity, to suit the stated goals.	See Section 4.2.1 . SEPA does not require a cost-benefit or economic analysis. Note that SDOT is currently in the process of developing the Seattle Transportation Plan which will integrate the City's modal plans into a comprehensive vision for the citywide transportation network centered around the following values and goals: equity, safety, mobility, sustainability, livability, and excellence.
95-8	Need intersectional/cumulative approach to assessing the alternatives. Need inclusion and entrepreneurship. Need this to meet climate goals and environmental justice.	See Chapter 1 of the EIS including Section 1.7.15 which summarizes equity and environmental justice and highlights results of the environmental evaluation including air quality and sea level rise.
95-9	Study how no action or proposed alternatives tangibly and directly improve economic, environmental, and health disparities or continue the historic disenfranchisement of the Duwamish communities.	The EIS focuses on environmental impacts and addresses some subjects important for health including air quality and noise.
95-10	Studying an expansion of housing into the Industrial areas as a means to preserve existing manufacturing and jobs, create new modern manufacturing and industrial jobs, increase residential and commercial affordability, bring environmental investments, increase safety, and bring better outcomes for our BIPOC communities, should be done as soon as possible so potential benefits can be incorporated into our Comprehensive Plan process.	The comments are noted. See Section 4.2.10 .
95-11	Convene the Strategy Council and Community Based Organizations to identify and recruit stakeholders from all constituencies to form and maintain the stewardship entity now so it can carry this work forward with authentic engagement.	The comments are noted and forwarded to City decision makers. See also Section 4.2.8 .

Number	Comment Summary	Response
96	Ramirez	Georgetown Community Council, King County International Airport Community Coalition
96-1	Significant change is needed to achieve a progressive, affordable, and sustainable strategy that meets the needs of Georgetown residents, small businesses, and workers. We look forward to your response, and we remain ready to collaborate on this effort.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
96-2	The UI zone has the potential for increased affordability, sustainability, and equitable outcomes. However, areas proposed for UI under the alternatives would make no material changes. Other zoning options to create bigger buffers should be considered including Commercial zones.	Please see Section 4.2.6 regarding the UI zone and adjustments made in response to community input in the Preferred Alternative.
96-3	Study expansion of Commercial or Mixed Use zoning for more areas in and around Georgetown. Connect the neighborhood. Decrease the amount of MML zoning in and around Georgetown.	Please see Section 4.2.5 regarding an enlarged Mixed Use area in Georgetown.
96-4	Much of the land the City has zoned as MML has—in reality—been full of mixed uses for decades. Create a meaningful buffer zone between our residential areas, thriving commercial core, and heavy industry.	Analysis of existing land uses is included in the Land Use chapter, including quantitative data and narrative description.
		Please see Section 4.2.6 regarding the UI zone and adjustments made in response to community input in the Preferred Alternative.
96-5	A fundamental flaw of the Draft EIS process is that the accompanying mitigation measures are merely suggestions, and will not be put forward as binding legislation eventually passed by the City Council.	Please see Section 4.2.7 regarding mitigation measures.
96-6	The Draft EIS makes zoning changes that need accompanying policy commitments in order to maximize their impact. For example, rezoning part of Airport Way from Industrial to Mixed Use has lots of potential benefits for the neighborhood. However, it requires accompanying policies from the City—such as commitments regarding historic preservation and affordable housing—to ensure the zoning changes align with the policy intent of the neighborhood, and don't exacerbate affordability and equity issues.	The Final EIS includes additional details about proposed development standards that would be unique to the Georgetown area to address concerns raised by community members in this and other comment letters. Please see Sections 4.2.5 and 4.2.6 .
96-7	The GCC supports the Duwamish River Community Coalition's request for a year-long extension to the Draft EIS to allow for meaningful engagement with impacted residents.	Please see Sections 4.2.8 and 4.2.9 .
96-8	Fold the Draft EIS process into the Comprehensive Plan update.	Please see Section 4.2.9.
97	Hampton-Clarridge	Georgetown Community Council, King County International Airport Community Coalition, Duwamish River Community

Number	Comment Summary	Response
		Coalition, Duwamish valley Affordable Housing Coalition, Duwamish Valley Safe Streets
97-1	The Draft EIS is deeply connected to the history of white settlement, heavy industrialization, and discriminatory housing policies that have left the Duwamish Valley community fighting for the advancement of environmental and climate justice for decades to come. The City must remain accountable to its actions and prioritize the wellbeing of the Duwamish Valley community over industry and profit in the Industrial and Maritime Strategy.	Thank you for your letter.
		The history of City Planning & Land Use Decisions is found in Section 3.8.1 including how expansion of industry affects residents in the Duwamish Valley.
		The EIS includes an evaluation of equity and environmental justice in Section 1.7.15 .
		Section 3.9 Housing addresses the relationship of housing and disparities including exposure to pollution. The EIS provides additional mitigation measures meant to address health and safety, (e.g., air quality, noise, light and glare, etc.). A complete list of mitigation measures is found in Final EIS Appendix J .
97-2	The Industrial and Maritime Strategy is an opportunity for the City of Seattle to right the wrongs set forth by the white settlement and early industrialists of the Seattle area, an issue of zoning and land use change.	The comment is noted and forwarded to City decision makers. Regarding engagement, please see Section 4.2.8 .
	In addition, the strategy presents a unique opportunity for the City to reconfigure processes for on-going, low-barrier, multilingual community engagement regarding land use updates for a more inclusive and fair engagement process. More so, the Industrial and Maritime Strategy should not move forward independently of the Comprehensive Plan, Seattle Transportation Plan and Freight Master planning.	
97-3	Long-standing advocacy on issues, such as industrial pollution, that remain unresolved and will be made worse by an increasing population and activities proposed by the Industrial and Maritime Strategy (alternatives 3 and 4).	Please see response to comment 97-1.
97-4	Concern that more housing will increase exposures to contaminants by more people. Encourages more legislation to increase environmental regulation standards.	Comment is noted. Refer to Section 3.5.1 that describes several robust regulatory frameworks (MTCA, CERCLA, RCRA) that converge to regulate site investigations and cleanup activities as well as proper use, handling, and offsite disposal of hazardous materials used by industry or generated during site cleanups. As experienced by this EIS section author, Ecology, EPA, and others are emphasizing careful review of all site cleanup and redevelopment projects near the Lower Duwamish Waterway superfund site to ensure that stormwater and dewatering water generated during construction are carefully managed, and site cleanup work meets the low cleanup levels necessary to prevent recontamination of areas previously cleaned up. Ecology is also highly engaged and aware of the importance of the public participation process.
97-5	To protect and support industry and Port operations without procedural justice and higher environmental standards for the residential communities of South Park and Georgetown ignores the reality of today and should not be acceptable to any of us.	Please see response to comment 97-1.

Number	Comment Summary	Response
97-6	This letter first explains why strong environmental	Please see response to comment 97-1.
	standards and meaningful engagement of the diverse Duwamish Valley community is necessary to eliminate negative cumulative health impacts experienced everyday, and why the Draft EIS must check the integrity of its data analysis and mitigation measures to eliminate bias and injustice towards a community that has long been affected by racism rooted in environmental and land use planning and policy.	
97-7	The significance of including the history of the Duwamish River and segregation in the City of Seattle is to shed light on the intersectional nature of land use and zoning change and its role in discriminatory practices that still impact Seattle today.	Please see response to comment 97-1.
97-8	Exposure to odors and noise.	ElS Section 3.2.1 acknowledges that industrial uses contribute to air quality emissions, including odors. That section also discusses the regulatory framework for limiting air emissions. Section 3.2.3 cites mitigation measures in the form of regional regulations by PSCAA for emission controls to minimize fugitive dust and odors during construction, permitting of stationary air pollutant sources. See Section 3.2 Air Quality & GHG .
		Section 3.6 Noise discusses potential noise impacts associated with implementing the alternatives; a description of noise and noise levels in general; regulatory standards for noise; noise sources and potential sensitive noise receptors in the maritime and industrial areas of Seattle; an assessment of noise impacts associated with each alternative, as well as potentially feasible noise mitigation measures where appropriate.
		Maps illustrating exposure to pollution are included in EIS Section 3.9 .
97-9	Comprehensive rules for increased environmental standards and protections from displacement driven by market forces must be enacted.	Comment is noted. See response to comment number 97-4.
97-10	Air quality and health.	Section 3.2.1 acknowledges that industrial uses, including associated diesel-related emissions from industrial use trucks, contribute to air quality emissions that can affect human health. That section also discusses the regulatory framework for limiting air emissions. Sections 3.2.3 and 3.9.1 include a discussion of the Duwamish Valley's ranking on the Washington State Department of Health (DOH) health disparities map (DOH 2021). Section 3.2.3 discusses mitigation measures for air emissions in the MICs, including the Duwamish Valley, that identify strategies to reduce the potential for exposure of existing and new employees, residents, and visitors to potential air emissions, including metals, in areas around arterials, along industrial buffers, and near port operations. See Section 3.2 Air Quality & GHG.
97-11	Inaccuracy on access to parks and open space in Georgetown and South Park.	Exhibit 3.12-10 referenced by the commenter is sourced from the Seattle Duwamish Valley Action Plan and indicates a relatively
Number	Comment Summary	Response
--------	---	---
		nigner percentage of access to public space in South Park and Georgetown than Citywide. That exhibit is followed by text acknowledging a need for improved parks and open space access in South Park and Georgetown: "While the neighborhoods have nearby parks, the total acreage per capita is half the citywide average and there may be park congestion caused by added population. Another factor related to park pressure and park access is being able to travel to and from the parks."
97-12	Air particulates and air quality monitoring network in the Duwamish Valley.	The Potential health impacts of particulate matter are discussed in Section 3.2.1 Air Quality & GHG, Pollutants of concern. Additional text has been added to include fugitive roadway dust as a source of particulate matter. The potential for fugitive dust emissions associated with soil-disturbing activities, demolition and construction work, and grading are discussed in general in Section 3.2.2, Construction Related Emissions. The potential for vehicle travel to generate PM2.5 from road dust is discussed in Section 3.2.2, Impacts of Alternative 1, Transportation Related Emissions. Discussion under Transportation Related Emissions for alternatives 2, 3, and 4 compare emissions to Alternative 1. Additional text is added in each of these sections to include the potential generation of dust associated with increased vehicle miles traveled. Additional text is added to Section 3.2.3 regarding increased street sweeping to prevent impacts from fugitive dust.
		This non-project EIS provides an assessment of the existing levels of regulated pollutants and compliance with the NAAQS, and anticipated air emissions associated with potential land use changes based on two sources of baseline ambient air quality conditions data: 1) from Ecology- and PSCAA-operated ambient air quality monitoring stations; and 2) from air quality data collected directly by The City of Seattle at eight sites within the BINMIC and Greater Duwamish MIC—selected due to the location of potential zoning changes in alternatives or due to their proximity to air quality emission sources. All data indicate that air pollutant concentration trends, and individual measurements, for these pollutants remain below the NAAQS when wildfire is excluded.
		As described in Section 3.2.1 , the Puget Sound Clean Air Agency (PSCAA) has local authority for setting regulations and permitting of stationary air pollutant sources and construction emissions. PSCAA and Ecology maintain and operate a network of ambient air quality monitoring stations measuring the levels of criteria pollutants found in the atmosphere throughout the region, with the Ecology- operated site at 10th and Weller the closest network station to the Interbay-Ballard subarea (https://secure.pscleanair.org/AirQuality/NetworkMap). In addition, PSCAA maintains an air quality senor map that displays calibrated data for a variety of pollutants, measured by lower-cost portable air quality devices, including dust, fine particulate matter, carbon dioxide, carbon monoxide, ozone, nitrogen oxide, and others (http://map.pscleanair.org/?lat=47.6768311&lon=- 122.4756425&z=9). These air sensors are intended to be educational and are non-regulatory, meaning that they cannot be used for permitting, compliance, policy, or interpretation of health effects. The data from these sensors are not owned by PSCAA. Text has been added to Section 3.2.3 to suggest consideration of a
		City-owned and operated air monitoring station in the Duwamish Valley to provide the public with access to daily air monitoring data.

Number	Comment Summary	Response
97-13	VMT related to air quality	Air Quality & GHG Section 3.2.1 acknowledges that industrial uses, including associated diesel-related emissions from industrial use trucks, contribute to air quality emissions. Section 3.2.2 discusses the anticipated VMT under each of the alternatives and the associated potential impacts on air emissions in Transportation Related Emissions. Baseline ambient air quality conditions data is presented: 1) from Ecology and PSCAA-operated ambient air quality monitoring stations; and 2) from air quality data collected directly by The City of Seattle at eight sites within the BINMIC and Greater Duwamish MIC—selected due to the location of potential zoning changes in alternatives or due to their proximity to air quality emission sources. Modeled vehicle VMT (see Section 3.10 Transportation) is used to project anticipated air emissions from transportation sources based on emission factors reflecting future improvements to the vehicle fleet using the AFLEET tool (2020 version) and data from the EPA MOVES2014b model. All data indicate that air pollutant concentration trends, and individual measurements, for these pollutants remain below the NAAQS when wildfire is excluded.
97-14	Concern about lack of meaningful engagement to reach diversity of Duwamish Valley Community.	See Section 4.2.8 .
97-15	It is concerning that mitigation recommendations for the Draft EIS are not true commitments considered by the Strategy.	See Section 4.2.7 .
97-16	Air Quality and increased GHG emissions.	Section 3.2 Air Quality & GHG discusses the significance of anticipated GHG emissions. It concludes that "through mitigation implementation, local and state climate actions, and expected continued regulatory changes, the alternatives may result in a decrease of the growth in GHG emissions [due to population and employment growth] such that the impacts from future development allowed by the changes in plans and zoning could be considered less than significant for SEPA. As proposed, the alternatives would not prevent or deter efforts to reduce emissions in comparison to local or regional goals or targets for GHG reductions."
97-17	Air Pollution and mitigation.	The comment is noted. Section 3.2.3 Air Quality & GHG discusses mitigation measures for air emissions in the MICs that identify strategies to reduce the potential for exposure of existing and new employees, residents, and visitors to potential air emissions in areas around arterials, along industrial buffers, and near port operations.
97-18	Displacement: The description of risk of displacement does not reflect community concerns regarding displacement pressures and affordability.	The displacement analysis in Section 3.9 Housing uses the City's Displacement Risk Index and Access to Opportunity Index. It also considers the limited housing within the MIC boundaries of around 413 dwellings across the nearly 7,000 acres. The compatibility concerns between industrial uses and abutting residential areas outside the boundaries is addressed in Section 2.8 together with mitigation measures.
		Section 3.9 Housing also provides for mitigation measures to address the potential for employment growth to shift housing demand, and apply MHA regulations, in the II zone.

Number	Comment Summary	Response
		I ne Preterred Alternative responds to concerns about Georgetown arts and culture displacement and housing needs. See Section 4.2.5 .
97-19	Impacts of sea level rise and additional threats of climate change must be taken more seriously throughout all mitigation areas.	Sea level rise is addressed through existing regulations as discussed in Section 3.3.2. Subareas sensitive to sea level rise are discussed in this section, along with mitigation measures in Section 3.3.3. Given the non-project nature of this EIS, Section 3.3 Water Resources provides an appropriate level of detail on the risk and impact of development related to sea level rise. Subsequent developments that may arise from the proposed land use changes in the Industrial and Maritime Strategy will be required to meet all applicable codes and regulations, and to conduct project-level SEPA review at that time, in which analysis will be conducted to assess site specific impacts and necessary mitigation measures.
97-20	Fairness in zoning: Increase mixed-use areas in Georgetown and South Park to allow for a larger percentage of community-driven anti- displacement efforts.	Alternatives 3 and 4 provide for areas of Mixed Use in Georgetown and South Park. The Mixed Use area is increased in the Preferred Alternative in Georgetown. See Final EIS Chapter 2 description of the Preferred Alternative as well as Section 4.2.5 .
97-21	Send a companion binding legislation to the City Council that codifies and funds recommended mitigation measures.	The comment is noted and forwarded to City decision makers. Please see Section 4.2.7 .
97-22	The Draft EIS must consider an additional alternative that reflects all the priorities of the community for a fair consideration of proposed alternatives. Commit to continued community engagement. Expand buffers and UI zoning. Commit to mitigation measures. Increase credibility of data. Slow down EIS process. Address pollution. Fold the EIS into the Comprehensive Plan.	 Please see the description of the Preferred alternative that expands UI buffers and adds Mixed Use. Regarding specific EIS topics and information please see responses 97-1 to 97-21. See also: Community Engagement (Section 4.2.8) Mitigation Measures Commitment (Section 4.2.7) Strategy and Comprehensive Plan (Section 4.2.9)
97-23	The community continues to wait for equitable safeguards from neighboring polluters while business as usual continues. This chronic issue must be addressed and land use change presents a unique opportunity to rezone more spaces for the community in order to restore environmental health and champion placekeeping, economic justice and resilience. We strongly recommend the City of Seattle commit to frequent and authentic community engagement around land use in order to strengthen environmental standards. Prioritize the recommendations of the Duwamish Valley community.	See response to comment 97-22.
98	Davidson	Georgetown Merchants Association
98-1	Concern about the public engagement process.	Thank you for your letter. The comment is noted. See Section 4.2.8 concerning public engagement.
98-2	Request for more specific information on proposed zoning boundaries and mitigations that could address displacement.	The comment is noted. The EIS contains a detailed zoning map for each of the proposed alternatives found in Appendix C , and reviewable in the <u>online story map</u> . Increased detail about proposed development standards is contained in the Final EIS in Appendix G , including a subsection describing development

Number	Comment Summary	Response
		standards specific to the mixed use area of Georgetown under the Preferred Alternative.
98-3	Consider systemic impacts.	For each element of the environment (EIS Chapter 3) consideration is given to cumulative impacts.
98-4	We ask for more focus on public safety, acknowledgement of public safety issues in Georgetown, and commitments to public safety as a part of any changes that are made.	Impacts from the proposal to public services including police response times are included in Section 3.13 Public Services . The City acknowledges that existing public safety concerns in industrial area are a high priority for many stakeholders in those areas. Although addressing existing public safety challenges is a part of the broader Industrial and Maritime Strategy, this topic is separate from the land use actions that are the focus of the proposed action.
99	Bookwalter	Georgetown Youth Council
99-1	Supports the Georgetown Community Council's comment letter.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See responses to letter 96.
100		Seattle Bicycle Advisory Board
100-1	Process has not included citywide outreach and is happening independently from Comprehensive Plan update and STP processes. Request that industrial zoning changes should be wrapped into Comprehensive Plan process and that safety is paramount.	Thank you for your letter. See response to comment 4-29 regarding the EIS's approach and findings regarding safety. See Section 4.2.9 concerning coordination with the Comprehensive Plan major update.
100-2	(1) Critical to have feedback from people who walk, roll, and bike through industrial areas.	See responses to comment 91-2
	(2) Planning for better access via non-auto modes opens opportunities to jobs and supports City mode shift goals.	
	(3) Changes to industrial zoning addresses pollution and climate change issues.	
100-3	Concern that changes to land use will be made before wider outreach around Comprehensive Plan and STP. Strategy assumes that the preservation of industrial land uses is the best and only outcome.	See response to comment 91-3. See Section 4.2.9 concerning coordination with the Comprehensive Plan major update process.
100-4	Industrial areas are of particular concern because of key cycling routes, lack of street improvements, conflicts with large trucks etc.	The commenter's concerns about the challenges in the study area are noted. See response to comment 91-4.
100-5	Safe bike routes are attractive to potential employees; all industrial jobs should be accessible by walking, biking, and transit.	See response to comment 91-5.
100-6	Development standards should be updated to require frontage improvements that increase safety for walking and biking and planting of trees to reduce heat island effects.	See responses to comment 91-6.
100-7	Conduct more detailed existing land use analysis and consider corridors that could have more UI zoning or other non-industrial uses which could support a safer biking corridor from Georgetown to downtown.	Section 3.8 Land & Shoreline Use includes analysis of existing land uses. See response to comment 91-5.

Number	Comment Summary	Response
100-8	Pollution and climate change are poorly addressed by all options.	See response to comment 91-8.
100-9	Concern that there are not commitments to mitigation measures.	Comment is noted. See Section 4.2.7 concerning mitigation measures.
101	Schwartz	South Park Neighborhood Association (SPNA)
101-1	Request for a year-long extension to the Draft EIS to allow for meaningful engagement with impacted residents. Outreach must also be accessible to non-native English speakers.	Thank you for your letter. The comment is noted and forwarded to City decision makers. Please see Section 4.2.8 .
102	Simson	Urban Systems Design
102-1	 The Draft EIS and overall strategy falls significantly short of meeting the needs and priorities of Georgetown and South Park residents, small businesses, and workers. Privileges future growth of industrial and maritime usages over actual creative industries proven to support and sustain local businesses; the consequences could mean the end of Seattle's legacy as an art and cultural center Insufficient study of impacts on existing vital arts and culture resources in the district All alternatives reduce or eliminate potential affordable housing Shows lack of consideration towards existing communities, families, and small businesss Threatens the future of core working art space which could sorely limit intrinsic creative resources 	Thank you for your letter. The EIS recognizes the lack of small or affordable space and housing for makers, creatives, and artists. Alternatives 3 and 4 and the Preferred Alternative address expanding allowances for limited industry-supportive housing such as caretakers' quarters and maker studios. Alternative 3 includes an estimated additional 610 limited industry supportive housing units in industrial zones. The Preferred Alternative would have an estimated 3,009 units across the full study area. The housing would be available to business owners or employees of an on-site business that is an industrial use, or available to artists/makers with a business license in live-work spaces. Live/workspaces contain area for production/art/making activities that are physically connected to residential space. The Preferred Alternative specifically addresses this issue with the new Mixed Use zone in the triangle area of Georgetown by creating incentives for retention, restoration, and reuse of historic-period buildings and arts organizations and/or art studios. See also Section 4.2.5 concerning retention of arts and cultural spaces in Georgetown.
102-2	UI has potential to increase affordability. Concern that the proposed UI zoned areas in Georgetown will not lead to material changes. Suggestion that more areas in Georgetown should be studied for a change to Commercial or mixed use zoning.	The comment is noted. See Section 4.2.6 concerning larger buffer areas and conversion of more land from MML zoning in and around Georgetown.
102-3	Suggestion to shift the Industrial and Maritime Strategy process into the Comprehensive Plan major update.	The comment is noted. See Section 4.2.9 concerning coordination with the Comprehensive Plan major update.
102-4	Requests rejection of all alternatives.	Comment is noted.
102-5	Increase study of and consideration for arts and cultural resources in Georgetown.	See response to 97-1 above, and response to frequent comment theme concerning arts and culture in Georgetown.
103	Benetua	Individual
103-1	Automatic vacation response.	Comment is noted.
104	Bookwalter, E.	Individual
104-1	Supports the Georgetown Community Council's comment letter and requests process be folded into the Comprehensive Plan update.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See responses to letter 96.

Number	Comment Summary	Response
105	Bookwalter, M.	Individual
105-1	Supports the Georgetown Community Council's comment letter and requests process be folded into the Comprehensive Plan update.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See responses to letter 96.
106	Bushue	Individual
106-1	Concern about conflict of interest with Ram Mounts/National Products.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
106-2	Concern about conflict of interest with Ram Mounts/National Products.	Comment is noted.
106-3	Concern about conflict of interest with Ram Mounts/National Products.	Comment is noted.
107	Carpenter	Individual
107-1	Supports the Georgetown Community Council's comment letter. Remove areas from MML to UI, MU, or Commercial: Orcas / E Marginal / Corson, Corson and Elysian Brewing, Airport Way S to S Lucille and other side of Airport Way. Have binding legislation to Council to codify mitigation measures. Commit to affordable housing and affordable housing.	Thank you for your letter. Comment is noted. See Section 4.2.6 concerning buffers and conversion of more MML land to other zones. See Section 4.2.7 concerning mitigation measures. See also responses to letter 96.
108	Claxton	Individual
108-1	Request for less heavy industrial and more housing. Supports the Georgetown Community Council's comment letter.	Thank you for your letter. The comment is noted and forwarded to City decision makers. The Action Alternatives propose targeted changes regarding housing, buffers to neighborhoods, and mitigation measures related to air quality, noise, sea level rise, and others. See also Sections 4.2.3 , 4.2.5 , 4.2.6 , 4.2.7 . See also responses to letter 96.
109	Cocking	Individual
109-1	Include the Draft EIS process with the upcoming Seattle Comprehensive Plan update.	Please see Section 4.2.9.
109-2	Clarify relationship of the Strategy with King County "sliver" annexation. What is role of the Port?	The sliver is identified as a possible future annexation area for the City; however, no timeline or specific plan for a possible future annexation is known at this time. The EIS considers existing conditions and existing plans and zoning within the sliver as a part of the affected environment. However, no changes to the sliver are proposed as a part of the alternatives.
109-3	This 'Plan' does nothing to help alleviate the toxic activities of industry and its encroachment into the healthier residential yards where homes exist.	Please see EIS Sections 3.8 Land & Shoreline Use, 3.9 Housing, 3.2 Air Quality & GHG, and 3.5 Contamination regarding industry and mitigation measures addressing compatibility and housing. The Preferred Alternative also expands the UI zone buffering uses, and mixed uses along boundaries of Georgetown and South Park. This is described in Chapter 2 of the Final EIS, as well as comment themes in Sections 4.2.3, 4.2.5 , and 4.2.6 .
109-4	Plan does not adequately address South Park zoning. It is not just a village. Residential UV was to stem rezoning to industrial. Work experiences by industrial workers are also missing in this study.	Please see the description of historic planning and inequity in Section 3.8.1 Land & Shoreline Use. Action alternatives including the Preferred Alternative do not expand MIC boundaries. Action alternatives make targeted adjustments to add Mixed Use in Georgetown and South Park. The

Number	Comment Summary	Response
		Preferred Alternative provides for more Mixed Use in Georgetown. The Preferred Alternative expands UI buffering near Georgetown and South Park. See Sections 4.2.5 and 4.2.6 . Please also see the commitment to mitigation measures in Section 4.2.7 .
100 5		Regarding workers in study area see response to comment 110-2.
109-5	called our shared subarea. Each community faces different impacts and is different. Georgetown is not a Residential Urban Village. Why are we? South Park has most IG zones and one or two buffer areas.	South Park are discussed in the Land Use chapter. For other aspects such as employment projection, it is not practical to disaggregate analysis to smaller geographies. South Park has been designated an urban village in the City's Comprehensive Plan for over 20 years. Designations as urban village growth areas are made as part of major comprehensive plan updates.
109-6	There should be emphasis on keeping the South Park residential area "green" to help mitigate the air quality and pollution here and there.	See EIS Sections 3.2 Air Quality & GHG and 3.12 Open Space & Recreation.
109-7	If South Park actually is an Urban Village then how convenient a motivation for you to shove more inappropriate dense housing into our green yards. correct existing zone designations in the residential area: (1) Remove Residential Urban Village status for South Park, Return to RS 5000 and include owner-occupied property be a must when making DADUor ADU on the property, and (2) do not allow Residential Small Lot zoning in South Park in order to avoid overbuilding on the already existing small lots here,	Regarding development in Urban Villages the City will consider housing needs across the City in the Comprehensive Plan Update as described in Section 3.9.3 . No changes to South Park residential zoning is proposed with the Industrial and Maritime Strategy.
109-8	Concerned about the proposed buffer zones between the industrial areas (UI, II, MML) and residential areas. Perhaps a more substantive buffer like Commercial 2 might be more effective for a transition between heavy industry and residential areas.	Comment noted. The alternatives study different combinations of zoning changes. Removal of land from MICs for placement in a non- industrial zone such as Commercial 2 are limited to focused locations to ensure consistency with the proposal's objectives.
109-9	Residents shouldn't have to monitor the developers and industrial neighbors but that is what it boils down to.	Comment noted. Discussion of increased coordination and effectiveness of enforcement by agencies is included in mitigations measures sections.
109-10	In order for 'urban industrial' to work, extensive testing and cleanup of buildings (reused) and land will need to be done. Environment and habitat will have to be healed through planting native plants and trees.	Please see Sections 3.4 Plants & Animals and 3.5 Contamination.
109-11	Treasure the RS 5000 lands and value them for their mitigation of the detrimental IG zones surrounding the yards and old homes in RS 5000. Don't apply inappropriate densities and MHA rezones.	No changes to South Park residentially zoned areas are proposed as a part of this action.
109-12	Stop trying to sacrifice South Park for new development either industrial or dense residential. Naturally occurring more affordable home ownerships that do not destroy the small	No changes to South Park residentially zoned areas are proposed as a part of this action.

Number	Comment Summary	Response
	town nistoric cnaracter of South Park should be encouraged	
110	Dae	Individual
110-1	Ul zone concept has potential to improve connectivity between residential and heavy industrial use areas.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
110-2	Proposed UI zoned areas in Georgetown are not likely to see land use changes.	See Section 4.2.6.
110-3	There aren't enough proposed UI zoned areas near Georgetown. More Commercial 2 zoned areas would provide a buffer.	The comment is noted. See Section 4.2.6 concerning buffer areas and reduction of MML zones in Georgetown.
110-4	Study expansion of more mixed use and Commercial 2 zoned areas.	The comment is noted. See Section 4.2.6 concerning buffer areas and reduction of MML zones in Georgetown.
110-5	Increase the area of zone changes around Georgetown to better connect the neighborhood. Create larger buffer areas.	The comment is noted. See Section 4.2.6 concerning buffer areas and reduction of MML zones in Georgetown.
110-6	Study replacing more MML zoned areas with Ul, Commercial or mixed use zoning. Specific areas are noted in the comment.	The comment is noted. See Section 4.2.6 concerning buffer areas and reduction of MML zones in Georgetown.
110-7	Study replacing more MML zoned areas with UI, Commercial or mixed use zoning. Specific areas are noted in the comment.	The comment is noted. See Section 4.2.6 concerning buffer areas and reduction of MML zones in Georgetown.
110-8	Accompany the Final EIS with legislation committing the City to fund mitigation measures. Concern that mitigation measures are only suggestions.	The comment is noted. See Section 4.2.7 concerning commitments to mitigation measures.
110-9	Accompany the Final EIS with legislation committing the City to fund mitigation measures. Concern that mitigation measures are only suggestions.	The comment is noted. See Section 4.2.7 concerning commitments to mitigation measures.
110-10	Extend the EIS process for a year. Concern that engagement has been with traditional stakeholders with power and influence.	The comment is noted. See Section 4.2.8 concerning the community engagement process.
110-11	Shift the process into the Comprehensive Plan major update.	The comment is noted. See Section 4.2.9 concerning coordination with the Comprehensive Plan major update.
111	Del Rio	Individual
111-1	Concerned the Draft EIS falls short of meeting needs of Georgetown residents, small businesses, and workers. Specifically, insufficient study of impacts on arts and cultural resources, reduction/elimination of existing affordable housing, lack of consideration towards existing communities, privileges future growth of industrial/maritime usages	Thank you for your letter. The EIS recognizes the lack of small or affordable space and housing for makers, creatives, and artists. Alternatives 3 and 4 addressed expanding allowances for limited industry-supportive housing such as caretakers' quarters and maker studios. Alternative 3 includes an estimated additional 610 limited industry supportive housing units in industrial zones, and Alternative 4 would have an estimated 2,195 units across the full study area. The housing would be available to business owners or employees of an on-site business that is an industrial use, or available to artists/makers with a business license in live-work spaces. Live/workspaces contain area for production/art/making activities that are physically connected to residential space.

Number	Comment Summary	Response
		I ne Preferred Alternative specifically addresses this issue with the new mixed use zone in the triangle area of Georgetown by creating incentives for retention, restoration, and reuse of historic-period buildings and arts organizations and/or art studios.
112	Facundo	Individual
112-1	Supports the Georgetown Community Council's comment letter.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See responses to letter 96.
113	Gallagher	Individual
113-1	Supports the Georgetown Community Council's comment letter and requests process be folded into the Comprehensive Plan update.	Comment is noted. See responses to letter 96.
114	Kirschenbaum	Individual
114-1	Disappointed by comment process. Proposed changes are marginal at best and favor the	Thank you for your letter. The comment is noted and forwarded to City decision makers.
	large industrial enterprises. The status quo has many current and future issues involving affordable housing, the lack of food and medical resources, traffic, pollution, crime, further effects of climate change, to mention just a few. None of these are seriously addressed in the zoning proposals.	The comment period was extended and more engagement opportunities were provided. Please see Section 4.2.8 . The Preferred Alternative integrates changes based on input from the South Park and Georgetown communities. See Chapter 2 of the Final EIS, as well as Sections 4.2.5 and 4.2.6 . The EIS addresses 14 environmental topics including traffic (Section 3.10), pollution (Section 3.2 Air Quality & GHG), Contamination (Section 3.5), Housing (Section 3.9) demand for police (Section 3.13), sea level rise (Section 3.4) and others. Mitigation measures are proposed. See also Section 4.2.7 .
114-2	Many other issues such as impact on cultural, historic, and archaeological resources and community character and quality are not adequately addressed.	See Section 3.11 Historic, Archaeological, & Cultural Resources. The City utilizes all applicable laws and ordinances with respect to impacts to cultural, historic, and archaeological resources. The SEPA process and/or cultural resources review, including architectural and archaeological survey, are completed prior to the start of many projects, and includes consultation with Tribes. Many federal, state, and local statutes and ordinance require notice and consultation with affected Tribes before, during, and after project review. The National Historic Preservation Act (NHPA) of 1966, was amended in 1986 with provisions for consultation with affected Tribes and 1992 to include and clarify the roles and responsibilities of Indian Tribes in Section 106 reviews. All cultural resources survey and archaeological work will follow best practices and standard archaeological techniques in the discovery and preservation of cultural and historical artifacts. The EIS scoping process and Draft EIS comment period included tribes. See also responses to letter 1 received from the Duwamish Tribe.
114-3	Arts and culture scene has grown due to affordable workspace and vision.	The comment is noted. Please also see Section 4.2.5 .
114-4	Expand the scope and vision of your efforts beyond just zoning to include plans and policies that encourage and support holistic growth for the whole community.	The comment is noted. The Preferred Alternative has been developed to respond to community needs and desires. Please see Chapter 2 of the Final EIS. Please also see proposed Comprehensive Plan policies in EIS Appendix C .
115	Knowles	Individual

Number	Comment Summary	Response
115-1	Supports the Georgetown Community Council's comment letter.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See responses to letter 96.
116	Krejci	Individual
116-1	We have before us an opportunity to do things differently, to address past and prevent future harm. Zoning dictates investment. Government's greatest role is that of convener and facilitator. Bold, innovative ideas are born in the differences of perspectives. I support the comments made by the Georgetown Community Council and the Duwamish River Cleanup Coalition.	Please see Chapter 2 of the Final EIS for a description of how the Preferred Alternative responds to community requests regarding zoning. Please also see Section 4.2.5 , 4.2.6 , and 4.2.8 .
		See also responses to letter 96.
116-2	Who works in our industrial areas? This requires a review of disaggregated data by race, gender,	Data about workers' home location is addressed in Section 3.9 Housing , see Exhibit 3.9-12 .
	age, and location to truly understand who works in the Duwamish MIC.	Additional information regarding worker race, gender, and age is
	Who benefits from ownership of industrial land? Who owns the land by race and gender?	Map information. Results show workers are primarily aged 30-54 (56.2%), earn more than \$3,333 (65%), two thirds white and one third persons of color (34.7%), and two thirds male and one third female (34.3%).
		Ownership of land by race and gender is not available.
116-3	Future expansion plans of the King County International Airport (KCIA) and the cumulative effect on the health of workers and residents in the Duwamish.	Future expansion plans of the KCIA are outside the scope of this proposal and would require their own review under the State Environmental Policy Act.
116-4	What is the current impact to industry of the current uses (not zoning) in the IG zones from Airport Way S to 1st Ave S? While the proposed industrial maritime zoning strategy recommendations are an improvement to the one-size-fits-all proposed in previous studies, they fail to provide a meaningful evaluation of Georgetown as an industrial neighborhood as a whole.	Existing land use conditions are described in the affected environment portion of the Land Use section. Effects on the Georgetown neighborhood are evaluated at the neighborhood scale to the extent that such analysis is practical. The Preferred Alternative includes unique development standards for new mixed use areas in Georgetown, and the City would continue to collaborate with community members on the content of those standards before adoption.
116-5	What is the future of industry? What does it look like—Amazon warehouses? Large-scale manufacturing? What are the wages of these jobs? Who benefits and who doesn't?	See the <u>Seattle Maritime and Industrial Strategy Updated</u> Employment Trends and Land Use Alternatives Analysis, December 22, 2020.
116-6	The Draft EIS makes zoning changes that need accompanying policy commitments in order to maximize their impact and enforce mitigation measures.	See Section 4.2.7 .
116-7	Allow for more engagement through the Comprehensive Plan and Seattle Transportation Plan.	See Sections 4.2.8 and 4.2.9.
117	Lanen	Individual
117-1	Concern about public engagement process.	Thank you for your letter. Comment is noted. See Section 4.2.8 concerning community engagement.
117-2	Concern that the alternatives are too limited or don't address Georgetown residents' needs.	Comment is noted. See Sections 4.2.5 , and 4.2.6 concerning Georgetown.

Number	Comment Summary	Response
117-3	Supports the GCC letter.	Comment is noted. See Sections 4.2.5 , and 4.2.6 concerning Georgetown and responses to letter 96.
118	Madison	Individual
118-1	Reject the EIS alternatives. Study impacts an arts and cultural resources. Alternatives would eliminate potential affordable housing. Increase engagement. Arts spaces are threatened. Privileges future growth of industrial and maritime uses over arts and cultural uses.	Thank you for your letter. The comment is noted. See Section 4.2.5 concerning arts and cultural communities in Georgetown.
119	Medina	Individual
119-1	Supports the Georgetown Community Council's comment letter.	Thank you for your letter. The comment is noted. See responses to letter 96.
119-2	Consider the flooding risks and consider the impact these plans could have to cause more environmental harm.	Section 3.3 Water Resources addresses potential for flooding risks and includes analysis of impacts in light of potential sea level rise.
119-3	Create possibilities for indigenous sovereignty and real environmental justice.	The comment is noted. Although the suggestion is beyond the scope of the EIS on the proposed action, equitable development measures targeted to supporting indigenous groups are discussed under Mitigation Measures in Section 3.8.3 Land & Shoreline Use .
120	Miller	Individual
120-1	Supports the Georgetown Community Council's comment letter.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See responses to letter 96.
121	Morrison	Individual
121-1	Concerned that proposed strategies do not take into account the economic and cultural value that the arts and artisans of Georgetown provide to Seattle.	Thank you for your letter. The EIS recognizes the lack of small or affordable space and housing for makers, creatives, and artists. Alternatives 3 and 4 addressed expanding allowances for limited industry-supportive housing such as caretakers' quarters and maker studios. Alternative 3 includes an estimated additional 610 limited industry supportive housing units in industrial zones, and Alternative 4 would have an estimated 2,195 units across the full study area. The housing would be available to business owners or employees of an on-site business that is an industrial use, or available to artists/makers with a business license in live-work spaces. Live/workspaces contain area for production/art/making activities that are physically connected to residential space. The Preferred Alternative specifically addresses this issue with the new mixed use zone in the triangle area of Georgetown by creating incentives for retention, restoration, and reuse of historic-period buildings and arts organizations and/or art studios. See also response to frequent comment theme concerning arts and culture in Georgetown.
121-2	Concern that the proposed alternatives would threaten arts space and affordable housing.	The comment is noted. See Section 4.2.5 concerning arts and culture in Georgetown and response to 121-1.
122	Neil	Individual
122-1	Concerns that the proposed alternatives do not include enough buffering between residential	Thank you for your letter. Comment is noted. See Section 4.2.6 about buffering and conversion of more MML zoned land.

Number	Comment Summary	Response
	areas ot Georgetown and neavier industrial areas.	
122-2	Concern that the proposal does not include commitments to mitigation and that there are not enough assurances that affordable housing will be provided or that historic resources will be retained.	The comment is noted. See Section 4.2.7 concerning mitigation measures, and Section 4.2.5 concerning arts and culture in Georgetown. See also response to 97-1.
122-3	Suggestion to shift the Industrial and Maritime Strategy process into the Comprehensive Plan major update.	The comment is noted. See Section 4.2.9 concerning coordination with the Comprehensive Plan major update.
123	Nyland, Kathy	Individual
123-1	Had similar comments in 2007. Draft EIS was issued in December 2021 during holidays, without people in mind.	Thank you for your letter. The comment period started in December 2021 and continued to the end of January 2022, and at that point the City extended the comment period to March 2, 2022. Further the City conducted extended engagements in the Georgetown and South Park communities until mid-April 2022. See Section 4.2.8 .
123-2	Georgetown is unique in a sea of IG zoning. Strategy promotes new economic opportunities but other companies, public and private, are no longer requiring degrees. Assumptions are outdated, and approach misguided.	The comment is noted and forwarded to City decision makers.
123-3	The focus of the entire EIS process was on economic impacts and opportunities. Urban Industrial (UI) zone was described as an innovative approach as a "safe and comfortable design". Question: why isn't safety and comfortable designed into ALL zones?	The EIS does not focus on economic impacts; see Section 4.2.1 . The EIS addresses 14 environmental elements addressing the natural and built environment. It identifies mitigation measures to address environmental impacts (e.g., air quality, noise, light and glare, open space/recreation, land use, housing, etc.). The action alternatives propose three new zones that are meant to improve the quality of development. The Preferred Alternative advances the conceptual code elements. See EIS Appendix G . The EIS mitigation measures can be applied across the zones, and the City can integrate them into policies and standards. See Section 4.2.7 and Appendix I .
123-4	What problems are being solved. One of the most pressing needs of Seattle is housing. Let's look at how industry AND mixed use AND residential can co-exist. The importance of livability should be applicable to everyone.	See response to comment 102-1.
123-5	Supports the Georgetown Community Council's comment letter.	Comment is noted. See responses to letter 96.
124	Nyland, Kelsey	Individual
124-1	Supports the Georgetown Community Council's comment letter.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See responses to letter 96.
125	Rajcich	Individual
125-1	Supports the Georgetown Community Council's comment letter and requests process be folded into the Comprehensive Plan update.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See responses to letter 96.
126	Rivera	Individual

Number	Comment Summary	Response
126-1	Supports the Georgetown Community Council's comment letter.	Thank you for your letter. The comment is noted. See responses to letter 96.
127	Ryan	Individual
127-1	Do not make zoning changes for Georgetown and South Park areas based on the industrial and maritime strategy process, and instead address the areas through the Comprehensive Plan update process.	Thank you for your letter. See Section 4.2.9 regarding the Strategy review and the Comprehensive Plan.
127-2	Concern that the alternatives studied threaten affordable arts and performance spaces.	Comment noted. See Section 4.2.5 concerning arts / cultural resources in Georgetown.
127-3	Concern that the alternatives would reduce or eliminate potential affordable housing.	The comments are noted. Action Alternatives expand housing allowances in currently industrially-zoned areas compared to the No Action Alternative. In Georgetown and South Park several areas are removed from industrially zoning and placed into a mixed use zone that would allow dense housing development in alternatives 3, 4 and the Preferred Alternative. The EIS discusses options for requiring that a portion of the housing be dedicated affordable housing. Section 3.9 Housing discusses impacts of alternatives on housing and displacement.
127-4	Concern that the Industrial and Maritime Strategy process did not adequately include engagement of Georgetown and South Park residential community members.	Comment is noted. See Section 4.2.8 concerning community engagement.
127-5	Concern that the proposed action prioritizes industrial and maritime uses over creative and cultural businesses.	The comment is noted. Multiple alternatives would change zoning in a portion of Georgetown from an industrial zone to a non- industrial zone. Additionally, the proposed UI designation would be intended to support small businesses, makers, and arts. Some aspects of the proposal intentionally support future viability of industrial and maritime uses in the regionally-designated MICs.
127-6	Study impacts on arts and creative communities.	The comment is noted. See Section 4.2.5 concerning arts / cultural resources in Georgetown.
127-7	Increase zoned buffer areas and decrease the amount of MML zoning.	The comment is noted. See Section 4.2.6 concerning buffer areas and reduction of MML zones in Georgetown.
127-8	Prioritize new affordable housing options.	The comment is noted. See response to 127-3.
127-9	Update zoning to reflect existing mixed uses and decrease the amount of MML zoning.	The comment is noted. See Section 4.2.6 concerning buffer areas and reduction of MML zones in Georgetown. Section 3.8 Land & Shoreline Use includes an analysis of existing land use.
127-10	Study expansion of buffer zoning such as more Commercial zoning and more mixed use zoned areas.	The comment is noted. See Section 4.2.6 concerning buffer areas and reduction of MML zones in Georgetown.
127-11	Enact changes that allow for more housing and more investment in maker and studio spaces.	The comment is noted. See response to comments 127-2 and 127-3.
127-12	Accompany the Final EIS with legislation committing the City to fund mitigation measures.	The comment is noted. See Section 4.2.7 concerning mitigation measures.
127-13	Add policy commitments to historic preservation and affordable housing for Georgetown and South Park.	The comment is noted. See response to frequent comments concerning mitigation measures.

Number	Comment Summary	Response
127-14	Conduct more community engagement.	The comment is noted. See Section 4.2.8 concerning community engagement.
127-15	Extend the EIS process for a year.	The comment is noted. See Section 4.2.8 concerning community engagement.
127-16	Shift the process into the Comprehensive Plan major update.	The comment is noted. See Section 4.2.9 concerning coordination with the Comprehensive Plan major update.
128	Schiffer	Individual
128-1	Supports the Georgetown Community Council's comment letter.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See responses to letter 96.
129	Smith	Individual
129-1	Supports the Georgetown Community Council's comment letter. Need more residential and commercial development and insulation from Industrial.	Thank you for your letter. The comment is noted. See responses to letter 96.
130	St John	Individual
130-1	Shared personal experience being impacted by noise and dangerous roads.	Comments noted. Thank you for sharing. The EIS reviews impacts of the proposed alternatives on numerous elements of the environment including noise and roadway safety.
130-2	Concern that the alternatives do not include enough conversion to UI zones or other mixed use zones that allow residential.	Comment noted. See Section 4.2.6 concerning buffer areas and conversion of more MML zoned land in and around Georgetown. See also Section 4.2.5 regarding Mixed Use in Georgetown.
130-3	Supports conversion of the Georgetown triangle area to a mixed use zone. The railroad spur in it should be removed.	Comments noted. See Section 4.2.5 regarding Mixed Use in Georgetown.
131	Sweet	Individual
131-1	Many artists located in Georgetown because they were displaced from other areas.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
131-2	Impacts on arts and cultural resources should be studied more. Mitigation measures should be described in more detail.	The comment is noted. See Section 4.2.5 concerning arts / cultural resources in Georgetown.
131-3	Proposals favor growth of industrial and maritime uses over existing creative industries.	The comment is noted. Multiple alternatives would change zoning in a portion of Georgetown from an industrial zone to a non- industrial zone. Additionally, the proposed UI designation would be intended to support small businesses, makers, and arts. Some aspects of the proposal intentionally support future viability of industrial and maritime uses in the regionally-designated MICs.
131-4	Existing communities, including artists, their workspaces and businesses, and the cultural life of Seattle, are threatened.	Comments noted. See Section 4.2.5 concerning arts / cultural resources in Georgetown.
131-5	Concerned that the EIS discloses that historical and cultural resources could be damaged or altered under any alternative.	The comment is noted. Alternatives include a No Action Alternative. See Section 4.2.5 concerning arts / cultural resources in Georgetown.
131-6	The environmental impact analysis is narrow and does not fully address core principles related to environmental justice and a fair community-driven process	The comment is noted. See Section 4.2.8 regarding community engagement, and Section 1.7.15 regarding equity and environmental justice.

Number	Comment Summary	Response
131-7	Concerns about process.	The comment is noted. A purpose of the EIS is to disclose potential impacts before any decisions are made.
131-8	Paraphrases text from the Draft EIS.	The comment is noted.
132	Terrenzio	Individual
132-1	Reject all the alternatives. Georgetown has a valuable and growing arts community and a need for affordable housing.	Thank you for your letter. The comment is noted. See Sections 4.2.5 and 4.2.10 .
133	Tilley	Individual
133-1	Supports the Georgetown Community Council's comment letter and requests process be folded into the Comprehensive Plan update.	Thank you for your letter. The comment is noted and forwarded to City decision makers. See responses to letter 96.
134	Veloria	Individual
134-1	Supports Coalitions letter.	Thank you for your letter. The comment is noted. See responses to letter 97.
135	White	Individual
135-1	Supports the Georgetown Community Council's comment letter.	Thank you for your letter. The comment is noted. See responses to letter 96.
136	Woo	Individual
136-1	Appreciate the opportunity to comment. Family built and operates the Georgetown Inn.	Thank you for your letter. The comment is noted and forwarded to City decision makers.
136-2	Request to extend the Mixed Use zone to include the Georgetown Inn (area between Harney St, Corson Ave, and Baily St). Would like an explanation of development standards for mixed use within the land use concept comparisons. Rezoning parts of Georgetown to mixed use offers many potential benefits but requires accompanying policies to ensure adequate historic preservation, affordability, and sustainability.	Alternatives 3 and 4 and the Preferred Alternative remove the triangular area of Georgetown bounded by Corson Avenue S, Carleton Avenue S and I-5 from the MIC and place it into a mixed-use zone. The area would likely develop with a high concentration of urban mixed-use structures with ground level retail and residential above, and by the end of the study time horizon the area would likely transition to mixed-use area similar to an urban village. Please see Section 4.2.5 regarding an enlarged Mixed Use area in Georgetown. Additional detail regarding development standards to address the unique conditions in the proposed mixed use zoning in Georgetown are included under the Preferred Alternative, in the development standards Appendix G . This includes features to incentivize the retention and restoration of historic character structures and arts organization and/or arts studios. The new Mixed Use zone in the triangle area of Georgetown would be Neighborhood Commercial with a 55 foot height limit (NC3-55) and a Mandatory Housing Affordability (M1) suffix would be applied to the zone.
137	Wright	Individual
137-1	Supports Coalitions letter.	Thank you for your letter. The comment is noted. See responses to letter 97.

4.3.2 Public Hearing Verbal Comments & Responses

Number	Comment Summary	Response
H1	Curtis, Josh	Washington State Ballpark Public Facilities District
H1-1	Appreciate the opportunity to provide comment and request an extension.	Thank you for your comment. The comment period was extended and more engagement opportunities were provided. Please see Section 4.2.8 .
H1-2	The stadium district is unique. Almost no industry is left in the area, but it is largely zoned industrial commercial. Most new development are offices because of the high price of land— the analysis should consider the transportation impact of a full office build out around the stadiums under the No Action Alternative.	The land use of each alternative is compared in the EIS and included in transportation modeling. Results on transportation networks inside and outside the STAOD are addressed at a non- project level of detail.
H2	Marchione, John	Washington State Public Stadium Authority
H2-1	EIS doesn't make clear how additional housing around the stadiums is out of character with what already exists here. Would like the Final EIS to separate the analysis of the stadium district and analyze the impacts to land use, transportation, and housing in particular.	The Preferred Alternative includes expanded flexibilities to address unique conditions of the stadium area through the STAOD. More information on these flexibilities is provided in the development standards Appendix G . The STAOD is part of the evaluation of the MIC in transportation,
		detail that is areawide, consistent with WAC 197-11-442.
		Alternatives' effects on transportation corridors in and near the STAOD are included; and the area is referenced in the land use evaluation and included on maps. The STAOD boundaries are added to the Preferred Alternative map to assist in viewing that portion of the study area.
H2-2	Concerned about antiquated restrictions on housing for land zoned industrial.	Consistent with the PSRC criteria for designating MICs to focus industrial uses in the MIC, the EIS does not study allowing residential uses in the majority of the study area. Alternatives 3 and 4 and the Preferred Alternative consider limited additional flexibility of existing allowances for caretakers' units and artist/studio quarters in the proposed UI zone only.
H3	Scott	Individual
H3-1	Did the City consider how other cities are addressing industrial lands (such as Tacoma, Vancouver, or Baltimore)?	The City reviewed other peer cities' initiatives related to industrial lands as part of background research and analysis for the proposed action.
H4	Williams Jr., Dennis	Individual
H4-1	Supports Alternative 4 and an increase in the maximum size of use for indoor sports and recreation uses.	Thank you for your comment. The comment is noted and forwarded to City decision makers. Alternative 4 and the Preferred Alternative incorporate an increase in the maximum size of use for indoor sports and recreation uses.
H5	Loe, Laura	Share The Cities Action Fund
H5-1	Organization and comments are focused on Ballard and Interbay industrial lands.	Comment is noted.

Exhibit 4.3-2 Public Hearing Verbal Comments and Responses

Number	Comment Summary	Response
H5-2	Supports Alternative 4, especially strengthened protections on industrials lands and the flexibility for industry supportive housing. Would like to see protections to ensure the housing is used as caretakers' quarters.	Comment is noted and forwarded to City decision makers.
H5-3	EIS should include more historical context of how redlining has aligned with Seattle's industrial lands and how growth patterns are rooted in past racial injustice.	See response to comment 59-7.
H5-4	Analysis should include more industrial areas within Seattle, such as at Madison or near light rail in North Seattle.	The community will have additional opportunities to provide input on the City's overall growth strategy as part of the Comprehensive Plan major update. The City considers the Industrial and Maritime Strategy to be a distinct subject area worthy of a topic-specific study and land use policy proposals because there are unique attributes and issues related to industrial lands and designated Manufacturing and Industrial Centers. See also Section 4.2.9 .
H5-5	Concerned about where the future vehicle traffic estimates come from.	Fehr & Peers applied a version of the PSRC regional trip-based travel demand model developed for the WSBLE project and the Ballard-Interbay Regional Transportation (BIRT) System project to develop the future forecasts for this project. The model estimates the demand for person and freight travel across a range of travel modes: private automobiles, trucks, transit vehicles, walking, and biking. The truck model defines a truck based on relative weight classes and separates medium and heavy trucks based on the definitions used by WSDOT for collecting truck counts.
		This version of the PSRC model is an appropriate tool for this project given its level of detail in the study area (in terms of both land uses and transportation network), assumptions for transit investments, and future land use assumptions that are consistent with growth anticipated through 2042. The model contains household and employment forecasts consistent with regional assumptions from PSRC and the City's MHA growth distributions.
		See also the analysis methodology in Section 5.10.2.
H5-6	Want to see more analysis by zone.	Chapter 2 of the EIS describes the alternatives, including the overall intent and themes for each. A Preferred Alternative is added in the Final EIS. All Action Alternatives are different variations of application of the UI, II, and MML zones. General locational criteria and intent is described for each of the three proposed new zones in Chapter 2 .
		Appendix C includes detailed maps depicting alternate zone changes with specific boundaries. A story map is also provided by the City which allows detailed review to a parcel-specific level. See the <u>storymap link here</u> .
H5-7	Concerned about the jobs to housing balance	The comment is noted and forwarded to City decision makers.
	and housing crunch for middle wage workers (which impacts middle and low wage workers and can lead to homelessness issues). Want to make sure the middle wage jobs are on the higher end of middle wage.	The City conducted research and analysis to project the type of jobs expected, including review of typical wages in different jobs. The analysis was conducted in parallel with the EIS and growth estimations in the proposal are based on the prior analysis.
H5-8	Happy with the Draft EIS and excited to see what folks in other parts of Seattle have to say.	Thank you for your comment. Comment is noted.

Number	Comment Summary	Response
H5-9	Make sure the future land uses and roads etc. are friendly for pedestrians.	SDOT is currently in the process of developing the Seattle Transportation Plan which will integrate the City's modal plans into a comprehensive vision for the citywide transportation network centered around the following values and goals: equity, safety, mobility, sustainability, livability, and excellence.
H6	Scott	Individual
H6-1	Strongly support Alternative 4.	Comment is noted and forwarded to City decision makers.
H6-2	Should be greater focus on general public benefit, such as better public access, ground level landscaping/green space, and sustainability (site and building features). Could offer height or density bonuses in exchange.	The UI zone would have higher standards for landscaping and tree planting with new development than the zone it would replace under the alternatives. The City's Parks, Recreation and Open Space Plan outlines the City's existing open space and recreational facilities, capital funding, and projects being funded and a vision for the future.
H6-3	Flexibility and affordability for artists and live/work opportunities, especially around light rail stations. New buildings on Salmon Bay have luxury caretaker units which doesn't seem equitable.	Comment is noted and forwarded to City decision makers. Alternatives 3 and 4 and the Preferred Alternative consider limited additional flexibility of existing allowances for caretakers' units and artist/studio quarters in the proposed UI zone only. The II zone, applied in alternatives 3 and 4 and the Preferred Alternative, would allow mixed use with industrial, technology, and office in proximity to light rail. See also Section 4.2.10 .
H6-4	Stadium area should be an urban entertainment and arts district. Heavy traffic during games has a major impact on industrial.	The Preferred Alternative includes expanded flexibilities to address unique conditions of the stadium area through the STAOD. More information on these flexibilities is provided in the development standards Appendix G . The STAOD is part of the transportation evaluation (see Section 3.10 Transportation), including the effects of each alternative on transportation corridors in and near the STAOD.
H6-5	Little to no investment to date near the stadiums or SODO light rail stations. Encourage affordable housing and small business opportunities near light rail stations (like in alternatives 3 and 4). Zone near light rail stations should have minimum residential height of 65 or 85 feet.	The II zone, applied in alternatives 3 and 4 and the Preferred Alternative, would allow for a significant amount of non-industrial uses—including technology and office—through a development bonus system. The II zone would be applied under multiple alternatives to the area around the SODO/Lander St. station in the SODO area.
H6-6	Important to consider the areas of underutilized or vacant industrial land.	Comment is noted and forward to City decision makers. Section 3.8 Land & Shoreline Use includes an analysis of existing land use, including narrative descriptions of subarea land use patterns under existing conditions.

4.4 Marked Comment Letters & Public Hearing Transcripts

The marked letters, online survey forms, and public hearing transcripts are available on the City's project webpage: <u>Industrial and Maritime Strategy—OPCD | seattle.gov</u>.



5.1 Acronyms

ADA	Americans with Disabilities Act
ALS	Advance Life Support
ARAR	Applicable or Relevant and Appropriate Requirements
ARPA	Archaeological Resources Protection Act
BINMIC	Ballard Interbay Northend MIC
BIRT	Ballard-Interbay Regional Transportation
BLS	Basic Life Support
BMP	Bicycle Master Plan
BMPs	Best Management Practices
BNSF	Burlington Northern Santa Fe
BPSA	Bicycle and Pedestrian Safety Analysis
BSOs	Buildings, Structures, or Objects
CAPCOA	California Air Pollution Control Officers Association
CARA	Critical Aquifer Recharge Area
CIP	Capital Improvement Program
CIP	Capital Improvement Program
CPPs	King Countywide Planning Policies
CPSC	Community Partners Steering Committee
CRPP	Cultural Resource Protection Plan
CSO	Combined Sewer Overflow
CTR	Commute Trip Reduction
CWA	Clean Water Act
DAHP	Washington State Department of Archaeology and Historic Preservation
dB	Decibel
dBA	A-weighted Sound Level
DNRP	Department of Natural Resources and Parks
DSL	Digital Subscriber Line
EA	Environmental Assessment
EEI	Equity and Environment Initiative
EHD	Environmental Health Disparities
EIS	Environmental Impact Statement
EMS	Emergency Medical Services
EPA	U.S. Environmental Protection Agency
fhwa	Federal Highway Administration
FMP	Freight Master Plan
FTA	Federal Transit Administration
GHG	Greenhouse Gas
GMA	Growth Management Act
GMPC	King County Growth Management Planning Council
HBMS	Hazardous Building Material Survey

HBMS	Hazardous Building Material Surveys
HCM	Highway Capacity Manual
HPI	Historic Property Inventory
HPP	King County Historic Preservation Program
HUD	U.S. Department of Housing and Urban Development
IDDE	Illicit discharge detection and elimination
	Industry and Innovation
ITS	Intelligent Transportation Systems
KCSWDM	King County Surface Water Design Manual
Ldn	Day-Night Average Sound Level
LEED	Leadership in Energy and Environmental Design
Leq	Equivalent Noise Level
Lmax	Maximum Noise Level
LOS	Level of Service
LTCP	Long-term Control Plan
MCPP	Micro-Community Policing Plans
mgd	Million Gallons per Day
MIC	Manufacturing/Industrial Center
MMDF	Maximum Month Design Flow
MML	Maritime, Manufacturing, and Logistics
MPD	Multiple Property Documentation
MPH	Miles per Hour
MSATs	Mobile Source Air Toxics
MTCA	Model Toxics Control Act
MW NHA	Maritime Washington National Heritage Area
NAAQS	National Ambient Air Quality Standards
NDS	Natural Drainage Systems
NEC	National Electric Code
NEPA	National Environmental Protection Act
NHL	National Historic Landmarks
NHPA	National Historic Preservation Act
NPDES	National Pollutant Discharge Elimination System
NRHP	National Register of Historic Places
NTHP	National Trust for Historic Preservation
NWI	National Wetlands Inventory
OPCD	Seattle Office of Planning and Community Development
OSE	Seattle Office of Sustainability and Environment
PMP	Pedestrian Master Plan
POSPD	Port of Seattle Police Department
PSCAA	Puget Sound Clean Air Agency
PSRC	Puget Sound Regional Council
RCO	Recreation Conservation Office
RCW	Revised Code of Washington

RGC	Regional Growth Center
RMP	Risk Management Plan
RPZ	Residential Parking Zone
SCL	Seattle City Light
SCWQP	Ship Canal Water Quality Project
Sdot	Seattle Department of Transportation
SFD	Seattle Fire Department
SLS	Seattle Library System
SMC	Seattle Municipal Code
SMP	Shoreline Master Program
SOV	Single Occupancy Vehicle
SPD	Seattle Police Department
SPR	Seattle Parks and Recreation
SPS	Seattle Public Schools
SPU	Seattle Public Utilities
SR	State Route
SWMP	Stormwater Management Program
TDM	Travel Demand Management
TMA	Transportation Management Association
TMDL	Total Maximum Daily Load
TMP	Transit Master Plan
TMP	Transportation Management Program
TSA	Transportation Security Administration
TSMO	Transportation Systems Management and Operations
UI	Urban Industrial
USGS	U.S. Geological Survey
USSG	U.S. Surveyor General
V/C	Volume to Capacity
VMT	Vehicle Miles Traveled
VOC	Volatile Organic Compounds
WAC	Washington Administrative Code
WHBR	Washington Heritage Barn Register
WISAAR	D Washington Information System for Architectural and Archaeological Records Data
WOTUS	Waters of the United States
WQ	Water Quality
WRIA	Water Resource Inventory Area
WSBLE	West Seattle and Ballard Link Extension
WSDOT	Washington State Department of Transportation
WTD	Wastewater Treatment Division
WTHP	Washington Trust for Historic Preservation
WWTP	Wastewater Treatment Plant

5.2 References

Photos sourced from the City of Seattle, Port of Seattle, Northwest Seaport Alliance, and BERK.

General

CAI. 2020. "Seattle Industrial Lands Technical Analysis Memo." Originally published in 2017, updated in 2020.

https://www.seattle.gov/Documents/Departments/OPCD/OngoingInitiatives/IndustrialM aritimeStrategy/SeattleIndustrialLandsEmploymentAnalysisMemo2020.pdf.

City of Seattle. 2016. Final Environmental Impact Statement for the Seattle Comprehensive Plan. May 5, 2016. Available:

http://www.seattle.gov/Documents/Departments/OPCD/OngoingInitiatives/SeattlesCom prehensivePlan/SeattleCPFEIS2016_0505.pdf.

- City of Seattle. 2017. *Racial and Social Equity Index.* <u>https://seattlecitygis.maps.arcgis.com/apps/Minimalist/index.html?appid=764b5d89885</u> <u>74644b61e644e9fbe30d1</u>. Posted <u>Seattle GeoData</u>: January 15, 2020.
- City of Seattle. 2018. "Duwamish Valley Action Plan: Advancing Environmental Justice & Equitable Development in Seattle." <u>https://www.seattle.gov/environment/equity-and-environment/duwamish-valley-program</u>.
- City of Seattle. 2020. "Seattle 2035." <u>http://www.seattle.gov/opcd/ongoing-initiatives/comprehensive-plan#projectdocuments</u>.
- City of Seattle. 2021. "Industrial and Maritime Strategy Council Recommendations." <u>https://www.seattle.gov/opcd/ongoing-initiatives/industrial-and-maritime-</u> <u>strategy#projectdocuments</u>.
- City of Seattle. 2021. *Seattle Municipal Code.* <u>https://library.municode.com/wa/seattle/codes/municipal_code</u>.
- PSRC. 2018. "Regional Centers Framework." <u>https://www.psrc.org/sites/default/files/final_regional_centers_framework_march_22_ver</u> <u>sion.pdf</u>.
- PSRC. 2020. "VISION 2050: A Plan for the Central Puget Sound Region." Adopted via Resolution PSRC-A-2020-02. <u>https://www.psrc.org/vision</u>
- Washington State Legislature. 2021. Revised Code of Washington. https://apps.leg.wa.gov/rcw/.

Washington State Legislature. 2021. *Washington Administrative Code*. <u>https://apps.leg.wa.gov/wac/</u>.

5-5

Soils/Geology

- City of Seattle. 2021. Environmentally Critical Areas mapping viewed July 9 at Seattle Department of Construction & Inspections GIS website: <<u>Seattle Department of</u> <u>Construction & Inspections GIS (arcgis.com)</u>>
- Galster R. W. and Laprade, W. T. 1991. Geology of Seattle, Washington United States of America. Bulletin of the Association of Engineering Geologists Vol. XXVII, No. 3.
- Rogers, G.C. 1988. An Assessment of the Megathrust Earthquake Potential of the Cascadia Subduction Zone. Canadian Journal of Earth Sciences, 25(6), pp. 844–852.
- Seattle-King County Department of Public Health. 1985. Abandoned landfill study in King County: Report prepared by Seattle-King County Department of Public Health, April 30.
- Troost, K.G., D.B. Booth, A.P. Wisher, and S.A. Shimel. 2005. The Geologic Map of Seattle— A Progress Report.
- Troost, Kathy G., Derek B. Booth, and William T. Laprade. 2003. Quaternary geology of Seattle. Field Guide 4. Geological Society of America, Boulder.
- Troost, Kathy G., and Derek B. Booth. 2008. Geology of Seattle and the Seattle area, Washington. In Landslides and Engineering Geology of the Seattle, Washington, Area, edited by Rex L. Baum, Jonathan W. Godt, and Lynn M. Highland, pp. 1-35. Reviews in Engineering Geology XX. The Geological Society of America, Boulder.
- USGS. 2014. National Seismic Hazard Model. US Geological Survey.
- USGS. 2020. M 9.3 Scenario Earthquake—Cascadia Megathrust Shake Map. Accessed May 4, 2020 at US Geological Society website: <<u>https://earthquake.usgs.gov/scenarios/eventpage/bssc2014cascadia_sub0_m9p34_se/shakemap_/intensity</u>>.
- WDNR. 2002. Geologic Map of Washington—Northwest Quadrangle. Washington Division of Geology and Earth Resources Geologic Map GM-50. Published by the Washington State Department of Natural Resources.
- WGCEP (Working Group on California Earthquake Probabilities). 2003. Earthquake Probabilities in the San Francisco Bay Region: 2002–2031. US Geological Survey Open-File Report 03-214.

Air Quality & GHG

CDC. 2019. National Asthma Data.

https://www.cdc.gov/asthma/most_recent_national_asthma_data.htm.

City of Seattle. 2020. "2018 Community Greenhouse Gas Emissions Inventory."

https://www.seattle.gov/Documents/Departments/OSE/ClimateDocs/2018_GHG_Invento ry_Dec2020.pdf.

- Ecology. 2021. "Washington State Greenhouse Gas Emissions Inventory 1990–2018." https://apps.ecology.wa.gov/publications/SummaryPages/2002020.html.
- PSCAA and Ecology. 2021. Air Monitoring Network. https://enviwa.ecology.wa.gov/home/map.
- PSCAA. 2019. "2019 Air Quality Data Summary." <u>https://pscleanair.gov/DocumentCenter/View/4164/Air-Quality-Data-Summary-2019</u>.
- PSMEI. 2018. "2016 Puget Sound Maritime Emissions Inventory", Puget Sound Maritime Air Forum, prepared by Starcrest Consulting Group, LLC, Poulsbo, WA. Revised October 2018.
- US Department of Transportation. 2015. Proximity To Major Roadways. August 24, 2015. Available: <u>https://www.transportation.gov/mission/health/proximity-major-roadways</u>.
- U.S. EPA. 2011. "Greenbook."
- U.S. EPA. 2021. "Climate Change and Social Vulnerability in the United States: A Focus on Six Impact Sectors." <u>https://www.epa.gov/cira/social-vulnerability-report</u>.
- WDOH. 2008. "Duwamish Valley Regional Modeling and Health Risk Assessment." https://www.doh.wa.gov/portals/1/Documents/Pubs/334-165.pdf.

Water Resources

CH2M. 2017. "Ship Canal Water Quality Project: Final Facility Plan." Prepared for Seattle Public Utilities, March 2017.

http://www.seattle.gov/util/cs/groups/public/@spu/@drainsew/documents/webcontent/ <u>1 065575.pdf</u>.

- City of Seattle. 2007. "Seattle Creeks State of the Waters Report." <u>https://www.seattle.gov/util/cs/groups/public/@spu/@conservation/documents/webcon</u> <u>tent/spu01_003413.pdf</u>.
- City of Seattle. 2017. "Preparing for Climate Change." August 2017. <u>https://www.seattle.gov/documents/Departments/Environment/ClimateChange/SEAClim</u> <u>atePreparedness_August2017.pdf</u>.
- Ecology. 2014. *Ecology Water Quality Assessment Database*. <u>https://apps.ecology.wa.gov/ApprovedWQA/ApprovedPages/ApprovedSearch.aspx</u>.
- Hart Crowser, Inc. 1998. "Duwamish Industrial Area Hydrogeologic Pathways Project: Duwamish Basin Groundwater Pathways Conceptual Model Report." Prepared for City of Seattle Office of Economic Development and King Country Office of Budget and Strategic Planning, April 1998. <u>http://cleanupsites.org/t108/documents/11.%20LDW-</u> <u>Related/Duwamish%20Industrial%20Area%20Hydrogeologic%20Pathways%20Project-%20Duw.pdf</u>.
- KPFF Consulting Engineers Shannon and Wilson, Inc. 2005. "Draft EIS Magnolia Bridge Replacement."

- Miller, I.M., Morgan, H., Mauger, G., Newton, T., Weldon, R., Schmidt, D., Welch, M., Grossman, E. 2018. "Projected Sea Level Rise for Washington State—A 2018 Assessment." A collaboration of Washington Sea Grant, University of Washington Climate Impacts Group, Oregon State University, University of Washington, and US Geological Survey. Prepared for the Washington Coastal Resilience Project, updated July 2019. https://cig.uw.edu/resources/special-reports/sea-level-rise-in-washington-state-a-2018-assessment/.
- USGS. 2019. "Assessment of Existing Groundwater Quality Data in the Green-Duwamish Watershed, Washington Report 2019-1131." <u>https://pubs.er.usgs.gov/publication/ofr20191131</u>.

WR Consulting. 2008. "Wolfe Creek Drainage Feasibility Study Final Report."

Plants & Animals

- City of Seattle. 2015. "Seattle Biological Evaluation." May 2015 Revision. <u>https://www.seattle.gov/utilities/construction-resources/design-standards/seattle-biological-evaluation/sbe-document.</u>
- City of Seattle. 2021. *Seattle GeoData*. Accessed October 28, 2021. <u>https://data-seattlecitygis.opendata.arcgis.com/search?tags=environment.</u>
- City of Seattle. 2010. "Shoreline Characterization Report." January 2010. <u>https://www.seattle.gov/Documents/Departments/SDCI/Vault/ShorelineMasterProgram/</u> <u>ShorelineCharacterizationReport.pdf.</u>
- DNR. 2021. Washington Department of Natural Resource GIS Open Data: Puget Sound Seagrass Monitoring. <u>https://data-wadnr.opendata.arcgis.com/apps/puget-sound-seagrass-</u> monitoring-web-application/explore.
- Ecology. 2021. *Draft Water Quality Atlas.* Accessed October 28, 2021. https://apps.ecology.wa.gov/waterqualityatlas/wqa/map.
- Encyclopedia of Puget Sound. 2020. "Types of Estuaries in Puget Sound." November 2020. https://www.eopugetsound.org/articles/types-estuaries-puget-sound.
- Hobbie, J. E. 2000. "Estuarine science: a synthetic approach to research and practice." Island Press, Washington D.C. 539 p.
- NMFS. 2020. "Endangered Species Act Section 7 Formal Programmatic Biological Opinion and Magnuson-Stevens Fishery Conservation and Management Act Essential Fish Habitat Consultation for the U.S. Department of Housing and Urban Development Housing Programs in Western Washington." Reference number WCRO-2020-00512. September 9, 2020.
- NMFS. 2021. ESA Critical Habitat Mapper. Accessed 10/29/2021. https://www.webapps.nwfsc.noaa.gov/portal7/apps/webappviewer/index.html?id=5c178 cec58504ca1a72c048eb552b029.

NWI. 2021. *National Wetland Inventory wetlands mapper*. Accessed July 2021. <u>https://www.fws.gov/wetlands/data/Mapper.html.</u>

WDFW. 2021. Priority Habitats and Species: PHS on the Web. Accessed October 27, 2021. https://geodataservices.wdfw.wa.gov/hp/phs/.

Contamination

- ASTM. 2013. Standard Practice for Environmental Site Assessments: Phase I Environmental Site Assessment Process. ASTM International Designation E1527-13. West Conshohocken, Pennsylvania.
- City of Seattle. 2021. Environmentally Critical Areas mapping viewed July 9 at Seattle Department of Construction & Inspections GIS website: <Seattle Department of Construction & Inspections GIS (arcgis.com)>
- Ecology. 2021. Washington State Department of Ecology Facilities/Sites of Environmental Interest geodatabase. Downloaded July 14, 2021, from <u>https://fortress.wa.gov/ecy/gispublic/DataDownload/ECY_ENV_FacilitiesSites_FacilitySitel_nteractions.htm</u>.
- Seattle-King County Department of Public Health. 1984. Abandoned Landfill Study in the City of Seattle. July 30.

Noise

- FHWA. 2006. "Highway Construction Noise Handbook." <u>https://www.fhwa.dot.gov/environment/noise/construction_noise/handbook/</u>.
- FHWA. 2021. "23 CFR Part 772, Procedures for Abatement of Highway Traffic Noise and Construction Noise." <u>https://www.ecfr.gov/current/title-23/chapter-l/subchapter-H/part-772</u>.
- FHWA. 2021. "23 CFR Part 772, Procedures for Abatement of Highway Traffic Noise and Construction Noise." <u>https://www.ecfr.gov/current/title-23/chapter-I/subchapter-H/part-772</u>.
- WSDOT. 2020. "Traffic Noise Policy and Procedures."
- U.S. Department of Housing and Urban Development. 2021. "24 CFR Part 51, Subpart B, Noise Abatement and Control. <u>https://www.ecfr.gov/current/title-24/subtitle-A/part-51/subpart-B?toc=1</u>.

Light & Glare

NASA. 2015. Orbital Imagery of Nighttime Lights. https://earthdata.nasa.gov/learn/backgrounders/nighttime-lights#data.

Land & Shoreline Use

CAI. 2021. "Estimated Industrial and Total Employment by Subarea and Alternative (2044)."

- City of Seattle. 1998. "Ballard Interbay Northend Manufacturing and Industrial Center." <u>http://www.seattle.gov/Documents/Departments/Neighborhoods/Planning/Plan/Binmic-plan.pdf</u>.
- City of Seattle. 2000. "Greater Duwamish Manufacturing and Industrial Center Plan." <u>http://www.seattle.gov/documents/Departments/Neighborhoods/Planning/Plan/Duwam</u> <u>ish-plan.pdf</u>.
- Honig, Doug. 2021. "Redlining in Seattle." HistoryLink Essay No. 21296. https://www.historylink.org/File/21296.
- King County. 2021. "Countywide Planning Policies." <u>https://kingcounty.gov/depts/executive/performance-strategy-budget/regional-planning/CPPs.aspx</u>.
- The Burke Museum. 2009. *The Waterlines Project.* <u>https://www.burkemuseum.org/static/waterlines/index.html</u>.
- University of Washington. 2020. *Seattle Civil Rights & Labor History Project.* <u>http://depts.washington.edu/civilr/index.htm</u>.

Housing

- City of Seattle. 2019. *Mandatory Housing Affordability (MHA).* <u>https://www.seattle.gov/hala/about/mandatory-housing-affordability-(mha).</u>
- City of Seattle. 2021. "Market Rate Housing Needs and Supply Analysis." <u>https://www.seattle.gov/Documents/Departments/OPCD/OngoingInitiatives/HousingCh</u> <u>oices/SeattleMarketRateHousingNeedsAndSupplyAnalysis2021.pdf</u>.
- U.S. Census Bureau. 2018. Longitudinal Employer-Household Dynamics (LEHD) Origin-Destination Employment Data.
- Washington Department of Health. 2021. *Washington Tracking Network*. <u>https://fortress.wa.gov/doh/wtn/WTNPortal/</u>.

Open Space & Recreation

City of Seattle. 2017. Racial and Social Equity Index.

https://seattlecitygis.maps.arcgis.com/apps/Minimalist/index.html?appid=764b5d89885 74644b61e644e9fbe30d1.

City of Seattle. 2018. "Duwamish Valley Action Plan: Advancing Environmental Justice & Equitable Development in Seattle." <u>https://www.seattle.gov/environment/equity-and-environment/duwamish-valley-program</u>.

King County and City of Seattle. 2021. *Results of heat mapping project show inequitable impact of hotter summers, will inform actions by King County and City of Seattle.* June 23. Accessed July 21, 2021.

https://kingcounty.gov/elected/executive/constantine/news/release/2021/June/23-heatmapping-results.aspx.

Seattle Parks and Recreation. 2016. "Community Center Strategic Plan."

https://www.seattle.gov/parks/about-us/policies-and-plans/community-center-strategicplan.

Seattle Parks and Recreation. 2016. "Recreation Demand Study."

https://www.seattle.gov/Documents/Departments/ParksAndRecreation/PoliciesPlanning /SeattleRecreationDemandStudy2016.pdf.

Seattle Parks and Recreation. 2017. "2017 Parks and Open Space Plan."

http://www.seattle.gov/Documents/Departments/ParksAndRecreation/PoliciesPlanning/ 2017Plan/2017ParksandOpenSpacePlanFinal.pdf.

Seattle Parks and Recreation. 2020. "A Strategic Plan for Seattle Parks and Recreation 2020-2032." <u>https://www.seattle.gov/parks/about-us/policies-and-plans/seattle-parks-and-recreation-strategic-plan</u>.

Historic, Archaeological, & Cultural Resources

- Ames, Kenneth M., and Herbert D. G. Maschner. 1999. *Peoples of the Northwest Coast, Their Archaeology and Prehistory*. Thames and Hudson Limited, London, England.
- Bagley, C. B. 1929. *History of King County*, Vols. 1 and 3. S. J. Clarke Publishing Company, Seattle, Washington.
- Blukis Onat, Astrida R., Lee A. Bennett, and Jan L. Hollenbeck. 1980. *Cultural Resource Overview and Sample Survey of the Skagit Wild and Scenic River.* Volume 1. Seattle: Institute of Cooperative Research, Mt. Baker-Snoqualmie National Forest.
- Boyd, R. T. 1990. "Demographic History, 1774–1874." In *Northwest Coast*, edited by Wayne Suttles, pp. 135–148. Handbook of North American Indians, Vol. 7, William C. Sturtevant, general editor, Smithsonian Institution, Washington, D.C.
- Caldbick, John. 2014. "Henry Yesler's Steam-powered Seattle Sawmill Cuts its First Lumber in Late March 1853." HistoryLink Essay No. 760. <u>http://www.historylink.org/File/760</u>.
- Caldbick, John. 2020a. "The Great Seattle Fire, Part I." HistoryLink Essay No. 21090. http://www.historylink.org/File/21090.
- Caldbick, John. 2020b. "The Great Seattle Fire, Part II." HistoryLink Essay No. 21099. http://www.historylink.org/File/21099.

Canaan, Leslie, Lisa Craig, Di Gao, Renee Kuhlman, Jim Lindberg and Bonnie McDonald. 2021. "Coming Next: The National Impact Agenda." *Forum Journal: Relevance and Resilience: Proceedings from PastForward Online 2020* 33(1).

https://forum.savingplaces.org/viewdocument/forum-journal-relevance-and-resili.

- Chesley, Frank. 2009. "Housebuilding in Seattle: A History." HistoryLink Essay No. 9116. http://www.historylink.org/File/9116.
- Chrzastowski, Michael. 1983. *Historical Changes to Lake Washington and Route of the Lake Washington Ship Canal, King County, Washington*. Department of the Interior, U.S. Geological Survey, Water Resources Investigation, Open-File Report 81-8112.
- Equity & Environment Initiative Community Partners Steering Committee (CPSC). 2016. "Equity & Environment Agenda."

https://www.seattle.gov/Documents/Departments/OSE/Equity/SeattleEEAgenda.pdf.

- Cullinan, Nicholas C. 1926. "History of the Seattle General Postoffice [*sic*]." *Washington Historical Quarterly* 17, no. 3 (July). The Washington University State Historical Society, Seattle.
- Dailey, Tom. 2020. "Coast Salish Villages of Puget Sound." http://coastsalishmap.org.
- Denny, Arthur A. 1888. Pioneer Days on Puget Sound. C. B. Bagley, Printer, Seattle, Washington.
- Dougherty, Phil. 2010. "Interstate 5 is completed in Washington on May 14, 1969." HistoryLink Essay No. 9393. <u>http://www.historylink.org/File/9393</u>.
- Douglass, John G., and Shelby A. Manney. 2020. "Creative Mitigation: Alternative Strategies for Resources, Stakeholders, and the Public. *Advances in Archaeological Practice* 8(3), pp. 213– 219. <u>https://doi.org/10.1017/aap.2020.25</u>.
- Duwamish Tribal Services. 2018. Who We Are. https://www.duwamishtribe.org/history.
- Fiset, Louis. 2001. "Seattle Neighborhoods: Cascade and South Lake Union—Thumbnail History." HistoryLink Essay No. 3178. <u>http://www.historylink.org/File/3178</u>.
- Gunther, Erna. 1945. "Ethnobotany of Western Washington." University of Washington Publications in Anthropology 10(1):1–62. Seattle.
- Hilbert, Vi, Jay Miller, and Zalmai Zahir. 2001. Puget Sound Geography: Original Manuscript from T.
 T. Waterman. Edited with additional material from Vi Hilbert, Jay Miller, and Zalmai Zahir.
 Lushootseed Press, Seattle, Washington.
- Jenkins, Dennis L., Loren G. Davis, Thomas W. Stafford Jr., Paula F. Campos, Bryan Hockett, George T. Jones, Linda Scott Cummings, Chad Yost, Thomas J. Connolly, Robert M. Yohe II, Summer C. Gibbons, Maanasa Raghavan, Morten Rasmussen, Johanna L. A. Paijmans, Michael Hofreiter, Brian M. Kemp, Jodi Lynn Barta, Cara Monroe, M. Thomas P. Gilbert, and Eske Willerslev. 2012. "Clovis Age Western Stemmed Projectile Points and Human Coprolites at the Paisley Caves." Science 337(6091):223–228.
- Kirk, R., and R. D. Daugherty. 2007. *Exploring Washington Archaeology*. University of Washington Press, Seattle.

- Kopperl, Robert, Charles Hodges, Christian Miss, Johonna Shea, and Alecia Spooner. 2016. *Archaeology of King County, Washington: A Context Statement for Native American Archaeological Resources*. SWCA Environmental Consultants, Seattle, Washington. Prepared for the King County Historic Preservation Program, Seattle, Washington.
- Lane, Barbara. 1975. Anthropological Report on the Indian Herring Fishery. Exhibit Presented in US Western District of Washington Court.
- Langloe, Lars. 1946. "Report on the Development of Industrial Sites in the Duwamish-Green River Valley." City Planning Commission, Seattle. <u>https://www.washington.edu/uwired/outreach/cspn/Website/Classroom%20Materials/C</u> urriculum%20Packets/Building%20Nature/Documents/43.html.
- Larson, Lynn L., and Dennis E. Lewarch (editors). 1995. The Archaeology of West Point, Seattle, Washington: 4,000 Years of Hunter-Fisher-Gatherer Land Use in Southern Puget Sound Volume 1, Parts 1 and 2. Larson Anthropological Archaeological Services, Ltd., Seattle, Washington. Prepared for King County Department of Metropolitan Services, Seattle, Washington. Submitted to CH2M Hill, Bellevue, Washington. On file at the Department of Archaeology and Historic Preservation, Olympia, Washington.
- Lewarch, Dennis E., Lynn L. Larson, Leonard A. Forsman, Guy F. Moura, Eric W. Bangs, and Paula Mohr Johnson. 1996. *King County Department of Natural Resources Water Pollution Control Division Alki Transfer/CSO Facilities Project Allentown Site (45KI431) and White Lake Site (45KI438 and 45KI438A) Data Recovery*. Larson Anthropological/Archaeological Services, Seattle, Washington. LAAS Technical Report #95-8. Prepared for King County Department of Natural Resources, Water Pollution Control Division, Seattle, Washington. Submitted to HDR Engineering, Bellevue, Washington.
- Link, Karin. 2004. 2003 Cascade Historic Survey Buildings, Objects & Artifacts: Context Statement. <u>https://www.seattle.gov/Documents/Departments/Neighborhoods/HistoricPreservation/</u> <u>HistoricResourcesSurvey/context-cascade.pdf</u>.
- MacIntosh, Heather M., and Walt Crowley. 1999. "Railroad Development in the Seattle/Puget Sound Region, 1872–1906." HistoryLink Essay No. 1683. <u>http://www.historylink.org/File/1683</u>.
- Marino, C. 1990. "History of Western Washington Since 1846." In *Northwest Coast*, edited by Wayne Suttles, pp. 169–179. Handbook of North American Indians, Vol. 7, William C. Sturtevant, general editor, Smithsonian Institution, Washington, D.C.
- McClary, Daryl C. 2013. "Fire destroys Great Northern Railway's grain and ore terminal at Smith Cove in Seattle on November 6, 1925." HistoryLink Essay No. 10653. <u>http://www.historylink.org/File/10653</u>.
- Mierendorf, Robert R. 1986. *People of the North Cascades*. National Park Service, Seattle, Washington.

National Trust for Historic Preservation (NTHP). 2021. "Historic Preservation Solutions to Build Back Better." March. <u>https://nthp-</u>

savingplaces.s3.amazonaws.com/2021/05/03/16/40/30/249/Historic%20Preservation%2 0Solutions%20to%20Build%20Back%20Better.pdf.

- Nelson, Charles M. 1990. "Prehistory of the Puget Sound Region." In *Northwest Coast*, edited by Wayne Suttles, pp. 481–484. Handbook of North American Indians, Vol. 7, William C. Sturtevant, general editor, Smithsonian Institution, Washington, D.C.
- Oldham, Kit. 2020. "Port of Seattle commissioners meet for the first time on September 12, 1911." HistoryLink Essay No. 9726. <u>http://www.historylink.org/File/9726</u>.
- Ruby, R. H., and J. A. Brown. 1992. *A Guide to the Indian Tribes of the Pacific Northwest, Revised Edition*. University of Oklahoma Press, Norman, Oklahoma, and London, England.
- Ryberg, Stephanie R. 2010. *Neighborhood Stabilization Through Historic Preservation: An Analysis of Historic Preservation and Community Development in Cleveland, Providence, and Seattle.* Dissertation, University of Pennsylvania. ProQuest Dissertations. <u>https://www.proquest.com/openview/16e9097389f6f2db6760ec4bdb7cc6e8/1?pq-</u>

origsite=gscholar&cbl=18750.

Rypkema, Donovan D. 2004. "The Oversimplification of Gentrification." *Forum Journal* 18(4). <u>https://forum.savingplaces.org/viewdocument/the-oversimplification-of-gentrific</u>.

Sanborn Map & Publishing Company. 1884. Seattle, King County, Washington Territory.

Sanborn Map & Publishing Company. 1888. Seattle, King County, Washington Territory.

Sanborn Map & Publishing Company. 1893. Seattle, King County, Washington.

Sanborn Map & Publishing Company. 1905. Seattle, King County, Washington.

- Sanborn Map & Publishing Company. 1928. Seattle, King County, Washington.
- Sanborn Map & Publishing Company. 1950. Seattle, King County, Washington.
- Seattle Office of Planning and Community Development (OPCD). 2020. "Equitable Development Community Indicators Report: Equitable Development Monitoring Program." September. <u>https://www.seattle.gov/Documents/Departments/OPCD/Demographics/communityindi</u> <u>catorsreport2020.pdf</u>.
- Suttles, W., and B. Lane. 1990. "Southern Coast Salish." In *Northwest Coast*, edited by Wayne Suttles, pp. 485–502. Handbook of North American Indians, Vol. 7, William C. Sturtevant, general editor, Smithsonian Institution, Washington, D.C.
- Swanton, John Reed. 1952. *Indian Tribes of Washington, Oregon and Idaho*. Bureau of American Ethnology Bulletin 145, Smithsonian Institution, Washington, D.C.
- Thompson, Nile, and Carolyn J. Marr. 2013. "Seattle Public Schools, 1862–2000: Cascade School." HistoryLink Essay No. 10478. <u>http://www.historylink.org/File/10478</u>.

- Thrush, C. 2007. *Native Seattle: Histories from the Crossing-Over Place*. University of Washington Press, Seattle.
- Updegrave, Cynthia. 2016. "Duwamish-Green Watershed." HistoryLink Essay No. 20272. <u>http://www.historylink.org/File/20272</u>.
- U.S. Surveyor General (USSG). 1856. *Township No 25 N. R. No 5 East Willamette Mer*. Surveyor General's Office, Olympia, Washington Territory. www.blm.gov/or/landrecords/survey/ySrvy1.php.
- U.S. Surveyor General (USSG). 1863. *Township No 25 N. R. No 5 East Willamette Mer*. Surveyor General's Office at Olympia, Washington Territory. <u>www.blm.gov/or/landrecords/survey/ySrvy1.php</u>.
- U.S. Surveyor General (USSG). 1871. *Township No 25 N. R. No 5 East Willamette Mer*. Surveyor General's Office at Olympia, Washington Territory. www.blm.gov/or/landrecords/survey/ySrvy1.php.
- Walton Potter, Elisabeth. 1975a. *Hill (Samuel) House* (KI00165). National Register of Historic Places Nomination Form. On file at the Department of Archaeology and Historic Preservation, Olympia, Washington.
- Walton Potter, Elisabeth. 1975b. *Volunteer Park* (DT00027). National Register of Historic Places Nomination Form. On file at the Department of Archaeology and Historic Preservation, Olympia, Washington.
- Walton Potter, Elisabeth. 1977a. *Merrill (R. D.) House* (KI00167). National Register of Historic Places Nomination Form. On file at the Department of Archaeology and Historic Preservation, Olympia, Washington.
- Walton Potter, Elisabeth. 1977b. Chittenden (Hiram M.) Locks and Related Features of the Lake Washington Ship Canal Historic District. National Register of Historic Places Nomination Form. On file at the Department of Archaeology and Historic Preservation, Olympia, Washington.
- Wessen, Gary C. 1988. *Prehistoric Cultural Resources of Island County*. Wessen & Associates, Seattle, Washington. Prepared for Washington State Department of Community Development.
- Wilma, David. 2001a. "Seattle Neighborhoods: Interbay—Thumbnail History." HistoryLink Essay No. 3418. <u>http://www.historylink.org/File/3418</u>.

Wilma, David. 2001b. "Seattle Neighborhoods: Georgetown—Thumbnail History." HistoryLink Essay No. 2975. <u>http://www.historylink.org/File/2975</u>.

Transportation

California Air Pollution Control Officers Association (CAPCOA). 2010. "Quantifying Greenhouse Gas Mitigation Measures."

City of Seattle. 2007. "Ordinance 122386: Complete Streets Policy."

http://clerk.ci.seattle.wa.us/~scripts/nph-

brs.exe?d=CBOR&s1=115861.cbn.&Sect6=HITOFF&l=20&p=1&u=/%7Epublic/cbor2.htm& r=1&f=G.

City of Seattle. Commute Trip Reduction Program.

https://www.seattle.gov/transportation/projects-andprograms/programs/transportation-options-program/commute-trip-reduction-program.

King County Metro. 2018. Fall 2018 Ridership Data.

- Seattle City Budget Office. 2020. "2021-2026 Adopted Capital Improvement Program." Accessed July 2021. <u>https://www.seattle.gov/city-budget-office/capital-improvement-program-archives</u>.
- Seattle Department of Transportation. 2010. "Intelligent Transportation Systems (ITS) Strategic Plan."

https://www.seattle.gov/Documents/Departments/SDOT/TechnologyProgram/ITSStrateg icPlan20102020.pdf.

Seattle Department of Transportation. 2014. "Seattle Bicycle Master Plan."

https://www.seattle.gov/transportation/document-library/citywide-plans/modalplans/bicycle-master-plan.

- Seattle Department of Transportation. 2015. "Move Seattle: 10-year Strategic Vision for Transportation." <u>https://www.seattle.gov/transportation/document-library/citywide-plans/move-seattle</u>.
- Seattle Department of Transportation. 2016. "Freight Master Plan." <u>https://www.seattle.gov/transportation/document-library/citywide-plans/modal-plans/freight-master-plan</u>.
- Seattle Department of Transportation. 2016. "Transit Master Plan." <u>https://www.seattle.gov/transportation/document-library/citywide-plans/modal-plans/transit-master-plan</u>.
- Seattle Department of Transportation. 2017. "Pedestrian Master Plan." <u>https://www.seattle.gov/transportation/document-library/citywide-plans/modal-plans/pedestrian-master-plan.</u>
- Seattle Department of Transportation. 2020. "Bicycle and Pedestrian Safety Analysis." <u>https://www.seattle.gov/documents/Departments/SDOT/VisionZero/SDOT_Bike%20and</u> <u>%20Ped%20Safety%20Analysis_Ph2_2420(0).pdf</u>.
- Seattle Department of Transportation. 2021. Collision Data, 2016-2020.

Sound Transit. 2020. "Service Implementation Plan."

https://www.soundtransit.org/sites/default/files/documents/2020-serviceimplementation-plan.pdf. Transportation Research Board. 2016. "Highway Capacity Manual, Sixth Edition: A Guide for Multimodal Mobility Analysis."

Public Services

- BNSF Railway. 2021. *Police Team.* Accessed July 2021. <u>https://www.bnsf.com/in-the-community/safety-and-security/police-team.page</u>.
- City of Seattle. 2018. 2018 Seattle Fire Code. City of Seattle. Accessed November 2021. <u>http://www.seattle.gov/sdci/codes/codes-we-enforce-(a-z)/fire-</u> <u>code#2018seattlefirecode</u>.
- Department of Finance and Administrative Services. 2021. *Fire Facilities and Emergency Response Levy.* Accessed July 2021. <u>https://www.seattle.gov/fire-facilities-and-emergency-</u> <u>response-levy</u>.
- Federal Bureau of Investigation. 2013. *Variables Affecting Crime*. Accessed October 2021. <u>https://ucr.fbi.gov/nibrs/2013/resources/variables-affecting-crime/view</u>.
- Port of Seattle Police. 2020. "2020 Annual Report." <u>https://www.portseattle.org/page/annual-reports#</u>.
- Port of Seattle Police. 2020. "Policy Manual." Accessed July 2021. https://public.powerdms.com/PORT/tree/documents/2056436.
- PSRC. 2020. "Who's affected by housing displacement?" *PSRC*. <u>https://www.psrc.org/whats-happening/blog/whos-affected-housing-displacement</u>.
- Seattle City Budget Office. 2020. "2021 Adopted Budget." Accessed July 2021. <u>https://www.seattle.gov/city-budget-office/budget-archives</u>.
- Seattle City Budget Office. 2020. "2021-2026 Adopted Capital Improvement Program." Accessed July 2021. <u>https://www.seattle.gov/city-budget-office/capital-improvement-program-archives</u>.
- Seattle City Budget Office. 2021. "2022 Proposed Budget." Accessed October 2021. <u>http://www.seattle.gov/city-budget-office</u>.
- Seattle City Budget Office. 2020. "2022-2027 Proposed Capital Improvement Program." Accessed October 2021. <u>http://www.seattle.gov/city-budget-office</u>.
- Seattle Department of Transportation. 2021. *West Seattle Bridge Program.* Accessed October 2021. <u>http://www.seattle.gov/transportation/projects-and-programs/programs/bridges-stairs-and-other-structures/bridges/west-seattle-bridge-program.</u>
- Seattle Fire Department. 2012. "Strategic Plan, 2012–2017." <u>https://www.seattle.gov/Documents/Departments/Fire/About/SeattleFireStrategicPlan.p</u> <u>df</u>.
- Seattle Fire Department. 2020. "2020 Annual Report." https://fireline.seattle.gov/2021/06/16/seattle-fires-2020-annual-report/.

- Seattle Fire Department. 2021. *Fire Station 31.* Accessed October 2021. <u>https://www.seattle.gov/fire/about-us/about-the-department/operations/fire-station-31.</u>
- Seattle Police Department. 2021. *Calls for Service Dashboard*. Accessed July 2021. <u>http://www.seattle.gov/police/information-and-data/calls-for-service-dashboard</u>.
- Seattle Police Department. 2021. *Department Fact Sheet*. Accessed July 2021. <u>http://www.seattle.gov/police/about-us/about-the-department/department-fact-sheet</u>.
- Seattle Police Department. 2021. *Precinct and Patrol Boundaries*. Accessed July 2021. <u>https://www.seattle.gov/police/about-us/about-policing/precinct-and-patrol-boundaries</u>.
- Seattle Police Department. 2021. *Seattle Police Department Manual.* 06 01. Accessed July 2021. <u>http://www.seattle.gov/police-manual</u>.
- The Northwest Seaport Alliance. 2021. *Terminal 5 Improvements*. Accessed October 2021. https://www.nwseaportalliance.com/about-us/planning/terminal-5-improvements.
- TSA. 2016. TSA Provides Support to Coast Guard to secure U.S. Ports. August 22. Accessed November 2021. <u>https://www.tsa.gov/news/press/top-stories/2016/08/22/tsa-provides-</u> <u>support-coast-guard-secure-us-ports</u>.

Utilities

- King County. 2012a. "King County's Long-term CSO Control Plan Amendment." King County Department of Natural Resources and Parks, Wastewater Treatment Division.
- King County. 2014a. "Updated Planning Assumptions for Wastewater Flow Forecasting." King County Department of Natural Resources and Parks, Wastewater Treatment Division. July.
- King County. 2018. "Combined Sewer Overflow Control Program, 2018 CSO Control Program Update." King County Department of Natural Resources and Parks, Wastewater Treatment Division.

Seattle City Light. 2019. "SCL 2019–2024 Strategic Plan." Seattle City Light.

- Seattle City Light. 2020. "SCL 2020 Integrated Resource Plan Progress Report." Seattle City Light.
- Seattle Public Utilities. 2015a. "2015 Plan to Protect Seattle's Waterways." Seattle Public Utilities. May.
- Seattle Public Utilities. 2015b. "2015–2010 Strategic Business Plan". Seattle Public Utilities.
- Seattle Public Utilities. 2015c. "SPU Long Term Control Plan." Seattle Public Utilities. May.

Seattle Public Utilities. 2015d. "SPU Integrated Plan." Seattle Public Utilities. May.

Seattle Public Utilities. 2019. "2019 Water System Plan." Seattle Public Utilities. August.

Starcrest. 2018. 2016 Puget Sound Maritime Emissions Inventory. Starcrest Consulting Group, LLC. October.


List of Appendices

- A Scoping Notice & Comment Summary
- B Industrial & Maritime Strategy Council Report
- C Alternative Future Land Use Zoning Maps
- D Draft Comprehensive Plan Goal & Policy Language
- E Industrial Development Regulations
- F Shoreline Master Program Development Regulations
- G Preferred Alternative Preliminary Regulations
- H Air Quality and Noise Technical Memo
- I Transportation Screenline Information
- J EIS Mitigation Measures List
- K Presentation Slides from the Draft EIS Public Hearings

Amendment A Version #1 to CB 120933

Sponsor: Council President Nelson

Limit the total number of residential units in Stadium Transition Area Overlay District

Effect: Council Bill 120933 would allow residential units as a conditional use in the Stadium Transition Area Overlay District (STAOD). Alternative 4 under the Seattle Industrial & Maritime Strategy Final Environmental Impact Statement (FEIS) estimated that 990 residential units could be built in the STAOD under the conditions placed on housing in the Urban Industrial (UI) zone, which is the underlying zoning in the STAOD.

This amendment would limit the total number of units in the STAOD to 990 units, consistent with the FEIS.

Amend Section 2 of Council Bill 120933 to amend subsection C of SMC section 23.74.008 as follows (new language in red with a double underline):

Section 2. Section 23.74.008 of the Seattle Municipal Code, last amended by Ordinance

126862, is amended as follows:

23.74.008 Uses.

Notwithstanding the use provisions of the underlying zone, the following use provisions apply:

* * *

C. In areas zoned Urban Industrial, residential uses are permitted as a conditional use

pursuant to the criteria contained in subsection 23.50A.062.C, ((except that)) only where the

following occur:

1. ((eriterion)) Criterion 23.50A.062.C.3 does not apply within the Stadium

Transition Area Overlay District; and

2. The total number of residential units permitted in the Stadium Transition Area

Overlay District may not exceed 990 units.

Amendment B Version #1 to CB 120933

Sponsor: Councilmember Saka State Council's intent to maintain industrial lands for industrial uses

Effect: This amendment would add seven recitals to CB 120933 that would state the Council's intent to strengthen policies and regulations and to state clearly that the Council will not further expand the amount of industrial areas where residential development will be permitted.

Add seven recitals to CB 120933 as follows:

- WHEREAS, Seattle's Comprehensive Plan includes policy LU 10.2, which states that the City will "Preserve industrial land for industrial uses, especially where industrial land is near rail- or water-transportation facilities to allow marine- and rail-related industries that rely on that transportation infrastructure to continue to function in the city." and
- WHEREAS, Seattle's Comprehensive Plan includes policy LU 10.3, which states that the City will "Ensure predictability and permanence for industrial activities in industrial areas by limiting changes in industrial land use designation. There should be no reclassification of industrial land to a non-industrial land use category except as part of a City-initiated comprehensive study and review of industrial land use policies or as part of a major update to the Comprehensive Plan." and
- WHEREAS, Seattle's Comprehensive Plan includes policy LU 10.8, which states that the City will "Prohibit new residential development in industrial zones except for certain types of dwellings, such as caretaker units and, in urban industrial zones, dwellings for workers,

that are related to the industrial area and that would not restrict or disrupt industrial activity." and

- WHEREAS, additional new housing in industrial areas outside of the limited industrial-related housing currently allowed could have significant impacts on the City's industrial areas; and
- WHEREAS, the Council intends to further strengthen the City's policies and regulations to further limit changes to the boundaries of industrial areas, and further limit rezones to non-industrial uses in the Urban Industrial zone within Manufacturing/Industrial Centers; NOW, THEREFORE,

BE IT ORDAINED BY THE CITY OF SEATTLE AS FOLLOWS:

* * *

Amendment C Version #1 to CB 120933

Sponsor: Councilmember Saka

Require covenants related to liquefaction zones be in place in perpetuity

Effect: The Stadium Transition Area Overlay District (STAOD) includes areas identified as liquefactionprone areas – areas where, during an earthquake, the earth becomes unstable due to a combination of loose soil structure and high water tables. Liquefaction-prone areas are identified as geologic hazard environmentally critical areas (ECAs) due to the likelihood of significant damage to structures if they are not properly designed to withstand ground movement during an earthquake. The ECA code (Seattle Municipal Code (SMC) 25.09) states that the Director of the Seattle Department of Construction and Inspections (SDCI) may require that a permanent covenant be recorded to the benefit of the City prior to permitting of development in geologic hazard areas. The covenant requires the property owner to:

- 1. Maintain their property in such a manner as will prevent harm to the public or occupants of the property;
- 2. Declare that they understand the risks of building in the liquefaction zone, and that they will affirmatively convey those risks to future occupants of the building; and
- 3. Waive any right to assert a claim against the City for damages to the structure, except when the City is solely at fault.

The model covenant language can be found here: <u>https://www.seattle.gov/Documents/Departments/SDCI/Forms/LiquefactionProneAreaCovenant.pdf</u>

This amendment to the proposed Stadium Transition Area Overlay District provisions requires that if a covenant is required, that the covenant will be required to be in place in perpetuity.

Amend Section 2 of Council Bill 120933 to amend subsection C of SMC section 23.74.008 as follows (new language in red with a double underline):

Section 2. Section 23.74.008 of the Seattle Municipal Code, last amended by Ordinance

126862, is amended as follows:

23.74.008 Uses.

Notwithstanding the use provisions of the underlying zone, the following use provisions apply:

* * *

C. In areas zoned Urban Industrial, residential uses are permitted as a conditional use

pursuant to the criteria contained in subsection 23.50A.062.C, except that:

1. ((eriterion)) Criterion 23.50A.062.C.3 does not apply within the Stadium

Transition Area Overlay District; and

2. If any site is determined to be a geologic hazard area by the Director, a

covenant shall be required and recorded to run with the land in perpetuity.

Amendment D Version #1 to CB 120933

Sponsor: Councilmember Saka Lower permitted noise levels in residential development

Effect: Excessive noise levels can result in impacts to residents' speech, sleep, and concentration, as well as annoyance, nuisance, and dissatisfaction in people in noisy environments. The Seattle Industrial & Maritime Strategy Final Environmental Impact Statement (2022) identified the Stadium Area as an area with excessive levels of noise with daytime levels between 61.5–69.0 decibels (dBA), nighttime levels between 55.7–68.0 dBA, and average day-night levels (DNL) of 69.2.

According to the Seattle Industrial & Maritime Strategy Final Environmental Impact Statement (2022), the United Stated Department of Housing and Urban Development (HUD) promulgates noise standards for federally-subsidized housing, as follows:

Under HUD standards, noise levels within residences should not exceed a DNL of 45 dB (typically expressed as dBA). Because interior noise levels in typical residential construction are about 20 dBA below exterior levels, HUD standards classify sites where community exterior noise levels exceed 65 dB as noise-impacted areas and require additional sound attenuation to bring interior noise levels within the 45 dBA standard.

Conditions on building housing in the Urban Industrial zone are based on the City's noise ordinance (Seattle Municipal Code (SMC) Chapter 25.08), which indicates that noise levels in residential areas receiving noise from industrial areas should not exceed 60 dBA.

For residential uses in residential areas, the City uses 45 dBA as the maximum permitted exterior nighttime noise level. See Seattle Municipal Code Sections 25.08.410 and .420.

This amendment amends the conditions for housing in the Urban Industrial zone in order to prohibit housing in the Stadium Transition Area Overlay District unless indoor noise levels can be reduced to 45 dBA.

Amend Section 2 of Council Bill 120933 to amend subsection C of SMC section 23.74.008 as follows (new language in red with a double underline):

Section 2. Section 23.74.008 of the Seattle Municipal Code, last amended by Ordinance

126862, is amended as follows:

23.74.008 Uses.

Notwithstanding the use provisions of the underlying zone, the following use provisions apply:

* * *

<u>C. In areas zoned Urban Industrial, residential uses are permitted as a conditional use</u> pursuant to the criteria contained in subsection 23.50A.062.C, except that:

1. ((eriterion)) Criterion 23.50A.062.C.3 does not apply within the Stadium

Transition Area Overlay District (STAOD); and

2. Criterion 23.50A.062.C.4. does not apply within the STAOD, and instead the

following criteria must be met: All dwelling units shall have sound-insulating windows sufficient

to maintain interior sound levels at 45 decibels or below in consideration of existing

environmental noise levels at the site. The applicant shall submit an analysis of existing noise

levels and documentation of the sound insulating capabilities of windows as part of the

conditional use permit application.

Amendment E Version #1 to CB 120933

Sponsor: Councilmember Saka Limit off-street parking access from Major Truck Streets

Effect: This amendment would add a condition to residential development in the Stadium Transition Area Overlay District that would prohibit access to off-street parking or loading facilities from Major Truck Streets. This would reduce conflicts between vehicles accessing residential structures in the STAOD and freight traffic using the City's designated truck routes. This amendment augments Seattle Municipal Code Section 23.74.010.A.2, which limits the number of curb cuts on block fronts in the district to three per block along each north-south street and two per block along east-west streets.

Amend Section 2 of Council Bill 120933 to amend subsection C of SMC section 23.74.008 as follows (new language in red with a double underline):

Section 2. Section 23.74.008 of the Seattle Municipal Code, last amended by Ordinance

126862, is amended as follows:

23.74.008 Uses.

Notwithstanding the use provisions of the underlying zone, the following use provisions apply:

* * *

C. In areas zoned Urban Industrial, residential uses are permitted as a conditional use

pursuant to the criteria contained in subsection 23.50A.062.C, ((except that)) only where the

following occur:

1. ((eriterion)) Criterion 23.50A.062.C.3 does not apply within the Stadium

Transition Area Overlay District (STAOD); and

2. No curb cuts serving required parking or required loading for a residential use

are allowed along a Major Truck Street unless no other access is possible. No curb cuts are

permitted from Major Truck Streets for any non-required residential parking or non-required

loading facilities.

Amendment F Version #1 to CB 120933

Sponsor: Councilmember Saka Require signage notifying residents of risks

Effect: This amendment would add a condition to residential development in the Stadium Transition Area Overlay District that would require owners to post clear and conspicuous notices regarding risks of living in the Stadium Transition Area Overlay District.

Amend Section 2 of Council Bill 120933 to amend subsection C of SMC section 23.74.008 as follows (new language in red with a double underline):

Section 2. Section 23.74.008 of the Seattle Municipal Code, last amended by Ordinance

126862, is amended as follows:

23.74.008 Uses.

Notwithstanding the use provisions of the underlying zone, the following use provisions apply:

* * *

C. In areas zoned Urban Industrial, residential uses are permitted as a conditional use

pursuant to the criteria contained in subsection 23.50A.062.C,

1. except that ((eriterion)) Criterion 23.50A.062.C.3 does not apply within the

Stadium Transition Area Overlay District (STAOD); and

2. only where the following occur: The building containing residential uses shall

have at least three signs in conspicuous locations, such as in the residential lobby, the leasing

office, and on the exterior of the building visible from the residential entry, that use clear

language to convey the following information:

a. That the project is located in an industrial area, and that residents, by

choosing to live in the area, accept the industrial character of the neighborhood and agree that

existing or permitted industrial uses do not constitute a nuisance or other inappropriate or

unlawful use of land, and

b, If the project has been determined to be in a liquefaction zone, that the building is in a liquefaction zone and that residents understand that there may be heightened risk <u>during earthquakes.</u>

Amendment G Version #1 to CB 120933

Sponsor: Councilmember Saka Extended effective date

Effect: This amendment would extend the effective date for CB 120933 to June 30 in order to provide time for the Port of Seattle to understand the extent and scope of the United States Coast Guard's expansion of their facilities in Seattle on Terminal 46. The effective date would be sixty days after the anticipated date of the United State Government's Record of Decision.

Amend Section 3 of Council Bill 120933 as follows (new language in red with a double underline):

Section 3. This ordinance shall take effect ((as provided by Seattle Municipal

Code Sections 1.04.020 and 1.04.070.)) on June 30, 2025.

Amendment H Version #1 to CB 120933

Sponsor: Councilmember Moore Continue to prohibit housing west of 1st Avenue S

Effect: Council Bill (CB) 120933 would allow residential uses as a conditional use in the Stadium transition Area Overlay District (STAOD) under the provisions of the Urban Industrial (UI) zone. Unlike in other UI zones, the provisions of CB 120933 would allow housing within 200 feet of a Major Truck Street, potentially increasing noise, vibration, and air pollution impacts to the residents of future structures in the STAOD, and also potentially impacting the movement of trucks through the STAOD.

This amendment would maintain the prohibition on housing on the west side of 1st Avenue S, but would allow housing on the east side of 1st Avenue S. This would allow housing on parcels further away from Port of Seattle facilities, potentially limiting impacts of housing on the Port's activities and vice versa.

Amend Section 2 of Council Bill 120933 to amend subsection C of SMC section 23.74.008 as follows:

Section 2. Section 23.74.008 of the Seattle Municipal Code, last amended by Ordinance

126862, is amended as follows:

23.74.008 Uses.

Notwithstanding the use provisions of the underlying zone, the following use provisions apply:

* * *

C. In areas zoned Urban Industrial, residential uses are permitted as a conditional use east

of 1st Avenue S pursuant to the criteria contained in subsection 23.50A.062.C, except that

criterion 23.50A.062.C.3 does not apply within the Stadium Transition Area Overlay District.

Residential uses otherwise allowed as an administrative conditional use in the Urban Industrial

zone pursuant to subsection 23.50A.062.C. are prohibited west of 1st Avenue S.

Amendment I Version #1 to CB 120933

Sponsor: Councilmember Moore

Ensure no City funding goes to housing projects in the Stadium Area

Effect: Council Bill (CB) 120933 would allow residential uses as a conditional use in the Stadium Transition Area Overlay District (STAOD) under the provisions of the Urban Industrial (UI) zone. One of the purposes of these provisions is to encourage mixed-use development near the stadiums that can provide housing affordable to workers in the Duwamish Industrial Area without public subsidies. This amendment would ensure that no City funding contributes to projects in this area by requiring that the application for a residential project in the area record an agreement not to seek or use funding for the development or operation of the project.

Amend Section 2 of Council Bill 120933 to amend subsection C of SMC section 23.74.008 as follows:

Section 2. Section 23.74.008 of the Seattle Municipal Code, last amended by Ordinance

126862, is amended as follows:

23.74.008 Uses.

Notwithstanding the use provisions of the underlying zone, the following use provisions apply:

* * *

C. In areas zoned Urban Industrial, residential uses are permitted as a conditional use

pursuant to the criteria contained in subsection 23.50A.062.C, except that:

1. ((eriterion)) Criterion 23.50A.062.C.3 does not apply within the Stadium

Transition Area Overlay District; and

2. A Master Use Permit application for a development containing residential uses

in the Stadium Transition Area Overlay District pursuant to the criteria contained in subsection

23.50A.062.C must contain an executed and recorded agreement stating that the development has not used City funding, will not use City funding and will not seek City funding for the construction of the project, or any environmental remediation of the site on which the development is located. The agreement shall be recorded on the title of the property on which that development is located.